	AD 645 775 QUESTIONS OF THE STATISTICAL THEORY OF
	RADAR (VOL II) P. A. Bakut, et al
trees containing to	
	National Tachnical Information Service
	U. S. DEPARTMENT OF COMMERCE

DISCLAIMER NOTICE

THIS DOCUMENT IS BEST QUALITY AVAILABLE. THE COPY FURNISHED TO DTIC CONTAINED A SIGNIFICANT NUMBER OF PAGES WHICH DO NOT REPRODUCE LEGIBLY.

This Document Contains Missing Page/s That Are Unavailable In The Original Document

THIS DOCUMENT CONTAINED BLANK PAGES THAT HAVE BEEN DELETED

REPRODUCED FROM BEST AVAILABLE COPY

DISCLAIMER NOTICE

THIS DOCUMENT IS BEST QUALITY PRACTICABLE. THE COPY FURNISHED TO DDC CONTAINED A SIGNIFICANT NUMBER OF PAGES WHICH DO NOT REPRODUCE LEGIBLY.



This document is a machine translation of Russian text which has been processed by the AN/GSQ-16(XW-2) Machine Translator, owned and operated by the United States Air Force. The machine output has been postedited to correct for major ambiguities of meaning, words missing from the machine's dictionary, and words out of the context of meaning. The sentence word order has been partially rearranged for readability. The content of this translation does not indicate editorial accuracy, nor does it indicate USAF approval or disapproval of the material translated.

N O T I C E

THIS DOCUMENT HAS BEEN REPRODUCED FROM THE BEST COPY FURNISHED US BY THE SPONSORING AGENCY. ALTHOUGH IT IS RECOGNIZED THAT CER-TAIN PORTIONS ARE ILLEGIBLE, IT IS BEING RE-LEASED IN THE INTEREST OF MAKING AVAILABLE AS MUCH INFORMATION AS POSSIBLE.

FTD-MT-65-47

EDITED MACHINE TRANSLATION

QUESTIONS OF THE STATISTICAL THEORY OF RADAR (VOL. II)

BY: P. A. Bakut and I. A. Bol'shakov, et al.

English Pages: 827

тм5500931

THIS TRANSLATION IS A RENDITION OF THE ORIGI-NAL FOREIGN TEXT WITHOUT ANY ANALYTICAL OR EDITORIAL COMMENT. STATEMENTS OR THEORIES ADVOCATED OR IMPLIED ARE THOSE OF THE SOURCE AND DO NOT NECESSARILY REFLECT THE POSITION OR OPINION OF THE FOREIGN TECHNOLOGY DI-VISION.

PREPARED BY:

TRANSLATION DIVISION FOREIGN TECHNOLOGY DIVISION WP-AFB, OHIO.

FTD-MT-65-47

Í

 \mathcal{N}

Date 28 June 19 66

P. A. Bakut, I. A. Bol'shake B. M. Gerasimov, A. A. Kuriksha, V. G. Repin, P. Tartakovskiy, V. V. Shirokov

ł

CONSTRUCTION OF A DESCRIPTION OF A DESCR

á 1

VOPROSY STATISTICHESKOY TEORII RADIOLOKATSII

pod obshchey redaktsiyey professora G. P. Tartakovskogo

Tom II

Izdatel'stvo "Sovetskoye Radio" Moskva ~ 1964

Pages 1-1079

.

1-a

TABLE OF CONTENTS

の一般の一般の一般の一般の一般の

「「「「「「「」」」の「「」」」

and the she

Gatidation and the

and the state of the second

ah huddhir da da da maran

1

U. S. Board	l on Geo	graphic Names Transliteration System	xi
Designation	ns of th	e Trigonometric Functions	xii
Chapter VI.	Gener	al Laws Governing Radar Measurements	1
\$ 6.:	Introd	uction	1
\$ 6.2	Genera	l Analysis of Tracking Meters	4
	6.2.1	Basic Features of Circuit Construction and Components of Errors of Measurement	5
	6.2.2	Characteristics of a Discriminator in the Case of Rapid Fluctuations	7
	6.2.3	Accuracy of Measurement in the Absence of Parametric Fluctuations	13
	6.2.4	Accuracy of Measurement in the Presence of Parametric Fluctuations	18
	6.2.5	Case of a Set of Rapid and Very Slow Fluctuations	21
§6.3	Questi	ons of Breakoff of Tracking	26
	6.3.1	Markovian Random Processes and Related Differential Equations	26
	6.3.2	Formulation of the Problem About Breakoff of Tracking	31
	6.3.3	Stationary Distribution with Limitations	3 5
	6.3.4	Average Time to Breakoff	38
§6.4	Genera	l Analysis of Nontracking Meters	41
	6.4.1	Basic Features of Circuit Construction and Components of Measurement Errors	41
	6.4.2	Errors of Measurement During Rapid Fluctuations	42
\$6.5	Select Meters Measur	ion of the Method of Synthesis of Rader Theory of Solutions in Problems of Theory	45
	6.5.1	Different Nonstatistical and Statistical Criteria for Synthesis	45
	6.5.2	Application of the Theory of Solutions to Problems of Measurements	48
	6.5.3	Theory of Statistical Estimates	54
	6.5.4	Theory of Optimum Linear Filtration	61
	6.5.5	Filtration of a Markovian Parameter by the Method of a Posteriori Probability	64
	6.5.6	Limitations of Theory. General Theory of Bayes Filtration	69

ť

STATE PINES

§6.6	Method Gaussi Accura	s of Synthesis of Optimum Meters with an Statistics of the Parameter. Potential cy of Measurement	73
	6.6.1	Properties of the Likelihood Function and Its Approximation	73
	6.6.2	Optimum Tracking Meter for Gaussian Statistics of the Measured Parameter	78
	6.6.3	The Optimum Tracking Meter in a Number of Specific Discrete Cases	90
	6.6.4	Possibility of One Simplification of the Meter Circuit	92
	6.6.5	Optimum Nontracking Meter	95
	6. 6.6	Potential Accuracy of Measurement	98
	6.6.7	Allowance for Side Peaks of the Likelihood Function	102
	6.6.8	Case of a Partially Assigned Discriminator	108
	6.6.9	Optimum Meter of a Linear Functional of a Parameter with Gaussian Statistics	110
	6 .6.1 0	Relationship of the Method of Synthesis with Other Branches of the Theory of Solutions	116
§6.7	Synthe Proper	sis of Discriminators for Different Statistical ties of Signals	119
	6.7.1	Regular Signals in Gaussian Noises	119
	6.7.2	Gaussian Coherent Signals in Gaussian Noises	125
	6.7.3	Independently Fluctuating Sendings. General Case of an Incoherent Signal	143
	6.7.4	Certain Forms of Non-Gaussian Signals in White Noises	151
	6.7.5	Approxiante Methods of Construction of Optimum Discriminators. Methods of Comparison of Performance of Circuits	154
§ 6.8	Synthe Accura	sis of Smoothing Circuits and Resultant cy of Measurements	157
	6.8.1	Parameter - Stationary Random Process	157
	6.8.2	Parameter — Random Process with Stationary Increasements	162
	6.8.3	Parameter — Linear Combination of Known Functions with Random Factors	17 0
	6.8.4	Mixed Cases	178
	6.8.5	Smoothing Circuits for Linear Functionals of Parameters	182

6

Star (Strange

÷.

. . . .

i

i

:

ï

§6.9	5.9 Synthesis of Meters with a Markovian Parameter and with Limited Knowledge of the Statistics of the Parameters			
	6.9.1	Optimum Meter of a Markovian Parameter	186	
	6.9.2	Synthesis of a Meter with Statistics Unknown. Minimax Solution	191	
	6.9.3	Description of a Parameter with Limited Knowledge of Its Statistics	1 94	
	6.9.4	Synthesis of a Meter with Limited Knowledge of the Statistics	196	
§ 6.10	Conclu	sion	201	
Chapter VII	. Rang	e Finding with a Coherent Signal	2 05	
§ 7.1	Introd	uction	205	
\$ 7.2	Optimu	m Discriminator	208	
	7.2.1	Operations of an Optimum Discriminator	208	
	7.2.2	Discriminator Characteristics	211	
\$ 7.3	Discri Respec	minator with Two Channels Detuned with t to Range	218	
	7.3.1	Block Diagram of the Discriminator	218	
	7.3.2	Identical Modulation of the Reference and Sounding Lignals	55 0	
	7.3.3	Influence of the Finite Magnitude of Detuning	227	
	7 .3. 4	Noncoincidence of Modulations of Reference and Sounding Signals	229	
\$ 7.4	Discri	minator with Switching of Reference Signals	232	
	7.4.1	Block Diagram of the Discriminator	232	
	7.4.2	Characteristics of the Discriminator	234	
	7.4.3	Optimum Filter for a Discriminator with Switching	238	
§ 7 . 5	Discri Signal	minator with Differentiation of the Reference	241	
	7.5.1	Block Diagram of the Discriminator	241	
	7.5.2	Characteristics of the Discriminator	242	
§ 7.6	Pulse	Radiation	245	
	7.6.1	Square Pulse	247	
	7.6.2	Gauesian Pulse	250	
	7.6.3	Discriminator with Differentiation of the Reference Signal	253	
\$ 7.7	Contin	uous Radiation	256	
	7.7.1	Frequency Modulation	256	

Strain an ad us

1

COLLES!

的成功的

0274

STATE OF STREET

ST PARTY OF

		and the second	
	7.7.2	Phase-Code Manipulation (FKM)	26 B
	7.7.3	Noise Phase Modulation	2 ⁶ 7
	7.7.4	Noise Signal	265
§7.8	Pulse	Radiation with Additional Mcdulation	268
	7.8.1	Linear Frequency Modulation	2 58
§7.9	Multi:	frequency Radiation Without Medulation	272
	7.9.1	Case of Arbitrarily Correlated Frequency Components of the Reflected Signal	274
§ 7.10	Analy	sis of Accuracy of Radar Range Finders	2
	7.10.1	Influence of a System of Automatic Gain Control (AGC) on Discriminator Characteristics	277
	7.10,2	Smoothing Circuits with Constant Parameters	280
	7.10.3	Optimum Smoothing Circuits for a Randomly Varying Parameter	5 ö0
	7.10.4	Smoothing Circuits with Variable Parameters	2 92
\$ 7.11	Nontr	acking Radar Range Finders	300
	7.11.1	Range Finder with Frequency Modulation	301
	7.11.2	Use of a Range Finder with a Fast Tracking Loop for Forming an Estimate	7 04
§ 7.12	Range	Finder with Discrete Measurement	3 06
§ 7 .1 3	Range Repet	Finding in a System with a Variable ition Frequency	311
§ 7.14	Influ Range	ence of Interferences on Johesent Radar Finders	314
	7.14.1	Noise Interference	315
	7.14.2	Pulse Chaotic Interference	316
	7.14.3	Return Interference	318
	7.14.4	Passive Interference	320
	7.14.5	Intermittent Interference	382
\$ 7.15	Nonli	near Phenomena in Tracking Range Finders	324
	7.15.1	Average Time to Breakoff of Tracking	326
	7.15.2	Fluctuation Error of Measurement Taking into Account Nonlinearity	329
\$ 7.16	Concl	usion	335
Chapter VI	II. Ra	nge Finding with an Incoherent Signal	339
§ 8.1	Intro	duction	339
§ 8.2	Optim	um Discriminator	340
	8.2.1	Operations of an Optimum Discriminator	342

1v

12013-00

The lot of the lot of

	8.2.2	Characteristics of an Optimum Discriminator	344
\$ 8.3	Discrin the Si	minators Using Correlation Processing of gnal	3 49
	8.3.1	Discriminator with Two Detuned Channels	3 50
	8.3.2	Discriminator with Switching of the Reference Signals	3 53
	8.3.3	Discriminator with Differentiation of the Reference Signal	3 55
\$ 8.4	Discri	minator with a "Shortening" Filter	356
	8.4.1	Block Diagram of the Discriminator	356
\$ 8.5	Incohe: Modul a	rent Pulse Radiation Without Additional	363
	8,5.1	Discriminator with a Broad-Band UPCh	364
	8.5.2	Square Pulse	365
	8.5.3	Gaussian Pulse	368
§ 8.6	Pulse .	Amplitude Modulation	378
§ 8.7	Freque	ncy Intrapulse Modulation	377
§ 8.8	Phase-	Code Intrapulse Manipulation	384
§ 8.9	Analys: Finder:	1s of Accuracy of Incoherent Range s	396
§ 8.10	Influe: Finder:	nce of Interferences on Incoherent Range s	394
	8.10.1	Active Noise Interference	394
	8.10.2	Pulse Chaotic Interference	395
	8.10.3	Return Interference	396
	8.10,4	Passive Interference	396
§ 8.11	Conclu	sion	403
Chapter IX.	Measu	rement of Speed	407
§ 9 . 1	Introd	uctory Remarks	407
§ 9 . 2	Synthe	sis of an Optimum Frequency Discriminator	409
§ 9 .3	Ch ara c Discri	teristics of an Optimum Frequency minator,	413
§ 9.4	Analys: Discri	is of Certain Circuits of Frequency minators	417
	9.4.1	Circuit with a Tuned Loop and a Phase Shifter	419
	9.4.2	Circuit with Mixers and Differentiation.,	424
	9.4.3	Circuit with Detuned Loops	425
	9.4.4	Circuit with Frequency Variation	429

ţ

i

ł

1

· 1

1

. **)**

ľ

÷

1

Ą

STORT STREET, GAL

100 100

	9.4. Inaccuracy of Reproduction of Modulation of the Signal During Reception	32
	9.4.0 Comparison of Circuits of Discriminators	35
§ 9.5	Smoothing Circuits and Acturacy of Measurement of Speed	37
	9.5.1 Smoothing Circuits with Constant Parameters 4	37
	9.5.2 Optimum Smoothing Circuits for Randomly Varying Speed	.4 4
	9.5.3 Smoothing Circuits with Variable Parameters 4	4 <u>'</u>)
	9.5.4 The Influence of Error: in a Priori Data on Accuracy of Measurement 4	56
\$ 5 . F.	Breakoff of Tracking in Doppler Speed Meters 4	60
5 9.7	Nontracking Speed Metors 4	65
§ 9.8	Measurement of Radial and Cangential Components of Speed as Derivatives of Distance and Angular Coordinates	70
	9.5.1 Accuracy of Measurement of Speed by a System with Constant respectors	71
	9.8.7 Optimum Speed Meter 4	82
§ 4.9	Speed Neters Under the Astim of Interferences	.91
\$ 9.10	Conclusion 4	.96
Chapter X.	Measurement of Angular Data with a Coherent Signal 4	.99
\$ 1.0.1	Introductory Remarks	99
\$ 10.2	Radar Methods of Direction H1 Sing	501
\$ 10.3	Optimum Radio for Goniometers with a Tracking Antenna	503
\$ 10. 4	Method of Scanning of the Firectional Pattern (Optimum Circuits)	518
	10.4.1 Synthesis of Optivum Circuits for the Method of Scanning the Directional Pattern	513
	10.4.2 Equivalent Spectral Density of the Optimum Circuit of the Method of Scanning the Directional Pattern	521
§ 10.5	Investigation of Synthesized Circuits for the Method of Scanning of Directional Patterns	524
	10.4.1 Introductory Remarks 5	524
	10.5.2 Investigation of the Quasi-Optimum Circuit and the Circuit with Narrow-Pand Filters	526
	10.5.7 Investigation of a Circuit with a Broad-Band Filter	534
§ 10. 6	Method of Scanning with Compensation and Instantaneous Amplitude Comparison of Signals (Optimum Circuits)	537
	10.6.1 Synthesis of Optimum Circuits for Methods of Scanning with Compensation and IAC	539

	10.6.2	Equivalent Spectral Densities of Optimum Circuits for Methods of Scanning with Compensation and IAC	542
§ 10.7	Invest: of Sca	igation of Synthesized Circuits for Methods nning with Compensation and for IAC	544
	10.7.1	Investigations of Circuits with Multiplication of Signals	544
	10.7.2	Investigation of the Circuit with Subtraction of Signals for the Method of Scanning with Compensation	546
	10.7.3	Investigation of a Circuit with Subtraction of Signals for IAC	549
§ 10.8	Method	of Phase Center Scanning	551
	10.8.1	Optimum Circuit for the Method of Phase Center Scanning	55 2
	10.8.2	Equivalent Spectral Density of the Optimum Circuit for the Method of Phase Center Scanning	553
	10.8.3	Investigation of the Synthesized Circuit for the Method of Phase Center Scanning	555
§ 10.9	Method	of Instantaneous Phase Comparison of Signals	553
	10.9.1	Optimum Circuits and Potential Accuracy for IPC	558
	10.9.2	Investigation of Synthesized Circuits for IPC	559
§ 10.10) The Me	thoá of Flat Scanning (Optimum Circuits)	561
	10.10.1	Synthesis of an Optimum Circuit for the Method of Flat Scanning	561
	10 .1 0.2	Equivalent Spectral Density of Optimum Circuits for the Method of Flat Scanning	568
§ 10.11	Invest Method	igation of Synthesized Circuits for the of Flat Scanning	570
	10 .11.1	Investigation of a Circuit with Differentiation of Gate Fulses	571
	10.11.2	Investigation of a Circuit with Detuned Gate Pulses	575
§ 10.12	Compar	ison of Methods of Direction Finding	578
§ 10 .1 3	5 Optimu Antenn	m Measurement of Angular Coordinates by as of Phased Array Type	581
§ 10.14	Analys	is of Accuracy of Tracking Radar Goniometers	586
	10.14.1	Influence of a System of Automatic Gain Control on Accuracy of Tracking Goniometers	587
	10.14.2	Errors of Goniometers with Smoothing Circuits with Constant Parameters	589
	10.14.3	Optimum Smoothing Circuits and Smoothing Circuits with Variable Parameters	599

!

日時日

;

ويتساحيكموا المناط يطونون

1

à.

f,

Ş	2.1	5 Nonline	ear Phenomena in Tracking Goniometers	6 02
6	10.1f	Nontra	cking Reder Gonigmeters	EOL
2	- - C - - - C	10 46 1	Ontimum Cincult of a Manimaking Contomaton	
		10.10.1	with IAC	605
		10.15.2	Optimum Circuit of a Nontracking Coniometer with IPC	611
Ş	10.1	7 Influer	nce of Interferences on Coherent Goniometers	615
		10.17.1	Influence of Broad-Bana Interferences on Side Lobes	615
		10.17.2	Influence of Broad-Band Interferences from the Target	6 16
		10.17.3	Influence of Active Interferences with L-F Modulation	619
		10.17.4	Influence of Powerful Intermittent Interferences from Target	622
		10.17. 5	Influence of Fassive Interference	524
ş	10.18	3 Conclu	sion	626
Chap	ter X	I. Messu: Incoh	rement of Angular Coordinates with an erent Signal	629
Ş	11.1	Introdu	uctory Remarks	6 2 9
Ş	11.2	Optimu	m Radio Channel of Incoherent Gonfometers	630
Ş	11.3	Method	of Scanning the Directional Pattern	635
ş	11.4	Method	s of Scanning with Compensation and IAC	644
		11.4.1	Optimum Circuits and Theory Characteristics	644
		11.4.2	Investigation of Circuits with Multiplication of Signals	645
		11.4.3	Investigation of Circuits with Subtraction of Signals	647
Ş	11.5	Method	of Instantaneous Phase Comparison of Signals	649
Ş	11.6	Method	of Flat Scanning of the Directional Pattern	651
Ś	11.7	Analys	is of Accuracy of Incoherent Goniometers	617
Ş	11.8	Influe	nce of Interferences on Incoherent Goniometers	ú58
		11.8.1	Influence of Broad-Band Interferences from Side Lobes	659
		11.8.2	Influence of Broaj-Band Interferences from the Target	659
		11.8.3	Influence of Active Interferences with Narrow-Band Amplitude Medulation	662
Ş	11.9	Conclu	sion	663

i i

)

ć

vili

Chapter XII. Joint Measurement of Several Coordinates	665
§ 12.1 Introduction	665
§ 12.2 Analysis of Multi-Dimensional Tracking Meters	666
12.2.1 Basic Features of Construction of Circuits and Components of Measuring Errors	666
12.2.2 Characteristic of Discriminators	669
12.2.3 Accuracy of Measurement	676
§ 12.3 Synthesis of Multi-Dimensional Meters of Several Parameters	679
12.3.1 Method of Synthesis of an Optimum Meter	679
12.3.2 Likelihood Function and Its Approximations	681
12.3.3 Optimum Tracking Multi-Dimensional Meter	683
12.3.4 Optimum Nontracking Multi-Dimensional Meter	691
12.3.5 Optimum Meter of Linear Functionals of Parameters	693
12.3.6 Optimum Meter of Markovian Parameters	696
§ 12.4 Synthesis of Multi-Dimensional Discriminators	700
12,4,1 Regular Signals in White Noises	701
12.4.2 Gaussian Coherent Signals	703
12,4,3 Gaussian Uncorrelated Sendings	707
12.4.4 Methods of Approximation for Construction of Discriminators and the Method of Comparison of Circuits	709
§ 12.5 Synthesis of Smoothing Circuits and Resultant Accuracy of Measurement	710
12.5.1 Parameters - Random Processes with Stationary Increments	, 710
12.5.2 Parameters — Linear Combinations of Known Functions with Random Factors	717
§ 12.6 Examples of Multi-Dimensional Meters	719
12,6,1 Measurement of Range and Speed by Coherent Radar	719
12.6.2 Measurement of Range and Speed by Incoherent Radar	723
§ 12.7 Conclusion	732
Chapter XIII, Resolving Power	735
§ 13.1 Introductory Remarks	735
§ 13.2 Synthesis of a System of Optimum Separation	11 1 1
of Signals	•••• 739
13.2.1 Solving the Problem in a General Form	•••• 739

.

• •

4 4

Į

「「「「「「」」」

nalan den de la contraction de la contr

	13.2.2	Separation of Signals by Range	746
Ş	13.3 Statis Resolu	tical Synthesis of Optimum Systems of	759
Ş	13.4 Optimu	m Detection of an Assigned Set of Targets	7 67
\$	13.5 Synthe of Mea	sis of Systems of Resolution in Conditions surement of Coordinates	770
	13.5.1	Formulation of the Problem	7 70
	13.5.2	Maximum Likelihood Estimates of Multi-Dimensional Farameters and Their Asymptotic Efficiency. General Properties of an Optimum Discriminator	773
	13.5.3	Optimum Discriminator in the Case of Many Targets with a Large Signal- to-Noise Ratio	782
	13.5.4	Optimum Discriminator in the Case of Two Targets	792
Ş	13.6 Analys with S	is of Systems of Detection and Measurement Suppression of Disturbing Signals	800
	13.6.1	System of Detection	800
	13 . 6 . 2	System of Measurement with Complete Suppression of Selected Interfering Signals	805
	13.6.3	System of Measurement of Parameters of Motion of Two Targets	809
	13.6.4	Meter of Range with Two Targets for One Particular Form of Signal Modulation	813
Ş	13.7 Conclu	usion	819
ter	ature for Vol	.ume II	823

х

1000

U. S. BOARD ON GEOGRAPHIC NAMES TRANSLITERATION SYSTEM

Block	Italic	Transliteration	Block	Italic	Transliteration
A a	A a	A, a	РР	Рр	R, r
Бб	Бб	B, b	Сc	C c	S, s
B 🛚	B •	V, v	Τт	T m	T,
Γr	5 7	G, g	Уу	Уу	U, u
Дд	Дд	D, ā	Φφ	Φφ	F, f
E e	Ë 🔹	Ye, ye; E, e*	XX	XX	Kh, kh
жж	Жж	Zh, zh	Цц	Ц 4	Ts, ts
3 3	3 3	Z, z	મુષ	4 4	Ch. ch
Ин	Ни	I, i	шш	Ши	Sh. sh
Яя	A 4	Y, y	Щщ	Щщ	Shch, shch
КК	Κĸ	K, k	Ъ ъ	Ъъ́	11
Лл	Лл	L, 1	ыы	Ыы	Y, y
Мм	Мм	M, m	Б 🖌	Бь	t
Нн	Нж	N, n	Э 🕨	э,	E, e
0 0	0 0	0, 0	юŨ	Юю	Yu, yu
Пп	ГГ п	P, p	R R	Яя	Ya, ya

* ye initially, after vowels, and after b, b; e elsewhere. When written as ë in Russian, transliterate as yë or ë. The use of diacritical marks is preferred, but such marks may be omitted when expediency dictates.

xi

FOLLOWING ARE THE CORRESPONDING RUSSIAN AND ENGLISH

DESIGNATIONS OF THE TRIGONOMETRIC FUNCTIONS

Russian	English
sin :	ain
005	COS
tg	tan
ctg	cot
80C	80C
COSEC	CSC
sh	sinh
ch	cosh
th	tanh
cth	coth
sch	sech
csch	cach
arc sin	sin ⁻¹
arc cos	cos ⁻¹
arc tg	tan-1
arc otg	\cot^{-1}
arc sec	sec ⁻¹
arc cosec	CBC ⁻¹
arc sh	sinh ⁻¹
arc ch	cosh-1
arc th	tanh-1
arc oth	coth-1
arc ach	sech-1
arc cach	csch-1
	<u></u>
rot	curl
1g	log
	3

.This second volume is dedicated to the theory of radar measurements and questions of target resolution.

There is developed a general theory of radar measurements, containing an analysis of tracking and nontracking measuring systems both in linear approximation, and also taking into account their nonlinearity, and also synthesis of optimum systems of measurement of separate time-variable parameters of motion of targets and sets of them.

On the basis of this theory there is conducted analysis and synthesis of range finding systems, systems of measurement of velocity and goniometrical systems. There are investigated cases of reception of both coherent, and also incoherent signals.

In examining questions of target resolution there are investigated possibilities of resolution of signals reflected from them, and there are found receivers optimum from this point of view, and also optimum systems of resolution in regimes of detection and measurement of coordinates.

In the course of development of theory there are revealed laws governing radar measurements and resolution of targets. $_{/\,\,\times}$

The book is intended for scientists and engineers studying questions of radar, and also for post graduates and students of corresponding specialities. Many questions of the developed general theory are also of interest to persons studying theoretical problems in all regions based on the theory of statistical solutions, in particular, in the area of automatic control.

í

MT-65-47 Questions of the Statistical Theory of Radar, Vol. II, "Soviet Radio," Moscow, 1964. Pages: Cover-1079

CHAPTER VI

GENERAL LAWS GOVERNING RADAR MEASUREMENTS

§ 6.1. Introduction

In the preceding chapters we considered questions of radar detection of targets. With this chapter we start a presentation of questions of radar measurements to which the subsequent chapters (7-12), too, are devoted. Before studying concrete schemes for construction of radars in a regime of measurement of coordinates of targets, dividing them by the form of measured coordinates (parameters of motion) of targets, we shall turn in this chapter to certain general laws, peculiar to radar measurements. The fact is that it is possible in fairly general form to theoretically give a basis for methods of analysis and synthesis of the most diverse radar meters. With certain, not too limiting, assumptions it is also possible to obtain basic performance characteristics of any meters and general laws governing radar measurements, we shall consider two large groups of questions, namely: analysis of meters whose circuit is known, and synthesis of meters ensuring the best possible performance characteristics for measurement.

In order to grasp the domain of applicability of subsequent results, it is necessary first of all to deal with the question of the interrelation between the regime of detection studied above and the regime of measurement. Ultimately, any radar is intended for measurement of various parameters of motion of targets. Therefore, measurement is a basic operation of radar and should start from the very beginning of work of the system.

However, under conditions when the very fact of the presence of a signal

reflected from target still has not been fixed, consistent carrying out of this principle in theory leads to difficulties both methodological and mathematical. Therefore, everywhere in radars there is implicitly assumed the presence of the devices of detection and lock-in, which were analyzed in Chapters 3-5. These devices in one way or another produce the initial determination of parameters with precision, allowing us to pass to precision measurement (tracking). From what follows it will follow that in the case of application of tracking meters for this transition we need lock-in for all coordinates with accuracy of the width of discrimination curves of the meters, comprising the radar set.

It is important to indicate that in reality establishment of the fact of the presence of the target for finite time of lock-in is possible only with a finite probability, differing from unity. Everywhere below we shall consider that if lock-in occurred, a signal from the target indeed exists, and all qualities of the meters will be calculated on this assumption.

We shall also discuss the statistical character of the studied problem. Coordinates of objects, measured by radars, are included ("coded") in parameters of the signal reflected from the target or radiated by the target. Thus, the range of a target is usually coded in the time delay of modulation; velocity in the Doppler frequency shift; angular coordinates in modulation introduced by the receiving antenna. In Chapter 1 we indicated that a reflected radar signal has a random character. Various interferences of radar also have a fluctuating nature. These circumstances already compel us to describe the process of measurement of coordinates in statistical categories. However, in the problem of measurement it is impossible to do without one more aspect of the problem — change of coordinates in time. Actually, in practical conditions interesting us the located object is always moving, and the necessity of measurement appears, essentially, when we do not exactly know parameters of this motion. Here, it is possible to suggest two methods of description of the varying parameters:

1) variation of parameters are considered statistically, given the distribution of probabilities of their values at different moments of time or more limited statistical evidence;

2) unknown functions describing motion are considered determined, and the available information is presented in the form of a series of mathematical conditions (limitations), imposed, for instance, on the derivatives of the

-2-

functions with respect to time.

Both approaches have their advantages and disadvantages. A disadvantage of the nonstatistical approach is the necessity to judge the quality of measurement by certain of the worst cases, which, possibly, will never be observed in practice. A basic deficiency of the statistical approach is the so-called "a priori difficulty" in determining statistical properties of parameters. The fact is that their variation is determined sometimes by a great number of factors, including psychological (for instance, behavior of the operator or pilot). Determination of statistical properties, implying mass trials, is difficult even in idealized conditions, and is sometimes simply impossible in the absence in the past of similar systems or conditions of their use. However, it is possible to give a series of no less convincing arguments in favor of the statistical approach:

1) there exist a whole series of applications where statistics can be considered fully given. An example can be the case of a Doppler meter of groundspeed of an aircraft, where stationary variations of speed during a large number of flights can be statistically studied in sufficient detail;

2) satisfactory results in the sense of accuracy of measurement are given by meters, built only with qualitatively correct allowance for statistical properties of the parameter, inasmuch as namely the qualitative aspect determines the meter's functional scheme. Quantitatively circuit elements can be regulated in the process of tuning and testing.

For meters it is possible to consider the statistical approach more consistent.

Subsequently we will use both methods of describing variation of parameters, leaning in most cases toward the statistical. Practically, during designing of each new system of concrete assignment there always exists a considerable amount of information and physical considerations, albeit not rigorous, which are permissibly interpreted as statistical properties of the measured quantity.

Thus, the problem of radar measurement of coordinates has a statistical sharacter both due to the random character of the mixture of the input signal and interferences, and also due to changes of measured quantities, unpredictable in general. In the course of the following presentation it is frequently necessary, however, to isolate an ensemble of parameters from the ensemble of the mixture of the signal and interferences. The latter, in turn, are sometimes conveniently divided into

-3-

subensembles.

During consideration of general questions of analysis and synthesis of radar meters in this chapter there is adopted the following order of presentation. First of all in § 6.2 there is conducted analysis of the most wide-spread tracking meters; there are introduced into consideration necessary characteristics of their elements; there is developed a method of investigation of the accuracy of measurements; and there are given results of analysis of accuracy.

In § 6.3 there are investigated these same meters under the action of intense interferences, leading to breakoff of tracking. In practice there are also applied nontracking meters; questions of their analysis are the contents of § 6.4.

Remaining sections of the chapter are devoted to questions of synthesis of meters. In § 6.5 there are discussed possible criteria and methods of synthesis. There are presented propositions of the general theory of statistical solutions and its separate branches necessary for problems of measurements. Further, in § 6.6, having for subsequent considerations very great importance, there are expounded methods, and there are presented general results of statistical synthesis of optimum meters, based on close approximation of the introduced probability characteristics.

The next two sections are devoted to synthesis of separate parts of an optimum meter — discriminators (§ 6.7) and smoothing circuits (§ 6.8), with different statistical properties of input signals and measured quantities. The physical nature of measured parameters, here, is not made concrete; these questions are saved for subsequent chapters.

In § 6.9 there is given a certain expansion of the method of synthesis in the direction of parameters with Markov variation. In the same place there is given a series of results of synthesis for a limited knowledge of the statistics. The conclusion (§ 6.10) gives a brief survey of methods and results and contains the formulation of a series of new problems.

§ 6.2. General Analysis of Tracking Meters

At present there exists a large number of various forms of radar measuring devices. High accuracy of measurement is usually attained by automatic-tracking meters, built on the principle of a servo system. Meters of this type will be studied in the present section.

-4-

6.2.1. Basic Features of Circuit Construction and Components of Errors of Measurement

In the general form tracking meters can be represented by the schematic of Fig. 6.1. On it there are marked two basic elements - element for separation of



Fig. 6.1. General schematic of a tracking meter: 1) discriminator; 2) smoothing circuits and drive unit.

语言:PSL化学于信兰·17 Warships

the signal of mismatch between current and measured values of a parameter, usually called a discriminator, and amplifyingsmoothing circuits together with drive units. Sometimes in the circuit there is also provided

insertion through an adder of a certain additional voltage $\lambda_{\rm B}$ (t), intended for compensation of clearly known components of the measured quantity. This can be carried out by timers, transducers of coordinates of the actual meter, by output data of roug¹ means of preliminary measurement, and so forth. Numerous examples of range finder: goniometers and speedometers, which can be reduced to the scheme of Fig. 6.1, are available in the literature, and also will be considered below in Chapters 7-11. Here we shall only indicate that the division of the whole circuit into two basic elements, to the second of which there belongs all the basic inertia, in the vast majority of cases can be carried out, at least in principle.

From the given description of the circuit of a mater it may be concluded that such types of automatic systems have very wide use in different areas of technology. Diverse control systems for industrial and military purposes, power systems, and so forth, include closed control circuits, separate elements of which are connected by communication circuits. During transmission of information in communication circuits there is produced its encoding and decoding in conditions of noises. If to investigation of such systems we bring the apparatus of the classical theory of automatic control, the devices for transmission of information usually are considered crudely simplified in the form of inertialess (sometimes linear) elements with additively imposed noises. For radar, however, as for many other contemporary areas, consideration of devices for processing data carried by a high-frequency signal (in this case, of discriminators) turns out to be absolutely obligatory. Here, this is no less important than to analyze smoothing and drive circuits. In this is the peculiarity of the subsequent analysis.

Tracked parameters in radar meters are angular coordinates, distance and speed,

-5-

concretely coded in time and in frequency shifts and other parameters of signal modulation. Input and output signals of a discriminator most frequently have an electrical nature, i.e., are certain voltages. Therefore, it is reasonable to introduce a binary system of designations, which will also be convenient in the theory of optimization of systems. Let us assume that $y(t; \lambda(t))$ and $z(t; \varepsilon(t)) =$ input and output voltages of the discriminator; $\lambda(t)$ and $\hat{\lambda}(t)$ are the input and output voltages of the discriminator; $\lambda(t) = \lambda(t) - \hat{\lambda}(t) - \text{current mismatch}$ (error). By $y(t; \lambda(t))$ we understand the whole mixture of useful signal and interferences at the meter input; an analogous mixture, obtained after processing in the discriminator, is implied by $z(t; \varepsilon(t))$.

In general to the discriminator input there proceed several input mixtures $y_{i}(t; \lambda(t))$ (i = 1, 2, ...), which, naturally, changes neither the further methods of analysis nor its results.

It is important to indicate that between $y_i(t; \lambda(t))$ (i = 1, 2, ...) and $z(t; \varepsilon(t))$, on the one hand, and their parameters $\lambda(t)$ and $\varepsilon(t)$, on the other, there does not exist a one-to-one correspondence. Thus $z(t; \varepsilon(t))$ is proportional to $\varepsilon(t)$ only on the average (over the ensemble of input signals), and in a narrow range of values of ε statistical properties of fluctuations in $z(t; \varepsilon(t))$ to a great degree depend on the intensity and form of the input interference included in mixtures $y_i(t; \lambda(t))$. The output value, i.e., result of measurement can be characterized by function $\hat{\lambda}(t)$.

The rest of this paragraph will be devoted to a basic characteristic of the described meters - accuracy of measurement. In this connection one should separate different components of measurement error.

In the first place fluctuating error is important. It a pears due to internal noises of the system (noises of transmitters, receivers, antennas), external interferences of varied origin, most frequently additively joined with the useful input signal (but not with the measured quantity), and also due to fluctuations of the reflected radar signal. Furthermore, the measured variable can be coded in the signal in sum with a certain spurious component. We have in mind, for instance, the phenomenon of random shifts of the center of reflection of an extended radar target in angular coordinates and distance, distortion of the beam and dispersed propagation of radio waves in the atmosphere, etc. Inasmuch as the goal is measurement of a certain undistorted value of a coordinate, these factors introduce

-6-

additional error, which can have both a fluctuating, and also a constant component.

The second component of meter error is dynamic error due to changes of the measured magnitude itself, both random and those which are regular, but not compensated inside the meter.

We also distinguish systematic error. This is the usually somewhat uncompensaled constant component of error, explained by the selected method of processing the signal.

In high-performance meters it is necessary to consider increase of fluctuating and systematic errors due to so-called instrument (equipment) errors, appearing due to oscillations of feed voltages, mechanical vibrations, fluctuations of temperature, gaps, dry friction, errors of quantization in digital variants of smoothing circuits, and so forth.

Henceforth, for simplicity we assume that there are no errors due to spurious components of the measured variable nor instrument errors where the forms of errors must be accounted for separately.

During the study of fluctuating and dynamic errors it turns out that an essential role in their calculation is played by the relationship between the rate of change of the measured parameter on the one hand, and on the other, the rate of change of random variables in the mixture of the signal with interferences (we call them for brevity fluctuations), which in principle there is no need to measure. In the most general case there certainly must exist rapid fluctuations due to internal noise of the receiver. Furthermore, there may exist random variables, the rate of change of which is commensurate with that for the measured parameter. Inasmuch as the approach during analysis and the method of calculation of error of measurement somewhat differ for various combinations of these variables, below we distinguish three separate cases:

- a) case of rapid fluctuations alone;
- b) case of a set of rapid and very slow fluctuations;
- c) general case.

6.2.2. Characteristics of a Discriminator in the Case of Rapid Fluctuations

Here, it is useful to introduce a single statistical ensemble of fluctuations, in general determined by fluctuations of varied physical nature (see Chapter 1). Width of the spectrum of fluctuation in $z(t; \epsilon(t))$ is assumed larger than the

-7-

effective transmission band of the whole closed system of the meter, and all processes in the discriminator are considerably faster than in the meter as a whole.* Therefore, function $z(t, \epsilon)$ can be considered to consist of two parts — a mean value and certain noise. For a continuous input signal the mean value is equal to

$$a(\mathbf{e}, \mathbf{i}) = \overline{z(\mathbf{i}, \mathbf{e})},$$
 (6.2.1)

and noise has correlation function

$$R(\tau; \bullet, t) = \overline{[z(t, \bullet) - \overline{z(t, \bullet)}] [z(t + \tau, \bullet) - \overline{z(t + \tau, \bullet)}]}.$$
(6.2.2)

Due to inertia of subsequent circuits this noise can be considered white. Its spectral density one should consider equal to the spectral density of output fluctuations of the discriminator at low frequencies, introducing here the characteristic

$$S(\mathbf{e}, t) \equiv S(\mathbf{e}; \mathbf{e}, t)|_{\mathbf{e}=0} = \int_{-\infty}^{+\infty} R(\tau; \mathbf{e}, t) d\tau.$$
 (6.2.3)

Averaging in (6.2.1) and (6.2.2), designated by the vinculum, is produced over the complete ensemble of fluctuations at the input, and magnitude ε here is considered fixed, which corresponds physically to slowness of its change. Possible dependence of $\mathbf{a}(\varepsilon, t)$, $\mathbf{S}(\varepsilon, t)$ on time is explained by the fact that statistical properties of input fluctuations may depend on time, but slowly and by known law.

For a pulse periodic signal it is possible to characterize $z(t, \varepsilon)$ by the same functions $a(\varepsilon, t)$, $S(\varepsilon, t)$, if we include in (6.2.1) and (6.2.2) additional averaging in time (for period of modulation T_n):

$$\boldsymbol{a}(\boldsymbol{\bullet},t) = \frac{1}{T_r} \int_{t-T_r} \overline{\boldsymbol{z}(t_1,\boldsymbol{\bullet})} dt_1, \qquad (6.2.4)$$
(6.2.4)

$$S(\bullet, t) = \frac{1}{T_r} \int_{t-T_r}^{t} dt_1 \int_{-\infty}^{\infty} d\tau R(\tau; \bullet, t_1).$$

If we now compose from characteristics $a(\varepsilon, t)$, $S(\varepsilon, t)$ a certain statistical equivalent of function $z(t, \varepsilon)$, this equivalent, obviously, will have form

$$z_{0}(t, 0) = a(i, 1) + \sqrt{S(0, 1)}\zeta(t), \qquad (6.2.6)$$

where $\zeta(t)$ - white noise of unit spectral density.

Actually, $z_{\alpha}(t, \epsilon)$ has identical properties at low frequencies with $z(t, \epsilon)$,

^{*}If smoothing circuits change in time, then the idea of an effective band is not applicable, but it nevertheless is possible to speak about a certain effective constant reaction time.

and this suffices for further analysis.

We introduce now for $a(\varepsilon, t)$ the name discrimination characteristic and for $S(\varepsilon, t)$, fluctuation characteristic of the discriminator. They are found by means of analysis of passage of the signal and different interferences through the discriminator for fixed mismatch ε . This is a separate problem, sometimes very bulky (see Chapters 7-11). Let us indicate that in nonstatistical consideration the discriminator characteristic is simply the dependence of output voltage of the discriminator on mismatch at a nominal level of input signal, and the fluctuation characteristic is not introduced in general.

If the signal has a pulse character, and fluctuating disturbances formed at the discriminator output in different periods are not correlated, it is sometimes more convenient to present the output of the discriminator in the k-th period in a form which is a discrete analog of (6.2.6):

$$\mathbf{z}_{k}(\mathbf{e}) = a_{k}(\mathbf{e}) + \sigma_{k}(\mathbf{e})\zeta_{k}, \qquad (6.2.7)$$

where $\zeta_k - a$ discrete random process with uncorrelated values and unit variance; $a_k(\epsilon) - mean value of z_k;$

 $\sigma_k^2(\epsilon)$ - variance of the fluctuating component in the k-th period (dependence of functions $a_k(\epsilon)$ and $\sigma_k^2(\epsilon)$ on k, as also of functions $a(\epsilon, t)$, $S(\epsilon, t)$ on time t, again is considered slow and regular).

Let us stress, however, that writing (6.2.7) on the assumption of smallness of the period of repetition T_r as compared to the time of change of the measured quantity not only does not lead to greater rigor of consideration as compared to (6.2.6), but is also valid in the more frequent case of absence of correlation between disturbances in separate periods. As it is easy to check, formal transition from (6.2.7) to (6.2.6) should be carried out by replacement of current subscript : by current time t, of process with uncorrelated values ζ_k by white noise $\zeta(t)$ of unit spectral density and of magnitude $\sigma_k^2(\varepsilon)$ by $S(\varepsilon, t_k) = \sigma_k^2(\varepsilon)T_r$.

Thus, the notation (6.2.6) in the considered conditions is sufficiently general, and it remains to consider in more detailed form the form of characteristics of the discriminator, assuming, for simplicity, that they do not depend on time.

With satisfaction of conditions of symmetry of the circuit and of characteristics of the mixture $y(t; \lambda(t))$ the discrimination characteristic is an odd, and the fluctuation characteristic is an even function of ε . The discrimination characteristic (Fig. 6.2) for small ε has a linear section. We usually seek

-9-



Fig. 6.2. Typical family of discrimination curves: 1) case of small interferences; 2) case of large interferences.

increases of the extent and slope of this section; however, this does not lead in all cases to lowering of errors of measurement. For large ε the discrimination characteristic has dips corresponding to emergence of the parameter beyond the limits of the region in which the discriminator is selective.

加加には、

「「「「「「「「」」」

Analysis of concrete circuits shows that intense interferences lead to decrease of the scale of the discrimination curve along the axis of ordinates and to amplification of dips. This is explained by the normalizing action of certain, generally speaking, nonlinear elements (system of automatic gain control (AGC), clippers, peak detectors, and so forth, see Chapter 2).

The fluctuation characteristic (Fig. 6.3) to a still greater extent depends on the form and level of input interferences. Output fluctuations of a discriminator



Fig. 6.3. Typical family of fluctuation characteristics: 1) case of small interferences; 2) case of large interferences.

consist of several components. They appear as a result of interaction among spectral components of interference, beats of interference with the signal, and, in the case of the presence of signal fluctuations, results of interaction of signal components.* Intensity of the last two components depends on ε , increasing where the control

signal, i.e., the modulus of the discrimination characteristic is great. Intensity of beats of interference components among themselves also depends on ε , but only

^{*}Strictly speaking, division of output fluctuations into the three shown components can be performed only when using in the discriminator circuit, as elements of nonlinear processing, square-law peak detectors and ideal multipliers. Qualitatively the picture also remains, however, true in other cases, at least for large signals.

due to the normalizing action of the discriminator, overwhelming noises for strong signals and small ε . We have in mind, for instance, decrease of gain of the receiver, and consequently also of the output level of noises due to action of an AGC system with sufficiently accurate tuning on a powerful signal. As a result the fluctuation characteristic frequently has the form of a double-humped curve with a uip near $\varepsilon = 0$, has rises near extremes of the discrimination characteristic and has a certain constant level for very great mismatch ε (Fig. 6.3). Here, S(ε) as $\varepsilon \to \pm \infty$ is determined only by internal noises of the receiver and interferences. With growth of interferences not carrying information about the measured variable, function S(ε) gradually is smoothed, until it becomes independent of ε and of the signal-to-interference ratio at the discriminator input.

The described dependences in subsequent chapters will be illustrated by a series of examples. They take place under the above-indicated conditions of symmetry. In other cases the form of the characteristics of the discriminator may vary somewhat. In particular, shift of both characteristics along the axis ε without change of the shape of the curves is possible, so that the new center of symmetry will be point $\Delta \neq 0$. Then we speak of systematic error, introduced by the discriminator. In general, shift is accompanied by distortion of the shape of the curves, so that they no longer have the shown simple form. This occurs, for instance, with the presence along with the useful signal of an interfering signal close in structure.

Presentation of discriminator output in form (6.2.6) should be used under conditions when mismatch ε can take large values, i.e., during the study of questions of lock-in and breakoft (§ 6.3). At a sufficiently low level of interferences random process $\varepsilon(t)$ with high probability takes small values, and approximation of characteristics of the discriminator by very simple functions near point $\varepsilon = 0$ is sufficient:

$$a(\mathbf{e}) = a_{\mathbf{e}} + K_{\mathbf{A}}\mathbf{e}, S(\mathbf{e}) = S_{\mathbf{e}} + S_{\mathbf{i}}\mathbf{e} + S_{\mathbf{i}}\mathbf{e}^{2},$$
(6.2.8)

where

$$a_0 = a(0), K_A = \frac{da(0)}{de}, S_A = \frac{1}{k!} \frac{d^A S(0)}{de^3} (k=0, 1, 2),$$

Ratio $\Delta = a_0/K_{ff}$ is the systematic error of the discriminator. When conditions of symmetry are satisfied, quantities a_0 and S_1 turn into zero and (6.2.8) takes the

-11-

form

 $a(e) = K_{A}e, \ S(e) = S_{e} + S_{g}e^{2}. \tag{6.2.9}$

Coefficient K_{μ} is called the amplification or gain factor of the discriminator. Quantity S_0 characterizes the fluctuating component, not depending on mismatch ε , and S_2 — the component, proportional to ε . For explanation we indicate that (6.2.9) corresponds to the following presentation of discriminator output voltage:

$$z_{b}(l; \bullet) = K_{A}\eta(l) + K_{A}(1 + \xi(l)) \bullet. \qquad (6.2.16)$$

In this case random process $\sqrt{S(\varepsilon)\zeta(t)}$ is replaced by two uncorrelated processes $K_{II}\eta(t)$ and $K_{II}\varepsilon\xi(t)$ with the same total intensity. Here $\eta(t)$ — white noise with dimensionality of the measured parameter and with spectral density proportional to S_{O} :

$$S_{q} \equiv S_{axa} = S_{q}/K_{A}^{2},$$
 (6.2.11)

henceforth called the equivalent spectral density, and $\xi(t)$ - a parametric input in the form of white noise with spectral density, proportional to S_0 :

$$S_{t} = S_{map} = S_{t} / K_{a}^{t}$$
, (6.2.12)

henceforth called parametric.

Thus, under the formulated conditions of symmetry output of the discriminator can be characterized by only three quantitie: $K_{\rm R}$, $S_{\rm SECO}$ and $S_{\rm Rap}$. Their dependence on the character of the input signal (signals) y(t; λ) and the method of construction of the discriminator in concrete examples will be clarified in subsequent chapters. Here, we indicate that the gain factor $K_{\rm R}$ drops with growth of interfences which are not contained in the coded form of the tracked parameter, simultaneously with the above-indicated decrease of the scale of curve $a(\varepsilon)$. Quantity $S_{\rm SEO}$ also depends on the level of interferences: for weak interferences it is close to zero, and with growth of interferences it grows without limit. Quantity $S_{\rm Rap}$ characterizes the coefficient of parameteric pulsations of control voltage, caused usually by amplitude fadings of the signal. In goniometric and range finding system $S_{\rm Rap}$ is considerably lowered when there are present amplitude-demodulators (AGC, clippers, ctc). The dependence of $S_{\rm Rap}$ on the level of input interferences is determined by the concrete nature of $\lambda(t)$ and the circuit of the discriminator.

It is easy to prove that the introduced magnitudes S_{axe} and S_{map} are two coefficients of the expansion with respect to ϵ^2 of function $S_{axe}(\epsilon) = S(\epsilon)/K_{\mu}^2$, which is sometimes convenient to use instead of $S(\epsilon)$, called the equivalent fluctuation

-12-

characteristic. This term indicates that $S_{acc}(\varepsilon)$ characterizes noises, recalculated in equivalent values of the tracked variable.

Relationship (6.2.10) is conveniently interpreted by the circuit of Fig. 6.4, which can be used during simulation of a complicated system, where a given



Fig. 6.4. Equivalent circuit of a discriminator: K_{π} - inertialess amplifier with gain factor K_{π} .

discriminator is applied. Note that S map characterizes that component of fluctuations, which in the given equivalent circuit parametrically governs amplification of the radio channel.

Thus, even in the case of linear smoothing circuits with constant parameters the radar tracking meter turns out to possess variable gain in the loop, i.e., one

variable parameter. In this connection the fluctuating component, proportional to mismatching, we call parametric. One should not, of course, confuse this component with random changes of the tracked parameter $\lambda(t)$.

It is necessary, however, to stipulate that in a number of combinations of parameters of the circuit and the input signal-to-noise ratio it turns out that $S_{p} < 0$, i.e., the level of noises upon the appearance of mismatch even decreases somewhat. Examples of this especially nonlinear effect will be given in subsequent chapters. When $S_{\gamma} < 0$ presentation (6.2.10) is already impermissible, and the linear model of Fig. 6.4 loses meaning. The problem can be studied only by a mathematical apparatus, adapted to nonlinear problems.

6.2.3. Accuracy of Measurement in the Absence of Parametric Fluctuations

On the basis of the introduced discriminator characteristics we shall calculate the error of measurement. Let us assume that parametric fluctuations are negligible. We shall study that important and sufficiently general case, when smoothing circuits are linear, but not necessarily with constant parameters, so that their input and output are connected by relationship

$$\hat{z}(t) = \int_{0}^{t} h(t, \tau) z(\tau) d\tau + \lambda_{\mu}(t),$$
 (6.2.13)

where $h(t, \tau)$ - pulse response of smoothing circuits;

 $\lambda_{B}(t)$ - quantity introduced into the meter (Fig. 6.1), not necessarily accurately coinciding with the a priori mean value of parameter $\lambda(t)$, which is considered a random process.

Substituting in (6.2.13) according to (6.2.10) quantity

-13-

$\mathbf{z}_{\mathbf{x}}(t) = K_{\mathbf{x}} \mathbf{e}(t) + K_{\mathbf{x}} \mathbf{e}(t)$

and considering that $\ddot{\lambda} = \lambda - \epsilon$, we have equation

$$*(l) + K_{\mathbf{x}} \int_{0}^{1} h(l, \tau) * (\tau) d\tau = \lambda(l) - \lambda_{\mathbf{x}}(l) - K_{\mathbf{x}} \int_{0}^{1} h(l, \tau) \eta(\tau) d\tau. \qquad (6.2.14)$$

We introduce the pulse response $g(t, \tau)$ of the closed system, considering $\hat{\lambda}(t)$ the output quantity. Function $g(t, \tau)$ is determined by integral equation

$$g(l, \tau) + K_{x} \int h(l, s) g(s, \tau) ds = K_{x}h(l, \tau).$$
 (6.2.15)

Additionally we introduce pulse response $v(t, \tau)$, considering mismatch $\varepsilon(t)$ the output quantity. Function $v(t, \tau)$ satisfies equation

$$v(l, \tau) + K_x \int h(l, s) v(s, \tau) ds = \delta(l-\tau).$$
 (6.2.16)

By direct substitution of expression (6.2.13) in (6.2.14), using (6.2.15) and (6.2.16), it is simple to prove that the solution of equation (6.2.14) has the form

$$e(t) = -\int_{t_0}^{t} g(t, s) \eta(s) ds + \int_{t_0}^{t} v(t, s) [\lambda(s) - \overline{\lambda(s)}] ds + \int_{t_0}^{t} v(t, s) [\overline{\lambda(s)} - \lambda_s(s)] ds = e_{\phi, \pi}(t) + e_{ABH_1}(t) + e_{ABH_2}(t), \qquad (6.2.17)$$

where function $\overline{\lambda(t)}$ is added and subtracted.

According to (6.2.17) current error is determined by the joint action of interference $\eta(t)$, passed through the filter with pulse response g(t, s), of random changes of parameter $\lambda(s) - \overline{\lambda(s)}$, and of uncorrected regular changes of parameter $\overline{\lambda(s)} - \lambda_{B}(s)$, passed through the filter with pulse response v(t, s). The first component

$$u_{\oplus \mathbf{z}}(t) = \int_{t_0}^{t} g'(t, s) \eta(s) \, ds$$

is naturally called fluctuation error; to terms $\varepsilon_{\text{ДИН 1}}(t)$, $\varepsilon_{\text{ДИН 2}}(t)$ we give the designation of dynamic errors. Let us note, 'nowever, that in the statistical approach to measured quantities introduction of the last term is somewhat conditional, inasmuch as $\varepsilon_{\text{ДИН 1}}(t)$ has a random character to the same extent as $\varepsilon_{\text{ФЛ}}(t)$.

Considering that the measured parameter is a random process, and that fluctuations are rapid, by (6.2.17) it is possible to easily calculate the mean square overall measuring error:

Here we use the property of 5-correlation of interference, and there is introduced correlation function $R_{\lambda}(t_1, t_2)$ of the random part of the measured parameter and we denote $\Delta(t) = \overline{\lambda(t)} - \lambda_{R}$ (t).

According to (6.2.18) for an arbitrary pulse response of smoothing circuits the variance of fluctuation error $\sigma_{\bar{Q}\bar{M}}^2(t)$ is proportional to the equivalent spectral density (6.2.11).

Further simplifications of expressions for separate components of expression (6.2.18) are possible upon concretization of the form of smoothing circuits and of the character of change of $\lambda(t)$.

a) Assume stationariness of the random part of parameter $R_{\lambda}(t_1, t_2) = R_{\lambda}(t_1 - t_2)$, full correction of its regular part ($\Delta(t) = 0$) and constancy of parameters of smoothing circuits. Fourier transforms from functions $g(t, \tau) = g(t - \tau)$ and $v(t, \tau) = v(t - \tau)$ with respect to (6.2.15) and (6.2.16) in this case are easily expressed in terms of the amplification factor of the discriminator and the frequency response of the smoothing circuits

$$H(lo) = \int_{0}^{\infty} h(\tau) e^{-i\omega \tau} d\tau$$

in the form

$$G(i\mathbf{o}) = \frac{K_A H(i\mathbf{o})}{1 + K_A H(i\mathbf{o})},$$

$$V(i\mathbf{o}) = \frac{1}{1 + K_A H(i\mathbf{o})}.$$
(6.2.19)

Finally instead of (6.2.18) we can obtain

$$e_{MAX}^{2} = e_{\phi A}^{2} + e_{AXM}^{2} = 2S_{2KB}\Delta f_{2\phi} + \frac{1}{2\pi} \int_{-\infty}^{+\infty} \frac{S_{\lambda}(\omega) d\omega}{(1 + K_{\lambda}H(i\omega))^{2}}$$
(6.2.20)

where $S_{\lambda}(\mathbf{e}) = \int_{\mathbf{e}}^{\infty} R_{\lambda}(\mathbf{t}) e^{-i\mathbf{e}\cdot\mathbf{t}} d\mathbf{t}$ - spectral density of the random part of parameter $\lambda(\mathbf{t})$;

 $\Delta f_{\partial \tilde{M}}$ - effective transmission band of a closed-cycle meter:

$$\Delta j_{ab} = \frac{1}{2\pi} \int_{0}^{+\infty} |G(i\omega)|^{a} d\omega = \frac{1}{2} \int_{0}^{\infty} g^{a}(t) dt. \qquad (6.2.21)$$
It would be more systematic to determine the effective band by normalized relationship

$$\Delta f_{ab} = \frac{1}{2\pi} \int_{-\infty}^{+\infty} \frac{|G(i\omega)|^2}{|G(0)|^2} d\omega, \qquad (6.2.22)$$

However, inasmuch as due to the large gain in the loop, usually being $G(0) \approx 1$, there is no difference between (6.2.21) and (6.2.22).

As follows from (6.2.20), variances of fluctuation and dynamic errors in the considered case do not depend on time, where the first is expressed by the simple and widely known formula

$$\mathbf{s}_{\phi,a}^{*} = 2S_{\mathsf{BRB}}\Delta f_{\mathsf{B}\phi}. \tag{6.2.23}$$

b) Let us consider an example, when smoothing circuits are constant, and the random part of the parameter is expressed by a polynomial in time t with random factors $\mu_{\rm L}$:

$$\lambda(t) = \sum_{k=0}^{m} \mu_{k} t^{k} (\overline{\mu}_{k} = 0, \ \overline{\mu}_{i} \mu_{k} = M_{i}, \ t > 0)$$
(6.2.24)

Fluctuation error is expressed as before by formula (6.2.23), and the mean square dynamic error according to (6.2.18) has the form:

$$\sigma_{\text{max}}^{*}(t) = \sum_{i, k=0}^{M} M_{ik} \epsilon_{i}(t) \epsilon_{k}(t), \qquad (6.2.25)$$

where

$$\mathbf{e}_{k}(t) = \int_{0}^{t} \nabla(t-\tau) \tau^{k} d\tau = \frac{1}{2\pi i} \int_{s-i\infty}^{s+i\infty} V(p) \frac{k!}{p^{k+1}} e^{pt} dp, \qquad (6.2.26)$$

and V(p) is given by formula (6.2.19).

With increase of the time of measurement, (6.2.26) passes to

$$\lim_{t \to \infty} \mathbf{s}_{k}(t) = \lim_{p \to 0} V(p) \frac{k!}{p^{k}}.$$
(6.2.27)

It follows from this that a regime with a strictly finite magnitude of variance of stationary dynamic error exists only in the presence in the numerator of V(p) of a factor p to a power, not smaller than the power of polynomial (6.2.24), depicting behavior of the parameter. The numerator of V(p), according to (6.2.19), is determined in turn by the denominator of the transfer function H(p) of the smoothing circuits. The latter, in the presence of factor p^n , are called astatic functions of the n-th order. Astaticism is attained by circuits, close in properties to

-16-

ideal integrators. Thus, a smoothing circuit in the form of a single integrator ensures astaticism of the first order:

$$H(p) = \frac{K_{R}}{p}, V(p) = \frac{p}{p + K_{R}K_{R}}, \qquad (6.2.28)$$

and with a double integrator with correction we have astaticism of the second order:

$$H(p) = \frac{K_{u}(1+p_{v})}{p^{2}}, V(p) = \frac{p^{2}}{p^{2} + K_{u}K_{x}(1+p_{v})}.$$
 (6.2.29)

In general form the mean square dynamic error in steady-state regime finally has the form

$$\sigma_{Am}^{2} = \sum_{i,k=0}^{m} M_{ik} \lim_{p \to 0} \frac{l! \, k! \, V^{*}(p)}{p^{i+k}}, \qquad (6.2.30)$$

Let us consider in more detail a series of simple examples.

1. The smoothing circuit has the form of an ideal integrator (6.2.28). The effective transmission band of a closed-loop meter here is equal to

$$\Delta f_{\bullet \bullet} = K_{\mathbf{x}} K_{\mathbf{x}} / 4. \tag{6.2.31}$$

For the case of a stationary parameter with spectral density $S_{\lambda}(\omega) = 2\sigma_{\lambda}^2 T/[1 + (\omega T)^2]$, where σ_{λ}^2 – variance, we have mean square dynamic error

$$\sigma_{ARH}^{2} = \frac{\sigma_{A}^{2}}{1 + 1/7K_{E}K_{A}}$$
(6.2.32)

For a parameter in the form of polynomial (6.2.24) establishment of error is observed only at m = 0 and 1. When m = 0 error $\sigma_{mWH}^2 = 0$, and when m = 1

$$\sigma_{ABB}^2 = \frac{M_{11}}{(K_B K_A)^3}.$$
 (6.2.33)

Error in position (i.e., due to the constant term of the random variable) is equal to zero, and error in acceleration or with respect to a higher derivative (in the case of the presence in the law of change of parameter of corresponding terms) grows without limit.

2. The smoothing circuit has the form of an RC-circuit:

$H(p) = K_0/(1 + p\tau).$

The effective band and dynamic error for a stationary parameter with the abovementioned spectral density have the form:

-17-

If we turn to the case of a parameter in the form of a polynomial, then it is easy to prove that establishment of error occurs only for constant parameter, and its variance is equal to $M_0 / (1 + K_0 K_{\pi})^2$. However, in the case of a large gain factor and large time constant of the smoothing circuit, for $M_{11} \neq 0$ we have in a sufficiently extended interval of time

$$\sigma_{ARE}^{2} \approx M_{LL} \left(\frac{T}{K_{e}K_{A}} \right)^{2}, \qquad (6.2.35)$$

i.e., the closed-loop system has approximately the same properties as in the case of an ideal integrator with gain factor K_0/T .

All that has been presented shows that in the absence of parametric fluctuations analysis of accuracy of meters is sufficiently simple and leads to results, widely known from classical automatic control theory.

6.2.4. Accuracy of Measurement in the Presence of Parametric Fluctuations

In the presence of parametric fluctuations it is possible to replace expression (6.2.14) by equation

$$\bullet(t) + K_{A} \int_{a}^{b} h(t, s) [1 + \tilde{t}(s)] \bullet(s) ds =$$

= $\lambda(t) - \lambda_{a}(t) - K_{A} \int_{a}^{b} h(t, s) \eta(s) ds.$ (6.2.36)

Again introducing pulse responses $g(t, \tau)$ and $v(t, \tau)$, according to (6.2.15) and (6.2.16), we can obtain

$$s(l) = -\int_{t_0}^{t_0} g(l, s) \eta(s) ds + \int_{t_0}^{t_0} v(l, s) [\lambda(s) - \lambda_n(s)] ds - - \int_{t_0}^{t_0} g(l, s) \xi(s) ds.$$
(6.2.37)

Formula (6.2.37) still is not the solution of equation (6.2.36), inasmuch as quantity $\varepsilon(t)$ enters into both parts of (6.2.37). Results convenient for use can be derived from (6.2.37), using smallness of variance of $\xi(t)$, usually observed in practice. The measure of this smallness will be shown below. According to the method of successive approximations we shall seek a solution of (6.2.37) in the form of the series

$\mathbf{s}(t) = \mathbf{s}_{\mathbf{a}}(t) + \mathbf{s}_{\mathbf{a}}(t) + \mathbf{s}_{\mathbf{a}}(t) + \dots,$

where $\varepsilon_0(t)$ has the zero, $\varepsilon_1(t)$ the first, $\varepsilon_2(t)$ the second, etc., order of smallness.

-18-

Then, for the zero approximation $\epsilon_0(t)$ we again have solution (6.2.17). The first correction is equal to

$$e_{1}(t) = \int_{0}^{t} \int_{0}^{t} g(t, s) \xi(s) g(s, \tau) \eta(\tau) ds d\tau - \int_{0}^{t} g(t, s) \xi(s) [e_{AUB_{1}}(s) + e_{AUB_{2}}(s)] ds. \qquad (6.2.38)$$

Analogously the second correction will be expressed in the form

$$\mathbf{e}_{\mathbf{g}}(t) = -\int_{t_0}^{t} ds \int_{t_0}^{s} d\tau g(t, s) \xi(s) g(s, \vartheta) \xi(\vartheta) g(\vartheta, \tau) \eta(\tau) + \int_{t_0}^{s} ds \int_{t_0}^{s} d\tau g(t, s) \xi(s) g(s, \tau) \xi(\tau) [\mathbf{e}_{AHH_1}(\tau) + \mathbf{e}_{AHH_2}(\tau)]. \qquad (6.2.39)$$

The mean square of $\varepsilon(t)$ during series expansion is equal to

$$\sigma_{\text{max}}^{\mathbf{g}}(t) = \overline{\mathfrak{e}_{0}^{2}(t)} + 2 \overline{\mathfrak{e}_{0}(t) \mathfrak{e}_{1}(t)} + [2 \overline{\mathfrak{e}_{0}(t) \mathfrak{e}_{1}(t)} + \overline{\mathfrak{e}_{1}^{2}(t)}] + \dots,$$

where averaging is conducted both for the ensemble of signals, and also for the ensemble of parameters. Averaging with the help of (6.2.17), (6.2.38) and (6.2.39), we obtain, in order:

1.

į į

$$\mathbf{e}_{0}^{2}(t) = S_{0KB} \int_{t_{0}}^{t} g^{2}(t, s) ds + \int_{t_{0}}^{t} \int_{t_{0}}^{t} v(t, s_{1}) v(t, s_{2}) R_{\lambda}(s_{1}, s_{2}) ds_{1} ds_{2} + \left[\int_{t_{0}}^{t} v(t, s) \Delta(s) ds\right]^{2} = \\ = \sigma_{0A}^{2} u(t) + \sigma_{ARH_{0}}^{2}(t) + \sigma_{ARH_{0}}^{2}(t).$$
(6.2.40)

2. $\overline{\varepsilon_0(t)\varepsilon_1(t)} = 0$ due to noncorrelation and zero mean values of functions $\xi(t)$, $\eta(t)$ and $\lambda(t) = \overline{\lambda(t)}$.

3.

The value in (6.2.41) of pulse response of the closed-loop system with coinciding arguments g(t, t) one should consider equal to zero, inasmuch as even in approximation of pulse response of smoothing circuits by functions breaking at these points (inertial link of the 1st order, link of the 2nd order with correction, and

-19-

so forth) it is impossible not to allow for the fact that the discriminator has inertia (delay), not smaller than the interval of correlation of that fluctuating voltage at its output, which we consider white noise. Consequently, there do not exist fluctuating components instantly returning through the feedback circuit and emphasizing fluctuations in the discriminator.

Considering this circumstance, we have $\overline{\epsilon_0(t)\epsilon_2(t)} = 0$. 4.

$$\overline{s_1^2(t)} = S_{\text{Hap}} S_{0\text{HB}} \iint g^*(t, s_1) g^*(\tau_1, s_2) ds_1 ds_2 + + S_{\text{Hap}} \iint g^*(t, \tau) [\sigma_{\text{ABH}_4}^2(\tau) + \sigma_{\text{ABH}_6}^2(\tau)] ds.$$
(6.2.42)

Considering (6.2.40) and (6.2.42) with an accuracy of terms of the second order of smallness we have total error of measurement:

$$\sigma_{\text{BMX}}^{2}(t) := \sigma_{\text{MXx}_{0}}^{2}(t) + S_{\text{H}\text{B}\text{P}} \int g^{2}(t, \tau) \left[\sigma_{A_{\text{B}\text{H}_{1}}}^{2}(\tau) + \sigma_{\text{H}\text{B}\text{H}}^{2}(\tau)\right] d\tau + S_{\text{H}\text{B}\text{P}} S_{\text{B}\text{H}\text{B}} \int g^{2}(t, \tau_{1}) g^{2}(\tau_{1}, \tau_{2}) d\tau_{1} d\tau_{2}, \qquad (6.2.43)$$

where $\sigma_{BHX_0}^2(t)$ - variance of output error for a meter when we disregard parametric fluctuations, expressed by formula (6.2.40).

Simpler relationships can be obtained in the case of a stationary parameter and constant smoothing circuits for $\Delta = 0$:

$$a_{max}^{3} = a_{max_{0}}^{2} (1 + a_{map}^{3}).$$
 (6.2.44)

Here, there is introduced $\sigma_{\pi a p}^2 = 2 s_{\pi a p} \Delta f_{\partial \phi}$ - variance of parametric fluctuations, smoothed in the loop.

After analogous calculations to terms of the fourth order of smallness we have

$$\sigma_{aux}^2 = \sigma_{aux}^2 (1 + \sigma_{aap}^2 + \sigma_{map}^4).$$
 (6.2.45)

(6.2.46)

According to (6.2.45) total output error due to parametric fluctuations is increased by a factor of $1 + \sigma_{\Pi \alpha p}^2 + \sigma_{\Pi \alpha p}^4$, i.e., the additional component of error is greater, the greater the error of measurement in the absence of parametric fluctuations. This circumstance corresponds to description of the circuit of Fig. 6.4, where parametric fluctuations characterize changes of gain of the discriminator, having no influence only when $\varepsilon = 0$.

Quantity $\sigma_{\Pi a p}^2$ in the stationary case can be expressed in terms of variance σ_{ξ}^2 and width of the spectrum Δf_{ξ} of function $\xi(t)$, assumed above to be white noise, in the form

$$\sigma_{\rm map}^2 = 2S_{\rm map}\Delta f_{\rm b\phi} = \sigma_{\rm t}^2 \frac{\Delta f_{\rm b\phi}}{\Delta f_{\rm t}}.$$

From (6.2.46) it follows that parametric fluctuations can be ignored when the effective band of a tracking meter is considerably narrower than the width of the spectrum of these fluctuations, and their variance is much less than unity. Only in this case, instead of (6.2.43) and (6.2.45), it is possible to use simpler formulas (6.2.18) and (6.2.20), and properties of the discriminator are determined only by quantities $K_{\rm H}$ and $S_{\rm AKB}$.

We shall indicate here that the obtained formulas, containing coefficient s_{map} , are not applicable if $s_{map} < 0$, and also upon reaching σ_{map}^2 of large values, when approximation, based on use of only the first terms of the series expansion and leading to formulas (6.2.44), (6.2.45), does not give a correct answer. In these cases the tracking meter should be considered a nonlinear system. Such a consideration will be given in § 6.3.

6.2.5. Case of a Set of Rapid and Very Slow Fluctuations Besides rapid fluctuations in the input signal there may exist disturbances, varying considerably more slowly than the measured parameter. We consider, for instance, very slow fadings of the signal, in which its amplitude and phase for very considerable time intervals can be considered constant, or stable interfering reflection. In these cases it is useful to write the input signal in the form $y(t, \lambda(t), \nu(t))$, where $\lambda(t)$ - tracked variable, and $\nu(t)$ - random, slowly varying parameter (or parameters) of the signal, which, in principle, there is no necessity to measure. We naturally introduce here the same characteristics of a discriminator as for the case of rapid fluctuations, but we consider them to additionally depend on $\nu(t)$. The equivalent of output voltage of the discriminator is presented in the form

$$z_{\bullet}(t) = \dot{a}(t, v) + \sqrt{S(t, v)} \zeta(t),$$
 (6.2.47)

where $s(\varepsilon, \nu)$, $S(\varepsilon, \nu)$, as before, are the discrimination and fluctuation characteristics.

Relative white noise $\zeta(t)$ here is formed only by rapid fluctuations, as distinguished from (6.2.6). For a low level of noises instead of (6.2.47) it is sufficient to use relationship

$$z_{\bullet}(l) = K_{A}(v) \cdot + \sqrt{S_{\bullet}(v)} \zeta(l). \qquad (6.2.48)$$

The basic performance characteristic of tracking - mean square error is found just as in Paragraph 6.2.3. In particular, in the absence of rapid

-21-

parametric fluctuations in the stationary case we have a formula, analogous to (6.2.20):

$$\mathbf{s}_{\text{BMR}}^{2}(\mathbf{v}) = 2S_{\text{BMR}}(\mathbf{v}) \Delta f_{\text{B}\Phi}(\mathbf{v}) + \frac{1}{2\pi} \int_{-\infty}^{+\infty} \frac{S_{\lambda}(\mathbf{w}) d\mathbf{w}}{|1 + K_{A}(\mathbf{v}) H(\mathbf{w})|^{2}}, \qquad (6.2.49)$$

and the second of the second second

where

$$S_{\Theta RB}(\mathbf{v}) = S_{\Theta}(\mathbf{v}) / K_{A}^{T}(\mathbf{v});$$

$$\Delta f_{\Theta \Theta}(\mathbf{v}) = \frac{1}{2\pi} \int_{0}^{\infty} \left| \frac{K_{A}(\mathbf{v}) H(i\omega)}{1 + K_{A}(\mathbf{v}) H(i\omega)} \right|^{2} d\omega.$$

According to (6.2.49), on ν there depend both dynamic error (through the gain factor in the loop), and also fluctuation error (through equivalent spectral density $S_{AKR}(\nu)$ and effective band width $\Delta f_{acc}(\nu)$).

In formula (6.2.49) we average the ensembles of rapid fluctuations and of parameter $\lambda(t)$. It is possible to use this formula when we are interested in the process of measurement at time intervals, in which random parameter $\nu(t)$ does not noticeably change. In another case it is useful to average σ_{BWY}^2 with respect to ν :

$$\sigma_{\text{MMX}}^{\sharp} = \overline{\sigma_{\text{MMX}}^{\sharp}(\nu)} + 2\overline{S}_{\text{JKB}}(\nu)\Delta f_{\phi\phi}(\nu). \qquad (6.2.50)$$

As a simple example we shall consider the case when slow fluctuations are expressed only in stationary amplitude modulation of the amplification factor of the discriminator, and S_0 does not change in time. In other words, there occur slow parametric fluctuations.

Introducing amplification factor $\overline{K}_{\rm II}$ averaged with respect to $\nu_{\rm r}$

 $K_{\mathbf{x}}(\mathbf{v}) = \overline{K}_{\mathbf{x}}(1 + \boldsymbol{\xi}(t)),$

where $\xi(t)$ - slowly varying stationary random function, we have fluctuation error

$$\mathbf{\hat{s}}_{\phi a}^{1} = \frac{S_{\phi}}{\overline{K}_{A}^{2}} \frac{1}{(1+\xi)^{1}} \int_{-\infty}^{+\infty} \left| \frac{\overline{K}_{A} (1+\xi) H (i\omega)}{1+\overline{K}_{A} (1+\xi) H (i\omega)} \right|^{2} d\omega \approx (6.2.51)$$

$$\approx \overline{S}_{0H_{0}} \int_{-\infty}^{+\infty} \overline{\overline{G}} (i\omega) \overline{\xi} + \overline{\overline{G}}^{2} (i\omega) \overline{\xi}^{1} |^{2} d\omega =$$

$$= \overline{S}_{0H_{0}} \overline{\Delta} \overline{f_{0\phi}} (1+ks_{H_{0}}^{2}).$$

Here there are introduced the averaged equivalent spectral density $\overline{S}_{\partial KD} = -S_0 / \overline{K}_{\pi}^2$, the frequency response of the closed loop

$$\ddot{G}(i \bullet) = \frac{K_{A} H (i \bullet)}{1 + K_{A} H (i \bullet)}$$

and effective bandwidth $\overline{\Delta}_{i=0}^{i} = \frac{1}{2\pi} \int_{0}^{\infty} |\overline{G}(i\omega)|^{2} d\omega$.

()

In relationship (6.2.51) we perform expansion with respect to ξ , as i there is given the designation $\sigma_{\text{map}}^2 = \overline{\xi^2} \ll 1$ for variance of the parametric input, and by k we denote the numerical coefficient

 $\int_{-\infty}^{+\infty} |\overline{\mathcal{O}}(i\omega)|^{2} [|\overline{\mathcal{O}}(i\omega)|^{2} - 2\overline{\mathcal{O}}^{2}(i\omega)| d\omega$ $k = \frac{-\infty}{\int_{-\infty}^{+\infty} |\overline{\mathcal{O}}(i\omega)|^{2} d\omega$ (6.2.52)

As we proved, the presence of very slow fluctuations does not lead to noticeable change of our approach to analysis of meters; it is only necessary to consider that S_{are} differs for rapid and slow fadings of the signal.

6.2.6. General Case

The most general case one should consider that in which, besides rapid interferences in the signal, there exist random variables with a rate of change, comparible with Paragraph 6.2.5, so that

$$z_{\bullet}(l) = a(e, v(l)) + \sqrt{S(e, v(l))}\zeta(l).$$
 (6.2.53)

However, $\nu(t)$ varies in characteristics of relationship (6.2.53) too fast, so that it is possible not to take this into account during determination of pulse response of the closed loop and of measurement error. Instead of (6.2.14), in this case we have equation

$$\bullet(t) + \overline{K}_{R} \int_{t_{0}}^{t} h(t, \tau) (1 + \varepsilon(\tau)) \bullet(\tau) d\tau = \lambda(t) - \lambda_{n}(t) - \lambda_{n}(t)$$

where $\mathbf{K}_{\mathbf{J}}$, $\mathbf{S}_{\mathbf{O}}$ - the amplification factor and the spectral density averaged for ν :

$$\xi(t) = \frac{K_{x}(v(t))}{K_{x}} - 1,$$

 $s(t) = S(\nu(t))/S_0$ - normalized random functions, characterizing change of amplification factor and spectral density in accordance with change of $\nu(t)$; and $\zeta(t)$, as before, is white noise of unit spectral density.

If we look for fluctuation error in the form

$$\phi_{\mathbf{z}}(t) = \sqrt{\overline{S}}_{\mathbf{0}\mathbf{E}\mathbf{z}} \int_{0}^{t} g(t, \tau) \zeta(\tau) d\tau \ (\overline{S}_{\mathbf{0}\mathbf{E}\mathbf{z}} = \overline{S}_{\mathbf{0}}/\overline{K}_{\mathbf{z}}^{2}),$$

assuming for simplicity that $\Delta(t) = 0$, for pulse response of a closed loop, $g(t, \tau)$, randomly depending on the input signal through $\nu(t)$, we can obtain equation

$$g(t, \tau) + K_{x} \int h(t, s) [1 + t(s)] g(s, \tau) ds = K_{x} h(t, \tau) / s(\tau), \qquad (6.2.54)$$

similar to (6.2.15), where $h(t, \tau)$ - pulse response of the smoothing circuits.

If we solve this equation for $g(t, \tau)$, fluctuation error will be expressed superficially by the known formula

$$\mathbf{s}_{q_n}^2(\mathbf{v}(l)) = \mathbf{\tilde{S}}_{nnn} \int \mathbf{g}^2(l, \mathbf{v}) d\mathbf{v}, \qquad (6.2.55)$$

and for a steady-state regime for stationary s(t) and $\xi(t)$, after averaging with respect to ν , we have

$$\overline{d_{\varphi_{R}}^{*}(\mathbf{v})} = \overline{S}_{\varphi_{R}} \int \overline{g^{*}(l, l-\xi)} d\xi. \qquad (6.2.56)$$

However, in common form (6.2.54) can not be solved, and it is necessary to limit oneself to consideration of the case of a small coefficient of modulation of the gain factor of the discriminator $(\xi^2(t) = \sigma_{\pi a p}^2 \ll 1)$ and constant smoothing circuits. Seeking our solution in the form

$$g(t, \mathbf{v}) = g_{\mathbf{v}}(t-\mathbf{v})\sqrt{s(\mathbf{v})} + g_{\mathbf{v}}(t-\mathbf{v}, \mathbf{v})\sqrt{s(\mathbf{v})} + g_{\mathbf{v}}(t-\mathbf{v}, \mathbf{v})\sqrt{s(\mathbf{v})} + \dots, \qquad (6.2.57)$$

where g_0 has the zero, g_1 the first, and g_2 the second order of smallness, for the zero approximation we obtain equation

$$g_{\bullet}(t) + \tilde{K}_{A} \int h(t-x) g_{\bullet}(x) dx = \tilde{K}_{A} h(t),$$

whence

$$g_{\bullet}(l) = \frac{1}{2\pi} \int_{-\infty}^{+\infty} \frac{\overline{K}_{A}H(l\omega)}{1 + \overline{K}_{A}H(l\omega)} e^{i\omega l} d\omega.$$

Further, for the first approximation we have equation

$$E_{1}(t, \tau) + \bar{K}_{R} \int_{0}^{t} h(t-x) g_{1}(x, \tau) dx =$$

= $-\bar{K}_{R} \int_{0}^{t} h(t-x) \xi(x+\tau) g_{0}(x) dx,$

which gives as a result

$$g_1(l-1, x) = -\int_{0}^{l-1} g_0(l-x-x) g_0(x) \xi(x+x) dx.$$

Analogously for the second approximation we obtain

$$E_{z}(t-\tau, \tau) ==$$

$$= \int_{0}^{t-\tau} dx \int_{0}^{z} dy g_{0}(t-\tau-x) g_{0}(x-y) g_{0}(y) \xi(x+y) \xi(y+\tau).$$

-24-

Finally, the solution of (6.2.54) has the form

()

$$g(t, \tau) = g(t, \tau) = -\frac{g(t, \tau)}{\int} g_{\bullet}(t - \tau) - \int_{0}^{t-\tau} g_{\bullet}(t - \tau - x) g_{\bullet}(x) \xi(x + \tau) dx + -\frac{1}{\int} dx \int_{0}^{\tau} dy g_{\bullet}(t - \tau - x) g_{\bullet}(x - y) g_{\bullet}(y) \xi(x + \tau) \xi(y + \tau) \bigg\}.$$
(6.2.58)

From this, by (6.2.56), limiting ourselves to terms of the second order of smalless and considering that $\overline{\xi(t)} = 0$, for variance of the mean fluctuation error we have

$$\overline{\sigma_{\phi x}^{2}} = \overline{S}_{9Ky} \int \left[g_{\phi}^{2}(s) + \int \int g_{\phi}(s - x) g_{\phi}(x) g_{\phi}(s - y) g_{\phi}(y) M(x, y) dx dy + 2 \int dx \int dy g_{\phi}(s) g_{\phi}(s - x) g_{\phi}(x - y) g_{\phi}(y) M(x, y) \right] ds, \qquad (6.2.59)$$

where we considered that $\overline{s(t)} = 1$, and

$$M(x, y) = s(:)\xi(t+x)\xi(t+y). \tag{6.2.60}$$

As can be seen from (6.2.59), only the first, the basic component of variance of mean fluctuation error can be rewritten in familiar form $2S_{BCD}\Delta f_{-2\phi}$, where in the definition of $\Delta f_{-2\phi}$ there enters average amplification factor. Other components have more complicated form and are expressed through the pulse response of the zero approximation $g_0(t)$ and the estimator M(x, y) of random functions s(t) and $\xi(t)$. In the particular case when the modulation of spectral density and of the gain factor of the discriminator due to slow interferences do not depend on one another additional components in $\sigma_{\phi\Pi}^2$ are expressed through spectral density $S_{Hap}(\omega)$ of parametric influence $\xi(t)$, inasmuch as then $M(x, y) = \overline{\xi(x)\xi(y)}$. This leads to relationship

$$\sigma_{\Phi_{a}}^{a} = \overline{S}_{\Phi K a} \left\{ 2 \overline{\Delta f_{\Phi \Phi}} + \frac{1}{(2\pi)^{a}} \int_{-\infty}^{+\infty} \int_{-\infty}^{+\infty} S_{\pi \varrho p}(\omega_{1}) \times \left[\overline{O} \left(i \omega_{a} \right) \right]^{a} \left[\left| \overline{O} \left(\omega_{a} - \omega_{1} \right) \right|^{a} + 2 \overline{O} \left(\omega_{a} \right) \overline{O} \left(\omega_{a} - \omega_{1} \right) \right] d\omega_{1} d\omega_{2} \right\}.$$

$$(6.2.61)$$

In the two extreme cases of very rapid and very slow parametric fluctuations from (6.2.61) we have

$$\lim_{\sigma_{\phi\pi}^*} = 2\overline{S}_{\sigma\pi\pi}\overline{\Delta f}_{\sigma\phi} [1 + 2S_{\pi\pi\pi}\overline{\Delta f}_{\sigma\phi}], \qquad (6.2.62)$$

$$S_{\pi\pi\pi}(\sigma) \rightarrow \text{const}$$

-25-

$$\lim_{\mathbf{\sigma} \to \sigma^2_{\text{map}}} \frac{2\overline{S}_{\text{DHB}} \overline{\Delta} f_{\text{D} \phi} \left[1 + k \sigma^2_{\text{map}} \right], \qquad (6.2.63)$$

يمشره بالأكرة للالكشار بالشرائيات سالطياب فالمشافرة فألقا بمقعد

والمقسيا وليلاف فللاطفان فالمنافق والمستعدة الاقالية المرابية

where k - numerical coefficient, given in (6.2.52). Formula (6.2.62) is already known from Paragraph 6.2.4, which is obvious upon comparison of it with (6.2.44). In its derivation we again used the fact that g(0) = 0. Formula (6.2.63) coincides with formula (6.2.51), derived earlier in Paragraph 6.2.5.

This investigation shows in what sense one should understand in every case the equivalent spectral density, how characteristics of the discriminator are introduced, and how by them we calculate errors of measurement. The given relationships are valid for tracking meters of any parameter of modulation of a radar signal and can serve, thereby, as a basis for analysis of accuracy of tracking radar meters.

§ 6.3. Questions of Breakoff of Tracking

We analyzed tracking meters for the influence on them of small noises and interferences, when errors of measurement are small, and the linear approximation introduced in the preceding section is valid for characterization of the discriminator. However, for large noises and interferences, error of measurement becomes comparable with the width of the linear section of the discrimination characteristic or even exceeds it. This, first, leads to insufficiency of linear approximation during calculation of accuracy both due to nonlinearity of the discrimination, so also due to the complicated form of the fluctuation characteristics. Secondly, here probability of breakoff of tracking sharply increases, where mismatch exceeds the width of the discrimination curve, and the useful signal, in general, ceases to act on the discriminator.

Investigation of nonlinear phenomena in tracking meters, occurring with intense interferences, is the subject of the present paragraph. It turns out that a suitable mathematical device for analysis in this case are diffusion and related differential equations, obtained during the study of Markovian random processes. Therefore, to facilitate understanding of the conducted analysis we give in Paragraph 6.3.1 certain information from the theory of Markovian processes and differential equations, describing their statistical properties.

6.3.1. Markovian Random Processes and Related Differential Equations The device of Markovian processes has a wide range of applications and recently has been widely used in theoretical radio engineering [20], being the best means

-26-

of investigation of properties of nonlower elements of radio channels, fluctuations in vacuum-tube oscillators, and so forth. The call Markovian a random process y(t), for which conditional probability density $P(y_n|y_{n-1}, \ldots, y_1)$ satisfies relationship

$$P(y_n | y_{n-1}, \dots, y_1) \equiv P(y_n | y_{n-1}) (y_i = y(t_i)), \qquad (6.3.1)$$

i.e., depends functionally only on the value of the process at one preceding moment of time. Conditional probability density $P(y_n|y_{n-1}) \equiv W(y_n|y_{n-1})$ we call probability density of transition from state y_{n-1} at time t_{n-1} to state y_n at time t_n . It is of basic value in the theory of Markovian processes, since by it we express ndimensional probability densities:

$$P_{\mathbf{x}}(y_{\bullet},\ldots,y_{n-1}) = P_{\mathbf{x}}(y_{\bullet}) \prod_{i=1}^{n-1} W(y_{i}|y_{i-1}), \qquad (6.3.2)$$

where $P_1(x)$ - one-dimensional distribution.

In the general theory it is proved [20] that for sufficiently broad conditions function $W(y|y_0)$ obeys two partial differential equations (Fokker-Planck-Kolmogorov equations). If differentiation is performed with respect to finite values of y, t, the equation is called direct and is recorded in the form

$$\frac{\partial \mathbf{v}}{\partial t} = -\frac{\partial}{\partial y} \{A(y, t) \mathbf{W}\} + \frac{1}{2} \frac{\partial^2}{\partial y^2} \{B(y, t) \mathbf{W}\}, \qquad (6.3.3)$$

if however, W is differentiated with respect to initial values of y_0 , t_0 , the equation is called reverse and has the form

$$\frac{\partial \Psi}{\partial t_0} = A(y_0, t_0) \frac{\partial \Psi}{\partial y} + \frac{B(y_0, t_0)}{2} \frac{\partial^2 \Psi}{\partial y^2}.$$
(6.3.4)

Functions A(y, t) and B(y, t) we call coefficients of drift and diffusion, respectively.

The method of finding these coefficients follows from another theoretical proposition of [20]. Let us assume that function y(t) obeys ordinary differential equation

$$\frac{dy}{dt} = f(y, t) + g(y, t)\xi(t), \qquad (6.3.5)$$

where f(y, t), g(y, t) - functions of a broad class;

 $\xi(t)$ — fluctuating influence with a wide spectrum, which it is possible to approximate by white noise with unit spectral density [the correlation function is equal to

 $\overline{\xi(t)\xi(t+\tau)} = \delta(\tau)].$

Then y(t) is a Markovian process, the probability density of transition of which obeys equations (6.3.3) and (6.3.4) with coefficients

$$A(y, t) = f(y, t), B(y, t) = g^{2}(y, t).$$
 (6.3.6)

Deeper mathematical consideration shows the presence in the coefficient of drift of an additional component $\frac{1}{2}g(y, t) \frac{\partial g(y, t)}{\partial y}$ [20]. However, proceeding from the physical meaning of the processes occurring in meters, which we will be interested in below, this component should not be considered. The explanation of this was the circumstance, already noted in § 6.2, that the discriminator has inertia or delay, not smaller than the interval of correlation of that fluctuating voltage at its output, which is approximated by white noise. As a result there does not occur an instantaneous return in the feedback circuit of fluctuating disturbances, which could correlate with fluctuations in the discriminator and thereby change the coefficient of drift.

It remains to explain how to pass to a differential equation of type (6.3.5) from the equation obtained during the analysis of certain meters, of form

$$\frac{dy}{dt} = \zeta(y, t), \qquad (6.3.7)$$

where the time of correlation of the random component in $\zeta(y, t)$ is small as compared to the effective time constant of the system. As shown [20], for this one should take as f(y, t) and g(y, t) functions

$$f(y, t) = \overline{\zeta(y, t)},$$

$$g(y, t) =$$

$$= \left\{ \int_{-\infty}^{+\infty} \overline{[\zeta(y, t) - \overline{\zeta(y, t)}][\zeta(y, t + \tau) - \overline{\zeta(y, t + \tau)}]d\tau} \right\}^{1/2}.$$
(6.3.8)

Equation (6.3.5) corresponds to transmission of white noise through an inertial link of the first order, nonlinear and with variable gain. Subsequently we will prove that namely (6.3.5) and related equations can be used for investigation of nonlinear phenomena in tracking meters of very simple form.

If the dynamic system, under the influence of fluctuations, has a more complicated form, it is not possible to describe it by the Markovian process of the 1st order, which was considered above. Here it is convenient to use the idea of a Markovian process of the n-th order. Let us introduce vector random process $\{y_1(t), \ldots, y_n(t)\} = y(t)$, components of which are scalar random processes $y_i(t)$. Let us assume that y(t) obeys system of equations

$$\frac{d\mathbf{y}(t)}{dt} = \zeta(\mathbf{y}, t),$$

where $\zeta(y, t)$ - vector function.

Then, if times of correlation of random components in $\boldsymbol{\zeta}(\boldsymbol{y}, t)$ are small as compared to the time scale of the system as a whole, the probability density of transition $W(\boldsymbol{y}, t|\boldsymbol{y}_0, t_0)$ of process $\boldsymbol{y}(t)$ from value \boldsymbol{y}_0 at time t_0 to \boldsymbol{y} at time t obeys equation

$$\frac{\partial \mathbf{W}}{\partial t} = -\sum_{i=1}^{n} \frac{\partial}{\partial g_i} \{A_i(\mathbf{y}, t) \mathbf{W}\} + \frac{1}{2} \sum_{i, j=1}^{n} \frac{\partial^2}{\partial g_i \partial g_i} \{B_{ij}(\mathbf{y}, t) \mathbf{W}\},\$$

where

$$A_{i}(\mathbf{y}, t) = \zeta_{i}(\mathbf{y}, t);$$

$$B_{ij}(\mathbf{y}, t) =$$

$$= \int_{-\infty}^{+\infty} [\zeta_{i}(\mathbf{y}, t) - \overline{\zeta_{i}(\mathbf{y}, t)}] [\zeta_{j}(\mathbf{y}, t+\tau) - \overline{\zeta_{j}(\mathbf{y}, t+\tau)}] d\tau.$$

As components $y_i(t)$ we can with equal success take different interconnected random processes, a random process and its derivatives to the (n-1)-th order inclusively, or the value of one and the same random process y(t) at different moments of time

$$y_1(t) = y(t + \tau_1), \quad y_2(t) = y(t + \tau_2), \dots, y_n(t) = y(t + \tau_n),$$

where $\tau_1, \ldots, \tau_n - \text{constants}$.

If we determine $y_1(t)$ by the last method, the probability density of transition

$$W'(y, t, y_0, t_0) = = \\ = W(y_1, t + z_1, \dots, y_n, t + z_n) |y_{01}, t_0 + z_1, \dots, y_{1n}, t_0 + z_n)$$

can be found by solution of equation (6.3.9). This means that the conditional probability density of values of the process at a certain n moments of time (and, consequently, of any one value of y(t) at one moment t) depends functionally on values of the process in n preceding moments of time. Such process is called an n-th order Markovian. Nonlinear phenomena in tracking meters, described by n-th order equations, lead to the necessity of investigating these processes and, correspondingly, to solution of equations of form (6.3.9). Considering, however, that such equations for n > 1 usually cannot be solved, let us turn to a more

-29-

detailed study of one-dimensional equations.

Solution of (6.3.3) reveals the evolution in time of the probability density of transition, which for an initial condition in the form of a δ -function simply repeats one-dimensional probability density.

In certain conditions [20] (primarily for independence of A(y) and B(y) from time) there is finally established stationary distribution, which it is possible to express through coefficients A(y) and B(y), equating $\partial W/\partial t$ in (6.3.3) to zero:

$$\boldsymbol{W}_{\text{stan}}(\boldsymbol{y}) = \frac{C}{B(\boldsymbol{y})} \exp\left\{2\int \frac{A(\boldsymbol{y}) d\boldsymbol{y}}{B(\boldsymbol{y})}\right\}, \quad (6.3.10)$$

where C - normalizing constant, and integral is taken as indefinite.

No less frequent, however, are cases when a stationary distribution does not exist or we are interested in the actual process of becoming stationary. Then, it is necessary to solve the complete equation (6.3.3), which is generally a very complicated mathematical problem. Here, it is usually necessary to be given so-called boundary conditions.

If, after the first time y(t) exceeds a certain magnitude y_1 further realization of y(t) does not interest us, by analogy with the process of diffusion, we talk about the presence at point y_1 of an "absorbing screen." With the help of mathematical condition

$$V(y_1, t) = 0$$
 (6.3.11)

realizations of y(t), even once touching the screen before moment t, are automatically ejected from consideration after this moment of time.

If, after arriving at point y_1 , any realization of y(t) immediately turns back, we talk of the presence of a "reflecting screen." The mathematical condition of reflection is equality

$$\frac{\partial f(y_1, f)}{\partial y} = 0. \tag{6.3.12}$$

Solving equation (t.3.3) under these conditions, we can find how W(y, t) changes in time. However, we are not always interested in the complete structure of probability density. Sometimes it is convenient to limit ourselves to cruder characteristics, as which we can consider in the first place variance of fluctuation error, during calculation of which, in distinction from the case of § 6.2, we must allow for nonlinear factors. In principle it is not difficult, for instance, to determine the variance of stationary distribution (6.3.10), if it exists.

We may be interested also in probability characteristics of another type, for instance average time of stay of a realization of y(t) in a given region on the condition of a certain concrete initial value $y_0 = y(t_0)$. For average time of stay of $T(y_0)$ as a function of the initial condition from equation (6.3.4) we can obtain differential equation

$$\frac{B(y_{*})}{2} \frac{d^{*}T}{dy_{*}^{2}} + A(y_{*}) \frac{dT}{dy_{*}} + 1 = 0, \qquad (6.3.13)$$

for which on boundaries of the region we have conditions

()

j

$$T(y_1)=0, T(y_2)=0.$$
 (6.3.14)

The physical meaning of these conditions is that a realization located next to the "absorbing" screen due to the purely random character of disturbing factors will immediately intersect it, and this is equivalent to a zero average time of stay.

6.3.2. Formulation of the Problem About Breakoff of Tracking Above we already indicated that interference of high intensity makes linear approximation for the discrimination characteristic invalid. Mismatch frequently goes into the nonlinear region and can attain such magnitudes, at which the useful signal no longer affects the discriminator. Here, there occurs breakoff of tracking. The shown nonlinearity of the discriminator characteristic is not some imperfection, inherent in certain concrete circuits. Further, we shall show that characteristics of optimum discriminators have similar form. This circumstance increases the importance of investigations of nonlinear regimes of meters and, in particular, questions of breakoff of tracking.

With a nonlinear characteristic of the discriminator the process representing the result of measurement in a number of conditions has variance growing in time. The fact is that the probability of departure from the linear section is always different from zero, so that during prolonged measurement short duration failure certainly sets in. Here, the meter practically becomes an open storage unit of noises, and mismatch $\varepsilon(t)$ experiences random changes, in no way connected with the parameter of the useful signal. The circumstance that in a certain time interval realization of $\varepsilon(t)$ can be similar to realization of the process in a linear system does not change the essence of the matter, since we are interested in the whole set of realizations, including those, in which nonlinear properties of the system have

-31-

already appeared. Since with passage of time the probability of failure increases, it is clear that variance of process $\varepsilon(t)$ increases. It is however, not the only possible crude characteristic. It is possible to apply other forms of probability characteristics, where their reasonable selection depends on the type and method of use of the meter.

Before considering such characteristics in reference to the phenomenon of breakoff of tracking, we will make the following remarks. From what has been presented it follows that with observance of certain conditions nonlinear phenomena in meters, including the phenomenon of breakoff of tracking, can be studied by solution of diffusion equations (and equations connected with them). As a result there are found various estimators of failure. Use of diffusion equations is based in this case on the practical inertialess nature of the discriminator as compared to the smoothing circuits, since only upon observance of this condition and during application of smoothing circuits of the first order for mismatch is there obtained an equation of form (6.3.5), where $\xi(t)$ is white noise.

In the case of a low level of noises or interferences the formulated condition is not satisfied sufficiently well in all cases. However, with growth of interference, not depending on the measured parameter (internal noises, interferences, etc.), the scale of the discrimination characteristics $\alpha(\epsilon)$ greatly decreases, and the transmission factor of the discriminator $K_{\mu} = \frac{da(0)}{d\epsilon}$ drops. Here, the inertia of the closed loop grows. Such a dependence on examples will be explained in subsequent chapters. From the point of view of questions of failure most interesting, namely, is a high level of interferences. Thus, conditions of applicability of diffusion equations are obviously satisfied where, in view of nonlinearity of the problem, it is most expedient to use this mathematical device.

Let us turn to estimators, which are best found from solution of diffusion equations for problems of failure. Let us assume, first of all that the meter is the only means of signal selection and tracking, there are no automatic means to shift to search upon disruption of tracking, and the final effect of work of the radar depends on the presence of tracking at every given moment of time. Then, if there are no limitations on output values of the measured variable, naturally as the characteristic of failure we take the probability density of mismatch, which, as a function of time, should be found from the solution of the diffusion equation without boundary conditions.

-32-

More frequently, however, in smoothing and control circuits there exist limiters of various types. Here, the phenomenon of breakoff of tracking leads to the following. In the process of tracking at a certain moment of time under the influence of some large fluctuating overshoot mismatch becomes larger than the width of the discrimination curve and tracking is disrupted. Then, for a certain random interval of time mismatch randomly varies under the influence of interference alone. Here, however, it remains finite due to the limitation, and therefore, under the influence of fluctuating overshoots of reverse sign it sooner or later decreases to a magnitude in which tracking again is renewed. This process is repeated again, and upon the expiration of a certain time establishments of it can be considered stationary. Therefore, if moments of measurement of the tracked variable are sufficiently removed from the beginning of tracking (observation time is great), and if we are interested in accuracy of measurement with breakoff of tracking, then as the characteristic we should take variance of measurement as t \rightarrow $\rightarrow \infty$. It can be found proceeding from stationary distribution (6.3.10) with idealization of the phenomenon of limitation in the form of a "reflecting screen" [boundary conditions are taken in form (6.3.12)]. As shown below, with growth of intensity of interference this stationary variance grows, and there occurs a threshold effect, corresponding to sharp increase of errors due to short duration failures when interference exceeds a certain critical level.

If the time of observation of the output quantity of the meter is small, the described approach to investigation of breakoff phenomena is unacceptable. Here characteristics of the first failure after beginning of tracking take on special interest, especially if we consider that short duration failure of a meter of one parameter of a signal usually also leads to vanishing of the signal in meters of its other parameters. In order to characterize the first short duration failure, we take the following definition. By short duration failure at time t we understand exceeding by the magnitude of mismatch $\varepsilon(t) = \lambda(t) - \hat{\lambda}(t)$ between the true and measured values of the parameter of certain fixed levels (- Δ , + Δ) under the condition that at the initial moment mismatch was sufficiently small [within limits (- Λ , + Δ)] and up to moment t was always in this interval. With smoothing circuits of the first order this leads to a boundary value problem for W(t, ε) in the form (6.3.3), where on the boundaries we are given conditions of absorbing screens.

-33-

Screens are considered located on boundaries of the discriminator range, immediately behind "slopes" of the discrimination characteristic (Fig. 6.5). These



Fig. 6.5. Discrimination characteristic and its approximations: 1) true relationship; 2) sinusoidal approximation; 3) linear approximation. boundaries are determined by the bandwidth of filters in the frequency discriminators, duration of gate pulses in time discriminators, the width of the radiation pattern in angular discriminators, etc. It is necessary to note that in selection of coordinates of screens there exists a certain arbitrariness. No matter how we choose these coordinates, realizations of mismatch are probable, which after intersecting the

screen almost immediately intersect it in the opposite direction, and there will be essentially no failure. It is clear that such phenomena are not considered during application of the above definition of failure. Therefore, the described approach to analysis of failure phenomena leads only to approximate solutions, allowing us, however, to find certain laws governing performance.

By direct solution of diffusion equation (6.3.3) with boundary conditions (6.3.11) we can find the probability density of mismatch by which it is easy to calculate the probability of absence of failure up to moment t. It is more expedient, however, to use a simplified characteristic of failure - average time to failure T_{CG} . This parameter is the solution of equation (6.3.13), ensuing from the diffusion equation. Average time to failure decreases with growth of intensity of interference. Just as for variance of stationary distribution, here there exists a threshold effect, expressed in the fact that with decrease of intensity of interference and related variance of error of a linearized system for certain values of these magnitudes there starts a sharp growth of average time to failure. This gives us the possibility of finding the cutoff value of intensities of interferences, at which there occurs short duration failure (in the above-indicated sense).

Direct solutions of diffusion equations for the problems interesting us in conditions of a high level of interferences are given in [21, 22, 23, 24].

-34-

Considering, however, the expediency of application for analysis of failure of criteria of stationary variance and average time to failure, we shall give a solution of two other problems, noted above. Here, we shall consider general laws; applications of the obtained solutions will be given in subsequent chapters devoted to investigation of meters of concrete parameters of radar signals.

6.3.3. Stationary Distribution with Limitations

Let us consider a tracking meter of the type in Fig. 6.1. Output voltage of the discriminator in the case of the presence of rapid fluctuations alone, according to § 6.2, can be recorded in the form

$$z(t, e) = a(e) + \sqrt{S(e)}\xi(t),$$
 (6.3.15)

where $a(\varepsilon)$, $S(\varepsilon)$ - discrimination and fluctuation characteristics, respectively; $\xi(t)$ - white noise of unit spectral density.

Analysis of phenomena interesting us can ultimately be reduced to the case of smoothing circuits in the form of a single integrator, where

$$\frac{d}{dt} \hat{\mathbf{z}}(t) = K_{\mathbf{x}} \mathbf{z}(t). \tag{6.3.16}$$

Considering that $\lambda - \hat{\lambda} = \epsilon$, from relationships (6.3.15) and (6.3.16) we find the differential equation, describing the meter as a whole:

$$\frac{d\epsilon}{dt} = \lambda - K_{\mathbf{x}} a(\mathbf{s}) - K_{\mathbf{x}} \sqrt{S(\mathbf{s})} \xi(t), \qquad (6.3.17)$$

where $\dot{\lambda} = d\lambda/dt$ - rate of change of the measured variable.

We assume that $\lambda = \text{const}$ and introduce equivalent discrimination $a_{333}(\varepsilon)$ and fluctuation $S_{3333}(\varepsilon)$ characteristics according to equalities

$$\boldsymbol{a}_{\text{DRB}}(\boldsymbol{\epsilon}) = \frac{\boldsymbol{\sigma}(\boldsymbol{\epsilon})}{K_{\text{A}}}, \quad \boldsymbol{S}_{\text{DRB}}(\boldsymbol{\epsilon}) = \frac{\boldsymbol{S}(\boldsymbol{\epsilon})}{K_{\text{A}}^2}. \tag{6.3.12}$$

Then (5.3.17) is rewritten in the form

$$\frac{de}{dt} = K_{\rm E}K_{\rm g}[e_{\rm V} - a_{\rm DNB}(e)] - K_{\rm E}K_{\rm g}V \overline{S_{\rm DNB}(e)}\xi(t), \qquad (6.3.19)$$

where $\epsilon_{\rm V} = \lambda/{\rm K_M}{\rm K_A}$ -- magnitude of dynamic error in a linearized system (see § 6.2). According to results of Paragraph 6.3.1 the corresponding diffusion equation

will be recorded in the form

$$\frac{\partial W}{\partial t} = -\frac{\partial}{\partial \epsilon} [K_{\pm} K_{\pm} (\epsilon_{V} - \omega_{\pm \pi \pm} (\epsilon)) W] + \frac{(K_{\pm} K_{\pm})^{2}}{2} \frac{\partial^{2}}{\partial \epsilon^{2}} [S_{\pm \pi \pm} (\epsilon) W]. \qquad (6.3.20)$$

-35-

Consequently, the coefficient of drift in tracking meters is determined by the discrimination, and the coefficient of fusion, by the fluctuation characteristics.

The stationary solution, if it exists, is determined by relationship $(6.3.1\circ)$. In the presence of limitations at point: l_1 , l_2 it should be normalized in this range, so that

$$W_{0YAR}(\mathbf{e}) = \frac{C_1}{S_{0KS}(\mathbf{e})} \exp\left[-\frac{2}{K_RK_R} \int \frac{a_{0KS}(\mathbf{e}) - \mathbf{e}_V}{S_{0KS}(\mathbf{e})} d\mathbf{e}\right] \\ (l_1 < \mathbf{e} < l_2), \\ \overline{\mathbf{e}^2} = C_1 \int_{l_1}^{l_2} \frac{\mathbf{e}^2}{S_{0KS}(\mathbf{e})} \exp\left[-\frac{2}{K_RK_R} \int \frac{a_{0KS}(\mathbf{e}) - \mathbf{e}_V}{S_{0KS}(\mathbf{e})} d\mathbf{e}\right] d\mathbf{e}, \\ C_1^{-1} = \int_{l_1}^{l_2} \frac{1}{S_{0KS}(\mathbf{e})} \exp\left[-\frac{2}{K_RK_R} \int \frac{a_{0KS}(\mathbf{e}) - \mathbf{e}_V}{S_{0KS}(\mathbf{e})} d\mathbf{e}\right] d\mathbf{e}. \end{cases}$$
(6.3.21)

Being limited to the case of a constant fluctuation characteristic $(S_{3KB}(\varepsilon) = S_{3KB})$, which is a good approximation during large noises, instead of (6.3.21), we have

$$\begin{aligned}
\Psi_{\text{STREW}}(\mathbf{e}) &= C_{s} \exp\left\{-\frac{1}{\sigma_{\pi}^{2}}\int \left[a_{\text{SKB}}(\mathbf{e}) - \mathbf{e}_{V}\right]d\mathbf{e}\right\},\\ \overline{\mathbf{e}^{3}} &= C_{s} \int_{I_{a}}^{I_{a}} \exp\left\{-\frac{1}{\sigma_{\pi}^{2}}\int \left[a_{\text{SKB}}(\mathbf{e}) - \mathbf{e}_{V}\right]d\mathbf{e}\right\}d\mathbf{e},\\ C_{y}^{-1} &= \int_{I_{a}}^{I_{a}} \exp\left\{-\frac{1}{\sigma_{\pi}^{2}}\int \left[a_{\text{SKB}}(\mathbf{e}) - \mathbf{e}_{V}\right]d\mathbf{e}\right\}d\mathbf{e},\\ \end{array}$$
(6.3.22)

where

$$a_{\mu}^2 = 2S_{\mu\kappa\mu}\Delta_{\mu\nu\phi}^{\mu} = S_{\mu\kappa\mu}\frac{K_{\mu}K_{\mu}}{2}$$

- variance of fluctuating error of measurement in a linearized system.

Further concretization of the solution requires assignment of an approximation for $a_{akB}(\epsilon)$. Let us assume that

$$a_{\text{ORB}}(s) = \frac{A}{s} \sin \frac{ss}{\Delta}, \quad l_1 = -l, \quad l_2 = l, \quad s_V = 0,$$

where as before Δ - half-width of the selected range.

Then*

*These results were obtained by I. P. Zelinikov.

-36-

$$W_{\text{eray}}(\mathbf{e}) = \begin{cases} \frac{1}{2\Delta} \frac{\exp\left[u\left(1 + \cos \pi \frac{\mathbf{e}}{\Delta}\right)\right]}{n - 1 + e^{u}I_{0}\left(u\right)} & \text{when } |\mathbf{e}| < \Delta; \\ \frac{1}{2\Delta} \frac{1}{n - 1 + e^{u}I_{0}\left(u\right)} & \text{when } \Delta < |\mathbf{e}| < l, \\ 0 & \text{when } |\mathbf{e}| > l, \end{cases}$$
(6.3.23)

where $n = \frac{1}{\Delta}$; $\mu = \frac{1}{n^2} \left(\frac{\Delta}{e_x}\right)^2 \approx 0.1 \left(\frac{\Delta}{e_x}\right)^2$ - dimensionless quantity, proportional

to the square of the ratio of the width of the selected range to mean square error in a linearized system.

For variance σ_{HJI}^2 of error, considering the phenomenon of breakoff and restoration, we obtain from (6.3.23) the general expression

$$\sigma_{nn}^{2} = \frac{A^{2}}{n-1+e^{n/2}} \left\{ \frac{1}{3} (n^{2}-1) + \frac{1}{3} \left(\frac{1}{3} (n^{2}-1) + \frac{1}{3} \left(\frac{1}{3} \right) + \frac{1}{n^{2}} \sum_{k=1}^{\infty} (-1)^{k} \frac{I_{k}(u)}{k^{2}} \right\} \right\}.$$
(6.3.24)

In particular, for very large noises, when u < 1 and the probability of breakoff is great,

$$\left(\frac{e_{nx}}{\Delta}\right)^{2} \approx \frac{1}{3} \frac{n^{4} - 1 + e^{u}/_{0}(u) \left[1 - \frac{12}{\pi^{2}} \frac{f_{1}(u)}{f_{0}(u)}\right]}{n - 1 + e^{u}/_{0}(u)}.$$
(6.3.25)

According to (6.3.25) when $u \ll 1$, $c_{\pi\pi}^2 \approx \frac{l^2}{3}$, i.e., variance corresponds to uniform distribution of ε in segment (-1, 1).

In the reverse case, when u > 1 (probability of breakoff is small),

$$\left(\frac{a_{na}}{\Delta}\right)^{2} \approx \frac{\frac{1}{3}(n^{2} - 1)\sqrt{2\pi u} e^{-2u} + \frac{1}{u} \frac{1}{\pi^{2}}}{1 + \sqrt{2\pi u} (n-1) e^{-2u}}.$$
 (6.3.26)

When $u \gg 1$, from (6.3.26) it follows that

 $\sigma^2_{_{_{\rm I\!I\!I\!I\!I}}}\approx\sigma^2_{_{_{\rm I\!I\!I\!I}}},$

i.e., breakoff practically does not occur, and the linear approximation in analysis is fully sufficient. Ratio $\sigma_{\rm R}/\Delta$, at which there occurs increase of the probability of breakoff, can be easily determined by the graph of Fig. 6.6, where there is plotted ratio $\sigma_{\rm HR}/\Delta$ as a function of $\sigma_{\rm R}/\Delta$. The phenomenon of breakoff occurs approximately at $\sigma_{\rm R}/\Delta \approx 0.08$ to 0.12. Here $\sigma_{\rm HR}/\Delta$ starts to increase sharply and is rapidly stabilized around level, determined by the limitation. Inasmuch as



ratio $\sigma_{\rm H}$ / Δ , at which the probability of breakoff sharply increases, depends little on n, it is possible to state that breakoff is a threshold phenomenon, scarting when $\sigma_{\rm H}$ / Δ exceeds a certain cutoff value, when fluctuation error comprises about 0.1 the nalf-width of the selected range.

Since with such a stationary distribution and ratios $\sigma_{\rm H}/\Delta$, somewhat below cutoff, there is implied a very large observation time, estimation of this cutoff value by the above method will be somewhat understated, which we shall prove in the following paragraph. A factor, acting in the opposite direction, is the presence of dynamic error, which we ignored in the considered example. Increased probability of breakoff here is explained by reduction of the distance to the edge of the discrimination characteristic, where restoring action of the discriminator ceases.

6.3.4. Average Time to Breakcif

Let us consider now the nonstationary phenomenon of breakoff, defined as emergence of the magnitude of mismatch



beyond the limits of a certain segment $(-\Delta, \Delta)$, where Δ - half-width of the range of selection (see Fig. 6.5). According to Paragraphs 6.3.1 and 6.3.3 average time to breakoff is determined by equation

$$\frac{(K_{\mathbf{n}}K_{\mathbf{A}})^{\mathfrak{s}}}{2}S_{\mathfrak{s}\mathfrak{n}\mathfrak{s}}(\mathfrak{s})\frac{d^{\mathfrak{s}}T_{\mathfrak{s}\mathfrak{s}}}{d\mathfrak{s}^{\mathfrak{s}}}-K_{\mathbf{n}}K_{\mathbf{A}}[a_{\mathfrak{s}\mathfrak{n}\mathfrak{s}}(\mathfrak{s})-\mathfrak{s}_{V}]\frac{dT_{\mathfrak{s}\mathfrak{s}}}{d\mathfrak{s}}+1=0,$$

-38-

我没知道你真正的好好的你的你的?""你想要你说的话,你们没有不可是要早吃

solving which, again on the assumption of zero dynamic error and constant fluctuation characteristic, for $\varepsilon = 0$ we have

$$T_{\mathfrak{e}\mathfrak{G}}(0) = \frac{1}{4\Delta f_{\mathfrak{e}\mathfrak{G}} \mathfrak{e}_{\mathfrak{A}}^{2}} \int dy \int \exp\left[-\frac{1}{\mathfrak{e}_{\mathfrak{A}}^{2}} \int \mathfrak{a}_{\mathfrak{e}\mathfrak{K}\mathfrak{b}}(\mathfrak{k}) d\mathfrak{k}\right] dz. \qquad (6.3.27)$$

To obtain a solution for an arbitrary magnitude σ_{π}^2 is difficult here. Therefore, we primarily investigate the range of intense noises, when it is possible to expand the exponential function under the integral sign in powers of the exponent, being limited to a small number of terms. Again given a sinusoidal approximation of the discrimination characteristic, we have asymptotic series

$$T_{06}(0) = \frac{1}{84f_{10}} \left(\frac{\Delta}{\epsilon_x}\right)^2 \left[1 + 0.08 \left(\frac{\Delta}{\epsilon_x}\right)^2 + 16 \cdot 10^{-4} \left(\frac{\Delta}{\epsilon_x}\right)^{4} + \frac{1}{\epsilon_x} + 0\left(\left(\frac{\Delta}{\epsilon_x}\right)^{6}\right)^{1}\right], \qquad (6.3.26)$$

convenient, however, only for comparatively large values of σ_{π} / Δ .

Graphic presentation of $\Delta f_{\text{Off}}T_{CG}$ as a function of σ_{π}/Δ in logarithmic and linear scales is given in Fig. 6.7. The linear scale helps us to understand the threshold effect of failure, in this case expressed in the fact that average time to failure sharply decreases (from a quantity of the order of 1000 to a quantity of order of units), when $\sigma_{\pi}/\Delta > 0.1$. The logarithmic scale permits us to definitize the concrete value of T_{CG} . However, for small σ_{π}/Δ formula (6.3.28) gives a somewhat lowered value of T_{CG} . More exact calculation, conducted by B. L. Karelov, showed that the threshold value of σ_{π}/Δ is approximately 0.2. In other words, threshold values, determined in terms of average time to breakoff and in terms of stationary variance, differ (approximately by a factor of 2). This is understandable, since with stationary variance there is implied a very large observation time, during which breakoff certainly occurs. Average time to breakoff is determined by realizations of mismatch, for which breakoff occurs for the first time.

Due to the approximate nature of analysis such a difference between results of two calculations should be considered immaterial, and, therefore, it is possible to use either of them during practical analysis. If smoothing circuits have a more complicated form, then, in general, critical ratio σ_{π} / Δ changes, but the obtained results can serve for approximate calculations even in these cases.

-39-



Fig. 6.7. Dependence of average time to failure on variance in linear approximation.

「「「「「「」」」

From analysis of tracking meters of concrete forms on the assumption of their linearity we can find σ_{π} for different forms of interferences. The value of the critical magnitude of ratio σ_{π} / Δ gives, thereby, a possibility of finding the intensity of interferences, leading to failure of these meters.

§ 6.4. General Analysis of Nontracking Meters

Besides the meters considered above, built on the principle of a servo system, in practice they also use nontracking measuring circuits. An example is a range finder with frequency modulation, in which the received signal is mixed directly with the radiated signal, and the measure of range is the number of zeroes of the resulting voltage [1]. Then there are Doppler speed meters built on the same principle of counting zeroes [2], angular data meters using two (or several) motionless radiation patterns of antennas for direct reading of the angle, multifilter systems and systems of gated amplifiers for measurement of range and speed, and so forth. For all such circuits it is possible to establish certain general properties of the method of measurement and to derive rules for determining accuracy.

> 6.4.1. Basic Features of Circuit Construction and Components of Measurement Errors

In general the considered meters, as also tracking meters, are divided into two parts (Fig. 5.8) — an inertialess unit for producing a signal, on the average proportional to the measured quantity (conditionally we call it the estimator unit), and open smoothing circuits. In practice it is rarely possible to immediately obtain





voltage which is on the average proportional to the measured parameter $\lambda(t)$. Thus, determination of angle by signals from two antennas with fixed patterns gives a nonlinear, and in a large interval even a monotonic, dependence of the mean value of output voltage of the coordinate; the average of zeroes in an FM range finder with time lags, comparable to the period of frequency modulation, is not proportional ' the range, etc. Therefore, processing of signals in the discriminator can often be divided into two stages.

In the first stage (unit 1 in Fig. 6.8) there will be formed a low-frequency voltage with a mean value, which is related to λ by a monotonic (not necessarily linear) dependence in necessary limits of values of the parameter:

$$\mathbf{z}_{1}(t, \lambda) = a(\lambda, t) + \xi(t, \lambda),$$
 (6.4.1)

where $a(\lambda, t) - mean value, for which <math>\frac{\partial a(\lambda, 1)}{\partial \lambda} > 0;$

$\xi(t) -$ fluctuating disturbance,

The second stage (unit 2 in Fig. 6.8) may consist of a certain inertialess nonlinear operation,

$$\mathbf{z}_{\mathbf{g}}(t, \lambda) = B\{\mathbf{z}_{\mathbf{i}}(t, \lambda)\},\$$

(2.4.2)

which has as its purpose obtaining of a magnitude, on the average as close as possible to λ . Smoothing circuits 3 in nontracking meters are usually linear and play, in principle, the same role as in tracking meters, but are constructed by an open-poop circuit so that their inertia directly determines the interval of smoothing.

In nontracking meters there are observed in general the same components of errors of measurement as in tracking meters: fluctuating, dynamic, systematic.

In view of the absence of a closed loop systematic errors in the considered circuits, appearing both due to insufficiently correct selection of nonlinear operator $B \$ in (6.4.2), and also due to incorrect selection of gain in any element of the meter, are not compensated during work of the meter. Appearance of such errors are an essential practical deficiency of the considered circuits. The dependence of these errors on selection of operator $B \$ will be studied below.

For nontracking meters it would have been possible to investigate the same set of questions appearing during investigation of accuracy as in tracking circuits. We will discuss below only the case when all fluctuating disturbances at the input are rapid as compared to the time of smoothing the meter.

6.4.2. Errors of Measurement During Rapid Fluctuations

In the shown conditions for $z_{i}(t, \lambda)$ notation, analogous to (6.2.6), is valid:

$$z_{1}(t, \lambda) = a(\lambda, t) + \sqrt{S(\lambda, t)} \xi(t),$$
 (6.4.3)

where $a(\lambda, t) - mean$ value;

 $C(\lambda, t)$ - spectral density at low frequencies;

L(t) - white noise of unit spectral density.

Functions $a(\lambda, t)$ and $S(\lambda, t)$ are obtained by averaging the whole ensemble of input fluctuations for a fixed value of λ . The obvious time-dependence of these functions is determined by regular changes of properties of the ensemble of fluctuations. Subsequently we shall consider $a(\lambda, t)$ and $S(\lambda, t)$ not time-dependent. Then we can write relationship (6.4.2), taking into account (6.4.3), in the form

-42-

$$\mathbf{z}_{\mathbf{a}}(t, \lambda) = B\{B^{-1}(k\lambda) + [a(\lambda) - B^{-1}(k\lambda)] - \sqrt{S(\lambda)}\zeta(t)\}, \qquad (6.4.4)$$

where $B^{-1}()$ - nonlinear inertialess operator, the inverse of B ();

k - proportionality factor.

Such coefficients are frequently encountered in the theory and practice of meters, inasmuch as measured quantities have the physical nature of distance, speed, angles, and the carriers of these quantities in circuits are voltages. After expansion of function B { } in (6.4.4) at point $B^{-1}(k\lambda)$; with preservation of only two members of the expansion, we have

$$z_{s}(t, \lambda) = B\{B^{-1}(k\lambda)\} + B'\{B^{-1}(k\lambda)\}(a(\lambda) - B^{-1}(k\lambda)) + B'\{B^{-1}(k\lambda)\}\sqrt{S(\lambda)}\zeta(t) =$$

$$= k\lambda + \frac{a(\lambda) - B^{-1}(k\lambda)}{\frac{\partial B^{-1}(k\lambda)}{\partial (k\lambda)}} + \frac{\sqrt{S(\lambda)}}{\frac{\partial B^{-1}(k\gamma)}{\partial (k\lambda)}}\zeta(t), \qquad (6.4.5)$$

where we use the formula for the derivative of an inverse function. If we translate $z_{0}(t, \lambda)$ into equivalent values of the measured variable, we obtain

$$\frac{z_{1}(t,\lambda)}{k} = \lambda + \frac{a(\lambda) - B^{-1}(k\lambda)}{k \frac{\partial B^{-1}(k\lambda)}{\partial (k\lambda)}} + \frac{V_{S}(\lambda)}{k \frac{\partial B^{-1}(k\lambda)}{\partial (k\lambda)}} \zeta(t).$$
(6.4.5)

The first term in $(6.4.5^{\circ})$ is the true value of the measured variable, the second is systematic error of a nontracking discriminator, and the third is fluctuating disturbance with intensity which depends on the current value of λ . We note that zero systematic error can be obtained only for $a(\lambda) = B^{-1}(k\lambda)$, i.e., when operator $B^{-1}()$ with an accuracy of a constant factor will convert $a(\lambda)$ into the true value of the parameter. Then, instead of $(6.4.5^{\circ})$ we have

$$\frac{\mathbf{z}_{1}(t,\lambda)}{h} = \lambda + \frac{\sqrt{S(\lambda)}}{\frac{\partial a'(\lambda)}{\partial \lambda}}\zeta(t).$$
(6.4.6)

However, such a relationship is difficult to achieve, inasmuch as the function $a(\lambda)$ in general is nonlinear and depends on the signal-to-noise ratio, and compensation of one nonlinearity by another is technically difficult to perform. Furthermore, as already indicated, error in the gain factor is in no way compensated in the circuit (if we do not take special measures, for instance, introduction of a periodically

-43-

fed control signal). Characteristics of the estimator unit are considered by us.

In view of linearity and openness of smoothing circuits further calculation of separate components of errors can be conducted independently. If, for instance, we are to determine fluctuation error in the circuit where there exists relationship (6.4.6), then for smoothing circuits with constant parameters, the frequency response of which* is equal to $G(i\omega)$ [pulse response g(t)], it is easy to obtain

$$\mathbf{d}_{\phi\pi}^{\mathbf{2}}(\boldsymbol{\lambda}) = 2G^{\mathbf{2}}(\mathbf{0}) \, S_{\mathbf{b}\mathbf{K}\mathbf{B}}(\boldsymbol{\lambda}) \, \Delta f_{\mathbf{b}\phi}, \qquad (\mathbf{0}, \mathbf{\lambda}, \gamma)$$

where $\Delta f_{p\bar{m}}$ - effective band, which is determined by relationship

$$\Delta \mathbf{j}_{\bullet \bullet} = \frac{1}{2\pi} \int_{0}^{\infty} \frac{|G(i\omega)|^{2}}{|G(0)|^{2}} d\omega = \frac{1}{2} \frac{\int_{0}^{\infty} g^{2}(l) dl}{\left|\int_{0}^{\infty} g(l) dl\right|^{2}}, \qquad (6.4.8)$$

and

$$S_{BRB}(\lambda) = S(\lambda) / \left| \frac{\partial a(\lambda)}{\partial \lambda} \right|^2$$

- equivalent spectral density of noises, depending on 2.

From (6.4.8) it follows that $S_{3KB}(\lambda)$ sharply increases in flat sections of the "discrimination characteristic" $a(\lambda)$, where $\left|\frac{\partial a}{\partial \lambda}\right| \rightarrow 0$. This phenomenon is explained by the fact that a small fluctuating disturbance is received here as a considerable deviation of the measured variable.

If the method of processing in the estimator unit in the first stage is already selected and dependence $a(\lambda)$ has been found, then no circuit manipulations can subsequently increase accuracy in flat sections of $a(\lambda)$, nor can we obtain uniqueness of measurement for those values of $a(\lambda)$ where monotonic dependence of this function on λ is disturbed. Inasmuch as in tracking circuits accuracy usually does not dependence on the absolute value of the measured variable, this circumstance can be considered one more deficiency of nontracking circuits.

In spite of the superficial simplicity of relationship (6.4.7), we indicate that $\sigma_{\phi\pi}^2(\lambda)$ is conditional variance, calculated, strictly speaking, under the condition that the true value of the parameter is constant and equal to λ . In the case of a variable λ , comparable in rate of change to the inertia of smoothing circuits,

*We have in mind a circuit whose input is a fictitious point, where $U_{BX}(t) = z_2(t, \lambda)/k$.

relationship (6.4.7) should be reexamined.

Other components of measurement of error - dynamic and systematic (the first may also be caused by incorrect input of the mean value $\overline{\lambda(t)}$), which in distinction from errors considered in § 6.2 were omitted here, are calculated analogously to how this was done earlier in § 5.2.

§ 6.5. <u>Selection of the Method of Synthesis of Radar Meters</u>. Theory of Solutions in Problems of Measurement

Before passing directly to synthesis of radar meters, it is necessary in brief to discuss different possible approaches to this problem. Below we discuss a series of nonstatistical and statistical methods for the purpose of finding such method of synthesis which would not simply correspond in meaning to the problem, but would be the most theoretically consistent. The desire to be freed to the greatest extent from preconceived ideas of construction of meters and to obtain their complete circuit analytically is natural. Such a preconceived idea, for instance, is the principle of feedback. If it indeed permits us to obtain highest performance in a number of conditions, then this should follow from theory, and not be postulated a priori.

6.5.1. Different Nonstatistical and Statistical Criteria for Synthesis

One of the most elementary nonstatistical criteria is simplicity of the circuit solution of one or another radio device, its cost, power consumption, and so forth. A related statistical criterion is reliability, determined by the probability of malfunction of the meter in a certain assigned time. It is natural that such criteria always have to be considered in the creation of meters; however, by themselves, at the present stage of development of theory they cannot come near the problem of initial selection of the circuit of a device. Moreover, the tendency of contemporary technical development consists in constant increase of complexity of devices for execution of ever more complicated problems, sometimes sacrificing realiability, costs, power consumption, and so forth. Thus, these very simple criteria evidently are not satisfactory, in any case in the first stage of synthesis.

It is absolutely clear that the criterion of synthesis should be intimately connected with the basic quality of the meter — accuracy of measurement. At first glance the matter is facilitated by the fact that there exists a large adjacent branch of science — the theory of control. Here there has been obtained a mass of results using different, mostly nonstatistical criteria (stability, invariance with respect to different disturbances, minimization of time of flow of the transient, and so forth) and various methods of mathematical solution (classical calculus of

-45-

variations, method of the Pontryagin principle of a maximum, linear, nonlinear and dynamic programming, the theory of games, and so forth; see, for instance, $[3, n]^4$, [65]).

Insufficiency of the mentioned criteria and methods in reference to synthesis of radar meters becomes evident, if we recall (see § 6.1) the circumstances that work of radar devices always occurs in the presence of random interferences, and selection of the class of possible laws of change of the measured variable a nonstatistical variables statistically. Moreover, in general, it is impossible to prove that a meter, possessing very high qualities with respect, for instance, to stability invariance and speed of flow of transients, simultaneously in good during measurement of the randomly varying parameter of mixture of the signal with noise.

Deficiencies of the above mentioned approaches, consisting in their nonstatistical nature, force us to turn to statistical criteria and methods.

First we shall discuss a mixed criterion widely used in the literature - the criterion of a minimum of fluctuating and zero dynamic error for a limited set-up time. According to this criterion we seek a minimum mean square random (fluctuating) error and zero (in steady-state regime) error from a definite change of measured variable. In application to meters here we consider the circuit of discriminator given, and we consider the object of synthesis the smoothing circuit. A deficiency of the criterion is the inconsistency in its approach to the measured variables, of which we already spoke above. Its result is that the meter, optimum according to this criterion, sometimes does not ensure minimum total mean square error.

A strictly statistical criterion can come from information theory. Thus, in [4, 5] it is shown that systems, ensuring minimum mean risk, simultaneously ensure a minimum of uncertainty of the reproduced signal. In [6] for serve systems of certain assigned structure there are introduced information theory concepts, in which accuracy is compared with a number of possible states of the system, and high speed operation is compared with the speed of transmission of information and carrying capacity. However, final relationships between accuracy of measurement and informational properties of complex input signals and methods of their processing, which would be useful for the problem of measurement, have yet to be fixed, so that to apply information theory criteria is difficult.

Very commonly used in radio practice and the practice of servo systems are

-46..

considerations about the signal-to-interference ratio at different points of electronic devices. The larger this ratio, the better, it is considered, the device works. This opinion, in general, is incorrect, and in meters this criterion is also difficult to use because it is unclear at what point of the circuit one should measure the signal-to-noise ratio. If, for instance, we consider the discriminator output, when, as is shown in § 6.2, it can be completely characterized by the equivalent spectral density of noises. Although the latter depends monotonically on the signalto interference ratio at the circuit input, introduced specially for the given forms of signal and interference, it cannot be expressed in terms of this ratio alone.

In a systematic statistical approach to random inputs and measured quantities most natural is the criterion of a minimum of some averaged monotonic function of total error of measurement. Averaging is conducted for the ensemble of noises and parameters. Such an approach in one or another form long ago was applied for finding a certain part of the circuit of a meter.

In the first place here one should note that the extreme case, when the whole circuit of the meter (discriminator, smoothing circuits and the method of their closing) are considered given, with the exception of a small number of circuit parameters (gain factors, time constants, and so forth), which can be selected in the process of development to increase accuracy. This very simple method, however, gives a small degree of confidence that we have obtained a system, close to optimum, inasmuch as during selection of the general idea of the construction it is necessary to lean only on preceding experience of construction and invitation. It is possible only to construct guesses whether a small change of the circuit will yield significanlty better results. 「シーム」の「「「「「「「「「「「」」」」」」「「「「「」」」」」」」」

Further it is possible to indicate an approach, in which the general construction of the meter is considered the same as that described in Paragraph 6.2.1, the distribuinator circuit is considered given, and only smoothing circuits are optimized. Additionally (sometimes implicitly) it is assumed that the discriminator has linear properties relative to the tracked quantity. During synthesis of smoothing circuits by the criterion of minimum total error there can be applied various statistical criteria. The greatest achievement of the statistical trend from works published until recently is the theory of filtrati [7], put forth in the works of Kolmogorov and Wiener. In this theory it was assumed that from statistical evidence we know only correlation function of the tracked quantity and interferences, and smoothing circuits were sought

-47-

in the class of linear filters. These methods are very customary for specialists in the theory of automatic control; however, they, in general, pass over the problem of construction of a radio discriminator. Meanwhile, it is possible to assume that, with respect to total error, optimization of the discriminator is no less important than optimization of smoothing filters. Everything suggests that the best device of statistical synthesis is that which minimizes total error of measurement with the least number of initial assumptions about the structure of the synthesized device, desirably, without singling out a priori the discriminator and smoothing circuits. As the object of optimum analysis here one should consider the existing mixture of the signal, depending on the measure quantity, with interferences. We know a priori statistical evidence on the mixture and measured quantity, and from considerations connected with the use of results of measurement we select the performance criterion. With such a combination of conditions we should be interested in the branch of mathematics which studies the method of obtaining statistical inferences from an available sampling (realization) of a random process and a priori information. Such a branch is the theory of "tatistical solutions - a branch of mathematical statistics. One should now turn to the theory of solutions.

6.5.2. Application of the Theory of Solutions to Problems of Measurements

In Chapter 3 we gave a summary of the general propositions of the theory of solutions. Here we will repeat certain propositions, having a direct relationship to problems of measurement. Let us remember that the object of optimum processing (finding a solution) is the mixture of signals with noises $y(t, \lambda)$ at the input, statistical evidence about which is given by the likelihood function

$P(\mathbf{y}|\boldsymbol{\lambda}),$

where \mathbf{y} - the whole set of observed values of the mixture.

 λ - the set of measured variables

When there is observed one mixture y(t) at moments t_1 , t_2 ,..., t_n , by **y** one should understand the column vector

$\{y_1,\ldots,y_n\}=\mathbf{y}.$

If here, for instance, y(t) obeys normal distribution, the likelihood function is recorded in the form of relationship (1.3.2), which is generalized for the case of many input mixtures:

-48-

$$P(\mathbf{y} | \boldsymbol{\lambda}) = (2\pi)^{-\frac{mn}{2}} [\det \mathbf{R}]^{-1/2} \exp\left\{-\frac{1}{2}(\mathbf{y}-\mathbf{a})^{+} \mathbf{W}(\mathbf{y}-\mathbf{a})\right\} =$$

$$= (2\pi)^{-\frac{mn}{2}} [\det ||^{(np)} R_{ij} ||]^{-1/2} \times$$

$$\times \exp\left\{-\frac{1}{2} \sum_{i, j=1}^{n} \sum_{a, b=1}^{m} {}^{(ab)} \mathbf{W}_{ij} ({}^{(a)}y_{i} - {}^{(a)}a_{i}) ({}^{(b)}y_{j} - {}^{(b)}a_{j})\right\}.$$
(6.5.1)

Here there is introduced matrix notation; sign + signifies transposition;

$$\mathbf{y} = \{ {}^{(1)}y_1^{(2)}, y_1, \ldots, {}^{(m)}y_1, \ldots, {}^{(1)}y_n^{(2)}, y_n, \ldots, {}^{(m)}y_n \}$$

is a block column vector of sampled values with n elements in the form of simple columns $\{{}^{(1)}y_1, \ldots, {}^{(m)}y_i\} = y_i$, composed of values of all m mixtures at the i-th moment of time; **a** - analogous column vector of mean values; **R** = $||\mathbf{R}_{ij}||$ - complex correlation matrix with matrix-elements $\mathbf{R}_{ij} = ||{}^{\alpha\beta}\mathbf{R}_{ij}||$, consisting of values of the function of cross-correlation of the α and β mixtures at moments t_i and t_j , respectively; $\mathbf{W} = \mathbf{R}^{-1}$ - matrix, the reciprocal of \mathbf{R} .

In addition to all this matrices a, R and W may depend on parameters λ of input mixtures.

For continuous realizations of y(t), as noted above, it is convenient to pass to functionals of probability density $P(y(t)|\lambda)$. Sometimes it is more convenient to pass to a functional limit already at the end of analysis, during interpretation of the obtained operations.

On the basis of the observed realization a radar set selects solution $\hat{\lambda}(\mathbf{y})$ from the possible set of solutions. In problems of measurement the space of solutions ("estimates") in structure is like the space of possible values of measured variables, which will become evident somewhat latter. Inasmuch as $\hat{\lambda}(\mathbf{y})$ does not depend functionally on λ [although statistically it does depend on λ through $\mathbf{y}(t, \lambda)$], the statistical nature of the observed mixture (mixtures) leads to the possibility of random errors.

For estimating the significance ("value") of allowed errors of solution there is introduced loss function $I(\hat{\lambda}, \lambda)$, depending on parameters of the signal included in the mixture and on the taken estimates. The more the accepted solution corresponds to the truth, the less the value of $I(\hat{\lambda}, \lambda)$. Performance of a device is determined by the average (for all realizations of y) value of the loss function

$$f(\lambda, \hat{\lambda}) == \int I(\hat{\lambda}, \lambda) P(\mathbf{y} \mid \lambda) d\mathbf{y}, \qquad (6.5.2)$$

-49-

called the function of conditional risk, inasmuch as this function is determine: on the condition of an assigned value of λ . As follows from §6.1, measured parameters of a signal also can reasonably be considered random variables, and we can assign them a prior distribution $P_{0}(\lambda)$. Then there is also introduced the function of mean risk

 $R(P_{\bullet}, \hat{\lambda}) = \int r(\lambda, \hat{\lambda}) P_{\bullet}(\lambda) d\lambda =$ $= \int \int I(\hat{\lambda}, \lambda) P(\mathbf{y} | \boldsymbol{\lambda}) P_{\bullet}(\boldsymbol{\lambda}) d\mathbf{y} d\boldsymbol{\lambda},$ (6.5.5)

۰۰ ما میشارد. دول در موالد با موالد از موالد میشاند. موالد موالد موالد موالد موالد موالد موالد موالد موالد موال

which is the averaging of conditional risk for a priori distribution of parameters. The optimum operation of estimation we find from the condition of a minimum mean or conditional risk for an arbitrarily selected loss function. Radio interpretation of operations of optimum estimation reveals the structures of an optimum radar meter.

If a priori distribution $P_0(\lambda)$ is given, and the optimum meter is found by minimization of mean risk min $R(P_0, \hat{\lambda})$, the estimate is called a Bayes estimate.

In sucher case it is necessary to seek the so-called minimax estimate. It minimizes (by all rules of estimation) the maximum value of conditional risk for all possible value: if the parameter. Cases when there are assigned no limitations on the value of the parameter are very rare, and usually one should seek the minimax estimate with certain limitations, statistical and nonstatistical.

Basically we shall use the Bayes methods of synthesis; however in § 0.9 certain attention will also be paid to minimax methods.

Properties of the theory of statistical solutions are such that in it it is possible to introduce certain a priori assumptions about the structure of the synthesized device, starting synthesis from a certain point of the circuit. However, it is possible also not to introduce any assumptions, considering as the input value the output signal of the antenna unit or even the electromagnetic field in its aperture [52]. Obviously, the last cases are the most interesting and to the largest extent characterize the merit of the theory of solutions. Chronologically, nowever, earlier there were offered such solutions of the problem of measurement, where a series of elements of the circuit of a meter were assumed assigned, and the theory of solutions was applied for vartial synthesis.

Thus, in the 1950's there was developed a method of synthesis of optimum meters on the basis of a branch of the theory of statistical solutions — the theory of estimation [10, 11]. It was assumed that the measured parameter of motion of the target, coded in the input signal, is constant for a certain time of observation.

--50-

Proceeding from complete statistical evidence on the mixture of the signal with interferences there was found an optimum circuit for determining the unknown parameter for this time. Such a method is completely applicable, for instance, for direction finders which determine angular position for a small interval of time.

()

If, however, we turn to the general case of measurement of variable coordinates, the theory of estimation, assuming invariability of parameters in time, strictly opeaking, cannot be applied. This does not exclude, however, certain compromise solutions of the problem, consisting in assigning a general idea of the construction of the meter in that form which is known in practice, considering smoothing circuits linear. Here, it is postulated that the discriminator should give an estimate of current minatch between the true and measured values of the tracked parameter for time intervals during which these mismatches almost do not change. In such an approach it was possible to use the ideas and methods of the theory of estimation for synthesis of discriminators in meters of the class interesting us.

With respect to partial synthesis of smoothing circuits on the assumption of linearity of the discriminator with respect to the measured quantity there is applied the theory of optimum filtration, but with a new conceptual basis. It seems that with an additive mixture of the measured quantity and interference, with Gaussian distributions of each of them, Wiener filters are absolutely optimum, i.e., there does not exist a class of operators executing this task more successfully. Unfortunately, the formulated conditions are so limiting that in practice in pure form they are almost never encountered. Therefore, it is clear that the theory of filtration should somehow be adapted to nonlinear problems.

Consequently, all presented methods of partial synthesis, even that conducted on the basis of the theory of solutions, left doubt about the correctness of selection of the general idea of construction of meters. We desire complete synthesis, emanating only from statistical evidence about signals and parameters. On the assumptions of Markovian and Gaussian parameters this problem was solved in [16, 18, 25, 60]. To these results we shall pass somewhat later.

Here we shall make a number of remarks, simultaneously showing the promising nature of the devices of the theory of solutions and its limitations, which, surely, will be surmounted with further development of the theory.

An essential limitation of the theory of solutions is the necessity for the presence of a sizeable quantity of statistical evidence about the signal and the parameter. This circumstance led certain authors to the pessimistic conclusion that

-51-
the theory of colutions simply chifts uncertainty during synthesis from the disstructure to statistical properties of input signals. However, in defense of the theory of solutions it is possible to repeat all the arguments of (0.1 concerning the statistical nature of measurements. In practice we always have some statistical evidence or physical considerations about them. For further firming op of the propositions of the theory of solutions it is ungent to investigate from all face the criticality of synthesized circuits with respect to statistics established in them and to find classes of optimum operations ensuring for a given class of statistic qualities which are lower than optimum by not more than a given quality. Extending, also, are quostions of optimization for limited statistical evidence.

With the question of apriorism in statistical evidence there are all elements of classes of interferences on the meter input. In a number of cases, especially where organized interferences are possible, it is impossible to decide beforehand what will be the exact statistical characteristics of input signals. Therefore, it is reasonable to consider construction of an optimum meter, primarily of its elements instar, from two directions.

1) Simpler is synthesis of an optimum meter for those interference, which are present always (for instance, internet notice) or with high probability (for instance, chaotic reflections), and whose influence can be decreased by sational selection of the form of the signal and construction of the circuit. Here, it is possible to consider the statistical evidence on the mixture of the signal with interferences completely accigned, to sidering, possibly, several parameters of interference unknown.

2) Somewhat more complicated is protection from interferences which may appear only with tow probability or with a probability which it is impossible to entire e. Among interferences of this kind we include these, protection from the for assigned encosteristics of radar system cannot even be ensured by optimes includes of construction of the receiver, but, in principle, such protection never-the exact can be found in the course of changing other parameters of the signal. (for instance, genreal construction of the presence of active interference). It is futile to synthesize an optimum receiver (meter) on the assumption of the presence of all such interferences. This would remove us from the optimum for the case of mixture of the signal with interferences of the first group, not to talk of extraordinary increase of receiver, most expedient is synthesize of the signal with interferences. Therefore, most expedient is reputied of the construction of the case of the first group, not to talk of extraordinary increase of receiver complexity. Therefore, most expedient is rynthesize of the signal complexity.

-52-

automatic devices for indication of interferences of the second group for the purpose of sharp variation of parameters of the system of processing of signals on an instruction from the automatic device.

1

A modified variant should also be synthesized, based as far as possible on statistical evidence about the given form of interferences. In this direction there have not been obtained many results, if we are considering complicated forms a^{-1} interferences.

Variation of parameters of a receiver in the course of work, in principle, is already self-tuning of the meter circuit in accordance with the character of the received signal. At present, problems of analysis and synthesis of automatic systems with self-tuning (true, of somewhat different type) are the subject of the widest attention [3]. However, in our applications, inasmuch as we are interested in the pr blem of measurement with full assignment of statistical evidence, optimum circuits are obtained without a priori reliance on ideas of self-tuning.

Sometimes circuits of optimum meters contain self-tuning elements, and in other cases they are absent; however, the method of synthesis always guarantees absolute optimality of a circuit if we keep initial data constant. If in the distribution of probabilities of a mixture we start to consider unknown some additional parameter, and assign it its own distribution of probabilities, the principle of synthesis and quality of work of the circuit will not be changed whether we treat measurement of the additional parameter as self-tuning or not. Thus, a priori introduction of ideas of self-tuning will most likely be useful when we have incomplete knowledge of when there is a change of statistical properties of the mixture of the signal, interferences and the parameter.

There is also possible a game-theory approach to meters. At present the theory of solutions sometimes is included in the theory of statistical games, considering that particular case when the "game" is conducted against an "unintelligent antor mist," whose behavior in a statistical sense is known beforehand. If, however, the enemy is intelligent and tries to change the form of interference to bring on maximum error of measurement, and there is the possibility of changing characteristics of the meter for constant preservation of its optimality, there is observed a nondegenerate game-theory situation, which is very interesting for practical applications. Unfortunately, only the first steps have been made in this direction.

Everything said speaks to the fact that the devices of the theory of solutions are completely timely and at present are a powerful means of solution of the problem

-53-

6.5.3. Theory of Statistical Estimates

We shall now discuss in greater detail the theory of statistical estimates of parameters of distributions [8]. This is what we call the branch of mathematical statistics in which the probability density of a sampling (functional of the densit, of distribution of realizations) depends on one or several constant, but unknown parameters. These parameters must be estimated from sampled values of the input variable.

The most well-developed and convenient methods of finding estimates are the method of maximum a posteriori probability and the maximum likelihood method. The first method consists in differentiation with respect to parameter λ (we shall now consider it unique) of the product of the a priori distribution of parameter $P_0(\lambda)$ and likelihood function $P(\mathbf{y}|\lambda)$. With an accuracy of a factor, not depending on λ , this product is the a posteriori, i.e., the knowledge-from-experience probability density, of the parameter, which follows from the known formula of Bayes

$$P(\lambda | \mathbf{y}) \equiv \widetilde{P}(\lambda) = \frac{P_{\bullet}(\lambda) P(\mathbf{y} | \lambda)}{\int P_{\bullet}(\lambda) P(\mathbf{y} | \lambda) d\lambda} = \frac{1}{k} P_{\bullet}(\lambda) P(\mathbf{y} | \lambda), \qquad (6.5.4)$$

where k is a factor, not depending on λ_*

The result of differentiation is equated to zero, and we neek the root of the obtained equation which explicitly depends on \mathbf{y} :

$$\frac{\partial}{\partial \lambda} \left[P_{\bullet}(\lambda) P(\mathbf{y}|\lambda) \right] = 0, \qquad (0.5.5)$$

which is taken as the very best estimate.

According to the maximum likelihood method we differentiate the probability function, which gives the so-called likelihood equation

$$\frac{\partial}{\partial \lambda} P(\mathbf{y}|\boldsymbol{\lambda}) = \mathbf{0}. \tag{6.5.0}$$

Its root is called the maximum likelihood estimate. Although the theory of estimation was developed before the general theory of solutions, with the appearance of the latter it was possible to interpret the theory of estimation from more general properlytions. Here, we consider that the space of solutions is a segment of a straight line, on which there are plotted possible "estimates" of parameter " \hat{a} ." As the loss functions we usually consider:

the simple loss function

$$I(\lambda, \hat{\lambda}) = C - \delta(\lambda - \hat{\lambda}) \quad (C = \text{const}), \quad (2.5.7)$$

the quadratic loss function

$$J(\lambda, \hat{\lambda}) = (\lambda - \hat{\lambda})^{2},$$
 (6.5.8)

a loss function with saturation

$$I(\lambda, \ \hat{\lambda}) = 1 - \exp\{-a(\lambda - \hat{\lambda})^{2}\}$$

or function

$$I(\lambda, \hat{\lambda}) = |\lambda - \hat{\lambda}|.$$

If we select simple loss function (6.5.7), mean risk (6.5.3) is equal to

$$R(P_{\bullet}, \hat{\boldsymbol{\lambda}}) = C - \int P(\boldsymbol{y} | \hat{\boldsymbol{\lambda}}) P_{\bullet}(\hat{\boldsymbol{\lambda}}) d\boldsymbol{y}. \qquad (6.5.9)$$

For its minimization, equivalent to maximization of the integral in the right part of (6.5.9), it is necessary that for any fixed \mathbf{y} there be found a solution $\hat{\lambda}$, corresponding to a maximum of the integrand. In other words, the optimum estimate corresponds to a maximum of the a posteriori probability, i.e., is a root of equation (6.5.5). Hence, it is clear in what sense the estimate of maximum a posteriori probability is optimum. It is easy to prove that such an estimate additionally maximizes the probability of a correct solution. We also note that with a wide a priori distribution, the estimate of maximum a posteriori probability passes to the estimate of maximum likelihood.

We now take quadratic loss function (6.5.8). Here, mean risk (6.5.3) will take the form

$$R(P_{\bullet}, \hat{\lambda}) = \iint (\lambda - \hat{\lambda})^{*} P(\mathbf{y} \mid \lambda) P_{\bullet}(\lambda) d\lambda d\mathbf{y}.$$
(6.5.10)

It has a physical meaning of error of measurement of λ , averaged over \mathbf{y} and λ . In order to minimize $R(P_0, \hat{\lambda})$, it is necessary and sufficient for any \mathbf{y} to minimize by selection of $\hat{\lambda}$ the integral over λ in (6.5.10). Differentiating this integral with respect to $\hat{\lambda}$ and equating the result to zero, we have an equation for finding the estimate

$$\int (\lambda - \hat{\lambda}) P(\mathbf{y} | \lambda) P_{\bullet}(\lambda) d\lambda = 0.$$
 (6.5.11)

Formal solution of (6.5.11) is the conditional mathematical expectation of λ for the given **y**, i.e., the mean value of the a posteriori distribution:

$$\hat{\lambda}(\mathbf{y}) = \overline{\lambda(\mathbf{y})} \equiv \frac{\int \lambda P(\mathbf{y} \mid \lambda) P_{\mathbf{x}}(\lambda) d\lambda}{\int P(\mathbf{y} \mid \lambda) P_{\mathbf{x}}(\lambda) d\lambda} \equiv \int \lambda \widetilde{P}(\lambda) d\lambda. \qquad (6.5.12)$$

In distinction from the operator of estimation of maximum a posteriori probability the structure of the operator of the optimum estimate for a quadratic loss function depends on the form of the whole function of the a posteriori probability.

If one assigns a loss function of any form differing from the considered one in general there will be obtained other operators. And although the criterion of minimum mean square error, ensuing from application of a quadratic loss function, is the most customary, the variety of optimum solutions causes certain dissetionfaction. We would like to obtain a solution which would be optimum for a sufficiently broad class of cases. It turns out that for certain limiting assumptions the theory of Bayes estimates permits such a solution. To explain this circumstance we first indicate that estimates for simple and quadratic loss functions coincide where the a posteriori distribution of probabilities is a function which is symmetric with respect to a certain point. This is clear from the expressions obtained. In the theory of estimation strict mathematical proof of this fact is connected with the property of efficiency of estimates, by which we mean reaching the lower bound of variance of an estimate relative to the true value. It appears [8] that in a certain class of cases estimates of maximum a posteriori probability are efficient. Especially frequently there is noted asymptotic efficiency, i.e., efficiency for a time of observation, considerably greater than the interval of correlation of the slowest random variables of the mixture y(t). Proof of efficiency thereby simultaneously establishes the fact that two classes of estimates are identical.

We shall prove that with symmetry of a posteriori distribution any symmetric loss function leads to the same optimu: estimate. Let u. assume that the a posteriori probability has the form

$$\overline{P}(\lambda) = F(\lambda - \lambda_{0}),$$

where $F(\mathbf{x})$ is a function, symmetric with respect to zero, and the loss function is also symmetric, so that

$$I(\lambda-\hat{\lambda})=I(\hat{\lambda}-\lambda), \quad \frac{\partial I(\lambda-\hat{\lambda})}{\partial \hat{\lambda}}=-\frac{\partial I(\hat{\lambda}-\lambda)}{\partial \hat{\lambda}}.$$

Induction as at the point of the optimum estimate the derivative of mean risk with respect to the estimate is equal to zero, at this point this equation should be satisfied:

$$J \equiv \int \frac{\partial I(\lambda - \hat{\lambda})}{\partial \hat{\lambda}} \tilde{P}(\lambda) d\lambda = 0. \qquad (4.5.12^{\circ})$$

We express $\widetilde{P}(\lambda)$ in the left part of this relationship through $F(\lambda - \lambda_{U})$ and set

~56-

 $\hat{\lambda} = \lambda_0$. Then, changing the sign of the argument of integration, it is easy to prove that J = -J, from which it follows that J = 0. This proves our statement.*

The value of λ_0 , determining the point of symmetry of a posteriori probability, and equal to any estimate using a symmetric loss function, is, obviously, already the familiar conditional mathematical expectation (6.5.12). Thereby, conditional mathematical expectation in a wide range coincides with the maximum value of a posteriori probability and can serve as a universal optimum estimate. Thus, the method of a maximum a posteriori probability, and for a wide a priori distribution, the method of maximum probability, too, under conditions very little limited give a single Bayes estimate of the measured parameter.

Further it is useful to note one mathematical law, proved in the theory of estimation. The fact is that with a large time of observation and with a comparatively low level of input interferences the logarithm of the likelihood function (as a function of the parameter) is close in form to a parabolic curve, located near the true value of the parameter:

$$\ln P(\mathbf{y}|\boldsymbol{\lambda}) \approx \ln P(\mathbf{y}|\boldsymbol{\lambda}_{\mathbf{x}}) + \frac{\partial}{\partial \lambda} \ln P(\mathbf{y}|\boldsymbol{\lambda}_{\mathbf{x}}) (\boldsymbol{\lambda} - \boldsymbol{\lambda}_{\mathbf{x}}) + \frac{1}{2} \frac{\partial^{2}}{\partial \lambda^{2}} \ln P(\mathbf{y}|\boldsymbol{\lambda}_{\mathbf{x}}) \frac{(\boldsymbol{\lambda} - \boldsymbol{\lambda}_{\mathbf{x}})^{2}}{2}, \qquad (6.5.13)$$

where λ_{M} - true value of the parameter of the mixture y(t).

From sampling to sampling the vertex of this parabola shifts along both axes, and also changes in width. Naturally this parabola can be expanded at any point, close to points of the true and maximum-probable value, limited in all to three terms of the expansion. In any case we arrive at a curve, coinciding with (6.5.13).

If we now pass from the logarithm to the actual likelihood function, the given dependences will be expressed in the fact that the likelihood function will be approximated by a Gaussian curve, located somewhere near the point of the true value λ_{K} , pulsating and shifting from sampling to sampling. Here, as it is easy to show [8], the width of the curve, characterized by the coefficient of sharpness of the parabola (6.5.13) - $\frac{\partial^2}{\partial \lambda^2} \ln P(\mathbf{y}|\lambda)$, on the average remains equal to the mean square scattering of the position of its maximum value from the true λ_{K} , and

^{*}In such a proof it is necessary to use the assumption of an infinite domain of determination of the function of the a priori distribution and the loss function, although in practical applications this condition is usually not satisfied. However, in the same applications interesting only are cases when the width of the a posteriori probability distribution is considerably less than that of the a priori. Here, finiteness of boundaries of the domain of determination of the a priori distribution is not important.

both mean values are connected with variance of efficient estimate $\sigma_{\varphi\varphi}^2$ according to the formula

$$\mathbf{a}_{\mathbf{s}\mathbf{\varphi}}^{-2} = -\left\langle \frac{\partial^{\mathbf{z}}}{\partial\lambda^{\mathbf{z}}} \ln P(\mathbf{y} \mid \mathbf{\lambda}) \right\rangle_{\mathbf{y}} = \left\langle \left[\frac{\partial}{\partial\lambda} \ln P(\mathbf{y} \mid \mathbf{\lambda}) \right]^{\mathbf{z}} \right\rangle_{\mathbf{y}}.$$
 (6.5.24)

Such laws occur for $P(\mathbf{y}|\lambda)$ near the true value of the parameter. Bosider the main peak of the probability function there are also parasitic peaks, occurring both due to nonideal properties of the signal, and also due to noise disturbances. In the theory of estimation, where the parameter is strictly constant, it is possible to show that the relative magnitude of parasitic peaks with increase of the three of observation (the size of the sampling) gradually decreases. We shall discuss the influence of parasitic peaks in somewhat greater detail in § 6.6.

It is useful to explain properties of joint estimates of several parameters. The likelihood function is determined in this case in the space of all measured quantities $\{\lambda_1, \ldots, \lambda_l\} = \lambda$. Expansion (4.5.13) is replaced by

$$\ln P(\mathbf{y} \mid \boldsymbol{\lambda}) \approx \ln P(\mathbf{y} \mid \boldsymbol{\lambda}_{H}) + \sum_{i=1}^{l} \frac{\partial}{\partial \lambda_{i}} \ln P(\mathbf{y} \mid \boldsymbol{\lambda}_{H}) (\lambda_{i} - \lambda_{Hi}) + \frac{1}{2} \sum_{i,j=1}^{l} \frac{\partial^{2}}{\partial \lambda_{i} \partial \lambda_{j}} \ln P(\mathbf{y} \mid \boldsymbol{\lambda}_{H}) (\lambda_{i} - \lambda_{Hi}) (\lambda_{j} - \lambda_{Hj}).$$

The matrix, composed from mean values of second derivatives,

 $\mathbf{A} = \left\| - \left\langle \frac{\partial^{\mathbf{v}}}{\partial \lambda_i \partial \lambda_j} \ln P(\mathbf{y} \mid \lambda_{\mathbf{H}}) \right\rangle \right\|$

characterizes "sharpness" of the multi-dimensional peak and sometimes is called a rischer information matrix. Its value is that reciprocal matrix $\Sigma = A^{-1}$ consists of mean second moments of minimum errors of measurement of the parameter attained in the case of the existence of jointly efficient estimates. If coding of separate parameters in the signal y(t) are independent, then Σ - diagonal matrix with element e^2_{30} , $\frac{1}{30}$, equal to variances of minimum attainable errors of measurement of separate parameters. In this case the meaning of a minimum does not require further explorations. If, however, forms of coding, and consequently also of error of measurement, are interrelated, it is necessary to introduce into consideration a multi-dimensional ellipsoid, second central moments of which are determined by the matrix of second moments of errors of measurement. Then the introduce ellipsoid of scattering for any method of estimation of parameters will wholly contain in its meaning of the minimum of matrix Σ or interrelipted by matrix Σ , which explains the meaning of the minimum of matrix Σ or interrelipted by matrix Σ , which explains the meaning of the minimum of matrix Σ or interrelipted by matrix Σ , which explains the meaning of the minimum of matrix Σ or interrelipted by matrix Σ .

-58-

Radar applications of the theory of estimation consist, primarily, of the following. In the course of reflection of the sounding radar signal from a target parameters of the probability density of the mixture of the reflected signal and interferences at the receiver input turn out to depend on the coordinates and velocity of the target. Measuring (estimating) the modulation delay of the signal, its corrage frequency or direction of arrival, it would be possible, thereby, to corrermine necessary target position data. If during a certain interval of time these coordinates can be considered constant, there appears the classical problem of the theory of estimation: for a given realization of the mixture of signal and interferences, observed in a given interval of time, determine with minimum errors unknown parameters of the signal in this mixture.

First results in the synthesis of optimum radar meters were obtained, namely, on the basis of the theory of estimation [10, 11]. Intervals of observation were selected in such a way that during them target position data indeed remained constant. Under conditions when most widely used were pulse radars, the structure of the pulse signal, i.e., the division of it into periods of pulse trains (or pulse packs), gave the possibility of conveniently dividing the whole interval of observation into time intervals, satisfying the shown requirement of theory. Not being interested in questions of unification of data of separate periods, they investigated potential properties of measurement in each of them.

It is possible to indicate the following methods of synthesis of devices for processing signals, obtained in the literature on the basis of the theory of estimation. In [10, 11] and in a considerable number of later works they explicitly or implicitly assume that there is the possibility of consecutive or parallel finding of values of the likelihood function in the whole a priori domain of the measured parameter, as a result of which there can be found a point of maximum value which is the estimate. It requires no explanation that, in general, obtaining of the whole "scan" of the likelihood function is technically complicated, with the exception of a parameter in the form of time delay of modulation.

In [12, 13] there is offered another, simpler method of finding the mathematical likelihood estimate, based on the assumption that we know the value of parameter λ_1 , removed from the true value of the parameter of the input mixture a distance less than the width of the basic peak of the likelihood function. We shall explain the idea of this method. We take the expansion of the logarithm of a likelihood function of type (6.5.13) at point λ_1 and, differentiating it with respect to λ_1 equate it

-59-

to zero. Then, transforming the likelihood equation

$\frac{\partial}{\partial \lambda} \ln P(\mathbf{y} \mid \lambda_i) + \frac{\partial^2}{\partial \lambda^2} \ln P(\mathbf{y} \mid \lambda_i)(\lambda - \lambda_i) = 0$

we give an explicit expression for the estimate of maximum likelihood $\hat{\Sigma}$:

$$\hat{\boldsymbol{\lambda}} = \boldsymbol{\lambda}_{1} - \frac{\frac{\partial \ln P(\mathbf{y} \mid \boldsymbol{\lambda}_{1})}{\partial \boldsymbol{\lambda}}}{\frac{\partial^{2} \ln P(\mathbf{y} \mid \boldsymbol{\lambda}_{1})}{\partial \boldsymbol{\lambda}^{2}}}.$$

(6.5.14)

(6.5.15)

Thus, to find $\hat{\lambda}$ we need to schematically realize the first and second derivatives of the logarithm of the function at point λ_1 , assumed to be close to the true value of λ_1 .

If one were to try to extend methods of synthesis based on the theory of estimation to the case of a variable parameter, then this can be done only by means of easing known rigor. Thus, we assume that the circuit of the meter is already given in the form of a tracking system of the type which was studied in § 6.2. With respect to the discriminator we postulate that it should constantly issue with maximum accuracy an estimate of mismatch between the true value of a parameter and that value λ_p , which as a result of measurement proceeds to the discriminator from the smoothing circuits. As follows from (6.5.141), the optimum estimate of difference $\lambda - \lambda_p = \epsilon$ is

$$\hat{\mathbf{s}} = -\frac{\frac{\partial \ln P(\mathbf{y} | \lambda_p)}{\partial \mathbf{\lambda}}}{\frac{\partial^2 \ln P(\mathbf{y} | \lambda_p)}{\partial \lambda^2}}.$$

Under conditions when in a subinterval there is realized sufficient accumulation, this approximate relationship is satisfied:

$$\frac{\partial^{a}}{\partial \lambda^{2}} \ln P(\mathbf{y} \mid \boldsymbol{\lambda}) \approx \frac{\partial^{a}}{\partial \lambda^{2}} \ln P(\mathbf{y} \mid \boldsymbol{\lambda}),$$

i.e., the second derivative may be replaced by a quantity which does not depend $\phi^{(1)}$, and the optimum estimate of mismatch is simply proportional to the first derivative of the logarithm of the likelihood function, taken at point $\lambda = \lambda_{p}$.

nowever, such methods of synthesis of tracking meters are inconsistent inasmuch us the new assumptions introduced are beyond the frames of the theory of estimation. At the same time we know that measuring systems are always synthesized from conditions of compromise between fluctuating and dynamic errors, where the appearance of the latter is explained namely by changeability of the parameter in time. It world be desirable to find the optimum measuring system immediately as a whole, proceeding from the requirement of a minimum of resultant (smoothed) errors. Such a possibility, in principle, is given by filtration theory.

Before presenting principles of filtration theory we shall discuss one more application of the theory of estimation to the problem of optimization of radar or other meters. We have in mind determination of parameters of orbits of objects, finited in ballistic curves [14]. It is known that the path of motion of the center

gravity of an arbitrary body in space with determined external forces is determined by six constants parameters μ , for which, in particular, we can consider coordinates of the object and the components of velocity at a certain moment of time. If to outputs of discriminators of meters we add values of coordinates of their tuning with coefficients, equal to the gain factors of discriminators, then, according to results of § 6.2, disregarding parametric fluctuations, we have

$$x(t) = K_{X}\lambda(t) + K_{X}\eta(t), \qquad (6.5, 10)$$

i.e., additive mixtures of measured coordinates $\lambda(t)$ (angles, distances) with noises. If one were to consider that coordinates $\lambda(t)$ depend in turn on constant parameters μ , then for any time of observation we arrive at a problem of the classical theory of optimation. After determining parameters of the orbit it is possible to calculate arbitrary functions of these parameters, for instance, the measured coordinates themselves. It is implicitly assumed that the best estimates of parameters of the orbit lead to the best measurements of coordinates of the body. This is indeed so if errors of separate measurements are small.

6.5.4. Theory of Optimum Linear Filtration

Filtration is continuous reproduction ("measurement," "finding of an estimate") of a certain variable which is a parameter of an observed random process.

The theory of optimum filtration already has a known history and has been given a foundation by the fundamental works of Kolmogorov and Wiener on filtration [7]. Initially this theory was not connected at all with the theory of solutions and which based on the following postulates:

1) parameter $\lambda(t)$ and interference n(t) are additively mixed stationary random processes having different correlation functions. Realization of mixture $y(t) = \lambda(t) + n(t)$ is assigned on an interval, starting from an infinitely removed moment up to the current moment;

2) the sought optimum system should carry out a linear operation on the mixture;
3) the system should be physically realizable, i.e., its pulse response h(t)

-61-

should turn into zero for negative t;

8.3 W.

4) the criterion of optimality is minimum mean square error of reproduction of the parameter or of any linear functional of the parameter — a derivative, an integral, a certain expected value of the parameter, and so forth.

Upon fulfillment of these requirements one can be certain that pulse response h(t) of an optimum Wiener filter for direct reproduction of the actual parameter satisfies the following integral equation (the Wiener-Hoof equation):

 $\int R_{yy}(t-\tau) h(\tau) d\tau = R_{y\lambda}(t),$

where $R_{yy}(\tau) = [\lambda(t) + n(t)][\lambda(t + \tau) + n(t + \tau)] - autocorrelation function of mixture y(t):$

 $R_{y\lambda}(\tau) = [\lambda(t) + n(t)]\lambda(t + \tau) - \text{cross-correlation function } c$ the mixture and the parameter.

(f. .). 473

If we immediately try to apply to (6.5.17) the Fourier transform, we easily find the frequency response of an optimum filter, but it will belong to the class of filters unrealizable in real time. By the latter we mean filters whose pulse response is equal to zero for negative time arguments. If we are talking about filtration with direct delivery of the result, this circumstance is equivalent to the appearance of a response before application of a disturbance and contradicts the causality principle. To find physically realizable filters mathematicians have had to apply much more refined methods of the theory of functions of a complex variable. As a result, the solution of equation (6.5.17) has the form [7]

$$H(i\omega) = \frac{1}{\Psi(i\omega)} \left[\frac{S_{p\lambda}(\omega)}{\Psi(-i\omega)} \right]_{+} =$$

= $\frac{1}{2\pi\Psi(i\omega)} \int_{0}^{\infty} e^{-i\omega t} dt \int_{-\infty}^{\infty} \frac{S_{p\lambda}(\omega)}{\Psi(-i\omega)} e^{i\omega t} du,$ (6.5.18)

where

$$H(i\bullet) = \int_{0}^{\infty} h(t) e^{-i\bullet t} dt; \quad S_{y\lambda}(\bullet) = \int_{-\infty}^{+\infty} R_{y\lambda}(\tau) e^{-i\bullet \tau} d\tau,$$

 nn^{d}

$$\Psi(i \bullet) \Psi(-i \bullet) = S_{yy}(\bullet) = \int_{-\infty}^{+\infty} R_{yy}(\tau) e^{-i \sigma \tau} d\tau, \qquad (6.5.19)$$

is a special expansion (factoring) of complex variable ω into factors with zeroes and poles, correspondingly, in the upper and lower half-planes. Such an expansion is always feasible, for instance, when $S_{yy}(\omega)$ is a rational-fraction function of ω^2 with real coefficients. (In practice, approximation of spectral densities by such

-62- 📜

functions is always possible). The operation $[]_+$ in (6.5.18) signifies taking that component of the expression in the brackets, which has poles in the upper half-plane of complex variable ω . Analytically, this operation can also be expressed in the form of two integrals.

With further development of filtration theory it was possible to weaken postulates 1 and 4. Thus, they considered certain forms of a nonstationary parameter, a finite interval of realization and other criteria of optimality.

In particular, it was found that the pulse response of a Wiener filter in more general cases is determined by integral equation

$$\int R_{st}(l, s) k(s, s) ds = R_{s\lambda}(l, s), \qquad (6.5.20)$$

where $R_{yy}(t, \tau)$, $R_{y\lambda}(t, \tau)$ have their former meaning, but due to nonstationariness of the input mixture they depend on two variables.

It is important to emphasize that the class of operators in which we sought the best remained in all cases given.

In spite of the indicated limitations, filtration theory has great methodological value, and in a whole series of practical cases gives an explicit analytical expression for pulse responses of different filters [7]. For spectral densities in the form of rational-fraction functions of ω these filters belong to systems with constant parameters and are executed in the form of a set of RC- and LRC-circuits and linear inertialess amplifiers.

After application of the theory of solutions it was shown that for an additive mixture of Gaussian measured quantities and interferences with Gaussian distributions Wiener filters are absolutely optimum, i.e., there does not exist a class of operators executing this task more successfully [5].

We note two other important results. From [61] it follows that for an additive mixture of processes with non-Gaussian distributions optimum filter-extrapolators give only a small improvement as compared to the Wiener ones, correctly calculated for the correlation function of the non-Gaussian components. Further, in [62] it is shown that for an additive mixture of Gaussian interference with a signal, for which we know only the correlation function, the Wiener filter is optimum in the minimax sense, i.e., ensures the least mean square error for the worst forms of signals with an assigned correlation function. Thereby, Wiener filters for additive mixtures possess optimum properties of a very general nature. Filters, close in their properties to Wiener ones, will also appear in the problem of nonlinear

-63-

filtration of especially nonadditive mixtures, for which in § 6.8 we contine their consideration.

Certrin authors also proceed along the same path of finding an operator from a given class in solving problems of nonlinear filtration [15]. However, inasmuch as each has the right to select arbitrary classes of nonlinear operators, this path leaves us unsatisfied: it is always possible to hypothesize the presence of a still better solution.

6.5.5. Filtration of a Markovian Parameter by the Method of a Posteriori Probability

Above we proved that in equations for optimum operators there enters the a posteriori probability of a parameter (parameters) for a given realization of the signal (signals). In other words, forming of the a posteriori probability is a sufficient primary operation of an optimum system, after which it only remains to take the solution, best from the point of view of some criterion. A receiver, forming the a posteriori probability for a constant parameter, is known in the literature as a Woodword-Davis receiver. The Bayes formula may, however, also be recorded for a parameter varying in time. Considering that the observation is conducted at discrete moments t_1, t_2, \ldots, t_n , it has formula

 $\widetilde{P}_{n}(\lambda_{1},\ldots,\lambda_{n})=C_{n}P_{n}(\lambda_{1},\ldots,\lambda_{n})P(y_{1},\ldots,y_{n}|\lambda_{1},\ldots,\lambda_{n}), \quad (6.5.21)$

where $\lambda_i = \lambda(t_i)$ (i = 1, ..., n);

 P_n , \widetilde{P}_n - multi-dimensional a priori and a posteriori distributions;

 $P(y_1, \ldots, y_n | \lambda_1, \ldots, \lambda_n)$ - multi-dimensional probability function;

 C_n - normalizing quantity, not depending on λ_i .

Investigation of the operator of a posteriori probability (6.5.21) without concretization of the character of change of the parameter in time is difficult. broad class of random processes, embracing the mass of cases interesting for application, is the class of Markovian processes (see § 6.3). The problem of corming the a posteriori probability for Markovian varying parameters, arbitrarily coded in the input signal y(t), was studied by R. L. Stratonovich in [16-18]. Although the method of R. L. Stratonovich essentially does not belong to the theory of solutions, it is important for the theory of measurements. We will present it, without holding to the ground work of the account in [16].

We assume that moments of observation of the input mixture y(t) are separate so that all random variables, on which we perform averaging when finding the

-64-

likelihood function, at various moments are not connected, so that $P(y_1, \ldots, y_n | \lambda_1, \ldots, \lambda_n)$ is broken up into the product

$$P(y_1, \ldots, y_n | \lambda_1, \ldots, \lambda_n) = \prod_{i=1}^n P(y_i | \lambda_i).$$
(6.5.22)

Using (6,3,2) and (6.5.22), we record (6,5.21) for two adjacent moments of size vation:

$$\widetilde{P}_{n}(\lambda_{1}, \ldots, \lambda_{n}) = C_{n}P_{1}(\lambda_{1})P(y_{1}|\lambda_{1})\prod_{i=2}^{n} W(\lambda_{i}|\lambda_{i-1})P(y_{i}|\lambda_{i}),$$

$$\widetilde{P}_{n+1}(\lambda_{1}, \ldots, \lambda_{n+1}) =$$

$$= C_{n+1}P_{1}(\lambda_{1})P(y_{1}|\lambda_{1})\prod_{i=2}^{n+1} W(\lambda_{i}|\lambda_{i-1})P(y_{i}|\lambda_{i}),$$

whence

t J

$$\widetilde{P}_{n+1}(\lambda_1, \ldots, \lambda_{n+1}) =$$

$$= C_{n+1}^0 \widetilde{P}_n(\lambda_1, \ldots, \lambda_n) W(\lambda_{n+1} | \lambda_n) P(y_{n+1} | \lambda_{n+1}), \qquad (6.5.23)$$

where $c_{n+1}^{0} \approx c_{n+1} | c_{n}$.

Let us now turn to final a posteriori probabilities, characterizing the a posteriori distribution of the parameter in the last step with observations in the preceding ones. This is carried out by integration of \widetilde{P}_k over all arguments λ_1 , except the last one λ_k . Then from (6.5.23) we have

$$\widetilde{P}_{n+1}(\lambda_{n+1}) = C_{n+1}^{0} P_{n+1}(y_{n+1}|\lambda_{n+1}) \int \widetilde{P}_n(\lambda_n) W(\lambda_{n+1}|\lambda_n) d\lambda_n. \quad (6.5.24)$$

The physical interpretation of relationship (6.5.24), subsequently important, is sufficiently clear from Fig. 6.9. For forming the a posteriori distribution in



Fig. 6.9. Filter of a posteriori protability: 1 - unit for formation of the likelihood function; 2 - unit for delay and processing of a posteriori probability. one step it is necessary to transform the a posteriori distribution of the preceding moment by a transition function, which permits us to consider the expect . character of change of the parameter in the step, and to multiply the result of the transformation by the value of the likelihood function of the next measure-

ment, characterizing the newly rriving information about the parameter.

Formula (6.5.24) was derived above on the assumption of discrete observation. However, it has a wider domain of applicability. If the useful component in y(t) has a pulse structure, where all random , puts (noises, interference, fluctuations)

-65-

of reflecting surface) are independent from pulse to pulse, then after passage to the limit of continuous observation for the likelihood function this relationship is valid:

 $P(y(l)|\lambda_i, \ldots, \lambda_n) = \prod_{i=1}^n P(y(l)|\lambda_i),$

analogous to (6.5.22). In (6.5.25) as elements there enter likelihood functions for all i-th cycles of observation, depending on the value of parameter λ_1 , which remains constant within the limits of the i-th pulse. In other words, if we are not interested in evolution of the a posteriori distribution in intervals between pulses, relationship (6.5.24) remains applicable, where quantization of the parameter is dictated by the character of the input signal y(t); and observation of the realization of the latter is considered continuous.

If, however, the signal has arbitrary structure, and we are interested in the form of the a posteriori distribution continuously, it is necessary to seek some new methods of mathematical description of the a posteriori probability, eliminating the discrete nature of relationship (6.5.24).

With passage to the limit of continuity the analog of relationship (6.5.22) is function

 $P(y(t)|\lambda(t)) = \exp \int_{0}^{t} l(y(s), \lambda(s), \tau) d\tau,$

(6.5.26)

(6.5.25)

whe re

 $(t_0, t) - interval of observation;$

 $l(y(s), \lambda(s), \tau)$ - certain functional of y(s) and $\lambda(s)$ for $s \in (t_0, \tau)$ (sometimes this is simply function $y(\tau)$, $\lambda(\tau)$, which we shall prove below with examples).

Then relationship (6.5.24) will pass to

$$P(\lambda, t+\Delta) = = C(t, \Delta) P_{t,\lambda}(y(t)|2) \int \vec{P}(\mathbf{p}, t) \Psi(\lambda, t+\Delta|\mathbf{p}, t) d\mathbf{p}.$$

(6.5.27)

He re

$$P_{l,\lambda}(y(l)|\lambda) = \exp \int_{-1}^{l+\lambda} l(y(s), \lambda(s), \tau) d\tau$$

- likelihood function in interval (t, t + β), in which we consider the parameter to vary so little that it is practically possible to identify $\lambda(t)$ with its value at the end of the interval.

We shall be interested in parameters in the form of diffusion processes with density of the probability of transition satisfying the Fokker-Planck equation

-66-

and the second second

(see § 6.3):

$$\frac{\partial}{\partial t} \mathbf{W}(\lambda, t | \lambda_{0}, t_{0}) = -\frac{\partial}{\partial \lambda} [A(\lambda) \mathbf{W}] + \frac{1}{2} \frac{\partial^{4}}{\partial \lambda^{2}} [B(\lambda) \mathbf{W}] \equiv Q \{ \mathbf{W}(\lambda, t | \lambda_{0}, t_{0}) \}, \qquad (6.5.28)$$

where Q[] - linear differential operator.

We consider that $W(\lambda, t | \lambda_0, t_0)$ satisfies the condition of continuity

$$\lim_{t \to t_0} \Psi(x, t \mid x_0, t_0) = \delta(x - x_0),$$

from (6.5.28) it follows that for small Δ

$$\int \vec{P}(\mu, t) \frac{1}{\Delta} [W(\lambda, t + \Delta | \mu, t) - - W(\lambda, t | \mu, t)] d\mu \approx Q\{\vec{P}(\lambda, t)\}$$

$$\int \vec{P}(\mu, t) W(\lambda, t + \Delta | \mu, t) d\mu \approx \vec{P}(\lambda, t) + Q\{\vec{P}(\lambda, t)\}\Delta. \qquad (6.5.29)$$

or

We now turn out attention to the fact that for small Δ there is satisfied

$$P_{i,\lambda}(y(i)|\lambda) \approx 1 + \Delta \left[\frac{1}{\Delta} \int_{i}^{i+\lambda} l(y(s), \lambda(s), \tau) d\tau\right], \qquad (6.5.30)$$

where the quantity contained in "mackets is a finite quantity as $\Delta \rightarrow 0$ and from considerations of statistical "malence can be identified with the integrand. We also consider that $C(t, \Delta)$ for mall values of Δ is close to 1, so that it is possible to consider

$$C(l, \Delta) \approx 1 - v\Delta, \qquad (6.5.31)$$

「「「「「「「」」」」」

where ν is still to be determined. Expanding $\widetilde{\mathbf{P}}(\lambda, t + \Delta)$ in a series with respect to Δ , substituting in (6.5.27) relationships (6.5.28)-(6.5.31) and being limited everywhere to first powers of Δ , we have equation

$$\frac{\partial}{\partial t} \widetilde{P}(\lambda, t) = Q\{\widetilde{P}(\lambda, t)\} + [I(\lambda, t) - v]\widetilde{P}(\lambda, t). \qquad (6.5.32)$$

If we integrate both parts of (5.5.32) over λ and consider decremss of \tilde{P} as $|\lambda| \rightarrow \infty$. it turns out that

$$\mathbf{v} = \int l(\lambda, t) \widetilde{P}(\lambda, t) d\lambda \equiv \overline{l(\lambda, t)},$$

from which we finally have the differential equation for a posteriori probability,

$$\frac{\partial}{\partial t} \widetilde{P}(\lambda, t) = Q\{\widetilde{P}(\lambda, t)\} + [t(\lambda, t) - \overline{t(\lambda, t)}] \widetilde{P}(\lambda, t). \qquad (6.5.33)$$

Thus, with continuous observation the ϵ volution of a posteriori probability is given by a differential equation of the Fokker-Planck type, the right side of which

-67-

contains linear operator Q, characterizing the transition function, and a component which depends on the input realization. The first term leads to "spreading" of $\widetilde{P}(\cdot, t)$ due to regular and random changes of the parameter; the second, conversity, leads to "narrowing" of peak of $\widetilde{P}(\lambda, t)$ due to newly arrived information. The measure of information is the proportional differences between the convert ($\alpha, \beta \in \mathbb{R}^{n}$) of the logarithm of the likelihood function and the value of the same function, averaged over the a posteriori distribution, which is formed by data arriving earlier.

In §6.3 we already said that to solve an equation of diffusion type is simileven in the absence of randomly varying coefficients in the right part. In the given case the second term contains random components of the type of white helps. But for forming the a posteriori probability it is sufficient to find a technicate convenient method of its simulation. Such a problem is solved, for instance, is [15]. Here we shall avoid simulation of general solutions and their investigation, results and differential equations for extrain basic characteristics of the a posteriori distribution.

Let us consider the case of a small a posteriori inaccuracy, when $\widetilde{\mathbb{P}}(\lambda, t)$ is a narrow (as compared to the a priori distribution), isolated peak on the axis of values of the parameter. In view of the imposition of a great number of rendom fractors it is reasonable to assume that $\widetilde{\mathbb{P}}(\lambda, t)$ with sufficient accuracy is a scribed by a Gaussian curve*:

$$\widetilde{P}(\lambda, t) = \frac{1}{|t| 2\pi \sigma(t)} \exp\left\{-\frac{[\lambda - \lambda_{0}(t)]^{2}}{2\sigma^{2}(t)}\right\}, \quad (1, 2, 2, 2)$$

where $v_{ij}(t) = the current vertex of the a posteriori distribution, which we cometime : the take as the optimum estimate of <math>\lambda(t)$;

 $\mathcal{F}_{1}(t)$ - the variance of the a posteriori distribution.

Exerciteting (0.5.34) in (5.5.33), expanding $A(\lambda)$, $B(\lambda)$, and $t(\gamma, \beta)$ is even the second $\gamma = -1$ and equating coefficients for identical powers (1, 1, 2, 3), we have equation

$$\frac{d}{dt} \lambda_{\bullet}(t) = A(\lambda_{\bullet}(t)) + l'(\lambda_{\bullet}(t), t) \sigma^{2} \langle \cdot \rangle_{t}$$

$$\frac{d}{dt} \sigma^{\bullet}(t) = 2\sigma^{\bullet}(t) A'(\lambda_{\bullet}(t)) + B(\lambda_{\bullet}(t)) + - + \sigma^{\bullet}(t) l''(\lambda_{\bullet}(t), t).$$
(*...)

"There are not sufficiently convinting proofs of an applicability of this opproximation for a Markovian parameter in [16-18]. A whole series of considerations

-68-

In the derivation of (6.5.35) we allowed for a small a posteriori scattering, in which coefficients A(λ) and B(λ) vary little, so that, e.g., $\sigma^2 B''(\lambda) \ll B(\lambda)$, $\sigma^2 A''(\lambda) \ll A(\lambda)$.

If $\lambda_0(t)$ is considered the optimum estimate, joint solution of equations (6.5.35) is a problem of optimum filtration for a small a posteriori inaccuracy. The solution, naturally, can be produced by a certain dynamic system, simulating

were first obtained, was not given. The analysis of measurement of varying frequency given in [17, 18] does not exhaust the problem, inasmuch as relationships (0.5.35) in that are not used there. Therefore, we delay schematic interpretation of these results to § 6.9 and give it after presenting basic results of the theory of optimum meters of Gaussian parameters. In the same place we will study recurrence relationships for curves of an a posteriori distribution, which for a pulse incoherent signal or a discrete observation can be derived directly from (6.5.24).

6.5.6. Limitations of Theory. General Theory of Bayes Filtration

We shall give results flowing from consideration and comparison of different branches of statistical theory in reference to the problem of meter synthesis. The basic deficiency of the theory of estimation is the requirement of constancy of the measured variable during the time of observation. The theory of linear filtration requires additivity of the measured variable and interferences, which, in practice, is usually not satisfied. The method of forming the a posteriori probability of Markovian processes, generally speaking, lacks these deficiencies, but it was developed independently of the theory of solutions, so that for its use in designing a meter one needs certain concretization of the results of this theory. At the point time the theory of solutions in principle permits formulating a general filteration problem, i.e., the problem of finding an optimum resolving filter, under constitions in no way limiting in engineering practice. A general theory of filteration is best constructed on the basis of the Bayes approach.

Again we consider random process y(t) a mixture of useful components and various kinds of interferences, depending on the measured function $\lambda(t)$, and is Paragraph 5.5.5, parameter $\lambda(t)$ is a random function of times. Let y(t) and $\lambda(t)$ take at moments $t_1 < t_2 \dots < t_n$ from interval (t_0, t) values $\{y_1, \dots, y_n\} = y$, $\{\lambda_1, \dots, \lambda_n\} = \lambda$, respectively. Problem of Bayer filtration consists of construction.

[FOOTNOTE CON'T FROM PRECEDING PAGE]

justifying this approximation for a Gaussian parameter will be given in [5,6,6].

-69-

of a vector function from samples $\hat{\lambda}(y)$ which is estimate λ , in some measure statistically close to the true realization of the parameter $\lambda(t)$.

The vector analog of the simple loss function is

$$I(\lambda, \hat{\lambda}) = C - \delta(\lambda - \hat{\lambda}) = C - \prod_{i} \delta(\lambda_{i} - \hat{\lambda}_{i}).$$

As it is easy to prove analogously to Paragraph 6.5.4, the optimum estimate here corresponds to a vector, turning into maximum a posteriori distribution (0.5.24). The system of equations of maximum a posteriori probability will be recorded

$$\frac{\partial}{\partial \lambda_j} \ln \tilde{P}(\lambda_1, \ldots, \lambda_n) = 0 \quad (j = 1, \ldots, n). \quad (5.5.35)$$

「日本市市市は正き

「中国市市でいっていてい

التابع معيقاته للم كالالتماطيات ... والمدت الالتامة

In that same case, when the estimate can use only the preceding data, the solution is sought in the form

$$\hat{\mathbf{\lambda}}_{j} = \hat{\mathbf{\lambda}}_{j} (y_{1}, \ldots, y_{j}) \quad (j = 1, 2, \ldots).$$

The a posteriori probability density takes the form

$$\widetilde{P}(\lambda_j) \equiv P(\lambda_j | y_1, \ldots, y_j) =$$

$$= \int_{(j-1)} \int P(\lambda_1, \ldots, \lambda_j | y_1, \ldots, y_j) d\lambda_1 \ldots d\lambda_{j-1} =$$

$$= k \int_{(j-1)} \int P(y_1, \ldots, y_j | \lambda_1, \ldots, \lambda_j) P_0(\lambda_1, \ldots, \lambda_j) d\lambda_1 \ldots d\lambda_{j-1},$$

and system of equations (6.5.36) is replaced by

$$\frac{\partial}{\partial \lambda_j} \ln \tilde{P}(\lambda_j) = 0.$$
 (6.9.37)

The analog of the scalar quadratic loss function leading to minimum mean square error of measurement for any interval of measurement is in the vector case the quadratic formula

$$I(\lambda, \hat{\lambda}) = (\lambda - \hat{\lambda})^{+} B(\lambda - \hat{\lambda}) = \sum_{i,j} B_{ij} (\lambda_{i} - \hat{\lambda}_{i}) (\lambda_{j} - \hat{\lambda}_{j}), \qquad (1, j, j\emptyset)$$

where B - certain symmetric non-singular matrix.

Here, mean risk is equal to

$$R(P_{\bullet}, \hat{\lambda}) = \iint (\lambda - \hat{\lambda})^{+} B(\lambda - \hat{\lambda}) P(y \mid \lambda) P_{\bullet}(\lambda) d\lambda dy$$

and is a linear combination of two moments of errors of measurement, allowed at all moments of observation. Thereby, we minimize the ellipsoid of scattering of 411 joint estimates, which is the most general requirement on it. In the particular case when matrix B is diagonal, mean risk is simply the weighted sum of variances of errors at separate moments of time. Variation of mean risk on the estimate vector $\hat{\lambda}$ gives a system of equations, analogous to (6.5.11), outside of dependence on the concrete form of matrix B:

$$\int (\lambda - \hat{\lambda}) P(\mathbf{y} | \lambda) P_{\bullet}(\lambda) d\lambda = 0.$$
 (6.5.39)

Again the formal solution of (6.5.39) is the conditional mathematical expectation of vector

$$\hat{\boldsymbol{\lambda}}(\mathbf{y}) = \frac{\int \boldsymbol{\lambda} P(\mathbf{y} \mid \boldsymbol{\lambda}) P_{\bullet}(\boldsymbol{\lambda}) d\boldsymbol{\lambda}}{\int P(\mathbf{y} \mid \boldsymbol{\lambda}) P_{\bullet}(\boldsymbol{\lambda}) d\boldsymbol{\lambda}}.$$
(6.5.40)

In expanded form one should understand expressions (6.5.39) and (6.5.40) as

$$\int_{(n)} \int (\lambda_j - \hat{\lambda}_j) P(y_1, \dots, y_n | \lambda_1, \dots, \lambda_n) \times \\
\times P_e(\lambda_1, \dots, \lambda_n) d\lambda_1, \dots, d\lambda_n = 0, \\
\hat{\lambda}_j = \frac{\int_{(n)} \int \lambda_j P(y_1, \dots, y_n | \lambda_1, \dots, \lambda_n) P_e(\lambda_1, \dots, \lambda_n) d\lambda_1, \dots, d\lambda_n}{\int_{(n)} \int P(y_1, \dots, y_n | \lambda_1, \dots, \lambda_n) P_e(\lambda_1, \dots, \lambda_n) d\lambda_1, \dots, d\lambda_n} \\
(j = 1, 2, \dots, n)$$
(6.5.41)

and use n relationships jointly.

1 }

When the operator of estimation uses only preceding data, there is taken the set of last relationships from (6.5.41):

$$\frac{\int \dots \int (\lambda_{j} - \hat{\lambda}_{j}) P(y_{1}, \dots, y_{j} | \lambda_{1}, \dots, \lambda_{j}) \times}{\times P_{\bullet}(\lambda_{1}, \dots, \lambda_{j}) d\lambda_{j}, \dots, d\lambda_{j} = 0,} \\
\hat{\lambda}_{j} = \frac{\int \dots \int \lambda_{j} P(y_{1}, \dots, y_{j} | \lambda_{1}, \dots, \lambda_{j}) P_{\bullet}(\lambda_{1}, \dots, \lambda_{j}) d\lambda_{1}, \dots, d\lambda_{j}}{\int \dots \int P(y_{1}, \dots, y_{j} | \lambda_{1}, \dots, \lambda_{j}) P_{\bullet}(\lambda_{1}, \dots, \lambda_{j}) d\lambda_{1}, \dots, d\lambda_{j}}.$$
(6.5.42)

Here the estimate for λ_j is found once, when $t = t_j$; more precise definition of this estimate after finding the new segment of realization of y(t) is not produced. Therefore the solution of (6.5.41) does not coincide with the solution of (1.42), with the exception of the last moment of observation, for which corresponding equations from (6.5.41) and (6.5.42) coin.ide.

For the multidimensional case all timese basic tencts which were considered in detail in the theory of estimation (Paragraph 6.5.3) remain basically in force. In particular, there remains in force that proposition that with a posteriori

distribution of form

$$P(\lambda) = F(\lambda - \lambda_{\bullet}),$$

where F(x) - arbitrary function, symmetric relative to zero, and for a symmetric loss function $(I(\lambda - \hat{\lambda} = I(\hat{\lambda} - \lambda))$ there exists a certain unique solution of the problem of Bayes filtration, coinciding with the operator of conditional matrix strictical expectation (6.5.40). Proof of this repeats that for the scalar case.

Besides the property that the operator of mathematical expectation (6.5, 40)minimizes on the average a large class of loss functions, it also always reduces to zero the value of errors of measurement, averaged in the ensemble of the measured parameter $\lambda(t)$ and signal y(t). This property is expressed by equality

$$\langle \hat{\mathbf{\lambda}} \rangle_{\mathbf{\lambda},\mathbf{p}} = \langle \mathbf{\lambda} \rangle_{\mathbf{\lambda}}, \qquad (6.5.43)$$

where brackets denote averaging. Relationship (6.5.43) is easy to obtain by multiplying both parts of (6.5,40) by the joint probability density $P(\mathbf{y}|\boldsymbol{\lambda})P_{O}(\boldsymbol{\lambda})$ of \mathbf{y} and $\boldsymbol{\lambda}$ and integrating over \mathbf{y} and $\boldsymbol{\lambda}$. The property of unbiasedness is especially useful in view of the presence of interfering signals structurally similar to the useful one. Complete absence of bias error in an optimum measuring system in these conditions is far from obvious.

All the given results emphasize the universal character of conditional mathematical expectation. The condition of symmetry of the a posteriori probability, formulated above, is satisfied in practice by a narrow level of signals, and the requirement of symmetry of the loss function is not a substantial limitation at all. In these conditions the theory of optimum filtration is freed from subjective elements connected with arbitrariness in selection of the loss function.

Considering the very attractive properties of the operator of conditional mathematical expectation, one would think it remains only to decipher the mathematical operation in (6.5, 40) and translate it into radio engineering lane curve. However, rigorous integration can be performed only in the additive case, when

$\mu(t) = n(t) + \lambda(t),$

where $\lambda(t)$, n(t) - random processes with normal distribution. Here, we arrive at the known Wiener filter. In other cases to digorously and sufficiently simply decode the operation of conditional mathematical expectation is impossible. Inasmuch as it appears certain that the method of construction of optimum meters should be based on the above-presented theory, the problem of technical creation of an optimum circuit has still greater importance than discovery of general overators.

-72-

§ 6.6. <u>Methods of Synthesis of Optimum Meters with Gaussian</u> <u>Statistics of the Parameter. Potential</u> <u>Accuracy of Measurement</u>

6.6.1. Properties of the Likelihood Function and Its Approximation

For the time being we shall assume that the measured parameter λ is constant. Inc. In the case of Gaussian noises, as was shown in [4, 12], the likelihood function of one parameter monotonically depends on the sum of the autocorrelation function of the useful component of the mixture y(t) and a certain random function, where both functions have as their argument the measured parameter. For illustration we shall consider the likelihood function of time delay of a regular signal u(t), received against a background of white noise with spectral density N_Q. The likelihood function here has the form

$$P(y(t)|\tau) = C \exp\left\{-\frac{1}{2N_{\bullet}}\int_{0}^{T} [y(t) - u(t-\tau)]^{2} dt\right\}, \qquad (6.6.1)$$

where y(t) - the observed realization;

 τ - time delay;

(_)

C - constant, not depending on y and τ .

After substitution in (6.6.1) of the form of realization

 $y(t) = u(t - \tau_0) + n(t),$

where $\tau_0 - \text{true}$ (unknown) value of delay, it is possible to separate members, explicitly depending on τ :

$$P(y(t)|s) = C' \exp\left\{\frac{1}{N_{\bullet}}\int_{0}^{T}u(t-s)u(t-s)dt + \frac{1}{N_{\bullet}}\int_{0}^{T}u(t-s)n(t)dt\right\}.$$

The first component under the sign of the exponential function is the autocorrelation function with respect to parameter τ , depending on difference $\tau - \tau_0$; the second is the noise disturbance, realizations of which are "signal-like" and stationary with respect to τ [4]. Analogous dependences occur, too, in more complicated cases [12]. With Gaussian input noises the autocorrelation function turns out to be in the exponent of the exponential function. The exponential function emphasizes the tenof the peak of the autocorrelation function and makes the slopes steeper. This reads to an approximately Gaussian curve of the likelihood function in conditions of a smooth top of the autocorrelation function, low level of its side peaks, of which we spoke in Chapter 1, and low level of interferences (Fig. 6.10).

With increase of time of observation the mean value of the exponent of the exponential function increases. As a result peak $P(\mathbf{y}|\lambda)$ narrows, and seathering of



Fig. 6.10. Likelihood function of a constant parameter and its approximation: $\hat{\lambda} = \text{varue of}$ optimum estimate; $\hat{\lambda} = \text{maximum likelihood value;}$ $\hat{\lambda}_{,i} = \text{true value of parameter; 1) true$ dependence; 2) Gaussian approximation.

its position relative to the true value $\lambda_{\rm H}$ decreases. This is explained by the fact that $\Gamma(\mathbf{y}|\lambda)$ characterizes a result averaged for all observations, in which random disturbances to a certain extent compensate one another.

If we turn now to the case of a parameter varying in time by arbitrar, law is general we can only ray a little about the form of the likelihood function.

In general, this function will be multi-dimensional, and with transition to continuous observation it will be a functional, and should be considered in the space of values of a parameter, occurring at separate moments of or servation. The produced measurements, forming for a constant parameter a narrowed peak, will the varying parameter give results depending both on the concretely occurring backer of the fluctuating disturbance, and also on the law of change of the parameter.

Sometimes it is possible, indeed, by selection of new parameter which is a function (functional) of the old parameter (for instance, speed, amplitude of the Fourier-harmonic, etc), to arrive at the case of a constant parameter. However, such cases are more the exception and cannot serve as a basis for analysis.

Meanwhile, on certain assumptions, made below, the form of the likelihood function found for a constant measured parameter permits us to qualitatively determine the form of this function for a varying parameter $\lambda(t)$ and on this basis

-74-

to find a satisfactory form of its approximate presentation.

In practice usually most interesting are the cases of sufficiently high-quality work of meters, when noises are not too great (but, ultimately, are also not too small, otherwise the problem of optimization loses its importance). Under this condition errors of measurement will be small and the a posteriori distribution will be marrow as compared to the a priori.

Frequently, the rate of change of the measured parameter is very slow as compared to the rate of change of all other random variables in the observed mixture, which are not subject to measurement and for which we realize averaging when finding the explicit form of likelihood function (henceforth we will call them immaterial parameters).

We choose time intervals considerably larger than the interval of statistical coupling of the slowest of the immaterial parameters of the input signal y(t), but considerably smaller than the interval of change of $\lambda(t)$. Upon fulfillment of the shown conditions the logarithm of the likelihood function, as a function of the value of the parameter frozen on these segments, near the main peak can be approximated by its own quadratic expansion for any point, knowingly close to the value of the measured variable. In distinction from the case of a constant parameter the peak of the likelihood function at every subinterval has a finite width, randomly depending on y(t) and the coded function $\lambda(t)$. The form of the likelihood function in the subintervals is illustrated in Fig. 6.11. The multi-dimensional likelihood function of likelihood functions obtained in separate subintervals due to the statistical independence of fluctuations of the immaterial parameters. Considering everything we have said and still holding to cases of discrete samplings, we can write the basic multi-dimensional approximation of $P(\mathbf{y}|\mathbf{\lambda})$ in the form

$$P(\mathbf{y} | \boldsymbol{\lambda}) = \exp \left\{ \ln P(\mathbf{y} | \boldsymbol{\lambda}_{0}) + \frac{1}{2} \sum_{i=1}^{n} \frac{\partial}{\partial \lambda_{i}} \ln P(\mathbf{y} | \boldsymbol{\lambda}_{0}) (\boldsymbol{\lambda}_{i} - \boldsymbol{\lambda}_{0i}) + \frac{1}{2} \sum_{i,j=1}^{n} \frac{\partial^{2}}{\partial \lambda_{i} \partial \lambda_{j}} \ln P(\mathbf{y} | \boldsymbol{\lambda}_{0}) (\boldsymbol{\lambda}_{i} - \boldsymbol{\lambda}_{0i}) (\boldsymbol{\lambda}_{j} - \boldsymbol{\lambda}_{0j}) \right\}, \quad (6.6.2)$$

where $\lambda_0(t)$ - an arbitrary function, considered at moments of observation t_1 , t_2 , ..., t_n and assumed close to the true value of the varying parameter.



Fig. 6.11. Likelihood "buttles of a variable parameter in the plane of the measured parameter and time.

An approximation of type (6.6.2) is used in the theory of estimation [see formula (6.5.13)] for finding operations of formation of an estimate and proof of asymptotic efficiency of maximum likelihood estimates. Here, there is proven the asymptotic convergence of the quadratic expansion of the $P(\mathbf{y}|\boldsymbol{\lambda})$ to its true visc.

Extension of such an expansion to the case of a variable parameter, in general, should be firmly grounded. However, at present a sufficiently satilizationy direct solution of this problem does not exist. Therefore, we will post-one proof of the applied presentation of the likelihood constitution until we investigate the assuracy of the obtained optimum circuits. ... this investigation we shall show that accuracy of meters synthesized by the shown expansion of the likelihood function practically coincides with the maximum possible (potential) accuracy, which proves the variably of this expansion.

Subsequently we will need two concrete expansions of type (6.6.2). The first is conducted at the point of the optimum estimate of parameter $\lambda_{C} = \hat{\lambda}$ (the meaning of optimality will be explained below). Designating

$$\ln P(\mathbf{y}|\boldsymbol{\lambda}) = L(\boldsymbol{\lambda}), \quad \frac{\partial L(\hat{\boldsymbol{\lambda}})}{\partial \lambda_i} = z_i, \quad -\frac{\partial^* L(\hat{\boldsymbol{\lambda}})}{\partial \lambda_i \partial \lambda_j} = A_{ij}$$

and introducing matrix notation with column-vectors $\mathbf{z} = \{\mathbf{z}_1, \ldots, \mathbf{z}_n\}, \lambda = \{\lambda_1, \ldots, \lambda_n\}$ and $\hat{\boldsymbol{\lambda}} = \{\hat{\boldsymbol{\lambda}}_1, \ldots, \hat{\boldsymbol{\lambda}}_n\}$ and square matrix $\mathbf{A} = \| \mathbf{A}_{i,j} \|$, we have instead of (6.6.2)

$$P(\mathbf{y}|\boldsymbol{\lambda}) = P(\mathbf{y}|\boldsymbol{\hat{\lambda}}) \exp\{\mathbf{z}^{+}(\boldsymbol{\lambda}-\boldsymbol{\hat{\lambda}}) - \frac{1}{2}(\boldsymbol{\lambda}-\boldsymbol{\hat{\lambda}})^{+} \mathbf{A}(\boldsymbol{\lambda}-\boldsymbol{\hat{\lambda}})\}.$$
(6.6.3)

The second expansion is conducted at the point $\lambda_0 = \tilde{\lambda}$ of the maximum value of $P(\mathbf{y}|\mathbf{\lambda})$ and has the form

$$P(\mathbf{y}|\boldsymbol{\lambda}) = P(\mathbf{y}|\boldsymbol{\lambda}) \exp\left\{-\frac{1}{2}(\boldsymbol{\lambda}-\boldsymbol{\lambda})^{\dagger}\boldsymbol{A}(\boldsymbol{\lambda}-\boldsymbol{\lambda})\right\}. \quad (6.6.4)$$

The point of the current position of maximum likelihood $\lambda(t)$ in general does not coincide with the point of the optimum estimate $\hat{\lambda}(t)$. Actually, according to § 6.5 the latter is the exact maximum of the a posteriori probability, calculated on the basis of the whole interval of observation, so that $\lambda(t)$ and $\hat{\lambda}(t)$ are correlated as the result of instantaneous measurement and as the smoothed output quantity.

The term with the first derivative in (6,6,4) is rejected as equal to zero, and matrix $\tilde{\mathbf{A}}$ is determined through derivatives at point $\tilde{\mathbf{\lambda}}$:

$$\mathbf{\tilde{A}} = \begin{bmatrix} \frac{\partial^2 L(\tilde{\lambda})}{\partial \lambda_i \partial \lambda_j} \end{bmatrix}. \tag{6.6.5}$$

Relationships (6.6.3) and (6.6.4) will be used below during the synthesis of basic circuits of optimum meters. For clarity in Fig. 6.10 we give a one-dimensional illustration of these expansions.

Let us assume that we nevertheless wanted to more closely account for the structure of the likelihood function. It is expressed with side peaks, determined as side peaks of the autocorrelation function of the signal (see Chapter I), and also purely noise overshoots, which by force of properties of the likelihood function are "signal-similar" in structure. Inasmuch as the exponential function always stresses maxima and suppresses minima of the function which is its argument, for these peaks it is possible to offer a Gaussian approximation. Side peaks of the

-77-

autocorrelation function of a signal are rigidly joined in position with the balls peak, and for them it is possible to offer two local approximations, $a_{t,a}|_{offed} = c$ (0.6.3) and (6.6.4):

$$P(\mathbf{y}|\boldsymbol{\lambda}) \approx P(\mathbf{y}|\boldsymbol{\hat{\lambda}} + \Delta_{(i)}) \exp\left\{ \boldsymbol{x}_{(i)}^{+} (\boldsymbol{\lambda} - \boldsymbol{\hat{\lambda}} - \Delta_{(i)}) - \frac{1}{2} (\boldsymbol{\lambda} - \boldsymbol{\hat{\lambda}} - \Delta_{(i)})^{+} A_{(i)} (\boldsymbol{\lambda} - \boldsymbol{\hat{\lambda}} - \Delta_{(i)}) \right\}, \qquad (\hat{\boldsymbol{U}}, (\boldsymbol{\nu}, \boldsymbol{\omega}))$$

$$P(\mathbf{y}|\boldsymbol{\lambda}) \approx P(\mathbf{y}|\boldsymbol{\hat{\lambda}}_{(i)}) \exp\left\{ -\frac{1}{2} (\boldsymbol{\lambda} - \boldsymbol{\hat{\lambda}}_{(i)})^{+} \boldsymbol{\hat{A}}_{(i)} (\boldsymbol{\lambda} - \boldsymbol{\hat{\lambda}}_{(i)}) \right\}, \qquad (\hat{\boldsymbol{U}}, (\boldsymbol{\nu}, \boldsymbol{\omega}))$$

Here $\Delta_{(1)} = \operatorname{known}$ displacement of the mildle of the 1-th peak from the context of the basic peak;

 λ - as before, the estimated value of λ ;

 $\lambda_{(1)}$ - measured position of i-th maximum;

$$\mathbf{z}(i) = \frac{\partial}{\partial \lambda} \ln P(\mathbf{y} | \hat{\boldsymbol{\lambda}} + \boldsymbol{\Delta}(i)),$$

and matrices $\mathbf{A}_{(1)}$, $\mathbf{\tilde{A}}_{(1)}$ consist of values or second derivatives of $-\ln P(\mathbf{y}|\mathbf{\lambda})$, one of for relationships (6.6.6) and (6.6.7), respectively at points $\mathbf{\tilde{\lambda}} + \Delta_{(1)}$ and $\mathbf{\tilde{\lambda}}_{(1)}$. Properties of quantities $\mathbf{z}_{(1)}$, $\mathbf{A}_{(1)}$, $\mathbf{\tilde{\lambda}}_{(1)}$ can be investigated for every concrete form of $P(\mathbf{y}|\mathbf{\lambda})$ absolutely the same as properties of quantities $\mathbf{z}_{\mathbf{A}}$, $\mathbf{\tilde{\lambda}}$ from (6.6.7) and (6.6.4).

If, however, we turn to analysis of fide noise peaks, then for them, obviously, only approximation (6.6.7) is suitable, and properties of functions $i(\mathbf{y}|\boldsymbol{\lambda}_{(1)})$ and $\mathbf{A}_{(1)}$ will be different here. Expression (6.6.0) is inapplicable, since the position of noise peaks is not connected with the true value of $\boldsymbol{\gamma}$.

Approximations (6.6.6) and (6.5.7) will be used below for deeper study of the structure of optimum meters.

6.6.8. Optimum Tracking Meter for Gaussian Statistics of the Measured Paramolor

Now, when there has been prepared a mathematical description of the likelihood function, it is necessary to give the concrete form of the multi-dimensional distribution of probabilities of the measured parameter.

Below right up to § 6.9 we will be interested in Gaussian distribution, which, besides mathematical convenience, is explained by a number of other factors.

First, in a considerable number of important applications the random character of change of $\lambda(t)$ is determined only by the randomness of values of certain initial conditions. The latter do not depend on the will of man and when there is a great

, -78-

number of disturbing factors they can be considered normally distributed, which leads to a normal (or close to it) distribution of $\lambda(t)$.

Second, change in time of the actual parameter $\lambda(t)$ frequently occurs under the action of a large number of independent random disturbances, which also leads to a normal distribution of probabilities. Finally, the statistics of the parameter may incompletely known, and may be given only by a correlation function. As we shall prove in § 6.9, with such limited knowledge the assumption of normal distribution ensures the best results as compared to any possible distributions with a given correlation function.

Thus, let us assume that the a priori distribution has the following form:

$$P(\lambda) = (2\pi)^{-n/2} \left[\det \mathbb{R} \right]^{-1/2} \exp \left\{ -\frac{1}{2} (\lambda - \bar{\lambda})^+ \mathbb{V} (\lambda - \bar{\lambda}) \right\}, \quad (6.6.8)$$

where $R = || R(t_i, t_j) || - correlation matrix of random component <math>\lambda(t)$; $\mathbf{V} = \mathbf{R}^{-1} - its$ inverse matrix [in interval (t_0, t_n)];

 $\overline{\lambda} = \{\overline{\lambda}_1, \ldots, \overline{\lambda}_n\} - \text{column-vector of mean value.}$

Let us turn to relationship (6.5, 42) for a physically realizable operator of filtration.

The first method of finding an optimum meter is direct integration. Substituting (6.6.3) and (6.6.8) in (6.5.42) and designating

$$\mathbf{x} = \mathbf{\hat{1}} - \mathbf{\hat{\lambda}}, \quad \mathbf{i} = \mathbf{\lambda} - \mathbf{\hat{\lambda}}, \quad (6.6.9)$$

under the sign of the integral performing transformation of variables $(\lambda \rightarrow \varepsilon)$ and rejecting factors not depending on ε , we have equation

$$\int \dots \int \mathbf{e}_{n} \exp \left\{ \mathbf{z}^{+} \mathbf{s} - \frac{1}{2} \mathbf{s}^{+} \mathbf{A} \mathbf{s} - \frac{1}{2} (\mathbf{s} + \mathbf{\chi})^{+} \mathbf{V} (\mathbf{s} + \mathbf{\chi}) \right\} d\mathbf{s}_{1} \dots d\mathbf{s}_{n} = 0.$$

After grouping terms, using symmetry of matrix V, we obtain from this

$$\int e_n \exp\left\{ [z - V\chi]^+ \varepsilon - \frac{1}{2} \varepsilon^+ (A - V) \varepsilon \right\} d\varepsilon =$$

$$= \frac{\partial}{\partial b_n} \int \exp\left\{ b^+ \varepsilon - \frac{1}{2} \varepsilon^+ (A - \varepsilon) \varepsilon \right\} d\varepsilon = 0$$

$$(d\varepsilon = d\varepsilon_1 \dots d\varepsilon_n), \qquad (6, 6, 10)$$

where there is designated

$$\mathbf{b} = \{b_1, \dots, b_n\} = \mathbf{z} - \mathbf{V} \mathbf{\chi}.$$
 (6.6.11)

Introducing matric $\mathbf{C} = \| \mathbf{C}_{ik} \| = [\mathbf{A} + \mathbf{V}]^{-1}$, the inverse of matrix $\mathbf{A} + \mathbf{V}$ in interval (t_0, t_n) , instead of (6.6.10), according to [8] we have equation

$$\frac{\partial}{\partial b_n} \exp\left\{-\frac{1}{2}\mathbf{b}^+\mathbf{C}\mathbf{b}\right\} = 0. \qquad (6.6.12)$$

Differentiation in (6.6.12) and substitution of the value of \mathbf{b} from (c.e.i.) then gives

$$\sum_{k=1}^{n} C_{nk} \left[z_k + \sum_{i=1}^{n} V_{ki} z_i \right] = 0.$$

From this, adding and subtracting from $V_{k,i}$ an element of equival, considering time $(\mathbf{A} + \mathbf{V})\mathbf{C} = \mathbf{I}$, where $\mathbf{I} =$ unit matrix, and spanding the value of χ , we finally to be

$$\hat{\boldsymbol{\lambda}}_{n} = \sum_{k=1}^{n} C_{nk} \Big[\boldsymbol{z}_{k} + \sum_{i=1}^{n} A_{ki} (\hat{\boldsymbol{\lambda}}_{i} - \overline{\boldsymbol{\lambda}}_{i}) \Big] + \overline{\boldsymbol{\lambda}}_{n}. \qquad (6.6.15)$$

Another method of finding of an optimum meter is based on direct study of the expression for the a posteriori probability of z(4). This method in this case is even more convenient, although in general it is not always applicable. After multiplication of the a priori distribution (6.6.8) by approximation of the likelihood function (6.6.3) we bring the logarithm of the formed function to form

$$\ln \tilde{P}(\lambda) = C_{o} - \frac{1}{2} (\lambda - \lambda_{o})^{+} C^{-1} (\lambda - \lambda_{o}), \qquad (0.0.4)$$

where scalar C_0 , vector $\boldsymbol{\lambda}_0$ and matrix \boldsymbol{C}^{-1} do not depend on the current argument of $\boldsymbol{\lambda}$. Determining coefficients for different powers of $\boldsymbol{\lambda}$, it is easy to prove that

$$\mathbf{C}^{-1} = (\mathbf{\Lambda} + \mathbf{V}), \qquad (1.0.11)$$
$$\mathbf{\lambda}_{\bullet} = \mathbf{C} [\mathbf{z} + \mathbf{A} (\mathbf{\hat{\lambda}} - \mathbf{\bar{\lambda}})] + \mathbf{\bar{\lambda}}, \qquad (6.0.10)$$

where \mathbf{C} -- matrix, the inverse of \mathbf{C}^{-1} .

At the same time from the very form of expression (6.6.14) it follows that λ has the sense of that point of symmetry (in multi-dimensional space) of the a costeriori distribution which we mentioned in § 6.5. Naturally, if should also correspond to the optimum estimate $\lambda_0 = \hat{\lambda}$. Then from relationship (6.6.16), considered at the last moment of observation t_0 , there again ensues expression t_0 , if , determining the optimum meter.

Relationship (6.6.12) relates the value of estimate $\hat{\lambda}_n$ at the last memory of observation with estimates $\hat{\lambda}_k$ is all the preceding moments and in estimate is a nonlinear equation for quantities $\hat{\lambda}_1, \ldots, \hat{\lambda}_n$. This nonlinearity is caused by the

--30**-**

fact that values of estimate λ enter the equation not only directly but also through the coefficient of expansion of the likelihood function $\mathbf{z} = \mathbf{z}(\hat{\lambda})$, $\mathbf{A} = \mathbf{A}(\hat{\lambda})$. If we are drawn away from this dependence, the optimum operator is linear with respect to vector $\mathbf{z} + \mathbf{A}(\hat{\lambda} - \overline{\lambda})$. The matrix of linear transformation **C**, according to (6.6.15), 260 is is equation

$$\mathbf{C}[\mathbf{I} + \mathbf{A}\mathbf{R}] = \mathbf{R} \tag{6.6.17}$$

or in expanded form

$$C_{nk} + \sum_{l,l=1}^{n} C_{nl} A_{ll} R_{lk} = R_{nk},$$
 (6.6.18)

where, according to the principle of using only the preceding data, it is assumed that $C_{nk} = 0$ when : > n.

Relationship (0.6.13) during discrete observation gives the algorithm for construction of a dynamic system, simulating the process of continuous solution or



(6.5.42) with use of former results of estimation and newly arriving data. All the prehistory of sampling y is depicted by estimates of the parameter made in earlier moments: for formation of new estimates there is no Such a property of operators

Fig. 6.12. Elementary intervals of observation: 1) envelope of realization of y(t).

need to turn again to former values of the sampling . Such a property of operators of processing are very convenient technically.

Certain simplification of expressions (6.6.13) and (6.6.18) can be produced if we consider that for slow changes of $\lambda(t)$ matrix **A** is diagonal, i.e., consists of elements $A_{ik} = K_i b_{ik}$. Actually, intervals of statistical coupling of all immaterial parameters of the input mixture y(t) are much smaller than the time of approximate constancy of the correlation interval of parameter $\lambda(t)$. As this is illustrated in Fig. 6.12, every moment of observation t_i corresponds to its own frozen value of the parameter $\lambda(t_i)$. First differentiation of the logarithm of the likelihood funct? With respect to λ_i actually is produced only on the subinterval near t_i . Second differentiation with respect to t_j , which will form matrix **A**, gives a result which differs from zero only when indices i = j. If we allow for this

-81-

circumstance, expression (0.0.13) and (0.0.18) will be rewritten in the Form

$$\hat{\boldsymbol{\lambda}}_{n} = \sum_{k=1}^{n} C_{nk} [\boldsymbol{z}_{k} + K_{k} (\hat{\boldsymbol{\lambda}}_{k} - \overline{\boldsymbol{\lambda}}_{k})] + \overline{\boldsymbol{\lambda}}_{n}, \qquad (1, 0, 1; 0)$$

$$C_{nk} + \sum_{i=1}^{n} C_{ni} K_i R_{ik} = R_{nk}. \qquad (\dots, \dots)$$

In general, as follows from analysis analogous to that given in [3], the diagonal nature of matrix **A** simultaneously sears witheds to the cost that diagonal nature of this matrix with a certain approximation can be replaced by their ownean value $(K_i \approx \overline{K}_i)$. However, below we shall show that this compared what decrements of measurement, so that such far-reaching simplification is inexpected.

By direct substitution we prove simply that (6.6.12) can be given a simpler form

$$\hat{\lambda}_n = \sum_{k=1}^n G_{nk} z_k + \overline{\lambda}_n, \qquad (6.6.21)$$

where triangular matrix \bm{G}_{nk} = 0 when n < k) is determined by equation

$$C_{nk} + \sum_{i=k}^{n} C_{ni} K_i G_{ik} = G_{nk}.$$
 (6.6.22)

We shall establish the sense of quantity z_k . For this we represent this magnitude by a Taylor expansion in the visitive point $\lambda_{(3)}$ coinciding with the free value of the parameter of the input value. It is obvious that

$$\boldsymbol{z}_{k} = \frac{\partial L\left(\boldsymbol{\lambda}\right)}{\partial \lambda_{k}} = \frac{\partial L\left(\boldsymbol{\lambda}_{n}\right)}{\partial \lambda_{k}} + \sum_{i=1}^{n} \frac{\partial^{2} L\left(\boldsymbol{\lambda}_{n}\right)}{\partial \lambda_{k} \partial \lambda_{i}} \left(\hat{\boldsymbol{\lambda}}_{i} - \boldsymbol{\lambda}_{n,i}\right) + \dots =$$

$$\frac{\partial L\left(\boldsymbol{\lambda}_{n}\right)}{\partial \lambda_{k}} - \sum_{i=1}^{n} A_{\mathbf{u}ki} \left(\hat{\boldsymbol{\lambda}}_{i} - \boldsymbol{\lambda}_{n,i}\right) = \boldsymbol{\xi}_{k} + K_{\mathbf{u}k} \left(\boldsymbol{\lambda}_{\mathbf{u}k} - \hat{\boldsymbol{\lambda}}_{k}\right), \qquad (0.5.25)$$

where matrix $\mathbf{A}_{\mu} = \left\| -\frac{\partial^2 L_{\mu} \left(\cdot \cdot \cdot \cdot \cdot \right)}{\partial \lambda_{\mathbf{i}} \partial \lambda_{\mathbf{k}}} \right\|$ and vector \mathbf{K}_{μ} differ from \mathbf{A}_{μ} and \mathbf{K}_{μ} only by the fact that values of their elements, we functions of the parameter, are taken, not at point $\hat{\mathbf{\lambda}}_{\mu}$ but at a point coinciding with the true value of the parameter λ_{μ} . Inasmuch as $\hat{\mathbf{\lambda}}$ and $\mathbf{\lambda}$ are assumed close, properties of \mathbf{A}_{μ} and \mathbf{K}_{μ} , on the one hand, and of \mathbf{A} and \mathbf{K} on the other hand, practically coincide. Quantities $\hat{\mathbf{k}}_{\mu}$ in formula (0.6.23) do not depend on values of the estimate, and since, according to definition,

-62-

$$\mathbf{\xi}_{\mathbf{A}} = \frac{\partial L(\mathbf{\lambda}_{\mathbf{n}})}{\partial \mathbf{\lambda}_{\mathbf{A}}} = \frac{\partial \ln P(\mathbf{y} \mid \mathbf{\lambda}_{\mathbf{n}})}{\partial \mathbf{\lambda}_{\mathbf{A}}}, \qquad (6.6.24)$$

it is easy to prove that ξ_k have zero (with respect to the ensemble of input signal y) mean values. In other words, discrete random process $\xi_k(k = 1, ..., n)$, taking even for exact coincidence of the estimate with the true value λ_M is a parely fluctuating disturbance.

The second component of z_k , presented in the form of (6.6.23), turns out to be proportional to current mismatch between the true and estimated values of the parameter. Proportionality factor K_k is a random variable with a certain positive mean value. Thus, according to (6.6.23) z_k is the measure of mismatch between the true and estimated values of the parameter. For small $\lambda_{||||||k|} - \hat{\lambda}_k$ this measure on average is linear. In general substantial too are terms with higher powers of $\lambda_{|||||k|} - \hat{\lambda}_k$, so that z_k can be presented by a Taylor expansion in powers of $\lambda_{|||||k|} - \hat{\lambda}_k$ with random coefficients depending only on the realization of the input signal. The described properties of z_k qualitatively coincide with properties of output signals of devices known in practice for detection of the error signal - discriminators (see § 6.2).

Let us turn now to clarification of the sense of quantity K_k , which is the current value of the second derivative of the logarithm of the likelihood function. If parameter λ is to be fixed, the mean value of this derivative, taken at the point of true value $\lambda_{\rm M}$, will completely characterize accuracy of measurement of the parameter. As already indicated, due to the proximity of $\lambda_{\rm M}$ and $\hat{\lambda}$ values of the second derivative at two these points practically coincide. Then it may be concluded that the value of $K_{\rm k}$ characterizes the current accuracy of the relative measurement of the parameter in the k-th moment of observation.

From the theory of estimation, and also the method of least squares we know that for unequal errors of separate measurements results of these measurements chould be used for calculation of the final result with weighting factors, inversely proportional to dispersions of separate independent measurements, considered known. Our analysis showed that allowance for unequal accuracy should also be made in the considered case of an optimum meter, where the measure of unequal accuracy, inversely proportional to dispersion, here is the current value of K_1 , formed in the actual circuit of the meter. Obtaining relationships (6,6,19) and (6,0,21) and howerthet needs the set $r_{\rm K}$ and ${\rm K}_{\rm K}$, it would be possible to pass to interpretation of the relation block diagrams realizing an optimum filter-moter. It setters this based on the form of block diagrams realizing an optimum filter-moter, it setters this based on the routly conveniently to do when we have disperte observation, headan as the setter measure as the setter of the setter of the setter measure as the setter measure as the setter of the se

With the help of passage to the limit $t_1 - t_{1-1} = \Delta \rightarrow 0$, $n \rightarrow c$ (1 - 1, 5, ... n; $n\Delta = t - t_0$) we obtain continuous manual of formulas (0.6.19)-(0.0.24). Designating $t_n = t$, $t_k = t$, we note that according to (0.6.20)-(0.0.22) the contraction of the tendency to a limit of matrix elements of **C** and **G** is the same as for elements of of matrix **R**, for which $R_{nk} = R(t_n, t_k) = P(t, \tau)$. Therefore, with passage to the limit one should introduce functions

$$c(t, v) = c(t_n, t_k) = C_{nk_1} g(t, v) = g(l_n, t_k) = G_{nk_2}$$
(6.0.25)

Then from formulas (6.6.19) and (6.6.25) it foll is that for the existence of limits it is necessary that z_k seek zero, when $A \rightarrow 0$, as Δ , and matrix element A performance as Δ^2 . This requirement will be satisfied if z_k and A_{1k} can be presented in the form

 $\dim_{\mathcal{O}}(t)$ and A(t, t) - certain random functions of time.

The form of these functions will be clarified in examples subsequently; where, if it useful to emphasize that relationship (6.6.26) determines a whole class of statistically equivalent, but generally not equal functions, which give in integration an identical result z_k . Thus, the continuous analog of z_k is determined ambiguously: any statistically equivalent functions will do.

In the same conditions in which matrix **A** is diagonal, i.e., $A_{ik} = K_i \delta_{ik}$, one should consider

$$\mathbf{A}(t, \tau) = K(t)\delta(t-\tau), \qquad (6.0.2c)$$

where K(t) - limit of quantity K_k/Δ as $\Delta \to 0$, and a δ -function will be formed at the limit from δ_{ik}/Δ . Then expressions (6.6.19)-(6.6.22) reduce to form:

$$\hat{\lambda}(t) = \int_{0}^{t} c(t, \tau) [z(\tau) + K(\tau)(\hat{\lambda}(\tau) - \overline{\lambda}(t))] d\tau + \overline{\lambda}(t), \qquad (6.6.29)$$

$$\hat{\boldsymbol{\lambda}}(l) = \int_{0}^{l} \boldsymbol{g}(l, \tau) \boldsymbol{z}(\tau) d\tau + \overline{\boldsymbol{\lambda}(l)}, \qquad (6.6.30)$$

$$c(t, \tau) + \int c(t, s) K(s) R(s, \tau) ds = R(t, \tau),$$
 (6.6.31)

$$c(t, \tau) + \int_{\tau}^{t} c(t, s) K(s) g(s, \tau) ds = g(t, \tau).$$
 (6.6.32)

Relationship (6.6.29) can be presented by the block-diagram of Fig. 6.13. Input signal y(t) is fed to nonlinear elements 1 and 2. They form, correspondingly, quantities z(t) and K(t). Quantity K(t) directly determines the gain of inertialess amplifier K and, furthermore, as control is fed to filter C. Quantity z(t) is added to output voltage of amplifier K and is fed to inertial filter C, whose pulse response $c(t, \tau)$ depends on the magnitude of K(t).

After addition to the a priori mean value of $\overline{\lambda(t)}$ there will be formed an estimate of the parameter, $\hat{\lambda}(t)$. It is used to check characteristics of elements 1, 2. One more loop of intercoupling closes output of filter **C** through amplifier **K**.

The described interpretation circuit is a complicated self-tuning system, containing tunable nonlinear elements 1, 2 and linear filters C and K, pulse



Fig. 6.13. Two-loop variant of the circuit of an optimum tracking meter: 1) discriminator; 2) accuracy unit; C = 1 inear rilter with pulse response $c(t, \tau)$; K = 1 inertialess amplifier with gain factor K.

responses of which change with change of input signal y(t), i.e., are determined by random factors.

Let us give names to nonlinear elements 1 and 2. Making for z(t) an expansion an analogous to (6.6.23), we obtain

$$\mathbf{z}(t) \approx \mathbf{\xi}(t) + K(t) (\lambda_{\mathbf{u}}(t) - \hat{\lambda}(t)),$$

(6.6.33)

where the first component $\xi(t)$ — white noise, intensity of which does not depend on the mismatch $\lambda_{\rm H}$ (t) - $\hat{\lambda}(t)$ between the the and measures values of the parameter, and the second component is preparated to mismatch with a variable proportionality factor K(t). These properties of the qualitatively coincide with properties of the discriminator output right. These property element 1 is rationally called the optimum discriminator. Common for provide known and optimum discriminators is a similar character of discrimination of fluctuation characteristics. The discrimination characteristic of element 1, we will follow from examples of Chapters VII-XI, has a linear section and at includin seeks zero.

Quantity K(t) according to (0.6.53) is the variable random gala in terms of all optimum discriminator, determined near zero mission of $a_{i,j}$ of $a_{i,j}$ and $a_{i,j}$ are value in encompletely (t) is the gain factor of the discriminator in the usual meaning.



Fig. 6.14. Single-loop variant of the circuit of an optimum tracking meter: 1) discriminator; 2) accuracy unit; G - linear filter with pulse response. element 2 to beleef a mane is somewhat normalifically inasmuch as an analogous element in practical meters is usually analogous elemently, noove we indicated that K_k , and consequently, the analogous function K(t) characterized widthickness the current accuracy of measurement. Therefore, we shall call element 2 the accuracy unit.

The relationship between the two interpretations of K(t) is obvious, intermedius decrease of current accuracy occur, namely due to lowering of the gain factor of the discriminator. In § 6.2 fluctuations of the gain factor of the discrimination were called parametric fluctuations, and it also was shown that they cause increase of error of measurement in practical circuits, in which there are not taken measured to increase their influence. Controlling of smoothing filters in optimum where it is increase to lower the influence of parametric fluctuations.

We pass now to interpreting relationship (6.6.30) in the form of a block alagraph. We pass now to interpreting relationship (6.6.30) in the form of a block alagraph. We had the obtained variant of an optimum meter constitutes a secondarily tership to with looss of feedback of aircov control (Fig. 6.14). Satisfielements of this system again are optimum discriminator based a single linear filter **G**, careging out smoothing.

Important for work of the system in principle, as also in the variant of Fig. 0.13, is feedback from its output to the discriminator, necessary to change discriminator tuning. In the course of this change at every moment of time the

-0f,-

zero of the discrimination characteristic, having a limited zone of linearity, is tuned approximately to the true value of $\lambda(t)$, coded in the input mixture. Thereby, the system realizes tracking of the realization of process $\lambda(t)$. The structure of linear filter G is determined by equations (6.6.31), (6.6.32), according to which it mississes variable randomly changing parameters. By the correlation function its merties depend on the a priori statistics of change of $\lambda(t)$, and through K(t) on the statistics and realization of input mixture y(t). Forming of K(t) again is carried out by accuracy unit 2, in turn controlled by output data of the system.

Meters close in structure already are used in practice. The circumstance that the optimum solution in any conditions is in principle close to the one practically utilized is very remarkable. Theoretical solution permits us to optimally select the characteristics of separate elements of the meter, in particular to find optimum operations of the discriminator. The need for an accuracy unit and special control of smoothing circuits is a new finding, ensuing from theory.

Let us indicate that the circuits of Figs. 6.13 and 6.14 are absolutely equivalent. Inasmuch as in the first of them there exist two basic coupling loops, closing to the discriminator and the accuracy unit and separately to the adaer, standing after the discriminator, and in the second there is a basic loop, we conventionally call the circuit of Fig. 6.13 two-loop, and the circuit of Fig. 6.14 - single-loop variants of an optimum meter.

Further concretization of the solution of the problem of optimum measurement for different statistical properties of input mixture consists of finding operations of formation of quantities z(t) and K(t), and instrumental realization of these operations. Furthermore, it is necessary to establish the algorithm of smoothing: in the linear filters for different forms of correlation function of the parameter and different forms of K(t).

For convenience of subsequently finding operations of the discriminator and its characteristics we shall discuss in somewhat greater detail quantities z(t) and K(t). It is obvious that in general likelihood function $P(\mathbf{y}|\mathbf{\lambda})$ can be expressed through the product of conditional probabilities in the form

$$P(\mathbf{y}|\boldsymbol{\lambda}) = \prod_{k=1}^{n} P_{k}(y_{k}|y_{k-1}, \dots, y_{1}, \boldsymbol{\lambda}_{k}, \dots, \boldsymbol{\lambda}_{1}) =$$
$$= \exp\left\{\sum_{k=1}^{n} l_{k}(y_{k}, \dots, y_{1}; \boldsymbol{\lambda}_{k}, \dots, \boldsymbol{\lambda}_{1})\right\}, \qquad (6.6.34)$$

おうと、「「「「「」」」」

-87-
where $\mathbb{P}_{\mathbb{R}}(\mathbf{y}_{k}|\mathbf{y}_{k-1}, \dots, \mathbf{y}_{1}, \mathbf{x}_{k}, \dots, \mathbf{x}_{1}) = \text{density of the conditional distribution }$ \mathbf{y}_{k} for assigned $\mathbf{y}_{k-1}, \dots, \mathbf{y}_{1}, \mathbf{x}_{k}, \dots, \mathbf{z}_{1}, \text{ and } \mathbf{z}_{k} = \ln \mathbf{\hat{v}}_{k}.$

Then, if there exists a tikelihood functional, according to (v,v,v_2) if v,v_3

$$P(y(t)|\lambda(t)) = \exp\left\{\int_{a}^{b} l(\tau, \lambda(s), y(s)) d\tau\right\}, \qquad (1.11)$$

and a static field in the second static state and a second in the second

וריה מו וללא הישמולים בישורים ביאלאנות אבור בעובל אים אלולאנון הישורים בישורים בישורים בישורים ובישורים לי בי

where $z(\tau, z(s), y(s)) = \text{certain functional of realizations and the first of the last <math>z(\tau, z(s), y(s)) = z(\tau, \tau)$.

In a particular case $t(\tau, \lambda(s), y(s))$ consider the a function of $y(\tau)$, $\lambda(\tau)$ at time τ . As example can be the functional of probability density of τ or s' of τ could be deckground of white noise (0.6.1), which in § 0.7 we should also us in greater detail.

Let us assume that the integral in formula (6.6.35) can be presented by the the second to be not integrals over intervals of duration 4. Here, normalize the the value of parameter $\lambda(t)$ practically uses not change, and segments of restlications of y(t) in neighboring intervals, if we exclude fringe effects, are not related to the value of $\chi(t)$ only in this interval. Then comparison of relationships (t,r,f_{1}) and (6.6.26) shows that output voltage of the discriminator is statistically for equivalent to

$$z(t) = \frac{\partial}{\partial \lambda} l(t, \lambda(s), y(s)) \Big|_{\lambda(s) = \hat{\lambda}(t)}, \qquad (v \in \mathbb{R}, \mathbb{R})$$

where the functional in $l(t, \lambda(s), y(z))$ is formed for a frozen value of farmater s(t), equal to the value of the estimate at time t, and the realization of y(z) is taken then s $\in (t_0, t)$.

A scatistically equivalent function is also obtained when, e.g., we extend to the fourtional dependence $l(t, \lambda(s), y(s))$ from realization y(s) to domain $\in \mathbf{C}(t, +\alpha)$, into oth as operator t possesses properties filtering with respect to moment s = t, i.e., acts in an interval considerably scatter than the time of sacked operators of $\lambda(t)$. The physical realizability of such operators can be achieved by means of insertion of a delay by the magnitude of effective action of the operator.

Analogously to (6.6.36) we determine the operation of the gap imply with

$$K(t) = -\frac{\partial^2}{\partial \lambda^2} l(\tau, \lambda(s), y(s)) \Big|_{\lambda(s) = \hat{\lambda}(t)}, \qquad (t, t', t')$$

by which we easily calculate the average gain factor $\overline{K(t)}$.

We turn now to characteristics of the resultant accuracy of measurement. First we shall study the case of discrete observation. It is obvious that any statistical moment of differences $\lambda = \hat{\lambda}$ is determined by relationship

$$\frac{(\lambda_n - \hat{\lambda}_n)^m (\lambda_k - \hat{\lambda}_k)^i}{(\lambda_n - \hat{\lambda}_n)^m (\lambda_k - \hat{\lambda}_k)^i P(\mathbf{y} | \lambda) P_{\mathbf{o}}(\lambda) d\lambda d\mathbf{y} = \\
= \int \left[\int (\lambda_n - \hat{\lambda}_n)^m (\lambda_k - \hat{\lambda}_k)^i \tilde{P}(\lambda) d\lambda \right] P(\mathbf{y}) d\mathbf{y}, \quad (6.6.38)$$

where $\widetilde{\mathbb{P}}(\lambda) \equiv \mathbb{P}(\lambda|\mathbf{y})$ is again the a posteriori distribution of λ . Substituting in (6.6.33) the a posteriori distribution

$$P(\lambda) = (2\pi)^{-n/2} \left[\det (V + A)^{-1/2} \right] \times \\ \times \exp \left\{ -\frac{1}{2} (\lambda - \hat{\lambda})^{+} (V + A) (\lambda - \hat{\lambda}) \right\},$$
(6.6.39)

. .

which is obtained by multiplication of $P(\mathbf{y}|\mathbf{\lambda})$ by $P_0(\mathbf{\lambda})$ and introduction of a momentum factor, we arrive at integrals with multi-dimensional Gaussian functions, eachdable in parts or with the help of a multi-dimensional characteristic function.

In particular, as it is easy to prove,

$$\overline{\lambda}_{n} = \overline{\lambda}_{n}, \quad \overline{\lambda}_{n}^{2m+1} = \overline{\lambda}_{n}^{2m+1}.$$
 (6.6.40)

This shows the unbiased nature of the estimate, which is a not unexpected eircumstance (see § 6.5). The correlation function of error of measurement is equal to

$$R_{nh \text{ BMX}} = \overline{\mathfrak{e}_{n}\mathfrak{e}_{h}} = \int_{-\infty}^{+\infty} \left[\int_{-\infty}^{+\infty} \mathfrak{e}_{n}\mathfrak{e}_{h} (2\pi)^{-n/2} (\det [V + A])^{-n/2} \times \exp\left\{-\frac{1}{2}\mathfrak{e}^{+} (V + A)\mathfrak{e}\right\} d\mathfrak{e} \right] \mathcal{P}(\mathbf{y}) d\mathbf{y} =$$

$$= \int_{-\infty}^{+\infty} [V_{nh} + A_{nh}]^{-1} P(\mathbf{y}) d\mathbf{y} =$$

$$= \int_{-\infty}^{+\infty} C_{nh}(\mathbf{y}) P(\mathbf{y}) d\mathbf{y} = \overline{C_{nh}(\mathbf{y})} \quad (n > k). \quad (6.6.41)$$

Variance of error of measurement is equal to

$$\mathbf{s}_{nut,n}^{2} = \overline{\mathbf{s}_{n}^{2}} = \overline{C_{nn}(\mathbf{y})}. \tag{6.6.42}$$

Thus, error is expressed through the value of $\sigma_{\rm nn}$, averaged over the ensets of input signals.

Formulas for the case of continuous observation are obtained by means of points, to the limit from (6, 6, 41) - (6, 6, 42):

 $R_{BMX}(t, \tau) = \overline{c(t, \tau)} \quad (t > \tau), \quad \sigma_{BMX}^2(t) = \overline{c(t, t)} = \overline{g(t, t)}. \quad (0, \dots, t_{i})$

6.6.3. The Optimum Tracking Metar is a cuber of Specific Discrete Class

In this paragraph we shall consider specific cases, when from " on theory" formulas of through 0.6.2 it is expected or powenies to again base to the traditionate analogs. Let us emphasize that as before we are considering continuous observation of mixture y(t).

First of all we shall study measurement of parameters of an incoherent parameters of an incoherent parameters of an incoherent parameter of a set of the presence of duration and form of intrapulse modulation, where in the case of the presence of fluctuations of the signal the values of amplitude and phase of pulses in departed, periods will be considered independent. With an incoherent signal for use of the given methods complete independence is not required. It is important only that during the time of action of the pulse the measured parameter does not vary. Heavier, shows of f(t) during the time of action of the pulse the measured parameter does not vary.

The specific character of this case consists in the fact that even after transition from sampling to continuous realization of y(t) the likelihood functional can be recorded in a form discrete relative to the parameter, somewhat differing from $(t, 0, \frac{1}{2}t)$:

$$P(y(l)|\lambda(l)) = \exp\left\{\sum_{k=1}^{n} \int_{(k-1)T_{\tau}}^{kT_{\tau}} l_{k}(\tau, \lambda_{k}, y(s)) d\tau\right\}, \qquad (1, 1, \dots, n)$$

Ì.

Wher $t_{k}(x, \lambda_{k}, y(z)) \rightarrow function$ (in general a functional) of $y \in I$, in many respects analogous to function $t(x, \lambda(z), y(z))$ from (6.6.44) and practically differing from zero only within limits of pulse duration.

The logaritism of function (6.6.44) can be expanded in powers of $\lambda_{\mathbf{k}} = \lambda_{\mathbf{k}}$ analogously to Paragraph 6.6.1. After multiplication by Gaussian a priori distribution (6.6.8) we formally arrive at the same operator of an optimum filter

-90-

(6.6.19) as during discrete observation, but by z_k and K_k one should understand (actually within limits of the pulse) the quantities:

()

$$z_{h} = \int_{(h-1)T_{r}}^{hT_{r}} \frac{\partial}{\partial \lambda} l_{h}(\tau, \hat{\lambda}_{h}, y(s)) d\tau, \qquad (6.6.45)$$

$$K_{h} = -\int_{(h-1)T_{r}}^{hT_{r}} \frac{\partial^{n}}{\partial \lambda^{1}} l_{h}(\tau, \hat{\lambda}_{h}, y(s)) d\tau. \qquad (6.6.46)$$

An optimum meter produces two types of operations — intraperiod (in discriminator and accuracy unit) and interperiod (in the discrete smoothing filter). Transition during smoothing to continuous quantities is formally and technically permissible only when during interval T_r the parameter changes insignificantly. In the general case the smoothing filter remains discrete and its pulse response C_{nk} (or G_{nk}) is determined by matrix equations (6.6.20) or (6.6.22). In distinction from the smoothing filter operations of the discriminator and accuracy unit within each pulse are always continuous. If we allow for the shown peculiarities, then in everything else the case of continuous observation of a pulse signal is formally similar to the case of discrete observation.

We shall discuss one case, when transition to discrete smoothing filters is already permissible from considerations of purely technical convenience. Receiver, more and more they realize smoothing circuits with the help of digital computers, possibilities of which are extraordinarily broad. Especially evident are advantaged of digital computer technology during designing of complicated radar complexes, where the problem of measurement is combined with the problem of further processing and transmission of information. If, in these conditions, outside of dependence on the structure of the carrier-signal of the parameter, we are to quantize in time the data proceeding from discriminators where we select an interval of quantization smaller than the time of noticeable measurement [sic ? change] of the parameter, the smoothing circuits can be synthesized with the help of matrix equations (0.0.20), (6.6.21) and are realized on that digital computer which can execute the set of operations of smoothing with the required speed.

In the particular case when the signal is pulsed, and of the type considered above, and when in a period the measured quantities practically do not change, to decrease the number of operations of smoothing it is permissible to divide smoothing.

-91-

into two stages. In the first there occurs grouping of data in intervals comparable to the time of change of the parameter, and in the second there occurs discrete smoothing by a filter synthesized according to the indicated matrix equations.

In practice we may encounter other interesting cases of close intertwining of questions of continuous (analog) and discrete (digital) processing of signals, the description of which is beyond the framework of the present section.

6.6.4. Possibility of One Simplification of the Meter Circuit

We shall give certain considerations, allowing us to produce an important simplification of circuits of the found optimum meters. Let us consider the case of continuous observation. With rapid changes of all immaterial parameters y(t) as compared to $\lambda(t)$ function K(t) is a rapidly changing function as compared to the correlation function of the parameter. Presenting the integral in (6.6.31) in the form

 $\int c(t, s) K(s) R(s, \tau) ds \approx$ $\approx \sum_{i=1}^{t_{i+1}} K(s) ds c(t, t_i) R(t_i, \tau),$

where duration of intervals $\Delta = t_{i+1} - t_i$ is much less than the time of correlation of R(t, τ), and assuming ergodicity of the random part of function K(t), thanks to which

$$\int_{t_i}^{t_i+\Delta} K(s) \, ds \approx \overline{K(t_i)} \, \Delta,$$

we obtain

$$\int c(t, s) K(s) R(s, \tau) ds = \int c(t, s) \overline{K(s)} R(s, \tau) ds.$$

Here $\overline{K(t)}$ is the mean value of K(t) in the ensemble of input signals y(t). Here, that is to smallness of error of measurement $\lambda(t) - \hat{\lambda}(t) = \varepsilon(t)$ the value of $\overline{K(t)}$ can be calculated, not at point $\lambda = \hat{\lambda}$, but at a point equal to the true value of the measured parameter, so that

-92-

$$\overline{K(l)} = -\frac{\partial^2 l(l, \lambda(l), y(s))}{\partial \lambda^2}.$$
(6.6.47)

Thus, function K(t), expressed through the limit of the matrix of second derivatives of the likelihood functional with respect to values of the measured parameter, in certain cases can be replaced by the a priori known mean value of this function. Equations (6.6...) and (6.6.32), determining characteristics of optimum filters, mere do not change their form, but instead of random function K(t) in them where now appears the known function of time $\overline{K(t)}$. The actual linear filters of block diagrams in Figs. 6.13 and 6.14 become here filters with nonrandom characteristics. The presence of an accuracy unit and circuits for controlling the smoothing filters becomes unnecessary, and elements of coupling, shown in Figs. 6.13 and 6.14 by the dotted line, disappear. In particular, the circuit of Fig. 6.14 takes the well-known form of a single-loop servo system. It is natural that, inasmuch as $\overline{K(t)}$ does not depend on the current value of y(t) in that same approximation in which it is possible to disregard fluctuations of K(t) during synthesis, variance and the correlation function of measurement error will be expressed in the form

$$\sigma_{aux}^{2}(t) = c(t, t) = g(t, t), \qquad (6.6.48)$$

$$R_{aux}(t, \tau) = c(t, \tau). \qquad (6.6.49)$$

From material of Paragraph 6.6.2 it is clear that the meaning of the replacement. of K(t) and $\overline{K(t)}$ consists in rejection of allowance for unequal accuracy of separate measurements or, which is the same, of allowance for parametric fluctuations. In Paragraph 6.8.6 we shall give an example showing that this refusal leads to certain increase of error of measurement, but, as it was shown in § 6.2 for practical circuits, this increase frequently is small and in a number of cases can be ignored. We note, only, that the introduced simplification deprives the optimum circuit of that flexibility of a system with self-tuning which is manifested upon change of the input signal level. When the average level of the signal changes as compared to design level the previously optimum circuits always become nonoptimal. However, in circuits with adjustment of smoothing circuits there occurs change of the gain factor in the feedback circuit (with respect to the accuracy unit) of the same nature which should occur during change of the design characteristic of the input quantity. At the same time in the absence of adjustment there remain no means for preservation of optimality in the described conditions, and impairment of the quality of measurement may be considerable.

-93-

In physical meaning, as follows from (6.6.33) and (6.6.47), $\overline{K(t)}$ is the (averaged) gain factor of the discriminator. On the other hand, considering the discrete case; analogously to [8] one can prove that

$$\overline{K}_{a}\delta_{ik} = -\frac{\partial^{2}\ln P(y|\lambda)}{\partial \lambda_{i}\partial \lambda_{k}} = \frac{\partial \ln P(y|\lambda)}{\partial \lambda_{i}} = \overline{\xi_{i}\xi_{k}}. \quad (6.6.50)$$

An analog of relationship (6.6.50) in the continuous case will be

$$\overline{K(t)}\delta(t-\tau) = \overline{\xi(t)}\xi(\tau), \qquad (6.6.51)$$

where $\xi(t)$ - fluctuating component in the discriminator output, taking place also with zero mismatch.

According to (6.6.51) $\overline{K(t)}$ is also the spectral density of the fluctuating component $\xi(t)$ at the discriminator output. If we translate this component into the equivalent value of the measured quantity by division by $\overline{K(t)}$ similarly to how this was done in § 6.2, the spectral density of the equivalent noise component $\eta(t) =$ = $\xi(t)/\overline{K(t)}$ will be quantity $1/\overline{K(t)}$:

$$\overline{\eta(l)} \overline{\eta(\tau)} = \frac{1}{\overline{K(l)}} \delta(l-\tau).$$
 (6.6.52)

Thus, $1/\overline{K(t)}$ is the variable spectral density of equivalent noise at the discriminator input. This interpretation of $\overline{K(t)}$ is exceedingly important, since multiplication of output voltage of an optimum discriminator by an arbitrary constant, immaterial from the point of view of producible operations, changes the gain factor and spectral density of output noise, but leaves constant the equivalent spectral density $1/\overline{K(t)}$. In the frames of these assumptions function $\overline{K(t)}$ (or $\overline{K_k}$) is the unique characteristic of performance of an optimum discriminator. Resultant errors of measurement always depend on this function monotonically: the larger $\overline{K(t)}$ (or $\overline{K_k}$), the less the error.

If statistical characteristics of immaterial parameters y(t) do not depend on time $\overline{K(t)}$ is a constant $[\overline{K(t)} = K]$. Then from (6.6.31)-(6.6.32) we have the following equations, determining pulse responses of filters $c(t, \tau)$ and $g(t, \tau)$:

$$c(t, \tau) + K \int_{t_0}^{t_0} c(t, s) R(s, \tau) ds = R(t, \tau), \qquad (6.6.53)$$

$$c(t, \tau) + K \int_{t_0}^{t_0} c(t, s) g(s, \tau) ds = g(t, \tau). \qquad (6.6.54)$$

-94-

Equation (6.6.53), considered the equation for function Kc(t, τ), coincides with the equation of an optimum Wiener filter, detecting a signal with correlation function R(t, τ) from its additive mixture with white noise with spectral density 1/K. This is understandable, inasmuch as for all ideal assumptions work of an



Fig. 6.15. Two-loop variant of an equivalent linear system. KC — linear filter with pulse response Kc(t, τ).



Fig. 6.16. Single-loop variant of an equivalent linear system. KG — linear filter with pulse response $Kg(t, \tau)$.

optimum meter in linear conditions is equivalent to work of a linear servo system, presented in any of the variants of Figs. 6.15 and 6.16, to which as input disturbance there proceeds the sum of a "signal" $\lambda(t) - \overline{\lambda(t)}$ and "noise" $\eta(t)$. Due to this all solutions of equation (6.6.53), determining the structure of smoothing circuits, simulta-

neously are solutions of the corresponding problems of optimum filtration.

6.6.5. Optimum Nontracking Meter

The considered type of optimum meters is not the only possible one in the framework of our assumptions. Another circuit of a meter, also realizing potential possibilities of measurement, can be obtained for Gaussian statistics of the parameter on the basis of approximation of the likelihood function by relationship (6.6.4). Cross multiplying (6.6.4) and (6.6.8) and producing transformations of the logarithm of a posteriori probability, analogous to those shown in Paragraph 6.6.2, we arrive during discrete observation at the relationship

$$\hat{\mathbf{\lambda}}_{n} = \sum_{k, j=1}^{n} C_{nk} \check{\mathbf{A}}_{kj} (\check{\mathbf{\lambda}}_{j} - \bar{\mathbf{\lambda}}_{j}) + \bar{\mathbf{\lambda}}_{n}, \qquad (6.6.55)$$

where $\mathbf{C} = [\mathbf{A} + \mathbf{V}]^{-1}$ — a matrix, determined in the current interval of observation, and matrix \mathbf{A} is determined according to (6.6.5) and with a high degree of accuracy coincides with matrices \mathbf{A} and \mathbf{A}_{\dots} .

In view of the diagonalness of matrix A relationship (6.6.55) takes the form

$$\hat{\lambda}_{n} = \sum_{k=1}^{n} C_{nk} K_{k} (\check{\lambda}_{k} - \bar{\lambda}_{k}) + \bar{\lambda}_{n}. \qquad (6.6.56)$$

If we examine the case of continuous observation, (6.6.56) passes into integral relationship

 $\tilde{\mathbf{z}}(t) = \int c(t, s) K(s) [\mathbf{z}(s) - \overline{\mathbf{z}(s)}] ds + \overline{\mathbf{z}(t)}. \quad (6.6.57)$

Relationships (6.6.56) and (6.6.57), giving a final solution to problems of meter synthesis, are basic, and their operations are illustrated by the Circuit in Fig. 6.17. Element 1, to which there proceeds realization y(t), is a nonlinear unit, constantly separating the value of the point of maximum likelihood $\hat{\lambda}(t)$ and function K(t), determined through the second derivative of the logarithm of the



Fig. 6.17. Nontracking variant of an optimum meta: 1) estimator unit; CK - linear filter with pulse response $c(t, \tau) K(\tau)$.

likelihood function, taken at this point. Then in the circuit there follows a subtractor and linear filter CK with pulse response c(t, s)K(s), to which as the object of filtration there proceeds the difference of the maximumlikelihood and mean value $\lambda(t) - \overline{\lambda(t)}$,

South Office & Office & States & States

and as an adjustment, function K(t), characterizing the current level of accuracy of measurement. An adder for reverse input of the mean value completes the circuit of the optimum meter.

In distinction from the circuit described in Paragraph 6.6.2, the new circuit of an optimum meter is not closed. The circuit similarity of the two variants consists in the following two factors. First, in both circuits in the first stage there are used low-inertia, nonlinear operations of processing input radio signals. In the tracking variant they are carried out by a discriminator and an accuracy unit; in the nontracking variant these units are replaced by a unit which it is possible to conditionally call an estimator unit.

In both circuits there exist linear smoothing filters; in the tracking variant they close the tracking loop, and here they are constructed on the basis of an open directly. The internal unity of variants becomes still clearer if we consider that the pulse response of the open-loop smoothing filter is the same as for a linear system equivalent to the closed loop of the tracking variant of a meter, if the latter works with small errors of reproduction of the parameter. This is especially easy to prove when we disregard the variable random part of K(t) (analogously to parameter 6.6.3) and have for a constant mean value of K(t):

 $K(l) \approx \overline{K(l)} \equiv K.$

Then the pulse response of the main part of the low-frequency filter of the circuit in Fig. 6.17 is determined by equation

 $c(t, \tau) + K \int_{t}^{t} c(t, s) R(s, \tau) ds = R(t, \tau),$

 \oplus \oplus \oplus \oplus \oplus \oplus (6, 6, 53).

By a method, analogous to that given in Paragraph 6.6.2, it is easy to show that the characteristics of accuracy of the new variant of meter are expressed by formulas

 $\overline{\hat{\lambda}(t)} = \overline{\lambda(t)}, \quad \sigma_{sMX}^2(t) = \overline{c(t, t)} = \overline{g(t, t)}.$

This nontrivial result shows that, in spite of differences in approximation of the likelihood function, in general determining different errors of measurements, both the considered approximations lead to identical errors. They give measuring systems which, for a low level of noises, are equally close to a truly optimum system, which in principle could be obtained by means of direct integration of relationship (6.5.42) without use of any approximations.

However, it must not be concluded that both variants of the circuit of an optimum filter are equally easily realized technically. A very important advantage of the tracking circuit is low criticality to change of a priori information feed into the circuit and relative simplicity of the discriminator as compared to the estimator unit. The discriminator will form the derivative of the logarithm of the likelihood function only at the point of the current result of measurement $\hat{\lambda}(t)$. This operation can be conducted continuously in real time. The estimator unit for issuing the point of maximum likelihood should investigate the whole range of values of λ . This can be carried out by a set of detection channels (without a terminal storage unit), constructed for assigned statistical properties y(t), for all possible values of the measured parameter λ . The circuit should be completed by a device for separation of the channel with the greatest output voltage. In principle such devices, due to their multichannel nature, for technical tool furcational can be realized.

An advantage of the new variant of the meter is basically lessening of requirements for width of the a priori distribution of measurement error in the initial moment of time. Previously this width had to be less than the section of linearity of the discriminator; now it must be somewhat wider than the peak of the likelihood

-97-

function. This circumstance is important in the stage of target lock-on. Analogously, the formula of accuracy, previously valid only during linear work of the discriminator, obtains a wider domain of applicability. The section of linearity in the nontracking variant as it) were is expanded, which is an advantage attained at the price of complication of the circuit of radio processing of signals.

In the case of a multichannel lock-on system, built from a great number of filters, gated amplifiers, etc., the estimator unit can be realized on the basis of this lock-on system. If channels are placed sufficiently closely, and the probability of appearance in the given region of large noise peaks or extraneous targets is small, such application of a lock-on system does not require additional commentaries.

6.6.6. Potential Accuracy of Measurement

Above we obtained an expression for variance of resultant error of measurement of a parameter changing in time. However, in its derivation from the very beginning we used a specific approximation of the likelihood function. Meanwhile, it is no less interesting to determine the potential minimum variance of measurement, as far as possible not using any approximations. If we could find coincidence of the true limit of accuracy with the previously derived expression, this would be a proof of the immaterial departure from the true optimum when using Gaussian approximations of the likelihood function.

Let us assume that the a priori distribution of probabilities for $\lambda(t)$ is known and can be described by density $P_0(\lambda)$. We consider that there exist derivatives $\frac{\partial}{\partial \lambda_k} F(\mathbf{y}|\boldsymbol{\lambda}) P_0(\boldsymbol{\lambda})$. We shall prove that for any estimate $\hat{\boldsymbol{\lambda}}(\mathbf{y}) = \{\lambda_1(\mathbf{y}), \ldots, \lambda_n(\mathbf{y})\}$ of vector $\boldsymbol{\lambda} = \{\lambda_1, \ldots, \lambda_n\}$, describing a realization of the variable parameter, we have the inequality

$$\sum_{i,k=1}^{n} \Sigma_{ik} u_i u_k \ge \sum_{i,k=1}^{n} C_{ik} u_i u_k,$$

(6.6.58)

where $\Sigma_{ik} = (\lambda_1 - \hat{\lambda}_i)(\lambda_k - \hat{\lambda}_k)$ - elements of a matrix of second moments of errors of $\Sigma = || \Sigma_{ik} ||$ of the order $(n \times n)$;

$$C = || C_{ik} || - symmetric matrix of order (n × n), the inverse of matrix $C^{-1} = || C_{ik}^{(-)} ||$ with elements$$

$$C_{i\lambda}^{(-1)} = \iint \frac{\partial}{\partial\lambda_i} \ln \left[P(\mathbf{y}|\lambda) P_{\mathbf{o}}(\lambda) \right] \frac{\partial}{\partial\lambda_\lambda} \ln \left[P(\mathbf{y})|\lambda) P_{\mathbf{o}}(\lambda) \right] \times \\ \times P(\mathbf{y}|\lambda) P_{\mathbf{o}}(\lambda) \, d\mathbf{y} d\lambda, \qquad (6.6.59)$$

 x_i (i = 1, ..., n) - arbitrary quantities, where the equality in (6.6.58) is where the for all i = 1, ..., n there are satisfied these relationships:

$$\hat{\lambda}_{i} - \lambda_{i} - \sum_{k=1}^{n} C_{ik} \frac{\partial}{\partial \lambda_{k}} \left[\ln P(\mathbf{y} | \lambda) P_{\mathbf{o}}(\lambda) \right] d\lambda = 0.$$
 (6.6.60)

For proof we shall consider functions

$$f_{i} = (\lambda_{i} - \hat{\lambda}_{i}) \sqrt{P(\mathbf{y} \mid \lambda) P_{\bullet}(\lambda)},$$

$$\mathbf{\hat{e}}_{i} = \frac{\partial}{\partial \lambda_{i}} \ln \left[P(\mathbf{y} \mid \lambda) P_{\bullet}(\lambda) \right] \sqrt{P(\mathbf{y} \mid \lambda) P_{\bullet}(\lambda)}$$
(6.6.61)

and form functions

$$\psi_i = \psi_i(\mathbf{y}, \lambda) = f_i + \sum_{k=1}^n C_{ik} \varphi_k. \qquad (6.6.62)$$

We introduce arbitrary quantities $u_i (i = 1, ..., n)$ and compose the sum

in $\lim_{\lambda \to 1} \psi_1(\mathbf{y}, \lambda) = \Psi(\mathbf{y}, \lambda)$. The integral of the square of function $\Psi(\mathbf{y}, \lambda)$, obviously, $\mathbf{i}=1$

is larger than or equal to zero. Therefore

$$0 < \iint \Psi^{3}(\mathbf{y}, \lambda) d\mathbf{y} d\lambda == \iint \left(\sum_{i=1}^{n} u_{i} \psi_{i} \right)^{2} d\mathbf{y} d\lambda =$$

$$= \sum_{i, k=1}^{n} u_{i} u_{k} \iint \psi_{i} \psi_{k} d\mathbf{y} d\lambda =$$

$$= \sum_{i, k=1}^{n} u_{i} u_{k} \iint \left[f_{i} f_{k} + 2 f_{i} \sum_{j=1}^{n} C_{kj} \varphi_{j} +$$

$$+ \sum_{i, m=1}^{n} C_{ij} C_{km} \varphi_{j} \varphi_{m} \right] d\mathbf{y} d\lambda =$$

$$= \sum_{i, k=1}^{n} u_{i} u_{k} \left[\sum_{i, k=1}^{n} C_{ik} + 2 \sum_{j=1}^{n} C_{ij} \iint \varphi_{j} f_{k} d\mathbf{y} d\lambda \right], \qquad (6.6.62)$$

since

$$\iiint_{i} \int_{\mathbf{A}} d\mathbf{y} d\lambda = \iint (\lambda_{i} - \hat{\lambda}_{i}) (\lambda_{k} - \hat{\lambda}_{k}) P(\mathbf{y} \mid \lambda) P_{\mathbf{y}}(\lambda) d\lambda = \Sigma_{ik}, \qquad (6.6.64)$$

$$\iint \varphi_{i} \varphi_{k} dy d\lambda = \iint \frac{\partial \ln P(y \mid \lambda) P_{0}(\lambda)}{\partial \lambda_{i}} \times \qquad (6.6.65)$$

-99-

$$\times \frac{\partial \ln P(\mathbf{y}|\boldsymbol{\lambda}) P_{\bullet}(\boldsymbol{\lambda})}{\partial \boldsymbol{\lambda}_{k}} P(\mathbf{y}|\boldsymbol{\lambda}) P_{\bullet}(\boldsymbol{\lambda}) d\mathbf{y} d\boldsymbol{\lambda} = C_{ik}^{(-1)}, \qquad (0,1,0)$$

Let us consider the integrals in the right part of inequality (G, U, C^{-1}) :

$$\iint f_i \varphi_h dy d\lambda = \iint (\lambda_i - \hat{\lambda}_i) \frac{\partial \ln P(y|\lambda) P_0(\lambda)}{d\lambda_h} P(y|\lambda) P_0(\lambda) d\lambda dy =$$
$$= \iint (\lambda_i - \hat{\lambda}_i) \frac{\partial P(y|\lambda) P_0(\lambda)}{\partial \lambda_h} dy d\lambda.$$

Is agrating in parts, we obtain

$$\iint f_i \varphi_k dy d\lambda = -\delta_{ik}, \qquad (c, c, c \in)$$

and then (6.6.7) reduces to form

$$0 < \sum_{i, k=1}^{n} u_{i}u_{k} \left[\Sigma_{ih} + C_{ih} - 2 \sum_{j=1}^{n} \delta_{ij}C_{jk} \right] =$$

$$= \sum_{i, k=1}^{n} u_{i}u_{k} \left(\Sigma_{ih} - C_{ik} \right), \qquad (C_{i}, C_{ij})$$

the equality in (6.6.63) and, consequently, in (0.6.98) is attained when $\psi_{\pm}(\mathbf{y}, \mathbf{\lambda}) = 0$, i.e., upon satisfaction of (6.6.60).

Inequality (6.6.58) shows that the efficients of scattering for any estimate λ whethy includes the ellipsoid corresponding to matrix C. Thanks to arbitrarian of quantities α_1 from (6.6.58) it follows that

$$\boldsymbol{\Sigma}_{\boldsymbol{k}\boldsymbol{k}} = \boldsymbol{\sigma}_{\boldsymbol{k}}^2 \geq \boldsymbol{C}_{\boldsymbol{k}\boldsymbol{k}}, \qquad (\boldsymbol{\upsilon}, \boldsymbol{\zeta}, \boldsymbol{7}\boldsymbol{\boldsymbol{\upsilon}})$$

The the mean square error of measurement of $\chi(b)$ at any moment of time f_{μ} . That is from below by the magnitude of $C_{\bf kk}$.

which is of variable parameter $\hat{\lambda}(y)$, for which in inequalities (1.4.5%), (1.4.5%) and (4.5.7%) there is achieved the sign of eachieve, by an approximation (4.5.7%) there is achieved the sign of eachieve, by an approximation (5.6.7%) and (4.5.7%) there is achieved the sign of eachieve. Then matrix $\| C_{ik} \|$ (5.6.7%) and (4.5.7%) there is achieved the sign of eachieve. Then matrix $\| C_{ik} \|$ (5.6.7%) and (4.5.7%) there is achieved the sign of eachieve. Then matrix $\| C_{ik} \|$ (5.7%) and (4.5.7%) there is achieved the sign of eachieves. Then matrix $\| C_{ik} \|$ (5.7%) and (1.5%) and characterizes potential accuracy of measurement. If (5.7%) accuracy of measurement of time-variable parameters of input signal. In the case of known a priori statistics. From the degree of difference of C_{ik} and quantities Σ_{ik} , valuated for any concrete rule of formation.

-100-

of estimate $\lambda(y)$, it is possible to judge the nearness of this rule, and of the device carrying out measurement corresponding to it, to the optimum one ensuring potentially possible accuracy of measurement. As the index of efficiency, quantitatively characterizing this nearness it is convenient to select the ratio of volume of errors of measurement in any moment of time interesting us t_k for efficient and efficient estimates, i.e.,

$$f_{A} = \frac{\Sigma_{AA}}{C_{AA}}.$$
 (6.6.71)

Figuression (6.6.59) for matrix \mathbf{C}^{-1} can be transformed to a form more convenient For calculations. Introducing again the logarithm of the likelihood function

$$L(\lambda) = \ln P(\mathbf{y} \mid \lambda)$$

and considering that

$$\int \frac{\partial L}{\partial \lambda_i} P(\mathbf{y}|\mathbf{\lambda}) d\mathbf{y} = \int \frac{\partial P(\mathbf{y}|\mathbf{\lambda})}{\partial \lambda_i} d\mathbf{y} = \frac{\partial}{\partial \lambda_i} \int P(\mathbf{y}|\mathbf{\lambda}) d\mathbf{y} = 0, \qquad (6.6.72)$$

we obtain

$$C_{is}^{(-1)} = \int \left[-\int dy \, \frac{\partial^{4} L(\lambda)}{\partial \lambda_{i} \partial \lambda_{k}} P(y \mid \lambda) + \frac{\partial \ln P_{\bullet}(\lambda)}{\partial \lambda_{i}} \frac{\partial \ln P_{\bullet}(\lambda)}{\partial \lambda_{k}} \right] P_{\bullet}(\lambda) d\lambda.$$
(6.6.73)

Thus, matrix C^{-1} is presented in the form of the sum of two matrices: matrix

$$\mathbf{A} = \left\| \int \frac{\partial \ln P_{\bullet}(\lambda)}{\partial \lambda_{t}} \frac{\partial \ln P_{\bullet}(\lambda)}{\partial \lambda_{h}} P_{\bullet}(\lambda) d\lambda \right\|, \qquad (6.6.74)$$

depending only on a priori distribution of λ , and matrix

$$\overline{\mathbf{A}} = \| \int \mathbf{A}_{ik}(\lambda) P_{\mathbf{0}}(\lambda) d\lambda \|, \qquad (5.6.75)$$

where matrix $\| A_{ik}(\lambda) \| = A(\lambda)$ is determined by expression

$$A_{ik}(\lambda) = -\int \frac{\partial L(\lambda)}{\partial \lambda_i \partial \lambda_k} P(\mathbf{y} | \lambda) d\mathbf{y} = \int \frac{\partial L(\lambda)}{\partial \lambda_i} \frac{\partial L(\lambda)}{\partial \lambda_k} P(\mathbf{y} | \lambda) d\mathbf{y}. \quad (6.6.76)$$

Matrix $\mathbf{\overline{A}}$ depends, in general, both on the method of encoding $\lambda(t)$ in y(t) and properties of input signal p(t), and also on the a priori distribution. As we shall subsequently prove with examples, in a whole series of practically interesting cases matrix $\mathbf{A}(\lambda)$ does not depend on λ . Then $\mathbf{\overline{A}} = \mathbf{A}$ and the a priori distribution of probabilities affects potential accuracy of measurement only by means of matrix \mathbf{A} . Matrix **A**, as is clear from the definition, is that matrix which we used in solving the problem of synthesis of an optimum system of measurement. Matrix **A** for Gaussian a priori statistics is equal to matrix $\mathbf{V} = || \mathbf{V}_{ik} ||$. Therefore, in this

10 10

CASE

and

 $C^{-1} = A + V$ C+CAR=R,

consequently, for Gaussian a priori statistics the matrix or efficient estimates coincides with the same matrix C which appeared in the process of solution of the problem of synthesis and which, as we showed earlier, characterized accuracy of the synthesized meter in linearized conditions.

It follows from this that the synthesized meter is actually optimum as long as conditions of its linearization are satisfied, inasmuch as it ensures potentially possible accuracy.

0.0.7. Allowance for Side Peaks of the Likelihood Function

Gaussian approximation of a likelihood function of type (6.6.3) or (6.6.4)remains true only near the true value of $\rangle(t)$, if noises are sufficiently low. In the general case during synthesis it is necessary to consider side peaks of the likelihood function, explained both by imperfect form of the autocorrelation function of the useful component y(t), and by purely noise disturbances. Let us show to what sort of circuit changes complication of the approximation of the likelihood function, aimed at more closely reflecting its structure, leads.

In the first place we shall study the case when side reaks of the likelihood function are explained by a specific imperfectness of the useful component of the signal. i.e., corresponding peaks of its autocorrelation function. The position of thece peaks is rigidly related to the position of the basic peak, so that the following complete approximation of the likelihood function, based on relationships (6.6.5) and (6.6.6) is convenient:

$$P(\mathbf{y}|\boldsymbol{\lambda}) = \sum_{i=-N}^{N} P(\mathbf{y}|\boldsymbol{\hat{\lambda}} + \Delta_{(i)}) \exp\left\{z_{(i)}^{+} (\boldsymbol{\lambda} - \boldsymbol{\hat{\lambda}} - \Delta_{(i)}) - \frac{1}{2} (\boldsymbol{\lambda} - \boldsymbol{\hat{\lambda}} - \Delta_{(i)})^{+} A_{(i)} (\boldsymbol{\lambda} - \boldsymbol{\hat{\lambda}} - \Delta_{(i)})\right\}.$$
(5.6.77)

Here $\Delta_{(i)}$ - separation of the basic and i-th side peaks; $\mathbf{z}_{(i)}, \mathbf{A}_{(i)}$ - matrices, expressed as per Paragraph 6.6.1 through the first and second derivatives of ln $P(\mathbf{y}|\boldsymbol{\lambda})$ with respect to $\boldsymbol{\lambda}$ at points $\boldsymbol{\lambda} = \hat{\boldsymbol{\lambda}} + \boldsymbol{\Delta}_{(i)}$, where

-102-

among the considered peaks 2N + 1 there are included those which exceed the noise background in level.

Multiplying (6.6.77) by Gaussian a priori distribution (6.6.8) and integrating by a method analogous to that presented in Paragraph 6.6.2, taking into account the dimension of all matrices $|| A_{(i)kj} || = || K_{(i)k}\delta_{kj} ||$, we have the following exclusion for the operator of an optimum meter:

$$\begin{aligned}
\hat{\lambda}_{n} &= \sum_{i=N}^{n} \tilde{A}_{n}^{(i)} \sum_{k=i}^{n} C_{(i)nk} [\mathbf{z}_{(i)k} + \mathbf{K}_{(i)k} (\hat{\lambda}_{k} + \Delta_{(i)k} - \overline{\lambda}_{k})] + \overline{\lambda}_{n}, \\
& C_{(i)} &= [A_{(i)} + \mathbf{V}]^{-1}; \quad \overline{A}_{n}^{(i)} = -\frac{A_{n}^{(i)}}{N}, \\
& \sum_{k=-N}^{n} A_{n}^{(k)}, \\
& A_{n}^{(i)} &= P(\mathbf{y} | \hat{\lambda} + \Delta_{(i)}) \left[\frac{\det (A_{(i)}^{-1} + \mathbf{R})}{\det A_{(i)}^{-1}} \right]^{-1/2} \times \\
& \times \frac{\exp\left\{-\frac{1}{2} [\mathbf{z}_{(i)} + A_{(i)} (\hat{\lambda} - \Delta_{(i)} - \lambda)] + (A_{(i)}^{-1} - C_{(i)}) [\mathbf{z}_{(i)} + A_{(i)} (\hat{\lambda} - \Delta_{(i)} - \hat{\lambda})]\right\}}{\exp\left\{-\frac{1}{2} \mathbf{z}_{(i)}^{+} A_{(i)} \frac{P_{en}(i)}{P_{n}(i)}.
\end{aligned}$$
(6.6.78)

Normal transition in (6.6.79) to continuous observation taking into account diagonalness of $A_{(1)}$ gives

WINT

$$\begin{split} \hat{\mathbf{\lambda}}(t) &= \sum_{i=-N}^{N} \overline{\mathbf{\Lambda}}^{(i)}(t) \int_{t_{0}}^{t} \mathbf{c}_{(i)}(t, \tau) [\mathbf{z}_{(i)}(\tau) + K_{(i)}(\tau) (\hat{\mathbf{\lambda}}(\tau) - \mathbf{\Delta}_{(i)}(\tau) - \\ -\overline{\mathbf{\lambda}}(\tau)] d\tau + \overline{\mathbf{\lambda}}(t), \quad (6.6.80) \end{split}$$

$$\begin{aligned} \hat{\mathbf{\Lambda}}^{(i)}(t) &= \mathbf{\Lambda}^{(i)}(t) / \sum_{k=-N}^{N} \mathbf{\Lambda}^{(k)}(t); \\ \mathbf{\Lambda}^{(i)}(t) &= \mathbf{\Lambda}^{(i)}(t) + \mathbf{\Delta}_{(i)} \exp\left\{\frac{1}{2}\int_{t_{0}}^{t} K_{(i)}(\tau) \mathbf{z}_{(i)}^{2}(\tau) d\tau - \\ -\frac{1}{2}\int_{t_{0}}^{t} K_{(i)}(\tau) [\mathbf{z}_{(i)}(\tau) + K_{(i)}(\tau) (\hat{\mathbf{\lambda}}(\tau) - \mathbf{\Delta}_{(i)} - \overline{\mathbf{\lambda}}(\tau)]]^{2} d\tau + \\ +\frac{1}{2}\int_{t_{0}}^{t} \int_{t_{0}}^{t} \mathbf{c}_{(i)}(t, \tau) [\mathbf{z}_{(i)}(t) + K_{(i)}(t) (\hat{\mathbf{\lambda}}(t) - \mathbf{\Delta}_{(i)} - \overline{\mathbf{\lambda}}(\tau)]] [\mathbf{z}_{(i}\mathbf{x}(\tau) + \\ + \mathbf{K}_{(i)}(\tau) (\hat{\mathbf{\lambda}}(\tau) - \mathbf{\Delta}_{(i)} - \overline{\mathbf{\lambda}}(\tau)]] dt d\tau + \int_{t_{0}}^{t} d\mu \int_{t_{0}}^{t} \mathbf{c}_{(i)}(t, \tau; \mu) dt \Big\}, \end{split}$$

 $c(t, t; \mu)$ - solution of equation (6.6.31), in which instead of K(s) we take μ K(s).

-103-

AMA.211 C 2324 S 10227-04



Fig. 6.18. Optimum meter with allowance for side peaks of the autocorrelation function: 1) tuned detection receiver; 2) discriminator; 3) accuracy unit; 4, 5) nonlinear units; 6) inertialess amplifier with gain factor $K_{(1)}$; 7, 10) adders for introducing

shift of side peak $\Delta_{(1)}$; 8) unit for formation of unnormalized coefficient $\Lambda^{(1)}(t)$; 9) linear filter with pulse response $c_{(1)}(t, \tau)$; 11) normalizing amplifier; 12) unit for producing normalized coefficients; Σ - adder.

Let us turn to interpretation of operator (6.6.81) in the form of the circuit of (Fig. 6.18). The overall circuit consists of 2N + 1 perallel-working circuits, similar to the optimum meter of Fig. 6.13. Each of them includes a discriminator 2, an accuracy unit 3, adder of the loop of intercommunication and linear filteramplifiers 9 and 6. As a new element in each i-th circuit there is coupled a device 1, essentially executing the operation of a detection receiver tuned to the i-th peak of the likelihood function. The position of the i-th peak is counted off from the a asured value $\hat{\lambda}(t)$ with help of known quantity $\Delta_{(i)}$. Furthermore, in the i-th circuit there are taps from the output of the discriminator and adder of intercommunication to nonlinear units 4 and 5. Output signals of all filters 9 through normalizing amplifiers 11 are fed to the common adder Σ , where they are combined with each other with the help of normalized weights $\tilde{\lambda}^{(i)}(t)$. The latter are expressed through unnormalized factors $\Lambda^{(i)}(t)$, formed finally in 12 with respect to the ideation of $\gamma(t)$ and data of measurement with the help of detection

-104-

receiver 1, nonlinear units 4, and cross-multipliers 8.

At the output of adder Σ there will form the estimated value of the parameter (after subtracting the mean value $\overline{\lambda(t)}$). This quantity is fed to internal feedback circuits through adders for input of shift 10. After addition to $\overline{\lambda(t)}$ there will be former, finally, the measured value $\widehat{\lambda}(t)$, controlled by selection of all units of this processing, which we consider detection receivers, discriminators and accuracy units. When $\widehat{\lambda}(t)$ is fed to the i-th point these devices produce shift of $\widehat{\lambda}$ by $\Delta_{(1)}$, similar to that noted above.

Let us discuss the physical meaning of factors $\Lambda_n^{(1)}$, expressed in the discrete case by formula (6.6.79). The first factor $P(\mathbf{y}|\hat{\boldsymbol{\lambda}} + \Lambda_{(1)})$, formed in circuit terms by the detection receiver, reflects the magnitude of some peak which characterizes the signal.

The second quantity $P_{cut}(1)$ can be reduced to form

$P_{em(i)} = \left[\det (\Lambda_{(i)}^{-1} + R) \right]^{-1/2} \exp \left\{ - \eta_{cm(i)}^{+} (\Lambda_{(i)}^{-1} + R)^{-1} \eta_{cm(i)} \right\},$

where vector quantity $\eta_{CU}(1) = \mathbf{A}_{(1)}^{-1} \mathbf{z}_{(1)} + \hat{\lambda} + \Delta_{(1)} - \bar{\lambda}$ of the dimensionality of the measured parameter constitutes (after subtracting $\bar{\lambda}$) the true value of the measured coordinates with correlation matrix \mathbf{R} , mixed with noise whose correlation matrix is equal to $\mathbf{A}_{(1)}^{-1}$. One can prove this by expanding $\mathbf{z}_{(1)}$ analogously to $(6.\ell.23)$ near the true value of the peak of the side overshoot.

Thus, $P_{cu(1)}$, formed by nonlinear unit 5, shows the "likelihood" of the given output of the i-th adder of intercommunication as a random variable with correlation matrix $A_{(1)}^{-1} + R$.

Analogously we can reduce factor P $_{\rm HI}$ (1) to the form

$P_{m(i)} = [\det A_{(i)}^{-1}]^{-1/2} \exp \{-\eta_{m(i)}^{+} A_{(i)} \eta_{m(i)} \}.$

where $\eta \equiv (i) = \mathbf{A}_{(1)}^{-1} \mathbf{z}_{(1)}$ with dimensionality of the measured coordinate constitute equivalent noise at the input of the i-th discriminator (with small mismatches).

Consequently, P $_{\rm H}$ (i), formed in 4, shows the likelihood of the given input of the i-th discriminator as a random variable with correlation matrix $\mathbf{A}_{(1)}^{-1}$. As a whole P $_{\rm off}(\mathbf{i})/P$ $_{\rm H}$ (i) constitutes as it were the likelihood ratio for separation of the parameter from its additive mixture with noise. The less a given peak is suppressed by noise, the bigger factors $P(\mathbf{y}|\hat{\boldsymbol{\lambda}} + \boldsymbol{\Delta}_{(1)})$ and $P_{\rm off}(\mathbf{i})/P$ $_{\rm H}$ (i). Inasmuch as in the process of work there constantly occurs accumulation at the output of detection receivers and nonlinear units, the true peak obtains weight $\chi_n^{(1)}$, more and more exceeding the weight of all other peaks, until this weight remains the only one differing from zero.

Thus, allowance for side peaks of the considered type in steady-state regime does not lead to circuit deviations from the meter considered earlier. Making the circuit more complex has, on the other hand, the advantage that it makes optional initial lock-on with accuracy equal to the width of the basic peak of the likelihood function; tracking is established even if initial error leads to a hit on a side peak, allowed for in synthesis.

Now let us turn to the case when the peak of the autocorrelation function is the only one, and side peaks of the likelihood function are connected only with the noise background. Here approximation of the likelihood function based on relationships (6.6.4) and (6.6.7) is convenient:

$$P(\mathbf{y}|\mathbf{\lambda}) = \sum_{i} P(\mathbf{y}|\mathbf{\dot{\lambda}}_{(i)}) \exp\left\{-\frac{1}{2}(\mathbf{\lambda} - \mathbf{\dot{\lambda}}_{(i)})\mathbf{\dot{\Lambda}}_{(i)}(\mathbf{\lambda} - \mathbf{\dot{\lambda}}_{(i)})\right\}.$$
 (6.6.82)

It is necessary to stipulate that the number of peaks of the likelihood function can be arbitrarily large with a large domain of definition of λ , so that it is reasonable to approximate by (6.6.63) those peaks which exceed in magnitude a certain threshold level. As we can see, such approximation, being simple in form, is not so in essence. Cross multiplying (6.6.82) with a priori Gaussian distribution (6.6.8) and integrating; for the operator of an optimum meter for the last moment of the time of observation we obtain the following expression

$$\hat{\boldsymbol{\lambda}}_{n} = \sum_{l} \overline{\boldsymbol{\lambda}}_{n}^{(l)} \sum_{k=1}^{n} C_{(l)nk} K_{(l)k} [\tilde{\boldsymbol{\lambda}}_{(l)k} - \overline{\boldsymbol{\lambda}}_{k}] + \overline{\boldsymbol{\lambda}}_{n}, \qquad (6.6.85)$$

where

$$C_{(i)} = [A_{(i)} + V]^{-1}; \quad A_{(i)} = || K_{(i)k}^{*}_{ki} ||; \quad \overline{\Lambda}_{n}^{(l)} = \Lambda_{n}^{(l)} / \sum_{j} \Lambda_{n}^{(j)};$$

$$\Lambda_{n}^{(l)} = P(y | \lambda_{(i)}) [\det(A_{(i)}^{-1} + R)]^{-1/2} \times$$

$$\times \exp\left\{-\frac{1}{2} (\lambda_{(i)} - \overline{\lambda})^{+} (A_{(l)}^{-1} + R)^{-1} (\lambda_{(i)} - \overline{\lambda}\right\}. \quad (6.6.8^{1/2})$$

With transition to continuous observations (6.6.83) reduces to the form

$$\tilde{\boldsymbol{\lambda}}(t) = \sum_{i} \tilde{\boldsymbol{\Lambda}}(t) (t) \int_{0}^{t} \boldsymbol{c}_{(1)}(t, \tau) K_{(1)}(\tau) [\tilde{\boldsymbol{\lambda}}_{(1)}(\tau) - \tilde{\boldsymbol{\lambda}}(\tau)] d\tau + \overline{\boldsymbol{\lambda}}(t). \quad (6.6.85)$$

A block diagram interpreting relationship (6.6.85) is presented in Fig. 6.19. It consists of separate parallel circuits, very similar to the nontracking meter of (Fig. 0.17). The estimator unit issues a set of points of maximum value of the

likelihood function and corresponding points of current values of "sharpness" of peaks of $P(y|\lambda)$. After subtracting the mean value from quantities $\hat{\lambda}_{(1)}(t)$ they are processed by linear filters with pulse responses $c_{(1)}(t, \tau)K_{(1)}(\tau)$, and then results are summed with normalized weighting factors $\lambda^{(i)}(t)$. After reverse input of the where some there will be formed estimate $\widehat{\lambda}(t)$.

Partial circuits of Fig. 6.19 have filters, absolutely similar to those considered in Paragraph 6.6.5, and therefore do not require explanation. We need only indicate the physical meaning of factors $\tilde{\Lambda}^{(1)}(t)$. In distinction from the past case, (esides the measure of the height of peak $\mathbb{P}(\mathbf{y}|\mathbf{\check{\lambda}}_{(1)})$ these factors have in their composition quantities

$$P_{\bullet=(i)} = [\det (A_{(i)}^{-1} + R)]^{-1/2} \times \\ \times \exp \left\{ -\frac{1}{2} (\dot{\lambda}_{(i)} - \overline{\lambda}) + (A_{(i)}^{-1} + R)^{-1} (\dot{\lambda}_{(i)} - \overline{\lambda}) \right\}, \qquad (6.6.86)$$

showing the "likelihood" that the position of the 1-th peak is a random function with correlation matrix $A_{(1)}^{-1}$ + R, i.e., consists of a variable parameter with correlation matrix **R** and a fluctuating disturbance with correlation matrix $A_{(1)}^{-1}$. Practically chl/ one of the peaks is true. The others, first, turn out to be of small height, which is necessarily revealed in the process of accumulation, and second, they



Fig. 6.19. Optimum meter with allowance for noise peaks: 1) estimator unit; 2) linear filter with pulse responses $c_{(1)}(t, \tau)K_{(1)}(\tau)$; 3) unit for producing normalizing factors; 4) general adder.

contain in their position a shift from the true value of λ , which being accumulated in inertial nonlinear unit 4, according to (6.6.86), leads to further decrease of the weight of this peak during formation of the estimate. As a result the circuit again passes into the known circuit of Fig. 6.17, working on one peak.

The conducted investigation shows that for a sufficiently

large observation time the meter circuit, synthesized on the basis of a "single-peak" approximation, in structure, and consequently also in performance, will differ little from systems built with finer allowance for the real structure of the likelihood pur-This indicates the sufficient universality of results of the preceding agraphs. points.

-107-

6.6.8. Case of a Partially Assigned Discriminator

Everywhere above we assumed that the input realization does not depend on results or estimation of the parameter, i.e., the method of forming the realization and the method of estimation do not have points of contact. In practice this is not always so. As an example we shall consider a radar goniometer system with a tracking antenna. From the antenna output there proceeds a signal, a certain parameter of modulation of which depends on the mismatch between the direction to the sounded target and the axis of the radiation pattern of the antenna. From physical considerations it is clear that for a maximum level of the reflected signal the axis of the antenna should to the best possible degree coincide with the target direction, but inasmuch as we have at our disposal no more exact judgement about the position of the target than the estimate of this position, the axis of the antenna should coincide with the current estimated position of the target.

Notice it is clear that the input (for the receiver) realization from which the estimate is produced directly depends on the estimate. We are primarily interested in cases when the dependence on the estimate is expressed through the difference between it and the true value of the parameter. It is reasonable to assign these bases to the class of systems with a partially assigned discriminator. Actually, the antenna system, during synthesis considered here to be assigned, to a considerable extent predetermines optimum construction of the subsequent circuits.

The indicated "feedback" between the realization and the estimate deprives us of the possibility of applying without reservations the apparatus developed in the preceding points. Let us start from the fact that conditional risk

$R(P_{\bullet}, \hat{\lambda}) = \iint [(\lambda, \hat{\lambda}) P(\mathbf{y} | \lambda - \hat{\lambda}) P_{\bullet}(\lambda) d\lambda d\mathbf{y}$

(6.6.87)

turns out to depend on $\hat{\lambda}$ both through the loss function and the likelihood function, so that, strictly speaking, its variation should be produced in full by both these functions. This leads to very tiresome calculations and a very complicate meter circuit.

Considering the case of discrete observation, we simplify the problem, assuming that the estimate should be produced for moment t_n from the sampling produced at moments of time $t_1, t_2, \ldots, t_{n-1}$. Then relationship (6.6.87) for a quadratic loss function will be rewritten in the form

$$R(P_{\bullet}, \hat{\lambda}) = \int \dots \int \sum_{i=1}^{n} (\lambda_{i} - \hat{\lambda}_{i})^{s} \times P(y_{1}, \dots, y_{n-1} | \lambda_{1} - \hat{\lambda}_{1}, \dots, \lambda_{n-1} - \hat{\lambda}_{n-1}) \times \times P_{\bullet}(\lambda_{1}, \dots, \lambda_{n}) d\lambda_{1}, \dots, d\lambda_{n} dy_{1}, \dots, dy_{n-1}.$$
(6.6.88)

The solution is produced as if with a delay of one step of quantization. Actually such delay necessarily appears due to the irremediable inertia of feedback. This assumption leads to great simplification of the problem of estimation. Variation must :- produced only for estimate $\hat{\lambda}_n$, contained in the loss function. Assuming again that conditions of approximation $P(\mathbf{y}|\mathbf{\lambda} - \hat{\lambda})$ are satisfied by the Gaussian function

$$P(\mathbf{y}|\boldsymbol{\lambda}-\hat{\boldsymbol{\lambda}}) \approx P(\mathbf{y}|\boldsymbol{0}) \exp\left\{\mathbf{z}^{+}(\boldsymbol{\lambda}-\boldsymbol{\lambda})-\frac{1}{2}(\boldsymbol{\lambda}-\hat{\boldsymbol{\lambda}})^{+}\boldsymbol{\Lambda}(\boldsymbol{\lambda}-\hat{\boldsymbol{\lambda}})\right\}, \quad (6.6.89)$$

where

$$z_{k} = \frac{\partial}{\partial \epsilon_{k}} \ln P(\mathbf{y} | \boldsymbol{\varepsilon}) \Big|_{\boldsymbol{\varepsilon} = 0}; \quad A_{ik} = -\frac{\partial^{2}}{\partial \epsilon_{i} \partial \epsilon_{k}} \ln P(\mathbf{y} | \boldsymbol{\varepsilon}) \Big|_{\boldsymbol{\varepsilon} = 0}, \quad (6.6.90)$$

we obtain a solution in two equivalent forms, very similar to (6.6.19) and (6.0.21):

$$\hat{\lambda}_{n} = \sum_{k=1}^{n-1} C_{nk} [z_{k} + K_{k} (\hat{\lambda}_{k} - \overline{\lambda}_{k})] + \overline{\lambda}_{n}, \qquad (6.6.91)$$

$$\hat{\lambda}_{n} = \sum_{k=1}^{n-1} G_{nk} z_{k} + \overline{\lambda}_{n}. \qquad (6.6.92)$$

Functions C_{nk} , G_{nk} and K_k introduced here do not require explanations. Relationships (6.6.91) and (6.6.92) differ from (6.6.19) and (6.6.21) only in their upper limit (n - 1 instead of n). It is obvious that for large n and a sufficient is small step of quantization the difference between the two cases is small and at the limit disappears completely, so that for the case of continuous observation

$$\int_{\mathbf{z}} c(t, \tau) [z(\tau) + K(\tau) (\hat{\lambda}(\tau) - \overline{\lambda(\tau)})] d\tau + \overline{\lambda(t)},$$

$$(6.6.93)$$

$$\hat{\lambda}(t) = \int_{\mathbf{z}} g(t, \tau) z(\tau) d\tau + \overline{\lambda(t)}.$$

$$(6.6.94)$$

-109-

Relationships (6.6.93) and (6.6.94) completely coincide with (6.6.29) and (6.6.31). However, it is worth recalling that operators for processing a signal in the discriminator and accuracy unit are obtained by differentiation not at the estimated point, but at zero, which is explained by assignment of part of the discriminator, usually carried out in the form of elements of the antenna system.

The shown specific character of solutions of (6.6.93) and (6.6.94) is also reflected on the form of the block diagrams of optimum meters. In Fig. 6.20 there is a circuit interpreting relationship (6.6.94). In the figure there are separated element 1, forming the realization depending upon the estimate, optimized parts of discriminator 2 and accuracy unit 3, a circuit for input of the mean value, and smoothing filter G. Specially marked is the signal fed to the input $y_{\text{EX}}(t; \lambda)$ which usually differs from $y(t; \lambda - \hat{\lambda})$ both in structure, and also in physical nature. Usually this is an electromagnetic wave, proceeding to the antenna aperture. Its parameters depend on true parameters of the target λ .



Fig. 6.20. Optimum meter for a partially assigned discriminator: 1) assigned part; 2) optimized part of the discriminator; 3) optimized part of the accuracy unit; G - linear filter with pulse response $g(t, \tau)$.

The result of the present paragraph is the fact that all subsequent results in synthesis of optimum discriminators (§ 6.7) completely pertain also to the case of partially assigned discriminator, if we consider the specific character of relationships (6.6.90).

6.6.9. Optimum Meter of a Linear Functional of a Parameter with Gaussian Statistics

In a whole series of cases of radar practice there arises the necessity of measurement not only of a parameter coded in the received mixture of signal with noise y(t) but of a certain linear or nonlinear function or functional of the coded quantity. An example is measurement of the speed of an object by readings of a range finder in an incoherent radar or measurement of Cartesian coordinates of the object by a signal in which there are coded the radio (usually spherical) coordinates of $t \to 0$ object.

-110-

In this section we consider methods of optimum construction of meters of linear functionals $\nu(t)$ of parameters $\lambda(t)$ with Gaussian distribution, directly coded in y(t). The relationship between $\nu(t)$ and $\lambda(t)$ is established by linear integral relationship

$$\mathbf{v}(l) = \int_{t_1}^{t_2} F(l, s) \lambda(s) ds$$
 (6.6.95)

und is wholly determined by the form of function F(t, s). In the particular case when F(t, s) is a k-th order derivative of 5-function

$$F(t, s) = \delta^{(k)}(t-s)$$
 (k = 0, 1, 2, ...),

 $\boldsymbol{\nu}(t)$ is the k-th derivative of $\lambda(t)$. If $F(t, s) = (t - s)^{k-1}$ (k = 1, 2, ...), $T_2 = t$, $\boldsymbol{\nu}(t)$ is the k-multiple integral of $\lambda(t)$. For other examples we can consider a functional which is a certain smoothed or expected value of $\lambda(t)$, etc. If certain functions $\lambda(t)$ and $\boldsymbol{\nu}(t)$ depend on constant random parameters $a_1(i = 1, ..., m)$, differing little from mean values \overline{a} , so that

$a_1 = a_1 + \mu_1$

it is possible to expand $\lambda(t)$ and $\nu(t)$ in powers of μ_1 , limiting ourselves to terms of the first order of smallness with respect to μ_1 . Then we obtain relationships

$$\lambda(t) \approx \lambda(t; \ \overline{a}_1, \ldots, \ \overline{a}_m) + \sum_i \mathbf{p}_i \frac{\partial}{\partial a_i} \lambda(t; \ \overline{a}_1, \ldots, \ \overline{a}_m),$$

$$\mathbf{v}(t) \approx \mathbf{v}(t; \ \overline{a}_1, \ldots, \ \overline{a}_m) + \sum_i \mathbf{p}_i \frac{\partial}{\partial a_i} \mathbf{v}(t; \ \overline{a}_1, \ldots, \ \overline{a}_m),$$

by which it is also possible to approximately establish a linear relationship between $\nu(t)$ and $\lambda(t)$, allowing us to consider $\nu(t)$ a linear functional of $\lambda(t)$.

The method of synthesis of optimum meters of linear functionals is closely connected with the method of synthesis of meters of quantity r(t) directly. As also in paragraph 6.6.2, it is convenient to quantize in time all the considered quantities. An analog to (5.6.95) in this case is relationship

$$\mathbf{v}_{\mathbf{m}} = \sum_{k=N_{\mathbf{n}}}^{N_{\mathbf{n}}} F_{\mathbf{m}k} \boldsymbol{\lambda}_{k}. \tag{6.6.9}$$

We shall vary the mean risk for \boldsymbol{v}_m with quadratic loss function

$$R(P_{\bullet}, \hat{\mathbf{v}}) = \int \dots \int (\mathbf{v}_{\mathsf{m}} - \hat{\mathbf{v}}_{\mathsf{m}})^{\bullet} P(y_1, \dots, y_n | \lambda_1, \dots, \lambda_n) \times \\ \times P_{\bullet}(\lambda_{N_1}, \dots, \lambda_{N_n}) d\lambda_{N_n}, \dots, d\lambda_{N_n} dy_1 \dots dy_n,$$

where we assume that signal y(t) is accessible to observation at moments t_1, \ldots, t_n , in no way connected with moment t_n which is the argument of functional $\boldsymbol{\nu}_n$. As a

-111-

result we have equation

$$(\delta, \delta, 97) \subseteq \mathcal{A} = \{ (\mathbf{v}_{\mathbf{m}} - \mathbf{v}_{\mathbf{m}})^{2} P_{(1,\mathbf{m})} (\mathbf{y} \mid \mathbf{\lambda}) P_{\mathbf{u}_{(N_{1},N_{2})}}(\mathbf{\lambda}) d\mathbf{\lambda} = 0, \text{ for } \mathbf{u} \in (\delta, \delta, 97) \}$$

where subscripts in the likelihood function and the a priori distribution signify

moments of reading. The formal solution of (6.6,97) is

وريه المجرد ويرد

$$\dot{\mathbf{v}}_{m} = \sum_{k=N}^{N_{e}} F_{mk} \frac{\int \lambda_{k} P_{(1, m)} (\mathbf{y} \mid \lambda) P_{0(N_{e}, N_{e})} (\lambda) d\lambda}{\int P_{(i, m)} (\mathbf{y} \mid \lambda) P_{0(N_{e}, N_{e})} (\lambda) d\lambda} = \sum_{k=N_{e}}^{N_{e}} F_{mk} \overline{\lambda}_{k} (1, n), \qquad (6.6.98)$$

where there is introduced the conditional mathematical expectation $\overline{\lambda}_k(1, n)$ of the parameter at the k-th moment of time, composed on the basis of observation of mixture y(t) at moments t_1, \ldots, t_n .

In § 6.5 we indicated that the conditional mathematical expectation in principle is the optimum estimate of the parameter. However, all methods of expanding the operator of conditional mathematical expectation above pertained to a case when the estimate was produced for the k-th moment of time for a sampling, obtained up to the k-th moment. In relationship (6.6.98) conditional mathematical expectation is taken for the sampling, ending with the n-th moment, which can exceed the k-th. As a result we cannot directly use the method of obtaining an optimum estimate developed in Paragraph 6.6.2. In other words, with the requirement of physical realizability of the operator of formation of an estimate of the actual parameter $\lambda(t)$ the linear functional of the estimate, taken according to (6.6.98), will not be the optimum estimate of the linear functional. In the practical sphere, for instance, it follows from this that simple differentiation of readings of the range finder in general will not give the value of speed of the object measured with the least error. Moreover, it is possible to indicate cases when this differentiation leads to great error of measurement, limited only by the passband of the discriminator.

For finding a physically realizable solution of equation (6.6.97) we again assume fulfillment of conditions in which the likelihood function can be approximpled by Gaussian curve (6.6.3), and we assume Gaussian statistics of parameter $\lambda(t)$. The expansion of the logarithm of the likelihood function we conduct at the point of the current physically realizable estimate, which can always be required beforehand. Substituting (6.6.3) and (6.6.8) in (6.6.97) and integrating according to the method of Paragraph 6.6.2, taking into account the diagonal nature of the matrix of second derivatives, we have

-112-

a man the second states and the second states and the second states and the second states and the second states

$$\hat{\mathbf{v}}_{m} = \sum_{k=N_{1}}^{N_{k}} F_{mk} \left\{ \bar{\mathbf{\lambda}}_{k} + \sum_{i=1}^{n} C_{ki} \left[z_{i} + K_{i} \left(\hat{\mathbf{\lambda}}_{i} - \bar{\mathbf{\lambda}}_{i} \right) \right] \right\} = \sum_{k=1}^{n} B_{mk} \left[z_{k} + K_{k} \left(\hat{\mathbf{\lambda}}_{k} - \bar{\mathbf{\lambda}}_{k} \right) \right] + \sum_{k=N_{1}}^{N_{n}} F_{mk} \bar{\mathbf{\lambda}}_{k}.$$
(6.6.99)

here $\mathbf{C} = [\mathbf{V} + \mathbf{Q}]^{-1}$ - inverse matrix; \mathbf{Q} - matrix of order $(N_2 - E_1)^2$, determined by relationship

$$Q_{ik} = \begin{cases} K_i \delta_{ik}, \ 1 \le i, \ k \le n, \\ 0, \ i, \ k > n, \\ 0, \ i, \ k \le 1. \end{cases}$$
(6.6.100)

Finally, matrix **B** is determined in expanded form by relationship

$$B_{mk} = \begin{cases} \sum_{i=N_{1}}^{N_{1}} F_{mi}C_{ik}, \ 1 \le k \le n, \\ 0, \qquad \{k \le 1, \\ k > n. \end{cases}$$
(6.6.101)

From (6.6.101), taking into account values of $\rm C_{ik}$ and $\rm V_{ik},$ we can obtain equation

$$B_{mk} + \sum_{i=1}^{n} B_{mi} K_{i} R_{ik} = \sum_{i=N_{i}}^{N_{i}} F_{mi} R_{ik}, \qquad (6.6.102)$$

connecting discrete linear operator B_{jk} with the correlation function of parameter R_{jk} and the operator of linear functional F_{jk} .

Another presentation of solution of (6.6.99) has the form

$$\hat{\Psi}_{m} := \sum_{h=1}^{n} H_{mh} z_{h} + \sum_{h=N_{h}}^{N_{h}} F_{mh} \tilde{\lambda}_{h}, \qquad (6.6.103)$$

where \boldsymbol{H}_{mk} is related to \boldsymbol{B}_{mk} by relationship

$$H_{mk} = B_{mk} + \sum_{l=k}^{n} B_{ml} K_{l} O_{lk}, \qquad (6, 6, 304)$$

and G_{ik} , in turn, is determined through R_{ik} by equations (6.6.20) and (6.6.22), describing the meter of the actual parameter $\star(t)$.

Transition to continuous observation gives

 $\hat{\mathbf{v}}(t_1) = \int_0^t b(t_1, s) \left[\mathbf{z}(s) + \overline{K}(s) (\widehat{\lambda}(s) - \overline{\lambda}(s)) \right] ds + \int_0^t F(t_1, s) \overline{\lambda}(s) ds,$

where the relationship between z(t), K(t) and F(t, s) and z_k , K_k and F_{ik} in the light of what was slated above does not require explanation; $t_i = m\Delta + t_0$; $t = m\Delta + t_0$; $t = m\Delta + t_0$; and $b(t_i, t_j) = B_{ij} - a$ function, connected with $R(t, \tau)$ by an integral equation, analogous to (6.6.102):

$$b(l_{1}, s) + \int_{0}^{t} b(l_{1}, s) R(s, s) ds = \int_{t_{1}}^{t_{2}} F(l_{1}, s) R(s, s) ds$$

$$[b(l_{1}, s) = 0 \text{ when } s > l]. \quad (6.6.106)$$

We shall interpret relationship (6.6.106) in the form of the block diagram of an optimum meter of the functional of the parameter. According to Fig. 6.21 this circuit, first, contains the complete circuit of an optimum meter of $\lambda(t)$, repeating the circuit of Fig. 6.13. From the adder, connecting the discriminator output to the internal loop of feedback in smoothing circuits, we make a tap to a linear filter with pulse response b(t, s). We pass the function depicting the a priori mean value of the parameter through filter F(t, s). Inasmuch as $\overline{\lambda(t)}$ is known beforehand, physically unrealizable elements here do not appear. Addition of the two formed output voltages gives the optimum estimate of the functional $\hat{\nu}(t)$.



Fig. 6.21. Optimum meter of a linear functional on basis of the two-loop variant of the meter of a parameter: 1) discriminator; 2) accuracy unit; C - linear filter with pulse response $c(t, \tau)$; K - amplifier with gain factor K(t); B - linear filter with pulse response $b(t, \tau)$; F - operator of the linear functional.

or the law of the tree water sectors with the sector of th



(6.6.105)

Fig. 6.22. Optimum meter of a linear functional on the basis of a singleloop variant of the meter of a parameter: 1) discriminator; 2) accuracy unit; G, H - linear filters with pulse responses $g(t, \tau)$, $h(t, \tau)$; F - operator of the linear functional. Inasmuch as $b(t, s) \neq \int_{T_1}^{T} F(t, s)c(s, \tau) ds$, the assumption that linear

1

functional $\int_{T_1}^{T_2} F(t, s)\hat{\lambda}(s) ds$ from the estimate of the parameter is not the optimum

estimate of the linear functional $\int_{T_1}^{T_2} F(t, s)\lambda(s) ds$ of this parameter was confirmed.

Another modification of the optimum circuit is obtained by means of transition to continuous observation in relationship (6.6.103):

$$\tilde{\mathbf{v}}(t_1) = \int_{t_1}^{t_2} h(t_1, s) z(s) ds + \int_{t_1}^{t_2} F(t_1, s) \overline{\lambda(s)} ds. \qquad (6.6.107)$$

It is conveniently combined with the single-loop variant of a meter of $\lambda(t)$ (Fig. 6.14). The complicated variant of a joint meter of $\lambda(t)$ and $\nu(t)$ is presented in Fig. 6.22. Output of the discriminator is processed by a linear filter H with pulse response h(t, s) satisfying a relationship, analogous to (6.6.104):

$$h(t_1, s) = b(t_1, s) \neg \int b(t_1, s) K(s) g(s, \tau) ds, \qquad (6.6.108)$$

where $g(t, \tau)$ is determined by the correlation function of the parameter according to equations (6.6.31) and (6.6.32).

For characterizing accuracy of measurement of the linear functional we can obtain the following formulas:

$$R_{\text{BMX}}(t_1, t_2) = \int_{t_0}^{t} b(t_1, s) F(t_2, s) ds, \qquad (6.6.109)$$

$$\sigma_{\text{BMX}}^2(t_1) = \int_{t_0}^{t} b(t_1, s) F(t_1, s) ds.$$

The given results show that for optimum measurement of an arbitrary linear functional $\nu(t)$ of parameter $\lambda(t)$ it is sufficient to introduce in the optimum meter of the actual parameter two additional linear filters, to one of which there is a tap from the smoothing circuits, and to the other, a tap from the feed firshif feeding the mean value of $\overline{\lambda(t)}$. Characteristics of filters are determined by the form of the functional, by the correlation function of parameter $R(t, \tau)$ and by the accuracy characteristic of the discriminator K(t).

In connection with this for every form of functional and correlation properties of $\lambda(t)$ determination of the form of filters constitutes a separate problem, the

-115-

concrete solution of which will be the subject of the corresponding paragraph in § 6.8. With respect to nonlinear processing of the input signal in discriminators and accuracy units for construction of optimum meters of functionals sufficient will be those results which in concrete cases are obtained later for different state trical properties of mixture y(t) and codings in it of $\lambda(t)$.

6.6.10. Relationship of the Method of Synthesis with Other Branches of the Theory of Solutions

In conclusion we deem it useful to discuss briefly interconnections of the developed method of synthesis with other branches of the theory of statistical solutions. In the first place one should stress the link with the theory of statistical estimates, observed in two spheres: with respect to producible operations and with respect to accuracy characteristics. It is important to indicate that the discriminator of tracking meters constantly issues an efficient estimate of current mismatch $\varepsilon = \lambda - \hat{\lambda}$ between the true and measured values of the parameter. It is especially convenient to use this for a discrete signal. Disregarding parametric fluctuations, according to Paragraph 6.6.3, we have

 $z_{k} = \tilde{K}_{k}(\epsilon_{k} + \eta_{k}),$

i.e., z_{ν} with some proportionality factor

$K_{k} = -\frac{\partial}{\partial \lambda^{2}} \ln P_{k}(y|\lambda)$

is equal to mismatch plus some fluctuating addition. Variance of addition is equal to $1/\overline{K_{k}}$, i.e., to variance of the efficient estimate of the constant parameter for the period of quantization. This proves the formulated proposition.

During continuous observation for proof of the same fact it is necessary to separate a certain segment Δ . Then it turns out that variance of the estimate of mismatch for that time is equal to $1/(\overline{K(t)}\Delta)$, i.e., again to the variance of the efficient estimate of a constant parameter.

However, for larger intervals of observation error of measurement of z varying quantity is not equal to $1/(\overline{K(t)}\Delta)$, and only monotonically decreases with growth of $\overline{K(t)}$. This is understandable, too, inasmuch as besides fluctuating ones it is necessary to track dynamic disturbances.

One more interrelationship with the theory of estimation will be revealed in § 6.8, when we consider measurement of a quantity, the law of change of which in turn depends on several constant coefficients [14].

-116-

We turn now to the relationship with the theory of linear filtration. It, obviously, follows from equations (6.6.31) and (6.6.53). They coincide with equations of Wiener filtration for separation of a signal from an additive mixture of it with white noise. Here, if K(t) is not subject to averaging, or if averaging different eliminate the dependence of $\overline{K}(t)$ on time in the equivalent Wiener problem

should consider interference in the form of white noise with variable intensity, and if $\overline{K}(t) = K$ does not depend on time, then white noise is stationary. The obtained result was a consequence of the assumption of Gaussian statistics of $\lambda(t)$, but let us note that the condition of normalness of noise at the output of the dimensionary turns out to be unnecessary.

The shown reduction to the Wiener case is valid for a Gaussian parameter. In general, as in § 6.9 the example of a Markovian parameter will convincingly show, operations of smoothing are nonlinear. Furthermore, one should remember that relative to the input realization the meter remains an especially nonlinear device.

Comparison of the developed method of synthesis of a meter of a Markovian parameter with the method of R. L. Stratonovich later (§ 6.9) will show their great similarity. A peculiarity of the method developed above, besides the assumption of Gaussian distribution of $\lambda(t)$, is the clear division into primary and secondary operations of processing, the introduction of concepts of an optimum discriminator, accuracy unit and smoothing circuits. This gives the possibility of studying separately questions of construction of these devices depending upon properties, correspondingly, of the signal and the parameter coded in it.

We shall discuss now the problem of forming the a posteriori probability in the whole domain of definition of $\lambda(t)$. As we proved in Chapter 3, this operation is necessary in detection regime, when a priori information is scant. We assume the presence of a unit of optimum receivers, covering the whole region with small quantization. At the same time discriminators and accuracy units are devices, marrowly selective with respect to λ , and technically their use is preferable with low a posteriori inaccuracy.

The same unit of optimum receivers (estimator unit) is necessary in nonclosed elevates of meters, even when there is carried out transition to forming of few characteristics of the a posteriori probability. This shows the technical advantages of closed circuits, in which the actual estimate controls the region selected by the receiver. However, in conditions of large a priori variances or high-level noises,

-117-

where there will be frequent preakoffs of tracking, use of multichannel units of receivers is inevitable.

§ 6.7. <u>Synthesis of Discriminators For Different Statistical</u> <u>Properties of Signals</u>

Results obtained in § 6.6 shows that as a basic element of an optimum met r we include discriminators, which form magnitude z(t) (or z_k), which is the measure of missered between the true and measured values of the parameter. In the same where we introduced the basic characteristic of a discriminator R(t) (or R_{μ}). moretely, both functions are determined by the statistics of the mixture of the signal with noise at the input y(t) and the method of encoding in it the parameter $\lambda(t)$. However, it is possible to reveal certain other general rules, allowing one to quickly find operations of a discriminator and its characteristic in the most interesting cases. In the present section we shall consider different forms of operations of a discriminator and explain characteristics of their accuracy. At the base of our classification, in distinction from subsequent chapters, we here place the form of statistics of y(t), more exactly its useful component, depending on parameter $\lambda(t)$. Separately we consider cases of regular signals, signals with random phases, Gaussian signals, and so forth. Here, we always consider continuous observation, when the likelihood functional, operations of the discriminator and its characteristics are expressed, correspondingly, by formulas (6.6.35)-(6.6.37) and (6.6.44)-(6.6.46).

6.7.1. Regular Signals in Gaussian Noises

For simplicity we first consider the case of one signal $v(t, \lambda(t))$, taken against a background of white noise with spectral density N_O. The logarithm of the likelihood functional in an elementary segment of observation $(t - \Delta/2, t + \Delta/2)$ here has the form

$$L(\boldsymbol{y},\boldsymbol{\lambda}) = C - \frac{1}{2N_0} \int_{t-\boldsymbol{\lambda}/2}^{t+\boldsymbol{\lambda}/2} [\boldsymbol{y}(\boldsymbol{z}) - \boldsymbol{v}(\boldsymbol{\tau},\boldsymbol{\lambda}(\boldsymbol{z}))]^* d\boldsymbol{z}, \qquad (6.7.1)$$

where C - constant, depending neither on realization y(t) nor the measured parameter $\lambda(t)$.

By comparison of (6.7.1) and (6.6.35) we prove that

$$l(\mathbf{x}, \boldsymbol{\lambda}(\mathbf{x}), \boldsymbol{y}(\mathbf{x})) = -\frac{1}{2N_0} [\boldsymbol{y}(\mathbf{x}) - \boldsymbol{v}(\mathbf{x}, \boldsymbol{\lambda}(\mathbf{x}))]^n, \qquad (6.7.2)$$

whence from (6.6.36) we have

()

$$z(t) = \frac{1}{N_0} \frac{\partial v(t, \hat{\lambda}(t))}{\partial \lambda} [y(t) - v(t, \hat{\lambda}(t))]. \qquad (0.7.3)$$

-119-

Formula (6.1.3) is illustrated in Fig. 6.23. For formation of z(t) it is necessary to subtract from realization y(t) a function describing a regular signal with the measured value of the parameter $\hat{z}(t)$ and to multiply the result by the derivative of this function with respect to the parameter, also taken for the estimated value $\hat{z}(t)$. For mismatch of $\lambda(t)$ and $\hat{\lambda}(t)$ subtraction of the regular part



Fig. 0.23. Discriminator for a regular signal in white noise.

of the mixture will be incomplete and the remainder, correlating with $\frac{3}{40}$ v(t, $\hat{\lambda}(t)$), will give at the output a quantity proportional to $x + \hat{\lambda}$. As we see, operation of the discriminator here was photony inertiateds and even linear relative to the insit signal (but not with respect to difference $\lambda + \hat{\lambda}$). From (6.6.37) it is easy to prove that the character-

istic of accuracy of discriminator $K = \overline{K(t)}$ is equal to

$$K = \lim_{T \to \infty} \frac{1}{TN_{*}} \int_{0}^{t} \left[\frac{\partial \sigma(t, \lambda(t))}{\partial \lambda} \right]^{*} dt. \qquad (t, t, 4)$$

If $v(t, \lambda)$ for a fixed λ is a periodic function, then according to (6.6.45) it is possible to consider function (6.7.3) integrates over period of reputation T_r a clustere signal at the discriminator output:

$$\mathbf{z}_{\mathbf{k}} = \frac{1}{N_{\theta}} \int_{(\mathbf{k}-1)T_{f}}^{\mathbf{k}T_{f}} \frac{\partial \sigma\left(\mathbf{r}, \hat{\lambda}_{\mathbf{k}}\right)}{\partial \lambda} \left[y\left(\mathbf{r}\right) - \sigma\left(\mathbf{r}, \hat{\lambda}_{\mathbf{k}}\right) \right] d\mathbf{r}. \qquad (0.7.5)$$

Here, in relationship (6.7.4) there is no need to carry out passage to the limit; it is sufficient that T embrace a whole number of periods T_{r} :

$$K = \frac{1}{T_r N_s} \int_{0}^{T_r} \left[\frac{\partial v(i \cdot \lambda)}{\partial \lambda} \right]^s dt. \qquad (6.7.5)$$

According to (6.7.4) and (6.7.6) accuracy of a discriminator is determined by the actio of the mean value of the square of the derivative of the regular signal with respect to the parameter to the spectral density of white noise. This result is well-known [19].

If we now generalize (6.7.1) for several input signals ${}^{(1)}y(t)$, ..., ${}^{(m)}y(t)$, consisting of regular signals ${}^{(1)}v(t, \lambda)$, ..., ${}^{(m)}v(t, \lambda)$ and generally correlated white noices ${}^{(1)}n(t)$, ..., ${}^{(m)}n(t)$, so that

$$\int_{0}^{(1)} \frac{(1)v(t, 2) + (1)n(t)}{n(t)(1)n(t+s)} ds = N_{ij} (i, j = 1, ..., m),$$

-120-

we can obtain function

$$l(\mathbf{x}, \lambda, y) = -\frac{1}{2} \sum_{i, j=1}^{m} W_{ij} [{}^{(i)}y(t) - {}^{(i)}\sigma(t, \lambda)] [{}^{(j)}y(t) - {}^{(j)}\sigma(t, \lambda)] = \frac{1}{2} [y(t) - v(t, \lambda)]^{+} W [y(t) - v(t, \lambda)],$$

the inverse of N. From this we have

$$z(t) = \sum_{j=1}^{n} \left(\sum_{i=1}^{n} W_{ij} \frac{\partial^{(i)} v(t,\lambda)}{\partial \lambda} \right) [^{(j)} y(t) - {}^{(j)} v(t,\lambda)].$$
(6.7.7)

Modification of (6.7.7) for periodic $v(t, \lambda)$ is obvious. According to Fig. 6.24, illustrating relationship (6.7.7), from each input mixture there is subtracted the corresponding regular signal for $\lambda = \hat{\lambda}$, and then results are multiplied by linear combinations of derivatives with respect to the parameter of all regular signals and are added. For uncorrelated noises matrices N and W are diagonal, and the



Fig. 6.24. Discriminator for a set of regular signals in white noise.

discriminator, according to (0.7.7), consists of m independent channels, similar to Fig. 6.23, which are summed with weights, inversely proportional to intensities of noises.

The accuracy characteristic of a discriminator for the considered case is characterized by the formula



and also is expressed through mean values of the derivative of separate signals with respect to the parameter.

How we turn to the more complicated case of correlated Gaussian noises. Considering the input mixture unique, we have

$$L(y,\lambda) = C - \frac{1}{2} \int_{t-\lambda/2}^{t+\lambda/2} W_{\pi}(t_1, t_2) [y(t_1) - v(t_1, \lambda)] [y(t_2) - \frac{1}{2} \int_{t-\lambda/2}^{t+\lambda/2} W_{\pi}(t_1, t_2) [y(t_1) - v(t_1, \lambda)] [y(t_2) - \frac{1}{2} \int_{t-\lambda/2}^{t+\lambda/2} W_{\pi}(t_1, t_2) [y(t_1) - v(t_1, \lambda)] [y(t_2) - \frac{1}{2} \int_{t-\lambda/2}^{t+\lambda/2} W_{\pi}(t_1, t_2) [y(t_1) - v(t_1, \lambda)] [y(t_2) - \frac{1}{2} \int_{t-\lambda/2}^{t+\lambda/2} W_{\pi}(t_1, t_2) [y(t_1) - v(t_1, \lambda)] [y(t_2) - \frac{1}{2} \int_{t-\lambda/2}^{t+\lambda/2} W_{\pi}(t_1, t_2) [y(t_1) - v(t_1, \lambda)] [y(t_2) - \frac{1}{2} \int_{t-\lambda/2}^{t+\lambda/2} W_{\pi}(t_1, t_2) [y(t_2) - v(t_2, \lambda)] [y(t_2) - \frac{1}{2} \int_{t-\lambda/2}^{t+\lambda/2} W_{\pi}(t_1, t_2) [y(t_2) - v(t_2, \lambda)] [y(t_2)$$

where W_{Π} (t_1, t_2) - a function, the inverse in interval Δ to the correlation function of noise R_{Π} (t_1, t_2) , i.e., satisfying equation

$$\int_{t-\Delta/2}^{t+\Delta/2} R_{\pi}(t_1, s) W_{\pi}(s, t_2) ds = \delta(t_1 - t_2).$$

It is implied that Δ exceeds the interval of correlation of interference (see [3,1,2]).

Several transitions from (6.7.9) to the operation of a discriminator are possible. Questions of physical realizability of the obtained circuits are easy to avoid, if, using the symmetry of the expression under the integral sign in (4.4.4) for arguments t_1 , t_2 we produce integration not over a square $t = \Delta/2 \le t_1$, $t_2 \le t_1$, $t_2 \le t_1 + \Delta/2$, but over a triangle $t = \Delta/2 < t_1 < t + \Delta/2$, $t_1 < t_2 < t - \Delta/2$, and then double the result:

$$L(y, \lambda) = C - \int_{t-\lambda/2}^{t+\lambda/2} dt_1 \int_{t-\lambda/2}^{t_1} dt_2 W_{II}(t_1, t_2) [y(t_1) - v(t_1, \lambda)] [y(t_2) - v(t_2, \lambda)].$$

Then differentiation with respect to λ and rejection of the outer integral gives the operation of the discriminator:

$$z(t) = \frac{\partial v(t; \hat{\lambda}(t))}{\partial \lambda} \int_{-\infty}^{t} W_{\pi}(t_{1}, t_{2}) [y(t_{3}) - v(t_{3}; \hat{\lambda}(t)] dt_{3} + [y(t) - v(t; \hat{\lambda}(t))] \int_{-\infty}^{t} W_{\pi}(t_{1}, t_{3}) \frac{\partial v(t_{3}; \hat{\lambda}(t))}{\partial \lambda} dt_{3}.$$
(6.7.10)

Intermuch as the interval of observation Δ exceeds the interval of correlation of intervences, we directed the lower limit to $-\infty$. Operations, producible according to (5.7.10), are illustrated in Fig. 6.25. From the realization at the input, as



Fig. 6.25. Discriminator for a regular signal in correlated noise (jot variant). W -- linear filters with pulse responses $W(t, \tau)$ $(t > \tau)$.

above, there is subtracted the expected form of the signal, and the difference is parked through a linear filter with pulse response $W_{\rm H}$ (t, τ) ($\tau <$ t) and is multiplied by the derivative of the expected form of the signal with respect to λ . Additionally this derivative is passed through the same filter and is multiplied by difference $y(t) - v(t; \hat{\lambda}(t))$. Both results of processing are added, forming the signal at the discriminator output. With such method there are made no approximations during the transition to z(t).

Another formally simpler method of transition from (6.7.9) to operation of the discriminator is differentiation directly of (6.7.9) with respect to λ , which, taking into account symmetry of the arguments, gives

$$\frac{\partial L}{\partial \lambda} = -\int_{t-\Lambda/2}^{t+\Lambda/2} \overline{W}_{\pi}(t_1, t_2) \frac{\partial \sigma(t_1, \lambda)}{\partial \lambda} \left[y(t_2) - \sigma(t_2, \lambda) \right] dt_1 dt_2, \qquad (6.7, 11)$$

and rejection of one of the integrals. Depending on what argument we choose for rejecting the integral in (6.7.11), it is possible to obtain different, but statistically equivalent expressions for operation of the discriminator:

$$\mathbf{r}(t) = \frac{\partial \sigma(t; \hat{\lambda}(t))}{\partial \lambda} \int_{-\infty}^{+\infty} [y(\tau) - \sigma(\tau, \hat{\lambda}(t))] \mathbf{W}_{\mathbf{L}}(t, \tau) d\tau \qquad (6.7.12)$$

or

$$z(t) = [y(t) - v(t; \hat{\lambda}(t))] \int_{-\infty}^{+\infty} \frac{\partial v(\tau, \hat{\lambda}(t))}{\partial \lambda} W_{\mu}(t, \tau) d\tau, \qquad (6.7.13)$$

where limits of integration are expanded to $\pm \infty$ for reasons already explained.

Operations (6.7.12) and (6.7.13) are carried out by the circuits of Figures 6.26 and 6.27, respectively, which repeat two channels of the circuit of Fig. 6.25,



with the exception of differences in properties of linear filters, appearing due to infinite limits of integration. In all cases filtration with weighting function W $_{\Pi}$ (t₁, t₂) has rejector properties with respect to interference, carrying out suppression of the most intense part of the spectrum of interference.

Fig. 6.26. Discriminator for a regular signal in correlated noise (2nd variant): W - linearfilter with pulse response $W(t, \tau)$.

We shall explain this circumstance in greater detail, considering a periodic signal with period T_r , larger than the interval of correlation of interference. If we consider interference stationary $R_{\pi}(t_1, t_2) = R_{\pi}(t_1 - t_2)$, then $W_{\pi}(t_1, t_2)$ one should seek also in the form $W_{\pi}(t_1 - t_2)$, from which, pushing the limits of integration in (6.7.9) to $\pm \infty$, we have the Fourier transform from the inverse function $W_{\pi}(t_1 - t_2)$ in the form

$$\widetilde{\boldsymbol{W}}_{\boldsymbol{\pi}}(\boldsymbol{\sigma}) = \int_{-\infty}^{+\infty} \widetilde{\boldsymbol{W}}_{\boldsymbol{\pi}}(\boldsymbol{\tau}) e^{-l\sigma t} d\boldsymbol{\tau} = \frac{1}{S_{\boldsymbol{\pi}}(\boldsymbol{\sigma})}, \qquad (6.7.14)$$
where S $_{+}$ (ω) - spectral density of interference.

The operation for the discriminator in the case of a periodic signal will be given directly by relationship (6.7.11), where limits embrace the period of reputition. From this, passing to Fourier transforms and using (6.7.14), we have

$$z_{k} = \operatorname{Re} \frac{1}{2\pi} \int_{-\infty}^{+\infty} \frac{Y_{k}(\omega) - V(\omega, \lambda_{k})}{S_{\pi}(\omega)} \frac{\partial V^{*}(\omega, \lambda_{k})}{\partial \lambda} d\omega, \qquad (u, v, v)$$

where $Y_k(\omega)$, $V(\omega, \lambda)$ — Fourier transforms of functions y(t), $z(\cdot, \cdot)$ in the k-inperiod of the signal. According to (6.7.15), in these sections of the spectrum shere interference is more intense, the transmization factor of the linear filter coupled in the circuit decreases. This also explains the rejector properties.

The orhematic of the discriminator for this case is shown in Fig. 6.28. It remains to clarify the quantion of physical realizability of operations (4.7.12), (6.7.13) and (6.7.15), which is pieced under doubt by the infinite upper



Fig. 6.27. Discriminator for a regular signal in correlated notes (for variant). $W \rightarrow$ Sinear filter with pulse response W(t, x).

limits of integration in (6.7.12) and (6.7.13) or, which is the same, the presence in (6.7.15) of function $1/S_{-1}(\infty)$, the Fourier original of which is symmetric with respect to zero time. The situation is eased by the fact that the internal of correlation of interference is usually considerably lost than the interval of noticeable change of signal function $v(2, \lambda)$. In these conditions it is possible to replace the Fourier original

$$K = \lim_{r \to \infty} \frac{1}{T} \iint_{\partial h} \frac{\partial v(t_1, \lambda)}{\partial \lambda} \frac{\partial v(t_2, \lambda)}{\partial \lambda} W_{\alpha}(t_1, t_2) dt_1 dt_2, \qquad (4.7.16)$$

where in the case of a periodic signal one should reject the sign of the limit and equate 1 to the period of repetition T_{μ} .

Although in subsequent chapters the case of a regular signal will not be

-124-





considered, results of the present paragraph will help us to comprehend certain laws governing construction of optimum circuits in cases more complicated and more interesting for practice.

6.7.2. Gaussian Coherent Signals in Gaussian Noises

First we shall consider that simple case when the carrier of the parameter is a single signal, received against a background of normal white noise n(t):

$$y(t) = E(t) u_{\mathbf{a}}(t, \lambda) \cos \left[\mathbf{w}_{\mathbf{a}} t + \psi(t, \lambda) - \psi(t) \right] + \pi(t) =$$

= Re {E(t) e^{tq(t)} u(t, \lambda) e^{tw(t)}} + \pi(t). (6.7.47)

Here $\omega_0 = \text{carrier frequency; E(t), \phi(t)} = \text{random amplitude and phase modulations,}$ distributed correspondingly by Rayleigh and uniform law, so that E(t) cos $\phi(t)$, E(t) sin $\phi(t)$ are stationary normal independent random processes with correlation function $r(t_1 - t_2)$. Then, $\omega_0(t, \lambda)$, $\psi(t, \lambda)$ are regular modulations, depending on the measured magnitude of $\lambda(t)$; in (6.7.17) there is introduced the complex modulation factor

$$\mu(t, \lambda) == \mu_{\mathbf{A}}(t, \lambda) e^{i\phi(t,\lambda)}, \qquad (i_{1}, 7, 16)$$

reflecting immediately both functions $u_{a}(t; \lambda)$ and $\psi(t, \lambda)$. The correlation function of mixture y(t) according to Chapter 1 (Vol. 1) has the form

$$\overline{y(t_i)y(t_s)} = R(t_i, t_s) =$$

$$= \operatorname{Re} \left\{ u(t_i, \lambda) r(t_i - t_s) u^{\bullet}(t_s, \lambda) e^{t_{\bullet}(t_i - t_s)} \right\} + N_{\bullet} \delta(t_i - t_s), \qquad (0, 7, 1)$$

where

$$r(t_{1}-t_{2}) = \frac{1}{2} \overline{E(t_{1})} e^{i \gamma(t_{1})} E(t_{2}) e^{-i \gamma(t_{2})}.$$

According to formula (6.5.1), for construction of the likelihood functions) for a normal process it is necessary to find function $W(t_1, t_2)$, the inverse of the correlation function, i.e., satisfying equation

$$\int_{t-A/2}^{t+A/2} W(t_1, s) R(s, t_s) ds = \delta(t_1 - t_s), \qquad (t_1, t_1, 2t_1)$$

where integration is conducted over the elementary interval (t = $\Lambda/2$, t + L/2), which we discussed in § 6.6. Similarly to how this was done in Chapter IV(Vol. 1), we find W(t₁, t₂) in the form

whence for auxiliary function $w(t_1, t_2)$ we find equation

$$\frac{1}{2} \int_{s-4/2}^{s+4/2} r(t_1 - s) \frac{|u(s,\lambda)|^2}{N_0} w(s,t_1) ds + w(t_1,t_2) = r(t_1 - t_2). \quad (6.7.22)$$

As the first example of solution of equation (i.7.22) we shall consider a signal without amplitude modulation, when $u_{a}(t) = u_{0} = \text{const.}$ Then, inasmuch as the interval of integration exceeds the interval of correlation of fluctuations of the signal, determined by function r(t), it is possible to consider limits of integration in (6.7.22) infinite and to seek the solution of $w(t_{1}, t_{2})$ in the form $w(t_{1} - t_{2})$. The method of Fourier transformations in these conditions gives

$$\int_{-\infty}^{+\infty} w(t) e^{-i\omega t} dt \equiv w(\bullet) = \frac{S_{\bullet}(\bullet)}{1 + \frac{S_{\bullet}(\bullet)}{2M_{\bullet}}} = \frac{P_{\bullet}}{4I_{\bullet}} \frac{S_{\bullet}(\bullet)}{1 + 4S_{\bullet}(\bullet)} \cdot (6.7.23)$$

Here as the condition of normalization we assume $u_a^2 = 1$, so that the correlation function of signal fluctuations r(t) must be normalized in the form

 $r(0) = P_{e},$

(6.7.24)

where P_c - mean signal power.

Further, $S_{c}(\omega) = Fourier$ transform from (1); and $S_{0}(\omega) = S_{c}(\omega)/S_{c}(0) = normalized$ spectrum of fluctuations, related to the width of this spectrum Δf_{c} by relationship

$$\Delta f_{\rm q} = \frac{1}{2\pi} \int S_{\rm q}({\bf n}) \, d{\bf n}.$$

(5.7.25)

Finally, quantity $h = P_C/2N_O\Delta f_C$, introduced in (6.7.23), is the signal-tonoise ratio encountered earlier (ChapterIV) in the theory of optimum reception of a coherent fluctuating signal.

wither interesting example of solution of equation (6.7.22), essentially reducible to the first one, is the case of rapid amplitude modulation. If we limit ourselves to modes of modulation with constant period T_r , this is equivalent to the condition of the interval of correlation of the signal greatly exceeding period T_r . In other words, the set of neighboring pulse trains fluctuate hermoniously. Then it is possible to divide the integral in (6.7.22) into the sum of integrals over separate periods, removing factors $r(t_1 - iT_r)$, $w(iT_r, t_2)$, constant in period T_r :

-126-

$$\int_{t-A/2}^{t+A/2} r\left(t_{1}-s\right) \frac{|s\left(s,\lambda\right)|^{2}}{N_{0}} w\left(s_{1}t_{3}\right) ds \sim \sum_{i} r\left(t_{1}-iT_{i}\right) \times \frac{|s_{r}|}{N_{0}} \left(iT_{r},t_{3}\right) \int_{t_{i}-I_{1}T_{r}}^{IT_{r}} \frac{|s\left(s,\lambda\right)|^{2}}{N_{0}} ds.$$

$$(6.7.26)$$

pring integration over separate periods we consider condition of normalization

$$\frac{1}{T_r} \int_{(k-1)T_r}^{kT_r} u_a^2(s,\lambda) ds = \frac{1}{T_r} \int_{(k-1)T_r}^{kT_r} |u(s,\lambda)|^s ds = 1, \qquad (6.7.27)$$

which is a generalization of the former normalization in the case of the presence of modulation. Passing then in (6.7.26) to an integral in the initial limits, as the solution of equation (6.7.22) we again have relationship (6.7.23).

Let us now turn to expression (1.4.5) for the derivative of the logarithm of the likelihood function P(y(t)|2) of a normal process y(t), taken with respect to arbitrary parameter $\lambda(t)$. Using (6.7.21) and (6.7.23) and considering h independent of λ , we have

$$\frac{\partial}{\partial \lambda} \ln P(y|\lambda) = -\frac{1}{2N_0^2} \int_{t-A/2}^{t+A/2} y(t_1) y(t_2) w(t_1 - t_2) \times \\ \times \left\{ u_A(t_3, \lambda) \frac{\partial}{\partial \lambda} u_A(t_3, \lambda) \cos \left[w_0(t_1 - t_3) + \psi(t_1, \lambda) - \frac{\partial}{\partial \lambda} (t_3, \lambda) \frac{\partial}{\partial \lambda} \sin \left[w_0(t_1 - t_3) + \frac{\partial}{\partial \lambda} + \frac{\partial}{\partial \lambda} \right] + \frac{\partial}{\partial \lambda} + \frac{\partial}{\partial \lambda} \sin \left[w_0(t_1 - t_3) + \frac{\partial}{\partial \lambda} + \frac{\partial}{\partial \lambda} \right] \right\}$$

In order to carry out transition to the operation of the discriminator, according to relationship (6.6.36), it is necessary to present (6.7.28) in the form of ssimple integral over the interval of observation. For this it is sufficient to assume that there exists such a function h(t), for which

$$\int_{-A/2}^{+a/a} h(\tau - t_1) h(\tau - t_2) d\tau = w(t_1 - t_2)/2N_0, \qquad (6.7.29)$$

or, in frequency presentation, that this relationship is satisfied:

$$|H(l\omega)|^{n} = \frac{\omega(\omega)}{2N_{\bullet}} = \frac{hS_{\bullet}(\omega)}{1 + hS_{\bullet}(\omega)}, \qquad (6.7, 30)$$

名で見ていたの

where H(ia) - Fourier transform from h(t).

Substitution of relationship (6.7.29) in (6.7.28) gives the possibility of

passing to the operation of the discriminator:

$$z(t) = \frac{1}{N_{\bullet}} \int_{-\infty}^{+\infty} y(t_{i}) u_{a}(t_{i}; \hat{\lambda}(t)) \cos \left[\omega_{\bullet}t_{1} + \psi(t_{1}; \hat{\lambda}(t))\right] \times \\ \times h(t-t_{i}) dt_{i} \int_{-\infty}^{+\infty} y(t_{\bullet}) \left[\frac{\partial}{\partial \lambda} u_{a}(t_{i}; \hat{\lambda}(t)) \cos \left[\omega_{\bullet}t_{1} + \psi(t_{i}; \hat{\lambda}(t))\right] - \\ - u_{a}(t_{i}; \hat{\lambda}(t)) \frac{\partial}{\partial \lambda} \frac{\partial}{\partial \lambda} \sin \left[\omega_{\bullet}t_{1} + \psi(t_{i}; \hat{\lambda}(t))\right] h(t-t_{i}) dt_{i} + \\ + \frac{1}{N_{\bullet}} \int_{-\infty}^{+\infty} y(t_{i}) u_{a}(t_{i}; \hat{\lambda}(t)) \sin \left[\omega_{\bullet}t_{1} + \psi(t_{i}; \hat{\lambda}(t))\right] h(t-t_{i}) dt_{i} \times \\ \times \left[\int_{-\infty}^{+\infty} y(t_{i}) \left[\frac{\partial}{\partial \lambda} u_{a}(t_{i}; \hat{\lambda}(t)) \sin \left[\omega_{\bullet}t_{2} + \psi(t_{i}; \hat{\lambda}(t))\right] h(t-t_{i}) dt_{i} \times \\ \times \left[\int_{-\infty}^{+\infty} y(t_{\bullet}) \left[\frac{\partial}{\partial \lambda} u_{a}(t_{i}; \hat{\lambda}(t)) \sin \left[\omega_{\bullet}t_{2} + \psi(t_{i}; \hat{\lambda}(t))\right] h(t-t_{i}) dt_{i} \right] \right] \\ + u_{a}(t_{i}; \hat{\lambda}(t)) \frac{\partial \psi(t_{i}; \hat{\lambda}(t))}{\partial \lambda} \cos \left[\omega_{\bullet}t_{2} + \psi(t_{i}; \hat{\lambda}(t))\right] h(t-t_{i}) dt_{a}. \quad (6.7.31)$$

The operation represented by relationship (6.7.31) is illustrated in Fig. 6.29. Input signal y(t) proceeds to two quadrature mixers, in one of which it is multiplied oy sinusoidal oscillation of the expected frequency ω_0 , modulated in phase by the law of phase (frequency) modulation with a value of the parameter equal to the measured, and in the second mixer it is multiplied by cosinusoidal oscillation of the expected frequency with the same modulation. Then there occurs multiplication of output signals of mixers by functions reflecting the expected form of amplitude modulation, the derivative of amplitude modulation, and the derivative of phase modulation with respect to parameter λ (see Fig. 6.29), all at the same measured value of $\lambda(t)$. There is conducted filtration of the obtained signals in low-frequency linear filters with pulse response h(t), satisfying relationship (5.7.29) or (6.7.30). Then the obtained signals are mutually multiplied in multipliers, after which all output signals proceed to an adder, completing the circuit of the discriminator.

Although the given circuit is complicated, it is possible to easily explain the work of all its elements. Input mixers and multipliers are elements of so-called correlation processing, by which we usually mean multiplication by a signal of expected form with subsequent accumulation of some type. The presence of sine-cosine channels is explained by fluctuations of phase of the signal, making it impossible to predict its concrete value. In any case the signal correlates (completely or partially) with heterodyne signals, in quadrature shift relative to each other,

-128-



Fig. c.29. Optimum discriminator for a fluctuating coherent signal (low-frequency variant): 1) quadrature mixers; 2) variablegain amplifiers; 3) linear low-frequency filters; 4) multipliers; 5) adder; 6) $\pi/2$ phase shifter. which also permits forming the output signal of the device.

١

In the corresponding circuit of an optimum receiver it would be sufficient further to have only elements of multiplication by the function of amplitude modulation, in the case of a pulse signal technically executed in the form of gated stages. In the circuit of the discriminator it is additionally necessary to use derivatives with respect to the parameter of amplitude and phase (frequency) modulations, taken for the measured value of parameter $\lambda(t)$. The useful

ALL AND A

signal appears at the output of amplifiers with a law of amplification in the form of derivatives only when there is mismatch between the measured and true values of λ . Here, the signal at the output of these amplifiers in sign and magnitude corresponds to difference $\lambda = \hat{\lambda}$, if this difference is small.

Then in the circuit there follow linear filters, whose pulse response is determined by the spectrum of fluctuations of signal and the signal-to-noise ratio. They carry out accumulation in time intervals which for a low level of the signal are equal to the correlation interval and upon increase of the signal decrease domewhat. Multiplication of signals carrying information about mismatch $\varepsilon = \lambda - \hat{\lambda}$ by signals carrying only information about the amplitude of the signal is produced in order to emphasize overshoots of the signal and to maximally filter off noises. Final addition helps to compensate phase fluctuations and unites effects caused by deviations of $\hat{\lambda}(t)$ from $\lambda(t)$ in channels sensitive to amplitude and phase modulation, respectively.

Physical realizability of operations (6.7.31), not obvious in view of the infinite upper limits of integration, is explained just as in Paragraph 6.7.1. The fact is that on an optimum filter there is imposed only condition (6.7.30) with

-129-

respect to the square of the modulus of its frequency response; the phase response can be selected arbitrarily, if, of course, there does not appear here delay comparable with the interval of change of $\lambda(t)$. Therefore, the first method of finding h(t) is factoring function $|H(iw)|^2 = hS_0(w)/(1 + hS_0(w))$, i.e., its expansion into factors having zeroes and poles in the upper and lower half-places. respectively, of complex variable w. Then one of the factors will give the sough physically realizable frequency response. When there is a rational-fraction spectral den ity of fluctuations of signal $S_0(w)$ such a possibility, as shown in 5 6.5, always exists.

A second method is direct determination of the Pourier original of function $\sqrt{hS_0(\omega)/(1 + hS_0(\omega))}$, which will give a physically unrealizable pulse response, and introduction then according to the method in Paragraph 6.7.1 of time delay an interval comparable with the duration of this response. Obviously, it is also possible to suggest other methods. All of them are equivalent in results and give filters of an approximately identical transmission band.

The given "low-frequency" interpretation of operations of the discriminator is not the only possible one. Another possible variant is transfer of operations of filtration to a certain intermediate frequency, allowing us to decrease the number of channels in processing (see Chapter IV, Vol. 1):

 $z(t) = \operatorname{const} \int y(t_1) u_n(t_1; \hat{\lambda}(t)) \cos \left[(\mathbf{w}_0 + \mathbf{w}_{np}) t_1 + \\ + \phi(t_1; \hat{\lambda}(t)) \right] h(t - t_1) \cos \mathbf{w}_{np}(t - t_1) dt_1 \int y(t_2) \times \\ \times \left[\frac{\partial}{\partial \lambda} u_n(t_2, \hat{\lambda}(t)) \cos \left[(\mathbf{w}_0 + \mathbf{w}_{np}) t_2 + \phi(t_2; \hat{\lambda}(t)) \right] + \\ + u_n(t_2, \hat{\lambda}(t)) \frac{\partial}{\partial \lambda} \phi(t_2; \hat{\lambda}(t)) \sin \left[(\mathbf{w}_0 + \mathbf{w}_{np}) t_2 + \\ + \phi(t_2; \hat{\lambda}(t)) \right] h(t - t_2) \cos \mathbf{w}_{np}(t - t_2) dt_2.$

(6.2.32)

Processing of y(t) according to (6.7.32) should be conducted by the circuit "Fig. 6.30. At the input of the circuit there are installed two miners to which there are fed heterodyne voltages with quadrature phase shift and the expected phase modulation. In view of noncoincidence of the middle frequency of heterodyne signals with the carrier frequency ω_0 simultaneously with "convolution" of phase (frequency) modulation the mixers shift the spectrum of the signal to intermediate frequency $a_{\rm HD}$. Then signals are fed to variable-gain amplifiers. Laws of change of

-130-

amplification in them are $u_{k}(t, \hat{\lambda}) = \frac{\partial u_{k}(t, \hat{\lambda})}{\partial \lambda}$, $u_{k}(t, \hat{\lambda}) \frac{\partial \psi(t, \hat{\lambda})}{\partial \lambda}$, respectively, i.e., are determined by the form of regular modulation of the signal. The mixers together with controlled amplifiers carry out correlation processing. In the circuit of Fig. 4.20 there occurs unification of outputs of amplifiers in two channels. The mixers together to controlled by identical band pass filters, tuned to controlled frequency ω_{np} and being high-frequency equivalents of low-frequency optimum filters. Naturally, frequency to which this shift is equal, is arbitrary and in no way follows from theory. During transition to (6.7.32) we pursued the goal of practical convenience of fulfillment of operations, where there was implied the circumstance that after mixing in subsequent filtration there does not participate a "mirror" signal.

As a result of bandpass filtration in the channel connected with the amplifiers, whose law of amplification is determined by the derivative of the modulating functions, there will form voltage proportional in amplitude to current mismatch ε and



Fig. 6.30. Optimum discriminator for a fluctuating coherent signal (high-frequency variant); 1) quadrature mixers; 2) variablejoin amplifiers; 3) linear bandpass filters; 4) multipliers; 5) adder; 6) $\pi/2$ phase shifter. corresponding in phase to the sign of this difference. On the output of the other filter voltage depends little on ε and will form as it were a reference signal, multiplication of the differential signal wi' which, carried out in the phase detector, completes the operation of the discriminator. As a result there is formed low-frequency voltage z(t), on the average proportional to $\lambda = \hat{\lambda}$ for small values of this quantity.

Let us turn to characterization of accuracy of the discriminator. In the work the stationariness of fluctuations of the signal the accuracy is characterized by a constant $\overline{K(t)} = K$. In the most general case of a normal random process y(t)by differentiation of the logarithm of the likelihood function and averaging it is possible to show that $\overline{K(t)}$ is expressed directly through the correlation function

-131-

of the signal $R(t_1, t_2)$ and it inverse $W(t_1, t_2)$ in the form

$$K = -\lim_{r \to \infty} \frac{1}{2r} \int_{-r/2}^{+r/2} \frac{\partial \Psi(t_1, t_2; \lambda)}{\partial \lambda} \frac{\partial R(t_1, t_2; \lambda)}{\partial \lambda} dt_1 dt_2, \qquad (6.7.33)$$

where limits for constant λ can be directed to $\pm \varpi$ by virtue of the factors shown earlier.

Substituting in (6.7.33) functions $R(t_1, t_2)$ and $W(t_1, t_2)$ from (6.7.19) and (6.7.21), we obtain

$$K = \lim_{T \to \infty} \frac{1}{2T} \iint_{T/2}^{T} \left\{ u_a^2(t_1, 2) \left[\left(\frac{\partial u_a(t_1, \lambda)}{\partial \lambda} \right)^a + u_a^2(t_2, 2) \left(\frac{\partial 4(t_1, \lambda)}{\partial \lambda} \right)^2 \right] + u_a^2(t_2, 2) \left[\left(\frac{\partial u_a(t_1, \lambda)}{\partial \lambda} \right)^a + u_a^2(t_1, 2) \left(\frac{\partial 4(t_1, \lambda)}{\partial \lambda} \right)^2 \right] + u_a^2(t_1, 2) u_a^2(t_2, 2) \frac{\partial 4(t_1, \lambda)}{\partial \lambda} \times \left\{ \frac{\partial 4(t_2, \lambda)}{\partial \lambda} \right\} r(t_1 - t_2) w(t_1 - t_2) dt_1 dt_2.$$

$$(6.7.34)$$

Relationship (6.7.34) is valid in the very general case. If, however, on function $\psi(t, \lambda)$ there are imposed certain limitations, then this convenient formula, having more modest limits of applicability, is valid:

$$K = J_1(u) J_2(h, S_0),$$
 (6.7.35)

where

$$J_{1}(u) = b_{1} - b_{1}^{2};$$

$$b_{2} = \lim_{T \to \infty} \frac{1}{T} \int_{-T/2}^{+T/2} \left[\left(\frac{\partial u_{a}(t,\lambda)}{\partial \lambda} \right)^{a} + u_{a}^{2}(t,\lambda) \left(\frac{\partial \psi(t,\lambda)}{\partial \lambda} \right)^{a} \right] dt,$$

$$b_{1} = \lim_{T \to \infty} \frac{1}{T} \int_{-T/2}^{+T/2} u_{a}^{2}(t,\lambda) \frac{\partial \psi(t,\lambda)}{\partial \lambda} dt,$$

$$I_{a}(h, S_{0}) = \frac{1}{\pi} \int_{-T/2}^{+\infty} \frac{h^{2}S_{0}^{2}(\omega)}{1 + hS_{a}(\omega)} d\omega.$$
(6.7.37)

Returally, during periodic modulation passages to the limit in $(6.7.3^{\circ})$ and (6.7.36) are replaced by averaging over one period of modulation.

Let us discuss formula (6.7.35). It contains two factors, the first of which J_1 is wholly determined by the form of regular modulation, and the other J_2 is determined only by the form of spectral density of fluctuations of signal $S_0(\omega)$ and the signal-to-noise ratio h. In other words, J_1 is determined by the form of coding

 λ in the signal, and J_{p} is determined by power properties of the latter. Therefore, detailed discussion of J_{1} can be postponed to subsequent chapters. For J_{2} it is useful to indicate limiting values. For very large values of h we have

$$I_{a} \approx 2h \frac{1}{2\pi} \int_{-\infty}^{+\infty} S_{\bullet}(\omega) d\omega = 2h\Delta j_{c} \quad (h \ge 1), \qquad (6.7.38)$$

wrwery small h

$$J_{a} \approx 2h^{a} \frac{1}{2\pi} \int_{-\infty}^{+\infty} S_{0}^{2}(\omega) d\omega = 2h^{a}k\Delta f_{c} (h \ll 1), \qquad (6.7.39)$$

where κ - constant, value of which is between one (for fast-dropping spectrum $S_O(\omega)$) and 1/2 (for a slow-dropping spectrum).

Thus, J_2 monotonically depends on signal-to-noise ratio h and is proportional to the width of the spectrum of fluctuations of the signal Δf_{c} .

The condition of invariability of the correlation function within limits of the period of modulation, assumed above, is not always observed in practice. Separate periods can be weakly correlated. The sphere of applicability of the developed methods can be considerably expanded if we assume that modulation has a pulse character with a sufficiently high off-duty factor, so that instead of u(t) we take

$$\sum_{k} u(t - kT_{r}) = \sum_{k} u_{k}(t - kT_{r}) e^{i\psi(t - kT_{r})}, \qquad (6.7.40)$$

where u(t) - complex law of modulation in a separate k-th period.

Correlation of signal fluctuations, divided by the period of repetition, we consider arbitrary, but within the limits of duration of each pulse train (or pack) we shall consider the correlation function and its inverse function constant, so that, for instance,

$$\int_{0}^{T_{r}} |u(t-\tau)|^{2} r(t) dt = r(\tau) \int_{0}^{T_{r}} |u(t)|^{2} dt.$$

As before, there remains in force the condition that in the interval of correlation of the signal the measured parameter does not vary. The correlation function of the mixture of the signal with noise has the form

$$R(t_{1}, t_{2}) = Re\left\{\sum_{k, j} u(t_{1} - kT_{r}) r(t_{1}, t_{2}) u(t_{2} - jT_{r})\right) e^{j\omega_{0}(t_{1} - t_{2})}\right\} + -\left| N_{0} f(t_{1} - t_{2})\right].$$
(6.7.41)

Function $W(t_1, t_2)$, the inverse of the correlation function in interval Δ , including many periods of repetition and intervals of correlation of the signal, we find in the form

-133-

$$W(t_{1}, t_{2}) = -\operatorname{Re}\left\{\sum_{k, j} u(t_{1} - kT_{r})^{\frac{w(t_{1}, t_{2})}{N_{0}^{2}}} u(t_{2} - jT_{r}) e^{j\omega_{0}(t_{1} - t_{2})}\right\} + \frac{1}{N_{0}}\delta(t_{1} - t_{2}).$$
(6.7.42)

This gives an equation for $w(t_1, t_2)$:

$$\frac{1}{2} \int_{t-h/2}^{t+h/2} w(t_1, s) \frac{\sum_{i} u^2(s-iT_i)}{N_0} r(s-t_2) ds + w(t_1, t_2) = -r(t_1-t_2). \qquad (6.7.43)$$

Considering the condition of a large off-duty factor, it is possible to produce integration over s, after which it is user'ul to quantize equation (6.7.43), i.e., to look for its solution $w(iT_r, jT_r) = w_{i,j}$ at discrete points coinciding with moments of action of pulses, since only at these points will it affect the form of the final solution of (6.7.42). This gives a discrete analog of equation (6.7.22):

$$\frac{T_r}{2N_{\bullet}} \sum_{i} w_{ij} r_{j-1} + w_{il} = r_{i-1}, \qquad (6.7.44)$$

م محمد نوم الخ من الح مح<mark>لية معض مع مارك المسالم المسالم المسالم المسالم المحارك المارك مسالم المسالم من مع الم</mark>

where it is assumed that the energy of pulses does not depend on time, and condition (6.7.27) is satisfied. Directing the limits of summation in (6.7.44) to infinity and seeking a solution in the form $w_{1l} = w_{1-l}$, it is easy to solve (6.7.44) by the method of discrete Fourier transformation:

$$\widetilde{w}(\Omega) = \frac{S_{cD}(\Omega)}{1 + \frac{T_r S_{cD}(\Omega)}{2N_{\bullet}}} = \sum_{k=-\infty}^{+\infty} r_k \frac{S_{vD}(\Omega)}{1 + h_D S_{0D}(\Omega)}.$$
(6.7.45)

Here

$$h_{D} = \sum_{k=-\infty}^{+\infty} r_{k} \frac{T_{r}}{2N_{\bullet}}$$
(6.7.46)

- signal-to-noise ratio, analogous to h and changing into h as $T_r \rightarrow 0$, insemuch as, here,

$$h_D \simeq \frac{\int_{-\infty}^{+\infty} r(\mathbf{r}) d\mathbf{r}}{2N_{\bullet}} = \frac{S_{\bullet}(0)}{2N_{\bullet}} = h; \qquad (6.7.47)$$

 Ω - dimensionless frequency, connected with usual frequency and the period of repetition by relationship

$$\Omega = \bullet T_r;$$

-134-

then,

()

$$S_{eD}(\Omega) = \sum_{k=-\infty}^{+\infty} r_k e^{-ik\Omega}, \qquad (6.7.48)$$
$$\widetilde{w}(\Omega) = \sum_{k=-\infty}^{+\infty} w_k e^{-ik\Omega}$$

- . screte Fourier transforms of the correlation function and the inverse function, and

$$S_{00}(\Omega) = \frac{S_{cD}(\Omega)}{S_{cD}(0)} = \frac{S_{cD}(\Omega)}{\sum_{k} r_{k}}$$

- normalized discrete spectrum.

Let us expand w_i analogously to w(t) in (6.7.29):

$$\frac{T_{\bullet}}{2N_{\bullet}}\widetilde{w}(\Omega) = \frac{h_D S_{0D}(\lambda)}{1 + h_D S_{0D}(\lambda)} = |H_D(i\Omega)|^{\bullet}$$

$$\frac{T_{\bullet}}{2N_{\bullet}} w_{i-j} = \sum_{i} h_{i-1}h_{j-1},$$
(6.7.49)

where $h_{\rm i}$ – a certain function of a discrete argument with Fourier transform ${\rm H}_{\rm D}(1\Omega)$.

Then, considering for simplicity the case when the signal-to-noise ratio h_D does not depend on parameter λ , we have a discrete analog of relationship (6.7.31):

$$\begin{aligned} z(jT_{r}) &= \frac{T_{r}}{N_{o}} \sum_{i} h_{j-i} \int_{(i-1)}^{H_{r}} y(t_{i}) u_{a}(t_{1} - iT_{r}; \hat{\lambda}_{j}) \cos\left[m_{o}t_{1} + \\ &+ \psi(t_{1} - iT_{r}; \hat{\lambda}_{j})\right] dt_{1} \sum_{i} h_{j-i} \int_{(i-1)}^{H_{r}} y(t_{2}) \left\{ \frac{\partial}{\partial \lambda} u_{a}(t_{2} - iT_{r}; \hat{\lambda}_{j}) \right\} \\ &\times \cos\left[m_{o}t_{2} + \psi(t_{2} - iT_{r}; \hat{\lambda}_{j})\right] - u_{a}(t_{2} - iT_{r}; \hat{\lambda}_{j}) \frac{\partial}{\partial \lambda} \psi(t_{2} - \\ &- iT_{r}; \hat{\lambda}_{j}) \sin\left[m_{o}t_{a} + \psi(t_{a} - iT_{r}; \hat{\lambda}_{j})\right] \right\} dt_{a} + \\ &+ \frac{T_{r}}{N_{o}} \sum_{i} h_{j-i} \int_{(i-1)}^{H_{r}} y(t_{i}) u_{a}(t_{1} - iT_{r}; \hat{\lambda}_{j}) \sin\left[m_{o}t + \\ &+ \psi(t_{1} - iT_{r}; \hat{\lambda}_{j})\right] dt_{1} \sum_{i} h_{j-i} \int_{(i-1)}^{H_{r}} y(t_{2}) \left\{ \frac{\partial}{\partial \lambda} u_{a}(t_{a} - \\ &- iT_{r}; \hat{\lambda}_{j}) \sin\left[m_{o}t_{a} + \psi(t_{2} - iT_{r}; \hat{\lambda}_{j})\right] + u_{a}(t_{a} - iT_{r}; \hat{\lambda}_{j}) \right\} \\ &\times \frac{\partial}{\partial \lambda} \psi(t_{a} - iT_{r}; \hat{\lambda}_{j}) \cos\left[m_{o}t_{a} + \psi(t_{2} - iT_{r}; \hat{\lambda}_{j})\right] dt_{a}. \end{aligned}$$

$$(6.7.50)$$

receiving of the signal according to (6.7,50) obviously breaks down into an interperiod and an interperiod part. The intraperiod processing completely repeats the first correlation part of processing in Fig. 6.29. Then, instead of low-frequency

filters there should follow integrators with drop, accumulating signals invice each period. Results of accumulation in discrete form are smoothed by discrete lines filters, realizing accumulation in the interval of fluctuations of the right. Technically, a circuit starting with these filter can be imagined in the form of a specialized digital computer. Further processing, consisting in multiplication the results of filtration and addition with each other, accurately repeats the operation of Fig. 6.29.

As before, let us assume transfer of operations (0.7.49) to intermediate frequency. This leads to a relationship, analogous to (6.7.32):

$$z(jT_{r}) = \operatorname{const} \int_{-\infty}^{+\infty} y(t_{1}) u_{a}(t_{1}, \hat{\lambda}_{j}) \cos \left[(v_{0} + \omega_{\pi\mu}) t_{1} + \frac{1}{2} \psi(t_{3}, \hat{\lambda}_{j}) \right] h(t_{j} - t_{1}) \cos \omega_{\mu\nu}(t_{j} - t_{1}) dt_{1} \int_{-\infty}^{+\infty} y(t_{3}) \times \left[\frac{\partial}{\partial \lambda} u_{a}(t_{2}, \hat{\lambda}_{j}) \cos \left[(w_{0} + \omega_{\pi\nu}) t_{2} + \psi(t_{2}, \hat{\lambda}_{j}) \right] + \frac{1}{2} + u_{a}(t_{3}, \hat{\lambda}_{j}) \frac{\partial \psi(t_{3}, \hat{\lambda}_{j})}{\partial \lambda} \sin \left[(\omega_{0} + \omega_{\pi\nu}) t_{3} + \psi(t_{2}, \hat{\lambda}_{j}) \right] \right] \times h(t_{j} - t_{3}) \cos \omega_{\mu\nu}(t_{j} - t_{2}) dt_{3}, \qquad (1.11)$$

where the envelope of pulse response h(t) at discrete moments $t = iT_p$ cathering condition (6.7.48), and in the remaining moments inside the period is arbitrary. The block diagram describing relationship (6.7.51) simply repeats the diagram of Fig. 6.30.

We shall show that, although operations (e.7.50) and (6.7.51) also back new into intra- and interperiod operations, interperiod processing of the shown form cannot be transferred to the output of the discriminator, inasmuch as phase detertion (multiplication) at the output is an operation which is especially nonlinear w⁽¹⁾ respect to y(1).

For characterization of accuracy of measurement K in the same condition. as in case (6.7.35) we have expression

$$\mathbf{K} = J_1(\mathbf{u}) J_{2\mathbf{p}}(\mathbf{h}_{\mathbf{p}}, \mathbf{S}_{\mathbf{p}\mathbf{p}}), \qquad (v, 7, ve)$$

where factor J_1 has the former value (6.7.36), and $J_{\rm 2D}(h_D, S_{\rm OD})$ is equal to

$$J_{2D} = \frac{1}{\pi T_r} \int_{-\pi}^{+\pi} \frac{h_D^2 S_{0D}^2(\Omega)}{1 + h_D S_{0D}(\Omega)} d\Omega.$$
 (0.11)

As factor J_2 from (6.7.37), J_{2D} monotonically depends on signal-to-noise ratio h_D . In the extreme case, when the correlation coupling between packages is very great, the range of values of Ω , where function $S_{0D}(\Omega)$ markedly differs from zero, is considerably less than interval $(-\pi, +\pi)$. Therefore, considering relationship (6.7.46), and also the fact that $\Omega = \omega T_r$ and $S_{0D}(\omega T_r) \rightarrow S_0(\omega)$ and directing to infinite moduling the integration in (6.7.53), we note that this relationship (6.7.37). The other extreme case of total absence of correlation will be considered in detail in the next paragraph.

Above we considered the case of one input mixture of signal with noise y(t). Let us try to generalize the preceding results to the case of many input mixtures $\binom{l}{y}(t)$ (l = 1, 2, ..., m), each of which can contain a whole series of components in general with different forms of modulation and correlation properties. An example of such a set of mixtures are outputs of a multiunit antenna system, in each of value there are included signals reflected from many targets and radiated by interference sources.

Here, carrier frequencies for all components of the mixture may differ. There is introduced a certain middle frequency ω_0 , and phase modulations $(1)\psi_k(t; \lambda)$ contain terms of form $(\omega_{0k} - \omega_0)t$, linearly increasing in time, reflecting differences of frequencies of separate components ω_k .

Let us assume the 1-th mixture has the form

We introduce a complex nearly-diagonal matrix of coefficients of modulation U(t), the *i*-th diagonal element of which is column $\binom{(l)}{u_1(t)} \dots \binom{(l)}{p_l} (t)^{j}$, where $\binom{(l)}{u_1(t)} = \binom{(l)}{u_{aj}(t)} e^{\binom{(l)}{y_j(t)}}$. Additionally we introduce complex column $\theta(t)$, the *i*-th subcolumn element of which is a column of complex elements

$${}^{(l)} \theta = \{{}^{(l)} E_1(l) e^{i(l)\varphi_1(l)}, \dots, {}^{(l)} E_{P_l}(l) e^{i(l)\varphi_{P_l}(l)}\},$$

reflecting random modulations. Then the whole set of mixtures (6, 8, 62) can be presented in matrix form

$$\dot{\mathbf{y}}(l) = \operatorname{Re} \{ \mathbf{U}^{+}(l) \, \mathbf{i}(l) \, \mathbf{e}^{l \cdot \mathbf{v}^{+}} \} + \mathbf{n}(l), \qquad (6.7.55)$$

-137-

where $\mathbf{y}(t)$, $\mathbf{n}(t)$ - simple column vectors of order (m × 1), composed of values of function ${}^{(i)}\mathbf{y}(t)$ and ${}^{(i)}\mathbf{n}(t)$ (i = 1, ..., m). The correlation matrix of the whete set of mixtures (6.7.55) is obtained in the form

$$\mathbf{R}(t_1, t_2) = \overline{\mathbf{y}(t_1) \mathbf{y}^+(t_2)} = \operatorname{Re} \{\mathbf{U}^+(t_1) \mathbf{r}(t_1 - t_2) \times \mathbf{U}^+(t_2) \mathbf{e}^{t_{\mathbf{u}_0}(t_1 - t_2)}\} + \operatorname{N\delta}(t_1 - t_2),$$

where

$$\mathbf{r}\left(t_{1}-t_{s}\right)=\frac{1}{2} \ \overline{\mathbf{0}^{*}\left(t_{1}\right)\mathbf{9}^{+}\left(t_{s}\right)}$$

- complex matrix function of order $\left(\sum_{i=1}^{m} p_i \times \sum_{i=1}^{m} p_i\right)$ with clements being substitute $(ij)_{\mathbf{r}(t_1 - t_2)} = \left| \left| {(ij)_{\mathbf{r}_{pq}}(t_1 - t_2)} \right| \right|$, where $r_{pq}(t_4 - t_2)$ is the function of each $r_{pq}(t_1 - t_2)$.

correlation of stationary random processive $\frac{(t)}{E_{t}} \left(\frac{t}{2}\right) \frac{\cos^{2}(1)}{\sin^{2}} \varphi_{F}(1)$ and

 $(\mathbf{j})_{\mathbf{E}_{\mathbf{q}}}(\mathbf{t}_{2})_{\sin}^{\cos(\mathbf{j})} \mathbf{\varphi}_{\mathbf{q}}(\mathbf{t}_{2})$. Matrix **N** in (6.7.57) is the matrix of spectral densities of white noises, not necessarily diagonal, ince white noises in channels may be interrelated. The likelihood functional of the whole set of mixture $\mathbf{y}(t)$ has a form analogous to that in the one-dimensional course:

$$P(\mathbf{y} \mid \boldsymbol{\lambda}) = C \exp \left\{ -\frac{1}{2} \liminf_{\boldsymbol{\lambda} \neq 0} \operatorname{Int} \operatorname{In} \det \mathbf{R} - \frac{1}{2} \iint_{\boldsymbol{\lambda} \neq 0} y^{+}(t_{i}) W(t_{i}, t_{i}) y(t_{i}) dt_{i} dt_{i} \right\},$$

where $W(t_1, t_2)$ - matrix, the inverse of $R(t_2, t_2)$, i.e., satisfying equation

$$\int_{t=A/2}^{t+A/2} W(t_1, s) R(s, t_1) ds = 1 \delta(t_1 - t_1). \qquad (t_2 + t_1)$$

We shall look for $W(t_1, t_2)$ in a form which is a multi-dimensional generalization of (6.7.21):

Then from relationships (6.7.56), (6.7.58) and (6.7.59) we arrive t^{-1} is following integral matrix equation for matrix function $\mathbf{w}(t_1, t_2)$, structurally similar to $\mathbf{r}(t_1, t_2)$:

$$\frac{1}{2}\int_{t-\Delta/2}^{t+\Delta/2} \mathbf{r}\left(t_{1}-s\right) \mathbf{B}_{\mathbf{0}}\left(s\right) \mathbf{w}\left(s, t_{s}\right) ds + \mathbf{w}\left(t_{1}, t_{s}\right) = \mathbf{r}\left(t_{1}, t_{s}\right). \qquad (t_{1}, t_{s})$$

Here $\mathbf{B}_{\mathbb{Q}}(s) = \mathbf{U}^{*}(s)\mathbf{N}^{-1}\mathbf{U}^{+}(s) - \operatorname{complex matrix}$, the character of whose dependence on s is determined by the form of regular modulation of all components of the mixture, not only of amplitude, but also of phase (frequency) modulation. Comparison of equations (6.7.60) and (6.7.21) shows their very complete analogy, from which there ensue the same cases of solutions as in the one-dimensional variant.

Considering, in particular, laws of modulation of all separate components of the mixture to be rapidly varying, it is possible to average $B_0(z)$ in time under the risk integral, introducing constant matrix

$$\mathbf{B}_{s} = \lim_{T \to \infty} \frac{1}{T} \int \mathbf{U}^{*}(s) \, \mathbf{N}^{-1} \mathbf{U}^{+}(s) \, ds. \qquad (6.7.61)$$

Then the Fourier transform of $w(t_1, t_2) = w(t_1 - t_2)$ will be expressed in the form

$$\widetilde{\mathbf{w}}(\boldsymbol{\omega}) = \left[\mathbf{1} + \frac{1}{2} \mathbf{S}_{\boldsymbol{\omega}}(\boldsymbol{\omega}) \widetilde{\mathbf{B}}_{\boldsymbol{\omega}}\right]^{-1} \mathbf{S}_{\boldsymbol{\omega}}(\boldsymbol{\omega}), \qquad (6.7, 62)$$

where $\mathbf{S}_{i}(x)$ - Fourier transform of matrix function $\mathbf{r}(t)$.

On the assumption that power characteristics of signals do not depend on 2, we take to it the elecation of the discriminator in the following turm, generalizing formula (1.7.52):

$$\begin{aligned} \mathbf{z}(t) &=: A \sum_{i_{1} \neq i_{1}}^{m} \sum_{k=1}^{p_{i_{1}}} \sum_{i_{2} \neq i_{1}}^{f} \sum_{i_{2}}^{(i)} \sum_{i_{2}}^{i} \sum_{i_{2} \neq i_{2}}^{(i)} \sum_{i_{2}}^{i} \sum_{i_{2} \neq i_{2}}^{(i)} \sum_{i_{2}}^{(i)} \sum_{i_{2} \neq i_{2}}^{(i)} \sum_{i_{2}$$

Here A - proportionality fmetors

1

ł

ł

$(i)_{\mathcal{K}_{h}}(l, \lambda) = (i)_{\mathcal{Y}}(l)^{(i)}_{\mathcal{H}_{hh}}(l, \lambda) \cos \left[(\omega_{n} + \omega_{hp}) l + (i) \psi_{h}(l, \lambda) \right] -$

condition conversions processing of the t-th mixture for the purpose of separation from it the k-th component of the signal, and envelopes of pulse responses $(4d)_{h_{k_{1}}}^{(1)}(r)$ and $(1)_{h_{k_{2}}}^{(2)}(r)$ and $(1)_{h_{k_{2}}}^{(2)}(r)$

$$\int_{-\infty}^{+\infty} (i) h_{kl}^{(1)}(l) e^{-lwl} dl \int_{-\infty}^{+\infty} (i) h_{kl}^{(1)}(l) e^{lwl} dl = (i) \widetilde{W}_{kl}(w), \qquad (i), \forall i \in \mathbb{N}$$

where $\frac{(1+i)}{w_{RI}}(x) = \text{element}(x)$ complex matrix w(x), determined by relationship (x_i, x_i, z) .

we could that the Fourier transform of a metric is the matrix of konsier transform of any of separate elements of the initial matrix,

Examples of optimum processing of several input signels will be given in subsequent chapters, and in view of the complexity of general case $(0.7, 63) \approx 0.3$ description of the circuit of the discriminator have.

For K(t) in the most general case of a set of normal processes with norm with matrix $R(t_2, t_2)$ and its inverse $W(t_1, t_2)$ there is a formula, analogous for , .25 μ

$$\mathcal{K} = -\lim_{T \to \infty} \frac{1}{2T} \iint_{-T/2}^{+T/3} \sup \left\{ \frac{\partial}{\partial \lambda} W(t_1, t_2) \frac{\partial}{\partial \lambda} R^+(t_1, t_2) \right\} dt_1 dt_2, \qquad (1 - t_1)$$

÷

)

where "spar" signifies the trade of the masriz.

In the considered case this formula, tasks is a contrability i^{2} , i^{2} (6.7.59), leads to this expression is. colliding (6.7.59):

$$\mathcal{K} = \lim_{T \to \infty} \operatorname{Re} \frac{1}{2T} \int_{-T/2}^{+T/2} \operatorname{spur} \{ \mathbf{B}_{s}(t_{1}) \le (t_{1} - t_{1}) \ \mathbf{B}_{0} \operatorname{r} (t_{1} - t_{1}) + \\ + \mathbf{B}_{i}(t_{1}) \le (t_{1} - t_{1}) \ \mathbf{B}_{i}^{*}(t_{1}) \operatorname{r} (t_{1} - t_{1}) \} dt_{i} dt_{1},$$

$$(0., 7, 0.0)$$

朝守した

. .

$$\mathbf{B}_{\mathbf{s}}(t) = = \frac{\partial}{\partial \lambda} \mathbf{U}^{\mathbf{s}}(t) \mathbf{N}^{-1} \frac{\partial}{\partial \lambda} \mathbf{U}^{+}(t); \qquad (1.7777)$$
$$\mathbf{B}_{\mathbf{s}}(t) = = \mathbf{U}^{\mathbf{s}}(t) \mathbf{N}^{-1} \frac{\partial}{\partial \lambda} \mathbf{U}^{+}(t)$$

— motified structurely similar to matrix $\mathbf{\hat{B}}_{12}$. In the rest of similar

then, taking into a count (1.7.15) we sharing constanting (6.7.66)

$$\mathcal{K}_{1}=\frac{1}{4\pi}\int_{-\infty}^{+\infty} \operatorname{spur}\left\{\left[1+\frac{1}{2}S_{\omega}(\omega)\widetilde{B}_{\varepsilon}\right]^{-1}S_{\sigma}(\omega)[\widetilde{B}_{\varepsilon}S_{\varepsilon}^{+}(\omega)\widetilde{B}_{\varepsilon}+\widetilde{B}_{\varepsilon}S_{\varepsilon}^{+}(\omega)\widetilde{B}_{\varepsilon}]\right\}d\omega.$$

bet us consider a simple example. Let us assume that every mixture contains only the signal component $(p_j \neq 4)$ with power P_{c_j} , where these some nexts are completely correlated, and white makes are independent, as that elements of matrices B_c and N have form

$$(ih)S_{\bullet}(\boldsymbol{\omega}) = \frac{\sqrt{P_{\bullet}/P_{eh}}}{\Delta_{\bullet}} S_{\bullet}(\boldsymbol{\omega}), \quad (ih)N = N_{\bullet}\delta_{ih}, \quad (i,i,j)$$

where $\Delta t_{\rm G}$, $t_{\rm G}$ - unique width of the spectrum is spectrum interactions of signal one spectral density of white notices.

--140-

In this simple case it is convenient to introduce column vector $\mathbf{p} = \{ V P_{e_1}, \dots, V P_{e_m} \}$, as a result of which it is possible to express the matrix of spectral densities of signals in the form

$$S_{c}(\omega) = \frac{S_{o}(\omega)}{\Delta f_{c}} pp^{+}.$$

Therefore \overline{B}_0 , \overline{B}_1 , \overline{B}_2 here are diagonal. \overline{B}_0 consists of elements $1/N_0$, inasmuch the force of (6.7.70) it is necessary to consider that for all forms of regular modulation this normalization is executed

$$\lim_{T\to\infty}\frac{1}{T}\int_{-T/2}^{+T/2}|(i)u(t)|^{a} dt = 1,$$

and diagonal elements of \overline{B}_{2} , \overline{B}_{1} contain elements

$$\frac{1}{N_{\bullet}} \lim_{T \to \infty} \frac{1}{T} \int_{-T/2}^{+T/2} \frac{\partial^{(1)}u(t,\lambda)}{\partial \lambda} \Big|^{*} dt =$$

$$= \frac{1}{N_{\bullet}} \lim_{T \to \infty} \frac{1}{T} \int_{-T/2}^{+T/2} \left(\frac{\partial^{(1)}u(t,\lambda)}{\partial \lambda} \right)^{*} + {}^{(1)}u_{a}^{2}(t,\lambda) \left(\frac{\partial^{(1)}\psi(t,\lambda)}{\partial \lambda} \right)^{*} \right)^{*} dt,$$

$$\frac{1}{N_{\bullet}} \lim_{T \to \infty} \frac{1}{T} \int_{-T/2}^{+T/2} {}^{(1)}u(t,\lambda) \frac{\partial^{(1)}u^{*}(t,\lambda)}{\partial \lambda} dt =$$

$$= \frac{1}{N_{\bullet}} \lim_{T \to \infty} \frac{1}{T} \int_{\bullet}^{T} \left\{ {}^{(1)}u_{a}^{2}(t,\lambda) \frac{\partial^{(1)}\psi(t,\lambda)}{\partial \lambda} \right\} dt. \qquad (F.7.71)$$

Finding in these conditions matrix $\left[1+\frac{1}{2}S_{c}(m)\times\overline{B}_{0}\right]^{-1}S_{c}(m)$ in the form App⁺, we have

$$A == \frac{S_{\bullet}(\omega)}{\Delta f_{\bullet}} \frac{1}{1 + h_{\Sigma} S_{\bullet}(\omega)},$$

$$h_{\rm I} = \sum_{k=1}^{m} P_{\rm ok}/2N_{\rm o}\Delta f_{\rm c}. \qquad (6.7.72)$$

「おいい」「「「「「「」」」」」

Finally, by formula (6.7.69), taking into account relationships (6.7.71), we obtain

$$K = J_1(U) J_1(h_1, S_0).$$
 (6.7.73)

Here

WHIFTO

 $J_1(U) = b_1 - b_1^2$,

$$b_{a} = \lim_{T \to \infty} \frac{1}{T} \int_{-T/2}^{+T/2} \sum_{k=1}^{m} a_{k} \left[\left(\frac{\partial^{(k)} u_{a}(t,\lambda)}{\partial \lambda} \right)^{2} + {}^{(k)} u_{a}^{2}(t,\lambda) \times \left(\frac{\partial^{(k)} \psi(t,\lambda)}{\partial \lambda} \right)^{a} \right] dt,$$

$$\sum_{k=1}^{+T/2} \lim_{T \to \infty} \frac{1}{T} \int_{-T/2}^{+T/2} \sum_{k=1}^{m} a_{k}^{(k)} u_{a}(t,\lambda) \frac{\partial^{(k)} \psi(t,\lambda)}{\partial \lambda} dt,$$

$$a_{k} = P_{ck} / \sum_{l=1}^{m} P_{cl},$$

and factor

$$J_{\mathbf{A}}(h_{\mathbf{z}}, S_{\mathbf{o}}) = \frac{1}{\pi} \int_{-\infty}^{+\infty} \frac{h_{\mathbf{z}}^2 S_0^2(\omega)}{1 + h_{\mathbf{z}} S_{\mathbf{o}}(\omega)} d\omega$$

repeats J_2 from relationship (6.7.37) with replacement of the signal-to-noise ratio h by quantity h_r , given by formula (6.7.72).

According to (6.7.73) and (6.7.74) the general form of quantity K remains the same as in the case of one input mixture, but in factor $J_1(0)$, determined by define of $\lambda(t)$ in regular modulations of the mixture, there is performed averaging of analogous coefficients for separate mixtures taking into account signal powers, and in factor J_2 there appears a new signal-to-noise ratio, which, in the of total correlation of useful signals in separate mixtures, is proportional to the ratio of total power of these signals to the spectral density of noise in one channel. This indicates that potential accuracy does not expend on the number of channels with identical noises levels into which total signal power is divided, if we concider the latter constant. With the same success we can consider more complicated examples.

Inverse matrix $W(t_1, t_2)$ can be effectively found also for pulse signals. The case of referrelated pulses we will consider in Paragraph 6.7.3.

Presentation of input mixtures in the form of (4,7,54) has a wide the standard application. In particular, as separate components there can be considered correlated unmodulated interferences, which on we touched in Paragraph 6.7.1. B., newsfer, correlated interferences have a complication form of modulation, which it is impossible to present in the form of (6.7.54) (for instance, passive interferences), then finding matrix functions $W(t_1, t_2)$, the interval of correlation functions, is complicated. Here methods developed in Chapter 4 for problems of detection may help.

All results given in this paragraph will be illustrated by numerous examples in subsequent chapters.

-142-

6.7.3. Independently Fluctuating Sendings. General Case of an Incoherent Signal

We shall now discuss in detail the case of independently fluctuating sendings of $\gamma_{\rm parse}$ signal. It is the limiting case both for coherent, and for incoherent callation, if the interval of fluctuations of the signal turns out to be less than the veriod of repetition.

Fassing in (6.7.45) to the limit $S_0(\Omega) \rightarrow 1$ and introducing the new signal-tonoise ratio

$$q = \lim_{s \ (\mathbf{R}) \to 1} h_D = P_c T_r / 2N_{e_1}$$

we have the following expression for the likelihood coefficient of a series of n sendings:

$$\Lambda(\lambda, \mathbf{y}) = \frac{1}{(1+q)^n} \prod_{k=1}^n \exp\left\{\frac{q}{1+q} \frac{|f_k(\mathbf{y}, \lambda)|^2}{N_e T_r}\right\}, \quad (5.7.76)$$

where

$$f_{k}(\mathbf{y}, \boldsymbol{\lambda}) = \int \boldsymbol{y}(t) \, \boldsymbol{\mu}(t - kT_{\tau}; \, \boldsymbol{\lambda}) \, \mathrm{e}^{i \boldsymbol{\omega}_{\mathbf{y}} t} \, dt$$

- cocrelation integral, components of which we already met above.

Thus, (6.7.76) is broken up into the product of n independent likelihood coefficients for each of the periods. Therefore, according to § 6 for analysis of elementary operations of primary processing it is sufficient to study the likelihood coefficient of a single sending

$$\mathbf{A}(\boldsymbol{\lambda}, \mathbf{y}) = \frac{1}{1+q} \exp\left\{\frac{q}{1+q} \frac{|f(\mathbf{y}, \boldsymbol{\lambda})|^2}{N_0 T_r}\right\}.$$
 (6.7.77)

Insimuch as in a wide class of cases intraperiod processing includes forming of two components of quantity $f(y, \lambda)$, it is useful to definitize the method of thi forming and to explain the physical meaning of $f(\mathbf{y}, \lambda)$ and $|f(\mathbf{y}, \lambda)|^2$.

According to Fig. 6.31 for formation of two components of $f(\mathbf{y}, \lambda)$ input signal $\mathbf{y}(t)$ should be fed to two mixers with quadrature shift of high-frequency neterodyne voltages fed to them. Phase (frequency) modulation of the heterodyne signal should correspond to the expected, and the middle frequency should be exactly equal to the frequency of the received signal. Additionally there is produced multiplication of signal by the expected form of amplitude modulation, i.e., gating by a matched gate pulse. Then there is produced integration of the results of processing within the

-143-



Fig. 6.31. Optimum receiver of an incoherent signal (correlation processing): 1) $\pi/2$ phase shifter; 2) quadrature mixers; 3) gated amplifiers; 4) integrators; 5) quadrature devices.

limits of a period (actually only within limits of duration of the sending). As already mentioned in Faragraph 6.7.2, such processing is calles correlation, inasmuch as in the circuit there will be formed the coefficient of correlation between the received realization and the expected form of the signal.

Shift to $|f(\mathbf{y}, \lambda)|^2$ is carried out by squaring and addition of the two formed components of the correlation integral.

For the purpose of simplifying operations of forming $|f(\mathbf{y}, \lambda)|^2$ we shall consider an expression for the output voltage of a certain bandpass filter with pulse response $h(t) \cos \omega_n t$, to whose input there is fed signal y(t). Then we have

$$\int y(\tau) h(t-\tau) \cos \alpha_0 (t-\tau) d\tau = \int y(\tau) h(t-\tau) \cos \alpha_0 \tau d\tau \times$$

$$\times \cos \alpha_0 t + \int y(\tau) h(t-\tau) \sin \alpha_0 \tau d\tau \sin \alpha_0 t =$$

$$= \sqrt{\left[\left[y(\tau) h(t-\tau) \cos \alpha_0 \tau d\tau \right]^2 + \left[\int y(\tau) h(t-\tau) \sin \alpha_0 \tau d\tau \right]^2 } \times$$

$$\times \sin \left[\alpha_0 t + \chi(t) \right],$$

$$(c.7.78)$$

$$ig \chi(t) = \frac{\int y(\tau) h(t-\tau) \sin \alpha_0 \tau d\tau}{\langle x(t) h(t-\tau) \sin \alpha_0 \tau d\tau]}$$

where

Comparison of the expression for $|f(\mathbf{y}, \lambda)|$ with the signal amplitude envelope, determined by relationship (6.7.78), shows that in the case of absence of regular phase (frequency) modulation $(\psi(t, \lambda) = 0)$ quantity $|f(\mathbf{y}, \lambda)|$ is proportional the envelope of output voltage of a bandpass filter with pulse response

$$t(1)\cos \omega_{s} t = u_{a}(t_{o} - t)\cos \omega_{s} t.$$
 (6.7.79)

Such a filter in the literature usually has the name optimum, matched or conjugate filter. The last two terms seem more suitable to us, inasmuch as we arrive at linear filters in a number of other optimizing problems (Wiener filters, see Paragraph 6.5.5, filters for accumulation of fluctuations of signal, see Paragraph 6.7.2).

The square of the signal envelope at the output of a matched filter is taken

-144-**

at moment t_0 , on the average corresponding to the fax mum value of the pulse envelope. t_0 , interted in the pulse response, is arbitrary and can be selected in an interval of the order of pulse duration $u_a(t)$ for conformity to the principle of physical realizability. Considering (6.7.79), we can rewrite $|f(y, \lambda)|^2$ in the form

$$|f(\mathbf{y}, \mathbf{\lambda})|^{2} = \int_{-\infty}^{+\infty} \left[\int_{t_{r}}^{t_{r}} y(\mathbf{x}) h(t - \mathbf{x}; \mathbf{\lambda}) \cos \omega_{0}(t - \mathbf{x}) d\mathbf{x} \right]^{2} \times \mathbf{y}(t - \mathbf{k}T_{r} - t_{0}) dt, \qquad (6.7.80)$$

where v(t) - function in the form of a single peak near point t = 0 with duration considerably less than duration of pulse $u_a(t)$, but embracing a large or an integral number of periods of high-frequency filling of the signal. By this function, similar in properties to a Dirac 5-function, there is "snatched-off" the tip of the filtered and detected sending.

The block diagram of an optimum receiver, issuing quantity $|f(\mathbf{y}, \lambda)|^2$ according to relationship (6.7.90), is shown in Fig. 6.32 and consists of a matched filter, a square-law detector and a gated amplifier. Processing of the signal, expressed in



Fig. 5.32. Optimum receiver of an incoherent signal (filtration processing): 1) matched (shortening) filter; 2) square-law detector; 3) gated amplifier.

form (6.7.80), usually carries in the literature the name of filtration. If the signal has phase (frequency) modulation, then next in the filtration method of processing will be insertion of

a so-called shortening filter with a pulse response which is the inversion in time of the expected form of the phase-modulated signal with arbitrary initial phase of its high-frequency filling φ . This leads to a somewhat complicated modification of relationship (6.7.79):

$$h(t)\cos[u_{0}t + \xi(t)] = u_{0}(t_{0} - t)\cos[u_{0}t + \psi(t_{0} - t) + \varphi], \qquad (6.7.81)$$

where $\xi(t)$ - phase modulation.

To technically realize such a filter with complex forms of modulation is sometimes difficult [66]. Then, more preferable will be mixed correlation-filtration processing, which during formation of $|f(\mathbf{y}, \lambda)|^2$ consists of initial multiplication of $\mathbf{y}(t)$ by heterodyne signal cos $[(\omega_0 + \omega_{\Gamma,P})t + \psi(t_\lambda \lambda)]$, leading to "convolution" of phase (frequency) modulation and transfer of the spectrum of the signal to a certain intermediate frequency $\omega_{\rm HP}$ analogously to how this was done in Faragraph 6.7.2. Then there is produced filtration of the signal by a bandpass filter tuned to frequency $\omega_{\rm HP}$, whose pulse response envelope depicts only the amplitude modulation of the sending according to relationship (6.7.79), detection and gating. A circuit for such processing is presented in Fig. 6.37. Let us emphasize that correlation, filtration and mixed processings, conducted optimally, are abcolutely equivalent in results during detection and measurement of parameters of a single sending.

If q does not depend on λ , the signal at the output of the discriminator is directly the derivative of $|f(\mathbf{y}, \lambda)|^2$, taken for the measured value of $\hat{\lambda}$. Forming of this derivative, as also of $f(\mathbf{y}, \lambda)$, can be produced by correlation, filtration or mixed methods.

In Fig. 6.34 there is presented a purely correlation method, which according to (6.7.77), can be expressed by the following formula:

$$\begin{aligned} z (kT_{r}) &= : \operatorname{const} \left\{ \int_{(k-1)}^{kT_{r}} y(t_{1}) u_{k}(t_{1} - kT_{r}; \hat{\lambda}_{k}) \cos \left[\omega_{e} t_{1} + \frac{kT_{r}}{2}; \hat{\lambda}_{k}\right] dt_{1} \int_{(k-1)}^{kT_{r}} y(t_{2}) \left[\frac{\partial}{\partial \lambda} u_{k}(t_{2} - kT_{r}; \hat{\lambda}_{k}) \times \right] \\ &\times \cos \left[\omega_{e} t_{1} + \psi(t_{2} - kT_{r}; \hat{\lambda}_{k})\right] - u_{k}(t_{2} - kT_{r}; \hat{\lambda}_{k}) \times \\ &\times \cos \left[\omega_{e} t_{1} + \psi(t_{2} - kT_{r}; \hat{\lambda}_{k})\right] - u_{k}(t_{2} - kT_{r}; \hat{\lambda}_{k}) \times \\ &\times \frac{\partial}{\partial \lambda} \psi(t_{2} - kT_{r}; \hat{\lambda}_{k}) \sin \left[\omega_{e} t_{1} + \psi(t_{2} - kT_{r}; \hat{\lambda}_{k})\right] dt_{2} + \\ &+ \int_{(k-1)T_{r}}^{kT_{r}} y(t_{1}) u_{k}(t_{1} - kT_{r}; \hat{\lambda}_{k}) \sin \left[\omega_{e} t_{1} + \psi(t_{1} - kT_{r}; \hat{\lambda}_{k})\right] dt_{1} \int_{(k-1)T_{r}}^{kT_{r}} y(t_{2}) \left[\frac{\partial}{\partial \lambda} u_{k}(t_{2} - kT_{r}; \hat{\lambda}_{k}) \sin \left[\omega_{e} t_{k} + \\ &+ \psi(t_{2} - kT_{r}; \hat{\lambda}_{k})\right] + u_{k}(t_{2} - kT_{r}; \hat{\lambda}_{k}) \sin \left[\omega_{e} t_{k} + \\ &- kT_{r}; \hat{\lambda}_{k}) \cos \left[\omega_{e} t_{2} + \psi(t_{2} - kT_{r}; \hat{\lambda}_{k})\right] \right] dt_{2} \right\}. \end{aligned}$$

The circuit of Fig. 6.34 is very close to the circuit of Fig. 6.29. The first part of the circuit (processing in quadrature mixers and multiplication by functions u_{a} , $\frac{\partial u_{B}}{\partial \lambda}$ and $u_{a}\frac{\partial \psi}{\partial \lambda}$) completely repeats the circuit of Fig. 6.29. Further as filters there follow integrators over the period of the pulses, actually integrating inside

-146-



the pulse. All the rest of the circuit - multipliers and adders - completely corresponds to Fig. 6.29. In other words, from processing in a discriminator of weakly correlated



coherent pulses (see Paragraph 6.7.2) processing (6.7.82) differs only in the absence of a different interperiod accumulation before terminal multiplication. Thereby, processing (6.7.82) is limiting upon complete disappearance of correlation between periods.

Unfortunately, we cannot offer a sufficiently convenient circuit representation of purely filtration processing, if we are talking about absolutely exact fulfillment of the operation of taking the derivative of $|f(\mathbf{y}, \lambda)|^2$. This is explained by the



fact that parameters of a matched filter depend on λ , so that it is necessary to interpret in circuit terms a filter, the "derivative" of a matched one.

Without concretization of coding of the parameter this can hardly be done. If, however, we are talking about approximate realization of the operation in the form of two channels, detuned with respect to the parameter (see Paragraph 6.7.5), then as the discriminator there

Fig. 6.34. Optimum discriminator for an incoherent signal (correlation processing): 1) quadrature mixers; 2) variable-gain amplifiers; 3) integrators; 4) multipliers; 5) adder; 6) $\pi/2$ phase shifter.

should be taken two circuits of type Fig. 6.32 or 6.33, defuned $\pm\Delta\lambda/2$ from the measured value of the parameter. In this case the filtration presentation of operations of the discriminator is obvious.

During calculation of the accuracy of a discriminator it 1.: possible to use formula (0.7.52), where $J_4(u)$ is given by (6.7.36), and as J_{2D} we take the limit of

(6.7.53) as $S_{OD}(\omega) \to 1$:

$$J_{2D}(q, S_{\bullet}) \rightarrow \frac{2q^2}{T_r(1+q)}.$$

Finally, we rewrite (6.7.52) in the form

$$\mathbf{K} = \left\{ \frac{1}{T_r} \int_{\mathbf{0}}^{T_r} \left[\frac{\partial u_a(t,\lambda)^2}{\partial \lambda} + u_a^2(t,\lambda) \left(\frac{\partial \psi(t,\lambda)}{\partial \lambda} \right)^2 \right] dt - \left[\frac{1}{T_r} \int_{\mathbf{0}}^{T_r} u_a^2(t,\lambda) \frac{\partial \psi(t,\lambda)}{\partial \lambda} dt \right]^2 \right\} \frac{2q^2}{T_r(1+q)}, \qquad (\dots, \dots)$$

where T_{μ} equals the period of repetition of sendings.

For independently fluctuating sendings there is possible a generalization of the preceding formulas to the case of correlated interference (with an interval of correlation, smaller than the repetition period). It turns out that in the period of repetition the functional of probability density as before is expressed by formula (6.7.77), but $|f(y, \lambda)|^2$ and q change their value:

$$\begin{aligned}
\mathbf{q}_{\mathbf{x}} &= q \left(1 - \frac{1}{2\pi T_{r}} \int_{-\infty}^{+\infty} |U(i_{\mathcal{D}})|^{2} \frac{h_{\alpha} S_{en}(\omega)}{|1 + h_{\alpha} S_{en}(\omega)} d\omega \right), \quad (0.7, 84) \\
& |I(\mathbf{y}, \lambda)|^{2} = \left[\int_{(k-1)}^{kT_{r}} y(t_{1}) W(t_{1} - t_{2}) u_{k}(t_{2} - kT_{r}; \lambda) \cos[w_{e}t_{2} + \frac{kT_{r}}{|t_{r}|^{2}} + \int_{(k-1)}^{kT_{r}} y(t_{1}) W(t_{1} - t_{2}) u_{k}(t_{2} - kT_{r}; \lambda) \sin[w_{e}t_{2} + \frac{kT_{r}}{|t_{r}|^{2}} + \int_{(k-1)}^{kT_{r}} y(t_{1}) W(t_{1} - t_{2}) u_{k}(t_{2} - kT_{r}; \lambda) \sin[w_{e}t_{2} + \frac{kT_{r}}{|t_{2}|^{2}} + \frac{kT_{$$

Here it is assumed that interference consists of white noise of spectral density N_0 and naturally correlated interference, the maximum of whose spectral density exceeds the density of white noise by a factor of $h_{\rm H}$, and the normalized spectrum is equal to $S_{0-{\rm H}}(\omega)$ ($S_{0-{\rm H}}(0) = 1$). Further, $W(t_1 - t_2) - a$ function, the inverse of the correlation function of interferences of both types; $q - {\rm signal-to-noise}$ ratio in the absence of a correlated component of interference; $Y(\omega)$, $\widetilde{W}(\omega)$, $\Psi(\omega)$, $\Psi(\omega) - {\rm Fourier transforms of functions of <math>y(t)$ (in the given period), $W(t_1 - t_2)$ and $u(t) = u_{\rm a}(t)e^{i\psi(t)}$, respectively. Naturally, as $h_{\rm H} \rightarrow 0$, according to (6.7.84), $q_{\rm H} \rightarrow 0$, and inasmuch as $W(t_1 - t_2) \Rightarrow \frac{1}{N_0} \delta(t_1 - t_2)$, relationship (6.7.85) pascer into (6.7.80).

A new processing element, appearing for correlated interference, is a rejector circuit with pulse response $W(t_1 - t_2)$, similar to that considered in Paragraph 6.7.1. Its minimum gain, equal to $[N_0(1 + h_{\Pi})]^{-1}$ as compared to N_0^{-1} during white noise, corresponds to the frequency of the carrier of interference, and the rejection band grows with growth of h_{Π} .

The formula for calculation of the accuracy characteristic here has the form

$$\mathcal{K} = \frac{2q_{\pi}^2}{T_r(1+q_{\pi})} (b_s + \operatorname{Re} b_1^2) - \frac{4q_{\pi}^3}{T_r(1+q_{\pi})^2} (\operatorname{Re} b_1)^2, \qquad (6.7.86)$$

where

$$b_{a} = \frac{1}{k_{q}} \frac{1}{2\pi T_{r}} \int_{-\infty}^{+\infty} \frac{\left|\frac{\partial}{\partial \lambda} U\left(i\omega;\lambda\right)\right|^{2}}{1 + h_{u}S_{ou}\left(\omega - \Delta\right)} d\omega;$$

$$b_{1} = \frac{1}{k_{q}} \frac{1}{2\pi T_{r}} \int_{-\infty}^{+\infty} \frac{U^{*}\left(i\omega;\lambda\right)\frac{\partial}{\partial \lambda} U\left(i\omega;\lambda\right)}{1 + h_{u}S_{ou}\left(\omega - \Delta\right)} d\omega;$$

$$k_{q} = \frac{1}{2\pi T_{r}} \int_{-\infty}^{+\infty} \frac{|U\left(i\omega;\lambda\right)|^{2}}{1 + h_{u}S_{ou}\left(\omega - \Delta\right)} d\omega,$$

(6.7.87)

 k_q - factor of reduction of q_{Π} in comparison with q; $\Delta = \omega_{\Pi} - \omega_0$ - separation by frequency of carriers of interference and the signal.

If we turn now to the general case of an incoherent signal against a background of white noise, it differs from the coherent appearance of uncontrollable (random) phase shifts θ_i in every period. The likelihood coefficient for this case can be found by introduction of these shifts in the expression for the likelihood coefficient of a coherent signal and by averaging all θ_i . In Chapter 5 (Vol. 1) it is shown that this averaging, besides the case of uncorrelated sendings, can be taken both for very large and very small noises. In all these cases the operation of detection is close to simple square accumulation of results of optimum intraperiod processing. Thus, from the point of view of operations of processing the whole analysis of construction of detection receivers and discriminators is also transferred to the case of an (correlated) incoherent signal, where it is more exact, the larger or smaller the q. This gives certain grounds to use the formulas derived with independent interperiods fluctuations as an approximation for the case of measurement of parameters of an arbitrary incoherent signal.

In completion we consider the case of many input signals $(i)_y(t)$ (i = 1, ..., m), each of which may contain a whole series of useful components generally with different forms of modulation. We assume, however, that in all of them fluctuations

-149-

of the signal are independent from period to period, being "harmonious" within ε period, and that interferences in all mixtures have the form of white noises. Construction of the functional of probability density then slmost completely with repeat the analogous case of Paragraph 6.7.2. In view of the importance for applications we will repeat the sequence of these computations. The column set of mixtures analogously to (6.7.55) in the k-th period has the form

$$y(t) = \operatorname{Re} \{ U^+(t) \in e^{i n_0 t} \} + n(t),$$

where U(t) - complex nearly-diagonal matrix of coefficients of regular modulation in the k-th period;

> θ - complex column vector of random modulations (in distinction from Paragraph 6.7.2 it does not depend on time):

n(t) - column vector of white noises.

We have the correlation matrix of mixtures and its reciprocal matrix in the form

$$\mathbf{R}(t_1, t_2) = \operatorname{Re} \{ \mathbf{U}^+(t_1) \, \mathbf{r} \, \mathbf{U}^+(t_2) \, \mathrm{e}^{h_{\mathbf{u}_0}(t_1 - t_2)} \} + N_{\mathbf{v}}^*(t_1 - t_3), \quad (1, \dots, 1, n) \}$$

$$W(l_1, l_2) = -\operatorname{Re} \{ N^{-1} U^+(l_1) W U^+(l_2) N^{-1} e^{i a_0 (l_1 - l_2)} \} + N^{-1} \delta(l_1 - l_2),$$

where $\mathbf{r} = \text{correlation matrix of order } \left\{ \sum_{i=1}^{n} p_i \cdot \sum_{i=1}^{n} p_i \right\}$. We describe relationships between

all components of all mixture;

K

w - a structurally similar matrix, which in this case is expressed in the same

$$W = \left[1 + \frac{1}{2} r \bar{B}_{0}\right]^{-1} r; \qquad (r \dots d)$$

$$(\bar{B}_{0} = \frac{1}{T_{r}} \int U^{*}(s) N^{-1} U(s) ds),$$

. . . . N

(. 7. 2)

(4.7.93)

analogous to (6.7.62). From this we can obtain the operations of the detection receiver and discriminator.

Finally, quantity K by formula (6.7.65) can easily be reduced to a form analogous to (6.7.69);

$$I = \frac{1}{T_r} \operatorname{spur} \left\{ \left[1 + \frac{1}{2} r \overline{B}_{\bullet} \right]^{-1} r \left[\overline{B}_{\bullet} r \overline{B}_{\bullet} + \overline{B}_{\bullet} r \overline{B}_{\bullet} \right] \right\},$$

where

$$\overline{\mathbf{B}}_{1} = \frac{1}{T_{r}} \int_{0}^{T_{r}} \mathbf{U}^{*}(s) \mathbf{N}^{-1} \frac{\partial}{\partial \lambda} \mathbf{U}^{+}(s) ds,$$
$$\overline{\mathbf{B}}_{3} = \frac{1}{T_{r}} \int_{0}^{T_{r}} \frac{\partial}{\partial \lambda} \mathbf{U}^{*}(s) \mathbf{N}^{-1} \frac{\partial}{\partial \lambda} \mathbf{U}^{+}(s) ds$$

-150-"

are matrices, reflecting regular modulation of components of mixtures (1)y(t),

In that particular case when the mixtures contain only one signal component, where these components are completely correlated, and white noises are independent, K again is expressed by formula (6.7.83), where J_{2} is replaced by

$$J_{a} = \frac{2q_{1}^{2}}{T_{r}(1+q_{1})}, \qquad (6.7.94)$$

and $q_z = \frac{\sum_{i} P_{ii}T_i}{2N_0} = \frac{\sum_{i} S_i}{2N_0}$ - ratio of total signal energy in all ...annels to the intensity of white noise in one of the channels.

From the given examples it is clear that between accuracy characteristics of discriminators intended for signals of different nature there is great similarity. These interrelationships will be shown in more detail in subsequent chapters with examples of concrete coordinates.

6.7.4. Certain Worms of Non-Gaussian Signals in White Noises

Till now in examples of statistical properties of mixturer of signals with noises we almost always assumed Gaussian distribution of interference, which is usually not a limitation. Therefore it is basically interesting to study change of processing of signals for non-Gaussian statistics of the useful component of mixtures. From the example of one form of signals, taken against a background of white noises, we will show that deviation in the distribution of probabilities, even very considerable, does not lead to noticeable circuit changes of optimum discriminators.

We shall discuss independently fluctuating sendings of a signal, similar to those which we studied in Paragraph 6.7.3. If the phase in each sending is distributed evenly, then averaging with respect to phase and amplitude E of the likelihood functional of the signal against a background of normal white noise in one period leads to expression

$$P(\mathbf{y}|\boldsymbol{\lambda}) = C \int I_{\bullet} \left(\frac{E}{N_{\bullet}} |f(\mathbf{y},\boldsymbol{\lambda})| \right) P_{\bullet}(E) e^{-\frac{ET_{\bullet}}{4N_{\bullet}}} dE. \qquad (v, f, v)$$

It is useful to show the maximum possible number of regularities in general, and then to consider concrete examples. Expanding $L_p(x)$ in a Maximum series, convergent for any x, we have the logarithm of the likelihood function in the form

-151-

$$\mathbf{P}(\mathbf{y}|\mathbf{\lambda}) = A_0 + A_1 |f(\mathbf{y},\mathbf{\lambda})|^2 + A_2 |f(\mathbf{y},\mathbf{\lambda})|^6 + \dots,$$

where

$$A_{a} = \frac{1}{4N_{0}^{2}} \cdot \frac{M_{a}}{M_{b}}, \quad A_{a} = \frac{1}{64N_{0}^{4}} \left[\frac{M_{a}}{M_{0}} - 2 \left(\frac{M_{a}}{M_{0}} \right)^{a} \right], \dots$$
$$M_{b} = \int_{0}^{\infty} E^{b} P_{0}(E) e^{-\frac{E^{4}T_{c}}{4N_{0}}} dE,$$

(6.7.97)

(6.7.96)

A_0° - constant.

According to (6.7.96) the logarithm of the likelihood function is a menotomorphic nondiminishing entire function of the square of the signal amplitude envelope at the output of a matched filter. If the level of the input signal is small, in expansion (6.7.96) it is sufficient to limit oneself to the first two terms, and the optimum receiver, and consequently also the discriminator, coincide with these for the case of a Gaussian signal. This circumstance is in force also in wider conditions and is intuitively intelligible in light of the fact that for a weak signal the mixture of it with noise is approximately normal for any properties of the signal cannot be made.

) Let us turn to the operation of the discriminator. According to (6, 7, 05)

$z(kT_r) = [A_1 + 2A_2|f(\mathbf{y}, \hat{\lambda}_h)|^2 + \dots] \frac{\partial}{\partial \lambda} |f(\mathbf{y}, \hat{\lambda}_h)|^2 =$ $= Q\{|f(\mathbf{y}, \hat{\lambda}_h)|\} \frac{\partial}{\partial \lambda} |f(\mathbf{y}, \hat{\lambda}_h)|^2.$

(6.7.98)

Thus, the operation of the discriminator for an arbitrary signal condicts of reproduction of operations of a "Gaussian" discriminator, forming according to Paragraph 6.7.3 quantity $\frac{\partial}{\partial \lambda} |f(\mathbf{y}, \lambda)|^2$, and of a "Gaussian" detector, forming quantity $|f(\mathbf{y}, \lambda)|^2$. According to Fig. 0.35 the output signal of a "Gaussian" discriminator is multiplied by a positive function Q of $|f(\mathbf{y}, \lambda)|$, concretely depending on statistical properties of the signal. As a result there will be formed the circuit of a discriminator for this signal. Inasmuch as proceederistics of accuracy in the general case cannot be investigated, we consider certain examples.

a) Case of a nonfluctuating signal, when

$$P_{\bullet}(E) = \delta(E - E_{\bullet}).$$

Then according to (6.7.95),

 $\ln P(\mathbf{y} \mid \boldsymbol{\lambda}) = C + \ln I_{\bullet} \left(\frac{E_{\bullet}}{N_{\bullet}} \mid f(\mathbf{y}, \boldsymbol{\lambda}) \mid \right)$

-152-



Fig. 6.35. Optimum discriminator for a "non-Gaussian" signal: 1) "Gaussian" discriminator; 2) optimum receiver. forming $|f(\mathbf{y}, \hat{\lambda})|^2$; 3) nonlinear inertialess converter; 4) multiplier. and the operation of the discriminator will take

form

$$f(kT_{\bullet}) = \frac{I_{\bullet}\left(\frac{E_{\bullet}}{N_{\bullet}} | f(\mathbf{y}, \hat{\lambda}_{\bullet})\right)}{I_{\bullet}\left(\frac{E_{\bullet}}{N_{\bullet}} | f(\mathbf{y}, \hat{\lambda}_{\bullet})\right)} \frac{E_{\bullet}}{2N_{\bullet} | f(\mathbf{y}, \hat{\lambda}_{\bullet})|} \frac{\partial | f(\mathbf{y}, \hat{\lambda}_{\bullet})|^{2}}{\partial \lambda}. \quad (6.7.99)$$

Function

$$Q\{|f(\mathbf{y}, \lambda)|\} = \frac{I_{0}\left(\frac{E_{0}}{N_{0}}|f|\right)}{I_{0}\left(\frac{E_{0}}{N_{0}}|f|\right)} \xrightarrow{E_{0}}{2N_{0}|f|} \approx \begin{cases} \left(\frac{E_{0}}{2N_{0}}\right)^{2} \text{ when } |\overline{f}| < 1, \\ \frac{E_{0}}{2N_{0}|f|} \text{ when } |\overline{f}| > 1 \end{cases}$$

in this case is monotonically diminishing. For lowlevel signals, as already proved in general, it seeks a constant. For high-level signals, as it is easy to see from (6.7.99),

$$z(kT_r) = \operatorname{const} \cdot \frac{\partial}{\partial t} |f(y, \hat{\lambda}_k)|.$$

Operation of an optimum receiver in these conditions reduces to linear detection of the signal on the output of a matched filter.

If, however, we turn to analysis of optimum operations of the discriminator, then it is easy to prove that, instead of terminal ideal multipliers (phase detectors) contained in the "Gaussian" discriminator, here we have to take multipliers with clipping of an introduced signal which does not carry information about mismatch. Most phase detectors utilized in practice possess just such a property (see Chapter 2, Vol. 1). The physical explanation of the need for certain clipping of peaks of the input mixture as compared to the case of a Gaussian discriminator is the circumstance that the signal does not fluctuate, and considerable peaks can be formed only due to noises, suppressing the useful signal at separate moments.

The characteristic of accuracy of measurement can be easily obtained only for a high signal level, when

$$K \approx 2q \frac{1}{T_{r}} \int_{0}^{T_{r}} \left[\left(\frac{\partial u_{a}\left(t,\lambda\right)}{\partial \lambda} \right)^{2} + \frac{2}{s} \left(t,\lambda\right) \left(\frac{\partial \phi\left(t,\lambda\right)}{\partial \lambda} \right)^{2} \right] dt, \qquad (6.7.100)$$

where $q = E_0^2 T_r / 4N_0 - signal-to-noise ratio.$

According to (6.7.100) K monotonically depends on the signal-to-noise ratio and on squares of the derivatives of modulating functions with respect to the measured quantity averaged over the period. We already met similar dependences above (Paragraphs 6.7.2 and 6.7.3).

-153-

b) The case of non-Gaussian distribution, having practical importance and presented in the form of the composition of several Rayleigh distributions for *amplitude E:

$$P_{\mathbf{e}}(E) = \sum_{i} p_{i} \frac{B}{P_{ei}} e^{-\frac{B}{2P_{ei}}}$$

Here $p_i = \text{weight of separate distribution} \left(\sum_{i=1}^{p_i} p_i = 1 \right)$. The likelihood functional, according to (6.7.90), is equal to

$$P(\mathbf{y}|\mathbf{\lambda}) = \sum_{i} \frac{p_i}{1+q_i} \exp\left\{\frac{q_i | f(\mathbf{y}, \mathbf{\lambda})|^2}{N_{\bullet}T_r (1+q_i)}\right\}, \quad \left(q_i = \frac{p_{ei}T_r}{2N_{\bullet}}\right), \quad (6.7.102)$$

(0.7.101)

and operation of the discriminator is expressed by formula

のと見ていていたがないないで、

26

$$z(kT_{*}) = \text{const} \frac{\sum_{i} \frac{p_{i}q_{i}}{(1+q_{i})^{2}} \exp\left\{\frac{q_{i}|f(y,\hat{\lambda}_{k})|^{2}}{N_{*}T_{*}(1+q_{i})}\right\}}{\sum_{i} \frac{p_{i}}{1+q_{i}} \exp\left\{\frac{q_{i}|f(y,\hat{\lambda}_{k})|^{2}}{N_{*}T_{*}(1+q_{i})}\right\}} \times \frac{\partial|f(y,\hat{\lambda}_{k})|^{2}}{\partial\lambda}, \qquad (6.7.103)$$

where we assumed independence of q, from λ . The coefficient in (6.7.103) with $\frac{\partial |f(\mathbf{y}, \hat{\lambda})|^2}{\partial \lambda}$ with change of $f(\mathbf{y}, \lambda)$ changes little, where members of series with small q, play the prevailing role if for all distributions p, in (6.7.101) they have the same order. In other words, the discriminator should be designed basically for a signal belonging to the distribution with the smallest q.

6.7.5. Approximate Methods of Construction of Optimum Discriminators. Method of Comparison of Performance of Circuits

By solution of the corresponding equations we found mathematical operations on the signal which optimum discriminators should execute. We also gave functional circuits corresponding to these operations. However, these circuits cannot always be practically realized exactly. Therefore, we give certain considerations about approximate technical realization of optimum discriminators.

The question reduces to approximate presentation of the first derivative of the logarithm of the likelihood function. The first natural method is replacement of the derivative by a finite difference. If exact operation of the discriminator is given by formula (6.6.36), the approximate operation will take form

$$\mathbf{z}(t) \simeq \frac{1}{\Delta \lambda} \left[l\left(y; \, \hat{\boldsymbol{\lambda}}(t) + \frac{\Delta \lambda}{2}; \, t \right) - l\left(y; \, \hat{\boldsymbol{\lambda}}(t) - \frac{\Delta \lambda}{2}; \, t \right) \right], \quad (\ell.7, 104)$$

and for a discrete signal of the type studied in Faragraph 6.6.3 it will take form

{ }

$$z_{k} \approx \frac{1}{\Delta \lambda} \left[\int_{(k-1)}^{\lambda T_{r}} \frac{\partial}{\partial \lambda} l_{k} \left(y; \hat{\lambda}_{k} + \frac{\Delta \lambda}{2}; \tau \right) d\tau - \int_{(k-1)T_{r}}^{\lambda T_{r}} \frac{\partial}{\partial \lambda} l_{k} \left(y; \hat{\lambda}_{k} - \frac{\Delta \lambda}{2}; \tau \right) d\tau \right].$$

$$(6.7.105)$$

In essence the shown method reduces to forming a discriminator from two detection channels (without terminal integration or summation), detuned $\pm \Delta \lambda/2$ from the measured value of the parameter. Quantity $\hat{\lambda}$, as it were, is "forked." Sometimes it is technically inconvenient to have two channels immediately. Therefore, the second method of approximate realization of a discriminator is based on alternate detuning ("swinging") one detection channel near $\hat{\lambda}$. Naturally, this swinging should be faster than noticeable change of $\lambda(t)$.

Concrete values of the frequency and amplitude of swinging are selected from a series of theoretical and technical considerations and, not least of all, from consideration of maximum proximity to the potential limit of accuracy. Mathematically, swinging can be presented in the form

$$z(l) \simeq \frac{1}{T_{\pi}} \int_{l-T_{\pi}/2}^{l+T_{\pi}/2} l(\tau; y; \hat{\lambda}(l) + f(\tau)) d\tau, \qquad (6.7.106)$$

where $\Gamma(t) - a$ function with period T_{K} and amplitude, usually not exceeding the extens of the linear section of the discrimination characteristic of an optimum circuit.

As the most used forms of f(t) we indicate a square or harmonic function.

All the shown approximations of the discriminator circuit, and also circuits, in which as compared to the optimum changes are experienced only by quantitative characteristics of separate elements, usually preserve the general idea of construction of the discriminator and therefore can be called quasi-optimum. The concept of quasi-optimality is imprecisely defined. Therefore, it is always necessary to have a quantitative comparison of these circuits, and also of any others which have appeared or will appear in practice for this function, with optimum circuits. Comparison of results of § 6.2 and § 6.6 shows that the basic characteristic of discriminators both optimum, and nonoptimal, is equivalent spectral density: (or variance of error of a

^{*}Besides S_{BKB} , we shall subsequently be interested in coefficient $S_{RB}p$, characterizing parametric fluctuations, cometimes very noticeable. Then it is necessary to compare $S_{RA}p$ with $S_{RA}p$ only corresponding to the optimum circuit.

unit measurement), designated correspondingly by $S_{\partial KB}$ and $S_{\partial KB \ O\Pi T} = 1/K(t)$. The coefficient of decrease of quality of the discriminator as compared to the optimum is introduced by relationship

(6.7.107)

In any case $n \ge 1$. If comparison of any constructed and practically convenient circuits with the optimum gives a small deviation n from 1, then it is possible to express confidence that discovery of any new circuits will not improve performance indices of the radar and is therefore inexpedient.

§ 6.8. <u>Synthesis of Smoothing Circuits and Resultant</u> <u>Accuracy of Measurements</u>

In § 6.6 it was shown that optimum meters contain smoothing circuits. In the present paragraph without concretization of the physical nature of the measured quantity we obtain and analyze optimum smoothing circuits for different statistical respectives of $\lambda(t)$. Pulse responses of these circuits, and also resultant error of measurements are determined by equations (6.6.31) and (6.6.32) or (6.6.53) and (6.6.54). Solutions of these equations depend on the correlation function of the parameter $R(t, \tau)$ and function K(t), which depends on statistical properties of the input signal y(t) and the method of encoding in it parameter $\lambda(t)$. We already indicated that for Gaussian statistics of λ (t) synthesis of smoothing circuits is analogous to synthesis of Wiener filters. Although there are solutions of the last problem in literature, the specific form of equivalent "interference" and presentation of the meter in the form of a closed tracking system permit us to obtain a series of new conclusions interesting for applications.

Below we shall consider parameters in the form of stationary random processes, processes with stationary increments, quasi-regular processes (linear combinations of known functions with random factors), and also mixed cases. We separately consider synthesis of smoothing filters for meters of linear functionals of parameters.

6.8.1. Parameter - Stationary Random Process

In a number of applications it is possible to consider the measured quantity a random stationary process. An example is steady fluctuations of speed in an aircraft Doppler meter, groundspeed meter, and so forth.

If the measured parameter is stationary, during an arbitrary time of observation equation (6.6.53) takes form

$$c(t, \tau) + K \int c(t, s) R(s - \tau) ds = R(t - \tau),$$
 (6.8.1)

i.e., turns out to be an integral equation with a nucleus, depending on the difference. of the arguments. Methods of colution of such equations have been studied in principle repeatedly [19]. An exact solution can be obtained if spectral density of the parameter has the form

$$S(\mathbf{o}) = \int_{-\mathbf{o}}^{+\infty} R(\mathbf{t}) e^{-i\mathbf{a}\mathbf{t}} d\mathbf{t} = \frac{|Q_m(i\mathbf{o})|^2}{|P_n(i\mathbf{o})|^2}, \qquad (i.3.2)$$

-157-

where $Q_m(i\omega)$, $P_n(i\omega)$ - polynomials from is of degree m and n > m.

Such presentation of $S(\omega)$ is very general and enbraces practically all real cases of stationary processes. Substituting in (...8.4) function $R(t - \tau)$, expressed through $S(\omega)$; and applying to both parts of this equation differential operator $P_n(d/d\tau) P_n(-d/d\tau)$, taking into account (6.8.2) it is possible to show that equation (6.8.1) is equivalent to differential equation

$$P_{n}\left(\frac{d}{d\tau}\right)P_{n}\left(-\frac{d}{d\tau}\right)c(l,\tau)+Q_{m}\left(\frac{d}{d\tau}\right)Q_{m}\left(-\frac{d}{d\tau}\right)Kc(l,\tau)=$$

$$=Q_{m}\left(\frac{d}{d\tau}\right)Q_{m}\left(-\frac{d}{d\tau}\right)\delta(l-\tau).$$
(4.8.3)

Additionally $c(t, \tau)$ should satisfy boundary conditions at the ends of the interval of observation (0, t), which are obtained if we apply to both parts of (6.8.1) operators $(d^k/d\tau^k)P_n(d/d\tau)$ and $(d^k/d\tau^k)P_n(-d/d\tau)$ (k = 0, 1, ..., n - 1) and use certain theorems of the theory of functions of a complex variable:

Here $\sum_{j=0}^{2m+4-n} A_j^{k}(in)^{j}$ - multinomial part of the fraction

$Q_{m}(im)|^{2}(im)^{A}P_{m}^{-1}(im).$

Time t in (6.8.3) and (6.8.4) is considered a fixed parameter.

Thus, solution of equation (6.8.1) is equivalent to finding Green's function of the corresponding linear differential equation with constant coefficients the boundary conditions (6.8.4). Solution of this problem is sometimes cumbersome, but does not present fundamental difficulties.

Certain simplification is attained when m = 0, $S(\omega) = |P_n(i\omega)|^2$. Boundary conditions here are simplified, since $A_i^{(k)} = \frac{1}{2}$.

We shall consider as an example a parameter in the form of a stationary random process, which it is formally possible to consider formed by passage of white noise through an inertial link with time constant T:

$$R(l) = s_{\lambda}^{2} e^{-j l l l T}, S(w) = \frac{2 s_{\lambda}^{2} T}{1 + (wT)^{2}}$$

Note that in this case

$$P\left(\frac{d}{dt}\right) = \frac{1}{\sqrt{2s_1^2T}} + \sqrt{\frac{T}{2s_1^2}} \frac{d}{dt},$$

then relationships (6.8.3) and (6.8.4) will take form

$$(1+2a_{h}^{2}TK)c(t,\tau)+T^{2}\frac{d^{2}}{d\tau^{4}}c(t,\tau)=2a_{h}^{2}T\delta(t-\tau),$$

$$[c(t,\tau)+T\frac{d}{d\tau}c(t,\tau)]_{\tau=0}=0,$$

$$[c(t,\tau)-T\frac{d}{d\tau}c(t,\tau)]_{\tau=0}=0.$$
(6.8)

System (6.8.6) has the following solution:

$$\mathcal{E}(l, \tau) = 2a_{\lambda}^{2} \frac{(\sqrt{1+\rho}+1)e^{\sqrt{1+\rho}\tau}}{(\sqrt{1+\rho}+1)^{2}e^{\frac{\sqrt{1+\rho}\tau}{T}} - (\sqrt{1+\rho}-1)^{2}e^{\frac{\sqrt{1+\rho}\tau}{T}}},$$

where $\rho = 2\sigma_1^2$ TK.

A circuit with pulse response (6.8.7) can be realized by a controlled variablegain amplifier, an integrator and one more controlled amplifier, coupled in series (Fig. 6.36).



Fig. 6.36. Smoothing circuits for a stationary parameter with small observation time. 1, 3 - variable-gain amplifiers; 2 - integrator.

Very interesting is the case of a large time of observation, when $t/T \gg 1$. Then in view of attenuation of pulse response one should consider small differences between moments of application of input disturbances and reading $(t - \tau)/T \sim 1$, when simultaneously $t/T \gg 1$, $\tau/T \gg 1$. It is easy to prove that in these conditions

$$c(t, \tau) \rightarrow c(t-\tau) = \frac{2\tau_0^2}{\sqrt{1+p+1}} \exp\left\{-\frac{\sqrt{1+p}}{T}(t-\tau)\right\}. \quad (6.8.8)$$

(6.8.5)

6)

(6.8.7)
Thereby, in the limiting case pulse response $c(t, \tau)$ depends only on the difference of arguments and coincides with the response of a certain link of the first order with effective time constant

$$T_{\bullet\bullet\bullet} = T/V + P.$$

Seeking from (6.8.8) solution of equation (6.6.54) in the form $g(t - \tau)$, we have function

$$g(t-\tau) = \frac{2\sigma_0^2}{\sqrt{1+\rho+1}} \exp\left\{-\frac{(t-\tau)}{T}\right\}$$
(6.8.10)

(6.8.9)

(6.8.13)

with Fourier transform

i er

$$G(io) = \frac{2s_0^2 T}{V_{1+\ell+1}} \cdot \frac{1}{1+ioT} \cdot (0.8.11)$$

Thus, the single-loop variant of smoothing circuit should be a circuit with exactly that time constant which corresponds to the inertial link forming the parameter of white noise. Multiplying G(ab) by the transmission factor of an optimum discriminator K, we note that the total gain factor in the loop of an optimum meter varies as

$$K_{g} = \frac{p}{\sqrt{1+p+1}} = \begin{cases} p/2 \text{ when } p < 1, \\ \sqrt{p} \text{ when } p > 1. \end{cases}$$
(0.8.12)

Finally, variance of resultant error of measurement of the parameter, according to (6.8.7), is equal to



Quantity ρ in relationships (6.8.7)-(6.8.13) actually is the equivalent signal-to-noise ratio in the meter, inasmuch as it is equal to the ratio of the maximum value of spectral density of the parameter $2s_{\chi}^{2}T$ to the equivalent spectral density of the discriminator 1/K. The monotone character of the dependence of

-160- *

maximum error in (6.8.13) on ρ does not require explanation. More interesting is the dependence of the overall gain factor in the loop on ρ . Physically the interconnection between K_{Σ} and ρ is explained by optimum selection of the transmission band of the system

$$\Delta f_{\bullet \bullet} = \frac{1}{2\pi} \iint \frac{KC(i \bullet)}{KC(0)} d \bullet = \frac{\sqrt{1+i}}{4T} \cdot \qquad (6.8, 14)$$

ういたいないないないであるのである

For high levels of noise it is useful to narrow the bandwidth to a matched width $(\Delta f_{a\phi})_{COTM} = 1/4T$ for maximum suppression of noises, sacrificing tracking of abrupt changes of the parameter. For low-level noises it is more useful to use a transmission band expanded by a factor of $\sqrt{1 + \rho}$, within limits of which the level of spectral density of the parameter exceeds the level of the noises.

Especially important for applications is the case of a large time of observation, when the process of measurement "becomes steady" and its characteristics cease to depend on the initial conditions of measurement. To find limiting values of $c(t - \tau)$ and $g(t - \tau)$ for the case of a large time of observation in general is considerably more convenient by the method of factoring, given in § 6.5. For the given case there is performed factoring of function

$$+KS(\bullet) = \Psi(i\bullet) \Psi(-i\bullet), \qquad (6.8.15)$$

and the final result is recorded in the form

$$C(in) = \frac{1}{\Psi(in)} \left[\frac{S(n)}{\Psi(-in)} \right]_{+}.$$
 (6.8.16)

Here, the smoothing filter of the equivalent single-loop system has frequency response

$$G(io) = C(io)/(1 - KC(io)),$$
 (6.8.17)

which is found from equation (6.6.54) by the method of Fourier transforms.

Stationary error of measurement at the limit is constant and is expressed by a simple formula, based on the theory of Fourier integrals:

$$\sigma_{\text{surr}}^2 = \lim_{e \to \infty} ieC(ie). \tag{6.8.18}$$

Especially convenient is the method of factoring for a rational-fractional spectral density of the parameter of form (6.8.2). Turning as an example again to the very simple case described by relationships (6.8.5) we have, concretely,

-161-

$$1 + KS(\mathbf{o}) = \frac{1 + \mathbf{p} + (\mathbf{o}T)^2}{1 + (\mathbf{o}T)^2} =$$
$$= \frac{\mathbf{y} + \mathbf{i} + \mathbf{i} + \mathbf{o}T}{1 + \mathbf{i} + \mathbf{o}T} \cdot \frac{\mathbf{y} + \mathbf{p} - \mathbf{i} + \mathbf{o}T}{1 - \mathbf{i} + \mathbf{o}T} = \Psi(\mathbf{i} = \mathbf{o}) \Psi(-\mathbf{i} = \mathbf{o})$$

Then from (6.8.16) we have



and, finally, the frequency response of the filter is equal to

$$C(io) = \frac{\rho}{K V \overline{1+\rho} (V \overline{1+\rho+1})} \cdot \frac{1}{1+\frac{ioT}{V \overline{1+\rho}}}$$

which coincides with the Fourier transform of (6.8.8). Increase of the orders of polynomials $P_n(i\omega)$ and $Q_k(i\omega)$ in (6.8.2) leads to complication of the factoring procedure, consisting of finding roots of algebraic much ions of a high order. In principle this is the same equation as for an arbitrary time of objective interval the limiting case nevertheless sharply complifies further calculations freeing us from the need to allow for boundary conditions.

6.8.2. Parameter - Random Process with Stationary Increasements

In engineering practice in a number of cases it impossible even to approximately consider the measured quantity stationary: however, it is possible to consider stationary a certain derivative of the parameter. Thus, if acceleration of the object changes chaotically and steadily, the coordinate of it is a random prove with a stationary second derivative. Such a type of random process we include in the class of processes with k-th stationary increments (k = 0, 1, ...) [59].

Mathematically a process with stationary increments of the k-th order is defined as process x(t), statistical characteristics of whose k-th difference

$$\Delta_{n}^{k} x(l) = \sum_{m=0}^{k} \binom{k}{m} (-1)^{m} x(l-m\tau)$$

are not time-dependent. Formally, it is possible equate such a process to a process

-162-

with a stationary k-th derivative. To this class there belong stationary processes and processes obtained from them by application of integro-differential operators with constant coefficients. Correlation function $R(t, \tau)$ of a process with stationary increments is easily expressed through the correlation function of the stationary k-th derivative of $\rho(t - \tau)$ in the form

$$R(t, \tau) = \begin{cases} \int_{t_0}^{t_0} dt_0 (t - t_0)^{k-1} (\tau - t_0)^{k-1} p(t_1 - t_0) \text{ when } k > 1, \\ p(t - \tau) & \text{ when } k = 0. \end{cases}$$
(6.8.19)

This relationship with a stationary process permits us to introduce the concept of spectral density $S(\omega)$ of the process with stationary increments, determining it according to the formula

$$p(\tau) = \frac{1}{2\pi} \int e^{+i\omega\tau} e^{2k} S(e) de.$$
 (6.8.20)

To find function $c(t, \tau)$ from equation (6.6.53) with arbitrary observation time in principle it would have been possible to use the method of reduction to a differential equation with a special right member, absolutely analogous to that presented in Paragraph 6.8.1. if the spectral density of the process is a rational-fraction function of i ω , i.e., can be presented in the form (6.8.2). We select a somewhat different way of presentation. By two important and very graphic examples we explain the method of reduction of equation (6.6.53) to differential equations of another type; with the help of the latter we effectively solve these particular problems; and then we expound a general method for obtaining limiting operators, i.e., smoothing filters for a large time of observation.

Let us start with the very simple example of a parameter in the form of a Wiener process, i.e., the integral of white noise. The correlation function of this proceis equal to

$$R(t, \tau) = B_1 \min(t, \tau),$$
 (6.8.21)

where $B_1 = \text{const}$ is the spectral density of the initial white noise. Substituting (6.8.21) in equation (6.6.53) for $t \ge \tau$, we have

$$c(l, s) + KB_{1} \int c(l, s) s ds + KB_{1} \int c(l, s) ds = B_{1} s,$$
 (6.8.22)

and successive differentiation with respect to τ gives

-163-

$$\frac{\partial c(t,\tau)}{\partial \tau} + KB_s \int c(t,s) ds = B_s, \qquad (C.8.87)$$

$$\frac{\partial^2 c(t,\tau)}{\partial \tau^2} - K B_{t} c(t,\tau) = 0. \qquad (6.6.2^{4})$$

The solution of equation (6.8.24), satisfying conditions (6.8.22) and (6.8.23) simultaneously has the form

$$c(l,\tau) = \sqrt{\frac{B_1}{K}} \frac{\operatorname{sh} \dot{\gamma} \overline{B_1 K \tau}}{\operatorname{ch} \sqrt{B_1 K t}}$$
(6.8.25)

From this, according to (6.8.25) and (6.8.45), we have

$$\mathbf{s}_{\text{sur}}^{2}(t) = \sqrt{\frac{B_{i}}{K}} \operatorname{th} \sqrt{B_{i}K} t. \tag{5.3.26}$$

The last expression shows that for small times of observation variance of error of measurement is equal to variance of the a priori distribution of R(t, t) == B_1T , and with growth of t passes to a stationary value $\sqrt{B_1/K}$, monotonically depending on intensity B_1 of white noise, from which the parameter is formed, and the equivalent spectral density of an optimum discriminator 1/K. The process of becoming steady is faster, the larger the product B_1K .

Function $c(t, \tau)$ for large t, τ also the limiting value

$$c(l,\tau) \rightarrow c(l-\tau) = \sqrt{\frac{B_1}{K}} e^{-\sqrt{B_1K}(l-\tau)},$$

corresponding to a RC-circuit with time constant $(P_1K)^{-1/2}$, which follows from the form of the frequency response

$$C(i\mathbf{o}) = \int_{-\infty}^{+\infty} c(l-z) e^{-i\alpha(l-z)} d(l-z) = \frac{\sqrt{B_1/R}}{\sqrt{B_1/R} + i\mathbf{o}}.$$

Fulse response of an open loop $g(t, \tau)$ is determined by equation (0.6.54), and for $c(t, \tau)$, given by formula (6.8.25), it is equal to

$$g(l, \tau) = \sqrt{\frac{B_1}{K}} \, lh \, \tau, \qquad (6.8.28)$$

i.e., in general, does not depend on t. According to (6.8.28) smoothing consists of amplification of the output signal of the discriminator by an amplifier with gain factor, varying according to the law (6.8.28), and subsequently by integration.

In view of the a priori known continuity of parameter $\lambda(t)$ and zero initial error of -1/2measurement the output signal of the discriminator z(t) for $\tau < (B_1K)$ is transmitted with a small, but increasing amplification. The build-up is explained by radual increase of variance of $\lambda(t)$. When $\tau >> (B_1K)$ smoothing passes into constant amplification and to integration.

It is interesting to note that a smoothing circuit with one integrator is very often used in practice. Theoretical consideration shows that without taking into account the transient regime such a smoothing filter is optimum for parameter $\lambda(t)$ in the form of the first integral of white noise. The magnitude of the overall gain factor in open loop K $\sqrt{B_1/K} = \sqrt{KB_1}$ here depends on the intensity of parameter B_1 , and also on the statistics of the input mixture of the signal with noise (through K). This result, according to Paragraph 6.8.1, is natural.

Let us consider now a process in the form of the second integral of white noise, the correlation function of which is equal to

$$R(l, \tau) = \begin{cases} B_0 \frac{\tau^2 (3l-\tau)}{6} & \text{when } \tau < l, \\ B_0 \frac{\tau^2 (3r-1)}{6} & \text{when } l < \tau. \end{cases}$$
(6.8.29)

Successively differentiating with respect to τ equation (6.6.53), in which there is placed the value of R(t, τ) from (6.8.29), we have differential equation

$$\frac{\partial e(l,s)}{\partial s^{*}} + a_{2}^{4} e(l,s) = 0, \ (a_{3} = (B,K)^{W})$$
(6.8.30)

with four integro-differential subsidiary conditions:

$$c^{a}(l, \tau) + a_{2}^{4} \int_{0}^{t} c^{a}(l, s) \frac{s^{a}(3\tau - s)}{6} ds + + a_{2}^{4} \int_{0}^{t} c^{a}(l, s) \frac{\tau^{a}(3s - \tau)}{6} ds = a_{2}^{4} \frac{\tau^{a}(3t - \tau)}{6}, \frac{\partial c^{a}(l, \tau)}{\partial \tau} + a_{2}^{4} \int_{0}^{t} c^{a}(l, s) \frac{s^{a} ds}{2} + + x_{2}^{4} \int_{0}^{t} c^{a}(l, s) \frac{2\tau s - \tau^{a}}{2} ds = a_{2}^{4} \frac{2\tau l - \tau^{2}}{2}, \frac{\partial^{a} c^{a}(l, \tau)}{\partial \tau^{a}} + a_{2}^{4} \int_{0}^{t} c^{a}(l, s)(s - \tau) ds = a_{2}^{4}(l - \tau), \frac{\partial^{a} c^{a}(l, \tau)}{\partial \tau^{a}} - a_{2}^{4} \int_{0}^{t} c^{a}(l, s) ds = -a_{2}^{4},$$

-165-

$$c^{\bullet}(t,\tau) = Kc(t,\tau).$$

Finally, we have a solution in the form

$$c^{\bullet}(l, \mathbf{v}) = 2\sqrt{2}a_{s} \left[\frac{c h \frac{d d}{\sqrt{2}} s h \frac{d \mathbf{v}}{\sqrt{2}} \cos \frac{d \mathbf{v}(l-\mathbf{v})}{\sqrt{2}}}{c h \sqrt{2} c d + 2 + \cos \sqrt{2}a_{s} l} - \frac{c \cos \frac{d d}{\sqrt{2}} s h \frac{d d}{\sqrt{2}} c h \frac{d \mathbf{v}(l-\mathbf{v})}{\sqrt{2}}}{\sqrt{2}} \right]$$
(6.8.31)

Variance of error of measurement is equal to

$$J_{sets}^{2}(l) = \frac{\sqrt{2}c_{s}}{k} \frac{t_{set}}{c_{s}} \frac{t_{set}}{2c_{s}} - \frac{t_{set}}{2c_{s}} \frac{1}{2c_{s}} \frac{$$

In particular, for large and small time of measurement

$$\sigma_{sus}^{2}(l) = \begin{cases} B_{g} t^{0} / 3 \text{ when } l < (B_{g} K)^{-1/l}, \\ \sqrt{2} (B_{g} / K^{0})^{1/l} \text{ when } l > (B_{g} K)^{-1/l}. \end{cases}$$
(6.8.33)

As follows from (6.8.33), for a small time observation variance of error of measurement, as also in the preceding example, increases as the a priori variance of parameter $R(t, \tau)$ from (6.8.29). The fact is that in the first moments inertial smoothing elements have not yet accumulated a signal, which testifies to the sppearing error of measurement. For a large time of observation variance of error seeks, as before, a stationary value, monotonically depending on spectral density B_2 of white noise, from which the parameters was formed, and on the equivalent spectral density of the discriminator 1/K.

It is interesting to note that for large time of observation $c^*(t, \tau) = Kc(t, \tau)$ passes to

$$Ke(l, \mathbf{r}) \rightarrow \frac{\mathbf{q}_{1}}{\sqrt{2}}e^{-\frac{\mathbf{q}_{1}}{\sqrt{2}}}\cos\frac{\mathbf{q}_{1}(l-\mathbf{r})}{\sqrt{2}}, \qquad (6.8.34)$$

i.e., describes a circuit with constant parameters and frequency response

$$KC(i=) = \frac{(i=) \sqrt{2} e_1 + e_2^2}{(i=)^2 + (i=) \sqrt{2} e_1 + e_1^2}.$$
(6.8.35)

In both the examples given, in spite of the nonstationariness of the parameter, operators of smoothing, analogously to Paragraph 6.8.1, belong to circuits with constant parameters.

where

The physical expansation of this is that during measurement of a process with stationary increments there is established dynamic equilibrium between the effect of the growth of uncertainty of the value of the parameter due to its random changes and the effect of more precise definition of its value due to the arrival of new data. These mutually opposite effects follow from relationships (6.5.35) and will be touched be again in § 6.9.

It is interesting that transmission of the mixture of a process with a stationary first or second derivative with white noise through a circuit with a rationalfraction frequency response of type (6.8.27) or (6.8.35), in which the order of the polynomial of the numerator is less than the order of the polynomial of the denominator by one, ensures minimum necessary difference of orders of these polynomials for smoothing white noise and simultaneously gives at the output a random process with a stationary derivative, i.e., a process, similar in properties to the measured quantity. This is especially graphic for a system of the first order, at whose output there will be formed a random process with spectral density E_1/x^2 , equal in accuracy to the spectral density of the tracked quantity.

It is natural to assume that for any processes with stationary derivatives there exists limiting operators of smoothing, belonging to circuits with a constant parameters. Therefore, we generalize the algorithm of finding limiting operators, based on factoring, to parameters in the form of random processes with stationary intermed. Let us substitute in equation (6.6.53) correlation function $R(t, \tau)$, expressed according to (6.8.19), and direct in this expression $t_0 \rightarrow \infty$, seeking a solution in the form $c(t, \tau) = c(t - \tau)$. Then, applying to both parts of the equation Fourier transformation and using relationship (6.8.20), one can prove that the solution equation (6.6.53) in this case is expressed by the same formula (6.8.16) as for stationary processes. The pulse response of the smoothing filter of a single-loc circuit $g(t, \tau) = g(t - \tau)$ is determined through the Fourier transformation with 's e help of relationship (6.8.17), and maximum error of measurement is given by formula (6.8.18).

It is necessary only to stipulate that factoring of (6.8.15) in this case is based on presentation of the factor of form ω^{2k} in the denominator of spectral density $S(\omega)$ in the form $(i\omega)^k(-i\omega)^k$, where 1/iw is formally considered a pole in the upper half-plane, and 1/(-iw) is considered a pole in the lower half-plane of complex variable ω .

For the purpose of illustration of the method of factoring we again consider

-167-

a parameter in the form of the first and second integrals of white noise. In the first case spectral density of the parameter according (6.3.19) should be defined as $S(\omega) = B_1/\omega^2$. Here,

$$\Psi(io)\Psi(-io) = 1 + KS(o) = \frac{\sqrt{B_1K} + io}{io} \frac{\sqrt{B_1K} - io}{-io}$$

and in accordance with (6.8.16)-(6.8.18) we have

$$C(i \bullet) = \frac{i \bullet}{\sqrt{B_1 K} + i \bullet} \left[\frac{B_1 (-i \kappa)}{\bullet^2 (\sqrt{B_1 K} + i \bullet)} \right]_+ = \sqrt{\frac{B_1}{K}} \frac{1}{\sqrt{B_1 K} + i \bullet},$$

$$G(i \bullet) = \sqrt{\frac{B_1}{K}} \frac{1}{i \bullet}, \quad \sigma_{sux}^2 = \sqrt{B_1/K},$$
(6.8.76)

i.e., results, already analyzed by us above [see (6.8.26) - (6.8.28)]. For the second example we have $S(\omega) = B_0/\omega^4$ and factoring



where again $a_2 = (B_2 K)^{1/4}$. From this we finally have a result, coinciding with (6.8.35). A most interesting circumstance is revealed when by formula: (6.8.47) and (6.8.35) we find the filter of sharthing in a single-loop system. It turns out that

$$G(im) = \sqrt{\frac{B_1}{K}} \frac{1 + (4/KB_2)^{1/1} im}{(im)^2}, \qquad (6.3.7)$$

i.e., the filter is an ideal double integrator with a correcting RC-circuit time constant $\tau_{RCP} = (4/KB_2)^{1/4}$ (Fig. 6.37). A filter with frequency - number of type (5.8.37) is, probably, the most widestread in practice: however the conditions of its optimality revealed here are by no means universal. We recall that there conditions require presence in the parameter of the first time derivative and a chaotically varying stationary second derivative, which it is possible to approximate by white noise.

Analogously we can obtain a solution of equations (5.6.53) and (6.6.54) a_{130} for a parameter in the form of the integral of any n-th order of white noise.

-168-

In particular, for the third order we have

$$KC(i \bullet) = \frac{2a_{s}(i \bullet)^{s} + 2a_{s}^{2}(i \bullet) + a_{s}^{3}}{(i \bullet)^{s} + 2a_{s}(i \bullet)^{s} + 2a_{s}^{2}(i \bullet) + a_{s}^{3}},$$

$$KG(i \bullet) = VKB_{s} \cdot \frac{1 + 2(i \bullet / a_{s}) + 2(i \bullet / a_{s})^{2}}{(i \bullet)^{2}},$$

$$a_{s \bullet u t}^{2} = 2\left(\frac{B_{s}}{K^{3}}\right)^{1/6},$$

$$a_{s} = (KB_{s})^{1/6}.$$

(6.8.38)

Calculation of an ever higher order of smoothness of parameter $\lambda(t)$, as follows from (6.8.36)-(6.8.38), leads to even greater complication of smoothing circuits. In general they contain n integrators and circuits of correction of the (n - 1)-th order, parameters of which depend on $a_n = (KB_n)^{1/2n}$. The gain factor of an open



Fig. 6.37. Block diagram of smoothing circuits for a parameter with stationary increments. 1 - high-gain amplifiers. loop taking into account the gain factor of the discriminator in all cases is proportional to $\sqrt{KB_n}$, which qualitatively coincides with the case of a stationary parameter and is dictated by the same factors. All synthesized filters provide sta-

bility to the closed system. In particular, in a system of the second order, according to (6.8.36), there are provided condition: of stability close to the threshold between aperiodic and oscillatory regimes. This general circumstance is characteristic for all optimum filters knowingly giving a limited response to disturbances. It is convenient in that the question of stability during synthesis again arises only when there arises the necessity of departing in the meter from characteristics which are in the statistic is sense optimum.

Let us give, finally, an expression for variance of stationary error of measurement of the parameter in the form of the n-th integral of white noise of intensity. B_n :

$$\Phi_{\text{sum}}^{2} = \frac{1}{\frac{1}{\sin\frac{\pi}{2\pi}}} \left(\frac{B_{n}}{K^{2n-1}}\right)^{1/2n} \xrightarrow{\frac{2\pi}{\pi}} \left(\frac{B_{n}}{K^{2n-1}}\right)^{1/2n} \cdot \left(\frac{1}{K^{2n-1}}\right)^{1/2n} \cdot$$

It is easy to prove that in particular cases (n = 1, 2, 3) from (6.8.39) there ensue the corresponding formulas, given above. As follows from formula (6.8.39), the potential mean square error with growth of smoothing of the parameter all the less depends on the intensity of the parameter (B_n is taken to the 1/4n power), and σ_{BEX}^{ϵ} as total error is close in value to its fluctuating component. However, to guarantee this result we should correctly select the band of transmission of the closed system by selection of the corresponding gain factor.

6.8.3. Parameter - Linear Combination of Known Functions with Random Factors

In a broad class of practical applications, for instance during measurement of coordinates of bodies flying by the laws of ballistics, the character of change of $\lambda(t)$ is known with an accuracy of certain parameters $\alpha_1, \alpha_2, \ldots, \alpha_m$, which are not time-dependent, the values of which by virtue of initial conditions of observation one should consider random. It is possible to present this dependence in the form

$$\mathbf{\lambda}(t) = F(t; \mathbf{a}_1, \dots, \mathbf{a}_m). \tag{6.8.40}$$

We consider that

$$a_b = a_b + \mu_b$$

(6.8.41)

where a_k - mean value;

 μ_k - small normally distributed deviation from the mean value.

Then function (6.8.40) can be approximately presented in the form

$$\lambda(t) \approx \overline{\lambda(t)} + \sum_{h=1}^{m} p_h j_h(t), \qquad (5.8.42)$$

where

$$\overline{\lambda(l)} = F(l; a_1, \dots, c_m), f_k(l) = \frac{\partial F(l; a_1, \dots, a_m)}{\partial a_k}$$

are known functions.

According to (6.8.42) parameter $\lambda(t)$ then is a normal random process. Its mean value is equal to $F(t; a_1, \ldots, a_m)$, and the correlation function is determined by the obvious relationship

$$R(l_1, l_2) = \sum_{l,k=1}^{m} \mu_{lk} f_1(l_1) f_k(l_2) = f^+(l_1) Mf(l_2), \qquad (0.0.43)$$

where $\mathbf{M} = \|\mathbf{M}_{\mathbf{i}\mathbf{k}}\| \|\mathbf{\mu}_{\mathbf{i}\mathbf{\mu}\mathbf{k}}\|$ - matrix of mixed moments of quantities $\mu_{\mathbf{i}}(\mathbf{i}, \mathbf{x} = 1, ..., \mathbf{m});$ $\mathbf{f}(\mathbf{t}) = \{(\mathbf{ft}_{\mathbf{i}}), ..., \mathbf{f}_{\mathbf{m}}(\mathbf{t})\}$ - column vector with elements $\mathbf{f}_{\mathbf{k}}(\mathbf{t}) = \partial/\partial \mathbf{a}_{\mathbf{k}} F(\mathbf{t}; \mathbf{a}_{\mathbf{i}}, ..., \mathbf{a}_{\mathbf{m}});$

sign "+" signifies transposition.

In general, a parameter of form (6.8.42) is degenerate random process, completely determined by an m-dimensional distribution of probabilities, so that a reciprocal matrix \mathbf{R}^{-1} for any n > m (n - number of moments of observation) does not exist for it.

-170-

This formal difficulty, allowing one to doubt the applicability in the given case of results obtained in § 6.6, can easily be by passed if we add to $\lambda(t)$ an arbitrarily small nondegenerate random process $\lambda_1(t) = \epsilon x(t)$. Then the correlation function of process $\lambda(t) + \epsilon x(t)$ allows inversion; the distribution is nondegenerate for any n > m, and in equation (6.6.53) there will be contained correlation function $R(t_1, t_2) + \epsilon^2 R_{\chi}(t_1, t_2)$. Directing ϵ to zero, we obtain an equation in which there enters correlation function (6.8.43).

We start to find smoothing circuits with the very simple case of m = 1:

$$2(l) = \overline{2(l)} + \mu(l), \qquad (6.8.44)$$

where $\overline{\lambda(t)}$ and f(t) - known functions;

 μ - normally distributed quantity, for which $\overline{\mu} = c$, $\overline{\mu}^2 = \sigma_0^2$, so that $R(t, \tau) = \sigma_0^2 f(t) f(\tau)$.

Equation (6.6.53) takes form

$$c(l, \tau) + K \sigma_{0}^{2} \int_{t_{0}}^{t} c(l, s) f(s) f(\tau) ds = \sigma_{0}^{2} f(l) f(\tau), \qquad (6.8.45)$$

i.e., is an integral equation with a degenerate kernel. Seeking its solution in the form $c(t, \tau) = \phi(t)f(\tau)$, it is simple to obtain finally

$$c(l, \mathbf{v}) = \frac{d_{1}^{2}(l) l(\mathbf{v})}{1 + K d_{2}^{2} \int l^{q}(\mathbf{v}) d\mathbf{v}}, \qquad (6.8.46)$$

from which error of measurement $\sigma_{BHX}^2(t)$ is equal to

$$\mathbf{e}_{\mathbf{pus}}^{\mathbf{2}}(l) = \frac{\mathbf{e}_{\mathbf{pus}}^{\mathbf{2}}(l)}{1 + K \mathbf{e}_{\mathbf{0}}^{\mathbf{2}} \int f(\mathbf{p}) d\mathbf{s}}$$
(6.8.47)

(6.8.48)

and when the value of the parameter does not grow faster than the power of t, it always seeks zero as $t \rightarrow \infty$. For instance, when

$$f(t) = \sum_{j=1}^{m} a_j t^j$$

 $\sigma_{sur}^2(l) \rightarrow \frac{2\pi+1}{Kl}$

we have

The tendency of error of measurement to zero is a specific peculiarity of the considered case and is caused by the fact that the law of change of $\lambda(t)$ is known with an accuracy of a constant factor μ . The problem of filtration actually consists of measurement (estimation) of this constant factor. From the theory of estimation, we know (§ 6.6) that measurement of a constant parameter can be performed with error tending to zero with increase of the number of measurements, where during measurement of the coefficient of an m-th order parabola mixed with white noise with spectral density 1/K the variance of error in the absence of any a priori data about this coefficient is exactly (2m + 1)/(Kt).

£⊁



Fig. 6.38. Smoothing circuits for a quasi-regular parameter (single-loop variant of a meter): 1, 3 - variable-gain amplifiers; 2 - integrator.

Seeking a solution of equation (6.6.54) in the form $g(t, \tau) = f(t)\xi(r)$ for $c(t, \tau)$, determined according to (6.8.46), we have

$$\mathbf{g}(l, \mathbf{r}) = \frac{\mathbf{r}_{0}^{2} l(l) f(\mathbf{r})}{1 + K \mathbf{q}_{0}^{2} \int_{0}^{1} l^{2}(\mathbf{s}) d\mathbf{s}}$$
(6.8.44)

According to (6.8.49) the smoothing filter of a single-loop system possesses variable parameters for any time of charaction. Smoothing (Fig. 6.38) consists of multiplication of the output signal of the discriminator by function $\sigma_0^2 t(t) / \left[1 + K\sigma_{0,t_0}^{2t} t^2(s) ds\right]$, subsequent integration and multiplication by function f(t). The output signal of the integrator is the current estimate of parameter μ and is used as a factor for the law of change of parameter f(t), generated in the circuit. Atter introduction of $\overline{\lambda(t)}$ there will form estimate $\hat{\lambda}(t)$.

Characteristic for a filter with characteristic (6.8.49) is gradual operation of Haddiminator output or, in other words, "freezing" of the estimated deficient, becausing faster, the less the equivalent spectral density 1/K. This is explained by the fact that the law of change of $\lambda(t)$ is known a priori so that the finite interval of the realization permits us with good accuracy to predict behavior of parameter $\lambda(t)$ in all subsequent moments of time. Upon the expiration of this interval error of measurement already becomes small, and only specially noticeable peaks of output voltage of the discriminator correct estimate μ .

-172-

With equal success smoothing can be conducted by a double-loop system of the type of Fig. 6.13, where pulse response of the filter $c(t, \tau)$ is given by relationship (6.8.46). As Fig. 6.39 illustrates, in this case output voltage of the adder of the loop of internal coupling which is approximately an additive mixture of the measured quantity and white noise, is multiplied by function f(t), and then is integrated.



Fig. 6.39. Smoothing circuits for a quasi-regular parameter (double-loop variant of a meter): 1, 3 -variable-gain amplifiers; 2 -integrator; 4 -constant amplifier.

Thereby there will be formed correlation between the input (for the system of smoothing) realization and the expected form of "signal." Further processing reduces to multiplication of the formed quantity by function $\frac{1}{5} \frac{1}{1+K_{c}} \frac{1}{5} \frac{1}{5$

Now let us turn to the general case of process (6.8.42) for an arbitrary m. Substituting (6.8.43) in (6.6.53), we again have an equation with a degenerate kernel, solution of which is best sought in the form

 $c(l, \tau) = \varphi^+(l) f(\tau).$ (6.8.50)

いたないないであり

As a result we have

$$c(t, \tau) = f^{+}(t) [M^{-1} + KU(t)]^{-1} f(\tau), \qquad (6.8.51)$$

where the square $(m \times m)$ -order matrix U(t) is determined by expression

$$\mathbf{U}(t) = \|U_{ik}(t)\| = \|\int_{0}^{\infty} f_{i}(s)f_{k}(s) ds\|. \qquad (6.8.52)$$

Expression (6.8.51) is a natural generalization of relationship (6.8.47). Analogously we have generalizations for pulse response of the filter and variance of

-173-

 $(t, -1) = f^{+}(t) [M^{-1} + KU(-1)]^{-1} f(-1), \qquad (6.8.53)$ $(t) = f^{+}(t) [M^{-1} + KU(t)]^{-1} f(-1). \qquad (6.8.54)$

According to (6.8.53) smoothing circuits in the single-loop variant consist of m parallel channels. In the k-th channel there is produced multiplication of z(t) by functions which depend on all $f_1(t)$ (i = 1, 2, ..., m) and on K, M_{1k} ; then results of multiplication are integrated, as a result of which there are formed estimates of m unknown coefficients. Then, from the estimates there is formed the value of the parameter according to formula (6.3.42).

The shown method of smoothing has the following peculiarities:

error

1. In it both the e priori accuracy of measurement of the unknown factor and also accuracy properties of an optimum discriminator, processing newly arriving data, are accounted for.

2. Any curvature of the functions which depict change of parameter $\lambda(t)$ in time is accounted for. For the purpose of minimizing errors during smoothing of especially nonlinear functions we use known laws of change of parameter $f_k(t)$ without any increase of the number of series-coupled integrating networks.

3. With increase of the time of observation the operators of smoothing end also the potential error of measurement all the less depend on a priori conditions, seeking magnitudes

$g(t,\tau) \rightarrow \frac{1}{K} f^+(t) U^{-1}(\tau) f(\tau),$	(6.8.55)
$\boldsymbol{c}(\boldsymbol{l},\boldsymbol{\tau}) \rightarrow \frac{1}{K} \boldsymbol{f}^{+}(\boldsymbol{t}) \boldsymbol{U}^{-1}(\boldsymbol{t}) \boldsymbol{f}(\boldsymbol{\tau}),$	(6.8.56)
$\mathbf{e}_{\mathtt{sux}}^{\mathtt{S}}(t) \rightarrow \frac{1}{K} \mathrm{f}^{+}(t) \mathrm{U}^{-1}(t) \mathrm{f}(t).$	(6. 8.57)

Formula (6.8.57) is met in literature devoted to measurements of coordinates of balliatic targets [14]. Here, the measured quantity is usually considered the constant parameters of the trajectory, but not the actual coordinated and they apply the theory of the maximum likelihood estimate, considering a priori information about the parameters generally absent. Let us explain this in greater detail.

We assume that radio processing in the discriminator is considered assigned, and it is assumed that the object of optimization is mixture x(t) of the law of motion $\lambda(t, \mu)$, depending on constant parameters μ , and noise disturbances n(t) with Guassian distribution. As results of §6.6 show, assuming linear work of the

-174-

discriminator and ignoring parametric fluctuations, by addition of the discriminator output (divided by its gain factor) to the measured value of the parameter it is indeed possible to obtain voltage having the properties of an additive mixture of equivalent "signal" and "noise." In conditions of § 6.6 noise is white, so that its distribution by virtue of inertia of subsequent circuits can be considered Ecussian. Then the likelihood functional of μ is recorded in the form

$$\mathcal{P}(\boldsymbol{x}(l)|\boldsymbol{\mu}) = C \exp\left\{-\frac{K}{L^2}\int_{0}^{l} [\boldsymbol{x}(\tau) - \lambda(\tau, \boldsymbol{\mu})]^2 d\tau\right\}.$$

If the a priori distribution of μ has large variance, and for $\lambda(t; \mu)$ expansion (6.8.43) is approximately valid, then for determination of μ we have a system of linear equations of maximum likelihood

$$\sum_{i=1}^{n} \mu_{j} \int [j(\tau)] f_{i}(\tau) d\tau = \int x(\tau) f_{i}(\tau) d\tau. \qquad (6.8.58)$$

Solution of the system in matrix representation has the form

$$\hat{\mathbf{M}} = \mathbf{U}^{-1}(l) \int_{0}^{l} \mathbf{1}(\tau) x(\tau) d\tau, \qquad (6.8.59)$$

by which it is possible, according to (6.8.43), to express the value of $\lambda(t, u)$:

$$\hat{\lambda}(l, \mu) = f^{+}(l)\mu = \int_{0}^{1} \left[\frac{1}{K}f^{+}(l)U^{-1}(l)f(\tau)\right] Kx(\tau)d\tau. \qquad (6.8.60)$$

Comparing (6.8.56) with (6.8.60), we note that for a large time of observation the operator of smoothing studied earlier coincides with the operator obtained by the maximum likelihood method. If we have the a priori distribution of μ , the estimates should be formed by the method of maximum a posteriori probability. Analogously to how we obtained a solution for (6.8.60), we can obtain a solution coinciding with (6.8.51) for any time of observation. This circumstance established one more interconnection between the theory of optimum filtration and the theory of the maximum likelihood estimate.

The given results on measurement of coordinates of objects with a ballistic law of motion do not give, however, an exhaustive solution of the problem. The fact is that when accounting for different disturbing factors the differential equations describing variation of the parameters in time become nonlinear, and an exact law of motion in the form of elementary functions cannot be obtained from them. In these conditions finding of functions $\overline{\lambda(t)}$ and $f_i(t)$ is hampered, and simultaneously with the problem of smoothing one should somehow solve the problem of current integration of equations of motion. Simple rules of carrying out the operation, especially in complicated cases, are absent here, and the important problem of synthesis of a convenient algorithm of smoothing, ensuring minimum errors and easily realized technically, remains open. Furthermore, complication of such algorithms occurs due to the need for accounting for additional components of errors (nee § 6.2). Regardless, however, of the complexity in construction of the algorithm of smoothing, sometimes hindering direct use of the above mentioned results, the latter nevertheless can also be used in:complicated cases if we are tabling about calculation of errors of measurement. In this calculation less accuracy of calculation than during synthesis of smoothing circuit: is permissible, so that the equation of motion can be integrated approximately, after which we conduct expandion of type (6.8.43) and find errors of measurement by formula (6.8.57).

The considered case of a quasi-regular parameter easily permits us to study, too, the case of self-tuning of smoothing direcuits for correction of unequal accuracy of measurements, or which is the same, the influence of parametric fluctuations. Let us analyze the most simple case of one unknown parameter. We note that equation (6.8, 45) in variable K(t) is replaced according to (6.6, 31) by

$$c(t, \tau) + a_0^2 \int c(t, s) f(s) K(s) ds f(\tau) = a_0^2 f(t) f(\tau).$$

As an equation with degenerate kernel, it is easily solved:

$$F(t, \tau) = \frac{c_0^2 f(t) f(\tau)}{1 + \sigma_0^2 \int_{0}^{t} f^{\tau}(s) K(s) ds},$$

from which, using equation (6.6.32), we obtain

$$g(l, \tau) = \frac{e_0^2 f(l) f(\tau)}{1 + e_0^2 \int_{l_0}^{2} f^2(s) K(s) ds}.$$

Error of measurement should be calculated according to (6.6.43) by averaging (6.8.62) over random variables, determined by input signal y(t) the random factor. We note that randomness of $c(t, \tau)$ is connected only with function $\xi(t)$, which is a random of modulation factor K(t):

$$K(t) = K(1 + \xi(t)), \quad K = \overline{K(t)}.$$

According to § 6.6 $\xi(t)$ has the properties of white moise:

$$\overline{(t_1)} \, \overline{\xi}(t_2) = \varkappa \delta(t_1 - t_2).$$

(6.8.64)

3. 11)

(6.8.62)

(6, ⁸. ·

Considering fluctuations of the gain factor of the discriminator small, it is possible to expand (6.8.62) in powers of $\xi(t)$ and after averaging to obtain

-176-

$$\sigma_{suz}^{2}(l) \approx \frac{\sigma_{0}^{2} f^{2}(l)}{1 + \sigma_{0}^{2} K \int_{I_{0}}^{l} f^{2}(s) ds} \left\{ 1 + \frac{\sigma_{0}^{4} K^{2} \int_{I_{0}}^{l} f^{4}(s) ds}{\left[1 + \sigma_{0}^{2} K \int_{I_{0}}^{l} f^{2}(s) ds \right]^{2}} \right\}, \quad (6.8.65)$$

which for a large time of observation gives

$$\sigma_{our}^{2}(l) \approx \frac{f^{2}(l)}{K \int_{l_{0}}^{l} f^{2}(s) ds} \left[1 + \frac{s \int_{l_{0}}^{l^{2}} f^{3}(s) ds}{\left(\int_{l_{0}}^{l} f^{2}(s) ds \right)^{2}} \right]. \quad (6.8.66)$$

Consequently, parametric fluctuations increase error of measurement. However, this increase is much more considerable if we do not allow for these fluctuations during synthesis of the circuit, when they, in fact exist. (Let us remember that in the past these fluctuations were not accounted for either during synthesis or during calculation of accuracy.) Then calculation of accuracy of measurement conducted by the method of § 6.2 gives

$$\sigma_{swx}^{2}(l) \approx \frac{e_{0}^{2f^{2}}(l)}{1 + e_{0}^{2}K\int_{l_{0}}^{l}l^{2}(s) ds} \left[1 + \frac{e_{0}^{4}xK\int_{l_{0}}^{l}l^{4}(s) ds}{1 + e_{0}^{2}K\int_{l_{0}}^{l}l^{2}(s) ds} \right]$$
(6.8.67)

or for a large time of observation

$$I_{sux}^{2}(l) \approx \frac{l^{2}(l)}{K \int_{l_{0}}^{l} l^{2}(s) ds} \left[1 + \frac{I_{0}}{\int_{0}^{l} l^{2}(s) ds} \right]$$
(6.8.68)

Inasmuch as for a constant or little-varying dimensionless function f(t) integrals

$$\frac{1}{t-t_{\bullet}}\int_{t_{\bullet}}^{t}f^{*}(s)\,ds, \quad \frac{1}{t-t_{\bullet}}\int_{t_{\bullet}}^{t}f^{*}(s)\,ds$$

are quantities of the same order, comparison of the second component in brackets in relationships (6.8.66) and (6.8.68) shows that in the case of allowance for parametric fluctuations during synthesis this addition decreases approximately according to the law $W/(t - t_0)$, and in the case of disregard it passes to a constant limit, approximately equal to $\sigma_0^2 W$. Remembering, that π and 1/K are equal to the spectral densities of the nondimensional modulation coefficient of the gain factor and equivalent input noise of an optimum ciscriminator, respectively, and producing

and at a present white second and the second of the second o

multiplication and division of $\sigma_0^2 \varkappa K$ by a certain transmission band Af, we prove that

 $e_{g}^{2} \times \mathcal{K} = \frac{e_{0}^{2} e_{map}}{e^{2}},$

i.e., increase of error of measurement is determined by the ratio of a priori variance µ to the variance of equivalent noise, multiplied by the variance of the modulation coefficient of the gain factor of the discriminator. In the case of largsignal-to-noise ratios the magnitude of (6.8.69) will be rather noticeable. Thereby, from this simple example we proved that disregard of parametric fluctuations during synthesis, i.e., rejection of an accuracy unit and self-tuning of smoothing circuits, leads to growth of errors of measurement, in general, very undesirable. This justfies the continuing interest in parametric fluctuations in the present and subsequent chapters.

6.8.4. Mixed Cases

is the most general case the measured quantity contains components of different origin, possessing therefore various correlation properties. Let us study measurement of a parameter in the form of the sum of a nondegenerate (purely random) process, for concreteness, of a stationary (see Paragraph 6.8.1) and a quasi-regular random process in the form of a linear combination of known functions with rend m factors (see Paragraph 6.8.3). In principle the results obtained can be generalized to more complicated cases.

Writing the correlation function of the stationary and quasi-regular processes in the form $r(t_1 - t_2)$ and $R(t_1, t_2) = \mathbf{f}^+(t_1)\mathbf{Mf}(t_2)$, respectively, we have for pulse response of the smoothing filter of a double-loop variant of a meter the integral eduction

$$c(t, \tau) + K \int_{t_0}^{t} c(t, s) [t^+(s) Mt(\tau) + r(s - \tau)] ds =$$

= t^+(t) Mt(\tau) + r(t - \tau).

(6.8.70)

(6.8.33)

Let us assume that the solution of (6.8.70) can be presented in the form

$$c(l, \tau) = c_0(l, \tau) + c_1(l, \tau),$$
 (6.8.71)

where $c_0(t, \tau)$ repeats the solution of the corresponding equation for a stationary process (see Paragraph 6.8.1):

$$C_{\bullet}(t, \tau) + K \int_{t_{\bullet}}^{t} C_{\bullet}(t, s) r(s - \tau) ds = r(t - \tau).$$
 (0.8.72)

-178-

Methods of solution of this equation were studied in detail above. Then for $c_1(t, \tau)$ we have equation

$$c_{1}(t, \tau) + K \int_{0}^{t} c_{1}(t, s) [t^{+}(s) Mt(\tau) + r(s - \tau)] ds =$$

= t^{+}(t) Mt(\tau) - K \int_{0}^{t} c_{0}(t, s) t(s) ds Mt(\tau). (6.8.73)

Let us assume that variance of the purely random component of the parameter is less than the variance of the quasi-regular component so that in equation (6.8.73) it is possible to seek out solution by the method of successive approximations:

$$c_{1}(l, s) = c_{10}(l, s) + c_{11}(l, s) + \dots$$
 (6.8.74)

Consequently, for the first approximation we have equation

$$c_{10}(l, \tau) + K \int_{0}^{1} c_{10}(l, s) f^{+}(s) Mf(\tau) ds = [f(l) - g_{1}(l)]^{+} Mf(\tau),$$
 (6.8.75)

where

$$q(l) = K \int_{0}^{1} c_{0}(l, s) f(s) ds$$
 (6.8.76)

is a column vector which is the result of processing of column vector f(s) by a linear filter-operator, designed for a purely random parameter.

For the following approximations we have equations of the same kind:

$$c_{1k}(l, \tau) + K \int_{l_{0}}^{l} c_{1k}(l, s) f^{+}(s) Mf(\tau) ds = -K \int_{l_{0}}^{l} c_{1(k-1)}(l, s) r(s-\tau) ds \quad (k=1, 2, \ldots).$$
(6.8.77)

Seeking the solution of equation (6.8.75), analogously to Paragraph 6.8.3, in the form $\mathbf{a}^{+}(t)\mathbf{f}(\tau)$, we have

$$c_{10}(t, \tau) = [f(t) - \varphi(t)]^{+} [M^{-1} + KU(t)]^{-1} f(\tau), \qquad (6.8.75)$$

where $\mathbf{U}(t)$, $\boldsymbol{\varphi}(t)$ are expressed by relationships (6.8.52) and (6.8.76), respectively.

For further approximation it is necessary to introduce into consideration quantity $\psi(t, \tau)$, equal to

$$\psi(t, \tau) = K \int_{0}^{t} c_{10}(t, s) r(s-\tau) ds = [f(t) - \varphi(t)]^{+} [M^{-1} + KU(t)]^{-1} K \int_{0}^{t} i(s) r(s-\tau) ds.$$

Then equation (6.8.77) for k = 1 will take form

$$c_{11}(t, \tau) + K \int_{0}^{t} c_{11}(t, s) f^{+}(s) Mf(\tau) ds = -\psi(t, \tau).$$
 (6.8.79)

Seeking its solution in the form

$$c_{11}(t, \tau) = g^+(t)f(\tau) - \psi(t, \tau),$$

as a result we have

$$K_{in}(t, \tau) = K \int_{0}^{t} \psi(t, s) f^{+}(s) ds [M^{-1} + KU(t)]^{-1} f(\tau) - \psi(t, \tau) =$$

$$= [f(t) - \psi(t)]^{+} [M^{-1} + KU(t)]^{-1} \times$$

$$\times \{K \int_{0}^{t} \int_{0}^{t} f(s_{i}) r(s_{i} - s_{i}) f^{+}(s_{i}) ds_{i} ds_{i} \times$$

$$\times [M^{-1} + KU(t)]^{-1} f(\tau) - \int_{0}^{t} f(s) r(s - \tau) ds \}.$$
(6.8.4)

Let us explain the physical meaning of the obtained solution. This is easy to do in the next approximation:

$$c(l, \tau) = c_0(l, \tau) + c_{10}(l, \tau). \qquad (c_1, 8, 81)$$

According to relationships (6.8.72) and (6.8.75) the smoothing filter consists in this case of two basic channels (Fig. 6.40). The first exactly repeats the smoothing filter for a stationary parameter. The second repeats the smoothing filter



Fig. 0.40. Smoothing circuits for the endedter in the form of a mixture of a purely random and a quasi-regular parameter: $C_0 =$ therefore for tracking the purely makes parameter: $C_{10} =$ channel for tracking the quasi-regular parameter. for a quasi-regular parameter in that part which concerns formation of estimated of unknown readom factors. However, during forming of a selinear combination of functions with measured values of coefficients known functions $f_i(t)$ decrease by quantities equal to the output responses of the first channel upon feeding $f_i(t)$ to its input. This is explained by the fact that the linear combination of known functions to some extent passes through the encorrect

Contracting the purely random process, causing necessity for correction. Existing while between f(t) and $\varphi(t)$ may differ. If the channel for tracking the stationary pured or is wide-band, it to a considerable extent also tracks the linear combinution of functions, so that estimated values of factors will be used only for compensation of dynamic errors of the first channel. In the case of narrow bandwidth of the first channel its influence on the tracking channel for the quasi-regular parameter is clight, so that each of the smoothing channels is designed for its own component. Successive approximations to pulse response $c_{1k}(t, \tau)$ (k = 1, 2, ...) give a more precise definition, ever better compensating the influence of the two tracking channels on one another. Usually these additions play a small role. This is conveniently estimated in the case of a rapidly varying stationary component, when this approximation is permissible:

$$r(t-s) = \frac{\epsilon_1^2}{\Delta f_\lambda} \delta(t-s).$$

Formulas (6.8.75) and (6.8.81) in these conditions give

$$c_{\mu\nu}(t, \tau) = [f(t) - \varphi(t)]^{+} [M^{-1} + KU(t)]^{-1} f(\tau), \qquad (6.8.82)$$

$$c_{\mu\nu}(t, \tau) = -[f(t) - \varphi(t)]^{+} [M^{-1} + KU(t)]^{-1} \times \left\{ \frac{K \sigma_{\lambda}^{2}}{4 I_{\lambda}} [I + MKU]^{-1} \right\} f(\tau). \qquad (6.8.83)$$

As it is easy to prove by comparing (6.8.82) and (6.8.83) the order of $c_{11}(t, \tau)$ with respect to $c_{10}(t, \tau)$ is characterized by matrix

$$\frac{K\sigma_{\lambda}^{2}}{A_{\lambda}^{2}}[1+KMU(t)]^{-1} \rightarrow \frac{\sigma_{\lambda}^{2}}{A_{\lambda}^{2}}M^{-1}U^{-1}(t).$$

For a constant or slowing varying parameter analogously to Paragraph 6.8.3 it is possible to consider matrix $\frac{1}{t-t_0} U(t) = \left\| \frac{1}{t-t_0} \int_{t_0}^{t} f_1(s) f_1(s) ds \right\|$ little time-depen-

dent. Then for $\Delta f_{\lambda}(t-t_{\bullet}) \to \infty$ we have $c_{\lambda}^{2} M^{-1} \left[\frac{1}{t-t_{\bullet}} U(t) \right]^{-1} \times (\Delta f_{\lambda}(t-t_{\bullet})^{-1} \to 0, \text{ i.e., the influence of } c_{11}$ is practically disregarded.

In particular, for one unknown factor with variance σ_0^2

$$\frac{c_{11}(t,\tau)}{c_{10}(t,\tau)} = -\frac{K\sigma_{\lambda}^{2}/\Delta f_{\lambda}}{1+K\sigma_{0}^{2}\int_{t_{0}}^{t}f^{2}(s) ds} \approx \frac{\sigma_{\lambda}^{2}/\sigma_{0}^{2}}{\Delta f_{\lambda}(t-t_{0})\left[\frac{1}{t-t_{0}}\int_{t_{0}}^{t}f^{2}(s) ds\right]} \rightarrow 0$$

Inasmuch as here $c_0(t, \tau) = c_0(t - \tau)$ is also a rapidly varying function

$$\int_{t_0}^{t} c_0(t, s) f(s) ds = f(t) \int_{t_0}^{t} c_0(\tau) d\tau,$$

and then the channel for tracking the regular part of the parameter has response

$$C_{10}(t, \tau) = \frac{\tau_0^2 f(t) f(\tau)}{1 + K \tau_0^2} \left[1 - K \int_0^\infty C_0(\tau) d\tau \right],$$

differing from the case of absence of a stationary part only by factor $1 - K \int c_0(\tau) d\tau$. Its meaning becomes completely clear if we consider that the integral is the value of the transmission factor of the filter for a stationary parameter at zero frequency.

Error of measurement according to (6.8.82) and (6.8.83) is expressed in the general formula

$$\sigma_{g_{M2}}^{3}(t) = c_{0}(t, t) + \left[f(t) - K \int_{t_{0}}^{t} c_{0}(t, s) f(s) ds\right]^{+} \times \\ \times \left[M^{-1} + KU(t)\right]^{-1} \left\{I + K^{2} \int_{t_{0}}^{t} \int_{t_{0}}^{t} f(s_{1}) r(s_{1} - s_{0}) f^{+}(s_{2}) ds_{1} ds_{2} \times \\ \times \left[M^{-1} + KU(t)\right]^{-1} \right\} \left[f(t) - K \int_{t_{0}}^{t} c_{0}(t, s) f(s) ds\right],$$

(6.8.84)

valid in that approximation when for pulse response we use relationship (6.8.81). In particular, for rapidly varying $r(t_1 - t_2)$

 $\sigma_{aux}^{2}(t) = c_{\bullet}(t, t) + [f(t) - \varphi(t)]^{+} [M^{-1} + KU(t)]^{-1} \times \left\{ 1 + \frac{K\sigma_{\lambda}^{2}}{\Delta f_{\lambda}} KU(t) [M^{-1} + KU(t)]^{-1} \right\} [f(t) - \varphi(t)] + c_{\bullet}(t, t) + (1 - \alpha(t))^{a} \left(1 + \frac{K\sigma_{\lambda}^{2}}{\Delta f_{\lambda}} \right) \sigma_{0}^{2}(t),$

where $a(l) = \int_{0}^{l} c_0(l, s) ds$; $c_0^2(l)$ - variance of measurement of the quasi-regular part of the parameter, if it was measured separately.

Inasmuch as $(1-a^2)\left(1+\frac{Ke_{\pi}^2}{\Delta f_{\pi}}\right) \rightarrow 1$, then in this case variance of error is equal to the sum of variances during measurement separately of the two components of the parameter.

6.8.5. Smoothing Circuits for Linear Functionals of Parameters

In § 6.6 it was shown that during measurement of linear functionals of a parameter primary processing of the input mixture of signal and noise, carried out by

-182-

discriminator, remains the same as during measurement of the actual parameter. Only there are added two special filters to the smoothing circuits. One of them will form the assigned functional from the a priori mean value of the parameters, and the second is used for processing data from the discriminator output. The latter filter is the only interesting element. Its concrete form depends on what variant of meter of the basic parameter we are calculating single or double-loop. Pulse responses in these two cases are determined by relationships (6.6.106) and (6.6.108), respectively. Basically below we consider the case of a double-loop system. Naturally we consider a correlation function of the same type as in Paragraphs 6.8.1 - 6.8.3.

a) If the parameter is a stationary process, and the linear functional also possesses stationary properties, i.e., can be presented in the form

$$\mu(l) = \int_{-\infty}^{+\infty} F(l-\tau) \lambda(\tau) d\tau,$$

where t -- the last moment of observation, equation (6.6.106) for a sufficiently large time of observation will take form $(b(t, \tau) = b(t - \tau))$:

$$b(t) + K \int_{0}^{\infty} b(t-\tau) R(\tau) d\tau = \int_{-\infty}^{+\infty} F(t-\tau) R(\tau) d\tau \quad (t > 0).$$
(6.8.85)

As also in the case of measurement of the actual parameter, equation (6.8.85) is an integral equation of the Wiener-Hopf type, solution of which is found by the factoring method. Considering again

$$1 + KS(\omega) = \Psi(i\omega)\Psi(-i\omega),$$

we have the Fourier transform of function b(t) in the form

$$B(i\omega) = \frac{1}{\Psi(i\omega)} \left[\frac{\tilde{F}(i\omega) S(\omega)}{\Psi(-i\omega)} \right]_{+} =$$

$$= \frac{1}{2\pi\Psi(i\omega)} \int_{0}^{\infty} e^{-i\omega t} dt \int_{-\infty}^{\infty} \frac{\tilde{F}(iu) S(u)}{\Psi(-iu)} e^{iu t} du, \qquad (6.8.86)$$

$$\tilde{F}(i\omega) = \int_{0}^{+\infty} F(i) e^{-i\omega t} dt.$$

where

Formula (6.8.86) for finding the limiting operator is completely preserved if the parameter is a process with stationary increments. Here, during factoring of $S(\omega)$ one should remember the generalizations of Paragraph 6.8.2. Accuracy of measurement of the parameter in both cases is expressed according to (6.6.109) by a formula analogous to (6.8.18):

$$\sigma_{\mu\nu\nu}^{2} = \lim_{i \to \infty} [i \otimes B(i \otimes) \widetilde{F}(i \circ)]. \qquad (0.8.57)$$

As an example we shall consider measurement of the first derivative of the second integral of white noise, where

$$S(\mathbf{o}) = \frac{B_1}{\mathbf{o}^{\mathbf{o}}}, \quad \widetilde{F}(i\mathbf{o}) = i\mathbf{o}.$$

From formula (6.8.86) we have

$$B(iw) = \sqrt{\frac{B_2}{K}} \frac{i\omega}{(i\omega)^2 + \sqrt{2}(iw)a_1 + a_2^2}, \quad o_2^2 = KB_2.$$

Very graphic in this case is the form of the filter for forming of the functional in a single-loop system:

$$H(i\bullet) = B(i\bullet) \left[1 + KG(i\bullet)\right] = \frac{\sqrt{B_1/K}}{i\bullet}.$$

The smoothing filter here has the same form as in the case of measurement of n Wiener process by a single-loop system.

b) If the parameter is a linear combination of functions with random factors, the problem is simply solved without any assumptions about the form of the functional (it can be of any form).



from modulator Fig. 6.41. Smoothing circuits for a linear functional of a quasti-regular parameter (in the single-loop variant): 1, 3, 4 - variablegain amplifiers; 2 - integrator.

Equation (6.6.106) takes in this case the form

$$b(t_1, \tau) + K \int_{t_0}^{t} b(t_1, s) t^+(s) dsMt(\tau) =$$

= $\int_{t_0}^{t_0} F(t_1, s) t^+(s) dsMt(\tau).$

Seeking its solution in the form $b(t_1, \tau) = g(t_1, \tau)f(\tau)$, we finally have for $b(t_1, \tau)$ and $h(t_1, \tau)$

$$b(t_1, \tau) = \int_{t_1}^{t_2} F(t_1, s) f^+(s) ds [M^{-1} + KU(t)]^{-1} f(\tau), \qquad (1.8.83)$$

$$h(t_{1}, s) = \int_{1}^{t} F(t_{1}, s) f^{+}(s) ds [M^{-1} + KU(s)]^{-1} f(s), \qquad (s, B, B)$$

-184-

where U(t) is expressed by formula (6.8.52).

According to (6.8.88) and (6.8.89) the circuit for formation of estimates of proportionality factors remains the same as for measurement of the actual parameter, but then the formed set of estimates is multiplied, not by original functions $f_i(t)$, but by functions $f_i(t)$, processed by linear operator F(t, s), characterizing the linear functional. These operations are illustrated for one coefficient μ in the single-loop variant in Fig. 6.41.

Error of measurement according to formula (6.6.109) is equal to

$$\sigma_{sur}^{2}(t_{1}, t) = \int_{t_{1}}^{t_{1}} F(t_{1}, s) f^{+}(s) ds [M^{-1} + KU(t)]^{-1} f(t).$$
(6.8.90)

In the particular case of a linearly varying parameter

$$\lambda(t) = V t, \quad \overline{V^i} = \sigma_1^2$$

and an advance (delay) operator

$$\mu(t_i) = V t_i \quad (t_i \neq l),$$

we have

$$\boldsymbol{\sigma}_{\boldsymbol{\sigma}\boldsymbol{\omega}\boldsymbol{x}}^{2}\left(t_{1}, t\right) = \frac{\boldsymbol{\sigma}_{1}^{2}t_{1}^{2}}{1 + K\boldsymbol{\sigma}_{1}^{2}t^{2}/3} \rightarrow \frac{3t_{1}^{2}}{Kt^{2}} = \frac{3}{Kt}\left(\frac{t_{1}}{t}\right)^{2} \quad \text{as} \quad t \rightarrow \infty.$$

Thus, error by a factor of approximately t_1/t exceeds error of measurement of the current value of the parameter.

§ 6.9. Synthesis of Meters with a Markovian Parameter and with Limited Knowledge of the Statistics of the Parameters

يسادينيا أتفاري سأناوشها بالشاقين سماليا والمسو

In §§ 6.6-6.8 we considered questions of synthesis of radar meters for a Gaussian distribution of the measured quantity $\lambda(t)$. This case is very interesting, but is not all-embracing. Therefore results obtained with other assumptions about the statistics of $\lambda(t)$, and also with limited knowledge of the statistics attract our attention. In the class of completely assigned a priori distributions besides the Gaussian case in the literature there is considered the case of synthesis for Markovian parameters, which we touched on in § 6.5. For the case of limited a priori knowledge investigations are extremely scanty (see, for instance, [62]). Below we give certain results on the synthesis of meters with a Markovian parameter and with limited knowledge of statistics of the parameter, generalizing and supplementing the known works. Results of solution with limited statistics are far from complete and can only serve as a basis for further investigations.

6.9.1. Optimum Meter of a Markovian Parameter

Passing to synthesis of a meter of a Markovian parameter, it would have been possible to compose its a priori distribution for all moments of observation and, multiplying by the Gaussian approximation of likelihood function (6.6.3), find the conditional mathematical expectation. More convenient, however, will be methods, based on the results of Paragraph 6.5.6.

We shall start with the case of discrete observation (or of an incoherent pulse signal), when it is permissible to use relationship (6.5.24). Assuming low a posteriori inaccuracy we approximate the a posteriori probability density of the value of $\lambda(t)$ at time t_{μ} of the Gaussian curve

$$\widetilde{P}_{n}(\lambda_{n}) = -\frac{1}{\sqrt{2\pi}e_{\mu\mu\lambda_{n}}} \exp\left\{-\frac{(\lambda_{n}-\overline{\lambda}_{n})^{2}}{2\sigma_{\mu\mu\lambda_{n}}^{2}}\right\}, \qquad (6.1.1)$$

where $\sigma_{\text{BHX},n}^2$ - a posteriori variance at the n-th moment;

 $\hat{\lambda}_n = \text{point of maximum a posteriori probability, according to results of Paragraphs 6.5.3 and 6.5.6 coinciding with the current optimum estimate.$

We also take the Gaussian approximation for likelihood function

$$P(y_h | \lambda_h) = P(y_h | \dot{\lambda}_h) \exp\left\{-\frac{(\lambda_h - \dot{\lambda}_h)^*}{2^{\sigma_h^2}}\right\}, \qquad (6.2.2)$$

where, analogously to (6.6.4), we expand the maximum likelihood of $\lambda_{\rm k}$ at the current

-186--

moment, and

$$\boldsymbol{\theta}_{A}^{2} = \left[-\frac{\partial^{2} L\left(\tilde{\boldsymbol{\lambda}}_{A}\right)}{\partial \boldsymbol{\lambda}^{2}} \right]^{-1}$$
(6.9.3)

- variance of error of the k-th measurement.

Finally, we shall consider the parameter a Markovian process of diffusion type, for which in small time intervals Δ the transition function is expressed by a Gaussian curve

$$W(\lambda_{A+1}|\lambda_{A}) = \frac{1}{\sqrt{2\pi\sigma}} \exp\left\{-\frac{[\lambda_{A+1}-\lambda_{A}-\sigma(\lambda_{A})]^{2}}{2\sigma_{A}^{2}(\lambda_{A})}\right\}, \quad (6.9.4)$$

where $a(\lambda_k)$, $\sigma_{\lambda}^2(\lambda_k)$ are determined through the coefficients of drift $A(\lambda)$ and diffusion $B(\lambda)$ in the form

$$a(\lambda) = A(\lambda) \Delta, \quad s_{\lambda}^{2}(\lambda) = B(\lambda) \Delta, \quad \Delta = t_{k+1} - t_{k}$$

and have the physical meaning of systematic bias and variance of random change of the parameter in the interval Δ between two successive measurements. In view of the low a posteriori inaccuracy, functions $A(\lambda)$ and $B(\lambda)$ vary little within the bounds of the width of the a posteriori peak, so that it is possible to consider $B(\hat{\lambda}) \approx B(\lambda)$ and

$$A(\lambda) \approx A(\hat{\lambda}) + A'(\hat{\lambda})(\lambda - \hat{\lambda}). \qquad (6.9.5)$$

The essential difference from the case of a Gaussian parameter is the fact that the Gaussian approximation for the transition function $W(\lambda|\mu)$ is valid only for small intervals between the moments of precise definition of values of $\lambda(t)$ and that variance and the mean value of $W(\lambda|\mu)$ even in the Gaussian approximation depend on the concrete value of the measured quantity. Substituting in these conditions (6.9.1), (6.9.2) and (6.9.4) in (6.5.24), integrating and equating coefficients for different powers of λ_{n+1} in logarithms of the right and left members, we obtain

$$\frac{\hat{\lambda}_{n+1}}{\sigma_{\text{MMX}(n+1)}^2} = \frac{\hat{\lambda}_n + \sigma(\hat{\lambda}_n)}{Q^2(\hat{\lambda}_n)\sigma_{\text{MXX}(n+1)}^2 + \sigma_{\lambda}^2(\hat{\lambda}_n)} + \frac{\check{\lambda}_{n+1}}{\sigma_{n+1}^2}, \qquad (6.9.6)$$

$$\frac{1}{\sigma_{\text{BMX}(n+1)}^2} \frac{1}{Q^2(\hat{\lambda}_n)\sigma_{\text{BMX},n}^2 + \sigma_{\lambda}^2(\hat{\lambda}_n)} + \frac{1}{\sigma_{n+1}^2}.$$
 (6.9.7)

As (6.9.6) shows, the (n + 1)-th estimate will be formed by the weighted addition of the n-th estimate, anticipated one step with the help of component $a(\hat{\lambda}_n)$, and the result of the newly performed measurement $\check{\lambda}_{n+1}$. Weights are determined for the first component by the a priori variance at the n-th moment, increased by the variance of the expected measurement of the parameter, and for the second component - by variance of the (n + 1)-th measurement. According to (9.6.7) the sum of these weights determines variance of the a posteriori distribution in the (n + 1)-th step. We intentionally did not make a distinction between variance, calculated according to $(\ell, 9.6)$ and (6.9.7) and true a posteriori variance, since with an optimum procedure of measurement they coincide with high accuracy. Here, it is possible to consider $o_{\lambda}^{2}(\hat{\lambda}) = o_{\lambda}^{2}(\lambda_{N}), Q(\hat{\lambda}) = Q(\lambda_{N})$, where λ_{N} — true value of the parameter.

It remains only to explain the appearance in front of the expression for variance of measurement on the preceding step $\sigma_{\text{BHX } h}$ of factor $Q^2(\hat{\lambda_{h}})$, where

$$Q(\hat{\lambda}_{n}) = 1 + \frac{\partial a(\hat{\lambda}_{n})}{\partial \lambda_{n}} = \frac{\partial}{\partial \lambda_{n}} \left(\lambda_{n} + A_{n}(\lambda_{n})\Delta\right)\Big|_{\lambda_{n} = \hat{\lambda}_{n}}$$
(6.9.8)

is a partial derivative of the value of the parameter extrapolated to the (n + 1)-th moment with respect to the value of the parameter at the n-th moment. This factor is needed because systematic change of parameter $A(\lambda)\Delta$ depends on λ , and depending upon its value inside the peak of the a posteriori probability parameter it will obtain various mean increments in the interval of extrapolation. Change of variance of the expected value is proportional to variance in the preceding step. Variance increases if to the large coordinate there corresponds (algebraically) high speed, since this leads to ever-increasing dispersion, and conversely. Formally, we calculate increment $\Delta\lambda_n$ to increment $\Delta\lambda_{n+1}$ with the help of partial derivative (6.9.8).

Now, analogously to (6.6.3), we expand the likelihood function at a point of estimation anticipated one step $\hat{\lambda}_n + a(\hat{\lambda}_n)$ [borore carrying out the (n + 1)-th measurement no more successful approximation to the value of the parameter λ_{n+1} exists]:

$$P(y_{n+1}|\lambda_{n+1}) = P(y_{n+1})|\bar{\lambda}_n + a(\bar{\lambda}_n)) \times \\ \times \exp\left\{z_{n+1}(\lambda_{n+1} - \bar{\lambda}_n - a(\bar{\lambda}_n)) - \frac{(\lambda_{n+1} - \bar{\lambda}_n - a(\bar{\lambda}_n))^2}{2\sigma_n^2}\right\}.$$
(6.9.9)

Here.

$$\mathbf{z}_{n+1} = \frac{\partial}{\partial \mathbf{x}} L(\hat{\mathbf{x}}_n + a(\hat{\mathbf{x}}_n)) \tag{6.9.10}$$

is the output signal of an optimum discriminator, and

$$\mathbf{s}_{g}^{2} = \left[-\frac{\partial^{2}L\left(\hat{\lambda}_{n} + a\left(\hat{\lambda}_{n}\right)\right)}{\partial \lambda^{2}}\right]^{-1}$$
(6.9.11)

is again variance of the n-th measurement, practically coinciding with (6.9.3). Substituting (6.9.1), (6.9.4), (6.9.5) and (6.9.6) in (6.5.24) and performing calculations, we have

$$\hat{\lambda}_{n+1} = \hat{\lambda}_n + a(\hat{\lambda}_n) + \sigma_{\text{max}(n+1)}^2 z_{n+1},$$
 (6.9.12)

where $\sigma_{\text{BHX}\ (n+1)}^2$ is given by relationship (6.9.7). The algorithm of formation of the (n + 1)-th estimate in this case is still simpler: to the expected value we add the output signal of the discriminator with a weight equal to the current a posteriori variance. The more exactly smoothed the data, the less we use the new information z_{n+1} , on the average proportional to current mismatch.

In both considered cases circuits of estimate formation are simple and possess the convenient quality that in the next step, besides the result of the new unit measurement, it is necessary to know only two magnitudes stored in the memory from the preceding measurement: the former result and its variance. In every step of the operation measurements are conducted by uniform recursion formulas. This is a result of the assumption that the measured parameter is a Markovian process of the first order.

Let us consider how the above mentioned relationships change with transition to continuous observation. Remembering the definition of the output signal z(t) of an optimum "continuous discriminator" (6.6.26), taking into account (6.9.4) we can rewrite (6.9.12) in the form

$$[\hat{\lambda}(t+\Delta) - \hat{\lambda}(t)]/\Delta = A(\hat{\lambda}(t)) + \gamma_{\text{star}}^2(t+\Delta) \frac{1}{\Delta} \int_{t}^{t+\Delta} Z(\tau) d\tau.$$

from which, passing to the limit $\Delta \rightarrow 0$, we have

$$\frac{d\bar{\lambda}(t)}{dt} = A(\hat{\lambda}(t)) + \sigma_{\text{obst}x}^{2}(t) z(t). \qquad (6.9.13)$$

Considering relationship (6.9.4) and the fact that $\sigma_n^{-2} = K(t_n) \Delta$, it is possible, analogously, to transform (6.9.7):

$$\frac{d\sigma_{BMX}^2(t)}{dt} = B(\hat{\lambda}(t)) - K(t) \sigma_{BMX}^4(t) - A'(\hat{\lambda}(t)) \sigma_{BMX}^2(t). \qquad (6.9.14)$$

The first components in (6.9.13) and (6.9.14) are the a priori known components of the rate of change of the parameter and variance of its random component, respectively. Second components reflect the influence on the estimate and its variance of newly arriving data. Let us note that in (6.9.14) the first component is always positive, and the second is always negative. Conditions with constant (or close to constant) variance occur with approximate equality of these two components, since the third component normally has little influence. Then the newly arriving data approximately compensate the decrease in accuracy due to change of the parameter.

The set of relationships (6.9.13) and (6.9.14) is simulated by the block diagram of Fig. 6.42. The input realization of y(t) is fed to discriminator 1 and accuracy



Fig. 6.42. Closed-loop optimum meter for a Markovian parameter; 1 - discriminator; 2 - accuracy unit; 3, 7, 9 - multipliers; 4, 12 - adders; 5, 10, 11 - nonlinear converters; 6, 13 - integrators; 8 - square-law generator. unit 2, the functions of which were already explained. Smoothing is produced by ratner complex circuits, in which it is possible to separate two basic groups of elements (divided by the dotted line). يموليان هو ياجي منهمان من من هذه هذه من المريخ المنهمان المالية المنهمان عن منهمان من المن من المن من المنام عن م

The first group smooths estimate $\lambda(t)$ and consists of multiplier 3, adder 4, a nonlinear converter 5 with "practeristic $A(\lambda)$ and integrator 6, directly carrying out accumulation. The second group smooths the measure of a posteriori variance $\sigma_{BHX}^2(t)$ and consists of multipliers 7, 9, square-law generator 8, nonlinear converters 10, 11 with characteristics $A'(\lambda)$ and $B(\lambda)$, adder

12 and integrator 13. The second group plays the auxiliary role of controlling the gain factor in 3 during smoothing of the estimate. The measured value $\hat{\lambda}(t)$ is fed to 1 and 2 to maintain selection. As we see, in distinction from a Gaussian parameter, the smoothing networks are nonlinear, but the general idea of smoothing in a closed loop is preserved here.

Transition to the continuous case in relationship (6.9.6) will lead to equation

$$\frac{d\hat{\lambda}(t)}{dt} = A(\hat{\lambda}(t)) + a_{\mu\nu\tau}^{2}(t) K(t) [\dot{\lambda}(t) - \hat{\lambda}(t)]. \qquad (6.9.15)$$

Operations, determined by (6.9.13) and (6.9.15), can be performed by the block diagram of Fig. 6.43. The nonlinear estimator unit 1 issues the maximum likelihood estimate $\hat{\lambda}(t)$. Further circuits again are divided into two groups. The first, intended for smoothing the estimate, consists of subtractor 2, two multipliers 3, 4, adder 5, nonlinear converter 6 and integrator 7. Unit 3 controls gain in the smoothing oop taking into account current unequal-accuracy of measurements, taking with greater weight the segments of the realization with a high signal level. Unit 4, as in the scheme of Fig. 6.42, decreases gain with increase of resultant accuracy. In everything else this group of elements, as also the second group, smoothing the a posteriori variance, repeats part of the scheme of Fig. 6.42. It is important only to note

-190-

that closing of the meter by the estimator unit is not produced. This again permits



Fig. 6.43. Open-loop optimum meter for a Markovian parameter: 1 - estimator unit; 2 - subtractor; 3, 4, 8, 10 - multipli-ers; 5, 13 - adders; 6, 11, 12 - non-linear converters; 7, 14 - integrators; 9 - square-law generator.

us to call such a variant of a meter open-loop.

Comparing (6.9.15) and (6.9.14) with (6.5.35) and considering that according to definitions $\iota'(\lambda_0, t) = z(t)$, $\iota''(\lambda_0, t) = -K(t)$, we with interest prove the coincidence of these equations. Although we did not use diffusion equations, the equations for characteristics of the a posteriori distribution during continuous observation were the same as in the works of R. L. Stratonovich [16, 17], by virtue of using the same approximations.

It would have been possible to continue our investigation here and consider a parameter in the form of a Markovian process of the k-th order. However, this leads to a problem so similar to measurement of several Markovian parameters that it is better to postpone its investigation until Chapter XII.

6.9.2. Synthesis of a Meter with Statistics Unknown. Minimax Solution.

The theory of statistical solutions usually has something to do with two extreme cases — complete a priori knowledge and complete a priori ignorance. The first of these cases was considered in detail above. In the last case synthesis of optimum resolvers is carried out on the basis of the minimax principle.

In reference to the problem of synthesis of an optimum meter we will consider as the case of complete a priori ignorance such a case, when with respect to function $\lambda(t)$ we know only that it remains practically constant for certain intervals of time Δ . We have here no information about the law of change of the measured parameter from one discrete value λ_4 to another $\lambda_{4\pm 4}$.

The minimax solution is constructed in the following way. For every possible estimate $\hat{\lambda}(\mathbf{y})$ we determine conditional risk as a function of λ , its maximum for all λ , and then select such an estimate $\hat{\lambda} = \hat{\lambda}_{M}$, for which this maximum would be minimum. The minimax estimate is determined by relationship

 $r(\hat{\lambda}_{\mathbf{x}}, \lambda) = \min_{\boldsymbol{\lambda}} \max r(\hat{\lambda}, \lambda).$ (6.9.16)

-191-

The character of its optimality consists in the fact that it minimizes the maximum value of conditional mathematical expectation of the loss function.

At present direct methods of construction of minimax solutions have not been developed, and finding them is based on Walde's [?] theorem [63]. The minimax solution is the Bayes solution relative to a certain a priori distribution, which gives a maximum magnitude of Bayes risk, i.e., satisfies relationship

$$R(\hat{\lambda}_{\bullet}, P_{\bullet}) = \max_{\substack{(P_{\bullet})\\ P_{\bullet}}} R[\hat{\lambda}_{\bullet}(P_{\bullet}), P_{\bullet}], \qquad (6.9.17)$$

يابي الاعتمالية مساعده والم

where $\hat{\lambda}_{\vec{0}}(P_0)$, $\hat{\lambda}_{\vec{0}}^*$ are the Bayes estimates with respect to a priori distributions $P_0(\lambda)$ and $P_0^*(\lambda)$, and gives conditional risk with a quantity not depending on λ for all values of λ which have, according to distribution $P_0^*(\lambda)$, non-zero a priori probability. The actual distribution with density $P_0^*(\lambda)$ is called the least preferable, and $\hat{\lambda}_{\vec{0}}^*(\mathbf{y}) = \hat{\lambda}_{\vec{M}}(\mathbf{y})$ is the minimax estimate, satisfying (6.9.17). This theorem, especially its second part, has fundamental importance, although its effective use for producing minimax solutions is rather difficult.

Let us consider one important case. Let us assume that the loss function is simple (6.5.5). Let us consider the Bayes estimate for a "uniform" a priori distribution. Mean risk in this case will be defined as

$$R(\hat{\lambda}, P_{\mathbf{j}}) = C - \int P(\mathbf{y} \mid \hat{\lambda}) d\mathbf{y}, \qquad (6.7.18)$$

From which it follows that the optimum estimate $\hat{\lambda}(\mathbf{y})$ is that value of λ which turns likelihood function $P(\mathbf{y}|\mathbf{\lambda})$ into a maximum. Thus, we arrive at the maximum likelihood estimate for vector parameter $\mathbf{\lambda}$, which with observance of conditions of analyticity is determined from the system of likelihood equations

$$\frac{\partial}{\partial \lambda_i} P(\mathbf{y} | \boldsymbol{\lambda}) = 0, \ (i = 1, ..., n).$$

If, then, the conditional risk $r(\hat{\lambda}, \lambda)$ for the maximum likelihood estimate turns out not to depend on λ , this estimate on the basis of the Walde theorem is the minimax estimate. It is not difficult to prove that the sufficient condition of this is the possibility of presenting the likelihood function in the form

$$P(\mathbf{y}|\boldsymbol{\lambda}) = F(\boldsymbol{\lambda} - \boldsymbol{\lambda}_{\mathbf{y}}) F_{\mathbf{x}}(\mathbf{y}), \qquad (6.9.19)$$

where $\boldsymbol{\lambda}_{O} = \boldsymbol{\lambda}_{O}(\boldsymbol{y})$;

 $F(\mathbf{x})$ - an even positive function of the vector argument;

 $F_1(\mathbf{y})$ - a function integrable with respect to \mathbf{y} , which by virtue of $P(\mathbf{y}|\mathbf{\lambda}) \ge 0$ and $F(\mathbf{\lambda} - \mathbf{\lambda}_0) \ge 0$ is nonnegative. Then the maximum likelihood estimate is equal to $\hat{\lambda}_{M}(\mathbf{y}) = \lambda_{O}(\mathbf{y})$. Furthermore, vector $\lambda_{O}(\mathbf{y})$ is the minimax estimate for any symmetric loss function, which is proved analogously to Paragraph 6.5.3.

Thus, the minimax solution of the problem of filtration under certain conditions consists of formation of a set of maximum likelihood estimates for values of λ_i at intervals of constancy of measured parameters Δ . Condition (6.9.19) leads actually to the existence of efficient estimates for parameters λ_i at intervals of time $[(t_0 + (i - 1)\Delta, t_0 + i\Delta)]$ of duration Δ . Practically, of course, it is sufficient to obtain approximate efficiency and, correspondingly, the possibility of presenting the likelihood function in the form of (6.9.19) with a satisfactory degree of approximation.

Let us consider one useful example. Let us assume that at the resolver input there is an additive mixture of the useful parameter $\lambda(t)$ and interference n(t):

$y(t) = \lambda(t) + n(t),$

and the loss function is quadratic. Such a mixture can be considered the output signal of the discriminator of the meter with which we sum the measured value of the parameter. Conditions of the validity of such a presentation were explained above. Then the optimum estimate of vector λ for any a priori distribution has the form

$$\hat{\lambda} = \frac{\int \lambda P(y|\lambda) P_{\bullet}(\lambda) d\lambda}{\int P(y|\lambda) P_{\bullet}(\lambda) d\lambda} = \frac{\int \lambda P(y-\lambda) P_{\bullet}(\lambda) d\lambda}{\int P(y-\lambda) P_{\bullet}(\lambda) d\lambda}, \qquad (6.9.20)$$

where $P(\mathbf{y}|\boldsymbol{\lambda}) = P(\mathbf{y} - \boldsymbol{\lambda}) - density$ of distribution of vector **n**.

Let us assume that a priori distribution $P_0(\lambda)$ is "uniform" in the whole infinite region of existence of λ . Then, replacing the variable of integration in (6.9.20) $y - \lambda = n$, we obtain

$$\hat{\lambda} = \mathbf{y} - \frac{\int \mathbf{n} P(\mathbf{n}) \, d\mathbf{n}}{\int P(\mathbf{n}) \, d\mathbf{n}} = \mathbf{y} - \overline{\mathbf{n}}, \tag{6.9.21}$$

states on shore is a still a state of the state of the states of the sta

i.e., estimate λ , with accuracy of a constant component not depending on \mathbf{y} , coincides with signal \mathbf{y} . Conditional risk for estimate (6.9.21) with uniform distribution is equal to

$$r(\hat{\lambda}, \hat{\lambda}) = \int (\hat{\lambda} - \hat{\lambda})^+ (\hat{\lambda} - \hat{\lambda}) P(\mathbf{y} - \hat{\lambda}) d\mathbf{y} =$$

= $\int (\mathbf{n} - \overline{\mathbf{n}})^+ (\mathbf{n} - \overline{\mathbf{n}}) P(\mathbf{n}) d\mathbf{n} = \text{spur } \mathbf{r},$

(where $\mathbf{r} - \text{correlation matrix of interference}$) and turns out not to depend on λ . Consequently, the solution of (6.9.21) is the minimax solution of the problem of filtration of signal $\lambda(t)$ from an additive mixture with interference n(t).

-193-

The obtained solution has a clear physical meaning. Actually, in the absence of any a priori information about $\lambda(t)$, realization of the input signal as estimate $\hat{\lambda}(t)$ will be the best that can be offered. Error of measurement here, as it should be, is simply equal to variance of interference n(t). Using the a priori assumed constancy of $\lambda(t)$ in intervals of length Δ , it is possible to decrease error of measurement somewhat. Nevertheless, the conclusion that in the absence of any information about trajectory properties of the target effective smoothing cannot be realized remains valid, and error of unit measurements are wholly recalculated into resultant errors.

6.9.3. Description of a Parameter with Limited Knowledge of Its Statistics

Assignment of a multi-dimensional distribution of probabilities of $\lambda(t)$ requires sufficiently detailed statistical evidence about the law of change of target position data. Frequently in practice such information is inaccessible; however certain statistical characteristics of $\lambda(t)$ nevertheless exist. Such a case is intermediate between the above-considered cases of complete a priori knowledge and complete a priori ignorance. With incomplete statistical description there appears a very great variety of cases, which in general reduce to the fact that we assign certain limitations on characteristics of $\lambda(t)$. The latter permit us to determine a certain set Λ of permissible functions $\lambda(t)$ and subsequently in problems of analysis and synthesis to limit ourselves to consideration of only such $\lambda(t)$ as belong to this cet $(\lambda(t) \in \Lambda)$. These limitations can be given in statistical form, and also in normal form.

We first consider the first case.

The simplest characteristic of $\lambda(t)$ is mathematical expectation $\overline{\gamma(t)}$. In most problems it is naturally assumed assigned. A more detailed description will be attained if we assign the variance at any moment of time and the moments of highest order right up to assignment of a one-dimensional law of distribution of probabilities. Further precise definition of the behavior of $\lambda(t)$ occurs with assignment, for instance, of variances of derivatives of $\lambda(t)$ up to a certain order at any moment of time. All these characteristics are more or less available in practice.

Very important both from the practical point of view, and also from the point of view of theory, is the case when there is assigned the correlation function of $\lambda(t)$. As we shall subsequently prove, under certain conditions of its assignment it is atready sufficient for synthesis of an optimum meter. A still more detailed

-194-

description of $\lambda(t)$ is attained with assignment of moment functions of higher order. Limitations of this type in general permit us to assign a set Λ of permissible realizations of $\lambda(t)$ as the set described by the a priori distribution of probabilities $P_{\Omega}(\lambda)$, obeying conditions

$$\int P_{\bullet}(\lambda) d\lambda = 1, \quad \int f(\lambda) P_{\bullet}(\lambda) d\lambda = \mu, \quad (6.9.22)$$

and otherwise being arbitrary. In (6.9.22) integrals are taken over the whole range of values of λ : $\mathbf{f}(\lambda)$ - certain multicomponent function of λ ; μ - the assigned value of the same structure. In the particular case when we are assigned variance of $\lambda(t)$ at any moment of time t_1 , we take $f_1(\lambda) = \lambda_1^2$, $\mu_1 = \sigma_1^2$, and $\overline{\lambda}_1$ for simplicity is considered equal to zero. With assignment of the correlation matrix we assume

$$i_{ij} = \lambda_i \lambda_j, \quad \mu_{ij} = R_{ij} \ (i, j = 1, ..., n), \quad (6.9.23)$$

with assignment of moments of the third order $f_{i,ik} = \lambda_i \lambda_j \lambda_k$, $\mu_{i,ik} = \lambda_i \lambda_j \lambda_k$, etc.

Other examples of partial statistical description are cases when multi-dimensional laws of distribution of probabilities are assigned with an accuracy to certain unknown parameters, or there are assigned distributions of probabilities of lowest orders - first, second, etc. Of special importance is the a priori comption of a Markovian character of change of coordinates without indication of the form of the initial distribution and transition probabilities. Actually, movement of radar targets is described by second order differential equations, so that the assumption of a Markovian process of the second order corresponds to ignorance of the concrete form of the equation determining the form of transition probabilities, and characterizes that minimum a priori information which we always have.

Sometimes statistical evidence about the character of change of $\lambda(t)$ is completely lacking. So that the problem of measurement has meaning, it is necessary to impose on $\lambda(t)$ some limitations, determining the set Λ of permissible realizations of $\lambda(t)$. The simplest limitation of this type is assignment of a certain interval of time Δ , during which $\lambda(t)$ remains practically constant, so that $|\lambda(1 + \Delta) - \lambda(1)| << \lambda(t)$ (see Paragraph 6.9.2). Analogous intervals can be assigned with respect to derivative $\frac{d\lambda(t)}{dt}$, etc. More detailed limitation is obtained if we know that $\lambda(t)$ and certain of its derivatives are limited by certain limits of variation, i.e.,

$$f_{k}(l) < \frac{d^{k}\lambda}{dl^{k}} < \varphi_{k}(l) \ (k = 0, 1, 2, ...),$$
 (6.9.24)

where $f_k(t)$, $\phi_k(t)$ - assigned functions.
Such a method of assignment is frequently used in practice. From it, in particular, we can obtain necessary intervals of constancy Δ for λ (t) and a certain quantity of derivatives.

In a number of problems there is possible still more detailed assignment of $\lambda(t)$. For instance, during measurement of coordinates of a ballistic target it is possible to assign $\lambda(t)$ with the help of known functions depending on certain unknown parameters, which in distinction from § 6.8 have an unknown probability distribution. Equivalent to this case is assignment of $\lambda(t)$ with the help of differential equations depending on a series of unknown parameters (for instance, unknown initial conditions). The difference is only in the fact that frequently solution of these equations and explicit expressions for $\lambda(t)$ cannot be obtained.

and the second of the second second second

0.9.4. Syntnesis of a Meter with Limited Knowledge of the Statistics Along with the Bayes solution, giving as it were an absolute optimum, of extraordinary interest from the practical point of view is solution of the problem of synthesis of a meter, optimum in conditions of limited a priori knowledge. Problems or this type in the theory of statistical solutions have generally not been formulated. A certain exception is the Wiener theory of filtration and its nonlinear generalisation in the works of Zadeh [15], in which the optimum operator of filtration is found when there are assigned only correlation functions of $\lambda(t)$ and n(t); however this operator is considered beforehand to belong to an assigned class - linear in Wiener theory - and a definite type of nonlinear ones in [15]. This last limitation naturally causes criticism, since it remains vague how close the synthesized circuits are to indeed optimum operators which one should obtain from the general theory of solutions without preliminary assignment of their structure. Certain results relative to solution of problem of synthesis with limited a priori knowledge are contained in [62]. Here, along with them, we will try to develop a general formulation and consider other particular cases.

Thus, let us assume that about $\hat{\lambda}(t)$ we have certain a priori statistical evidence, for instance of the type considered in Paragraph 6.9.4, which permits us to assign the set of permissible realizations of $\lambda(t)$. This set is described by an a priori probability distribution with density $P_0(\lambda)$, belonging to a given class of distributions, which we designate by P_0 . A particular case is the total absence of the tistical evidence, when P_0 is the set of all normalized, positive, integrable in an infinite range, values λ of the function.

-196-

The rule for finding estimates $\hat{\lambda}(\mathbf{y})$ in this case can be ontained by proper generalization of the rule based on the least preferred distribution. Here, we naturally consider that the least preferred distribution also obeys the limitations, and we seek it among distributions of class \mathbf{P}_0 . The general rule for selection of the optimum solution can be formulated as follows:

$$R(\hat{\lambda}_{s}^{\bullet}, P_{0}^{\bullet}) = \max \min R_{\delta}(\hat{\lambda}_{\delta}(P_{\bullet}), P_{\bullet}), \qquad (6.9.25)$$

where $P_{O}(\lambda)$ - distribution, belonging to the assigned class $P_{O}(P_{O} \in P_{O})$;

 $P_0^*(\lambda)$ - least preferred distribution of this class; $\hat{\lambda}_6^*$ - Bayes estimate with respect to $P^*(\lambda)$.

During construction of the solution of equation (6.9.25) we first for any density $P_0(\lambda) \in P_0$ find the Bayes estimate $\hat{\lambda} = \hat{\lambda}(P_0)$, which minimizes mean risk and is a functional of P_0 . Then by selection of function $P_0 = P_0^*$ we maximize mean risk, calculated for this estimate. This function P_0^* is the density of the least preferred distribution from class P_0 . Substituting it in the expression for estimate $\hat{\lambda}(P_0)$, we obtain the optimum solution $\hat{\lambda}(P_0^*) = \hat{\lambda}_5^*$.

The direct method of finding the least preferred distribution and the corresponding solution, as we can see, is extraordinarily difficult since it requires skill to solve the Bayes problem for a sufficiently broad class of partially assigned a priori distributions and to calculate for every solution the mean risk. However, in a number of cases it is possible to formulate the rule for finding the least preferred distribution by generalizing the rule, which follows from the the Walde theorem (Paragraph 6.9.2) for the case of complete a priori ignorance.

Let us assume that some statistical characteristics of $\lambda(t)$, allowing us to determine the class of assigned a priori distributions $P_0(\lambda)$, are assigned by conditions (6.9.22). Let us renumber components of function $f(\lambda)$ and of the given quantity μ so that

$f(\lambda) = \{f_1(\lambda), \ldots, f_k(\lambda), \ldots\}, \quad \mu = \{\mu_1, \ldots, \mu_k, \ldots\}, f_j(\lambda)$

are functions which depend on all or certain values of λ_i (i = 1, ..., n), and index j runs through as many values as conditions assigned.

Let us assume that $\lambda(\mathbf{y}, \mathbf{P}_0)$ is the Bayes estimate with respect to distribution $\mathbf{P}_0 \in \mathbf{P}_0$, which is the solution of equation (6.9.25). We use the Lagrange method for

finding the least preferred distribution of $P_0^*(\lambda)$. Here, function $P_0^*(\lambda)$ is determined from the condition of the maximum of functional

$$I = R(\hat{\lambda}(\mathbf{y}, P_{\mathbf{o}}), P_{\mathbf{o}}) - \mathbf{x}_{\mathbf{o}} \int P_{\mathbf{o}}(\lambda) d\lambda - \sum_{j} \mathbf{x}_{j} \int f_{j}(\lambda) P_{\mathbf{o}}(\lambda) d\lambda =$$

= $\int [\int w(\lambda - \hat{\lambda}) P(\mathbf{y}|\lambda) d\mathbf{y} - \mathbf{x}_{\mathbf{o}} - \sum_{j} \mathbf{x}_{j} f_{j}(\lambda)] P_{\mathbf{o}}(\lambda) d\lambda,$ (6.9.26)

where n, n_{j} - indefinite factors, found from conditions

$$\int P_{\bullet}(\lambda) d\lambda = 1, \quad \int f_{j}(\lambda) P_{\bullet}(\lambda) d\lambda = \mu_{j} \quad (j = 1, 2, ...). \quad (6.9, 27)$$

Calculating variation of functional $I = I(P_0)$ with respect to function P_0 , we obtain

$$\delta_{P_0} I = \int \left[\int w \left(\lambda - \hat{\lambda} \right) P(\mathbf{y} | \lambda) d\mathbf{y} - \mathbf{x}_0 - \sum_{I} \mathbf{x}_{i} f_{i}(\lambda) \delta P_0(\lambda) d\lambda - - \sum_{i=1}^{n} \int \left[\int \frac{\partial w(\lambda - \hat{\lambda})}{\partial \hat{\lambda}_i} P(\mathbf{y} | \lambda) P_0(\lambda) d\lambda \right] \delta_{P_0} \hat{\lambda}_i d\mathbf{y}, \qquad (6.9.28)$$

where $\delta_{P_0} \hat{\lambda_i}$ - variation of the vector component of the estimate with respect to distribution P_0 .

The second component in (6.9.28) turns into zero by virtue of equality $\frac{\partial w}{\partial \tilde{\lambda}_i} = -\frac{\partial w}{\partial \lambda_i}$, the evident independence of $\hat{\lambda}_i$ and, consequently, of $\hat{b}_{P_0}\hat{\lambda}_i$ from $\hat{\lambda}$ and equations (6.5.12'). Therefore, from the requirement of equality to zero of variation $\hat{b}_{P_0}^{-1}$ for an arbitrary δP_0 there follows the equation for conditional risk

$$r(\hat{\lambda}; \lambda) = \int w(\lambda - \hat{\lambda}) P(\mathbf{y} | \lambda) d\mathbf{y} = \mathbf{x}_{0} + \sum_{j} \mathbf{x}_{j} f_{j}(\lambda), \qquad (6.9.29)$$

which permits us to formulate the following rule for finding $P_{0}^{*}(\mathcal{V})$.

So that the a priori distribution with density $P_0^*(\lambda)$ is the least preferred in the class of distributions P_0 , satisfying assigned conditions (6.9.22), the Bayes solution built with this distribution should give conditional risk $r(\hat{\lambda}, \lambda)$, the functional dependence of which on λ is determined by expression (6.9.29).

This rule generalizes the requirement of independence of conditional risk from λ , valid for total absence of a priori information. The concrete form of $P_0(\lambda)$ besides the assigned a priori information depends on the method of coding $\lambda(t)$ and y(t) and statistics of y(t), i.e., on the form of likelihood function $P(\mathbf{y}|\lambda)$.

An important particular case is assignment of the correlation matrix **R** of vector $\boldsymbol{\lambda}_i$, Here $\mathbf{f}(\boldsymbol{\lambda}) = \{\lambda_i \lambda_j\}$ and the equation for $r(\hat{\boldsymbol{\lambda}}, \boldsymbol{\lambda})$ has the form

$$r(\hat{\lambda}, \lambda) = \int w(\lambda - \hat{\lambda}) P(\mathbf{y} \mid \lambda) d\mathbf{y} = \mathbf{x}_0 + \sum_{i,j=1}^n \mathbf{x}_{ij} \lambda_i \lambda_j, \qquad (0.9.30)$$

-198-

where factors n_0 and $n_{i,i}$ are determined from conditions

$$\int P^{*}_{\bullet}(\lambda) d\lambda = 1, \quad \int \lambda_{i} \lambda_{j} P^{*}_{\bullet}(\lambda) d\lambda = P_{ij}, \quad (6.9.31)$$

and $\hat{\boldsymbol{\lambda}}(\mathbf{y})$ - the Bayes estimate with respect to distribution $P^{*}(\boldsymbol{\lambda})$.

Thus, estimate $\hat{\lambda}(\mathbf{y})$, corresponding to assignment of only the correlation matrix of the measured parameters, is optimum when it is the Bayes estimate with respect to the a priori distribution of probabilities for which the correlation matrix is equal to the assigned one, and conditional risk for this estimate is the sum of a constant and the quadratic form of λ of form

$$\boldsymbol{\lambda}^{+}\boldsymbol{x}\boldsymbol{\lambda} = \sum_{i,j} \boldsymbol{x}_{ij} \boldsymbol{\lambda}_{i} \boldsymbol{\lambda}_{j}. \tag{6.9.32}$$

We shall show that under certain conditions the least preferred distribution for the given case is a Gaussian distribution. Let us assume that the loss function is quadratic (6.5.8), and likelihood function $P(\mathbf{y}|\mathbf{\lambda})$ is presented in the form of (6.9.19), where function $F(\mathbf{\lambda} - \mathbf{\lambda}_0)$ is approximated by a Gaussian curve

$$F(\lambda - \lambda_0) = \exp\left\{-\frac{1}{2}(\lambda - \overline{\lambda}_0)^+ A(\lambda - \overline{\lambda}_0)\right\}, \qquad (6.9.33)$$

where $A = ||A_{i,i}||$ - matrix of order $(n \times n)$, similar in structure to matrix R.

Let us note that function $F(\lambda - \lambda_0)$ in (6.9.19) with the corresponding normalization is the density of the probability distribution for vector $\lambda_0 = \lambda_0(\mathbf{y})$ which is the maximum likelihood estimate for λ . Let us consider the Gaussian a priori distribution of $\lambda(t)$ with correlation matrix **R** and find the estimate of conditional mathematical expectation corresponding to the quadratic loss function. Then

$$\hat{\lambda}(\mathbf{y}) = -\frac{\int \lambda \exp\left\{-\frac{1}{2}(\lambda - \lambda_{0}) + \Lambda (\lambda - \lambda_{0}) - \frac{1}{2}\lambda + V\lambda\right\} d\lambda}{\int \exp\left\{-\frac{1}{2}(\lambda - \lambda_{0}) + \Lambda(\lambda - \lambda_{0}) - \frac{1}{2}\lambda + V\lambda\right\} d\lambda} = C\lambda_{0}(\mathbf{y}), \qquad (6.9.34)$$

where matrix $\mathbf{C} = \|\mathbf{C}_{i,i}\|$ is determined from equation

$$C[R-rA^{-1}] = R,$$
 (6.9.35)

i.e., estimate $\hat{\lambda}(\mathbf{y})$ is expressed linearly in terms of vector $\lambda_0(\mathbf{y})$. Substituting (6.9.34) in the expression for conditional risk, we obtain

$$r(\hat{\lambda}, \lambda) = \int w(\lambda - \hat{\lambda}) P(y|\lambda) dy =$$

= $\int (\lambda - C\lambda_0)^+ B(\lambda - C\lambda_0) F(\lambda - \lambda_0) d\lambda_0 =$
= $\int [\lambda - C\lambda + Cx]^+ B[\lambda - C\lambda + Cx] F(x) dx =$ (6.9.36)

-199-

 $= \lambda^{+} (\mathbf{I} - \mathbf{C})^{+} \mathbf{B} (\mathbf{I} - \mathbf{C}) \lambda \int F(\mathbf{x}) d\mathbf{x} + \int \mathbf{x}^{+} \mathbf{C}^{+} \mathbf{B} \mathbf{C} \mathbf{x} F(\mathbf{x}) d\mathbf{x} =$ $= \lambda^{+} \mathbf{x} \lambda + \mathbf{x}_{0}.$ (6.9.37)
cont 1d

As we see, conditional risk has the required form (6.9.30), i.e., the Gaussian distribution indeed is the least preferred, and the optimum estimate itself has the form of (6.9.34).

Representation (6.9.19) is strictly satisfied for an additive mixture $y(t) = \lambda(t) + n(t)$; in this case $\lambda_0 = y$ and $\hat{\lambda} = Cy$. This result permits us to comprehend the character of optimality of solutions of the linear theory of filtration. It turns out that if measured parameter $\lambda(t)$ is additively mixed with Gaussian interference, and we are given the correlation function of $\lambda(t)$, there is no better (from the point of view of a quadratic loss function) operator of filtration than the linear operator, determined by the Wiener theory of filtration, and only more detailed statistical evidence can lead to change of the structure of the optimum operator.

Representation (6.9.19) is approximately realized in a rather large number of cases, of which we already spoke in Paragraph 6.5.1, and which occur in problems of radar measurement. This circumstance emphasizes the importance of (aussian a priori distribution, which turns out to be interesting not only for itself, but as the least preferred distribution from the class of distributions with an assigned correlation function. In this connection we paid the greatest attention above to Gaussian a priori distribution.

One more interesting example of a problem with partially assigned statistical characteristics is the case when we are assigned the value of only the matrix of second moments of the derivative of $\lambda(t)$ at any moment of time. To this case we can approximately reduce the case when there are assigned a priori limits of variation of the derivative at any moment of time. In the given example

$$f(\lambda) = \{(\lambda_i - \lambda_{i-1})^2\}, \ \mu = \{(\lambda_1 - \lambda_{i-1})^2\} = \{M_i\}$$

and equation (6.9.29) for conditional risk has the form

$$r(\hat{\lambda}, \lambda) = \varkappa_{0} + \sum_{i=2}^{n} (\lambda_{i} - \lambda_{i-1})^{2} \varkappa_{i}. \qquad (6.9.37)$$

If there is also assigned the second moment for the value of $\lambda(t)$ at the initial moment of time $\sigma_1^2 = \overline{\lambda_0^2}$, to the right side of (6.9.37) one should add component $\lambda_1^2 n_1$. One can prove that under certain conditions, similar to those considered above, the least preferred distribution for the given case is the distribution for a Markovian

-200-

Gaussian process with transition probability

$$\boldsymbol{w}(\boldsymbol{\lambda}_{i} | \boldsymbol{\lambda}_{i-1}) = \frac{1}{V 2\pi M_{i}} \exp\left\{-\frac{(\lambda_{i} - \lambda_{i-1})^{2}}{2M_{i}}\right\}$$
(6.9.38)

and with uniform distribution for λ_1 when σ_1^2 is not assigned, or with a Gaussian distribution for λ_1 with variance σ_1^2 for an assigned σ_1^2 .

Analogous results are obtained also when along with statistical characteristics of first derivatives there are assigned statistical characteristics of the second derivatives, so that the following moments are determined:

$$M_{i}^{(11)} = (\overline{\lambda_{i} - \lambda_{i-1}})^{2}, \quad M_{i}^{(12)} = (\overline{\lambda_{i} - \lambda_{i-1}}) (\overline{\lambda_{i} - 2\lambda_{i-1} + \lambda_{i-2}}),$$
$$M_{i}^{(21)} = (\overline{\lambda_{i} - 2\lambda_{i-1} + \lambda_{i-2}}) (\overline{\lambda_{i} - \lambda_{i-1}}),$$
$$M_{i}^{(22)} = (\overline{\lambda_{i} - 2\lambda_{i-1} + \lambda_{i-2}})^{2}. \quad (6.9.39)$$

In this case the least preferred distribution turns out to correspond to a Markovian Gaussian process of the second order with Gaussian transition probability, depending only on the quadratic form of variables λ_i , λ_{i-1} , λ_{i-2} , completely determined by assigned $M_1^{(11)}$, $M_1^{(12)}$, $M_1^{(21)}$, $M_1^{(22)}$.

Therefore, the optimum operator of estimation obtained for such a distribution is simultaneously optimum in the case of assignment of only the matrix of second moments for the first and second derivatives of measured parameters $\lambda(t)$. These statements show that the examples of measurement of Markovian and simultaneously Gaussian parameters given in Paragraph 6.8.2 have more universal meaning than supposed earlier.

§ 6.10. Conclusion

Basic results of the theory of radar meters developed in Chapter VI reduce to the following.

1. In the analysis of tracking meters it is convenient to divide their circuits into discriminators and smoothing circuits. Discriminators are completely described by two functions of mismatch - discrimination and fluctuating characteristics. The minimum parameters of these functions necessary for analysis of accuracy of meters are equivalent spectral density S_{OKB} , the gain factor of the discriminator $K_{\rm H}$, and in a number of cases spectral density of parametric fluctuations $S_{\rm HAF}$. If we know these characteristics and statistical properties of the measured quantity, error of

-201-

measurement usually can be calculated easily. Formulas for calculating errors for cases of rapid and slow fluctuations of the signal are given in § 6.2.

2. Breakoff of tracking can be characterized by two characteristics — average time to breakoff and magnitude of variance of errors of measurement in steady-state operating conditions, which in the presence of breakoff's of tracking increases sharply. In any case it turns out that breakoff is a threshold phenomenon, the probability of which increases with the ratio of mean square error (in a linearised system) to the width of the discriminator curve, exceeding a certain number which depends only on the type of smoothing circuits. This gives us the possibility of finding critical intensities of noises and interferences, which lead to breakoff of tracking. 11.25

3. In the analysis of nontracking meters it is convenient to divide them into estimator units and smoothing circuits. In §6.3 we obtained formulas for calculation of accuracy of measurement of nontracking circuits on the simplified assumptions of rapid fluctuations and a slowly changing measured quantity.

4. The most adequate means of synthesis of optimum meters at present one should consider the theory of nonlinear filtration. By application of this theory we solved the problem of synthesis of an optimum meter of a slowly changing parameter with Gaussian approximation of the likelihood function.

For a parameter with Gaussian distribution we found three variants of construction of optimum meter: single-loop and double-loop circuits of a tracking meter and a nontracking meter. All of them are identical in performance in identical conditions of work, but the tracking variants are usually more convenient technically.

Optimum tracking meters as basic elements contain nonlinear units (a discriminator and an accuracy unit) and smoothing circuits which for Gaussian statistics of the parameter are linear. An optimum discriminator, just as analogous devices known in practice, issues the measure of mismatch between current and measured values of the parameter, and the accuracy unit issues the measure of current accuracy of separate measurements, which, after smoothing, form the final result of measurement.

The type of discriminator and accuracy unit is determined by statistics of the input signal and the method of encoding the measured parameter in it. Therefore, without concretization of the form of this parameter we could only find certain general rules poculiar to discriminators, and we discussed possible types of them. Final synthesis of optimum discriminators, just as analysis of different closely

-202-

related circuits, will be conducted in subsequent chapters, in which we concretize the measured parameter (distance, speed, angles).

The form of smoothing circuits is determined by statistics of the measured parameter and a certain generalized characteristic of accuracy of the discriminator. Therefore, synthesis of smoothing circuits, and also analysis of accuracy of the optimum meter as a whole could be conducted in Chapter VI in sufficiently general form.

5. It should be noted that an optimum discriminator constructed with application of the developed theory coincides in certain assumptions with the discriminator found by the theory of estimation. Thus, we defined the meaning of optimality of devices synthesized by the theory of estimation.

6. Characteristics of optimum smoothing filters with a Gaussian law of distribution of measured parameters obey essentially the same equations as follow from filtration theory, which indicates the meaning of optimality determined by this wide-spread theory.

7. The developed method of synthesis is very universal. Thus, it remains in force for parameters with unknown statistics, if we are only given the correlation function of the parameter. With small changes it can be transferred to the case of a Markovian parameter. Here, the discriminator and accuracy unit remain constant, and only the smoothing circuits are subjected to certain complication.

Although the above findings concerning general rules of radar measurements pertain to a very broad class of radar systems, they nevertheless require further development. Besides problems of simultaneous measurement of several parameters and resolution of signals, to which we devote Chapters XII and XIII, it is possible to outline the following directions for further investigations.

During the analysis of meters above we considered fluctuating errors (due to noises, interferences and fluctuations of the signal) and dynamic errors. Meanwhile, in precision radar systems of importance, too, may be errors of different origin, for instance instrument errors. It would be desirable subsequently to expand the class of errors considered during analysis and synthesis.

If we are concerned with nonlinear phenomena in meters (phenomena of breakoff), results given in § 6.3 indicate a promising mathematical technique of investigation and give practical results for cases of systems with small astaticism. Solution of the problem of breakoff for more complicated forms of smoothing circuits, including

-203-

those with variable parameters, runs into greater mathematical difficulties, inacmuch as methods of solution of multi-dimensional diffusion equations have not been developed. This then poses a task for mathematicians, important in its application. Here fully permissible are formulations of problems, differing from these considered in 5 6.3, inasmuch as actual definition of the concept of breakoff, obviously, depends on the purpose of the meter.

It is also important to consider that meters can be equipped with failure indicators, for instance, in the form of threshold circuits and "insuring" devices with "matchguard" channels, detuned relative to the measured value of the parameter. Questions of analysis of joint work of meters and such circuits remain practice by unconsidered.

Further development of the theory of Bayes synthesis in the direction of non-Gaussian parameters, and also synthesis on the basis of the minimax method with limited statistics and the game-theory method is desirable. As initial propositions it is possible to consider in this case the material of § 6.9.

During concretization of the form of discriminators it is practically useful to consider non-Gaussian signals and interference of a fairly broad class, and also the case of slow fluctuations of the signal. Although the method of coupling such discriminators in optimum meters is known, their circuit construction at present is not completely clear. Analogous questions of concrete realization of circuit construction for non-Gaussian parameters are important also for smoothing circuits.

Although the direction " synthesis of smoothing circuits for quasi-regular parameters are outlined, it is necessary to generalize them for the real case when laws of variation of parameters are unknown and we are assigned only complex differential equations, when properties of errors are complicated, and from smoothing devices there is required maximum accuracy in combination with technical convenience.

Results of analysis and synthesis of radar meters, unboubtedly, may be useful also in those regions of technology where there are used information-bearing signals: in optics, infrared engineering, ultrasonic engineering, communications, broadcasting and television. In those places where the idealizations made above are not fully suitable the method of analysis and synthesis developed for radar systems remains in force.

-204-

CHAPTER VII

÷.,

RANGE FINDING WITH A COHERENT SIGNAL

§ 7.1. Introduction

One of the most important problems of contemporary radar systems of various assignment is measurement of the distance to a target. The need to measure range appears both in survey radars and all the more so in acquisition and guidance radars, and also in navigation systems. Range finding can be based on different physical principles; in particular, methods based on integration of velocity and the triangulational method have been used; however, in extent of application and importance of greatest significance is the direct radar method, based on encoding distance to target in delay of the received signal and measurement of this delay [1, 26, 27].

We shall consider questions connected with range finding by this last method. The first two methods are completely based on measurement of velocity and angles, and all their consideration reduces only to application of results of the corresponding chapters of this book. Furthermore, of all systems finding range by delay of a reflected signal we shall be interested basically in the most widely used automatic tracking meters, carrying out range tracking of a selected target, and only to a smaller extent in other types of range finders.

Statistical analysis and synthesis of systems of range finding are the subject of a comparatively small number of published works. Most works known to the authors pertain actually only to analysis of the influence of random signals, noises, and certain interferences for the most widespread (until recently) pulse incoherent automatic range finder with various limiting assumptions [28-30]. To a considerably lesser extent they have touched on questions of analysis of range finders using other forms of signals [1, 31], and also questions of synthesis of an optimum system

~205-

of range finding. Certain particular solutions pertaining to the last question were obtained in [12,32].

Almost simultaneously with works on analysis of the fluctuating accuracy of range-only radars of a very specific type under the influence of noises and interferences there were undertaken attempts at investigation of the potential accuracy of range finding by radar. Apparently, the first work on this question is [4], where for very limiting assumptions there is found the variance of the efficient estimate of delay of a reflected signal for the case of a motionless target. The same problem with less limiting assumptions relative to the received signal was solved in [33]. Further advance on the given question was attained with application of the general results presented in Chapter VI. These results, in particular, give the possibility of obtaining a sufficiently correct solution of the problem of synthesis of an optimum system of range finding for a very broad and practically important class of cases, when the rate of change of the measured distance is shall as compared to the rate of fluctuations of the received signal.

As already shown in Chapter VI, the optimum meter is realized here in the form of a closed-loop tracking system, including a nonlinear discriminator and smoothing



Fig. 7.1. Block diagram of a tracking range finder: 1 - discriminator; 2 - smoothing and control circuits.

circuits, controlling tuning of the discriminator (Fig. 7.1). Solution of the problem of synthesis permits us to establish the algorithm for finding operations of the discriminator and characteristics of smoothing circuits for an arbitrary form of modulation of the received signal and a broad class of cases of statistics of the change of distance, and also to conduct detailed investigation of potential accuracy of range finding without the limiting assumption of immobility of the target.

Development of the theory of statistical solutions permits us to present statistical questions related to range finding by the deductive method. In accordance with this we shall first consider questions of the synthesis of optimum range finders and of the potential accuracy of range finding for an arbitrary case, and then also in general investigate basic methods of approximate realization of optimum operations and, finally, analyze more specifically concrete circuits of range finders for the most wide-spread and characteristic forms of signal molulation. Inasmuch as all results have great mathematical completeness for the case of a coherent signal, we shall start namely with it, and in the next chapter we shall

-206-

consider the case of an incoherent fluctuating signal.

In this and subsequent chapters both from the point of view of analysis, and also from the point of view of synthesis, we shall be interested basically in discriminators of range finding systems. This is explained by the fact that synthesis of smoothing circuits is conducted identically for any measured parameter of the signal. The structure of smoothing circuits is determined here only by the statistics of change of the parameter and does not depend on its physical nature. Such a problem of synthesis is considered in detail in Chapter VI, and here we can now use the available results. If we have determined basic characteristics of the discriminator (discrimination and fluctuation), further analysis of the measuring system, and in particular, determination of errors of measurement also no longer depend on the physical nature of the measured parameter of the signal, in which some target position datum is coded. This gives us the possibility of calculating errors of measuring of coordinates by a single method using previously found characteristics of the discriminator.

For instance, in the case of linearized consideration of the tracking system with smoothing circuits with constant parameters the calculation of the variance of error of measurement reduces to multiplication of the equivalent spectral density of the discriminator by the effective passband of the closed-loop tracking system. In other cases these methods are more complicated, but in general they have already been presented in Chapter VI, and for final calculations of accuracy of range finding systems we can use the available results. In contrast to this, questions pertaining to the construction of discriminators and their analysis have a specific character, characteristic to range finding systems, and require, of course, detailed consideration.

As shown in Chapter VI, the most complete characteristics of a discriminator are the discrimination and fluctuation characteristics. Such characteristics in the case when the measured distance changes slowly as compared to all random components of the signal completely determine, together with characteristics of smoothing circuits, the accuracy of range finding.

The discrimination characteristic is the dependence of the Mathematical expectation of output voltage of the discriminator on the magnitude of mismatch between the true and measured value of delay; the fluctuation characteristic is the dependence of the spectral density of the output voltage of the discriminator at zero frequency on the same mismatch. In general these characteristics are arbitrary nonlinear functions, possessing definite symmetry, and in accordance with

-207-

this, the range finder is a nonlinear system with feedback. Only with small mismatch, when with high probability errors of measurement are sufficiently small, can this system be linearized (see Chapter VI).

For analysis of the linearized system it is sufficient to have simpler characteristics of the discriminator - null shift Δ_0 , slope K_{II} of the discrimination characteristic and equivalent spectral density S_{ORB} , i.e., the ratio of spectral density of the output voltage of the discriminator at zero frequency with zero mismatch to the square of the gain factor. Calculation of parametric fluctuations in the measuring system requires additional calculation of their spectral density, which is determined by the corresponding formulas of Chapter VI.

Subsequently in most cases we shall limit ourselves to linearized consideration of a range finder without taking into account parametric fluctuations, in accordance with which analysis of the discriminator and comparison of various discriminators lead to determination of magnitudes Δ_0 , $K_{\rm H}$, $S_{\rm HKB}$ and their comparison for various discriminators. Such factors, as the signal-to-noise ratio, width of the spectrum of fluctuations of the received signal, the law of modulation and the method of processing the signal affect accuracy of range finding only through these three quantities.

In this chapter we first shall consider operation of a discriminator for a fluctuating arbitrarily modulated coherent signal in noises and investigate its characteristics, determining potential accuracy of range finding. Then we shall generally investigate basic methods of approximate realization of these operations and, more specifically, final circuits of discriminators for certain of the most important forms of modulation. Then, for various smoothing circuits, corresponding to the various forms of statistics of change of the measured range, we shall analyze the range finder as a whole. And, finally, at the end of the chapter we shall consider questions connected with the influence of organized interferences on coherent range finders.

§ 7.2. Optimum Discriminator

7.2.1. Operations of an Optimum Discriminator

In accordance with the definition given in Chapter I, the received coherent signal is a nonstationary normal random process with a correlation function which, according to (1.4.3), is equal to

$$\begin{aligned} \mathbf{t}_{i}, \ \mathbf{t}_{i}; \ \mathbf{\tau} &= P_{o} \operatorname{Re} u \left(\mathbf{t}_{i} - \mathbf{\tau} \right) u^{*} \left(\mathbf{t}_{i} - \mathbf{\tau} \right) \times \\ & \times \mathbf{P} \left(\mathbf{t}_{j} - \mathbf{t}_{j} \right) e^{i \mathbf{\tau} d \left(\mathbf{t}_{i} - \mathbf{t}_{i} \right)} - N_{o} \delta \left(\mathbf{t}_{j} - \mathbf{t}_{j} \right), \end{aligned} \tag{7.2.1}$$

-208-

where $u(t) = u_{a}(t)e^{i\psi(t)}$ - complex notation of the law of modulation, in general describing the amplitude $(u_{a}(t))$ and phase $(\psi(t))$ modulation of the sounding signal;

$$\tau = 2d/c - delay$$
 of the reflected signal;

- P_c mean signal power;
- $\rho(t)$ correlation coefficient of fluctuations of the signal reflected from the target;
 - $N_{\mbox{O}}$ spectral density of noise, and function u(t) is normalized so that

$$\lim_{T \to \infty} \frac{1}{T} \int |u(t)|^2 dt = 1.$$
 (7.2.2)

With periodic modulation in (7.2.2) it is possible to be limited to averaging over one period.

All statistical properties of the received signal y(t) are characterized by the functional of the density of the distribution of probabilities of process y(t), which on the basis of (4.3.7) in the case interesting is presented in the following form:

$$P(y|\tau) = \operatorname{Cexp}\left\{-\frac{1}{2N_{o}}\int_{0}^{T}y^{2}(t) dt + \frac{1}{N_{o}}\int_{0}^{T}|Q(t,\tau)|^{2}dt\right\},$$
(7.2.3)

where C does not depend on y(t), and $|Q(t, \tau)|^2$ is defined as

$$|Q(t, \tau)|^{s} = \left| \int_{-\infty}^{t} h(t-s) u(s-\tau) y(s) e^{t-s} ds \right|^{2}.$$
 (7.2.4)

The pulse response of the filter h(t) is given by its own Fourier transform $H_O(i\omega)$, the square of the modulus of which is equal to

$$|H_{\bullet}(l\bullet)|^{a} = \frac{hS_{\bullet}(\bullet)}{1+hS_{\bullet}(\bullet)}, \qquad (7.2.5)$$

where h = $P_c/2\Delta f_c N_0$ - signal-to-noise ratio, repeatedly used in the preceding chapters;

 $S_{O}(\omega)$ - normalized spectral density of fluctuations of the signal $(S_{O}(0)$ = 1);

 Δf_c — effective width of the spectrum of signal fluctuations. Function h(t) describes the pulse response of an optimum filter of a coherent receiver and appears in all problems of optimum detection and measurement in noises. Its properties were already discussed in sufficient detail in Chapter IV.

Assignment of functional $P(y|\tau)$ permits us to find operations which should be

performed on the received signal y(t) by an optimum discriminator. As shown in Chapter VI, these operations consist of formation of the derivative with respect to τ of the integrand in the exponent of $P(y|\tau)$, i.e., the output signal of an optimum discriminator z(t) is defined as

$$\mathbf{z}(t) = \frac{1}{N_0} \frac{\partial |Q(t, \mathbf{x})|^2}{\partial \mathbf{x}} =$$
$$= \frac{1}{N_0} \frac{\partial}{\partial \mathbf{x}} \left| \int_{\infty}^{t} h(t-s) u(s-\mathbf{x}) y(s) e^{i\mathbf{x} \cdot \mathbf{x}} ds \right|^2. \tag{7.2.6}$$

Derivative $\frac{\partial |Q(t, \tau)|^2}{\partial \tau}$ is taken for a value of τ , equal to the measured value of delay obtained at the output of the range finder.

The latter is inserted in the discriminator by feedback and controls its tuning. Quantity r(t), describing the output voltage of the optimum discriminator, turns out here to be a function of the mismatch $\tau_0 - \tau$ between the true value of delay of the signal τ_0 and its estimated (measured) value τ .

Differentiating in (7.2.6), we obtain the following final expression, determining the operation of an optimum discriminator:

$$z(t; \mathbf{x}_{0} - \mathbf{z}) = -\frac{2}{N_{0}} \operatorname{Re} \int_{-\infty}^{t} \int_{-\infty}^{t} h(t - s_{1}) h(t - s_{2}) \times \\ \times u(s_{1} - \mathbf{z}) u^{s_{1}}(s_{2} - \mathbf{z}) e^{i \mathbf{z}_{1} ds_{1} - s_{2}} y(s_{1}) y(s_{2}) ds_{1} ds_{2}.$$
(7.2.7)

These operations can be realized by the block diagram of Fig. 7.2. The received signal y(t), possibly, after preamplification in amplifiers, broad-band as



Fig. 7.2. Block diagram of an optimum discriminator: 1 - preamplifier; 2 - controlled coherent local oscillator; 3 - mixer; 4 narrow-band filter; 5 - phase detector (multiplier). compared to the width of the spectrum of modulation, proceeds to two mixers. As reference signals of the mixers we use the expected signal and its derivative with respect to delay. Generators of reference signals (modulated local oscillators, strobing-pulse generators, and so forth) are controlled by the output value of

measured delay. Converted signals from the output of the mixers enter optimum filters with pulse response $h(t) \cos \omega_{\Pi D}(t)$, and then a phase detector, whose

-210-

output is the discriminator output. Processing of the signal in the first channel of the discriminator, obviously, coincides with predetection processing of a rapidly fluctuating coherent signal during detection (see Chapter IV), and processing in the second channel corresponds to calculation of correlation between the received signal and the derivative with respect to delay of the expected signal. By direct analysis of the block diagram of Fig. 7.2 it is simple to prove that, if only the magnitude of the intermediate frequency satisfies the usual requirement of elimination of image frequencies, output voltage of the phase detector indeed is described by formula (7.2.7), and, consequently, this circuit exactly executes optimum operations.

7.2.2. Discriminator Characteristics

Let us find now basic characteristics of an optimum discriminator, necessary for consideration of the measuring system as a whole. Averaging (7.2.7), replacing s - τ_0 by s, introducing designation $\Delta = \tau_0 - \tau$ and using the slowness of change of function h(t) in comparison with u(t), we obtain

$$\overline{z(t, \Delta)} = -\frac{2}{N_{\bullet}} \operatorname{Re} \int_{-\infty}^{t} \int_{-\infty}^{t} h(t-s_{1})h(t-s_{9}) \times \\ \times u(s_{1}+\Delta) u^{*'}(s_{9}+\Delta) e^{i\omega_{9}(s_{1}-s_{9})} [\operatorname{Re} P_{c}u(s_{1}) \times \\ \times u^{*}(s_{1}) e^{i\omega_{9}(s_{1}-s_{1})} \rho(s_{1}-s_{9}) + N_{\bullet}\delta(s_{1}-s_{1})] ds_{1} ds_{9} = \\ = -\operatorname{Re} \left\{ \lim_{T \to \infty} \frac{1}{T} \int_{0}^{T} u^{*}(s) u(s+\Delta) ds \lim_{T \to \infty} \frac{1}{T} \int_{0}^{T} u(s) \times \\ \times u^{*'}(s+\Delta) ds \frac{P_{\circ}}{N_{\bullet}} \int_{-\infty}^{t} \int_{-\infty}^{t} h(t-s_{1})h(t-s_{1})\rho(s_{1}-s_{2}) ds_{1} ds_{2} + \\ + 2\lim_{T \to \infty} \frac{1}{T} \int_{0}^{T} u(s) u^{*'}(s) ds \int_{-\infty}^{t} h^{2}(t-s_{1}) ds_{1} \right\} = \\ = -\frac{1}{*} \int_{-\infty}^{\infty} \frac{hS_{\bullet}(\omega)}{1+hS_{\bullet}(\omega)} \operatorname{Re} \left\{ C^{*'}(0) + hC(\Delta) C^{*'}(\Delta) S_{\bullet}(\omega) \right\} d^{*\circ}, \qquad (7.2.8)$$

where $C(\Delta)$ - autocorrelation function of the law of modulation, determined by relationship

$$C(\Delta) = \lim_{r \to \infty} \frac{1}{r} \int_{0}^{r} u(s + \Delta) u^{*}(s) ds, \qquad (7.2.9)$$

where in accordance with the adopted normalization (7.2.2) $C(\psi) = 1$. Function C(i) from (7.2.9) is equal, obviously, to function $C(\Delta, 0)$, determined by formula (1.2.2).

For convenience we shall subsequently use the shorter notation. With a periodic modulation function, $C(\Delta)$ is periodic. However, in examining range finders mismatches Δ exceeding the period of repetition of the signal are not of practical interest. In accordance with this, subsequently, in examining all concrete examples it is sufficient to consider only one period of function $C(\Delta)$, corresponding to mismatches close to zero.

Function $C(\Delta)$ is the single characteristic of modulation of the sounding radar signal. Through it we find a description of the dependence of accuracy of range finding on the mode and parameters of modulation of the signal both in optimal, and also (we will prove this later) in nonoptimal systems. Furthermore, function $C(\Delta)$ is the characteristic of the potential resolution capability with respect to distance for a given signal [4]. Let us note certain important properties of function $C(\Delta)$. Separating in (7.2.9) the real and imaginary part

$$C(\Delta) = \lim_{T \to \infty} \frac{1}{T} \int_{0}^{T} u_{\mathbf{a}}(s) u_{\mathbf{a}}(s + \Delta) \cos \left[\psi(s + \Delta) - \psi(s)\right] dz + + i \lim_{T \to \infty} \frac{1}{T} \int_{0}^{T} u_{\mathbf{a}}(s) u_{\mathbf{a}}(s + \Delta) \sin \left[\psi(s + \Delta) - \psi(s)\right] ds,$$

it is simple to prove that Re C(L) is an even function of L, and Im C(L) is an odd function. Therefore, $C^{*}(0)$ is a purely imaginary quantity,* and the final expression for the discrimination characteristic $\overline{Z(t, \Delta)}$ on the basis of (7.2.8) takes the form

$$\overline{z(l, \Delta)} = -\operatorname{Re} C(\Delta) C^{**}(\Delta) \frac{1}{\pi} \int_{-\infty}^{\infty} \frac{\hbar^2 S_0^2(\omega) d\omega}{1 + \hbar S_0(\omega)}. \qquad (7.2.10)$$

Furthermore, thanks to the fact that C'(0) is purely imaginary, z(t, 0) = 0, i.e., there is no null shift of the discrimination characteristic in an optimum circuit.

Important parameters of function $C(\Delta)$ are values of its first and second derivatives at zero. Determining them by (7.2.9), we obtain

$$C^{r}(0) = + ia = + i \lim_{T \to \infty} \frac{1}{T} \int_{0}^{T} u_{a}^{2}(l) \psi(l) dl =$$
$$= + \frac{1}{2\pi} \int_{-\infty}^{\infty} u \lim_{T \to \infty} \frac{|U_{T}(lw)|^{2}}{T} du, \qquad (7.2.11)$$

*Here and henceforth C'(x) - derivative of function C(x).

-212-

$$C''(0) = -b = -\lim_{T \to \infty} \frac{1}{T} \int_{0}^{T} \{ [u'_{a}(t)]^{2} + u_{a}^{2}(t) [\psi'(t)]^{2} \} dt =$$

= $-\frac{1}{2\pi} \int_{-\infty}^{\infty} e^{2} \lim_{T \to \infty} \frac{|U_{T}(i\infty)|^{2}}{T} d\infty,$ (7.2.12)

whe **re**

$$U_r(i \bullet) = \int_0^t u(t) e^{-i \bullet t} dt$$
 (7.2.13)

is the spectrum of a finite segment of complex signal $u(t) = u_{g}(t)e^{i\psi(t)}$.

With periodic modulation, which occurs in most practical cases, it is sufficient to produce averaging in (7.2.11) and (7.2.12) in one period of modulation T_r . Here $U_T(i\alpha)$ is the spectrum of one period of modulation. With stationary ergodic random modulation function $\lim_{T\to\infty} \frac{|U_T(i\alpha)|^2}{\pi}$ coincides with the spectral density of modulation.

Positive values of a and b in (7.2.11) and (7.2.12) have the meaning of the mean frequency and the mean square of the frequency of the spectrum of modulation [4], and quantity $\sqrt{b} - a^2$, which, as we shall see presently, determines the accuracy of range finding, constitutes the mean square width of the spectrum of modulation, characterizing the rate of change of the modulating function. Quantity a differs from zero only with an asymmetric spectrum of modulation. In most real cases, for instance with only amplitude or only phase modulation, with simultaneous modulation in amplitude and phase by symmetric modulating functions, and in certain other cases, a = 0. Here, accuracy of range finding is determined only by the parameter of the law of modulation - quantity b.

For analysis of a linearized measuring system without taking into account parametric fluctuations (see Chapter VI) the sufficient characteristic of the discriminator is the equivalent spectral density with zero mismatch, determined by relationship (6.2.5):

$$S_{\mathbf{0}K\mathbf{0}} = \frac{S_{\mathbf{0}K\mathbf{0}}}{K_{A}^{2}} = \int_{-\infty}^{\infty} \overline{\left[z\left(t, 0\right)z\left(t+\tau, 0\right)} - \frac{\overline{z}\left(t, 0\right)\overline{z}\left(t+\tau, 0\right)}{\overline{z}\left(t+\tau, 0\right)}\right]} d\tau \left[\left.\frac{\overline{\partial z\left(t, \Delta\right)}}{\overline{\partial \Delta}}\right|_{A=0}\right]^{2}.$$

In Chapter VI we show that for an optimum discriminator the magnitude of the equivalent spectral density S_{court} in

-213-

$$S_{OET} = \frac{1}{K_{OET}} = \frac{1}{\frac{\partial z(l, \Delta)}{\partial \Delta}}, \qquad (7.2.14)$$

where K_{ORT} - slope of the discrimination characteristic of an optimum discriminator. Substituting in (7.2.14) $\overline{z(t, \Delta)}$ from (7.2.10) and using (7.2.11) and (7.2.12), we obtain

$$S_{\text{ORY}} = \left[(b - a^2) \frac{1}{\pi} \int_{-\infty}^{\infty} \frac{h^2 S_0^2(\omega) d\omega}{1 + h S_0(\omega)} \right]^{-1} = \frac{F(h)}{2\Delta f_c (b - a^2)}, \quad (7.2.15)$$

where F(h) is a dimensionless function of the signal-to-noise ratio h, depending only on the form of spectral density of fluctuations of the received signal $(h = P_0/2\Delta f_0 N_0).$

Thus, the dependence of S_{OIT} and potential errors of range finding on the parameters of the law of modulation and the signal-to-noise ratio has a fairly simple character. Quantity S_{OIT} , and, consequently, variance of fluctuating error of measurement with linear smoothing circuits are proportional to S_{350} , are inversely proportional to the mean square width of the spectrum of the modulating signal. This signifies, in particular, that from the point of view of accuracy of range finding all other characteristics of the modulating signal besides spectrum width do not play a role, and, applying various modulating signals with identical spectrum width, we, in principle, should obtain identical accuracy of range finding. Therefore, being interested only in accuracy of range finding, it is impossible to give preference to one or the other form of sounding signal, of course, with identical spectrum width, and final selection of the method of modulation of the sounding signal requires us to turn to other considerations, such as uniqueness, resolution capability, possibility of simple measurement of Doppler frequency, technical realizability, and so forth.

The dependence of S_{OHT} on the signal-to-noise ratio is described by function F(n), which has the following general properties. It is obvious that for large values of h

$$F(h) = \frac{1}{\frac{1}{2\pi\Delta/e} \int_{-\infty}^{\infty} \frac{h^2 S_0^2(\omega) d\omega}{1 + h S_0(\omega)}} \xrightarrow{h \to \infty} \frac{1}{\frac{1}{2\pi\Delta/e} \int_{-\infty}^{\infty} h S_0(\omega) d\omega} (7.2.10)$$

since, thanks to normalization of $S_{0}(\omega)$

$$\frac{1}{2\pi}\int_{-\infty}^{\infty}S_{\bullet}(\bullet)\,d\bullet=\Delta f_{\circ},$$

and for small h

$$F(h) \xrightarrow{}_{h \to 0} \frac{1}{\frac{1}{2\pi \Delta f_*} \int_{-\infty}^{\infty} h^2 S_0^2(\omega) d\omega} = \frac{1}{ah^2}, \qquad (7.2.17)$$

where α - numerical coefficient of order ($\alpha = 0.5-1.0$).

The exact form of function F(h) depends on the form of spectrum $S_0(\omega)$; however, this dependence is not very essential, especially with such magnitudes of h, which



Fig. 7.3. Dependence of function F(h) on the signal-to-noise ratio. --- square spectrum of fluctuations; --- exponential correlation function of fluctuations.

in measuring systems, i.e., h > 5-10. For the two extreme cases (of the spectral density $S_0(\omega) = \left[1 + \left(\frac{\omega}{2\Delta f c}\right)^2\right]^{-1}$ dropping most slowly to $\pm \omega$, corresponding to exponential correlation function, and a rectangular spectrum of

can be considered working magnitudes

fluctuations with the same effective width $\Delta f_{\rm C}$) in Fig. 7.3 there are given dependences which have the form:

for the first case

$$F(h) = \frac{\sqrt{1+h}(1+\sqrt{1+h})}{h^3}, \qquad (7.2.18)$$

for second case

$$F(h) = \frac{1+h}{h^2}$$
 (7.2.19)

These curves simultaneously show the dependence of fluctuating error of range finding by a system with an optimum discriminator on the signal-to-noise ratio.

We shall also discuss the influence of the width of the spectrum of fluctuations of the signal on accuracy of measurement. It is obvious that for a sufflciently large h, when $F(h) \approx 1/h$, ratio $\frac{F(h)}{2\Delta f_{C}} \approx \frac{N_{O}}{P_{O}}$ does not depend on Δf_{C} . However, for not very large h expansion of the spectrum of fluctuations worsens accuracy of measurement, since components of equivalent spectral density, caused by beats of noises, and having an order of $1/h^2$, increase with growth of Δf_c .

Let us now find the fluctuation characteristic of an optimum discriminator. Here, we recalculate fluctuations of output voltage of the discriminator into fluctuations of the measured parameter (delay) by dividing the output spectral density by the square of the slope. Then, in accordance with (6.2.8)

$$S_{\text{OBT}}(\Delta) = \frac{1}{K_{\text{OBT}}^2} \int_{-\infty}^{\infty} \overline{[z(t, \Delta) z(t+\tau, \Delta) - \frac{1}{Z(t, \Delta)}]} d\tau.$$

Substituting (7.2.7) in this expression and averaging analogously to (7.2.8), we obtain

$$S_{offr}(\Delta) = \frac{1}{K_{offr}^2} \operatorname{Re}\left\{ \left[|C(\Delta)|^2 |C'(\Delta)|^2 + C^{*2}(\Delta) C^{*2}(\Delta) \right] \frac{1}{\pi} \times \left\{ \sum_{-\infty}^{\infty} \frac{h^4 S_0^4(\omega) d\omega}{(1 + h S_0(\omega))^2} + [b|C(\Delta)|^2 + C^{*2}(\Delta)|^2 + 2iaC^*(\Delta) C^*(\Delta) \right] \frac{1}{\pi} \int_{-\infty}^{\infty} \frac{h^4 S_0^3(\omega) d\omega}{(1 + h S_0(\omega))^2} + (b - a^2) \frac{1}{\pi} \int_{-\infty}^{\infty} \frac{h^4 S_0^3(\omega) d\omega}{(1 + h S_0(\omega))^2} \right\}.$$

$$(7.2.20)$$

From this expression it follows that when $\Delta \neq 0$ quantity $S_{OUT}(\Delta)$ no longer tends to zero as $h \rightarrow \infty$, i.e., even with zero noises the output voltage of the



Fig. 7.4. Fluctuation characteristic of an optimum discriminator.

discriminator contains random components, caused by fluctuations of the reflected signal and depending on mismatch. For illustration in Fig. 7.4 there is constructed a family of fluctuation characteristics, corresponding to a rectangular spectrum of fluctuations of the signal and the autocorrelation function of the law of modulation يتحيث بالقافة الملاعية أماحيا يحور

÷.

-216-

 $C(\Delta) = \exp \{-\frac{1}{2}b\Delta^2\}$ for various values of h. From the figure it is clear that irregularities of the fluctuation characteristic start to appear when $h \sim 1$. A stationary level of equivalent spectral density is attained already when $\sqrt{b\Delta} =$ = 1.5-3, i.e., with mismatch exceeding the width of the basic lobe of autocorrelation function by a factor of one and a half to three.

Besides S_{DKB} , of greatest practical interest for analysis of a range finder is the spectral density of equivalent parametric fluctuations $S_{\Pi A B}$, which determines parametric error of measurement (see Chapter IV) and is equal to the coefficient of Δ^2 in the expansion of $S_{DKB}(\Delta)$ in powers of Δ . Then, as follows from (7.2.20), this quantity is equal to

$$S_{\mu a p} = \frac{\frac{1}{2\pi} \int_{-\infty}^{\infty} \frac{h^{4} S_{0}^{4}(\mathbf{w}) d\omega}{[1 + hS_{0}(\mathbf{w})]^{2}}}{\left[\frac{1}{2\pi} \int_{-\infty}^{\infty} \frac{h^{4} S_{0}^{2}(\mathbf{w}) d\omega}{1 + hS_{0}(\mathbf{w})}\right]^{2}}$$
(7.2.21)

and depends only on beats of signals in the two discriminator channels. The presence of noises does not increase parametric fluctuations in an optimum discriminator. The influence of parametric fluctuations on accuracy of the range finder as a whole and on the relationship between usual and parametric fluctuating errors we shall consider subsequently.

In conclusion let us discuss one more interpretation of quantity S_{OILT} . In Chapter VI it is shown that if one were to assume constancy of the measured parameter throughout an interval of observation of duration T and construct some estimate of the parameter from values of the accepted realization of the signal, variance of this estimate, characterizing accuracy of measurement of the parameter, cannot be less than a certain quantity, called variance of the efficient estimate. Here, for instance, estimation of the parameter by the maximum likelihood method ([18], Chapter VI) ensures variance, practically equal to variance of the efficient estimate. In Chapter VI it is shown that variance of the efficient estimate is connected with quantity S_{OILT} in the following way:

$$\sigma_{\bullet\bullet}^2 = \frac{S_{\bullet\bullet\uparrow}}{T}.$$
 (7.2.22)

Consequently, maximum accuracy of measurement of delay of the signal received from the target (assuming its constancy during the time during which we produce this measurement) for the case when we make no assumptions with respect to statistical

-217-

characteristics of measured delay, but produce a simple estimate from the functional of likelihood, is determined by

$$\mathbf{g}_{ab}^{2} = \frac{S_{ost}}{T} = \frac{F(h)}{2\Delta f_{c}T(b-a^{2})}.$$
 (7.2.23)

This formula is valid when $\Delta f_{C} T \gg 1$. In the opposite case, when the time of observation is small as compared to the time of correlation of the signal($\Delta f_{C} T \ll \ll 1$), which may be of interest for survey radars in which range finding is produced during every cycle of survey for a small time, we must return to the general formula for variance of the efficient estimate (6.7.33), from which it follows that

$$\boldsymbol{\sigma}_{\boldsymbol{\lambda}\boldsymbol{\varphi}}^{\boldsymbol{z}} = \left[\frac{1}{2}\int_{0}^{T}\int_{0}^{T}\frac{\partial R\left(t_{1}, t_{2}; \tau\right)}{\partial \tau}\frac{\partial W\left(t_{1}, t_{2}; \tau\right)}{\partial \tau}dt_{1}dt_{2}\right]^{-1}, \qquad (7.2.24)$$

where $R(t_1, t_2; \tau)$ - function of correlation of the signal with noise (7.2.1);

 $W(t_1, t_2; \tau) - its$ reciprocal in interval (0, T) (Chapters I, IV).

In accordance with (4.3.9) for $\Delta f \sim T \ll 1$

Performing the calculation in (7.2.24), we obtain

$$a_{ab}^{2} = \frac{1+\mu}{2\mu^{1}(b-a^{2})},$$
 (7.2.25)

where the signal-to-noise ratio $\mu \approx P_c T/2N_0$ is equal to the ratio of energy of the received signal during the time of observation T to the physical (one-way) spectral density of noise $2N_0$. Let us note that formula (7.2.25) formally coincides with the expression for $\sigma_{g\phi}^2$ from (7.2.23), if we consider the spectrum of fluctuations of the signal rectangular, with width $\Delta f_c = \frac{1}{T}$.

§ 7.3. Discriminator with Two Channels Detuned with Respect to Range

7.3.1. Block Diagram of the Discriminator

Exact technical fulfillment of an optimum discriminator circuit is hardly possibly. In practice, modulation of the reference signal utilized in the discriminator will somewhat differ from modulation of the sounding signal, the frequency response of the filter will differ from the optimum, etc. In this and subsequent paragraphs we shall consider the influence of different deviations from optimum

-218-

processing of the signal on accuracy of range finding.

One approximate method of realizing an optimum discriminator is directly indicated by formula (7.2.6) and consists of replacement of the derivative in the expression for z(t) by a finite difference. Since

$$\frac{\partial |Q(t;\tau)|^2}{\partial \tau} \approx \frac{1}{2\partial} \{ |Q(t,\tau+\delta)|^2 - |Q(t,\tau-\delta)|^2 \},\$$

output voltage of the optimum discriminator with an accuracy of an immaterial constant coefficient is approximately determined by the following quantity:

$$z(t, \Delta) \approx |Q(t, \tau+\delta)|^{s} - |Q(t, \tau-\delta)|^{s} = = \left| \int_{-\infty}^{t} h(t-s) u(s-\tau-\delta) y(s) e^{t-\sigma^{s}} ds \right|^{2} - - \left| \int_{-\infty}^{t} h(t-s) u(s-\tau+\delta) y(s) e^{t-\sigma^{s}} ds \right|^{2}.$$
(7.3.1)

It is obvious that the degree of approximation in general is better, the less the magnitude of detuning δ ; quantitative characteristics of the degree of approximation of operation (7.3.1) to (7.2.6) can be obtained in the course of analysis.

Operations (7.3.1) are realized by two channels, each of which is built just as an optimum receiver of a rapidly fluctuating signal in conditions of detection



Fig. 7.5. Block diagram of a discriminator with two staggered channels: 1 - preamplifier; 2 - controlled modulated coherent local oscillator; 3 - mixer; 4 - narrow-band filter; 5 - square-law detector; 6 - subtractor.

 $s_1(t) = \operatorname{Re} u_1(t-\tau-\delta) e^{\frac{1}{2}(\omega_0-\omega_{\mu\nu})t} s_2(t) = \operatorname{Re} u_1(t-\tau+\delta) e^{\frac{1}{2}(\omega_0+\omega_{\mu\nu})t}.$

(Chapter IV). Reference signals utilized in the channels have a difference in delays of 2δ . The corresponding block diagram is shown in Fig. 7.5. The received signal y(t) is fed to two mixers, in which it is multiplied by the expected signal, shifted in frequency the magnitude of intermediate frequency $\omega_{\rm RP}$ and in delay by quantities $\tau_0 - \Delta + 5$ and $\tau_0 - \Delta - \delta$, respectively

 $(\Delta = \tau_0 - \tau)$. In practice heterodyning of the signal and its multiplication by a modulating signal, for instance, a delayed strobe in pulse modulation, can be carried out separately, but for analysis this is not important. Output signals of mixers are filtered in narrow-band filters, tuned to the intermediate frequency, and, after detection by square-law detectors, they are subtracted. The physical

essence of all these operations, besides subtraction, obviously, is the same as during detection (Chapter IV), and needs no further discussion.

In practice, reference signals in such a circuit may differ somewhat from sounding signals (for instance, noncoincidence of the form of the strobe and the pulse), and also due to any kind of nonidentity may differ among themselves. Therefore, in general, in one of the channels there is used reference signal $u_1(t) \neq i(t)$, and in other, $u_2(t) \neq u(t)$. Furthermore, the frequency response of the narrow-band filters may also differ from optimum (7.2.5). Taking into account these circumstances output voltage of the discriminator of Fig. 7.5, in general, is described by the following expression:

$$z(t, \Delta) = \left| \int_{-\infty}^{t} h_{1}(t-s) u_{1}(s-\tau_{0}+\Delta-\delta) y(s) e^{i\omega_{0}s} ds \right|^{2} - \left| \int_{-\infty}^{t} h_{2}(t-s) u_{1}(s-\tau_{0}+\Delta+\delta) y(s) e^{i\omega_{0}s} ds \right|^{2}.$$
(7.3.2)

where to pulse responses $h_1(t)$ and $h_2(t)$ there correspond frequency responses $H_1(i\omega)$ and $H_2(i\omega)$, and the true value of delay in the formula for the correlation function of y(t) (7.1.1) is designated τ_0 .

7.3.2. Identical Modulation of the Reference and Sounding Signals Of all idealizations of real discriminators, probably, the most fundamental is the difference of the frequency responses of narrow-band filters from optimum, although quantitively this distinction is not necessarily the most important. The fact is that in an optimum system the frequency response of the filter depends on the magnitude of the signal-to-noise ratio and should, in principle, be corrected in the course of operation as the latter changes.

It is of considerable interest to consider a discriminator with filters not depending on the signal-to-noise ratio h, and to investigate its behavior during change of this ratio. In particular, it is interesting to consider that case when for a certain h we achieve coincidence of the frequency responses of optimum and real filters, but for all others they differ.

In order to investigate the influence of the difference of frequency responses of filters from optimum we shall first consider a somewhat idealized case, when $u_1(t) = u_2(t) = u(t)$, $h_1(t) = h_2(t) \neq h(t)$. Presenting in (7.3.2) the squares of moduli in the form of the product of complex conjugate integrals and averaging the the product $y(s_1)y(s_2)$, we obtain the following expression for the discrimination characteristic:

-220-

$$\overline{z(t, \Delta)} = \int_{-\infty}^{t} \int_{-\infty}^{t} h_{1}(t-s_{1}) h_{1}(t-s_{2}) [u(s_{1}-\tau_{0}+\Delta-\delta) \times \\
\times u^{*}(s_{2}-\tau_{0}+\Delta-\delta) - u(s_{1}-\tau_{0}+\Delta+\delta) \times \\
\times u^{*}(s_{2}-\tau_{0}+\Delta+\delta)] e^{i\pi_{0}(s_{1}-s_{0})} [N_{0}\delta(s_{1}-s_{2}) + \\
+ \operatorname{Re} P_{c} u(s_{1}-\tau_{0}) u^{*}(s_{2}-\tau_{0}) p(s_{1}-s_{2}) e^{i\pi_{0}(s_{1}-s_{0})}] ds_{1} ds_{2} = \\
= \begin{cases} \left| \lim_{T \to \infty} \frac{1}{T} \int_{0}^{T} u(s+\Delta-\delta) u^{*}(s) ds \right|^{2} - \\
- \left| \lim_{T \to \infty} \frac{1}{T} \int_{0}^{T} u(s+\Delta+\delta) u^{*}(s) ds \right|^{2} \right\} \times \\
\times \frac{P_{c}}{2} \int_{-\infty}^{t} \int_{-\infty}^{t} h_{1}(t-s_{1}) h_{1}(t-s_{2}) p(s_{1}-s_{3}) ds_{1} ds_{2} = \\
= \frac{P_{0}}{2\Delta f_{0}} \left[|C(\Delta-\delta)|^{2} - |C(\Delta+\delta)|^{2} \right] \frac{1}{2\pi} \int_{-\infty}^{\infty} |H_{1}(i\infty)|^{2} S_{0}(\infty) d\omega, \quad (7.3.3)$$

from which it follows that null shift $\Delta_{_{\rm O}} \approx$ 0, and steepness of the discrimination characteristic

i }

$$K_{\mathbf{X}} = -\operatorname{Re}\left[C'(\delta) C^{*}(\delta)\right] \xrightarrow[\pi \Delta f_{\circ}]{\infty} \int_{-\infty}^{\infty} |H_{1}(i\omega)|^{2} S_{o}(\omega) d\omega. \qquad (7.3.4)$$

This equality is satisfied thanks to properties of the real and imaginary parts of $C(\Delta)$, which guarantees that $C(-\Delta) = C^*(\Delta)$ and $C'(-\Delta) = -C^{*'}(\Delta)$.

Likewise, we can obtain an expression for the correlation function, and through it an expression for the spectral density of the output voltage. We shall do this in some detail:

$$R_{s}(t_{1}, t_{2}) = \overline{z(t_{1}, \Delta) z(t_{2}, \Delta)} - \overline{z(t_{1}, \Delta) z(t_{2}, \Delta)} = \int_{t_{1}}^{t_{1}} \int_{t_{1}}^{t_{2}} \int_{t_{2}}^{t_{2}} \int_{t_{2}}^{t_{2}} \int_{t_{2}}^{t_{2}} e^{i\omega_{0}(s_{1}-s_{2}+s_{2}-s_{1})}h_{1}(t_{1}-s_{1})h_{1}(t_{1}-s_{2}) \times \\ \times h_{1}(t_{2}-s_{2})h_{1}(t_{2}-s_{4})[u(s_{1}-\tau_{0}+\Delta-\delta)u^{*}(s_{2}+\tau_{0}+\Delta-\delta) - u(s_{1}-\tau_{0}+\Delta+\delta)u^{*}(s_{2}-\tau_{0}+\Delta+\delta)] \times \\ \times u(s_{2}-\tau_{0}+\Delta-\delta)u^{*}(s_{4}-\tau_{0}+\Delta-\delta)] - \\ - [u(s_{4}-\tau_{0}+\Delta-\delta)u^{*}(s_{4}-\tau_{0}+\Delta-\delta)] - \\ - [u(s_{4}-\tau_{0}+\Delta+\delta)u^{*}(s_{4}-\tau_{0}+\Delta+\delta)] \times \\ \times [\overline{y(s_{1})y(s_{2})y(s_{4})y(s_{4})} - \overline{y(s_{1})y(s_{2})y(s_{4})}] ds_{1}ds_{2}ds_{2}ds_{4}.$$
(7.3.5)

Further calculations can be made with the help of the known expression for the fourth mixed moment of a multi-dimensional normal distribution

$$\begin{array}{l} \overline{y}(\overline{s_{1}})\overline{y}(\overline{s_{2}})y(\overline{s_{2}})y(\overline{s_{4}}) = y(\underline{s_{1}})y(\underline{s_{2}})y(\underline{s_{4}}) + \\ + \overline{y}(\underline{s_{1}})\overline{y}(\underline{s_{2}})\overline{y}(\underline{s_{4}}) + \overline{y}(\underline{s_{1}})\overline{y}(\underline{s_{4}})\overline{y}(\underline{s_{4}})y(\underline{s_{4}}) = \\ = R(\underline{s_{1}}, \underline{s_{2}})R(\underline{s_{2}}, \underline{s_{4}}) + R(\underline{s_{1}}, \underline{s_{3}})R(\underline{s_{2}}, \underline{s_{4}}) + \\ + R(\underline{s_{1}}, \underline{s_{4}})R(\underline{s_{2}}, \underline{s_{3}}). \end{array}$$

$$(7.3.6)$$

During calculation of statistical characteristics of quantity $z(t, \Delta)$ one should consider the following circumstance, appearing due to use of complex notation, and essentially simplifying the calculations. Output voltage of the discriminator $z(t, \Delta)$ in optimum and nonoptimal circuits is in the form of the sum of a product of form $\xi(t)\eta^*(t)$, where $\xi(t)$ and $\eta(t)$ are complex high-frequency normal random processes, whose total spectrum width is small as compared to the magnitude of the carrier frequency, identical for both processes.

A known property of such processes is equality to zero of the mathematical expectation of products $\overline{\xi(t_1)\xi(t_2)}$ and $\overline{\xi(t_1)\eta(t_2)}$ for any t_1 and t_2 . Then from (7.3.6) it follows that

$\overline{\xi(t_1) \eta^*(t_1) \xi(t_2) \eta^*(t_3)} = \overline{\xi(t_1) \eta^*(t_1) \xi(t_2) \eta^*(t_3)} + \frac{1}{1 + \overline{\xi(t_1) \eta^*(t_3) \xi(t_2) \eta^*(t_3)}}$

This circumstance gives us the possibility during substitution of (7.3.6) in (7.3.5) to reject product $R(s_1, s_3)R(s_2, s_4)$, integration of which with factor exp $\{i\omega_0(s_1 - s_2 + s_3 - s_4)\}$ all the same gives zero thanks to rapid oscillation of the integrand. Substituting the remaining terms in the integral, replacing $s - \tau_0$ by s, and rejecting the remaining high-frequency terms, we obtain

$$R_{z}(t_{1}, t_{2}) = \int_{-\infty}^{t_{1}} \int_{-\infty}^{t_{1}} \int_{-\infty}^{t_{2}} \int_{-\infty}^{t_{2}} h_{1}(t_{1} - s_{1}) h_{1}(t_{1} - s_{2}) \times \\ \times h_{1}(t_{2} - s_{2}) h_{1}(t_{2} - s_{4}) [u(s_{1} + \Delta - \delta) u^{*}(s_{2} + \Delta - \delta) - u(s_{1} + \Delta + \delta)] [u(s_{2} + \Delta + \delta)] [u(s_{3} + \Delta - \delta) u^{*}(s_{4} + \Delta - \delta) - u(s_{4} + \Delta + \delta)] [u(s_{3} + \Delta + \delta)] \times \\ \times \left[\frac{1}{4} P_{z}^{2} u^{*}(s_{1}) u(s_{2}) u^{*}(s_{2}) u(s_{4}) \rho(s_{1} - s_{4}) \rho(s_{2} - s_{4}) + h_{0}^{2} \delta(s_{1} - s_{4}) \delta(s_{2} - s_{4}) + \frac{1}{2} P_{c} N_{s} u^{*}(s_{1}) u(s_{4}) \times \\ \times \rho(s_{1} - s_{4}) \delta(s_{2} - s_{4}) + \frac{1}{2} P_{c} N_{s} u^{*}(s_{2}) \rho(s_{2} - s_{3}) \times \\ \times \delta(s_{1} - s_{4}) \left[ds_{1} ds_{2} ds_{4} ds_{4} ds_{4} \right]$$

Averaging in this expression modulating functions u(t) over the time under the sign of the integral, integrating the expression for $R_z(t_1, t_2)$ over $t_2 - t_1$, and dividing the obtained expression for spectral density at zero by the square of steepness, for the fluctuation characteristic of the discriminator we have

$$S_{0 HB}(\Delta) = \frac{1}{2\pi} \int_{-\infty}^{\infty} |H_{1}(i\omega)|^{4} \left\{ \frac{1}{2} [|C(\Delta + \delta)|^{2} - \frac{|C(\Delta - \delta)|^{2}}{2\pi} + \frac{|C(\Delta - \delta)|^{2}}{2\pi} - \frac{|C(\Delta - \delta)|^{2}}{2\pi} + \frac{|C(\Delta - \delta)|^{2}}{2\pi}$$

It is not difficult to prove that with small detunings δ and optimum frequency response this expression completely coincides with (7.2.20). From (7.3.7) for equivalent spectral density $S_{3KB} = S_{5KB}(0)$ for zero mismatch we can obtain the following expression:

$$\frac{\frac{1}{2\pi}\int_{-\infty}^{\infty} |H_{1}(i\omega)|^{4} \{1-|C(2\vartheta)|^{2}+2hS_{\theta}(\omega)[|C(\vartheta)|^{2}-\operatorname{Re}C^{*}(2\vartheta)C^{2}(\vartheta)]\}d\omega}{8h^{2}\left[\operatorname{Re}C'(\vartheta)C^{*}(\vartheta)\frac{1}{2\pi}\int_{-\infty}^{\infty} |H_{1}(i\omega)|^{2}S_{\theta}(\omega)d\omega\right]^{2}}.$$
(7.3.8)

Investigating the dependence of this expression on ô, it is simple to prove (this is easily done, considering small δ , when $C(\delta) \approx 1 + ia\delta - \frac{1}{2}b\delta^2$), that $S_{\partial KB}$ monotonically decreases with decrease of 5, passing to the limit as $\delta \rightarrow 0$, which for $|H_1(i\omega)|^2 = |H_0(i\omega)|^2$ from (7.2.5) is equal to S_{OUT} from (7.2.15). This fact leads to the requirement of realization of small detuning between channels of the discriminator. However, with decrease of δ proportionally to $(b - a^2)\delta$ steepness of the discrimination characteristic decreases. This means that to preserve the required high speed operation of the closed measuring system it is necessary proportionally to decrease of 5 to increase gain in the open circuit of the range finder. This circumstance forces us to take in real systems, built on the considered scheme, such δ for which steepness K_{π} is not very small or is even maximum. Fortunately, as we shall subsequently prove in concrete examples, the value of S_{AHB} with increase of δ up to values of $\delta,$ turning into a maximum the steepness $K_{\rm g},$ differs little from the value of $S_{_{\mbox{\tiny SKB}}}$ with zero detuning δ . This gives us the possibility, without specifying the form of modulation, to investigate the dependence of Samme on characteristics of filters.

For δ , close to zero, using the expansion for $C(\delta)$, from formula (7.3.8) we obtain:

$$S_{\mathbf{0}\mathbf{X}\mathbf{R}} = \frac{1}{2h^{2}(b-a^{2})} \frac{\frac{1}{2\pi} \int_{-\infty}^{+\infty} |H_{1}(i\omega)|^{4} [1+hS_{4}(\omega)]d\omega}{\left[\frac{1}{2\pi} \int_{-\infty}^{+\infty} |H_{1}(i\omega)|^{4}S_{4}(\omega)d\omega\right]^{2}} \cdot (7.3.9)$$

Let us calculate the magnitude of S_{3KB} for certain of the most commonly used forms of spectral density of fluctuations and frequency response of a filter. If spectral density $S_{\Omega}(\omega)$ is uniform in band Δf_{C} , and the filter has a rectangular frequency response with band $\Delta f_{\tilde{\Phi}}$, then

$$S_{\bullet XB} = \begin{cases} \frac{1+h}{2(b-a^2) \Delta f_{\bullet} h^2}, & \Delta f_{\bullet} < \Delta f_{\circ}, \\ \frac{\Delta f_{\bullet}}{\Delta f_{\bullet}} + h & (7.3.10) \end{cases}$$

If the correlation function of the signal is approximated by an exponential function, i.e.,

$$S_{\bullet}(\bullet) = \frac{1}{1 + \left(\frac{\bullet}{24/\bullet}\right)^2},$$
 (7.3.11)

and narrow-band filtration is carried out by an LRC-filter with the response of a low-frequency equivalent

 $|H_1(i\bullet)|^2 = \frac{1}{1 + \left(\frac{\Phi}{2\Delta f_0}\right)^2}, \qquad (7.3.12)$

where $\Delta f_c = cffective$ width of the spectrum of fluctuations; $\Delta f_{\phi} = effective$ pass band of the filter, then

$$S_{0HB} = \frac{\left(1 + \frac{\Delta f_{\phi}}{\Delta f_{\bullet}}\right)^2 + \hbar \left(1 + 2 \frac{\Delta f_{\phi}}{\Delta f_{\bullet}}\right)}{4 \left(b - a^2\right) \hbar^2 \Delta f_{\phi}}.$$
 (7.3.13)

If spectral density $S_0(\omega)$ has the form (7.3.11), and filtration is carried out by a filter with a rectangular frequency response, then

$$S_{\text{DEED}} = \frac{\frac{\pi \Delta f_{\phi}}{2\Delta f_{o}} + h \operatorname{atctg} \frac{\pi \Delta f_{\phi}}{2\Delta f_{o}}}{\frac{4h^{2}}{\pi} \Delta f_{o}(b - a^{2}) \left(\operatorname{arctg} \frac{\pi \Delta f_{\phi}}{2\Delta f_{o}}\right)^{2}}, \qquad (7.3.14)$$

-224-

This expression practically coincides for all values of $\Delta f_{ij}/\Delta f_{ij}$ with (7.3.10).

Let us further consider the case of a spectrum of fluctuations, approaching $\pm \infty$ somewhat more rapidly than (7.3.11), when

$$S_{\bullet}(\mathbf{s}) := \frac{1}{\left[1 + \left(\frac{\mathbf{s}}{4\Delta f_{\bullet}}\right)^{2}\right]^{2}}, \qquad (7.3.15)$$

and that of the corresponding filter

$$|H_1(i\boldsymbol{\omega})|^2 = \frac{1}{\left[1 + \left(\frac{\boldsymbol{\omega}}{4\Delta f_{\phi}}\right)^2\right]^2}.$$
 (7.3.16)

In this case

$$S_{\text{PRB}} = \frac{5(1+x)^4 + h[5+20x+29x^2+16x^3]}{16(b-a^2)\Delta f_{\text{PR}}(1+2x)^3}, \qquad (7.3.17)$$

where $x = \frac{\Delta f_{\phi}}{\Delta f_{\phi}}$.

Comparing these expressions with formulas for the case of an optimum filter (7.2.15) for a uniform spectrum of fluctuations in band Δf_c and a rectangular frequency response, we obtain

$$\frac{S_{ans}}{S_{out}} = \begin{cases} \frac{1}{x}, & x < 1, \\ \frac{h+x}{1+h}, & x > 1. \end{cases}$$
(7.3.18)

For spectrum (7.3.11) and filter (7.3.12) this ratio is equal to

$$\frac{S_{ms}}{S_{ms}} = \frac{(1+x^{s}) + h(1+2x)}{2x \sqrt{1+h}(1+\sqrt{1+h})}.$$
(7.3.19)

Dependences (7.3.18) and (7.3.19) are shown in Fig. 7.6. Analysis of them shows that, as one should have expected, ratio S_{HB}/S_{ORT} for a certain $x = \frac{\hbar f_0}{\Delta f}$ attains a minimum, equal to one. For a rectangular spectrum this minimum is reached at x = 1, and for the spectrum of (7.3.11) it is when x = $\sqrt{1 + h}$. Expansion of the passband of the filter, which is most interesting from the technical point of view, leads to especially unpleasant consequences for a small signal-tonoise ratio. When h << 1 ratio S_{HE}/S_{ORT} has the order of x, i.e., fluctuation error grows proportionally to the passband of the filter. For larger values of h expansion of the band plays a considerably smaller role. Ratio S_{HE}/S_{ORT} here has

-225-

the order 1 + (0.5 to 1) $\frac{x}{h}$, and with growth of F it approaches one. At working signal-to-noise ratios (h ~ 3 to 10) and expansion of the band to x ~ 10 increase of S₃₄₈ does not exceed 50% for the spectrum of (7.7.11) and 3% for a rectangular spectrum.



Fig. 7.6. Influence of the passband of the filter on accuracy of range finding: ---- rectangular spectrum of fluctuations and rectangular frequency response of the filter; --- exponential correlation function of fluctuations and exponential pulse response of the filter.

Let us give another expression for the spectral density of parametric fluctuations. Expanding $S_{SKB}(\Delta)$ from (7.3.7) in the vicinity of $\Delta = 0$, we obtain



This expression shows that, in distinction from an optimum circuit, in the given case parametric fluctuations increase without limit with decrease of h;

-226-

however, since as $h \rightarrow 0$ normal fluctuations grow as $1/h^2$, they play a considerably large role. With high signal-to-noise ratios h, when the influence of parametric fluctuations is substantial, the term with 1/h in S_{map} is small and does not play a role. Furthermore, the coefficient of this term, depending on δ , for not very large values of δ is small. As $\delta \rightarrow 0$ it has the order of δ^2 , increasing with increase of δ up to values, ensuring maximum steepness of the order of one. Let us note that the tendency of S_{map} for small δ to reach a finite limit, not depending on h, is one more basis for selection in real circuits of sufficiently small detunings.

It is interesting to note that the main component of parametric fluctuations, not depending on the signal-to-noise ratio in the two-channel circuit, depends neither on the magnitude of detuning nor the form of modulation in general. This component is equal to

$$\lim_{\Lambda \to \infty} S_{\mu \alpha p} = \frac{\frac{1}{2\pi} \int_{-\infty}^{\infty} |H_1(i\omega)|^4 S_0^2(\omega) d\omega}{\left[\frac{1}{2\pi} \int_{-\infty}^{\infty} |H_1(i\omega)|^2 S_0(\omega) d\omega\right]^2} = \frac{z}{\Delta f_0}, \qquad (7.3.21)$$

where coefficient α has the order of one, insignificantly changing with change of spectral density of the signal and of frequency response of the filter. In particular, for rectangular $S_0(\omega)$ and $|H_1(i\omega)|^2 (\Delta f_{\tilde{\Phi}} \geq \Delta f_c)$ coefficient $\alpha = 1$, and for $S_0(\omega)$ and $|H_1(i\omega)|^2$ from (7.3.11) and (7.3.12)

$$a = \frac{1+x}{x} + \frac{1}{1+x} \quad (x = \Delta f_{\phi} / \Delta f_c).$$

7.3.3. Influence of the Finite Magnitude of Detuning

Without specifying yet the form of modulation, we can make certain important conclusions about the influence of the finite magnitude of detuning δ on characteristics of the discriminator. Turning to formula (7.3.8) for S_{GKB} with arbitrary detuning, it is easy to see that it reduces to formula (7.3.9) for equivalent spectral density with zero detuning if we introduce certain equivalent magnitudes of the signal-to-noise ratio $h_{\partial KB}$ and mean square spectrum width (b - a^2)_{BKB} and takes the form

$$S_{3RB} = \frac{1}{2k_{3RB}^2 (b - a^3)_{3RB}} \times \frac{\frac{1}{2\pi} \int_{-\infty}^{\infty} |H_1(i\omega)|^4 [1 + k_{3RB} S_0(\omega)] d\omega}{\left[\frac{1}{2\pi} \int_{-\infty}^{\infty} |H_1(i\omega)|^2 S_0(\omega) d\omega\right]^2}, \quad (7.3.22)$$

where h_{BHB} and $(b - a^2)_{BHB}$ depend on detuning and are equal to

$$h_{\mu KB} = h \frac{2 [[C(4)]^2 - \text{Re} C^*(2) C^2(4)]}{1 - [C(20)]^4}, \qquad (7.3.23)$$

$$(b - a^2)_{\mu KB} = \frac{[\text{Re} C^*(6) C^*(6)]^2 [1 - [C(23)]^2]}{[[C(4)]^2 - \text{Re} C^*(6)]^2 (1 - [C(23)]^2]}.$$

$$-a^{2})_{9K8} = \frac{1}{\|C(\hat{a})\|^{2} - \text{Re}C^{*}(2\hat{a})C^{2}(\hat{a})\|^{2}}.$$
(7.3.24)

Thereby, the nature of the dependence of accuracy of measurement on characteristics of narrow-band filters, on the ratio of bands of fluctuations and of the filter; studied at zero detuning, completely is preserved with arbi+rary detuning, and quantitative calculations for the last case reduce to substitution in the formulas (7.3.9), (7.3.10), (7.3.13), (7.3.14) and (7.3.17) obtained above of quantities $h_{\rm SKB}$ and (b - a²)_{SKB}. Similarly we can also use graphs giving the dependence of $S_{\rm SKB}/S_{\rm OHT}$ and $S_{\rm SKB}$ on the ratio of bands $\Delta f_{\rm ch}/\Delta f_{\rm c}$ and the signal-to-noise ratio h.

Selection of a finite magnitude of detuning leads to a certain change of the frequency response of the filter, ensuring minimum equivalent spectral density. If from formula (7.3.9) for the case of zero detuning this characteristic is equal to the optimum (7.2.5), which it is possible to show by direct variation of expression (7.3.9), with finite detuning the minimum of S_{aux} , obviously, is at

$$[H_{j0Hy}(im)]^{2} = \frac{h_{BHB}S_{0}(m)}{1 + h_{BHB}S_{0}(m)}, \qquad (7.3.25)$$

In particular, for a spectrum of form (7.3.11) this leads to change of the optimum band to

$$\Delta f_{cora} = \Delta f_o \sqrt{1 + h_{sks}}.$$

The nature of the dependence of h_{BHB} and $(b - a^2)_{BHB}$ on the magnitude of detuning we shall study in examining concrete forms of modulation. However, it is possible to indicate that, in general, the magnitude of ratio $h_{BHD}/h < 1$, and ratio

-228-

 $(b - a^2)_{REP}/(b - a^2)$ may be larger or less than one, where $(b - a^2)_{REP}/(b - a^2) < 1$ only for such signals for which C(5) does not have zeroes for finite 5 and function C(5) decreases to infinity no faster than its derivative C'(5). In most real cases $(b - a^2)_{REP}/(b - a^2) > 1$, but so that the product $\frac{(b-a^2)_{REP}}{b-a^2} \cdot \frac{h_{REP}}{h}$, inversely proportional. to which spectral density increases due to finite detuning for large h, and all the more so product $\frac{(b-a^2)_{REP}}{b-a^2} \cdot \frac{h_{REP}^2}{h^2}$, characterizing this increase for small h, is always smaller than one.

As we shall subsequently prove, in working conditions, i.e., for δ , smaller or of the order of δ_{MARY} , both h_{ACB}/h , and also $(b - a^2)_{ACB}/(b - a^2)$ in many cases differ so insignificantly from one that it is unnecessary to talk of any influence of detuning.

7.3.4. Noncoincidence of Modulations of Reference and Sounding Signals Let us consider now the difference in the forms of modulating voltages of the sounding and reference signals. It is obvious that, assuming $u_1(t) \neq u_2(t)$, we cover also the most simple case of imbalance of gain in two channels, for which $u_1(t)$ and $u_2(t)$ coincide in form, but differ in amplitude. Returning to formula (7.3.2) for output voltage of the discriminator in general, considering again $h_1(t) = h_2(t)$ (the most important nonidentity of filters - the difference in gain factors - we already allowed for, considering $u_1(t) \neq u_2(t)$) and producing necessary operations of averaging and integration just as in the preceding paragraph, it is possible similarly to (7.3.3) to obtain the following expression for the discrimination characteristic:

$$\frac{\overline{z(i, \Delta)} = N_{\bullet}\Delta_{1}^{\bullet} [C_{11}(0) - C_{12}(0)] + \\
+ [|C_{10}(\Delta - \delta)|^{2} - |C_{20}(\Delta + \delta)|^{2}] \times \\
\times \frac{P_{0}}{2\Delta_{1}^{\bullet}} \frac{1}{2\pi} \int_{-\infty}^{\infty} |H_{1}(i\infty)|^{2} S_{0}(\infty) d\infty,$$
(7.3.26)

where

$$C_{ik}(\delta) = \lim_{T \to \infty} \frac{1}{T} \int_{0}^{T} u_{i}(t+\delta) u^{*}_{k}(t) dt, \qquad (7.3.27)$$

and $u_0(t) = u(t)$, i.e., subscript 0 pertains to the sounding signal. From (7.3.26) we can obtain the following formulas for steepness and systematic error:

$$K_{R} = \frac{P_{e}}{\delta f_{e}} \frac{1}{2\pi} \int_{-\infty}^{\infty} |H_{1}(i\omega)|^{2} S_{e}(\omega) d\omega \times \times \\ \times \operatorname{Re} [C_{1e}^{*}(-\delta) C_{1e}^{*}(-\delta) - C_{1e}^{*}(\delta) C_{1e}^{*}(\delta)], \qquad (7.3.28) \\ A_{e} = \begin{cases} \frac{|C_{11}(0) - C_{12}(0)|}{2\pi \delta f_{e}} + |C_{1e}(-\delta)|^{2} - |C_{1e}(\delta)|^{2} \\ \frac{1}{2} \int_{-\infty}^{+} |H_{1}(i\omega)|^{2} S_{e}(\omega) d\omega \end{cases} \\ : \operatorname{Re} [C_{1e}^{*}(-\delta) C_{1e}^{*}(-\delta) - C_{2e}^{*}(\delta) C_{2e}^{*}(\delta)]. \qquad (7.3.29) \end{cases}$$

Likewise, we can find the expression for equivalent spectral density (we shall limit ourselves to the formula for zero mismatch):

$$S_{0555} = \left\{ \frac{1}{2\pi} \int_{-\infty}^{\infty} |H_{1}(i\omega)|^{4} [|C_{11}(0)|^{4} + |C_{33}(0)|^{4} - \frac{|C_{13}(2\delta)|^{4} - |C_{33}(2\delta)|^{4} + 2hS_{0}(\omega)(C_{11}(0)|C_{10}(-\delta)|^{4} + C_{35}(0)|C_{30}(\delta)|^{4} - \operatorname{Re} C_{15}(-2\delta)C_{10}^{*}(-\delta)C_{30}^{*}(-\delta) - - \operatorname{Re} C_{31}(2\delta)C_{30}^{*}(\delta)C_{30}^{*}(\delta)) + h^{4}S_{0}^{2}(\omega)(|C_{10}(-\delta)|^{4} - \frac{|C_{30}(\delta)|^{4}}{\pi}\operatorname{Re}[C_{10}^{*}(-\delta)C_{10}^{*}(-\delta) - C_{30}^{*}(\delta)C_{30}^{*}(\delta)] \times \left\{ \frac{h}{\pi}\operatorname{Re}[C_{10}^{*}(-\delta)C_{10}^{*}(-\delta) - C_{30}^{*}(\delta)C_{30}^{*}(\delta)] \times \int_{-\infty}^{\infty} |H_{1}(i\omega)|^{4}S_{0}(\omega)d\omega \right\}^{4}.$$

$$(7.3.30)$$

These results cover a broad class of cases and permit us for all concrete examples to investigate the dependence of systematic and fluctuating errors on the form of the reference signals. Let us consider certain particular cases. We separate first of all conditions in which there is no systematic error. As follows from (7.3.29), $\Delta_0 = 0$ for all h, if

1)
$$C_{11}(0) = C_{12}(0);$$
 2) $|C_{10}(-\delta)|^2 = |C_{10}(\delta)|^2.$ (7.3.31)

The first condition requires equality of powers of reference signals and includes the condition of equality of gain factors in the channels. The second condition requires symmetry of reference signals, expressed in equal correlation of "lagging" and "leading" reference signals with the expected signal. Both conditions are normal specifications, fulfillment of which should be sought during practical developments.

If (7.3.31) is satisfied, in the formula for S_{3KB} the component not depending on the signal-to-noise ratio turns into zero, and $S_{3KB} \rightarrow 0$ as $h \rightarrow \infty$, as also for coinciding modulation.

The formula for S_{reen} is considerably simplified if we assume that laws of

-230-

modulation of reference signals are identical. If the second of conditions (7.3.31)is simultaneously satisfied (the first is always satisfied when $u_1(t) = u_2(t)$), from (7.3.30) there follows:

$$S_{\Theta NB} = \left\{ \frac{1}{2\pi} \int_{-\infty}^{\infty} |H_{1}(i\omega)|^{4} ||C_{11}(0)|^{4} - |C_{11}(2\delta)|^{4} + \frac{1}{2hS_{0}}(\omega)(C_{11}(0)|C_{10}(-\delta)|^{4} - \frac{1}{2hS_{0}}(-\delta)C_{10}(-\delta)C_{10}(\delta)|^{4} + \frac{1}{2}hS_{0}(\omega)(C_{11}(-\delta)C_{10}(-\delta)C_{10}(\delta)|^{4} + \frac{1}{2\pi}\int_{-\infty}^{\infty} |H_{1}(i\omega)|^{2}S_{0}(\omega)d\omega \right\}$$

$$:2h^{2} \left\{ \operatorname{Re}\left[C_{10}^{\prime}(-\delta)C_{10}^{\ast}(-\delta)-C_{10}^{\prime}(\delta)C_{10}^{\ast}(\delta)\right] \times \frac{1}{2\pi} \int_{-\infty}^{\infty} |H_{1}(i\omega)|^{2}S_{0}(\omega)d\omega \right\}.$$

$$(7.3.32)$$

If detuning of reference signals with respect to delay is so small that it is possible to be limited to first terms in the expansion of functions $C_{ik}(\delta)$, i.e.,

$$C_{ik}(\delta) = |C_{ik}(0)| \left(1 + ia_{ik}\delta - \frac{1}{2}b_{ik}\delta^2\right),$$

from (7.3.30) we can obtain a formula generalizing (7.3.9) for the case of noncoincident modulation:

$$S_{0RB} = \frac{\frac{1}{2\pi} \int_{-\infty}^{\infty} |H_{1}(i\omega)|^{4} [|C_{11}(0)|^{1} (b_{11} - a_{11}^{2}) + \frac{1}{2h^{2} (\operatorname{Re} b_{10} - a_{10}^{2})^{2} |C_{10}(0)|^{4} \times \frac{1}{hS_{0}(\omega) C_{11}(0) |C_{10}(0)|^{4} (b_{11} - 2a_{11}a_{10} - a_{10}^{2})] d\omega}{\times \left[\frac{1}{2\pi} \int_{-\infty}^{\infty} |H_{1}(i\omega)|^{2} S_{0}(\omega) d\omega\right]^{1}}, \qquad (7.3.33)$$

Analysis of formulas (7.3.30), (7.3.32) and (7.3.33) shows that the presence of any nonidentities and noncoincidences leads to increase of the equivalent spectral density. In general there appears a component of S_{3KB} , caused only by fluctuations of the signal and not vanishing with infinite increase of the signal-to-noise ratio.

If this component is absent, the presence of nonidentities changes the distribution between components of S_{0KF} caused by herets of the signal with noise and of noise with noise. In the simplest case, when $\sqrt{b\delta} \ll 1$ and $b_{11} \approx \text{Re } b_{10} \approx b$, $a_{11} \approx a_{10} \approx a$, the presence of nonidentities reduces to equivalent decrease of the signal-to-noise ratio to
$$h_{1} = h \frac{|C_{10}(0)|^{2}}{C_{11}(0)}$$
(7.3.34)

and (7.3.33) passes into (7.3.9) with replacement of h by h_1 .

In the more general case we are able, just as in Paragraph 7.3.3, to reduce formulas (7.3.32) and (7.3.33) to a very simple formula for the case of coinciding modulations and zero detuning (7.3.9) by introducing the equivalent signal-to-noise ratio h_{3KB} and the mean square spectrum width (b - $a^2)_{3KB}$. Then for arbitrary modulation of reference signals $u_1(t) = u_2(t) \neq u(t)$ and arbitrary detuning we again have formula (7.3.22), where

$$h_{0KB} = h \frac{2[|C_{10}(-\delta)|^2 - \text{Re} C_{11}(-2\delta) C_{10}(-\delta) C_{10}(\delta)]}{1 - |C_{11}(-2\delta)|^2}, \quad (7.3.35)$$

$$(b - a^{*})_{0 \times n} = \frac{\{\operatorname{Re} [C'_{10} (-\delta) C^{*}_{10} (-\delta) - C'_{10} (\delta) C^{*}_{10} (\delta)]\}^{2} [1 - [C_{11} (-2\delta)]^{2}]}{4 [[C_{10} (-\delta)]^{3} - \operatorname{Re} C_{11} (-2\delta) C^{*}_{10} (-\delta) C_{10} (\delta)]^{2}}.$$
(7.3.36)

Reference signal $u_1(t)$ for simplicity of recording in these formulas was normalized so that

$$C_{11}(0) = \lim_{T \to \infty} \frac{1}{T} \int_{0}^{T} |u_{1}(t)|^{2} dt = 1. \qquad (7.3.37)$$

Thus, the already considerably more complex case of noncoincident modulations and finite detuning reduces to the very simple case of zero detuning with coinciding modulation. This gives us the possibility of using for calculation of final quantitative results the same few formulas and graphs of Paragraph 7.3.2, which together with calculation of quantities $(b - a^2)_{\text{RGB}}$ and h_{RGB} permit us actually to answer all questions related to analysis of accuracy of a discriminator with detuned channels in linear approximation.

§ 7.4. Discriminator with Switching of Reference Signals

7.4.1. Block Diagram of the Discriminator

Solution of the problem of identification of channels in a two-channel circuit is connected with great technical difficulties. Therefore, preserving the principle of approximate replacement of the derivative of functional $|Q(t, \tau)|^2$ by a finite difference, it is possible to try, instead of simultaneous formation in the meter of quantities $|Q(t, \tau + \delta)|^2$, $|Q(t, \tau - \delta)|^2$ and their difference, to form the shown

-232-

quantities consecutively, periodically changing tuning of the discriminator relative



Fig. 7.7. Block diagram of a discriminator with switching of reference signals: 1 - preamplifier; 2 - controlled coherent local oscillator; 3 mixer; 4 - narrow-band filter; 5 square-law detector; 6 - phase commutator $s(t) = \operatorname{Ke} u_t |t-t-ij(t)| e^{i(w_0+w_{op})t}$. to the measured value of τ. This is attained by application of the one channel of Fig. 7.5, to the mixer of which there is alternately fed "leading" and "lagging" reference signals, and the sign of output voltage of the detector correspondingly is changed by a phase switcher (Fig. 7.7). During analysis of the discriminator

of Fig. 7.7 we naturally assume that

reference signals are identical, although they may differ from the sounding signals, and that the time constant of the narrow-band filter is small as compared to the period of switching T_{Π} . Nonfulfillment of last condition would lead to averaging of the error signal in the filter and to sharp decrease of the gain factor of the discriminator. With these assumptions output voltage of the discriminator will be

$$z(t, \Delta) = \left| \int_{-\infty}^{t} h_{1}(t-s) u_{1}(s-\tau_{0}+\Delta-\delta) y(s) e^{i\omega_{0}s} ds \right|^{2} f(t) - \left| \int_{-\infty}^{t} h_{1}(t-s) u_{1}(s-\tau_{0}+\Delta+\delta) y(s) e^{i\omega_{0}s} ds \right|^{2} f\left(t-\frac{T_{x}}{2}\right), \quad (7.4.1)$$

where f(t) is a sequence of meanders with period T_{Π^*}

Voltage $z(t, \Delta)$ constitutes a nonstationary random process with periodic nonstationariness. In smoothing circuits of the servo system of a range finder, due to their inertia, there occurs averaging of this nonstationariness. In connection with this the magnitude of fluctuating error is determined by the spectral density corresponding to the time-averaged correlation function of voltage $z(t, \Delta)$. Then, in accordance with (6.2.4)-(6.2.5), the discrimination characteristic, the gain factor and the fluctuation characteristic are determined by the following expressions:

$$\begin{aligned}
\widetilde{z}(t, \Delta) &= \frac{1}{T_{\pi}} \int_{0}^{T_{\pi}} \widetilde{z}(t, \Delta) dt = \frac{1}{2} \widetilde{z}(t, \Delta), \\
K_{\pi} &= \frac{\widetilde{\partial z}(t, \Delta)}{\partial \Delta} \Big|_{\Delta=0} = \frac{1}{2} \frac{\widetilde{\partial z}(t, \Delta)}{\partial \Delta} \Big|_{\Delta=0}, \\
S_{\partial R_{\pi}}(\Delta) &= \frac{1}{K_{\pi}^{2}} \int_{0}^{\infty} d\tau \frac{1}{T_{\pi}} \int_{0}^{T_{\pi}} [\widetilde{z}(t+\tau, \Delta) \widetilde{z}(t, \Delta) - - - \widetilde{z}(t+\tau, \Delta) \widetilde{z}(t, \Delta)] dt,
\end{aligned}$$
(7.4.2)

where $\overline{z(t, \Delta)}$ corresponds to the case of a two-channel discriminator where $u_1(t) = u_2(t)$ and $h_1(t) = h_2(t)$ (7.3.15).

7.4.2. Characteristics of the Discriminator Formulas (7.4.2) permit us to directly use for a discriminator with switching of channels the results obtained earlier for a two-channel discriminator. In particular, the discrimination characteristic and the gain factor differ, other things being equal, by only a factor of 1/2. Systematic error is absent when there is symmetry of the reference signal, where

$$|C_{i0}(-\delta)|^2 = |C_{i0}(\delta)|^2.$$

In other cases Δ_0 is determined by formula (7.3.18) with $C_{10}(\delta) = C_{20}(\delta)$.

Thus, as in preceding paragraphs, with the given assumptions with respect to the frequency of switching we can obtain an expression for the equivalent spectral density.

$$\frac{\frac{1}{2\pi}\int_{-\infty}^{\infty} |H_{1}(i\omega)|^{\alpha} ||C_{11}(0)|^{\alpha} + 2hS_{0}(\omega) |C_{10}(-\delta)|^{\alpha} |d\omega + S_{0}(\omega)|^{\alpha} |Re[C_{10}(-\delta)C_{10}(-\delta) - C_{10}'(\delta)C_{10}'(-\delta)] \times + h^{\alpha}\Delta f_{0}A\left(\frac{1}{\Delta f_{0}T_{u}}\right) ||C_{10}(-\delta)|^{\alpha} + |C_{10}(\delta)|^{\alpha} ||C_{10}(\delta)|^{\alpha} + |C_{10}(\delta)|^{\alpha} ||C_{10}(-\delta)|^{\alpha} + |C_{10}(\delta)|^{\alpha} ||C_{10}(-\delta)|^{\alpha} ||C_{10}(-\delta)|^{\alpha} ||C_{10}(\delta)|^{\alpha} ||C_{10}(\delta)||C_{10}(\delta)|^{\alpha} ||C_{10}(\delta)|^{\alpha} ||C_{10}(\delta)||$$

where

$$A\left(\frac{1}{\Delta f_{\bullet}T_{a}}\right) = \frac{2}{\Delta f_{\bullet}} \sum_{k=-\infty}^{\infty} \frac{B\left(2\pi \frac{2k+1}{T_{a}}\right)}{\pi^{2} \left(2k+1\right)^{2}}; \qquad (7.4.4)$$

 $B(\omega)$ - a quantity, proportional to the spectral density of fluctuations at the detector output,

$$B(\bullet) = \frac{1}{2\pi} \int_{-\infty}^{\infty} |H_1(i\bullet - is)|^2 S_{\bullet}(\bullet - s) |H_1(is)|^2 S_{\bullet}(s) ds. \qquad (7.4.5)$$

An essential peculiarity of the considered discriminator is the presence in S_{OKB} of a component not depending on the signal-to-noise ratio but proportional to $A(\frac{1}{T_{\text{II}}\Delta f_{\text{C}}})$. This means that in a range finder with such a discriminator fluctuating error does not approach zero with unlimited decrease of noises. Physically this is explained by incomplete correlation of those components of output voltage of the discriminator in various half-periods of frequency switching which are

-234-

caused by the presence of fluctuations of the reflected signal, and the related impossibility of their compensation, as occurs in a two-channel system carrying out instantaneous comparison with "lagging" and "leading" reference signals. Such components of S_{3KB} occur in all measuring systems where the discrimination characteristic is created by consecutive measurement of correlation between the received and expected signals for various values of the measured parameter. A characteristic example, as we shall subsequently see, is angle measurement by scanning. The presence of such components is a serious deficiency of such systems.

With slow switching, when $\Delta f_{C} T_{\Pi} >> 1$,

$$A\left(\frac{1}{\Delta f_{\sigma}T_{\pi}}\right) = \frac{B(0)}{2\Delta f_{\sigma}} = \frac{1}{4\pi\Delta f_{\sigma}} \int_{-\infty}^{\infty} |H_1(i\omega)|^4 S_0^2(\omega) d\omega. \qquad (7.4.6)$$

The case of switchings, rapid as compared to signal fluctuations, when $A(\frac{1}{\Delta f_C} T_{\Pi}) \rightarrow 0$, can be presented only if the band of the filter exceeds by many times the width of the spectrum of fluctuations Δf_C . Here, the period of switching



Fig. 7.8. Dependence of the relative magnitude of the residual equivalent spectral density on the product $\Delta f_C = T_{\Pi}$.

may be small as compared to the time constant of the filter, but large as compared to the time of correlation of fluctuations. For exponential correlation and an LRC-filter with band Δf_{\oplus} calculation of function $A(\frac{1}{\Delta f_{C} T_{\Pi}})$ can be conducted for an arbitrary switching frequency $F_{\Pi} = \frac{1}{T_{\Pi}}$. In the case interesting us $x = \frac{\Delta f_{\oplus}}{\Delta f_{C}} >> 1$:

$$A\left(\frac{1}{\Delta f_{o}T_{\pi}}\right) = \frac{1}{2} \left(1 - \frac{1}{\Delta f_{o}T_{\pi}} \operatorname{th} \Delta f_{o}T_{\pi}\right). \tag{7.1.7}$$

With increase of F_{Π} function $A(\frac{1}{\Delta \Gamma c T_{\Pi}})$ decreases rather slowly. Its graph is constructed in Fig. 7.8.

With slow switchings and fulfillment of the condition of the obsence of systematic error the formula for S_{OKB} is simplified and takes the form

-235-

$$S_{0KB} = \frac{\frac{1}{2\pi} \int_{-\infty}^{\infty} |H_1(i\omega)|^4 [1 + k |C_{16}(\partial)|^2 S_0(\omega)]^2 d\omega}{\left\{ \operatorname{Re}[C_{16}'(-\partial) C_{16}^0(-\partial)] \cdot \frac{h}{\pi} \int_{-\infty}^{\infty} |H_1(i\omega)|^4 S_0(\omega) d\omega \right\}^2}, \qquad (7.4.8)$$

where we again normalized $u_1(t)$ so that $C_{11}(0) = 1$. With coinciding modulations function $C_{10}(\delta)$ is replaced by $C(\delta)$. Noncoincidence of modulations of the reference and sounding signals again can be described by introduction of the equivalent signalto-noise ratio, depending on δ :

$$\mathbf{h}_{\mathbf{s} \circ \mathbf{x} \mathbf{s}} = \mathbf{h} \frac{|\mathbf{C}_{\mathbf{s} \circ}(\mathbf{\delta})|^{\mathbf{s}}}{|\mathbf{C}(\mathbf{\delta})|^{\mathbf{s}}}.$$
 (7.4.9)

toth description of the state o

Then, if we introduce for the case of coinciding modulation the special designation

$$S_{\mu\mu\nu}^{(0)}(h) = \frac{\frac{1}{2\pi} \int_{-\infty}^{\infty} |H_{1}(i\omega)|^{4} [1+h|C(\delta)|^{2} S_{0}(\omega)]^{2} d\omega}{\left[\operatorname{Re} C'(\delta) C^{4}(\delta) \frac{h}{\pi} \int_{-\infty}^{\infty} |H_{1}(i\omega)|^{2} S_{0}(\omega) d\omega\right]^{2}}, \qquad (7.4.10)$$

the equivalent spectral density with noncoincident modulations is

$$S_{\text{DKB}} = M(\delta) S_{\text{DKB}}^{(0)}(h_{10 \text{KB}}),$$
 (7.4.11)

where

$$\mathcal{M}(\mathbf{\delta}) = \frac{|C_{10}(-\mathbf{\delta})|^4 |\text{Re} C'(\mathbf{\delta}) C^*(\mathbf{\delta})|^3}{|C(\mathbf{\delta})|^4 |\text{Re} C'_{10}(-\mathbf{\delta}) C^*_{10}(-\mathbf{\delta})|^3}$$
(7.4.12)

is the equivalent change of the square of the slope of the discrimination characteristic,

In formulas (7.4.8), (7.4.9), (7.4.11) and (7.4.12) and henceforth we assume that the condition of the absence of systematic error is satisfied.

We also give an expression for the fluctuation characteristic and spectral density of parametric fluctuations for the case

$$u_{1}(t) = u(t), \quad C_{10}(\delta) = C(\delta):$$

$$S_{0KB}(\Delta) = \frac{\frac{1}{2\pi} \int_{-\infty}^{\infty} |H_{1}(i\omega)|^{4} \{[1+h|C(\Delta+\delta)]^{2} S_{0}(\omega)]^{1} + \delta h^{4} \left[\operatorname{Re} C'(\delta) C^{*}(\delta) \times \frac{1}{8h^{4} \left[\operatorname{Re} C'(\delta) C^{*}(\delta) \times \frac{1}{2\pi} \int_{-\infty}^{\infty} |H_{1}(i\omega)|^{2} S_{0}(\omega) d\omega \right]^{1}}{\chi \frac{1}{2\pi} \int_{-\infty}^{\infty} |H_{1}(i\omega)|^{2} S_{0}(\omega) d\omega}$$

$$S_{\pi \Delta p} = \left[1 + \frac{\operatorname{Re} [C''(\delta) C^{*}(\delta) + C'(\delta) C^{*}'(\delta)] |C(\delta)|^{4}}{2[\operatorname{Re} C'(\delta) C^{*}(\delta)]^{4}} \right] \times \frac{1}{2\pi} \int_{-\infty}^{\infty} |H_{1}(i\omega)|^{4} S_{0}^{2}(\omega) d\omega} \times \frac{1}{\left[\frac{1}{2\pi} \int_{-\infty}^{\infty} |H_{1}(i\omega)|^{2} S_{0}(\omega) d\omega} \right]^{4}} + \frac{\operatorname{Re} [C''(\delta) C^{*}(\delta)]^{4}}{2h[\operatorname{Re} C'(\delta) C^{*}(\delta)]^{4}} \frac{1}{2\pi} \int_{-\infty}^{\infty} |H_{1}(i\omega)|^{4} S_{0}(\omega) d\omega}$$

$$(7.4.14)$$

Formulas (7.4.8) and (7.4.14) show that both nonparametric and parametric fluctuations essentially depend on the form of signal modulation and the magnitude of detuning δ . For any finite values of h and any $C_{10}(\delta)$ quantity $S_{\rm MB}$ changes monotonically depending upon δ , turning for a certain δ into a minimum. The magnitude of this optimum value of detuning essentially depends on the form of the signal and the magnitude of the signal-to-noise ratio α . For small values of h the minimum of $S_{\rm HO}$ is attained at value of δ , which turns the slope of the discrimination characteristic into a maximum; for larger h the optimum value of δ increases. We shall investigate the character of the dependence of $S_{\rm HOB}$ on δ in more detail when we examine concrete forms of modulation; however, in general it is possible to indicate that in circuits with switching there are all grounds on which to select detuning equal to that value of it which maximizes the gain factor, or somewhat larger.

In distinction from $(7.3.1^4)$, in this case both components of $S_{\Pi AP}$ depend on the form of modulation and detuning 5. This dependence is such that, for small 5, $S_{\Pi AP}$ may be negative, i.e., the fluctuation characteristic may have its maximum in the vicinity of $\Delta = 0$.

-237-

Such a character of the dependence $S_{\rm SKP}(\Delta)$ means that in the considered case, if $S_{\rm TAp} < 0$, there is no sense in talking about parametric fluctuations. In examining linearized conditions we should consider fluctuations not to depend on mismatch. If, however, we try to allow for the dependence of $S_{\rm SKP}(\Delta)$ on mismatch, then, even being limited to small Δ , we have to solve a nonlinear problem, considering also the dependence of the discrimination characteristic on Δ . For large δ the dependence of $S_{\rm SKP}(\Delta)$ on Δ has a normal character, and analysis of the range finder as a whole can be conducted by the usual means. As follows from (7.4.10), for δ , turning the gain factor $K_{\rm II}$ into a maximum, the component of $S_{\rm TAP}$ of the order 1/h disappears, just at the addition to the basic term, and $S_{\rm TAP}$ turns out to coincide with (7.3.14), i.e., $S_{\rm TAP} = \frac{\alpha}{\Delta f_{\rm C}}$. For a certain smaller value of δ depending on h, the magnitude of $S_{\rm TAP}$ turns into zero. The considered character of change of $S_{\rm SKB}$ and $S_{\rm TAP}$ from δ is rather unique and essentially distinguishes a circuit with switching from the optimum and two-channel circuits.

7.4.3. Optimum Filter for a Discriminator with Switching The dependence of $S_{ax\,B}$ on the form of frequency response of the filter has an interesting character. Let us consider this dependence for the case of low frequency switching. Composing the variation of S_{aKB} from (7.4.8) with respect to $|H_1(i\omega)|^2$ and equating it to zero, we obtain the following equation, determining the optimum frequency response:

$$|H_{1}(i\bullet)|^{2}[1+h|C_{1\bullet}(\hat{o})|^{2}S_{\bullet}(\omega)] \int_{-\infty}^{\infty} |H_{1}(ix)|^{2}S_{\bullet}(x)dx - hS_{\bullet}(\bullet) \int_{-\infty}^{\infty} |H_{1}(ix)|^{4}[1+h|C_{1\bullet}(\hat{o})|^{2}S_{\bullet}(x)]^{2}dx = 0, \qquad (7.4.15)$$

which, obviously, is satisfied when

$$|H_{1}(i\omega)|^{2} = |H_{ORT}(i\omega)|^{2} = \frac{h|C_{10}(\delta)|^{2} S_{0}(\omega)}{[1+h|C_{10}(\delta)|^{2} S_{0}(\omega)]^{2}}.$$
(7.4.16)

This result shows that an optimum filter for a discriminator with switching, ensuring a minimum equivalent spectral density, may substantially differ in its characteristics from the usual optimum filter (7.2.5). With fulfillment of (7.4.16)

$$S_{0RB} = \frac{1}{4A^{2} [\operatorname{Re} C'_{10}(\delta) C^{2}_{10}(\delta)]^{2} \frac{1}{2\pi} \int_{-\infty}^{\infty} \frac{S_{0}^{2}(\omega) d\omega}{[1 + \hbar |C_{10}(\delta)|^{2} S_{0}(\omega)]^{2}}$$
(7.4.17)

-238-

For any real spectrum of fluctuations $[S_{O}(\omega)$ is different from a rectangular function $]S_{3KB} \rightarrow 0$ as $h \rightarrow \infty$. The order of decrease of S_{KKB} depends on the form of the spectral density and is always slower than 1/h.

The difference between optimum filter (7.4.16) and a usual one from the theoretical point of view is explained by the fact that in this case during synthesis of the optimum discriminator we impose an additional technical condition. This condition is that at every moment of time we can form the likelihood functional [actually functional $Q(t, \tau)$] for only one value of τ . In principle in an optimum system values of this functional should be formed for all τ ; however, as we proved in the example of a discriminator with staggered channels, with observance of certain conditions it is practically sufficient to form this functional simultaneously only for two differing values of parameter τ . The additional technical limitation, naturally, leads to change of the form of the optimum system and, as any limitation, leads to loss in characteristics of the synthesized system. This loss may be slight, but it always exists in principle, being the cost of the attained technical simplification.

Let us investigate in more detail the frequency response of an optimum filter. For small $h|C_{10}(\delta)|^2$ this characteristic coincides with the form of the spectrum of



 $\{ \}$





Fig. 7.10. Fundamental circuit of an optimum filter.

fluctuations and with the usual optimum frequency response for small h. With growth of $h|C_{10}(5)|^2$ the difference between (7.4.16) and (7.2.5) becomes essential. For frequency response curve (7.4.16) there is a dip at $\omega = 0$, the depth of which decreases as $1/h|C_{10}(5)|^2$, and two maxima, the separation of which increases with increase of $h|C_{10}(5)|^2$. In the particular case of an exponential correlation of fluctuations

$$|H_{OBT}(i\omega)|^{2} = \frac{h|C_{1v}(\delta)|^{2} \left(1 + \frac{\omega^{2}}{4\Delta f_{c}^{2}}\right)}{\left[1 + h|C_{1v}(\delta)|^{2} + \frac{\omega^{2}}{4\Delta f_{c}^{2}}\right]^{2}} \cdot (7.4.18)$$

The graph of this function is constructed in Fig. 7.9. Frequency response (7.4.18), taking into account transfer to intermediate frequency -239 $\omega_{\!_{\rm TD}},$ can be formed by the circuit of Fig. 7.10 when

$$\frac{1}{\sqrt{L_{1}C_{1}}} = \frac{1}{\sqrt{L_{1}C_{2}}} = \frac{1}{\sqrt{L_{2}C_{2}}} = \alpha_{up}, \quad \frac{1}{R_{1}C_{1}} = 2\Delta f_{o},$$

$$\frac{1}{R_{1}(C_{1} + C_{2})} = \frac{1}{R_{2}C_{2}} = 2\Delta f_{o}\sqrt{1 + \frac{1}{R_{1}(C_{1} + C_{2})}} = \frac{1}{R_{1}C_{2}} = \frac{$$

In a system with optimum filter (7.4.18) S_{9K8} is expressed by the formula

 $S_{\text{BKB}} = \frac{1}{2\Delta f_0} \frac{\left[1 + h\right] C_{10}(\delta) \left[^3\right]^{3/2}}{h^2 \left[\text{Re} C'_{10}(\delta) C^*_{10}(\delta)\right]^3}.$ (7.4.19)

This expression, as also (7.3.8), for a certain 5 turns into a minimum, but in distinction from the general case it approaches zero as $h \rightarrow \infty$ as $1/\sqrt{h}$. Thus, even with an optimum illter the circuit with switching for a large signal-to-noise ratio has a loss as compared to a two-channel circuit, growing as \sqrt{h} . Detailed comparison of circuits with detuning and with switching requires specification of the form of function $C_{10}(\delta)$, and therefore, we postpone it to subsequent paragraphs. Here we can only note that in general both for an optimum filter (7.4.16) and for an arbitrary one loss in accuracy during use of a circuit with switching with respect to an optimum and a two-channel one is greater, the greater h. For small h it may be more or less immaterial, but for large h, when fluctuating error of the range finder is caused basically by the randomness of the signal, it attains considerable magnitudes. For a normal filter loss grows proportionally to h; for an optimum filter it is somewhat slower, for instance, for filter (7.2.18), as \sqrt{h} . This peculiarity is an unpleasant property of a circuit with switching.

Let us estimate now what gain application of an optimum filter gives in a circuit with switching. Let us consider for this again the spectral density of a signal of form (7.3.11) and frequency response of form (7.3.12). Substituting these in the formula for (7.4.8), we obtain

$$S_{\text{DKB}} = \frac{(1+x)^2 + h |C_{10}(\delta)|^2 (1+2x) + h^2 |C_{10}(\delta)|^4 \left(2x + \frac{1}{1+x}\right)}{2\Delta f_0 \cdot 4h^2 x \left(\operatorname{Re} C'_{10}(\delta) C^*_{10}(\delta)\right)^2}, \qquad (7.4.20)$$

where x, as before, is $\Delta f_{\tilde{m}} / \Delta f_{c}$.

Then ratio γ of equivalent spectral densities during use of LRC-filter (7.3.12) and optimum filter (7.4.18) is

$$\gamma = \frac{(1+x)^{2} + h |C_{10}(\delta)|^{2}(1+2x) + h^{2} |C_{10}(\delta)|^{4} \left(2x + \frac{1}{1+x}\right)}{4x \left[1 + h |C_{11}(\delta)|^{2}\right]^{3/2}}.$$
 (7.4.21)

Dependence $\gamma(x)$ for various $h|C_{16}(b)|^2$ is shown in Fig. 7.11. For small $h|C_{10}(b)|^2$ this dependence has a clearly expressed minimum at $x \approx 1$, where the magnitude of the minimum is close to one, which is naturally explained by practical coincidence when $h|C_{10}(b)|^2 \ll 1$ of characteristics of an optimum and an LRC-filter when $x \approx 1$. With increase of $h|C_{10}(b)|^2$ the minimum is shifted in the direction of large x, and curve $\gamma(x)$ in the vicinity of the minimum is all the more shallow, so that there appears a zone within which change of the band of the LRC-filter practically does not affect S_{3KB} . The loss as compared to an optimum circuit in this zone comprises approximately a factor of $|C_{10}(b)|\sqrt{h}/2$. Width of the zone grows



rapidly with increase of $h|C_{10}(\delta)|^2$. If for $h|C_{10}(\delta)|^2 =$ = 10 error of measurement almost does not change with triple change of band, then for $h|C_{10}(\delta)|^2 =$ 100 error remains practically constant with change of the band by a factor of 200.

The gain during use of an optimum filter is explained by the fact that in a circuit with switching useful components of

Fig. 7.11. Influence of nonoptimality of the filter on equivalent spectral density.

the spectrum are transferred to the frequency of switching, and therefore, suppression of signal fluctuations concentrated near zero frequency and playing, at low noises, the basic role in formation of fluctuating error, which is attained by use of a frequency response curve with a dip near zero, turns out to be useful. The magnitude and width of the dip by formula (7.4.16) is automatically selected in such a way as to ensure minimum equivalent spectral density.

§ 7.5. Discriminator with Differentiation of the Reference Signal

7.5.1. Block Diagram of the Discriminator

In § 7.2 we already saw that an optimum discriminator can be built by the circuit of Fig. 7.2 and consist of two channels, to one of which there proceeds the reference signal, and to the other, its derivative with respect to τ . The advantage of the circuit with differentiation of the reference signal as compared to a circuit with staggered channels is that operations performed by it can be made identical to optimum operations without decrease of the slope of the discrimination characteristic.

-241-

. م

One of the basic difficulties connected with realization of such a circuit consists in forming reference voltage for the channel in which the received signal should be mixed with the derivative of the sounding signal. However, in certain cases this difficulty is surmountable. For instance, with frequency modulation reference signals of the local oscillator are equal, respectively, to

 $\operatorname{Re} u (t-\tau) e^{i(\omega_0+\omega_{\pi p})t} = \cos \left[(\omega_0+\omega_{\pi p}) t + \psi (t-\tau) \right],$ $\frac{\partial}{\partial t} \operatorname{Re} u (t - \tau) e^{i(\mathbf{e}_0 + \mathbf{e}_{ap})t} =$ $= \Delta \omega (t - \tau) \sin \left[(\omega_0 + \omega_{\pi \mu}) t + \psi (t - \tau) \right].$

It is possible to form these signals with the help of the circuit of Fig. 7.12. The signal of the master oscillator, delayed τ , enters the frequency-modulated local



Fig. 7.12. Block diagram of forming of reference signals with frequency modulation: 1 - controlled master oscillator; 2 - frequency-modulated local oscillator; $3 - \pi/2$ phase shifter; 4 - amplitude modulator.

oscillator at whose output there is obtained the first reference signal. The second reference signal is obtained from the first as a result of a shift of $\pi/2$ and amplitude modulation by the law $\Delta\omega(t - \tau)$. In the case of sinusoidal frequency modulation ($\Delta\omega(t) = \omega_m \cos \Omega t$) the second reference signal is obtained by simple shift of the first in frequency by $\pm\Omega$ and

by phase shift.

During technical realization of the block diagram of Fig. 7.2 there are possibly all sorts of deviations from optimum operations, the influence of which we shall now estimate. Denoting, as in § 7.3, by $u_1(t)$ and $u_2(t)$ the laws of modulation of the reference signals in the first and second channels and considering the filters identical, we obtain the following expression for discriminator output, generalizing (7.2.7):

$$z(t, \Delta) = \operatorname{Re} \int_{-\infty}^{t} \int_{-\infty}^{t} h_{i}(t-s_{i})h_{i}(t-s_{s})u_{i}(s_{i}-s_{o}-\Delta) \times \\ \times u^{*}_{s}(s_{s}-s_{o}-\Delta)y(s_{i})y(s_{s})e^{i\omega_{s}(s_{i}-s_{o})}ds_{i}ds_{s}.$$

$$(7.5.1)$$

7.5.2. Characteristics of the Discriminator

Averaging expression (7.5.1) as usual, we obtain the following formula for the discrimination characteristic:

-242-

$$\overline{z(t, \Delta)} = \frac{N_{\bullet}}{2\pi} \int_{-\infty}^{\infty} |H_{i}(i\omega)|^{s} \operatorname{Re} [C_{12}(0) + hS_{\bullet}(\omega)C_{10}(-\Delta)C_{\infty}^{*}(-\Delta)] d\omega, \qquad (7.5.2)$$

where $C_{ik}(\Delta)$ are determined by formula (7.3.16), from which it follows that

$$K_{\mu} = -\frac{N_{*}A}{2\pi} \int_{-\infty}^{\infty} |H_{1}(i \bullet)|^{2} S_{0}(\bullet) d \bullet \times \times \\ \times \operatorname{Re} [C'_{10}(0) C^{*}_{10}(0) + C^{*}_{10}(0) C'_{10}(0)], \\ \Delta_{0} = \left\{ -\operatorname{Re} C_{10}(0) C^{*}_{10}(0) + \operatorname{Re} C_{10}(0) \frac{\frac{1}{2\pi} \int_{-\infty}^{\infty} |H_{1}(i \bullet)|^{2} d \bullet}{\frac{1}{2\pi} \int_{-\infty}^{\infty} |H_{1}(i \bullet)|^{2} S_{0}(\bullet) d \bullet} \right\}; \\ \operatorname{Re} [C'_{10}(0) C^{*}_{10}(0) + C^{*}_{10}(0) C'_{10}(0)].$$

$$(7.5.3)$$

From the last expression it follows that systematic error is absent for any h, if

$$\operatorname{Re} C_{11}(0) = \operatorname{Re} C_{10}(0) C^{*}_{10}(0) = 0.$$
(7.5.4)

These conditions, obviously, lead to the requirement that when there is a symmetric sounding signal one reference signal be symmetric, and the other, antisymmetric.

Determining the correlation function of voltage $z(t, \Delta)$ and subjecting it to the required transformations, it is possible to obtain an expression for equivalent spectral density

$$S_{\bullet Ks} = \left\{ \frac{1}{2\pi} \int_{-\infty}^{\infty} |H_{1}(i\sigma)|^{4} [C_{11}(0) C_{22}(0) + \operatorname{Re} C_{12}^{2}(0) + \\ + hS_{\bullet}(\sigma) (C_{11}(0) |C_{\bullet 2}(0)|^{2} + \\ + 2\operatorname{Re} C_{1s}(0) C_{\bullet 2}(0) C_{\bullet 1}^{*}(0)) + h^{2}S_{0}^{2}(\sigma) (|C_{01}(0) C_{02}(0)|^{2} + \\ + \operatorname{Re} C_{01}^{2}(0) C_{02}^{2}(0)) d\omega \right\} : 2h^{2} \left\{ \operatorname{Re} [C_{1s}'(0) C_{\bullet 2}^{*}(0) + \\ + C_{1s}^{*}(0) C_{\bullet 2}'(0)] \frac{1}{2\pi} \int_{-\infty}^{\infty} |H_{1}(i\sigma)|^{2} S_{0}(\omega) d\omega \right\}^{2}.$$

$$(7.5.5)$$

From this expression it again follows that the presence of nonidealnesses leads to the appearance of a component of S_{AGB} not depending on h. As also in the circuit of Fig. 7.5, this component disappears together with disappearance of systematic error, i.e., with fulfillment of conditions (7.5.4). In general, formula (7.5.5) with an accuracy of coefficients coincides with the corresponding formula (7.3.19)

-243-

for a discriminator with staggered channels. If, in particular, differentiation of the reference signal is sufficiently accurate, which in certain cases is fully practicable, i.e., $u_2(t) = u'_1(t)$, and the main reference signal $u_1(t)$ differs from the sounding signal u(t) due to the difference of modulation characteristics of the local oscillator and the transmitter, with fulfillment of (7.4.4) the spectral density of S_{3KB} is determined by a formula which coincides with (7.3.33) for a discriminator with staggered channels with small detuning. Obviously, expression (7.3.9), satisfied for $u_1(t) = u(t)$, also remains valid.

All these circumstances indicate that the dependence of equivalent spectral density for the considered discriminator on the signal-to-noise ratio h and the band of the filter actually has already been investigated earlier in § 7.3, and it remains for us only to repeat the conclusions presented there.

It is essential that in the considered circuit nonidentity of gain factors in channels does not lead to the appearance of stationary error, and that density of phase responses plays the basic role. If $u_1(t)$ and $u_2(t)$ coincide with u(t) and u'(t) with an accuracy of constant phase shift, for instance:

$$u_{1}(t) = u(t) e^{t\varphi}, \quad u_{1}(t) = u'(t),$$

where ϕ - phase delay in one channel relative to the other, then, as it is easy to prove,

$$C_{10}(0) C^{*}_{30}(0) = C_{12}(0) = iae^{i\varphi},$$

$$\Delta_{0} = \frac{a \sin \varphi}{b - a^{2}} \cdot \frac{\frac{1}{2\pi} \int_{-\infty}^{\infty} |H_{1}(i\omega)|^{2} [1 + hS_{0}(\omega)] d\omega}{h \frac{1}{2\pi} \int_{-\infty}^{\infty} |H_{1}(i\omega)|^{2} S_{0}(\omega) d\omega}.$$
(7.5.0)

The condition of the absence of systematic error consists, thus, of the equality to zero of quantity a, i.e., the requirement of symmetry of the power spectrum of modulation. Phase shift leads also to increase of equivalent spectral density. When a = 0 quantity S_{330} increases as compared to (7.3.9) by a factor of $1/\cos^2 \varphi$.

In conclusion let us give the expression for the fluctuation characteristic of a discriminator with differentiation of the reference signal. Limiting curselver to the case $u_1(t) = u(t)$, $u_2(t) = u'(t)$, we obtain, completely analogously to (7.2.20)

$$S_{\Theta KB}(\Delta) = \operatorname{Re}\left\{\frac{1}{2\pi}\int_{0}^{\infty} |H_{1}(i\omega)|^{4} [h^{2}S_{0}^{2}(\bullet)(|C(\Delta)|^{2}|C^{4}(\Delta)|^{2} + C^{4}(\Delta)C^{*'}(\Delta)) + hS_{0}(\omega)(b|C(\Delta)|^{2} + |C'(\Delta)|^{2} - iaC(\Delta)C^{*'}(\Delta)) + (C'(\Delta)|^{2} + iaC(\Delta)C^{*'}(\Delta)) + b - a^{2}d\omega\right\}: 2h^{2}(b - a^{2})^{2}\left[\frac{1}{2\pi}\int_{-\infty}^{\infty} |H_{1}(i\omega)|^{2}S_{0}(\omega)d\omega\right]^{2}.$$

$$(7.5.7)$$

This expression, obviously, differs from (7.2.20) only by replacement of $hS_0(\omega)|[1 + hS_0(\omega)]$ by an arbitrary frequency response $|H_1(i\omega)|^2$. Calculating the second derivative of $S_{\text{DKB}}(\Delta)$ from (7.5.7) when $\Delta = 0$, it is simple to prove that spectral density of parametric fluctuations of S_{HAP} is determined by the very simple formula (7.3.21), occurring for a circuit with staggered channels as $h \to \infty$.

With this we complete our general consideration of time discriminators. In the following paragraphs we shall turn to concrete forms of modulation of the sounding signal. Inasmuch as the dependence of characteristics of discriminators on the signal-to-noise ratio and characteristics of filters was considered in general form, subsequently we shall be interested basically in the dependence of $S_{\rm DKB}$ and $K_{\rm H}$ on detuning of channels with respect to delay and on nonidentity of the reference and sounding signals. Characteristics of filters we consider optimum. Corrections for noncoincidence of bands and frequency responses of filters with the optimum can be introduced by the above results. Furthermore, we subsequently consider that spectral density of the fluctuating signal is determined by formula (7.3.11).

§ 7.6. Pulse Radiation

Pulse radiation, thanks to a series of well-known technical advantages, is widely used in contemporary radar. With pulse radiation we can use, obviously, all the schemes for construction of a discriminator considered earlier. Furthermore, in this case there is one more possibility — multiplication of the received signal by the expected can be replaced by transmission of it through a filter physically realizable due to finite pulse duration, with frequency response conjugate with the spectrum of the signal, and with subsequent gating by a very short delta-shaped pulse. Equivalence of these methods of processing was proven in [4] and at present is well-known. An advantage of replacement of multiplication by filtration with subsequent gating is that one filter can be used both in the channel of detection and in tracking channels for measurement of coordinates of one or several targets. In more detail questions of realizing an optimum coherent receiver on the basis of the principle of filtration were discussed previously in Chapter IV in examining

-245-

the problem of detection.

The circuit of a coherent time discriminator with two detuned channels, using filtration, has the form shown in Fig. 7.13. After the mixer, the signal enters a



Fig. 7. 13. Block diagram of an optimum discriminator with a matched intermediate-frequency amplifier (UFCh): 1 - preamplifier; $2 - \text{coherent local oscil$ $lator; } 3 - \text{mixer}$; 4 - controlled generator of gatepulses; 5 - filter (UPCh) with a frequency response matched with the spectrum of modulation; 6 - gatedamplifier; 7 - narrow-band filter; 8 - square-lawdetector; 9 - subtractor. filter with frequency response $\frac{1}{2}$ {U*(i ω - i $\omega_{\rm Hp}$) + + U(-i ω - i $\omega_{\rm hp}$)} and is gated by two gate pulses, detuned $\pm\delta$ from the measured value of delay. After gating, the signals are filtered by narrow-band filters, detectors and subtracted. Analogously, we can convert a circuit with differentiation of the reference signal. date pulses

and a second and the state and the second in the second second second second second second second second second

in this case coincide, and the filter corresponding to the channel with the derivative has frequency response

$$\frac{1}{2} \{ i (\boldsymbol{\omega} - \boldsymbol{\omega}_{\pi p}) U^* (i \boldsymbol{\omega} - i \boldsymbol{\omega}_{\pi p}) - i (\boldsymbol{\omega} + \boldsymbol{\omega}_{\pi p}) U (-i \boldsymbol{\omega} - i \boldsymbol{\omega}_{\pi p}) \}.$$

Likewise we realize a circuit with switching. In the case of pulse signals without additional intrapulse modulation it is sufficient that the band of the filter coincide with the band of the signal.

From equivalence of circuits with filtration and with multiplication by the expected signal (such a realization of an optimum circuit for processing is often called a correlation receiver), it follows that analysis of their accuracy can be performed by the same formulas. In this paragraph we investigate accuracy of rangfinding during pulse radiation without additional modulation for different forms of the pulse and gate pulse (or, which is the same, for different signal spectra and frequency response of the filter in circuits using the principle of optimum filtration). In the absence of intrapulse modulation and with arbitrary forms of pulses and gate pulses all functions $C_{ik}(\delta)$ are real, and general formulas of the preceding paragraphs are somewhat simplified.

Most practicable for pulse modulation are circuits with detuned channels and with switching; therefore, we shall basically consider discriminators of these forms, but at the end of the paragraph we shall briefly discuss the application of a discriminator with differentiation of the reference signal.

-246-

7.6.1. Square Pulse

Let us consider first modulation of the signal by square, periodica ly repeated pulses of duration $\tau_{\rm N}$. It is obvious that the autocorrelation function for such a signal is

$$C(\delta) = 1 - \frac{|\delta|}{\epsilon_{\pi}}, \quad |\delta| < \epsilon_{\pi}, \qquad (7.6.1)$$

$$C(\delta) = 0, \qquad |\delta| > \epsilon_{\pi}.$$

A characteristic feature of such a signal is infinite root mean square spectrum width \sqrt{r} , which formally leads to infinite accuracy of range finding for any signalto-noise ratio. Actually this does not take place, since the pulse edge always has finite duration τ_{A} . Thanks to this the vertex of function O(6) does not have a break and the magnitude of b is finite. In examining an optimum circuit and its characteristics allowance for these circumstances is necessary; however, during the analysis of a circuit with detuned channels and a circuit with switching we with full right can use function C(5) in form (7.6.1), calculated without taking into account the edge, if we always consider that detuning δ exceeds the edge duration, and equality of δ to zero is understood in the sense that $\delta << \tau_{\mu}$, but $\delta > \tau_{\tilde{m}}$. If we consider the hypothetical case of a square pulse of strictly zero edge duration, then, in general, the approach utilized by us becomes inapplicable. In this case the logarithm of the likelihood functional does not have a finite second derivative, and operations, calculated on the basis of replacement of this functional by first terms of a Taylor expansion, do not have meaning. Such a nonanalytical case requires special consideration, which leads anew to finite accuracy of measurement. Such a consideration, however, does not have any essential practical interest.

Substituting (7.6.1) in the corresponding general formulas for K_{A} and S_{AKB} , for the gain factor and equivalent spectral density we obtain the following expressions:

$$K_{\mathrm{R}} = D \frac{1}{\tau_{\mathrm{H}}} \left(1 - \frac{\vartheta}{\tau_{\mathrm{H}}} \right), \qquad (7.6.2)$$

where $D = 2P_c \sqrt{1 + h}/[1 + \sqrt{1 + h}]$ - designation, which we introduce to shorten notation;

for a two-channel circuit

$$S_{0KB} = \frac{\partial}{\tau_{\pi}} \frac{(1 + \sqrt[4]{1 + \hbar})^2 + \hbar (1 + 2\sqrt[4]{1 + \hbar}) \left(1 - \frac{\partial}{\tau_{\pi}}\right)}{4\hbar^2 \sqrt{1 + \hbar} \Delta f_c \left(1 - \frac{\hbar}{\tau_{\pi}}\right)} \tau_{\pi}^2; \qquad (7.6.3)$$

for a circuit with switching

(7.0.5)



Let us remember that characteristics of filters are considered optimum. In order to pass in expressions pertaining to the two-channel circuit to the case of an LRC-filter with an arbitrary band $\Delta f_{\phi} = x \Delta f_c$ here and subsequently one should replace in the formulas $\sqrt{1 + h}$ by x.

The dependence of $K_{\rm H}$ on 6 is very simple; the gain factor decreases proportionally to 5. Dependences (7.6.3) and (7.6.4) are constructed in Figures 7.14 and 7.15, respectively. For 5 close to $\tau_{\rm H}$, spectral density in both cases increases without limit; as 5 \rightarrow 0 for a two- hannel circuit it approaches zero (actually a value, determined by the duration of one edge); and for a circuit with switching it approaches $(1 + h)^{3/2}\tau_{\rm H}^2/2\Delta f_{\rm C}$ h². For the last circuit S_{3KB}(5) has a minimum, the existance of which one should use when selecting parameters of the discriminator. This question already was discussed in § 7.4. For the considered case the minimum is reached when

 $\delta_{\bullet} = \tau_{\pm} \left(1 - \sqrt{\frac{2}{h}} \right).$







Fig. 7.15. Dependence of equivalent spectral density on detuning for a discriminator with switching of reference signals.

When h < 2 quantity $\delta_0 < 0$. This means that when h < 2 one should select detuning close to zero. When h > 2 the magnitude of optimum detuning is greater, the larger h. All this confirms the conclusions of Paragraph 7.4.2 about the character of the dependence of optimum 6 on the signal-to-noise ratio. In normal conditions (when h = 10 to 30) detuning should comprise (0.45 to 0.8) τ_v . If $5 = b_0$,

-248-

$$S_{\text{DRD}} = \frac{1.3r_{\text{R}}^2}{4f_0h}$$
 (7.6.6)

Now let the duration of gate pulses τ_{c} differ from pulse duration. We shall consider for definitiveness $\tau_{c} > \tau_{\mu}$ (the opposite case if not of interest for practice). Designating $\frac{\delta}{\tau_{\mu}} = z$, and $\frac{\tau_{c}}{\tau_{\mu}} = \mu$, in accordance with (7.2.16) we obtain

$$C_{11}(\delta) = 1 - \frac{|\delta|}{\tau_{0}} = 1 - \frac{|z|}{\mu}, \quad C_{10}(\delta) = \sqrt{\frac{1}{\mu}} (1 - |z|) = \sqrt{\frac{1}{\mu}} C_{00}(\delta), \quad (7.6.7)$$

Substituting these expressions in formulas (7.3.17) and (7.3.21) for the two-channel circuit, we prove that the gain factor decreases as compared to (7.6.2) by a factor of μ , and

$$S_{\text{9KB}} = \mu z \frac{(1 + \sqrt{1 + h})^2 \left(1 - \frac{z}{\mu}\right) + h (1 + 2\sqrt{1 + h}) (1 - z)^2}{4h^2 \Delta f_0 \sqrt{1 + h} (1 - z)^2} \tau_a^2.$$
(7.6.8)

The dependence of ratio $S_{3KB}(\mu, z)/S_{3KB}(1, z)$ on μ for various relative detunings z is constructed in Fig. 7.16. This ratio varies little with change of h in a

Syn8 (H.Z)/SJK8(1.Z)



Fig. 7.16. Influence of duration of the gate pulse on S_{3KB} for a square pulse.

sufficiently wide range, especially if detuning is small. From the graph it is possible to establish the expansion of the gate pulse and under what conditions it can be considered permissible. With sufficiently good approximation the loss in magnitude of spectral density S_{3KB} can be considered equal to μ in all operating conditions.

Results for the circuit with switching are still simpler. Since coefficient $M(\delta)$ from (7.4.12) for $C_{10}(\delta)$ from (7.6.7) is equal to one for all δ , and h_{13KB} from (7.4.9) is h/μ , everything reduces to equivalent decrease of the signal-to-noise

ratio by a factor of μ . Here, the magnitude of optimum detuning decreases to

$$\delta_{i} = \tau_{n} \left(1 - \sqrt{\frac{2\mu}{h}} \right),$$

-249-

, and equivalent spectral density for any μ , as before, is characterized by formula (7.6.4) and the graph of (7.15) with replacement of $h_{1.0HB} = h/\mu$ for h.

7.6.2. Gaussian Pulse

Very often the shape of the pulse essentially differs from recompular. There a convenient approximation, corresponding well to reality in many cases, is a Gaussian curve of form

$$\boldsymbol{u}(t) = \sqrt{\frac{T_{\tau}}{\tau_{\pi}}} e^{-\frac{\pi t^{2}}{2\tau_{\pi}^{2}}}, \qquad (7, 0, 9)$$

where τ_{μ} - duration of the equivalent square pulse of the same amplitude $u_0 = \sqrt{\frac{m_0}{\tau_{\pi}}}$, introduced generally by the following relationship:

$$\mathbf{v}_{\mathbf{g}} = \int_{-\infty}^{\infty} \frac{u^{\mathbf{t}}(t)}{u_{\text{MARC}}^{2}} dt = \int_{-\infty}^{\infty} \frac{u^{\mathbf{t}}(t)}{u^{\mathbf{t}}(0)} dt. \qquad (7.6.10)$$

Quantity $\tau_{\rm N}$ is connected with the duration of a pulse at level 0.5 by the obvious relationship

$$\tau_{0.6} = \sqrt{\frac{5.6}{\pi}} \tau_{g} = 1,33\tau_{g}.$$
 (7.6.11)

We shall consider that gate pulses (reference signals) $u_{\pm}(t)$ also have Gaussian form and duration $\tau_{c} = \mu \tau_{g}$, i.e.,

$$u_{1}(t) = \sqrt{\frac{T_{r}}{\tau_{0}}} e^{-\frac{\pi t^{2}}{2\tau_{0}^{2}}} = \sqrt{\frac{T_{r}}{\mu\tau_{0}}} e^{-\frac{\pi}{2\mu^{2}}\frac{t^{4}}{\tau_{0}^{2}}}.$$
 (7.6.12)

"non-

$$C_{so}(\delta) = C(\delta) = e^{-\frac{\kappa \pi^{4}}{4}}, \quad C_{11}(\delta) = e^{-\frac{\kappa \pi^{4}}{4\mu^{3}}};$$

$$C_{10}(\delta) = \sqrt{\frac{2\mu}{1+\mu^{3}}} e^{-\frac{\kappa \pi^{4}}{2(1+\mu^{3})}}, \quad (7.0.13)$$

where, as before, $z = \delta/\tau_{H^*}$.

The mean square spectrum width for a Gaussian pulse, obviously, is equal to

$$b = -C''(0) = \frac{\pi}{2\tau_{\rm m}^2}.$$
 (7.6.14)

The slope of the discrimination characteristic for a two-channel circuit according to (7.3.28) is determined by the following expression:

-250-

$$K_{\rm A} = D \frac{1}{\tau_{\rm B}} \sqrt{\frac{2\mu}{1+\mu^2}} \frac{\pi z}{1+\mu^2} e^{-\frac{\pi z^2}{1+\mu^2}}.$$
 (7.6.15)

'Its maximum is attained, obviously, when

$$z_{\text{MARC}} = \sqrt{\frac{1+\mu^2}{2\pi}} = \sqrt{\frac{1+\mu^2}{4b\tau_{\text{H}}^2}}, \qquad (7.6.16)$$

which increases with growth of μ . Substituting (7.6.13) in (7.3.35) and (7.3.36), we obtain expressions, determining h_{BKB} and b_{BKB} for a circuit with detuned channels

$$\frac{h_{2KB}}{h} = \frac{4\mu}{1+\mu^2} \frac{e^{-\frac{\mu}{1+\mu^2}}}{e^{-\frac{\pi}{\mu_1}}} \xrightarrow{\bullet} \frac{2\mu}{1+\mu^2}, \qquad (7.6.17)$$

$$\frac{b_{0\pi\pi}}{b} = \frac{2\pi z^2}{(1+\mu^2)^2} \frac{1+e^{-\frac{\pi z^2}{\mu^2}}}{1-e^{-\frac{\pi z^2}{\mu^2}}} + \frac{4\mu^2}{e^{-\frac{\pi z^2}{\mu^2}}}$$
(7.6.18)

The dependence of ratios $K_{\mu}(\delta)/K_{\mu MSKC}$, $h_{SKB}(\delta)/h$, $b_{SKB}(\delta)/b$ and $S_{SKB}(\delta)/S_{OHT}$ (for small and large h) on detuning is shown in Fig. 7.17 for the case of coinciding





signal and gate pulses ($\mu = 1$). This figure graphically shows now weakly the finite quantity of detuning affects accuracy of range finding in this case. Actually, for $z = z_{MPHC} b_{PKB}/b = 1.09$, $h_{HCD}/h =$ = 0.88, $bh/b_{PKB} h_{HCB} = 1.04$, $bh^2/b_{PKB}h_{PKB}^2 = 1.18$, i.e., in the most unfavorable case the equivalent spectral density is increased by not more than 20%, and at working signal-to-noise ratios this increase does not exceed 5%.

For μ , differing from one (we

are interested in practically realizable values of $\mu > 1$), the nature of the dependence of b_{3GP}/b and h_{3GP}/h on detuning does not change, and we are equally justified for values of $z \leq z_{\text{MEKC}}$ in using the formulas for zero detuning. Ratios b_{3GP}/b and h_{3GP}/h are given here by limiting forms of expressions (7.6.17) and (7.6.18), allowing us easily to analyze the influence of mismatch of durations of the pulse and gate pulse on accuracy of range finding. In particular, for small h ratio

$$\frac{S_{0 \times 0}(\mu)}{S_{0 \times 0}(1)} = \frac{(1 + \mu^{0})^{4}}{16\mu^{4}},$$
(7.6.19)

for sufficiently large $h_{\rm SKB}\approx$ $2h\mu/(1+\mu^2)$ the spectral density increases in accordance with

$$\frac{S_{0RB}(\mu)}{S_{0RB}(1)} = \frac{(1+\mu^2)^3}{8\mu^3}, \qquad (7.55)$$

Dependences (7.6.19) and (7.6.20) are shown in Fig. 7.18. Formulas (7.6.17)-(7.6.20) and Fig. 7.18 show that in this case expansion of the gate pulse leads to



Fig. 7.13. Influence of duration of the gate pulse on S_{SKB} with a Gaussian pulse envelope: — two-channel discriminator; --- discriminator with switching of reference signals. considerably more unpleasant consequences than in the case of a square pole and gate pulse. For sufficiently large μ spectral density increased as $\mu^4/16$ for small h and as $\mu^2/3$ for lars h, whereas for a square pulse we had an approximately linear dependence on μ . The physically stronger dependence $S_{\rm SKB}(\mu)/S_{\rm SKB}(1)$ for a Gaussian pulse and gate pulse is explained by the fact that along with increase of power of noise proceeding to the input of the narrow-band filter, which occurs with expansion of the gate pulse, there simultaneously occurs a sharp decrement

of the gain factor, leading immediately to increase of equivalent spectral density. All this imposes more stringent requirements on duration of gate pulses in the case of a Gaussian pulse.

Analogous dependences are simply obtained for a circuit with switching, and this consideration leads to analogous conclusions. Being limited to the case of the spectrum of (7.3.11) and of optimum filter (7.4.18) for a circuit with cwitching, we obtain the following expression for equivalent spectral density:

$$S_{9KB} = \frac{\tau_{H}^{2}}{2\Delta f_{0}} \frac{\left[1 + \frac{2\hbar\mu}{1+\mu^{2}} e^{-\frac{\pi z^{2}}{1+\mu^{2}}}\right]^{3/2} (1+\mu^{2})^{4}}{4\hbar^{4}\mu^{3}\pi^{2}z^{3} \exp\left(-\frac{2\pi z^{2}}{1+\mu^{2}}\right)}.$$
 (7.1.1)

The dependence of S_{BBP}/S_{OHP} on z for various values of h and μ is shown in Fig. (.19. It turns out that the optimum value of z, minimizing S_{BBP} comparatively weakly depends on h and approximately coincides with the value of detuning which ensures the maximum gain factor (7.6.16). Noncoincidence of durations of the pelse

-252-



Fig. 7.19. Influence of parameters of a discriminator with switching of the reference signal on S_{3KB} :

 $---- \mu = 1; ---- \mu = 3.$

sequences as in a two-channel circuit.

In general, if only detuning $z = \frac{\delta}{\tau_{M}}$ is selected close to the optimum for any μ , spectral density increases as:

for small h

and gate pulses leads in a circuit with switching approximately to the same con-

 $\frac{S_{\text{DEB}}(\mu)}{S_{\text{DEE}}(1)} = \frac{(1+\mu^{3})^{2}}{4}; \quad (7.6.22)$

for large h

$$\frac{S_{\text{DRB}}(\mu)}{S_{\text{DRB}}(1)} = \mu^{9}. \qquad (7.6.23)$$

These dependences are shown in Fig. 7.18 by the dotted line.

7.6.3. Discriminator with Differentiation of the Reference Signal

Technical realization of such a discriminator during pulse radiation will

cause, obviously, serious difficulties. If narrow-band filtration of the signal is carried out at intermediate frequency, multiplication by the differentiated reference signal u'(t), always having positive and negative values (Fig. 7.20a), can be carried



Fig. 7.20. Derivative of the modulating signal and its modulus.

out by means of multiplication of the signal from the output of the mixer by function |u'(t)|(Fig. 7.20b) and change of the phase of the amplified signal at the time of passage of function u'(t) through zero.

Possibly simpler from the technical point of view will be

the circuit of a discriminator with sine and cosine channels and filtration of the received signal at low frequency, analogous to the corresponding circuits of detectors (Chapter IV). This circuit is shown in Fig. 7.21, and in the operations executed by it is completely equivalent to the circuit of Fig. 7.2. Formation of correlation integrals (multiplication by reference signals and filtration) is carried out in it at low frequency. The necessity of using two channels with quadrature mixers was discussed in Chapter IV. In each of these channels a discrimination characteristic

can be created, of course, both with the help of two delay-detuned channels and



Fig. 7.21. Block diagram of an optimum discriminator with quadrature channels: 1 - preamplifier; 2 - coherent local oscillator; 3 - phase detector; $4 - \nu/2$ phase shifter; 5 - controlled generator ofreference signals; 6 - multiplier; 7 - narrow-bandlow-frequency filter; 8 - circuit of addition.

with the help of switching of reference signals. The advantage of this kind of circuit for the case of coherent radiation as compared to the circuit of Fig. 7.2 and approximations to it can be possibility of measurement of the distance simultaneously of several targets using only one filter in each channel. Such a filter, carrying out accumulation of the signal

immediately for all distances may be realized by a charge-storage tube.

Since operations of the circuits of Figures 7.2 and 7.21 completely coincide, for the latter there are preserved all results obtained earlier. This pertains also to all possible methods of approximate realization of the circuit of Fig. 7.21 analogous to those considered earlier for the case of filtration in one channel at intermediate frequency. Let us consider for illustration of a circuit with differentiation the case when the signal pulse has cosine form

$$u(t) = \sqrt{\frac{T_r}{\tau_{\pi}}} \cos \frac{\pi t}{2\tau_{\pi}}, \qquad (7.6, 24)$$

where $\tau_{ji}=duration$ of the equivalent square pulse, connected with the duration of a pulse of zeroes τ_0 by relationship

We assume that the reference signal of the first channel $u_1(t)$ coincides in form with the signal pulse $u_1(t) = u(t)$, and the reference signal of the second channel constitutes two adjoining identical square pulses of duration τ_c/ϵ , one of which has positive, and the other, negative polarity. In the variant of construction of a discriminator with filtration at intermediate frequency such a signal is generated in the form of a square pulse of duration τ_c , in the middle of which there is carried out change of phase of the filtered signal, or of heterodyne voltage, by π . If the reference pulses of the second channel have, not rectangular form,

-254-

but are two adjoining quarters of a sine wave, then, obviously, we have complete matching of the form of mc.ulations of the reference and sounding signals. In the considered case functions $C_{ik}(\delta)$ have the form

$$C_{so}(\delta) = C(\delta) = \left(1 - \frac{|\delta|}{\tau_{s}}\right) \cos \frac{n\delta}{2\tau_{s}} + \frac{1}{\pi} \cos \frac{n\delta}{2\tau_{s}} \sin \frac{|\delta|}{\tau_{s}} = C_{11}(\delta) = C_{10}(\delta),$$

$$C_{so}(\delta) = 1 - 3\frac{|\delta|}{\tau_{s}} \quad C_{so}(\delta) = C_{s1}(\delta) = \frac{\delta}{s} \sqrt{\frac{\tau_{s}}{\tau_{s}}} \sin^{2} \frac{\pi\tau_{s}}{\pi_{s}} \sin \frac{\pi\delta}{2\tau_{s}}.$$
(7.6.26)

The mean square spectrum width b for a cosinusoidal pulse

$$b=\frac{n^2}{4n_{\pi}^2},$$

and quantities necessary for substitution in the formula for spectral density are

$$C_{10}(0) = 1, \quad C_{21}(0) = C_{12}(0) = 0, \quad C_{22}(0) = 1, \quad C_{11}(0) = 1.$$

$$C_{10}(0) = 0, \quad C_{20}(0) = \frac{4}{\sqrt{\tau_{\pi}\tau_{\pi}}} \sin^2 \frac{8\tau_{\pi}}{8\tau_{\pi}}.$$

As follows from (7.5.3), there is no systematic error in this case, and equivalent spectral density according to (7.5.5) is

$$S_{0 H B} = \frac{\tau_{B} \tau_{0}}{16 \sin^{4} \frac{\pi \tau_{0}}{8 \tau_{B}}} \frac{\frac{1}{2\pi} \int |H_{1}(i\omega)|^{2} [1 + hS_{0}(\omega)] d\omega}{2h^{2} \left[\frac{1}{2\pi} \int |H_{1}(i\omega)|^{2} S_{0}(\omega) d\omega\right]^{2}}.$$
 (7.6.27)

Expression (7.6.27) differs from S_{ave} for the case of coinciding modulations only by factor

$$\frac{b}{[C_{30}(0)]} = \frac{\pi^2 \tau_e}{64 \tau_{\pi} \sin^4 \frac{\pi \tau_e}{8 \tau_e}},$$
 (7.6.28)

characterizing the decrease of the slope of the discrimination characteristic and the related increase of equivalent spectral density. The dependence of $S_{\overline{ACP}}/S_{OIIT}$ on $\frac{\tau_c}{\tau_n} = \mu$ is shown in Fig. 7.22. With sufficiently good selection of duration of the gate pulse $[\tau_c = (2 \text{ to } 5)\tau_n]$ the loss caused by noncoincidence of modulations is filter the considered circuit almost does not differ from the optimum.



Fig. 7.22. Influence of mismatch of the reference signal with the sound-ing signal on S_{ayu} .

§ 7.7. Continuous Radiation

In the case of a coherent signal of special interest is application of continuous radiation. This is caused by the fact that in a number of cases it may be the most profitable in terms of power, since it permits us for a fixed peak power of the transmitter to ensure maximum average power and, consequently, the greatest accuracy of range finding. Another advantage of continuous radiation is simplicity of realization of frequency selection, which is the most effective means of combatting passive interferences

[34]. During continuous radiation we can use the most diverse forms of modulation: amplitude, frequency, phase-code manipulation, random amplitude, phase or frequency modulation [1, 12, 27, 35-37]. Below we shall consider some of the possible forms of modulation.

7.7.1. Frequency Modulation

Let us consider first a signal with periodic frequency modulation according to the law $\Delta\omega(t)$ (sinusoidal and triangular). We shall consider the amplitude of function $\Delta\omega(t)$ equal to ω_m , so that deviation (maximum change) of frequency is equal to $2\omega_m$. The corresponding laws of modulation are shown in Fig. 7.23. A



Fig. 7.23. Change of frequency during frequency modulation: a) triangular modulation; b) sinusoidal modulation. discriminator with frequency modulation can be constructed on any of the previously considered circuics. It is possible that the most suitable from the technical point of view will be the circuit with differentiation of the reference signal. We consider first the case when modulation of reference signals is not distorted. Autocorrelation functions C(5) for FM signals are determined for values of 8 which are small as compared to the period of modulation by the following relationships:

for sinusoidal modulation

$$C(\delta) = J_{\bullet}(\omega_{m}\delta); \qquad (7.7.1)$$

いたいたちまた、ためにないなど、第二日のないないないないであるという

いいのないとの情報の時間になっていたいです。

「「「「「「「」」」」

for triangular modulation

$$C(\delta) = \frac{\sin \omega_m \delta}{\omega_m \delta}, \qquad (7.7.2)$$

where $J_0(x)$ - Bessel function.

Equivalent spectral density for a circuit with differentiation is determined by formula (7.3.9), where in both cases a = 0, and $b = \frac{\omega_m^2}{2}$ for sinusoidal modulation and $b = \frac{\omega_m^2}{3}$ for triangular modulation. Thus, frequency modulation by triangular law in principle by a factor of $\sqrt{1.5}$ gives greater fluctuation error than sinusoidal modulation with the same deviation of frequency. Remaining dependences, essential for a circuit with differentiation (dependences of S_{END} on the signal-to-noise ratio, bands and frequency response of filters) were already studied earlier and need no further discussion.

We shall investigate now how qualities of a discriminator change depending upon detuning δ when we use a two-channel circuit and a circuit with switching. The slope of the discrimination characteristic in both cases is proportional to $J_0(\omega_m \delta)J_1(\omega_m \delta)$

for sinusoidal FM and to
$$\frac{\sin \omega_m^0}{\omega_m^2 \delta} \left(\frac{1}{\omega_m^0} - \operatorname{ctg} \omega_m^0 \delta\right)$$
 for triangular FM. For a circuit

with two detuned channels quantities h_{3RB} and b_{RRB} , characterizing change of properties of the discriminator with finite detuning, in accordance with (7.3.23) and (7.3.24), are expressed in the following way:

with sinusoidal modulation

$$h_{0XD} = h \frac{2f_0^2(\omega_m \delta)}{1 + f_0 (2\omega_m \delta)}, \qquad (7.7.3)$$

$$b_{0RB} = \Phi_m^2 \frac{J_1^2 (\Phi_m \delta) [1 + J_0(2\Phi_m \delta)]}{J_0^2 (\Phi_m \delta) [1 - J_0(2\Phi_m \delta)]}; \qquad (7.7.4)$$

with triangular modulation

$$h_{\text{sws}} = h \frac{2 \left(\frac{\sin \omega_{m} \delta}{\omega_{m} \delta}\right)^{4}}{1 + \frac{\sin 2\omega_{m} \delta}{2\omega_{m} \delta}}$$
(7.7.5)
$$b_{\text{sws}} = \Phi_{m}^{2} \frac{\left[\frac{1}{\omega_{m} \delta} - \operatorname{ctg} \omega_{m} \delta\right]^{2} \left[1 + \frac{\sin 2\omega_{m} \delta}{2\omega_{m} \delta}\right]}{1 - \frac{\sin 2\omega_{m} \delta}{2\omega_{m} \delta}}.$$
(7.7.6)

The magnitude of equivalent spectral density is determined here by formula (7.3.22) with substitution in it of (7.7.3)-(7.7.6). In particular, for small h for sinusoidal FM

$$\frac{S_{000}(\delta)}{S_{000}(0)} = \frac{h^{4}b}{h_{000}^{2}b_{000}} = \frac{1 - J_{0}^{2}(2\omega_{m}\delta)}{8J_{1}^{2}(\omega_{m}\delta)J_{0}^{2}(\omega_{m}\delta)}; \qquad (7.7.7)$$

for triangular FM

$$\frac{S_{\text{one}}(\theta)}{S_{\text{one}}(\theta)} = \frac{1 - \left(\frac{\sin 2\omega_m \theta}{2\omega_m \theta}\right)^2}{12 \left(\frac{\sin \omega_m \theta}{\omega_m \theta}\right)^4 \left\lfloor \frac{1}{\omega_m \theta} - \operatorname{ctg} \omega_m \theta \right\rfloor^2}.$$
(7.7.8)

With sufficiently large have the corresponding formulas have the form

$$\frac{S_{000}(l)}{S_{000}(0)} = \frac{1 - J_0^2(20_m l)}{4J_1^2(0_m l)},$$
(7.7.9)

$$\frac{S_{\text{ens}}(\theta)}{S_{\text{ens}}(\theta)} = \frac{1 - \frac{1}{2\omega_{\text{ens}}\theta}}{6\left(\frac{\sin\omega_{\text{ens}}\theta}{\omega_{\text{ens}}\theta}\right)^{2} \left[\frac{1}{\omega_{\text{ens}}\theta} - \operatorname{ctg}\omega_{\text{ens}}\theta\right]^{2^{n}}}$$
(7.7.10)

The dependence of ratios $b_{ZKB}(\delta)/b$, $h_{SKB}(\delta)/n$, $K_{II}(\delta)/K_{IIMENC}$ and $S_{SKB}(\delta)/S_{OHT}$ according to (7.7.7) and (7.7.9) on detuning δ with sinusoidal FM is shown in Fig. 7.24. Analogous dependences for triangular FM are shown in Fig. 7.25. From these graphs it is clear that the magnitude of detuning plays practically an identical role in both cases. Selection of detuning, just as in the previously considered case of modulation by Gaussian pulses, is best produced from condition of a maximum gain factor i.e., $\delta_0 \approx \frac{1}{\omega_m}$ for sinusoidal FM and $\delta_C \approx \frac{1.2}{\omega_m}$ for triangular FM. The equivalent spectral density increases here to a still smaller degree than in Paragraph 7.6.2.

For large h_{3KB} in both cases $S_{3KB}(\delta_0)/S_{OHT}$ is practically equal to one, but for small h spectral density S_{3KB} with such detuning is increased in all by 4-6%. To

find the character of relationship $S_{BKD}(5)/S_{OHT}$ for intermediate values of h in Fig. 7.26 we show this ratio during sinusoidal FM for various h. The frequency response of filter H₁(im) is assumed to coincide with the optimum (7.2.5) for spectrum (7.3.11). It is obvious that here $S_{BKD}(0) = S_{OHT}$.



Fig. 7.24. Influence of detuning on the characteristic of a two-channel discriminator with sinusoidal FM:

 $\begin{array}{c} ----K_{A} \left(b X K_{A} \right)_{A = 0} \left(b Y S_{B = V}$



Fig. 7.26. Influence of detuning on S_{3RB} for a two-channel discriminator with sinusoidal FM.





Thus, the examples considered here once again confirm the fact that with any reasonable selection of detuning a two-channel circuit practically does not differ from the optimum when there are coinciding laws of modulation of the reference and sounding signals.

To find the character of $S_{313}(5)$ in a circuit with switching we shall limit ourselves to the case of spectrum (7.3.11) and the corresponding optimum filter (7.4.18). Then, in accordance with (7.4.19), with sinusoidal modulation

$$S_{9RB}(\delta) = \frac{\left[1 + kJ_{9}^{2}(\omega_{m}\delta)\right]^{3/2}}{2\Delta f_{e}k^{2}\omega_{m}^{2}J_{1}^{2}(\omega_{m}\delta)J_{0}^{2}(\gamma_{m}\delta)}.$$
 (7.7.11)

Ratio $S_{3KB}(\delta)/S_{ONT}$ for the given circuit is shown for various values of h in Fig. 7.27. This dependence is more substantial than with modulation by Gaussian pulses and an analogous discriminator circuit. The minimum, occurring for a certain

-259-



Fig. 7.27. Influence of detuning on Same for a discriminator with switching of reference signals during sinusoidal FM.

graph, increases with increase of the signal-to-noise ratio h. For small h this value coincides with δ , for which the gain factor will be maximum; for larger h it increases, approaching a value, corresponding to the first root of function $J_0(x)$.

Presence of a minimum for $S_{3KB}(\delta)$ should be allowed for during practical construction of a system. Incorrect selection of detuning, as follows from Fig. 7.27, leads to impermissibly great losses of accuracy. Inasmuch as normally the same system should be used for

different values of signal power (at different distances), one should select b equal to 50, corresponding to such a value of h at which the most stringent requirements are made on accuracy of the system; this usually takes place at maximum ranges to the target, when the value of h has the order of 5-10. The optimum value of δ_{A} here is about $1.5/\omega_m$, and fluctuation error is 1.5-2 times larger than the minimum. The dependence of $S_{AKB}(\delta)/S_{OIIT}$ on h has a normal character. For small h the loss is comparatively small; with growth of h quantity $S_{and}(\delta)/S_{ORT}$ increases, which is caused by a different order of decrease of Smp for large h in an optimum circuit and in a circuit with switching.

Results for triangular FM turn out to be completely analogous; optimum values of δ , at which $S_{300}(\delta)$ will be minimal, and the actual curves for $S_{300}(\delta)/S_{OI}$ practically coincide with the curves for sinusoidal FM.

In real circuits with FM the reference and sounding signals are obtained by means of frequency modulation of separate generators. With such a construction due to noncoincidence of modulation characteristics of the transmitter and local oscillator the laws of modulation may not coincide. A very simple case of notcoincidence is difference in frequency deviations. The influence of this phenomenon on accuracy can be calculated by the formulas of the preceding paragraphs. We shall limit ourselves to circuits with two channels and with differentiation of the reference signal. In § 7.6 with the example of pulse radiation we already proved that the influence of nonidentity in the two-channel circuit practically does not

-260-

depend on detuning for not very large values of 5. Then, considering sufficiently small detunings 5, with the help of formulas (7.3.33) and (7.5.5) we can simply prove that all results for both circuits are identical.

If the difference of frequency deviations of the sounding and reference signals is equal to Δx_m , with sinusoidal FM

$$C_{ii}(0) = 1, |C_{ii}(0)| = |J_{0}(n)|, a_{ii} = a_{ii} = 0, \qquad (7.7.12)$$

$$b_{ii} = \frac{a_{ii}^{3}}{2}, b_{i0} = \frac{J_{1}(n)}{p |J_{0}(n)|} a_{m}^{2},$$

where $\mu = \frac{\Delta \omega_m}{\omega_r} = \frac{\Delta \omega_m T_r}{2\pi}$ - ratio of the difference of deviations to the frequency of repetition of modulation.

With triangular FM

$$C_{ii}(0) = 1, C_{ii} = \int \cos \frac{a_{ii}}{4} (1-z^2) dz, a_{ii} = a_{ii} = 0,$$

$$b_{11} = \frac{\omega_{11}^2}{3}, \ b_{10} = \omega_{10}^2 \cos \frac{\omega_{11}}{4} \frac{\int_0^1 z^0 \cos \frac{\omega_{11}}{4} (1-z^0) dz}{\int_0^1 \cos \frac{\omega_{11}}{4} (1-z^0) dz}.$$
 (7.7.13)

Substituting (7.7.12) and (7.7.13) in (7.3.33), we can obtain the dependence of equivalent spectral density on μ . This relationship is shown in Fig. 7.28 on the



 assumption that the frequency response of the filters is optimum. As follows from the graphs, when $\mu > 1$ there occurs rapid increase of spectral density. When $\mu \approx$ ≈ 2.4 for sinusoidal FM and $\mu = 2$ for triangular, the reference and sounding signals become orthogonal and S_{3KB} turns into infinity. Deviations $\Delta \omega_m > (1 \text{ to}$ $1.5)\omega_r$, apparently, already are impermissible. Noncoincidence of deviations is somewhat more evident for triangular modulation and, as in the general case of any noncoincidences, leads to greater impairments, the less the signal-to-noise Fatio.

7.7.2. Phase-Code Manipulation (FKM)

Characteristics of different forms of FKM are considered in Chapter I. In the same place it is shown that with any reasonable selection of the code, ensuring a sufficiently low level of side lobes, the autocorrelation function of a phase-codemanipulated signal within the main lobe with a properly selected code is described by the following approximate expression:

 $C(\delta) \approx 1 - \frac{|\delta|}{\tau_{\rm s}}, |\delta| \leq \tau_{\rm K}, \qquad (7.7.14)$

which becomes more exact, the larger the numbers of code intervals n in the period of repetition of the signal. Here τ_{χ} - duration of the code interval. For certain codes (e.g., for binary code with values of phase 0 and π - arccos $\frac{n-1}{n+1}$) this expression is exact.

Function (7.7.14) for values of 5 from $-\tau_R$ to τ_R , which are only of interest for the work of an automatic range finder, coincides with the autocorrelation function for square pulse modulation. Since all dependences of characteristics of the discriminator on the form and parameters of modulation are completely determined by function C(5), it is obvious that it remains for us only to repeat everything said in Paragraph 7.6.1 with respect to range finding with modulation by square pulses with duration $\tau_R = \tau_R$ and with coincidence of forms of the sounding and reference signals. With FKM there can be used either a two-channel discriminator, or a discriminator with switching, and in both cases all results of Paragraph 7.6.1 are simply repeated.

We consider that duration τ_{Π} of switching of phase of an FKM signal from one value to the other is finite. This is necessary when the magnitude of detuning is less than the duration of switching. The coefficients of the expansion of function C(5) in a series of powers of 5 are equal to

$$a=0, b=\frac{1}{T_r}\int_{0}^{T_r} [\Psi(t)]^p dt = \frac{l}{T_r}\int_{0}^{T_r} [\Psi(t)]^p dt,$$

where l - number of changes of phase, and the integral in the final formula is taken over the edge of a code pulse.

if phase changes by linear law, $[\forall'(t)]^2 = \frac{\pi^2}{\tau_{\pi}^2}$, and taking into account $\iota \approx \frac{n}{2}$

-262-

It is interesting to note that the maximum value of b is reached when the phase changes at the end of every interval, i.e., l = n. Quantity b here is increased by a factor of 2; however, simultaneously, in function C(5) there appear large side lobes. Thus, increase of fluctuation error is the price we pay for the high resolution capability provided by FKM.

 $b=\frac{\pi^2}{2}\frac{\tau_n}{\tau_n}\frac{1}{\tau_n^2}.$

7.7.3. Noise Phase Modulation

Let us consider a signal whose phase changes proportionally to the instantaneous value of a normal stationary random process

$\phi(l) = \alpha(l).$

If process $\xi(t)$ is ergodic, time-averaging, produced during calculation of function $C(\delta)$, is equivalent to ensemble-averaging and

$$C(\mathbf{a}) = e^{\frac{1}{\mathbf{a}} \left[\frac{1}{2} \left(\frac{1}{2} - \frac{1}{2} \right) - \frac{1}{2} \left(\frac{1}{2} \right) \right]} = e^{-\frac{1}{2} \left[\frac{1}{2} - \frac{1}{2} \right]}$$
(7.7.16)

Children with States States and a constraint of the states of the states

(7.7.15)

where σ_m^2 and $\rho_m(\delta)$ - variance and correlation coefficient of modulating process $\xi(t)$. The mean square spectrum width of modulation in accordance with (7.2.11) and (7.2.12) is equal to

$$b = -C^{*}(0) = -e^{2\sigma_{ab}^{2}} (0) (a = 0), \qquad (7.7.17)$$

i.e., is $\alpha^2 \sigma_m^2$ times greater than the mean square spectrum width of the modulating process $\xi(t)$. In particular, if the spectral density of $\xi(t)$ is uniform in frequency band $(-\omega_m, \omega_m)$,

$$C(\mathbf{i}) = \mathbf{e}^{-\mathbf{a}_{m}^{2}} \left(1 - \frac{\mathbf{a}_{m}^{2} \mathbf{a}_{m}^{2}}{\mathbf{a}_{m}^{2}}\right)$$

$$b = \mathbf{e}^{2} \mathbf{a}_{m}^{2} \frac{\mathbf{a}_{m}^{2}}{3}.$$
(7.7.18)

A characteristic feature of the considered form of modulation is the aperiodicity of function $C(\delta)$ and the related absence of ambiguity, and also the fact that $C(\delta)$ does not decrease to zero as $\delta \rightarrow \infty$, but has a finite limit $C(\infty) = e^{-\alpha^2 \sigma_m^2}$. The magnitude of this limit, determining the possibility of resolution of targets, sufficiently removed in distance, depends on the magnitude of product $\alpha \sigma_m$. For instance, -0.86 $(\alpha \sigma_m)^2$ for C(5) from (7.7.18) the first spurious maximum has magnitude e

means that for a fixed magnitude of b it is more expedient to use narrow-band modulating voltage with a larger magnitude of $\alpha \sigma_m$ than broad-band modulating

voltage with smaller aom.

In the considered case the slope of the discrimination characteristic changes proportionally to

 $\frac{\sin \omega_m \delta}{\omega_m \delta} \left(\frac{1}{\omega_m \delta} - \operatorname{cig} \omega_m \delta \right) e^{-2(\omega \omega_m F \left(1 - \frac{\sin \omega_m A}{\omega_m A} \right)}.$

and all formulas necessary for calculations $(h_{args}, b_{args}, S_{args}(\delta)/S_{args}(0))$ are obtained,



Fig. 7.29. Dependence of the gain factor on detuning with noise phase modulation.

for instance, by replacement in the formulas for sinusoidal FM of $J_0(\omega_m \delta)$ by (7.7.18), and of $\omega_m J_1(\omega_m \delta)$ by

This

 $\mathbf{e}_{m} \frac{\sin \mathbf{e}_{m}\delta}{\mathbf{e}_{m}\delta} \left(\frac{1}{\mathbf{e}_{m}\delta} - \operatorname{ctg} \mathbf{e}_{m}\delta \right) (a\sigma_{m})^{2} \times \mathbf{e}_{m}\delta$ $\left| \left| \left| \left| \exp \left\{ - \left(\cos_{m} \right)^{2} \left[1 - \frac{\sin \omega_{m} \delta}{\omega_{m} \delta} \right] \right| \right| \right| \right|$

or by the general formulas of §§ 7.3-7.5. Dependences of $K_{II}/K_{II MANC}$ and of $S_{IKB}(5)/S_{IKB}(0)$ on detuning for various values of h for the

two-channel circuit and the circuit with switching are shown in Figures 7.29-7.31, respectively. Quantity $a\sigma_m$ in the graphs is taken equal to five. In accordance with general affirmations of Paragraph 7.3.3 the influence of detuning for a two-channel circuit in this case is more substantial than in those considered earlier. When $\delta = \delta_{MENC} \approx \frac{0.2}{\sigma_m}$ the equivalent spectral density is increased by a factor of 1.3 when h = 100 and a factor of 2 when h = 0.1. In the circuit with switching the dependence of $S_{MENC}(\delta)/S_{MEN}$ on detuning has a rather complicated character; minimum values of this ratio are rather large, and increase of it for 5, differing from δ_0 , occurs comparatively rapidly. The optimum value δ_0 essentially depends on the signal-to-noise ratio and for large h greatly differs from δ_{WENC}^* .

-264-



Fig. 7.30. Influence of detuning on S_{3KB} in a two-channel discriminator with noise phase modulation.



Fig. 7.31. Influence of detuning on $S_{\rm HB}$ in a discriminator with switching of reference signals with noise phase modulation.

7.7.4. Noise Signal

We next consider range finding with purely noise radiation. Such a signal can be formed by transmission of broad-band noise through a bandpass filter with subsequent amplification, transfer to the frequency of radiation and amplification at this frequency. Modulating voltage u(t) here is a normal stationary random process, and the autocorrelation function C(5) coincides with the correlation function of this process, so that

$$C(\delta) = \frac{1}{2\pi} \int_{-\infty}^{\infty} S_m(\omega) e^{i\omega \delta} d\omega = \frac{1}{2\pi} \int_{-\infty}^{\infty} e^{i\omega \delta} \frac{[H_{\bullet}(i\omega)]^2}{\delta f_{1\bullet}} d\omega, \qquad (7.7.19)$$

where $S_{u}(\omega)$ - spectral density of process u(t)

$$\left(\frac{1}{2\pi}\int\limits_{-\infty}^{\infty}S_{m}(\boldsymbol{\omega})\,d\boldsymbol{\omega}=1\right);$$

H $_{\mbox{$\dot{\Phi}$}}$ (iw) - frequency response of the low-frequency equivalent of the shaping filter;

 Δf_{ab} - its passband.

In spite of the fact that technical methods of delaying such a signal, just as the signal with phase noise modulation considered in Paragraph 7.7.3, at present are far from clear, it is of definite interest both from the point of view of simplicity of generation, and also from the point of view of absence of ambiguity.

-265-

In the optimum system and in a circuit with differentiation accuracy of range finding is determined by the mean square spectrum width

$$b = \frac{1}{2\pi} \int_{-\infty}^{\infty} \omega^2 S_m(\omega) \, d\omega,$$

and the influence of finite detuning in the two-channel circuit and in the circuit with switching is determined by the form of function C(5). If the spectral density of modulation $S_m(\omega)$ is described by a Gaussian curve or has rectangular form, function C(5) has the form of (7.6.13) or (7.7.2), and all results of Paragraphs 7.4.2 and 7.7.1, pertaining to pulse modulation by Gaussian pulses and frequency modulation by triangular law, remain in force. We next consider the case, when the shaping filter is a sequency of two decoupled LRC-filters, i.e., frequency response H_{ϕ} (i ω) has the form of (7.3.16). Substituting (7.3.16) in (7.7.19), we obtain

> $C(\delta) = (1 + 4\Delta f_{a \oplus} |\delta|) e^{-\Delta f_{a \oplus} |\delta|}, \qquad (7.7, 20)$ $a = 0, \ b = -C''(0) = 16\Delta f_{a \oplus}^{2}.$

Dependences of $K_{\Pi}(\delta)/K_{\Pi MARG}$, $b_{3KB}(\delta)/b$, $h_{3KB}(\delta)/h$ and $S_{3KB}(\delta)/S_{3KB}(0)$ on detuning, calculated for a two-channel circuit by formulas of § 7.3, pertaining to the case of coinciding modulations of autocorrelation determined by formula (7.7.20), are constructed in Fig. 7.32. The influence of detuning in this case is somewhat more



Fig. 7.32. Influence of detuning on sharacteristics of a two-channel discriminator with a noise signal: $-K_{A}(\delta)/K_{A}|_{MARC}$; $---\delta_{MR}(\delta)/\delta$; $---\delta_{MR}(\delta)/\delta$; $---S_{MR}(\delta)/S_{ORT}$ $\forall A>1$; $-\times-\times-S_{MR}(\delta)/S_{ORT}$

substantial than for other forms of $C(\delta)$. When $\delta = \delta_{MBRC} S_{REB}(\phi_{MERC})/C_{REB}(\phi)$ increases already by 1.7 times for large h and by 2.5 times for small h. This difference is explained by the fact that due to the showness of decrease of $C(\delta)$ ratio $\frac{b_{REB}}{b} \approx 1$ for all 6.

7.7.5. Amplitude Modulation Amplitude modulation during continuous radiation has the deficiency that for a fixed peak power of the sounding signal it provides, other things being equal, lower mean power and, consequently, lower accuracy of measurement than frequency or phase modulation. However, in certain cases its application, possibly, is sensible, and we shall briefly discuss it, limiting ourselves to the very simple case of sinusoidal modulation by one frequency. The modulating signal u(t) here has the form

$$u(t) = \frac{1 + m \cos \omega_m t}{\sqrt{1 + \frac{m^2}{2}}},$$
 (7.7.21)

and its autocorrelation function and mean square spectrum width are

$$C(\delta) = \frac{1 + \frac{m^2}{2} \cos \omega_m \delta}{1 + \frac{m^2}{2}},$$
 (7.7.22)

$$b = \frac{m^2 \omega_m^2}{2\left(1 + \frac{m^2}{2}\right)}, a = 0.$$
 (7.7.23)

The most reasonable scheme of construction of a discriminator in this case is, in all probability, a circuit with differentiation of the reference signal. Multiplication by reference signals reduces to heterodyning the received signal with the signal amplitude-modulated by law (7.7.21) in the first channel and with signal $\sin [(\omega_0 + \omega_{\Pi p} + \omega_m)t - \omega_m^{\tau}] - \sin [(\omega_0 + \omega_{\Pi p} - \omega_m)t + \omega_m^{\tau}]$ in the second channel. The required coincidence of laws of modulation in this case can be ensured without great difficulties. The magnitude of the equivalent spectral density is determined here by expression (7.3.9).

The form of formulas (7.7.22) and (7.7.23) already indicates the essential defleciences of such a signal. It is characterized by low resolution capability and great ambiguity. The period of ambiguity and accuracy of range finding are determined by the same magnitude — the frequency of modulation ω_m , where $1/\sqrt{b}$, having the meaning of a certain equivalent pulse duration, for small m is even larger than the period of ambiguity. Accuracy of range finding with sinusoidal AM essentially depends on the modulation percentage. For small m fluctuation error is proportional to 1/m, and with increase of m to infinity it approaches a finite limit, determined only by the frequency of modulation. Physically the signal with m > 1 signifies radiation at three frequencies ω_0 , $\omega_0 - \omega_m$ and $\omega_0 + \omega_m$, where with increase of m the relative intensity of the center frequency decreases, while that of the sidebands increases. The case $m = \omega$ signifies radiation at two frequencies, $2\omega_m$ apart and

-267-
having identical intensity. As $m \rightarrow \infty$ autocorrelation function $C(\delta) = \cos \omega_m \delta$ and $b = \omega_m^2$.

§ 7.8. Pulse Radiation with Additional Modulation

The effort to ensure high power potential in pulse radars leads to the requirement of decreasing the off-duty factor. At the same time in certain cases the only method of eliminating ambiguity remaining is selection of a sufficiently low frequency of repetition. These two requirements, in turn, lead to the necessity of increase of pulse duration to such limits that in a number of cases it is impossible to ensure the needed accuracy of range finding. With selection of a high frequency of repetition with ambiguity with respect to range pulse duration can be so great that the requirements on accuracy of measurement and resolution capability with respect to range are not satisfied. Therefore, it is of interest to consider intrapluse modulation. Use of pulse signals with additional modulation permits us to immediately obtain high power potential, great accuracy of range finding, and good resolving power [38-39].

As additional modulation we can use all those forms of modulation which were considered for continuous radiation. We shall limit ourselves to only two examples — linear frequency modulation and phase-code manipulation within limits of a pulse. Signals with such forms of modulation possess, apparently, the greatest advantages. As discriminators here we can use all the circuits considered earlier, both discriminators of correlation type of all three forms, and also discriminators with optimum filtration and pulse shortening (§ 7.5).

7.8.1. Linear Frequency Modulation

In the given case additional modulation consists of linear change of frequency within the limits of a pulse. Without loss of generality one may assume that this



Fig. 7.33. Pulse signal with linear frequency modulation.

change occurs symmetrically with respect to the center of the pulse with speed $a = \frac{d\omega}{dt}$ (Fig. 7.33). Then signal u(t) has the form

$$u(l) = u_{a}(l)e^{\frac{tal^{2}}{2}},$$
 (7.8.1)

where $u_a(t)$ is a function, describing the pulse shape, where, as before,

 $\frac{1}{T_r} \int_{T_r}^{\frac{T_r}{2}} u_4^2(l) dl = 1, \quad u_2(0) = \sqrt{\frac{T_r}{\tau_1}}, \quad (7.8.2)$

 τ_{μ} - duration of an equivalent square pulse of the same energy.

Subsequently we assume that $u_a(t)$ is a function symmetric with respect to zero. Furthermore, we consider that the width of the spectrum of the modulated pulse is much greater than the width of the spectrum of the unmodulated pulse, i.e., $a\tau_N^2 >> 1$. This condition means that in practice $u_a(t - \frac{1}{a\tau_N}) = u_a(t)$. Then the autocorrelation function of the signal is

$$C(\hat{s}) = \frac{1}{T_r} \int_{-T_r/2}^{T_r/2} u_a(t) u_a(t+\hat{s}) e^{iaMt + \frac{iaM}{2}} dt \approx \frac{1}{T_r} \int_{-T_r/2}^{T_r/2} u_a^2(t) \cos a^{\hat{s}}t \, dt, \qquad (7.8.3)$$

or, introducing frequency deviation for pulse duration $2\omega_m = a\tau_{\mu}$ and considering finiteness of pulse $u_a(t)$, we have

$$C(\mathfrak{F}) = \frac{1}{T_r} \int_{-\infty}^{\infty} u_{\mathfrak{a}}^2(t) \, \mathrm{e}^{i\mathfrak{a}\mathfrak{M}} dt = \frac{1}{T_r} \int_{-\infty}^{\infty} u_{\mathfrak{a}}^2(t) \, \mathrm{e}^{i - \frac{2\mathfrak{m}_{\mathfrak{a}}}{\mathfrak{r}_{\mathfrak{a}}}} dt. \tag{7.8.4}$$

Let us consider several examples for modulation by pulses of different shape. If, for instance, $u_a(t)$ is a square pulse,

$$C(\hat{o}) = \frac{\sin \frac{a\tau_n \hat{\sigma}}{2}}{\frac{a\tau_n \hat{\sigma}}{2}} = \frac{\sin \omega_n \hat{\sigma}}{\omega_m \hat{\sigma}}, \qquad (7.12)$$

which coincides for identical deviations with (7.72) for triangular FM during continuous radiation.

During modulation by Gaussian pulses (7.6.9)

$$C(2) = e^{-\frac{a^{1}r^{2}}{4a}} = e^{-\frac{a^{2}b^{2}}{a}}, \quad b = \frac{2\omega_{m}^{2}}{a}.$$
 (7.3.6)

which coincides with the case of absence of intrapulse modulation, if pulse duration has equivalent magnitude

$$\tau_{0XB} = \frac{\pi}{2\omega_m} \frac{\pi}{a\tau_B}.$$
 (7.8.7)

In this case it is also possible to easily obtain an exact expression for $C(\delta)$ without the above-indicated assumption about the relationship of spectrum width of modulated and unmodulated pulses. Expressions for $C(\delta)$, 2 and the have the form

-269-

$$C(\delta) = \exp\left\{-\frac{\omega_{m}^{2}\delta^{2}}{\pi}\left(1+\frac{\pi^{2}}{4\omega_{m}^{2}\tau_{g}^{2}}\right)\right\}, \qquad (7.8.8)$$

$$b = \frac{2\omega_{m}^{2}}{\pi}\left(1+\frac{\pi^{2}}{4\omega_{m}^{2}\tau_{g}^{2}}\right), \quad \tau_{BKB} = \frac{\pi}{2\omega_{m}}\frac{1}{\sqrt{1+\frac{\pi^{2}}{4\omega_{m}^{2}\tau_{g}^{2}}}}.$$

Thus, from the given example it is clear that corrections to results (7.3.0) and (7.6.7) have the order $\frac{\pi^2}{4\omega_m^2\tau_M^2} = \frac{\pi^2}{a^2\tau_M^4}$. If we require that duration of the shorter is pulse $\tau_{\rm SNS}$ be less than the duration of the basic pulse by at least a factor of 5-10, the correction for allowance for finite duration of the basic pulse will have the order (1-5)10⁻² and can be considered immaterial.

From comparison of (7.8.5) and (7.8.6) it follows that the shape of the pulse envelope comparatively weakly affects accuracy of range finding. With a given rate of change of frequency a and effective pulse duration $\tau_{\rm M}$ or with assigned frequency deviation for a pulse duration $2\omega_{\rm m} = a\tau_{\rm M}$ coefficient b is approximately identical both for the Gaussian and for the square pulse. However, the shape of the pulse envelope essentially affects behavior of function C(5) with large 0, i.e., the resolving power with respect to distance. With a square envelope we distin comparatively large side lobes of C(5), and with a Gaussian envelope C(5) monotonizatly decreases. For pulses of sufficiently great duration to recreate a Gaussian envelope which ensures high resolving power may be difficult. Therefore, it is useful to consider a more realistic envelope shape, which at the same time would ensure good resolving power. If, for instance, the envelope has the shape of a cosine wave (7.6.24),

 $C(\delta) = \frac{\sin 2\omega_m \delta}{2\omega_m \delta} \frac{1}{1 - \frac{4}{\pi} \omega_m^2 \delta^2} \quad (2\omega_m - a \tau_{\mu}),$ $b = \frac{4 (\pi^2 - b)}{3\pi^4} \omega_m^2 \approx 0.52\omega_m^2.$

This function decreases in the vicinity of zero somewhat faster than $\frac{\sin \omega_m \delta}{\omega_m \delta}$, and more slowly than Gaussian (7.8.7). Its first zero coincides with the first zero of $\sin \omega_m \delta/\omega_m \delta$ when $\delta = \pi/\omega_m$, and further zeroes occur with twice the frequency. Already the first spurious maximum of this function comprises in absolute value only 2.7% of C(0) in distinction from the case of a square envelope, where this maximum is 22%. Thus, modulation by cosinusoidal pulses ensures practically the same resolving power as with a Gaussian envelope.

-270-

It should be noted that relationship (7.8.4) permits us to find the shape of the pulse envelope necessary for producing a signal with any assigned autocorrelation function $C(\delta)$. Therefore, we have the possibility of assigning any shape of shortened pulse and here must only correctly select the shape of the envelope of the fundamental pulse. Froducing in (7.8.4) reverse Fourier transformation, we obtain the following expression, establishing the relationship between the assigned function $C(\delta)$ and the required form of the pulse envelope $u_{a}(t)$

$$u_{a}(t) = \sqrt{\frac{T_{ra}}{2\pi}} \int_{-\infty}^{\infty} C(\delta) e^{-iaM} d\delta. \qquad (7.8.9)$$

따 다구하는 다. 바람이 하는 것은 마람이 있는 다. 아무리 가 다 아무리 가 다 아무리 가 다 아무리 가 다 다 다 가 다 다 다 가 다 다 가 다 다 가 다 다 가 다 다 가 다 다 가 다 다 가

It is necessary, of course, to assign $C(\delta)$ in such a manner that it indeed is an autocorrelation function and possesses the required properties for this (the Fourier transform of $C(\delta)$ should be positive everywhere). Furthermore, for an assigned power and effective pulse duration $u_a(t)$, i.e., with fulfillment of (7.8.2), function $C(\delta)$ should obey normalizing condition

$$\int_{-\infty}^{\infty} C(\delta) d\delta = \frac{2\pi}{a\tau_{\pi}} = \frac{\pi}{\omega_{\pi}}, \qquad (7.8.10)$$

which for an assigned rate of change of frequency a fixes the effective duration of $C(\delta)$, and for an assigned duration of $C(\delta)$ assigns the required magnitude of a.

For instance, if we require that $C(\delta)$ be a triangular function of form (7.6.1), condition (7.8.10) gives

$$C(\delta) = 1 - \frac{\Phi_{m}[\delta]}{\pi}, \quad |\delta| < \frac{\pi}{\omega_{m}}, \quad (7.8.11)$$

and the shape of the pulse envelope has the form

$$\mathcal{H}_{\alpha}(l) = \sqrt{\frac{T_{r}}{\tau_{\alpha}}} \left| \frac{\sin \frac{\tau_{\alpha}}{\tau_{\alpha}}}{\frac{\pi l}{\tau_{\alpha}}} \right|, \qquad (7.8.12)$$

i.e., is described by a curve of form $(\sin x)/x$. Then a signal with such an envelope and linear change of frequency will be equivalent to a square pulse of duration $\frac{\pi}{\omega_c}$.

Functions C(0), corresponding to (7.8.5), (7.8.6), (7.8.8) and (7.4.12), are shown in Fig. 7.34. Questions of range finding for a signal with linear change of frequency and a Gaussian envelope, and for a square signal and of a shape of form (7.8.12) we considered earlier. A signal with a cosinusoidal envelope have as



Fig. 7.34. Autocorrelation functions for a pulse signal with linear frequency modulation:



follows from Fig. 7.34. an autocorrelation function $\sigma(b)$ so close in form to the Gaussian Gauwithout essential differences forit we have all the results of Paragraph 7.5.2. 「「「「「「「「」」」」」

7.8.2. Pulse Radiation with MRM Phase-code manipulation of a signal can also be used during pulse radiation. Here, on codes we make the same requirement of high resolving power as during continuous radiation. Different only is selection of the code, and this is caused by the fact that

to extreme code intervals in a shifted signal, occurring beyond the limits of the initial pulse, there correspond in the initial signal empty places, and not the start of the next or the end of the preceding period, as in the continue of the .

With a sufficiently large number of code intervals in a pulse the autocorrelation function practically coincides with the autocorrelation function of a continuous signal (see Chapter I), and all results in the study of accuracy completely coincide with results of Paragraph 7.6.1.

§ 7.9. Multifrequency Radiation Without Modulation

As we know [1], range finding, in principle, is possible in the presence in the spectrum of the sounding signal of only two frequency components. Here, field is measured by the difference of phase advances of these frequency components. A deficiency of such a method is low accuracy. It can be improved by increasing the number of radiated sine waves and rational selection of frequencies. In the process of measurement one should use here difference of phases of all sine waves, where the minimum difference is used for rough but also unambiguous range finding, and other differences larger in magnitude, are for more exact measurement. In order to ensure tuning of frequency away from passive interference the frequencies should be selected sufficiently far from each other, and for simple range finding one should use a difference of phases of the second order (the difference of differences of phases).

-272-

In the general case of m radiated oscillations of different frequencies with arbitrary amplitudes a_j and phases ψ_j sounding signal u(t) can be presented in the form

$$u(t) = \sum_{j=1}^{m} A_j e^{i \omega_j t}, \qquad (7.9.1)$$

where $A_j = a_j e^{i\psi_j}$; $\omega_j = difference$ between the j-th radiated frequency and an arbitrarily selected carrier frequency ω_0 , and $\sum_{j=1}^{m} |A_j|^* = 1$. It is obvious that all results obtained earlier are applicable to a signal of

such form, and for range finding using a multifrequency signal we can use all the above considered discriminator circuits. The mean square spectrum width of the signal, determining accuracy of range finding, is equal to

$$b - a^{2} = \sum_{j=1}^{m} \omega_{j}^{2} |A_{j}|^{2} - \left(\sum_{j=1}^{m} \omega_{j} |A_{j}|^{2} \right), \qquad (7.9.2)$$

and function C(5) is recorded in the form

$$C(\delta) = \sum_{j=1}^{m} |A_j|^* e^{i - j\delta}. \qquad (7.9.3)$$

Resolving power, ensured with use of a set of a small number of sine waves, is very low. As an example we shall consider the case of three sinusoids of identical amplitude. Here,

$$|C(\delta)|^{2} = \frac{1}{3} \left\{ 1 + \frac{2}{3} \left[\cos \left(\omega_{2} - \omega_{1} \right) \delta + \left(7.9.4 \right) \right] \right\}.$$

$$(7.9.4)$$

The dependence of $|C(5)|^2$ on product $(\omega_2 - \omega_1)^5$ for various values of $\alpha = (\frac{\omega_3 - \omega_2}{\omega_2 - \omega_1})^6$ is shown in Fig. 7.35. As can be seen from the figure, the considered relationship has spurious maxima, attaining a magnitude of 0.6-0.8.

Thus, he advantage connected with the possibility of tuning frequency away from party interference with simultaneous simple measurement of large distances (as compared to the case when tuning is achieved by increase of the frequency of repetition), is accompanied for a given signal by considerable impairment of resolving

-273-

power. With increase of the number of frequencies utilized and proper selection



Fig. 7.35. Autocorrelation function of a three-frequency signal: $-\alpha = 0.8; ---\alpha = 1.2.$

of them the resolving power can be increased, and here in its properties the signal, apparently. inevitably nears normally considered forms of modulation.

7.9.1. Case of Arbitrarily Correlated Frequency Components of the Reflected Signal

Till recently, in examining different Vortes of modulation of the sounding signal we implici 17 assumed that the width of the spectrum of model tion is limited from above by the condition complete correlatedness of the most distant frequency components of the spectrum. Physiccally, this means that the interval of resolution

with respect to distance exceeds the distance to the irradiated target. Here, all frequency components of the reflected signal fluctuate harmoniously, and we have its usual presentation in the form of a normal honstationary random process with correlation function (7.2.1). 100

For signals presented in the form of (7.9.1) it is possible to obtain a nation of results pertaining to the case of nonharmoniously fluctuating frequency coup be ents. It should be noted that presentation (7.9.1) is possible for a very broad class of signals. In particular, these can be periodic signals or signals of any form, considered in a finite time interval and represented in this interval by \sim Fourier series. Therefore, results obtained for a signal of form (7.9.1), are very gereral.

Nonharmonious fluctuations of separate spectral components of the refluctions signal are caused by target extent and start to appear when the wavelength of the difference frequency for the considered components becomes comparable with dimensions of the target. If the considered frequencies are sufficiently close, so that it is possible to disregard the difference of spectral properties of fluctuation at these frequencies, then, as it is easy to show, the correlation function of the signal reflected from an extended target, considered as a set of "brilliant" points, is recorded in the form

$$R(t_1, t_s; \tau) = P_o \rho(t_1 - t_s) \operatorname{Re} \sum_{j,k=1}^{m} A_j A^*_k \rho_{jk} e^{i\omega_j(t_s - \tau) - i\omega_k(t_s - \tau)}, \quad (7.9.5)$$

-274-

where $\rho(t)$ - correlation function of fluctuations, which was used in all the preceding cases;

- τ delay, determined by distance to a certain median point of target;
- P_c mean power of the reflected signal;
- $\rho_{jk} = \begin{array}{c} coefficient \ of \ mutual \ correlation \ of \ oscillations \ with \ frequencies \ \omega_{j} \end{array}$

$$P_{jk} = \int_{-\infty}^{\infty} \sigma(x) e^{i(\bullet_j - \bullet_k)x} dx, \qquad (7.9.6)$$

 $\sigma(x)$ - normalized density of distribution of reflectors making up the target with respect to distance ($\rho_{1,1} = 1$).

Functional $|Q(t, \tau)|^2$, determining the character of optimum operations and accuracy of range finding, is obtained for the given case by a certain generalization of results from the case of harmoniously fluctuating frequency components and has the following form (see (4.2.9) and (4.2.11)):

$$Q(t, \tau)|^{s} = \sum_{j,k=1}^{m} A_{j}A^{*}_{k} \int_{-\infty}^{t} \int_{-\infty}^{t} y(s_{1}) y(s_{2}) e^{i\omega_{j}(s_{1}-\tau)-i\omega_{k}(s_{2}-\tau)} \times \\ \times h_{jk}(t-s_{1}) h'_{jk}(t-s_{2}) ds_{1}ds_{2}, \qquad (7.9.7)$$

where $h_{jk}(t)$ and $h'_{jk}(t)$ - pulse responses of filters, frequency responses of which satisfy relationship

$$H_{jk}(i\omega)H'^{*}_{jk}(i\omega) = V_{jk}(\omega), \qquad (7.9.8)$$

and functions $V_{ik}(\omega)$ are determined from the following equation

$$\sum_{l=1}^{m} V_{jl}(\mathbf{o}) \left[\partial_{lk} + h S_{\mathbf{o}}(\mathbf{o}) | A_{l} |^{\mathbf{a}} \rho_{lk} \right] = h S_{\mathbf{q}}(\mathbf{o}) \rho_{jk}, \qquad (7.9.9)$$

during derivation we use the practically always realized assumption that the difference of any two frequencies if great as compared to the width of the spectrum of fluctuations.

The discriminator of a range finder, carrying out exact or approximate formation of the derivative of $|Q(t, \tau)|^2$ with respect to τ , can be realized by any of the previously considered methods. In particular, with exact fulfillment of optimum operations the output of the discriminator is defined as

$$z(t, \Delta) = \frac{1}{N_{0}} \operatorname{Re} \sum_{j,k=1}^{m} i(\omega_{j} - \omega_{k}) A_{j} A^{*}_{k} \int_{-\infty}^{t} \int_{-\infty}^{t} h_{jk} (t - s_{i}) \times \\ \times h'_{jk} (t - s_{s}) e^{i\omega_{j}(s_{i} - 1) - i\omega_{k}(s_{0} - 1)} y(s_{i}) y(s_{s}) ds_{i} ds_{s}$$

$$(7, 9, 10)$$

and can be realized by a complicated set of filters and phase detectors, where characteristic of the filters for all pairs of frequencies are different. The equivalent spectral density in the optimum case, determining potential accuracy of range finding is found from (7.2.14) and is determined by expression

$$\frac{1}{S_{\text{our}}} = K_{\text{our}} =$$

$$= \hbar \operatorname{Re} \sum_{j,k=1}^{m} (\bullet_{j} - \bullet_{k})^{2} |A_{j}A^{*}_{k}|^{2} \rho^{*}_{jk} \frac{1}{2\pi} \int_{-\infty}^{\infty} V_{jk}(\omega) S_{\bullet}(\omega) d\omega. \qquad (7.9.11)$$

With harmonious fluctuations ($\rho_{jk} = 1$ for all j, k) (7.9.10)-(7.9.11) pass into the corresponding formulas of § 7.2 with coefficient b - a², determined by formula (7.9.2). With independently fluctuating signals ($\rho_{jk} = 0$ when $j \neq k$) $K_{OHT} = 0$ and $S_{3KB} = \infty$, i.e., range finding becomes impossible. Since squares of differences of frequencies in (7.9.11) enter into a sum with factors, proportional to coefficients of mutual correlation ρ^*_{jk} , those components for which $\rho_{jk} = 0$ drop out, and the presence of corresponding pairs of frequencies in no way affects accuracy of measurement of distance to a target.

The simplest results are obtained for a low signal-to-noise ratio h. In this case, as follows from (7.9.9),

$$V_{jk}(\omega) \approx h \rho_{jk} S_{\alpha}(\omega), \qquad (7.9.12)$$

and frequency responses of all filters turn out to be identical with an accorregation



Fig. 7.36. Block diagram of a discriminator with multifrequency radiation: 1 - mixer; 2 - narrow-band filter; 3 - phase detector; $4 - \text{amplifier with gain factor } \rho_{jk}$; 5 - adder. $s_j = \text{Re} A_j e^{i(m_j + m_{kp} + m_k)(t-\tau)}$.

factor $\sqrt{\rho_{jk}}$. One possible block diagram of such a discriminator is shown in Fig. 7.36 for the case when $\sigma(x)$ is an even function and ρ_{jk} are real numbers. The received signal enters m mixers, where it is mixed with radiated oscillations, delayed and shifted in frequency. After filtration oscillations are

-276-

fed to $\frac{m(m-1)}{2}$ phase detectors, outputs of which after multiplication by ρ_{jk} are summed and form the output voltage of the discriminator.

Formula (7.9.11) for h << 1 will be transformed to

$$\frac{1}{S_{ont}} = \frac{h^{a}}{2\pi} \int_{-\infty}^{\infty} S_{a}^{2}(\mathbf{o}) d\mathbf{o} \cdot \sum_{j,k=1}^{m} (\omega_{j} - \omega_{k})^{a} |\rho_{jk}|^{a} |A_{j}A_{k}|^{a}.$$
(7.9.13)

Coefficient of correlation ρ_{jk} , as follows from (7.9.6), depends on difference $|\omega_j - \omega_k|$, decreasing with increase of this difference. If ρ_{jk} decreases faster than $1/|\omega_j - \omega_k|$, there exists an optimum value of difference $|\omega_j - \omega_k|$, at which product $(\omega_j - \omega_k)^2 \rho_{jk}$ is maximum. We can find the optimum form of spectrum, ensuring the greatest accuracy of range finding. It is obvious that the best result will be obtained if all differences of frequencies have optimum value, but in this case there can be only two frequencies. Power should be distributed equally between these frequencies, as one may see from (7.9.13).

The obtained result cannot, of course, serve as a basis for selection of the shape of the sounding signal, since here accuracy of measurement presents the latter with requirements of uniqueness and resolving power with respect to distance. How-ever, this result permits us to conclude that the width of the spectrum of modulation of the signal should be taken close to $2\pi c/l$, where l is the linear dimensions of the target, and c is the speed of light.

§ 7.10. Analysis of Accuracy of Radar Range Finders

The above analysis of discriminators of various types and general results of Chapter VI permit us to investigate accuracy of range finders as a whole. As it was shown in Chapter VI, with linearized consideration and disregard of parametric fluctuations, total error of a tracking meter can be broken down into three components - fluctuation, dynamic and systematic. Calculation of the dependence of spectral density of noise at the discriminator output on mismatch leads to additional error, owing its origin to parametric fluctuations. Corresponding expressions for errors of measurement, which we shall use subsequently, are given in § 2, Chapter VI. In this section, from a series of examples of smoothing circuits and discriminators we shall calculate errors of range finding giving quantitative examples.

> 7.10.1. Influence of a System of Automatic Gain Control (AGC) on Discriminator Characteristics

The presence of an AGC system in general does not lead to essential change of our analysis of a discriminator. As it was shown in Chapter II, with good

-277-

approximation an AGC system can be described as a linear inertial amplifier of the received signal envelope with parameters which depend on signal level. basically, its action reduces to normalization of the output signal of the preamplifier, covered by an AGC loop. Such an amplifier always is present in circuits of discriminators of distance and usually precedes the narrow-band filter. Its band width Δf_y , so a rule, substantially exceeds bandwidth of fluctuations of the signal ($\Delta r_y = (5-20)\Delta f_c$) and at the same time is small as compared to spectrum width of modulation.

In two-channel discriminator circuits amplifiers are used in both channels, but the AGC system is common and works from the output of one of the amplifiers. Another factor which must be accounted for in circuits with automatic gain conducis demodulation of amplitude fluctuations of the signal, leading in our equivalent consideration to a steep slope of low frequencies in the signal amplitude envelope. This factor leads to decrease of parametric fluctuations at the discriminator output and decreases errors of measurement for small noises. Furthermore, the Add system, in general, produces modulation of the signal by noises, which pass through a fee iback circuit. This phenomenon significantly increases the difficulty of analysis. However, with a demodulation band which is not too broad (AdC is not too high-speed), they can be ignored.

Although all these factors, in practice, do not influence discriminator characteristics, during the analysis of dynamic properties of a closed-loop measuring system they must be accounted for inasmuch as, due to the presence of an Add system, its parameters (gain factor of the open loop and its fluctuations caused by parametric noises, the band of the closed system, etc) become functions of the signal-to-noise ratio and vary together with changes of it. In view of the great complexity in completely accounting for the influence of AGC (modulation of the signal by noise, cross modulation, nonlinear effects of high order, etc) and terms, to the fact that in normal circumstances these phenomena are weakly manifested, we shall limit ourselves to a very simple description of the AGC system and shall henceforth consider only changes of discriminator gain factor due to the normalizing action of AGC and decrease of the level of parametric fluctuations.

In accordance with the results of Chapter II for sufficiently great inertia of the filter in the feedback network the AGC system reduces the mean value of voltage at the output of the amplifier covered by the AGC loop to a constant level. As a result the gain factor of the receiver, and consequently, the discriminator gain factor, Vary with change of the mean input signal level. The law of change of the

-278-

gain factor is determined from conditions of normalization of the output voltage, which for receivers with a linear and a square-law detector, respectively, have the following form:

$$K_{i} E(t) = \text{const},$$

$$K_{i}^{2} \overline{E^{*}(t)} = \text{const},$$
(7.10.1)

where $K_1 - gain factor of the amplifier covered by the AGC loop;$ E(t) - amplitude of input signal.

Since

$$\overline{E(t)} = \sqrt{\frac{\pi}{2}(2N_{\bullet}\Delta f_{y} + P_{\bullet})},$$

$$\overline{E^{*}(t)} = 2(2N_{\bullet}\Delta f_{y} + P_{\bullet}),$$
(7.10.2)

from either of these relationships it follows that

$$K_{1} = K_{10} \frac{1}{\sqrt{1 + \frac{1}{k} \frac{\Delta f_{\gamma}}{\Delta f_{\bullet}}}}, \qquad (7.10.3)$$

where $K_{1,0}$ - gain factor in the absence of noise.

In the considered circuits with square-law detectors or phase detectors, realizing multiplication of signals, the discriminator gain factor is proportional to K_1^2 . Therefore,

$$K_{\rm A} = K_{\rm A0} \frac{1}{1 + \frac{1}{h} \frac{\Delta f_{\rm Y}}{\Delta f_{\rm 0}}} = K_{\rm A0} \frac{1}{1 + \frac{g}{h}}, \qquad (7.10.4)$$

where $y = \Delta f_y / \Delta f_c$ - ratio of amplifier bandwidth to signal bandwidth; K_{IO} - nominal value of discriminator gain factor, corresponding to the case of no noise.

The influence of all other discriminator parameters on gain factor, studied in the preceding paragraphs, is unchanged; relationship (7.10.4) should be understood in the sense that the gain factor of the open system depends on the signal-to-noise ration h according to the law (7.10.4), decreasing with strong noises and approaching a limit with increase of the signal. Expressions (7.10.3) and (7.10.4) are valid, of course, only for sufficiently large values of h, since for small signal levels it does not reach the delay level of AGC, the AGC system is opened and produces no normalization. Thus, for h smaller than a certain value h_0 , corresponding to the delay level, $K_1 = \text{const}$ and does not depend on the signal-to-noise ration.

-279-

Decrease of parametric fluctuations, caused by the demodulating effect of AGC. can be calculated with the help of results of Chapter II. As is clear from analysis of discrimators, parametric fluctuations are caused by random variations of the envelope of the useful signal, the spectral density of which in the low-frequency range at the output of an amplifier with AGC are determined by expression (2.7.23). Considering that the spectral density of parametric fluctuations is proportional to the spectral density of the signal amplitude envelope in the low-frequency range, from expression (2.7.23) we obtain the relationship between spectral densities of parametric fluctuations with $(S_{RBP}^{(1)})$ and without (S_{RBP}) AGC

$$\mathbf{S}_{\mathbf{n}\mathbf{a}\mathbf{p}}^{(1)} = \frac{1}{n^2} S_{\mathbf{n}\mathbf{a}\mathbf{p}}, \qquad (7.12).$$

where n^2 - coefficient depending on parameters of AGC and the signal-to-noise ratio. According to (2.7.23) this coefficient is presented in the form

$$n^{9} = n_{0}^{2} \frac{h}{h_{0}},$$
 (7.10)

where h_0 - the signal-to-noise ratio which corresponds to the delay level $E(t) = E_0$; $u_0 = bk_1 E_0$ (b - slope of the controlled characteristic, $k_1 = rain$ is tor in the reedback network of AGC).

The value of n_0 in practice is normally great as compared with unity, and thus even for signals close to the delay level parametric fluctuations caused by change of signal amplitude substantially decrease.

7.10.2. Smoothing Circuits with Constant Parameters

Until recently, in smoothing circuits of radar range finders they almost exclusively used linear filters with constant parameters of a fairly low level. The most widely applied types of such filters are the single integrator and the double integrator with correction. Furthermore, when, for a number of reasons, it was impossible to obtain ideal integration, they use smoothing filters in the form of a single RC-circuit and two series-coupled RC-circuits with correction.

According to (6.2.20) fluctuation error of measurement in smoothing circuits with constant parameters is determined by the simple relationship

$$d_{\phi x}^{\dagger} = 2S_{\phi x a} \Delta f_{\phi \phi}, \qquad (7.10.7)$$

where S_{3KB} - determined by the previous analysis of discriminators;

 $\Delta f_{\partial \phi}$ - effective band width of the closed-loop servo system, determined by expression (6.2.19).

For smoothing circuits in the form of two RC-circuits with correction the

-280-

transfer function of the closed circuit is

$$H(p) = \frac{K(1+pT_{i})}{(1+pT_{i})(1+pT_{i})},$$
(7.10.8)

where

K - gain factor of the smoothing filter;

 T_1 , T_2 - time constants of the RC-circuits;

 T_3 - time constant of the correcting circuit. The effective band width in this case is

$$\Delta f_{00} = \frac{1}{2\pi} \int_{0}^{\infty} \left| \frac{K_{A}H(i\omega)}{1+K_{A}H(i\omega)} \right|^{2} d\omega =$$

$$= \frac{1 + \frac{K_{A}KT_{3}^{2}}{T_{1}T_{3}}}{4\left[T_{0} + \frac{T_{1}+T_{3}}{K_{A}K}\right]} \approx \frac{K_{A}KT_{3}}{4T_{1}T_{3}}.$$
(7.10.9)

This approximate equality is valid for normal relationships of parameters of the servo system, since the gain factor of the open system $K_{\mu}K$ usually is very great. Quantities in expression (7.10.8) have the following order:

 $K_{\rm m}K \sim 10^3 + 10^5$, $T_1T_2 \sim 10 \div 100$ see,

where one of the time constants is larger than the other by a factor of 3-10, and $T_3 \sim 10^{-2}$ to 10^{-1} sec. Selection of the time constant of the correcting circuit is sometimes carried out in such a manner that for assigned $K_{\rm H}K$, $T_{\rm 1}T_{\rm 2}$ it ensures a minimum effective bandwidth

Here,

$$T_{a} \approx \sqrt{\frac{T_{1}T_{a}}{2K_{A}K}},$$
$$\Delta_{i}^{t} = \frac{3\sqrt{2K_{A}K}}{8\sqrt{7}, T_{a}}.$$

This selection is produced at the nominal gain factor of the open circuit, i.e., when $K_{\pi} = K_{\pi 0}$. Then the dependence of effective bandwidth on the signal-to-noise ratio approximately has the form

$$\Delta f_{0.0} = \Delta f_{0.0} \frac{2}{3} \frac{1 + \frac{0.5n}{h+h}}{1 + \left(1 + \frac{\mu}{h}\right) \sqrt{\frac{2(7_1 + 7_3)^2}{K_{A0}KT_1T_1}}},$$
(7.10.10)

where $\Delta f_{\partial D} = effective bandwidth with the nominal gain factor.$

-281-

The dependence of $\Delta f_{3\phi} / \Delta f_{3\phi 0}$ on h with y = 3, $T_1 = 4T_2$ and for various values of product $\Delta f_{3\phi 0} T_2$ is shown in Fig. 7.37. With noises which are not too weak and



sufficiently large $\Delta f_{2\Phi} O_{2}^{T}$ this dependence is not very obsential.

Formula (7.10.9) also permits us to find effective bandwidth for other types of smoothing filters, enumerated above. In particular, ton a double integrator with correction, i.e., when

the expression for $\Delta f_{-\partial d c}$ is obtained

$$H(p) = \frac{K_{\pi}(1 + pT_{\kappa})}{p^{3}},$$
 (7.10.11)

Fig. 7.37. Dependence of effective bandwidth of the servo system on the signalto-noise ratio h.

by driving T_1 and T_2 to infinity in such a manner that $K/T_1T_2 = K_R$, where $K_R = gain factor (measured) of the two integrators. Considering also that <math>T_3 = T_R$, we obtain

$$\Delta f_{ie\phi} = \frac{1 + K_{\rm R} K_{\rm R} T_{\rm R}^2}{4T_{\rm R}}.$$
 (7.10.12)

Minimum Δf_{ab} is attained when

$$T_{\mathbf{x}} = \frac{1}{\sqrt[4]{K_{\mathbf{x}}K_{\mathbf{x}}}}$$
(7.10.13)

and is equal to $\sqrt{K_{\chi}} - K_{\chi}/2$. The dependence of effective bandwidth on the signal-tonoise ratio with selection of T_{χ} in accordance with (7.10.13) for $K_{\chi} = K_{\chi()}$ has the form:

$$\Delta f_{00} = \frac{\Delta f_{000}}{2} \left(1 + \frac{h}{h+y} \right). \qquad (7.10.14)$$

For illustration in Fig. 7.38 there is shown the dependence of fluctuation error of measurement of delay, referred to pulse duration, on the signal-to-noise ratio h for a range finder with modulation of the signal by square pulses and a discriminator with two detuned channels for H(p) from (7.10.11), 5 = 0.5, $\tau_{\rm R}$ and for different values of y and $\Delta f_{\rm perc}/\Delta f_{\rm C}$. Since effective bandwidth $\Delta f_{\rm perc}$ depends on h, the path of curves in Fig. 7.38 differs from dependence $S_{\rm OKE}(h)$. Bands of filters of the discriminator in accordance with (7.6.3) are assumed to be matched with signal spectrum width.

-282-



Fig. 7.38. Dependence of fluctuation error of measurement on the signal-to-noise ratio: $-1.T_{0} = 10^{-1}$;

--- $\ln_{so} o / \Delta f = 10^{-2}$.

By means of further simplifications from formula (7.10.9) we obtain the expression for effective bandwidth of a system with smoothing circuits in the form of one RC-filter

$$H(p) = \frac{K}{1 + \rho T_1}, \qquad (7.10.15)$$

and a single integrator

$$H(p) = \frac{K_{\pm}}{p}$$
. (7.10.16)

Corresponding expressions have form

$$\Delta f_{\bullet \Phi} = \frac{KK_{\star}}{4T_{\bullet}}, \qquad (7.10.17)$$

$$\Delta f_{\bullet \bullet} = \frac{K_{\bullet}K_{\star}}{4}.$$
 (7.10.18)

In both cases the dependence of $\Delta t_{g\bar{q}}$, on the signal-to-noise ratio has the form

$$\Delta f_{\bullet \bullet} = \Delta f_{\bullet \bullet}, \frac{\hbar}{h+y}. \tag{7.10.19}$$

This dependence is more sharply expressed.

As another example, in Fig. 7.39 we show the dependence of $\sigma_{\phi\pi}^2 (b - a^2) \frac{\Delta \Gamma_c}{\Delta f_{a\phi}}$ on h for an optimum discriminator with signal spectrum (7.3.11), various y and



error of measurement on the signal-tonoise ratio. or (7.10.16).

We shall now consider other components of error of measurement. In tracking meters there exists double interpretation of dynamic error of tracking. In the statistical approach to measured quantities by dynamic error, in accordance with Chapter VL, we understand the mean square of the component of error caused by random changes of measured parameter. Its magnitude is determined

smoothing circuits of form (7.10.15)

by formula (6.2.14). Along with this in such a consideration there is introduced

-283-

the concept of systematic error, which is caused by inaccuracies of input of the mean value of the measured parameter, known a priori. Let us remember that in an optimum measuring system such insertion is obligatory, and the tracking loop is designed to process a priori unknown random changes of the parameter. The rationale for such input is obvious, too, for a normal tracking systems. The magnitude of systematic error is determined by formula (6.2.14).

In the absence of statistical evidence about the measured quantity determination of dynamic error in accordance with Chapter VI is impossible. In this case dynamic error is determined by the widely known method as the error of processing a certain model disturbance - an input varying linearly, by square-law, etc. Formal mathematical determination of such error, obviously, coincides with the expression for systematic error (6.2.14) with replacement of $\Delta\lambda(t)$ by the corresponding disturbance. Therefore, calculation of systematic error for various $\Delta\lambda(t)$ gives simultaneously the magnitude of dynamic error corresponding to the nonstatistical approach to the measured parameter.

Let us consider several examples of calculation of dynamic error in a system with smoothing circuits of form (7.10.8), (7.10.11), (7.10.15), and (7.10.16). If the random part of the measured distance changes in time as a stationary random process with spectral density $S(\omega)$, the stationary value of dynamic error, on the basis of (6.2.17), is

$$\mathbf{s}_{A}^{s} = \frac{1}{2\pi} \int_{-\infty}^{\infty} \frac{S(\omega) \, d\omega}{\left[1 + K_{A} \, H(\omega)\right]^{1}}.$$
 (7.10.20

Stationary random changes of distance correspond to cases when the target carries out random maneuvers or experiences certain irregular disturbances, possessing a stationary character. Examples of such disturbances may be random oscillations of engine thrust, change of drag due to irregularity of the atmosphere, noises in drive assemblies of control systems of the target, and so forth. In the presence of sufficient damping such disturbances lead to stationary change of distance. If, in particular, spectral density $S(\omega)$ has the form

$$S(\mathbf{o}) = \frac{3n_0^2 T}{1 + \mathbf{o}^2 T^2},$$
 (7.10.21)

where σ_0^2 - variance of the random component of the measured distance; T - correlation time, and the smoothing filter is a single integrator; dynamic error is equal to

-284-

$$\sigma_{A^{\text{PFT}}}^{2} = \frac{\sigma_{0}^{2}}{1 + K_{X}K_{X}T} = \frac{\sigma_{0}^{2}}{1 + 4\Delta_{1}^{2} + \Phi^{T}}.$$
 (7.10.22)

This error will decrease with increase of product $K_{I\!I}K_{I\!I}$ T, i.e., with broadening of the effective bandwidth of the system. With decrease of gain factor due to lowering of the signal-to-noise ratio dynamic error increases, approaching a limiting value, equal to the a priori variance of the measured quantity σ_0^2 .

Likewise, with smoothing filter (7.10.11) and selection of $T_{\rm R}$ in accordance with (7.10.13)

$$\sigma_{ABM}^{2} = \frac{\sigma_{0}^{2} [1 + (K_{X} K_{B} T^{0})^{3/2}]}{1 + K_{X} K_{B} T^{0} + K_{A}^{2} K_{B}^{2} T^{0}} = \sigma_{0}^{2} \frac{1 + 8\Delta f_{Sb}^{3} T^{0}}{1 + 4\Delta f_{Sb}^{2} T^{0} + 16\Delta f_{Sb}^{4} T^{0}}.$$
 (7.10.23)

In this case the magnitude of dynamic error also is determined by product $\Delta f_{gamma}T$. The dependence of σ_{MMH}/σ_0 on this product is shown in Fig. 7.40. The



dependence of dynamic error on the signal-to-noise ratio can be estimated by formulas (7.10.14) and (7.10.23).

Of essential interest is the case when the measured distance changes as a linear combination of known functions with random coefficients of the form

$$d(l) = \sum_{i=1}^{m} \mu_i f_i(l) + \overline{d(l)}, \quad (7.10.24)$$

Fig. 7.40. Dependence of dynamic error on effective bandwidth.

where $\overline{\mu_{i}} = 0$; $\overline{\mu_{i}\mu_{k}} = M_{ik}$; $\overline{d(t)} - mean$ value of distance.

Change of distance according to (7.10.24) is realized when the law of motion of the target is known with an accuracy of certain constant parameters, i.e.,

$$d(t) = F(t; a_1, \ldots, a_m) = F(t; \overline{a_1 + \mu_1}, \ldots, \overline{a_m + \mu_m}),$$

where α_i (i = 1, ..., m) - unknown parameters with certain mean values $\overline{\alpha_i}$ and random deviations from them μ_i .

In the overwhelming majority of practically interesting cases these deviations are sufficiently small to permit presenting d(t) in the form of (7.10.24), where

$$\overline{d(t)} = F(t; \overline{a_1}, ..., \overline{a_m})$$
, and $f_1(t) = \frac{\partial F(t; \overline{a_1}, ..., \overline{a_m})}{\partial_{\alpha i}}$. Parameters a_1 can have

various meaning. With change of d(t) according to (7.10.24)

$$\mathbf{z}_{am}^{a} = \sum_{i, h=1}^{m} M_{ih} \int_{\tau_{0}}^{t} v(t, \tau) f_{i}(\tau) d\tau \int_{t_{0}}^{t} v(t, \tau) f_{h}(\tau) d\tau = \sum_{i, h=1}^{m} M_{ih} \mathbf{e}_{i}(t) \mathbf{e}_{h}(t), \qquad (7.10.25)$$

where $v(t, \tau)$ - pulse response, corresponding to the transfer function of error, which is determined by equation (6.2.13).

With smoothing circuits with constant parameters the Fourier transform from $v(t - \tau)$ is determined by the expression

$$V(i_{\bullet}) = \frac{1}{1 + K_{\lambda} H(i_{\bullet})}.$$
 (7.10.26)

Expression (7.10.25) when $M_{ik} = 1$ simultaneously determines the square of dynamic error, corresponding to the nonstatistical approach, which takes place with influence on the servo system of a disturbance assigned by relationship (7.10.24) when $\mu_i = 1$ (i = 1, 2, ...), and also the square of systematic error, corresponding to the same difference between the real and introduced mean values of the measured distance.

With smoothing circuits with constants parameters and $t_0 = 0$, function $\varepsilon_j(t)$, obviously, is determined by relationship

$$\bullet_{j}(l) = \frac{1}{2\pi l} \int_{Q-lim}^{Q+lim} V(p) F_{j}(p) e^{pt} dp, \qquad (7.10.27)$$

where $F_{i}(p)$ is the Laplace transform of function $f_{i}(t)$.

From formulas (7.10.26) and (7.10.27) it follows that with a smoothing filter in the form of a single integrator (7.10.16) function $\epsilon_j(t)$ asymptotically for sufficiently large values of t can be approximately presented in the form

•
$$_{j}(t) = \frac{\dot{f}_{j}(t)}{K_{A}K_{A}},$$
 (7.10.28)

where $f_j(t)$ - derivative of $f_j(t)$.

This expression shows that stationary error in this case exists only if all $f_j(t)$ are functions with a bounded derivative. In particular, if $f_1(t) = a_0 + a_1 t$, $f_1(t) = 0$, j > 1, relationship (7.10.28) as $t \rightarrow \infty$ is exact and

$$e_{\text{ABR}} = \lim_{t \to \infty} e_1(t) = \frac{a_1}{K_A K_B}.$$
 (7.10.29)

Dynamic error, determined by formula (7.10.29), is used most frequently in practice for evaluating the servo system.

Likewise, in a system with two integrators with correction function $\epsilon_j(t)$ approximately is presented in the form

$$j(l) \approx \frac{\tilde{l}_{j}(l)}{K_{\rm a}K_{\rm a}},$$
 (7.10.30)

and, in particular, when $f_1(t) = a_0 + a_1 t + \frac{1}{2}a_2 t^2$ we obtain exact expression

$$e_{\text{ABE}} = \lim_{t \to \infty} e_j(t) = \frac{a_2}{K_R K_R},$$
 (7.10.31)

and there is no error from a linear disturbance in steady-state operating conditions. If the measured distance has components, growing faster than t in the first case, and faster than t^2 in the second, as $t \rightarrow \infty$ dynamic error increases without limit. With smoothing circuits in the form of a single RC-filter and two filters with correction, a stationary value of error exists only with a constant disturbance at the input. In both cases it is equal to

$$R_{A} = \frac{a_0}{1 + K_A K}.$$
 (7.10.32)

Relationships (7.10.28)-(7.10.32) show that in general the magnitude of dynamic error is inversely proportional to the gain factor of the open circuit of the servo



Fig. 7.41. Dependence of the gain factor of the discriminator on the signal-to-noise ratio.

system and increases with decrease of gain factor of the discriminator due to decrease of the **signal**-tonoise ratio. In Fig. 7.41 there is shown the dependence of $K_{\rm H}/K_{\rm HO}$ on h, characterizing change of dynamic error with change of the **signal**-tonoise ratio. The magnitude of error in the range of working values of h, depending upon selection of the magnitude of y, may vary by a factor of 1.2 to 10. With wider

bands of the amplifier with AGC the influence of normalizing properties of the AGC

THE PAST AND A DESCRIPTION OF THE PAST OF

system, naturally, are more noticeable.

Increase of the gain factor of the open loop has a different effect on magnitudes of fluctuation and dynamic error. Therefore, when designing smoothing circuits of a range finder there is required a definite compromise selection of their parameters. Let us consider for example the case of a double integrator with correction. Since the time constant of the correcting circuit does not affect the magnitude of dynamic error, it should be selected in accordance with (7.10.13), i.e., from the condition of minimum effective bandwidth. Then total error of measurement with a distance varying in time by square law will have the form

$$\sigma^{2} = \sigma_{\phi A}^{2} + \sigma_{\mu \mu \mu}^{2} = \sqrt{K_{\mu} K_{\mu}} S_{\sigma K \sigma} + \frac{\sigma_{\pi}^{2}}{(K_{\pi} K_{A})^{2}}$$
(7.10.33)

If mean square acceleration a_2^2 is unknown, it can be replaced by a certain quantity given by tactical conditions. From (7.10.33) it follows that the optimum magnitude of gain is

$$(K_{\rm m}K_{\rm g})_{\rm outr} = \left(\frac{4\sigma_{\rm g}^2}{S_{\rm outr}}\right)^{2/5}$$
 (7.10.34)

and the second second

Total error here comprises

$$\sigma^{3} = 1,68 \left(\overline{a_{2}^{3}}\right)^{1/5} S_{MS}^{4/5},$$
 (7.10.35)

where fluctuation error is approximately 4 times greater than dynamic. Inasmuch as S_{3KB} depends on h, the gain in the open circuit, according to (7.10.34), should vary with change of the signal-to-noise ratio. In particular, for optimum or nearly-optimum discriminators and a square spectrum of fluctuations of the signal, when S_{3KB} is proportional to $\frac{1+h}{h^2}$, the required law of change has the form

$$(K_{\rm B}K_{\rm A})_{\rm OBT} = K_0 \left(\frac{h^1}{1+h}\right)^{2/5}$$
. (7.10.36)

By proper selection of the amplifier band and parameters of the AGC system it is possible to achieve automatic approximation of this dependence in a sufficiently wide range of values of h.

In Fig. 7.42 is the dependence of the optimum gain factor on h and curves of change of gain factor for various y. All these curves are joined for point h = 10. From this figure it is clear that with selection y = 20 the required dependence of the gain factor of an open circuit with satisfactory accuracy is reproduced by an

-288-



Fig. 7.42. Dependence of the optimum gain factor of an open system on the signal-to-noise ratio: ---- required dependence; ---- real dependence for various y.

AGC system in the range from $h \sim 3$ to $h \sim 200$.

For illustration of formulas (7.10.34)-(7.10.35) in Table 7.1 we give errors of range finding with a continuous signal frequency modulated by sinusoidal law with deviation $2\omega_{\rm m} = 2\pi \cdot 10^6$ rad/sec for a range finder with a two-channel discriminator matched with spectrum (7.3.11) by a filter. Spectrum width $\Delta f_{\rm c}$ is taken equal to 20 cps. Values

of errors correspond to various signal-to-noise ratios and different mean square accelerations developed by the target. In that same table there are given values of optimum gain factor of an open circuit.

Monn square acceleration of the target	e, 11	K _{II} K _Y sec ⁻²	7 , N	^к д ^к и _{seo} -2	•, #	^К Д ^К И; sec ⁻²	σ, <u>μ</u>	^К Д ^К И _{sec} -2
	h.	=3	h =	10	۸ =	: 30	h =	: 100
0,1 1,0 3,0 10 30	4,6 7,3 9,2 11,5 14,5	0,05 0,31 0,74 2,0 4,72	2,5 4 5 6,3 8	0,1 0,57 1,37 3,6 8,55	1,4 2,2 2,8 3,5 4,5	0,16 1,05 2,44 6,5 15,2	0,9 1,3 1,7 2,1 2,7	0,25 1,75 4 10,8 25,4

Table 7.1

The presence of parametric fluctuations in the servo system leads to increase of errors of measurement. With constant smoothing circuits this increase is expressed by simple relationship (6.2.39), proceeding from which by (7.10.5) and previously obtained formulas for spectral density of parametric fluctuations (7.3.20), (7.3.21) and (7.4.14) we can be easily estimate the influence of these fluctuations. In the most interesting case of fairly large signal-to-noise ratios and for circuits of discriminators with detuned channels or differentiation of the reference signal, increase of error due to parametric fluctuations is determined by the following simple expression:

$$\sigma^{2} = \sigma_{0}^{2} \left(1 + \frac{2\pi \Delta f_{00}}{n^{2} \Delta f_{0}} \right), \qquad (7.10.37)$$

where

 σ_0^2 - variance of error;

 $\Delta f_{\vartheta \varphi}$ — bandwidth of the system, calculated without taking into account parametric fluctuations;

 $\alpha = 0.5-1.0 - numerical coefficient (see Paragraph 7.3.2);$

n - coefficient of suppression of fluctuations by the AGC system. In cases of practical interest ratio $\Delta f_{\partial \Phi} / \Delta f_{\rm C}$ has an order of 10^{-3} to 1; therefore, already at n ~ 5-10 error is increased by no more than 10%, and in most cases this increase is, in general, immaterial.

7.10.3. Optimum Smoothing Circuits for a Randomly Varying Parameter

In ChapterVI it is shown that when the measured parameter is a nondegenerate random process, stationary or with stationary increments, the smoothing filter is a filter with constant parameters, depending on statistical properties of the measured coordinate, the signal-to-noise ratio and parameters of modulation of the received signal. Let us consider one example of such a process. Let us assume the the radar range finder is designed for measurement of distance to a target moving with random uncorrelated accelerations with spectral density B_2 (quantity B_2 is numerically equal to the mean square of the speed developed by the target in 1 sec and has dimension $[m^2/sec^3]$). Here, the distance to the target is a random process in the form of the double integral of white noise of spectral density B_2 , and the optimum smoothing filter according to results of Chapter VI is a double integrator with correction, the transfer function of which is given by formula (7.10.11). Moreover, in the optimum circuit

$$K_{\rm R}K_{\rm R} = \sqrt{B_{\rm s}/S_{\rm OHT}(h)},$$

$$T_{\rm R} = \sqrt{4S_{\rm OHT}(h)/B_{\rm s}},$$

(7.10.38)

and total error of measurement is given by formula :

$$\sigma_{\text{ont}}^2 = \sqrt{2} (B_a S_{\text{ont}}^3(h))^{1/4}.$$
 (7.10.39)

From (7.10.38) it follows that quantities $K_{_{N}}$ and $T_{_{K}}$ should change with change of h. We shall consider that these quantities are fixed, and change of the transmission factor of the open circuit occurs only due to normalizing properties of AGC. Furthermore, we shall assume that the real spectral density of the target differs from that used during synthesis and is equal to B_{201}

The gain factor of the integrator and the time constant of the correcting circuit we select corresponding to some $h = h_0$. Then, total amplification in the open loop is equal to

$$K_{\rm m}K_{\rm m} = \sqrt{B_{\rm m}/S_{\rm out}(h_{\rm s})} \frac{1 + \frac{y}{h_{\rm s}}}{1 + \frac{y}{h_{\rm s}}},$$

(7.10.40)

where $y = \Delta f_y / \Delta f_c$, and the effective bandwidth, according to (7.10.12), is

$$\Delta f_{ab} = \frac{1}{4T_{a}(h_{a})} \left[1 + 2 \frac{1 + \frac{y}{h_{a}}}{1 + \frac{y}{h}} \right]. \qquad (7.10.41)$$

Then, variance of fluctuation error

$$S_{\phi,\pi}^{*} = \frac{S_{\phi,\pi}(h)}{2} \sqrt[4]{\frac{B_{\pi}}{4S_{\phi,\pi^{+}}(h_{\phi})}} \left[1 + 2\frac{1 + \frac{1}{h_{\phi}}}{1 + \frac{1}{h_{\phi}}} \right], \qquad (7.10.42)$$

and variance of dynamic error is obtained from (7.10.20) when $S(\omega) = B_{20}/\omega^4$,

$$\sigma_{\rm gm}^2 = \frac{B_{20}}{8} \left(\frac{4S_{\rm out}(h_0)}{B_2} \right)^{3/4} \left[\frac{1 + \frac{y}{h}}{1 + \frac{y}{h_0}} \right]^2.$$
(7.10.43)

We select $h_0 = 5$, when for an exponential function of correlation of fluctuations $B_2/S_{0\Pi T}(h_0) = 23.7B_2b\Delta f_c/c^2$, which for instance, when $B_2 = 10^2m^2/sec^3$, rms modulation spectrum width $\sqrt{b} = 2\pi \cdot 10^6 sec^{-1}$, and width of the spectrum of fluctuati.ns $\Delta f_c = 30$ cps, gives the following value of parameters of the servo system:

$$K_{\rm m} = 5.6 \frac{1}{{}_{\rm sec}^{-2}}, T_{\rm m} = 0.6 {}_{\rm sec}, \Delta f_{\rm b \phi}(h_{\rm o}) = 1.25 {}_{\rm cu},$$

and error of measurement for $h = h_0$ is about 3.5 m. The dependence of the ratio of total error $\sqrt{\sigma_{Q,I}^2 + \sigma_{Q,IH}^2}$ to quantity $\sigma_{O,IIT}$ from (7.10.39) for values of h_0 , B_2 , b, and Δf_c selected above, on h is shown for various values of B_{20} and two values of y(y = 20 and y = 100) in Fig. 7.43. In order to obtain an idea of the absolute values of error of measurement, in Table 7.2 there are given values of $\sigma_{O,IIT}$ for various h with the same values of parameters B_2 , b, and Δf_c .

	Tal	ble	-7	•2
--	-----	-----	----	----

h	0,3	1	3	10	30	100
• _{opt} , #	18	8,4	4,5	2,5	1,6	0,95

Figure 7.43 shows that in general the investigated radar range finder with a criterion for selection of its discriminator close to optimum sufficiently well approximates properties of an optimum system. With coincidence of assumed B_2 and

-291-

real B_{20} increase of error occurs practically only for the smallest values of h, which actually are not the working values. In the working range increase of error is several per cent.



If the real value of B_{20} is less than the assumed, in the considered circuit for certain values of h, depending on selection of quantity y, error is less than in an optimum circuit, calculated for spectral density of accelerations B_2 . This circumstance is not surprising, since measurement of distance of a target maneuvering less, naturally, should be produced with less error.

If, on the other hand, value of the spectral density of accelerations during synthesis is lowered, error

of measurement for all values of h is greater than σ_{ONT} ; however, in the working range of values of the signal-to-noise ratio this increase does not exceed a factor of 2-3. Selection of y, as comparison of curves in Fig. 7.43 shows, is not very critical. Best results in general are obtained for smaller y (in this case it is reasonable to select y approximately within the range from 10 to 30); however, if there is a danger that during synthesis the value of the spectral density of accelerations is understated and the range finder is designed, basically, for work at a comparatively high signal-to-noise ratio, it is somewhat better to increase y, expanding the bandwidth of the preemplifier.

7.10.4. Smoothing Circuits with Variable Parameters

In Chapter VI it is shown that in many cases requirements of optimality of a meter lead to the necessity of application of smoothing circuits with variable parameters. Most characteristic is the case when the measured quantity is presented in form (7.10.24). The pulse response of optimum smoothing circuits in this case is given by formula (6.8.53), and error of measurement is presented in the form of (6.8.54). The smoothing filter here is a set of variable-gain amplifiers, integrators and generators of known functions of time with controlled gain.

A characteristic peculiarity of a meter designed for a law of change of the

-292-

parameter of type (7.10.24) is the asymptotic tendency of error of measurement to zero, which is a consequence of the quasi-regularity of change of the measured quantity. Smoothing circuits of the meter are constructed here in such a way that the range finder, with the help of suppression of output signals of the discriminator, with the passage of time more and more shifts to work from memory, and change of the output quantity become smoother and smoother, ever more exact than change of the measured distance.

The pulse response of an optimum smoothing filter designed for measurement of distance which changes according to the law (7.10.24), according to (6.8.53), is determined in the following way:

$$G(t, \tau) = \sum_{i, k=1}^{m} A_{ik}(\tau) f_i(t) f_k(\tau), \qquad (7.10.44)$$

where $A_{ik}(t) = A_{ik}(t; h) - matrix elements;$

$$\mathbf{A}(t; h) = [\mathbf{M}^{-1} + \mathbf{U}(t; h)]^{-1}; \qquad (7.10.45)$$

$$U(t; h) = \left\| \frac{1}{S_{ouv}(h)} \int f_i(s) f_k(s) ds \right\|.$$
(7.10.46)

According to (7.10.44)-(7.10.46) parameters of the optimum smoothing filter are functions of the signal-to-noise ratio and should be corrected with change of it. Let us assume first that such correction is produced, and we also assume that the discriminator is sufficiently close to optimum, and change of its gain factor is by the law required in an optimum meter, $K_{II}(h) = 1/S_{OIIT}(h)$. Then, according to equation (6.8.53) pulse response of the closed-loop servo system is

$$g(t, \tau) = \frac{1}{S_{our}(h)} \sum_{l, h=1}^{m} A_{ih}(l) f_i(l) f_h(\tau), \qquad (7.10.47)$$

and pulse response of error, according to (6.2.13), is

$$v(t, r) = \delta(t - r) - g(t, r).$$
 (7.10.48)

If real changes of distance correspond to law (7.10.24), used during the synthesis of smoothing circuits, then, as it was shown in Chapter VI, total error of measurement is expressed by the simple dependence

$$\sigma^{*}(t) = G(t, t).$$
 (7.10.49)

Corresponding examples of calculation of errors of measurement of a general character were already considered in ChapterVI. Here we give one more concrete

-293-

example. Let us assume that the target moves evenly toward the radar, so that the distance to it is equal to

$$d(l) = d_0 + \Delta d - (V + \Delta V) t, \qquad (7.10.50)$$

where d_0 - distance at the time of tracking lock-on, known with an accuracy of error of lock-on Δd of the order of the magnitude of the resolution capability with respect to distance.

Target velocity V is also known with certain error ΔV , where errors Δd and ΔV are normal and uncorrelated. Designating $\Delta d^2 = \sigma_0^2$ and $\Delta V^2 = c_1^2$, we obtain, according to (6.8.63), for the pulse response of the smoothing circuit the expression

$$G(t, \tau) = \frac{\sigma_0^2 \left(1 - \frac{\sigma_1^2}{S_{out}(h)} \frac{\tau^2}{6}\right) + \sigma_1^2 t \left(1 + \frac{\sigma_0^2}{S_{out}(h)} \frac{\tau}{2}\right)}{1 + \frac{1}{S_{out}(h)} \left(\sigma_0^2 \tau + \sigma_1^2 \frac{\tau^2}{3}\right) + \frac{\sigma_0^2 \sigma_1^2 \tau^4}{12S_{out}^2(h)}},$$
(7.10.51)

which when $\tau = t$ also determines variance of total error of measurement. In formula (7.10.51) and subsequently we have in mind equivalent spectral density, recomputed to values of distance with coefficient $e^2/4$, where c = velocity of light. This variance changes in time both due to the obvious dependence of G(t, t) on t, and also due to change of the signal-to-noise ratio in time, which occurs according to the law

$$h(l) = h_0 \frac{1}{\left(1 - \frac{Vl}{d_0}\right)^4}.$$
 (7.10.52)

Considering the spectrum of signal fluctuations square, and using expression (7.2.15) for $S_{OUT}(h)$, we obtain the following expression, characterizing change of error of measurement in time:

$$\frac{\mathbf{q}^{2}(t)}{\mathbf{q}_{0}^{2}} = \frac{1 + \frac{h^{2}(t)}{1 + h(t)} \frac{8\Delta f_{0}t^{4}bc_{1}^{2}}{3c^{4}} + \frac{e_{1}^{2}t^{4}}{e_{0}^{2}}}{1 + \frac{h^{2}(t)}{1 + h(t)} \frac{8\Delta f_{0}tbu_{0}^{2}}{c^{4}} \left(1 + \frac{e_{1}^{2}t^{2}}{3e_{0}^{2}}\right) + \frac{16h^{4}(t)\Delta f_{c}^{2}u_{0}^{2}u_{1}^{2}t^{4}b^{4}}{3(1 + h(t))^{3}c^{4}}}.$$
(7.10.53)

The dependence of $\sigma(t)/\sigma_0$ on Δf_c t for various $x_0 = d_0 \Delta r_c / V = T_0 \Delta r_c$, where $T_0 - time$ to target impact, and various $y = \frac{\sigma_1}{\sigma_0 \Delta f_c} = \sqrt{\Delta V^2 / \Delta d^2 \Delta f_c^2}$, is shown in Fig. 7.44. The curves are constructed for the signal-to-noise ratio at maximum range $h_0 = 3$ and for product $\frac{4\omega\sigma_0^2}{c^2} = 0.5$.

and the second second



 $---y = 10^{-3}; ---y = 10^{-1}; ----y = 10.$

Curves of Fig. 7.44 show that error of measurement sufficiently rapidly decreases in time. Here, there first occurs decrease of error, then a certain increase of it, and then decrease to zero. Maximum error is attained faster, the larger y = $= \sigma_j / \sigma_0 \Delta f_c$, i.e., the greator the variance of velocity V, and is expressed more strongly, the larger the magnitude of x_0 , i.e., the smoother the change of the signalto-noise ratio. For small values of x_0 and y there is no maximum, and in general its magnitude is greater, the larger y. Here, for small y the magnitude of this maximum is less than one, and for large y it may exceed one; however, even at y = 10it doer not exceed two.

Ex: Gion (7.10.53) characterizes total error of measurement only when the mean value of measured distance is inserted in the circuit of the servo system. If such insertion is not specially anticipated and there is introduced only value d_0 , which occurs automatically upon lock-on, there appears additional error. Let us find the systematic error, appearing due to the difference between the real and inserted values of distance. Function $g(t, \tau)$ in this case, as it is simple to prove, is

$$g(l, s) = \frac{s_0^2 \left(1 - \frac{s_1^{2/2}}{6S_{our}}\right) + s_1^2 ls \left(1 + \frac{s_0^{2}}{2S_{our}}\right)}{1 + \frac{1}{S_{our}} \left(s_0^2 l + \frac{1}{3} s_1^2 l^2\right) + \frac{1}{12S_{our}^3} s_0^{2} s_1^{2/4}}.$$
(7.10.54)

Considering that difference $\Delta\lambda(t)$ in this case is equal to Vt, such the help of

(6.2.14) and (7.19.48) we obtain the following expression for systematic error:

$$\sigma_{GHOT} = Vt \frac{1 + \frac{e_0^2 t}{2S_{out}}}{1 + \frac{1}{S_{out}} \left(e_0^2 t + \frac{1}{3} e_1^2 t^2\right) + \frac{1}{12S_{out}^2} e_0^2 e_1^2 t^4}$$
(7.10.55)

This expression shows that systematic error for small t grows as Vt, and then rapidly decreases. Asymptotically, it changes as

and its maximum is attained for times t of the order of $1/\Delta f_{\rm C}$, i.e., practically



Fig. 7.45. Dependence of systematic error of range finding in a system with variable parameters, related to magnitude d_0 , on Δf_0 t:

$$---y = 10^{-3}; ---y = 10^{-1}; ----y = 10.$$

during the time equal to the inertia of the discriminator. The dependence of systematic error, related to magnitude d_0 , on time, with the same conditions as for Fig. 7.44, is shown in Fig. 7.45. This analysis shows that in this case the influence of systematic error is of little substance.

In practice sometimes it is not possible to correctly determine statistics of the magnitude to be determined and it is necessary to make more or less arbitrary assumptions. Here, it may be that smoothing circuits are calculated for one

law of change of the parameter, but in reality another takes place. Furthermore, it may be that the real law of change of parameter is so complicated that from considerations of technical convenience it is necessary consciously to replace it by a simpler one. The general case of measurement of a nonstationary parameter, varying according to the law

$$2(l) = \sum_{k=1}^{N} \alpha_{k} \varphi_{k}(l) + \overline{\lambda(l)}, \qquad (7.10.56)$$

-296-

where $\overline{\alpha_k} = 0$; $\overline{\alpha_1 \alpha_k} = A_{1k}$, by a system whose smoothing circuits are designed for law of change (7.10.24) is considered in Chapter VI. We shall limit ourselves to a simple example, when smoothing circuits are designed for law of change of distance (7.10.50) with variances σ_0^2 and σ_1^2 , and their pulse response has form (7.10.51), while, in fact, distance changes according to law

$$d(t) = d_0 + \Delta d_1 - (V + \Delta V_1)t + \frac{1}{2}\Delta at^2, \qquad (7.10.57)$$

where Δd_1 , ΔV_1 , Δa - independent normal random variables with variances $\overline{\Delta d_1^2} = \sigma_{01}^2$, $\overline{\Delta V_1^2} = \sigma_{11}^2$, $\overline{\Delta a^2} = \sigma_{21}^2$.

Fluctuation error remains the same as in the case of coincidence of the assumed and real laws, and is determined by expression (6.2.14)

$$\sigma_{\phi n} = S_{ont} \int g^{*}(t, \tau) d\tau,$$
 (7.10.58)

and its magnitude for sufficiently large t practically coincides with the magnitude of total error σ^2 of (7.10.49) and (7.10.53), calculated earlier.

Let us find dynamic error of range finding. According to (6.2.14)

$$S_{AHH}^{2} = \sum_{i=0}^{2} \frac{\sigma_{i1}^{2}}{(i!)^{s}} \left[t^{i} - S_{ont} \left[\frac{t}{g}(t, \tau) \tau^{i} d\tau \right]^{2} = \frac{\sigma_{0i}^{2} \left(1 - \frac{1}{6} \frac{\sigma_{i1}^{2} t^{s}}{S_{ont}} \right)^{2} + \sigma_{11}^{2} t^{s} \left(1 + \frac{\sigma_{0}^{2} t}{2S_{ont}} \right)^{2} + \frac{1}{1 + \frac{1}{S_{ont}}} \left(\sigma_{0}^{2} t + \frac{1}{3} \sigma_{1}^{2} t^{s} \right) + \frac{+ \frac{\sigma_{21}^{2} t^{4}}{4} \left(1 + \frac{\sigma_{11}^{2} t^{s}}{12S_{ont}} + \frac{\sigma_{0}^{2} \sigma_{11}^{2} t^{s}}{72S_{ont}^{2}} \right)^{2} + \frac{1}{12} \frac{\sigma_{0}^{2} \sigma_{11}^{2} t^{s}}{S_{ont}^{2}} \right]^{2}$$
(7.10.59)

From this expression it follows that different errors in the hypothesized statistics differently affect dynamic error. Inaccuracy of knowledge of corresponding variances does not lead to change of the character of the dependence of dynamic error on time. If, for instance, σ_{21}^2 , i.e., acceleration of the target, is absent, components of dynamic error in position and speed change correspondingly by a factor of σ_{01}/σ_0 and σ_{11}/σ_1 .

The expression for error in speed (component with coefficient o_{11}^2), obviously, coincides with the expression for systematic error (7.10.55) studied above with replacement of V by σ_{11} . We already proved that its influence on total error of

measurement is immaterial.

Error in position (component with coefficient σ_{10}^2) also decreases fairly rapidly in time. Asymptotic error in position is determined by $2\sigma_{01}S_{0\Pi T}/\sigma_0^2 t$.

In the conditions considered above, and for $t \approx \frac{10}{\Delta f_c}$, its values are given in Table 7.3. Influence of this error is also immaterial.

Table 7.3							
#e	10*	109	304	104			
Error in position	0,055¢.1	0,085°,1	0, 09 # _{#1}	0,094,1			

More essential is error caused by incorrect selection of the degree of the polynomial describing change of distance. If σ_{21}^2 differs from zero, dynamic error increases without limit in time. Its asymptotic value is given by expression

$$\sigma_{\rm XBB} \approx \frac{\sigma_{\rm s1} t^{\rm t}}{12}.$$
 (7.10.60)

This means that understating of the degree of the approximating polynomial is permissible only when error of approximation in the whole interval of observation is included in limits assigned by tactical requirements.

The opposite error, connected with overstatement of the degree of the approximating polynomial, does not present any danger. From formula (7.10.59) it follows that dynamic error consists of the sum of the squares of components, corresponding to various degrees of the polynomial, and if any of the coefficients is in reality equal to zero, dynamic error will only be less than that on which we counted. Certainly, here, dynamic and fluctuation errors will increase as compared to their values with correct selection of the degree of the polynomial, but this increase is not very substantial, since both errors decrease without limit with passage of time.

Let us consider one more example of smoothing circuits with variable parameters, when forced change of their parameters with change of the signal-to-noise ratio is absent, and there occurs only change of the gain factor of the open loop due to the AGC system. We shall limit ourselves to the simple case when the measured distance changes according to the law

$$d(t) = pf(t), \ \overline{\mu^3} = \sigma_0^2. \tag{7.10.61}$$

Considering that the smoothing filter is designed for a certain signal-to-noise

ratio h_0 , we obtain

$$G(l, *) = \frac{e_0^2 f(l) f(s)}{1 + \frac{e_0^2}{S_{out}(h_0)} \int_0^s l^2(s) ds}.$$
 (7.10.62)

Then, from equation (6.2.12),

$$= K_{\rm A}(h) \, \sigma_0^2 \, f(t) \, f(\tau) \frac{\left[1 + \frac{\sigma_0^2}{S_{0\rm BT}(h_0)} \int_0^t f^{\rm a}(s) \, ds\right]^{K_{\rm A}(h) \, S_{0\rm BT}(h_0) - 1}}{\left[1 + \frac{\sigma_0^2}{S_{0\rm BT}(h_0)} \int_0^t f^{\rm a}(s) \, ds\right]^{K_{\rm A}(h) \, S_{0\rm BT}(h_0)}} .$$
(7.10.63)

Selecting at point $h = h_0$ the optimum value of the gain factor $K_{\mu}(h_0) = 1/S_{OUT}(h_0)$, we obtain the final expression

$$g(l, \tau) = \alpha^{2} K(h) f(l) f(\tau) \frac{\left[1 + \alpha^{2} \int_{0}^{t} f^{2}(s) ds\right]^{K(h)-1}}{\left[1 + \alpha^{2} \int_{0}^{t} f^{2}(s) ds\right]^{K(h)}}, \qquad (7.10.64)$$

where $K(h) = K_{\mu}(h)/K_{\mu}(h_0)$; $\alpha^2 = \sigma_0^2/s_{0NT}(h_0)$.

Here, fluctuation and dynamic errors are determined by relationships

$$\sigma_{\phi,a}^{a} = \frac{S_{\text{DRB}}(h) K^{2}(h) a^{3} f^{2}(l)}{2K(h) - 1} \frac{\left[1 + a^{2} \int_{0}^{l} f^{2}(s) ds\right]^{2K(h) - 1} - 1}{\left[1 + a^{2} \int_{0}^{l} f^{2}(s) ds\right]^{2K(h)}},$$
(7.10.65)
$$\sigma_{ABH}^{2} = \sigma_{0}^{2} f^{a}(l) \frac{1}{\left[1 + a^{2} \int_{0}^{l} f^{2}(s) ds\right]^{2K(h)}}.$$
(7.10.66)

Asymptotic change of fluctuation error when $\alpha^2 \int_0^t f^2(s) ds >> 1$ coincides with the case of optimum selection of parameters of smoothing circuits, and dynamic error decreases faster when K(h) > 1 and slower when K(h) < 1.

The dependence of variance of total error of measurement $\sigma_{BMX}^2 = \sigma_{\Phi\Pi}^2 + \sigma_{\Pi\Pi H}^2$, related to the a priori variance of the measured distance $\sigma_0^2 f^2(t)$, on the quantity



 $z = \alpha^2 \int_{c}^{t} f^2(s) ds$ when $h_0 = 10$ and $y = \Delta f_y / \Delta f_c = -30$ is shown in Fig. 7.46. Quantity z can be considered dimensionless time. The physical meaning of it is that it is the ratio of variance of measured distance, averaged for time t, $\frac{1}{t}\sigma_0^2 \int f^2(s) ds$ to variance of the efficient estimate of a constant parameter during the time t for a signal-tonoise ratio ho, equal to $S_{our}(h_0)/t$. For instance,

Fig. 7.46. The influence of nonoptimality of smoothing of circuits of a range finder on

smoothing of circuits of a range finder on accuracy of measurement. $z = o^{2}t^{2n+1}/(2n + 1)S_{OHI}(h_{O}), \text{ and when } t(t) = \sin(\omega_{O}t + \phi), z \approx \frac{\sigma_{O}^{2}}{2S_{OHI}(h_{O})}.$ During construction of the graphs of Fig. 7.46 we assumed that the discriminator

of the range finder is sufficiently close to optimum, so that $S_{OPT}(h) \approx S_{OPT}(h)$. As can be seen from the figure, curves corresponding to various values of h do not differ much from each other, and on the whole error of measurement sufficiently rapidly decreases with increase of z.

§ 7.11. Nontracking Radar Range Finders

In the practice of radar measurements they sometimes use nontracking range finders. They are applied usually when there is not required high accuracy of measurement or when there is required measurement simultaneously of coordinates of a large number of targets and application in the radar station of a large number of advanced, but complex tracking meters is connected with excessively great technical complications. Such situations are encountered in early warning and target acquisition radars [1, 26, 27].

In nontracking meters the estimate of the distance to the target is produced from the realization of the received signal in a comparatively small time interval.

Fig. 7.47. Functional diagram of a nontracking range finder: 1 - unit, producing rough estimate of distance $\hat{H}(t)$; 2 smoothing circuits.

Further data, obtained as a result of the estimate, either are used directly, or are additionally smoothed to decrease fluctuation errors of measurement. A functional diagram of the construction of a nontracking range finder is shown

-300-

in Fig. 7.47. Estimation of the distance to the target can be made by various devices or by the operator, reading data on the distance to the target from the screen of the radar set.

If calculation of the estimate is produced by a method, sufficiently close to optimum, for instance, the method of maximum likelihood, its variance is determined by formulas (7.2.22) for the case $\Delta f_{\rm C}$ T >> 1, and (7.2.25) for the case $\Delta f_{\rm C}$ T << 1, where T is the time during which the estimate is formed. Quantities $\sigma_{g\bar{q}}^2$ from (7.2.22) and (7.2.25) characterize accuracy of measurement then the estimate of distance is used directly. If there is applied additional smoothing of data obtained by the estimator unit, variance of the estimate characterizes intensity of components of interferences at the input of smoothing circuits. Here, if the time for formation of the estimate T and the time of correlation of the received signal are sufficiently small as compared to inertia of the smoothing circuits, these components can be considered white noise with spectral density

$$S_{gx} = \sigma_{so}^{*} T. \tag{7.11.1}$$

With rapid fluctuations of the signal it is obvious that S_{BX} coincides with $S_{ON T}$. Therefore, tracking and nontracking meters with identical structure of smoothing circuits (in the case of a tracking meter we have in mind the pulse response of the closed-loop servo system) will give identical accuracy of measurement.

Actually, all real estimator units give estimates which are less than efficient. Furthermore, real estimates are not unbiased. All these circumstances can lead to considerable worsening of accuracy of measurement by nontracking systems. Nevertheless, for a number of applications of radar results obtained here may be satisfactory. Let us consider more concretely several examples of nontracking meters.

7.11.1. Range Finder with Frequency Modulation

In radio altimeters and certain other cases there is applied a range finder constructed according to the block diagram of Fig. 7.48 [27, 31]. A transmitter radiates the frequency modulated signal. The received signal is mixed with the signal of the transmitter, is passed through a low-frequency filter, and is fed to a frequency measuring unit. If the frequency is modulated according to the law $\Delta \omega(t)$, frequency of the signal at the mixer output varies according to the law $-i(t) = \Delta \omega(t) - \Delta \omega(t - \tau)$.

For small τ frequency $\omega_1(t) \approx \tau \frac{\partial \Delta \omega(t)}{dt}$; therefore, knowing the law of change of frequency and measuring the frequency of the signal at the mixer output, it is

-301-

possible to measure dalay of the reflected signal. For any periodic frequency modulation the magnitude of $\frac{\partial \Delta \omega(t)}{\partial t}$ changes in time. For instance, with triangular and sawtooth modulation function $\omega_1(t)$ varies in accordance with Fig. 7.49. From the figure it is clear that, depending upon the magnitude of τ , the whole shape of curve $\omega_1(t)$ changes; for instance, for sawtooth modulation not only the level of flat sections, but also their location changes. Usually in nontracking radio range finders with frequency modulation the influence of delay τ on the shape of the curve is not taken into account, and measurement of distance is based on measurement of frequency in the flat section (Fig. 7.49).



Fig. 7.48. Block diagram of a nontracking range finder with frequency modulation: 1 - transmitter; 2 - mixer; 3 - low-frequency filter; 4 frequency measuring unit.

With triangular modulation,



Fig. 7.49. Change of frequency in time; a) sounding signal; b) received signal; c) signal, at mixer output; 1 - section utilized for measurement.

for instance, bandwidth of the filter is selected such that frequencies, corresponding to the interval of time from kT_r to $kT_r + \tau$, in general, do not pass through the filter. Such construction of a meter is permissible, obviously, only when the required range of change of delay is considerably less than the period of repetition of the signal. Otherwise, there will take place excessively large power losses, which are proportional to τ/T_r . Subsequently, we shall consider, in accordance with the way this is done in practice, that $\tau_{MENC}/T_r <<$ 1, and we shall ignore losses connected with this.

With fulfillment of condition $\tau_{MAKO}/T_{r} \ll 1$ the considered range finder with triangular and sawtooth modulation, from the point of view of analysis of accuracy, is equivalent to a frequency meter. Fluctuation error of measurement is determined here by the following expressions for variance:

for sawtooth FM



-302-

(7.11.2)

for triangular FM

 $\sigma_{\tau}^2 \sim \frac{T_r^2}{16\omega_{\perp}^2} \sigma_{\omega}^2 \,,$

where σ_{m}^{2} - variance of error of measurement of frequency;

 $2\omega_m$ - frequency deviation.

During calculation of σ_{ω}^2 one should consider that coherence of the signal from period to period is disturbed. We explain this by the example of sawtooth modulation. Since the filter does not pass frequencies corresponding to negative values of $\omega_1(t)$ (Fig. 7.49), the signal at the filter output in every period has a new value of phase, depending on the duration of the negative section, equal to τ . If this advance of phase from period to period is not compensated, signals in various periods will be incoherent. Compensation of phase advance can be carried out only using the measured value of delay, which, obviously, signifies anew return to a tracking meter. In the absence of compensation coherence of the signal is disturbed, which leads to lowering of accuracy of measurement. Analogous phenomena take place also for any other law of frequency modulation. This means that for any method of measurement of frequency accuracy of the considered range finder will be worse than the accuracy of a tracking range finder.

It is possible to show that with an optimum method of measurement of frequency and low signal-to-noise ratios accuracy of the considered range finder is approximately identical to the precision of an incoherent tracking range finder using a signal with the same parameters.

Actually, as is shown in ChapterIX, variance of the efficient estimate of frequency, formed over one period T_r for not too large a signal-to-noise ratio, is approximately equal to

$$\sigma_{\bullet}^{2} = \frac{6}{T_{r}^{2}q^{2}} (1+q), \qquad (7.11.4)$$

(7.11.3)

where q — the ratio of energy of the signal for the period to the spectral density of noise $(q = P_c T_{\gamma}/2N_0)$.

Then, for instance, for sawtooth FM, according to the formula (7.11.2)

$$\sigma_{q}^{2} = \frac{3}{\sigma_{q}^{3}} \frac{1+q}{2q^{2}}.$$
 (7.11.5)

With triangular FM measurement should be produced by half-periods $T_{\rm p}/2$, and we again obtain the same expression for σ_{τ}^2 . Quantity σ_{τ}^2 from (7.11.5) characterizes accuracy of measurement during single measurement. If measurements are produced
over many periods, then quantity σ_{τ}^2 determines the spectral lensity of fluctuation error

المالية المستقدمات المالية المحالية المستقدمة المستقدمة المحالية المستقدية المالية المحالية ا

$$S_{gKB} = \sigma_v^2 T_r = \frac{3T_r}{\omega_m^2} \frac{1+q}{2q^2}.$$

Since $\omega_m^2/3$ is equal to parameter b = -C''(0) for the considered forms of modulation, S₃₆₀ coincides with the expression for equivalent density of an incoherent range finder using a pulse signal with the same mean power and with linear change of frequency with deviation within limits of the pulse $2\omega_m >> 1/\tau_M$ (see Chapter VIII).

In reality, due to the specific dependence of accuracy of measurement of frequency on the signal-to-noise ratio [for large q formula (7.11.4) is invalid; σ_{ω}^2 here approaches a finite quantity], actual accuracy of a nontracking meter will be still worse. Furthermore, in practice in such range finders there is not produced measurement of frequency by a method even faintly similar to optimum, but there is carried out counting of zeroes of output voltage of the filter. Since the passband of this filter, designed for the range of measured distances, considerably exceeds the frequency of repetition, and counting of zeroes is produced by nonlinear transformations, then due to worsening of the signal-to-noise ratio, connected with expansion of the passband, accuracy of measurement sharply decreases. Simultaneously with increase of fluctuation error, the method of counting zeroes leads to the appearance of systematic errors of measurement.

7.11.2. Use of a Rangel Finder with a Fast Tracking Loop for Founding on Estimate

In the beginning of this section we said that if the estimate of distance in a nontracking range finder is close to the efficient one, its accuracy is close to the accuracy of a tracking range finder with identical structure of its smoothing circuits. Optimum estimator units can be synthesized just as discriminators. Nowever, a satisfactory solution of the problem of optimum formation of an estimete is application of a high-speed tracking range finder with very simple structure of its smoothing circuits. The output of this range finder should be smoothed in inertial circuits, matched in their characteristics with the law of change of the measured distance. As a whole such a range finder is "nontracking" in the sense that for control of tuning of the discriminator in it there is used, not the final output value of distance, but a certain intermediate value, which may repeat the true value of distance with considerably greater errors.

Requirements on the high-speed range finder ensue from requirements on the

-304-

optimum estimate. This estimate should be unbiased and efficient. From the requirement of unbiasedness there follow requirements of the absence of systematic and dynamic errors of tracking. These requirements will be carried out sufficiently satisfactorily if inertia of the high-speed range finder is small with respect to the rate of change of the measured distance.

The requirement of efficiency leads to the requirement of optimality of the discriminator of the high-speed range finder from the point of view of minimum equivalent spectral density. Then the output of the high-speed range finder in the absence of noises will repeat the law of change of the measured distance, and fluctuation error will be practically equal to the variance of the efficient estimate, built from the realization of the signal during the time $T = 1/2\Delta f_{\partial\Phi}$, where $\Delta f_{\partial\Phi} - \Delta f_{\partial\Phi}$ where $\Delta f_{\partial\Phi} - \Delta f_{\partial\Phi}$ action of the closed-loop system is low or high as compared to the time of correlation of fluctuations ($\Delta f_{\partial\Phi}/\Delta f_{C} \ll 1$ and $\Delta f_{\partial\Phi}/\Delta f_{C} \gg 1$), are determined by formulas (7.2.22) and (7.2.25), respectively. They characterize accuracy of measurement when the resultant estimate of distance is used directly.

If, however, subsequent smoothing is used, these variances characterize the intensity of interferences at the input of the smoothing circuits. Moreover, due to the low inertia of the closed-loop meter forming the estimate these components may be considered white noise with spectral density

$$S_{ax} = \frac{e_{a\phi}^2}{2\Delta f_{a\phi}}, \qquad (7.11.6)$$

which, for rapid fluctuations coincides with quantity $S_{O\Pi\Pi}$ or S_{BHB} in the case of a nonoptimal discriminator, and when $\Delta f_{ab}/\Delta f_{c} >> 1$ gives

$$\mathbf{S}_{\mathbf{a}\mathbf{x}} = \frac{4N_0^2\Delta f_{\mathbf{a}\mathbf{b}}}{(b-a^2)P_c^2} \left(1 + \frac{P_c}{4N_{\mathbf{a}\mathbf{a}f_{\mathbf{a}\mathbf{b}}}}\right). \tag{7.11.7}$$

Further analysis of the meter does not differ, in principle, from that performed above. Dynamic and fluctuation errors of the meter as a whole are determined by the very same formulas, in which by $g(t, \tau)$ we mean the pulse response of the open-loop smoothing filter. Pulse response $v(t, \tau)$, as before, is determined by formula (7.10.42). During calculation of fluctuation errors one should use the magnitudes of spectral density from (7.11.6) and (7.11.7). With optimum selection of the smoothing filter and a comparatively low level of noises errors of range finding, obviously, in this case coincide with errors of a tracking meter. With a high level of noises due to the high speed of operation of the preliminary meter,

-305-

forming the estimate, in the considered system breakoffs of tracking become more probable, the influence of parametric fluctuations increases, and as a whole errors of measurement increase. Furthermore, due to the high speed of operation application of a single-channel discriminator with switching of reference signals is hampered, the influence of all sorts of intermittent interferences is increased, etc., which, due to inevitable nonlinearities and parametric connections, may lead to swinging of the system, its excitation and loss of the target in the unit forming the estimate. All this, in general, leads to undesirability of application of such meters when we require from the radar range finder the very highest accuracy of measurement.

§ 7.12. Range Finder with Discrete Measurement

In a number of cases requirements on the width of the spectrum of modulation are dictated, not by permissible errors of range finding, but by the resolution capability. Modulation in certain cases is so broad-banded that potential accuracy of range finding, calculated from the concept of a point target, cannot be realized. The resolution range here is comparable to or even less than dimensions of the target, and accuracy of range finding is determined already practically by dimensions of the target. In these conditions is is sometimes senseless to apply complex range finders with special discriminators, and it is possible to limit ourselves to simpler systems.

Usually radars are equipped with detection channels designed for a certain range of distances, where often detection is produced simultaneously over the whole range of distances, for which there is a set of parallel channels of identical type, each of which is designed for a certain distance. Methods of construction of such channels are considered in Chapters I7 and 7.

With optimum processing the output voltage of such a channel coincides with the logarithm of the likelihood ratio of the received signal for the given distance. It is obvious that the unit of target detection channels, in one way or another present in a radar, can be used for range finding and also for simultaneous measurement of the distance to many targets. This unit can be used for various tasks, including creation of discrimination characteristics of normal range finders. Here we consider a case where taking of data on the distance to a target from the unit of detection channels is realized discretely with accuracy of the number of the channel. Possibilities of increase of the accuracy of determination of the distance already were considered in Chapters THI and IV.

One very simple functional circuit of such a range finder, designed for

-306-

measurement of the distance to one target, is shown in Fig. 7.50. The received



Fig. 7.50. Functional circuit of a range finder with discrete measurement: 1 - unit of detection channels; 2 - unit for selection of maximum voltage at the output of the unit 1; 3 - converter of channel number into distance; 4 - smoothing circuits.

signal passes through the unit of range detection channels, at the cutput of which is a unit which selects at definite intervals of time of duration T_0 the channel whose output voltage is maximum. Then, there follows a unit for transformation of the number of the channel into a

discrete distance, and there occurs smoothing of the obtained data by linear smoothing circuits. If there arises the task of tracking several targets, the selection unit should determine the number of channels in which voltage attains a maximum, the maximum of the remaining channels, etc. It is also possible to select all channels in which output voltage exceeds a certain threshold, determined by the level of noises, and to carry out smoothing of all data by separate smoothing filters. The circuit of Fig. 7.50 can be modified and turned into a servo system. Tracking here will be carried out not by zero of the discrimination characteristic, but by the channel number. It is possible to carry out such tracking, for instance, in the following way.

After preliminary determination of the channel with the target, from its output voltage the output voltages of the two adjacent channels are subtracted. If one of these differences is less than a threshold selected in a certain definite form, as the measure of distance there is selected the channel for which this difference is less than the threshold, and its voltage is compared with voltages of the two adjacent channels. Discrete data are smoothed by the usual method, and they control switching of the channels, subsequently selecting the number of the channel with the target. For low signal-to-noise ratios, when errors of measurement can be great and exceed the magnitude of one discrete value of distance, one should anticipate comparison of voltage in the selected channel not only with adjacent, but with more distant channels.

Let us consider in more detail with very simple assumptions a nontracking meter of discrete type with a fundamental circuit of the type of Fig. 7.50. We shall consider that adjacent channels of the detection unit do not overlap, so that signals at their output are orthogonal and statistically independent. Furthermore, we assume that selection of number is carried out in such a time T_0 , at which voltages in any of the channels at the end of adjacent intervals of duration T_0 are statistically independent. We note that formation of the logarithm of the likelihood

-307-

ratio may occur, not for all time To, but during a certain part of it.

On these assumptions the input signal of the smoothing filter is a sequence of random variables d_k discretely varying with period T_0 , which can take one of the discrete values $d^{(1)}$, $d^{(2)}$, ..., $d^{(m)}$, corresponding to the 1st, 2nd, ..., m-th channel. Quantities $d^{(1)}$, $d^{(2)}$, ..., $d^{(m)}$ is every period T_0 have the probability distribution $p_{1j}^{(k)}$, depending on the real position of the target in the k-th period of T_0 . Probability $p_{1j}^{(k)}$ is the conditional probability that in the k-th period of T_0 maximum voltage will be in the i-th channel on the condition that the target in this period is in the j-th channel.

The mean value and variance of the input signal of the smoothing filter are determined by the following relationships:

$$\overline{d_{A}(i)} = \sum_{i=1}^{M} d^{(i)} p_{ii}^{(A)}, \qquad (7.12.1)$$

$$\sigma_{k}^{2}(j) = \sum_{l=1}^{m} d^{(i)} p_{ij}^{(k)} - \left(\sum_{j=1}^{m} d^{(i)} p_{ij}^{(k)}\right)^{3}.$$
 (7.12.2)

Both quantities are calculated under the condition that the target is at distance d(j).

If smoothing circuits are a discrete filter, fluctuation error of measurement will be determined by expression

$$\sigma_{\max,k}^{2} = \sum_{k=1}^{n} g_{nk}^{2} \sigma_{k}^{2}, \qquad (7.12.3)$$

where g_{nk} - response of the filter at the time nT_{ij} to a disturbance applied at time kT_{ℓ} .

Correspondingly, dynamic error in the nonstatistical approach is equal to

$$\mathbf{a}_{\text{RHM},n} := d_{\mathbf{o}n} - \sum_{k=1}^{n} g_{nk} \sum_{i=1}^{m} d^{(i)} p_{ij}^{(k)}, \qquad (7.42.4)$$

where d_{OD} - true distance at time nT_{O} .

With the statistical approach, when we know the probability distribution of real distance at any moments of time, we must find the mean square dynamic error $\varepsilon_{\Pi NH}$ n^{*}

If inertia of the smoothing filter is great as compared to the interval of quantization T_0 , it may be continuous, and then fluctuation error, obviously, is given by the following formula:

4). 4 a sahahan dalam dara sahar sahar saharan dalam dalam dalam saharan saharan saharan saharan saharan saharan saha

$$\sigma_{\phi,\tau}^2 = T_{\phi} \int_0^t g^2(t,\tau) \sigma^2(\tau) d\tau, \qquad (7.12.5)$$

where $\sigma^2(t)$ - value of variance from (7.12.2) at time t = kT_0 .

The expression for dynamic error varies analogously.

Probabilities $p_{i,j}^{(k)}$ with the accepted assumptions about the independence of output voltages of various channels, as it is simple to prove, are determined by the following expressions:

$$p_{jj}^{(k)} = \int_{-\infty}^{\infty} w_{\text{cm}}(x) \left[\int_{-\infty}^{x} w_{\text{m}}(z) dz \right]^{m-1} dx,$$

$$p_{jj}^{(k)} = \int_{-\infty}^{\infty} w_{\text{m}}(x) \int_{-\infty}^{x} w_{\text{cm}}(z) dz \left[\int_{-\infty}^{x} w_{\text{m}}(z) dz \right]^{m-2} dx \qquad (7.12.6)$$

and, as follows from these expressions, depend on k only through the value of j, corresponding to true distance at moment kT_0 . Functions $w_{CHI}(x)$ and $w_{HI}(x)$ are probability densities for output voltage of the detection channel under the condition that on it there act a signal mixed with noise and noise alone, respectively (see Chapters IV and V). If, at the output of each of the detection channels, as in practice, there is also a certain threshold unit allowing us to eliminate from consideration those channels, voltage in which in the given period does not attain its trigger level C, expressions (7.12.6) are somewhat changed — the lower limit of integration over x is changed in them from - ∞ to c. Transforming the second of expressions (7.12.6) by integration in parts, it is easy to prove the validity of the following relationship:

$$p_{ij}^{(k)} = \frac{1}{m-1} \left[1 - p_{jj}^{(k)} \right]$$
(7.12.7)

which gives us the possibility to limit ourselves to determination only of probability $p_{i,i}^{(k)}$.

With fluctuations which are slow as compared to the period of duty of a detection channel in one period distributions $w_{CUI}(x)$ and $w_{III}(x)$ are equal, respectively, (see Chapter N), to

$$w_{\text{om}}(x) = \frac{1}{1+\mu} e^{-\frac{x}{1+\mu}},$$

$$w_{\text{m}}(x) = e^{-x},$$
 (7.12.8)

where μ is the ratio of energy accumulated in one period T_0 of the signal to the spectral density of noise.

Then, from formula (7.12.6), we have

$$p_{jj}^{(k)} = \sum_{k=0}^{m-1} C_{m-1}^{k} (-1)^{k} \frac{1}{1+k(1+\mu)} \approx 1 - \frac{m-1}{m\mu}. \qquad (7.12.9)$$

The approximate expression is valid when $\mu \gg 1$, which is usually realized in practice. Analogously, probability $p_{jj}^{(k)}$ can be calculated with certain approximations also for fluctuations which are rapid as compared to the period of duty of the detection channel. Probabilities of erroneous selection of the maximum on the basis of (7.12.7) are equal to

$$\boldsymbol{p}_{ij}^{(k)} \approx \frac{1}{m\mu}. \tag{7.12.10}$$

. is its active and its .

Due to the equality of all probabilities $p_{ij}^{(k)}$ when $i \neq j$, formulas (7.12.1) and (7.12.2) can be simplified and reduced to the form

$$\overline{d_{h}(j)} = \frac{1 + m p_{jj}^{(k)}}{m-1} d^{(j)} + \frac{mN}{m-1} (1 - p_{jj}^{(k)}) \Delta d, \qquad (7.12.11)$$

$$\overline{d_{k}^{1}(j)} = \frac{1 + m p_{jj}^{(k)}}{m-1} d^{(j)*} + (1 - p_{jj}^{(k)}) \frac{m \Delta d^{*}}{m-1} \left[N^{*} + \frac{m^{*} - 1}{6} \right], \quad (7.12.12)$$

where Δd - interval of distances covered by one detection channel;

N - distance, expressed in intervals Ad, corresponding to the middle of the range covered by the unit of detection channels.

From formulas (7.12.11) and (7.12.12) it follows that when m >> 1

$$\sigma_{k}^{2} = p_{jj}^{(k)} (1 - p_{jj}^{(k)}) (d^{(j)} - N\Delta d)^{2} + (1 - p_{jj}^{(k)}) \frac{m^{2} \Delta d^{2}}{6}. \qquad (7.12.13)$$

A characteristic feature of the considered range finder is the dependence of errors of measurement on the magnitude of the real distance [see formulas (7.12.2) and (7.12.13)]. This peculiarity is inherent to a certain extent in all nontracking meters, but in the given case it is manifested especially distinctly. Fluctuation error reaches its minimum if the target is in the middle of the range occupied by the detection unit, and increased with shift of the target towards its edges. As follows from (7.12.13), this increase occurs by square law. With probability $p_{d,d}^{(k)}$ close to one, variance of errors on the edges of the range is doubled as compared to the minimum magnitude. Here, from expression (7.12.13) it follows that variance on the edge of the range, with accuracy to coefficient 1 - $p_{d,d}^{(k)}$, coincides with variance

-310-

of a uniform a priori distribution of width $\Delta d_{\Omega} = m \Delta d$.

If, in particular, probability $p_{jj}^{(k)}$ is determined by formula (7.12.1°), the magnitude of variance is

$$\sigma^{*}(l) = \frac{1}{l^{*}} \left[(d(l) - d_{0})^{*} + \frac{1}{6} \Delta d_{0}^{2} \right], \qquad (7.12.14)$$

where d_0 - distance, corresponding to the middle of the range $(d_0 = N\Delta d)$;

d(t) - true distance at time t.

Then, for great inertia of the smoothing filter and constancy of the measured distance d(t) = d, and considering that $\mu = P_c T_0/2N_0$, we obtain in steady-state operating conditions the following expression for variance of fluctuation error:

$$\sigma_{\phi,s}^{2} = T_{\phi}\sigma_{\phi}^{a}\int_{0}^{\infty} g^{a} (t-s) ds = T_{\phi}\sigma^{2}2\Delta f_{\phi\phi} =$$

$$= \frac{N_{\phi}\Delta f_{\phi\phi}}{P_{\phi}} \left[(d-d_{\phi})^{a} + \frac{1}{6}\Delta d_{0}^{2} \right]. \qquad (7.12.15)$$

This expression shows that a range finder of the considered type has essentially worse accuracy than a tracking range finder. The magnitude of fluctuation error with such a method of range finding increases by ratio $\Delta d_0/\Delta d$ and is no longer determined by the magnitude of the resolution capability, but by the magnitude of the a priori interval.

§ 7.13. Range Finding in a System with a Variable Repetition Frequency

As it was shown in Chapter IV, to increase noise immunity with respect to passive interferences it is useful to increase the frequency of repetition. Here, the advantages of unmodulated radiation (speed resolution) can be combined with selection of the target with respect to distance. However, when using sufficiently high frequencies of repetition there may arise ambiguity with respect to distance and a number of difficult circumstances connected with the necessity of suppressing powerful signals from closely located local objects and the necessity of work of the receiver and transmitter on one antenna. In the last case, when the transmitter is operating its signal proceeds to the input of the receiver, and even if the tuning frequency of the receiver, due to Doppler shift, is different from the transmitter frequency, due to transmitter noises, occupying a wide frequency range, at the input of the receiver during this time there will be powerful interference, essentially exceeding signal level [40]. This circumstance leads to the necessity of cutoff of the receiver during radiation, which in turn causes the appearance of "blind" zones with respect to

-311-

distance. The number of these zones with a high frequency of repetition may be great, and the total extent depends on the off-duty factor. In particular, with an off-duty factor equal to two, half the range is covered by "blind" zones.

With movement of the target the signal from it alternately passes through "blind" zones. Therefore, for a significant part of the time it is inaccessible for observetion, and the radar works with interruption equal to the time of passage through the "blind" zone. In such interruptions there are power losses and essential impairment of accuracy of measuring of distance and other coordinates. In order to eliminate these losses, it is useful to produce an abrupt or smooth change of the frequency of repetition, reducing to a minimum the times when the target is in a "blind" zone. Most useful is continuous change of the frequency of repetition, which permits us in principle to shift the interval between "blind" zones right behind the target and reduce the time of signal cutoff to zero. Here, by change of frequency we realize target tracking.

In the course of such tracking of the target the system should change its period of repetition $T_r(t)$ in such a manner that error is minimal:

• (l) =
$$\left(n + \frac{1}{2}\right)T$$
, (l) - • (l), (7.13.1)

where $\tau(t)$ - changing magnitude of delay;

 $T_r(t)$ - period of repetition;

n - integer.

This system can consist of a discriminator and smoothing circuits the same type as in an ordinary range finder. It is possible to show that the optimum system, ensuring minimum mean square error $\varepsilon^2(t)$, remains in this case the same as the circuit of an optimum range finder. The difference is that in the given case, not the magnitude of delay, but $(n + \frac{1}{2})T_r(t)$ should be controlled. We note that it is sometimes more convenient technically to control proportionally to control voltage, not the pericd, but the frequency of repetition f_r .

It is easy to prove that with use of the error signal for control of f_r and linear smoothing circuits the system does not ensure the required quality of tracking: with movement of the target the equation of the system does not have a solution in the form

$$f_r(t) = \frac{n+\frac{1}{2}}{\tau(t)} + \xi(t),$$

where $\xi(t) - small$ error which does not increase.

-312-

In practice the quantity permitting control is not $(n + \frac{1}{2})T_r(t)$, but the period of repetition $T_r(t)$. In this connection there arises the necessity of controlling the gain factor of the open circuit proportionally to $1/(n + \frac{1}{2})$. Such control is especially necessary if the distance changes in the course of tracking by a factor of several times. Inasmuch as linear change of the period is possible only in a limited range, in this case jumps of period T_r by a nominal amount after passage by the target through several "blind" zones with respect to distance are inevitable. Thus, quantity n changes in the course of tracking, and in order to keep the speed of operation of the range tracking system constant it is necessary to change the gain factor of the open circuit K_0 .

To realize control of K_0 and to properly, i.e., without losing the target, choose moments of jumps it is necessary to measure the distance to the target. Unambiguous measurement of distance can be effected with the use of outputs of the period control and speed tracking systems. The latter always is present in coherent single-target radars. Let $\varepsilon(t) = 0$; then from (7.13.1) $(n + \frac{1}{2})T_r(t) = \tau(t) = \frac{2d(t)}{c}$. Differentiating both parts of this equality, we obtain

$$\left(n+\frac{1}{2}\right)\dot{T}_{r}(l) = -\frac{2V(l)}{c} = -\frac{\lambda}{c}f_{x}(l), \qquad (7.13.2)$$

where V(t) - speed of approach;

 $f_{II}(t)$ - Doppler frequency.

Eliminating n, we can obtain

$$d(l) = -\frac{1}{2} \frac{T_r(l)}{\dot{T}_r(l)} \lambda_{I_R}(l) = -\frac{T_r(l)}{\dot{T}_r(l)} V(l). \qquad (7.13.3)$$

Calculation of range by formula (7.13.3) can be done by analog or digital computers. To decrease fluctuations of voltage proportional to $T_r(t)$ one can use additional smoothing circuits after differentiation of voltage at the output of the period control system.

Let us find the magnitude of fluctuation error of range finding. We shall consider fluctuations of functions $T_r(t)$, $T_r(t)$ and V(t) small as compared with their mean values. Then error of unambigous measurement of distance is

$$\Delta d(t) = d(t) \left[\frac{\Delta T_{r}(t)}{T_{r}(t)} - \frac{\Delta T_{r}(t)}{T_{r}(t)} + \frac{\Delta V(t)}{V(t)} \right] = \frac{c}{2} \left[e_{\Phi}(t) + \frac{d(t)}{V(t)} (\dot{e}_{\Phi}(t) + \frac{2\Delta V(t)}{c}) \right], \qquad (7.13.4)$$

where $\varepsilon_{\phi}(t)$ - fluctuation error of the system for range tracking of a target; $\Delta V(t)$ - fluctuation error of the speed tracking system.

-313-

Random variables ϵ_{Φ} (t), $\dot{\epsilon}_{\Phi}$ (t), and $\Delta V(t)$ can be considered uncorrelated: ϵ_{Φ} (t) and $\dot{\epsilon}_{\Phi}$ (t) are uncorrelated with one another in view of the symmetry of the spectrum of fluctuation error ϵ_{Φ} (t) with respect to zero frequency; while ϵ_{Φ} (t) and $\dot{\epsilon}_{\Phi}$ (t) are not correlated with $\Delta V(t)$ in view of the unconnectedness of coding of distance and speed in the received signal.

As a result the variance of fluctuation error is equal to

$$\mathbf{\sigma}^{\mathbf{a}}(t) = \frac{c^{\mathbf{a}}}{4} \left[\sigma_{\mathbf{a}}^{\mathbf{a}} + t_{0}^{\mathbf{a}}(t) \left(\sigma_{\mathbf{a}}^{\mathbf{a}} + \frac{4}{c^{\mathbf{a}}} \sigma_{\mathbf{V}}^{\mathbf{a}} \right) \right], \qquad (7.13.5)$$

يە يەرەب مەلەر مەرەب مەرەپ بەرەپ بەرەپ بەرەپ مەرەپ بەرەپ بەر

where $t_0(t) = d(t)/V(t) - time$ remaining until target impact.

Error of tracking of distance σ_e^2 can be calculated from the results of the preceding paragraphs; error of measurement of speed can be defined by formulas of Chapter IX, and σ_e^2 is equal to

$$P_{\mu}^{3} = \frac{1}{2\pi} \int_{-\infty}^{\infty} \Phi^{3} S_{\mu \pi \mu} |H_{1}(i\omega)|^{2} |H_{2}(i\omega)|^{2} d\omega, \qquad (7.13.6)$$

where $H_1(i\omega)$ - frequency response of the range tracking system;

- $H_2(i\omega) frequency response of the filter intended for decreasing fluctuations <math>\dot{T}_r(t)$;
 - S_{3HB} equivalent spectral density of the discriminator of the range tracking system.

Quantity σ_V during speed tracking can comprise 0.1-0.5 m/sec or more, and σ_e^2 with correct choice of smoothing circuits will coincide in order with $\frac{4}{c^2}\sigma_V^2 \approx \approx 4.10^{-18}$ to 10^{-16} . Then, for instance, when $t_0 = 10$ min

$$\frac{c}{2}$$
 $\sqrt{\frac{t_0^2}{t_0^2} \left(\frac{a_i^2 + \frac{4}{c^4} a_V^2}{a_V^2} \right)} \approx 25 - 100 \text{ min.}$

This means that error obtained due to inaccuracy of measurement of speed and magnitude $\dot{T}_{r}(t)$ may significantly exceed error of tracking for distance σ_{c} .

Thus, in the considered system accuracy of unambiguous range finding is determined, basically, not by the usual tracking range finder, tracking the target, but by a computer.

§ 7.14. Influence of Interferences on Coherent Radar Range Finders

As it was shown in Chapter I, application of contemporary radar means may be accompanied by the influence on them of natural and man-made external interferences. The presence of such interferences impedes work of the radar and, in particular, decreases accuracy of range finding, and in certain cases generally leads to cessation of work of a radar range finder [69-70]. All this creates the necessity to consider

-314-

the influence of external interferences on range finders,

In the case of coherent radio range finders such consideration is facilitated by the very principle of their construction. Due to the presence in discriminators of range finders of narrow-band elements, preceding amplitude or phase detection, any interference having a random character and a spectrum width relatively large as compared to the transmission bandwidth of the narrow-band filters turns out to be statistically equivalent to normal white noise with the corresponding spectral density. Therefore, the influence of interferences can be described by means of corresponding increase of the level of noises at the input of the receiver of the range finder.

Analysis of the influence of interferences leads then to determination of the spectral density of such equivalent white noise and calculation of the new signalto-noise ratio. In other respects all results obtained in the preceding paragraphs do not change. It is necessary only to substitute in all formulas for the previously used signal-to-noise ratio h its new magnitude

$$h_{\rm IIII} = \frac{P_{\bullet}}{2\Delta I_{\bullet} (N_{\bullet} + N_{\rm I})} = \frac{h}{1 + \frac{N_{\rm I}}{N_{\bullet}}}, \qquad (7.14.1)$$

where h_{nm} - signal-to-noise ratio in the presence of noise and interference;

- spectral density of white noise, which is equivalent to interference recalculated at the input of the receiver of a coherent radio range finder.

Let us consider now examples of the most common external interferences.

7.14.1. Noise Interference

Active noise interference, radiated from the tracked or another target, by virtue of its usual broad-banded nature, obviously, is equivalent to white noise without any reservations. The spectral density of it at the input of the receiver is determined from the power of the jamming transmitter P_{π} and width of the spectrum of interference Δf_{π} by the distance formula

$$2N_{\pi} = \frac{P_{\pi}G_{\theta\pi}G_{\pi\nu}\lambda^{*}}{(4\pi)^{3}\Delta f_{\pi}d_{\pi}^{2}}, \qquad (7.14.2)$$

- antenna gain of the station of interferences; where G_{nn}

Gпр

- gain of the receiving antenna of the radar in the direction to the source of interference;

 λ - wavelength;

 d_{π} - distance to the source of interference.

Decrease of the signal-to-noise ratio is determined by relationship

(7.14.3)

 $\frac{h_{\text{RW}}}{h} = \frac{1}{1 + \frac{P_{\text{B}}G_{\text{RR}}}{2N_{0}\Delta/a} \frac{G_{\text{RP}}\lambda^{3}}{(4\pi)^{3}d_{-}^{2}}}$

and is stronger, the larger the ratio of the spectral density of radiated interference $P_{\Pi} = G_{\Pi\Pi} = /\Delta f_{\Pi}$ to the spectral density of internal noise and the larger the ratio of the area of the receiving antenna to the square of the distance to the source of interferences.

From formula (7.14.3) it follows that the signal-to-noise ratio $h_{\Pi\Psi}$ is determined in many cases by the magnitude of external interference, and not the



Fig. 7.51. Decrease of the signalto-noise ratio under the influence of active noise interference.

internal noises of the receiver. Therefore, for the case of a source of interferences combined with the target the signal-tonoise ratio increases with approach to the target approximately proportionally to $1/d^2$, and not proportionally to $1/d^4$ as in the absence of interference. Curves of the dependence of the signal-to-noise ratio on G $_{\Pi_1}$ $\lambda^2/(4\pi)^2 d_{\Pi}^2$ for various values of P_Π G_{ΠΠ} /2N₀ Δf_{Π} are shown in Fig. 7.51.

7.14.2. Pulse Chaotic Interference Properties of such interference were

considered in Chapter I. Here, we additionally assume incoherence of pulse chaotic interference, i.e., we consider that separate pulses of interference have independent random initial phases. Then the signal of interference proceeding to the receiver input can be recorded in the form

$$\mathbf{r}(t) = \sum_{\mathbf{k}} u_{\mathbf{n}}(t - t_{\mathbf{k}}) \cos\left[(\omega_{\mathbf{0}} + \omega_{\mathbf{p}})t + \varphi_{\mathbf{k}}\right], \qquad (7.14.4)$$

where u_{II} (t) - function, describing the shape of the pulse of interference;

 t_k - random moment of appearance of the k-th pulse;

 φ_k - random phase of the high-frequency charge of the k-th pulse;

 ω_p - frequency separation between carrier frequencies of the radar and the jamming station.

After multiplication of interference by a reference signal in the receiver (heterodyning and gating) it takes the form

-316-

where A(t), B(t) - periodic nonstationary random processes with zero mathematical expectation and spectral density corresponding to the averaged correlation function at zero frequency, equal to

$$2N_{\pi} = \frac{\mathbf{v}}{2\pi} \int_{-\infty}^{\infty} |U_{\pi}(i\mathbf{e})|^{\mathbf{e}} \int_{-\infty}^{\infty} C(\mathbf{v}) e^{i(\mathbf{e}-\mathbf{e}_{p})\mathbf{v}} d\mathbf{v} d\omega, \qquad (7.14.6)$$

where U_{Π} (i ω) - Fourier transform of u_{Π} (t);

 $\mathtt{C}(\tau)$ - autocorrelation function of the sounding signal of the radar;

 ν - average frequency of interference pulses.

Quantity N_{Π} is the spectral density of equivalent white noise, which is contained in formula (7.14.1) and characterizes decrease of the signal-to-noise ratio. The permissibility of the idealization of pulse chaotic interference by white noise during action on a coherent radio range finder is determined by the relationship between the prolonged nature of the interference pulse and bandwidth of the narrow-band filter. If $\Delta f_{\oplus} \tau_{\Pi} \ll 1$, where $\tau_{\Pi} = duration$ of the interference pulse, then such idealization is fully justified.

Formula (7.14.6) allows certain simplifications with the proper assumptions. If, in particular, width of the spectrum of interference pulses is great as compared to the width of the spectrum of modulation, and detuning of the carrier $\omega_{\rm F}$ is small as compared to the width of the spectrum of modulation, from expression (7.13.6) it follows that

$$2N_{\pi} \approx \frac{v}{2\pi} \int_{-\infty}^{\infty} |U_{\pi}(0)|^{s} \int_{-\infty}^{\infty} C(\tau) e^{i\omega\tau} d\tau d\omega = v |U_{\pi}(0)|^{s} C(0) =$$
$$= v |U_{\pi}(0)|^{s} = v \Big[\int_{-\infty}^{\infty} u_{\pi}(t) dt \Big]^{s} = v u_{0}^{2} \tau_{\pi}^{2} = v \tau_{\pi}^{2} P_{\pi}, \qquad (7.14.7)$$

where P_{π} - power of interference in the pulse,

u_o - pulse amplitude.

In the other extreme case, when the width of the spectrum of the pulse of interference is small as compared to the width of the modulation spectrum,

$$2N_{\rm H} \sim v_{\rm H} \frac{1}{2\pi} \int_{-\infty}^{\infty} |U_{\rm H}(i\omega)|^2 d\omega \int_{-\infty}^{\infty} C(\mathbf{t}) d\mathbf{s} = v_{\rm H} P_{\rm H} \frac{1}{\Delta I_{\rm H} \mathbf{s} \mathbf{s}}, \qquad (7.14, 8)$$

-317-

where $\Delta f_{m} = \hat{\Phi} - \hat{\Phi}$ effective width of the spectrum of modulation, determined by the usual method and equal to $1/\int_{-\infty}^{\infty} C(\tau) d\tau$ on the basis of the relationship between spectrum width and correlation time.

In general, proceeding from the form of the integral in formula (7.14.6) one should expect that

$$2N_{\rm H} = \frac{v r_{\rm h}^2 P_{\rm H}}{\sqrt{1 + 4J_{\rm H}^2 + \delta_{\rm h}^2}}.$$
(7.14.9)

日のというないので、「などのない」のです。「こう」

Effectiveness of the influence of pulse chaotic interference, except for the case when formula (7.14.8) is valid, depends not only on its mean power $\nu \tau_{\Pi} P_{\Pi}$, but also on pulse duration τ_{Π} , where with lengthening of the pulse its effectiveness increases.

Decrease of the signal-to-noise ratio under the influence of pulse chaotic interference on a coherent range finder can be characterized by an expression similar to (7.14.3),

$$\frac{A_{nm}}{h} = \frac{1}{1 + \frac{v\tau_n^2 P_m}{2N_0 \sqrt{1 + \Delta f_{max}^2 \sigma_n^2}} \frac{G_{ny}\lambda^1}{(4\pi)^2 d_n^2}},$$
(7.14.10)

7.14.3. Return Interference

Effectiveness of return interference in reference to range finders essentially depends on its concrete form. It is obvious that the return signal of a station of interferences, radiated from an object without delay relative to the reflected signal, is not actually interference, but only helps to measure target position data.

If the responder is located not on the target, so that the distances between the radar and the target and between the radar and the station of interferences differ by more than the magnitude of the resolution capability, thanks to the selective ability of discriminator with respect to range the signal of return interference does not affect work of the measuring system, of course, under the condition that preliminary selecting of the required target was already performed. "Multiplication" of the return signal also cannot lead to essential increase of the effectiveness of return interference, since for this it is required that the period of repetition of the multiplied signal be commensurate with the magnitude of the range resolution capability of the radar, which for a sufficiently high resolution

-318-

capability requires huge mean power of interference.

Interference with multiplication can cause a range finder to malfunction only if the signal of interference in every period of repetition of modulation of the signal appears at different, but sufficiently small as compared to resolution capability, delays with respect to the signal from the target. In this case interference will pass to the discriminator output, and with fair effectiveness it will either suppress the signal from the target with the help of an AGC system, or due to shifts of the signal of the responder in range it will lead to swinging of the tracking system and breakoff of tracking. If, however, one of the multiplied return signals constantly coincides with the signal from the target, it only increases accuracy of measurement, introducing, possibly, a certain systematic error upon nonprecise coincidence of signals of the station of interferences and those reflected from the target.

Additional circumstances, decreasing effectiveness of return interference with multiplication, are:

1) the preference for complicated forms of modulation in coherent radars and the connected difficulties of undistorted multiplication of signals with such modulation;

2) difficulties of preserving coherence of the return signal, which is necessary for accumulation of interference in narrow-band filters.

An effective means of affecting radio range finders using discriminators with switching of reference signals can be any return interference with low-frequency modulation of a regular of random character. Beats of spectral components of this modulation with the frequency of switching can give low-frequency components which are capable, without essential suppression, of passing through a closed-loop tracking system and causing it to swing.

An effective means of combatting all range finders can be return interference with variable delay, which with sufficient intensity and a reasonably selected rate of change of delay distracts the range finder from the target and forces it to track the interference. Physically, the action of return interference with variable delay, obviously, is equivalent to a single disturbing target, moving relative to the selected target, and the question of the character and results of action of the interference belongs, in point of fact, to the problem of resolution of targets in radar, more exactly to the problem of dynamic resolution in the process of tracking by a serve system. Here, one should establish the possibility of transition of the

-319-

process of transition, the dependence of these characteristics on the intensity and and rate of movement of the interfering signal, etc. Unfortunately, due to the necessity of solution of nonlinear nonstationary problems this question to this day has not been studied. It is possible to indicate only certain qualitative considerations about improvements of characteristics of range finders in the presence of such interferences. Rather obvious is the requirement of indication of the presence of an interferences. Rather obvious is the requireful and interfering signals, and subsequent compensation of the latter by various methods. Further improvement of the quality of work of the range finder is possible using distinctions between laws governing change of distance to the target and distance corresponding to delay of the interfering signal known a priori and studied in the process of measurement. All these questions need systematic investigation.

7.14.4. Passive Interference

In Chapter IV it is shown that during coherent reception a signal from a passive interference after multiplication by the reference signal and with narrowband filtration turns out to be, from the point of view of its influence on narrowband filters and its further transformations, statistically equivalent to a stationary normal random signal whose spectral density on the basis of results of Chapter IV is equal to

$$\mathbf{S}_{\mathbf{\pi}}(\boldsymbol{\omega}) = \frac{P_{\bullet}}{2\Delta f_{\bullet}} \cdot \frac{\mathbf{e}_{\mathbf{\pi}}}{\mathbf{e}_{\mathbf{\pi}}} \sum_{k=-\infty}^{\infty} f(\boldsymbol{\omega} - \Delta \boldsymbol{\omega}_{\mathbf{\pi}} + k\boldsymbol{\omega}_{\mathbf{r}}), \qquad (7.14.11)$$

where σ_{ii} - reflecting surface of the target;

 $\sigma_{\rm ff}$ - reflecting surface of the interference in the resolution volume of the radar;

 $\Delta \omega_{\rm p} = {\rm difference}$ between Doppler shifts of the signal from the target and free the interference;

$\mathbf{e}_r = 2\pi f_r;$

f(x) - describes the form of the spectrum of interference, where f(0) = 1.

In deriving this formula it was assumed that Doppler shifts are small as compared to the width of the spectrum of modulation.

As follows from (7.14.11), the intensity of interference, besides the total reflecting surface of all dipole reflectors, essentially depends on the difference of Doppler shifts and the frequency of repetition. Summation in this formula emphasizes the fact that interference is dangerous not only when $\Delta \omega_{\rm R} \approx 0$, but also when $\Delta \omega_{\rm R} \approx k\omega_{\rm r}$. Since in the overwhelming majority of practical cases the width of

-320-

the spectrum of interference is small as compared to the frequency of repetition, in formula (7.14.11) actually only one term is essential, the one for which difference $\Delta \omega_{\rm H} - k \omega_{\rm T}$ is minimal. In many cases selection of the frequency of repetition is made so that this difference for all $k \neq 0$ is great, and in (7.14.11) one should consider only the zero term. It is obvious that such selection of the frequency of repetition in general ensures the greatest immunity for a coherent radar with respect to passive interference, since here there decrease to a minimum the number of situations in which to the input of the narrow-band filter there proceed the most intense components of the spectrum of interference, corresponding to a maximum of function

$$f(\omega - \Delta \omega_{II} + k\omega_{r})$$
 when $\Delta \omega_{II} - k\omega_{r} \approx 0$,

The width of the spectrum of each of the bands of the signal reflected from the cloud of interferences is determined by irregularity of movement of the dipoles and the transmitter wavelength, and in the case of a moving radar also by the speed of the radar with respect to the cloud of interferences and the width of its radiation pattern. Depending upon wavelength, for a motionless radar the width of the spectrum is 20-100 cycles, and for a moving radar it is from tens to thousands of cycles per second. Such a width of the spectrum, other things being equal exceeds the width of the spectrum of the reflected signal and the matened transmission bandwidth of the narrow-band filters. Therefore, with certain approximations in coherent radio range finders passive interference also is equivalent to white noise with spectral density

$$2N_{\rm H} = \frac{P_{\rm o}}{2\Delta f_{\rm o}} \frac{\sigma_{\rm H}}{\sigma_{\rm H}} f(\Delta \omega_{\rm H}), \qquad (7.14.12)$$

where the frequency of repetition is assumed sufficiently great as compared to all possible values of $\Delta\omega_{\pi}$. The signal-to-noise ratio decreases here by a factor of

$$\frac{A_{\text{BH}}}{A} = \frac{1}{1 + A \frac{\sigma_{\text{B}}}{\sigma_{\text{B}}} i (\Delta \omega_{\text{A}})} \text{ times.}$$
(7.14.13)

Ratio (7.14.13) essentially depends on the ratio of the difference of Doppler shifts $\Delta f_{\Pi} = \Delta \omega_{\Pi}/2\pi$ to the width of the spectrum of interference Δf_{Π} . For large $\Delta f_{\Pi}/\Delta f_{\Pi}$ function $f(\Delta \omega_{\Pi})$ takes small values, and we achieve tuning away from passive interference at the expense of frequency selection during coherent reception. For small $\Delta f_{\Pi}/\Delta f_{\Pi}$, when there is no tuning, function $f(\Delta \omega_{\Pi}) \approx 1$ and the discriminator of the range finder is affected by the most intense components of the spectrum of interference. Here, by virtue of the relation existing in real situations

-321-

hs $_{\rm H}/\sigma_{\rm H} >> 1$, the signal-to-noise ratio sharply decreases and takes value h_{H,H} = $\sigma_{\rm H}/\sigma_{\rm H}$, depending only on the reflecting surfaces of the target and the interference. The magnitude of $\sigma_{\rm H}/\sigma_{\rm H}$ is frequently considerably less than unity. This means that with coincidence of speeds of the target and of the interference the meter with high probability may be caused to malfunction. Speeds of the target or, correspondingly, the angles between the speeds of the target and the interference with respect to the radar at which the difference of Dopplor frequencies of the target and the interference are close to zero are sometimes called "blind." Their location and magnitude depend on the concrete situation and are specially calculated in every case interesting us.

7.14.5. Intermittent Interference

Any of the active interferences may be radiated, not continuously, but from time to time interrupted at preselected or random moments of time. Interruption is used for concentration of high power of interference in definite intervals of time, and also for causing parametric action on tracking systems of the radar and swinging them, achieving breakoff of tracking.

The frequency of interruptions may be most diverse, and depending upon it the results of the interference on the range finder will differ somewhat. With fast-intermittent (as compared to the inertia of smoothing circuits of the range finder) interference it generally turns out to be equivalent to continuous interference of the same mean power. For instance, with periodic interruption with times of the action and absence of interference $T_1 = 0$ T_2 the power of radiation of interference should be multiplied by $T_4/(T_1 + T_2)$.

However, one should note that such equivalence of intermittent interference to continuous interference requires taking special measures to eliminate different parasitic factors in the radar receiver.

In particular, there should be taken measures to preventing overload in the limited range of the linear amplifiers of the receiver, so that after cessation of the action of interference the signal is passed by these amplifiers without distortion or suppression.

A second important element of the receiver, requiring special attention, is the AGC system. After powerful interference is switched on, the AGC system should rapidly develop its level, lowering amplification of the receiver, fixed before this from the signal level. Otherwise there will be different conditions during transmission of the useful signal and of interference, and the signal-to-noise ratio

-322-

will decrease as compared to the case of a linear receiver without automatic gain control.

After the interference is turned off, the AGC system, conversely, should rapidly increase amplification of the receiver, restoring its sensitivity to the signal. Only when the first process is fast as compared to the time of action of the interference and the second is fast as compared with the time of its absence is there no worsening of the signal-to-noise ratio as compared to the case of continuous interference of the same mean power.

A certain exception, for any frequency of interruption of interference, is a range finder using a discriminator with switching of reference signals. Application of such discriminators is generally undesirable from the point of view of the action of intermittent interferences, since beats of the frequency of switching with the frequency of interruption of interference may yield has some passing through the servo system and leading to its excitation.

With slow interruption of interference all processes in the range finder during the time of its action or absence can be established, and on the whole the influence of interference is characterized by the fact that the range finder works alternately, first with signal-to-noise ratio h, (when the interference is turned off) and then with signal-to-noise ratio h_{II M} (when the interference is on). Under the action of interference errors of measurement increase and can exceed permissible levels, and with sufficiently great interference strength there may occur breakoff of tracking.

Another characteristic feature is the fact that as a result of the dependence of the gain factor of the open loop of a range finder on the signal-to-noise ratio due to the normalizing properties of the AGC system during interruption of interference the range finder as a whole experiences a parametric influence. With sufficiently complicated smoothing circuits this influence will lead to parametric excitation of the tracking system and total disruption of its operation.

With an arbitrary frequency of switching of the interference calculation of its influence on a range finder requires special, rather complex investigation. Analyzing the serve system, one should consider that the gain factor of the discriminator and the equivalent spectral density of noise at the input of the range finder are functions of time; K = K(h(t)) and $S_{RCB}(t) = S_{OKTB}(h(t))$, taking values corresponding to the signal-to-noise ratio h at those moments of time when there is no interference and to the signal-to-noise ratio h $_{\Pi \ \square}$ when there is interference. We will indicate only the method of solution of the problem of calculation of dynamic and fluctuation errors in this case.

-323-

The pulse response of the closed-loop system $g(t, \tau)$ and the related pulse response $v(t, \tau)$ are determined as before by equations (6.2.12) and (6.2.13), in which, however, one should consider $t^{\mu} = \pm i n$ factor $K_{\mathbf{A}}$ is a function of time. Besides this, during calculation of the tion error $S_{\mathbf{SKB}}$ no longer can be removed from under the integral sign, and one should consider its dependence on time. For instance, in the case of a very simple smoothing filter in the form a single integrator the solution of equation (6.2.12) has the form

$$g(t, s) = K_{\mathbf{x}}K_{\mathbf{x}}(t) \exp \left\{-\int K_{\mathbf{x}}K_{\mathbf{x}}(s) ds\right\}. \qquad (7.14.14)$$

▲村田村村村村市でです。2 後についていたけの村村村でい

From this expression it follows that for any regular law of interruption of interference parametric excitation of a system with one integrator does not occur, since the pulse response remains limited. In this respect the considered example is not very informative. With chaotic interruptions with a certain probability even a first-order system may be excited. Fluctuation error in the considered example is determined by the following expression:

$$g_{\phi,x}^{2}(t) = \int_{0}^{t} g^{*}(t, \tau) S_{\theta K B}(\tau) d\tau =$$

$$= K_{\pi}^{2} K_{\pi}^{2}(t) \int_{0}^{t} e^{-2 \int_{0}^{t} K_{\pi} K_{\pi}(s) ds} S_{\theta K B}(\tau) d\tau, \qquad (7.14.15)$$

from which in this particular case there follow all the conclusions about the relative character of the influence of fast-intermittent and slowly intermittent interferences. Analysis of the effect of interferences on range finders with smoothing circuits of higher order can, in principle, be conducted analogously; however, it requires solutions of differential equations of the corresponding ord r with variable coefficients.

§ 7.15. Nonlinear Phenomena in Tracking Range Finders

The whole preceding analysis was based on a linearized presentation of the tracking radar range finder. Here, we considered that the signal at the discriminator output is proportional to the average current mismatch, and noises in most cases were considered independent of mismatch. In fact, as follows from the results of the first paragraphs of this chapter, the discrimination and fluctuation characteristics of discriminators of coherent range finders meet these requirements only with limited ranges of change of mismatch, which take place only with comparatively high

-324-

signal-to-noise ratios. In real conditions, due to the influence of interferences, which essentially decrease the magnitude of the signal-to-noise ratio, these requirements can not be satisfied, and it is necessary therefore to consider nonlinearity of the discriminator.

Subsequently we shall limit ourselves to a simplified situation, when it is possible to consider the fluctuation characteristic constant in required limits and equal to quantity S_{BGB} used everywhere earlier. This assumption is justified by the fact that calculation of nonlinearity of the discriminator is most interesting for comparatively small signal-to-noise ratios, and here irregularity of the fluctuation characteristic become less and less noticeable. Furthermore, due to demodulation of signal fluctuations by the AGC system there occurs additional leveling of the fluctuation characteristic, so that in real coherent receivers it does not differ greatly from a constant magnitude even for comparatively large signal-tonoise ratios. The discrimination characteristics of range finders is expressed by formulas for $\overline{z(t, \Delta)}$, obtained in § 7.2-7.5, according to which, for detunings 5 which are not very great, the discrimination characteristic of practically any of the range finders considered above is

$$m(\Delta) \doteq -KC(\Delta)C'(\Delta), \qquad (7.15.1)$$

where K - proportionality factor.

Henceforth, we shall consider only real $C(\Delta)$; therefore, in (7.15.1) we omit the sign of the real part.

The general approach to investigation of nonlinear phenomena in tracking measuring systems and certain results of more concrete content in reference to systems with a smoothing filter in the form of a single integrator are presented in Chapter VI. Basic attention there was paid to phenomenon of breakoff of tracking, by which we understood mismatch between the real and measured values of the measured coordinate reaching certain boundary-points of the discrimination characteristic. We calculated basic statistical characteristics of breakoff — the probability of reaching a boundary-point in a certain time t and the mean time to the first reaching of a boundary-point — by which it is possible to judge the critical magnitude of equivalent, spectral density and, correspondingly, of the signal-to-noise ratio, already impermissible for normal work of the measuring system.

Along with such an approach to the investigation of nonlinear phenomena in a tracking meter, based on the assumption of the absence in it of steady-state operating conditions, we may also use a solution with reflecting screens at the points

-325-

corresponding to limitation of the output of the measuring system. Such limitation practically always exists in tracking meters, and its approximate calculation with the help of reflecting boundaries, valid for a signal-to-noise ratio which is not too small, permits us to obtain the stationary distribution of probabilities and with its help the errors of measurement in steady-state operation. We shall consider several examples of calculation of various characteristics for concrete forms of discriminators of range finding meters.

7.15.1. Average Time to Breakoff of Tracking

Using formulas (6.3.26) and (7.15.1), it is possible with the help of simple transformations to obtain the following exact expression for the mean value of the time of the first attainment of a boundary-point of the discrimination characteristic, equal to Δ_0 , under the condition that initial mismatch was zero:

$$T = \frac{1}{44 \int_{-\infty}^{0} \sigma_{x}^{2}} \int_{0}^{\infty} dy \int_{0}^{\pi} dx e^{\frac{C^{2}(y) - C^{2}(x) - \frac{2V}{K_{x}K_{x}}(y-x)}{2\delta \sigma_{x}^{2}}}, \qquad (7.15.2)$$

where $\sigma_{\pi}^2 = 2\Delta f_{0} \phi S_{0}$;

- $\Delta f_{\partial \phi}$ variance of fluctuation error and the effective bandwidth of the tracking system in the linearized consideration;
 - V speed of the tracked target, assumed constant in time intervals \sim the order of T;

$$b = -C^{n}(0)$$
 - mean square width of the spectrum of modulation of the sound-
ing signal.

Considering that function $C(\Delta)$ actually depends on $\sqrt{b}\Delta$, we replace $\sqrt{b}y$ by η and $\sqrt{b}x$ and ξ , designate the relative magnitude of variance of fluctuation error in the linearized system and the relative magnitude of dynamic error

$$b\sigma_A^2 = \mu^2, \quad \frac{\sqrt{b}V}{K_BK_A} = a$$

and introduce designation $C(\Delta) = a(\sqrt{b}\Delta)$. Then the final expression for the average time to breakoff can be given the form

$$T = \frac{1}{44 \int_{0.00}^{0.00} d\eta} \int_{0}^{0} d\xi \exp\left\{\frac{a^{0}(\eta) - a^{0}(\xi) - 2a(\eta - \xi)}{2\mu^{0}}\right\}, \qquad (7.15.3)$$

where one should consider that a''(0) = -1. With initial mismatch Δ differing from zero, the lower limit of integration over η is replaced by $\sqrt{b}\Delta$.

Analytic calculation of quadratures in (7.15.3) for practically interesting

- 326 --

functions a(x) is impossible; therefore the average time to breakoff must be calculated by numerical or graphical integration. From formula (7.15.3) it follows that the average time to breakoff, expressed in units of the time constant of the tracking system $1/\Delta f_{g\bar{\Phi}}$, depends only on the relative magnitude of fluctuation error of the linearized system, the boundary value of the discrimination characteristic and the relative magnitude of dynamic error. Qualitative investigation of this formula shows that the magnitude of T rapidly decreases with increase of dynamic error and variance of fluctuation error. In order to estimate the influence of dynamic error on breakoff, we shall consider the asymptotic case of high-level noises, when $\mu^2 = b\sigma_A^2$ is great (of the order of unity or larger). Then, from (7.15.3) it follows that

$$T = \frac{V \delta \Delta_0}{4 \alpha \Delta f_{0,0}} = \frac{\Delta_0}{4 \epsilon_{\mu} \Delta f_{0,0}} = \frac{\Delta_0}{V}, \qquad (7.15.4)$$

i.e., the average time to breakoff is equal to the time of movement of the target the magnitude of the interval from zero to the boundary-point of the discrimination characteristic.

From expression (7.15.3) we can obtain the approximate formula, valid for noises at not too low a level and zero dynamic error,

$$T = \frac{1}{8\Delta_{1}^{\prime} \cdot \phi} \left(\frac{\Delta_{0}}{\sigma_{\pi}}\right)^{4} \left[1 + 0_{1} \frac{\partial 8}{\partial (\sigma_{\pi})}^{4} + 16 \cdot 10^{-4} \left(\frac{\Delta_{0}}{\sigma_{\pi}}\right)^{4}\right], \qquad (7.15.5)$$

which was already given in Chapter VI. This expression depends only on the ratio of the width of the assigned zone of the discrimination characteristic to the fluctuation error of the linearized system. Graphs of this expression and the investigation of it given in Chapter VI show that the average time to breakoff rapidly increases with growth of Δ_0/σ_{π} , where as the threshold value of this ratio, ensuring work of the tracking system for a sufficiently large number of its time constants, we can select quantity $\Delta_0/\sigma_{\pi} \simeq 5$ to 7. With further increase of this ratio the phenomenon of breakoff practically is not observed.

Assigning magnitude $\Delta_0 \sim 2/\sqrt{b}$, we obtain for the threshold value of μ a quantity of the order of 0.3-0.4, which permits us to find the permissible magnitude of the signal-to-noise ratio in a coherent range finder. In particular, with a discriminator close to optimal, when $S_{\partial KB} \sim S_{OHT}$, the critical signal-to-noise ratio is determined by expression

$$\frac{\Delta f_{e0}}{\Delta f_{e}} F(h_{\pi p}) = 0,1 + 0,15, \qquad (7.15.6)$$

where F(h) is given by formula (7.2.15).

-327-

Dependence of the critical signal-to-noise ratio $h_{\rm KP}$ on the ratio of bandwidths $\Delta f_{\rm c} / \Delta f_{\rm KD}$ for the case of an exponential correlation function for fluctuations is





shown in Fig. 7.52. For other correlation functions the magnitude of $h_{\rm KF}$ experiences only immaterial changes relative to the given law. The magnitude of $h_{\rm KF}$, naturally, decreases with decrease of ratio $\Delta f_{\rm C} / \Delta f_{\rm BOO}$, and for practically interesting values of this ratio it is in the range of 0.1 to 3.

The dependence of the average time to breakoff, calculated by exact formula with the help of numerical integration, on μ is shown in Fig. 7.53

for $\sqrt{b}\Delta_0 = 2$ and a Gaussian autocorelation function for the sounding signal $a(x) = \frac{x^2}{2}$

This relationship very exactly coincides with that obtained by approximate formula (7.15.5).



Fig. 7.53. Dependence of average time to breakoff on error of measurement in a linearized system. Use of other approximations of a(x), corresponding to other methods of modulation of the sounding signal, does not lead to essential changes, since for all smooth functions of a(x) the febehavior within limits of the basic maximum is approximately identical. Comparison of the curve of Fig. 7.53 with the curve in Fig. 6.7, constructed by the approximate formula, shows the good coincidence and the satis-

factoriness of approximation of the expression for average time to breakoff by formula (7.15.5).

Let us consider for illustration one concrete example of analysis of phenomena

of breakoff of tracking in a coherent range finder. Let us assume that on the range finder there acts an active noise interference with spectral density $P_{\Pi} = G_{\Pi\Pi} / \Delta f_{\Pi}$ and distance $d_{\Pi} = 300$ km. The receiving antenna of the radar has diameter 1.5 m, and the noise factor is 5. Let the signal-to-noise ratio h in the range finder receiver in the absence of interference by equal to 50. Then the signal-to-noise ratio in the presence of interference is

 $h_{\rm HHH} = \frac{50}{1+100\frac{P_{\rm H}G_{\rm HH}}{\Delta I_{\rm H}}},$

where $P_{\Pi} = G_{\Pi\Pi} = /\Delta f_{\Pi}$ in watts/Mc.

Assuming that the discriminator of the range finder is close in characteristics to optimum, the spectrum of signal fluctuations is square with width $\Delta f_c = 30$ cycles, the effective bandwidth of the system is equal to 1 cycle, and taking boundary value of mistuning $\Delta_0 = 2/\sqrt{b}$, we obtain

$$\frac{\Delta_0}{\sigma_{\rm s}} = \frac{111}{\sqrt[\gamma]{(1+100P_{\rm s}G_{\rm ss}/\Delta f_{\rm s})}(1+2P_{\rm s}G_{\rm ss}/\Delta f_{\rm s})}}.$$

This relationship permits us to find the average time to breakoff as a function of the spectral density of interference. This function is shown in Fig. 7.54, from



Fig. 7.54. Dependence of average time to breakoff on the spectral density of interference.

This function is shown in Fig. 7.54, from which it is clear that a spectral density of interference near 1 watt/Mc in certain cases still gives an acceptable value of the average time to breakoff of the order of 50 sec. Slight decrease or increase of the spectral density of interference leads correspondingly to sharp increase or decrease of average time. With change of spectral density in any direction by a factor of two there occurs more than a tenfold change of average time to breakoff. These results emphasize the great criticality of a range finder to the level of interferences acting on them.

7.15.2. Fluctuation Error of Measurement Taking into Account Nonlinearity As already noted above, breakoff of tracking, understood as the first time the magnitude of mismatch reaches a certain conditional point on the falling section of

-329-

the discrimination characteristic, cannot be, of course, an adequate description of the behavior of a tracking system during allowance for its nonlinearity.

Reaching a boundary-point, mismatch with noticeable probability may again take a small value, and the system will work normally, but, of course, with large errors. Therefore, the magnitude of mismatch, going beyond certain limits, still does not signify breakoff of tracking in the direct sense. Experimental study of nonlinear conditions of tracking meters also emphasizes the applicability of an approach - based on the assumption of the absence in a nonlinear system of steady-state operating conditions.

In tracking range finders such an approach gives the correct answer for very large interferences and when tracking in comparatively small intervals of time is of interest. In examining the behavior of a system in large time intervals, due to the practically inevitable limitation on output in it, there sets in steady-state operating conditions, characteristics of which in a number of cases are of the greatest practical interest. Calculation of statistical characteristics of these conditions, in particular variance of fluctuation error, permits us to find the critical magnitude of the signal-to-noise ratio, at which fluctuation error still is within permissible limits. Comparison of this critical magnitude with the corresponding quantity, calculated in the preceding paragraph, permits us to characterize more fully nonlinear operating conditions of the meter and more completely find conditions when linearization of a *racking radar range finder is permissible.

Using formulas (6.3.22) and (7.15.4), it is possible to reduce the expression for variance of fluctuation error of measurement, found during idealization of the limitation at the system output by a reflecting screen at points $\pm \Delta_1$, to the following form:

 $\mathbf{c}^{\mathbf{a}} = \frac{1}{\mathbf{b}} \frac{-\mathbf{v}_{\mathbf{b}\mathbf{a}_{1}}}{\mathbf{v}_{\mathbf{b}\mathbf{a}_{1}}} \frac{\mathbf{c}^{\mathbf{a}}(y)}{\mathbf{c}^{\mathbf{a}}} , \qquad (7.15.7)$

where all designations are the same as in Paragraph 7.15.1.

For low-level noises, when μ is small, from this expression it follows that

12.

$$\sigma^{2} = \frac{1}{\delta} \mu^{2} = \sigma_{a}^{2}$$
 (7.15.8)

For high-level noises, when $\mu >> 1$,

$$s^{2} = \frac{4^{3}}{3},$$
 (7.15.9)

i.e., the magnitude of variance here is determined only by the level of limitation on the output of the system, and the distribution of probabilities for mismatch becomes uniform.

Expression (7.15.7) allows us for certain forms of a sounding signal to perform analytic calculation. For instance, for a Gaussian $C(\Delta)$

$$b\sigma^{2} = \frac{\frac{1}{3} (\sqrt[y]{\delta \Delta_{1}})^{2} + \frac{\sqrt{\pi}}{4} \sum_{A=1}^{\infty} \frac{1}{(2\mu^{2})^{A} A^{2}/4 A!}}{\sqrt[y]{\delta \Delta_{1}} + \frac{\sqrt{\pi}}{2} \sum_{A=1}^{\infty} \frac{1}{(2\mu^{3})^{A} A^{2}/4 A!}}, \qquad (7.15.10)$$

where it is assumed that $\sqrt{b}\Lambda_1$ in accordance with real conditions is much larger than



Fig. 7.55. Dependence of error of range finding on $\mu = \sqrt{b}\sigma_{\pi}$ for various levels of limitation $\sqrt{b}\Delta_{4}$.

one. A similar expression can be obtained for cosinusoidal function $C(\Delta)$, and also for a series of other autocorrelation functions. Relative variance of error of measurement as a function of μ is shown in Fig. 7.55 for various values of the relative level of limitation $\sqrt{b}\Delta_1$. From formulas (7.15.7) and (7.15.10) and this figure it follows that already at $\mu = 0.3$ and at $\sqrt{b}\Delta_1 > 5$ to 10 the magnitude of error is determined to a significant degree by the

level of limitation. For larger μ it is practically equal to the asymptotic value from formula (7.15.9).

The curves of Fig. 7.55 show that the dependence of fluctuation error on the signal-to-noise ratio (error of a linearized system) also has a clearly expressed threshold character. Critical values of μ , corresponding, for instance, to fluctuation error $\sigma = 2/\sqrt{b}$, are in this case somewhat smaller than the same values found from the average time to breakoff of tracking (of the order of 0.15-0.3). In accordance with this permissible signal-to-noise ratios h increase somewhat. However,

-331-

this increase is not so considerable that one should take it into account.

Formulas (7.15.7) and (7.15.10) and the curves of Fig. 7.55 permit investigation of another interesting question — the limits of applicability of the linearized approach to a tracking range finder. From Fig. 7.55 it follows that equality of c and σ_{π} is preserved approximately to values $\mu \approx 0.12$ -0.15. This range of values is very narrow so that at practically any level of limitation nonlinearity of the discriminator of a range finder starts to appear at the same signal-to-noise ratio. The magnitude of the signal-to-noise ratio at which linearized consideration of a tracking range finder with a discriminator close to optimum is still permissible is determined by this magnitude and can be found from equation

$$\frac{\Delta I_{00}}{\Delta I_{0}}F(h_{xp}) = 0.017. \tag{7.15.11}$$

The dependence of h_{RP} on the ratio of bandwidths $\Delta f_c / \Delta f_{pq}$ for the case of a square spectrum of fluctuations is shown in Fig. 7.56. For real relationships of



Fig. 7.56. Dependence of the critical signal-to-noise ratio on $\Delta f_{c} = /\Delta f_{ab}$.

bandwidth of the system and the width of the spectrum of fluctuations h_{KP} varies from 0.2 to 6, which approximately exceeds by a factor of 2 the corresponding values calculated from the critical magnitude of the average time to breakoff. Thus, breakoff of tracking in the sense of the preceding paragraph occurs at approximately the same signal-to-noise ratios at which allowance for nonlinearities of the discriminator becomes necessary for calculating fluctuation error.

Steady-state operation of a tracking range finder in nonlinear conditions was

studied experimentally. A range finder was simulated by an analog computer. In the simulation we reproduced a nonlinear discrimination characteristic and established limitation on output of the system. Analysis of results of the experiment showed satisfactory coincidence with the theoretically found probability distributions for mismatch obtained by solution of the boundary value problem for a Fokker-Planck equation with reflecting screens. In the course of the experiment we also studied characteristics of nonstationary conditions. We investigated change of the distribution of probabilities of mismatch with time and found the dependence of

-332-

fluctuation error on time. Data of the experiment, in general, well confirm conclusions of theoretical analysis, but for high-level noises there are certain divergences, caused by the approximate nature of the allowance for limitation on output. For illustration, in Fig. 7 7 are theoretically calculated and experimentally taken probability densities for mismatch in a range finder with one integrator with an effective bandwidth of the tracking system $\Delta f_{g\bar{\Phi}} = 1.5$ cycles and the discrimination characteristic shown in Fig. 7.58. Curves correspond to the ratio of error of a linearized system to the half-width of the discrimination characteristic σ_{π} / Δ_{0} , equal to 0.44 and 1.1.





-333-



Fig. 7.58. Discrimination characteristic of the range finder in the experiment.

We shall consider one illustrating example of calculation of errors of range finding taking into account nonlinearity in a coherent range finder. We consider that the sounding signal of the radar is modulated by pulses close in form to Gaussian with linear frequency modulation, where frequency

deviation due to the prolonged nature of the pulse is great as compared to the width of its spectrum. Then the autocorrelation function of the sounding signal is described by a Gaussian curve with parameter

$$b = \frac{2 \cdot \frac{3}{n}}{n}, \bullet_{m} = \frac{a \cdot \frac{1}{2}}{2},$$
 (7.15.12)

where a - rate of change of frequency;

 $\tau_{_{\rm H}}$ - effective pulse duration.

Let frequency deviation $2\omega_{\rm m} = 2\pi \cdot 10^6$ rad/sec. Then, $\sqrt{b} = 2.5 \cdot 10^6$ sec⁻¹. We consider that limitation on output of the range finder is determined by natural factors — mismatch reaching a magnitude equal to the period of repetition of the signal. Considering the frequency of repetition equal to 25 kilocycles, we find $\sqrt{b}\Delta_1 = 100$. We consider that the signal-to-noise ratio $h_{\Pi\Pi}$ does not depend on distance and is equal to

$$h_{\text{fig}} = \frac{h_0}{\frac{P_a G_{BB}}{2\Delta f_B N_0} - \frac{G_{BB} \lambda^2}{(4\pi d_0)^3}}, \qquad (7.15.13)$$

where $h_0 - signal-to-noise$ ratio in the absence of interferences at distance d_0 , corresponding to the moment of switching on of interference.

Assigning values $h_0 = 100$, $d_0 = 25$ km, the diameter of the antenna equal to 30 cm, and noise factor of the receiver of the radar equal to 10, we obtain

$$h_{\rm max} = \frac{\Delta f_{\rm m}}{P_{\rm m}G_{\rm max}},$$

where spectral density $P_{\Pi} = G_{\Pi\Pi} = /\Delta f_{\Pi}$ is expressed in watts/Mc. Assuming that the discriminator of the range finder is close to optimum, the width of the spectrum of the reflected signal is 100 cycles, and the bandwidth of the tracking system is 4 cycles, we obtain the following expression for variance of fluctuation error of the linearized system:

$$\mu^{2} = b\sigma_{a}^{2} = 0.04 \frac{P_{a}G_{ua}}{\Delta f_{a}} \left(1 + \frac{P_{a}G_{au}}{\Delta f_{u}}\right),$$

which gives the possibility of determining with the help of the graphs of Fig. 7.55



error of range finding taking into account nonlinearity. Dependence of this error on the spectral density of the radiated interference is shown in Fig. 7.59. As follows from this figure, the dependence of accuracy of range finding on the spectral density of interference has a clearly expressed threshold character. With change of $P_{\Pi} G_{\Pi\Pi} / \Delta f_{\Pi}$ from 0.5 watts/Mc to 1 watt/Mc error increases by more than 60 times. The critical value of spectral density of interference comprises in the considered example about 0.6-0.7 watt/Mc.

For high levels of interference accuracy of range finding becomes too low.

§ 7.16. Conclusion

This consideration of coherent range finders showed that in a significant majority of cases

the above-described methods of realizing optimum operations permit us to obtain accuracy close to that which is potentially possible. This particularly pertains to discriminators of tracking radio range finders, which with sufficiently careful selection of their parameters, turn out to be close to optimal in their characteristics. Analysis in this chapter allowed us to find the influence on characteristics of a discriminator and on accuracy of range finding in general of parameters of modulation of the sounding signal, power features of the received signal and imperfections of its generation in the circuit of the radio range finder. Not dwelling on tiresome, detailed enumeration of the results of this chapter, we note only those directions in which there remain definite gaps and where there is required additional investigation.

Questions of the analysis of discriminators can be considered sufficiently exhaustively investigated. Here we should only consider in more detail the influence of the AGC system on discriminator characteristics for comparatively low signal-tonoise ratios, when the cross interaction of the signal with noise in the AGC system is substantial, and there occurs modulation of the signal by noises passing through the feedback circuit. Another question is more detailed consideration of the case of slow fluctuations, when during the time of processing of the signal in the

-335-

Fig. 7.59. Dependence of error of range finding on the spectral density of noise interference.

discriminator signal amplitude remains constant.

The case of fluctuations which are slow or comparable in speed with the rate of variation of parameters, in general, is of great interest from the point of view of synthesis of measuring systems and, in particular, of ranging meters. Very interesting is allowance for the non-Gaussian character of signal fluctuations, which in a number of cases takes place.

The next important question requiring investigation is consideration of the range finders in § 7.2 of the type using for measurement a unit of detection channels. Here, it would be useful to study the influence of overlapping of channels and nonorthogonality of their output signals, to calculate characteristics of such a meter allowing for disharmony of reference signals in the unit of detection channels with the sounding signal and change of the measured distance.

Many problems of concrete content still remain in the question of analysis of a range finder as a whole. It would be interesting to consider examples of smoothing circuits of more complex structure, especially optimum smoothing circuits, corresponding to a quasi-regular law of change of the measured distance. Here, the most important question is a sufficiently deep explanation of the influence of errors in the statistics used during synthesis as compared the real statistics. Especially interesting is investigation of cases when such errors have a qualitative character; however, interesting, too, would be more detailed investigation of the influence of quantitative errors. An important question is consideration of range finders with smoothing circuits in the form of filters with a finite memory. The requirement of finiteness of memory can become essential in optimum smoothing circuits with variable parameters when their structure is very complicated, and it causes great difficulties in technical realization.

Above we already discussed the importance of investigation of operating conditions of the tracking range finder, when in its selection zone there are several targets or a target and delayed signals of return interference. At present this, in point of fact, is an open domain. At the same time investigation of the process of resolution of targets by a tracking range finder permitted us to establish new laws, to find requirements on the sounding signal and on methods of construction of range finders.

And, finally, problems connected with the investigation of nonlinear conditions of tracking range finders also are very far from complete solution. In this area we need more precise definition both of already obtained results with consideration of a large number of quantitative examples, and also solution of new problems. It

-336-

would be very interesting to obtain other characteristics of nonstationary operating conditions, and also definitized characteristics of steady-state operating conditions with allowance for limiting. It is important to consider nonlinear phenomena in systems with smoothing circuits of higher order and with smoothing circuits in the form of filters with variable parameters.

All these questions are not, however, decisive, and the level of understanding of problems pertaining to methods of construction of range finders and characteristics of accuracy of their work already attained permits us to competently approach their practical development. Available knowledge gives development engineers the possibility in a well-founded and reasonable manner of selecting the structure both of the discriminator, and of smoothing circuits of a range finder, of correctly selecting their parameters and of fully intelligently making various simplifications and technical compromises, having estimated their influence beforehand.

CHAPTER VIII

RANGE FINDING WITH AN INCOHERENT SIGNAL

§ 8.1. Introduction

An incoherent pulse signal at present is widely used in various radars. Along with normal pulse modulation there are used signals of more complicated form with additional intrapulse modulation of various forms [35-39]. Properties of an incotrent signal are described in Chapter I. Theoretical synthesis of range discriminators for the case of an incoherent fluctuating signal is hampered by the fact that for such a signal the general form of the likelihood functional is unknown. Therefore, during solution of the problem of synthesis it is necessary to limit ourselves only to certain particular cases and to make definite assumptions, sometimes not covering all cases in practice. In contrast to this the problem of analysis of existing discriminators can be solved with the same degree of correctness as in the case of a coherent signal. This to an identical degree pertains both to the work of the range finder with only its own noises, and also to cases of the influence of various kinds of interferences.

As for problems of analysis of a range finder as a whole and synthesis of smoothing circuits, it is obvious that the form of the signal introduces nothing new. Actually, as soon as we have determined characteristics of the discriminator and we have assigned statistics of the measured distance, it already makes absolutely no difference to what these characteristics correspond. Therefore, after we have obtained a solution of the corresponding problems for coherent range finders, their solution for incoherent range finders reduces to trivial replacement of characteristics of the one discriminator by characteristics of the other. In connection with this the present chapter to a still lesser degree than the preceding one will deal

-339-

Preceding page blank

·加利加利用。如何是在自己的人们的人们的最高级。如何就是有这些有效的。""这个你是我们的是是是有些这些人的。""你们是我们的,我们就是我们就是我们就是我们的。"

with questions of analysis of range finders as a whole. Desic attention will be allotted namely to discriminators of incoherent range finders. 1

ar an ha deal a faith and a second

The present chapter is organized on the same plan as Chapter VII. First, we consider possibilities and certain particular results of synthesis of an optimum discriminator. Then in general form we investigate basic methods of creating a discrimination characteristic taking into account a series of inevitable technical deviations from optimality. Then, we consider different concrete forms of modulation, give examples of analysis of range finders as a whole, study the influence of interferences on incoherent distances and touch on nonlinear phenomena in such range finders. As will be clear subsequently, many results will permit broad analogies with, and will sometimes even coincide with the coherent case.

§ 8.2. Optimum Discriminator

To find operations of an optimum discriminator it is necessary to have the logarithm of the likelihood functional for an incoher nt signal. In § 5.2 it is shown that it can be presented in the form of a certain function [see (5.2.3)] of the voltage envelope, corresponding to the value of the correlation integral formed in the j-th period of repetition of the cignal, i.e.,

$$P(y/\tau) = C \int_{0}^{2\pi} \cdots \int_{0}^{2\pi} \exp\left\{\sum_{j,k=0}^{n} \sigma_{jk} |f_j| |f_k| e^{i(\alpha_j - \alpha_k)}\right\} \times P(\alpha_1, \ldots, \alpha_n) d\alpha_1, \ldots, d\alpha_n,$$

wl.ere

 $a_k - value of the random phase in the k-th period;$ $<math>p(a_1, a_2, ..., a_n) - distribution of probabilities for phases;$

> v_{jk} - elements of a matrix equal to the difference of matrices which are reciprocal to correlation matrices for noise and for a signal mixed with noise in the absence of modulation;

$$\|f_{j}\| = \left| \int_{(j-1)T_{r}}^{T_{r}} y(s) u(s-r) e^{i m_{0} s} ds \right|.$$
 (3.2.1)

Here all designations are the same as in the precoding chapter.

Quantity $|f_j| = |f(jT_r)|$ obviously coincides with the value of envelope $|Q(t, \tau)|$ from (7.2.4), if pulse response h(t - s) from (7.2.4) is

$$h(t-s) = 1 \text{ when } (j-1) T_r < t-s < jT_r,$$

$$h(t-s) = 0 \text{ when } t-s < (j-1) T_r, \ t-s > jT_r.$$
(3.2.2)

This definition of pulse response n(t - s) corresponds to the fact that due to properties of the incoherent signal its coherent processing is possible only in an interval of time equal to one period, and further accumulation of the signal can be carried out only incoherently. Pulse response (8.2.2) defines an integrator over

-340-
period T_r with clearing [?].

The form of function (5.2.3), determining the logarithm of the likelihood functional, can be made concrete only with additional assumptions about the character of fluctuations of the signal and the magnitude of the signal-to-noise ratio.

In § 5.2 it is shown that with certain conditions the logarithm of the functional of probability density reduces to the sum of the squares of envelopes $|f_j|$, taken in all periods corresponding to the time of observation. This takes place:

a) for small signal-to-noise ratios and arbitrary speed of fluctuations;

b) for fluctuations, fast as compared to period T_r , and an arbitrary signal-tonoise ratio;

c) for large signal-to-noise ratios and arbitrary speed of fluctuations, if the spectral density of fluctuations of the signal differs from square.

In general the sum of squares is replaced by a certain inertial nonlinear processing which considers the interperiod correlation of values of the envelope. Accumulated experience in designing systems based on simple summation of the squares of the envelope permits us to state that consideration of complicated systems considering interperiod correlation will lead only to slight improvement of their characteristics. This statement is based on the fact that with a sufficiently large signal-to-noise ratio incoherent range finders built without taking into account interperiod correlation ensure, as will be shown later, practically the same accuracy as coherent ones, and with a small signal-to-noise ratio, when impairment of accuracy in the incoherent case is considerable, simple processing is optimum.

From this point of view, complication of processing of the received signal, ensuing from calculation of interperiod correlation, is hardly desirable, since it increases technical complexity of realizing range finders without leading to essential improvement of accuracy. Therefore, subsequently, we shall consider that the logarithm of the likelihood functional is presented in the form of the sum of squares of envelope $|f_{i}|$, i.e.,

$$P(y/\tau) = C \exp\left\{K \sum_{j=1}^{n} |f_j|^{s}\right\} = C \exp\left\{K \sum_{j=1}^{n} \left|\int_{(j-1)T_{r}}^{T_{r}} y(s) u(s-\tau) e^{i\omega_{s}s} ds \right|^{s}\right\}, \quad (3.2.3)$$

where

C — factor, not depending on delay τ ;

 $n = [T/T_n]$ - number of periods of the signal accumulated during time T;

K - a certain coefficient, depending on the signal-to-noise ratio. Such a presentation is sufficiently accurate in the above-considered cases, and, of course, for n = 1, i.e., for the case of a single sending of the signal. In

-341-

accordance with expression (8.2.3) we shall consider only such a construction of discriminators which does not take into account interperiod correlation of the envelope. Namely thus are existing incoherent range discriminators built - processing of the signal for one period is carried out in them coherently, and interperiod accumulation is carried only with respect to the envelope without taking into account correlation of its values in various periods.

8.2.1. Operations of an Optimum Discriminator

In accordance with results of Chapter VI, output of the discriminator in this case is defined as

$$z(t) = K \frac{\partial}{\partial \tau} |f_j|^2 = K \frac{\partial}{\partial \tau} \left| \int_{(j-1)}^{j_{\tau}} y(s) u(s-\tau) e^{i\omega_0 \tau} ds \right|^2.$$
(8.2.4)

It is obvices that z(t) - discrete random process, taking values $z_j = z(jT_r)$ in the j-th period. The concrete form of this process depends on the method of formation of correlation integral f_j , determining optimum processing of the signal in one period.

As we already noted above, this integral can be formed by two methods — the correlation method and by an optimum filter with frequency response matched with the spectrum of modulation. In the first case there is produced multiplication of the received signal by the sounding signal, shifted in frequency and in delay, and integration over the period with the help f filter (8.2.2), tuned to intermediate frequency. Actually, of course, integration can be carried out by any filter with a passband considerably larger than the frequency of repetition of the signal, and considerably smaller than the width of the spectrum of modulation. In this case the signal at the output of an optimum detection channel is a periodic sequence of . quarks of the envelopes of responses of this filter in every period under the lift. ence at the input of the filter of a signal consisting of the product

$$\boldsymbol{y}(t) \boldsymbol{\mu}(t-\tau) e^{i (\boldsymbol{u}_{0}+\boldsymbol{\omega}_{gp})t}.$$

During formation of z(t) in the range discriminator, the square of the envelope is replaced by the product

$$z(l) = -K2\text{Re} \int_{(l-1)T_r}^{lT_r} y(s) u(s-\tau) e^{l\omega s} ds \times \int_{(l-1)T_r}^{lT_r} y(s) u^{\omega r}(s-\tau) e^{-l\omega s} ds, \qquad (3.2.5)$$

which corresponds to multiplication of the received signal in one channel by $u(t - \tau)$, and in the other by $u^{*'}(t - \tau)$ and subsequent multiplication in the phase detector of output voltages of these channels.

-342-

Thus, in this respect there is complete analogy with the coherent case. The whole difference reduces to difference of pulse responses of filters (8.2.2) and (7.2.5). The actual output voltage of the discriminator z(t), as a function of time, can, as before, be presented by formula (7.2.7), where h(t) - pulse response of any filter whose inertia is low as compared to the duration of the period, so that the response of this filter to an incoming pulse attenuates by the moment of appearance of the next pulse.

「「「「二」」

With the correlation method of processing creation of a discrimination characteristic is possible only in circuits preceding detection. Here we can use both an exact method — multiplication of the received signal in two channels by a delayed sounding signal and its derivative — and also approximate methods — either simultaneous, or alternate multiplication of the received signal by two reference signals detuned with respect to delay.

With the second method of processing a pulse signal the received signal is passed through a filter with pulse response h(t) = u(-t) (such a filter is frequently called optimum for the given form of modulation, or "shortening," or matched with the spectrum of modulation). It is obvious here that the value of the output voltage envelope of this filter at time $t = (j - 1)T_r + \tau$ gives the needed value of $|f_j|$. These values should be selected by gating the envelope by a narrow gate pulse and then should be accumulated discretely.

Thus, in this case output voltage of the discriminator is a discrete random process consisting in the optimum case of infinitely narrow pulses appearing at moments $t_j = (j - 1)T_r + \tau$. Operations of the optimum discriminator in this case consist in transmission of the received signal through filters with pulse responses u(-t) and $u^{*'}(-t)$, subsequent phase detection of output voltages of these filters and gating of the output voltage at moments $t_j = (j - 1)T_r + \tau$, where $\tau -$ output value of delay. Approximate formation of optimum operations in the envelope of output voltage of a "shortening" filter gives the possibility of forming the likelihood functional for all values of τ . Creation of the discrimination characteristic here is carried out by two delay-detuned gates, tuned in accordance with the output value of delay. In practice, thanks to the great technical simplicity in real discriminators differentiation of the logarithm of the likelihood functional is carried out with respect to the envelope (see Chapter IV and § 7.6).

-343-

With optimum fulfillment of all operations in (3,2.4) and (3,2.5) both methods of construction of the discriminator are absolutely identical. This follows directly from (3.2.5), since both methods are the realization of one and the same mathematical expression. In real conditions, when the required operations are executed imperfectly, these methods can lead to somewhat differing results due to the varying influence of any kind of deviations from optimality. Therefore, subsequently, when necessary, we shall conduct parallel consideration of them.

8.2.2. Characteristics of an Optimum Discriminator

For calculation of characteristics of a discriminator, we record the input signal in the form

$$y(t) = \sqrt{P_e} \operatorname{Re} \sum_{k} E_{k} u_{e}(t - \tau_{e} - kT_{r}) e^{i t_{k} + i m_{e} t} + n(t), \qquad (8.2.6)$$

where $E_k = E(kT_r)$ - the envelope of the reflected signal in the k-th period ($\overline{E}_k^2 = 2$);

 α_k - independent random phases, evenly distributed in the interval (0 to 2π);

 τ_0 - true value of delay.

Function $u_0(t)$ is no longer periodic, but describes a single pulse and is normalized as before by unit mean power.

In all expressions describing real discriminators there is product $y(t)e^{-0}$; therefore, we introduce complex signal

$$\eta(t) = y(t) e^{i u_{t} t}$$
 (8.2.7)

and seek necessary statistical characteristics namely for it. Signal $\eta(t)$ according to (8.2.6) contains a low-frequency component and a component of frequency $2\omega_0$. With any fulfillment of the discriminator the last component does not pass to its output; therefore, in expressions for statistical characteristics of $\eta(t)$ we retain only low-frequency components, considering that integrals containing high-frequency components of the correlation function of signal $\eta(t)$ all the same turn into zero.

Let us find the low-frequency component of the correlation function. Dropping high-frequency components, we obtain

$$\overline{[\mathfrak{g}(t_1)\mathfrak{g}^{\bullet}(t_2)]}_{\mathfrak{h}\mathfrak{h}} = \frac{P_{\bullet}}{4} \sum_{j,k} u^{\bullet}_{\bullet j}(t_1) u_{\bullet h}(t_2) \overline{E_j E_h} e^{-I(\overline{u}_j - \overline{u_h})} + \overline{\pi(t_1)\pi(t_2)} e^{Iu_0(t_1 - t_2)} = \frac{P_c}{2} \sum_k u^{\bullet}_{\bullet h}(t_1) u_{\bullet h}(t_2) + N_{\bullet}\delta(t_1 - t_2), \qquad (3.2.3)$$

where $u_{0k}(t) = u_0(t - \tau_0 - kT_r)$.

Averaging in (8.2.8) for amplitudes and phases is conducted separately due to their independence. Furthermore, by virtue of the independence and uniformity of the distribution of phases α_k in different periods all members in the double sum turn into zero when $j \neq k$.

For calculation of the fluctuation characteristic and the equivalent spectral density we will need a fourth mixed moment for values of $\eta(t)$. As, too, during calculation of the correlation function, it is possible to show that

$$\begin{aligned} \overline{(\eta(t_{1})\eta^{*}(t_{2})\eta(t_{3})\eta^{*}(t_{4})} &- \overline{\eta(t_{1})\eta^{*}(t_{3})\eta^{*}(t_{4})\eta^{*}(t_{4})}_{\eta(t_{3})\eta^{*}(t_{4})]_{B}\eta} = \\ &= \frac{P_{a}^{2}}{4} \sum_{k,j} P_{kj}^{2} \mathcal{U}_{ak}^{a}(t_{1}) \mathcal{U}_{ak}(t_{3}) \mathcal{U}_{aj}^{a}(t_{3}) \mathcal{U}_{aj}(t_{4}) + \\ &+ \frac{P_{a}N_{a}}{2} \sum_{k} \mathcal{U}_{ak}^{a}(t_{3}) \mathcal{U}_{ak}(t_{4}) \delta(t_{3} - t_{3}) + \\ &+ \frac{P_{a}N_{a}}{2} \sum_{k} \mathcal{U}_{ak}^{a}(t_{3}) \mathcal{U}_{ak}(t_{3}) \delta(t_{1} - t_{4}) \delta(t_{3} - t_{3}). \end{aligned}$$

$$(8.2.9)$$

When deriving formula (8.2.9), we consider that we are interested in integrals of this expression over all t_i , where integration over t_1 and t_2 , and correspondingly over t_3 and t_4 , will always be produced over the same period.

Let us find the expression for the discrimination characteristic. We note, first of all, that since in formulas (8.2.1), (8.2.3)-(8.2.5) integration is produced only over one period, in these formulas there actually enter values of the modulating function $u(t - \tau_0)$ and only in this period. Therefore,

$$\int_{(j-1)T_{r}}^{jT_{r}} u(s-\tau)y(s)e^{i\omega_{r}s}ds = \int_{(j-1)T_{r}}^{jT_{r}} u_{o}(s-\tau-(j-1)T_{r})y(s)e^{i\omega_{r}s}ds. \qquad (8.2.10)$$

Using the given circumstance, dropping immaterial numerical factor 2K in formula (8.2.5), replacing variable s by s + τ and introducing former designation $\Delta = \tau_0 - \tau$, we obtain

$$z_{k}(\Delta) = \overline{z(kT_{r}, \Delta)} = -\operatorname{Re} \int_{(k-1)T_{r}}^{kT_{r}} \int_{(k-1)T_{r}}^{kT_{r}} u_{0}(s_{1}-(k-1)T_{r}) u^{0'}_{0}(s_{2}-(k-1)T_{r}) \overline{\eta(s_{1})\eta^{*}(s_{2})} ds_{1} ds_{2} =$$

$$= -\operatorname{Re} \left\{ \sum_{i} \frac{P_{e}}{2} \int_{(k-1)T_{r}}^{kT_{e}} u_{0}(s_{1}-(k-1)T_{r}) u^{*}_{0}(s_{1}-\Delta - (k-1)T_{r}) ds_{1} - (k-1)T_{r} \right\} u_{0}(s_{2}-\Delta - jT_{r}) ds_{1} + N_{0} \int_{(k-1)T_{r}}^{kT_{r}} u_{0}(s_{1}-(k-1)T_{r}) u^{*'}_{0}(s_{1}-(k-1)T_{r}) ds_{1} \right\} =$$

$$= -\operatorname{Re} \left\{ \frac{P_{e}T_{r}^{2}}{2} C(\Delta) C^{*'}(\Delta) + N_{0}T_{r}C^{*'}(0) \right\} = \frac{P_{0}T_{r}^{2}}{2} \operatorname{Re} C(\Delta) C^{*'}(\Delta), \qquad (5.2.11)$$

~345-

where $C(\Delta)$ - autocorrelation function of one period of modulation, defined in Chapters I and VII [see, for instance, (7.2.9)].

Η,

Replacement of variables of integration in (8.2.11) without change of the limits of the integrals is permissible due to the periodicity of the integrands.

Comparison of expression (8,2.11) with (7.2.10) shows that the discrimination characteristic in this case coincides with the discrimination characteristic of an optimum coherent discriminator with the accuracy of a factor. Therefore, its dependence on the form of modulation will have the same character as before. The gain factor of the discriminator will be determined by expression

$$K_{R} = \frac{\partial z_{k}(\Delta)}{\partial \Delta} \bigg|_{A=0} = -\frac{P_{0}T_{r}^{2}}{2} \operatorname{Re} |C(0)C^{*\prime\prime}(0) + |C^{\prime}(0)|^{2} = \frac{P_{c}T_{r}^{2}}{2} (b-a^{2}), \qquad (8.2.12)$$

coinciding with the accuracy of a constant factor with the expression for the gain factor in the coherent case.

Let us find now the expression for the fluctuation characteristic. Just as when calculating $\overline{z_k(\Delta)}$, we can obtain

$$R_{ik} = \overline{z_i(\Delta) z_k(\Delta)} - \overline{z_k(\Delta) \overline{z_i(\Delta)}} =$$

$$= \frac{N_0^2 T_r^2}{2} \operatorname{Re} \left\{ q^a p_{ik}^2 \left[|C(\Delta)|^a |C'(\Delta)|^2 + C^{**}(\Delta) C'^a(\Delta) \right] +$$

$$+ q \delta_{ik} \left[b |C(\Delta)|^a + |C'(\Delta)|^2 + 2iaC^*(\Delta) C'(\Delta) \right] + \left(b - a^a \right) \delta_{ik} \right\}, \qquad (5.2.13)$$

and the second second state and the second second

where $\rho_{ik} = \rho((i - k)T_r)$, $\rho(t_1 - t_2)$ - correlation function of fluctuations;

 $q = P_{c} T_{r}/2N_{0} - the ratic of the energy of the received sig$ nal for a period to the one-sided spectraldensity of noise, already used earlier inChapter V as the signal-to-noise ratio in $the incoherent case; <math display="block"> \delta_{ik} - Kronecker \ delta \ \begin{pmatrix} \bar{b}_{ik} = \begin{cases} 1 \ i = k \\ 0 \ i \neq k \end{cases} },$

Expression (8.2.13) characterizes the fluctuation component at the discrimiretor output in the discrete consideration. In the majority of real cases increte of the smoothing curcuits is great as compared to the period of repetition of the signal. In these cases the discrete process at the discriminator output can be replaced by a continuous one. The rule of replacement follows from the presentation of the output quantity of the range finder. In the discrete case it is equal to

Egnh2ki.

where $g_{nk} = g(nT_r, kT_r) - values$ of the pulse response of the smoothing filter at moments $t_n = nT_r, t_k = kT_r$.

-346-

If function $g(t, \tau)$ changes slowly as compared to T_r ,

$$\sum_{k=n_0}^{n} g(nT_r, kT_r) z_k = \sum_{k=n_0}^{n} g(nT_r, kT_r) \frac{z_k}{T_r} \cdot T_r = \int_{t_0}^{t} g(l, \tau) z(\tau) d\tau,$$

where continuous process z(t) is defined as

$$\int_{(k-1)T_r}^{kT_r} z(l) dl = z_k, \ z(l_k) = \lim_{T_r \to 0} \frac{z_k}{T_r}, \ t_k = kT_r.$$

In accordance with this, during continuous consideration the expression for the gain factor found above should be divided by T_r , and the one for the correlation function R_{ik} , by T_r^2 , having defined the correlation function of the continuous process by relationship

$$R(t_i, t_h) = R(iT_r, kT_r) = \lim_{T_r \to 0} \frac{R_{ih}}{T_r^2}.$$
(8.2.14)

Then on the basis of formulas (8.2.12)-(8.2.14) the correlation function of an equivalent fluctuating disturbance at the input of the range finder, determining equivalent spectral density, is equal to

$$R(t_{1}, t_{2}) = \lim_{T_{r} \to 0} \frac{R_{t_{h}}}{K_{h}^{2}} = \frac{1}{2q^{2}(b-a^{2})^{2}} \operatorname{Re} \{q^{2}p^{2}(t_{1}-t_{2}) \times [|C(\Delta)|^{2}|C'(\Delta)|^{2} + C^{2}(\Delta)C'^{2}(\Delta)] + T_{r}q\delta(t_{1}-t_{2})[b|C(\Delta)|^{2} + |C'(\Delta)|^{2} + 2iaC^{2}(\Delta)C'(\Delta)] + T_{r}(b-a^{2})\delta(t_{1}-t_{2})\}, \qquad (5.2,15)$$

where $t_1 = kT_r$, $t_2 = iT_r$,

$$\lim_{T_r\to 0}\frac{\delta_{ik}}{T_r}=\delta(t_1-t_2).$$

Formula (8.2.15) determines the fluctuation characteristic of the discriminator:

$$S_{\Theta KB}(\Delta) = \int_{-\infty}^{\infty} R(t_1 - t_2) d(t_1 - t_3) = \frac{T_r}{2q^3 (b - a^3)^3} \left\{ b - a^3 + q \left[b | C(\Delta) \right]^2 - [C'(\Delta)]^2 + \frac{q^3}{2\pi T_r \Delta f_c^2} \int_{-\infty}^{\infty} S_0^2(\infty) d\omega \left[|C(\Delta)|^3 | C'(\Delta)|^3 + C^{\Theta 2}(\Delta) C'^3(\Delta) \right] \right\}, \quad (3.2.16)$$

where as before $S_{\bullet}(\bullet) = \Delta f_{\bullet} \int_{-\infty}^{\infty} p(\tau) e^{-i \sigma \tau} d\tau$ - normalized spectral density of fluctuations

of the reflected signal.

The dependence of functions $S_{OUD}(\Delta)$ on characteristics of modulation completely coincides with the corresponding dependence for the coherent case (7.2.20), which already was investigated in Chapter VII. The dependence on the signal-to-noise ratio and characteristics of fluctuations of the signal is found differently. Let us

consider, first of all, the value of $S_{OKB}(\Delta)$ for $\Delta \approx 0$, i.e., equivalent spectral density of the discriminator S_{OKE} . From formula (8.2.16) it follows that

$$S_{offr} = \frac{7 \cdot (1+q)}{2q^{2} (4-q^{2})}.$$
 (8.2.17)

This expression, obviously, coincides with expression (7.2.15) for the coherent case if it is considered that spectral density of fluctuations of the signal has the form of a square function with width $\Delta f_c = f_r = 1/T_r$. Such coincidence is completely natural, since with spectrum width $\Delta f_c = 1/T_r$ the coherent signal coincides with the incoherent one. The dependence of $S_{\rm SKB}$ on q was already investigated earlier in Chapter VII. It is obvious that the curve of Fig. 7.3 for a square spectrum of fluctuations describes simultaneously the dependence on q of quantity $S_{\rm OFT}$ from (8.2.17) with replacement of h by q. For large signal-to-noise ratios q, quantity $S_{\rm OFT}$ is equal to $N_0/P_c(b - a^2)$, which coincides with the limiting value of $S_{\rm OFT}$ for a coherent signal with large h. This means that with optimum construction and high signal levels coherent and incoherent range finders ensure identical accuracy.

For signal-to-noise ratios which are not very large incoherence of the signal leads to definite loss of accuracy, the magnitude of which for the case of a square spectrum of fluctuations of the signal can be determined by the following expression:

$$\frac{S_{our nexor}}{S_{our ner}} = \frac{1+q}{4 \left[cT, +q \right]} = \frac{k + \frac{1}{4 \left[cT, -h + 1 \right]}}{h+1}, \qquad (8.2.13)$$

which is valid when $\Delta f_{c} = T_{r} < 1$.

When $\Delta f_{\rm C} T_{\rm r} \ge 1$, this ratio turns into one. Formula (3.2.16) shows that for identical accuracy of coherent and incoherent range finders it is necessary that q is great as compared to unity, and h is great as compared with $1/\Delta f_{\rm C} T_{\rm r}$. The dependence of ratio (8.2.18) on h for various values $\Delta f_{\rm C} T_{\rm r}$ is shown in Fig. 7. . From this figure it follows that with high frequencies of repetition and values of the signal-to-noise ratio h ~ 10 to 100, the loss due to incoherence may be rather substantial. Presence of such loss is one more ground for selection in incoherent radars of a low frequency of repetition (see Chapter V).

Let us find now the spectral density of parametric fluctuations. From (3,2,16) it follows that

$$S_{\text{Map}} = \frac{1}{2} \frac{\partial^2 S_{\text{over}}(1)}{\partial \Delta^4} \bigg|_{\Delta=0} = \frac{1}{2\kappa \Delta f_c^2} \int_{-\infty}^{\infty} S_0^2(\omega) d\omega = \frac{\epsilon}{\Delta f_o}, \qquad (3.2.19)$$

i.e., it coincides with an accuracy of numerical factor α of the order of 0.5 to 1

-348-

with the magnitude of $1/\Delta f_{\rm C}$ and thus, as in an optimum coherent range finder, does not depend on the form of modulation of the sounding signal. With a square spectrum of fluctuations $\alpha = 1$, and in the case of an exponential correlation function $\alpha = 0.5$.



Fig. 8.1. Loss in accuracy of incoherent range finders.

Thus, the optimum incoherent discriminator both in the character of dependences of the discrimination and fluctuation characteristics on the form of modulation and the magnitude of mismatch, and also in the magnitude of equivalent spectral density with sufficiently large signal-to-noise ratios does not differ from a coherent one. Substantial difference in characteristics of accuracy can occur only for signal-to-noise ratios q which are small as compared to unity, where the loss in accuracy is approximately equal (for small Δf_c T_r) to quantity (1 + q)/q.

§ 8.3. Discriminators Using Correlation Processing of the Signal

Discriminators with correlation processing of the signal in methods of construction and calculation of characteristics are the closest to coherent discriminators. Considering analogies which exist here, we shall start our consideration with them. Approximate fulfillment of optimum operations during correlation processing can be realized by any of three methods: with the help of two delay-detuned channels, in the form of a single-channel discriminator with switching of reference signals, with the help of a two-channel discriminator using a reference signal.

8.3.1. Discriminator with Two Detuned Channels

المحاجب المحاجب

The block diagram of a discriminator with two range-detuned channels is shown in Fig. 8.2. In its functional structure it coincides with the corresponding diagram



Fig. 8.2. Block diagram of a correlation discriminator with two delaydetuned channels: 1 - local escillator; 2 - mixer; 3 - controlled generator of reference signals; 4 multiplier; 5 - intermediate-frequency filter; 6 - detector; 7 subtractor. of a coherent discriminator. The received signal is multiplied after transfer to intermediate frequency or before this transfer by two reference signals $u_1(t)$ and $u_2(t)$, which to some degree of accuracy repeat the law of modulation of the sounding signal and have a corresponding high-frequency filling. Signals $u_1(t)$ and $u_2(t)$ are delayed relative to the sounding signal by $\tau + \delta$ and $\tau - \delta$, respectively. After multiplication the

signals are passed through filters with pulse response $h(t) \cos u_{np}t$, are detected and subtracted, forming output voltage of the discriminator. The difference between the incoherent and the coherent cases is only the inertia of the filters, which in the incoherent case should be small as compared to the period of repetition of signal, in such a manner that responses of the filter to adjacent pulses do not overlap. In the optimum case this filter should represent an integrator for a period with clearing; however, as we shall prove subsequently, final results do not depend on the form of the characteristic of the filter if only it possesses integrating properties during a time equal to the pulse duration of the signal. Let us assume, first, that this condition is satisfied, i.e., that $\Delta f_{ij} = \tau_{ij} \ll 1$, where $\tau_{ij} =$ pulse duration of the signal, and $\Delta f_{ij} =$ bandwidth of the filter. Then we shall consider the case when this condition is not satisfied.

Output voltage of the discriminator, analogously to § 7.3, is presented in the form

$$z(t, \Delta) = \left| \int_{-\infty}^{t} h(t-s) u_1(s-\tau-\delta) y(s) e^{i\omega_0 s} ds \right|^2 - \left| \int_{-\infty}^{t} h(t-s) u_1(s-\tau+\delta) y(s) e^{i\omega_0 s} ds \right|^2, \quad (8.3.1)$$

where it is assumed that pulse responses of filters in both channels are identical. but to nonstationariness of the input signal and small (as compared to the period) inertia of the filter $z(t, \Delta)$ is a nonstationary random process with periodic nonstationariness. Therefore, the discrimination and fluctuation characteristics are determined by expressions

$$\overline{z(t,\Delta)} = \lim_{T \to \infty} \frac{1}{T} \int_{0}^{T} \overline{z(t,\Delta)} dt, \qquad (8.3.2)$$

$$S_{\text{SKB}}(\Delta) = \frac{1}{K_{A}^{2}} \int_{-\infty}^{\infty} \frac{ds \lim_{T \to \infty} \frac{1}{T} \int_{0}^{T} \overline{[z(t, \Delta)z(t+s, \Delta) - (8.3.3)]}}{-\overline{z(t, \Delta)} \overline{z(t+s, \Delta)}] dt, \qquad (8.3.3)$$

where

$$K_{\mathbf{X}} = \frac{\partial \overline{z} \, \overline{(1, \Delta)}}{\partial \Delta} \Big|_{\mathbf{A} = \mathbf{0}} \tag{8.3.4}$$

is the gain factor of the discriminator (see § 7.4).

Expanding expression (8.3.1) and substituting in it the expression for the $i\omega_0 t$ correlation function of signal $\eta(t) = y(t)e^{-i\omega_0 t}$ from (8.2.8), we obtain

$$\begin{aligned} \overline{u_{1}(t, \Delta)} &= \int_{-\infty}^{\infty} \int_{-\infty}^{t} h(t-s_{1}) h(t-s_{2}) [u_{1}(s_{1}-\tau-\delta) u^{*}_{1}(s_{2}-\tau-\delta) - u_{2}(s_{1}-\tau+\delta) u^{*}_{2}(s_{2}-\tau+\delta)] \left[\frac{P_{\sigma}}{2} \sum_{k} u^{*}_{0}(s_{1}-\tau_{0}-\delta) - u_{2}(s_{2}-\tau_{0}-kT_{\tau}) + N_{0}\delta(s_{1}-s_{0}) \right] ds_{1} ds_{2} = \\ &= \sum_{k} T_{\tau}h^{2} (t-kT_{\tau}) \left\{ \frac{P_{\sigma}T_{\tau}}{2} [|C_{10}(\Delta-\delta)|^{2} - u_{10}(\Delta+\delta)|^{2} - |C_{20}(\Delta+\delta)|^{2}] + N_{0} (C_{11}(0) - C_{22}(0)) \right\}, \end{aligned}$$

$$(8.3.5)$$

where functions $C_{ik}(x)$ are determined by relationship

$$C_{ib}(x) = \frac{1}{T_r} \int_{0}^{T_r} u_i(s+x) u^{*} (s) ds,$$

$$C_{ib}(x) = \frac{1}{T_r} \int_{0}^{T_r} u_i(s+x) u^{*} (s) ds,$$
(8.3.6)

i.e., are crosscorrelation and autocorrelation functions for the totality set of sounding and reference signals.

Thus, as in the preceding chapter, we shall be interested only in values of these functions within the limits of one period, since mismatch exceeding the period of repetition of the signal is not of practical interest.

Averaging (8.3.5) in time, for the discrimination characteristic we obtain the following expression:

$$\frac{1}{2} \underbrace{(t, \Delta)}_{t} = \{q \{ [C_{10} (\Delta - \delta)]^{2} - [C_{10} (\Delta + \delta)]^{2} \} + C_{11} (0) - C_{10} (0) \} N_{0} \int_{0}^{\infty} h^{2} (t) dt = \{q \{ [C_{10} (\Delta - \delta)]^{2} - [C_{10} (\Delta + \delta)]^{2} \} + C_{11} (0) - C_{10} (0) \} N_{0} \Delta f_{0}, \quad (3.3.7)$$

-351-

where

$$\Delta j_{\phi} = \int_{0}^{\infty} h^{*}(t) dt = \frac{1}{2\pi} \int_{-\infty}^{\infty} |H(i\omega)|^{*} d\omega. \qquad (3.3.3)$$

and we assume that H(10) = 1.

From comparison with (7.3.26) it is clear that according to the character of the dependence of the discrimination characteristic on the laws of modulation of the sounding and reference signals the considered discriminator is completely equivalent to the corresponding coherent discriminator. In particular, slope and null shift of the discrimination characteristic are equal

$$K_{\mathbf{x}} = 2q N_{\bullet} \Delta f_{\bullet} \operatorname{Re} \left[C'_{10} (-\delta) - C'_{10} (\delta) C^{*}_{10} (\delta) \right], \qquad (3.3.9)$$

$$\Lambda_{0} = \frac{z(t,0)}{K_{A}} = \frac{C_{11}(0) - C_{11}(0) + \mathcal{J}[C_{10}, -C_{10}(0)]^{-1}}{q \operatorname{Re}[C_{10}(-d) C_{10}^{+}(-d) - C_{10}^{+}(d) C_{10}^{+}(d)]}, \qquad (8.3.10)$$

which differs from (7.3.2d) and (7.3.29) only by numerical factors. Null shift is absend for all q 11

$$\begin{array}{c} C_{11}(0) = C_{11}(0), \\ |C_{10}(-\delta)|^{4} = |C_{10}(\delta)|^{4}, \end{array}$$

$$(5.3.11)$$

i.e., if conditions of symmetry of reference signals relative to sounding signals and of equality of gains in both channels are satisfied. The dependence of the gain factor on detuning between the reference signals is precisely the same as in the coherent discriminator.

Let us find now the equivalent spectral 6 nuity. We assume that conditions (8.3.11) are satisfied. Here, as in the observed discriminator, the component of Σ_{BRS} which does not depend on the signal-to-noise ratio will turn into zero, and in the expression for S_{BRS} there will remain only terms with 1/q and 1/q². Calculating correlation function z(t, 0) with the help of (β, ϵ_{CC}) , executing transformations similar to those which were used in deriving (7.3.5), averaging the obtained expression in time and integrating in accordance with (8.3.3), we can obtain the following expression:

$$S_{012} = T_r \{2 | C_{11}(0) |^{\circ} - |C_{12}(2\delta)|^{\circ} - |C_{11}(-2\delta)|^{\circ} + + 2q [2C_{11}(0) | C_{12}(-\delta) |^{\circ} - - Re C_{11}(-2\delta) C_{12}^{\circ}(-\delta) C_{12}^{\circ}(-\delta) - - Re C_{11}(2\delta) C_{22}^{\circ}(\delta)]^{\circ} \{4q \operatorname{Re} [C_{12}(-\delta) C_{12}^{\circ}(-\delta)]\}^{\circ}.$$
(7.3.12)

From comparison of this expression with the corresponding expression for the coherent discriminator (7.3.30, it follows that expression (2.3.17) differs true if only by coefficients which depend on the signal-to-noise ratio. The structure of

-352-

these coefficients in this case is still simpler — they depend neither on the form of spectral density of fluctuations of the signal nor on the frequency response of the filter. For small detunings 5 expression (8.3.12) approaches S_{ORT} from (8.2.17), and with arbitrary detuning the dependence on its magnitude and the form of the sounding and reference signals can be described just as in § 7.3 by introduction of the equivalent signal-to-noise ratio q_{0KB} and the equivalent width of the spectrum of modulation b_{SKB} . Formula (8.3.12) here takes the form

$$S_{\mu KB} = \frac{T_{*} (1 + q_{2KB})}{2q_{\mu KB}^{2} b_{\mu KB}}, \qquad (8.3.13)$$

(8.3.15)

where

$$=2\frac{2C_{11}(0)|C_{:0}(\delta)|^{2}-\operatorname{Re} C_{12}(-2^{1})C^{*}_{10}(-\delta)C^{*}_{01}(-\delta)-}{2C_{11}^{2}(0)-|C_{12}(2\delta)|^{2}-} - \frac{-\operatorname{Re} C_{21}(2^{1})C^{*}_{10}(\delta)C^{*}_{10}(\delta)}{-|C_{11}(2\delta)|^{2}}; \qquad (8.3.1^{1})$$

$$=\frac{2[\operatorname{Re} C_{10}'(-\delta)C^{*}_{10}(-\delta)]^{2}[2C_{11}^{2}(0)-}{b[2C_{11}(0)C_{10}(-\delta)]^{2}-\operatorname{Re} C_{12}(-\delta)C^{*}_{10}(-\delta)C^{*}_{10}(-\delta)-}$$

coincide with the corresponding ratios
$$h_{330}$$
/h and b_{330} /b from § 7.3, which already were investigated in the preceding chapter. These relationships permit us with the

were investigated in the preceding chapter. These relationships permit us with the help of graphs for $h_{\rm HED}/h$ and $b_{\rm HED}/b$ for various concrete forms of modulation from Chapter VII and formula (8.3.13) for Fig. 8.1 to calculate the magnitude of $S_{\rm HED}$ for any case interesting us.

8.3.2. Discriminator with Switching of the Reference Signals

The block diagram of an incoherent discriminator with switching of reference signals differs from the corresponding concrent diagram (Fig. 7.7) only in the form of the filter h(t). In all other respects, as in the preceding case, all operations and the expression for the output voltage of the discriminator coincide with those given in § 7.4. Due to the great (as compared to $\Delta f_{\rm C}$) bandwidth of the filter in the incoherent case there is possible selection of a frequency of switching of reference signals, rather high as compared to $\Delta f_{\rm C}$. As shown in § 7.4, cach selection is very desirable from the point of view of decreasing the component of equivalent spectral density which does not depend on the signal-te-noise ratio but is caused by incomplete correlation of the useful signal in adjacent half periods of the frequency of switching. This component dusappears only when the frequency

-353-

of switching is great as compared to Δf_{c} (see 7.4.7).

. In coherent discriminators possibilities of selection of the frequency of switching are limited by the passband of the filter, matched with the spectrum of fluctuations of the signal; rapid switching and elimination of the corresponding component of S_{3KB} are possible only with a wide filter passband Δf_{ϕ} , so that the frequency of switching f_{μ} should satisfy condition $\Delta f_{\phi} \gg f_{\pi} \gg \Delta f_{\pi}$.

During coherent processing this condition can be satisfied only in rare cases, while in incoherent discriminators, due to the great bandwidth of the filter $(\Delta r_{e_i} \ge 1/T_r)$, fulfillment of this condition does not present any difficulty, and the frequency of switching can be only half the frequency of repetition of the signal. We assume that conditions on the speed of switching, leading to elimination of the component of equivalent spectral density which does not depend on the signal-to-noise ratio, are satisfied.

The gain factor of the discriminator with switching of reference signals and its discrimination characteristic, as also in the coherent case, coincide with an accuracy of coefficient 1/2 with the corresponding characteristics of a two-channel discriminator when $C_{10}(0) = C_{20}(0)$. This equality corresponds to the natural assumption of coincidence in form of the reference signals, alternately utilized as the "leading" and "lagging" signals. The condition of absence of systematic error here is the equality $|C_{10}(0)|^2 = |C_{10}(-0)|^2$. Then, the gain factor will be proportional to Re $(C_{10}^{\dagger}(0) C_{10}^{\dagger}(0))$.

Just as in Paragraph 8.3.1, we can obtain the following expression for equivalent spectral density:

$$S_{0NB} = \frac{T_r}{4q^8} \cdot \frac{1 + 2q |C_{10}(\hat{d})|^8}{[\text{Re } C'_{10}(\hat{d}) C^*_{10}(\hat{d})]^8}.$$
 (8.3.16)

In particular, with coincidence of modulations of the sounding and the refer-

$$S_{0,XR} = -\frac{T_r}{4q^3} \cdot \frac{1+2q |C(\delta)|^3}{[Re C'(\delta) C^4(\delta)]^3}.$$
 (8.3.17)

Let s quantity, depending upon the selection of δ , may differ substantially from S_{OFT} , where there exists a certain optimum value of detuning ensuring minimum S_{OFT} . The physical reasons for such a dependence of S_{OFT} on detuning are the same as in the case of a coherent discriminator (§ 7.4); however, the actual optimum value S_{OFT} is found somewnat differently. This quantity is determined from equation

and depends on signal-to-noise ratio q. Detailed investigation of the dependence of S_{SHE} on detuning 6 and comparison of the considered discriminator with a two-channel one will be conducted later in examining concrete forms of modulation.

8.3.3. Discriminator with Differentiation of the Reference Signal

Such a discriminator in its principles of construction is the closest to the optimum one. Its block diagram coincides with the block diagram of the corresponding coherent discriminator (Fig. 7.2). The difference, as also in the preceding cases, is only in the bandwidth of the filter. Merits and deficiencies of such a method of realizing optimum operations already were discussed in Chapter VII. Incoherence of the signal in this respect introduces nothing new.

Designating reference signals in the two channels of discriminator $u_1(t)$ and $u_2(t)$ and assuming that the frequency responses of filters in both channels are identical, for output voltage of the discriminator we obtain an expression coinciding with (7.5.1). From the latter, by transformations of § 8.2, Paragraphs 8.3.1 and 8.3.2, and averaging in time we can find the following expressions for the gain factor and equivalent spectral lensity:

$$K_{\rm A} = N_{\rm e} \Delta f_{\rm e} q \operatorname{Re} \left[C'_{10} (0) C^*_{10} (0) + C^*_{10} (0) C'_{20} (0) \right], \qquad (8.3,18)$$

$$S_{\rm BKB} = T_{\rm e} \frac{C_{11}(0)C_{22}(0) + \operatorname{Re} C^2_{12}(0) + q \left[\zeta_{11} (0) \left| C_{es} (0) \right|^2 + 2q^2 \left| \operatorname{Re} \left(C'_{10} (0) C^*_{es} (0) + \frac{1}{2q^2} \left| \operatorname{Re} \left(C'_{10} (0) C^*_{es} (0) + \frac{1}{2q^2} \left| \operatorname{Re} \left(C'_{10} (0) C^*_{es} (0) + \frac{1}{2q^2} \left| \operatorname{Re} \left(C'_{10} (0) C^*_{es} (0) + \frac{1}{2q^2} \left| \operatorname{Re} \left(C'_{10} (0) C^*_{es} (0) + \frac{1}{2q^2} \left| \operatorname{Re} \left(C'_{10} \left(0 \right) C^*_{es} (0) + \frac{1}{2q^2} \left| \operatorname{Re} \left(C'_{10} \left(0 \right) C^*_{es} (0) \right) \right|^2 \right], \qquad (8.3.19)$$

where $C_{ik}(x)$ are determined by formulas (8.3.6), and we assume that there is no systematic error. The condition of its absence is

$$\operatorname{Re} C_{11}(0) = \operatorname{Re} C_{10}(0) C^{*}_{10}(0) = 0. \qquad (8, 3, 20)$$

With optimum reference signals $u_1(t) = u(t)$ and $u_2(t) = u'(t)$ formula (8.3.19) changes into the expression for S_{OH1} (8.2.17). With arbitrary $u_1(t)$ and $u_2(t)$, just as in Paragraph 8.3.1, it is possible to introduce equivalent values of the signal-tonoise ratio q_{OHB} and the mean square width of the spectrum of modulation (b - $a^2)_{OH3}$. If, in particular, reference $s_{ABA}(t)$ differs from the sounding signal, and $u_2(t) = u'_1(t)$, i.e., differentiation of the reference signal is produced exactly,

$$S_{\text{DKB}} = T_r \frac{b_{11} - a_{11}^2 + q |C_{10}(0)|^2 (b_{11} - 2a_{11}a_{10} + a_{10}^2)}{2q^2 |C_{10}(0)|^4 (\text{Re } b_{10} - a_{10}^2)^2}, \qquad (8.3.21)$$

where $a_{ik} = \operatorname{Re} C'_{ik}(0); b_{ik} = -C''_{ik}(0).$

Thus, consideration of incoherent discriminators with correlation processing of the received signal shows that the character of the dependence of the gain factor and equivalent spectral density on the method of construction of discriminator and the form of modulation of the sounding signal and characteristics of the reference signals remain the same as in the case of a coherent signal. The dependence on the magnitude of detuning in two-channel discriminators also does not change, and in discriminators with switching of reference signals the quantitative form of the dependence changes somewhat, although qualitatively all remains as before.

The structure of the formulas determining characteristics of incoherent discriminators does not differ from the corresponding coherent cases. Only, here, in formulas for $S_{\rm SCC}$ pertaining to coherent range finders all integrals containing the frequency response of the filter and spectral density of fluctuations are replaced by $1/T_r$, and the signal-to-noise ratio h is replaced by q. This gives us the possibility of using for calculation the characteristics of incoherent discriminators, calculation formulas and graphs already given in Chapter VII, by a simple change of scale.

A distinctive feature of incoherent discriminators with correlation processing is independence of their characteristic of the form of the particular characteristic of the filter and the magnitud. Its passband, if only the latter satisfies relationships $1/\tau_{\mu} \gg \Delta f_{\oplus} \gg 1/T_{r}$.

§ 3.4. Discriminator with a "Shortening" Filter

In incoherent range finders using a pulse signal in many cases it is more rational to use the principle of optimum filtration of the received signal in a ("scortening") filter [4, 35, 38]. This gives us the possibility to form the likelihorm functional and to measure the distance to many targets using the high-frequency part of the radar set common for all meters and the detection unit.

Even in single-target radars use of a "shortening" filter can lead to great technical simplicity as compared to correlation receivers.

8.4.1. Block Diagram of the Discriminator

A "shortening" filter, in principle, can be realized on a carrier of intermediate or low frequency. In practice, they most often use filtration at intermediate frequency. However, selection of the frequency at which the "shortening" filter is realized does not affect analysis of the discriminator and its results. For definitiveness in Fig. 8.3 we show the block diagram of a discriminator with a

-356-



Fig. 8.3. Block diagram of ε discriminator with a "shortening" filter: 1 - local oscillator; 2 - mixer; 3 - "shortening" filter; 4 - detector; 5 - gating stage; 6 - controlled generator of gates; 7 - subtractor. "shortening" filter at intermediate frequency.

The received signal y(t) after transfer to intermediate frequency is passed through a shortening filter and detected. Output voltage of the detector is fed to two gating stages, to which there are fed gating pulses (gates), detuned relative to the output value of delay by ± 5 . Output voltages of the gating stages are subtracted, forming the output voltage of the discriminator.

The derivative of the logarithm of the likelihood functional in the given circuit will be formed approximately by the two detuned gate pulses. This approximation will be all the better, the narrower the gate pulses and the less the detuning between them. At the limit they should be delta-shaped and located with detuning $5 \rightarrow 0$. In practice gates have finite width and are detuned a finite quantity, where both have a magnitude of the order of the duration of the pulse at the detector output.

The considered circuit represents an analog of the correlation discriminator with two range-detuned channels. It is obvious that the corresponding analog for a circuit with switching of reference signals is a circuit in which gates are not fed simultaneously, but one-by-one during a certain time $T_{\rm H}/2$. Such a change in this case is not a technical simplification, and at the same time, as in correlation discriminators, it leads to certain loss depending on the degree of overlapping of the gates, their form and the ratio of the width of the spectrum of signal fluctuations to the frequency of switching.

The circuit with differentiation in this case requires two "shortening" filters with different characteristics, which, because of their complexity, is hardly an acceptable technical solution. Therefore, subsequently we shall consider only the block diagram of Fig. 8.3, which, as we shall prove subsequently, in a broad range of conditions gives results sufficiently close to those potentially attainable.

Pulse response $h_0(t)$ of the "shortening" filter in the block diagram of Fig. 3.3 should be sufficiently close to Re $u_0(-t)e^{-i\omega_{\rm HP}t}$, where $u_0(t)$ — a function describing one period of modulation. Only in the absence of additional intrapulse modulation can this pulse response be presented, as for all filters considered earlier, in the form $h(t) \cos \omega_{\rm HT}t$, where h(t) is the pulse response envelope.

-357-

In general, with intrapulse modulation this pulse response can be presented by low-frequency functions only in the form

$h_{1}(t)\cos\omega_{xy}t + h_{1}(t)\sin\omega_{xy}t$,

where $h_1(t)$, $h_2(t)$ - sine and cosine components of pulse response.

For instance, for the pulse described by function u_{a0} , with phase modulation according to the law $\psi_0(t)$ for the case of an optimum shortening filter

$$h_1(t) = u_{a0}(-t) \cos \psi_0(-t), h_2(t) = -u_{a0}(-t) \sin \psi_0(-t).$$

With arbitrary modulation the pulse response of the filter can be recorded in the form Re h(t)e $i\omega_{np}t$, where h(t) - the complex pulse response envelope of the filter. In the above-mentioned example

$$h(t) = u_{aa}(-t)e^{i\phi_a(-t)}.$$

Taking into account this circumstance we can record output voltage of the filter in the form

$$u_{\phi}(t) = \operatorname{Re} \int_{-\infty}^{t} h(t-s) e^{i\omega_{x,p}(t-s)} y(s) \cos(\omega_{\phi} + \omega_{n,p}) s ds =$$

$$= \frac{1}{2} \operatorname{Re} e^{i\omega_{x,p}t} \int_{-\infty}^{t} h(t-s) y(s) e^{i\omega_{x,p}t} ds, \qquad (5, 4, 1)$$

where there are omitted components with doubled frequency $\omega_{\Pi P}$. Then, on the basis of (3.4.1) we can describe output voltage of the discriminator by the following expression:

$$\mathbf{z}(t) = \left| \int_{-\infty}^{s} h(t-s) \mathbf{y}(s) e^{i \omega_{s} t} ds \right|^{s} v(t-s), \qquad (P, 4, 2)$$

where there are omitted immaterial numerical coefficients, and v(t) - a function describing the sequence of gate pairs, so that

$$v(t) = \sum_{k} v_{e}(t - kT_{r}), \qquad (3.4.3)$$

 τ - measured value of delay.

Function $\mathbf{v}_0(t)$ must have negative and positive values, since only in this case is creation of a discrimination characteristic possible. In accordance with the block diagram of Fig. 8.3 function $\mathbf{v}_0(t)$ is the difference of functions describing gating pulses fed to the various gating stages. For example, with square gates $\mathbf{v}_0(t)$ has the form shown in Fig. 8.4. This form depends on the amplitudes of the gates

-358-

and detuning between them. By selection of the form of the gates and parameters of channels one should try to ensure that function $v_{\Omega}(t)$ is odd.



Fig. 8.4. The form of function $v_0(t)$ depending upon the amplitude of the gates and detuning between them.

8.4.2. Discriminator Characteristics

Let us find characteristics of the considered discriminator. Substituting in formulas (3.4.2) the expression for correlation function $q(t) = y(t)e^{-\frac{i\omega_0 t}{t}}$ from (3.2.1), we obtain the mean value

$$\begin{aligned} \overline{x(t)} &= \int_{-\infty}^{s} \int_{-\infty}^{t} h(t-s_{1}) h^{*}(t-s_{2}) \left[\frac{P_{\tau}}{2} \sum_{k} u^{*} (s_{1}-\tau_{0}-kT_{\tau}) \times u_{0}(s_{2}-\tau_{0}-kT_{\tau}) + N_{0} h(s_{1}-s_{1}) \right] ds_{1} ds_{2} \cdot v(t-\tau) = \\ &= \left[\frac{P_{\tau}}{2} \sum_{k} |f(t-\tau_{0}-kT_{\tau})|^{2} + N_{0} \int_{0}^{\infty} |h(t)|^{*} dt \right] \cdot v(t-\tau), \end{aligned}$$

$$(5.4.4)$$

$$f(t) = \int_{0}^{s} h^{*}(s) u_{*}(t-s) ds \qquad (3.4.5)$$

is the result of passage of signal $u_{\rho}(t)$ through a filter with pulse response $h^{*}(t)$, which it is possible to treat the same as the function of crosscorrelation between the pulse response of the filter and the function describing the law of modulation of the signal in one period. Such treatment is useful for finding an analogy with the correlation method of processing a signal. In particular, with an optimum "shortening" filter $[h(t) = u_{\rho}(-t)]$

$$f(t) := T_r C(t),$$
 (5.4.6)

where C(t), as hefore, is the autocorrelation function of the signal. To produce the discrimination characteristic we average expression (8.4.4, in time. Then

-359-

$$\frac{1}{z(t, \Delta)} = \frac{1}{T_r} \int_0^T \frac{1}{z(t, \Delta)} dt = \frac{P_o}{2T_r} \int_0^T |f(t-\tau_o)|^2 v_o(t-\tau) dt + N_o \int_0^\infty |h(t)|^2 dt \frac{1}{T_r} \int_0^\infty v_o(t-\tau) dt = \frac{P_c}{2T_r} \int_{-\infty}^\infty |f(s)|^2 v_o(s+\Delta) ds + N_o \int_0^\infty |h(t)|^2 dt \frac{1}{T_r} \int_{-\infty}^\infty v_o(t) dt. (8.4.7)$$

Hence, the gain factor of the discriminator

$$K_{\mathrm{R}} = \frac{P_{\mathrm{e}}}{2T_{\mathrm{r}}} \int_{-\infty}^{\infty} |f(s)|^{\mathrm{s}} v'_{\mathrm{e}}(s) \, ds.$$

During calculation of discriminator characteristics it is often more convenient to use, not a time, but a frequency presentation of functions. Passing in expression (8.4.7) to spectra, we obtain

$$\frac{\widetilde{z(i, \Delta)}}{\widetilde{z(i, \Delta)}} = \frac{P_e}{2T_r} \frac{1}{(2\pi)^3} \int_{-\infty}^{\infty} \int_{-\infty}^{\infty} U^* [i(\omega + x)] \times \\
\times H [-i(\omega + x)] U(ix) H^* (-ix) V_{\bullet}(i\omega) e^{i\omega \Delta} d\omega dx + \\
+ \frac{N_e}{T_r} \int_{-\infty}^{\infty} v_{\bullet}(i) dt \frac{1}{2\pi} \int_{-\infty}^{\infty} |H(i\omega)|^2 d\omega, \qquad (6.4.9)$$

where $U(i\omega)$, $H(i\omega)$, $V_0(i\omega)$ - Fourier transforms if the corresponding functions. Likewise, the gain factor of the discriminator

$$K_{\mathbf{x}} = \frac{P_{\mathbf{o}}}{2T_{\mathbf{r}}} \cdot \frac{1}{(2\pi)^{\mathbf{o}}} \int_{-\infty}^{\infty} \int_{-\infty}^{\infty} i \omega U^{\mathbf{o}} [i(\omega + x)] H [-i(\omega + x)] \times U(ix) H^{\mathbf{o}} (-ix) V_{\mathbf{o}} (i\omega) d\omega dx. \qquad (3.4.10)$$

From formule (4.4.7) it follows that the conditions of the absence of systematic error are the following:

$$\int_{-\infty}^{\infty} v_{\bullet}(t) dt = 0, \qquad (8.4.12)$$

$$\int_{-\infty}^{\infty} |f(t)|^{*} v_{\bullet}(t) dt = 0. \qquad (3.4.12)$$

The first of these conditions consists of the requirement of equality of the areas of the positive and negative parts of $v_0(t)$, and the second is the requirement of orthogonality of the square of the shortened pulse envelope with function $v_0(t)$. If the gates have identical shape, condition ($^{8}, 4, 11$) is the condition of identity of amplifications of gating stages. Then the second of conditions (3, 4, 12) reduces to the corresponding determination of the reference point for delay. With symmetric form of the shortened pulse envelope this reference point coincides with its center,

-360-

but with asymmetric form it should be determined in such a manner that (8.4.12) is satisfied. In practice conditions (8.4.11) and (8.4.12) usually are satisfied. Moreover, gates, as a rule, are selected identical, so that

$$P_{s}(t) = v(t+\delta) - v(t-\delta),$$
 (8.4.13)

where v(t) - function describing the form of the gate;

 δ - detuning relative to their center.

Subsequently, we shall consider systematic error absent.

In discriminators of the considered type they very often use square gates of certain duration $\tau_{\Pi C}$ Detuning 5 here can, without loss of generality, be considered larger than $\tau_{\Pi C}/2$, since when $\delta < \tau_{\Pi C}/2$ the form of function $v_0(t)$ coincides with the form of this function for $\delta > \tau_{\Pi C}/2$ with an accuracy of quantitative parameters (Fig. 8.4). With such a form of gates

$$v'_{\bullet}(t) = -\delta\left(t+\delta+\frac{\tau_{ac}}{2}\right) - \delta\left(t-\delta-\frac{\tau_{ac}}{2}\right) + \delta\left(t+\delta-\frac{\tau_{ac}}{2}\right) + \delta\left(t-\delta+\frac{\tau_{ac}}{2}\right)$$

and the expression for the gain factor has the form

1 4

$$K_{\mathbf{A}} = \frac{P_{\mathbf{a}}}{2T_{\mathbf{r}}} \left\{ \left| f \left(\delta - \frac{\tau_{\mathbf{a}} c}{2} \right) \right|^{2} + \left| f \left(-\delta + \frac{\tau_{\mathbf{a}} c}{2} \right) \right|^{2} - \left| f \left(\delta + \frac{\tau_{\mathbf{a}} c}{2} \right) \right|^{2} - \left| f \left(-\delta - \frac{\tau_{\mathbf{a}} c}{2} \right) \right|^{2} \right\}.$$

$$(8.4.14)$$

In particular, for the most wide-spread case when gates are located end-to-end $(\delta = \tau_{\rm H}/2)$

$$K_{\rm x} = \frac{P_{\rm c}}{2T_{\rm r}} \{ 2 | f(0) |^{\rm s} - | f(\tau_{\rm nc}) |^{\rm s} - | f(-\tau_{\rm nc}) |^{\rm s} \}.$$
(8.4.15)

Let us find now the correlation function of cutput voltage of the discriminator. Using formula (8.2.9) and assuming conditions of the absence of systematic error to be satisfied, for zero mismatch we obtain

$$R_{\theta}(l, \theta) = \int_{-\infty}^{t} \int_{-\infty}^{t} \int_{-\infty}^{t+\theta} h(l-s_{1}) h^{*}(l-s_{0}) h(l+\theta-s_{0}) \times \\ \times h^{*}(l+\theta-s_{0}) v_{0}(l-\tau_{0}) v_{0}(l+\theta-\tau_{0}) \times \\ \frac{\times [\eta(s_{1})\eta^{*}(s_{0})\eta(s_{0})\eta^{*}(s_{0})] ds_{1} ds_{0} ds_{0} ds_{0} = \\ -\overline{\eta(s_{1})\eta^{*}(s_{0})\eta(s_{0})\eta(s_{0})\eta^{*}(s_{0})] ds_{1} ds_{0} ds_{0} ds_{0} ds_{0} = \\ \{N_{0}P_{0} \operatorname{Re} \sum_{h} f^{*}(l-\tau_{0}-kT_{n})f(l+\theta-\tau_{0}-kT_{n}) \times \\ \times \int_{0}^{\infty} h^{*}(s) h(s+\theta) ds + N_{0}^{2} |\int_{0}^{\infty} h(s) h^{*}(s+\theta) ds|^{*} \} \times \\ \times v_{0}(l-\tau_{0}) v_{0}(l+\theta-\tau_{0}). \end{cases}$$
(8.4.10)

Averaging this expression over time t, we obtain

$$\widetilde{R_{s}(0)} = \frac{N_{0}^{2}}{T_{r}} \int_{0}^{\infty} h^{*}(s) h(s+1) ds \Big|^{s} \int_{-\infty}^{\infty} v_{\bullet}(t) v_{\bullet}(t+1) dt + \frac{N_{1}P_{\bullet}}{T_{r}} \operatorname{Re} \int_{-\infty}^{\infty} j^{*}(t) v_{\bullet}(t) j(t+1) v_{\bullet}(t+1) dt \times \int_{-\infty}^{\infty} h^{*}(s) h(s+1) ds. \qquad (0.7.17)$$

Integrating the expression for $R_{z}(\theta)$ over θ and dividing by the square of the gain factor, we find the following expression for equivalent spectral density:

$$S_{\theta K \theta} = T_r \left\{ \int_{-\infty}^{\infty} \left[\frac{2q}{T_r^3} \operatorname{Re} \int_{-\infty}^{\infty} f^*(l) v_{\theta}(l) f(l+\theta) v_{v}(l+\theta) dl \times \right] \right\}$$

$$\times \int_{0}^{\infty} h^*(s) h(s+\theta) ds + \left| \frac{1}{T_r} \int_{0}^{\infty} h^*(s) h(s+\theta) ds \right|^2 \times \left[\int_{-\infty}^{\infty} v_{\theta}(l) v_{\theta}(l+\theta) dl \right] d\theta : \left[\frac{q}{T_r} \int_{-\infty}^{\infty} |f(s)|^2 v_{\theta}(s) ds \right]^2. \quad (3.4.18)$$

Just as in formula (8.4.10), integration over time in (8.4.18) can be replaced everywhere by integration over frequency; then in the expression for S_{RB} there will be only spectra of the corresponding functions. Producing the necessary transformations, consisting of multiple application of Parseval's formula, we arrive at the following expression for S_{RHF}

$$S_{\Theta KB} = \left\{ 2q \frac{1}{(2\pi T_r)^3} \operatorname{Re} \int_{-\infty}^{\infty} \int_{-\infty}^{\infty} \int_{-\infty}^{\infty} U(ix) H^*(-ix) U^*(iy) \times \\ \times H(-iy) V_{\bullet}[i(\omega-x)] V^{\bullet}_{\bullet}[i(\mathbf{e}-y)] H^*(-i\mathbf{e}) \times \\ \times H(-i\omega) d\omega dx dy + \frac{1}{(2\pi T_r)^2} \int_{-\infty}^{\infty} \int_{-\infty}^{\infty} |V_{\bullet}(i\mathbf{e})|^2 H^{\bullet}(i\mathbf{e}) H(i\mathbf{e}) \times \\ \times H[i(\mathbf{e}+x)] H^*[i(\omega+x)] d\omega dx \right\} : \left\{ \frac{q}{(2\pi T_r)^2} \int_{-\infty}^{\infty} \int_{-\infty}^{\infty} i\omega U^* \times \\ \times [i(\mathbf{e}+x)] H[-i(\mathbf{e}+x)] U(ix) H^*(-ix) V_{\bullet}(i\mathbf{e}) d\omega dx \right\}^2.$$
 (2.4.19)

Let us consider the case when the duration of gates is small as compared to duration of the pulse at the detector output. Then function $v_{\rm C}(t)$ can be presented in the form

$$v(t) = \delta(t+\delta) - \delta(t-\delta), \qquad (8, 5, 20)$$

र केंद्र में जिसके केंद्र के प्रदेश के प

where $\delta(t) = delta$ -function.

-362-

 $\overline{\boldsymbol{z}(\Delta)} = \frac{P_{\bullet}}{2T_{\tau}} \{ \| \boldsymbol{f}(-\boldsymbol{\delta} - \Delta) \|^{\bullet} - \| \boldsymbol{f}(\boldsymbol{\delta} - \Delta) \|^{\bullet} \}$ (8.4.21)

and the gain factor of the discriminator will be equal to

Here

$$K_{\rm A} = \frac{P_{\bullet}}{2T_{\rm r}} 2 \operatorname{Re} \{ f'(-\delta) f^{*}(-\delta) - f'(\delta) f^{*}(\delta) \}.$$
(8.4.22)

Substituting (8.4.20) in formula (8.4.18) for equivalent spectral density, we obtain

$$S_{9NB} = T_{\varphi} \left\{ \frac{2q}{T_{r}^{3}} \operatorname{Re} \left[(|f(-\delta)|^{2} + |f(\delta)|^{2}) \int_{0}^{\infty} |h(s)|^{2} ds - f^{*}(-\delta) f(\delta) \int_{0}^{\infty} h^{*}(s) h(s+2\delta) ds - - f^{*}(\delta) f(-\delta) \int_{0}^{\infty} h^{*}(s) h(s+2\delta) ds \right] + \frac{2}{T_{r}^{2}} \left[\left(\int_{0}^{\infty} |h(s)|^{2} ds \right)^{2} - (8.4.23) - \left(\int_{0}^{\infty} h^{*}(s) h(s+2\delta) ds \right]^{4} \right] \right\} : \left[\frac{2q}{T_{r}^{2}} \operatorname{Re} \left(f'(-\delta) f^{*}(-\delta) - f'(\delta) f^{*}(\delta) \right) \right]^{4}.$$

From comparison of this formula with (8,3.12) it follows that both expressions coincide if we set

$$u_1(t) = u_2(t) = h(-t).$$
 (8.4.24)

Thus, the considered discriminator with gates which are narrow as compared to the duration of the shortened pulse is completely equivalent to a two-channel discriminator with correlation signal processing, in which reference signals are identifical and are determined in accordance with (8.4.24). In particular, with an optimum "shortening" filter h(t) = u(-t) and detunings 5, small as compared to the duration of the shortened pulse, formula (8.4.23) becomes expression (8.2.17) for S_{ONT} . When h(t) = u(-t) (8.4.6) is satisfied, and

$$\int h^{\bullet}(s) h(s-l) ds = T_{\tau}C(l). \qquad (5.4.5\%)$$

Further investigation of the discriminator without specifying the form of modulation of the sounding signal and characteristics of the optimum filter is difficult. The obtained general results permit us to conduct such an investigation in any concrete case. We shall pursue this in subsequent paragraphs.

§ 5.5. Incoherent Pulse Radiation Without Additional Modulation

Simple pulse radiation, probably, is the most wide-spread in practice [1, 2t = 27, 36]. The matched filter in this case can be called "shortening" only in quotes. L

-363-

reality it is a normal i-f amplifier (UPCh) with a bandwidth, approximately matched with the spectrum width of the pulse. Requirements on the form of the frequency response of the UPCh are not very critical; in practice exact matching of it with the form of the pulse spectrum is not achieved [30, 41]. With a pulse signal without additional modulation most rational from the technical point of view is the application of discriminators with "shortening" filters; however, in principle, there can also be used correlation discriminators. We will consider, basically, discriminators with a "shortening" filter; however, we shall also touch on correlation discriminators.

8.5.1. Discriminator with a Broad-Band UPCh

We consider first one particular case of a discriminator using filtration, without specifying the form of the pulse.

Quite often bandwidth of the UPCh (VIIY)

$$\Delta f_{\gamma \Pi \mathbf{u}} = \int_{0}^{\infty} h^{*}(t) dt = \frac{1}{2\pi} \int_{0}^{\infty} |H(i\boldsymbol{\omega})|^{*} d\boldsymbol{\omega} \qquad (5.5.1)$$

considerably exceeds the matched bandwidth, so that $\Delta f_{y \prod i} \tau_{ii} >> 1$ (pulse response of the filter h(t) during pulse radiation bitmout additional modulation is real). Then, the pulse passes to the UPCh output without distortion, so that

$$i(t) = u_{\bullet}(t)$$
 (3.5.2)

and formulas for the gain factor and spectral density take the form

$$K_{R} = \frac{P_{\bullet}}{2T_{r}} \int_{-\infty}^{\infty} u_{\bullet}^{2}(s) v_{\bullet}^{*}(s) ds, \qquad (8.5.3)$$

$$S_{\bullet H \bullet} = T_{r} \frac{2q \frac{1}{T_{r}} \int_{-\infty}^{\infty} u_{0}^{2}(t) v_{0}^{2}(t) dt + \int_{-\infty}^{\infty} \sigma_{0}^{2}(t) dt \frac{1}{2\pi} \int_{-\infty}^{\infty} |H(i \bullet)|^{*} d\omega}{q^{*} \left[\frac{1}{T_{r}} \int_{-\infty}^{\infty} u_{0}^{2}(t) \sigma_{\bullet}^{*}(t) dt\right]^{*}}, \qquad (5.5.3)$$

where frequency response of the filter is normalized so that H(i0) = 1, and integral

$$\frac{1}{2\pi}\int_{-\infty}^{\infty}|H(i\infty)|^4 d\infty$$

has the order of $\Delta f_{y||y|}$ and is equal to $\Delta f_{y||y|}$ in the case of a square frequency response, $\frac{1}{2}\Delta f_{y||y|}$ with a frequency response of the UPCh coinciding with the frequency response of a single LRC-circuit, and $(1/\sqrt{2})\Delta f_{y||y|}$ with a Gaussian frequency response.

For validity of formula (8.5.4) there is required, in general, not only condition $\Delta f_{y\Pi \Psi} \tau_{\Pi} >> 1$, but also the condition that bandwidth of the UPCh is great as compared to the width of the spectrum of gates. From expressions (8.5.3) and (8.5.4) by means of calculations we can obtain particular results for any concrete form of the signal pulse and of gates, in particular for square gates of duration $\tau_{\Pi C}$ and detuning $\delta = \tau_{\Pi C}/2$:

$$S_{0KB} = T_{r} \frac{q \frac{1}{T_{r}} \int_{-\tau_{R0}}^{\tau_{R0}} u_{0}^{2}(t) dt + u\Delta f_{Y\Pi q} \tau_{R0}}{2q^{2} \left[\frac{u_{0}^{2}(0)}{T_{r}} + \frac{u_{0}^{2}(\tau_{R0}) + u_{0}^{2}(-\tau_{R0})}{2T_{r}} \right]^{4}}, \qquad (3.5.5)$$

where α - coefficient of the order of 0.5-1, characterizing the difference of integral $\frac{1}{2\pi} \int_{-\infty}^{\infty} |H(i\omega)|^4 d\omega$ from $\Delta f_{y\Pi \Psi}$.

8.5.2 Square Pulse

Let us consider first of all the case when the frequency response of the UPCh is matched with the spectrum of the pulse, and the gates are narrow as compared to its duration. Then

$$f(t) = \int_{0}^{\infty} h(s) h(s-t) ds = T_r \left(1 - \frac{|t|}{\tau_{s}}\right), |t| < \tau_{s}$$
(8.5.6)

and from formula (8.4.23) it follows that

$$S_{\theta R \theta} = T_{\tau} \delta \tau_{R} \frac{1 + q \left(1 - \frac{\delta}{\tau_{R}}\right)}{2q^{\delta} \left(1 - \frac{\delta}{\tau_{R}}\right)}.$$
 (3.5.7)

Just as in the case of a coherent discriminator, this expression approaches zero as $\delta \rightarrow 0$. Actually S_{DMB} seeks a value determined by the duration of the pulse edge. The dependence of S_{DMB} on detuning δ is identical with (7.6.3), describing the same dependence in a coherent discriminator.

With a broad-band UPCh and square gates of duration $\tau_{\rm HC}$ with dotuning ± 5 (5 \neq $\pm \tau_{\rm HC}/2 \geq \tau_{\rm A}/2$) from formula (3.5.5) we obtain

$$S_{\text{ORB}} = \frac{T_r \tau_{\text{R}}^2}{2q^5} (q + \alpha \Delta j_{y_{\text{Flu}}} \tau_{\text{RC}}). \qquad (12.23)$$

From this formula it follows that with finite duration of the gates ($\tau_{\rm HC} >> > 1/\Delta f_{\rm VEV}$) accuracy of range finding is finite for all 6 and is determined by pulse duration. The dependence of $S_{\rm SHE}$ on q for different $\Delta f_{\rm VEV}$ $\tau_{\rm IE}$ is shown in Fig. 8.5, from which it follows that with sufficiently large signal-to-noise ratios



Fig. 3.5. Equivalent spectral density for a discriminator with a broad-band UPCh: $--- \alpha \Delta f y_{\Pi H} \tau_{\Pi O} = 3; --- \alpha \Delta f y_{\Pi H} \tau_{\Pi O} = 6;$ $--- \alpha \Delta f y_{\Pi H} \tau_{\Pi O} = 10.$

q the considered discriminator is uncritical to the width of the UPCh passband and duration of the gates.

Let us further consider the case when the frequency response of the UPCh is matched with the pulse spectron, and gates have finite width and a square shape. In accordance with how this is most frequently done in practice, we shall consider that gates are located end-to-end, i.e., $5 = t_{\rm H}/c$. The gain factor of the discriminator here is proportional to

$$\frac{2\tau_{\pi e}}{\tau_{\pi}} \left(2 - \frac{\tau_{\pi e}}{\tau_{\pi}} \right) \qquad (3, 5, 9)$$

and reaches a maximum value at $\tau_{\rm HC} = \tau_{\rm H}$. Further expansion of the gates has no $g_{\rm HC}$ is since pulse duration with respect to zeroes at the output of the matched UPCh like $C\tau_{\rm HC}$ and at $\tau_{\rm HC} > \tau_{\rm H}$ the power of the useful signal at the discriminator output does not increase, and the power of noise components grows, inasmuch as integration of noise occurs over the whole large interval of time. Therefore we shall consider $\tau_{\rm HC} \in \tau_{\rm H}$. Substituting expression (3.5.6) and function $v_{\rm O}(t)$, describing square gates, in formula (3.4.13) we obtain as a result of integration the following $-z_{\rm H}$ restor:

$$S_{0KB} = \frac{T_r \epsilon_a^2}{2q^2} \left[a_s \left(\frac{\epsilon_{a,s}}{\epsilon_a} \right) q + a_s \left(\frac{\epsilon_{a,s}}{\epsilon_a} \right) \right], \qquad (3.5.10)$$

where a_1 , a_2 - functions describing the dependence of components of spectral density on the ratio of the durations of gates and of the pulse. These functions have the form

$$a_{i} = \frac{\frac{4}{3} \frac{\tau_{ue}}{\tau_{u}} - \frac{5}{3} \left(\frac{\tau_{ae}}{\tau_{u}} \right)^{2} + \frac{8}{15} \left(\frac{\tau_{ue}}{\tau_{u}} \right)^{4}}{\left(2 - \frac{\tau_{ae}}{\tau_{u}} \right)^{2}}, \qquad \frac{\tau_{ue}}{\tau_{u}} \le \frac{1}{2},$$

$$(8.5.11)$$

$$a_{1} = \frac{\frac{43}{240} - \frac{4}{3} \frac{\tau_{\alpha o}}{\tau_{\alpha}} + \frac{11}{3} \left(\frac{\tau_{\alpha c}}{\tau_{\alpha}}\right)^{3} - \frac{10}{3} \left(\frac{\tau_{\alpha c}}{\tau_{\alpha}}\right)^{3} + \left(\frac{\tau_{\alpha o}}{\tau_{\alpha}}\right)^{4}}{\left(\frac{\tau_{\alpha o}}{\tau_{\alpha}}\right)^{3} \left(2 - \frac{\tau_{\alpha o}}{\tau_{\alpha}}\right)^{3}}, \frac{\tau_{\alpha o}}{\tau_{\alpha}} > \frac{1}{2},$$

$$a_{s} = \frac{\frac{4}{3} \frac{\tau_{sc}}{\tau_{u}} - \left(\frac{\tau_{sc}}{\tau_{u}}\right)^{s}}{\left(2 - \frac{\tau_{sc}}{\tau_{u}}\right)^{s}}, \quad \frac{\tau_{sc}}{\tau_{s}} < \frac{1}{2},$$

$$a_{s} = \frac{\frac{1}{12} - \frac{2}{3} \frac{\tau_{sc}}{\tau_{u}} + 2\left(\frac{\tau_{sc}}{\tau_{u}}\right)^{s} - \frac{4}{3}\left(\frac{\tau_{uc}}{\tau_{u}}\right)^{s} + \frac{1}{3}\left(\frac{\tau_{uc}}{\tau_{u}}\right)^{s}}{\left(\frac{\tau_{sc}}{\tau_{u}}\right)^{s}}, \quad \frac{\tau_{sc}}{\tau_{u}} > \frac{1}{2}.$$

$$(8.5.12)$$

In particular, when $\tau_{\rm HC}/\tau_{\rm M} = 1$ coefficients a_1 and a_2 take values $a_1 = 0.13$ and $a_2 = 0.42$. For small $\tau_{\rm HC}/\tau_{\rm M}$ both coefficients have the order of $\tau_{\rm HC}/\tau_{\rm M}$. The dependence of a_1 and a_2 on $\tau_{\rm HC}/\tau_{\rm M}$ is shown in Fig. 8.6. For small signal-to-noise ratios q dura-



Fig. 8.6. Influence of ratio $\tau_{\rm HC}/\tau_{\rm H}$ for a discriminator with a matched UPCh on $s_{\rm HC}'$

tion of the gate rather strongly affects accuracy of measurement. For large signal-to-noise ratios this influence is less essential if duration of the gates is not very small — in the domain 1/2 < $<\tau_{nc}/\tau_{M} < 1$ coefficient a_{1} varies little. For small τ_{nc}/τ_{M} the dependence of a_{1} on τ_{nc}/τ_{M} is approximately linear. Thus, with a matched UPCh it is useful to decrease duration of the gates; in spite of decrease of the gain factor of the discriminator quantity S_{BKB} decreases with decrease of τ_{nc}/τ_{M} . For very small τ_{nc}/τ_{M} results of analysis, of course, do not reflect reality — we need allowance for the finite

duration of the edge and nonlinearity of the discrimination and fluctuation characteristics.

We shall briefly discuss discriminators with correlation processing. For a two-channel discriminator with detuned channels we need only repeat what was already said in examining coherent range finders (see Paragraphs 7.6.1 and 8.3.1). For a

-367-

discriminator with switching of reference signals of interest is the dependence of equivalent spectral density on detuning. Considering that the duration of the gate pulse, which in this case is the reference signal, $\tau_{\rm RC} < \tau_{\mu}$, on the basis of formulas (7.6.7) and (8.3.16) we obtain

$$S_{\text{OKB}} = \frac{T_r \tau_c^2}{4q^2} \frac{1 + 2q \frac{\tau_u}{\tau_c} \left(1 - \frac{\delta}{\tau_u}\right)}{\left(1 - \frac{\delta}{\tau_u}\right)^2}, \qquad (8.5.13)$$

As follows from this expression, equivalent spectral density monotonically decreases with decrease of detuning; however, even as $\delta \rightarrow 0$ accuracy remains finite. With magnitudes of detuning close to $\tau_{\rm M}$, spectral density increases without limit, which is caused by decrease of the gain factor. Here increase of $S_{\rm OHB}$ is faster that in a two-channel discriminator. An optimum for detuning, which existed in the coherent case, is absent here.

It is interesting to compare the accuracy of the discriminator with switching of reference signals with the accuracy of a two-channel correlation-type discrimina-



rig. 3.7. The influence of detuning 5 and the signal-to-noise ratio q on S_{3KR} for a discriminator with switching of reference signals. tor, for which quantity S_{BKB} with coinciding denotions of the pulse and gate pulses is determined, obviously, by formula (8.5.7). The ratio of quantities S_{BCB} for both cases is equal to

$$\frac{S_{\text{BRB1}}}{S_{\text{BRB2}}} = \frac{1 + 2q\left(1 - \frac{\delta}{\tau_{\text{H}}}\right)}{2\frac{\delta}{\tau_{\text{H}}}\left(1 - \frac{\delta}{\tau_{\text{H}}}\right)\left[1 + q\left(1 - \frac{\delta}{\tau_{\text{H}}}\right)\right]}, \quad (8, 5, 1^{1/2})$$

where $S_{\partial KB1}$ is expressed by formula (3.5.13) when $\tau_{c} = \tau_{w}$.

For small q the discriminator with switching of reference signals gives a loss of at least one half, and for large q, at least one fourth. In both cases minimum loss is attained at $\delta = 0.5\tau_{\rm N}$. The dependence of

ratio (8.5.14) on detuning δ for various q is shown in Fig. 8.7.

8.5.3. Gaussian Pulse

Let us consider now the case when the form of the pulse is approximated a Gaussian curve of form (7.6.9). In the case of square gates and a UPCh with a

broad band as compared to $1/\tau_{N}$ on the basis of (8.5.5) we have

$$S_{0KB} = T_{r} \epsilon_{a}^{2} \frac{a \Delta f_{V\Pi u} \epsilon_{a} \epsilon + q \left[2\Phi \left(\sqrt{2\pi \epsilon_{a}} e^{/\epsilon_{a}} \right)^{-1} \right]}{2q^{2} \left[1 - \exp \left(- \pi \epsilon_{a}^{2} / \epsilon_{a}^{2} \right) \right]^{2}}, \qquad (8.5.15)$$

where $\Phi(x)$ - integral of probability.

For small $\tau_{\Pi C}$ this expression is valid if $\Delta f \gamma_{\Pi C} > 3$ to 5. For larger $\tau_{\Pi C} / \tau_{M}$ and q quantity S_{BKB} is determined by asymptotic expression

$$S_{\text{BKB}} \approx \frac{T_r s_{\pi}^2}{2q}, \qquad (8.5.16)$$

which is $\pi/2$ times larger than the magnitude of S_{ONT} from (8.2.17) for large q (let us remember that for a Gaussian pulse a = 0, $b = \pi/2\tau_M^2$).

Analysis of formula (8.5.15) shows that duration of the gates should be selected sufficiently great as compared to $\tau_{\rm N}$, since increase of $\tau_{\rm ND}^{\prime}/\tau_{\rm N}$ increases the gain factor to a greater degree than the spectral density of output voltage of the discriminator. For very small $\tau_{\rm ND}^{\prime}/\tau_{\rm N}$ the asymptotic expression for S_{3KB} has the form

$$S_{\text{prp}} \approx \frac{T_r \tau_n^6 \, \epsilon \Delta f_{\text{ying}}}{2\pi^2 \tau_{\text{pc}}^2 \, q^2}. \tag{S.5.17}$$

Fractically, ratio $\tau_{\rm IE}^{\prime}/\tau_{\rm N}$ should be near 1.0. Further increase of duration of the gates already almost does not change the gain factor of the discriminator but



Fig. 8.8. The dependence of ratio S_{attr}/S_{ONT} on q for the case of a Gaussian pulse and a broad-band UPCh: $\alpha \Delta f y_{\Pi Q} \tau_{\Pi C} = 3; - - \alpha \Delta f y_{\Pi Q} \tau_{\Pi C} = 10.$

increases the influence of the noise term in S_{3KE} The dependence of ratio $S_{3KE} S_{OUT}$ on q for different $\alpha\Delta f y_{\Pi Q} \tau_{\Pi C}$ and τ_{1C} / τ_{1V} is shown in Fig. 8.8.

We shall consider now an example of a more general nature. Let us assume that frequency response of the UPCh has the form of a Gaussian curve of arbitrary width, i.e.,

and gates also are Gaussian with width γ_{IIC} and detuning $\delta,$ i.e.,

Designating
$$\tau_{\mu}/\tau_{\mu c} = y$$
, $\Delta f_{y \mu \mu}/\Delta f_{corv} = 2\Delta f_{y \mu}$, $z = x (\Delta f_{corv} = 1/2\tau_{\mu})$, $z = 1/2\tau_{\mu}$

= $\sqrt{2\pi}\delta/\tau_{\rm p}$ and performing necessary calculations by formulas (8.4.18) and (8.4.19), for equivalent spectral density we obtain the following expression:

$$S_{9KB} = \frac{T_{\tau}\tau_{K}^{2}(1+x^{3})[2x^{3}+y^{4}(1+x^{2})]^{3}}{\pi q^{2} \cdot 16z^{8}x^{4}y^{4}\exp\left\{-\frac{z^{2}x^{2}y^{2}}{2x^{2}+y^{2}(1+x^{2})}\right\}} \times \left\{\frac{1-\exp\left\{-\frac{x^{2}x^{2}y^{2}}{yx^{4}\sqrt{2x^{2}+y^{2}}}\right\}}{yx^{4}\sqrt{2x^{2}+y^{2}}}\right\} + \frac{4q\exp\left\{-\frac{z^{4}x^{2}y^{4}}{2(x^{2}+y^{2}+x^{2}y^{2})}\right\}}{x\sqrt{(x^{2}+y^{2}+x^{2}y^{2})}} \times \left[1-\exp\left\{\frac{-\frac{-z^{2}x^{2}y^{4}(1+x^{2})^{2}}{2(x^{2}+y^{2}+x^{2})[y^{2}+x^{2}(2+y^{2}+x^{2})]}\right\}\right\}}{(y^{4}+x^{4}(2+y^{2}+x^{4})]}\right\}.$$
 (8.5.20)

When x = 1 ($\Delta f_{Y\Pi \Psi} = \Delta f_{COIM}$), $y \to \infty$ ($\tau_{\Pi C} \to 0$), $z \to 0$ ($\delta \to 0$) this expression turns into S_{OH1} with $b = \pi/2\tau_R^2$, reaching its minimum. Investigation of formula (8.5.20) shows that S_{BCB} comparatively weakly depends on detuning z, especially if the passband of the UPCh is close to the matched band. For small z the expression for equivalent spectral density takes the form

$$S_{\mathbf{3}\mathbf{K}\mathbf{B}} = \frac{T_{\mathbf{r}} \mathbf{x}_{\mathbf{R}}^{2}}{\pi q^{3}} \cdot \frac{(1+x)^{3} [2x^{2} + y^{2} (1+x^{2})]^{3}}{16x^{2} y^{3}} \cdot \left\{ \frac{1}{yx (2x^{2} + y^{3})^{3/2}} + \frac{2qy^{2} (1+x^{2})^{2}}{(y^{2} + x^{2} + x^{2} y^{3})^{3/2} (2x^{2} + y^{2} + y^{2} x^{2} + x^{4})^{3/2}} \right\}, \qquad (6.5, 21)$$

which is considerably more convenient for calculations.

The dependence of S_{are}/S_{OUV} on q for various x and y is shown in Fig. 3.9. Curves of this figure show that loss in accuracy increases with decrease of the



Fig. 8.9 The dependence of ratio S_{acc}/S_{off} on q for Gaussian gates and pulse: ---x = 0.3; ---x = 1; --x = 3; --x = 10.

signal-to-noise ratio q and increase of duration of the gate and the difference between Δf_{JTT} and Δf_{COTM} . This loss for small q can reach substantial magnitude. Analysis of formulas (8.5.20), (8.5.21), and Fig. 8.9 permits us to draw the following conclusions:

1. For $\Delta f_{y\Pi Y} \leq \Delta f_{OOTM}$ the magnitude of S_{GOB} is less, the larger y, i.e., the less the duration of the gate. When $\Delta f_{Y\Pi Y} >$ > Δf_{COTM} duration of the gate should be increased to a magnitude of the order of τ_{u} (y ~ 1). The necessity for this is the faster drop of the gain factor with reduction of the gates as compared to decrease of the spectral density at the discriminator output in the case of an expanded passband of the UPCh.

2. The influence of expansion of the passband, in particular, on the component of S_{336} with factor 1/q is stronger, the shorter the gates.

3. For large x and y the magnitude of S_{BRB} decreases with increase of detuning to values ensuring maximum slope of the discrimination characteristic, i.e., to $z = \sqrt{(2x^2 + y^2 + x^2y^2)/xy}$. For instance, when x = 3 and y = 10, selecting z from this condition, we obtain for S_{BRB}/S_{ORT} the following values: $S_{BRB}/S_{ORT} = 1.51$ for q = 0.1; $S_{BRB}/S_{ORT} = 1.41$ for q = 3; $S_{BRB}/S_{ORT} = 1.25$ for q = 100, which is considerably smaller than corresponding values for small values of z.

4. In general, selection of parameters of the discriminator should be carried out in such a manner that x = 1 to 3, y = 1 to 3, and the magnitude of detuning ensures a maximum gain factor. Then, for all signal-to-noise ratios q the magnitude of S_{area} will not exceed S_{ORT} by more than a factor of two, and, in practice, the range finder will realize its potential accuracy.

Characteristics of a two-channel discriminator with correlation processing of the signal for the considered case were actually already found in Chapter VII. Equivalent spectral density here is determined by formula (8.3.13), where quantities $b_{\rm SMB}$ and $q_{\rm SMB}$, determined by formulas (8.3.14) and (8.3.15), are given for an arbitrary duration of gates of Gaussian form and arbitrary detuning by expressions (7.6.17) and (7.6.18) and the curves of Fig. 7.17. Let us remember that ratio $q_{\rm SMB}/q = h_{\rm SMB}/h$. Therefore, it remains only to repeat what was said in Paragraph 7.6.2 about the given case.

We shall investigate further the influence of the magnitude of detuning on accuracy of a correlation-type discriminator with switching of reference signals. We shall consider that durations of gate pulses coincide with pulse duration. Then

$$S_{aHa} = \frac{T_r \tau_{H}^2}{\pi q^2} \cdot \frac{1 + 2qe}{\frac{\pi \delta^2}{2\tau_{H}^2}} - \frac{\frac{\pi \delta^2}{2\tau_{H}^2}}{\frac{\pi \delta^2}{\tau_{H}^2}}.$$
 (8.5.22)

For both large and shall $5/\tau_{\rm H}$ the magnitude of $S_{\rm HB}$ for all q increases without limit. The minimum of $S_{\rm HB}$ for small q is reached with detuning δ which ensures a maximum gain factor, i.e., at $\xi = \sqrt{\tau_{\rm H}^2/\pi}$, and for large q at $\delta = \sqrt{2\tau_{\rm H}^2/\pi}$. The ratio

-371-



1.00.100.0

Fig. 8.10. The dependence of S_{310}/S_{DH1} on $\delta/\tau_{\rm E}$ for a discriminator with switching of reference signals with a Gaussian form of the pulse.

of S_{abb} from (d.5.22) to S_{OUT} , as a function of detuning 5, is shown for various q in Fig. 8.10. As can be seen from this figure, loss can attain substantial magnitudes; even for optimum selection of detuning ratio S_{abb}/S_{abb} is 3. For small detunings the dependence of loss on the signal-to-noise ratio is not very substantial; for large δ/τ_{w} loss is greater, the smaller q.

§ 8.6. Pulse Amplitude Modulation

With incoherent radiation more frequently than with coherent there arises the necessity of expanding the spectrum of the sounding signal by intrapulse modulation. In this and the following paragraphs we shall consider several examples of such modulation and corresponding schemes for construction of discriminators. Furthermore, we shall consider subsequently that in accordance with requirements of practic the width of the spectrum of intrapulse modulation is great as compared to the width of the pulse spectrum.

Let us consider, first, amplitude modulation of a pulse by sinusoidal law with frequency $\omega_{\rm m} \left(\omega_{\rm m} \gg \frac{1}{\tau_{\rm p}} \right)$. Obviously, most convenient from the technical point of view for the given case is the scheme of a discriminator with correlation processing and differentiation of the reference signal. Since for large $\omega_{\rm m} \tau_{\rm M}$ it is possible to disregard the possibility of additional increase of accuracy of measurement by using the form of the pulse envelope,* which gives a relative contribution to

-372-

^{*}We have in mind a pulse envelope without sinusoidal modulation.

accuracy of the order of $1/(\omega_m \tau_H)^2$, then it is sufficient to carry out differentiation of the reference signal only with respect to the additional amplitude modulation, not differentiating the pulse envelope. Then the operation of differentiation is realized by shift of the modulating voltage in phase and heterodyning.

The block diagram of the discriminator allowing for this circumstance is shown in Fig. 8.11. The received signal after transfer to intermediate frequency and preamplification proceeds to two mixers, to which there are fed voltages m cos $\omega_m \times (t - \tau)$ and m sin $\omega_m (t - \tau)$ with phase $\omega_m \tau$, controlled by the output quantity of the range finder. The signal from the output of the cosine mixer is added to the input signal of the mixer. Then both signals are gated, are approximately matched with the duration of the signal pulse by gate pulses, are passed through filters, from which there is required sufficient identity of phase responses, and they are mixed in the phase detector, forming output voltage of the discriminator. With respect to the bandwidth of the filters all remarks in §§ 8.2 and 8.3 are valid. However, in this case we assume that bandwidth of the filter does not necessarily satisfy condition $\Delta f_{\oplus} - \tau_{\mu} << 1$, but has arbitrary magnitude.



Fig. 8.11. Block diagram of a discriminator for a signal with additional sinusoidal mc⁻¹ lation: 1 - local oscillator; 2 - mixer; 3 - p.eamplifier; 4 - generator of sinusoidal oscillations with controlled phase; 5 - mixer; 6 - adder; 7 - controlled generator of gate pulses; 8 broad-band gated amplifier; 9 - i-f filter; 10 - phase detector.

If condition $\Delta f_{\phi} \tau_{N} \ll 1$ is satisfied, the equivalent spectral density is determined by formula (8.3.19). The law of modulation of the sounding signal can be recorded in this case in the form

$$u_{0}(l) = u_{a0}(l) \frac{1 + m \cos \omega_{al} l}{\sqrt{1 + \frac{m^{2}}{2}}}, \qquad (8.6.1)$$

and for reference signals

$$u_{1}(l) = v(l) \frac{1 + m \cos u_{m} l}{\sqrt{1 + \frac{m^{2}}{2}}}, \qquad (8.6.2)$$

$$u_{s}(l) = v(l) \frac{m\omega_{m} \sin \omega_{m} l}{\sqrt{1 + \frac{m^{2}}{2}}}, \qquad (3.6.3)$$

ي يتول شد خان شكرته من من من من من من من المعمد المار من الم

where it is assumed that gate pulses in both channels are identical and are described by function v(t), which we for convenience can normalize so that

$$\frac{1}{T_r} \int_{-\infty}^{\infty} \dot{\sigma}^{t}(t) dt = 1.$$
 (8.6.4)

Such normalization is valid, too, for the pulse envelope $u_{a0}(t)$.

If durations of pulses $u_{a0}(t)$ and v(t) are great as compared to $1/\omega_m$, the cross-correlation functions $C_{ik}(x)$ will have the form:

$$C_{10}(x) = \frac{1 + \frac{m^2}{2} \cos \omega_m x}{1 + \frac{m^2}{2}} \cdot \frac{1}{T_r} \int_{-\infty}^{\infty} u_{A0}(t) v(t) dt, \qquad (8.6.5)$$

$$C_{e_{1}}(x) = -C_{1e}(x) = C'_{10}(x) = -\frac{\omega_{m}m^{2}\sin\omega_{m}x}{2\left(1+\frac{m^{2}}{2}\right)} \cdot \frac{1}{T_{r}} \int_{-\infty}^{\infty} u_{1e_{1}}(t) v(t) dt, \qquad (8.6.1)$$

$$c_{11}(x) = -\frac{\omega_m m^2 \sin \omega_m x}{2\left(1 + \frac{m^2}{2}\right)}, \qquad (8.6.7)$$

$$C_{11}(x) = -\frac{1 + \frac{m^2}{2} \cos \omega_m x}{1 + \frac{m^2}{2}}, \qquad (3.6.8)$$

$$C_{s2}(x) = \frac{m^2 \omega_m^2 \cos \omega_m x}{2\left(1 + \frac{m^2}{2}\right)}.$$
 (0.6.9)

Then, on the basis of formula (8.3.13) for equivalent spectral density we obtat the following expression:

$$S_{\text{DNB}} = T_{r} \frac{2\left(1+\frac{m^{2}}{2}\right)}{m^{2}\omega_{m}^{2}} \cdot \frac{1+q\left(\frac{1}{T_{r}}\int_{-\infty}^{\infty}u_{a,0}(t)\,v(t)\,dt\right)^{2}}{2q^{2}\left(\frac{1}{T_{r}}\int_{-\infty}^{\infty}u_{a,0}(t)\,v(t)dt\right)^{4}} = \frac{1+q_{0,R,0}}{2bq_{0,K,0}^{2}}, \quad (3.6.10)$$

$$b = \frac{m^2 \Theta_m^2}{2\left(1 + \frac{m^2}{2}\right)}$$
(8.6.11)

-374-

where '

is the mean square width of the spectrum of modulation for the considered case, and

$$q_{\text{DRB}} = q \left(\frac{1}{T_r} \int_{-\infty}^{\infty} u_{\text{Ab}}(t) v(t) dt \right)^t$$
(8.6.12)

is the equivalent signal-to-noise ratio, decrease of which is caused by mismatch of durations of the sounding pulse and the gate pulses. With sufficiently large τ_c/τ_n ratio $q_{\rm SEP}/q$ for any form of pulse and gate pulses is equal to τ_n/τ_c ; for a square pulse and square gate pulses there is exact equality when τ_c/τ_n , and, conversely, $q_{\rm SEP}/q = \tau_c/\tau_n$ when $\tau_c < \tau_n$.

Thus, nonoptimality of processing, consisting of the mismatching of durations of the sounding and reference signals, leads in this case simply to corresponding decrease of the signal-to-noise ratio. If in circuits of this type we cannot achieve identity of phase responses of the two channels, then, just as in corresponding circuits of coherent discriminators, there occurs additional increase of errors of measurement. The equivalent spectral density increases proportionally to $1/\cos^{2} \varphi$. where φ — phase delay in one channel relative to the other.

Now let us assume that the bandwidth of the filter satisfies requirement $\Delta f_{\oplus} \gg 1/T_r$, but is not limited from above. Then, using the expression for output voltage of the discriminator and performing the same transformations as in §§ 3.2 and 8.3, for equivalent spectral density we can obtain the following expression:

$$S_{9RB} = \frac{T_{*}\left(1+\frac{m^{2}}{2}\right)}{s^{2}\omega_{m}^{2}q^{2}} \frac{\int_{-\infty}^{\infty}\int_{-\infty}^{\infty}|H(i\omega_{1})|^{2}|H(i\omega_{2})|^{2}\{|V[i(\omega_{1}+\omega_{2})]|^{2}+\frac{1}{s^{2}\omega_{m}^{2}q^{2}}\left|\int_{-\infty}^{\infty}|H(i\omega)|^{2}\times\frac{1}{s^{2}}+\frac{qF(i\omega_{1})F(i\omega_{2})V^{*}[i(\omega_{1}+\omega_{2})]\}d\omega_{1}d\omega_{2}}{\times 1F(i\omega)|^{2}d\omega}\right|^{2}}, \quad (6.6.13)$$

where $H(i\omega)$ - frequency response of the filter;

 $V(i\omega)$ - Fourier transform of function $v^2(t)$;

 $F(i\omega)$ - Fourier transform of function $u_{a0}(t)v(t)$.

If function $|H(i\omega)|$ is narrow as compared to functions $|V(i\omega)|$ and $|F(i\omega)|$, expression (8.6.13) changes into (8.6.10), and for $v(t) = u_{a0}(t)$ into the expression for S_{OIIII} with coefficient b, determined by formula (8.6.11). In general, e.g., for a Gaussian form of pulses, of gat's pulses and of the frequency-response curve of filters with passband $\Delta f_{di} = \Delta f_{VIIII}$ from formula (8.6.13) it follows that

$$S_{yxy} = \frac{T_r \left(1 + \frac{m_s}{2}\right)}{m^2 \omega_m^2 q^4} \left\{ \frac{(1+y)^2 \left(1 + x^2 + y^2\right)}{4y \sqrt{x^2 + y^4}} + q \frac{(1+y)^2 \left(1 + x^2 + y^2\right)}{\sqrt{(x^2 + 2y^2 + 2)(x^2 + 2y^2 + 2y^2 x^2 + 2y^4)}} \right\}, \quad (3.6.14)$$

where all designations are the same as in Paragraph 8.5.3. This formula permits of to investigate a series of limiting cases. For instance, with expansion of the passband of the filter $(x \rightarrow \infty)$ and $y \ll 1$

$$S_{0RB} = \frac{T_r \left(1 + \frac{m^2}{2}\right)}{m^2 \omega_m^2 q^4} \left(q + \frac{x}{4y}\right) = \frac{T_r \left(1 + \frac{m^2}{2}\right)}{m^2 \omega_m^2 q^4} \left(q + 0.5\Delta f_{y_{\Pi H}} \tau_0\right), \qquad (5.6.15)$$

which, with the exception of the proportionality factor, depending on the form of modulation, coincides with the expression for equivalent spectral density of a discriminator for a simple pulse signal with a broad-band UPCh.

If duration of the gate pulse considerably exceeds pulse duration $\left(y = \frac{\tau_{12}}{\tau_{0}} < 1\right)$,

$$S_{0RB} = \frac{T_r \left(1 + \frac{m^2}{2}\right)}{m^2 \omega_{\mu}^2 q^{0^2}} \left(\frac{1 + x^2}{4yx} + q \frac{1 + x^4}{x \sqrt{2 + x^2}}\right), \qquad (8.6.46)$$

and, finally, with a filter which is narrow-band as compared to the width of the spectrum of the pulse formula (8.6.10) is valid, where in this case

$$\frac{q_{\text{DMB}}}{q} = \frac{2\frac{\tau_{\text{M}}}{\tau_{0}}}{1 + \left(\frac{\tau_{\text{M}}}{\tau_{0}}\right)^{3}}.$$
(8.6.17)

From formulas (8.6.14) and (8.6.16) it follows that expansion of the filter paraband and of duration of the gate pulse for not very small values of x basically affects the noise term in S_{SFG} (i.e., the term with $1/q^2$) and, consequently, is especially undesirable with a small signal-to-noise ratio.

The dependence of S_{acc}/S_{ORF} on q for various x and y. illustrating the influence of basic parameters of the discriminator on accuracy, is shown in Fig. ².12.

The scheme of a discriminator with correlation processing of the signal and differentiation is, of course, not unique for the given case. For instance, it is also possible to use a discriminator with a "shortening" filter, which is realized in this case by three i-i amplifiers matched with the spectrum of the pulse envelope, tuned to frequencies $\omega_{\rm EF}$ and $\omega_{\rm EF} \pm \omega_{\rm m}$, respectively. Gain factors of the UPCh's tuned to frequencies $\omega_{\rm EF} \pm \omega_{\rm m}$, should be a factor of m/2 less than the gain factor of the basic UFCh. Output voltages of all three UFCh's are added, detected and fed

-376-


to gating stages. With sufficiently rational selection of parameters the characteristics of this discriminator will be the same as those considered above.

§ 8.7. Frequency Intrapulse Modulation

One of the most studied forms of intrapulse modulation at present is linear frequency modulation [38, 39]. The form of the pulse here, as a rule, is selected symmetric. With the given form of modulation various methods of construction of discriminators may be reasonable from the technical point of view. Let us consider first a discriminator with a shortening filter. The shortening filter for the case of frequency modulation can be made in different ways. The most well-known is use of delay lines with taps [12, 38]. Amplitudes and phases of signals taken from the different taps will be selected in such a manner that as a whole the frequency response of the filter is conjugate with the spectrum of the signal.

ومقاربة ومقابط معلقات فالقر فللا الالان فللمان التحرر عرب خاطات الكلالة

Influence of the form of the pulse envelope on accuracy and range resolution capability was already discussed in Chapter VII (see Paragraph 7.8.1); in this case, as follows from general expressions of §§ 8.2, 8.3, and 3.4, in this respect everything remains constant. Therefore, we will be interested basically only in the influence of nonoptimality of processing on accuracy of range finding.

Let us assume first that the shortening filter is optimum. Then for gates which are narrow as compared to the duration of the shortened pulse the equivalent spectral density is determined by expression (8.2.17) for S_{OIII} . Parameter b in this

-377-

formula is determined by the following expressions:

for a square pulse envelope

$$b = \frac{a^{4}\tau_{\pi}}{12} = \frac{b_{\pi}^{2}}{3}$$
. (3.7.1)

for a Gaussian envelope

$$b = \frac{a^{2}r_{u}^{2}}{2\pi} = \frac{2u_{m}^{2}}{\pi}, \qquad (3.7.2)$$

for a cosinusoidal envelope

$$b = \frac{\pi^{4} - 6}{3\pi^{2}} a^{2} \tau_{x}^{2} = \frac{4(\pi^{2} - 6)}{3\pi^{2}} \omega_{m}^{2} = 0,52\omega_{m}^{2}, \qquad (3.7.3)$$

where a - speed of change of frequency within limits of pulse;

 $n_{\rm u}$ - effective pulse duration;

 $\omega_{\rm p}$ = at $_{\rm p}$ - effective frequency deviation for the duration of a pulse.

With finite duration of gates accuracy somewhat worsens. Let us consider, for instance, the case when the pulse envelope has the form of a Gaussian curve. $\Im(r, 3, 6)$ the form of the shortened pulse is also described by a Gaussian curve of form

$$\mathbf{f}(t) = \mathbf{e}^{-\frac{\mathbf{e}^{2}t^{2}}{\mathbf{e}}}.$$
(3.7)

÷)

This means that the considered case with m optimum shortening filter is equivalent to simple pulse modulation by Gaussian p areas with duration

and a discriminal r with a matched UPCh. Assuming the form of the gates Gaussian (1,2,...), we see that for calculation of S_{GKB} it is possible to use expression (1,...,1) in which one should set x = 1 (Af_{JIE} = Af_{COLT}). Then for the considered discriminator formula (3.5.21) is also valid, corresponding to small defuning between gates; curves of Fig. 8.9 and all conclusions of Paragraph 8.5.3 pertain to the case x = 1.

Sometimes in practice one can use a realization of the shortening filter, simpler in certain respects, which ensues directly from the presentation of output voltage of the discriminator in the optimum case. As we already said above, this voltage is determined through the modulus of quantity f_j from (8.2.1), which for any period, when we use optimum shortening filtration is recorded in the form

$$f(\mathbf{s}) = \left| \int_{\mathbf{s}} \mathbf{u}_{\mathbf{s}} \left(\mathbf{t} - \mathbf{s} \right) \mathbf{y}(s) e^{i \mathbf{s} \cdot \mathbf{s}} \, ds \right|,$$

-378-

where moment τ in the upper limit of the integral coincides with delay of the reflected signal, $u_0(t)$ in this case is $u_{a0}(t)e^{iat^2/2}$, a - rate of linear change of frequency.

Then

$$f(\tau) = \left| \int_{-\infty}^{\tau} u_{s_0}(s - \tau) e^{\frac{ias^{n}}{2} - iaw + \frac{iav^{n}}{2} + i\omega_{s}\tau} y(s) ds \right| = \\ = \left| \int_{-\infty}^{\tau} u_{s_0}(s - \tau) e^{\frac{iaw}{2} + av_{1}(\tau - s)} y(s) e^{\frac{i}{2}(\omega_{s} + \omega_{np})s + i\frac{as^{n}}{2}} ds \right| = \\ = \left| \int_{-\infty}^{\tau} h_{s}(\tau - s) e^{\frac{i}{2}(\omega_{np} + av_{1})(\tau - s)} \eta(s) ds \right|.$$
(8.7.6)

This means that the output voltage of the optimum shortening filter at time $t = \tau$ can be formed by heterodyning the received signal y(t) with voltage of a frequency-modulated local oscillator and passing it through a filter with a frequency response matched with the spectrum of the pulse envelope. This filter should be tuned to frequency $\omega_{\pi\pi} + a\tau$, depending on the distance interesting us.

In order to ensure full equivalence with the usual shortening filter, forming the integral of $f(\tau)$ for all delays τ_i , we need a unit of identical filters tuned to frequencies $\omega_{\rm EF} + a\tau_i$, and a unit of lagging gate pulses by which there should be carried out tapping of output voltages of filters, each at the corresponding moment τ_i . The set of filters and gate pulses gives at the output a shortened pulse.

Optimality of processing is preserved, of course, only when the pulse response of the filter of the UPCh satisfies the condition of matching $h_1(t) = h_0(t) = u_{a0}(-t)$, and gate pulses are narrow as compared to the duration of the shortened pulse. Such a realization of a shortening filter may be more convenient technically when the radar is intended for simultaneous work in a small range of distances. Then the number of filters can be comparatively small; gate pulses, if the usual condition of small duration of the shortened pulse as compared to the duration of the sounding pulse envelope is satisfied, can be fed simultaneously, and voltage of the local oscillator can also be pulsed. Here it is necessary only that the pulse duration of the local oscillator be sufficient for complete overlapping of the signal pulse. The possibility of using pulse mode in the local oscillator frees us from the necessity of providing too large deviations of its frequency.

For realization of a discriminator sufficient are two channels, tuned to frequencies $\omega_{\rm HT}$ + at ± a5, where τ - measured value of delay. For tuning a range

-379-

discriminator we can use change of the intermediate frequency, so that $\omega_{\rm PP} = \omega_{\rm DP} + 4\pi$. From this point of view such a discriminator is like a discriminator of correlation type in which the reference signal is changed in accordance with change of the measured value of delay. In this case the change consists not in shift of the reference signal, but in change of the local oscillator frequency. The block diagra of the discriminator is shown in Fig. 8.13.



Fig. 8.13. Block diagram of a discriminator for frequency-modulated pulses. 1 - preamplifier; 2 - amplitude- and frequency-modulated local oscillator with controlled delay of amplitude modulation; 3 - mixer; h = 100 oscillator with controlled frequency; h = 100 with frequency of tuning $\omega_{\rm H} = \alpha \delta$; $\delta = 000$ with frequency of tuning $\omega_{\rm H} = \alpha \delta$; $\delta = 000$ with frequency of tuning $\omega_{\rm H} = \alpha \delta$; $\beta = 000$ with frequency of tuning $\omega_{\rm H} = \alpha \delta$; $\beta = 000$ with frequency of tuning $\omega_{\rm H} = \alpha \delta$; $\beta = 000$ with frequency of tuning $\omega_{\rm H} = \alpha \delta$; $\beta = 000$ with frequency of tuning $\omega_{\rm H} = \alpha \delta$; $\beta = 000$ with frequency of tuning $\omega_{\rm H} = 000$ mp and $\beta = 000$ with frequency of tuning $\omega_{\rm H} = 000$ mp and $\beta = 000$ mp for the second second

To find characteristics of the discriminator we can use general formulas of § 3.4. We shall limit ourselves to the ε self narrow gate pulses, but we shall consider the UPCh frequency response $H_{d_{1}}(\varepsilon_{2})$ arbitrary. From comparison of (3.4.1) with (3.7.6) it follows that the complex pulse response of the filter h(t) in this case should be set equal to

$$h(l) = h_1(l) e^{\frac{lal^2}{2}}$$
. (2.7.7)

コート 一日 御堂 アート

Then, assuming duration of the pulse envelope great as compared to duration of the shortened pulse, we obtain

$$a(\delta) = \frac{1}{T_r} \int_{-\infty}^{\infty} h(t+\delta) u^*(t) dt \approx \frac{1}{T_r} \int_{-\infty}^{\infty} h_1(t) u_{ab}(t) e^{iaM} dt, \qquad (3.7.6)$$

$$\boldsymbol{b}(\boldsymbol{\delta}) = \frac{1}{T_r} \int_{-\infty}^{\infty} h^{\boldsymbol{\bullet}}(t) h(t+\boldsymbol{\delta}) dt \approx \int_{-\infty}^{\infty} h_i^2(t) e^{i\boldsymbol{a}\boldsymbol{\delta}t} dt \qquad (9.7.9)$$

and the expression for equivalent spectral density will take the form

-- 380--

$$S_{\theta R \theta} = T_{\tau} \frac{b^{4}(0) - |b(2\delta)|^{2} + \frac{2q[b(0)|a(\delta)|^{2} - \text{Re} a^{4}(\delta) b(2\delta)]}{8q^{4} [\text{Re} a^{4}(\delta)_{j}a'_{j}(\delta)]^{2}}.$$
(8.7.10)

In particular, for a Gaussian form of the pulse and of frequency-response curves of the UPCh

$$a(\mathbf{\delta}) = \sqrt{\frac{2}{1+4\Delta/^{2}_{y\Pi q}\tau^{2}_{\pi}}} e^{-\frac{4A}{2\pi(1+4A/^{2}_{y\Pi q}\tau^{2}_{\pi})}} = \sqrt{\frac{2}{1+x^{2}}} e^{-\frac{2m^{2}t^{2}}{\pi(1+x^{4})}}, \quad (8.7.11)$$

 $b(\delta) = \frac{1}{2\Delta f_{y_{\Pi}q^{q_{x}}}} e^{-\frac{a^{q_{y_{\Pi}}}}{16\pi \Delta f_{y_{\Pi}q}^{2}}} = \frac{1}{x} e^{-\frac{a^{2}y}{\pi x^{4}}}, \qquad (3.7.12)$

where, as before, $x = \Delta f_{y\Pi U} / \Delta f_{cor \pi} = 2\Delta f_{y\Pi U} \tau_{u}$, and $\omega_{m} = a \tau_{u} / 2$.

Then according to (8.7.10) equivalent spectral density is equal to

$$S_{0HB} = = \frac{T_{\pi}}{2\omega_{m}^{2}} \frac{(1+x^{2})^{4} e^{\frac{4x^{2}}{1+x^{4}}}}{128z^{4}} \times \left\{ \frac{-\frac{4u^{4}}{z^{4}}}{x^{4}} + \frac{4qe^{-\frac{2z^{4}}{1+z^{4}}}}{x(1+x^{4})} \left(1-e^{-\frac{2z}{z^{4}}}\right) \right\}, \quad (3.7.13)$$

where $z = \sqrt{55} = \left(\sqrt{\frac{2\omega_m^2}{m}}\right)^5$ - relative magnitude of detuning.

When x = 1 and $z \to 0$ formula (8.7.13) changes into the expression for S_{OUT} with coefficient b (from 8.7.2). As also in other cases, the dependence on z is relatively weak; practically up to values $z/x \sim 1$ in formula (8.7.13) it is possible to consider detuning zero, especially with a sufficiently large signal-to-noise rate. The exact dependence of S_{OUT}/S_{OUT} on detuning with a matched band (x = 1) is shown in Fig. 8.14, from which it is clear that increase of detuning to a value providing a maximum gain factor of the discriminator ($z \sim 0.7$) increases S_{OUT} even when $q \sim 0.1$ by not more than 15%. The influence of mismatching of the band of the UPCh for various q is illustrated in Fig. 8.15. This dependence turns out to be rather strong — expansion of the passband in the considered circuit leads, probably, to not anpleasant consequences than in any other one.







Fig. 5.15. Influence of mismatch of the passband of the UPCh on $S_{\rm OHB}$ in an FM range finder.

The block diagram of Fig. 8.13 allows for a definite modification, in its essence corresponding to correlation processing of the received signal. With such processing the received signal become being sent to the filters should be multiplied by a gate pulse approximately matched with the signal pulse envelope. This can be attained, for instance, if voltage of the FM local oscillator is

, N

modulated in amplitude by a certain pulse v(t) with duration of the order of duration of the sounding pulse.

Filters of the UPCh in this case need not be matched with the width of the spectrum of the pulse envelope; conversely, operation of the discriminator is nearer to optimum, the less the magnitude of $\Delta f_{Y_{11}} \tau_{y_1}$. Gating of output voltages of the UPCh also is not necessary; these output voltages should be detected and subtracted. Changes in the block diagram of Fig. 8.13 reduce in this case to medulating the local oscillator in amplitude, and having no gating stages in general. Such a scheme is simpler in that there is no need to carry out control of the narrow gate pulses fed to gating stages; however, in place of this there is required control of the position of the local oscillator pulse

Thus, in both cases there are required two control circuits — for delay and for frequency — but in the second case requirements on the circuits for control of the reference signal in delay and on accuracy of this control are considerably weaker.

Spainum processing in the given circuit is attained when the form of the pulse envelope of the local oscillator coincides with the form of the sounding pulse envelope and bandwidth of the filters of the UPCh satisfies relationships $1/\tau_{\rm H} \gg$ $\gg \Delta f_{\rm VEH} \gg 1/T_{\rm r}$. If the last condition is satisfied, and the shape of the pulse of the local oscillator v(t) differs from $u_{\rm aO}(t)$, the equivalent spectral density, as before, is given by formula (8.7.10), where in expressions (8.7.3) and (8.7.3) for a(5) and b(5) one should replace pulse response of the filter $h_1(t)$ by a function describing the form of the local oscillator pulse v(t).

In the more general case, when both conditions of optimality are not satisfied, there is dependence of S_{OKB} on the form and parameters of the frequency response of the filter of the UPCh. The expression for equivalent spectral density here can be obtained by normal means and has the form

-382-

$$S_{\theta H_B} = \frac{T_r}{2q^2} \left\{ \int_{-\infty}^{\infty} \int_{-\infty}^{\infty} \{|V(i\omega_1)|^2 | H(i\omega_2)|^2 [|H(i(\omega_1 - \omega_2))|^2 - - |H(i(\omega_1 - \omega_2 - 2a\delta))|^2] + + 2qF(i\omega_2)F(i\omega_3)V^2(i(\omega_1 + \omega_2))|H(i(\omega_1 - a\delta))| \times \\ \times [|H(i(\omega_1 + a\delta))|^2 - |H(i(\omega_2 - a\delta))|^2] d\omega_2 d\omega_2 \right\};$$
$$: \left[\frac{\partial}{\partial \delta} \int_{-\infty}^{\infty} |F(i\omega)|^2 | H(i(\omega - a\delta))|^2 d\omega_2 \right], \qquad (9.7.14)$$

where, just as in § 8.6, $V(i\omega)$ - Fourier transform of function $v^2(t)$, and $F(i\omega)$ - the transform of function $v(t)u_{a0}(t)$.

Let us give final results for a Gaussian form of the pulse envelope, of the pulse envelope of the local oscillator v(t) and of frequency responses of the UPCo. Substituting the corresponding functions in (8.7.14), we obt

$$S_{9NB} = \frac{T_{r}\pi (1+y^{s})(1+y^{2}+x^{s})}{2m_{R}^{2} \cdot i28z^{2}q^{2}} \cdot e^{\frac{4z^{4}}{1+x^{3}+y^{4}}} \times \left\{ \frac{1-\exp\left\{-\frac{4z^{4}}{x^{3}+y^{3}}\right\}}{yVx^{3}+y^{3}} + \frac{8q\exp\left\{-\frac{4z^{4}}{x^{3}+2(1+y^{3})}\right\}}{V[x^{3}+2(1+y^{3})][x^{3}+2y^{2}(1+x^{2}+y^{2})]} \times \left\{ 1-\exp\left\{-\frac{8z^{4}}{[x^{4}+2(1+y^{3})][x^{3}+2y^{3}(1+x^{2}+y^{3})]}\right\} \right\}, \qquad (8.7.15)$$

where $x = 2\Delta f y_{\Pi II} \tau_{II}$; $z = \sqrt{2\omega_{III}^2 \delta^2 / \pi}$; $y = \tau_{II} / \tau_{IIC}$.

As also in other cases, this expression weakly depends on detuning z, only slightly increasing with increase of it. Therefore, the magnitude of z should be selected such as to ensure a maximum discriminator gain factor, which corresponds to values of $z \sim 0.6$ to 0.8. In place of formula (8.7.15) here it is possible with error not exceeding 25-30% to use the expression for S_{OVB} when z = 0, which has the form

$$S_{9K8} = \frac{T_{r} \pi (1 + y^{2}) (1 + y^{2} + x^{2})^{a}}{64m_{m}^{2}q^{2}} \left\{ \frac{1}{y (x^{2} + y^{2})^{3/2}} + \frac{16q (1 + y^{2})^{2}}{(x^{2} + 2(1 + y^{2}))^{3/2} (x^{2} + 2y^{2} (1 + x^{2} + y^{2}))^{3/2}} \right\}$$
(6.7.16.)

and is considerably more convenient for practical calculation.

When $y = 1, z \rightarrow 0, x \rightarrow 0$ the quantity S_{PKB} changes into S_{OTT} . Mismatch of durations of the sounding pulse and of the local oscillator pulse leads to the same consequences as in a circuit with amplitude modulation. Bandwidth of the UPCh for $\Delta f_{yHY} < \Delta f_{corn}$ practically does not affect the magnitude of S_{atB} , and further expansion

-383-



स्त्रीय के कॉलिंग्रिंग कर कि के कि निर्माह के कि मिलकार के कर देखि कोंकिन के न कर ने न कर के का को के कि कि कि क

rte a analdada

i dela i

Fig. 8.16. Dependence of $S_{\text{HED}}/S_{\text{OUT}}$ on q for a range finder with frequency modulation: ---- y = 0.3; ---- y = 1.

of the band leads to increase of S_{BKP} basically due to the noise term. Curves of the dependence of S_{BKP}/S_{GRD} on q for various values of x and y, illustrating the influence of parameters on accuracy of measurement, are shown in Fig. 3.10.

§ 8.8. Phase-Code Intrapulse Manipulation

ł

With incoherent radiation there can also be used phase-code manipulation inside the pulse. Questions of selection of codes and general properties of the autocorrelation function of the phase-code manipulated pulse signal were already discussed in Chapter J, and measurement of distance by the delay of such signals in the coherent case was considered in Chapter VII. Here it remains to add only very little to what has already been said above.

The optimum discriminator in this case can be realized both by correlation processing, and also by a shortening filter [12]. One possible block diagram of ε discriminator with a shortening filter is shown in Fig. 8.17. The received signal after conversion of frequency and transmission through a UPCh, matched in bandwidth with duration of the code interval τ_k is fed to a delay line with taps, corresponding to delay $k\tau_k$ (k = 1, 2, ..., n). Delayed signals are fed to an adder with a plus or minus sign depending upon what value (0 or π) the phase of the signal in the corresponding code interval has. At the output of the adder there will form a pulse which is then processed just as in a discriminator for a normal pulse signal. Such a circuit is completely equivalent to the usual pulse range finder (see § 8.5),

-384-



Fig. 8.17. Block diagram of a dis-criminator with PCM and a shortening filter: 1 - filter matched with τ_k ; 2 - delay line with taps; 3 - adder; 4 - detector; 5 - gating stage; 6 controlled generator of gate pulses; 7 -- subtractor.

working on a square pulse of duration τ_{ν} . Therefore, for calculation of characteristics of the range finder it is possible to use all the findings of Paragraph 8.5.2.

Characteristics of discriminators with correlation processing, for instance, with two detuned channels or with switching of reference signals, also actually were already found above. It is necessary only to use formulas relating to the case of a square pulse everywhere replacing its duration by the duration of the code interval τ_{ir} . Here one should consider that at present in examining possibilities of using phase-code manipulation with a pulse signal they almost exclusively consider a pulse with a square envelope.

This is explained, first, by the simplicity of technical realization of circuits for processing a signal with a square pulse and, second, by the fact that for an assigned peak power of the signal selection of a square envelope ensures maximum pulse energy. At the same time with a square envelope in discriminators of correlation type it is possible without special difficulties to ensure coincidence of the reference and sounding signals. Therefore it is possible to consider only the case of coinciding modulations of the sounding and reference signals. Here in the formulas determining the discriminator characteristics $C_{ik}(x) = C_{OO}(x) = C(x)$, i.e., the cross-correlation functions can be assumed equal to the autocorrelation functions of the sounding signal. This circumstance simplifies calculations.

The above investigation of discriminators of incoherent radar range finders shows that, as for a coherent signal, accuracy of range finding is determined by characteristics of modulation of the sounding signal and the magnitude of the signat to-noise ratio. For a sufficiently large signal-to-noise ratio and other conditions being equal accuracy of range finding is not worse than in the case of a coherent signal. For small signal-to-noise ratios there is obtained a loss, precisely depending on relationship $\Delta f_{c} / f_{r} = \Delta f_{c} T_{r}$. The dependence of accuracy on characteristics of modulation of the sounding signal, on the form of the reference signals and the magnitude of detuning in discriminators using correlation processing also is identical in both cases. Mismatch of characteristics of "shortening" filters with the spectrum of the sounding signal in incoherent discriminators leads to the same

-385-

consequences as noncoincidence of modulations of reference and sounding signals in correlation discriminators.

Discriminator characteristic found above can be directly used for analysis of tracking range finders as a whole with any smoothing circuits.

§ 8.9. Analysis of Accuracy of Incoherent Range Finders

As was shown above, the problem of analysis of the accuracy of tracking meters is solved by a single method regardless of what magnitude is measured and how the discriminator of the meter is constructed. If characteristics of the discriminator are found, it remains only to use general formulas of Chapter VI, substituting in them the corresponding expressions for characteristics of the discriminator. According to this the whole problem of analysis of accuracy of incoherent range finders reduces to simple replacement in formulas of § 7.10, characterizing error of measurement, of the expressions for $K_{\rm II}$ and $S_{\rm SHB}$ found for coherent discriminators by the corresponding expressions for $K_{\rm II}$ and $S_{\rm SHB}$ found for coherent discriminators by the corresponding expressions for $K_{\rm II}$ and $S_{\rm SHB}$ found in the preceding paragraphs for incoherent discriminators. This means that all laws governing accuracy of range finding which were investigated in § 7.10 are completely preserved and do not need repeated discussion. Therefore, we shall limit ourselves in this chapter only to consideration of a series of illustrative examples.

We shall discuss the influence of Automatic Gain Centrol (AGC) on accuracy of yeal maters. In incoherent range finders, in general, there are the same regularities caused by the normalizing action of the A.C system; however, the concrete form of the dependence of discriminator gain on the signal-to-noise ratio is somewhat different. In discriminators with correlation processing of the signal the AGC system chould be closed from the output of the integrating filter. Then the average power of cutput voltage of a receiver with AGC is

$K_1^2 (P_o + 2N_o \Delta j_{\phi})$

and discriminator gain varies as

$$K_{x} = K_{x_{0}} \frac{1}{1 + \frac{2N_{0}\Delta f_{0}}{P_{0}}} = K_{x_{0}} \frac{1}{1 + \frac{\Delta f_{0}T_{r}}{q}}, \qquad (8.9.1)$$

where, as before K_{AO} — the gain factor of the discriminator in the absence of noises. In discriminators with a "shortening" filter the AGC system is closed from the output of the detector, which follows after the "shortening" filter, where to decrease noise components in the control voltage of the system of AGC output voltage of the detector is gated by a pulse travelling after the pulse of the signal. In

-386-

practice gating is frequently carried out directly in the "shortening" filter - in normal pulse receivers, as a rule, the UPCh is gated. Duration of the gate pulse here is several times greater than the duration of the "shortened" pulse. Gating leads to decrease of intensity of noise with respect to τ_c/T_r , and thanks to this in such discriminators

$$K_{R} = K_{R_{0}} \frac{1}{1 + \frac{\Delta f_{y \Pi u^{\tau_{0}}}}{q}},$$
 (8.9.2)

where τ_c - duration of the gating pulse;

Δf y_{ll}y - bandwidth of the "shortening" filter (UPCh without intrapulse modulation).

If in the correlation discriminator the passband of the filter $\Delta f_{\tilde{\Phi}}$ is great, to decrease noise components of the control voltage in this case, too, it is possible to use gating of the output voltage of the filter. Formulas (8.9.1) and (8.9.2), obviously, can be reduced to the form of (7.10.4) where quantity y is equal to $\Delta f_{\tilde{\Phi}} /\Delta f_{c}$ and $\Delta f_{y\Pi\Psi} \tau_{c} /\Delta f_{c}$ T_r, respectively; however, in the given case it is more convenient to consider the dependence of K_A on q. Both formulas (8.9.1) and (8.9.2) can be recorded in a single form

$$K_{\mathbf{X}} = K_{\mathbf{X}}, \frac{q}{q+y_1}, \qquad (8.9.3)$$

where $y_1 = \Delta f_{\tilde{m}} T_r$ or, accordingly, $y_1 = \Delta f_{y_{\Pi II}} \tau_c$.

Thanks to this the dependence of the effective bandwidth of closed-loop tracking systems with incoherent discriminators and smoothing filters with constant parameters on the signal-to-noise ratio q coincides with the dependence of effective bandwidth on signal-to-noise ratio h in coherent range finders (see Paragraph 7.10.2).

Let us consider a range finder with a smoothing circuit in the form of a single integrator with gain factor $K_{\rm M}$. We consider that measurement of distance is carries out by a pulse signal without additional modulation. Let us assume that the discriminator of the range finder has a UPCh, matched with pulse duration $\tau_{\rm M}$, and gates are located end-to-end and have duration $\tau_{\rm HC} = \tau_{\rm M}$. Then, according to the formula (8.5.10), equivalent spectral density is

$$S_{\text{DRB}} = T_r \tau_x^2 \left(\frac{0.09}{q} + \frac{0.21}{q^1} \right). \tag{8.9.4}$$

The dimensional gain factor of the open loop of a tracking meter is

$$K = K_{\rm X}K_{\rm X} = K_{\rm X,0}K_{\rm X}\frac{q}{q+y_1} = K_{\rm 0}\frac{q}{q+y_1}, \qquad (3.9.5)$$

-387-

where K_0 - nominal value of the gain factor in the absence of noises, and y_1 in this case is $\Delta f y_{\Pi} q \tau_0$.

Since the effective passband of the closed-loop tracking system $\Delta f_{p\bar{q}}$ is equal to K/4, fluctuation error of measurement is determined by expression

$$\sigma_{\phi_{\pi}}^{2} = 2\Delta f_{\phi\phi}S_{\phi\pi\pi} = K_{\phi}T_{r}\tau_{\pi}^{2} \left(\frac{2.25 \cdot 10^{-1}}{q+y_{1}} + \frac{5.25 \cdot 10^{-1}}{q(q+y_{1})}\right) = \Delta f_{\phi\phi\phi}T_{r}\tau_{\pi}^{2} \cdot 10^{-1} \left(\frac{9}{q+y_{1}} + \frac{21}{q(q+y_{1})}\right), \qquad (8.9.0)$$

where $\Delta f_{g\bar{q}=0} = K_{C}/4$ - nominal value of the effective passband of the closed-loop tracking system.

The dependence of the relative magnitude of fluctuation error $\sigma_{q_{11}}/\tau_{11}\sqrt{\Delta f} \frac{1}{S\phi}$ on q for various y_1 is shown in Fig. 8.18. With an assigned value of $\Delta f_{S\phi}$ o fluctuation error decreases with increase of y_1 ; this corresponds to the fact that with increase of y_1 decrease of q leads to still more considerable narrowing of the



Fig. 8.12. Fluctuation error of range finding in a system with one integrator.

 y_1 is more essential, the less the magnitude of the signal-to-noise ratio. For large q change of y_1 does not lead to noticeable change of fluctuation error. The magnitude of product $\Delta f \partial \phi = 0^T r$ is usually 10^{-3} to 10^{-1} ; then error of measurement may comprise tenths and hundredths of the pulse duration.

effective bandwidth. The influence of

Dynamic error, corresponding to a nonstatistical approach (see § 7.10), in the considered case in accordance with (7.10.29) is equal to

$$\sigma_{\mathbf{g}} = \frac{\sigma_1}{\Lambda} = \frac{\sigma_1}{K_0} \left(1 + \frac{y_1}{q} \right) = \frac{\sigma_1}{4\Delta f_{1000}} \left(1 + \frac{y_1}{q} \right), \quad (3.9.7)$$

where $a_1 - speed$ of the target.

Dynamic error increases with decrease of the signal-to-noise ratio, where this increase is stronger, the larger y_1 .

Let us consider now the example of a range finder with a smoothing filter in the form of a double integrator with correction, whose transfer function is determined by expression (7.10.11). Let us assume that measurement of distance is produced with the help of a pulse signal with a Gaussian envelope and intrapulse

-388-

frequency modulation with frequency deviation $2\omega_m$. We consider that the discriminator is close to optimum, so that

$$S_{\text{DKB}} = S_{\text{OKE}} = \frac{T_{\text{rR}}(1+q)}{4m_{\text{R}}^2 q^2}.$$
 (8.9.8)

Assuming in the circuit of the discriminator the presence of an AGC system, for the effective bandwidth we obtain the following expression:

$$\Delta f_{ab} = \frac{1 + K_a T_a^2 \frac{q}{q + p_1}}{4T_a}, \qquad (8.9.9)$$

where K_{0} — as before, the gain factor of the open system in the absence of noises $(K_{0} = K_{\pi}K_{II 0}$ has dimensionality of $1/\sec^{2})$.

We saw in Paragraph 7.10.3 that the minimum of errors of measurement which is obtained in an optimum system with double integrator is attained under condition $KT_{K}^{2} = 2\{K = K_{0}[q/(q + y_{1})]\}$. Consider that this condition is satisfied for a certain value of q_{0} , i.e.,

$$K_{\bullet}T_{\mu}^{2}\frac{q_{\bullet}}{q_{\bullet}+\mu_{1}}=2.$$

Then

$$\Delta f_{00} = \frac{1}{4T_{\rm m}} \left[1 + 2 \frac{1 + \frac{g_1}{q_0}}{1 + \frac{g_1}{q}} \right]. \tag{8.9.10}$$

Using formula (8.9.8) and (8.9.10), for variance of fluctuation error we obtain the following expression:

$$\sigma_{\phi,a}^{2} = \frac{T_{\phi,a}}{8\omega_{aa}^{2}T_{a}} \cdot \frac{1+q}{q^{2}} \left[1 + 2\frac{1+\frac{y_{1}}{q_{0}}}{1+\frac{y_{1}}{q}} \right].$$

The dependence of the relative magnitude of error $\sigma_{q_M} / \sqrt{T_r \pi / 8 \omega_m^2 T_K}$ on the signal-to-noise ratio q is shown in Fig. 8.19 for different y_1 and q_0 . Investigative of this dependence shows that fluctuation error of measurement with a fixed magnitude of T_K rather essentially depends on selection of q_0 and y_1 . It turns out that for fixed q_0 error increases with increase of y_1 if $q > q_0$, and decreases if $q < q_0$.

On the whole fluctuation error for any q and y_1 decreases with increase of q_0 ; first this decrease is rather fast (with change of q_0 from 0.1 to 1, $\sigma_{\rm QR}$ decreases for various q and y_1 by a factor of 1.5-3), and then slows, so that increase of q_0 above values $q_0 \sim 10$ no longer practically leads to change of error for values of y_1 considered in the example. More exactly, dependence on q_0 disappears if $y_1 < q_0$.

-389-

Decrease of $a_{0,\Pi}^2$ with increase of q for values of q_0 which are not very small is somewhat slower than analogous decrease of equivalent spectral density. In order to obtain an idea about absolute values of fluctuation error, we consider the example when frequency deviation is $2a_{\rm m} = 2\pi \cdot 10^6$ mad/sec, frequency of repetition of pulses is $f_{\rm r} = 1/T_{\rm r} = 100$ eps, and the time constant of the correcting circuit $T_{\rm H} = 0.1$ then the scale factor $\frac{c}{2} \sqrt{\pi T_{\rm r}/8\omega_{\rm m}^2 T_{\rm R}}$ is equal to 9.4 m, and fluctuation error, cor的现在分词 化二十分 化合合物 化合合物 化合合物

Then the scale factor $\frac{1}{2} \bigvee \pi T_r / \delta \omega_m^{T_R}$ is equal to 9.4 m, and fluctuation error, for responding to Fig. 8.19, will be in the range 170 to 1.7 m.



Fig. 3.19. Fluctuation error of range finding in a system with a double integrator: $y_1 = 1; ---y_1 = 3; ----y_1 = 10.$

Let us consider now the dependence of dynamic error on the signal-to-noise ratio. With a smoothing filter with two integrators error in speed in steady- $e^{-i\omega x}$ operating conditions is absent, and stationary error is

$$\sigma_{\mathbf{x}} = \frac{a_{1}}{K} = \frac{a_{2}(q+y_{1})}{K_{0}q} = a_{2}T_{\mathbf{x}}^{2}\frac{q_{0}(q+y_{1})}{q(q_{0}+y_{1})}.$$
(8.9.11)

In formula (8.9.11) $a_2 = acceleration = 1$ the target, and it is assumed that condition $KT^2_{\mathbf{R}}$ is satisfied when $q = q_0$.

The dependence of the relative magnitude of dynamic error $\sigma_{\rm H} / a_2 T_{\rm H}^2$ on q for various q₀ and y₁ is shown in Fig. 8.20. For all q₀ and y₁ dynamic error decreases with increase of q, where this decrease is rather sharp.



Fig. 8.20. Dynamic error of range finding in a system with a double integrator: $y_1 = 1$; ---- $y_1 = 3$; ---- $y_1 = 10$.

Dependence of $\sigma_{\rm H}$ on q_0 and y_1 has a sharper character than in the case of $c_{\rm H}\pi$. For all y_1 and q quantity $\sigma_{\rm H}$ decreases with decrease of q_0 . The dependence on y_1 is such that for $q < q_0$ quantity $\sigma_{\rm H}$ increases with increase of y_1 , and when $q > q_0$ it decreases with increase of y_1 . Since for large q dynamic error is thus small, for its decrease for small q we must select sufficiently small values of y_1 , the more so since the quantity y_1 comparatively weakly affects the magnitude of fluctuating \sim error.

Selection of small values of q_0 is impermissible from the point of view of quantity σ_{QDT} ; obviously, an acceptable compromise in many cases will be selection of a magnitude of q_0 of the order of 1, and y_1 of the order of 2-5. If, as above, $T_R = 0.1 \text{ sec}$, at $a_2 = 100 \text{ m/sec}^2$ product $a_2 T_R$ is 1 m, and dynamic error in accordance with Fig. 8.20 will not exceed 50 m in the most unfavorable case.

Let us consider one more example, when the measured distance varies as a polynomial of degree n, i.e.,

-391-

$$d(t) = \sum_{k=0}^{n} \mu_{k} t^{k}, \qquad (8.9.12)$$

and in the range finder there is used an optimum smoothing filter, the structure of which is determined by expression (7.10.44), i.e., a filter with pulse response

$$O(t, \tau) = \sum_{i, k=0}^{n} A_{ik}(\tau) t^{i} \tau^{k}, \qquad (3.9.13)$$

where $A_{ik}(\tau)$ - matrix elements;

$$A(\tau) = ||A_{ik}(\tau)|| = [M^{-1} + U(\tau)]^{-1}; \qquad (8.9.14)$$

 $\mathbf{M} = \|\overline{\mu_{i}\mu_{k}}\| - \text{matrix of second moments for the coefficients of the polynomial} \\ (\overline{\mu_{k}} = 0);$

$$U(\mathbf{r}) = \| u_{ik}(\mathbf{r}) \| = \left\| \int_{S_{BRB}}^{t_{i+k}} \frac{t_{i+k}}{t_{i+k}} \right\| = \frac{1}{S_{BRB}(q)} \left\| \frac{t_{i+k+1}}{t_{i+k+1}} \right\|. \tag{8.9.15}$$

For a sufficiently large time of measurement

$$\mathbf{A}(\mathbf{r}) \approx \mathbf{U}_{1}^{-1}(\mathbf{r}), \qquad (3.9.16)$$

where this equality is fulfilled factor, the greater the uncertainty in knowledge of coefficients of the polynomial; the condition of validity of this approximate equality is

$$\sum_{k=0}^{n} M_{ik} \tau^{i+k} = \sum_{i,k=0}^{n} \mu_{ik} \tau^{i+k} \gg \frac{S_{uk}}{\tau}.$$
 (8.9.17)

Then, from (8.9.15) it follows that

$$A_{ik}(\tau) = S_{0KB}(q) \frac{a_{ik}}{\tau^{i+k+1}}, \qquad (3.9.18)$$

where $a_{\pm K}$ - elements of a matrix which is not time-dependent and is the reciprocal of matrix

$$\mathbf{a}^{-1} = \begin{bmatrix} 1 \\ i+k+1 \end{bmatrix}. \tag{8.9.19}$$

As it was shown in Chapter VI, error of measurement is equal to G(t, v); therefore in this case for large t we have the following equality:

$$\sigma_{\text{sizer}}^{2}(l) = \sum_{l, k=0}^{n} A_{ik}(l) t^{l+k} \approx \frac{S_{\text{size}}(q)}{l} \sum_{l, k=0}^{n} a_{ik}. \tag{8.9.20}$$

It is possible to show that for matrix **a**, defined by relationship (8.9.19), the sum of all its elements $\sum_{i,k=0}^{n} a_{ik}$ is equal to $(n + 1)^2$; therefore the final expression for variance of error of range finding has the form

$$\sigma_{\text{BMX}}^{2}(t) = \frac{(n+1)^{2}}{t} S_{\text{BKB}}(q), \qquad (8.9.21)$$

so that $\sigma_{\rm BHX}^2$ with an accuracy of coefficient $(n + 1)^2/t$ coincides with the equivalent spectral density. Error decreases without limit with time, and for any fixed t it decreases more, the higher the degree of the polynomial. For small t expression (8.9.21) gives an overstated value of error.

We shall touch on nonlinear phenomena in incoherent tracking range finders. It is obvious that in the framework of the theory of these phenomena developed in Chapter VI incoherence introduces nothing new. Actually, with the idealizations used in Chapters VI and VII characteristics of breakoff of tracking and variance of error, taking into account nonlinearity of the discriminator, are completely determined by the form of the discrimination characteristic and the magnitude of fluctuation error in a linearized system, where the dependence on the detailed form of the discrimination characteristic is not very essential.

In incoherent range finders, sufficiently close in their structure to optimum, the form of the discrimination characteristic, as before, is determined by the autcorrelation function of the sounding signal, i.e., remains the same as in the case of a coherent range finder. Therefore, from this point of view nothing changes.

Variance of fluctuation error of a linearized system is completely determined by the equivalent spectral density. This means that the whole difference reduces to corresponding replacement of some expressions for S_{ONB} by others. Taking this circumstance into account, for determination of the average time to breakoff of tracking and of variance of fluctuation error of range finding we can use formulas (7.15.3)-(7.15.5), (7.15.7)-(7.15.10).

Most interesting is the question of the critical magnitude of the signal-tonoise ratio, at which still it is possible to ignore nonlinearity of the discriminator.

As it was shown in § 7.15 the critical magnitude of the signal-to-noise ratio is determined by the critical value of parameter μ^2 , equal to the product of variance of fluctuation error of a linearized system σ_{π}^2 and the mean square width of the spectrum of modulation of the sounding signal, i.e., $\mu^2 = b\sigma_{\pi}^2$. The critical value

-393-

of μ varies somewhat, depending upon determination of the limit of applicability of the linearized consideration - by average time to the first breakoff of tracking or by the magnitude of variance of fluctuation error taking into account nonlinearity. However, as shown in § 7.15, this difference leads only to slight change of the critical value of the signal-to-noise ratio. Oriented to the worst case, it is pos sible to consider (see § 7.15) that $\mu_{\rm NP} \approx 0.12$ to 0.15. Then with a discriminator sufficiently close from the point of view of the magnitude of equivalent spectral density to optimum the critical value of the signal-to-noise ratio will be determined from equation

$$2S_{\bullet K \bullet}(q_{K p}) \Delta f_{\bullet \phi} b = \frac{T_r \Delta f_{\bullet \phi}(1 + q_{K p})}{q_{K p}^2} = \mu_{K p}^2 \approx 0.02, \qquad (8.9.27)$$

which coincides with the corresponding equation for $h_{\rm KL}$ with a square spectrum of fluctuations and replacement of $\Delta f_{\rm c}$ by $1/T_{\rm r}$. Solving equation (8.9.22), we obtain

$$q_{\mathtt{x}\mathtt{p}} = 25 [T_{\mathtt{r}} \Delta f_{\bullet \phi} + \sqrt{(T_{\mathtt{r}} \Delta f_{\bullet \phi})^{2} + 0.02T_{\mathtt{r}} \Delta f_{\bullet \phi}}]. \qquad (8.9.23)$$

This dependence is shown in Fig. 3.21. From this figure it follows that linear conditions in a tracking meter are preserved for magnitudes of signal-to-noise ratio q exceeding 0.03-10, depending on inertia of the tracking system and the pulse



Fig. 8.21. Dependence of q_{cr} on $\Delta f_c T_r$.

repetition period.

§ 8.10. Influence of Interferences on Inconerent Range Finders

The influence of interferences on incoherent tracking range finders [69-70]leads to results which in many respects are similar to the case of coherent range finder This especially pertains to active interferences. The character of the influence of passive interferences in this case is somewhat different, and noise immunity of incoherent range finders with respect to it is

lower. We shall consider these questions in greater detail.

8.10.1. Active Noise Interference

As also in the case of coherent radars, the influence of active noise interference, due to its broadband nature as compared to the width of the spectrum of the sounding signal, is equivalent to the influence of natural noises of the receiver. Therefore, the presence of noise interference can be accounted for by introduction of a new signal-to-noise ratio $q_{\Pi\Pi}$, equal to the ratio of the energy of the signal received for a period to the sum of spectral densities of natural noise and interference at the input of the receiver of the radar. This quantity is connected with the signal-to-noise ratio q in the absence of interferences by an expression similar to (7.14.1) and (7.14.3), i.e.,

$$q_{nm} = q \frac{1}{1 + \frac{N_{n}}{N_{0}}} = \frac{q}{1 + \frac{P_{n}G_{nn}}{2N_{0}\Delta f_{n}} \frac{G_{np}\lambda^{2}}{(4\pi)^{4}d_{n}^{2}}},$$
 (2.10.1)

where all designations are the same as in Paragraph 7.14.1.

From comparison with (7.14.3) it follows that ratio $q_{\Pi III} / q$ coincides with ratio $h_{\Pi III} / h$ for a coherent radar; therefore $q_{\Pi III} / q$ in various situations, as before, is given by the curves of Fig. 7.51, and all conclusions made in Paragraph 7.14.1 are preserved.

Thus, the relative influence of active noise interference in both cases is equal, but, in general, accuracy of range finding by an incoherent range finder under the influence of interference is lowered more than in a coherent one. For low intensities of interference this difference may be immaterial; however, if the intersity of interference is great, even for q >> 1 and identical accuracy of coherent and incoherent range finders in the absence and in the presence of interferences quantity $q_{\Pi \, W}$ may be less than 1. Here, along with lowering of accuracy of range finding due to decrease of the signal-to-noise ratio there appears loss of accuracy due to incoherence (see § 8.2).

Thereby, an incoherent range finder, providing identical accuracy with a coherent one in the absence of interferences, will give greater errors of measurement in the presence of interferences, in spite of the fact that in both cases the signal-to-noise ratio is lowered by an identical factor.

8.10.2. Pulse Chaotic Interference

In this case there also is an analogy with a coherent range finder, and the influence of interference, true, with a somewhat worse approximation than during noise interference, can be described as the influence of equivalent white noise. This analogy is evident for incoherent range finders with correlation processing of the signal, when the passband of the integrating filter is small as compared to the width of the spectrum of the pulse envelope of the signal.

-395-

Actually, for pulses of interference which are not very long there is no qualitative difference between coherent and incoherent range finders — in both cases pulses of interference after multiplication by the reference signal are passed through narrowband filters and are expanded. As a result they form a random process equivalent in statistical characteristics to a random process at the output of this filter upon feeding its input white noise with spectral density N_{Π} , which is determined by formulas (7.14.6)-(7.14.9).

Then the influence of interference, as before, can be quantitatively described by introduction of a new signal-to-noise ratio $q_{\Pi \Pi}$ and use of the former formulas. Ratio $q_{\Pi \Pi}$ /q, as with noise interference, coincides with $h_{\Pi \Pi}$ /h and is given by formula (7.14.10) and curves of Fig. 7.51. It is obvious that here the finding of the preceding paragraph about the worse noise immunity of incoherent range finders remains in effect, since quantitative characteristics of pulse chaotic interference are such that $q_{\Pi \Pi}$ in real conditions may be less than one.

In incoherent discriminators with "shortening" filters, in general, there are less grounds for replacing pulse chaotic interference with equivalent white noise, especially if the pulse duration of interference is small as compared to the duration of the pulse envelope. However, even in this case pulse chaotic interference can be replaced by a continuous random process with the same level of spectral density at the maximum, but now of finite spectrum width. Therefore, all results are approximately valid for such discriminators, but the formula for $q_{\Pi,\Pi}$ gives a somewhat understated value, i.e., estimation of noise immunity by $q_{\Pi,\Pi}$ is obtained with a certain safety margin.

8.10.3. Return Interference

The character and results of the influence of return interference on incoherce. range finders completely coincide with the case of coherent range finders. Therefore, here there remain all the same problems which we discussed in Paragraph 7.14.3.

8.10.4. Passive Interference

Well-known is the fact that noise immunity of incoherent radars without special means of protection from passive interferences is very low. This has a clear physical foundation and is illustrated in detail in Chapter V in examining incoherent detection systems. Analogously, the influence of passive interferences on incoherent range finders leads to sharp increase of errors of measurement, and with sufficient intensity of interference, to breakoff of tracking and cessation of the regime of

-396-

tracking. Let us consider as confirmation of this one example of the influence of interference on an incoherent range finder.

In order to be sure that low noise immunity of incoherent range finders to passive interferences has a fundamental character, we consider a range finder with an optimum discriminator. For definitiveness we consider that the discriminator is realized by a "shortening" filter ideally matched with the sounding signal, a squarelaw detector and two gates, narrow as compared to the duration of the "shortened" pulse, with detuning 25, which we now will consider finite, and in final results we pass to the limit $\delta \rightarrow 0$, i.e., we turn to the case of an optimum discriminator. Modulation of the signal we consider arbitrary. For narrow triangular gate pulses the process at the output of the discriminator can be considered a discrete random process with a period of repetition of the signal T_r .

If the value of this process in the j-th period is equal to $v_{,i}(\Delta)$, where Δ is mismatch, it has mean value $\overline{v_{,i}(\Delta)}$ and correlation function

$$R_{\mathtt{R}j\mathtt{k}}(0) = [v_{j}(0) - \overline{v_{j}(0)}] [v_{\mathtt{k}}(0) - \overline{v_{\mathtt{k}}(0)}],$$

the gain factor and the equivalent spectral density of the equivalent continuous process are determined by relationships

$$K_{\mathbf{x}} = \frac{\overline{\partial \sigma_{j}(\Delta)}}{\partial \Delta} \Big|_{\Delta = 0}, \qquad (8.10.2)$$

$$S_{0K0} = T_{\tau} \sum_{j=k}^{K_{A}j=k} \frac{K_{A}j=k}{K_{A}^{2}}.$$
 (3.10.3)

Output voltage of the discriminator $v_j(\Delta)$ can be recorded through the square of the envelope of output voltage of the "shortening" filter in the j-th period $z_j(\Delta)$ in the following way:

$$D_{j}(\Delta) = Z_{j}(\tau_{\bullet} + \delta - \Delta) - Z_{j}(\tau_{\bullet} - \delta - \Delta). \qquad (8.10.4)$$

Here τ_0 - true value of delay; τ_0 - $\Delta = \tau$ - its measured value,

$$z_{j}(t) = \left| \int_{-\infty}^{\infty} u_{0}(s-t) y_{j}(s) e^{t - s t} ds \right|^{2}$$
(8.10.5)

- the square of the envelope of output voltage of the "shortening" filter with pulse response $h(t) = u_0(-t)$ ($u_0(t)$ describes modulation in one period); $y_j(t)$ - the signal received in the j-th period.

Then

$$K_{A} = 2 \frac{\overline{\partial z_{j}(s_{0}+\vartheta)}}{\partial \vartheta}, \qquad (3.10.6)$$

$$R_{R,jk} = 2R_{sjk}(\tau_0 + \delta, \tau_0 + \delta) - 2R_{sjk}(\tau_0 + \delta, \tau_0 - \delta), \qquad (8, 10, 7)$$

where $R_{zjk}(t_1t_2) = [\overline{z_j(t_1) - \overline{z_j(t_1)}}][z_k(t_2) - \overline{z_k(t_2)}]$ - correlation function of values of the square of the envelope in the j-th and k-th periods.

Let us find the mean value of the square of the envelope. It is obvious that

$$\overline{z_j(t)} = \int_{-\infty}^{\infty} \int_{-\infty}^{\infty} u_0(s_1 - t) u^{s_0}(s_2 - t) e^{i \omega_0(s_1 - s_1)} R_{yjk}(s_1, s_2) ds_1 ds_2, \qquad (8.10.3)$$

where $R_{yjk}(t_1, t_2)$ - correlation function of the received signal, consisting of a mixture of the signal reflected from the target, internal noise and the signal from passive interference. As it was shown in Chapter I, this correlation function is equal to

$$\begin{split} R_{\text{Wjk}}(l_1, l_2) &= P_c \, \text{Re}\, u_0(l_1 - \tau_0) \, u_0^* (l_2 - \tau_0) \, \text{P}_{jk} e^{lu_0(l_1 - l_1)} + \\ &+ N_0 \delta(l_1 - l_2) \, \delta_{jk} + \text{Re} \int_{-\infty}^{\infty} \sigma(\tau) \, u_0(l_1 - \tau) \, u_0^* (l_2 - \tau) \, \times \, d\tau r_{jk} e^{lu_0(l_1 - l_1)} e^{l_0} (l_1 - k), \quad (8.10.9) \\ &\quad \rho_{jk} - \text{coefficient of interperiod correlation of signal;} \\ &\quad r_{jk} - \text{coefficient of interperiod correlation of interperiod correlation of interperiod;} \\ &\quad \theta_{jk} = \Delta \omega_{jk} T_r - [\Delta \omega_{jk} T_r/2\pi] 2\pi - \text{phase shift for a period, caused by difference of Doppler frequencies of the signal and interference;} \\ &\quad [\Delta \omega_{jk} T_r/2\pi] - \text{integral part of ratio } \Delta \omega_{jk} T_r/2\pi; \end{split}$$

 $\sqrt{2\pi}$] - integral part of ratio $\Delta \omega_{\mu} T_{r}^{2\pi}$; $\Delta \omega_{\mu}$ - difference of Doppler frequencies of signal and interference.

Phase shift $\theta_{\rm II}$ is conveniently presented in the form

where

$\Phi_{\rm A} == \Delta \omega'_{\rm A} T_{\rm r},$

where $\Delta n_{\Pi}^{\dagger} = \Delta \omega_{\Pi} - [\Delta n_{\Pi} T_{\Gamma}/2\pi] 2\pi/T_{\Gamma}$ -- difference of the difference Doppler frequency and the nearest frequency to it, which is a multiple of the frequency of repetition of the signal.

Function $\sigma(\tau)$ in formula (8.10.9) is the distribution density of power with respect to range reflected from passive interference, expressed, for instance, in $w/\mu sec$. This function is connected with the distribution density of range reflectors by the simple relationship

$$\Theta(\tau) = P_{0} \frac{\sigma_{0}}{\sigma_{\pi}} \frac{c}{2} n \left(\frac{c\tau}{2}\right), \qquad (8.10.10)$$

where
$$\sigma_0$$
 - effective reflecting surface of one reflector;

 $\sigma_{\rm H}$ - effective reflecting surface of target;

n(r) - distribution density of range reflectors (number of reflectors in a layer of unit thickness).

-398-

Substituting expression (8.10.9) in formula (8.10.8) and integrating, we obtain

$$\overline{z_{j}(l)} = \frac{T_{r}^{2}}{2} \left[P_{c} |C(\tau_{o}-l)|^{2} + \int_{-\infty}^{\infty} \sigma(\tau) |C(\tau-l)|^{2} d\tau + \frac{2N_{o}}{T_{r}} \right]. \quad (8.10.11)$$

where C(x) - autocorrelation function of one period of modulation.

If density of reflectors $\sigma(\tau)$ is constant within limits of the interval of range resolution, the component caused by the presence of passive interference is

$$\int_{-\infty}^{\infty} \sigma(\tau) |C(\tau-t)|^2 d\tau \approx \sigma(t) \int_{-\infty}^{\infty} |C(\tau-t)|^2 d\tau \approx \sigma(\tau_0) \int_{-\infty}^{\infty} |C(x)|^2 dx, \quad (8.10.12)$$

i.e., does not depend on t. Therefore, the presence of passive interference, as also the presence of internal noises, does not affect the gain factor of the discriminator, and magnitude of K_{Π} remains the same as before (§ 8.2).

Note that in a nonoptimal discriminator, having in the absence of interference null shift of the discrimination characteristic due to noises, in the presence of passive interference due to imbalance of channels there will appear additional systematic error, the magnitude of which may be very significant due to the high level of interference.

Likewise, calculating the correlation function, as we have done many times for the case when there are only a reflected signal and noise, we can obtain the following expression:

$$R_{AJA}(t_{1}, t_{2}) = \frac{\dot{P}_{c}^{2}}{4} T_{r}^{1} \rho_{JA}^{2} |C(\tau_{0} - t_{1})C(\tau_{0} - t_{2})|^{2} + + N_{0}^{2} T_{r}^{3} |C(t_{1} - t_{2})|^{2} \delta_{JA} + \operatorname{Re} \left\{ N_{0} T_{r}^{3} P_{0} C(t_{1} - t_{2}) C^{*}(\tau_{0} - t_{1}) C(\tau_{0} - t_{2}) + + N_{0} T_{r}^{3} \circ (\tau_{0}) \int_{-\infty}^{\infty} C^{*}(x) C(x + t_{1} - t_{2}) dx \right\} \delta_{JA} + \frac{T_{r}^{4}}{4} r_{JA}^{2} \delta^{2}(\tau_{0}) \left| \int_{-\infty}^{\infty} C(x - t_{1}) C^{*}(x - t_{2}) dx \right|^{2} + + \operatorname{Re} \left\{ \frac{P_{0}}{2} T_{r}^{1} \rho_{JA} r_{JA}^{0} e^{t \theta_{A}(t - h)} \sigma(\tau_{0}) C(\tau_{0} - t_{1}) \times C^{*}(\tau_{0} - t_{2}) \int_{-\infty}^{\infty} C^{*}(x) C(x + t_{1} - t_{2}) dx \right\}, (8.10.13)$$

where we again assumed that the density of reflectors within limits of the range resolution interval near point τ_0 is constant and equal to $v(\tau_0)$, i.e., its value at the point of location of the target. This assumption sufficiently well is realized in practice already for comparatively low range resolution capability of the radar.

Substituting the found expression for the correlation function in (8.10.7), summing in accordance with formula (8.10.3) and passing in the obtained expression to limit $\delta \rightarrow 0$, for equivalent spectral density we find the following expression:

$$S_{BKB} = \frac{T_r}{2bq^1} \left\{ 1 + q + \frac{e(\tau_1) T_r}{2N_e} \int_{-\infty}^{\infty} \left[|C(x)|^2 + \frac{1}{b} |C'(x)|^2 \right] dx + \frac{1}{b} |C'(x)|^2 \right] dx + \frac{1}{b} |C'(x)|^2 + \frac{1}{b} |C'($$

$$+\frac{e^{2}(\tau_{0})T_{r}^{2}}{4N_{0}^{2}b}\int_{-\infty}^{\infty}|C(x)|^{2}dx\int_{-\infty}^{\infty}|C'(x)|^{2}dx+\frac{qe(\tau_{0})T_{r}}{2N_{0}b}\int_{-\infty}^{\infty}|C'(x)|^{2}dxf\left(\Delta f_{\pi}T_{r}\frac{\Delta \omega'_{\pi}}{\Delta f_{\pi}}\right)\right\}, (8.10.14)$$

where it is assumed that the width of spectral band of passive interference Δf_{μ} is great as compared to the width of the spectrum of fluctuations of the signal Δr_{c} , and function $f(\Delta f_{\Pi} - T_{r}, \Delta u_{\Pi}^{\prime}/\Delta f_{\Pi})$ is equal to

$$f\left(\Delta f_{\mathfrak{a}}T_{r}, \frac{\Delta \omega_{A}}{\Delta f_{\mathfrak{a}}}\right) = \sum_{j=k} r_{jk} e^{i\theta_{A}(j-k)}.$$
(8.10.15)

For great width of the spectrum of interference as compared to the frequency for repetition, when $r_{jk} = \delta_{jk}$, this function is equal to one, and in the opposite case, when $\Delta f_{II} = T_r \ll 1$, summation in formula (0.10.15) can be replaced by integration, and then

$$f\left(\Delta f_{\mathbf{n}}T_{\mathbf{r}}, \frac{\Delta \omega_{\mathbf{A}}}{\Delta f_{\mathbf{n}}}\right) = \frac{1}{\Delta f_{\mathbf{n}}T_{\mathbf{r}}} S_{\mathbf{n}}\left(\Delta \omega_{\mathbf{A}}'\right), \qquad (3.10.1c^{2})$$

where $S_{n}(\omega)$ - spectral density of interference, normalized so that $S_{11}(0) = 1$.

In the particular case of exponential correlation of interference, when $r_{jk} = e^{-2\Delta f} = \frac{T_r |i-k|}{r_r}$, the exact expression for $f(\Delta f_{\Pi} T_r, \Delta \omega_{\Pi}/\Delta f_{\Pi})$ has form

$$f\left(\Delta f_{n}T_{r}, \frac{\Delta \omega_{A}}{\Delta f_{n}}\right) = 1 + \frac{1}{\Delta f_{n}T_{r}\left[1 + \left(\frac{\Delta \omega_{A}}{2\Delta f_{n}}\right)^{2}\right]}.$$
(8.10.17)

Considering in greater detail the expression for $S_{\partial KB'}$ we see that the first 'the terms are caused by the presence of internal noises and remain in the absence of interference, characterizing accuracy of measurement with internal noises alone.

The last term is caused by the interaction of the signal with interference and depends, at least for a frequency of repetition high as compared to Δf_{Π} , on the difference of Doppler frequencies of the signal and interference. This term coincides with the corresponding component of S_{BNF} caused by the presence of passive interference; in coherent systems and with selection of a high frequency of repetition it is determined only by the magnitude of the difference Doppler frequency. Therefore, for all "nonblind" speeds (see Paragraph 7.14.4) this term does not lead to

-400-

increased errors of measurement.

The third term in formula (8.10.14) is caused by interaction of interference with noise and has an order of 1/q (let us remember that in accordance with (8.10.10) $\sigma(\tau_0)$ has an order of P_c). This term has a specific incoherent form - it does not depend on $\Delta f_{\Pi} T_r$ and $\Delta \omega_{\Pi}^{\dagger}$. In coherent systems, thanks to the presence of narrowband filtration of noise at the signal frequency, this term also depends on the aifference Doppler frequency and for sufficiently great detuning turns out to be small as compared to q.

An analogous state of affairs occurs with respect to the fourth term, which is caused by the interaction of interference with interference. This term gives the main component of spectral density - it does not depend on the signal-to-noise ratio q. In coherent systems, thanks to the fact that to the discriminator output there pass only lateral components of the spectrum of interference, this term also depends on detuning.

Integrals in formula (8.10.14) have the order of magnitude of the effective range resolution interval $\Delta \tau_{ac} \sim 1\sqrt{b}$, i.e.,

$$\int_{-\infty}^{\infty} |C(x)|^{s} dx = \Delta \tau_{s\phi} \approx \frac{1}{\sqrt{b}}, \qquad (8.10.18)$$

$$\frac{1}{b}\int_{-\infty}^{\infty} |C'(x)|^{b} dx \approx \Delta \mathfrak{r}_{b\phi}. \tag{8.10.19}$$

Then

$$\int_{-\infty}^{\infty} |C(x)|^{s} dx \cdot \sigma(\tau_{0}) = P_{0} \frac{\sigma_{0}}{\sigma_{R}} \frac{c \lambda \tau_{00}}{2} n\left(\frac{c \tau_{0}}{2}\right) = P_{0} \frac{\sigma_{n}}{\sigma_{R}}, \qquad (8.10.20)$$

where $\sigma_{\Pi} = \sigma_0 \frac{c\Delta \tau}{2} \frac{\Delta \phi}{n} \left(\frac{c\tau_0}{2}\right) - \text{total reflecting surface of interference in the resolving volume of the radar.}$

With an accuracy of a numerical coefficient of the order of unity, equal to the quantity, too, is the product

$$s(\tau_0) - \frac{1}{b} \int_{-\infty}^{\infty} |C'(x)|^2 dx \simeq P_0 \frac{\epsilon_n}{\epsilon_n}. \qquad (8.10.21)$$

More exactly, integrals (8.10.18) and (8.10.19), for instance, for the Gaussian autocorrelation function

$$C(x) = \exp\left(-\frac{bx^2}{2}\right)$$

are equal to

-401-

$$\Delta \tau_{00} = \int_{-\infty}^{\infty} |C(x)|^{2} dx = \sqrt{\frac{\pi}{b}}, \qquad (8.10.22)$$

$$\frac{1}{b} \int_{-\infty}^{\infty} |C'(x)|^{2} dx = \frac{1}{2} \sqrt{\frac{\pi}{b}} = \frac{1}{2} \Delta \tau_{00}. \qquad (5.10.23)$$

Thus, taking into account (8.10.20) and (3.10.21), we can give equivalent spectral density the form

$$S_{\alpha \kappa \alpha} = \frac{T_r}{2b} \left\{ \frac{1}{q^3} + \frac{1}{q} + \frac{1}{q} \frac{\sigma_{\alpha}}{\sigma_{\alpha}} (1 + \alpha) + \left(\frac{\sigma_{\alpha}}{\sigma_{\alpha}} \right)^2 \alpha + \alpha \frac{\sigma_{\alpha}}{\sigma_{\alpha}} \int \left(\Delta f_{\alpha} T_r \frac{\Delta \omega'_{\alpha}}{\Delta f_{\alpha}} \right) \right\}, \quad (8.10.24)$$

where α - numerical coefficient of the order of unity (for Gaussian C(x) it is equal to 1/2).

Consequently, even for a large signal-to-noise ratio q, a narrow spectrum of interference, and great detuning in frequency $(\Delta \omega_{\rm H}^{'}/\Delta f_{\rm H}^{'} \gg 1)$, the equivalent spectral density may be very great. Its limiting value is determined by $(\sigma_{\rm H}^{'}/\sigma_{\rm H}^{'})^2$. If it is considered that in the resolved volume there are reflectors with total reflecting surface $\sigma_{\rm H}^{'} \approx 50 \ {\rm m}^2$, then even for the largest air targets with a reflecting surface of the order of $\sigma_{\rm H} = 10$ to $20 \ {\rm m}^2$ [26] the spectral density of $S_{\rm HKB}$ is (C to 13)T_r/t, i.e., is the same as in the absence of interferences and for a signal-to-noise ratio q ~ 0.4 to 0.2. Here, as follows from results in § 8.9, in most cases there occurs disturbance of linear conditions in the tracking system, errors of measurement essentially increase, and breakoff of tracking becomes possible. In many cases ratio $\sigma_{\rm H}^{'}/\sigma_{\rm H}$ can reach considerably larger values. Therefore, the influence of passive interference on an incoherent range finder leads, actually, to the impossibility of range finding.

The only specifically incoherent means of combatting passive interferences knew at present is incoherent alternating-period compensation [1, 34]. It consists of the following: "shortened" pulses after detection are delayed a period and are subtracted from the undelayed sequence (see Chapter IV). Then, the difference signal is fed to gating stages for formation of the signal of range mismatch.

If the width of the spectrum of interference is sufficiently small as compared to the frequency of repetition, the components of output voltages, caused by the presence of interference, due to their correlatedness are partially compensated, and the signal-to-interference ratio is improved. With incoherent compensation there occurs, basically, decrease of the basic component of equivalent spectral density, which is caused by interaction of interference with itself (the term in

-402-

formula (8.10.24) which is proportional to $(\sigma_{\Pi} / \sigma_{\Pi})^2$). The magnitude of suppression essentially depends on the degree of interperiod correlation of interference and the form of its spectral density. The relative magnitude of suppression, i.e., the degree of increase of the signal-to-interference ratio, depends also on the speed of the target.

On the basis of estimates given in Chapter IV one may assume that suppression of interference has an order of $1/\Delta f_{\Pi} T_r$. For normally used frequencies of repetition and width of the spectrum of interference of several tens of cycles per second, which is characteristic of fixed, e.g., ground radars, this suppression can attain 10-20 db. In the case of moving airborne radars, due to great width of the spectrum of passive interference suppression turns out to be essentially smaller or may, in general, be absent.

Thus, incoherent compensation of passive interference is not a radical means of protecting radars and in many cases cannot ensure reliable range finding. Therefore, the only truly effective means of protecting pulse range finders from passive interferences is application of coherent techniques.

Most frequently this is attained by introduction of coherent alternating-period compensation (Chapter IV), which for sufficiently high multiplicity ensures approximately the same results as the coherent range finders considered in the preceding chapter. Theoretically, it turns out that all terms in formula (8.10.24) depend on the difference Doppler frequency and for sufficiently great detuning are small. In practice, due to the presence of blind speeds and nonoptimality of processing, efficiency of a system with alternating-period subtraction is lower than efficiency of a radar with a coherent range finder (see Chapter IV); however, in many important cases suppression of interference is sufficient to guarantee normal efficiency of the range finder.

§ 8.11. Conclusion

The conducted investigation of incoherent tracking radar range finders and comparison with results pertaining to coherent meters shows that from many points of view differences in the form of sounding radiation of the radar and in methods of processing the received signal do not lead to any essential difference in characteristics of accuracy of measurement.

In particular, with external interferences and a sufficiently large signal-tonoise ratio potential accuracy of range finding is identical for coherent and incoherent signals. Just as in coherent systems, different approximate methods of

-403-

realization of optimum operations considered in this chapter ensure accuracy close to the potential, if only we reasonably select parameters of the corresponding circuits. The dependence of accuracy of measurement on the form of modulation in incoherent discriminators is the same as in coherent. The influence of various kirds of nonidealnesses of processing also, basically, remains the same.

Essential difference in the characteristics of accuracy of coherent and incoherent radar range finders can take place with external interferences, especially passive ones. In this respect incoherent systems give considerable loss, and in a number of cases the presence of passive interferences leads to impossibility of their application. Another peculiarity of the investigated incoherent range finders is worse accuracy for low signal levels and, connected with this, the necessity of allowing for nonlinearity of discriminators and phenomena of breakoff of tracking.

Regarding questions requiring further investigation and more detailed working out, here it is possible to repeat everything said in § 7.16 with respect to problems concerning coherent range finders. Actually, problems enumerated there have noless urgency and interest in this case. This list can be supplemented by a seried of specifically "incoherent" problems. From the practical point of view one of the most interesting such problems is more detailed study of interference resistance of incoherent range finders with respect to passive interferences. We consider correct the investigation of the system of incoherent alternating-period compensation, conseing and studying other possible means of protection from passive interferences, a combined system using for protection from possive interferences coherent compensation with external and internal coherence, etc.

Another interesting question is more precise definition of solution of the problem of synthesis of an optimum incoherent discriminator. Here we should consider the presence of interperiod correlation of the received signal, find operation by which there is replaced accumulation of squares of the envelope, and analyze their influence on accuracy of measurement. Although physical considerations and consideration of the limiting cases force us to think that the influence of interperiod correlation is hardly essential, nonetheless it would be interesting to obtain strict analytic confirmation of this circumstance.

One more important question is investigation of nontracking incoherent range finders. Practically, most interesting are meters using a unit of detection channels. This problem obtains special urgency in connection with application of incoherent signals with high range resolution capability, provided by intrapulse

-404-

modulation. In the given question one should seek optimum operations of processing the signal from the output of the unit of channels, and study characteristics of accuracy for these and other possible operations.

In incoherent range finders we can use quantization of the output signal by level. (This quantization may be, in principle, applied in coherent range finders, too.) Investigation of systems with quantization is also interesting for practice. Here the most important question is the question of the influence of the magnitude of quantization on accuracy of measurement and of finding conditions when quantization can be disregarded.

However, in spite of the fact that there are a number of unsolved questions, results of the present chapter, in general, permit us with sufficient knowledge of the matter to approach the problem of construction of incoherent radar range finders with arbitrary intrapulse modulation and of investigation of their accuracy in various conditions. These results give the possibility of correctly selecting the functional circuit of the discriminator of a range finder, to estimate the influence of various kinds of deviations from optimality of processing, to calculate characteristics of discriminators taking into account these deviations, and t.: analyze accuracy of range finding with different laws of motion of the target and different us to correctly select the required magnitudes of width of the spectrum of modulation and of the energy of the received signal.

-405-

CHAPTER IX

MEASUREMENT OF SPEED

§ 9.1. Introductory Remarks

Requirements on contemporary radars frequently lead to the necessity of the measurement not only of current coordinates of targets, but also rates of their change. In certain radar systems it is desirable to measure both radial, and also tangential components of velocity and to find as a final result the velocity vector of the target with respect to the radar. Moreover, there exist technical problems leading to the necessity of measurement of the second derivative of coordinates of targets for discerning the latter and for exact prolongation of their trajectories.

However, of greatest interest is measurement of the radial component of velocity. It is needed, for instance, for navigational systems, where there are used Doppler groundspeed meters (dead reckoners). Measurement of radial velocity in the variant of automatic tracking of frequency of reflected signal is necessary during construction of any coherent radars, since they contain narrow-band filters (see Chapters IV and VII), which it is necessary to tune upon change of frequency of the signal.

Thus, from the principle of construction of coherent radars there ensues the necessity of measurement of Doppler shift of frequency of the signal

$$f_{\rm A}=\frac{2V}{\lambda}$$

where V - radial velocity;

 λ - wavelength of the generated oscillations.

Namely for this of greatest interest is investigation of Doppler speed meters, which is the basic content of this chapter. Besides direct measurement of radial velocity using the Poppler effect there are applied other methods of finding it.

-407-

Preceding page blank

based on use of range-finding systems, since voltages at certain points of filters of range finders are proportional to radial velocity. Such a method is essentially the only one for incoherent pulse radars with short pulse duration, since attempts to measure speed by Doppler shift of frequency, determined during a pulse duration. usually do not stand up to criticism in connection with low accuracy of measurement. Measurement of tangent components of velocity can be performed using goniometrical systems of radars. Questions of measurement of speed with the help of range finders and goniometers also will be seen in this chapter. As also in preceding chapters, we will be interested first of all in finding optimum methods of construction of speed meters (their discriminators and smoothing circuits), allowing us to obtain minimum errors of measurement in the presence of noises and with allowance for fluctuations of the signal reflected from the target. Quantitative estimation of these errors, just as of errors of measurement of speed with various applied methods of construction of discriminators and smoothing circuits, is the next problem of this chapter. Furthermore, we analyze nonlinear phenomena in speed meters, taking place with intense noises and leading to breakoff of tracking of automatic tracking meters, and also questions of the influence on them of certain forms of active and passive interferences.

As also during investigation of range systems, we will widely use results of Chapter VI in accordance with which during synthesis of optimum speed meters which are tracking systems of definite form, we shall be basically interested in synthesis of the optimum discriminator. Smoothing circuits are synthesized for any meters in Chapter VI. Correspondingly, being interested in Doppler speed meters, we shall first determine operations of an optimum frequency discriminator, judge different technical methods of construction of the discriminator, giving results close to optimum, and make certain recommendations ensuing from comparison of circuits of discriminators. Smoothing circuits will be considered from the point of view of application of results of Chapter VI to measurement of speed giving physical treatments and examples, illustrating accuracy of the obtained meters as a whole. Special attention will be paid to the question of deviations from the optimum method of construction of meters and their influence on accuracy of measurement of speed.

Estimates of the influence of strong noises and interferences on Doppler tracking speed meters are given on the basis of a consideration of the phenometon of breakoff of tracking, which is conducted from tenets of the two criteria of breakoff considered in Chapter VI.

-408-

In a number of cases they apply nontracking speed meters, in particular, meters containing a unit of filters tuned to various Doppler frequenci(. Analysis of potentialities of such meters is also given in this chapter.

Investigating properties of meters of speed, determined by differentiation of a coordinate (range, angle), we shall find optimum circuits of meters for certain particular forms of statistics of change of speed, and also we shall analyze accuracy of measurement during application of usual forms of linear filters with constant parameters.

All these investigations give us the possibility to estimate performance of speed meters not only in the presence of internal noises of systems, but also under the influence of certain wide-spread forms of interferences, whose effect. As will be shown, is equivalent to the effect of noise.

§ 9.2. Synthesis of an Optimum Frequency Discriminator

We turn first of all to measurement of radial velocity on the basis of the Doppler effect. The measured parameter of the received radar signal here is Doppler shift of frequency $\omega_{\rm H} = \frac{4\pi V}{\lambda}$ which varies, in general, randomly in time. The function of the discriminator of a tracking radar meter is singling out voltage proportional to current mismatch between the true magnitude of $\omega_{\rm H}(t)$ and its measured value, which is the output variable of the tracking system. Here, as follows from Chapter VI, an optimum discriminator, ensuring minimum spectral density of the random component of voltage at its output, should form the derivative $\partial |Q(t, \omega_{\rm H})|^2 / \partial \omega_{\rm H}$ for $\omega_{\rm H}$, equal to the estimated value (output variable of the tracking meter). Here $Q(t, \omega_{\rm H})$ is determined by the presentation of the logarithm of the functional of the probability density of the received signal y(t) for an accidence value of parameter $\omega_{\rm H}$ in the form

$$\int_{0}^{T} |Q(t, \omega_{\rm H})|^{2} dt + C,$$

where C does not depend on ω_{π} .

To find the functional of probability density $P[y(t)/\omega_{II}(t)]$ necessary for further calculations it is necessary to assign the form of signal y(t). In accordance with the description of the signal, given in Chapter I, the reflected signal can often be considered a normal random process. We shall be interested in its reception in white Gaussian noise of spectral density Π_0 . This corresponde to cases of the presence of internal noises of the receiver, of broad-band active interferences, and, with certain assumptions, stated in § 9.9, passive interferences.

-409-

Then the received signal (mixture of reflected signal and interference) $y(t_i)$ is a normal random process whose correlation function, according to (1.4.3), is determined by expression

$$R(t_{1}, t_{g}) = P_{c}u_{A}(t_{1} - \tau)u_{A}(t_{2} - \tau)\rho(t_{1} - t_{g}) \times \\ \times \cos \left[\mathbf{e}_{0}(t_{1} - t_{g}) + \mathbf{e}_{A}(t_{1})(t_{1} - t_{g}) + \psi(t_{1} - \tau) - \psi(t_{a} - \tau) \right] + \\ + N_{g} \mathcal{R}(t_{1} - t_{g}), \qquad (9.2.1)$$

where, as before, $u_a(t)$ and $\psi(t)$ - laws of amplitude and phase modulation of the signal, respectively; τ - delay of the reflected signal; P_0 - its mean power; f(t) - correlation function of fluctuations of the signal; $\varpi_{\mathbf{n}}(t)$ - a function, representing Doppler shift of the frequency of the signal, slow as compared to $\rho(t)$.

Calculation of the functional of probability density of a normal random signal with correlation function of form (9.2.1) already is given in Chapter IV during synthesis of an optimum detector of a coherent signal. According to (4.3.7) the logarithm of the likelihood function (functional of probability density of signal y(t)) is presented in the form just now mentioned:

$$L(\mathbf{w}_{R}) = C + \frac{1}{N_{\bullet}} \int_{0}^{1} |Q(t, \mathbf{w}_{R})|^{\bullet} dt, \qquad (9.2.2)$$

*Strictly speaking, with a random signal and radial velocity of the target varying in time one should talk about random changes of phase of the signal, connected with radial velocity. These changes can be presented in the form

$$\Psi(l) = \int_{l_0}^{l} \omega_{\lambda}(s) \, ds.$$

in the expression for the correlation function of the signal there is the difference ϵ

$$\Delta \varphi \simeq \varphi(f_1) - \varphi(f_2) = \int_{f_1}^{f_2} \omega_{\lambda}(s) ds.$$

In connection with the essumption of the quickness of change of components of the random signal as compared to changes of speed of the target for intervals $t_1 - t_2$, corresponding to the interval of correlation of the shown components, this approximate equality is valid:

$$\Delta \varphi = \omega_A (l_1) (l_1 - l_2).$$

It directly leads to formula (9.2.1) for the correlation function and to the aboveindicated interpretation of the function of the discriminator. which gives the possibility of direct interpretation of operations of an optimum discriminator. Here $Q(t, \omega_{II})$ is the result of passage of the product of the observed realization of signal y(t) and a reference signal, which in complex form is expressed as

$$\mathcal{U}_{a}\left(t-\tau\right)e^{i\left(u_{0}+u_{A}\right)i+\frac{1}{2}\left(t-\tau\right)},$$

through a filter, the square of the modulus of whose frequency response, according to (4.3.8), is equal to

$$|H_{\bullet}(i\omega)|^{*} = \frac{hS_{\bullet}(\omega)}{1+hS_{\bullet}(\omega)}.$$
(9.2.3)

Quantity h is the signal-to-noise ratio and is defined as

$$h = \frac{P_c}{2N_{\rm e}\Delta/c},$$

 $\Delta f_c = -$ effective band of fluctuations of the signal; $S_0(\omega) = -$ normalized spectral density of fluctuations.

Correspondingly,

$$|Q(t, \bullet_{\pi})|^{2} = |\int_{-\infty}^{t} y(s) u_{k}(s-\tau) e^{t[(\bullet_{0}+\bullet_{k})s+\psi(s-\tau)]} h_{0}(t-s) ds|^{2}, \qquad (9.2.4)$$

where $h_0(t)$ — pulse response of a filter, whose frequency response is $H_0(i\omega)$. This filter is a low-frequency equivalent of the optimum filter of a coherent receiver, encountered already in problems of detection and range finding. The obtained result is valid in the case of rapid fluctuations of the signal.

Output voltage of the optimum frequency discriminator is defined as

$$z(t) = \frac{1}{N_{\theta}} \frac{\partial |Q(t, \omega_{R})|^{2}}{\partial \omega_{R}} \Big|_{\omega_{R} = \omega_{R}} = \frac{1}{N_{\theta}} \frac{\partial}{\partial \omega} \left\{ \left| \int_{-\infty}^{t} h_{\theta}(t-s) \times \left(9.2.5 \right) \right. \right\} \right\}$$

$$\times e^{t\omega_{R} p(t-s)} y(s) u_{\theta}(s-\tau) e^{t[\omega_{R} s + \phi(s-\tau)]} ds \Big|^{2} \right\}, \qquad (9.2.5)$$

where $\omega_{\Pi_{1}}$ - fixed frequency (h₀(t) cos $\omega_{\Pi_{1}}$ t - pulse response of the optimum filter);

 $\mathbf{e}_{\mathbf{f}} = \mathbf{w}_{\mathbf{0}} + \hat{\mathbf{e}}_{\mathbf{g}} + \mathbf{w}_{\mathbf{g}_{\mathbf{f}}};$

 $\hat{\omega}_{\mathrm{I}}(\mathrm{t})$ - estimated value of Doppler frequency, obtained from the output of the meter.

Recording of output voltage of the discriminator z(t) in the form of (9.2.5) permits us to easily give the following physical interpretation of operations of the optimum discriminator. The expression in braces is the square of the envelope of voltage, formed by mixing the received signal y(t) with reference signal $u_{a}(t - \tau) \cos [\omega_{p} t + \psi (t - \tau)]$ and passing this signal through an optimum filter, tuned to intermediate frequency $\omega_{H\Gamma}$. Here, delay of modulation of the reference signal τ should correspond to the true distance to the target; it is determined practically by output voltage of the range system. The frequency of the hear prodyne oscillator should be tuned to $\omega_{0} + \hat{\omega}_{\mu}(t) + \omega_{\mu\mu}$, which is attained with the help of the voltage from the output of the synthesized speed meter. Mixing of the recieved signal with the shown reference signal ensures convolution of phase modulation, as a result of which there is obtained a signal unmodulated in phase and the best separation of the amplitude-modulated signal from noises.

Naturally, convolution of phase modulation during mixing with heterodyne voltable and multiplication by function $u_a(t - \tau)$, expressing the law of amplitude modulation (gating during pulse modulation of the signal) by no means must necessarily be produced by giving to heterodyne voltage the corresponding law of amplitude modulation. These two functions can also be realized separately.

Formation of the square of the envelope of voltage at the output of the filter can be carried out by a square-law detector. In order to produce voltage, proportional to the derivative with respect to $\omega_{\rm H}$, it is possible to realize two channels of the described form, filters of which are detuned relative to $\omega_{\rm HP}$ by $\frac{i\Delta\omega}{2}$, and to subtract the output voltages of these channels. As a result we replace differentiation by calculation of the difference and as $\Delta\omega \rightarrow 0$ output voltage of the resulting circuit seeks z(t). Optimum speed meter is seen in Fig. 9.1.



Fig. 9.1. Optimum speed meter: 1 - controlled local oscillator; 2 - mixer; 3 filter; 4 - square-law detector; 5 smoothing circuit. Comparison of the obtained form of optimum processing of a signal with results of Chapter IV shows that an optimum frequency discriminator consists of two channels of optimum detection, detuned with respect to the measured parameter (frequency), which facilitates construction of radars optimum both in conditions ij

and the state of t

of detection and in conditions of measurements.

We emphasize that the optimum circuit is synthesized for an arbitrary form of radiation, both continuous and also pulse. Operations of processing the signal, starting with narrow-band filtration, do not depend on the method of radiation and form of modulation. This will facilitate our further consideration of different

-412-

deviations from the optimum method of construction of the frequency discriminator, allowing us not to turn to concrete forms of modulation.

§ 9.3. Characteristics of an Optimum Frequency Discriminator

The found frequency discriminator is optimum from the point of view of obtaining maximum accuracy of measurement of speed. Therefore, its basic characteristics are magnitudes characterizing accuracy, i.e., in accordance with ChapterVI, equivalent spectral density S_{ORT} and a coefficient determining the intensity of parametric fluctuations S_{Rer} .

Equaivalent spectral density with an optimum discriminator can be found according to (6.7.33) by the formula

$$S_{our}^{-1} = -\frac{1}{2T} \int_{0}^{T} \int_{0}^{T} \frac{\partial F(t_1, t_2, \omega_{\lambda})}{\partial \omega_{\lambda}} \frac{\partial R(t_1, t_2, \omega_{\lambda})}{\partial \omega_{\lambda}} \frac{\partial R(t_1, t_2, \omega_{\lambda})}{\partial \omega_{\lambda}} dt_1 dt_2, \qquad (9.3.1)$$

where $R(t_1, t_2, \omega_{II}) = correlation function of the received signal, determined by formula (9.2.1);$ $<math>W(t_1, t_2, \omega_{II}) = function$, found from solution of equation (6.7.20), equal to

$$W(t_{1}, t_{2}, \omega_{\pi}) = -u_{a}(t_{1} - \tau) u_{a}(t_{2} - \tau)\cos[(\omega_{0} + \omega_{\pi})(t_{1} - t_{2}) + \psi(t_{1} - \tau) - \psi(t_{2} - \tau)] \frac{1}{\pi N_{0}} \int_{-\infty}^{\infty} \frac{hS_{0}(\omega)}{1 + hS_{0}(\omega)} e^{i\omega\tau} d\omega + \frac{1}{N_{0}} \delta(t_{1} - t_{2}).$$
(9.3.2)

Substituting (9.2.1) and (9.3.2) in (9.3.1) and performing calculations, we obtain

$$S_{\sigma\sigma\sigma}^{-1} = \frac{1}{2\pi} \int_{-\infty}^{\infty} \left[\frac{hS_{\sigma}'(\omega)}{1 + hS_{\sigma}(\omega)} \right]^2 d\sigma, \qquad (9.3.3)$$

where the stroke is the sign of differentiation.

The same result can be obtained by another method, if we consider that for an optimum discriminator

$$S_{\text{OHT}} = \frac{1}{K_{\text{OHT}}}, \qquad (9.3.4)$$

where $K_{\rm opt}$ - the gain factor of the discriminator, defined as

$$K_{\text{OUT}} = \left| \frac{\partial z(t, \omega_{\text{A}})}{\partial \omega_{\text{A}}} \right|_{\boldsymbol{w}_{\text{A}}} = \hat{\boldsymbol{w}}_{\text{A}}^{*}$$
(9.3.5)
Averaging in set (9.2.5), we easily obtain for the discrimination characteristic of an optimum frequency discriminator

$$a(\delta) = \overline{z(t, \delta)} = -\frac{\hbar}{2\pi} \int_{-\infty}^{\infty} |H_{\bullet}(i\omega)|^{*} S'_{\bullet}(\omega + \delta) d\omega, \qquad (9.3.6)$$

where $\delta = \omega_{\Pi} - \hat{\omega}_{\Pi}$, and $|H_0(i\omega)|^2$ is determined by expression (9.2.3).

From (9.3.5) and (9.3.6) we can obtain formula (9.3.3) for equivalent spectral density. Moreover, we incidentally obtain the expression for the discrimination characteristic of the optimum discriminator, needed during the analysis of processes occurring with strong noises and interferences.

For finding the coefficient of parametric fluctuations $S_{\pi\sigma\Gamma}$ we calculate the equivalent fluctuation characteristic of the discriminator

$$S_{\text{DRB}}(\delta) = \frac{1}{K_{\text{ORT}}^2} \int_{-\infty}^{\infty} \left[\overline{z(l, \delta) z(l+\tau, \delta)} - \overline{z(l, \delta) z(l+\tau, \delta)} \right] d\tau. \qquad (9.3.7)$$

Performing calculations, we obtain

$$S_{\text{org}}(\delta) = \frac{1}{2\pi K_{\text{curr}}^2} \int \left[\frac{\partial}{\partial \omega} \left[H_{\bullet}(i(\omega + \delta)) \right]^2 \right]^4 [hS_{\bullet}(\omega) + 1]^4 d\omega.$$
(9.3.8)

Coefficient S_{Hat} is now calculated as

$$S_{\text{sap}} = \frac{1}{2} \frac{d^4 S_{\text{sab}}(\hat{\theta})}{d\hat{\theta}^4} \bigg|_{\theta=0}.$$
 (9.3.9)

From the obtained expressions it follows that both equivalent spectral density S_{OHT} , and coefficient S_{H8T} do not depend on the form of modulation of the sounding signal but are determined only by the signal-to-noise ratio h and the form of the spectrum of fluctuations of the reflected signal $S_O(\omega)$.

Of basic importance for determining error of the speed meter is the equivalent spectral density S_{OBT} . Let us find it for certain characteristic forms of $S_O(\omega)$.

If spectral density of the signal corresponds to white noise passing through an RC-filter, then

$$S_{\bullet}(\omega) = \frac{1}{1 + \left(\frac{\omega}{2\Delta f_{\bullet}}\right)^{1}}$$
(9.3.10)

where Δf_{c} - effective band of fluctuations of the signal.

Substituting (9.3.10) in (9.3.3), we obtain

$$S_{out} = 2\Delta f_o \frac{\sqrt{1+h}(1+\sqrt{1+h})^3}{h^3}$$
 (9.3.11)

When the spectral density of the signal corresponds to white noise passing through two series-coupled RC-filters

$$S_{\bullet}(\bullet) = \frac{1}{\left[1 + \left(\frac{\Theta}{4\Delta f_{\bullet}}\right)^{2}\right]^{2}}.$$
(9.3.12)

From (9.3.3) we obtain

$$S_{oHT} = \Delta f_{c} \left[1 - 3 \sqrt{\frac{V_{1+h-1}}{2h}} + \frac{1}{2} \sqrt{\frac{h}{2(1+h)(V_{1+h-1})}} \right]^{-1} \cdot (9.3.13)$$

If, however, the spectrum of the signal has a Gaussian form,

$$S_{o}(\omega) = \exp\left[-\frac{1}{4\pi}\left(\frac{\omega}{\Delta f_{c}}\right)^{s}\right],$$
 (9.3.14)

it is possible to calculate \mathbf{S}_{OHI} approximately, obtaining here

$$S_{ont} \approx \begin{cases} 4\sqrt{2\pi} \frac{\Delta f_o}{h^3} & h < 1, \\ \frac{\Delta f_o}{(\ln h)^{3/2} - \frac{3}{2} (\ln h)^{1/2}} & h > 1. \end{cases}$$
(9.3.15)

Dependences of $S_{0\Pi T} / \Delta f_{C}$ on the signal-to-noise ratio h for the three considered cases are shown in Fig. 9.2.

These three cases do not exhaust, of course, all possibilities. However, they belong to one family of functions $S_{\Omega}(\omega, n)$, depending on integral parameter n,

$$S_{\bullet}(\mathbf{w}, n) = \frac{1}{[1 + (\mathbf{w}\tau_n)^*]^n}, \qquad (9.3.16)$$
$$s_n = \frac{1}{\Delta f_o} \frac{(2n-2)!}{2^{2n-1} [(n-1)!]^3}.$$

where

Generalizing the obtained results for an arbitrary n, we can obtain limiting relationships for $S_{0,0,0}$ for small and large signal-to-noise ratios:

$$S_{\text{URT}}(n) = \frac{\Delta f_{\sigma}}{h^{2}} \frac{2^{n-1} (2n+1) [(n-1)!]^{2} [(2n)!]^{2}}{n^{2} (4n)! (2n-2)!} \quad h \to 0,$$

$$S_{\text{URT}}(n) = \Delta f_{\sigma} \frac{2^{2n-2} [(n-1)!]^{2}}{n^{2} (2n-2)!} \quad h \to \infty.$$
(9.3.17)

From these relationships and from Fig. 9.2 it follows that for small h the magnitude of S_{OUT} depends little on the form of the spectrum, being proportional

-415-



Fig. 9.2. Dependence of equivalent spectral density of an optimum frequency discriminator S_{OR1} on the signal-to-noise ratio h: ---- $S_O(\omega)$ is expressed by formula (9.3.10); ---- $S_O(\omega)$ is expressed by formula (9.3.12); ---- $S_O(\omega)$ is expressed by formula (9.3.14); to $1/h^2$. For large h(h > 100)the magnitude of S_{ORT} essentially depends on the form of the spectrum, varying as $h \rightarrow \omega$ from $2\Delta f_{C}$ ($\alpha = 1$ to 0 ($n \rightarrow \omega$). In Table 9.1 is given the dependence of limiting values of S_{ORT} on the approximation of the spectrum for family (9.3.16).

For very large signal-to-noise ratios the question of optimization of a meter usually does not arise. Therefore, working signal-to-noise ratios should be considered magnitudes of h = 0.1 to 106. In this range of h the dependence of S_{OHT} on the form of the spectrum of fluctuations

of the signal is not great and it often can be ignored, considering for simplicity approximation (9.3.10).

It should be noted that one should not especially trust the obtained values of $S_{\rm OHT}$ for very large h, since the calculating formulas used here become insufficiently exact.

n	1	2	3	4	5	 80
$\lim_{h\to 0} \frac{h^2 S_{obt}}{\Delta f_o}$	16	18,3	18,35	18,32	18,3	 17,7
$\lim_{h\to\infty} \frac{S_{ont}}{\Delta f_o}$	2	1	0,59	0,4	0,29	 0

Table 9.1

Besides consideration of the dependence of S_{OID} on h one should emphasize the proportionality of S_{OID} , and consequently, and of variance of error of measurement of speed, to the effective width of the spectrum of fluctuations of the signal $\Delta f_{\rm C}$. Here there exists a fundamental difference between results obtained for Doppler speed meters and meters for range and coordinates, and using range finders

-416-

and goniometers. It consists in the fact that there exists a limiting error of measurement of speed for any size signal-to-noise ratio, while in range finders and goniometers error decreases without limit with increase of this ratio. Physically this is explained by the fact that in a speed meter subjected to measurement is the derivative of the phase of the signal, which for the accepted idealizations contains, due to fluctuations of the signal, an additive random addition, not depending on the signal-to-noise ratio.

Thus, the method of encoding the measured parameter (speed) in the signal in principle does not permit us to compensate the influence of fluctuations of the signal, in distinction from range finders and goniometers where such compensation is possible.

Assigning the simplest form of spectral density of fluctuations of the signal (9.3.10), by formulas (9.3.8) and (9.3.9) we find S_{max} . Results of rather cumbersome





calculations are presented in the form of the curve of Fig. 9.3. As $h \rightarrow \infty$ quantity S_{map} reaches its maximum and is equal to $1/2\Delta f_c$. We do not give values of S_{map} for small h, since here conditions of optimality of the synthesized circuit are no longer realized, and the model of the tracking meter accepted for analysis of accuracy loses its meaning, which leads during formal application of the expression for S_{map} to physically inexplicable results. We shall postpone estimation of the increase of fluctuation error due to parametric fluctuation until our consideration of smoothing filters and of errors of the opeed mater as p whele.

§ 9.4. Analysis of Certain Circuits of Frequency Discriminators

Let us turn to analysis of frequency discriminators, whose circuits differ from the above-considered optimum circuit. Analyzing properties of these discriminators and characterizing them by parameters S_{BKB} and S_{BAB} , we have the possibility of comparing different discriminators with one another and with an optimum discriminator. As a result we shall find conditions in which certain of the circuits are close in performance to optimum ones, i.e., accuracy of measurement of speed provided by them is close to potential accuracy belonging to an optimum meter.

For convenience of consideration we introduce certain nonessential idealizations. During the analysis of different circuits we assume that convolution of the spectrum of an arbitrarily modulated received signal is realized ideally. This assumption means that the arbitrarily modulated signal will be converted into a signal, containing only one spectral component of width Δf_c , by which we measure the Doppler frequency, without change of the signal-to-noise ratio with respect to the input signal of the receiver. As a result characteristics of the considered circuits will not depend on the form of modulation similarly to how this took place for the optimum circuit. Subsequently, we shall take into account those changes to which imperfectness of convolution of the spectrum of the signal leads. It turns out that this is possible to do in sufficiently general form without turning to consideration of the concrete forms of modulation.

We shall subsequently assume that the receiver in the considered discriminators, has an automatic gain control [AGC] system, which reduces the average level of voltage at the output of the amplifier covered by it to a constant level. Thus, assumptions about the function executed by the AGC system remain the same as in Chapter VII. With application of square-law detectors or multipliers in circuits of discriminators this leads, naturally, to the same dependence of the discriminator gain factor K_n on the signal-to-noise ratio h

$$K_{\mathbf{X}} = K_{\mathbf{X}} \left[1 + \frac{1}{h} \left(\frac{\Delta f_{\mathbf{Y}}}{A f_{\mathbf{C}}} \right) \right]^{-1}, \qquad (9.4.1)$$

where $K_{j_{1}j_{2}}$ - nominal value of the gain factor in the absence of noises;

 Δf_y — effective bandwith of the amplifier covered by the AGC loop (Δf_y >> Δf_c).

When there is no AGC system, or it does not work due to smallness of the amplitude of the signal, the dependence of $K_{\rm H}$ on h changes. Here, the gain factor of the discriminator is proportional to h. There may be other dependences, determined by suppression of the signal by noise due to various nonlinearities in the radio channel. However, we henceforth take dependence (9.4.1).

Let us consider in the above-described plan some of the most widely used circuits of frequency discriminators.

-418-

9.4.1. Circuit with a Tuned Loop and a Phase Shifter

The diagram of a frequency discriminator is shown in Fig. 9.4. Input signal y(t) is fed to a mixer. Heterodyne voltage, proceeding to the same mixer, is



Fig. 9.4. Diagram of a frequency discriminator with a tuned loop and a phase shifter: 1 - mixer; 2 - controlled localoscillator; 3 - amplifier; 4 - modulator; 5 - amplifier with controlled gain; 6 - AGC circuit; 7 - tuned loop; 8 - phaseshifter; 9 - phase detector. modulated in phase in such a way that there is ensured phase demodulation of the signal. Heterodyne frequency is controlled by the output voltage of the speed meter. After heterodyning the signal is subjected to amplification, where the gain factor changes in accordance with the law of amplitude modulation of the received signal.

In the case of a pulse signal the shown operations simply signify gating of the amplifier with the proper form of gate pulse. Delay of phase modulation of the local oscillator and amplitude modulation in the amplifier is controlled from the output of the range finder. Then the signal is passed through an amplifier with band $\Delta f_y \gg \Delta f_c$, which is subject to automatic gain control. Note that in practice preceding stages (gated stages) may also have gain control.

We assume that band Δf_y is sufficiently small to ensure suppression of sidebands of amplitude modulation of the signal. This assumption is made for convenience of further analysis, although it is obvious that results are not changed if it is not realized. The described part of the circuit, ensuring demodulation of the signal and reduction of it to a level corresponding to work of subsequent stages in a linear regime, is common both for the considered, and also for the following circuits of discriminators.

After demodulation the signal proceeds to the tuned loop of the discriminator with passband $\Delta f_{\rm R}$ and from its output it is fed to a phase detector; simultaneously to the phase detector there is fed voltage from the output of the amplifier, shifted $\pi/2$ by a phase shifter. Output voltage of the phase detector, which multiplies these voltages, is the output variable of the discriminator, which after amplification and smoothing controls the frequency of the local oscillator. The principle of action of the circuit is based on the fact that the phase-frequency response of a loop tuned to a certain frequency is an odd function of detuning relative to this frequency, approximately linear with small detuning. As a result during detuning of middle frequencies of the filter and of the signal beats between

-419-

components of the signal at the phase detector input, shifted correspondingly by the phase shifter and filter, form a constant component, approximately proportions: to detuning.

The mixture of the signal and noise at the input of the discriminator $1 \le j$ tuned to frequency $\alpha_{j,r}$ can be presented in the form

$$u_{1}(t) = A(t) \cos \omega_{np} t + B(t) \sin \omega_{np} t, \qquad (9.4.2)$$

(9.4.3)

where

$$A(t) = a(t)\cos\delta t + b(t)\sin\delta t + \xi(t),$$

$$B(t) = -a(t)\sin\delta t + b(t)\cos\delta t + \eta(t),$$

 δ - detuning between frequencies of the local oscillator and the signal: a(t) and b(t) - independent normal random processes, characterizing fluctuations of the received signal, with spectral density $S_0(\omega)$;

 $\xi(t)$ and $\eta(t)$ - noises with spectral density, uniform in the passband of the loop and equal to 1/h.

Strictly speaking, voltage at the output of the loop is not equal, but proportional to $u_1(t)$ with a proportionality factor which depends on amplification of the channel and the power of the signal at the input. However, subsequently this proportionality factor has no significance, since we introduced characteristics of the discriminator which do not depend on constant proportionality factors.

Output voltage of the discriminator $u_{jl}(t, \delta)$ is obtained by multiplying the result of passage of $u_1(t)$ through a filter with pulse response $h(t) \cos \omega_{hp} t$ and the result of shift $u_1(t)$ in phase by δ/ϵ . Performing the shown operations, we have, with an accuracy of a constant coefficient,

$$u_{\pi}(t, b) = \frac{1}{2} \int_{-\infty}^{b} h(t-\tau) [A(\tau) B(t) - A(t) B(\tau)] d\tau. \qquad (9.4.4)$$

Letermining by usual method the gain factor of the discriminator $K_{\!\!\rm II}$, we have

$$K_{\mathbf{x}} = \left| \frac{\partial \overline{u_{\mathbf{x}}(\mathbf{i}, \mathbf{\delta})}}{\partial \mathbf{\delta}} \right|_{\mathbf{b}=0} = \frac{1}{2\pi} \int_{-\infty}^{\infty} H^{*}(i\omega) S'_{\mathbf{0}}(\omega) d\omega, \qquad (9.4.5)$$

where $H(i\omega)$ - Fourier transform of h(t); * - designate: a complex conjugate magnitude; and the stroke - sign of differentiation.

Equivalent spectral density S_{RKB} determined by the value of the fluctuation characteristic for $\delta = 0$, is equal to

$$S_{0KB} = \frac{1}{K_A^2} \frac{1}{4\pi} \int_{-\infty}^{\infty} \left[|H(in)|^2 - H^{*2}(in)] \left[S_0(n) + \frac{1}{\hbar} \right]^2 dn. \qquad (9.4.6)$$

For the simplest form of spectral density (9.3.10) and a loop whose

low-frequency equivalent has frequency response

$$H(i\omega) = \frac{1}{1 + \frac{i\omega}{2\Delta I_{\star}}}, \qquad (9.4.7)$$

expressions (9.4.5) and (9.4.6) take form

$$K_{R} = k \frac{\Delta f_{c} \Delta f_{R}}{2 (\Delta f_{R} + \Delta f_{c})^{3}} = k \frac{\alpha}{2 (1 + \alpha)^{3}}, \qquad (9.4.8)$$

$$S_{3K_{3}} = 2\Delta f_{c} \left(1 + \frac{\Delta f_{c}}{\Delta f_{R}}\right) \left[1 + \frac{2}{h} \left(1 + \frac{\Delta f_{K}}{\Delta f_{c}}\right) + \frac{1}{h^{3}} \left(1 + \frac{\Delta f_{c}}{\Delta f_{c}}\right)^{2}\right] = 2\Delta f_{c} \left(1 + \frac{1}{a}\right) \left[1 + \frac{2}{h} (1 + \alpha) + \frac{1}{h^{3}} (1 + \alpha)^{3}\right]. \qquad (9.4.9)$$

Here $\alpha = \Delta f_{\rm K} / \Delta f_{\rm C}$; k - proportionality factor, not affecting the magnitude of $S_{\partial \rm KB}$, but, in general, depending on the signal-to-noise ratio due to the influence of the AGC system, described above. As a result the dependence of $K_{\rm H}$ on h is determined by formula (9.4.1). Formula (9.4.8) determines the dependence of $K_{\rm H}$ on the pessband of the loop $\Delta f_{\rm K}$. From (9.4.8) it follows that maximum $K_{\rm H}$ is attained at $\alpha = \Delta f_{\rm K} / \Delta f_{\rm C} = 1$, but as in other meters this still does not minimize $S_{\rm BKB}$.

Quantity S_{gKB} consists of three components, caused by beats of components of the signal at inputs of the phase detector (component not depending on h), beats of the signal with noise (component containing 1/h) and beats of noise with noise (component with 1/h²). For low-level noises the first component predominates; for high levels, the third. For any value of h quantity S_{gKB} has a minimum at a certain value of the ratio of bandwidth of the loop and signal α . This testifies to presence, as in the optimum circuit, of an optimum value of bandwidth Δf_{R} . Minimum β_{gKB} is obtained at

$$\mathbf{a} = \frac{\Delta f_{\pi}}{\Delta f_{\sigma}} \approx \begin{cases} \frac{1}{3} + \frac{h}{4} & h < 1, \\ \sqrt{\frac{h}{2}} & h > 1; \end{cases}$$
 (7.4.1

in the optimum circuit, according to results obtained above, the filter should have the same form of frequency response with the relationship of bandwidths

$$a_{out} = \sqrt{1+h}.$$
 (9.4.11)

Thus, in a circuit with a tuned loop and a phase shifter the best bandwidth of the loop is approximately 1/3 as broad as in the optimum circuit for small h and $\frac{1}{\sqrt{2}}$ as broad as in the optimum for large n. This is explained by the difference be

-421-

the character of processing of the signal. It is useful to estimate qualities of the considered elecuit, comparing it with the optimum circuit. Therefore, of interest is comparison of S_{OLF} , def by formula (9.4.9), with S_{OLF} , expressed by (9.3.11), for various values of

Results of such comparison are presented in Fig. 9.5. For low-level noises and large α from formula (9.4.9) and from the given curves it is clear that the



Fig. 9.5. Dependence of S_{409}/S_{005} on the signal-to-noise value h for discriminators with a toped loop and a phase shifter and when mixers and differentiation for various ω .

magnitude of $S_{0BB} \rightarrow 2\Delta f_{c}$, i.e., the circuit in its properties do not differ from the optimum one. For high-level noises and $\alpha \gg 1$ the magnitude of Same is approvimately proportional to a2. This is explained both by lowering of the gain factor of the discriminate tor, and also by increase of the power of noises at the output of the loop without eccential increase of signal power. At the same time narrowing of Afg with respect to the optimum value of the loop passband also leads to growth of S due to decrease of the gain factor. From the given curves we see that in the range of signal-to-noise ratio.

h = 0.1 to 100 loss in the considered circuit as compared to the optimum can orreduced by selection of the passband of the loop for small h to 20% or less, and for large h practically to zero.

Loring the analysis of nonlinear phenomena, occuring for high-level noises, it may be useful to know the form of the discrimination charactoristic of the considered circuit. By averaging (9, 4, 4) we obtain for it the expression

$$a(\delta) = \overline{u_{\mathbf{A}}(t, \delta)} = \frac{1}{4\pi t} \int_{-\infty}^{\infty} H(i\mathbf{m}) \left[S_{\mathbf{a}}(\mathbf{m} - \delta) - S_{\mathbf{a}}(\mathbf{m} + \delta) \right] d\mathbf{\omega}. \qquad (9.4.12)$$





For spectral density of signal (9.3.19) and frequency response (9.4.7) this expression, after calculations, gives

$$k_1(\delta) = k_1 \frac{1}{1+s_1+s_2},$$
 (9.4.13)

where

$$\mathbf{x} = \overline{\mathbf{2}(\Delta \mathbf{i} + \Delta \mathbf{i})}, \qquad (9.4.14)$$

and $k_1 - proportionality factor.$

Calculation of the fluctuation characteristic

of the considered discriminator on the same

on h for a discriminator with a tuned loop and a phase shifter for different α .

assumptions with respect to the form of the spectrum of fluctuations and frequency response of the loop leads to expression

$$S_{0 \times n}(x) = 2\Delta \int_{\varphi} \frac{1+\alpha}{\alpha} \left\{ \frac{1+((1+\alpha)^{n}+3\alpha)x^{n}+\alpha(1+\alpha)x^{n}}{(1+x^{n})^{n}} + \frac{2(1+\alpha)}{k} \frac{1+(1+2\alpha)x^{n}}{(1+x^{n})^{n}} + \frac{(1+\alpha)^{n}}{k^{n}} \right\}.$$
(1.11)

Coefficient \mathbf{S}_{max} is found from this as

$$S_{\text{map}} := \frac{1}{2} \frac{S''_{\text{ma}}(3)}{[2(\Lambda/a + \Delta/a)]^2} := \frac{1}{2} \frac{1}{\Delta/a} \frac{1}{1 + a} \left[5 + a - \frac{2}{a} + \frac{2(1 + a)(21 - 1)}{a\hbar} \right],$$
(').4.2(')

where the two strokes designate the second derivative of S_{aKB} with respect to \sim .

In Fig. 9.6 are dependences of $S_{\text{HO}I}$ on the signal-to-noise ratio h for various values of the ratio of bandwidths of the loop and signal. As $h \to \infty$ the magnitude of $S_{\text{HO}I}$ seeks a limiting value, the dependence of which on α is presented in Fig. 9.7. With decrease of h the magnitude of $S_{\text{HO}I}$ grows inversely proportionally to h.

9.4.2. Circuit with Mixers and Differentiation

During presentation of circuits of frequency discriminators here and subsequently we shall not depict circuits of demodulation of the signal and of preliminary processing, which remain the same as in the circuit of Fig. 9.4. The part of the circuit determining the principle of action of the considered discriminator is shown in Fig. 9.8. Output voltage of amplifier enters two mixers to which as the



reference voltages there are fed the voltage of a local oscillator tuned from the output of the meter and the same voltage, shifted in phase $\pi/2$ (with the help of a phase shifter). Output voltage of one of the mixers is fed to a lowfrequency filter, and from the other to a differentiating circuit with the same time constant. Output voltages of the filter and the differentiating circuit are multiplied in a phase detector.

forming the output voltage of the discriminator.

The principle of action of this discriminator is based on the fact that with detuning b between frequencies of the local oscillator and the signal at the output of mixers there occur variable components of frequency b, shifted in phase $\pi/2$. The differentiator removes this phase shift, forming voltage whose amplitude is determined by detuning. As a result, after multiplication in the phase detector there is obtained a quantity which is proportional to detuning b.

In connection with the fact that the local oscillator is tuned from the frequency of the input signal of the mixers, voltage at the output of one of them is equal to A(t), and at the output of the other is B(t) (9.4.3). At the discriminabor output there will be formed

$$u_{\mathbf{A}}(t, \delta) = \int_{-\infty}^{t} \int_{-\infty}^{t} h(t-t_{1}) h_{1}(t-t_{2}) A(t_{1}) B(t_{2}) dt_{1} dt_{2}, \qquad (9.4.17)$$

where h(t) - pulse response of a low-frequency filter;

 $h_1(t)$ - pulse response of the differentiating circuit, whose frequency response is related to the frequency response of the low-frequency filter $H(i\omega)$ by relationship

$$H_{1}(i\omega) = \frac{i\omega}{2\Delta f_{\pi}} H(i\omega). \qquad (9.4.18)$$

Using formula (9.4.17) and performing proper averaging, we obtain the gain factor of the discriminator

$$K_{\mathbf{x}} = \left| \frac{\partial \overline{u_{\mathbf{x}}(t, \delta)}}{\partial \delta} \right|_{\mathbf{t}=0} = -\frac{i}{2\pi} \int_{-\infty}^{\infty} H(i\omega) H^{*}_{\mathbf{x}}(i\omega) S'_{\mathbf{0}}(\omega) d\omega \qquad (9.4.19)$$

and equivalent spectral density

$$S_{\theta R \theta} = \frac{1}{K_{A}^{2}} \int_{-\infty}^{\infty} [\overline{u_{R}(t, 0) u_{R}(t + \tau, 0)} - \frac{1}{u_{R}(t, 0)} \frac{1}{u_{R}(t, 0)} \frac{1}{u_{R}(t + \tau, 0)}] d\tau = \pm$$
(9.4.20)
$$= \frac{1}{K_{A}^{2}} \frac{1}{2\pi} \int_{-\infty}^{\infty} |H(i\omega)|^{2} |H_{1}(i\omega)|^{2} \left[S_{\theta}(\omega) + \frac{1}{\hbar}\right]^{2} d\omega.$$

In order to have the possibility of comparing properties of the considered discriminator with properties of the preceding circuits, we consider that the low-frequency filter in this case has frequency response $H(i\omega)$, determined by (9.4.7), i.e., is the l-f equivalent of the bandpass filter of the preceding circuit. Then, as it is easy to show, the discrimination and fluctuation characteristics of the considered discriminator and of the discriminator with a tuned loop and a phase shifter coincide. In particular, expressions for K_{μ} and $S_{\rm 3KB}$ obtained from (9.4.19) and (9.4.20) coincide with formulas (9.4.8) and (9.4.9), respectively. The coefficient of parametric fluctuations is determined by formula (9.4.16). This means that although in the method of formation of the signal of error the circuits of Figs. 9.4 and 9.8 differ from each other, their characteristics completely coincide (with satisfaction of the imposed conditions), and all dependences obtained above remain in force for the given discriminator.

Thus, these two discriminators are equivalent. The basis for application of the circuit of Fig. 9.8 instead of the preceding one, in spite of greater complexity, is the possibility of using low-frequency filters instead of bandpass filters, creation of which causes technical difficulties in cases when the signal has a very narrow band. In these cases there is required high stability of the bandpass filters, obtained by application of crystal filters, whereas during creation of low-frequency filters no difficulties arise.

9.4.3. Circuit with Detuned Loops

The considered circuit without networks intended for convolution of the spectrum of the signal is presented in Fig. 9.9. The signal from the output of the

-425-



Fig. 9.9. Circuit of a frequency discriminator with actuned loops: 1 - 100p; 2 - detector.

amplifier proceeds to two mutually detuned (by $\Delta \omega$) loops. Output voltages of the loops are detected and substracted, forming the output quantity of the discriminator. It is clear that as $\Delta \omega$ approaches zero and with matching of frequency responses of the loops with the spectrum of the signal the

- Shouth the state of the

circuit exactly executes optimum operations. However, here the transmission factor of the discriminator seeks zero, which should be compensated by a high-gain amplifier, coupled in series with the depicted part of the discriminator circuit. In order to avoid considerable loss in amplification, in the real circuit detuning of loops is produced by a finite quantity $\Delta \omega$, comparable with the width of the spectrum of the signal. Furthermore, there is practically selected some transmission band of the loops $\Delta f_{\rm R}$, which does not change with change of the signal-to-noise ratio h. Frequently, due to technical conditions it is impossible to select this band in such a way as to ensure matching of the loops with the spectrum of the signal.

Considering the shown peculiarities, we shall analyze qualities of the circuit with detuned loops. Voltage to the input of the analyzed part of the circuit is determined, as before, by expression (9.4.2). This voltage is passed through filters with pulse responses h(t) cos ($\omega_{\rm HP} - \Delta\omega/2$)t and h(t) cos ($\omega_{\rm HP} + \Delta\omega/2$)t. Squares of the moduli of voltages obtained at the output of the filters, formed by square-law detectors, are subtracted. Considering these operations, for the output of the discriminator, with an accuracy of a constant coefficient, we obtain

$$\mathbf{x}_{\mathbf{x}}(t, \mathbf{\delta}) = \int_{-\infty}^{t} \int_{-\infty}^{t} h(t-t_{1}) h(t-t_{2}) \sin \frac{\Delta \mathbf{\omega}}{2} (t_{2}-t_{3}) \times \\ \times [A(t_{2}) B(t_{1}) - A(t_{3}) B(t_{2})] dt_{1} dt_{3}.$$
(9.4.21)

Calculating by the usual method the slope of the discrimination characteristic and equivalent spectral density, we have

$$K_{\rm A} = -\frac{1}{\pi} \int |H(l\omega)|^2 S_{\rm e} \left(\omega + \frac{\Delta \omega}{2}\right) d\omega \qquad (9.4.22)$$

and

$$S_{gRB} = \frac{1}{K_{A}^{2}} \frac{1}{2\pi} \int_{-\infty}^{\infty} \left\{ \left[\left| H\left(i\left(\omega + \frac{\Delta \omega}{2}\right) \right) \right|^{2} - \left| H\left(i\left(\omega - \frac{\Delta \omega}{2}\right) \right) \right|^{2} \right] \left[S_{\bullet}(\omega) + \frac{1}{h} \right] \right\}^{2} d\omega.$$
(9.4.23)

The discrimination characteristic of the given form of a frequency discriminator is expressed as

$$a(\delta) = \overline{u_{R}(t, \delta)} = \frac{1}{2\pi} \int_{-\infty}^{\infty} |H(i\omega)|^{2} \left[S_{\bullet} \left(\omega + \frac{\Delta \omega}{2} - \delta \right) - S_{\bullet} \left(\omega + \frac{\Delta \omega}{2} + \delta \right) \right] d\omega, \qquad (9.4.24)$$

whereas its equivalent fluctuation characteristic is found by formula

$$S_{RRB}(\delta) = \frac{1}{K_{A}^{d}} \frac{1}{2\pi} \int_{-\infty}^{\infty} \left[\left| H\left(i\left(\omega + \frac{\Delta \omega}{2}\right) \right) \right|^{2} - \left| H\left(i\left(\omega - \frac{\Delta \omega}{2}\right) \right) \right|^{2} \right]^{2} \left[S_{0}\left(\omega - \delta\right) + \frac{1}{\hbar} \right]^{2} d\omega.$$

$$(9.4.25)$$

The obtained characteristics determine all properties of the considered discriminator interesting us. As also for the preceding circuits, we calculate them with spectral density $S_0(\omega)$ having the form of (9.3.10) and frequency response $H(i\omega)$ of form (9.4.7). Then the transmission factor of the discriminator is expressed by formula

$$K_{\rm A} = k \frac{2a (1 + a) \mu_i}{(\mu^2 + (1 + a)^2)^2}, \qquad (9.4.26)$$

where k, as before, is a coefficient which depends on h due to the influence of automatic gain control;

$$q = \frac{\Delta f_{\pi}}{\Delta f_{e}}; \mu = \frac{\Delta \omega}{4\Delta f_{e}},$$

Equivalent spectral density takes the form

$$S_{0K0} = \frac{\Delta f_{a}}{4a} \frac{\mu^{a} + (1+a)^{2}}{(\mu^{a} + a^{2})(1+a)^{2}} \left\{ [\mu^{a} + (1+a)(1+5a+8a^{2})] + \frac{2}{h} [\mu^{a} + (1+a)^{2}] (\mu^{a} + 1+4a+5a^{4}) + \frac{1}{h^{2}} [\mu^{a} + (1+a)^{2}]^{a} \right\}.$$
(9.4.27)

Quantity S_{0KB} , as also for the preceding circuits, has three terms, variously depending on the signal-to-noise ratio h and explained by beats of different components of the mixture of the signal with noise. The dependence of S_{0KB} on h, α , and μ can be investigated most vividly by graph. In Fig. 9.10 there is shown the dependence of S_{0KB}/S_{0HT} on h for different values of α and μ . From these curves we see that the dependence of S_{0KB}/S_{0HT} on the ratio of bandwidths $\alpha = \Delta f_K/\Delta f_C$ is not monotonic, and as for the preceding circuits there exists an optimum value of the bandwith of the loops Δf_K , minimizing S_{0KB} . For small μ this optimum band



Fig. 9.10. Dependence of S_{0107}/S_{0117} on the signal-to-noise ratio h for a discriminator with detuned loops: ---- $\mu = 0.3$; ---- $\mu = 1$; ---- $\mu = 5$.

is close to $\Delta f_{c} \sqrt{1 + h}$, as one should have expected, since as $\mu \rightarrow 0$ and with such selection of the loop bandwidth the circuit approaches its optimum form. For large μ the value of the optimum bandwidth of the loops changes.

If, fixing band of the loops and the signal-to-noise ratio, we investigate the dependence of the magnitude of S_{GCB} on detuning of loops μ , it is easy to see that for large α it increases with growth of detuning μ , while for small α in a certain range of variation of h there is the minimum spectral density for a certain optimal value of detuning μ .

Thus, in this circuit, just as in the others, deviation from optimum value by one of the parameters

(bandwidth of loops) leads to change of the optimum value of the other parameter (detuning). From Fig. 9.10 it follows that with some, at first glance, fully reasonable selection of parameters of the circuit its accuracy may be an order or two hower than the potential. This testifies to the necessity of careful selection of bands of the loops and their detuning in the considered circuit, and also the great pensitivity of this form of discriminator to its parameters.

For a more complete judgement of accuracies which are provided by the given circuit it is also necessary to calculate the coefficient of parametric fluctuations $S_{\rm mAT}$. It, as earlier, is calculated from the fluctuation characteristic and is presented graphically in Fig. 9.11. Quantity $S_{\rm H01}$ contains a component not depending on h, owing its origin to beats of signal components. With decrease of h coefficient $S_{\rm H01}$ grows inversely proportionally to h. For very small h there is no sens in using the obtained dependences, since error of the system of measurement no longer is expressed correctly through $S_{\rm H01}$.

-428-



Fig. 9.11. Dependence of S_{map} on h for a discriminator with detuned loops: $-\mu = 0.3$; $--\mu = 1$; $--\mu = 5$.

9.4.4. Circuit with Frequency Variation

Part of the considered discriminator circuit, determining the principle of its action, without circuits of preliminary processing of the signal, is presented in Fig. 9.12. The input signal is mixed with output voltage of a local oscillator controlled from the output of the meter. The output of the meter determines the mean value of the frequency of the local

oscillator, periodically varying (near this mean value) according to a law assigned by the generator of reference voltage. From the output of the mixer the signal proceeds to a bandpass filter (tuned loop), is detected by a square-law amplitude detector and enters a stage which multiplies it by the reference voltage. Output of this stage is the output of the discriminator.

The control voltage will be formed by frequency variation of the signal at the output of the mixer near the frequency of tuning of the loop. The reference voltage,



Fig. 9.12. Circuit of a frequency discriminator with frequency variation: 1 - mixer; 2 - tunedloop; 3 - square-law detector; 4 - stage ofmultiplication; 5 - generator of reference voltage; 6 - filter; $7 - \text{controlled local oscil$ $lator.}$ varying the frequency of the local oscillator, and consequently also of the signal at the input of the loop, may have different laws of chapte. Most frequently this voltage has a square form, so that it constitutes a sequence of positive and negative pulses of identical amplitude, of duration $T_{\pi}/2$, with period T_{π} . As a result frequencies of the local oscillator and the signal at the input of the loop deviate from face values in successive halfperiods by $\pm \frac{\Delta \omega}{2}$, respectively. With such form of reference voltage the element multiplying output voltage of the detector by the reference signal is simply a commutation switch.

To ensure normal work of the circuit the frequency of pulses of reference voltage should be great as compared to the effective bandwidth of the tracking speed meter. At the same time it should be small as compared to bandwidth of the loop of the discriminator.

In this case output voltage of the discriminator $u_{\mu}(t, \delta)$ can be found if we present the signal at the input of the depicted circuit in the form

$$u_{1}(l) = E(l) \cos[(\omega_{r} + \omega_{o}) l + \varphi(l)], \qquad (9.4.28)$$

where

E(t) and $\phi(t)$ - random functions, where

 $E(t) \cos \varphi(t)$ and $E(t) \sin \varphi(t)$ - independent normal random processes with spectral density $S_0(\omega) + 1/h$;

 $m_{\rm p}$ - center frequency of tuning of the heterodyning oscillator;

 $\omega_0 = \omega_{\rm III} + \delta$ - frequency of tuning of the loop.

Designating the pulse response of the loop $h(t) \cos \omega_{np} t$, for output voltage of the discriminator we can obtain the following expression;

$$u_{\mathrm{A}}(t, \delta) = f(t) \int_{-\infty}^{t} \int_{-\infty}^{t} h(t - \tau_{1}) h(t - \tau_{2}) E(\tau_{1}) E(\tau_{2}) \times \\ \times \cos \left[\left(\frac{\Delta \omega}{2} + \delta \right) (\tau_{1} - \tau_{2}) + \varphi(\tau_{1}) - \varphi(\tau_{2}) \right] d\tau_{1} d\tau_{2} - \\ - f\left(t - \frac{T_{\pi}}{2} \right) \int_{-\infty}^{t} \int_{-\infty}^{t} h(t - \tau_{1}) h(t - \tau_{2}) E(\tau_{1}) E(\tau_{3}) \times \\ \times \cos \left[\left(\frac{\Delta \omega}{2} - \delta \right) (\tau_{1} - \tau_{3}) - \varphi(\tau_{1}) + \varphi(\tau_{3}) \right] d\tau_{1} d\tau_{3}, \qquad (9.4.29)$$

where $f\left(t\right)$ - periodic function with period $T_{_{\rm H}}$, determined as

$$f(t) = \begin{cases} 1 & nT_{n} < t < \left(n + \frac{1}{2}\right) T_{n}, \\ 0 & \left(n + \frac{1}{2}\right) T_{n} < t < (n+1) T_{n}, \end{cases}$$
(9.4.30)

n - integers.

Considering the inertia of subsequent smoothing circuits, the characteristic of the considered circuit interesting us can be found by averaging in period T_{Π} the results of averagings of the set, connected with use of (9.4.29). Then for

the gain factor and equivalent spectral density of the discriminator it is possible to write

$$K_{\mathrm{R}} = \frac{1}{T_{\mathrm{R}}} \int_{0}^{T_{\mathrm{R}}} \frac{\partial u_{\mathrm{R}}(t, \theta)}{\partial \theta} \Big|_{\theta=0} dt =$$

$$= \frac{1}{4\pi} \int_{-\infty}^{\infty} |H(t \cdot \theta)|^{2} \left[S_{0}' \left(\omega + \frac{\Delta \omega}{2} \right) - S_{0}' \left(\omega - \frac{\Delta \omega}{2} \right) \right] d\omega, \qquad (9.4.31)$$

$$S_{\Theta RB} = \frac{1}{T_{\pi}} \int_{0}^{\pi} \frac{1}{K_{\pi}^{2}} \int_{-\infty}^{\infty} [\overline{u_{\pi}(t, 0)} u_{\pi}(t + \tau, 0) - - -\overline{u_{\pi}(t, 0)} u_{\pi}(t + \tau, 0)] d\tau dt =$$

$$= \frac{1}{K_{\pi}^{2}} \frac{2}{\pi} \int_{-\infty}^{\infty} \left\{ |H(i\omega)|^{2} \left[S_{0} \left(\omega - \frac{\Delta \omega}{2} \right) + \frac{1}{h} \right] \right\}^{2} d\omega, \qquad (9.4.32)$$

where as before $H(i\omega)$ - Fourier transform of h(t).

For a spectral density of the signal $S_0(\omega)$ of form (9.3.10) and frequency response H(i ω) of form (9.4.7) calculation by formulas (9.4.31) and (9.4.32) gives

$$K_{\rm A} = k \frac{\alpha (1 + \alpha) \mu}{[\mu^{2} + (1 + \alpha)^{2}]^{2}} \qquad (9.4.33)$$

and_

$$S_{aBB} = \frac{2\Delta f_{a} \left[\mu^{2} + (1+\alpha)^{2}\right]}{a\mu^{2} \left(1+\alpha\right)^{2}} \left\{ (1+\alpha) \left[\alpha + \mu^{2} \left(1-\alpha\right)^{3} + (1+\alpha)^{2} \left(\alpha + (1+\alpha)^{2}\right)\right] + \frac{2}{h} \left[\mu^{2} + (1+\alpha)^{2}\right] \left[\mu^{2} + (1+\alpha)^{2} \left(1+\alpha\right)^{2} \left(1+2\alpha\right)\right] + \frac{1}{h^{2}} \left[\mu^{2} + (1+\alpha)^{2}\right]^{2} \right\}, \qquad (9.4.34)$$

where $\alpha = \Delta f_{\rm K} / \Delta f_{\rm C}$; $\mu = \Delta \omega / 4 \Delta f_{\rm C}$.

The transmission factor of the discriminator here depends on parameters of the circuit just as in the circuit with detuned loops; however, the equivalent spectral density has another magnitude. Investigation of formula (9.4.34) and the graphic presentation of the dependence of S_{3HB}/S_{OBT} on h for different α and μ , shown in Fig. 9.13, show that the spectral density of fluctuations at the output of the given form of discriminator considerably exceeds that for the optimum discriminator. This occurs for any combinations of parameters $\Delta \omega$; Δf_{μ} ; Δf_{c} and h.

In distinction from the circuit with detuned loops, as $\Delta \omega \rightarrow 0$ (with simultaneous increase of amplification in the circuit) the magnitude of S_{OKB} for a circuit with frequency variation grows proportionally to $1/\mu^2$. Increase of frequency deviation $\Delta \omega$ also leads to growth of S_{OKB} , proportional to $(\Delta \omega)^6$. With increase of h, if



Fig. 9.13. Dependence of S_{ator}/S_{CH1} on the signal-to-noise ratio h for a discriminator with frequency variation: $\mu = 0.3$; --- $\mu = 1$; --- $\mu = 5$.

 $\Delta t_{\rm K}^*/\Delta f_{\rm C}^*$ and $\Delta f_{\rm K}^*/\Delta \omega$ are great, this equality is satisfied:

$$\lim_{h \to \infty} S_{\mu \kappa a} = \frac{32\Delta f_{\kappa}^{*}}{\Delta f_{\kappa} \Delta \omega^{*}}, \qquad (9.4.35)$$

i.e., for large signal-to-noise ratios the spectral density of fluctuations at the output of a discriminator with frequency variation will be in $40(\omega f_R/\omega f_C^{-1})^2 \times$ $\times (\omega f_R/\omega)^2$ greater than in an optimum discriminator and all discriminators previously considered.

Thus, the considered circuit gives sharp worsening of results as compared to the preceding circuits. The physical explanation of nonoptimality of the discriminator with frequency variation is

the following. Signals in various half-periods $T_{\rm m}/2$ are uncorrelated, since for normal work of the circuit there should be satisfied the relationship $T_{\rm m}/2 \gg 1/\Delta r_{\rm c}$. Therefore, any averaging of results of sequential amplitude detection of signals in various half-periods cannot compensate fluctuations, whereas in the previous circuits such compensation occurred.

Considering nonoptimality of the considered circuit, we shall not investigate it in more detail. In particular, we shall not calculate the coefficient of parametric fluctuations $S_{\rm max}$.

It is necessary to note that a circuit with frequency variation is sensitive to interferences modulated by the frequency of switching c its odd harmonics. In this respect the given discriminator is analogous to goniometric discriminators with conical or square scanning, which will be considered in detail in subsequent chapters. Therefore, we will not discuss this question in d-tail here.

9.4.5. Inaccuracy of Reproduction of Modulation of the Signal During Reception

During the analysis of different frequency discriminators we assumed that in

-432-

them there is performed exact demodulation in phase (frequency) and exact mutiplication by a function expressing the law of amplitude modulation of signal, taking into account the true value of its delay (matched gating in the case of a pulse signal). In real radar receivers exact fulfillment of these functions is impossible. Besides technical tolerances on laws of gating and change of frequency of heterodyne oscillators, here more essential circumstances play a role. Thus, delay of the function expressing the law of modulation of the signal reproduced during reception never coincides exactly with true delay of the reflected signal due to the presence of errors of range finders; during gating we often specially apply gate pulses which are longer than the pulse of the signal, proceeding from considerations connected with breakoff of tracking during intense noises, etc.

We shall show that mismatch of laws of modulation of the reflected signal and of signals used for processing the received signal in the receiver leads simply to decrease of the signal-to-noise ratio. Actually, if in the receiver as laws of amplitude and phase modulation of the signal we use functions $u_{a1}(t - \tau - \Delta)$ and $\psi_1(t - \tau - \Delta)$ instead of functions $u_a(t - \tau)$ and $\psi(t - \tau)$, the signals at the input of the narrow-band filters of the discriminators have the form

$$\boldsymbol{v}(t) = \frac{1}{2} \nabla \overline{P_{\sigma}} \boldsymbol{E}(t) \boldsymbol{u}_{a}(t-\tau) \boldsymbol{u}_{ai}(t-\tau-\Delta) \cos[\boldsymbol{\omega}_{np}t + \frac{1}{2} \boldsymbol{\psi}(t-\tau) - \boldsymbol{\psi}_{i}(t-\tau-\Delta) + \boldsymbol{\varphi}(t)] + \frac{1}{2} \boldsymbol{\psi}(t-\tau-\Delta) \boldsymbol{n}(t) \cos[\boldsymbol{\omega}_{r}t - \boldsymbol{\psi}_{i}(t-\tau-\Delta)], \qquad (2.4.36)$$

where

 Δ - time shift caused by inaccuracy of range tracking;

- τ delay of the function describing the law of modulation of the reflected signal;
- E(t) and $\varphi(t)$ random functions representing fluctuations of the reflected signal;

n(t) - white noise with spectral density N_0 ;

P_c - mean signal power.

We assume the following normalization of signals:

$$\overline{E^{\mathbf{a}}(t)} = 1, \frac{1}{T_r} \int_{0}^{t} u_a^2 (t-\tau) dt = 1,$$

where $T_r = period$ of functions $u_a(t)$, $\psi(t)$, $u_{a1}(t)$, $\psi_1(t)$.

Due to narrow-bandedness of filters of the discriminators through them there only pass components of the signal spectrum which are concentrated near frequency $\omega_{\rm RP}$. Therefore, according to the character of their influence on the discriminator, signals with periodic modulation are equivalent to continuous unmodulated signals. As it was shown in Chapter IV, the signal-to-noise ratio for such an equivalent

-433-

unmodulated signal with ideal matching of $u_{a1}(t)$ with $u_a(t)$ and $\psi_1(t)$ with $\psi(t)$ is equal to $h = P_c/2\Delta f_c N_D$. With imperfect matching of these functions at the input of the narrow-band filter there acts v(t), determined by (9.4.36). Averaging the result of passage of v(t) through the filter and calculating the power of the signal and of the components, we find the new signal-to-noise ratio

$$h_{1} = h \frac{\left| \frac{1}{T_{r}} \int_{0}^{T_{r}} u_{a} \left(t - \tau \right) u_{a1} \left(t - \tau - \Delta \right) e^{t \left[\psi \left(t - \tau \right) - \psi_{1} \left(t - \tau - \Delta \right) \right]} dt \right|^{a}}{\frac{1}{T_{r}} \int_{0}^{T_{r}} u_{a1}^{2} \left(t - \tau - \Delta \right) dt}$$
(31.4.37)

All formulas obtained for the above-considered discriminators remain valid upon replacement of h_1 for h. It should be noted that in such an approach we consider assigned both the form of functions $u_{a1}(t)$ and $\psi_1(t)$, and also quantity Δ . By virtue of the random nature of the output of the range finder, Δ , in general, randomly varies in time. It is possible, however, to fix some, for instance, the worst value of Δ and, by calculating the corresponding quantity h_1 , to find how much accuracy of measurement of speed will decrease.

For illustration of magnitudes of change of the signal-to-noise ratio we shall consider two examples.

1. Let us assume that the signal is modulated only in amplitude by square pulses of duration τ_{N} , and that the gale pulse also has square form and duration $\tau_{C} \geq \tau_{N}$. From (9.4.37) we obtain

$$h_{1} = \begin{cases} h \frac{\tau_{u}}{\tau_{0}} & \text{at } \Delta \leq \frac{\tau_{0} - \tau_{u}}{2} \\ h \frac{(\tau_{u} + \tau_{0} - 2\Delta)^{2}}{4\tau_{u}\tau_{0}} & \text{at } \frac{\tau_{0} - \tau_{u}}{2} \leq \Delta \leq \frac{\tau_{0} + \tau_{u}}{2}. \end{cases}$$
(9.4.38)

This relationship shows the power loss connected with increase of duration of the gate as compared to τ_{μ} . Corresponding loss in spectral density $S_{\rm SMP}$, and concequently, in accuracy of measurement of speed can be estimated from the preceding formulas and graphs.

2. Let us assume that the signal is modulated in frequency by a sinusoidal law with frequency Ω and deviation $\omega_{\rm m}$, and heterodyne voltage has different frequency deviation $\omega_{\rm mi}$. Then, with zero mismatch Δ calculations by (9.4.37) give

$$h_1 = h J_0^2 \left(\frac{\boldsymbol{\omega}_m - \boldsymbol{\omega}_{m1}}{\boldsymbol{\Omega}} \right), \qquad (9.4.39)$$

where $J_{O}(x)$ - Bessel function of zero order.

For sufficiently large values of $\omega_{\rm m} - \omega_{\rm m1}/\Omega$ the signal-to-noise ratio rapidly drops, and for certain values of $\omega_{\rm m} - \omega_{\rm m1}/\Omega$ the signal of the mixer, in general, is absent. From relationship (9.4.39) there ensues the expediency of selection of a low frequency of modulation Ω at which a comparatively large difference in deviations $\omega_{\rm m}$ and $\omega_{\rm m1}$ does not lead to substantial increase of errors of measurement.

9.4.6. Comparison of Circuits of Discriminators

As follows from the conducted analysis, the first three of the four considered forms of discriminators with proper selection of their parameters give results very close to an optimum discriminator. The fourth circuit, with frequency variation of the heterodyne oscillator, is the farthest from optimum. It gives substantially less accuracy of measurements, is sensitive to easily created interferences with periodic modulation and, furthermore, on it there must be stringent requirements of identity of the positive and negative half-periods of the reference voltage. Nonfulfillment of these requirements and also instability of the frequency of tuning of filters lead to the appearance of systematic errors of measurement of speed. By virtue of these circumstances it is doubtful whether one should recommend application of this scheme of a frequency discriminator, if, of course, requirements on accuracy of measurement are rather high.

The circuit with detuned loops ensures accuracy of measurements close to the potential (β_{300} is close to S_{OET}), even if we select detuning of loops proceeding from the condition of a maximum gain factor of the discriminator K_{II} . However, with small variation of parameters of the circuit its errors can increase rather considerably, i.e., among deficiencies of the circuit are criticality with respect to the parameters. One should note especially that with discrimination at sufficiently high frequencies it is necessary to have two filters, tuned to close, but, in principle, different frequencies. Instability of these frequencies can lead here to appearance of systematic error of measurement, which is characteristic for other circuits, to substantial change of the fluctuation error of measurement and even to cessation of work of the discriminator if the frequencies of tuning become identical er, conversely, go too far apart. Therefore, although in principle the circuit with detuned here is also close to optimum, it is not always possible to recommend its use.

The most acceptable of those considered for proctical application are chronics of

-435-

discriminators with a tuned loop and a phase shifter and those with mixers and differentiation. These circuits are identical in their characteristics, and with correct selection of parameters, in particular the bandwidth of the loops, they give results sufficiently close to an optimum discriminator. In particular, equivalent spectral density for them does not exceed S_{ORT} by more than 20%. The influence of instability of parameters for such circuits is considerably less than for the ones discussed above. Choice between these two circuits should be made from considerations of technical convenience. The circuit with a tuned loop and a phase shifter in principle is less bulky, but requires creation of a narrow-band (in most cases, quartz) filter. The circuit with mixers and differentiation does not possess this deficiency; during its realization it is not necessary to think about stability of tuning of the loop, but it is sensitive to inaccuracy of the phase shifter of the heterodyne voltage. Furthermore, this circuit contains two filters instead of the one in the circuit with a tuned loop.

We can, of course, use other schemes of frequency discriminators. Their qualities should be compared with qualities of the optimum circuit to find the expedienc. I their application.

Let us discuss briefly selection of the transmission band of filters in the two best circuits. Usually meters on which there are requirements of high accuracy work at a high signal-to-noise ratio h. Here, the selected transmission band should be changed with change of h and depending upon the width of the spectrum of the signal. However, in practice neither the signal-to-noise ratio nor the width of the spectrum of the signal are ever known exactly. Therefore, it is necessary to select bandwidth of the filters proceeding from the worst case, considering that spectrum of the signal. Considering these circumstances, it is necessary to select filter bandwidth from the highest possible width of the spectrum of the signal and from a minimum signal-to-noise ratio, occurring at the maximum range, determined by conditions of lock-on and transition to automatic tracking. With such selection of bandwidth of the filters with decrease of distance and with narrowing of the width of the spectrum of the signal the system of measurement of speed will be further from the optimum than under the assumed conditions; however, its error will always be within permissible limits.

Often from technical considerations it is necessary to expand the bandwidth of the filters with respect to its optimum value, since the signal spectrum has a

-436-

2 Er

width of the order of units of cycles per second or even fractions of cycles per second, and stabilities of frequencies obtainable relatively simple are such that the bandwidth of filters cannot be made smaller than ten cycles per second. Usually this does not lead to essential impairment of qualities of the discriminator.

§ 9.5. Smoothing Circuits and Accuracy of Measurement of Speed

Studying properties of discriminators of Doppler speed meters, we turn to estimation of accuracy of these meters. For this it is necessary to assign the second component part of the tracking meter — the smoothing circuits. Very often there are applied linear smoothing circuits with constant parameters of certain very simple forms. Let us consider, first of all, error of speed meters with such smoothing circuits. This can be done by applying formulas of § 6.2, determining fluctuation, dynamic and systematic errors of tracking meters. We shall specially discuss cases in which optimum smoothing circuits have constants parameters, and we shall estimate accordingly accuracies of optimum speed meters. However, in other cases optimum smoothing circuits, as it was shown in Chapter VI, have variable parameters. Therefore, we shall also investigate accuracy of speed meters with variable parameters of their smoothing circuits.

9.5.1. Smoothing Circuits with Constant Parameters

In speed meters there are applied smoothing circuits of the same forms as in range finders. Therefore, many formulas and results of § 7.10 (Paragraph 7.10.2) remain in force for speed meters.

If we disregard parametric fluctuations, fluctuation error of the meter is expressed as before by formula

$$\sigma_{\phi \pi}^{2} = 2S_{\phi \pi}\Delta f_{\phi \phi}, \qquad (9.5.1)$$

in which $S_{\partial KB}$ is determined by the above-mentioned analysis of frequency discriminators, and Δf_{ab} - effective bandwidth of the closed tracking speed meter.

As it was shown in § 7.10, for a smoothing circuit in the form of an integrator

$$\Delta f_{i\phi} = \frac{K_{e}K_{A}}{4}, \qquad (9.5.7)$$

where K_{μ} - gain factor (dimensional) of the integrator;

 K_{π} - discriminator gain factor.

For an RC-filter

$$\mathbf{M}_{\bullet \phi} = \frac{\mathbf{K} \mathbf{K}_{\mathbf{A}}}{4 \mathbf{T}_{\mathbf{I}}}, \qquad (9.5.3)$$

-437-

where $T_1 = RC$, and K = gain factor.

For a smoothing circuit in the form of a double integrator with correction

$$\Delta f_{00} = \frac{1 + K K_{\rm A} T_{\rm A}^2}{4 T_{\rm A}}, \qquad (9.5.4)$$

where K - gain factor of the two integrators;

 T_{ν} - time constant of the correcting network.

The value of Δf_{ab} reaches a minimum, equal to $\sqrt{KK_{II}}/2$, at $T_{H} = 1/\sqrt{KK_{II}}$.

Finally, if the smoothing filter consists of two RC-circuits (inertial links) with correction

$$\Delta f_{100} = \frac{KK_{\pi}T_{\pi}}{T_{1}T_{\pi}}, \qquad (9.5.5)$$

where T_1 and T_2 - time constants of the inertial links.

In the last case the minimum of $\Delta f_{\partial \Phi}$ is reached at $T_{K} = \sqrt{T_1 T_2/2KK_{II}}$ and is

$$\Delta I_{00} = \frac{3\sqrt{2KK_{\pi}}}{8\sqrt{T_{1}T_{0}}}.$$
 (9.5.6)

If we select the mirimum effective bandwidth in the absence of noise, the dependences of the effective bandwidth on the signal-to-noise ratio h, determined by formulas (7.10.19), (7.10.14) and (7.10.10), also remain in force. Then with use of an optimum discriminator [with the simplest form of spectral density of fluctuations (9.3.10)] for smoothing circuits of the first order we have

$$\sigma_{\phi_{\#}}^{3} = 4\Delta f_{e} \Delta f_{\phi_{\phi}} \frac{\sqrt{1+h}\left(1+\sqrt{1+h}\right)^{s}}{h} \frac{1}{h+y}, \qquad (9.5.7)$$

where $\Delta f_{\partial \Phi} = 0$ - value of the effective bandwidth of a tracking speed meter in the absence of noises, and

$$y = \frac{\Delta l_y}{\Delta l_0} \tag{9.5.8}$$

is the ratio of bandwidth of the amplifier covered by the AGC loop to the signal bandwidth. In the case of application of a smoothing circuit in the form of a double integrator with correction

$$\mathbf{e}_{\phi,\eta}^{2} = 2\Delta f_{\phi} \Delta f_{\phi\phi\phi} \frac{\sqrt{1+h} \left(1+\sqrt{1+h}\right)^{2}}{h^{2}} \left(1+\frac{h}{h+y}\right)^{4} \left(1+\frac{h}{h+y}\right)^{4}$$
(9.5.9)

Curves of the dependence of variance of fluctuation error on the signal-tonoise ratio h for various y are shown in Fig. 9.14. Consideration of them leads to the conclusion that with the selection of an identical effective bandwidth in the absence of noise Δf_{pot} 0 flucuation error with noise in the case of application of filters of the 2nd order is greater than with application of filters of the 1st order. This is explained by the varying dependence of the effective bandwidth of the system on the gain factor of the discriminator in the two considered cases.



Fig. 9.14. Dependence of variance of fluctuation error of a speed meter $\sigma_{\rm full}^2$ on the signalto-noise ratio h during application of an optimum discriminator and smoothing circuits with constant parameters: --- for circuits of the 1st order; --- for circuits of the 2nd order.

For smoothing circuits of the 1st order there is noticeable dependence of fluctuation error on selection of amplifier bandwidth Δf_y (or y). With broadening of this band error decreases due to increase of intensity of noise at the output of the amplifier, suppression of the gain factor, and consequently also of the effective bandwidth of the system. For smoothing circuits of the 2nd order the dependence of fluctuation error on Δf_v is immaterial.

Increase of fluctuation errors with application of nonoptimal discriminators is easily found by multiplication of the obtained values of σ_{en}^2 by ratio S_{BKE}/S_{ONT} , found for various discriminators in the preceding paragraph.

It should be noted that from consideration of Fig. 9.14 one should not make conclusions about the advantages of filters of the 1st order, since it is also necessary to consider other components of errors of measurement (dynamic and systematic errors), which turn out to be larger for smoothing circuits of the 1st order.

Passing to consideration of dynamic and systematic errors, let us note that all results obtained in Chapter VII for systems of range finding remain valid for speed meters. In particular, considering that speed of the target has a component which varies randomly, and assuming that this component is a stationary random process with spectral density $S(\omega)$, for dynamic error, according to (7.10.20), we have formula

-439-

$$\sigma_{A=0}^{2} = \frac{1}{2\pi} \int_{|1+K_{A}H(i\omega)|^{2}}^{\infty} (9.5.10)$$

where K_{μ} - transmission factor of the discriminator; H(iw) - frequency response of the smoothing circuit. With a very simple form of spectral density of speed

$$S(\mathbf{e}) = \frac{3\mathbf{e}_0^2 T}{1 + \mathbf{e}^2 T^2}$$
(9.5.11)

(where σ_0^2 - variance of the random component of speed, and T - time of correlation of the process of its change) formulas (7.10.22) and (7.10.23) for $\sigma_{\rm L}^2$ turn out to be valid, from which, in particular, it follows that dynamic error increases with decrease of the effective bandwidth of the tracking meter, taking place with decrease of the signal-to-noise ratio h.

If the measured speed varies as a linear combination of assigned functions with unknown coefficients

$$\mathbf{V}(l) = \sum_{i=1}^{n} \mu_i f_i(l) + \overline{V(l)}, \qquad (9.5.12)$$

where $\overline{\mu}_1 = 0$, $\overline{\mu_1 \mu_k} = M_{1k}$, $\overline{V(t)}$ — mathematical expectation of speed, according to (7.10.25) we have

Here

•(1) =
$$\int \Phi(t, \tau) f_1(\tau) d\tau$$
, (9.5.14)

 $b(t, \tau)$ is determined by equation (6.2.6), and by virtue of constancy of parameters of smoothing circuits the Fourier transform of $b(t - \tau)$ is found as

$$= \frac{1}{1 + K_{\rm A} H \, (lm)} \,. \tag{9.5.15}$$

With a nonstatistical approach to the question of changes of speed dynamic error can be obtained if in (9.5.13) we set $M_{1k} = 1$. Then, using as the smoothing circuit an integrator and with change of speed according to the law $V(t) = v_0 + v_1 t$ dynamic error as $t \rightarrow \infty$ we determine as

$$e_{\text{guss}} = \frac{v_1}{K_{\text{g}}K_{\text{g}}}.$$
 (9.5.16)

Analogously, with smoothing circuits with two integrators and $V(t) \approx v_0 + v_1 t + v_2 t^2$,

 $a_{ABB} = \frac{\sigma_{B}}{K_{A}K}$

Dynamic error is inversely proportional to the gain factor of the open circuit of the tracking speed meter and increases with decrease of the signal-to-noise ratio.

Dynamic errors in the nonstatistical approach to measurement of speed are determined in the same manner as systematic errors, so that it is possible to consider we have simultaneously discussed the method of finding the latter.

By virtue of the different dependences of fluctuati and dynamic errors on the gain factor, and thus on the signal-to-noise ratio h, as, too, for range finders, a compromise selection of the gain factor of the open loop is reasonable. Let us consider two examples of such selection.

1. Let us assume that the smoothing circuit is an integrator, and that the speed of the target varies as $V(t) = v_0 + v_1 t$. Then, the square of total error of measurement of speed in accordance with (9.5.1) and (9.5.16) is

$$\sigma^{3} = \sigma_{\phi \pi}^{2} + \sigma_{A \pi \pi}^{2} = \frac{K_{\pi} K_{A}}{2} S_{\phi K B} + \frac{\sigma_{1}^{2}}{(K_{\pi} K_{A})^{3}}.$$
 (9.5.18)

In the statistical approach to dynamic error v_1^{\succeq} must be replaced by \overline{v}_1^2 . Minimizing σ^2 by selection of $K_{\mu}K_{\mu}$, we obtain the optimum value of the gain factor of the open loop

$$(K_{\rm L}K_{\rm A})_{\rm OHT} = \sqrt[3]{\frac{4\sigma_{\rm i}^2}{S_{\rm OHT}}}, \qquad (9.5.19)$$

at which total error is

$$\sigma^{*} == 1,19 \left(v_{1} S_{PKB} \right)^{*/}. \tag{9.5.20}$$

Substituting in (9.5.19) the dependence of $S_{\Theta KB}$ on h, it is easy to find how the gain factor of the open loop should be changed with change of the signal-tonoise ratio. In particular, for an optimum frequency discriminator and spectral density of fluctuations of the signal of form (9.3.10)

$$(K_{\rm H}K_{\rm H})_{\rm OHY} = K_0 \sqrt[3]{\frac{h^2}{\sqrt{1+h}(1+\sqrt{1+h})^2}},$$
 (9.5.21)

where K_{ij} - the value of the gain factor when $h \rightarrow \infty$, i.e., in the absence of noise. The dependence of the optimum gain factor on h is shown in Fig. 9.15.

2. Let us consider a smoothing circuit in the form of a double integrator with correction and the case of a square-law change of speed. The square of total error in this case is determined by expression (7.10.33) and

-441-

$$(\mathcal{K}\mathcal{K}_{R})_{OXY} = \left(\frac{4\sigma_{\mathcal{I}}^{2}}{S_{BRR}}\right)^{1/s},$$

so that with optimum gain

$$\bullet^{\bullet} = 1,68\sigma_2^{*}S_{\bullet^{H_{\bullet}}}^{*}$$

In the statistical approach to dynamic error in the formulas instead of v_2^2 there enters \overline{v}_2^2 . With the same dependence of S_{3KB} on h as in the preceding case the optimum gain factor should vary with change of the signal-to-noise ratio as

$$(KK_{k})_{env} = K_{e} \left[\frac{h^{2}}{V^{1+h}(1+V^{1+h})^{e}} \right]^{V_{e}}.$$
 (9.5.24)

This change is also presented in Fig. 9 15, where it is easy to see that dependences of the gain factor on the signal to-noise ratio for smoothing circuits in the form of one integrator and of two segrators with correction are very close.



Fig. 9.15. The dependence of the gain factor of the open loop K_p on the signalto-noise ratio h: 1 - optimum gain for circuits of the lat order; 2 - optimum gain for for a circuit of the 2nd order. --- change of gain from the circuit of automatic gain control. In Fig. 9.15 there are also given curves of the dependences on h of the gain factor of the open loop, obtained due to the normalizing action of the AGC system. These curves are constructed on the assumption that in the absence of noise $(h \rightarrow \infty)$ the gain factor is equal to the optimum. From the figure it is clear that when h > 1 and we have small ratios $\cup f$ bandwidths of the amplifier covered by automatic gain control and of the signal (y = 2 to 3) necessary change of the gain factor with sufficient accuracy is provided by the AGC system. For small h the system of automatic gain control leads to unnecessary lowering of the gain

(9.5.22)

(9.5.23)

factor. Thus, in spite of the fact that with decrease of y fluctuation error grows (see Fig. 9.14), from the point of view of providing minimum total error it is necessary to select sufficiently small values of y.

The obtained relationships permit us to find error of speed meters under

-442-

conditions when it is possible to disregard parametric fluctuations. However in a number of cases such disregard is impermissible, since the spectral density of parametric fluctuations, as we have seen, can attain for the considered forms of frequency discriminators considerable magnitudes. Therefore, calculation of parametric fluctuations is of interest.

In accordance with formula (6.2.44) variance of total error during calculation of parametric fluctuations is defined as

$$\sigma^{\mathbf{a}} = \sigma_0^2 \left(1 + 2S_{\pi \mathbf{a} p} \Delta f_{\mathbf{a} \mathbf{b}} \right), \qquad (9.5.25)$$

where σ_0^2 - the variance of error in the absence of parametric fluctuations, found above.

Considering that for frequency discriminators $S_{\Pi ap} = \varkappa / \Delta f_c$, where $\varkappa - a$ coefficient, varying in a wide range depending on the form and parameters of the frequency discriminator, and also on the magnitude of the signal-to-noise ratio h, we obtain

$$\sigma^{a} = \sigma_{0}^{2} \left(1 + 2 \times \frac{4f_{*0}}{4f_{*}} \right). \tag{9.5.26}$$

For large signal-to-noise ratios $(h \rightarrow \infty)$ coefficient π takes the following value:

- for an optimum discriminator n = 0.5;

- for a discriminators with a tuned loop and a phase shifter, and also with mixers and differentiation \varkappa varies from 0.5 to 1 depending up $\alpha = \Delta f_{\mu}/\Delta f_{c}$:

- for a discriminator with detuned loops \varkappa varies in the same range, depending upon the magnitude of detuning.

Therefore, with rapid fluctuations, when ratio $\Delta f_{e\bar{\Phi}}/\Delta f_e$ is small (it may attain magnitudes of the order of 10^{-3}), there are conditions in which parametric fluctuations can be disregarded. With slower fluctuations and for smaller h, at which π grows; these conditions may not be observed. It is necessary, however, to recall that with $2\pi \frac{\Delta f_{e\bar{e}}}{\Delta f_e}$ comparable with unity formula (9.5.26) ceases to be valid. Therefore, it gives the possibility of obtaining corrections due to parametric fluctuations only under the condition of smallness of these corrections, and the basic meaning of its application consists in determining conditions in which linear approximations are still valid.

Certain Lowering of the influence of parametric fluctuations occurs from the AGC system. However, for Doppler speed meters this lowering is no longer found by formulas of Chapter VII inasmuch as parametric fluctuations (during rapid

-443-

fluctuations of the signal) are determined here not only by the random nature of changes of amplitude, but also by the random nature of changes of phase, not compensated by the AGC system.

9.5.2. Optimum Smoothing Circuits for Randomly Varying Speed Linear smoothing circuits with constant parameters in certain cases analyzed in Chapter VI are optimum, i.e., ensure with application of the proper discriminater minimum error of measurement. This occurs when speed is either a stationary random process, or a random process with stationary increments.

Speed is a stationary random process, for instance, in the case of measurement of groundspeed by a Doppler dead-reckoner. With stationary random oscillations of the flying object relative to the horizontal line of flight, small stationary changes of its speed and sufficiently uniform relief of the locale, the measured components of groundspeed, just as the reflected signal, can be considered stationary processes.

The method of finding both the optimum filter and also errors of measurement obtained during its application is presented in § 6.8 for an arbitrary form of spectral density of the measured parameter. Let us consider the particular case of spectral density of speed of the form of (9.5.11), for which the frequency response of the optimum smoothing filter for a large time of observation, according to (6.8.11), has the form

$$G(i\omega) = \frac{2\sigma_0^2 T}{1 + \sqrt{1 + 2\sigma_0^2 T K_{out}}} \frac{1}{1 + i\omega T},$$
(9.5.27)

where σ_0^2 -- variance;

T - time of correlation of the process of change of speed;

 K_{curv} - gain factor of the optimum discriminator.

Variance of the total error of measurement of speed according to (6.8.13) is equal to

$$\sigma_{\text{ours}}^{2} = \frac{2\sigma_{0}^{2}}{1 + \sqrt{1 + 2\sigma_{0}^{2}TK_{\text{ours}}}}.$$
(9.5.28)

Considering that in (9.5.28) there is implied application of an optimum frequency discriminator, and considering that its equivalent spectral density is determined by expression (9.3.11), we have

$$K_{onv} = \frac{1}{S_{ouv}} \left(\frac{4\pi}{\lambda}\right)^{2} = \left(\frac{4\pi}{\lambda}\right)^{2} \frac{1}{2\Delta f_{o}} \times \frac{\hbar^{2}}{\sqrt{1+\lambda}\left(1+\sqrt{1+\lambda}\right)^{2}},$$

$$(9.5.29)$$

whence

$$\sigma_{g_{MX}}^{2} = \frac{2\sigma_{0}^{2}}{1+\sqrt{1+\frac{T}{\Delta f_{0}}\left(\frac{4\pi\sigma_{0}}{\lambda}\right)^{2}\frac{A^{2}}{\sqrt{1+\Lambda}\left(1+\sqrt{1+\Lambda}\right)^{4}}}}$$
(9.5.30)

For very small signal-to-noise ratios $(h \rightarrow 0)$ quantity σ_{BHX}^2 is equal to the a priori variance of change of speed; for large signal-to-noise ratios $(h \rightarrow \infty)$

$$q_{aux}^{2} = \frac{2q_{0}^{2}}{1+\sqrt{1+\frac{T}{\Delta f_{0}}\left(\frac{4\pi a_{0}}{\lambda}\right)^{2}}}.$$
(9.5.31)

The dependence of the mean square error of measurement of speed on the signalto-noise ratio we shall show with an example. Considering $\sigma_0 = 10 \frac{\text{m}}{\text{sec}}$, $\Delta f_c = 1000$ cps, T = 10 sec, and $\lambda = 4$ cm, for high-level noises $(h \rightarrow 0)$ we obtain $\sigma_{BHX} = 1.0 \frac{\text{m}}{\text{sec}}$; for low-level noises $(h \rightarrow \infty)$ we obtain $\sigma_{BHX} \approx 0.8$ m/sec; the dependence of σ_{BHX} on h is shown in Fig. 9.16.

It is necessary to note that the obtained values of error of measurement of speed correspond to an optimum system, the gain factor of whose open loop should be equal to

$$K_{\rm p} = \frac{2\sigma_0^2 T K_{\rm out}}{1 + \sqrt{1 + 2\sigma_0^2 T K_{\rm out}}}$$
(9.5.32)

and takes different values with change of the signal-to-noise ratio h. The graph of the dependence of K_n on h for the given example is shown in Fig. 9.17.



Fig. 9.16. Dependence of the mean square error of measurement of speed $\sigma_{\rm BHX}$ on the signal-to-noise ratio h for the example of an optimum meter when speed is a stationary process.

In the same figure there are shown curves of the dependence of the gain factor on h, determined by action of the automatic gain control system for three values of $y = \Delta f_y / \Delta f_c$. From



这些最低的最优的年1月1日日日日 - 150日的外部的 150日的

「「「大学生」「「「「「「「「」」」」



consideration of these curves it may be concluded that satisfactory reproduction of the required law of change of amplification by automatic gain control is attained for values of h = 1 to 100 for y = 4 to 6.

In most cases the assumption of stationary random change of speed is not realized. Thus, in many cases the radar target can at any moment start to maneuver and never return to the past direction of motion, due to which the radial component of speed will change. However, sufficiently often change of speed can be considered a process with stationary increments. In particular, if changes of engine thrust, and consequently also of acceleration of the target are stationary, speed is nonstationary, but has a stationary first derivative, i.e., can be included in processes with stationary increments.

If we idealize the process of change of acceleration, considering it white noise with spectral density $B_1[m^2/sec^3]$, the speed is a Wiener process. For large time of observation the optimum smoothing filter, according to (6.8.28), is an integrator with a gain factor equal to $\sqrt{B_1/K_{OHT}}$, where K_{OHT} — gain factor of the discriminator. The gain factor of the open loop here is defined as $\sqrt{B_1K_{OHT}}$ and should be different for different signal-to-noise ratios h. Thus, for an optimum discriminator in the same conditions as in the preceding case

$$K_{\rm p} = \frac{4\pi}{\lambda} \sqrt{\frac{B_{\rm i}}{2\Delta f_{\rm a}}} x = K_{\rm a} x. \tag{9.5.33}$$

-446-

The magnitude of variance of error of the meter turns out to be equal to

$$\mathbf{g}_{aux}^{2} = \sqrt{\frac{B_{1}}{K_{our}}} = \frac{\lambda}{4\pi} \sqrt{2\Delta \int_{\mathbf{0}} B_{1}} \frac{1}{x}, \qquad (9.5.34)$$

where

$$x = \frac{h}{(1+h)^{1/4} (1+\sqrt[4]{1+h})^{3/4}}.$$
 (9.5.35)

Relationship (9.5.35) is shown in Fig. 9.18. The dotted c we in this figure shows that the required dependence is sufficiently well reproduced by the system of automatic gain control.

In order to illustrate the magnitude of errors of measurement of speed in the case when acceleration can be considered white noise we shall give the following example. Let us assume that wavelength of the radar $\lambda = 3.4$ cm, and the bandwidth



Fig. 9.18. Dependence of $x = K_p / K_0$ on the signal-tonoise ratio h when speed is a process with a stationary 1st derivative: — optimum change of gain; --- change of gain from automatic gain control.

of fluctuations of the signal $\Delta f_c = 30$ cps. Errors of measurement of speed for a large signal-to-noise ratio $(h \rightarrow \infty)$ and for different mean square values of speed developed in 1 sec are given in Table 9.2. In it there are also given corresponding errors of

و ملكول الاست

measurement of frequency and values of the effective bandwidth of the tracking meter, calculated from (9.5.2).

Table 9.2.

Mean square value of speed developed by the target in 1 sec, m/sec	v . 5	5	50
$B_1, m^2/sec^3$	0.25	25	2500
Mean square error of measurement of frequency $\sigma_{f}^{}$, cps	6	19	60
Mean square error of measurement of speed $\sigma_{\rm BHX}$, m/scc	0.2	0.32	1
Effective bandwidth of the system $\Delta f_{\partial \tilde{\Phi}}$, cps	6	60	600

Acceleration of the target cannot always be assumed uncorrelated. However, if acceleration is finite, and its derivative is approximately stationary and changes rapidly, speed can be considered the double integral of white noise with a certain spectral density $B_2[m^2/sec^5]$. For a large time of observation according to (6.8.37) the optimum smoothing filter is a double integrator with correction. The gain factor of the open loop turns out to be equal to

$$K_{p} = \bigvee \overline{B_{s}K_{out}}$$
(9.5.36)

and with the same the optimum frequency discriminator depends in the same way on the signal-to-noise ratio h as in the case of uncorrelated accelerations. The time constant of the correcting circuit is defined as

$$T_{\rm H} = \sqrt[4]{\frac{4}{B_{\rm p}K_{\rm out}}} \tag{9.5.37}$$

and, finally, variance of error of measurement of speed according to (6.8.33) is equal to

$$\sigma_{oux}^{3} = \sqrt{2} \left(\frac{B_{3}}{K_{our}^{3}} \right)^{\eta_{4}}.$$
 (9.5.38)

The dependence of ratio $\sigma_{\rm BHX}^2/\sigma_{\rm BHX}^2$ on h (where $\sigma_{\rm BHX,0}^2$ - variance as h $\rightarrow \infty$) obtained from (9.5.38) is shown in Fig. 9.19. It is valid for change of K_p and $T_{\rm H}^2$ with change of h in accordance with the above-mentioned formulas. In the



Fig. 4.19. Dependence of variance of error of measurement of speed σ^2_{BHX} on the signal-to-noise ratio h when speed is a process with a stationary 2nd derivative.

case of fixed selection of K_p and T_k the system ceases to be optimum.

For illustration of the possibilities of an optimum system of measurement of speed when the derivative of acceleration is white noise, we turn again to the example of a radar for which $\lambda = 3.4$ cm, and the bandwidth of the received signal is $\Delta f_{\rm C} = 30$ cps. Assigning the magnitudes of the mean square value of acceleration developed in 1 sec and calculating mean square error of measurement of speed as $h \rightarrow \infty$, we obtain the results presented in Table 9.3. In this table, as also in the preceding example, there are given values of error of measurement of frequency and the effective bandwidth of a tracking meter, calculated by formula (9.5.4).

-448-

Table 9.3.

Mean square value of acceleration developed by the target in 1 sec, m/sec	0.5	5	50
B ₂ , m ² /sec ⁵	0.25	25	2500
Mean square error of measurement of frequency σ_{f} , cps	3.3	5.8	10.5
Mean square error of measurement of speed o _{BMX} , m/sec	0.056	0.1	0.18
Effective bandwidth of system $\Delta f_{\partial \tilde{q}}$, eps	2.6	8.2	20

From the table it is clear that the dependence of error of measurement or dynamic properties of the target, characterized by coefficient B_2 , in this case is very weak.

9.5.3. Smoothing Circuits with Variable Parameters

In many cases the law of change of speed can be sufficiently approximated by a function of assigned form, out depending on a certain number of unknown parameters. Here, fairly frequently there are satisfied conditions in which approximation (6.3,42) is valid, and the measured speed is presented in the form

$$\mathbf{V}(l) = \overline{\mathbf{V}(l)} + \sum_{k=1}^{n} \mu_k f_k(l), \qquad (9.5.35)$$

where V(t) - machematical expectation of the law of enange of speca;

 $r_{\chi}(z)$ - assigned functions;

 μ_{μ} - normally distributed random variables

$(\overline{\mu}_h = 0, \ \overline{\mu}_i \mu_h = M_{ih}).$

An it was shown in Chapter VI, optimum amouthing circuits in such cases possess variable parameters, and their pulse responses in the plane - we variable and error of menourement are determined by expressions (0.5.6) and (0.7.6), respectively. By virtue of the fact that these capressions include the gain factor of the disperiments $K_{\rm BL}(h)$ merameters of the smoothing circuits depend on the signal-to-

 $u_{10} + u_{10} = 1$

$$V(t) = V(t) + \mu t(t).$$

····)
Considering $\mu^2 = \sigma_0^2$ we have for the pulse response of the smoothing circuit according to (6.8.49)

E (<i>l</i> , =) ==	(9.5.41)	
a (0=	€0 ² f² (ℓ)	
1+Kor	$= \sigma_0^2 \int_{z_0}^{z_0} f^2(s) ds$	(9.5.42)

Let us consider the following examples.

and for error of measurement

1. After lock-on the radar should pass to automatic target tracking. Let us assume that after lock-on there exists a certain initial error of tuning in frequency, and consequently also of measurement of speed. Let us assume also that speed varies according to a law, known with the error of this initial error (for instance, speed is constant during the time of measurement). Then $f(t) \approx 1$, and σ_0^2 - variance of the initial error of measurement of speed. Considering $t_0 = 0$, we obtain

$$g(l, \tau) = \frac{q_0^2}{1 + K_{ont} q_0^2 \tau}, \quad q_{out}^2(l) = \frac{q_0^2}{1 + K_{ont} q_0^2 t}. \quad (9.5.43)$$

The smoothing circuit consists of an ampultier with a variable gain factor $\sigma_0^2/1 + K_{\rm OHT}\sigma_0^2$ t and an integrator. Variance of error decreases in time for large times of observation as $1/K_{\rm OHT}$ t. In the given and subsequent examples we shall not give physical explanations of the meaning of optimality of smoothing circuits, since this question was generally considered in Chapter VI.

With application of an optimum trequency discriminator with a gain factor deterion by (9.5.23) error of measurement of speed is defined as

$$\mathbf{ME}(t) = \frac{\mathbf{v}_{\mathbf{0}}}{\sqrt{1 + \left(\frac{4\pi}{\lambda}\right)^{2} \frac{\mathbf{v}_{\mathbf{0}}^{2} t}{2\Delta f_{\mathbf{0}}} \mathbf{v}(\mathbf{A})}}, \qquad (1 - t^{1}, \dots, t^{1})$$

WINCTO

$\Psi(h) = \frac{h^2}{1 + h (1 + 1 + h)^4}.$

For the example already repeatedly conditioned $\lambda = 3.5$ cm, $\Delta f_{\rm C} = 3$ cp., in 11., 9.20 are surves $\sigma_{\rm BWX}(t)$ for three values of n and for values of initial error of measurement of speed corresponding — between d Δt by 40, 27 and 1 mps. In the given example error decreases ration rapidly, naiving powers, there is human either of a second.



Fig. 9.20. Dependence of error of measurement of speed on time t for the example of an ptimum meter with f(t) = 1: $-\Delta f = 10 \text{ cps}$: $--\Delta f = 20 \text{ cps}$; $---\Delta f = 40 \text{ cps}$.

.2. Let us assume that acceleration of an object whose speed is measured by a radar is constant in a certain stage of observation. Due to scattering of thrust of engines we can only know the mean value of acceleration and the magnitude of scattering of acceleration near this average. Here f(t) = t and from (9.5.41), (9.5.42) for $t_0 = 0$ we have

$$g(t, s) = \frac{\sigma_0^2 ts}{1 + K_{ont} \sigma_0^2 \frac{\tau^2}{3}},$$

$$\sigma_{out}^2(t) = \frac{\sigma_0^2 ts}{1 + K_{out} \sigma_0^2 \frac{\tau^2}{3}},$$
(9.1.45)

where σ_{Ω}^2 - variance of acceleration.

The optimum smoothing circuit here consists of a series-coupled amplifier with a gain factor varying in time according to the law $\frac{4}{1+K_{out}}$, an integrator and $\frac{1+K_{out}}{3}$, an integrator and introduce as σ_c t, und for a large time of observation it decreases as $3/K_{out}$ t. At time

error takes its maximum value, equal to

-451-

$$\mathbf{G}_{\mathbf{m}\mathbf{M}\mathbf{x}} = \frac{1}{\sqrt{3}} \left(\frac{\mathbf{\delta} \sigma_{\mathbf{0}}}{\mathbf{K}_{\mathbf{0}\mathbf{x}\mathbf{T}}} \right)^{\gamma_{\mathbf{x}}}, \qquad (9.5.47)$$

and with the same assumptions about the frequency discriminator it is determined by the formula

$$\mathbf{g}_{\mathbf{s}\mathbf{u}\mathbf{x}} = \mathbf{1}, \mathbf{G}\mathbf{J}'_{\mathbf{a}} \left[\frac{2\Delta f_{\mathbf{a}} \cdot \mathbf{r}_{\mathbf{a}}}{\Psi(h)} \right]^{1/a}. \tag{9.5.48}$$

For case $\lambda = 3.4$ cm, $\Delta f_{C} = 30$ cps in Fig. 9.21 are curves of the dependence of $\sigma_{BHX m}$ on h for three mean square values of acceleration. It turns out that mean square error of measurement of speed in the given optimum system in a wide range of accelerations and signal-to-noise ratios (h = 1 to 100) is a quantity in its frequency expression corresponding to only part of the width of the signal spectrum.



Fig. 9.21. Dependence of maximum error of measurement of speed $z_{BEX} = o_{i} f(t)$ signal-to-noise ratio h for the example of an optimum meter when f(t) = t.

3. If the target is a body moving rapidly and experiencing some drag in the atmosphere, then with observance of certain hardly limiting conditions one may assume that

$$V(l) = \overline{V(l)} - \mu \cdot e^{al}. \qquad (9.5.49)$$

Considering as before $\overline{\mu} = 0$ and $\overline{\mu^2} = \sigma_{1,1}^{2}$ according to (9.5.41) and (9.5.42) we obtain



$$g(t, \tau) = e^{at} \frac{\sigma_0^2 e^{a\tau}}{1 + \frac{K_{000} \sigma_0^2}{2a} (e^{2q\tau} - 1)}$$

$$\sigma_{0000}^2(t) = \frac{\sigma_0^2 e^{2at}}{1 + \frac{K_{000} \sigma_0^2}{2a} (e^{2at} - 1)}$$
(9.5.50)

and

The smoothing circuit consists of a series-coupled amplifier with variable gain $\frac{e_0^{2e^{\alpha t}}}{1+\frac{K_{0ax}e_0^2}{2e^{\alpha t}}}$ an integrator and an amplifier with gain $e^{\alpha t}$.

For large times of observation,

$$g(t, s) \rightarrow \frac{2a}{K_{our}} e^{s(t-s)},$$
 (9.5.51)

and pulse response of the closed tracking meter

$$K_{ont}c(t, \tau) \rightarrow 2ae^{-\kappa(t-\tau)}$$
 (9.5.52)

Here variance of error of measurement of speed is

$$\sigma_{\text{sux}}^2(t) \rightarrow \frac{2\alpha}{K_{\text{ORT}}}.$$
 (9.5.53)

Thus, if we are interes in error for large times of observation, the optimum tracking meter possesses cons parameters and is equivalent to an inertial link (RC-circuit) with time constant $1/\alpha$. Variance of error of the meter is equal here to $2\alpha S_{OHT}(h)$, which determines the form of its dependence on the signal-to-noise ratio h. It is necessary to note that for realization of an optimum meter possessing the shown properties it is necessary in the smoothing circuit to have an amplifier whose gain factor is proportional to $1/K_{OHT} = S_{OHT}(h)$. If in the preceding examples the optimum gain factor of the open circuit decreased with decrease of h, and with some degree of accuracy this change was reproduced by the system of automatic gain control, in this case with a system of automatic gain control it is necessary to constant.

The radial component of the speed of a target, measured by a Doppler speed meter, in many cases contains both components of assigned form, but depending in random parameters, and also purely random components. The form of smoothing circuits optimum for such cases already was discussed in Chapter VI. In the twoloop variant they contain two channels. One of these channels is intended basically for production of the component of the law of change of the measured variable which has the assigned form for finding true values of its parameters. The other channel is intended basically for tracking the purely random part of the measured variable. However, other method of realizing smoothing circuits are possible. We meet one of them in the following example.

Let us find optimum smoothing circuits and corresponding potential accuracy of a speed meter when speed is

$$\mathbf{V}(l) = \overline{\mathbf{V}(l)} + \boldsymbol{\mu} l + \boldsymbol{\mathfrak{t}}(l), \qquad (9.5.54)$$

where V(t) - known mathematical expectation of speed;

 μ - random, normally distributed variable ($\overline{\mu} = 0$, $\overline{\mu^2} = \sigma_0^2$);

 $\xi(t)$ - Wiener process with variance Bt.

Then, composing the correlation function of process V(t), substituting it in equations (6.6.53) and (6.6.54) and solving them, for the pulse response of smoothing circuits we obtain



In connection with the fact that $g(t, \tau)$ turns out to be equal to the product of a certain function of variable t and the function of variable τ , the smoothing circuits in the single-loop variant of the king meter (Fig. 9.22) contain coupled in series:

an amplifier with variable gain

$$f_1(t) = \sqrt{\frac{B}{K_{out}}} \operatorname{sh} \sqrt{BK_{out}} t \cdot e^{-F(\sqrt{BK_{out}}t)}, \qquad (9.5.56)$$

an integrator and amplifier with variable gain

$$i_{s}(l) = \frac{e^{P(\sqrt{BK_{out}}l)}}{ch \sqrt{BK_{out}} - sh \sqrt{BK_{out}}l (\sqrt{BK_{out}}l + B \sqrt{BK_{out}}/\sigma_{0}^{2})^{-1}}, \qquad (0.0.07)$$

where

$$F(s) = \int \frac{\sinh s \, ds}{\cosh s - \sinh s \, (s + B \, V \, B \, K_{s-r/s}^2)^{-1}}.$$

For small times of observation $(t \rightarrow 0 \text{ and } \tau \rightarrow 0)$ (win factors of amplifiers, depicted in Fig. 9.22, vary as

$$f_1(t) \sim \sqrt{\frac{B}{K_{\text{ext}}}} t \ge f_1(t) \approx \frac{1}{t}.$$

For large times of observation $(t \rightarrow \alpha \text{ and } t \rightarrow \alpha)$

-454-



Fig. 9.22. Diagram of an optimum speed meter when

$V(t) = \overline{V(t)} + \mu t + \xi(t):$

1 - discriminator; 2 - amplifier with gain factor $f_1(t)$; 3 - integrator; 4 amplifier with gain factor $f_2(t)$; 5 generator of law of change of $\overline{V(t)}$; 6 adder.

$$f_1(l) = \sqrt{\frac{B}{K_{out}}} \text{ and } f_2(l) = 1;$$

so that the smoothing circuit contains only an integrator with gain factor $\sqrt{B/K_{ORT}}$ It is easy to find that for intermediate values of t the gain factor $f_1(t)$ has a maximum at value $t = t_1$, determined by equation

$$\frac{1}{2} \sin 2 \sqrt{BK_{\text{out}}} t = \sqrt{BK_{\text{out}}} \left(t + \frac{B}{c_0^2} \right),$$

and $f_2(t)$ has a minimum at $t = t_2$, determined from equation

th
$$\sqrt{BK_{out}} t = \sqrt{BK_{out}} \left(t + \frac{B}{\Phi_0^2}\right)$$

Laws of change of gain factors of the amplifiers are shown in Fig. 9.23.



Calculating by (6.6.48) variance of error of the considered optimum meter, we obtain

$$d_{ous}^{2}(t) = \sqrt{\frac{B}{K_{out}}} \times (9.5.58)$$

$$\times \frac{1}{\operatorname{cth} \sqrt{BK_{out}} t - (\sqrt{BK_{out}} t + B\sqrt{BK_{out}}/\epsilon_{0}^{2})^{-1}}$$

As t $\rightarrow \omega$ this variance takes value

$$a_{\text{RMX}}^{2}(\infty) = \sqrt{\frac{B}{K_{\text{ONY}}}} \quad (9.5.59)$$

Fig. 9.23. Form of laws of change of gain factors $f_1(t)$ and $f_p(t)$.

Quantity $\nu = \frac{B}{\sigma_0^2} \sqrt{BK_{OUT}}$ in the formulas is the ratio of $\sigma_{BHX}^2(\omega)$ to variance of the linear component μt in (9.5.54) over interval of time (from moment of beginning) of observation) 5 = $(BK_{OUT})^{1/2}$, equal to the set-up time in the circuit, calculated for a Wiener process.

In Fig. 9.24 according to (9.5.58) there are shown curves of the dependence of $\sigma_{BHX}^2(t)/\sigma_{BHX}^2(\omega)$ on time for different values of parameter ν . The dotted line shows the asymptotic value and the locus of maximum values,



Fig. 9.24. Dependence of variance of error on time t for the example of a meter of speed, determined as $V(0 = \overline{V(t)} + \mu t + \xi(0)$.

The dependence of $\sigma_{BHX}(t)$ on the signal-to-noise ratio h is easy to find, assigning K_{ONT} , for instance, in the form of (9.5.29). For $\sigma_{BHX}(\omega)$ here results obtained above for an optimum meter in case of speed which is a Wiener process are valid.

9.5.4. The Influence of Errors in & Prior1 Data on Accuracy of Measurement

Optimum speed meters, errors of which are analyzed above, can be realized only for known a priori statistics both or right and also of their parameters. Change of a priori statistics of the signal leads to variation of parameters of the optimum discriminator, whereas change of a priori statistics of the measured variable (speed) leads to change of smoothing circuits. Usually "a priori difficulty" with respect to statistics of the measured parameter are stronger than with respect to the received signal. Therefore, consideration of the case of a discriminator which does not differ in its properties from the optimum and of nonoptimal smoothing circuits is of intervat. Here we assume that smoothing circuits are synthesized preceeding from certain y priori statistics of the law of change of speed, but in fact these statistics are different.

We shall consider this question with exampled.

Let us assume that accelerations of the target are random, stationary and uncorrelated. Then speed is a Wiener process. However, variance of this process B_{10} t is difficult to know exactly. Therefore, we consider that smoothing circuits

-451-

are synthesized counting on some other variance B_1t . As it was shown above, as the smoothing filter in this case one should apply an integrator with gain factor $\sqrt{B_1/K_{OHT}}$. Error of tracking in steady-state operating conditions will be composed of fluctuation and dynamic errors. Variance of the first of them is defined as

$$a_{\phi,a}^{2} = 2S_{\phi,\kappa_{a}}\Delta f_{\phi\phi} = \frac{1}{2\pi} \int_{-\infty}^{\infty} \frac{1}{K_{out}} \left| \frac{K_{out}G(i\omega)}{1 + K_{out}G(i\omega)} \right|^{2} d\omega =$$
$$= \frac{1}{2} \sqrt{\frac{B_{i\phi}}{K_{out}}} \sqrt{\frac{B_{i}}{B_{j_{1}}}}, \qquad (9.5.60)$$

where $G(i\omega)$ - frequency response of the integrator.

Variance of dynamic error* can be found, proceeding from the following reasonings. Dynamic error is defined as

$$\pi_{\mathbf{X}\mathbf{H}\mathbf{H}}(t) = \int \vartheta(t-\tau) V(\tau) d\tau = \int \vartheta(t-\tau) d\tau \int \xi(s) ds =$$

= $\int \mathbf{x} (t-\tau) \xi(\tau) d\tau,$ (9.5.61)

where V(t) - law of change of the random component of speed, consisting of the integral of white noise $\xi(t)$, having spectral density B_{10} : $\vartheta(t) - pulse$ response of the closed system, if we consider mismatch the output variable

$$\kappa(t) = \int_{\delta}^{t} \vartheta(\tau) d\tau.$$

Variance of dynamic error

$$\sigma_{ABB}^{2} = \overline{s_{ABB}^{2}(t)} = B_{10} \int_{t_{0}}^{t} x^{2} (t - \tau) d\tau \approx$$

$$\approx \frac{1}{2\pi} \int_{-\infty}^{\infty} \frac{B_{10}}{\omega^{2}} \left| \frac{1}{1 + K_{0BT}G(i\omega)} \right|^{2} d\omega =$$

$$= \frac{1}{2} \sqrt{\frac{B_{10}}{K_{0BT}}} \sqrt{\frac{B_{10}}{B_{1}}}, \qquad (9.5.67)$$

where we use the circumstance that the Fourier transform of $\vartheta(t)$ is $1/1 + K_{CHT}G(1m)$. Total error has variance $\sigma_{BHX}^2 = \sigma_{QJH}^2 + \sigma_{QJH}^2$. Its ratio to minimum possible $\sigma_{BHX}^2 = \sqrt{B_{10}/K_{OHT}}$, obtained for $B_1 = B_{10}$, is equal to

$$\frac{\mathbf{e}_{\mathbf{p}_{\mathsf{eff}}}^{2}}{\mathbf{e}_{\mathsf{p}_{\mathsf{eff}}}^{2}} = \frac{1}{2} \sqrt{\frac{B_{1}}{B_{1e}}} + \frac{1}{2} \sqrt{\frac{B_{1e}}{B_{1}}}.$$
(9.5.63)

This formula also determines increase of error due to a priori data incorrectly presented during synthesis of the system.

^{*}Here, in accordance with ChapterVI, we applied the statistical approach to dynamic error.

If the measured speed is the double integral of white noise, absolutely analogously it is possible to derive formula

$$\frac{\mathbf{q}_{BMX}^{2}}{\mathbf{q}_{BMX}^{2}} = \frac{3}{4} \left(\frac{B_{3}}{B_{24}} \right)^{1/4} + \frac{1}{4} \left(\frac{B_{24}}{B_{3}} \right)^{1/4}, \qquad (9.5.64)$$

where B_{20} - true spectral density of white noise, the double integral of which is speed, and B_2 - its value assumed during synthesis of the filter.



Fig. 9.25. Curves of the dependence of increase of error of measurement of speed on error in a priori data presented during synthesis: $-B_1/B_{10}$; --- B_2/B_{20} .

In Fig. 9.25 formulas (9.5.63) and (9.5.64) are presented graphically. From the graphs we see the low criticality of the system of measurement of speed to selection of magnitudes of B_1 and B_2 , respectively. Thus, for $B_1 = 10B_{10}$ or $B_1 = 0.1B_{10}$ (which are equivalent to expansion or narrowing of bandwidth of the system by a factor of 3.2 as compared to the optimum) error increases by 32%, and for $B_2 = 16B_{20}$ (i.e., with expansion of the band by a factor of 2) this growth comprises 24% in all.

With decrease of B_2/B_{20} quantity σ_{BHX}^2 grows somewhat more strongly, so that in systems of the second order certain expansion of bandwidth is preferable if dynamic properties of the target are not known exactly.

Let us consider now the case when speed varies as

$$V(t) = \overline{V(t)} + \mu t + \xi(t),$$

where $\xi(t)$ - Wiener process with variance Bt;

 μ - normally distributed random variable ($\overline{\mu} = 0$, $\overline{\mu^2} = \sigma_0^2$).

Let us assume that this process does not influence the optimum system of Fig. 9.22, but a meter designed for the fact that speed is a Wiener process. Then the smoothing filter is an integrator with gain $\sqrt{E/K_{OUC}}$. Calculation of variance of total error for large times of observation $(t \rightarrow \infty)$ gives

$$\sigma_{a_{MX}}^{3} = \sigma_{\phi,a}^{2} + \sigma_{A_{XX}}^{2} = \sqrt{\frac{B}{K_{0,XY}}} \left(1 + \frac{1}{v}\right),$$

from which the ratio of variance of error to its minimum possible value (9.5.59), peculiar to an optimum system, is defined as

$$\frac{q_{\text{BMX}}^2}{q_{\text{BMX}}^2} = 1 + \frac{1}{y},$$

where

$$v = \frac{B}{\sigma_0^2} \sqrt{BK_{our}}$$

When $\nu < 1$ growth of error is substantial. If we assume that in the considered speed meter there is not introduced the mean value of $\overline{V(t)}$, for $\overline{V(t)}$ = at error will increase additionally magnitude $a/\sqrt{BK_{DUT}}$.

For the above-considered example of a radar with wavelength $\lambda = 3.4$ cm with signal bandwidth $\Delta f_{c} = 30$ cps and a large signal-to-noise ratio ($h \rightarrow \infty$) increase of errors due to disregarding the quasi-regular component of speed and failure to insert the mean value is illustrated by Table 9.4.

₽.	8	Minimum Serror		Error appli of integra	with Nation an tor for	Error with application of an integrator and without insertion of the mean value				
				\$1000	thing	a = 1 m/sec ²		a = 10 m/sec		
m ² /sec ³ m/sec ²	m/\$e0	cps	m/sec	cps	a/seq	cps	m/sec	cps		
0,03 0,25·10~* 0,1 0,3	0,0324 0,1 0,324	0,0324	1,9	0,035 0,053 0,14	2.06 3.1 8,2	0,455 0,473 0,56	26.8 27.8 33	4,235 4,253 4,34	250 250 255	
0,25 10-1	0,1 0,324 1	0,0576	3,4	0,059 0,072 0,014	3,5 4,25 8,55	0,192 0,205 0,278	11,3 12,1 16,3	1,39 1,4 1,47	81.8 82,2 86,5	
0,25	0,324 1 3,24	0,105	6	0,106 0,113 0,175	6,04 6,48 10	0,148 0,155 0,217	8,7 9,1 12,8	0,526 0,533 0,595	31 31,4 35	
			I. I				,-			

Table 9.4

As can be seen from the table, for comparatively low rates of change of the mean value of measured speed (acceleration of 1 and 10 m/sec²) errors of measurement, expressed in units of Doepler frequency shift, attain substantial magnitudes, exceeding the width of the spectrum of fluctuations of the signal. Due to the narrow bands of filters of the discriminator breakoff of tracking is possible here. To avoid it, it is necessary to wider the effective band of the system, increasing fluctuation error of measurement, or to switch to smoothing circuits containing two integrators. Introduction of the mean value and use of a smoothing filter with variable parameters permits us to narrow the effective bandwidth and to obtain minimum errors given in the table.

(9.5.65)

§ 9.6. Breakoff of Tracking in Doppler Speed Meters

With intense noises and interferences in speed meters there occur nonlinear phenomena, leading ultimately to breakoff of tracking. Such phenomena were studied in Chapter VI for tracking meters of any parameters of motion of a target. There we discussed two criteria of breakoff of tracking: the criterion of sharp decrease of average time to the first breakoff and the criterion of sharp increase of fluctuation error of measurement in steady-state operating conditions. With sinusoidal approximation of the discrimination characteristic we found the critical value of the ratio of the half-width of the selected domain Δ to the mean square value of fluctuating error of a linearized system $\sigma_{\rm m}$ at which breakoff of tracking occurs.

For the criterion of the average time to breakoff this ratio is $(\Delta/\sigma_{\pi})_{\rm Kp} \approx 5$, and for the criterion of steady-state error it is $(\Delta/\sigma_{\pi})_{\rm Kp} \approx 10$.

Let us consider the phenomenon of breakoff of tracking for Doppler speed meters, considering application of the forms of frequency discriminators investigated above. Here, besides using results of Chapter VI just now mentioned we analyze nonlinear phenomena in certain speed meters without using a sinusoidal approximation of the discrimination curve, which gives us the possibility of estimating accuracy of finding critical intensities of interferences.

We turn, first of all, to a speed meter with an optimum frequency discriminator. Its discrimination characteristic is determined by expression (9.3.6).

Considering that the spectral density of fluctuations of the signal is described by formula (9.3.10), after calculations we obtain

$$a(\delta) = \frac{1}{2\Delta_{0}} h^{2} \frac{1 + \sqrt{1+h}}{\sqrt{1+h}} \frac{2}{\left[\left(\frac{2}{2\Delta_{0}}\right)^{2} + (1 + \sqrt{1+h})^{2}\right]^{2}}.$$
 (9.6.1)

We assume that there is realized a discriminator completely optimum only for a signalto-noise ratio $h = h_1$ and untunable with change of h.

Then from (9.6.1) and (9.3.11) the equivalent discrimination characteristic is

$$a_{\text{OKB}}(\delta) = \frac{a(\delta)}{K_{\mathcal{A}}} = a_1^4 \frac{\delta}{\left[\left(\frac{\delta}{2\Delta f_{\sigma}}\right)^2 + a_1^2\right]^2}, \qquad (9, 6.7)$$

where

$$a_1 = 1 + \sqrt{1 + h_1}.$$
 (9.6.3)

In the case of application of a sinusoidal approximation the equivalent discrimination characteristic is

$$a_{\text{UKB}} \cdot (\delta) = \frac{\Lambda}{\pi} \sin \frac{\pi \delta}{\Lambda}. \qquad (9.6.4)$$

Quantity \triangle can be found by equating areas of curves (9.6.2) and (9.6.4) in the domain $\delta > 0$:

$$\int_{\delta}^{\infty} a_{\mathfrak{d} \mathsf{K} \mathfrak{s}}(\delta) d\delta = \int_{\delta}^{\infty} a_{\mathfrak{d} \mathsf{K} \mathfrak{s}} \mathfrak{s}(\delta) d\delta,$$

from which we obtain

 $\Delta = \star \Delta f_{c} a_{1}. \tag{9.6.5}$

There can be applied other methods of approximation of the discrimination characteristic by a sinusoid. In particular, it is possible to equate abscissas of maxima.

Then

$$\Delta = \frac{4}{\sqrt{3}} a_1 \Delta f_c = 2,32 \Delta f_c a_1. \tag{9.6.5'}$$

Applying the criterion of average time to breakoff, we have

Δ'=255.

Substituting in this formula expression (9.7.5) for variance of fluctuation error of a linearized speed meter with application of an integrator as the smoothing circuit, it is easy to find the relationship for the critical signal-to-noise ratio. For the first method of approximation it has the form

$$\mathbf{p}(\mathbf{h}_{\mathbf{K}\mathbf{p}}) = \frac{\mathbf{a}_{1}^{2}}{10} \frac{\mathbf{A}_{fo}}{\mathbf{A}_{food}}, \qquad (9.6.6)$$

for the second method of approximation it has the form

$$\Psi(h_{\rm xp}) = \frac{a_1^2}{18.7} \frac{\Delta f_o}{\Delta f_{o+o}}.$$
 (9.6.01)

Here $\Delta f_{\partial \Phi} = 0$ - effective bandwidth of the tracking meter in the absence of noises, and dependence

$$\Psi(h) = \frac{\sqrt{1+h}(1+\sqrt{1+h})^{4}}{h} \frac{1}{h+y}$$
(9.6.7)

is shown in Fig. 9.14.

Let us assume for instance, that $\Delta f_{C} = 30 \text{ cps}$, $\Delta f_{3\oplus -C} = 1 \text{ cps}$, y = 3 and $h_{I} = 1$; then from the curve for y = 3 in Fig. 9.14 we obtain $h_{K_{I}-I} = 0.17$ with the first method of approximation and $h_{K_{I}-II} = 0.35$ for the second method.

When using the criterion of steady-state error

the equation for h_{KI} takes the form

-461-

$$\varphi(h_{Hp}) = \frac{a_1^2}{40} \frac{\Delta f_0}{\Delta f_{0,0}}$$
(9.6.8)

for the first method of approximation, and

$$\boldsymbol{\varphi}(h_{\mathrm{Kp}}) = \frac{\boldsymbol{u}_{\mathrm{i}}^{2}}{75} \frac{\Delta f_{\mathrm{c}}}{\Delta f_{\mathrm{i}} + \bullet}$$
(9.6.81)

for the second method of approximation.

For the preceding example we obtain $h_{KP} I = 1$ and $h_{KP} II = 6$. In connection with the fact that between the values of h_{KP} obtained using different criteria there exists a marked difference in cases when from conditions of work of system it is impossible to give obvious preference to the criterion of average time to breakoff, one should use equations (9.6.8).

Let us find the equation for h_{KP} without resorting to sinusoidal approximation of the discrimination characteristic. Using the criterion of steady-state error on the basis of the general formulas (0.3.22) assuming the absence of dynamic error, for variance of fluctuating error we have

$$\sigma^{0} = \frac{\int d^{2} \exp\left\{-\frac{1}{\sigma_{A}^{2}}\int a_{\partial x a}(d') dd'\right\} dd}{\int \exp\left\{-\frac{1}{\sigma_{A}^{2}}\int a_{\partial x a}(d') dd'\right\} dd}$$
(9.6.9)

where l - level of limitation.

Substituting (9.6.2) in (9.6.9), we obtain

 $\mathbf{G}^{2} = \mathbf{G}_{a}^{2} \frac{\int_{l_{1}}^{l_{1}} x^{3} e^{\frac{|\mathbf{p}|^{2}}{1+x^{3}/2\mu}} dx}{\int_{l_{1}}^{l_{1}} e^{\frac{|\mathbf{p}|^{2}}{1+x^{3}/2\mu}} dx}, \qquad (9,6,10)$

where

$$\mu = \frac{a_1^2 (2\Delta f_0)^2}{2\sigma_{\mu}^2},$$
(9.6.11)
$$I_1 = \frac{i \sqrt{2\mu}}{a_1 2\Delta f_0}.$$

Calculating (9.6.10) approximately for large a (low level of noises), we have

 $\underbrace{ \begin{array}{c} \mathbf{e}^{1} \\ \mathbf{e}^{2} \\$



increase of variance of error of measurement of speed on parameter μ in the case of application of an

optimum frequency discriminator.

Dependence (9.6.12) is shown in Fig. 9.26. Considering that for $\mu = 10$ this dependence already gives clearly minimized values, it is necessary to consider that $u_{\rm KF} =$ = 15 to 20. If we consider $\mu_{\rm KF} = 20$, for $h_{\rm KF}$ we again obtain equation (9.6.8). Let us analyze breakoff of tracking during application of discriminators with a tuned circuit and a phase shifter and also with mixers and differentiation. According to (9.4.13) the equivalent discrimination characteristic of these discriminators has the form

$$a_{\delta K \mathfrak{g}}(\delta) = \frac{\delta}{1 + \left[\frac{\delta}{2(\Delta f_{\mathfrak{g}} + \Delta f_{\mathfrak{g}})}\right]^{2}}.$$
 (9.6.13)

Approximating (9.6.13) by a sinusoid, with the help of equating of abscissas of maxima* we have

$$\Delta = 4(\Delta f_0 + \Delta f_{\rm K}), \qquad (9.6.14)$$

from which, during application of the criterion of steady-state error ($\omega = 100 \frac{1}{\pi}$) and using expression (9.4.9) for equivalent spectral density, we obtain

$$F(h_{Rp}) = 0.04a (1 + a) \frac{\Delta f_0}{\Delta f_{000}},$$
 (9.6.15)

where

$$F(h) = \frac{h}{h+s} \left[1 + \frac{2}{h} (1+a) + \frac{1}{h^{s}} (1+a)^{s} \right], \qquad (9.6.16)$$

and $\alpha = \Delta f_{\rm R} / \Delta f_{\rm C}$.

Relationship F(h) for y = 3 and $\alpha = 1$ is shown in Fig. 9.27. For the abovementioned example from the curve of Fig. 9.27 we find $h_{KT} = 1.5$.

If we do not use sinusoidal approximation of the discrimination characteristic, the law of distribution of probabilities for mismatch is defined as



*The method of approximation based on equating areas of curves is inapplicable in this case, since the integral of (9.6.13) goes to infinity.

-463-



Fig. 9.27. Relationship F(h), illustrating formula (9.6.16).

Dependence

$$\frac{\Phi^2}{\Phi^2} = \frac{1}{1 - \frac{3}{4} \Psi^2}, \qquad (9.6.20)$$

where

is shown in Fig. 9.28.

Froceeding from this dependence, critical value of $\nu_{\rm KF}$ = 0.8 to 0.9. Considering $\nu_{\rm KF}$ = 0.8, we find

 $v = \frac{\sigma_{\pi}}{\Delta I_0 + \Delta I_{\pi}}$

$$F(h_{K_{\rm F}}) = 0,16u(1+a) \frac{\Delta f_c}{\Delta f_{0,0}}.$$
(9.6.21)

For the example considered in this paragraph from the curve of Fig. 9.28 we obtain h_{ν} . ≈ 0.3 .



2018月1日,其他为111月1日,在11月1日,在11月1日,11月1日,11月1日,11月1日,11月1日,11月1日,11月1日,11月1日,11月1日,11月1日,11月1日,11月1日,11月1日,11月1日,1

Fig. 9.23. Dependence of increase of variance of error of measurement of speed on parameter ν when using a discriminator with a tuned circuit and phase shifter or a discriminator with mixers and differentiation.

From the given examples it is clear that even in the framework of one criterion of breakoff of tracking, depending upon the form and method of approximation of the discrimination characteristic we can obtain

-464-

等于可能能能需要的分析,并且对其正式的意义的行为。但在自然是有些问题的主义的任何的意思。

Designating

and

$$\frac{r+1}{2} = \frac{4(\Delta f_c + \Delta f_w)^2}{2\sigma_n^2}$$

$$\frac{x^3}{n} = \frac{3^n}{4(\Delta f_c + \Delta f_w)^2},$$
(9.6.18)

we reduce (9.6.17) to a Student distribution [8].

Then, with no limitation in the smoothing circuits, we obtain

$$\mathbf{o}^{\mathbf{a}} = \overline{\mathbf{\delta}^{\mathbf{a}}} = \frac{4(\Delta f_{\mathbf{a}} + \Delta f_{\mathbf{x}})^{2}}{n-2} = \frac{\sigma_{\pi}^{2}}{1 - \frac{3\sigma_{\pi}^{2}}{4(\Delta f_{\mathbf{c}} + \Delta f_{\mathbf{x}})^{2}}}.$$

$$(9.6.19)$$

्रा

rather substantial scattering in determination of the critical value of the signalto-noise ratio $h_{\rm KP}$. Therefore, when we have formulas for $h_{\rm KP}$ obtained without any approximations, one should use them. Results obtained by sinusoidal approximation of the discrimination characteristic can serve only for estimating the order of quantity $h_{\rm KP}$.

§ 9.7. Nontracking Speed Meters

In multipurpose radars, to which class most detection and acquisition radars belong, they often use nontracking meters. Therefore, consideration of the question of accuracy attainable when using nontracking Doppler speed meters, and of methods of constructing such meters merits our attention.

In accordance with Chapter VI (Paragraph 6.6.5) optimum nontracking meters should be constructed by the scheme of Fig. 6.17 and should consist of an estimator unit, issuing the maximum likelihood estimate of measured speed at every given moment of time, an accuracy unit which with normally satisfactory limiting assumptions can be absent, and a certain linear filter. The estimator unit can be approximately presented in the form of a multichannel system for processing the preceived radio signal, channels of which are mutually detuned in frequency (theoretically there should be infinitely many of them). Every channel should be constructed in such a way that the voltage at its output, formed during time T, for which the measured speed (Doppler shift of frequency) does not change, is proportional to the value of the likelihood function for some value of ω_{μ} . Selection of the maximum of the obtained voltages gives the maximum likelihood estimate of speed.

The form of each channel here is determined in Chapter IV and is illustrated by Fig. 9.29. As also in discriminators of tracking meters, in the channel there occurs



Fig. 9.29. Scheme of a channel of an optimum nontracking speed meter: $1 - \text{mixer}; 2 - \text{amplifier}; 3 - \text{filter}; 4 - \text{square-law de-}^{-1}$ tector; 5 - integrator (absent with slow fluctuations).

shifting of phase modulation by heterodyning and multiplication by a function expressing the law of amplitude modulation (gating - for pulse modulation). Thus processed, the signal is subjected to narrowband filtration and then square-law

detection. The form of the filter is different for fact and slow fluctuations $t_{1,2}$ low-frequency equivalent of its frequency response is determined by (9.2.3); for slow fluctuations this low-frequency equivalent corresponds to an integrator. In

-465-

practice, parameters of such systems of course differ from the optimum, which leads to lowering of accuracy. We, however, shall limit ourselves to analysis of the potential accuracy peculiar to the described optimum circuit of signal processing.

In accordance with the analysis in Chapter VI (Paragraph 6.6.9), with measurement by the method of maximum likelihood of speed, constant in a certain interval T, variance of error of measurement with broad assumptions is close to variance of the efficient estimate and is defined as

$$\sigma_{\bullet\bullet}^2 = \frac{S_{OBT}}{T}, \qquad (9.7.1)$$

where S_{OHT} - equivalent spectral density of the optimum discriminator of the tracking speed meter.

According to (9,3.1)

$$\mathbf{g}_{\mathbf{a}\mathbf{b}}^{-2} = -\frac{1}{2} \int_{\mathbf{0}}^{\mathbf{T}} \int_{\mathbf{0}}^{\mathbf{T}} \frac{\partial \mathbf{W}\left(t_{1}, t_{2}, \mathbf{w}_{R}\right)}{\partial \mathbf{w}_{R}} \frac{\partial \mathbf{R}\left(t_{1}, t_{2}, \mathbf{w}_{R}\right)}{\partial \mathbf{w}_{R}} dt_{1} dt_{2}, \qquad (9.7.2)$$

where $R(t_1, t_2, \omega_{II})$ - correlation function of the signal, determined by (9.2.1), and $W(t_1, t_2, \omega_{II})$ in accordance with (6.7.20) is found from solution of equation

$$R(t_1, s) W(s, t_2) ds = \delta(t_1 - t_2).$$
 (9.7.3)

Fresenting $W(t_1, t_2)$ in the form

$$W(t_{1}, t_{2}) = -w(t_{1}, t_{2}) \operatorname{Re} \left\{ u(t - t_{1}) u^{*}(t - t_{2}) e^{i(t_{2} - t_{2})} + \frac{1}{N_{a}} \delta(t_{1} - t_{2}), \right\}$$

$$(9.7.4)$$

where $u(t) = u_a(t)e^{i\psi(t)}$, we transform (3.7.3) and (9.7.8) to

$$\frac{P_{e}}{2}\int_{0}^{T} p(t_{1}-s)w(s, t_{2}) ds + N_{e}w(t_{1}, t_{2}) = \frac{P_{e}}{N_{e}}p(t_{1}-t_{2})$$
(9.7.5)

and

$$\mathbf{e}_{s0}^{-2} = \frac{P_0}{4} \int_0^T \int_0^T (t_1 - t_2)^2 \rho \left(t_1 - t_2\right) w \left(t_1, t_2\right) dt_1 dt_2, \qquad (9.7.6)$$

where

P____mean signal power;

N_O - spectral density of noise;

 $\rho(t_1 - t_2)$ - correlation function of fluctuations.

Approximate solution of (9.7.5) and calculation of (9.7.6) can be performed immediately in asymptotic cases of fast and slow fluctuations. For fast fluctuations $(\text{Lf}_{C}|T>>1)$ this solution is given by formula (9.7.1) with substitution in it of $S_{\text{cl.c.}}$, found in § 9.3. As a result

$$\sigma_{\bullet\bullet}^{-2} = \frac{T}{2\pi} \int_{-\infty}^{\infty} \left[\frac{hS'_{\bullet}(\omega)}{1 + hS_{\bullet}(\omega)} \right]^2 d\omega \qquad (9.7.7)$$

-466-

and with spectral density of fluctuations $S_0(\omega)$, determined by (9.3.10),

$$\sigma_{a\phi}^{2} = \frac{2\Delta f_{c}}{T} \frac{\sqrt{1+h} (1+\sqrt{1+h})^{3}}{h^{a}}, \qquad (9.7.8)$$

where Δf_{c} - effective bandwidth of signal fluctuations.

For slow fluctuations ($\Delta f_c T \ll 1$), assuming that during time T the coefficient of correlation $\rho(t_1, t_2) = 1$, we obtain

$$\sigma_{a\phi}^2 = \frac{6}{T^2} \frac{1+\mu}{\mu^2}, \qquad (9.7.9)$$

where

$$\mu = \frac{P_{o}T}{2N_{o}} = \Delta f_{o}Th. \qquad (9.7.10)$$

However, asymptotic results are insufficient, first, because we are interested in error of measurement in intermediate cases and, secondly, formula (9.7.9) needs more precise definition of conditions of its applicability, since as $\mu \rightarrow \infty$ according to this formula $\sigma_{\sigma\tilde{m}}^2 \rightarrow 0$, while for fast fluctuations $\sigma_{\sigma\tilde{m}}^2$ differs from zero. To find $\sigma_{\sigma\tilde{m}}^2$ for arbitrary time of correlation of the signal it is necessary to assign the concrete form of the correlation function. Producing calculations for exponential correlation

$$\mathbf{p}(t) = \mathbf{e}^{-2\Delta t_0(t)}, \qquad (9.7.11)$$

we obtain

$$a_{s,\phi}^{-2} = \frac{\hbar^{2}}{4\Delta f_{c}^{2}} \left\{ \frac{1}{\lambda(\lambda+1)^{s} \left[1 - \left(\frac{\lambda-1}{\lambda+1}\right)^{s} e^{-2\lambda a} \right]} \times \left[a - \frac{5\lambda^{s} - 9\lambda^{s} + 3\lambda + 1}{2\lambda(\lambda-1)^{s}(\lambda+1)} + 2\lambda e^{-(\lambda+1)a} \frac{\lambda-3+(\lambda^{s}-1)a}{(\lambda-1)^{s}(\lambda+1)} + \frac{e^{-2\lambda a}}{2\lambda} \left(\frac{\lambda+1}{\lambda-1}\right)^{s} \right] - \frac{1}{\lambda(\lambda-1)^{s} \left[\left(\frac{\lambda+1}{\lambda-1}\right)^{s} \frac{2\lambda a}{e^{-1}} - 1 \right]} \times \left[a + \frac{5\lambda^{s} + 9\lambda^{s} + 3\lambda - 1}{2\lambda(\lambda+1)^{s}(\lambda-1)} - 2\lambda e^{(\lambda-1)a} \frac{\lambda+3-(\lambda^{s}-1)a}{(\lambda+1)^{s}(\lambda-1)} - \frac{e^{2\lambda a}}{2\lambda} \left(\frac{\lambda-1}{\lambda+1}\right)^{s} \right] \right\},$$
(9.7.12)

where $\lambda = \sqrt{1 + h}$,

$$a = 2\Delta f_0 T.$$
 (9.7.13)

As $h \to \infty$ this expression leads to $\sigma_{\partial f}^2 = 2\Delta f_C / T$ for any Δf_C , i.e., variance of the efficient estimate of frequency (speed) in the case of absence of noises both for fast and for slow fluctuations of the signal is identical. In the case of slow fluctuations $\lambda a = 2\Delta f_C T / T + h$ is a small quantity. From (9.7.12) approximately for $h \gg 1$ we obtain

$$s_{0}^{-2} = \frac{T^{2}\mu^{2}}{6(1+\mu)} \left(1 - \mu - \frac{16\Delta f_{c}T}{5}\right), \qquad (9.7.14)$$

from which it follows that asymptotic formula (9.7.9) is valid only when $\mu \ll \ll 5/16\Delta r_c$ T, i.e., for not too large a signal-to-noise ratio h.

For large λa , occurring either in the case of fast fluctuations or during sufficiently large signal-to-noise ratios h, formula (9.7.12) is simplified, taking form

$$\sigma_{\bullet\phi}^{-2} = \frac{\hbar^{9}T}{2\Delta f_{\bullet}} \frac{1}{\lambda(\lambda+1)^{2}} \left[1 - \frac{2\lambda^{9} - 3\lambda^{9} + 1}{2\Delta f_{c}T\lambda(\lambda+1)(\lambda-1)^{3}} + \frac{2\lambda^{9}e^{-(\lambda+1)2\Delta f_{c}T}[\lambda-3+2\Delta f_{c}T(\lambda^{9}-1)]}{2\lambda\Delta f_{c}T(\lambda-1)^{9}(\lambda+1)} \right].$$
(9.7.15)

The obtained expressions determine potential accuracy of nontracking Doppler speed meters in case of absence of averaging of data from the output of the estimator unit. For illustration of errors of measurement of speed, obtained for different values of Δf_{c} T and h, in Fig. 9.30 we give a graph of

$$\frac{\mathfrak{e}_{\mathfrak{p}\varphi}^{2}(\Delta f_{\mathfrak{e}}T,h)}{\frac{2\Delta f_{\mathfrak{c}}}{T}} = \frac{\mathfrak{e}_{\mathfrak{p}\varphi}^{2}(\Delta f_{\mathfrak{c}}T,h)}{\mathfrak{e}_{\mathfrak{p}\varphi}^{2}(\infty,\infty)}$$

depending upon the signal-to-noise ratio h for various Δf_{0} T. As already noted, with increase of h quantity $\sigma_{\partial \Phi}^{2}$ seeks $2\Delta f_{0}/T$. However, for finite h the obtained



Fig. 9.30. Dependence of variance of error of an optimum nontracking speed meter on the signal-to-noise ratio for different products of bandwidth of fluctuations of signal by the time of observation $\Delta f_{\rm c}$ T.

variance of error of measurement strongly depends on $\Delta f_{\rm C}$ T if $\Delta f_{\rm C}$ T << << 1. Therefore, one should select time of estimate T in such a manner that there is observed the condition $\Delta f_{\rm C}$ T >> 1. As can be seen from the graph, considerable increase of $\Delta f_{\rm C}$ T as compared to unity is not required. When for measurement of speed

-468-

there are used data from the output of the estimator unit, experiencing additional smoothing by some filter, the obtained variances of efficient estimates characterize the intensity of the random components at the input of this filter. Here if the timof formation of the estimate T and the time of correlation of signal fluctuations are small as compared to the inertia of the filter, then, as also in the case of measurement of distance, it is possible to approximately consider that at the input of the filter there acts white noise with spectral density

$$S_{\mu\nu} = \sigma_{\mu\nu}^2 T.$$
 (9.7.16)

For fast fluctuations $S_{BX} = S_{OUT}$, and, consequently, with proper selection of smoothing circuits tracking and in nontracking optimum meters ensure identical accuracy, which was noted in Chapter VI.

The obtained results permit us to estimate the accuracy with which it is possible to measure speed on the basis of the Doppler effect with incoherent pulse radiation. If for pulse duration $\tau_{_{H}}$, as normally occurs, fluctuations of signal are considered slow, then variance of the efficient estimate of Doppler frequency shift, obtained during the time of the pulse, is, according to (9.7.9), found as

$$\mathbf{q}_{\mathbf{s}\mathbf{\phi}}^{2} = \frac{6}{\mathbf{r}_{\mathbf{s}}^{2}} \frac{1+q}{q^{\mathbf{s}}}, \qquad (9.7.17)$$

where

$$q = \frac{P_{\mathrm{H}} \varsigma_{\mathrm{H}}}{2N_{\mathrm{o}}} = \frac{P_{\mathrm{c}} T_{\mathrm{r}}}{2N_{\mathrm{o}}},$$

P₁, - power in pulse;

 T_r - period of repetition of pulses.

Expression (9.7.17) determines accuracy of a unit measurement of Doppler frequency. If these measurements are cross-correlated and we average them for time T,

$$\mathbf{o}_{\mathbf{a}\phi}^2 = \mathbf{o}_{\mathbf{a}\phi}^2 \frac{T_r}{T}.$$
 (9.7.18)

For large signal-to-noise ratios $(q \gg 1)$

$$\sigma_{\bullet\bullet}^{2} = \frac{6}{\tau_{\bullet}^{2}} \frac{1}{c} \frac{T_{\bullet}}{T} = \frac{6}{\tau_{\bullet}^{2}} \frac{1}{\mu}.$$
 (9.7.19)

The obtained accuracy of measurements for the usual relationships of parameters is very low. Thus, for instance, when $\tau_{\rm M} = 1~\mu{\rm sec}$, $q = 10^3$ and $T/T_{\rm r} = 10^3$ we obtain $c_{\rm sch} = 390$ cps, i.e., error is great even for a very large signal-to-noise ratio. Therefore, for sufficiently accurate measurement of radial speed by the Doppler effect it is necessary to have a coherent signal.

§ 9.8. <u>Measurement of Radial and Tangential Components of Speed</u> as Derivatives of Distance and Angular Coordinates

In the preceding paragraphs we considered in detail questions of measurement of speed on the basis of the Doppler effect. Not always, however, is this method of measurement of speed applicable. As we have seen, in the case of widely-used incoherent pulse radiation with the usual combinations of parameters there is unsatisfactory accuracy of measurement. Furthermore, we are not alway interested in the radial component of speed alone; in certain cases it is also necessary to measure its tangential component. Therefore, we shall investigate methods of measurement of components of speed of target, based on the fact that these components are derivatives of distance and angular coordinates.

We shall consider that there are applied tracking range meters and goniometers (Fig. 9.31), general analysis of which was conducted in Chapter VI. We designate $\lambda(t)$ - a parameter varying in time (distance, angle), to be measured. The component



of speed interesting us is the derivative of this parameter $\lambda^{\dagger}(t)$. Farameter $\lambda(t)$ is coded in the received radar signal, mixed with noise $y[t, \lambda(t)]$. As a result of the influence of this signal on the discriminator there will be formed mismatch signal $z[t, \varepsilon(t)]$, where $\varepsilon(t) = \lambda(t) - \hat{\lambda}(t)$, and $\hat{\lambda}(t)$ is the output of the range meter or goniometer. The mismatch signal influences smoothing circuits of some form, forming $\hat{\lambda}(t)$.

Such a meter can be both optimum, or nonoptimal depending upon selection of the discriminator and smoothing circuits. Investigation of different range discriminators is conducted in Chapters VII and VIII, and angular discriminators are investigated in Chapters X and XI. In all cases for low levels of noises or interferences there are small mismatches, for which the discriminator can be linearized, i.e., $z(t, \varepsilon(t))$ can be presented in the form

$$\mathbf{z}(t, \mathbf{u}(t)) \approx K_{\mathbf{A}}[\mathbf{\varepsilon}(t) + \mathbf{\xi}(t)], \qquad (9 \ 5.1)$$

where K_{II} - gain factor of the discriminator;

 $\xi(t)$ - equivalent noise, which, thanks to the inertia of the smoothing circuits, can be considered "white" and is characterized by its spectral density s_{ake} .



Fig. 9.32. A speed meter based on differentiation of a coordinate: 1 - discrim-

inator; 2 - smoothing circuits; 3 - dif-

ferentiator.

It is required to measure not only coordinate $\lambda(t)$, but also speed $\lambda'(t)$, this can be done by passing the output quantity of the tracking meter $\hat{\lambda}(t)$ through a filter which to some degree of accuracy carries out

differentiation (Fig. 9.32). Often the voltages at certain points of filters of range finders and goniometers are proportional to the derivative of the measured coordinate; in these cases there is no need for additional differentiating filters. Of interest is investigation of accuracy of measurement of speed by the shown method for various forms of smoothing circuits; also of interest is synthesis of an optimum meter of speed as the derivative of a coordinate and finding the potential accuracy of such measurements. These problems are the content of the present paragraph.

9.3.1. Accuracy of Measurement of Speed by a System with Constant Parameters

As follows from the preceding chapters, in many cases both the ones most widespread in practice, and also optimum smoothing filters are linear and possess constant parameters. We shall investigate accuracy of measurement of speed by such systems.

If we designate by G(p) the transfer function of a smoothing filter of a range meter or goniometer, with observance of condition (9.8.1) the Laplace transform of the output variable of the meter will be recorded in the form

$$\hat{\boldsymbol{\Lambda}}(\boldsymbol{p}) = \frac{K_{A}G(\boldsymbol{p})}{1+K_{A}C(\boldsymbol{p})} [\boldsymbol{\Lambda}(\boldsymbol{p}) + \boldsymbol{\Xi}(\boldsymbol{p})], \qquad (9.8.2)$$

where $\Lambda(p)$ and $\Xi(p)$ - Laplace transforms of $\lambda(t)$ and $\xi(t)$, respectively.

We consider that formation of the derivative is also carried out by a linear filter with constant parameters with transfer function H(p). The Laplace transform of the output variable of this filter takes the form

$$\hat{\boldsymbol{\lambda}}(\boldsymbol{p}) = \frac{\boldsymbol{K}_{\boldsymbol{x}}\boldsymbol{G}(\boldsymbol{p})\boldsymbol{H}(\boldsymbol{p})}{1 + \boldsymbol{K}_{\boldsymbol{x}}\boldsymbol{G}(\boldsymbol{p})} [\boldsymbol{\Lambda}(\boldsymbol{p}) + \boldsymbol{\Xi}(\boldsymbol{p})]. \tag{9.5.3}$$

Considering that $\dot{\Lambda}(p) = p\Lambda(p)$, for error of measurement of speed transformed according to Laplace we have

$$D(p) = \dot{\lambda}(p) - \hat{\lambda}(p) = \frac{p + (p - H(p))K_{A}G(p)}{1 + K_{A}G(p)} \Lambda(p) - \frac{K_{A}G(p)H(p)}{1 + K_{A}G(p)} \Xi(p).$$
(9.3.4)

The first term in the direct part of (9.8.4) depicts dynamic error, and the

second - fluctuation error. Then dynamic error of measurement of speed can be found by the formula

$$\Delta_{ABR}(l) = \int_{0}^{l} v(t-\tau) \lambda(\tau) d\tau, \qquad (9.8.5)$$

where v(t) - reverse Laplace transform of

$$V(p) = \frac{p + (p - H(p)) K_{x}G(p)}{1 + K_{x}G(p)}.$$
 (9.8.6)

Variance of fluctuation error of measurement of speed in steady-state operating conditions $(t \rightarrow \omega)$, according to (9.8.4), is defined as

$$\overline{\Delta_{\phi,x}^2} = \frac{1}{2\pi} \int_{-\infty}^{\infty} \left| \frac{K_{a}G(i\omega) H(i\omega)}{1 + K_{a}G(i\omega)} \right|^2 S_{\mu} d\omega.$$
(9.8.7)

As also earlier, dynamic error can be considered statistically, or for unknown statistics of $\lambda(t)$ we can determine it from model disturbances. In the statistical approach we consider assigned the mathematical expectation of coordinate $\overline{\lambda(t)}$ and its correlation function $R_{\lambda}(t, \tau)$. Then statistical characteristics of dynamic error of measurement of speed can be calculated by formulas known from the theory of random processes [9]. In particular, when $R_{\lambda}(t, \tau) = R_{\lambda}(t - \tau)$ (condition of stationariness of the random part of $\lambda(t)$) variance of dynamic error in steady-state operating conditions is

$$J_{\mu\nu\nu}^{2} = \frac{1}{2\pi} \int_{-\infty}^{\infty} |V(i\omega)|^{2} S_{\lambda}(\omega) d\omega, \qquad (9.8.8)$$

where $S_{\lambda}(\omega)$ - Fourier transform of $h_{\lambda}(\omega)$.

The mathematical expectation of dynamic error is

$$\overline{\Delta_{Auu}(t)} = \int v(t-\tau)\overline{\lambda(\tau)}d\tau. \qquad (9.8.9)$$

In the nonstatistical approach to dynamic error of measurement of speed we present coordinate $\lambda(t)$ in the form of a polynomial of a certain degree n. We shall limit our consideration to steady-state operating conditions. Then, using the formula known from theory of dynamic systems [44], we shall present dynamic error in the form

$$\Delta_{\text{ARK}}(l) := \sum_{k=0}^{\infty} v_k \lambda^{(k)}(l), \qquad (9.8.10)$$

where $\lambda^{(k)}(t) = d^k \lambda(t) / dt^k$ - derivatives of some assigned model law of change of coordinate $\lambda(t)$; coefficients $\sigma_k = \frac{1}{k!} \frac{d^k V(p)}{dp^k} \Big|_{p=0}$.

-472-

With presentation of $\lambda(t)$ by a polynomial of degree n series (9.8.10) includes only the first n + 1 terms, and the requirement of finiteness of $\Delta_{\text{MMH}}(t)$ as $t \to \infty$ leads to condition $\mathbf{v}_0 = \mathbf{v}_1 = \dots = \mathbf{v}_{n-1} = 0$.

Usually to guarantee smallness of dynamic errors of measurement of coordinate $\lambda(t)$ there are applied smoothing circuits possessing astaticism of a certain order m, i.e.,

$$G(p) = \frac{G_1(p)}{p^m},$$
 (9.8.11)

where $G_1(p)$ does not contain factors 1/p.

Substituting (9.8.11) in (9.8.6), we have

$$V(p) = \frac{p^{m+1} + [p - H(p)] K_{\Lambda}G_{1}(p)}{p^{m} + K_{\lambda}G_{1}(p)}.$$

We require that

$$p - H(p) = p^{m+1}H_1(p), \qquad (9.8.12)$$

where $H_1(0)$ - finite quantity; then

$$V(p) = p^{m+1} \frac{1 + K_{\lambda} H_{1}(p) G_{1}(p)}{p^{-} + K_{\lambda} G_{1}(p)}$$
(9.8.13)

and the first m + 1 coefficients in (9.8.10) are equal to zero ($v_0 = v_1 = \dots = v_m = = 0$).

If the polynomial by which we present coordinate $\lambda(t)$ has a degree not higher than m, there is ensured zero dynamic error of measurement of speed. When n = m + 1dynamic error is determined by coefficient

$$v_{m+1} = \frac{1 + K_A H_1(0) G_1(0)}{K_A G_1(0)}, \qquad (9.8.14)$$

Condition (9.8.12) is satisfied, in particular, with application of a purely differentiating filter, when $H_1(p) = 0$ and H(p) = p. Here, c efficient v_{m+1} , and consequently, dynamic error, too, reaches its minimum at n = m + 1. This, however, does not mean that it is necessary to apply ideal differentiation of distance and angles, since from (9.8.7) it follows that fluctuation error decreases if $[H(i\omega)/i\omega]$ has a steep drop at high frequencies. When $H(i\omega) = i\omega$ this dip, in general, is absent, and for certain forms of frequency response of the filter of the tracking meter of coordinate $G(i\omega)$ variance of fluctuating error of measurement of speed is formally obtained as infinite. Turning to filters, not carrying out ideal differentiation, one should note that for large m filters can be rather complicated. Then, proceeding to certain increase of dynamic errors, it is reasonable to impose on the filter forming the estimate of speed the condition

-473-

$$p - H(p) = p^{k+1}H_{1}(p), \qquad (9.8.15)$$

where $H_1(0)$ - finite quantity, and k < m. Here, $v_0 = v_1 = \ldots = v_k = 0$, and

$$v_{k+1} = H_1(0).$$
 (9.8.1c)

(9.8.17)

The question of selection of the filter for formation of the speed estimate we shall consider later in the course of synthesis of an optimum meter; however, we first shall examine with examples errors of measurement of speed for certain normal forms of filters.

1. Let us assume that the meter of distance or angle as its filter contains an integrator, i.e., $K_{\rm A}G(p) = K/p$, where K — overall gain factor of the open trackic system, having dimensionality 1/sec. Let us assume that as the filter intended for measurement of speed there is used a differentiating RC-circuit, i.e., H(p) == p/1 + pT. Consideration of this case gives us the possibility of obtaining all the necessary results for pure differentiation, too, if we set T = 0.

According to (9.8.7) variance of fluctuating error

$$\overline{\Delta_{\phi \pi}^{2}} = \frac{1}{2\pi} \int_{-\infty}^{\infty} \frac{K^{2} \omega^{2} S_{0KB} d\omega}{(K^{2} + \omega^{2})(1 + \omega^{2}T^{2})} = \frac{K S_{\pi KB}}{2} - \frac{K}{T(1 + KT)} = \frac{c_{\phi \pi}^{2} K^{2} \frac{1}{KT(1 + KT)}}{c_{\phi \pi}^{2} K^{2} \frac{1}{KT(1 + KT)}},$$

where $\sigma_{\psi\pi}^2 = S_{BBB} \frac{K}{2} = 2S_{BBB} \Delta f_{BW} - \text{variance of fluctuating error of measurement of coordinate b(t);}$ $\Delta f_{BW} = \frac{K}{4} - \text{effective banawidth of the closed tracking system.}$

From expression (9.8.17) it follows that fluctuating error of measurement of sided decreases with growth of the time constant T of the differentiating filter. With pure differentiation (T = 0) variance of fluctuating error turns into infinity. This is obtained, of course, due to the accepted idealizations, since we considered $\xi(t)$ white noise. In reality, due to contain inertia of the discriminator this is not white noise. We can approximately assume that the spectral density of $\xi(t)$ is constant in the band of the discriminator $\Delta f_{\rm H}$, the magnitude of which is determined by the form of operations and parameters of every given discriminator, and is equal to $S_{\rm HEP}$ and beyond this band it is equal to zero. Then integration in (9.8.17) must be produced within limits (-2# $\Delta f_{\rm H}$, +2# $\Delta f_{\rm H}$), and

$$\overline{\Delta_{\phi \pi}^{2}} = \sigma_{\phi \pi}^{2} K^{2} \frac{1}{KT (1 - K^{2} T^{2})} \left(\frac{2}{\pi} \operatorname{arc} \operatorname{tg} 2\pi \Delta f_{\pi} T - KT\right). \quad (9.8.16)$$

With ideal differentiation, i.e., when T = 0, we obtain

$$\overline{\Delta_{\phi a}^{2}} = s_{\phi a}^{2} K (4\Delta f_{A} - K). \qquad (9.8.19)$$

Thus, variance of fluctuating error of measurement of speed in this case, too, is finite and is determined by the bandwidth of the discriminator $\Delta f_{\rm I}$. With a coherent signal $\Delta f_{\rm I}$ is a quantity of the order of the bandwidth of the narrow-band filters of the discriminator; for an incoherent signal $\Delta f_{\rm I} = F_{\rm r}$, where $F_{\rm r} = f$ requency of repetition.

In Fig. 9.33 are curves of the dependence of the relative fluctuation error of measurement of speed $\sqrt{\Delta_{q,n}^2}/\sigma_{q,n}K$ on KT for various ratios of bandwidths of the



Fig. 9.33. Dependence of relative fluctuating error of measurement of speed on the product of the gain factor of an open system K for the time constant of the differentiator T with various $\Delta f_{\rm II}/\Delta f_{\rm aff}$.

discriminator and the closed system of measurement of the coordinate $\Delta f_{\Pi} / \Delta f_{\Theta \Phi} = 4\Delta f_{\Pi} / K$. Curves are constructed for values $\Delta f_{\Pi} / \Delta f_{\Theta \Phi} >> 1$. To estimate the order of magnitudes of fluctuating errors of measurement of speed we consider that very often $\Delta f_{\Theta \Phi} = (1 \text{ to } 10) \text{ cps}, \text{ i.e.},$ (K = 4 to 40) 1/sec. besides, if we express $\sigma_{\Phi \Pi}$ in meters, to obtain $\sqrt{\Delta f_{\Pi \Pi}}$ m/sec we must multiply $\sigma_{\Phi \Pi}$ by a coefficient, usually Thus, for instance, when K = 10 graph of Fig. 9.33 we determine

numerically equal to several units or even tens. Thus, for instance, when K = 10 1/sec ($\Delta f_{gff} \approx 2.5$ cps) and with KT = 1 from the graph of Fig. 9.33 we determine that $\sqrt{\Delta_{fff}^2} = 70$ [m/sec].

We now turn to consideration of dynamic errors. Substituting expressions (G(p) and H(p) in (9.8.6), we have

$$V(p) = p^{a} \frac{1 + KT + pT}{(K + p)(1 + pT)}, \qquad (9.8, 20)$$

from which $v_0 = v_1 = 0$, and coefficient v_2 in formula (9.8.10) for dynamic error is equal to

$$\boldsymbol{o}_{1} = \frac{1+KT}{K}.$$

$$(9.6.71)$$

$$\lambda_{1}t + \frac{1}{2}\lambda_{2}t^{2}$$

Then, for instance, when $\lambda(t) = \lambda_0 + \lambda_1 t + \frac{1}{2} \lambda_2 t$

$$\Delta_{\text{HH}} = \frac{\lambda_2}{K} (1 + KT). \qquad (9.8.22)$$

From the obtained expressions it follows that $\Delta_{\rm HVH}$ decreases with decrease of T. However, this dependence is relatively weak, therefore one should not select magnitudes of KT which are too small, in order not to increase fluctuating error to a considerable degree. Selection of the magnitude of KT is best produced proceeding. from the requirement of minimum total error of measurement of speed. For instance, if $\lambda(t)$ is quadratic, $\Delta_{\rm HVH}$ is determined by expression (9.8.22), and the optimum 'value of KT is found from condition $\Delta^2 = \Delta_{\rm HZH}^2 + \overline{\Delta_{\rm CM}^2} = \min$. Here we obtain the dependence of (KT)_{CHT} on $x = \sigma_{\rm CH} R^2/\lambda_2$, shown in Fig. 9.34. For small x quantity (KT)_{CHT} is approximately proportional to x; for large x it is proportional to \sqrt{x} . In the same figure we show the dependence of the relative magnitude of total error $\sqrt{\lambda_{\rm HMH}^2 + \Delta_{\rm CH}^2}$ Ko_{dAl} on x. If in this system gain of the open loop is selected optimally from the condition of minimum error of measurement of coordinate $\lambda(t)$, then, as it is easy to show,

$$K = \left(\frac{4\lambda_1^2}{S_{3\,8\,8}}\right)^{1/3}$$

and

 $\sigma_{\phi n} K = V \overline{2} \lambda_{\mu}.$

Then from Fig. 9.34 it is clear that for small x, taking place with high socializations of the target λ_p , error of seasurement of speed can exceed λ_1 , i.e.,



Fig. 9.34. Dependence of the optimum value of product (KT) (solid curve) and relative total error of measurement of speed (dotted curve) on parameter $\mathbf{x} = \sigma_{\mathbf{f},\mathbf{h}} \mathbf{k}^2 / \lambda_0$.

measurement of speed loses meaning. For small accelerations, i.e., large x and with optimum selection both of K, and also of KT, it is easy to find that error of measurement of speed is equal to

$$\lambda \approx 1,35\lambda_1^{1/6}\lambda_2^{1/2}S_{ska}^{1/6}$$
. (9.8.22)

The magnitude of error has considerably stronger dependence on acceleration λ_2 than on λ_1 and S_{2KB} . If, for instance, in a goniometer system $S_{2KB} = 0.1 \ \text{deg}^2/\text{cps}$ and $\lambda_1 = 2 \ \text{deg/sec}$, $\Delta = 0.3 \ \text{deg/sec}$ with acceleration

-475-

 $\lambda_2 = 0.1 \text{ deg/sec}^2$; $\Delta = 0.2 \text{ deg/sec}$ with $\lambda_2 = 0.04 \text{ deg/sec}^2$ and $\Delta = 0.1 \text{ deg/sec}$ with $\lambda_2 = 10^{-2} \text{ deg/sec}^2$.

2. Let us assume that the smoothing filter of the tracking meter c: distance or angle is a double integrator with correction. Then

$$K_{R}G(p) = \frac{K(1+pT_{R})}{p^{2}},$$
 (9.8.24)

where $K = gain factor of the open loop, having dimensionality <math>1/sec^2$;

 T_v - time constant of the correcting circuit.

We shall consider that for formation of the estimate of speed, as also in the preceding example, there is used a differentiating RC-circuit, i.e., $H(p) = p/1 + \frac{1}{4}T$. Below we shall show that the greatest accuracy of measurement of speed is attained at $T = T_{\rm K}$; therefore, we assume that this condition is realized. Considering equaltity $K_{\rm R}G(p)H(p) = K/p$, we note that with observance of the mentioned conditions there is no need to install a special differentiating filter. As the estimate of speed $\lambda^{\dagger}(t)$ it is possible to use the output of first integrator of the smoothing circuits of the tracking system (Fig. 9.35). Here, formulas errors of measurement,



obtained above, remain valid. According to (9.8.7) variance of fluctuating error of measurement of speed

$$\overline{\Delta_{\phi\eta}^{2}} = S_{\theta KB} \frac{1}{2\pi} \int_{-\infty}^{\infty} \frac{K^{3} \omega^{3} d\omega}{(\omega^{3} - K)^{2} + K^{3} T_{K}^{2} \omega^{2}} = \frac{K S_{\theta KB}}{2T_{K}} =$$

$$= 16 \sigma_{\phi\eta}^{2} \Delta f_{\theta \phi}^{2} \frac{K T_{K}^{3}}{(1 + K T_{K}^{2})^{3}}, \qquad (9.8.25)$$

where

 $\sigma_{\eta,\eta}^2 = 2S_{arg} \Delta f_{afg} - variance of fluctuating error of measurement of coordinate <math>\lambda(t)$;

 $\Delta f_{gg} \approx (1 + KT_{R}^{2})/4T_{R}$ - effective bandwidth of the closed tracking system.

The dependence of the relative magnitude of fluctuating error $\sqrt{\Delta_{q,n}^2} \delta_{q,n}^2 \delta_{q,n}^2$



fluctuating error of measurement of

the speed of a target by a system with a smoothing filter consisting of two integrators with correction

on parameter KT2.

instance, when $\Delta f_{g\bar{q}\bar{h}} = 2.5 \text{ cps}$ and $\mathrm{KT}_{\mathbf{K}}^{2} = 4$, which corresponds to transition of the tracking system from oscillatory to aperiodic, $\sqrt{\Delta_{q\bar{q}\bar{n}}^{2}} = 1.8 \sigma_{q\bar{q}\bar{n}}$; then, as in the example given for the first-order system, this error composes $7 \sigma_{q\bar{q}\bar{n}}$.

Calculating dynamic error, we have

$$V(p) = p^{*} \frac{p + KT_{*}}{p^{*} + K(1 + pT_{*})}, v_{0} = v_{1} = 0, v_{2} = T_{K}.$$

$$(9.8.26)$$

With a square law of change of $\lambda(t)$ steadystate error is equal to

$$\mathbf{X}_{\mathbf{K}\mathbf{K}\mathbf{S}} \coloneqq \mathbf{T}_{\mathbf{K}} \boldsymbol{\lambda}_{\mathbf{s}}. \tag{9.8.27}$$

We will not study finding of optimum relationships between parameters of system

here, leaving this question to synthesis of optimum speed meters.

Of interest is consideration of an ideal differentiating filter for the same system. In this case

$$V(p) = \frac{p^{2}}{p^{2} + K(1 + pT_{s})}, \ v_{0} = v_{1} = v_{2} = 0, \ v_{3} = \frac{1}{K}.$$
(9.8.2b)

Dynamic error decreases as compared to the preceding case; and for quadratic $\lambda(t)$ it is generally absent. However, : Luctuating error increases and, if we do not consider finiteness of the width of spectrum $\xi(t)$, it turns into infinity. Allowing as before for the form of finiteness of the discriminator band, we find

$$\overline{\Delta_{\phi n}^2} = K^* T_{\kappa}^2 S_{\mu \kappa \mu} \Delta f_{\pi} = \sigma_{\phi n}^2 \Delta f_{s\phi}^2 \frac{\Delta f_{\pi}}{\Delta f_{s\phi}} \frac{8 (\kappa T_{\kappa}^2)^s}{(1 + \kappa T_{\kappa}^2)^s}.$$
(9.8.29)

Fluctuating error is increased as compared to the case $T = T_R$ by a factor of $\sqrt{0.5KT_R^2} (1 + KT_R^2) \Delta f_R / \Delta f_{\partial \Phi}$. With the above-mentioned selection of magnitudes of ET_R^2 this comprises (1 to 5) $\sqrt{\Delta f_R} / \Delta f_{\partial \Phi}$, which corresponds to large loss in the accuracy of measurement of speed.

3. A form of smoothing circuits very wide-spread in practice is a circuit with transfer function

$$K_{\mathbf{x}}G(p) = \frac{K(1+pT_{\mathbf{x}})}{p(1+pT_{\mathbf{x}})(1+pT_{\mathbf{x}})}.$$
 (9.8.30)

Time constants here usually satisfy condition $T_1 > T_R \to T_2$ at which stability

-478-

of the system is ensured for any K. Frequently ratio T_2/T_K is selected in the range 0.1 to 0.5 and T_1/T_K in the range 2 to 50. The order of magnitudes of KT_K and T_1/T_2 is usually identical. The effective bandwidth of the closed system turns out to be equal to

$$\Delta f_{ab} = \frac{K}{4} \frac{T_1 + T_2 + KT_K^2}{T_1 + T_2 + K(T_1T_K + T_2T_K - T_1T_2)}.$$
(9.8.31)

Finding errors of measurement of speed for such a system on the assumption that there is executed ideal differentiation (H(p) = p), we have

$$\overline{\Delta_{\phi n}^{2}} = \frac{1}{2\pi} \int_{-\infty}^{\infty} \frac{K^{2}\omega^{2} (1 + \omega^{2}T_{K}^{2}) S_{3KB} d\omega}{[i\omega (1 + i\omega T_{3})(1 + i\omega T_{2}) + K (1 + i\omega T_{K})]^{2}} = \frac{K^{2}S_{9KB} [T_{3}T_{3} + T_{K}^{2} (1 + KT_{K})]}{2T_{3}T_{3} [(T_{3} + T_{3}) (1 + KT_{K} - KT_{3}T_{3}]} = \frac{g_{\phi n}^{2} K^{2} \frac{1 + KT_{n} + \frac{T_{1}}{T_{N}} \frac{T_{2}}{T_{N}}}{KT_{n} (KT_{n} + \frac{T_{1}}{T_{N}} + \frac{T_{3}}{T_{N}})}.$$
(9.8.32)

Calculating coefficients of dynamic error, we obtain

$$v_0 = v_1 = 0, v_0 = \frac{1}{K},$$
 (9.8.33)

i.e., for quadratic $\lambda(t)$ dynamic error is

$$\Delta_{\text{REM}} = \frac{\lambda_{\text{e}}}{K}, \qquad (9.8.34)$$

as also in a system with a single integrator. However, by selection of time constants T_1 , T_2 and T_R in the considered case with an assigned effective bandwidth we can obtain a large value of coefficient K and thereby decrease dynamic error. In practice it is possible to increase gain K by a factor of approximately T_1/T_R as compared to a system with one integrator. The dependence of relative fluctuating error on parameter KT_R for different T_1/T_R and T_2/T_R is shown in Fig. 9.37. It is easy to see that with change of KT_R in a wide range $\sqrt{\Delta_{QT}^2} = (0.1 \text{ to } 0.5)\sigma_{QT}K$. The magnitude of error decreases with growth of T_1/T_R and with decrease of T_2/T_R ; however, these dependences are weak. Although the considered system with ideal differentiation for formation of the estimate of speed ensures higher accuracy than a system with a single integrator, this accuracy is nevertheless low. Thus, for instance, when K = (100 to 500)1/sec fluctuating error $\sqrt{\Delta_{QT}^2} = (10 \text{ to } 250)\sigma_{dT}$, which is usually absolutely unacceptable.



Fig. 9.37. Dependence of relative fluctuating error of measurement of speed by a system with a smoothing filter of the second order and ideal differentiation on parameter KT_{R} : --- T_{2}/T_{R} = 0.1; ---- $T_{2}/T_{H} = 0.5.$

Fluctuating error can be substantially lowered by applying, e.g., a differentiating RC-circuit [H(p) = $= p/1 + pT_{R}$]. Then

$$\overline{\Delta_{\phi n}^2} = \sigma_{\phi n}^2 K^4 \frac{1}{KT_{\pi} \left(KT_{\pi} + \frac{T_1}{T_{\pi}} + \frac{T_2}{T_{\pi}} \right)}.$$

(9.8.35)

In Fig. 9.38 there are shown curves of the dependence of relativ flu 'unting error on KT_{H} for various T_1/T_R and for $T_2/T_R = 0.2$. With those same values of parameters as in the preceding case fluctuating error is much less $\sqrt{\Delta_{m\pi}^2} = (0.01 \text{ to})$ $0.2)\sigma_{m\pi}K$; however, dynamic error increases. The first coefficient of error differing

from zero $v_2 = 1 + KT_K / K$, i.e., with quadratic $\lambda(t)$, error Δ_{RSH} increases by $1 + KT_{H}$ as compared to the preceding case.

Let us consider numerical examples:

(a) Let us assume that we are measuring radial speed as the derivative of the range. Let us assume that the range firster has a smoothing filter of the lastconsidered type with parameters K = 200 1/sec, $T_1 = 1$ sec, $T_2 = 0.01$ sec,



Fig. 9.38. Dependence of relative fluctuating error of measurement of speed by a system with a smoothing filter of the second order and a differentiating RC-circuit on parameter KT_K.

 $T_{v} = 0.1$ sec, and the transfer function of the differentiating filter is $H(p) = p/1 + pT_{w}$. Let us assume that radial speed of the target can attain a value

 $\lambda_1 = V_r = 1000 \text{ m/sec}$, and radial acceleration can reach $\lambda_2 = 10 \text{ m/sec}^2$. For the selected values of the parameters $\Delta f_{ab} = 8 \text{ cps}$,

$$a_{\phi\pi} = 4 \sqrt[4]{S_{\phi\pi}} [m], \sqrt{\overline{\Delta_{\phi\pi}^2}} = 32,6 \sqrt{S_{\pi\pi\pi}} \left[\frac{m}{sec} \right],$$

where $S_{3K_{23}}$ is expressed in m²/cps. Dynamic errors of measurement of distance and speeds are equal to

 $s_{\text{RMM}} = (5 + 0.05t) \text{ m}, \Delta_{\text{RMM}} = 1.05 \frac{\text{m}}{\text{sec}},$

Table 9.5a S_{3KB}, m² cps 100 3 10 30 1 4 6.9 12.6 21.8 40 ¶đn, ^m $V = \frac{1}{\Delta_{(1)}^2}, \frac{m}{sec}$ 32. 6 56,5 103 178 326

and values of fluctuating errors are given in Table 9.5a.

1 1

b) Let us assume that the tangential component of speed is measured by differentiation of the angular coordinate, and filters have the same form and the same parameters as in the preceding example. Let us assume that angular velocity is characterized by quantity $\lambda_1 = 2 \text{ deg/sec}$, and angular acceleration is characterized by $\lambda_2 = 0.1 \text{ deg/sec}^2$. Here, dynamic errors of measurement of the angle and angular velocities are equal to

 $e_{Aux} = (1 + 0.05t) 10^{-1} \text{ deg}, \quad \Delta_{Aux} = 1.05 \cdot 10^{-1} \frac{\text{deg}}{\text{sec}},$

and fluctuating errors are found by the same formulas, in which S_{3KB} is expressed in deg²/cps. Their values are given in Table 9.5b.

20020 9190					
Sake, deg ²	10~*	3.10-1	10,-*	3.10-*	10-1
o _t , deg	0,126	0,218	0,4	0,69	1,26
V 32 deg	1,03	1,78	3,26	5,65	10,3

Table 9.5b

From the given examples it is clear that in a number of cases accuracy of measurement of speed by the considered method is low. However, it essentially

-481-

depends on the method of construction of systems and selection of their parameters. For estimating potential possibilities of measurement of speed as the derivative of the coordinate it is necessary to turn to an optimum meter and its properties.

9.8.2. Optimum Speed Meter

Solution of problem of synthesis of an optimum system of measurement of speed as the derivative of a coordinate with Gaussian statistics of the coordinate $\lambda(t)$ ensues from results obtained in Chapter VI (Paragraphs 6.6.9 and 6.8.5) concerning optimum systems of measurement of functionals of parameters of a random process.

According to these results the optimum speed meter is the scheme of Fig. 9.39. It consists of an optimum meter of the actual coordinate $\lambda(t)$, the principle of construction and action of which already has been repeatedly discussed, and an additional filter, serving to form the estimate of speed $\hat{\lambda}'(t)$, to whose input there is fed voltage from the output of the discriminator z(t). Generally speaking, there should also take place formation of the mathematical expectation of speed to which the output quantity of the filter is added.



Fig. 9.39. Optimum meter for measuring speed as the derivative of the coordinate: 1 - discriminator; 2 - smoothing circuits;3 - filter estimating speed; 4 - adder.

Of greatest interest is synthesis of the optimum filter for estimating speed, since remaining operations of the meter either are already known from synthesis of the meter of coordinate $\lambda(t)$, or are trivial. Pulse response $h(t, \tau)$ of the filter estimating speed can be found simultaneously with the pulse response of the filter

of the tracking meter of the coordinate $g(t, \tau)$ from solution of equations (6.6.10b), (6.6.108), (6.6.31), and (6.6.32), if we consider that function F(t, s) in this case is defined as

$$F(t, s) = \delta'(t-s),$$
 (9.8.36)

where the stroke is the sign of differentiation.

Then these equations have the form

$$h(t, \tau) = b(t, \tau) + K_{out} \int_{0}^{t} b(t, s) g(s, \tau) ds, \qquad (9.8.37)$$

 $b(t, \tau) + K_{on\tau} \int b(t, s) R(s, \tau) ds = \frac{1}{dt}$ (b(t, \tau) = 0 when t < \tau), (9.8.38)

-482-

$$g(t, \tau) = c(t, \tau) + K_{out} \int_{0}^{t} c(t, s) g(s, \tau) ds,$$
 (9.8.39)

$$c(t, \tau) + K_{OHT} \int_{t_0}^{t_0} c(t, s) R(s, \tau) ds = R(t, \tau)$$

(c(t, \tau)=0 when t < \tau), (9.8.40)

where $K_{0\Pi T}$ - gain factor of the optimum discriminator for measurement of coordinate $\lambda(t)$;

 $R(t, \tau)$ - correlation function of coordinate $\lambda(t)$.

Equations (9.8.39) and (9.8.40) determine the optimum structure of smoothing circuits of the meter of $\lambda(t)$, and equations (9.8.37) and (9.8.38) determine the structure of additional circuits, needed for estimating the derivative $\lambda^{\dagger}(t)$.

Actually filters with pulse responses $g(t, \tau)$ and $h(t, \tau)$ usually have many common elements, which simplifies the circuit of the optimum meter, making, at the same time, somewhat conditional the diagram of Fig. 9.39. Furthermore, it is possible for production of the estimate of speed to carry out filtration, not of voltage from the discriminator output, but the output quantity $\hat{\lambda}(t)$ of the coordinate-tracking meter, but here there should be applied a filter with pulse response $\vartheta(t, \tau)$ determined from equation

$$h(t, \tau) = \int \vartheta(t, s) g(s, \tau) ds. \qquad (9.8.41)$$

In accordance with (6.6.109) variance of error of measurement of speed by the optimum system is defined as

$$\mathbf{a}_{\mathbf{\lambda}}^{2} = \frac{db\left(t, \mathbf{x}\right)}{d\mathbf{x}}\Big|_{\mathbf{x}=t}.$$
(9.8.42)

After giving these general results we consider certain examples of optimum speed meters. We assume that the form and basic characteristic $K_{ONT} = 1/S_{ONT}$ of the optimum discriminator are known. Let us calculate here the pulse responses of smoothing filters and error of measurement of speed.

1. Let us assume that $\lambda(t)$ be a stationary random process or a process with stationary increments and that we are interested in moments of time very far removed from the beginning of observation t_0 . Then in accordance with results of Paragraph. 6.8.5 solutions of equations (9.8.38) and (9.8.40) can be presented by their Fourier transforms

$$C(i\omega) = \frac{1}{\Psi(i\omega)} \left[\frac{S(\omega)}{\Psi(-i\omega)} \right]_{+},$$

$$c(l-\tau) = \frac{1}{2\pi} \int_{-\infty}^{\infty} C(i\omega) e^{i\omega(l-\tau)} d\omega, \qquad (9.8.4\%)$$

-483-

$$B(i\omega) = \frac{1}{\Psi(i\omega)} \left[\frac{i\omega S(\omega)}{\Psi(-i\omega)} \right]_{+};$$

$$b(t-\tau) = \frac{1}{2\pi} \int_{-\infty}^{\infty} B(i\omega) e^{i\omega(t-\tau)} d\omega, \qquad (9.8.44)$$

where $S(\omega)$ - spectral density of process $\lambda(t)$;

 Ψ (im) Ψ (- iw) = 1 + K_{ont} S (w),

where $\Psi(i\omega)$ has zeroes and poles in the upper half-plane of complex variable ω , and $\Psi(-i\omega)$ has them in the lower. Operation []₊ signifies singling out of that part of the expression in parentheses which has poles in the upper half-plane of ω . A solution of the problem of measurement of the derivative exists if $S(\omega)$ at infinity seeks zero not slower than $1/\omega^4$. Functions $g(t - \tau)$ and $h(t - \tau)$ are determined according to (9.8.39) and (9.8.37) by their Fourier transforms with the help of expressions

$$G(i\omega) = \frac{C(i\omega)}{1 - K_{our}C(i\omega)}, \quad H(i\omega) = B(i\omega) [1 + K_{our}G(i\omega)]. \quad (9.8.45)$$

If, for instance, $\lambda(t)$ corresponds to motion of the target with uncorrelated random accelerations, caused by various random disturbances, where the mean square speed of the target developed in 1 sec due to the random disturbances is equal to B_0 , then $\lambda(t)$ - double integral of white noise with spectral density B_0 , and spectral density $S(\omega) = B_0/\omega^4$. Direct application of the written equalities here gives

$$G(i = p) = G(p) = \frac{K(1 + pT_{\mu})}{p^{4}},$$

$$H(p) = \frac{K}{p},$$

$$K = \sqrt{\frac{B_{0}}{K_{0}}}, T_{\mu} = \left(\frac{4}{B_{0}K_{0}}\right)^{1/4}.$$
(9.8.46)

where

Thus, the optimum smoothing filter of the meter tracking coordinate $\lambda(t)$ is a double integrator with correction, which was noted for the case of the $\lambda(t)$ -double integral of white noise and earlier, and the optimum filter for estimating speed is an integrator with gain K. As a result the signal which is the estimate of speed $\lambda'(t)$ can be taken from the output of the first integrator of the filter of the tracking system, or be formed by means of transmission of the output of the coordinate-tracking meter $\hat{\lambda}(t)$ through a differentiating RC-circuit with transfer function $\Theta(p) = p/1 + pT_{R}$. Consequently, the system analyzed in the preceding

-484-

paragraph, with $T = T_{R}$ is optimum for the case when the target moves with uncorrelated accelerations. Parameters of the optimum system K and T_{R} are not arbitrary, but according to (9.8.46) are functions of the intensity of accelerations B_{Q} , characterizing maneuverability of target, and of the intensity of equivalent noises at the input of the linearized discriminator $S_{OUT} = 1/K_{OUT}$.

Using the fact that variance of measurement of coordinate $\sigma_{\lambda}^2 = c(t, t)$, while variance of measurement of speed is determined by expression (9.8.42), we obtain

$$\sigma_{\lambda}^{2} = \left(\frac{4B_{o}}{K_{onr}^{3}}\right)^{1/4}, \quad \sigma_{\lambda'}^{2} = \left(\frac{4B_{0}^{3}}{K_{onr}}\right)^{1/4}.$$
(9.8.47)

From formulas (9.8.47) it is clear that error of measurement of speed depends more on maneuverability of the target (B_0) and less on the level of noises $1/K_{OHT}$ than error of measurement of the coordinate. For illustration we calculated error of measurement of distance (σ_d) and of radial speed (σ_d) of an optimum system for two values of equivalent spectral density S_{OHT} and several values of B_0 .

Results of the calculation are given in Table 9.6. In this table there are given values of the gain factor of the open loop, the time constant of the correcting circuit and the effective bandwidth of the system.

Table	9,6
-------	-----

$S_{oxe} = \frac{i}{K_{oxe}} \left[\frac{A^{*}}{Cps} \right]$	104				104					
$B_{\phi}\left[\frac{M^{b}}{\sec^{1}}\right]$	1	10	10 ^e	10*	104	1	10	10ª	10•	10*
$K_{\bullet \pi \tau} K = \gamma \overline{B_{\bullet} K_{\bullet \pi \tau}} \left[\frac{1}{\sec^{\pi}} \right]$	0,1	0,316	1	3,16	10	0.01	0,0316	0,1	0,316	1
7 к (зес)	4,5	2,8	1,41	0,79	0,45	14,1	7.9	4,5	2.5	1,41
. Δ ί •• [cps]	0,17	0,3	Q, 5 3	0,95	1,7	0,053	0,095	0,17	0.3	0,53
₹d [M]	6,7	9	12	16	21	37,5	50	67	90	118
$\mathbf{e}_{d}, \left[\frac{M}{\text{sec}}\right]$	1,6	3,8	9	21,4	50	2,9	6,8	16,2	38,5	90

-485-
Errors given in the table characterize both fluctuation and dynamic components; moreover the latter is understood in the statistical sense. From the table it follows that change of the magnitude of S_{OUT} two orders leads to change of error of measurement of distance by a factor of 5.6 and of error of measurement of speed by a factor of 1.8. Even with high maneuverability of the target accuracy of measurement of speed is relatively high.

The obtained results give the possibility of comparing potential accuracy of measurement of radial speed by Doppler frequency and by the derivative of distance in the case of coherent radiation. In the considered example speed is the integral of white noise, and according to (9.5.34) variance of error of its measurement by the Doppler frequency shift is equal to

$$\mathbf{o}_{\mathbf{p}}^{2} = \sqrt{\frac{B_{\mathbf{q}}}{K_{\mathbf{q}}}}, \qquad (9.8.48)$$

where $1/K_v = S_{v \text{ OFT}}$ - equivalent spectral density for an optimum frequency discriminator. For large signal-to-noise ratios, according to (9.5.29)

$$\frac{1}{K_{\bullet}} = S_{\nu \text{ our}} = \frac{\lambda^{\bullet} \Delta f_{\bullet}}{8\pi^{\bullet}}.$$
 (9.8.49)

Here λ - wavelength; Δr_c - effective width of the spectrum of fluctuations of the signal.

Variance of error of measurement of speed as the derivative of distance is determined by (9.8.47) and depends on K_{OUT} . For large signal-to-noise ratios h according to (7.2.15) and (7.2.16)

$$K_{ont} = \frac{1}{S_{ont}} = \frac{8\beta^2 h \Delta f_o}{c^2},$$
 (9.8.50)

where o - speed of light;

 β - mean square width of the spectrum of modulation of the signal utilized for measurement of distance.

Thus, the ratio of errors is

$$\frac{\mathbf{e}_{d'}}{\mathbf{e}_{\mathfrak{p}}} = \left(\frac{4B_{\mathfrak{p}}K_{\mathfrak{p}}^{2}}{K_{\mathfrak{o}\mathfrak{n}\mathfrak{r}}}\right)^{1/8} = \left(\frac{\pi^{2}B_{\mathfrak{p}}}{\Delta f_{\mathfrak{c}}^{3}\lambda^{2}}\frac{8\omega_{\mathfrak{p}}^{2}}{\beta^{2}\hbar}\right)^{1/8}, \qquad (9.8.51)$$

where ω_0 - carrier frequency of the emitted signal.

In order to determine the usual order of magnitudes of ratios of errors, we consider that product $\Delta f_{\rm C}$ λ weakly depends on the wavelength and frequently is a magnitude of the order of 1 m/sec. The ratio of the carrier frequency to the width of the spectrum of modulation is usually very great, so that even for large

signal-to-noise ratios $\omega_0^2/h\beta^2 > 10^5$ to 10^6 . Then, considering that usually the width of the spectrum of the reflected signal has an order of units or tens of cycles per second, we find that even with small maneuverability of targets $(B_0 \sim 1 \text{ m}^2/\text{sec}^3)$ ratio $\sigma_{d1}/\sigma_v > 3$ to 5. Thus, measurement of radial speed by Doppler frequency shift of the signal ensures great accuracy where, with increase of maneuverability of the target, the gain increases. It is necessary to note that ratio σ_{d1}/σ_v weakly depends on its determining parameters; therefore even with rather strong change of them the relationships of accuracies of measurement of speed by the two discussed methods changes little.

2. Let us assume that the law of change of coordinate $\lambda(t)$ is known with an accuracy of certain random coefficients $\mu_k(k = 0, 1, 2, ..., n)$. Thus are matters, for instance, when a certain body moves in a ballistic curve. Coefficients μ_k here are random due to the randomness of the initial conditions. As we already have repeatedly indicated, frequently there are observed conditions where presentation of $\lambda(t)$ in this form is valid:

$$\boldsymbol{\lambda}(t) = \sum_{k=0}^{n} \boldsymbol{\mu}_{k} \boldsymbol{f}_{k}(t), \qquad (9.8.52)$$

where $f_k(t)$ - assigned functions.

If, for instance, $f_k(t) = (t - t_0)^k$, coefficients μ_k constitute values of $\lambda(t)$ and n its first derivative at the initial moment $t = t_0$. We assume that $\overline{\mu}_k = 0$. This does not disrupt generality, since the known mathematical expectation can always be accounted for during measurement. Then the correlation function of process $\lambda(t)$ is

$$\mathsf{R}(t,\tau) = \sum_{i,k=0}^{n} \mathcal{M}_{ik} f_i(t) f_k(\tau) = \mathsf{f}^+(t) \operatorname{Mf}(\tau), \qquad (9.8.55)$$

where $M_{ik} = \overline{\mu_i \mu_k};$

M - symmetric matrix with elements M_{ik};

$$\label{eq:f_t_n} \begin{split} \mathbf{f}(t) &= \{\mathbf{f}_0(t), \ \mathbf{f}_1(t), \ \ldots, \ \mathbf{f}_n(t)\} - \text{column vectors, and}^+ \text{ signifies transposition} \\ & \text{position } (\mathbf{f}^+(t) - \text{line vector}). \end{split}$$

Equations (9.8.37)-(9.8.40) here are equations with a degenerate nucleus and their solutions are easy. They have form:

$$c(t, \mathbf{1}) == \mathbf{f}^{+}(t) \left[\mathbf{M}^{-1} + \mathbf{U}(t) \right]^{-1} \mathbf{f}(\mathbf{\tau}) = \sum_{i,k=0}^{n} A_{ik}(t) f_{i}(t) f_{k}(\mathbf{\tau}), \qquad (9.8.54)$$

$$g(l, \tau) = l^{+}(l) [M^{-1} + U(\tau)]^{-1} f(\tau) = \sum_{l, k=0}^{n} A_{lk}(\tau) f_{l}(l) f_{k}(\tau), \qquad (9.8.55)$$

$$b(t, \tau) = \left[\frac{d}{dt}f(t)\right]^{+} [M^{-1} + U(t)]^{-1}f(\tau) = -\sum_{i,k=0}^{n} A_{ik}(t) \frac{df_{i}(t)}{dt}f_{k}(\tau), \qquad (9.8.56)$$

$$h(t, \tau) = \left[\frac{d}{dt}f(t)\right]^{+} [M^{-1} + U(\tau)]^{-1}f(\tau) = -\sum_{i,k=0}^{n} A_{ik}(\tau) \frac{df_{i}(t)}{dt}f_{k}(\tau), \qquad (9.8.57)$$

where matrix $A(t) = [M^{-1} + U(t)]^{-1}$, and matrix $U(t) = ||U_{1k}(t)||$ is determined by expression

$$U_{ik}(t) = K_{ont} \int_{a}^{b} f_{i}(s) f_{k}(s) ds. \qquad (9.8.58)$$

The block diagram of an optimum meter of coordinate $\lambda(t)$ and the rate of its change $\lambda'(t)$, corresponding to the obtained expressions, is presented in Fig. 9.40.



Fig. 9.40. Optimum speed meter with quasi-regular change of coordinate $\lambda(t)$: 1 - discriminator; 2 - variable-gain amplifier; 3 - integrator; 4 - variable-gain amplifier; 5 - adder.

To the input of the discriminator of the meter there is fed a mixture of the signal with noise $y(t, \lambda(t))$. From the output of the optimum discriminator voltage z(t) is fed to n + 1 amplifiers with variable gain factors.

The law of change of the gain factor of the i-th amplifier is determined by expression $\sum_{i=0}^{n} A_{ik}(i) f_{ik}(i)$. Output voltages of amplifiers are integrated, as a result of which there are formed estimates of coefficients μ_i . Then the voltages representing the estimates are passed through two groups of amplifiers with gain factors $f_i(t)$ and $df_i(t)/dt$ respectively. The formed voltages are added by groups. As a result of summation of voltages from outputs of the amplifiers with gain factors $f_i(t)$ there will be formed the estimate of coordinate $\hat{\lambda}(t)$, which controls tuning of the discriminator, and from those with gain factors $df_i(t)/dt$ we get the estimate of speed $\hat{\lambda}'(t)$.

Let us consider with what accuracy speed is measured by such a system. Variance of error of measurement of speed according to (9.8.42) has the form

$$\sigma_{\lambda'}^{2} = \left[\frac{d}{dt} f(t)\right]^{+} [M^{-1} + U(t)]^{-1} \frac{dt(t)}{dt} = \\ = \sum_{i, h=0}^{n} A_{ih}(t) \frac{df_{i}(t)}{dt} \frac{df_{h}(t)}{dt}.$$
(9.8.52)

Let us turn to the particular case n = 0, $f_0(t) = f(t)$. Then, designating $M_{00} = \overline{\mu_0^2} = \sigma_0^2$, we obtain

$$A(t) = \frac{1}{\frac{1}{s_0^2} + K_{our}} \int_{t_0}^{t_1} f^2(s) ds},$$

$$\sigma_{\lambda'}^2 = \frac{\left[\frac{dl}{dt}\right]^2}{\frac{1}{s_0^2} + K_{our}} \int_{t_0}^{t_1} f^2(s) ds},$$
(9.8.60)

For large t

 $\sigma_{\lambda}^{2} \simeq \frac{\left[\frac{df(t)}{dt}\right]^{2}}{K_{our} \int_{0}^{t} f^{2}(s) ds} \qquad (9.8.61)$

Usually the asymptotic value of (9.8.61) is reached rather fast. From the obtained expressions it is clear that for functions f(t) growing with unlimited increase of t no faster than any finite power of t error of measurement, as $t \rightarrow \infty$, approached zero. If, for instance, $f(t) = t^n$ and $t_0 = 0$, then $\sigma_{\lambda_1}^2 = n^2(2n + 1)/K_{OBT}t^2$. In general, when variance $\sigma_{\lambda_1}^2$ is determined by formula (9.8.2), for large times of observation of matrix

-489-

$\mathbf{A} = [\mathbf{M}^{-1} + \mathbf{U}(t)]^{-1} \cong \mathbf{U}^{-1}(t)$

and the expression for variance of error of measurement is simplified.

Thus, for instance, if $\lambda(t)$ is a polynomial of the n-th degree and $t_0 = 0$, i.e., $f_1(t) = t^1$, calculating by the given formulas, we have

$$\sigma_{\lambda}^{2} = \frac{a_{n}}{K_{0\pi\tau}t^{3}} = \frac{a_{n}}{t^{3}} S_{0\pi\tau},$$
 (9.8.62)

where values of a_n for various n are given in Table 9.7.

Table 9.7							
R ·	1	2	3	4	5.	6	7
a.,	12	192	1 200	4 800	14 700	37 600	83 800

As $t \rightarrow \infty$ error in the considered cases approaches zero. Therefore, with a quasi-regular law of change of $\lambda(t)$ by the optimum system of measurement of speed there is attained substantially higher accuracy of measurement than with a purely random law of change of $\lambda(t)$, if only the time of measurement can be sufficiently large. Here, however, more strongly than for purely random change of $\lambda(t)$, error is influenced by the equivalent intensity of noise S_{ORT} . Error turns out to be proportional to $\sqrt{S_{ORT}}$. In spite of this, for large times of measurement error is small and the given method of measurement can exceed all others in accuracy, including measurement of Doppler frequency shift. Thus, for instance, in the case of change of distance according to the law $d(t) = \mu t^n (\mu - random coefficient)$ for large times of distance, was just now found, and variance of measurement of speed by Doppler frequency shift is determined by formula (9.5.42), and, if we consider that $V(t) = n\mu t^{n-1}$, it is equal to

$$s_{p}^{2} = \frac{2n-1}{K_{*} \cdot s_{1} \cdot l}$$
 (9.8.63)

The ratio of variances is

$$\frac{q_{a}^2}{q_{a}^2} = \frac{n^2 (2n+1) K_{0.017}}{(2n-1) K_{d.017}} \frac{1}{l^4}$$
(9.8.64)

* and as $t \to \infty$ it approaches zero. In the last expressions $K_{V \text{ OHT}}$ — the gain factor of a optimum discriminator for measurement of speed by Doppler frequency shift, $K_{d \text{ OHT}}$ — the gain factor of the optimum range discriminator. It should be noted that such an effect occurs only with quasi-regular change of range.

-490-

§ 9.9. Speed Meters Under the Action of Interferences

A CONTRACTOR OF A CONTRACTOR O

Speed meters, just as other devices, in a radar, can experience not only natural noises of the system, but also all the possible interferences which were mentioned in Chapter I. In the presence of interferences accuracy of measurement of speed decreases or even the process of measurement may be disturbed [70]. This is connected with different nonlinear phenomena in elements of the meter, in particular with the phenomenon of breakoff of tracking in tracking speed meters discussed above. Disruption of the process of measurement of speed may also take place in meters using differentiation of distance or angles. Only in these forms of meters it occurs due to breakoff of tracking of range finders or goniometers. Let us consider briefly questions of the influence of interferences on speed meters.

In most cases speed meters are coherent and use during processing of signals narrow-band filters up to detection. Thus stands the matter with respect to Doppler speed meters, since, as was shown above, for an incoherent pulse signal these meters do not ensure the usually acceptable accuracy. Meters of speed as the derivative of a coordinate also in many cases are based on coherent processing of the signal. In connection with this circumstance many forms of stochastic interferences, having a spectrum width much larger than the passband of filters in the meter, are normalized during passage through filters and their action turns out to be equivalent to the action of Gaussian white noise. Here, the action of interference reduces to equivalent increase of the level of noises at the input of the radar receiver or to decrease of the signal-to-noise ratio. Therefore, in the case of action on speed meters by such interferences we can use all the preceding results for accuracy of measurement and for estimating failures of tracking speed meters. The only difference as compared to the preceding formulas is the fact that the signal-to-noise ratio h should be replaced by quantity

$$h_{\rm HH} = \frac{P_c}{2\Delta f_c \left(N_0 + N_{\rm H}\right)} = \frac{h}{1 + \frac{N_{\rm H}}{N_c}},$$
(9.9.1)

where $h_{\pi,m}$ - signal-to-noise ratio in the presence of noise and interference;

Νп

- spectral density of white noise, equivalent in its action to interference.

Further calculation of noise immunity of speed meters thus reduces to finding quantity N $_{\rm R}$, which was already done in Chapter VII during the analysis of the influence of interferences on coherent range finders. We shall give the basic formula obtained there. For noise interference, according to (7.14.3),

-491-

$$\frac{1}{h} = \frac{1}{1 + \frac{P_{\rm B}G_{\rm BR}}{2N_{\rm B}\Delta f_{\rm B}}}, \qquad (9.9.2)$$

where

 $\lambda - wavelength;$

 P_{π} - power of the jamming transmitter;

 Δf_{π} - width of the spectrum of interfernce;

 ${\tt G}_{\pi\pi}$ — antenna gain of the station of interferences;

 ${}^{G}{}_{\Pi p} = \mathop{\rm gain}_{}$ of the receiving antenna of the radar in the direction to the source of interference;

 d_{π} - distance of the source of interferences.

Magnitudes of lowering of the signal-to-noise ratio with noise interference are illustrated by graphs of Fig. 7.51.

During calculation of noise immunity of radar it is necessary to determine the critical strength of interference, and consequently also quantity $h_{\Pi\Pi} = h_{RP}$, at which one or another undersirable change of properties of system sets in. In this case such a change may be error of the speed meter reaching a certain assigned impermissible magnitude or breakoff of tracking, determined by one of the above-indicated methods. Critical intensity of noise interference $(P_{\Pi}G_{\Pi\Pi}/\Delta f_{\Pi})_{RP}$ [w/cps] can be found from (9.9.2) if we consider that $N_{O} = P_{C}/2\Delta f_{C}$ h and according to the radar formula

$$\boldsymbol{P}_{\mathbf{G}} = \frac{\boldsymbol{P}_{\mathbf{g} \in \mathbf{p}} \boldsymbol{G}^{2} \boldsymbol{\sigma}_{\mathbf{g}} \boldsymbol{\lambda}^{\mathbf{g}}}{(4\pi)^{3} d_{\mathbf{g}}^{4}}, \qquad (9.9.3)$$

where $o_{\rm H}$ - reflecting surface of the target;

G - antenna gain of radar in the direction to the target;

 $P_{\rm mer}$ - mean power of the transmitter of the radar;

 d_{ii} - distance to the target.

Under the condition of spatial coincidence of the source of interferences with the target and with the condition normally observed in conditions of measurements in the presence of only natural noise $h \gg h_{\rm KD}$ it is easy to find that

$$\begin{pmatrix} \frac{P_{a}G_{aa}}{\Delta f_{a}} \end{pmatrix}_{Rp} \approx \frac{1}{h_{Rp}} \frac{P_{Rap}G_{a}}{4\pi\Delta f_{c}d_{u}^{2}}, \qquad (9.9.4)$$

where $h_{K_{L}}$ is found either by formulas of §§ 9.5 and 9.8 for accuracy of measurement of speed (if we assign permissible accuracy), or by formulas of § 9.6 if we apply criteria connected with breakoff of tracking.

For illustration we give the following example.

Let us assume that the radar is characterized by the following data: $P_{\Pi \in P} = 100 \text{ w}, G = 9 \cdot 10^{-3}, \lambda = 10 \text{ cm}, \Delta f_{C} = 30 \text{ cps}$ and should work for a target where $\sigma_{\mu} = 10 \text{ m}^2$ at distance $d_{\mu} = 100 \text{ km}$. Let us assume that according to one of the criteria there is obtained the critical value of the signal-to-interference ratio $h_{\text{KP}} = 1$. Then from (9.9.4), $(P_{\Pi}G_{\Pi\Pi}/\Delta f_{\Pi})_{\text{KP}} = 2.37 \text{ w/Mc}$, i.e., for the given sufficiently typical values of parameters it composes a fully attainable quantity in contemporary conditions. If, however, we take $G = 25 \cdot 10^3$ (increasing the diameter of the antenna from 3 to 5 m), raise the power of the transmitter to $P_{\Pi \in P} \approx 1 \text{ kw}$, and assume the presence of a target with $\sigma_{\mu} = 50 \text{ m}^2$, then $(P_{\Pi}G_{\Pi\Pi}/\Delta f_{\Pi})_{\text{KP}} = 340 \text{ w/Mc}$, i.e., composes a normally unattainable magnitude.

For pulse chaotic interference with the assumptions made in Paragraph 7.14.2

$$\frac{1}{h} = \frac{1}{1 + \frac{v\tau_{\pi}^{2} P_{\pi} G_{\pi\pi}}{2N_{0} \sqrt{1 + \Delta f_{\pi\pi}^{2} + \sigma_{\pi}^{2}}}, (9.9.5)}$$

مىتىغى ئىلغان بالىلىغان بىرى بىرى بىرىغىغانغان بىرىغى يەركىيە يەركىيە بىرىغان بىلىرىغى بىلغان بىلغان بىلغان بىر

where

 P_{Π} - power in a pulse of the jamming transmitter;

 τ_{π} - pulse duration of interferences;

 ν - average frequency of pulses of interference;

 $\Delta f_m \rightarrow \phi$ - effective width of spectrum of modulation of the signal; the remaining designations are as before.

If we again assume spatial coincidence of the source of interference with the target and fulfillment of inequality $h \gg h_{\rm KP}$, the critical value of the magnitude of intensity of interference $\nu \tau_{\Pi}^{2} P_{\Pi} G_{\Pi\Pi}$ [w/Mc] according to (9.9.3) and (9.9.5) is defined as

$$(\mathbf{v}\tau_{\mathbf{g}}^{2} P_{\mathbf{n}} G_{\mathbf{n}\mathbf{n}})_{\mathbf{H}\mathbf{p}} = \frac{1}{h_{\mathbf{H}\mathbf{p}}} \frac{P_{\mathbf{n} \in \mathbf{p}} G_{\mathbf{n}}}{4\pi \Delta f_{\mathbf{0}} d^{2}_{\mathbf{n}}} \sqrt{1 + \Delta f_{\mathbf{m} \circ \mathbf{0}}^{2} \tau_{\mathbf{n}}^{2}}, \qquad (9.3.6)$$

i.e., composes a magnitude $\sqrt{1 + \Delta f_m^2} = \phi \overline{\tau}_{\Pi}^2$ -times larger than the critical intensity of noise interference, which shows the lesser effectiveness of pulse chaotic interference in the conditions formulated above. If, for instance, duration of pulses of interference and of the signal coincide, then $\Delta f_m = \phi \overline{\tau}_{\Pi}$ is a quantity of the order of unity. When $\Delta f_m = 0$, for creation of the same effect as from noise interference it is necessary to have $\sqrt{2}$ more power of pulse chaotic interference.

For passive interference, broad in band as compared to the signal, according to (7.14.13)

$$\frac{h_{\text{RW}}}{h} = \frac{1}{1 + h \frac{\theta_{\text{R}}}{\theta_{\text{R}}} f(\Delta \omega_{\text{A}})}, \qquad (9.9.7)$$

where p_{ij} - reflecting surface of target;

- $\sigma_{\rm re}$ reflecting surface of interference in the resolution cavity of the radar;
- $\Delta \omega_{\rm g} = difference of Doppler frequencies of signals reflected from the target and interference;$
- $f(\omega)$ function describing the form of the spectrum of the signal reflected from interference (f(0) = 1); and it is assumed that there is selected a sufficiently large frequency of repetition F_p .

From (9.9.7) for $h >> h_{H\!O}$ we obtain

$$(\sigma_{\rm M})_{\rm MP} \simeq \frac{1}{h_{\rm MP}} \frac{\sigma_{\rm M}}{f(\Delta \omega_{\rm A})}, \qquad (9.9.8)$$

from which it is easy to calculate the density of passive interference necessary to "swamp" radar (to knock out the speed meter). It is clear that with great detuning of Doppler frequencies of signals reflected from target and from interference $f(\Delta \omega_{\rm II}) << 1$, noise immunity with respect to passive interference is high; in particular, when $h_{\rm Rp} \approx 1$ it turns out that $(\sigma_{\rm II})_{\rm Rp} >> \sigma_{\rm II}$. If however $\Delta \omega_{\rm II} \approx 0$, $f(\Delta \omega_{\rm II}) \approx 1$ and for $h_{\rm Rp} \approx 1$ we have relationship $(\sigma_{\rm II})_{\rm Rp} \approx \sigma_{\rm II}$, which signifies for attainable densities of interferences absolutely unsatisfactory noise immunity.

Further conclusions, ensuing from lowering of the signal-to-interference ratio, according to formulas (9.9.2), (9.9.5), and (9.9.7), do not differ at all from those given in § 7.14; therefore we will not repeat them here.

Intermittent active interferences act on coherent speed meters just as on coherent range meters. In the presence of rapid intermittent (as compared to inertia of the speed meter) interferences and during application of high-speed AGC systems interferences act just as continuous interferences of the same mean power.

With a slow AGC system we start to see nonlinear phenomena, which appear in various ways for different schemes of processing the signal and for their quantitative estimating require carrying out of corresponding investigations of the separate forms of bircuits.

Slowly intermittent interferences lead to parameters of the system of measurement of speed intermittently changing in connection with the fact that there exist intervals of time in which the signal-to-noise ratio is equal to h, and intervals in which this ratio is equal to $h_{\Pi\Pi}$. Besides the fact that in intervals of action of interference there can occur the same phenomenon as during continuous interference, there also occurs parametric influence, which may cause parametric excitation of the tracking system. These questions require further study; their formulation does not differ at all from that given in § 7.14.

-494-

As we already noted during the analysis of frequency discriminators, there exists a type of discriminator — with commutation switching of reference voltages — . which is especially susceptible to the influence of intermittent interferences with frequencies of switching close to the frequency of the reference voltage or to one of the frequencies which is a multiple of it. As a result of appearing beats between the reference voltage and interference there can appear low-frequency oscillations, passing to the output of the tracking system and leading to the appearance of large errors or even to breakoff of tracking.

÷

It is necessary to make several remarks about the influence on coherent speed meters of return interference. This interference is applied mainly during pulse radiation of the radar and, as was indicated in Chapter I, can be created both by starting by pulses of the radar of a certain transmitter, and also by amplification and reradiation of the radar pulses themselves. In the first case there appear difficulties in ensuring coherence of return interference, but the incoherent pulse interference for coherent radar is not very effective. In the second case coherence is ensured automatically. However, if there is absent some additional modulation of the pulses of return interference, being radiated from the irradiated target, it permits us more exactly than by the reflected signal to measure parameters of motion of the target, in particular speed.

Such a phenomenon takes place due to the absence of fluctuations and to the great power of pulses of interference as compared to the reflected signal. Therefore, for a system of measurement of speed by Doppler frequency return interference is only effective with additional frequency modulation, where changes of frequency of the interference should be so slow that they are tracked by the tracking system. In this respect we have complete analogy with the influence of return interference on range finders with this difference only, that variable delay of modulation, necessary for misleading the range finder, is replaced by a variable carrier frequency of pulses of interference for misleading the speed meter. When the jamming transmitter is not on the target, just as during breeding of pulses of interference on coherent range finders remain valid.

Considering the influence of interferences on meters of speed as the derivative of a coordinate, one should turn first of all to noise immunity of range finders and goniometers. Short duration failure of meters tracking coordinates leads to

-495-

cessation of measurement of speed. In cases when there is no short duration failure and increase of errors of measurement of coordinates is determined only by change of signal-to-noise ratio (h $_{\Pi \ \amalg}$ /h - in coherent case and q $_{\Pi \ \amalg}$ /q - in the incoherent case), increased error of measurement of speed also is determined by the formulas obtained above, if in them we replace h by h $_{\Pi \ \amalg}$ (or q by q $_{\Pi \ \amalg}$).

§ 9.10. Conclusion

Resuming the basic results of theoretical research of questions of measurement of speed in the given chapter we first of all should emphasize that the basic method of measurement of the radial component of speed should be considered measurement by Doppler frequency shift. This method for coherent radiation ensures in the vast majority of cases greater accuracy than the other considered method, based on differentiation of the distance to the target.

Being interested in the most wide-spread tracking Doppler speed meters, we with not very limiting assumptions found an optimum frequency discriminator and optimum smoothing circuits of such meters. The optimum frequency discriminator cannot be realized exactly; however two of the considered practically realizable schemes are: a discriminator with a tuned circuit and phase shifter and a discriminator with two mixers and differentiation, very close to the optimum discriminator in its properties. Here as the criterion of optimality we, as in preceding chapters, used accuracy of measurement.

When using a tracking meter for tracking purely random components of change of speed optimum smoothing circuits frequently are linear with constant parameters and in a number of cases coincide in form with practically applied filters. In a case when speed is a quasi-regular process, optimum filters possess variable parameters and frequently are characterized by gradual cutoff of the discriminator and by accuracy of measurement unlimitedly increasing in time. It is necessary to note that in practice speed will always have a certain purely random component. Therefore, very small errors are not obtained, even if we ignore instrument errors, disregard of which is justified only with sufficiently large fluctuation and dynamic errors.

With intense noises and interferences tracking meters are subject to failure (breakoff of tracking). The formulas obtained for quantitative estimate of this phenomenon can be, unfortunately, used only for rough estimation of the order of magnitudes of intensities of noises and interferences at which failure begins. One should make such a conclusion, first, in connection with the fact that analysis of

-496-

breakoff of tracking could be conducted only for smoothing circuits of very simple form and, secondly, the obtained results rather strongly differ for various criteria of breakoff and for various methods of approximation of the discrimination characteristic.

Investigation of potentialities of nontracking Doppler speed meters leads to the conclusion that in principle these meters can ensure the same accuracy as tracking ones if we consider measurements with sufficiently fast signal fluctuations.

During attempts to measure speed by Doppler frequency shift for an incoherent pulse signal we usually obtain very low accuracy of measurement, due to which with this form of radiation measurement of speed as the derivative of distance is preferable. The tangential component of speed can be measured as the derivative of an angular coordinate.

Accuracies of measurement of speed as the derivative of a coordinate found in a number of cases have acceptable magnitudes for accelerations which are not too high; with high accelerations accuracies of measurement of speed are low. This also pertains to optimum meters in the case of purely random change of the coordinate. With quasi-regular change of it potential error of measurement of speed approaches zero as $t \rightarrow \infty$, for radial speed in certain cases even faster than error of measurement by Doppler shift of the signal frequency. Consequently, in these cases instrument error predominates.

Quantitative estimation of the influence of interferences on speed meters is determined by the fact that these meters in most cases are coherent. Therefore, $tl_{i\epsilon}$ influence of the most wide-spread interferences reduces simply to the influence of equivalent noise and, consequently, to change of the signal-to-noise ratio.

Thus, we have conducted a rather detailed analysis of speed meters from positions of accuracy of measurement, including investigation of the question of cessation of these measurements under the influence of intense noises and interferences. In order of formulating still unsolved problems, connected with measurement of speed, we should first of all pay attention to investigation of the resclusion capability of speed meters, including synthesis of meters optimum from this point of view. Currently urgent questions of resolution of targets still have not found sufficiently complete theoretical solution and need careful consideration.

It is desirable, besides frequency, to investigate phase discriminators, which

「「「「「「「「」」」」

-497-

can be applied with success in Doppler speed meters, giving, possibly, no worse results than frequency discriminators. The number of analyzed types of frequency discriminators also should be increased.

Very urgent, for meters of other parameters of motion of the targets, too, is more detailed study of questions of breakoff of tracking. In particular, it should be interesting to find solution of these questions with application of smoothing circuits of high order. Apparently, here we will not manage to obtain all necessary regularities purely theoretically, and it is necessary to utilize appropriately formulated experiment.

In these investigations we assumed Gaussianness both of the reflected signal, and also the process of change of speed. Although this assumption corresponds to a very large number of practical cases, it nonetheless is of interest to investigate accuracy of speed meters with other assumptions. It would also be very interesting to conduct a whole complex of investigations for slow fluctuations of the signal, the speed of which is commensurate with the speed of processes of set-up in the systems. The questions found in this chapter are only a very superficial reflection. Questions of the influence on speed meters of various forms of interferences, including synthesis of meters optimum in the presence of the interferences, need more careful consideration.

-498-

CHAPTER X

MEASUREMENT OF ANGULAR DATA WITH A COHERENT SIGNAL

§ 10.1 Introductory Remarks

Measurement of angular data is one of the most important functions executed by radar sets of various assignment. Continuously increasing requirements on accuracy of measurement of angles has led to a need for thorough theoretical research of possibilities of radar measurement of angular data. Especially important is the problem of finding potential accuracy of goniometry in conditions of a fluctuating radar signal received against a background of internal noises, and also the problem of synthesis of circuits of optimum goniometers realizing this potential accuracy. However, in contemporary radar literature there exists a comparatively small number of works on these problems.

The first works in which there are attempts at optimization of radar goniometers are [10 and 11]. In them there is considered the problem of optimum measurement of angular data by a surveillance radar. Here there is introduced a series of very limiting assumptions. Circuits of goniometers are assumed assigned up to and including the amplitude detector, and only post-detector processing is optimized. The target is assumed either not fluctuating at all, for fluctuating so that its reflecting sucface does not change during the time of passage over the target of the directional pattern of the antenna, but takes independent values in different periods of standard. For solution of problems of optimization in these works there is used the theory of statistical maximum likelihood estimates.

We note further the work [45], in which, in point of fact, the radio channel of a tracking radar goniumeter is optimized. We consider the method of direction finding with a scanning directional pattern. The circuit of the radio channel again

-499-

is assumed assigned up to and including the amplitude detector. The signal is assumed to be fluctuating so that its amplitude is constant within the duration of pulses, but is independent from pulse to pulse. For solution of problems of optimization there is used the method of inverse probability.

The very limiting initial assumptions used in [10, 11, 45] lower the value of the results obtained in them. However, already in these works there is shown the connection between the considered problems and mathematical statistics, although its powerful techniques are evidently used insufficiently.

Recently there appeared works [12, 40] where there is solved the problem of estimating angular data on the basis of the observed realization of the signal. The signal is assumed fluctuating, but rigidly correlated during the time of observation. Consideration was conducted, for example, for methods of instantaneous comparison of signals, both in amplitude and phase. In [12, 46] they used estimates of maximum a posteriori probability. The essential fact is that in these works with respect to the circuit of the goniometer there were introduced no preliminary assumptions, and it was completely optimized.

The assumption of rigid correlatedness of the signal for the time of observation in these works is rather limiting. More common in practice is the case when the time of correlation of fluctuations of the signal is comparable to or considerably less than the time of observation.

In the present and subsequent chapters we shall give a detailed and systematic study of the problem of radar measurement of angular data relying on this last case. The account will be conducted basically in the same plan as in all the preceding. chapters: first we synthesize and study optimum circuits; then we analyze circuits chose to optimum. During the synthesis of optimum goniometers we will make no preliminary assumptions about the possible circuit of the goniometer. However, we shall assume assigned the method of direction finding (i.e., the structure of the antenna system) and the method of coding angular data in the radar signal, determined by this. Here we shall systematically consider existing or possible methods of direction finding.

Synthesis of radar goniometers without assignment of the method of direction finding is a difficult problem which can be solved sufficiently accurately only for antennas of the phased arrey type. This question is the subject of § 10.13.

In the present chapter we shall study the problem of radar measurement of angular data with a coherent signal. The material here is divided into three basic

-500-

groups of problems: goniometers with a tracking antenna, goniometers in which tracking is carried out electronically (not by the antenna), and, finally, nontracking goniometers (not containing a feedback circuit).

The main part of the chapter is devoted to goniometers with a tracking antenna, since this class of goniometers is the broadest. First we synthesize the circuit of an optimum radio channel for these goniometers with very general assumptions about the method of direction finding utilized. Then this circuit is specified for the most important methods of direction finding utilized in practice: the method of pattern scanning, the method of scanning with compensation, of instantaneous amplitude comparison of signals, the method of phase center scanning and of instantaneous phase comparison of signals. The most attention is paid to the question of the influence on accuracy of synthesized goniometer circuits of various deviations from idealness in them, inevitable during practical realization of the circuits.

Then we study radar goniometers with antennas of phased array type. For these goniometers together with the optimum radio channel we also synthesize the optimum method of direction finding (optimum processing of the field in the aperture of the antenna system).

Then we investigated total error of a closed-loop goniometer tracking system in linear conditions (with small mistuning) and touch briefly on question of nonlinear phenomena in tracking goniometers, occurring with high noise levels (breakofr of tracking). Approximately in the same aspect we consider goniometers in which tracking is carried out electronically. Here we study the so-called method of two-dimensional scanning of the directional pattern.

At the end of the chapter we study nontracking goniometers. Here basic attention is paid to synthesis of the "estimator unit" (or "unit of primary processing"), which issues the maximum likelihood estimate of an angular coordinate from the realization of a signal of finite duration, within limits of which the angular coordinate of the target does not change. We study methods of instantaneous amplitude and phase comparason of signals most frequently used in nontracking goniometers.

The last section is devoted to study of the influence of certain forms of interferences on radar goniometers.

§ 10.2. Radar Methods of Direction Finding

In radar goniometers the measured parameters (angular coordinates of the target) are coded in the received radar signal. On angular target coordinates (on current angular mismatches in goniometers with a tracking antenna) there may depend both the

-501-

「「「「「「「」」」」

amplitude and the phase of the received signal. The form of this dependence is determined by the structure of the antenna system or, so to speak, the method of direction finding applied in the goniometer. Obviously, the structure of optimum goniometer systems will essentially depend on the method of direction finding; therefore, before solving the problem of synthesis of optimum goniometers we shall give a description or methods of direction finding. Our description will be rather short, inasmuch as radar methods of direction finding are well-known.

At present there have been proposed a large number of methods of direction finding, using different physical principles. The most well-known of the methods of direction finding is the so-called method of scanning the directional pattern of the receiving antenna [47]. Reception of the signal in this method is by an antenna whose directional pattern scans with respect to the axis of scanning in an angular sector, narrow as compared to the width of the pattern. On the received signal here there is superimposed a signal, modulated in amplitude according to a law which depends on the angular divergence of the target from the axis of scanning.



Fig. 10.1. The method of seaming of the directional pattern of a receiving antenna.

We shall find the form of amplitude modulation of the received signal introduced by the scanning pattern. Here and subsequently we shall limit ourselves to the case of measurement of one angular coordinate of the target, considering tracking of the other coordinate exact. The grounds for such consideration will be given in ChapterXII. devoted to multi-dimensional meters. o at lakes a state a sector of the sector of the state of the state of the sector of the sector of the sector of

Consider Fig. 10.1, in which there is depicted a section of a sphere, passed from an antenna

through a target (with large distances to the target this section can be considered two-dimensional). If we designate the measured angular divergence from the axis of scanning by α , the angle between the axis of the directional pattern and the axis of scanning by γ , and the polar coordinate of the current position of the axis of the directional pattern by \$(t), then, as it is easy to see from Fig. 10.1, the angle between the direction to the target and the axis of the directional pattern will be equal to

 $\varphi(t) = \sqrt{\gamma^2 + a^2 + 2\gamma a \cos \vartheta(t)}.$

Let us assume that the directional pattern (for voltage) has axisymmetric

-502-

form $Gg(\phi)$. In this case the received signal will be amplitude-modulated according to the law

$$\mathcal{V}_{a}(t, a) = g\left\{ \left| \sqrt{a^{2} + \gamma^{2} + 2a\gamma \cos \vartheta(t)} \right| \right\}.$$
(10.2.1)

Here we assume that target is irradiated by a certain nonscanning pattern. For definitiveness we shall consider this assumption carried out if nothing further is said. More complicated cases can enable be considered if desired, and the difference of them will be quite immaterial.

In expression (10.2.1) function $g(\varphi)$ in a definite way is normalized. Its normalization will be selected later. Coefficient G accounts for the gain of the directional pattern.

For small α expression (10.2.1) can be rewritten in the following form:

$$U_{a}(t, a) = g(\gamma) [1 + \mu_{a} a \cos \vartheta(t)], \qquad (10.2.2)$$

where

$$\mu_{a} = g'(Y)/g(Y)$$
(10.2.3)

is the gain factor of the directional pattern along the axis of scanning.

Obviously, when the target is located on the scan axis, amplitude modulation of the received signal disappears; therefore, the direction determined by the scan axis is also frequently called the equisignal direction.

The law of change of angle (t) may differ. When $(t) = \Omega_{CK}t + \vartheta_0$ we obtain so-called conical scanning of the directional pattern (with angular frequency Ω_{CK}); if $\vartheta(t)$ takes values ϑ_0 , $\vartheta_0 + \pi/2$, $\vartheta_0 + \pi$, $\vartheta_0 + 3\pi/2$, etc., changing over intervals of time $T_{CK}/4$, we obtain so-called quadrant scanning (with period T_{CK}).

The next method of direction finding which we shall consider is the method of scanning with two directional patterns, often called the method of scanning with compensation. (As we shall see later, this method provides compensation of the haveful influence of fluctuations of the signal.) In the given method reception of the reflected signal is effected by two directional patterns scanning with respect to one axis. Angles of inclination of axes of the patterns to the scan axis may be different, In particular, one of the angles may be equal to 0, which corresponds to one fixed pattern.

The mutual location of the target and the patterns in the method of scanning with compensation is depicted in Fig. 10.2, where by α we again designate angular divergence of the target from the scan axis; by γ_i - the angle between the scan

-503-



axis and the axis of the i-th pattern (i = 1, 2); by $\$_i(t)$ - the current value of the polar coordinate of the i-th pattern. From this figure it is easy to see that the angle between the direction to the target and the axis of the i-th pattern is equal to 「「「「「「「「」」」」」」」「「「「」」」」」」」

$$\varphi_i(t) = \sqrt{a^2 + \gamma_i^2 + 2a\gamma_i \cos \vartheta_i(t)}.$$

If we now designate by $G_{j}g_{j}(\Phi)$ the form of the i-th pattern (for voltage), we find that the signal

Fig. 10.2. Method of scanning with compencation.

received in this pattern will be amplitude-modulated according to the haw

$$U_{ai}(t, a) = g_i \{ \sqrt{a^2 + \gamma^2}_i + 2a\gamma_i \cos \vartheta_i(t) \}. \qquad (10.2.4)$$

From (10.2.4) it is clear that if the target is located on the scan axis modulation of signals received in both patterns disappears, i.e., the scan axis determines the equisignal direction. We note also that if one of the directional patterns is fixed, i.e., for it $\gamma_i = 0$, the signal received in this pattern will have amplitude $U_{ai}(t, \alpha) = g_i(\alpha)$, depending on the angular divergence of the target from the sca axis, but not depending on time.

For small α , (10.2.4) can be recorded in the form

$$U_{ai}(t, a) = g_i(\gamma_i) [1 + \mu_{ai} \cos \vartheta_i(t)], \qquad (10.2.5)$$

where

$$\mathbf{Fni} = \mathbf{g}'_i (\mathbf{Y}_i) / \mathbf{g}_i (\mathbf{Y}_i) \tag{10.1.0}$$

is the gain factor of the directional pattern.

Methods of seanning of a directional pattern without compensation and with compensation are usually used in goniometers in which tracking is carried out directly by the antenna. In these goniometers the measured value of angular divergence of the target from the scan axis is used to control the antenna, which turns in such a way that the scan axis passes through the target. Consequently, the angular position of the scan axis, obviously, is also the measured value of the angular coordinate of the target, and angular divergence of the target from the scan axis is the difference

-504-

between the true value of the angular coordinate and i.s measured value.

Let us consider now the method of direction finding called the method of instantaneous amplitude comparison of signals (IAC) [27, 48, 49]. In this method, as we know, during measurement of angle α reception of the reflected signal is conducted by two patterns, the intersection of which by the plane of angle α is depicted in



Fig. 10.3. If we designate by γ_i the angle between the axis of the i-th pattern (i = 1, 2) and the equisignal direction; by α - the angula divergence of the target from the equisignal axis; and by $Gg_i(\phi)$ - the form of the i-th pattern in the considered plane

Fig. 10.3. Method of instantaneous amplitude comparison of signals.

(voltage), then it is clear that amplitude of signal received by the 1st and 2nd patterns will be equal, respectively, to,

$$U_{a_1}(t, a) = g_1(Y_1 - a), \qquad (10.2.7)$$

$$U_{a_2}(t, a) = g_1(Y_1 + a).$$

When $\alpha = 0$, i.e., when the target is located on the equisignal direction, amplitudes of the received signals become identical, since $g_1(\gamma_1) = g_2(\gamma_2)$ (Fig. 10.3).

Let us note that from the formal point of view the IAC method can be considered a particular case of the method of scanning with compensation, in which the directional patterns are motionless and have coordinates $\vartheta_1(t) = 0$ and $\vartheta_2(t) = \pi$.

For small α expressions (10.2.7) can be rewritten in the form

$$U_{a_1}(t, a) = g_1(\gamma_1)(1 - \mu_{a_1}a), \quad U_{a_2}(t, a) = g_2(\gamma_2)(1 + \mu_{a_2}a), \quad (10.22)$$

where gain factors μ_{a_4} are determined analogously to (10.2.6).

The IAC method is used both in goniometers with a tracking antenna and in nentracking goniometers. In the last case the directional patterns are selected sufficiently wide so that the sector in which they intersect covers the sector of possible values of the angular coordinate of the target. Here the patterns are motionless, and the angular coordinate of the target is measured by the difference of amplitudes of the received signals.

Next we consider the so-called method of linear or two-dimensional scanning $[h_{\mathcal{F}}]$. In this method a rather narrow directional pattern periodically passes over a certain

-505-



Fig. 10.4. Method of two-dimensional scanning.

angular sector, covering the sector of possible values of the angular target coordinate. If the angular target coordinate in this sector is α (Pig. 10.4), the intersection of the pattern with the considered plane has with respect to power the form $Gg(\varphi)$ (we assume that irradiation of the target is by this pattern, and we disregard the propagation time of the

radar signal to the target and back), and Ω — angular velocity of motion of the directional pattern over the sector, then at the input of the receiver there will a received a periodic sequence of signal packs (pulses), having form Gg(Ω t) and delayed relative to the beginning of the period time $\tau_3 = \alpha/\Omega$. In other words, the received signal will be amplitude-modulated according to the law

$$U_{\mathbf{a}}(t, \mathbf{a}) = \sum_{n} g \left[\Omega \left(t - \tau_{\mathbf{a}} + nT_{\mathbf{u}} \right) \right], \qquad (10.2.9)$$

where T $_{\pi}$ - period of survey of the sector by the directional pattern.

Here it is assumed for simplicity that the pattern has no reverse movement over the sector. If, upon reaching the boundary of the sector the pattern instantly jumps to its beginning, obviously,

$$T_{\rm g} = = \Phi/\Omega_{\rm r}$$

where Ψ - angular dimension of the sector.

Goniometers using the method of two-dimensional scanning are usually trackingtype. However, tracking here is carried out by an electronic circuit, which tracks the signal pulses. Therefore, the method of two-dimensional scanning is offer called the method of pulse tracking. This method we shall consider subsequently.



Let us now turn to consideration of the so-called method of phase center scanning. In this method reception of the reflected signal is by an antenna whose phase center shifts (scans) in the aperture plane of the antenna. Then the received signal

Fig. 10.5. Method of phase center scanning.

becomes phase-modulated according to a law which depends on the acceptance angle of the redar.signal, i.e., on the angular position of the target.

~506-

Let us find the form of this modulation. We designate by α the angle between the target direction and the normal to the aperture plane of the antenna (which determines the equiphase direction), and by $\delta(t)$ — the coordinate of the projection of the phase center on the considered plane (Fig. 10.5). Then, obviously, the phase of the received signal is time-dependent and is equal to

$$\Phi(l, a) = \varphi_0 + \frac{2\pi \delta(l)}{\lambda} \sin a,$$
 (10.2.10)

where φ_0 - phase of the received signal when the target is on the equiphase direction; λ - wavelength of the radar signal.

For small α , sin α in (10.2.10) can be replaced by α . The method of phase center scanning is usually used in goniometers with a tracking antenna. The angle α here is the difference between the true and measured values of the angular target coordinate.

A considerably more wide-spread method of direction finding is the method of instantaneous phase comparison of signals (IPC) [27, 48, 49]. Here during measurement



()

Fig. 10.6. Method of instantaneous phase comparison of signals. of angle α reception of the reflected signal is by two antennas whose phase centers lie in the plane of angle α and are separated distance d, called the base of the antenna system. By α we shall understand the angle between the target direction and the normal to the plane in which the phase centers lie (Fig. 10.6). It is easy to see that phases of signeds received by the 1st and 2nd antennas are equal, respectively,

to

$$\Phi_{1}(l, a) = \varphi_{0} - \frac{\pi d}{\lambda} \sin a; \Phi_{1}(l, a) = \varphi_{0} + \frac{\pi d}{\lambda} \sin a,$$
 (10.2.11)

where all designations are the same as in (10.2.10).

The IPC method is used both in geniometers with a tracking antenna, and also with a nontracking antenna. In nontracking geniometers the antenna system is methon-less, and its base d is not too great for angular sector $|\alpha| < \arcsin \lambda/2d$ [in which phases $\Phi_1(t, \alpha)$ and $\Phi_2(t, \alpha)$ (10.2.11) unambiguously determine α]

-507-

to cover the sector of possible values of angular coordinates of the target. Angular target coordinate α is determined in this case by the difference of phases of the received signals.

Thus, we have characterized all basic methods of direction finding applied at present in radar. We note the possibility of combined use of various methods of direction finding. Generalizations in this direction are obvious. For instance, it is possible to use a method with simultaneous scanning of the directional pattern and phase center scanning.

The received signal in this case becomes modulated in amplitude and phase with laws of modulation which depend on the angular position of the target, i.e., becomes, as it were, more informative, which should lead, ultimately, to increase of accuracy of measurement of angular data. Also there can be used the method of instantaneous amplitude-phase comparison of signals, consisting of the reception of the reflected signal in two antennas with separated phase centers and directional patterns opened a certain angle to one another. The received signals in this method will differ both in amplitudes and in phases, the difference of which depends on the acceptance angle of the radar signal. Continuing generalizations in this direction, we arrive a) a radar method of direction finding which consists of reception of a radar signal in an arbitrary number of antennas, directional patterns and phase centers of which deep by certain laws. Any of the methods of direction finding is a particular case of such a generalized method and is obtained upon specifying the number of receiving antennas and laws of scanning of their patterns and phase centers.

Subsequently we shall consider, as far as possible, goniometers with the constatized method of direction finding, specifying, where it is necessary, final results. In particular, this approach is applicable to goniometers with a tracking antenna.

§ 10.3. Optimum Radio for Goniometers with a Tracking Antenna

In the following sections we will study the most important class of radar goniometers - goniometers with a tracking antenna. The present section is devoted to synthesis of an optimum radio channel for such goniometers.

As we already said, among goniometers with a tracking antenna we include goniometers for which target tracking is produced directly by the antenna array. Goniometers of this type are very widely used, inasmuch as they possess best the power characteristics (power is always radiated in the direction of the target).

In the case of goniometers with a tracking antenna the optimum radio channel

-508-

can be synthesized for the generalized method of direction finding described in the preceding paragraph, when reception of the reflected signal is conducted in any number of antennas, directional patterns and phase centers of which scan by arbitrary laws. Here we shall assume that there exists a single equisignal direction: if the target is located on this direction, the amplitude and phase modulations of the received signals introduced during scanning of the directional patterns and phase centers of the antennas disappear. This assumption is very significant subsequently.

Most methods of direction finding applied in practice — scanning of the directional pattern without compensation and with compensation, IAC, phase center scanning, IPC — are applicable to this case.

Thus, we assume that the goniometer has n antennas, and the received signal at the output of the i-th antenna due to scanning of its directional pattern and phase center is modulated in amplitude and phase by laws $U_{ai}(t, \alpha)$ and $\Phi_{i}(t, \alpha)$, depending on the angular divergence α of the target from the equisignal axis. Obviously, $U_{ai}(t, 0) = \text{const}$ and $\Phi_{i}(t, 0) = \text{const}$, since by condition coordinate $\alpha = 0$ determines the equisignal axis. We assume that there is carried out normalization

$$U_{ai}(t, 0) = 1.$$
 (10.3.1)

We introduce the designation

()

$$U_i(t, a) = U_{ai}(t, a) e^{i \Phi_i(t, a)}$$
. (10.3.2)

Quantity $U_{i}(t, \alpha)$ we shall call the complex directional pattern of the i-th antenna of the goniometer.

The received signal at the output of the i-th antenna can be recorded in the form

$$\begin{aligned} y_{i}(l) &= V P_{ei} U_{ai}(l, a) u_{a}(l) \{a(l) \cos [w_{e}t + \psi(l) + \\ &+ \Phi_{i}(l, a)] + b(l) \sin [w_{e}t + \psi(l) + \Phi_{i}(l, a)] \} + \\ &+ V \overline{N_{ei}} n_{i}(l) = \sqrt{P_{ei}} \operatorname{Re} U_{i}(l, a) E(l) u(l) e^{l \cdot a l} + \sqrt{N_{ei}} n_{i}(l). \end{aligned}$$
(10.2.3)

Here $u_a(t)$ and $\psi(t)$ are laws of amplitude and phase modulation of the sounding signal; function $u_a(t)$ is normalized

$$\frac{1}{T_r} \int_0^{T_r} u_a(t)^a dt = 1, \qquad (10.3.4)$$

where T_r - the period of repetition of the sounding signal:

$$u(l) = u_a(l) e^{l\phi(l)}; \qquad (1 \cdot . \cdot . \cdot . \cdot)$$

-509-

a(t) and b(t) - independent stationary random processes with zero mean value and with correlation function $\overline{a(t)a(s)} = \overline{b(t)b(s)} = \rho(t - s)$, normalized so that

$$p(0) = 1;$$

 $E(l) = a(l) + ib(l);$ (10.3.6)

 $n_i(t)$ - white noise with unit spectral density; noises $n_i(t)$, i = 1, 2, ..., n are independent;

 N_{OI} - spectral density of noise added to the signal at the input of the i-th antenna;

 $P_{c,i}$ - mean power of this signal.

We assume that the time position and carrier frequency of signal (10.2.3) are known exactly. This means that we know exactly the delay of the received signal (distance to the target) and Doppler shift of frequency of the reflected signal (speed of the target). Being interested exclusively in the problem of measurement of angular data, we can consider these assumptions realized. The influence of inaccurate knowledge of the delay and frequency of the received signal on measurement of angles will be considered subsequently during the analysis of goniometer circuits.

The set of signals (10.3.3) will form an n-dimensional normal process, characterized by a matrix of correlation functions $\mathbf{R}(t_1, t_2, \alpha) = |[\mathbf{R}_{ij}(t_1, t_2, \alpha)]|$. From (10.3.3) we can easily find that

$$R_{ij}(l_1, l_2, a) = \sqrt{P_{ci}P_{cj}} \varepsilon(l_1 - t_2) \operatorname{Re} U_i(l_1, a) U^*_j(l_2, a) \times u(l_1) u^*(l_2) e^{i \omega_0 (l_1 - u_1)} + \delta_{ij} \sqrt{N_{vi}N_{vj}} \delta(l_1 - l_2), \qquad (10.3.7)$$

where o_{i,j} - Kronecker delta, and by 2* there is designated the complex conjugate of 2.

To finding the operation executed by the optimum radio channel we need to know the joint functional of the probability density of the set of signals (10.3.3). Such a functional was already given in § 6.7. In this case it is equal to

$$P[y_{1}(\tau), y_{1}(\tau), \dots, y_{n}(\tau), a] = K \exp\left\{-\frac{1}{2}\int_{0}^{T}\int_{0}^{T} y^{+}(t_{1})W(t_{1}, t_{2})y(t_{2})dt_{1}dt_{2}\right\}, \quad (10.3.8)$$

where

(0, T) - interval observation of the process; Y(1) - column-vector of signals (10.3.3); Y⁺(t) - row vector of these signals;

 $W(t_1, t_2) = ||W_{1j}(t_1, t_2)||$ -- matrix of so-called inverse-correlation functions, satisfying equation

$$W(t_1, t_2) R(t_1, t_3, a) dt_2 = 18(t_1 - t_3), \qquad (10.3.9)$$

(I - unit matrix).

-510-

In more expanded form matrix equation (10.3.9) can be written in the form of the following system of integral equations:

$$\sum_{i=1}^{n} \int_{Q} W_{ij}(t_{1}, t_{2}) R_{jk}(t_{3}, t_{3}, a) dt_{3} = \delta_{ik} \delta(t_{1} - t_{3}).$$
(10.3.10)

Coefficient K in (10.3.8) is determined by the relationship

$$\frac{\partial \ln K}{\partial \mathbf{x}} = -\frac{1}{2} \int_{\mathbf{x}}^{\mathbf{r}} \int_{\mathbf{x}}^{\mathbf{r}} S_{p} \frac{\partial \mathbf{R}(t_{1}, t_{2}, \mathbf{a})}{\partial \mathbf{x}} \mathbf{W}(t_{1}, t_{2}) dt_{1} dt_{2} =$$

$$= -\frac{1}{2} \sum_{i,j=1}^{n} \int_{\mathbf{x}}^{\mathbf{r}} \int_{\mathbf{x}}^{\mathbf{r}} \frac{\partial R_{ij}(t_{1}, t_{2}, \mathbf{a})}{\partial \mathbf{x}} \mathbf{W}_{ij}(t_{1}, t_{2}) dt_{1} dt_{2}.$$
(10.3.11)

We introduce likelihood function $L(\alpha)$, which is simply the functional of the distribution of probabilities, considered as a function of parameter α .

In Chapter VI it was shown that operation z(t) of an optimum radio channel of a meter of angle α is determined by relationship

$$\int_{I-M} z(\tau) d\tau = \frac{\partial \ln L(a)}{\partial a} \Big|_{a=0}.$$
(10.3.12)

Here likelihood function $L(\alpha)$ is taken in interval (t, t- Δ t), where Δ t considerably exceeds the time of correlation of fluctuations of the signal, but during Δ t the angular coordinate of the target α can be considered constant.

Substituting (10.3.8) in (10.3.12) and considering (10.3.11), we can obtain

$$\frac{\partial \ln L(\mathbf{a})}{\partial \mathbf{a}}\Big|_{\mathbf{a}=\mathbf{0}} = -\frac{1}{2} \sum_{i,j=1}^{n} \int_{\mathbf{a}=\mathbf{0}}^{t} \int_{\mathbf{a}=\mathbf{0}}^{t} \left\{ \frac{\partial R_{ij}(t_{1},t_{2},\mathbf{a})}{\partial \mathbf{a}} W_{ij}(t_{1},t_{2}) \Big|_{\mathbf{a}=\mathbf{0}} + \frac{\partial W_{ij}(t_{1},t_{2})}{\partial \mathbf{a}} \Big|_{\mathbf{a}=\mathbf{0}} y_{i}(t_{1}) y_{j}(t_{2}) \right\} dt_{1} dt_{2}.$$

$$(10.3.13)$$

Thus, to find the operation of an optimum radio channel we need to find values of functions $W_{ij}(t_1, t_2)$ and their derivatives with respect to α when $\alpha = 0$. We shall look for a solution of system of equations (10.3.9) in the form

$$W_{ij}(t_1, t_2) = \frac{\sqrt{P_{0i}P_{0j}}}{N_{0i}N_{0j}} v(t_1, t_2) \operatorname{Re} U_i(t_1, a) U^*_j(t_2, a) \times \\ \times u(t_1) u^*(t_2) e^{i \omega_0 (t_1 - t_2)} + \frac{\vartheta_{ij}}{\sqrt{N_{0i}N_{0j}}} \vartheta(t_1, t_2), \qquad (10.3.1^{l_1})$$

where function $v(t_1, t_2)$ is slow in comparison with $e^{i\omega_0 t}$. Substituting expressions (10.3.7) and (10.3.14) in equations (10.3.10) and replacing under the signs of intervals rapidly oscillating functions by their time-averaged values, we obtain for $v(t_1, t_2)$ the equation

$$\Delta f_{o} \int_{t-\Delta t}^{t} |u(t_{s})|^{a} \left\{ \sum_{i=1}^{n} h_{i} |U_{i}(t_{s}, a)|^{a} \right\} \rho(t_{1} - t_{s}) v(t_{s}, t_{s}) dt_{s} + \rho(t_{1} - t_{s}) + v(t_{1}, t_{s}) = 0, \qquad (10.3.15)$$

where Δf_{c}^{\prime} - effective width of the spectrum of signal fluctuations

$$\Delta f_{c} = \frac{1}{\int_{-\infty}^{\infty} P(\eta) dt} \qquad (10.3.15!)$$

 $h_i = P_{c_i}/2N_{0i}\Delta f_{c_i}$ - ratio of mean signal power at the output of the i-th antenna to the power of the corresponding noise in the band of signal fluctuations.

Equation (10.3.15) must be solved during synthesis of circuits of optimum gonders oters. In the case of goniometers with a tracking antenna it can be solved. As we determined, we need to find only the values of function $v(t_1, t_2)$ and its first derivatives with respect to α when $\alpha = 0$. However, for goniometers with a tracking antenna $U_i(t, 0) = 1$ [by virtue of normalization of (10.3.1)]. Substituting $\alpha = 0$ in (10.3.15) we obtain

$$\begin{split} h_{2} \Delta f_{0} \int_{t-\Delta t}^{t} & |u(t_{2})|^{2} \rho(t_{1}-t_{3}) \sigma(t_{2},t_{3})|_{a=0} dt_{3} + \\ & + \rho(t_{1}-t_{3}) + \sigma(t_{1},t_{3})|_{a=0} = 0, \end{split}$$
 (10.3.16)

where

$$h_{1} = h_{1} + h_{2} + \dots + h_{n}. \tag{10.3 16'}$$

We introduce now the assumption of smallness of the period of repetition of the signal as compared to the time of constantion of its fluctuations, which was proven and widely used in the preceding chapters. With this assumption function $|u(t_2)|^2$ will be rapidly oscillating as compared to other functions under the sign of the integral in (10.3.16), and therefore it can be replaced there by its time-averaged value. Using normalization (10.3.4), instead of (10.3.16) we obtain equation

$$h_{2}\Delta f_{0} \int_{-\Delta t} f(t_{1}-t_{2}) \dot{v}(t_{1},t_{2})|_{t=0} dt_{2} + f(t_{1}-t_{2}) + \dot{v}(t_{1},t_{2})|_{t=0} = 0, \qquad (10.3.17)$$

coinciding with equation (4.3.5) (with accuracy of a coefficient). Solution of this equation, as it was shown in Chapter IV, has the form

$$v(t_1, t_2)|_{\alpha=0} = -\frac{1}{2\pi\Delta f_0} \int_{-\infty}^{\infty} \frac{S_0(\omega)}{h_2 S_0(\omega) + 1} e^{i\omega(t_1 - t_2)} d\omega, \qquad (10.3.15)$$

where

$$S_{\bullet}(\boldsymbol{\omega}) = \Delta f_{\boldsymbol{\omega}} \int_{-\infty}^{\infty} P(t) e^{t \boldsymbol{\omega} t} dt \qquad (10.3.19)$$

is the spectrum of fluctuations of the signal, normalized in such a way that $S_0(0) = 1$.

We determine now function $v'(t_1, t_2)|_{\alpha=0}$. Differentiating equation (10.3.15) with respect to α and considering $\alpha = 0$, we obtain

$$\begin{split} \Delta f_{0} \int_{t-\Delta t}^{t} |\mu(t_{2})|^{2} \left\{ \sum_{l=1}^{n} h_{i} \frac{\partial}{\partial \alpha} |U_{i}(t_{2}, \alpha)|^{2} \right\} \Big|_{\alpha=0} \rho(t_{1}-t_{2}) \times \\ \times \sigma(t_{2}, t_{3})|_{\alpha=0} dt_{2} + \sigma'(t_{1}, t_{3})|_{\alpha=0} + \\ + h_{2} \Delta f_{\alpha} \int_{t-\Delta t}^{t} |\mu(t_{3})|^{2} \rho(t_{1}-t_{2}) \sigma'(t_{2}, t_{3})|_{\alpha=0} dt_{3} = 0. \end{split}$$

$$(10.3.20)$$

Not stopping to discuss methods of solution of this equation, we immediately give the final result

$$\boldsymbol{\boldsymbol{v}}^{t}(t_{1}, t_{2})\Big|_{\boldsymbol{\alpha}=0} = \Delta f_{\boldsymbol{\alpha}} \int_{\boldsymbol{t}-\boldsymbol{\Delta t}}^{\boldsymbol{t}} |\boldsymbol{\boldsymbol{u}}(t_{2})|^{2} \left\{ \sum_{i=1}^{\boldsymbol{\alpha}} h_{i} \frac{\partial}{\partial \boldsymbol{x}} |\boldsymbol{\boldsymbol{U}}_{i}(t_{2}, \boldsymbol{\alpha})|^{2} \right\} \times \\ \times \boldsymbol{\boldsymbol{v}}(t_{2}, t_{2}) \boldsymbol{\boldsymbol{v}}(t_{2}, t_{2}) dt_{2} \Big|_{\boldsymbol{\alpha}=0}.$$

$$(10.3.21)$$

It is easy by simple substitution of this expression in (10.3.20) to prove that this really is its solution.

We introduce subsequently the assumption that $T_r \ll T_{0K}$ (T_{0K} - period of scanning of directional patterns and phase centers), which practically always is realized. Here $|u(t_2)|^2$ under the integral in (10.3.21) changes considerably more rapidly that the remaining functions, and can be replaced by its time-averaged value, equal to 1. The expression for $v'(t_1, t_2)|_{\alpha=0}$ then takes the simple form

$$\begin{aligned} \boldsymbol{v}'(t_1, t_2) \Big|_{\boldsymbol{u}=0} &= \Delta f_0 \int_{1-\Delta t}^{t} \left\{ \sum_{i=1}^{n} h_i \frac{\partial}{\partial \boldsymbol{u}} \left| U_i(t_2, \boldsymbol{u}) \right|^2 \right\} \times \\ &\times \boldsymbol{v}(t_1, t_2) \boldsymbol{v}(t_2, t_2) dt_2 \Big|_{\boldsymbol{u}=0}. \end{aligned} \tag{10.5.11}$$

and the second

Thus, we have found all the basic characteristics necessary for construction of the operation of an optimum radio channel.

Using (10.3.21'), (10.3.8) and (10.3.11), it is possible to expand expression

(10.3.12) in the following way:

If, as this is done in Chapter IV, we introduce function $h_{\rm 1-OIIT}(t)$, related to $v(t_1,\,t_2)$ by equation

$$\int_{-\infty}^{\infty} h_{100T}(t_1 - t_2) h_{100T}(t_2 - t_3) dt_3 = -h_2 \Delta f_c v(t_1, t_3) \Big|_{s=0}, \quad (10.3.23)$$

and designate

$$h_{\text{cont}}(l) = \sqrt{h_1} \Delta f_c v(l_1, l_1 + l) \Big|_{a=0}$$
(10.3.23')

[as can be seen from (10.3.18) $v(t_1, t_1 + t)|_{\alpha=0}$ depends only on t], then from (10.3.22) we can obtain an explicit expression for operation of an optimum radio channel:

$$z(\tau) = \frac{4\Delta f_{c}}{P_{ci}} \left\{ \frac{\partial}{\partial \tau} \left| \int_{t-\Delta t}^{t} h_{10}\pi\tau (\tau - s) \left[\sum_{i=1}^{n} \sqrt{\frac{P_{ci}}{P_{ci}}} h_{i}U_{i}(s, a) \times \right. \right. \\ \left. \times u(s)y_{i}(s) \right] e^{i\omega_{s}s} ds \left|^{2} - \sum_{i=1}^{n} h_{i} \frac{\partial}{\partial a} \left| U_{i}(t_{1}, a) \right|^{2} \times \\ \left. \times \left| \int_{t-\Delta t}^{t} h_{son\tau}(\tau - s) \left[\sum_{i=1}^{n} \sqrt{\frac{P_{ci}}{P_{ci}}} h_{i}u(s)y_{i}(s) \right] e^{i\omega_{s}s} ds \left|^{2} \right\} \right|_{s=0} \right.$$

$$(10.3.2^{h})$$

Expression (10.3.24) shows what operations the optimum radio channel of a radar goniometer should produce on the received signals. The most essential here are operations of optimum linear filtration. Pulse responses of the obtained optimum filters in a definite way are matched with the spectrum of signal fluctuations. The filter with pulse response $h_1(t)$, determined by relationship (10.3.23), already has been repeatedly encountered in preceding chapters. The square of its gain-frequency response has the form

$$|H_{1011}(i\omega)|^{2} = \frac{h_{z}S_{0}(\omega)}{1 + h_{z}S_{0}(\omega)} \cdot (10.3.25)$$

Phase-frequency response of this filter is arbitrary. If we approximate $S_{\rm O}(\omega)$ by function

$$S_{0}(\mathbf{0}) = \frac{1}{1 + \left(\frac{\mathbf{0}}{2\Delta f_{0}}\right)^{2}},$$
 (10.3.26)

then for $|H_{10\Pi T}(i\omega)|^2$ we obtain

$$H_{100T}(i\oplus)|^{2} = \frac{h_{2}}{1+h_{2}} \frac{1}{1+\left(\frac{\Phi}{2\Delta f_{0}\sqrt{1+h_{2}}}\right)^{2}},$$
 (10.3.27)

i.e., the form of the square of the gain-frequency response of this filter coincides with the form of the spectrum of signal fluctuations, and the width of its transmission band is larger than the width of the spectrum of fluctuations by a factor of

 $\sqrt{1 + h_{\Sigma}}$. The physical meaning of this fact was discussed in Chapters IV and VII.

The second filter contained in (10.3.24) has pulse response $h_{200T}(t)$ (10.3.23), i.e., frequency response

$$H_{\text{sont}}(i\omega) = \frac{\sqrt{h_{\text{L}}}S_{\bullet}(\omega)}{h_{\text{L}}S_{\bullet}(\omega) + 1}.$$
 (10.3.28)

This filter, obviously, is not realizable physically. The question of construction of a physically realizable equivalent of this filter will be considered below in § 10.5.

For construction of a block diagram realizing operation (10.3.24) it is userul to rewrite the latter in real form. For this we introduce intermediate frequency $\omega_{\rm np}$, considerably exceeding the width of the frequency spectrum of signals $y_i(t)$ (10.3.3). Then operation (10.3.24) can be reduced to form

$$z(\mathbf{r}) = \frac{8\Delta I_{e_{i}}}{P_{e_{i}}} \int_{t-\Delta t}^{t} h_{10\pi\pi} (\tau - s) \cos \omega_{\pi p} (\tau - s) \sum_{i=1}^{n} \sqrt{\frac{P_{e_{i}}}{P_{e_{i}}}} \times \\ \times \left[h_{i}U'_{ai}(s, 0) y_{i}(s) u_{a}(s) \cos \left[\omega_{r}s + \psi(s)\right] \right] + \\ + \Phi_{i}(s, 0) y_{i}(s) u_{a}(s) \sin \left[\omega_{r}s + \psi(s)\right] ds \left\{ \int_{t-\Delta t}^{t} h_{10\pi\pi} (\tau - s) \times \right. \\ \times \cos \left. u_{i}(\tau - s) \sum_{i=1}^{n} \sqrt{\frac{P_{e_{i}}}{P_{e_{i}}}} h_{i}y_{i}(s) u_{a}(s) \cos \left[\omega_{r}s + \psi(s)\right] ds - \\ \left. - \sum_{i=1}^{n} h_{i}U'_{ai}(s, 0) \left[\int_{t-\Delta t}^{t} h_{a0\pi\pi} (\tau - s) \cos \omega_{\pi \mu} (\tau - s) \times \right. \\ \times \sum_{i=1}^{n} \sqrt{\frac{P_{e_{i}}}{P_{e_{i}}}} h_{i}u_{a}(s) y_{i}(s) \cos \left[v_{r}s + \psi(s)\right] ds \right]^{2} \right\},$$

$$(10.3.79)$$

where $\omega_{\rm p} = \omega_{\rm O} - \omega_{\rm HD}$ heterodyne frequency.

The block diagram of the unit realizing operation (10.3.29) (with an accuracy of immaterial coefficient $8\Delta f_{\rm C}/P_{\rm C-1}$), is shown in Fig. 10.7. First, in the circuit



Fig. 10.7. Optimum circuit of the radio channel of a radar goniometer with a generalized method of direction finding: 1) adder; 2) optimum filters with frequency response $H_{1 \text{ OHT}}(i\omega)$; 3) optimum filter with frequency response $H_{2 \text{ OHT}}(i\omega)$; 4) square-law detector.

there occurs normalization of the received signals in order to eliminate effects related to nonidentical gain of directional patterns and unequalness of noises acting in each channel (if this takes place). If the gain of the patterns and noises are identical, the shown normalization is absent.

After normalization, the signal from the output of each antenna is separated into two channels, conditionally called subsequently the first and second, and in each channel it is heterodyned. Here signals of the heterodyne, directed towards the second channels (see Fig. 10.7), are shifted with respect to signals directed towards the first channels, by $\pi/2$. Heterodyne signals should coincide in form with the emitted radar signal.

đ

After heterodyning, the signals in the first channels are multiplied by derivatives with respect to the measured angle in the equisignal direction of moduli of complex gains of corresponding directional patterns, and signals in the second channels are multiplied by the derivatives of the arguments of complex gains of these patterns. After that all signals are added, filtered by an optimum filter with characteristic (10.3.25) and are fed to one input of the multiplier (phase detector). As a reference signal for this phase detector there is used the sum of signals of the first channels (see Fig. 10.7), pa sed through precisely the same optimum filter. This same sum, furthermore, proceeds to an optimum filter with characteristic (10.3.27), is detected by a square-law detector and is multiplied by the sum of derivatives in the equisignal direction of moduli of complex gains of all directional patterns, (taken with weights equal to the signal-to-noise ratio in the corresponding channels). The obtained signal is subtracted from the output signal of the phase

-516-

detector.

Thus, all operations produced by the optimum circuit are sufficiently simple and are encountered in one form or another in radar goniometer technology. However, the sequence of operations and responses of the filters are new.

The physical meaning of transformations of the signal in the optimum circuit we shall discuss later, in examining concrete methods of direction finding.

Let us turn to the question of potential accuracy of measurement of angles in goniometers of the considered type. As it was shown in Chapter VI, the equivalent spectral density of the optimum radio channel is equal to

$$S_{our} = \frac{\Delta t}{\left[\frac{\partial \ln L(a)}{\partial a}\right]^{2}}, \qquad (10.7.70)$$

where likelihood function $L(\alpha)$ is constructed as before in an interval of duration Δt , small as compared to the time of correlation of angular shifts, but sufficiently great as compared to the time of correlation of fluctuations of the reflected signal [in this case expression (10.3.30), as it is possible to show, does not depend on Δt].

Calculation by the formula (10.3.30) using expressions (10.3.6) and (10.3.11) gives

$$S_{017} = \Delta t \left[-\frac{1}{2} \int_{t-\Delta t}^{t} \int_{1-\Delta t}^{t} \sum_{i,j=1}^{n} W'_{ij}(t_1, t_2) \times R'_{ij}(t_1, t_2, a) dt_1 dt_2 \right]^{-1} \Big|_{a=0}.$$
(10.3.31)

This formula is very important subsequently.

Using expressions (10.3.7), (10.3.8) and (10.3.14), it is possible to reduce (0.3.31) to the following form:

$$S_{\text{off}} = \left\{ \frac{2\Delta f_{e}}{\Delta t} \int_{t-At}^{t} |1+v(t,t)|_{s=0} \right] dt \times \\ \times \frac{1}{\Delta t} \int_{t-At}^{t} \sum_{i=1}^{n} h_{i} |U'_{i}(t,0)|^{s} dt - \frac{2\Delta f_{c}^{2}}{\Delta t} \int_{t-At}^{t} \int_{t-At}^{t} \int_{t-At}^{t} \rho(t_{1}-t_{s}) \times \\ \times v(t_{i},t_{s})|_{s=0} \left[\operatorname{Re} \sum_{i=1}^{n} h_{i} U'_{i}(t_{1},0) \sum_{i=1}^{n} h_{i} U^{*i}_{i}(t_{2},0) \right] dt_{1} dt_{1} + \\ + \frac{4\Delta f_{c}^{2}}{\Delta t} \int_{t-At}^{t} \int_{t-At}^{t} \left[\rho(t_{1}-t_{s})+v(t_{1},t_{s})|_{s=0} \right] v(t_{1},t_{s})|_{s=0} \times \\ \times \left[\sum_{i=1}^{n} h_{i} |U_{i}(t,0)|^{t} \sum_{i=1}^{n} h_{i} |U_{i}(t_{s},0)|^{t} \right] dt_{1} dt_{s} \right\}^{-1}.$$

$$(10.3.3.3)$$

-517-

Expression (10.3.32) is very cumbersome, since it contains uncalculable integrais. Calcula: integrals in this formula in such a general form, without specifying funct. $J_1(t, \alpha)$, i.e., the method of direction finding, is difficult. These calculations, and also investigation of potential Accuracy we shall postpone until our consideration of concrete methods of direction finding. Formula (10.3.32) will be the starting point for these considerations.

Thus, we succeeded in a very general form in synthesizing the circuit of an optimum radio channel of a radar goniometer with a tracking antenna. The results obtained here are basis for a systematic study of this class of goniometers from the point of view of their potential and real properties, which will be conducted in subsequent paragraphs.

§ 10.4. <u>Method of Scanning of the Directional</u> <u>Pattern (Optimum Circuits)</u>

The method of scanning of the directional pattern has been used in radar for a long time. Experience in the use of this method testifies to its low noise immunity to active interferences which are modulated by frequencies close to frequencies of scanning, and to decrease of accuracy of measurement of angles by this method in the presence of amplitude fluctuations of the reflected signal, the spectrum of which contains as components the frequencies of scanning. In particular, in the property of such fluctuations of the signal in circuits of the method of scanning the pattern utilized in practice even with complete absence of noises there exists a final fluctuating error of target tracking. these factors have a simple physical explanation: when using the method of pattern scanning the useful information about the angular coordinates of the target is included in the amplitude of the received signals, and any distortion of their amplitude leads directly to distortion of the useful information and, ultimately, to error or measurement. In this connection study of optimum circuits of the method of pattern scanning which have, obviously, means of optimum suppression of the harmful influence of amplitude fluctuations of reflected signal is of great theoretical and practical interest.

10.4.1. Synthesis of Optimum Circuits for the Method of Scanning the Directional Pattern

The optimum circuit for the method of pattern scanning is easily obtained from the general circuit of Fig. 19.7. For this method, obviously, n = 1 (one antenna), $V(t, \alpha) = 0$ (the phase center of the antenna is motionless), and $U_g(t, \alpha)$ is expressed by Formula (10.2.2). Using (19.3.29), we obtain the optimum circuit for measurement

-518-

of angle α in the form depicted in Fig. 10.8.

The circuit of Fig. 10.8 contains three channels to which the signal proceeds after heterodyning. In the first channel there first is produced multiplication of



Fig. 10.8. Optimum circuit of the radio channel of a goniometer with a scanning pattern: 1-optimum filters with frequency response $H_{1 \text{ OHT}}(i\omega)$; 2- optimum filter with frequency response $H_{2 \text{ OHT}}(i\omega)$; 3- square-law detector.

the signal by a function varying according to the law of its amplitude modulation, introduced by the scanning pattern, and then optimum filtration (10.3.25). The output signal of this channel enters the phase detector. Useful information about angular coordinates of the target is contained in the side spectral components of the received signal, and in the absence of the shown multiplication we would be forced to further insert a sufficiently

「「「「「「」」」」」

broad-banded filter, which would not cut off the side spectral components of the signal. This would lead to unnecessary increase of noises at the filter output. The operation applied in the optimum circuit produces transfer of the side spectral components of the signal to the center frequency, after which there can be carried out filtration of the signal by a sufficiently narrow-band filter, passing the signal and noise in the optimum relationship.

As the reference signal to the phase detector there proceed center spectral components of the received signal, separated by an optimum filter in the second channel. The filter here is the same as in the first channel.

At the output of the phase detector there will be formed, obviously, the basic component of the signal of error. The third channel, as it is easy to see, is intended exactly for compensation of the harmful influence of amplitude fluctuation of the reflected signal. At first the signal in this channel is filtered by an optimum filter with response (10.3.28), and then it is detected by a square-law detector, at the output of which there are obtained in pure form amplitude fluctuations of the signal. Multiplying them by a function which varies according to the law of pattern scanning, we separate the information about angular coordinates contained in the amplitude which is false. The false information is then extracted from the total information separated by the first two channels. Thus there occurs optimum compensation of the harmful influence of amplitude fluctuations of the signal.

-519-

Having considered in general form the work of the optimum circuit, let us turn to investigation of certain limiting cases of practical interest. The first of these cases is the case of a high frequency of scanning, considerably exceeding the width



Fig. 10.9. Optimum circuit of the radio channel of a goniometer with a scanning pattern with high frequencies of scanhing. 1 — optimum filters with frequency response $H_{1,011}(i\omega)$.

of the spectrum of signal fluctuations. Here, as it is easy to see, the third channel in the circuit of Fig. 10.8 disappears. The circuit of the optimum radio channel takes the form depicted in Fig. 10.9. The physically obtained simplification is understandable, since with high frequencie of scanning amplitude fluctuations of the signal which are low-frequency cease to

affect accuracy of direction finding. The circuit of Fig. 10.9 is rather simple and is very attractive, but in the practice of radar goniometry it has not been encountered. More detailed study of this circuit is of interest and will be made in the following section.

Let us turn now to another extreme case, when the frequency of scanning is considerably smaller than the transmission band of the filter (10.3.25).

Here multiplication by $\cos \delta(t)$ in the first channel can be transferred to the obtained of the filter. The circuit of the optimum radio channel takes the form depicted in Fig. 10.10, where the filter has the square of the gain-frequency response

$$|H_{\text{sour}}(i,\upsilon)|^2 = \frac{4\hbar S_0(\omega)}{(1+\hbar S_0(\omega))^2}.$$
 (10.4.1)

We already met a filter of such form in Chapter VII. As was shown there $\left[d_{2^{-1}(1)}\left(\frac{4\omega}{2}\right)\right]^2$ is a double-humped curve with maxima at frequencies ω_1 and ω_2 , for which $hS_Q(\omega) = 4$, and a dip to $4h/(4 + h)^2$ at zero frequency. Thanks to this form e^{-1} the definition frequency response of the filter, we in the very best manner suppress center components of the signal, emphasizing to the greatest degree its side components, hearing the useful information. Here we achieve the maximum possible decrease of the harmful influence of amplitude fluctuations of the signal on accuracy of direction finding. When h < 4 the response of the filter becomes single-humped. Physically this is explained by the fact that for high noise levels it becomes more profitable to cut off natural noises more strongly than to worry about decrease of the influence of signal fluctuations.

We note that the circuit of the method of pattern scanning usually applied in

-520-

practice coincides in structure with the circuit of Fig. 10.10. The results above show in what cases this circuit turns out to be optimum, i.e., realizes the poten-



Fig. 10.10. Optimum circuit of the radio channel of a goniometer with a scanning pattern with low frequencies of scanning: 1 - optimum filter with frequency response $H_{3 \text{ ORT}}(i\omega)$; 2 - square-law detector. tialities of the method of pattern scanning. More detailed study of this circuit and comparison of it with the circuit of Fig. 10.9 will be given in the next section. In conclusion let us note that optimum circuits for measurement of the same, only everywhere multiplication by

angle in another plane will be precisely the same, only everywhere multiplication by $\cos (t)$ will be replaced by multiplication by $\sin (t)$.

10.4.2. Equivalent Spectral Density of the Optimum Circuit of the Method of Scanning the Directional Pattern

Let us turn to the question of the potential accuracy of the method of pattern scanning.

Considering n = 1 in (10.3.31), substituting there expression (10.2.2) and producing calculation of integrals, we obtain

$$S_{\omega RT} = \frac{*}{\mu_{4}^{2} h^{2}} \left\{ C^{2} \int_{-\infty}^{\infty} \frac{S_{0}(\omega)^{2}}{hS_{0}(\omega) + 1} d\omega + \sum_{k=-\infty}^{\infty} |c_{k}|^{2} \int_{-\infty}^{\infty} \frac{1 - hS_{0}(\omega)}{1 + hS_{0}(\omega)} \cdot \frac{S_{0}(\omega) + I}{1 + hS_{0}(\omega + k\Omega)} d\omega \right\}^{-1}, \qquad (10.4.2)$$

where

1]

$$\boldsymbol{c}^{a} = \frac{1}{T_{en}} \int_{0}^{T_{en}} \left[\cos \boldsymbol{\Phi}(t) \right]^{a} dt, \ \boldsymbol{c}_{k} = \frac{1}{T_{en}} \int_{0}^{T_{en}} \cos \boldsymbol{\Phi}(t) e^{ik\theta t} dt; \qquad (10.4.5)$$

 T_{CK} - period of scanning; $\Omega = 2\pi/T_{CK}$ - angular frequency of scanning.

Let us note now that s_{OHT} for measurement of the angle in the other plane will have the same form as (10.4.2), only coefficients c^2 and c_k , should be replaced by s^2 and s_k , respectively:

$$s^{0} = \frac{1}{T_{on}} \int_{0}^{T_{on}} [\sin \vartheta(t)]^{s} dt, \quad s_{h} = \frac{1}{T_{on}} \int_{0}^{T_{on}} \sin \vartheta(t) e^{t/h \theta t} dt. \quad (1.5, 4.5)$$

The complicated form of formula (10.4.2) loes not permit us to see directly from it the law governing change of $S_{\rm OHI}$ depending upon h and other quantities. Therefore we will consider the most interesting limiting cases, as we did in examining the optimum circuit.

-521-
In the case of a rather high frequency of scanning, when $S_0(\omega + \Omega) \approx 0$, the second component in parentheses in (10.4.2) vanishes and we obtain

$$S_{0117} = \frac{\pi}{\mu_0^2 h^2} \left\{ C^2 \int_{-\infty}^{\infty} \frac{S_0(\omega)^2}{1 + h S_0(\omega)} d\omega + C_0^2 \int_{-\infty}^{\infty} \frac{[1 - h S_0(\omega)] S_0(\omega)^2}{[1 + h S_0(\omega)]^2} d\omega \right\}^{-1}.$$
 (10.4.5)

In the usually occurring range of variation of h the second integral in (10.4.5) is negative [in approximation (10.3.26) this occurs when h > 1]. Then S_{ORT} will be the smallest if

$$c_{\bullet} = \frac{1}{T_{\bullet x}} \int_{0}^{x} \cos \vartheta(l) dl = 0, \qquad (10.4.0)$$

and Sour is equal to

$$S_{\text{OHY}} = \frac{\pi}{\mu_a^2 \hbar^2 c^a} \int_{-\infty}^{\infty} \frac{S_a(\omega)^2}{1 + \hbar S_a(\omega)} d\omega \qquad (10.4.7)$$

Accuracy in this case, obviously, does not depend on the form of the law of scanning and is determined only by c^2 . If it was problem of measurement of only one angular coordinate, then the best scanning would be jumps of the directional pattern to two extreme positions in the plane of this angle. Here $\cos \vartheta(t) = \pm 1$, and $c^2 = 1$. However, such scanning is unsuitable if it is necessary to measure angular coordinates of a target in two planes. Roughly speaking c^2 determines that part of the total received power which goes to measurement of the angle in one plane. Coefficient s^2 (10.4.4) has analogous meaning (obviously, $c^2 + s^2 = 4$). In this connection the relationship of c^2 and s^2 should be selected based on the required relationship of accuracies of measurement of angles in the two planes. In particular, if these accuracies must be identical, which normally is the case then we naturally consider $c^2 = s^2 = 1/2$. This we shall assume subsequently.

Let us now turn to the limiting case of low frequencies of scanning. If the frequencies of scanning are so low that for those values of k for which $c_k \approx 0$, we still have $S_0(\omega + k\Omega) \approx S_0(\omega)$, then instead of (10.4.2) we obtain (considering, as a agreed, $c^2 = 1/2$)

$$S_{\text{OIII}} = \frac{\pi}{\mu_a^2 \hbar^3} \frac{1}{\int \limits_{-\infty}^{\infty} \left[\frac{S_{\bullet}(\omega)}{1 + \hbar S_{\bullet}(\omega)} \right]^3 d\omega}$$
(10.4.8)

For comparison of accuracies at high and low frequencies of scanning let us note that during transition from low frequencies of scanning to high quantity $2\pi/\mu_{\rm B}^2 {\rm h}^2 {\rm S}_{\rm OHT}$ changes by

$$\int_{-\infty}^{\infty} \frac{S_{\bullet}(\boldsymbol{\omega})^{*} [1 - hS_{\bullet}(\boldsymbol{\omega})]}{[1 + hS_{\bullet}(\boldsymbol{\omega})]^{*}} d\boldsymbol{\omega}.$$
(10.4.9)

~522~

This integral coincides with the second integral in (10.4.5). As we already established, for sufficiently large values of h [in approximation (10.3.26), when h > 1] this integral is positive, i.e., accuracy of high-frequency scanning is higher than accuracy of low-frequency scanning. For small h everything will be the opposite. This fact is curious. It is explained by the fact that for small h the basic influence on accuracy of direction finding is rendered by natural noises. At low frequencies of scanning the total transmission band of the filters must be smaller,





which leads to decrease of natural noises. Let us now perform calculation by

formula (10.4.2) with approximation of the spectrum (10.3.26). For simplicity we shall consider the case of conical scanning of the directional pattern, i.e., $c_{\pm 1} = 1/2$, and at $i \neq \pm 1$, $c_{\pm} = 0$. Performing the necessary calculations, we obtain

$$S_{oht} = \frac{1}{\mu_{a}^{2}\Delta f_{c}} \cdot \frac{\sqrt{1+h}\left(1+\sqrt{1+h}\right)}{h^{2}} \times \frac{\zeta^{a}+2\left(3h+4\right)\zeta^{4}+}{\zeta^{a}+(5h+10+2\sqrt{1+h})\zeta^{a}+} + \frac{(3h+4)^{2}\zeta^{2}+4h^{2}(h+1)}{4\left(h^{2}+4h+6+2\sqrt{1+h}\right)\zeta^{2}+4h\left(1+h\sqrt{1+h}\right)},$$

$$(1?.4,1')$$

where $\zeta = \Omega/2\Delta f_{c}$.

The graph of the dependence of S_{OHI} on h, calculated by formula (10.4.10), is shown in Fig. 10.11. All laws governing change of S_{OHI} are easily perceived from this graph: with growth of h, S_{OHI} rather

sharply drops; for values of h normal in measuring systems (when h > 1) with decrease of the scan frequency S_{OHI} increases.

+

We note the very simple formulas for S_{OHI} ensuing from (10.4.10) for large and small values of the signal-to-noise ratio for high and low frequencies of scanning:

$$S_{O\Pi T} = \frac{1}{\mu_a^2 \Delta I_o \hbar} \quad \hbar \gg 1, \quad \zeta \gg 1; \\ S_{O\Pi T} = \frac{1}{\mu_a^2 \Delta I_o \hbar} \quad \hbar \gg 1, \quad \zeta \ll 1; \\ S_{O\Pi T} = \frac{1}{\mu_a^2 \Delta I_o \ell \hbar} \quad \hbar \gg 1, \quad \zeta \ll 1; \\ S_{O\Pi T} = \frac{1}{\mu_a^2 \Delta I_o \ell \hbar} \quad \hbar \gg 1, \quad \zeta \ll 1; \\ S_{O\Pi T} = \frac{1}{\mu_a^2 \Delta I_o \ell \hbar} \quad \hbar \gg 1, \quad \zeta \ll 1; \\ S_{O\Pi T} = \frac{1}{\mu_a^2 \Delta I_o \ell \hbar} \quad \hbar \gg 1, \quad \zeta \ll 1; \\ S_{O\Pi T} = \frac{1}{\mu_a^2 \Delta I_o \ell \hbar} \quad \hbar \gg 1, \quad \zeta \ll 1; \\ S_{O\Pi T} = \frac{1}{\mu_a^2 \Delta I_o \ell \hbar} \quad \hbar \gg 1, \quad \zeta \ll 1; \\ S_{O\Pi T} = \frac{1}{\mu_a^2 \Delta I_o \ell \hbar} \quad \hbar \gg 1, \quad \zeta \ll 1; \\ S_{O\Pi T} = \frac{1}{\mu_a^2 \Delta I_o \ell \hbar} \quad \hbar \gg 1, \quad \zeta \ll 1; \\ S_{O\Pi T} = \frac{1}{\mu_a^2 \Delta I_o \ell \hbar} \quad \hbar \gg 1, \quad \zeta \ll 1; \\ S_{O\Pi T} = \frac{1}{\mu_a^2 \Delta I_o \ell \hbar} \quad \hbar \gg 1, \quad \zeta \ll 1; \\ S_{O\Pi T} = \frac{1}{\mu_a^2 \Delta I_o \ell \hbar} \quad \hbar \gg 1, \quad \zeta \ll 1; \\ S_{O\Pi T} = \frac{1}{\mu_a^2 \Delta I_o \ell \hbar} \quad \hbar \gg 1, \quad \zeta \ll 1; \\ S_{O\Pi T} = \frac{1}{\mu_a^2 \Delta I_o \ell \hbar} \quad \hbar \gg 1, \quad \zeta \ll 1; \\ S_{O\Pi T} = \frac{1}{\mu_a^2 \Delta I_o \ell \hbar} \quad \hbar \gg 1, \quad \zeta \ll 1; \\ S_{O\Pi T} = \frac{1}{\mu_a^2 \Delta I_o \ell \hbar} \quad \hbar \gg 1, \quad \zeta \ll 1; \\ S_{O\Pi T} = \frac{1}{\mu_a^2 \Delta I_o \ell \hbar} \quad \hbar \gg 1, \quad \zeta \ll 1; \\ S_{O\Pi T} = \frac{1}{\mu_a^2 \Delta I_o \ell \hbar} \quad \hbar \gg 1; \quad \zeta \gg 1;$$

The simplicity of these formulas permits us to recommend them as basic during appraisals of potential accuracy of the method of pattern scanning. From them it is clear that with low frequencies of scanning, the equivalent spectral density with growth of the signal-to-noise ratio increases considerably slower than with high frequencies of scanning. However, even with low frequencies of coanning as $h \to \infty$ quantity $S_{OHT} \to 0$. This contradicts the established opinion that with low frequencies of scanning even without noises accuracy will be limited thanks to the influence of amplitude fluctuations of the signal. Such an opinion developed only as a result of operations of monoptimal circuits. With optimum processing of signals, as can be seen from what has been said, we succeed in sufficiently completely compensation the harmful influence of amplitude fluctuations of the signal. bet us note that formulas (10.4.11), although calculated for a rather particular approximation for S_{OHT} in more general cases.

§ 10.5. Investigation of Synthesized Circuits for the Method of Scanning of Directional Patterns

10.5.1. Introductory hemarks

In the present section we shall pursue a steay of a group of questions connected with practical realization of circuits, close to optimum. As a result of these investigations we should find how well it is possible to realize the potential accuracy of the actual of pattern scanning. Furthermore, it is necessary to obtain sufficiently simple formulas for conjugation of accuracy of real circuits.

Pirot of all 0 is not solvy to find what features distinguish real circuits from theoretical. Circuits synthesized in the preceding section in the form in which they are depicted in Figs. 10.8, 10.9 and 10.10 are, obviously, very simplified. They reflect only the basic, fundamental operations produced on the signal. In the practical realization these circuits will require additional technical components. The basic ones of these components, of course, should be considered during analysis of the real accuracy of the considered circuits. In itself appraisal of the complexity of technical construction of different circuits is also one of the important characteristics of real circuits.

In Fig. 10.17 as an example there is depicted a circuit of the type of Fig. 10.10 taking into account basic technical components, inevitable during construction of this circuit. The signal in this circuit first enters a sizer, carrying out identical transfer of signal to the intermediate frequency. This mixer is the basic

-524-

source of natural noises added to the signal. Then there follows an UPCh [i-f amplifier] whose passband is usually very great (considerably exceed the width of the



Fig. 10.12. Practical circuit of the radio channel of a goniometer with a scanning pattern: 1 - mixer of r-f amplifier; 2 heterodyne oscillator; 3 - UPCh; 4 - mixer with phase modulation; 5 - controlled heterodyne with phase modulation; 6 - amplitude modulator; 7 - controlled generator of amplitude modulation; 8 - narrow-band filter; 9 system of automatic gain control; 10 - amplitude detector; 11 - phase detector; 12 - GON [reference voltage generator]. spectrum of signal fluctuations). After the UPCh the signal enters a mixer with phase modulation. Here the signal is transferred to the second intermediate frequency with inversion of phase modulation. Then there is an amplitude modulator, carrying out multiplication of the signal by a function varying accoraing to the law of amplitude modulation introduced by the antenna. The frequency of the signal of the heterodyne with phase modulation is

continuously tuned by the system of speed measurement, and its phase modulation, as well as the modulating signal from the amplitude modulator, is tuned by the system of range measurement.

After these operations the signal enters a narrow-band filter whose frequency response is matched as far as possible with the spectrum of fluctuations of the signal (usually, of course, matching concerns only the transmission bandwidth of the filter). In view of the limitedness of the range of linearity of amplifiers in practical circuits there always are used AGC systems, maintaining the necessary signal level in the circuit. From the output of the narrow-band filter the signal enters an amplitude detector and then a phase detector. Reference voltage proceeds to the phase detector from a generator of reference voltages (GON); the same voltage controls scanning of the directional pattern.

Analogous variants of circuits of Figs. 10.8 and 10.9 complicated by back: technical components we shall not give, since they are very similar to the circuit considered just now, from the example of which we clearly see the basic for the constant such complication.

Thus, practical circuits rather greatly differ from their theoretical prototype. Now it is easy to establish in what components of practical circuits difference of the operations produced on signals from idealized optimum operations is possible. First of all, due to the limited accuracy of measurement of speed there will be

-525-

produced insufficiently exact tuning of the frequency of the heterodyne with phase modulation, and after transformation the signal will have a frequency differing somewhat from that to which the narrow-band filter is tuned. Due to the limited accuracy of measurement of range the phase modulation of the heterodyne signal will lag somewhat relative to the phase modulation of the received signal (or outstrip it). The same pertains to the signal of the amplitude modulator.

In the case of pulse radiation to avoid losses of the useful signal pulses of the amplitude modulator (strobes) are made, as a rule, even several times longer than pulses of the received signal. Finally, the form of the spectrum of fluctuations of the reflected signal, and also the signal-to-noise ratio in the circuit, necessary for calculation of the frequency response of the narrow-band filter, are known by us very approximately. This will lead to uncontrollable divergence of the frequency response of the narrow-band filter from the calculated, which may be considerable.

Thus, to exactly realize an optimum circuit in practice is impossible. It is possible to exactly maintain the structure of the circuit, i.e., the sequence of basic operations produced on the received signal; however parameters of the circuit will differ to greater or lesser extent from their optimum values.

We shall make several remarks about comparative appraisal of circuits of Figs. 10.3, 10.9 and 10.10 from the point of view of complexity of their technical construction. It is clear that the circuit of Fig. 10.8 is the most complicated of them, and we shall subsequently allot to it the minimum attention.

Circuits of Figs. 10.9 and 10.10 can be considered identical in complexity; in any case, the difference between them is insignificant. The fact is that the most complicated components of these circuits (UPCh's, mixers with inversion of phase modulation, amplitude modulators, phase detectors) are contained in both circuits in identical number.

Subsequently the circuit of Fig. 10.8 (but with parameters, possibly differing from their optimum values) we shall call a quasi-optimum circuit; the circuit of Fig. 10.2^{-1} - a circuit with narrow-band filters; and the circuit of Fig. 10.10^{-1} a circuit with a broad-band filter. These names correctly reflect the essence of the matter and will be used as basic terms.

10.5.2. Investigation of the Quasi-Optimum Circuit and the Circuit With Narrow-Band Filters

We start our investigation of circuits with the circuit of Fig. 10.8. Let us assume that the heterodyne signal has the form $v_a(t) \cos [\omega_p + \phi(t)]$, where $v_a(t)$

-526-

differs, in general, from the amplitude modulation of the received signal $u_a(t)$, and $\varphi(t)$ differs from its phase modulation $\psi(t)$. Thus, the received signal is multiplice by

The pulse response of filters in the first two channels we designate $h_1(t) \cos [\omega_{\Pi p}t + \theta_1(t)]$. These filters we shall consider identical, which substantially will simplify further calculations. The pulse response of the filter in the third channel we shall designate by $h_2(t) \cos [\omega_{\Pi p} t + \theta_2(t)]$. We set

$$\dot{h}_{i}(t) = h_{i}(t)e^{i\theta_{i}(t)}, \quad H_{i}(i\omega) = \int_{0}^{\infty} \dot{h}_{i}(t)e^{i\omega t}dt, \quad i = 1, 2,$$
 (10.5.2)

i.e., $H_1(i\omega)$ - frequency response of the low-frequency equivalents of the considered filters.

For definitiveness during the analysis of circuits we shall limit ourselves only to consideration of uniform conical scanning of the directional pattern. Modulation of the received signal $U_a(t, \alpha)$ is expressed in this case by formula (10.2.2). Obviously, the basic laws obtained here are qualitatively preserved with other laws of scanning.

The signal at the output of the circuit of Fig. 10.8 with these assumptions can be recorded in the following rather bulky form:

$$z(t) = \int_{-\infty}^{t} \operatorname{Re} \dot{h}_{1}(t-\tau) e^{i\omega_{\pi}v^{(t-\tau)}} y(\tau) \operatorname{Re} v(\tau) e^{i\omega_{\pi}\tau} \cos \Omega \tau d\tau \times \\ \times \int_{-\infty}^{t} \operatorname{Re} \dot{h}_{1}(t-\tau) e^{i\omega_{\pi}v^{(t-\tau)}} y(\tau) \operatorname{Re} v(\tau) e^{i\omega_{\pi}\tau} d\tau - \\ - \left[\int_{-\infty}^{t} \operatorname{Re} \dot{h}_{1}(t-\tau) e^{i\omega_{\pi}v^{-(t-\tau)}} y(\tau) \operatorname{Re} v(\tau) e^{i\omega_{\pi}\tau} d\tau\right]^{2} \cos \Omega t.$$

$$(40.5.3)$$

Producing in reverse order the reasonings by which we performed transition from expression (10.3.24) to (10.3.29), it is possible to reduce (10.5.3) to the form most convenient for further calculations, namely:

$$z(t) = \operatorname{Re} \int_{-\infty}^{t} \dot{h}_{1}^{*} (t-\tau) v(\tau) y(\tau) \cos \Omega \tau e^{i\omega_{0}\tau} d\tau \times \int_{-\infty}^{t} \dot{h}_{1} (t-\tau) v^{*} (\tau) y(\tau) e^{-i\omega_{0}\tau} d\tau - \left| \int_{-\infty}^{t} \dot{h}_{2}^{*} (t-\tau) v(\tau) y(\tau) e^{i\omega_{0}\tau} d\tau \right|^{2} \cos \Omega t.$$

$$(10.5.4)$$

The first stage of calculation of accuracy of the circuit is calculation of the slope of the discrimination characteristic (or transmission factor of the radio channel)

-527-

$$K_{\rm II} = \frac{d}{da} \lim_{T \to \infty} \frac{1}{T} \int_{0}^{T} \overline{z(t)} dt \Big|_{a=0}$$

and of systematic error

$$\Delta = \frac{1}{K_{\Pi}} \lim_{r \to \infty} \frac{1}{r} \int_{0}^{r} z(l) dl \Big|_{q=0}.$$
 (10.5.6)

(10.5.5)

We shall calculate these quantities in detail, since the calculations produced here are typical for calculation of goniometer systems. Multiplying in (10.5.4) the integrals, averaging under the sign of integrals the rapidly oscillating terms, and producing also statistical averaging, we obtain

$$\overline{z(t)} = \int_{-\infty}^{t} \int_{-\infty}^{t} P_{c} U_{a}(t_{1}, \alpha) U_{a}(t_{2}, \alpha) \cos \Omega t_{1} u^{*}(t_{1}) u(t_{2}) \times \\ \times \rho(t_{1} - t_{2}) \dot{h}_{1}^{*}(t - t_{1}) \dot{h}_{1}(t - t_{2}) v(t_{1}) v^{*}(t_{2}) dt_{1} dt_{2} + \\ + N_{0} \int_{-\infty}^{t} |\dot{h}_{1}(t - \tau)|^{a} |v(\tau)|^{s} \cos \Omega \tau d\tau - \\ - \left[\int_{-\infty}^{t} \int_{-\infty}^{t} P_{0} U_{a}(t_{1}, \alpha) U_{u}(t_{2}, \alpha) u^{*}(t_{1}) u(t_{2}) \rho(t_{1} - t_{2}) \times \\ \times \dot{h}_{a}^{*}(t - i_{1}) \dot{h}_{a}(t - t_{2}) v(i_{1}) v^{*}(t_{2}) dt_{1} dt_{2} - \\ - N_{0} \int_{-\infty}^{t} |\dot{h}_{a}(t - \tau)|^{a} |v(\tau)|^{a} d\tau \right] \cos \Omega t.$$
(10.5.7)

Then we allow for the circumstance that functions u(t) and v(t) vary considerably more rapidly than the remaining functions under the signs of integrals in (10.5.7). Averaging them, we have

$$\overline{z(t)} = \varkappa_{1} \int_{-\infty}^{t} \int_{-\infty}^{t} P_{c} U_{a}(t_{1}, a) U_{a}(t_{2}, a) \cos \Omega t_{1} \rho(t_{1} - t_{2}) \times$$

$$\times \dot{h}_{1}^{*} (t - t_{1}) \dot{h}_{1} (t - t_{2}) dt_{1} dt_{2} + \varkappa_{2} N_{0} \int_{-\infty}^{t} |\dot{h}_{1}(t - \tau)|^{2} \cos \Omega t d\tau -$$

$$- \cos \Omega t \left[\varkappa_{1} \int_{-\infty}^{t} \int_{-\infty}^{t} P_{c} U_{a}(t_{1}, a) U_{a}(t_{2}, a) \rho(t_{1} - t_{2}) \times$$

$$\times \dot{h}_{0}^{*} (t - t_{1}) \dot{h}_{0} (t - t_{2}) dt_{1} dt_{2} - N_{0} \varkappa_{2} \int_{-\infty}^{t} |\dot{h}_{2} (t - \tau)|^{2} d\tau \right], \qquad (10.5.8)$$

where

$$\mathbf{x}_{1} = \left| \frac{1}{T_{r}} \int_{0}^{T_{r}} u(t) v^{*}(t) dt \right|^{2}, \quad \mathbf{x}_{1} = \frac{1}{T_{r}} \int_{0}^{T_{r}} |v(t)|^{2} dt. \quad (10.5.9)$$

Finding the derivative of (10.5.8) with respect to α for $\alpha = 0$ and averaging it in time, we find

$$K_{JI} = \frac{P_{e}\mu_{a}\pi_{1}}{2} \left[\int_{0}^{\infty} \int_{0}^{\infty} \dot{h}_{1}^{*}(t_{1}) \dot{h}_{1}(t_{2}) \rho(t_{1} - t_{3}) (1 + \cos \Omega (t_{1} - t_{3})) \times \right]$$

$$\times dt_{1} dt_{3} - 2 \operatorname{Re} \int_{0}^{\infty} \int_{0}^{\infty} \dot{h}_{3}^{*}(t_{1}) \dot{h}_{3}(t_{3}) \rho(t_{1} - t_{2}) \cos \Omega t_{1} dt_{1} dt_{3} =$$

$$= \frac{P_{e}\mu_{a}\pi_{1}}{2\pi\Delta f_{e}} \int_{-\infty}^{\infty} [|H_{1}(i\omega)|^{2} + |H_{1}(i\omega + i\Omega)|^{2} -$$

$$-2 \operatorname{Re} H_{3}(i\omega) H_{3}^{*}(i\omega + i\Omega)] S_{0}(\omega) d\omega. \qquad (10.5.10)$$

Considering $\alpha = 0$ directly in expression (10.5.8), we obtain $\Delta = 0$, i.e., systematic error is absent.

Now we shall study calculation of the basic accuracy characteristic of a circuit - the equivalent spectral density. This quantity is defined, as it is known, in the following way:

$$S_{BKB} = \frac{1}{K_{\Pi}^2} \lim_{T \to \infty} \frac{1}{T} \int_0^T dt \int_{-\infty}^{\infty} \overline{z(t) z(\tau)} d\tau \Big|_{\alpha=0}.$$
 (10.5.11)

Omitting the calculations, which are very similar to those just now made, we obtain a result in the following form:

$$S_{\Theta g_{B}} = \frac{2\pi}{\mu_{a}^{2}h^{\prime a}} \int_{-\infty}^{\infty} [|H_{1}(i\omega)|^{a} + |H_{1}(i\omega)|^{a} |H_{1}(i\omega + i\Omega)|^{a} + \frac{42}{\mu_{a}(i\omega)}|^{a} |H_{1}(i\omega + i\Omega)|^{a} - 4H_{1}(i\omega + i\Omega) \times \\ \times \operatorname{Re} H_{a}(i\omega) H_{a}^{*}(i\omega + i\Omega)] [1 + h'S_{0}(\omega + \Omega)] \times \\ \times [1 + h'S_{0}(\omega)] d\omega \left\{ \int_{-\infty}^{\infty} [|H_{1}(i\omega)|^{a} + |H_{1}(i\omega + i\Omega)|^{a} - \frac{2}{\mu_{a}} (i\omega) H_{a}^{*}(i\omega + i\Omega)] S_{0}(\omega) d\omega \right\}^{-2},$$

$$(10.5.17)$$

where

$$h' = h \frac{\left| \frac{1}{T_r} \int_0^r u(t) v^*(t) dt \right|^2}{\frac{1}{T_r} \int_0^r |v(t)|^2 dt} = h \frac{\pi_1}{\pi_2}.$$
(17.4.13)

Expression (10.5.12) is rather bulky. However, from it we can draw certain general conclusions with respect to the influence on S_{0KB} of various factors. In particular, from (10.5.12) it is clear that imperfectnesses of heterodyning lead to equivalent change of the signal-to-noise ratio according to the formula (10.5.13). It is easy to see that



where the sign of equality is attained only in case v(t) = u(t). For instance, when $u_a(t)$ are pulses of duration τ_n , $v_a(t)$ - pulses of duration τ_c , and there is no phase modulation of the signal (or it is exactly inverted), we obtain:

$$\frac{\left|\frac{1}{T_r}\int_{0}^{T_r}u(t)v^*(t)dt\right|^2}{\frac{1}{T_r}\int_{0}^{T_r}|v(t)|^2dt} = \min\left(\frac{\tau_{\rm H}}{\tau_{\rm g}},\frac{\tau_{\rm o}}{\tau_{\rm H}}\right).$$
(10.5.14)

utuli militikan a dun

Thus, any imperfectness of heterodyning leads to an effect, equivalent to increase of natural noises of the receiver.

This phenomenon has a very general character. In the preceding chapters it was shown that it occurs, for instance, in range and frequency metering radar circuits. Imperfectness of heterodyning in all cases led to equivalent decrease of the signalto-noise ratio, where this decrease is expressed by a single formula (10.5.13). It is obvious that this phenomenon will take place in all circuits of radar goniometers. Therefore, subsequently, during investigation of goniometer circuits we for simplicity shall consider heterodyning ideal. Imperfectness of heterodyning if desired can easily be accounted for by means of replacement of signal-to-noise ratio h by h'.

Let us consider now in greater detail expression (10.5.12) for S_{BKB} . It contains three components: one which does not depend on signal-to-noise ratio h, a second which is proportional to 1/h and, a third which is proportional to $1/h^2$. The first component arises as a result of nonlinear transformation of the useful signal, carried out in the circuit. This component is the result of incorrect selection of characteristics of filters. In the optimum circuit, as we noted in the preceding section, such a component of error is absent. The other two components in (10.5.8) are explained, respectively, by interaction of the useful signal with noise and nonlinear transformation of noise. These components of error cannot be completely eliminated, and it is possible only to decrease them by proper selection of characteristics of filters.

-530-

By direct variation of expression (10.5.12) with respect to frequency responses of filters $H_1(i\omega)$ it is easy to prove that $S_{\rm OVE}$ reaches a minimum, when $|H_1(i\omega)|^2$ is determined by formula (10.3.25), and $H_2(i\omega)$ is determined by formula (10.3.28), where h_{Σ} will be replaced, of course, by h. Then $S_{\rm OVE}$ will become equal to $S_{\rm OVE}$. This confirms the correctness of the obtained results.

Let us return now to the question of the physically realizable equivalent of an optimum filter with frequency response (10.3.28), left unsolved in § 10.3. From formula (10.5.12) it is clear that frequency response $H_2(i\omega)$ is contained in it in the form of expressions Re $H_2(i\omega)H_2(i\omega + i\Omega)$. This means that S_{GKB} will not vary if instead of a filter with characteristic $H_{2 \text{ ORT}}(i\omega)$ we use a filter with characteristic $H_{2 \text{ ORT}}(i\omega)$ we use a filter with characteristic $H_{2 \text{ ORT}}(i\omega)e^{-ik\omega T}cK$, where k is any integer. Selecting k so large that $h_{2 \text{ ORT}}(kT_{\text{ck}}) = \sqrt{h_{\Sigma}} \Delta f_c v(t, t + kT_{\text{CK}})|_{\alpha=0} \approx 0$, it is possible to make a filter with characteristic $H_{2 \text{ ORT}}(i\omega)e^{-ik\omega T}cK$ physically realizable.

We turn to consideration of the circuit of Fig. 10.9. It, as it was established, is optimum for high frequencies of scanning. Therefore, besides allowance for nonidealnesses of processing in this circuit it is of interest to estimate also the worsening of accuracy as compared to the optimum which this circuit gives for other face values of the frequency of scanning. If filters in the circuit have frequency response $H_1(i\omega)$, and the heterodyne signal has the form (10.5.1), the expression for equivalent spectral density is obtained immediately from (10.5.12), if there we set $H_2(i\omega) \equiv 0$. Then

$$S_{\Theta R \Theta} = \frac{\pi}{2\mu_{\sigma}^{2}h^{4}} \int_{-\infty}^{\infty} [|H_{1}(i\omega)|^{4} + |H_{1}(i\omega)|^{4}|H_{1}(i\omega + i\Omega)|^{4}] \times \\ \times [1 + hS_{\Theta}(\omega)] [1 + hS_{\Theta}(\omega + \Omega)] d\omega \times \\ \times \left\{ \int_{-\infty}^{\infty} [|H_{1}(i\omega)|^{4} + |H_{1}(i\omega + i\Omega)|^{4}] S_{\Theta}(\omega) d\omega \right\}^{-2}.$$

$$(4\omega, 4, 10)$$

In spite of the considerable simplification of this formula as compared to (10.5.12), it is difficult to perceive the dependence of accuracy of measurement on the form of the frequency response of the filter or the magnitude of the frequency of scanning directly from it. For investigation of these dependences it is necessary to produce calculation of integrals in (10.5.13), using suitable approximations of

-531-

the frequency response of the filters and of the spectrum of fluctuation.

Let us produce this calculation, approximating frequency response of the filter $H_4(i\omega)$ by expression

$$H_1(ie) = -\frac{1}{1 + \frac{ie}{2\Delta_1^i + \Delta_1^i}},$$
 (10.5.16)

and $S_0(\omega)$ by expression (10.3.26). Here we obtain

$$S_{0BB} = \frac{\xi^{2} + (1+x)^{3}}{\mu_{a}^{2} \delta I_{0} [\xi^{2} + 2(1+x)^{3}]^{3}} \left\{ \begin{bmatrix} (1+x) [\xi^{2} + (1+x)^{2}(1+2x)] \\ 2(1-x) x [\xi^{2} + (1+x)^{2}] \end{bmatrix} + \frac{2x (1+x) [\xi^{4} + 5\xi^{2} (1+x^{2}) + 4(x-1)^{2} (1+x) (1+2x)]}{[\xi^{2} + (x-1)^{2}] (\xi^{2} + 4) (\xi^{2} + 4x^{2})} - \frac{x^{4} [\xi^{4} + \xi^{2} (5x^{2} + 4x + 3) + 2(x-1)^{2} (x+1) (2x+1)]}{(1-x) [\xi^{2} + (x-1)^{2}] (\xi^{2} + 4x^{2})} \right] + \frac{1}{h} \left[\frac{2x^{2} (1+x) [\xi^{4} + \xi^{2} (5x^{2} + 4x + 3) + 2(x-1)^{2} (x+1) (2x+1)]}{[\xi^{2} + (x-1)^{2}] (\xi^{2} + 4x^{2})} + \frac{\xi^{4} (1+2x) + \xi^{2} (1+x^{2}) (4x+3) + 2(x+1)^{4} (1+2x)}{2x [\xi^{2} + (1+x)^{2}]} \right] + \frac{1}{h^{2}} \frac{(x+1)^{2} [\xi^{2} + (x+1)^{2}] (\xi^{2} + 8x^{2})}{2x (\xi^{2} + 4x^{2})} \right\},$$

$$(10.5.17)$$

where $\zeta = \Omega/2\Delta f_c$; $x = \Delta f_{\dot{C}}/\Delta f_c$.

Let us investigate formula (10.5.17). Although it is obtained with rather particular approximations of the frequency response of the filters and of the spectrum of fluctuations of the signal, the laws governing change of S_{BKB} expressed by this formula are preserved, too, in more general cases. First of all we shall consider limiting cases when the laws of variation of S_{BKB} are considerably simplified. For small x instead of formula (10.5.17) we obtain

$$S_{\text{BKB}} = \frac{1}{2\mu_{\text{B}}^2 \Delta f_{\text{BK}}} \frac{\zeta^2 + 1}{(\zeta^2 + 2)^2} \left(1 + \frac{\zeta^2 + 2}{h} + \frac{\zeta^2 + 1}{h^2} \right). \tag{10.5.18}$$

Thus, the spectral density of error with decrease of x, i.e., with marrowing of the passband of the filter, grows inversely proportionally to x. For small x the component of error from nonlinear transformation of the useful signal, monotonically drops with growth of the frequency of scanning. The remaining components of error almost do not depend on ζ .

For large x, i.e., with expansion of the passband of the filters, formula (10.5.17) takes the very simple form

$$S_{0KB} = \frac{x}{2\mu_s^2 \Delta / ch^2}$$
 (10.5.19)

From this it is clear that for sufficiently broad-banded filters the spectral density of error of measurement is caused basically by nonlinear transformation of the signal and grows proportionally to x. Error in this case does not depend on the frequency of scanning.

For large $\boldsymbol{\zeta}$ equivalent spectral density is equal to

$$S_{0K0} = \frac{1}{2\mu_0^2 \Delta f_0} \left(\frac{2+1/x}{\hbar} + \frac{x+2+1/x}{\hbar^2} \right). \tag{10.5.20}$$

From (10.5.20) one may see that S_{BKB} reaches its minimum at $x = \sqrt{1 + h}$. This is understandable, since the considered circuit for large ζ is optimum. Decrease or increase of x in comparison with $\sqrt{1 + h}$ leads to increase of the spectral density. For small ζ we also obtain a very simple formula:

$$S_{aKa} = \frac{1}{4\mu_a^2 \Delta f_e} \left(2 + \frac{1}{x} + 2 \frac{2 + \frac{1}{x}}{h} + \frac{x + 2 + \frac{1}{x}}{h^3} \right). \tag{10.5.21}$$

in shakes.

In this case the optimum for x is attained when x = 1 + h (S_{3KB} reaches its minimum value), and not when $x = \sqrt{1 + h}$, since for small ζ this circuit is not optimum, and the laws will be different here.

In order to judge the relative change of accuracy of the considered circuit as compared to the optimum, it is of interest to investigate ratio S_{3KG}/S_{OHT} . Spectral density S_{OHT} with approximation of the spectrum of fluctuations by formula (10.3.76) was calculated in the preceding section and was presented by formula (10.4.16). The graph of the dependence of ratio S_{3KG}/S_{OHT} on h for different values of x and ζ is shown in Fig. 10.13. From this figure it is clear that at high frequencies of scanning ($\zeta \approx 100$) accuracy of the considered circuit is close to optimum in a very broad range of variation of x (for x = 2-10 ratio $S_{3KG}/S_{OHT} \approx 1-1.5$). For low frequencies of scanning accuracy of the considered circuit rather strongly differs frethe optimum; the difference is even greater, the larger h. This is explained by the fact that at low frequencies of scanning the equivalent spectral density of the considered circuit as h $\rightarrow \infty$ seeks a finite magnitude (since there is a component of S_{0KB} caused by nonlinear transformation of the useful signal); for an optimum circuit equivalent spectral density as h $\rightarrow \infty$ tends to zero.

We shall now discuss the spectral density of parametric fluctuations. As it was shown in Chapter VI, it has the form

$$S_{\text{Hap}} = \frac{1}{2K_{\text{H}}^2} \frac{d^2}{da^2} \lim_{T \to \infty} \frac{1}{T} \int_{0}^{T} dt \int_{0}^{\infty} \overline{z(t)z(s)} \, ds \Big|_{a=0}. \qquad (10.....22)$$

-533-



Fig. 10.13. The dependence of S_{212}/S_{OUT} on h for a circuit with narrow-band filters of the radio channel of a goniometer with a scanning pattern: $\zeta = 1.00; - -\zeta =$ 10; -. -. $\zeta = 1.$ We shall calculate $S_{\rm hep}$ for the case of high frequencies of scanning, when $S_{\rm DHB}$ is already small, and a considerable share of total error is caused by parametric fluctuations. Calculations in finding $S_{\rm HBL}$ are very similar to those by which we calculated $K_{\rm H}$ and $S_{\rm DHB}$. Therefore, we shall immediately give the final result



(10.5.23)

Thus, S_{HSP} does not depend on the signal-to-noise ratio and frequency of scanning. Calculation by (10.5.23) with approximation (10.5.16) and (10.3.26) gives

$$S_{\text{gap}} = \frac{1}{\Delta f_{\bullet}} \frac{1+3x+x^{*}}{x(1+x)}, \qquad (10.5.24)$$

For large x quantity $S_{\Pi\Theta\uparrow} \rightarrow 1/\Delta f_{\Theta}$; for small x spectral density $S_{\Pi\Theta\uparrow} \rightarrow 1/x\Delta f_{\Theta} = 1/\Delta f_{\oplus}$. Laws of this kind are preserved qualitatively also for other forms of frequency response and of the spectrum of fluctuations.

10.5.3. Investigation of a Circuit with a Broad-Band Filter

Let us consider now the circuit of Fig. 10.10. If we designate the frequency response of the filter in this circuit $H(i\omega)$, the expression for the equivalent spectral density of such a circuit could be obtained from (10.5.8), if we set $H_1(i\omega) \equiv 0$, and $H_p(i\omega) = H(i\omega)$.

However let us note that for not too low frequencies of scanning modulation of received signal introduced by scanning pattern will be delayed during passage through

-534-

the filter. This delay should be, of course, introduced in the reference signal from the GON in order to ensure a maximum value of the transmission factor of the radio channel. Calculations of characteristics of a circuit in these conditions have to be redone.

The signal at the output of the circuit of Fig. 10.10 has the form

$$z(l) = \cos\Omega (l - \tau_a) \left| \int_{-\infty}^{t} \dot{h} (l - \tau) y(\tau) u(\tau) e^{l \cdot \tau_a \tau} d\tau \right|^{2}, \qquad (10.5.25)$$

where $\dot{h}(t) - \text{complex amplitude of the pulse response of the filter in this circuit [determined as in (10.5.20)], and delay <math>\tau_3$ should be selected from the condition of maximum K_{μ} . Using formula (10.5.5), we can obtain (calculations are similar to those already made, and therefore we omit them)

$$\Delta = 0,$$

$$\max_{\mathbf{x}} \mathcal{K}_{\mathbf{x}} = \frac{P_{c}\mu_{a}}{2\pi\Delta f_{e}} \Big| \int_{-\infty}^{\infty} H(i\mathbf{o}) H^{*}(i\mathbf{o} + i\Omega) S_{e}(\omega) d\omega \Big|.$$
(10.5.26)

In an analogous way from formula (10.5.11) we have

$$S_{\Theta K \mu} = \frac{\pi}{\mu_{a}^{2}} \int_{-\infty}^{\infty} |H(i \otimes |^{2} |H(i \otimes + i \otimes |^{2})|^{2} \times \left\{ S_{\bullet}(\bullet) S_{\bullet}(\bullet + \Omega) + \frac{1}{h} [S_{\bullet}(\bullet) + S_{\bullet}(\bullet + \Omega)] + \frac{1}{h^{2}} \right\} d\bullet \times \\ \times \left[\int_{-\infty}^{\infty} |H(i \otimes H^{*}(i \otimes + i \Omega)/S_{\bullet}(\bullet) d\bullet \right]^{-2}.$$

$$(10.5.27)$$

We shall perform our calculation from this formula, using approximation (10.4.10) for $|H(i\omega)|^2$ and (10.3.26) for $S_0(\omega)$. As a result of calculation of integrals (10.5.27) we obtain

$$S_{0RB} = \frac{x+1}{\mu_{a}^{2} \Delta f_{0} x} \left\{ \frac{1}{A^{0}} \frac{\left[\zeta^{0} + (1+x)^{2}\right](1+x)}{\zeta^{0} + 4\left(1+x\right)^{2}} + \frac{1}{A} \frac{\zeta^{0} + \zeta^{0} \left(5x^{0} + 4x + 3\right) + 2\left(x-1\right)^{2} \left(x+1\right)\left(2x+1\right)}{\left[\zeta^{0} + (x-1)^{2}\right]\left[\zeta^{0} + 4\left(x+1\right)^{2}\right]} + \frac{\zeta^{0} + 5\zeta^{0} \left(1+x\right)^{2} + 4\left(x-1\right)^{2} \left(x^{2} + 3x+1\right)}{\left[\zeta^{0} + (x-1)^{2}\right]\left(\zeta^{0} + 4\right)\left[\zeta^{0} + 4\left(x+1\right)^{2}\right]} \right\}, \qquad (10.5.28)$$

where as before $\zeta = \Omega/2\Delta f_{\rm C}$; $x = \Delta f_{\rm c}^{*}/\Delta f_{\rm C}^{*}$

Let us consider different limiting cases for ζ and x. For small x from $(1^{(j_1,j_2,j_3)})$

we have

$$S_{ana} = \frac{1}{\mu_a^2 \Delta f_o x} \frac{1}{\xi^2 + 4} \left(1 + \frac{\xi^2 + 2}{\hbar} + \frac{\xi^2 + 1}{\hbar^3} \right). \tag{10.5.29}$$

From this it is clear that the equivalent spectral density for small x is inversely proportional to x and monotonically drops with increase of ζ (the drop is basically due to the component of error caused by nonlinear transformation of the useful signal). Comparison of (10.5.29) with (10.5.18) shows that for small x the circuit of Fig. 10.10 has spectral density less by a factor of $(\zeta^2 + 1) (\zeta^2 + 4)/(\zeta^2 + 2)^2$ than the circuit of Fig. 10.9. However, let us note that

$$1 < \frac{(\zeta^{2}, +1)(\zeta^{2}, +4)}{(\zeta^{2}+2)^{2}} < 1, 1,$$

so that both circuits for small x are practically identical.

For large x from (10.5.28) we obtain

$$S_{a \times B} = \frac{x}{\mu_a^2 \Lambda f_c h^2}.$$
 (10.5.30)

Comparison of (10.5.30) and (10.5.19) shows that for large x the circuit of Fig. 10.9 has a spectral density half as large as the elecuit of Fig. 10.10. We now consider the case of large 5. From (10.5.28) we obtain

$$\dot{S}_{BRD} = \frac{1}{\mu_a^2 \Delta f_a} \left(\frac{1+1/x}{\hbar} + \frac{x+2+1/x}{\hbar^2} \right).$$
(10.5.31)

Here, obviously, there is an optimum at $x = \sqrt{1 + h}$. Comparison of (10.5.31) and (10.5.20) shows that for large ζ the circuit of Fig. 10.9 has the best accuracy (for small h, twice as good, for large h - better by a factor of $\frac{1}{2} \frac{2x + 1}{x + 1}$). Finally, for low frequencies of scanning (small ζ) from (10.5.28) we obtain

$$S_{\text{ONE}} = \frac{1}{4\mu_a^2} \frac{1}{\Delta f_o} \left(2 + \frac{1}{x} + 2\frac{2 + \frac{1}{x}}{h} + \frac{x + 2 + \frac{1}{x}}{h^2} \right), \quad (10.137)$$

which completely coincides with (10.5.21). Thus, comparison of circuits of $F1_6$. 10.9 and Fig. 10.10 shows that these circuits for identical parameters (x and ζ) have approximately identical accuracy; the circuit of Fig. 10.9 is less critical to widening of the pass band of the filters and ensures for wide filters twice the accuracy (with respect to equivalent spectral density) of the circuit of Fig. 10.10.

For more detailed study of accuracy of the circuit of Fig. 10.10 in Fig. 10.14 there is constructed the graph of the dependence of S_{HB}/S_{OHT} on h for this circuit, calculated with the same approximations as in the preceding case. From this graph it is clear that the considered circuit, in general, realizes the potential accuracy

-536-



ι j

Fig. 10.14. The dependence of $S_{\text{2HB}}/S_{\text{CHT}}$ on h for the circuit with a broad-band filter of the radio channel of a goniometer with a scanning pattern: $\zeta = 100; - - \zeta = 10; - - - \zeta = 1.$

of the method of pattern scanning worse than the preceding circuit. For low frequencies of scanning ($\zeta < 10$) only in the range h = 1-10 does it ensure accuracy worse than the potential by a factor of 1-1.5 (with respect to equivalent spectral density). With further growth of h ratio $S_{\rm BKF}/S_{\rm ORT}$ rather sharply increases (due to the component of error caused by nonlinear transformation of the signal). For high frequencies of scanning ($\zeta \approx 100$) accuracy of the circuit nears the optimum ($S_{\rm BKF}/S_{\rm ORT} = 1-1.5$) when h > 10.

We note that ratio S_{3KB}/S_{OIII} for both the considered circults in a wide range of change of their parameters does not exced 4.

§ 10.6. <u>Method of Scanning with Compensation and</u> <u>Instantaneous Amplitude Comparison</u> <u>of Signals (Optimum Circuits)</u>

Besides the method of scanning the directional pattern of the receiving antenna in goniometrical practice there are widely applied the method of scanning with compensation and the method of instantaneous amplitude comparison of signals (IAC). In the use of these methods there are simultaneously observed several signals, received

-537-

by different directional patterns. Inasmuch as fluctuations of amplitudes of the signals received by the various patterns will be identical, this is the prerequisite for their elimination or, at least, partial compensation. Therefore from these methods there is anticipated heightened, as compared to method of pattern scanning, accuracy of measurement during a fluctuating signal, and also increased noise immunity against amplitude-modulated active interferences.

Finally, a very important consideration in favor of the method of scanning with compensation and IAC is connected with the efficiency of use of the aperture of the antenna system. The fact is that during use of the method of pattern scanning one may approximately assume that there is used power from only part of the antenna aperture. The section f om which power is selected shifts over the aperture of the antenna in accordance to a certain periodic law. In the presence of angular divergence of the target form the equisignal direction the distribution of the field in the aperture of the antenna is asymmetric, and, consequently, there occurs periodic change of amplitude of the received signal. If all the power from the aperture of the antenna system was used, the received signal would have constant amplitude and would not contain information about angular coordinates of the target. Thus, during use of the method of pattern scanning the received power comprises only a certain share of the power reaching the aperture of the antenna system (usually half). This circumstance is fundamental.

During use of scanning with compensation and LAC one may approximately assume that in one channel there is used power from one part of the antenna aperture, and in other - power from the remaining part, so that the total received power is equal to all the power entering the aperture of the antenna system. In the case of asymmetric distribution of the field in the aperture (with angular divergence of the target from the equisignal direction) the powers entering each of the channels are unequal and their difference depends on the angular coordinates of the target. Thus, the compensation method and IAC ensure more efficient use of the aperture of the antenna system than the method of pattern scanning: total power received during use of these methods is approximately twice the average power received with the method of scanning the directional pattern of the antenna system.

In the present section we shall study potentialities of the method of scanning with compensation and IAC and their optimum circuits. We combine consideration of these two methods in view of their very great similarity physically and mathematically.

-538-

From the technical point of view these methods, of course, rather greatly differ, which is not decisive in this case.

10.6.1. Synthesis of Optimum Circuits for Methods of Scanning with Compensation and IAC

The optimum circuit for the method of scanning with compensation can easily be obtained from the general circuit of Fig. 10.7, where it is necessary to set n = 2 (two antennas), $\Phi_i(t, \alpha) = 0$ (phase centers of antennas coincide and are fixed), and $U_{ia}(t, \alpha)$ is determined by formula (10.2.4). Here, using in (10.2.4) different concrete values for laws of scanning of patterns $\hat{s}_i(t)$, it is possible to consider a large number of particular cases of the method of scanning with compensation. However we immediately will separate methods of scanning in which this method has the best potential properties and subsequently will limit our consideration to only such cases.

From consideration of the general circuit of Fig. 10.7 it is clear that the optimum circuit of the method of scanning with compensation, besides the two basic channels, contains, in general, an additional, third channel which, as was shown in § 10.3, carries out singling out of false information from the amplitude fluctuations of the signal, which permits us subsequently to partially compensate the harmful influence of these fluctuations. It is easy to see that with fulfillment of condition

 $h_{\mu}\mu_{a_1}\cos\vartheta_1(t) + h_{\mu}\mu_{a_2}\cos\vartheta_1(t) = 0 \qquad (10.1.1)$

in the circuit measuring angle α the third channel disappears. Consequently, with fulfillment of condition (10.6.1) in this circuit there occurs automatic compensation of amplitude fluctuations of signals, i.e., there is realized that concept which was pursued during development of the method of scanning with compensation.

Let us consider conditions (10.6.1) in greater detail. They signify a definition symmetry of receiving channels. If $h_1 = h_2$ and the directional patterns of the antennas are identical, these conditions will be satisfied when

$$\boldsymbol{\vartheta}_{1}(t) = \boldsymbol{\vartheta}_{1}(t) + \boldsymbol{\pi}. \tag{12.6.7}$$

Expression (10.6.2) means that the directional patterns occupy the extreme possible positions and scan without changing their mutual location. The directional patterns forming such a figure are sometimes called "waitzing."

Thus, only with fulfillment of condition (10.6.2) does the method of scanning with compensation ensure complete compensation of amplitude fluctuations of the

-539-

signal. Any other forms of scanning one should recognize as unsatisfactory, since they do not ensure realization of the potential of this method. In particular, the widely known variant of the method of scanning with compensation with one fixed and one scanning pattern should be recognized as a failure. Henceforth we shall limit ourselves only to consideration of this method with "waltzing" patterns, considering here spectral densities of noises in the channels to be identical.

We shall make a series of remarks about the optimum circuit of scanning with compensation. Making the proper simplifications in formula (10.3.29), we can record the optimum operation of the radio channel for this method in the form

$$z(\tau) = \int_{t-At}^{t} h_{10\pi\tau} (\tau - s) \cos \omega_{\pi\mu} (\tau - s) [y_1(s) - y_1(s)] \times \\ \times \cos \vartheta (s) u_n(s) \cos [\omega_r s + \psi(s)] ds \int_{t-At}^{t} h_{10\pi\tau} (\tau - s) \times \\ \times \cos \vartheta_{\pi\mu} (\tau - s) [y_1(s) + y_2(s)] u_n(s) \cos [\omega_r s + \psi(s)] ds,$$
(10.6.3)

where $y_1(s)$ - received signals, expressed by formulas (10.3.3).

Let us designate the useful components of these signals by $\tilde{y}_i(s)$, i.e., $\tilde{y}_i(s) = y_i(s) - \sqrt{N_0 n_i(s)}$. Then

$$y_{1}(s) - y_{2}(s) = \overline{y}_{1}(s) - \overline{y}_{1}(s) + \sqrt{N_{o}}n_{1}(s) - \sqrt{N_{o}}n_{2}(s) =$$

$$= \overline{y}_{1}(s) - \overline{y}_{2}(s) + \sqrt{N_{o}}n_{-}(s), \quad n_{-}(s) = n_{1}(s) - n_{1}(s), \quad (10.6.4)$$

$$y_{1}(s) + y_{2}(s) = y_{1}(s) + \overline{y}_{1}(s) + \sqrt{N_{o}}n_{1}(s) + \sqrt{N_{o}}n_{2}(s) =$$

$$= \overline{y}_{1}(s) + \overline{y}_{1}(s) + \sqrt{N_{o}}n_{2}(s), \quad n_{+}(s) = n_{1}(s) + n_{2}(s).$$

Inasmuch as $n_1(z)$ and $n_2(z)$ are independent, independent, too, will be noises $n_+(z)$ and $n_-(z)$. It follows from this that from the point of view of statistical characteristics it is possible to form the sum and difference of signals after their mixing with the two independent noises or it is possible to add independent noises already to the "sum" and "difference" signal. This remark is very significant, since technically it is often more convenient to form the sum and difference signals directly at the outputs of the antenna system. Using this remark, it is possible to present the circuit of the optimum radio channel in the compensation method in the form depicted in Fig. 10.15.

Let us discuss in greater detail the operation of the circuit of Fig. 10.15. In this circuit signals from outputs of the antenna system first enter a unit which forms their sum and difference (hybrid ring junction, T-junction, etc). The sum and difference signals are heterodyned. Then the difference signal is multiplied

-540-



Fig. 10.15. Optimum circuit of the radio channel of a goniometer using scanning with compensation: 1) optimum filters with frequency response $H_{1 \text{ OHT}}(i\omega)$.

the attentive study which will be conducted in the next section. When the frequency of scanning of the directional patterns is sufficiently small as compared to the width of the passband of the filters in the circuit of Fig. 10.15, multiplication of the difference signal by a function varying according to the law of scanning can be carried out after filtration. Here the circuit can be transformed to the form depicted in Fig. 10.16. Here signals directly from the output of the ancenna system



Fig. 10.16. Optimum circuit of the radio channel of a goniometer using scanning with compensation for low frequencies of scanning: 1) optimum filters with frequency response $H_{1 \ ORT}(i\omega)$; 2) square-law detectors.

by a function varying according to the law of scanning of the directional patterns. After these transformations the signals are filtered by optimum filters with response (10.3.25) and are multiplied. The optimum circuit of Fig. 10.15 is not more complicated than known circuits of the method of scanning with compensation and therefore deserves 1

u.

are heterodyned, are filtered by optimum filters with response (10.3.25), are detected by squarelaw detectors, and are subtracted; the obtained difference is multiplied by a function varying according to the law of scanning. This circuit, well-known in goniometer practice,

is close to the optimum only for low-frequency scanning of the directional patterns.

The optimum circuits for the IAC method are very similar to the circuits of scanning with compensation. If we take as initial prerequisites identity of directional patterns of the antennas in the goniometer with IAC and identity of noises added to the signals, the optimum circuit for IAC will be obtained from the circuit synthesized just now for the method of scanning with compensation (Fig. 10.1) if we there set \$(t) = 0. The optimum circuit for IAC is shown in Fig. 10.17. Conjously it can be converted identically to the form in Fig. 10.18. Both variants of circuits for IAC are known and have been described in the literature [49].

Let us note that with unequalness of directional patterns of the antenna system and unequalness of noises in optimum circuits for IAC besides the two basic channels there would appear, just as for the method of scanning with compensation, an additional channel intended for compensation of harmful effects caused by the shown

-541-



Fig. 10.17. Variant of an optimum circuit of the radio channel of a goniometer with IAC (circuit with multiplication of signals). 1) optimum filters with frequency response curve $H_{1-OBT}(1\omega)$.



Fig. 10.18. Variant of an optimum circuit of the radio channel of a goniometer with IAC (circuit with subtraction of signals): 1) optimum filters with frequency response curve $H_{1-CHT}(i\omega)$; 2) square-law detectors.

of these cases does not have practical value. In conclusion we note that for

disymmetry. However, consideration

measurement of an angular coordinate of a target in another plane the optimum circuits of the method of scanning with compensation, otviously, do not change, only multiplication by $\cos \vartheta(t)/\overline{\text{De}}$ replaced by multiplication by $\sin \vartheta(t)$. Optimum circuits of IAL, in general, do not depend on what angular coordinate of the target is measured: IAC circuits will measure one or the other angular coordinate of the target depending on the pair of

outputs of the antenna to which it is joined.

10.6.2. Equivalent Spectral Densities of Optimum Circuits for Methods of Scanning with Compensation and IAC

Let us turn to the question of potential accuracy c? the method of scanning with compensation and IAC. We first consider the compensation method. Considering in (10.3.31) n = 2, $\Phi_i(t, \alpha) = 0$, introducing $U_{ia}(t, \alpha)$ according to expression (10.2.4), considering here all assumptions of symmetry made in the beginning of this section, in particular (10.6.2), and then calculating integrals in (10.3.31), for equivalent spectral density of the optimum circuit we obtain the following expressi

$$S_{\text{ony}} = \left[\frac{h_{2}^{2} \mu_{a}^{2} c^{4}}{\pi} \int_{-\infty}^{\infty} \frac{S_{a}(\omega)^{a}}{1 + h_{z} S_{a}(\omega)} dw\right]^{-1}, \qquad (10.6.5)$$

where $h_{\Sigma} = h_1 + h_2$ (i.e., $h_{\Sigma} - ratio$ of total signal power received by both directional patterns, to the power of noise of one channel in the band of signal fluctuations), and coefficient c^2 is the same as in formulas of accuracy for the method of pattern scanning (10.4.3). Equivalent spectral density during measurement of an angular coordinate in another plane is expressed by the same formula as (10.6.5), where it is necessary to replace c^2 by s^2 (10.4.4). As also in the method of pattern scanning, coefficients c^2 and s^2 , the sum of which is equal to 1, characterize,

roughly speaking, the distribution of received power for measurement of angles in various planes.

Usually there is required identical accuracy of measurement of both angles; therefore we naturally select the law of scanning so that

$$c^2 = s^2 = \frac{1}{2}.$$
 (10.6.6)

This we shall assume subsequently. However, note that if we need to measure only one of the angles, best accuracy will be rendered by switching of the directional patterns in the plane of this angle or even measurement with fixed patterns, ideated in the given plane; in these cases $\cos i(t) = \pm 1$ and $c^2 = 1$. For motionless patterns, we obtain nothing other than a two-dimensional variant of the IAC method.

It is easy to show that potential accuracy of IAC when there is required measurement of angles in two planes with identical accuracy is expressed by formula (10.0.5), where $e^2 = 1/2$. From this it follows that potential accuracy of IAC and of the method of scanning with compensation are identical.

Comparison of formulas (10.6.5) and (10.4.9) shows also that potential accuracy of IAC (or of the method of scanning with compensation) and of the method of pattern scanning (with use in the latter of high frequencies of scanning) are identical if we consider identical the total powers received in each case. However, as was already shown, the power received in the method of pattern scanning is approximately half the total power received in the method of scanning with compensation or with IAC (for an antenna with the same total aperture area). Thus, if we fix the aperture area of the antenna potential accuracy of the method of pattern scanning, even with high frequencies of scanning, will be less than the accuracy of IAC (of the method of scanning with compensation). However, we note that this is connected with the weight use of the aperture of the antenna system during the method of pattern scanning.

Let us consider formula (10.6.5) in greater detail. For small noises, i.e., when $h_7 >> 1$, from (10.6.5) we obtain the very simple expression

$$S_{\text{OHV}} = \frac{1}{k_{\text{O}}^2 h_{\text{I}} \Delta f_{\text{O}}} \qquad (\cdot \cdot \cdot . 7)$$

「「「「「「「」」」」」

(remember that we set $c^2 = 1/2$).

Inasmuch as $h_{\Sigma} = (P_{C-1} + P_{C-2})/2N_0/f_c$, equivalent spectral density in this call does not depend on statistical characteristics of fluctuations of the signal.

In general we make our calculation by formula (10.6.5), using approximation (10.3.20) of the spectrum of fluctuations. Here, we obtain

$$S_{\text{outr}} = \frac{1}{\frac{1}{\mu_a^2 \Delta f_o}} \frac{\sqrt{1 + h_z} (1 + \sqrt{1 + h_c})}{h_z^2}.$$
 (10.6.8)

We already have the graph of function (10.6.3) in Fig. 10.11 (case $\zeta = \infty$).

We note also the formula for S_{ONT} for small values of h_{Σ} , easily obtained from (10.5.8):

$$S_{\text{ORT}} = \frac{2}{\mu_{a}^{2} \Delta f_{o} h_{z}^{2}}.$$
 (10.6.9)

2

Subsequently for brevity we shall call the elecuits of Figs. 10.15 and 10.17 eircuits with multiplication, and the circuits of Figs. 10.16 and 10.15 - circuits with subtraction of signals.

§ 10.7. Investigation of Synthesized Circuits for Methods of Scanning with Compensation and for IAC

We shall investigate real accuracy of synthesized circuits for the method of accuracy with compensation and, correspondingly, for IAC. The meaning of this investigation is to account for the influence on accuracy of different deviations from optimality in parameters of these circuits, inevitable during their practical realization.

Peculiarities appearing during practical realization of theoretically synthesized circuits were described in sufficient detail in § 10.5. All the facts mentioned there, obviously, directly apply to the considered circuits as well. However here there are certain additional peculiarities which must be considered during calculation of real accuracy. Basically they reduce to the fact that in each channel of these circuits there are mixers and UFCh's , and these devices are complicated and difficult to make identical; from this there follows nonidentity of the frequency responses of the channels inevitable in practical realization of such circuits, the influence of which on accuracy we must consider.

Thus, we shall assume subsequently that filters in channels have different pulse responses

$h_1(t)\cos[\omega_{ny}t+\theta_1(t)]$ and $h_n(t)\cos[\omega_{ny}t+\theta_n(t)]$.

For frequency responses we preserve designations of (10.5.2).

10.7.1. Investigations of Circuits with Multiplication of Signals

The output signal from the circuit of Fig. 10.15 can, obviously, be recorded in the form

$$z(t) = \operatorname{Re} \int_{-\infty}^{\infty} \dot{h}_{1}(t-\tau) [y_{1}(\tau) + y_{2}(\tau)] u(\tau) e^{i\omega_{0}\tau} d\tau \times \int_{-\infty}^{\infty} \dot{h}_{2}^{*}(t-\tau) [y_{1}(\tau) - y_{2}(\tau)] u^{*}(\tau) e^{-i\omega_{0}\tau} \cos \vartheta(\tau) d\tau, \qquad (10.7.1)$$

where $y_i(t)$, (for i = 1, 2) are determined by formula (10.3.3), in which, in turn, $y_{ai}(t, \alpha)$ are expressed by formula (10.2.4).

The output signal from the circuit of Fig. 10.17 has a simpler form:

$$(t) := \operatorname{Re} \int_{-\infty} \dot{h}_{1} (t - \tau) [y_{1}(\tau) + y_{1}(\tau)] u(\tau) e^{i u_{0} \tau} d\tau \times$$

$$\times \int_{-\infty} \dot{h}_{2} (t - \tau) [y_{1}(\tau) - y_{1}(\tau)] u^{*}(\tau) e^{-i u_{0} \tau} d\tau,$$

$$(10.7.2)$$

where $y_i(t)$ is also given by formula (10.3.3), in which $U_{ai}(t, \alpha)$ are expressed by formulas (10.2.5).

Then substituting the obtained expressions in (10.5.5) and (10.5.6) and acting by the rules already described, it is easy to find that in both cases $\Delta = 0$, and K_R will be expressed by the same formula

$$K_{\rm R} = \frac{P_{\rm cr}}{2\pi\Delta f_{\rm s}} \int_{-\infty}^{\infty} \operatorname{Re} H_{\rm s}(im) H^{\rm s}_{\rm s}(im) S_{\rm s}(m) dm. \qquad (40, 1.5)$$

Using formula (10.5.11), we obtain in both cases the following expressions:

$$S_{5\pi 3} = \frac{\int_{-\infty}^{\infty} |I_{1}(l\omega)|^{2} |H_{1}(l\omega)|^{2} [1] + h_{1} S_{0}(\omega)] d\omega}{\frac{\mu_{0}^{2} h_{1}^{2}}{2\pi} \int_{-\infty}^{\infty} \operatorname{Re} H_{1}(l\omega) H^{0}_{1}(l\omega) S_{0}(\omega) d\omega}$$
(10.7.4)

From (10.7.4) it follows that the minimum of S_{SKB} (which it is easy to prove by simple variation) takes place when $H_1(i\omega) = H_2(i\omega)$ and $|H_1(i\omega)|^2$ are expressed by formula (10.3.25). This minimum is equal simply to S_{CHT} . Furthermore, from (14.43) and (15.7.4) it is easy to perceive the influence of nonidentity of frequency is a sponses of filters. Obviously, nonidentity of the gain-frequency responses alterts C_{WTB} little. Nonidentity of phase-frequency responses lead, to essential increase of S_{GRT} (as a result of decrease of K_{II}). Actually, with difference of the phase-frequency responses $\Delta \Phi(\omega)$

$$K_{\mathbf{A}} = \frac{P_{cs}F_{\mathbf{a}}}{2\pi\Delta f_{\mathbf{a}}} \int_{-\infty}^{\infty} |H_{\mathbf{a}}(i\mathbf{a})|^{2} |H_{\mathbf{a}}(i\mathbf{a})|^{2} S_{\mathbf{a}}(\mathbf{a}) \cos \Delta \varphi(\mathbf{a}) d\mathbf{a}, \qquad (1 \le i \le 1)$$

from which we clearly see the cause of decrease of $K_{\rm H}$. In particular, if $L_{\rm H}(\omega)$ is sufficiently great (reaches limits $2r_{\rm HC}$ still in the vicinity of the maximum of

-545-



Fig. 10.19. The dependence $S_{OHT} = S_{OHT} = n h_{\lambda}$ for circuits with multiplication of signals in gonloweters using scanning with compensation or IAC.

 $|H_{i}(i\omega)|$), K_{μ} may be zero or even change sign. The circuit will not work here. Thus, of basic importance for the considered circuits is making the phase-frequency responses of filters identical.

Let us calculate S_{346} by formula (10.7.4), considering filters identical and approximating their gain-frequency response by formula (10.5.16). Here we obtain

$$S_{\text{9KB}} = \frac{1}{2\mu_{4}^{2} \delta f_{c}} \left(\frac{2+1/x}{h_{z}} + \frac{x+2+1/x}{h_{z}^{2}} \right),$$
(10.7.1)

where as before $x \approx \Delta f_{\phi} / \Delta f_{c}$.

This formula is very simple, and from it, it is easy to perceive all the laws governing variation of $S_{\rm DKB}$. Curves of the dependence of $S_{\rm DKB}/S_{\rm OHT}$. In h_2 are shown in Fig. 10.19. From this figure it is clear short change of the filter in very wide limits

(x = 0.5-10) for $h_Z > 1$ leads to increase of spectral density as compared to the optimum by not more than 2 times.

The spectral density of parametric fluctuations for the considered direction with identical filters is easily calculated by formula (10.5.22) and turns out to rectarately coincide with expression (10.5.23). Analysis of this expression was made in 240.5.

10.7.7. Investigation of the Circuit with Subtraction of Signals for the Method of Scanning with Compensation

Let us turn to consideration of the circuit for the method of scanning with compensation (Fig. 10.16). As we established, this circuit is close to the opticum only with low frequencies of scanning. However, a circuit of this type is used in practice, and it is of interest to analyze its accuracy for face values of the frequency of scanning which are not too low.

The output signel of the circuit of Fig. 17.10 has the form

-546-

$$z(t) = \cos \theta \left\{ (t - \tau_{a}) \left\{ \left| \int_{-\infty}^{t} \dot{h}_{1}(t - \tau) y_{1}(\tau) u(\tau) e^{i \omega_{a} \tau} d\tau \right|^{a} - \left| \int_{-\infty}^{t} \dot{h}_{a}(t - \tau) y_{a}(\tau) u(\tau) e^{i \omega_{a} \tau} d\tau \right|^{a} \right\}, \qquad (10.7.6)$$

where $y_1(\tau)$ and $y_2(\tau)$ are the same as in (10.7.1), and τ_3 — delay of the reference signal, selected from the condition of maximum K_{π} (see § 10.5).

This condition exceptionally hampers calculation in general, and we subsequently shall limit our consideration to the case of conical scalining of directional patterns. In this case all calculations are considerably simplified, and after simple calculations we obtain

$$\max_{\mathbf{x}_{g}} K_{g} = \frac{P_{cz}\mu_{a}}{2^{r}J_{e}} \left| \int_{-\infty}^{\infty} [H_{1}(i\omega) H^{*}_{1}(i\omega + i\Omega) + H_{g}(i\omega) H^{*}_{g}(i\omega + i\Omega)] S_{\phi}(\omega) d\omega \right|, \qquad (10.7.7)$$

$$S_{\Theta \mathbf{x}_{B}} = \frac{4\pi}{\mu_{a}^{2}h_{z}^{2}} \int_{-\infty}^{\infty} \left\{ [|H_{1}(i\omega)|^{2} |H_{1}(i\omega + i\Omega)|^{2} + |H_{a}(i\omega)|^{2} |H_{a}(i + i\Omega)|^{2} \right] \left[1 + \frac{h_{z}}{2} S_{\phi}(\omega + \Omega) \right] \times \left[1 + \frac{h_{z}}{2} S_{\phi}(\omega) \right] \cdot -\frac{h_{z}}{2} |H_{1}(i\omega)|^{2} |H_{g}(i\omega + i\Omega)|^{2} \times \left[1 + \frac{h_{z}}{2} S_{\phi}(\omega) \right] \cdot -\frac{h_{z}}{2} |H_{1}(i\omega)|^{2} |H_{a}(i\omega + i\Omega)|^{2} \times \left[S_{\phi}(\omega) S_{\phi}(\omega + \Omega) \right] d\omega \times$$

 $\times \left| \int_{-\infty}^{\infty} \left[H_{1}(i\omega) \, i I^{*}_{1}(i\omega + l\Omega) + H_{2}(i\omega) \, H^{*}_{3}(i\omega + i\Omega) \right] S_{0}(\omega) d\omega \right|^{-2}, \qquad (1 \dots 1)^{-1}$

where h_{Σ} - ratio of the total signal power to the power of noise of one channel in the band of signal fluctuations. Systematic error Δ in the considered case is also equal to 0.

Let us consider formula (10.7.8) in greater detail. From it we can easily perceive that in general for different gain-frequency responses of filters there exists a component of equivalent spectral density caused by nonlinear transformation of the useful signal. This component in the given circuit causes error of measurement even with complete elimination of noises. With identical gain-frequency enaracteristics of filters such a component of equivalent spectral density disappears. The influence of nonidentity of filters on accuracy of the considered circuit leads basically to appearance of this component of error.

We shall conduct further investigation of formula (10.7.8), considering finiteral identical and using usual approximations (10.3.26) and (10.5.10). Here we obtain

$$S_{\text{DNB}} = \frac{1+x}{\mu_{\text{B}}^{2}\Delta f_{\text{D}}x} \times \left\{ \frac{1}{h_{\text{L}}} \frac{\xi^{4} + \xi^{2} (5x^{2} + 4x + 3) + 2(x - 1)^{3} (x + 1) (2x + 1)}{(\xi^{4} + (x - 1)^{2}] [\xi^{2} + 4(x + 1)^{2}]} + \frac{2}{h_{\text{L}}^{2}} \frac{(1+x) [\xi^{2} + (1+x)^{2}]}{\xi^{2} + 4(x + 1)^{2}} \right\}, \quad (10.7.9)$$

where as before $\zeta = \Omega/2\Delta f_c$; $x = \Delta f_{tb}/\Delta f_c$.

For high frequencies of scanning, i.e., for large ζ from (10.7.9) we obtain

$$S_{\text{PHB}} = \frac{1}{\mu_a^2 M_a} \left(\frac{1+1/x}{h_x} + \frac{x+2+1/x}{h_x^2} \right). \tag{10.7.10}$$

Comparison of this formula with $(1^{\circ},7,5)$ shows that for high frequencies of scanning the considered circuit gives accuracy identical to the accuracy of the circuit of Fig. 10.15 only when x >> 1 and $h_{\Sigma}^2 >> 1$. In the remaining cases accuracy of the considered circuit is approximately half the accuracy of the circuit of Fig. 10.15 (in equivalent spectral density).

For low frequencies of scanning from (10.7.9) we obtain for S₃₆₀ an expression exactly coinciding with (10.7.5), ...e., the accuracy of the considered circuit in this case is the same as accuracy of the circuit of Fig. 10.15.

-548-



Curves of the dependence of S_{3KB}/S_{OPT} on h_{Σ} , calculated with approximations (10.3.26) and (10.5.16), for different ζ are shown in Fig. 10.20. From this figure it is clear that in the range of changes $\zeta = 1-3$ the considered circuit sufficiently well realizes the potential of the method of scanning with compensation, giving for x = 1-5 equivalent spectral density exceeding S_{opt} by not more than in 2.5 times

Fig. 10.20. The dependence of S_{ACE}/S_{OHT} on h_{Σ} for a circuit with subtraction of signals of a goniometer using scanning with compensation: $-\zeta = 3; --\zeta = 1;$ $-\cdot -\cdot \zeta = 0.01.$ (and for $h_{\Sigma} > 1$, S_{RKB} exceeds S_{OHT} by not more than 1.8 times). For higher frequencies of scanning properties of this circuit worsen.

10.7.3. Investigation of a Circuit with Subtraction of Signals for IAC

Let us analyze, finally, the last circuit of Fig. 10.18 for IAC. The output signal of this circuit is equal to

$$z(t) = \left| \int_{-\infty}^{t} \dot{h}_{1}(t-\tau) y_{1}(\tau) u(\tau) e^{i u_{0} \tau} d\tau \right|^{2} - \left| \int_{-\infty}^{t} \dot{h}_{2}(t-\tau) y_{1}(\tau) u(\tau) e^{i u_{0} \tau} d\tau \right|^{2}, \quad (10.7.11)$$

where $y_i(t)$ are the same as in (10.7.2). Substituting this expression in (10.5.4) and (10.5.11), we obtain, respectively,

$$K_{\mathrm{R}} = \frac{P_{\mathrm{cE}}\mu_{\mathrm{a}}}{2\pi\Delta f_{\mathrm{c}}} \int_{-\infty}^{\infty} \left[\left| H_{\mathrm{a}}(i\omega) \right|^{2} + \left| H_{\mathrm{a}}(i\omega) \right|^{2} \right] S_{\mathrm{o}}(\omega) \, d\omega, \qquad (10.7.12)$$

$$S_{2R_{B}} = -\frac{4\pi}{\mu_{a}^{2}h_{x}^{2}} \int_{-\infty}^{\infty} \left\{ [|H_{1}(i\omega)|^{4} + |H_{a}(i\omega)|^{4}] \left[1 + \frac{\hbar}{2} S_{0}(\omega) \right]^{4} - \frac{h_{x}^{2}}{2} |H_{1}(i\omega)|^{4} |H_{3}(i\omega)|^{2} S_{0}^{2}(\omega) \right\} d\omega \times \\ \times \left\{ \int_{-\infty}^{\infty} [|H_{1}(i\omega)|^{2} + |H_{3}(i\omega)|^{2}] S_{0}(\omega) d\omega \right\}^{-4}.$$
(10.7.13)

It is easy to see that (10.7.7) and (10.7.8) for small Ω are transformed into (10.7.12) and (10.7.13), respectively, i.e., accuracy of the circuit of Fig. 10.14 for IAC and accuracy of the circuit of Fig. 10.16 for the method of scanning with compensation for low frequencies of scanning are identical. From (10.7.13) it is clear that nonidentity of the gain-frequency responses of filters leads in the given circuit to the same effect as in the circuit of Fig. 10.16 of the method or comming with compensation: the appearance of a very undesirable component of error, caused by nonlinear transformation of the signal. Furthermore, in the considered circuit for different gain-frequency responses there will exist certain systematic error differing from zero, calculable by formula (10.5.6):

-549-

$$\Delta = \frac{1!}{2\mu_{a}h_{2}} - \frac{\int_{-\infty}^{\infty} [|H_{1}(i\omega)|^{2} - |H_{1}(i\omega)|^{2}] \left[1 + \frac{h_{2}}{2} S_{0}(\omega)\right] d\omega}{\int_{-\infty}^{\infty} [|H_{1}(i\omega)|^{2} + |H_{1}(i\omega)|^{2}] S_{0}(\omega) d\omega} - (10.7.14)$$

In particular, when $|H_1(1\omega)|^2 = K|H_2(1\omega)|^2$ (i.e., if only gains of the channels differ) and with approximation (10.5.12) we will have a quite simple expression

$$\Delta = \frac{1}{2\mu_{a}} \frac{1-K}{1+K} \left(\frac{1}{2} + \frac{x+1}{h_{g}} \right).$$
(10.7.12)

Caculation of S_{BKB} for approximations (10.3.26) and (10.5.16) will lead, etviously, to formula (10.7.5), since S_{BKB} as we already noted, coincides with (10.7.7) when $\zeta \ll 1$, which in turn reduces to (10.7.5).

The spectral density of parametric fluctuations we shall consider again for identical filters (when systematic error is equal to 0, equivalent spectral density of error is also small, and parametric fluctuations give an appreciable contribution to total error of measurement). In this case it completely coincides with (10.5.23).

We sum up our investigation. First we see that with identical channels the circuit of Fig. 10.15 for the method of scanning with compensation and the circuit for the method of IAC (Figs 10.17 and 10.18) have absolutely identical accuracy, rather close to the potential accuracy of these methods. The circuit of Fig. 10.16 for the method of scanning with compensation is close in accuracy to these circuits only at low frequencies of scanning. With increase of the frequency of scanning accuracy of this circuit monotonically worsens, attaining a limit of approximately half its magnitude at low frequencies of scanning.

Nonidentity of channels influences, in general, different circuits differently. In the circuits of Figs. 10.15 and 10.17 with nonidentical channels accuracy remainabsolutely identical. In these circuits the basic influence on accuracy is rendered only by nonidentity of the phase-frequency responses of the filters, leading to decrease of the transmission factor of the radio channel and through it to increase of equivalent spectral density.

In the circuits of Figs. 10.16 and 10.16 the basic influence on accuracy comes from nonidentity of the gain-frequency responses of the filters. In the circuit of Fig. 10.16 for the method of scanning with compensation this nonidentity loads to the appearance of a component of error caused by nonlinear transformation of the signal, and in the circuit for the IAC method (Fig. 10.1%), furthermore, it leads

-550-

to the appearance of systematic error.

Now we introduce considerations about the technical complexity of the considered circuits.

As we already said in § 10.5, the most complicated components of goniometer circuits are the UPCh's, mixers with inversion of phase modulation, and amplitude modulators. It is easy to see that circuits with compensation contain the least number of such components: each of these circuits has 2 channels with such components and provides measurement of angular target data in two planes. It is easy to see that the circuit of the IAC method (Fig. 10.17) for measurement of angular coordinates in two planes will already require 3 such channels (since here there should be 1 channel with the sum signal and 2 channels with the difference signal), and the circcuit of Fig. 10.18 even requires 4 channels (2 channel for measurement of the angle in one plane and 2 in the other). Thus, the circuits of the method of scanning with compensation are the simplest, the circuit of the IAC method of Fig. 10.17 is more complicated, and the circuit of Fig. 10.18 is the most complicated.

If we assume that making both the phase-frequency responses of filters and their gain-frequency responses identical is a difficult task, then among the considered set of circuits the circuit of Fig. 10.15 of the method of scanning with compensation will have obvious advantages over the other circuits: naving the greatest accuracy and simplicity, it is least critical to nonidentity of channels. However, in many cases it is easier to ensure identity of gain-frequency response of filters. In this case the advantage, for the same reasons, will lie on the side of the circuit (Fig. 10.16) of the method of scanning with compensation with use in it of low frequencies of scanning. However, in view of the small difference of accuracies of the different circuits these recommendations one should consider very relative, and the best solution can be realized only by taking into account the concrete situation.

§ 10.8. Method of Phase Center Scanning

Let us turn to investigation of phase methods of direction finding, using for direction finding of a target effects connected with the dependence (under contain conditions) of phases of arriving radar signals on angular coordinates of the target. The simplest of the phase methods of direction finding is the method of phase center scanning. The given method, if it is possible to say so, is the "phase snate;" of the method of pattern scanning.

The deficiency of this method is lowering of the acturate of direction (inclusion) with low frequencies of phase denter scamping, since us oil is retting by a line is the second second

--551-

included in the law of measurement of the phase of the arriving signal, and at low frequencies of scanning this law will be distorted by phase fluctuations. Here intuition does not suggest any measures of compensating phase fluctuations. Study of the potential of this method, therefore, is of special interest.

10,8,1. Optimum Circuit for the Method of Phase Center Scanning

The optimum circuit for the method of phase center scanning is easily obtained from the general circuit of Fig. 10.7 if we there set n = 1 (one receiving antenna), $U_{a}(1, \alpha) = 1$ (i.e., the signal during reception does not obtain amplitude received. and $\Phi(t, \alpha)$ is expressed by formula (10.2.7). Substituting these date in (10.3.72), we obtain the optimum circuit of the method of phase center scanning in the form depicted in Fig. 10.21. The circuit for measurement of the angle in the other plane



Fig. 10.21. Optimum circuit of the radio channel of a gonicmeter using phase center scanning. 1) optimum filters with frequency response $\Pi_{1 \to \Pi_{1}}(i\omega)$.

will be obviously, the same, only multiplication by $\delta(t)$ must be replaced by multiplication by a function varying according to the law of motion of the projection of the phase center in the other plane. The output signal of the antenna system here immediately enters two channels, in which there is realized heterodyning. A peculiarity of heterosyning in the considered case is the

circumstance that the signal of the heterodyne proceeds to one of the channels through a power shifter shifting the phase of this signal $\pi/2$. Then the signal in this multiplied by a function varying according to the law of scanning of the phase center (more exactly, by the projection of this function in the phase of the measured angle). After such transformations signals of both channels are filtered by an optimum filter with response (10.3.25) and are multiplied.

The physical meaning of these transformations is approximately the same as in the optimum circuit of the method of pattern scanning. Multiplying the received signal by a function which varies by the law of phase center scanning, we transferred the side spectral components of the signal, containing useful information, to the center frequency, after which it is possible to fifter the signal by a sufficiently narrow-band fifter, passing a minimum of noises without loss of useful information of the signal. It is curious that there is not any compensation of phase fluctuations

--552-

of the signal. Inasmuch as this circuit is optimum, then from this there follows the conclusion about the impossiblility of compensation of phase fluctuations of the signal. In order to grasp the peculiarities distinguishing phase from amplitude fluctuations, partial compensation of which turns out to be possible, we note the following: addition of a constant to amplitude will change its law of distribution of probabilities, but addition of a constant to phase will not change its distribution of probabilities. This graphically shows that fluctuations of phase lead to more significant distortion of information included in the phase than occurs for amplitudes.

10.8.2. Equivalent Spectral Density of the Optimum Circuit for the Method of Fhase Center Scanning

Let us consider potential accuracy of measurement of angles by the method of phase center scanning. Considering in formula (10.3.31) n = 1, substituting there expression (10.2.7) and producing the necessary calculations, we obtain

$$S_{\text{OHT}} = \frac{\pi}{\left(\frac{\pi d}{\lambda}\right)^2 h^2} \left\{ c_{\phi}^2 \int_{-\infty}^{\infty} \frac{S_{\phi}(\omega)^2}{1 + hS_{\phi}(\omega)} d\omega - \frac{S_{\phi}(\omega)}{1 + hS_{\phi}(\omega)} d\omega \right\}^{-1},$$

$$- \sum_{h=-\infty}^{\infty} |c_{h\phi}|^2 \int_{-\infty}^{\infty} \frac{S_{\phi}(\omega) S_{\phi}(\omega + h\Omega)}{1 + hS_{\phi}(\omega)} d\omega \Big\}^{-1},$$
(10.8.1)

where

$$\boldsymbol{c}_{\phi}^{2} = \frac{1}{T_{ond}^{2}} \int_{0}^{T_{ond}} \delta(t)^{2} dt; \quad \boldsymbol{c}_{k\phi} = \frac{1}{T_{ond}} \int_{0}^{T_{ond}} \delta(t) e^{ik\theta t} dt; \quad (10.3.2)$$

 T_{CK} - period of phase center scanning; $\Omega = 2\pi/T_{CK}$ - angular frequency of scanning;

 λ - wavelength;

d - maximum deviation of the phase center from its center positon.

In the other plane S_{0DT} will have the same form, only c_{ϕ}^2 and $|c_{k\phi}|^2$ muct be replaced by s_{ϕ}^2 and $|s_{k\phi}|^2$, expressed analogously to (10.8.2) through the law or motion of the projection of the phase center in this plane.

Let us consider certain limiting cases. With high frequencies of scanning we obtain

$$S_{\text{OHT}} = \frac{\pi}{\left(\frac{\pi d}{\lambda}\right)^3 h^2 \left(c_{\phi}^2 - |c_{\phi\phi}|^2\right)} \left[\int_{-\infty}^{\infty} \frac{S_{\phi}(\phi)^2}{1 + hS_{\phi}(\phi)} d\phi\right]^{-1}.$$
 (10.26)

From this it is clear that the best scanning is symmetric scanning of the phase center, where $c_{\phi 0} = 0$. Then (10.8.3) takes form

$$S_{\text{OHT}} = \frac{\pi}{\left(\frac{\pi d}{\lambda}\right)^2 h^2 c_{\phi}^2} \left[\int_{-\infty}^{\infty} \frac{S_{\phi}(\omega)^2}{1 + h S_{\phi}(\omega)} d\omega \right]^{-1}, \qquad (10.8.4)$$

coinciding with (10.4.8) with replacement of μ_{a} by $\pi d/\lambda$, and of c² by c_{ϕ}.

Regarding magnitudes of coefficients c_{ϕ} and s_{ϕ}^2 it is possible to conduct reasoning analogous to the reasoning regarding coefficients c^2 and s^2 for the method of pattern scanning (§ 10.4). Here it will become clear that if it is required to measure only one angle, the best scanning will be jumps of the phase center to two extreme positions in the plane of this angle, where $c_{\phi}^2 = 1$ and $s_{\phi}^2 = 0$. If, however, it is required to measure angles in two planes with identical accuracy, the law of scanning of the phase center one should select in such a manner that $c_{\phi}^2 = s_{\phi}^2 = 1/2$. Subsequently, we assume namely the latter.

At low frequencies of scanning, obviously, $S_{CNT} \rightarrow \infty$, i.e., measurement of angular coordinates is impossible.

As $h \rightarrow \infty$ (decrease of noises) we obtain

$$S_{\text{OHT}} = \frac{\pi}{\left(\frac{\pi d}{\lambda}\right)^{2}} \left\{ \sum_{k=-\infty}^{\infty} |c_{k\phi}|^{2} \int_{-\infty}^{\infty} \frac{S_{\bullet}(\omega + k\Omega) - S_{\bullet}(\omega)}{S_{\bullet}(\omega)} d\omega \right\}^{-1}.$$
 (10.8.5)

Thus, for the method of phase center scanning due to fluctuations of the phase of the signal for finite Ω there exists error of measurement differing from zero even with complete elimination of hoises.

For investigation of the general case we produce by formula (10.8.1) a calculation using approximation (10.3.26) for $S_{\rm C}(\infty)$. Let us consider the case of uniform circular scanning of the phase center, i.e., with $c_{\rm g\pm1} = 1/2$, and with $i \neq \pm 1$, $c_{\rm gi} = 0$. Here we obtain

$$S_{\text{ext}} = \frac{1}{\left(\frac{\pi d}{\Lambda}\right)^2 \Delta f_{\text{e}}} \frac{\sqrt{1+\lambda}(1+\sqrt{1+\lambda})}{\lambda^2} \left[1 + \frac{(1+\sqrt{1+\lambda})^2}{\zeta^4}\right], \quad (10.8.6)$$

where $\zeta = \Omega/2\Delta f_{0}$.

Curves of the dependence of S_{OFF} on h, calculated by this formula (for different values of ζ), are shown in Fig. 10.22. From the curves it is clear that with growth of h spectral density S_{OFF} monotonically drops to a certain finite quantity, seeking 0 as $\zeta \rightarrow \infty$. With decrease of ζ it monotonically increases, increasing to infinity.



Fig. 10.22. The dependence of S_{OHT} on h for the method of phase center scanning.

10.8.3 Investigation of the Synthesized Circuit for the Method of Phase Center Scanning

Let us consider now the question of the real accuracy of the circuit of Fig. 10.21. We assume that filters in this circuit have pulse response $h(t) \cos [\omega_{np} t + b(t)]$. We introduce also designations analogous to (10.5.2). The output signal of the circuit of Fig. 10.21 here will have the form

$$z(t) = \operatorname{Re} \int_{-\infty}^{\infty} \dot{h}^{*} (t-\tau) e^{i\omega_{0}\tau} y(\tau) u^{*}(\tau) d\tau \times \int_{-\infty}^{\infty} \dot{h} (t-\tau) e^{i\omega_{0}\tau} y(\tau) iu(\tau) \delta(\tau) d\tau, \qquad (10.37)$$

where $y(\tau)$ - received signal.

Subsequently we shall limit ourselves to the case of uniform circular scanning of the phase center in the aperture plane, where $\sigma(t) = d \cos \Omega t$.

Calculation by the usual rules using formula (10.5.5) and (10.5.11) gives

$$K_{\rm R} = \frac{P_{\rm e}\left(\frac{\pi d}{\lambda}\right)}{2\pi\Delta/\epsilon} \int_{-\infty}^{\infty} \operatorname{Re} H_{\rm 1}(i\omega) H^{*}_{\rm a}(i\omega) [S_{\rm e}(\omega) - S_{\rm e}(\omega + \Omega)] d\omega, \qquad (10.17)$$

-555-

$$S_{\theta R \theta} = \frac{2\pi}{\left(\frac{\pi d}{\lambda}\right)^{2} h^{2}} \int_{-\infty}^{\infty} [|H_{1}(i\omega)|^{2} |H_{2}(i\omega)|^{2} + |H_{1}(i\omega)|^{2} \times |H_{1}(i\omega + i\Omega)|^{2}] [1 + hS_{0}(\omega)] [1 + hS_{0}(\omega + \Omega)] d\omega \times [\int_{-\infty}^{\infty} \operatorname{Re} H_{1}(\omega) H^{*}_{3}(i\omega) [S_{0}(\omega) - S_{0}(\omega + \Omega)] d\omega]^{-2}.$$

$$(10.8.9)$$

al a histocratabilitata ana ana a 12

・1911年、1911

It is also easy to calculate that systematic error $\Delta = 0$. From (10.8.9) it is clear that nonidentity of gain-frequency responses little affects accuracy of the circuit; nonidentity of phase-frequency responses leads to decrease of $K_{\rm R}$ and, consequently, to increase of $S_{\rm SKB}$. Here considerable divergence of phase responses can even lead to change of the sign of $K_{\rm R}$.

Calculation by formula (10.8.9) we perform on the assumption of identity of the filters, approximating their frequency responses by formula (10.5.12). Here we dotain for S_{BHB} an expression coinciding with (10.5.17), only factor

 $\zeta^{2}(1 + x)^{2}/\mu_{a}\Delta f_{c} [\zeta^{2} + 2(1 + x)^{2}]^{2}$ before the braces must be replaced by the simpler $\zeta^{2} + (1 + x)^{2}/(\pi d/\lambda)^{2}\Delta f_{c} \zeta^{4}$. This will introduce definite changes in the law of the dependence of S_{BAB} on h, x and ζ . In particular, for small x we will have the following expression for S_{BAB} :

$$S_{\rm DKB} = \frac{1}{2\left(\frac{\pi d}{\lambda}\right)^2} \Delta f_{\rm o} x \frac{\zeta^2 + 1}{\zeta^4} \left(1 + \frac{\zeta^2 + 2}{\hbar} + \frac{\zeta^2 + 1}{\hbar^2}\right), \qquad (10.5.1c)$$

from which we see infinite growth of $S_{g,\eta}$ as $\zeta \to 0$. For large x we obtain

$$S_{0KB} = \frac{2\kappa^4}{\left(\frac{\pi d}{\lambda}\right)^5 \Delta f_0 h^2 \zeta^4}.$$
 (10.8.11)

Here we see the exceptional criticality of the circuit to expansion of the passband of the filters. Equivalent spectral density increases proportionally (x'). However, the dependence on the frequency of scanning is preserved. Increase of the frequency of scanning leads to considerable decrease of the equivalent spectral density.

With respect to limiting cases for ζ we note that for large ζ the expression for S_{DKB} will coincide with (10.5.20) (with replacement of μ_{a} by $\pi d/\lambda$), i.e., in this case accuracy of the considered circuit coincides with the accuracy of the circuit of Fig. 10.8 for the method of pattern scanning. For small ζ we have

$$S_{0x0} = \frac{(1+x)^4}{\left(\frac{\pi d}{\lambda}\right)^6 \Delta f_0 \zeta^4} \left(2 + \frac{1}{x} + 2\frac{2+\frac{1}{x}}{h} + \frac{x+2+\frac{1}{x}}{h^5}\right), \quad (12.0, 12)$$

-550-



Fig. 10.23. The dependence of S_{HB}/S_{GRT} on h for a quasi-optimum circuit of the radio channel of a goniometer with phase center scanning: $\zeta = 100$; $-\zeta = 10$.

i.e., the drop of accuracy is proportional to $\zeta^{4'}$.

Curves of the dependence of $S_{\rm HE}/S_{\rm ORT}$ on h, calculated with these same approximations, are shown in Fig. 10.23.

As can be seen from these curves, with high frequencies of scanning ($\zeta \approx 100$) the considered circuit sufficiently well realizes potential accuracy of the method in a wide range of variation of x(x = 2-10). For lower frequencies of scanning ($\zeta = 10$) accuracy of the circuit already will sharply depend on x (for h = 5 change of x from 2 to 10 leads to increase of S_{BIB} by approximately 20 times).

Thus, results of investigation of the method of phase center scanning can be summarized in the following way: this method is applicable only

with sufficiently high frequencies of scanning ($\xi \approx 100$), where it ensures accuracy equal to the accuracy of high-frequency scanning of the directional pattern (or the method of scanning with compensation or IAC), only $\mu_a = \pi d/\lambda$ and the received powers are identical; the synthesized circuit of the method of phase scanning is not very critical to the degree of matching of filters with the spectrum of fluctuations of signal only at very high frequencies of scanning; already at $\zeta \approx 10$ it becomes are critical that it is possible to place in doubt the expediency of its practical realization.

From the obtained results it is clear that in many cases the method of place center scanning is less desirable in comparison with the above-considered amplitude methods of direction finding.
§ 10.9. <u>Method of Instantaneous Phase Comparison of Signals</u> 10.9.1. Optimum Circuits and Potential Accuracy for IPC

The method of instantaneous phase comparison of signals (IPC) is the phase analog of the IAC method. In IPC circuits the presence of several outputs of the antenna array with separated phase centers means that phase fluctuations of the received signals differ only by a constant component, and, consequently, there exist the prerequisites for compensation of phase fluctuations. This indicates that from the IPC method we are justified in expecting higher accuracy of measurements during operation on a fluctuating signal, than from the method of phase center scanning.

The circuit of the optimum radio channel for IPC is easy to obtain from the general circuit of Fig. 10.7 if we there set n = 2 (two receiving antennas), $U_{ai}(t, \alpha) = 1$ (amplitudes of the received signals are identical), and we introduce $\Phi_i(t, \alpha)$ according to expressions (10.2.8). The circuit of the radio channel of IPC is



Fig. 10.24. Variant of the optimum circuit with formation of the sum and difference signals of the radio channel of a goniometer for IPC. 1) optimum filters with frequency response $H_{1 \text{ OUT}}(i\omega)$. obtained here in the form depicted in Fig. 10.24, which can easily be absolutely identically transformed to the form depicted in Fig. 10.25.

In both variants of optimum IPC circuits signals from the output of the antenna system after transformations of the type of addition or subtraction of these signals enter two channels. There the signals are heterodyned, where heterodyne signals proceeding into these channels are shifted in phase $\pi/2$.

Then the signals are filtered by an optimum filter of type (10.3.25) and are multiplied. Both circuits are very simple and are similar to the circuit of the radio channel with phase center scanning (if in the latter we exclude multiplication by the function varying according to the law of scanning).

Potential accuracy of IPC can be calculated by formula (10.3.31) if there we set n = 2, $U_{pi}(t, \alpha) = 1$, and introduce $\Phi_i(t, \alpha)$ by formula (10.2.8). Then

$$S_{0HT} = \frac{2\pi}{\left(\frac{\pi d}{\lambda}\right)^2 h_T^2} \left[\int_{-\infty}^{\infty} \frac{S_0(\omega)^2}{1 + h_T S_0(\omega)} d\omega \right]^{-1}.$$
(10.9.1)

-558-



e_(t)cor[in_t+\$/t]

Fig. 10.25. Variant of the optimum circuit (without formation of the sum and difference signals) of the radio channel of a goniometer for IPC. 1) optimum filters with frequency response $H_{1 \text{ OIIT}}(i\omega)$. This formula will completely coincide with (10.6.5), characterizing the potential accuracy of the method of scanning with compensation and IAC, if there we replace μ_a on by $\pi d/\lambda$ (as we agreed, $c^2 = 1/2$). It follows from this that potential accuracy of IPC is identical with potential accuracy of IAC (or the method of scanning with compensation), if the total received powers in both cases are identical and if $\mu_a = \pi d/\lambda$. Accuracy of the method of phase center scanning nears the accuracy of IPC only at high frequencies of

scanning (if the received total powers in both cases are identical).

The dependence of $S_{O\Pi T}$ on different factors was studied in sufficient detail in § 10.6.

10.9.2. Investigation of Synthesized Circuits for IPC

Let us analyze real accuracy of synethesized circuits of IPC. All assumptions about circuits and designations we keep the same as in the preceding paragraph. We consider first the circuit of Fig. 10.24. The output signal of this circuit has the form

$$z(t) = \operatorname{Re} \int_{-\infty}^{t} \dot{h}_{1}^{*}(t-\tau) [y_{1}(\tau) + y_{2}(\tau)] u^{*}(\tau) e^{-i\omega_{0}\tau} d\tau \times \\ \times \int_{-\infty}^{t} \dot{h}_{2}(t-\tau) [y_{1}(\tau) - y_{2}(\tau)] iu(\tau) e^{i\omega_{0}\tau} d\tau, \qquad (10.9.2)$$

where $y_1(t)$ - are the received signals.

Substituting this expression in (10.5.5), (10.5.6), (10.5.11) and (10.5.22), we can calculate $K_{\rm H}$, Δ , $S_{\rm HGB}$ and $S_{\rm HAP}$. It is interesting to note that the results here are absolutely the same as for the circuit of Fig. 10.15 or for the circuit of Fig. 10.17, only $\mu_{\rm g}$ is replaced by $\pi d/\lambda$, i.e., $K_{\rm H}$ is expressed by formula (10.7.2), $S_{\rm HFB}$ by formula (10.7.3), $\Delta = 0$, and $S_{\rm HAP}$ (with identical filters) is expressed by formula (10. 7. 6). Investigation of accuracy of these circuits was conducted in detail in § 10.7. We recall the basic results of this investigation: with identical channels the circuit well realizes potential accuracy of measurement; the degree of matching of filters with the spectrum of signal fluctuations weakly influences accuracy of the circuit; the greatest danger to identity of channels is nonidentity of phase-frequency responses of filters, leading to decrease of K_{II} and increase of $\frac{c_{\rm BKB}}{c_{\rm BKB}}$

The circuit of Fig. 10.25 in its properties, as it turns out, considerably differs from the circuit of Fig. 10.24. The output signal of the circuit of Fig. 10.25 has, obviously, the form (10.9.2), only under the sign of the first integral instead of the combination of received signals there should be $y_1(\tau)$, and under the second integral - $y_2(\tau)$. Substitution of this expression in (10.5.5) after simple calculations gives for $K_{\rm I}$ again the expression (10.7.2) (with replacement of $\mu_{\rm a}$ by $\pi d/\lambda$). However, the other characteristics of the circuit turn cut to be different. In particular, $\xi_{\rm BKB}$ for this circuit equals

$$S_{\Theta RB} = \frac{\pi}{\left(\frac{\pi d}{\lambda}\right)^2} h_2^2 \left\{ \int_{-\infty}^{\infty} [|H_1(i\omega)|^2 |H_1(i\omega)|^2 (1+h_2 S_0(\omega))^2 - -\frac{1}{4} h_2^* S_0(\omega)^2 \operatorname{Re} H_1(i\omega)^2 H_1^*(i\omega)^2 d\omega + +\frac{1}{2} h_2^* \int_{-\infty}^{\infty} \operatorname{Im} H_1(i\omega) H_2^*(i\omega) S_0(\omega) d\omega \right\} \times \left[\int_{-\infty}^{\infty} \operatorname{Re} H_1(i\omega) H_2^*(i\omega) S_0(\omega) d\omega \right]^{-2}.$$

$$(10.9.5)$$

Furthermore, systematic error Δ for this circuit is different from to ro and is equal to ∞

$$\Delta = -\frac{1}{2 \frac{\pi d}{\lambda}} \int_{-\infty}^{\infty} \operatorname{Re} H_{1}(i\omega) H^{\bullet}_{1}(i\omega) S_{\bullet}(\omega) d\omega \qquad (10.9.4)$$

With identical filters, obviously, accuracy of both circuits of Fig. 10.24 and Fig. 19.25 are identical. However, nonidentity of filters in the circuit of Fig. 19.25 are identical. However, nonidentity of filters in the circuit of Fig. 19.24. Analysis of expression (10.9.3) shows that with nonidentical filters in the circuit of Fig. 10.25 there occurs decrease of $K_{\rm H}$ and, consequently, increase of S_{2KB} to the same measure as for the circuit of Fig. 10.24. However, nonidentity of filters here leads to the appearance of a component of equivalent spectral density causes in particular, error of measurement in the considered circuit even with total ubsence of noises. Finally, a very negative property of the considered circuit is the presence of systematic error with nonidentical filters. Let us note that the basic influence on accuracy of the circuit is consequent only by nonidentity.

-560-

of phase-frequency responses of the filters. To the above we must add that the circuit of Fig. 10.25 is more complicated technically than the circuit of Fig. 10.24, since it, like the circuit of Fig. 10.18, requires for measurement of the angles in two planes four UPCh's, four mixers with inversion of phase modulation, etc., while the circuit of Fig. 10.23 in this respect is similar to the circuit of Fig. 10.17 and requires only three UPCh's, etc. On the basis of this we can draw conclusions concerning the obvious advantages of the circuit of the IPC method of Fig. 10.24 over the circuit of Fig. 10.25. Let us note that investigation of real properties of circuits leads to this conclusion. In ideal conditions both circuits are identical.

Thus, the IPC method has the same potential accuracy as IAC or the method of scanning with compensation, if μ_a and $\pi d/\lambda$ are identical. Accuracy close to the potential for the IPC method can be realized with success in practice by the circuit of Fig. 10.24.

§ 10.10. The Method of Flat Scanning (Optimum Circuits)

Let us study the method of flat scanning or the method of tracking by pulse packs. With this method, as was shown in § 10.2, there is realized tracking of pulse packs of the received signal by an electronic circuit. Pulse packs are received at the moments of passage of the directional pattern through the target during its scanning in a certain angular sector, in which this target is located.

Thus, the method of tracking by pulse packs no longer belongs to methods with a tracking antenna, and the preceding results, in particular, general results of § 10.3, here are not directly applicable. Synthesis of the optimum circuit here should be done anew.

10.10.1. Synthesis of an Optimum Circuit for the Method of Flat Scanning

Let us assume that the directional pattern (for power) of the antenna array of a goniometer using the method of tracking by pulse packs has in plane of the measured angle α symmetric form Gg(φ). For g(φ) we select the following normalization:

$$\frac{1}{2T_{s}} \int_{-\infty}^{\infty} g(\varphi)^{s} d\varphi = 1, \qquad (10.10.1)$$

where Ω - angular velocity of motion of the pattern over the sector; T_{rr} - period of repetition of pulse packs.

-561.-

Antenna gain is accounted for by coefficient G. Here the received signal can be recorded in the form (10.3.3), where $\Phi(t, \alpha) = 0$ (the antenna array during reception does not introduce in the received signal any phase modulation), and $U_a(t, \alpha)$ is determined by formula (10.2.6). The amplitude and phase modulations of the sounding signal are arbitrary.

We introduce one assumption with respect to the character of fluctuations of the received signal. The assumption we make here reduces to the fact that reflecting properties of the target do not change during the time during which the directional pattern passes over the target. This will lead the signal inside the pack to be rigidly correlated or, so to speak, the pack will be "harmoniously fluctuating." The degree of correlatedness of different packs may, in general, be arbitrary, since the period of repetition of the packs almost always is comparible with the time of correlation of fluctuations of the signal. The assumption of "harmoniously fluctuating" packs is very significant later.

Furthermore, we shall subsequently consider that the period of repetition T_r is considerably less than the duration of a pulse pack. This assumption, in general, is more limiting than the preceding one; however, in many cases it to greater or lesser extent is realized. Furthermore, it also will considerably facilitate our finding of optimum operations for the method of tracking by pulse packs.

To find the operation of the optimum radio channel for the method of tracking by packs we need, as usual, to first construct the likelihood functional $L(\alpha)$ of parameter α . For this we must solve integral equation (10.5.9) cr, ultimately, equation (10.3.14). Solution of the last equation for the method of tracking by pulse packs involves considerable difficulties, since here that method which we used in examining goniometers with a tracking antenna is unapplicable. However here very useful is our assumption of "harmoniously fluctuating packs." Really, function $U_a(t_2, \alpha)^2$, under the sign of the integral in (10.3.14) differs from zero only in the environments of moments iT $_{\Pi}$ + α/Ω , determined by the width of the directional pattern and the rate of scenning (i - integers). We assume for definitiveness that

$$t - \Delta t \leq (n - m) T_{u} + a/\Omega < (n - m + 1) T_{u} + a/\Omega < \dots < n T_{u} + a/\Omega < t.$$
(10.10.2)

The assumption of harmonicus fluctuations of the signal inside the packs means that during a time of the order of duration of one pack the correlation function of fluctuations of signal $\rho(t)$ practically does not vary. We also consider the assumption of smallness of the period of repetition of the signal T_r as compared to the

-562-

duration of the pack, thanks to which it is possible under the sign of the integral in (10.3.15) to replace $|u(t_2)|^2$ by its time-averaged value, which is equal to 1. Here from equation (10.3.15) we can obtain the following expression (see also Chapter VI):

$$\Delta f_{oh} \sum_{k=n-m}^{n} \rho_{ik} \sigma_{ki} + \rho_{ii} + \sigma_{ii} = 0, \qquad (10.10.3)$$

where h, as before, is the ratio of mean signal power to the power of noise in the band of fluctuations of the signal,

$$P_{ik} = P[(l-k)T_{II}], \ \sigma_{ik} = \sigma(iT_{II} + a/Q, \ kT_{II} + a/\Omega). \tag{10.10.1}$$

This equation is solved using discrete Fourier transforms. Considering

$$S_{0T_{\rm H}}(\mathbf{x}) = T_{\rm H} \Delta f_0 \sum_{k=-\infty}^{\infty} P(kT_{\rm H}) e^{-i\mathbf{x}kT_{\rm H}}$$
(10.10.5)

it is simple to obtain

$$\mathcal{D}_{jl} = -\frac{1}{2\pi\Delta f_o} \int_{-\pi/T_m}^{\sigma/T_m} \frac{S_{0T_m}(\omega)e^{i\omega(l-l)T_m}}{1+\hbar S_{0T_m}(\omega)} d\omega. \qquad (10.10.6)$$

We note that, as can be seen from (10,10.6), v_{jl} does not depend on α . Thus, we could find, not the whole function $v(t_1, t_2)$, but only a set of its values in discrete moments of time (10.10.2). However, this very set of values is necessary to us for construction of the operation of the optimum radio channel.

The operation of the optimum radio channel z(t) for the considered case should be found from relationship

$$\int_{\mathbf{r}-\mathbf{A}} \mathbf{z}(\mathbf{r}) d\mathbf{r} = \frac{\partial \ln L(\mathbf{e})}{\partial \mathbf{a}} \Big|_{\mathbf{e}=\hat{\mathbf{e}}}, \qquad (10.10.7)$$

where \hat{a} - estimated value of angle a, introduced in the radio channel from the output of the tracking circuit [this is the distinction between (10.10.7) and (10.3.12)]. Hence

$$\int_{t-M}^{t} z(\tau) d\tau = -\frac{\partial}{\partial u} \frac{\sqrt{P_{\sigma}}}{N_{\sigma}} \operatorname{Re} \int_{t-M}^{t} \int_{t-M}^{t} U_{\mathbf{x}}(t_{1}, \mathbf{a}) U_{\mathbf{x}}(t_{3}, \mathbf{a}) \times (t_{1}) u^{*}(t_{3}) v(t_{1}, t_{3}) y(t_{1}) y(t_{3}) e^{i u_{0}(t_{1}-t_{3})} dt_{1} dt_{3}|_{u=\tilde{u}}$$
(10.8)

e to the presence of function $U_{a}(t_{1}, \alpha)U_{a}(t_{2}, \alpha)$ in (10.10.8) under the sign of the integral the integrand will differ from zero only in the environment of moments of time (10.10.2) where function $v(t_{1}, t_{2})$ takes value v_{1} . In connection with this (10.10.8) can be rewritten in the form

$$\int_{t-At}^{t} z(t) dt = -\frac{\partial}{\partial u} \frac{\sqrt{P_u}}{N_u} \sum_{i,k=n-m}^{n} v_{ik} \operatorname{Re} z_i z^*_{ki}, \qquad (10, 10, 9)$$

where

$$Z_{i} = \int_{U_{a}(t, a)} U_{a}(t, a) u(t) y(t) e^{i a y} dt; \qquad (10.10.10)$$

 δt - intervals of time of the order of the duration of a pulse pack. Consider now quantity h_{Ok} , related to v_{ik} by relationship

$$T_{\mathbf{R}} \sum_{k=-\infty}^{\infty} h_{\mathbf{0}(i-k)} h_{\mathbf{0}(k-1)} = -v_{il} h \Delta f_{\mathbf{0}}. \qquad (10.1.1)$$

If we introduce the discrete Fourier transform of sequence h_{Ok}

$$H_{0T_{\pi}}(i\omega) = T_{\pi} \sum_{k=-\infty}^{\infty} h_{kk} e^{i\omega kT_{\pi}}, \qquad (10.10.12)$$

from (16.10.11) it is easy to find that

$$|H_{0T_{a}}(i\omega)|^{2} = \frac{hS_{0T_{a}}(i\omega)}{1 + hS_{0T_{a}}(\omega)},$$
 (10.10.13)

where S_{OT} (a) is expressed by formula (10.10.15). With the help of h_{Ok} expression (10.10.9) can be rewritten (with accuracy to a proportionality factor) in the form

$$\int_{t-At}^{t} z(\mathbf{r}) d\mathbf{r} \sim \frac{\partial}{\partial \mathbf{r}} \sum_{k=n-m}^{n} T_{\pi} \left[\sum_{l=n-m}^{n} h_{h-l} z_{l} \right]^{l}, \qquad (10.10.14)$$

Then, considering (10.10.10), we can write

$$\sum_{l=n-m}^{n} h_{k-i} z_{l} = \sum_{l=n-m}^{n} h_{k-i} \int_{iT_{r}+a/R-M}^{iT_{r}+a/R+M} U_{a}(t,a) u(t) y(t) e^{ia_{0}t} dt =$$

$$= \int_{t-M}^{t} h_{0}(kT_{r}+a/\Omega-t) U_{a}(t,a) u(t) y(t) e^{ia_{0}t} dt, \qquad (1.0, 10.1)$$

where function $h_0(t)$ is introduced in such a way that $h_0(kT_r) = h_{0k}$; values of this function in the remaining moments of time are not essential and can be anything, inasmuch as due to filtering properties of function $U_a(t, \alpha)$ they simply disappear. Substituting (10.10.15) in (10.10.14), we obtain

$$\int_{t-\Delta t}^{t} z(\tau) d\tau \sim \frac{\partial}{\partial a} \sum_{k=n-m}^{n} T_{\pi} \left| \int_{t-\Delta t}^{t} h_{0} (kT_{\pi} + a/2 - \tau) U_{a}(\tau, a) \times u(\tau) y(\tau) e^{i\omega_{0}\tau} d\tau \right|^{2} \approx \frac{\partial}{\partial a} \int_{t-\Delta t}^{t} ds \left| \int_{t-\Delta t}^{t} h_{v}(s-\tau) U_{a}(\tau, a) u(\tau) y(\tau) e^{i\omega_{0}\tau} d\tau \right|^{2}$$

-564-

$$\boldsymbol{z}_{\boldsymbol{o}}(\boldsymbol{s}) \approx \frac{\partial}{\partial \boldsymbol{a}} \left| \int_{\boldsymbol{t}-\boldsymbol{A}\boldsymbol{f}}^{\boldsymbol{s}} h_{\boldsymbol{o}}(\boldsymbol{s}-\boldsymbol{\tau}) \boldsymbol{U}_{\boldsymbol{a}}(\boldsymbol{\tau},\boldsymbol{a}) \boldsymbol{u}(\boldsymbol{\tau}) \boldsymbol{y}(\boldsymbol{\tau}) \mathrm{e}^{\boldsymbol{i}\boldsymbol{\omega}\cdot\boldsymbol{\tau}} \, d\boldsymbol{\tau} \right|^{2}. \tag{10.10.16}$$

Thus, in the considered case the optimum operation already has a familiar form, derived earlier (see Chapter VI). However, the pulse response of the optimum filter $h_0(t)$ is assigned by the set of its values h_{OK} in moments of time kT_{II} by relationships (10.10.12) and (10.10.13). In the remaining moments of time values of pulse response are arbitrary.

Let us consider this optimum filter in greater detail. We introduce for the correlation function of fluctuations of the signal the approximation $\rho(t) = e^{-2\Delta f} c |t|$, i.e., we approximate the spectrum of fluctuations by expression (10.3.26). Here, from (10.10.5) we obtain

$$S_{0T_{\rm R}}(\omega) = \frac{\Delta f_{\rm c} T_{\rm R}(1-\rho_{\rm c}^2)}{\rho_{\rm c}^2 - 2\rho_{\rm c} \cos \omega T_{\rm R} + 1}, \qquad (10.10.17)$$

where $\rho_{c} = e^{-2\Delta f c} T_{\Pi}$.

By virtue of (10.10.13) we have

$$[H_{0T_{n}}(i\omega)]^{2} = \frac{hT_{n}\Delta f_{o}(1-\rho_{c}^{2})}{hT_{n}\Delta f_{o}(1-\rho_{c}^{2})+\rho_{c}^{2}-2\rho_{o}\cos\omega T_{n}+1}.$$
 (10.10)

It is of interest to find the pulse response $h_{0}(t)$ of the optimum filter in the form

$$h_{\bullet}(l) = c e^{-24/_{000} l}, \qquad (10.10.19)$$

i.e., the gain-frequency response of this filter has the form

$$|H_0(i\omega)|^4 = \frac{c}{1 + \left(\frac{\omega}{2\Delta f_{out}}\right)^2}.$$

Here $|H_{OT}|_{\Pi}$ (iw) $|^2$, defined by relationship (10.10.18), should be the Fourier -2\Delta f_{OTT}^{KT} we bring (10.10.18) to form

$$\begin{aligned} H_{0T_{g}}(i\omega)|^{2} &= \frac{hT_{g}\Delta f_{0}(1-\rho_{c}^{2})}{A^{6}} / \left| 1 - \frac{B}{A} e^{-h\sigma T_{g}} \right|^{2}, \qquad (1 \le 1 \le i \le 1) \\ A &= \frac{1}{2} \sqrt{hT_{n}\Delta f_{0}(1-\rho_{c}^{2}) + (1+\rho_{c})^{2} + \cdots + \sqrt{hT_{n}\Delta f_{0}(1-\rho_{c}^{2}) + (1-\rho_{c})^{2}}; \\ B &= \frac{1}{2} \sqrt{hT_{n}\Delta f_{0}(1-\rho_{c}^{2}) + (1+\rho_{0})^{2} - \cdots + \sqrt{hT_{n}\Delta f_{0}(1-\rho_{c}^{2}) + (1-\rho_{0})^{2}}. \end{aligned}$$

7

where

or

-565-

From this it is easy to find that the pulse response can be taken namely in the form (10.10.19),

where

$$\Delta f_{\text{ORY}} = \frac{\ln A/B}{2T_{\pi}}, \qquad (10.10.22)$$

A and B are determined by formulas (10.10.21), and

$$c = \frac{\sqrt{hT_{\pi\Delta f_c}(1-p_c)}}{A}$$

Thus, the optimum filter, as also in the preceding cases, has a frequency response similar to the spectrum of fluctuations of the signal; however the width of the filter passband has a more complicated dependence on h. The passband also essentially depends on the degree of correlatedness of neighboring pulse packs, i.e., on quantity $T_{\Pi} \Delta f_{C}$. In particular, when $T_{\Pi} \Delta f_{C} \gg 1$ (neighboring pulse packs fluctuate independently) and $\tau_{\Pi} \ll T_{\Pi}$ the passband of the optimum filter should satisfy condition

$$\frac{1}{T_{u}} < \Delta f_{\text{OBT}} < \frac{1}{\tau_{u}}, \qquad (10.10.23)$$

where τ_{π} - duration of the pack.

Consequently, the optimum filter in this case should integrate every pack, where response of the filter to a given pack should attenuate before the arrival of the following pack. In Fig. 10.26 there is depicted the dependence of Δf_{OIIT} on h for different values of T_{Π} Δf_{C} . As can be seen from this figure, with growth of h, Δf_{OIIT} increases; however the rate of this growth drops with increase of T_{Π} Δf_{C} (i.e., with constant h with growth of the number of correlated packs the passband of the filter narrows). Physically these laws are understandable.

Operation (10.10.6) can easily be reduced to a real form analogously to how we obtained expression (10.3.29). Here there become clear the block diagrams of the optimum radio channel realizing operation (10.10.16). We can offer two block diagrams of the optimum radio channel, depicted in Figs. 10.27 and 10.28. The circuit of Fig. 10.27 is obtained when in expression (10.10.6) we carry out exact differentiation with respect to parameter α . In this circuit the output signal from the antenna is first heterodyned, as in other optimum circuits, and then enters two channels. In these channels the signal is gated. Gate pulses in one channel in form should coincide with the form of the pack, and in the other channel - with the derivative of the pack. The time position of the gate pulses is established in

-566-



Fig. 10.20. Dependence of the bandwidth of the optimum filter Δf_{OHI} on h for goniometers using the method of flat semning.

accordance with the measured delay of the packs (i.e., in accordance with the measured angular position of the target). After gating, the signals are filtered by optimum filters with response (10.10.18) and are multiplied.

The block diagram of Fig. 10.28 is obtained if we replace calculation of the derivative in (10.10.6) approximately by calculation of the difference. The bignal in this circuit after networdyning as in the preceding case, enters two channels where it is gated. Gate pulses in both channels coincide in form with the form of the pulse pack; however, they are detuned relative to each other in time by a certain time interval. The time position of these gate pulses is obtatlished in accordance with the measured delay of the packs; however their relative.

detuning is preserved. Then the signals are filtered, detected by a square-law detector, and one signal is subtracted from the other.



Σ g[R(L+nT_n)-k]

Fig. 10.27. Optimum circuit with differentiation of gates of the radio channel of a goniometer with flat scanning of the directional pattern: 1) optimum filters with discrete frequency response H_{OT} (1 ω).



Fig. 10.28. Optimum circuit with detuned gates of the radio channel of a goniometer with flat scanning of the directional pattern: 1) optimum filters with discrete frequency response $H_{OT_{eff}}$ (10); 2) square-

law detectors.

The obtained circuits are very similar to the circuits of the optimum radio channel of a range finder, if we ignore the form of the optimum filters and of the gate pulses. This is understandable, since in both cases the matter ultimately leads to measurement of delay of the signal relative to the beginning of the period. Filtration was different due to the different fluctuating properties of the received signal: if in range finders pulses of the signal whose delay was measured were rigidly correlated, the correlation coefficient of the equivalent packs of pulses in the method of angular tracking by packs no longer is equal to unity. The form of gate pulses is determined in the given case by the form of the pulse packs.

10.10.2. Equivalent Spectral Density of Optimum Circuits for the Method of Flat Scanning

Let us consider now the potential accuracy of measurements using the method of flat scanning. Equivalent spectral density in this case can be presented in the form

$$S_{OIIT} = \left\{ \frac{2\Delta f_{C}}{\Delta t} \int_{t-\Delta t}^{t} [1 + \sigma(\tau, \tau)] |u(\tau)|^{2} \times \left[\sum_{t} u'_{a} \left(\tau - iT_{u} - \frac{c}{u}\right) \right]^{2} d\tau \right\}^{-1} = \left| \frac{2\pi}{e^{2\Delta t}} \left\{ \int_{-\pi/T_{u}}^{\pi/T_{u}} \frac{{}^{1}S_{0T_{u}}^{2}(\omega)}{1 + \hbar S_{0T_{u}}(\omega)} d\omega \right\}^{-1}, \qquad (10.10.24)$$
$$c^{2} = \frac{1}{\Omega T_{u}} \int_{0}^{\infty} g'(\varphi)^{2} d\varphi. \qquad (10.10.25)$$

where

The form of the directional pattern affects accuracy through quantity c^2 . Let us calculate this magnitude for some approximation of the directional pattern. If w take a Gaussian directional pattern

$$g(\mathbf{r}) = 1,15 \sqrt{\frac{97\pi}{4\gamma}} e^{-2.4 \left(\frac{\gamma}{4\gamma}\right)^4}$$
(10.10.20)

[taking into account normalization (10.10.1)], where $\Delta \phi$ — width of the directional pattern of a level of half power, then it is easy to find that

$$C^{a} = \frac{2.8}{\Delta \gamma^{a}}.$$
 (10.10.27)

From (10.10.25) it is clear that c^2 increases infinitely with increase of the steepness of the leading edge of the pack.

Let us consider formula (10.10.24) in greater detail. Obviously, as $h \to \infty$ quantity $S_{OHT} \to 0$, i.e., in the absence of noises there does not exist fluctuating error of target tracking, in spite of fluctuations of the received signal. As $h \to 0$

$$S_{out} = \frac{1}{c^{a}h^{3}T_{u}\Delta f_{c}^{2}} \left[\sum_{k=-\infty}^{\infty} p(kT_{u})^{a} \right]^{-1}, \qquad (10.10.28)$$

i.e., accuracy decreases inversely proportionally to the square of the signal-tonoise ratio. We also give the formula occurring for T $_{\Pi}$ $\Delta f_{C} \gg 1$, i.e., with independently fluctuating pulse packs:

$$S_{ont} = \frac{1}{c^2 h \Delta f_o} \left(1 + \frac{1}{h T_n \Delta f_o} \right).$$
 (10.10.28')

To more fully judge the laws governing variation of S_{OHT} , we calculate it by formula (10.10.24) with approximation of the spectrum of signal fluctuations by expression (10.3.26). Here, as was shown, S_{OT} (ω) is expressed by formula (10.10.5). Substituting (10.10.5) in (10.10.24), we obtain

$$S_{ont} = \frac{1}{\Delta f_{o}c^{2}h^{2}} \sqrt{(hT_{n}\Delta f_{c})^{2} + 2hT_{n}\Delta f_{c}\frac{1+\rho_{c}^{2}}{1-\rho_{c}^{2}} + 1} \times \frac{1+\sqrt{(hT_{n}f_{c})^{2} + 2hT_{n}\Delta f_{c}\frac{1+\rho_{c}^{2}}{1-\rho_{c}^{2}} + 1}}{h(T_{n}\Delta f_{c})^{2} + 2T_{n}\Delta f_{c}\frac{1+\rho_{c}^{2}}{1-\rho_{c}^{2}}}.$$

Curves of the dependence of S_{OHT} on h for different values of T_{II} Δf_{C} are shown in Fig. 10.29. As can be seen from this figure, S_{OHT} drops rather rapidly with increase of h. Very curious is the dependence of S_{OHT} on T_{II} Δf_{C} . With growth of T_{II} Δf_{C} for an assigned magnitude of h quantity S_{OHT} drops, i.e., accuracy of angle measurement increases. The physical explanation of this phenomenon is the following. Increase of T_{II} Δf_{C} for a fixed h means that there occurs decrease of the frequency of repetition of packs with increase of the energy in each pack. As a result there is increased, if it is possible to express it thus, the energy of signal sections which retain their coherence during reflection from a fluctuating target. Increase of accuracy due to this effect depends on h. For large h it is insignificant; for h = 1, S_{OHT} will decrease approximately by a factor of 4 (with change of T_{II} Δf_{C} from 0.3 to 100).

In conclusion we note that for the method of tracking by packs it is sometimes profitable to take as the power characteristic the ratio of maximum signal power (which would exist if the directional pattern of the antenna always "looked" at the target) to the power of noise in the band of signal fluctuations. Designating this characteristic by h_m , it is easy to find its relationship to h [with approximation

-569-



(10.10.26) of the directional pattern]

$$h_{\rm m} = 1,32 \frac{27_{\rm m}}{\Delta_{\rm P}} h.$$
 (10.10.40)

If we designate by Φ the magnitude of the examined angular sector and consider that $\Omega T_{\Pi} = \Phi$, from (10.10.30) we obtain

$$h_{\rm m} = 1.32 \frac{\Phi}{\Delta y} h.$$
 (10.17.21)

We note that the order of magnitude of h_m for the method of tracking by pull packs is comparable with the order of magnitude h for methods with a tracking antenna (since with these methods the directional pattern of the antenna system always "looks" at the target). Inasmuch as usually $\Delta \phi \ll \phi$, for the method of tracking by pulse packs h $\ll h_m$, i.e., the method of tracking by pulse packs in terms

Fig. 10.29. Dependence of $S_{0\Pi^{\rm H}}$ on h for the method of flat scanning of the directional pattern.

of power is considerably inferior to methods with a tracking antenna. However, here it is necessary to consider other tactical functions of the method of tracking by pulse packs (tracking of many targets), compelling us to put up with this loss in terms of power.

§ 10.11. Investigation of Synthesized Circuits for the Method of Flat Scanning

Let us pursue the study of real accuracy of the synthesized circuits of Figs. 10.37 and 10.28 for the method of flat scanning. We consider here certain basic factors of the type of various nonidentities and imperfectnesses affecting accuracy of measurement of angles. First of all, the gate pulses fed to different channel: of the synthesized circuits are practically formed each separately, and therefore will not be matched in form both with each other, and also with the form of the pulse pack. Thus, the gates must be considered different and differing in form from the pulse pack.

Filters in the circuits should be matched with the spectrum of signal fluctuations. However, as already repeatedly noted, this matching cannot be realized

-570-

sufficiently accurately practically, and during the analysis of the frequency response of the filters it should be considered, in general, arbitrary. Regarding identity of filters in each of the circuits, it is possible to consider it sufficiently good.

10.11.1. Investigation of a Circuit with Differentiation of Gate Pulses

Let us consider first the circuit of Fig. 10.27. We introduce for characteristics of the filters designation (10.5.2). Urthermore, trains of gate pulses entering different channels of this circuit we designate by $u_1(t)$ and $u_2(t)$. Then the output signal can easily be reduced to the form:

$$z(l) = \operatorname{Re} \int_{-\infty}^{t} h(l-\tau) u_{1}(\tau-\tau) u(\tau) y(\tau) e^{i\omega_{1}\tau} d\tau \times \int_{-\infty}^{t} h^{*}(l-\tau) u_{1}(l-\tau) u^{*}(\tau) y(\tau) e^{-i\omega_{1}\tau} d\tau, \quad (10,11,1)$$

where τ - delay of gate pulses issued by the tracking system.

By formula (10.5.5) it is easy to find that the slope of the discrimination characteristic of the considered circuit is determined by

$$K_{\rm R} = -\frac{P_{\bullet}}{4\pi\Delta f_{\rm c}} \int_{-\pi/T_{\rm R}}^{\pi+1/T_{\rm R}} |H_{T_{\rm R}}(i\omega)|^{\bullet} S_{0T_{\rm R}}(\omega) d\omega \times [C'_{10}(0) C_{10}(0) + C_{10}(0) C'_{10}(0)], \quad (10.11.2)$$

where $H_{T_{a}}(i\omega) = T_{a} \sum_{k=-\infty}^{\infty} \dot{h}(kT_{a}) e^{ik\omega T_{a}}$ discrete Fourier transform of the pulse re-

 $C_{ij}(\tau) = \frac{1}{T_a} \int_{0}^{T_a} u_i(t) u_j(t-\tau) dt; \qquad (10.11.3)$ $u_0(t) = u(t);$

strokes designate derivatives (note that results obtained here are very similar to results of Chapter VII, dealing with range meters).

Furthermore, in this case there will exist systematic error equal to

$$\Delta = \frac{C_{10}(0) C_{30}(0) + \frac{C_{13}(0)}{h} \int_{-\pi/T_{g}}^{\pi/T_{g}} |H_{T_{g}}(i\omega)|^{2} d\omega}{C'_{10}(0) C_{30}(0) + \frac{C_{10}(0) |^{2} C_{30}(0) |^{2} C_{30}(0)}{+ C_{10}(0) C'_{30}(0)}$$

$$(100, 100)$$

As can be seen from (10.11.4), systematic error is absent if $C_{12}(0) = C_{12}(0)C_{\infty} < \infty$ (0) = 0. This condition, obviously, means that one of the gate pulses should be symmetric and the other, asymmetric (with a symmetric pulse pack).

Equivalent spectral density in this case turns out to be equal to

 $\mathbf{S}_{\text{ORB}} = \frac{\mathbf{u}}{h^{2}} \int_{-\pi/2}^{\pi/2} |H_{T_{\text{B}}}(i\mathbf{e})|^{4} \{C_{11}(0) C_{12}(0) + C_{12}^{2}(0) + C_{12}^{2}(0) \}$ $+ AS_{er_{a}}(0) [C_{11}(0) C_{20}^{2}(0) + C_{aa}(0) C_{10}^{2}(0) + C_{ab}(0) + C_{ab}(0) C_{10}^{2}(0) + C_{ab}(0) +$ $+ 2C_{10}(0) C_{0}(0) C_{10}(0) + 2h^{2}S_{0T_{a}}(0) C_{01}^{2}(0) C_{20}^{2}(0) d \omega \times$

$$\times \left\{ [C_{30}(0) C_{00}(0) + C_{10}(0) C_{10}(0)] \times \int_{-\pi/T_{0}}^{-\pi/T_{0}} |H_{T_{0}}(i_{0})|^{2} S_{0T_{0}}(0) d_{0} \right\}^{-2}.$$
(10.11.5)

From this expression it is clear that when $C_{01}(0) C_{02}(0) \neq 0$ there exists a component of error caused by nonlinear transformation of the signal; it does not depend on h and will give error of measurement with complete elimination of noises. This component of error disappears together with systematic error.

It is difficult to study in more detail the dependence of S_{3KB} on different parameters in view of the complexity of expression (10.11.5). We will conduct this study in two stages. First, we assume gating to be ideal; equivalent spectral density during ideal gating we shall designate by S_{WKH} . It is easy to find that

$$S_{MBB} = \frac{\int_{-\infty/T_{B}}^{\infty/T_{B}} |H_{T_{B}}(i\omega)|^{4} [1 + hS_{0T_{B}}(\omega)] d\omega}{\int_{-\infty/T_{B}}^{\infty} |H_{T_{B}}(i\omega)|^{4} S_{0T_{B}}(\omega) d\omega}$$
(10.11.6)

We produce by formula (10.11.6) calculation for the most typical approximations of characteristics of the filter and of the spectrum of fluctuations of the signal. We shall approximate the response envelope of the filters by expression $h(t) = e^{-2\Delta f} \tilde{\phi}^t$. Then by the formula (10.11.2) we have

$$H_{T_{u}}(im) = \frac{T_{u}(1-t_{\phi}^{2})}{t_{\phi}^{2}-2t_{\phi}\cos mT_{u}+1},$$
 (10.11.7)
$$P_{\phi} = e^{-3Mt_{\phi}T_{u}} = e^{-2Mt_{a}T_{u}T_{u}};$$

where

 $x = \frac{\delta f_{\phi}}{\delta f_{\phi}}$ ratio of the effective passband of the filter to the width of the band of fluctuations of the signal.

For $S_{OT_{\Pi}}(\omega)$ we introduce approximation (10.10.17). Substituting (10.10.17) in (10.11.6), we obtain

 $S_{max} = \frac{1}{3e^{4}A^{0}T_{m}\Delta f_{m}^{2}} \left\{ \frac{1+s_{\phi}^{2}}{1-s_{\phi}^{2}} + hT_{m}\Delta f_{\phi} \frac{1-s_{\phi}^{2}}{(1-s_{\phi})(s_{\phi}-s_{\phi})} \right\}$ $\times \left[\mathbf{P}_{0} \frac{1 + \mathbf{P}_{0}\mathbf{P}_{0}}{1 - \mathbf{P}_{0}\mathbf{P}_{0}} - \mathbf{P}_{0} \frac{(1 + \mathbf{P}_{0}^{2})(1 - \mathbf{P}_{0}^{2})}{(1 - \mathbf{P}_{0}^{2})^{2}} \right] \right] \left(\frac{1 + \mathbf{P}_{0}\mathbf{P}_{0}}{1 - \mathbf{P}_{0}\mathbf{P}_{0}} \right)^{-2}.$ (10.11.8)

The curve of the dependence of S_{MNH}/S_{ORT} on h, where S_{ORT} is calculated with these same approximations and is given by formula (10.10.29), is shown in Fig. 10.30.



Fig. 10.30. The dependence of S_{MMH}/S_{ORT} on h for circuits of the method of flat scanning:

----- $T_{g} A f_{g} = 10; ---- T_{g} A f_{g} = 3; ---- T_{g} A f_{g} = 1.$

As can be seen from the figure, nonmatching of filters with the spectrum of fluctuations has greater influence, the larger quantity $T_{\Pi}\Delta f_{C}$. The difference between S_{MKH} and $S_{O\Pi T}$ can attain a very considerable magnitude (a factor of 10 or more).

Knowing the dependence of S_{MMH}/S_{CHT} on basic parameters of the circuit, we consider now ratio S_{3KB}/S_{MMH} . This ratio essentially depends on the form of the gate pulses. For simplicity we shall consider that gate pulses possess the necessary symmetry. The influence of asymmetry of gate pulses was already estimated qualitatively. If we consider, for instance, gate pulses $u_1(t)$ symmetric, and gate pulses $u_2(t)$ asymmetric, in (10.11.5) we have the following simplification: $C_{12}(0) = 0$, $C_{20}(0) = 0$, $C_{10}'(0) = 0$. Taking into account this simplification, it is easy to see that for sufficiently large h (h \ge 10)

ratio S /S will not depend on characteristics of the filters, but will depend on characteristics of the gate pulses:

$$\frac{S_{010}}{S_{010}} = C^3 \frac{C_{11}(0) + C_{10}(0) C_{10}^2(0)}{C_{10}^2(0) C_{10}(0)^3}$$
(10.11.9)

We produce calculation by formula (10.11.9). For this we approximate the directional mattern of the antenna system by formula (10.10.26). Pulse packs here have form

$$u_{0}(t) = \sqrt{\frac{T_{0}}{\tau_{m}}} 1,15 e^{-2.3 \left(\frac{t}{\tau_{m}}\right)^{2}},$$

where $\tau_{\Pi} = \Omega \Delta \phi - duration$ of the pack at level 0.5.

Gate pulses $u_1(t)$ we approximate also by a Gaussian function.

$$u_{1}(l) = e^{-2.8 \left(\frac{l}{\tau_{1e}}\right)^{2}}, \qquad (10.11.11)$$

where $\tau_{i,a}$ - duration of the gate pulse at level 0.5.

Gate pulses $u_2(t)$ we consider derivatives of Gaussian pulses, i.e.,

$$u_{s}(t) = te^{-2.8 \left(\frac{t}{y_{s}}\right)^{2}}.$$
 (10.11.1)

With such approximations all coefficients in (10.11.9) are easily calculated, as a result of which we obtain



Fig. 10.31. The dependence of S_{avg}/S_{MNH} on the relationship of durations of the pack and gate pulses in a circuit with differentiation of gate pulses.

$$\frac{S_{010}}{S_{M10}} = \left(\frac{1+t^2\eta^2}{2\xi\eta}\right)^5,$$
 (10.11.13)

where $\xi = \frac{\varepsilon_{1}}{\varepsilon_{1e}}; \eta = \frac{\varepsilon_{1o}}{\varepsilon_{1e}}$.

Thus, always $S_{AFG}/S_{MNH} \ge 1$. The sign of equality is attained when $\xi \eta = 1$. The graph of the dependence of S_{AFG}/S_{MUH} on ξ for different values of η is shown in Fig. 10.31. As can be seen from this figure, imperfectness of gating can lead to very great worsening of accuracy of measurement.

Having dependence S_{MNH}/S_{OHT} (Fig. 10.30) and S_{2KE}/S_{MNH} (Fig. 10.31), we can, if we so desire, find the ratio for S_{2KE}/S_{OHT} . We recall once again that Fig. 10.31 is constructed for large h.

The spectral density of parametric fluctuations will be given for the case of ideal gating. It turns out to be equal to

(10.11.10)

$$S_{\text{map}} = 2\pi \frac{\int_{-\pi/T_{\text{m}}}^{\pi/T_{\text{m}}} |H_{T_{\text{m}}}(i\omega)|^{4} S_{0T_{\text{m}}}^{2}(\omega) d\omega}{\left(\int_{-\pi/T_{\text{m}}}^{\pi/T_{\text{m}}} |H_{T_{\text{m}}}(i\omega)|^{4} S_{0T_{\text{m}}}(\omega) d\omega\right)^{4}}, \qquad (10.11.14)$$

As can be seen from this, it does not depend on h. If the filter has a sufficiently wide band, $S_{\pi a p}$ turns out to be equal to $T_{\pi} \operatorname{coth} T_{\pi} \Delta f_{c}$. With decrease of Δf_{c} quantity $S_{\pi a p}$ grows. When $T_{\pi} \Delta f_{c} \ge 1$ we have $S_{\pi a p} \approx T_{\pi}$. If the filter has a sufficiently narrow band, then $S_{\pi a p} \approx T_{\pi}$ coth $\Delta f_{c} T_{\pi} x$, where $x = \Delta f_{\phi} / \Delta f_{c}$. Lawe governing the change of $S_{\pi a p}$ here are as before, only the approximation of $B_{\pi a p}$ to T_{π} with increase of Δf_{c} is slower.

10.11.2. Investigation of a Circuit with Defuned Gate Fulses

Let us turn to consideration of the circuit of Fig. 10.28. All designations we leave the same. The output signal of this circuit, obviously, has form

$$z(t) = \left| \int_{-\infty}^{t} h(t-s) u_{s}(s-\hat{s}-\delta) y(s) u(s) e^{i\omega_{0}s} ds \right|^{2} - \left| \int_{-\infty}^{t} h^{4}(t-s) u_{s}(s-\hat{s}+\delta) y(s) u^{*}(s) e^{-i\omega_{0}s} ds \right|^{2}.$$
(10,11,10)

From formula (10.5.5) it is easy to find that the slope of the discrimination characteristic of the given radio channel is equal to

$$K_{\rm R} = \frac{P_{\rm e} \left[C_{1e}^{\prime}(b) C_{1e}(b) - C_{1e}^{\prime}(-b) C_{1e}(-b) \right]}{2\pi \Delta f_{\rm e}} \times \int_{-e/T_{\rm g}}^{e/T_{\rm g}} |H_{T_{\rm g}}(ie)|^{e} S_{0T_{\rm g}}(e) de. \qquad (10, 11, 10)$$

Systematic error with the general assumptions with respect to the form of gate pulses differs from zero and is equal to

$$\Delta = \frac{C_{10}(\hat{s})^2 - C_{20}(-\hat{s})^2 + C_{11}(0) - C_{21}(0)}{C_{20}(\hat{s}) - C_{20}(\hat{s}) - C_{20}'(-\hat{s})} \times \frac{A}{2mA_{f_0,K}} \int_{-\pi/T_{\pi}}^{\pi/T_{\pi}} |H_{T_{\pi}}(i\omega)|^2 S_{0T_{\pi}}(\omega) d\omega \,. \quad (10, 11, 17)$$

As in the preceding case, systematic error is caused by nonidentity of gating. Systematic error will be absent if $C_{11}(0) = C_{22}(0)$ and $C_{10}(b)^2 = C_{20}(-b)^2$. These conditions in general are rather rigid and practically signify identity of gate pulses in both channels. Obviously, to ensure identity of gate pulses in two channels is practically more difficult than simply to ensure symmetry of gate pulses in each channel, which was required for elimination of systematic error in the preceding case. Thus, from this point of view the circuit of Fig. 10.28 one should consider somewhat more critical to imperfectnesses of gating than the circuit of Fig. 10.27.

~575~

Equivalent spectral density for the considered circuit is equal to

$$S_{0HB} = \frac{\pi}{2\hbar} \int_{-\pi/r_{H}}^{\pi/r_{H}} |H_{r_{H}}(i\omega)|^{4} \{C_{11}(0)^{6} + C_{12}(0)^{6} - C_{12}(2\delta)^{5} - C_{12}(2\delta)^{5} + 2\hbar S_{0r_{H}}(\omega) [C_{12}(0) C_{10}(\delta)^{5} + C_{23}(0) C_{20}(-\delta)^{5} - C_{12}(2\delta)^{5} + 2\hbar S_{0r_{H}}(\omega) [C_{10}(\delta)^{5} - C_{21}(-2\delta) C_{20}(-\delta)^{5}] + - C_{12}(2\delta) C_{10}(\delta) C_{10}(\delta) - C_{21}(-2\delta) C_{20}(-\delta) C_{10}(\delta)] + - \hbar^{6} S_{0r_{H}}^{2}(\omega) [C_{10}(\delta)^{5} - C_{20}(-\delta)^{5}] d\omega \times \\ \times \left\{ \int_{-\pi/r_{H}}^{\pi/r_{H}} |H_{r_{H}}(i\omega)|^{5} S_{0r_{H}}(\omega) d\omega \times \left\{ \int_{-\pi/r_{H}}^{\pi/r_{H}} |H_{r_{H}}(i\omega)|^{5} S_{0r_{H}}(\omega) d\omega \right\} \right\}^{-5},$$
(10.11.14)

From the obtained expression it is clear that for arbitrary imperfectness of yating in the expression for S_{3EP} there exists a component caused by nonlinear transtormation of the signal. It will give fluctuating error even with complete absence of noises. This component of error disappears when $C_{10}(5)^2 = C_{20}(-5)^2$, i.e., when there is no systematic error.

In investigating accuracy of the considered circuit it is also important to ostablish the influence on basic characteristics of the elevation of detuning of gate pulses 5. Here it is necessary to note the following: theoretically, the considered circuit more exactly approximates the optimum, the less 5; hewever, in practice, for small 5 we will obtain small values of the clope of the discrimination characteristic $K_{\rm R}$, which will lead to workening of dynamic characteristics of the system. Consequently, detuning 5 must be selected, in general, noticeably differing from zero. In order to investigate the dependence of $K_{\rm R}$ on 5, we introduce approximation (10.11.10) For $u_{\rm G}(5)$, and gate pulses $u_{\rm I}(t)$ and $u_{\rm R}(t)$ we assume identical and approximate them by expression (10.11.11). Here, as it is easy to calculate,

$$C'_{10}(\delta) C_{10}(\delta) - C'_{10}(-\delta) C_{10}(-\delta) = \frac{22.4\xi^3 x}{(1+\xi^3)^4 \Lambda y} e^{-5.6\frac{\xi^2 x^3}{1+\xi^3}}, \quad (10.11.19)$$

.

where $t = \frac{\tau_0}{\tau_0}$; $z = \frac{|\delta|}{\tau_0}$; $\Delta \phi$ - width of the directional pattern.

The graph of the dependence of $C_{10}^{*}(5)C_{10}(5) = C_{20}^{*}(-5)C_{20}(-5)$ on z when $\xi = 1$ is in Fig. 7.17 (there this dependence is given also for other approximations of the pulse pack and gate pulses; however, for understanding qualitative laws one approximation suffices). Analysis of Fig. 7.17 of Chapter VII shows that $C_{10}^{*}(5)C_{10}(5) = -C_{20}^{*}(5)C_{20}(-5)$, and consequently, also the slope of the discrimination characteristic K₀ attain a maximum at z = 0.4, i.e., when time detuning of gate pulses is

approximately 40% of the duration of the pack. Thus, namely such detuning is desirable in practice.

Investigation of the dependence of S_{OKB} on different parameters, as in the preceding case, we produce in two stages. First we consider cases of ideally matched gate pulses and zero detuning. The value of S_{3KB} here we designate by S_{MNH} . It is not difficult to see that S_{MNH} in this case is expressed again by formula (10.11.3).

Studying ratio S_{MNH}/S_{OUT} , we examine, as in the preceding case, ratio S_{3KB}/S_{MNH} . For simplicity we shall consider that gate pulses are identical. Influence of nonidentity of gate pulses we have already studied qualitatively. Here, as it is easy to see, for sufficiently large h (h \geq 10) we have

$$\frac{S_{010}}{S_{010}} = c^{2} \left[C_{11} (0) C_{10} (\delta)^{2} + C_{11} (0) C_{10} (-\delta)^{2} - C_{11} (2\delta) C_{10} (\delta) - C_{11} (-2\delta) C_{10} (-\delta) C_{10} (\delta) \right] \times \left[C_{10} (\delta) C_{10} (\delta) - C_{10} (-\delta) C_{10} (-\delta) \right]^{-2} \cdot c^{-2}$$
(10.11.20)

Using the earlier Gaussian approximations for pulse packs and gate pulses, we obtain

$$\frac{S_{0KS}}{S_{KRW}} = \left(\frac{1+\xi^2}{2\xi}\right)^4 \frac{1-e^{-5,6\xi^2z^2}}{5,6\xi^2z^2} e^{-5,6\frac{\xi^2z^2}{1+\xi^2}}.$$
 (10.11.21)



Fig. 10.32. The dependence of S_{BKB}/S_{MMH} on the relationship of durations of the pack, gate pulses and detuning between gate pulses in a circuit with detuned gate pulses.

The graph of the dependence of $S_{\rm HKH}/S_{\rm MMH}$ on ξ for different values of z is shown in Fig. 10.32. As can be seen from this figure, the value z = 0.4 is characteristic in that approximately at this z there occurs the local minimum of $S_{\rm HKH}/S_{\rm MMH}$, very close to unity. Thus, mismatch of gate pulses, if only gate pulses in the various channels are identical, with detuning $\beta \approx 0.4 \tau_{\rm H}$ leads to considerable worsenin, if accuracy as compared with the case of ideal gating.

Finishing the present section, we shall say several words on comparison

of the circuits of Figs. 10.27 and 10.28 for the method of tracking by pulse packs.

In accuracy the circuits are almost identical. If, however, we talk about those small differences which nevertheless take place, then it is possible to note somewhat lower criticality of the circuit of Fig. 10.27 to imperfectness of gating: to eliminate systematic error in the circuit of Fig. 10.27 it is necessary to ensure only symmetry of gate pulses, while in the circuit of Fig. 10.26 gate pulses must be identical.

§ 10.12. Comparison of Methods of Direction Finding

For the practice of radar goniometry it is extraordinarily interestingly to have comparative characteristics both of methods of angular direction finding and also of separate circuits of radar goniometer radio channels. Such comparison and basiconclusions from it were produced in the preceding sections; however it is of interest to systematically expound the questions pertinent here.

Potential accuracy corresponding to different methods of direction finding in most cases is expressed by formulas of the same time. Namely for methods of pattern scanning with high frequencies of scanning, scanning with compensation, IAC, phase center scanning with high frequencies of scanning, IPC, and, finally, flat scanning (when h > 1) minimum equivalent spectral densities are expressed by the single formula

$$S_{047} = \left[c^{2}h^{2} \frac{1}{\pi} \int_{-\infty}^{\infty} \frac{S_{0}^{2}(\omega)}{1 + hS_{0}(\omega)} d\omega \right]^{-1}.$$
 (10.12.1

In (10.12.1) by h it is necessary to non-rotative the ratio of the total mean signal power received by the antenna system of the gondometer to the power of noise acting in one channel, and coefficient c^2 is expressed through parameters of the gondometer antenna system and differs for different methods of direction finding;

for methods of pattern scanning, scanning with compensation and IAC

$$C^{1} == \mu_{a}^{2},$$
 (10,12,2)

where μ_a - gain factor of the directional pattern in the direction of the axis of axis of scanning or the equisignal axis;

for the method of phase center scanning and TPC

$$\boldsymbol{c}^{*} = \left(\frac{nd}{\lambda}\right)^{*}, \qquad (10, 12.3)$$

where d — amplitude of oscillations of the phase center or the distance between phase centers, λ — wavelength;

for the method of flat scanning

-578-

$${}^{*} = \frac{1}{\Omega T_{\pi}} \int_{-\infty}^{\infty} g'(\varphi) \, d\varphi, \qquad (10, 12, 4)$$

where Ω - angular velocity of pattern scanning;

 T_{π} - period of repetition of packs;

 $g(\varphi) \rightarrow power directional pattern (appropriately normalized);$

in particular, for a Gaussian directional pattern

$$C^{a} = \frac{2.8}{4y^{1}},$$
 (10.17.17)

where $\Delta \phi$ - width of the directional pattern for the level of half power.

In all the enumerated cases, when equivalent spectral densities are expressed by formula (10.12.1), it is rather simple to compare methods or direction finding. For this it is necessary to compare the signal-to-noise ratios h and coefficients e^2 for the different methods of direction finding. During comparison of signal-tonoise ratios it is possible to indicate the following: if the up stury area of the antenna system of the goniometer is fixed, then in the method of beaming with compensation, IAC and IPC there is received all the power reaching the specture of the antenna system; in the methods of pattern and phase scanning only half the power is received, i.e., the signal-to-noise ratio for these methods is approximately half that for the above-mentioned methods.

In the method of flat scanning, as already noted, the method received power is usually a minute share (of the order of $/\phi/\phi$, where $\Delta\phi - width$ of the directioned pattern; ϕ - angular dimensions of the scan sector) of the power which would be received by an antenna with the same aperture using the IAC method. However, it comparison of the method of flat scanning with methods IAC, THE and others, it is necessary to consider other tactical functions of the method of flat scanning, $\beta u_{2} =$ fying the indicated power losses. Thus, if we ignore the method of flat scanning, $\beta u_{2} =$ to-noise ratio best are the method of scanning with componsation, and the scale of β^{-1} IAC and TPC, where all the power resching the aperture of the united ay the scale $\phi = -\phi^{-1}$.

Now we produce comparison of gain factors e. Obviously, a depende on the genmatric charactoristics of the aperture of the antenna system. Here, for the same form and area of the aperture gain factors e for the methods of pattern scandar, scanning with compensation and IAC completely coincide. If, for instance, we have a square aperture with dimension d in the plane of the angle interesting us, for

-579-

these methods

$$c^{2} = \mu_{a}^{2} \approx \left(\frac{\pi d}{2\lambda}\right)^{2},$$

where λ - wavelength.

Thus, comparison of the methods of pattern scanning, scanning with compensation and IAC absolutely definitely testifies to the advantages of IAC and the method of scanning with compensation over the method of pattern scanning. Here the method of scanning with compensation and IAC are equivalent.

We produce now comparison of the phase method with IAC. In a number of cases the advantage of the phase method is obvious. This occurs when the antenna system aperture consists of two parts, separated a considerable distance d from each other Here quantity $(\pi d/\lambda)^2$ considerably exceeds the possible gain factors of the directional patterns, which can be provided due to the dimensions of the aperture of the antenna system.

Less clear is the question of comparison of methods of IAC and IPC if the aperture is of simple form, for instance square. If the dimension of the aperture in the plane of the angle interesting is d, the gain factor of the directional pattern which we will have for the IAC method is expressed by formula (10.12.6). But it is absolutely obvious that if we use the IPC method, dividing the aperture into two parts, we obtain a distance between phase centers equal to d/2. Putting d/2 instead of d in (10.12.3) and comparing the obtained expression with (10.12.6), we see that IAC and IPC are equivalent. Thus, with an aperture of the assigned simple form it is possible to use IPC or IAC.

Thus, consideration of slopes c did not change our conclusion concerning the equivalence of IAC, IPC and scanning with compensation and their advantage with respect to methods of pattern scanning and phase center scanning, which in the same conlitions are equivalent to one another.

For comparison of the method of flat scanning, for instance with IAC, it is necessary to compare the square of the gain factor of the directional pattern in IAC μ_a^2 with c^2 (10.12.4). With a Gaussian approximation of the directional pattern instead of (10.12.4) we have expression (10.12.5) for c^2 . With such an approximation of the directional for IAC we have $\mu_a^2 \approx 2/\Delta \varphi^2$. Thus, for the method of flat scanning on the assumption of equality of the mean received powers there is a somewhat better result. However, as already noted, comparison of the method of flat scanning with IAC, IPC and other methods is not quite just in view of the difference of the

-580-

(10.12.6)

tactical function of these methods.

We have till now left from consideration methods of pattern scanning and phase scanning in cases when the frequencies of scanning are not too high. However, this gap is easily filled if we recall results of comparison of these methods in the shown cases with cases of high-frequency scanning. Such comparison was conducted earlier and showed that accuracy of methods of pattern scanning and phase scanning rather sharply drops with decrease of the frequency ' scanning. Numerically, the drop of accuracy with decrease of frequency of scanning can be found from graphs of Figs. 10.11 and 10.22.

Thus, we have compared the considered methods of direction finding. The best one should recognize is the methods of scanning with compensation, IAC and IPC. The worst results come from methods of pattern scanning and phase center scanning, especially at low frequencies of scanning.

§ 10.13. Optimum Measurement of Angular Coordinates by Antennas of Phased Array Type

Till now we were occupied with synthesis and study of optimum radar meters of angular coordinates with assigned methods of direction finding. By this we understood that there is assigned the structure of the antenna array of the goniometer, i.e., there are determined operations produced on the electromagnetic field in the aperture of the antenna array. Such a formulation of the problem is limited. It does not give a complete answer to this question: if we are assigned the geometric form and dimensions of the aperture of the antenna array, what method of direction finding will be best?

Certainly, it is possible by comparing all methods of direction finding to establish the best of them. However there is no guarantee that there will not be found a certain method of direction finding, not considered by us, which gives greater accuracy of angle measurement. In order to obtain an answer to this question, it is necessary to synthesize an optimum goniometer, originating, not from an assigned method of direction finding, but from an assigned aperture of the antenna syntem.

In one case, very important and interesting, the technique developed in the preceding sections permits us to carry out complete synthesis of the goniometer circuit, including operations produced in the entenna system. We are talking about the case when there are used antennas of the phased array type.

Such antennas, as it is known, consist of a set of discrete radiators (usually isotropic), each of which is fed current with its own amplitude and phase. Selection

-581-

of amplitudes and phases of feed of the radiators permits us to obtain from the system of radiators various forms of directional patterns. Clearly, the sum of the effective areas of the radiators is approximately equal to the area occupied by all the radiators so that power losses due to the discrete structure comma arrays are insignificant. Interest in antennas of the phased accupied acception

has sharply increase.



During use of antennas of the phased array type the received signals from every element of the antenna usually are first processed separately up to and including transformation of frequencies, and only then are they cross-processed. This cross-processing determines the structure of the device, which should be selected optimally.

Let us assume that the antenna array

Fig. 10.33. Finding the form of signals received by elements of an antenna array.

contists of n elements, having (in a certain system of coordinates) polar coordinates R_i , α_i , β_i (i = 1, 2, ..., n). Vector **n** of the normal to the incident wave (which we consider planar in the vicinity of the antenna) we assume has coordinates α and β (Fig. 10.33). If we designate by ϕ_0 the phase of the incident wave at the origin of coordinates, the phase of the

signal received by the 1-th element of the antenna will be equal to

$$\Phi_i(t, a, \beta) = \Psi_0 + \frac{2\pi}{\lambda} (R_i \sin a_i \sin a \cos(\beta_i - \beta) + R_i \cos a_i \cos a).$$

then the received signals are recorded in the form

 $y_i(t) = \sqrt{P_{0i}} \operatorname{Re} E(t) u(t) \times \frac{I[a_0 t + \frac{2\pi}{4} (R_i \sin a_i \sin a \cos(\beta_i - \beta) + R_i \cos a_i \cos a]}{+ \sqrt{N_{ai}} n_i(t)},$

where $\frac{1}{100}$ - power of the signal received by the i-th element of the antenna,

 \mathbb{H}_{O1} -spectral density of noises acting in the channel of this element.

Subsequently to facilitate calculation we shall limit our consideration to the most important practical case of a flat square uniform array with identical elements (Fig. 10.34). Here for definitiveness we shall consider the variant with a tracking antenna, i.e., in the process of target tracking the actual array turns to match the normal to the array plane with the direction to the target. Considering here mismatch with respect to one of the coordinates equal to zero, as we did in all cases,

-582-



we can consider the array oriented so that one of its sides (we call it transverse) is parallel to the leading edge of the incident wave. Then signals received by elements of transverse rows will be identical. This is equivalent to observing n signals of power mP_{c-1} , where P_{c-1} power received by one element; m number of elements in the transverse row; n - number of elements in the longitudinal row. Denoting by d the distance between elements in the

Fig. 10.34. Square uniform antenna array.

longitudinal row of the array, we see that the phase in the i-th row of radiators (i = 1, 2, ..., n, Fig. 10.34) is equal to

$$\Phi_i(t, a) = \frac{2\pi}{\lambda} d\sin a. \qquad (10.13.1)$$

Thus, the received signals turn out to be equal to

$$y_{j}(l) = l\left(\frac{mP_{o_{1}}}{mP_{o_{1}}}\operatorname{Re} E(l)u(l)e^{l\left(\frac{m}{2}+\frac{2m}{2}d\operatorname{aln}e\right)} + \sqrt{N_{o}}n_{j}(l).$$
(10.13.2)

We now use a general expression for the operation of the optimum radio channel (10.3.29). Substituting (10.3.1) in this formula, we obtain

$$z(\tau) = \frac{26\Delta f_c h^2}{mP_{e_1}\lambda} \pi d \int_{t-\Delta t}^{t} h_{10\pi\tau} (\tau - s) \cos \omega_{\pi\mu} (\tau - s) \times \sum_{i=1}^{n} iy_i(s) u_a(s) \sin [\omega_{\Gamma}s + \psi(s)] ds \times \\ \times \int_{t-\Delta t}^{t} h_{10\pi\tau} (\tau - s) \cos \omega_{\pi\mu} (\tau - s) \times \sum_{i=1}^{n} y_i(s) u_a(s) \cos (\omega_{\Gamma}s + \psi(s)) ds. \quad (1.15.5)$$

The block diagram realizing this operation is depicted in Fig. 10.35. As can be seen from this diagram, the signal from each element of the antenna is separated into two channels, in which it is heterodyned by voltages shifted relative to each other in phase by $\pi/2$. Signals heterodyned with one phase are added, are filtered by an optimum filter with gain-frequency response curve (10.3.29) and enter the



Fig. 10.35. Block diagram of an optimum receiver during measurement of angular coordinates with the help of an antenna of phased array type: 1) adder; 2) optimum. filters with frequency response $H_{1 \text{ OFT}}$ (iw).

c is any number, including a negative number. Operation of the circuit in no way is changed by this.

phase detector as reference voltage. Signals heterodyned by veltage with the other phase, are added with scales 1, 2, ..., n, are filtered by the same filter and enter the other input of the phase detector. Summation with scales has a simple physical meaning: the further a given radiator is removed from the 1-th, the greater the information about angular coordinates in the signal from this radiator and the greater the weight with which this signal should be considered. Note that the weights with which we consider signals from different radiators are determined ambiguously: as sealed there can be used quantities c, c + 1, ..., c + n - 1, where

We now calculate potential accurcay of measurement of angular coordinates by a phased array. For this we use general formula (10.3.32). Substituting there expression (10.13.1) for the equivalent spectral density of measurement of one angle (taking into account the usual assumption that we have to measure both angles with identics accuracy), we obtain the following formula:

$$S_{\text{OHT}} = \frac{1}{h_{\Sigma}^{2} \left(\frac{2\pi d}{\lambda}\right)^{2} \frac{1}{n} \left[\sum_{i=1}^{n} i^{3} - \frac{1}{n} \left(\sum_{i=1}^{n} i\right)^{2}\right] \frac{1}{2\pi} \int_{-\infty}^{\infty} \frac{S_{\bullet}(\omega) d\omega}{1 + h_{\Sigma} S_{\bullet}(\omega)}, \qquad (10.13.3)$$

where $h_2 = ratio$ of total signal power received by all elements of the array to the spectral density of noise in one channel;

d - distance between neighboring radiators;

 λ - wavelength.

We calculate the sum in (10.13.3); finally we find

$$S_{\text{off}} = \frac{1}{\frac{\dot{h}_{\Sigma}^{2}}{2\pi} \int_{-\infty}^{\infty} \frac{S_{\bullet}(\omega) d\omega}{1 + h_{\Sigma} S_{\bullet}(\omega)} \frac{n^{2} - 1}{3} \left(\frac{\pi d}{\lambda}\right)^{2}}.$$
(10.13.4)

It is interesting for comparison to consider analogs of the methods of IAC and IPC during use of phased arrays as antennas and to compare accuracy with the potential



Fig. 10.36. Coupling of radiators of the array with the IAC method.

accuracy found by us.

In the TAC method all signals received by radiators are added, and there will be formed a difference signal, representing the difference between the sum of the signals of one half of the radiators and the sum of the signals of the other half of the radiators (Fig. 10.36). Further processing will only be on these two signals. Hore, 15 is easy, producing the shown operations on the signals (10.13.2), to fight that the sum signal is proportional.

$$g_+(a) = \cos \frac{n}{2} \frac{\pi d}{\lambda} \sin a$$
;

(sum directional pattern), and the difference is proportional to

$$g_{-}(a) = \sin \frac{n}{2} \frac{\pi d}{\lambda} \sin a$$

(difference directional pattern). Comparing these expressions with (46.4.6) and considering normalization (10.3.1), it is easy to calculate the gain factor of the directional pattern entering all formulas for accuracy of the IAC method:

$$\mu_{a} = \frac{n}{2} \frac{\pi d}{\lambda}, \qquad (1, 1; ...)$$

i.e., the equivalent spectral density of the IAC method is equal where the

$$S'_{\text{outr}} = \frac{i}{\frac{h_{L}^{2}}{2\pi} \int_{-\infty}^{\infty} \frac{S_{\bullet}(\omega) d\omega}{1 + h_{L}S_{\bullet}(\omega)} \frac{n^{2}}{4} \left(\frac{\pi d}{\lambda}\right)^{2}}.$$
 (1.14.17)

Comparing this expression with (10.13.4), we see that the optimum method of measurement of angles ensures somewhat higher accuracy than IAC; however this gain is insignificant. When n > 6

$$\frac{S'_{000}}{S_{000}} = \frac{4}{3} \frac{n^{0} - 1}{n^{0}} \approx \frac{4}{3}.$$

We shall now consider the IPC method. With this method radiators are coupled as shown in Fig. 10.37, i.e., the signals of two groups of radiators are added separately, and subsequently only the two sum signals thus obtained are processed.



These two signals are easily found by corresponding summation of signals (10.13.2). It is easy to find that one of the sum signals differ. from the second only by complex factor in $\pi d/\lambda$ sin α e . From this it is clear that

the equivalent base of the considered IPC method is equal to nd/2. Consequently, potential accuracy of this method, too, is expressed by

Fig. 10.37. Coupling of radiators of the array in the IPC method.

formula (10.13.6). Thus, accuracies of IAC and IPC are identical and somewhat lower than the accuracy of the optimum method of direction finding.

In the obtained results on synthesis of the optimum method of direction finding using antennas of phased array type it would have been possible to pass to the limit $n \rightarrow \infty$, d ~ 1/n. Here, a discrete array is turned into an antenna with a continuous aperture [51].

Thus, from consideration of antennas of phased array type it is clear that there exists an optimum method of processing the field in the aperture of an antenna system (method of direction finding) which does not reduce to the widely applied IAC or IFC methods. However, accuracy of measurement provided by the optimum method insignificantly exceeds accuracy of IAC or IFC (by 18% for equivalent spectral density), which has no special practical value.

§ 10.1". Malysis of Accuracy of Tracking Radar Contometers

In the preceding sections we studied in detail radio channels of tracking radar goniometers. However accuracy of measurement of angular coordinates of a target by a tracking goniometer is determined to a considerable extent by the structure of the smoothing circuits. In Chapter VI it was shown that mean square fluctuating error

-580-

is determined by the characteristic of the radio channel $S_{\partial KB}$ and characteristics of the smoothing circuits. Thus, for instance, during application of linear smoothing circuits with constant parameters error is determined by effective bandwidth of the system $\Delta f_{\partial \phi}$, which depends basically from the form of the smoothing circuits. Furthermore, by smoothing circuits we determine dynamic error, constituting a considerable share of total error of tracking a target by angles.

The present section considers different types of smoothing circuits and analyzes accuracy of measurement of angular coordinates by tracking radar goniometers.

10.14.1. Influence of a System of Automatic Gain Control on Accuracy of Tracking Goniometers

In analysis of accuracy of tracking radar goniometers it is necessary first of all to consider the influence of the system of automatic gain control. From results of Chapter II it follows that the AGC system can be considered in first approximation a linear inertial amplifier of the signal amplitude envelope, the gain factor of which depends on the level of amplitude of the signal. Action of AGC leads basically to normalization of the output signal of the amplifier covered by the AGC loop, so that the mean value of signal amplitude remains constant. In this approximation the AGC system, as it is easy to see, does not affect the magnitude of equivalent spectral density at the input of the discriminator.

However for analysis of accuracy of a tracking goniometer as a whole the AGC system must be taken into account. Actually, due to automatic gain control there is established a definite gain factor of the radio channel, depending on the signalto-noise ratio. This leads to dependence of the effective bandwidth and other characteristics of the closed tracking system on the signal-to-noise ratio, which in a certain way affects fluctuating and dynamic error of a tracking goniometer.

In goniometers using one antenna the gain factor of the radio channel with automatic gain control is calculated just as in § 7.10. The AGC loop in these geniometers covers the UPCh, whose passband is sufficiently wide as compared to the band of subsequent filters. The input signal of the radio channel has the form (10.3.3). Jain of the UPCh K_v is established in such a way that

 $\mathcal{K}_{\tau}^{2} \lim_{T \to \infty} \frac{1}{T} \int_{0}^{T} \overline{y_{t}(l)^{2}} dl = \text{const},$

1.0.,

 $K_{y}2(2N_{o}\Delta i_{y}+P_{o})=\mathrm{const},$

where $N_{\rm o}$ - spectral density of noise;

 $\Delta f_{\rm V}$ - bandwidth of the amplifier covered by the AGC loop.

Hence, as it is easy to see, the slope of the discrimination characteristic is equal to

$$K_{\mathbf{A}} = \frac{K_{\mathbf{0}\mathbf{A}}}{1 + \frac{1}{h} \frac{\Delta/\gamma}{\Delta/\epsilon}} = \frac{K_{\mathbf{0}\mathbf{A}}}{1 + \frac{y}{h}}, \qquad (10.1^{1} \cdot 1)$$

where $y = \Delta f_y / \Delta f_c$ - ratio of the band of amplification of the UPCh to the band of the signal;

- K_{OA} slope of the discrimination characteristic in the absence of noises;
 - h ratio of mean signal power to the power of noise in the task of signal fluctuations.

In circuits of goniometers using several antennas there exists a variety of methods of coupling in automatic gain control. Most frequently automatic gain control is coupled in one of the channels, simultaneously controlling amplification of the remaining channels. In this case the slope of the discrimination characteristic of the goniometer has the form (10.14.1); however, by h it is necessary to understand the signal-to-noise ratio of only the one channel in which the AGC system is coupled. In a number of cases to the AGC system there is simultaneously inserted voltage from several channels, taken with certain scales. The slope of the discrimination characteristic will again be expressed by formula (10.12.1), where by h it is necessary to understand the weighted size of the signal-to-noise ratios in the channels to which the AGC system is coupled. guantity $K_{\rm p}$ is calculated analogously in more complicated cases, too.

Besides the influence on the transmission factor of the radio channel automatic gain control also leads to decrease of the spectral density of parametric fluctuations becrease of the spectral density of parametric fluctuations is caused by the decadulating action of the AGC system. This decrease can be easily calculated with the help of results of Chapter II. According to (2.7.54) the spectral density of parametric fluctuations $S_{\rm HEL}^{(1)}$ with the presence of an AGC system is connected with the magnitude of $S_{\rm HEL}$ calculated in the preceding sections by relationship

$$S_{\mu_{0}\nu}^{(1)} = S_{\mu_{0}\nu} \frac{1}{n^{2}}, \qquad (10.14.2)$$

where n - a coefficient which depends on parameters of the AGC system and the signalto-noise ratio, where

$$n^2 = n_0^2 \frac{h}{h_0}.$$

-588-

Here h_0 - the signal-to-noise ratio corresponding to a signal equal to the delay level of the AGC system; n_0 - the value of parameter n corresponding to the delay level (see also Paragraph 7.10.1).

10.14.2. Errors of Goniometers with Smoothing Circuits with Constant Parameters

Usually in smoothing circuits of radar goniometers there are used filters with constant parameters. Here it is essential that the structure of the smoothing circuits of tracking goniometers in distinction from range finders and speed meters turns out in a number of cases to be rather complicated, inasmuch as smoothing circuits include such elements as actuator motors, power amplifiers, etc., necessary for rotation of the antenna.

Changes of smoothing circuits here can be carried out only by introduction of additional correcting links.

Smoothing circuit transfer functions used in practice have one of the following forms:

$$H_{1}(p) = \frac{D}{p} \frac{1+pT_{1}}{(1+pT_{1})(1+pT_{2})}, \quad H_{11}(p) = \frac{D}{p} \frac{(1+pT_{2})^{4}}{(1+pT_{1})^{4}(1+pT_{2})},$$
$$H_{11}(p) = \frac{D}{p} \frac{1+pT_{2}}{(1+pT_{1})(1+pT_{2})^{2}}, \quad H_{1V}(p) = \frac{D}{p} \frac{(1+pT_{2})^{4}}{(1+pT_{1})^{4}(1+pT_{2})^{4}}, \quad (10, 14.3)$$

where $T_1 > T_2 > T_3$.

Smoothing circuits of a very simple type in the form of an integrator (RC-filter) or a double integrator (double RC-filter) with correction, studied in detail in the preceding chapters, in the practice of radar goniometers are almost never met. Therefore, we will pursue our study of meters with smoothing circuits of type (10.14.3), using methods of approximation for calculation, developed in radar automation, since obtaining of exact results here is very difficult in view of the complexity of these expressions.

Smoothing circuits are characterized by Q-factor D and time constants $T_{\rm 1}$, $T_{\rm 2}$, $T_{\rm 3}$.

However, in practice, smoothing circuits are better characterized by secondles generalized parameters, having considerably greater physical and technical meaning: Q-factor D, cutoff frequency ω_c and reserve phase stability ψ_c .

Cutoff frequency $\omega_{\rm c}$ is determined by relationship

$$H(i_{0})|^{2} = 1, \qquad (10.1^{h}, 4)$$

the reserve of phase stability

$$\varphi_c = \pi + \arg H(i \sigma_c).$$

These parameters usually are assigned.

The relationship of time constants T_1 , T_2 and T_3 with parameters $D_1\omega_c$ and ϕ_c , in general, is rather complicated; however with accuracy sufficient for practical applications it can be fixed as follows.

For an example we shall consider only function $H_{I}(p)$ from (10.14.3). Results for the remaining transfer functions are established absolutely analogously. From (10.14.4) we have

$$|H_{1}(i\boldsymbol{\omega}_{c})| = \frac{D}{\boldsymbol{\omega}_{c}} \frac{V_{1} + \omega_{c}^{2} T_{2}^{2}}{V_{1} + \omega_{c}^{2} T_{1}^{2} V_{1} + \omega_{c}^{2} T_{3}^{2}}$$

It is easy to see that $1/T_c < \omega_c < 1/T_3$; considering $\sqrt{1 + x^2} \approx x$ for x > 1 and

 $\sqrt{1 + x^2} \approx 1$ for x < 1, for $H_{I}(p)$ we obtain

$$\boldsymbol{\omega}_{\mathbf{0}} = \frac{T_{\mathbf{0}}}{T_{\mathbf{1}}} \boldsymbol{D}. \tag{10.14.6}$$

The same result is obtained for transfer function $H_{IIIT}(p)$. For transfer functions of type $H_{TT}(p)$ and $H_{IV}(p)$ in an analogous way we find

$$\mathbf{e}_{\mathbf{c}} = D\left(\frac{T_{\mathbf{t}}}{T_{\mathbf{t}}}\right)^{\mathbf{t}}.$$
 (10.14.7)

We consider now the phase-frequency componies. We have

$$\arg H_1(i\omega_c) = -\frac{\pi}{2} - a_i c \operatorname{tg} \omega_c T_1 + \operatorname{acc} \operatorname{tg} \omega_c T_1 - \operatorname{acc} \operatorname{tg} \omega_c T_1,$$

Considering that are tg x $\approx \pi/2 - 1/x$ hen x > 1 and are tg x $\approx x$ when x < 1, it is easy to find that for $H_T(p)$

 $\boldsymbol{\beta}_{0} = \frac{1}{\boldsymbol{\omega}_{0}T_{1}} \left(1 - \frac{\boldsymbol{\omega}_{0}}{D} \right) + \boldsymbol{\omega}_{0}T_{1}, \qquad (10.14.7)$

where

$$\beta_0 = \frac{\pi}{2} - \gamma_0. \tag{10.14.9}$$

For transfer functions $H_{II}(p)$, $H_{III}(p)$ and $H_{IV}(p)$ analogous expressions will

have the form

$$\beta_{0} = \frac{2}{\omega_{c}T_{1}} \left(1 + \sqrt{\frac{\omega_{c}}{D}} \right) + T_{s} \omega_{c}, \quad \beta_{0} = \frac{1}{\omega_{c}T_{1}} \left(1 - \frac{\omega_{c}}{D} \right) + 2T_{s} \omega_{c},$$

$$\beta_{0} = \frac{2}{\omega_{c}T_{1}} \left(1 - \sqrt{\frac{\omega_{c}}{D}} \right) + 2T_{s} \omega_{c}.$$

$$(10.14.10)$$

Analysis of errors of the used approximations shows that errors here are fully permissible [52].

-590-

(10.14.5)

As can be seen from expressions (10.14.6) and (10.14.10), time constants T_1 , T_2 and T_3 are determined by parameters D, ω_c and φ_c ambiguously, i.e., there exists a broad class of systems with assigned D, ω_c and φ_c , but with different time constants T_1 , T_2 , and T_3 . Inasmuch as basic parameters D, ω_c and φ_c are fixed, these systems possess approximately identical dynamic properties; the difference of time constants T_1 , T_2 and T_3 plays a less essential role. However, from the point of view of operation most profitable is the system for which ratio T_2/T_3 is minimum [52]. Here, in particular, the system has a phase-frequency response changing little near the cutoff frequency, which ensures constancy of φ_c with change of amplification. Therefore in practice there are used basically systems possessing for assigned D, ω_c and φ_c a minimum value of ratio T_2/T_3 . We subsequently will limit our consideration to such systems.

Using expressions (10.14.8) and (10.14.10), in Table 10.1 we give minimum values of ratio T_2/T_3 for fixed D, ω_c and β_c .

From this table it is easy to obtain expressions for time constants T_1 , T_2 and T_3 in terms of basic parameters of the smoothing circuits, D, ω_c and Φ_c , given in

Table 10.1	
H (p)	$\left(\frac{T_1}{T_2}\right)_{n=1}$
H ₁ (p)	$\frac{4}{\boldsymbol{\beta}_{c}^{2}}\left(1-\frac{\boldsymbol{\omega}_{c}}{D}\right)$
H ₁₁ (p)	$\frac{\frac{1}{2}}{\frac{1}{2}}\left(1-\sqrt{\frac{1}{D}}\right)$
Η ₁₁₁ (ρ)	$\frac{\frac{8}{1}}{\frac{1}{1}}\left(1-\frac{\omega_{4}}{D}\right)$
$H_{1V_{a}}(p)$	$\frac{16}{\mathfrak{h}_{\mathfrak{c}}^{2}}\left(1-\sqrt{\frac{\omega_{\mathfrak{c}}}{D}}\right)$

()

Table 10.2.

What has been presented can be summarized in the following way. Smoothing circuits of goniometer systems are obtractorized by basic parameters b, $\omega_{\rm C}$ and $\beta_{\rm C}$. Transfer functions of smoothing circuits have form (10.14.3), where T₁, T₂ and T₃ are expressed through b, $\omega_{\rm C}$ and $\phi_{\rm C}$, with the help of Table 10.2.

Numerical values of basic parameters for real tracking systems have different values. The reserve phase stability usually is selected in such a manner that 20° f ϕ_{c} f 60° . Usually $\phi_{c} \approx 5^{\circ}$, and

 ω_c and b_c can have different values depending upon the assignment of the radar gontometers $b_c \approx 150-500$ 1/sec, $\omega_c \approx 8-400$ 1/sec. Hence with Table 10.7 it is easy to establish the order of time constants T_1 , T_2 , and T_3 .

Table 10,2			
71	T _s	T,	
$\frac{2}{\Theta_0 \beta_0} \left(\frac{D}{\Theta_0} - 1 \right)$	$\frac{2}{\omega_{e}\beta_{e}}\left(1-\frac{\omega_{e}}{D}\right)$	<u>β.</u> 2ω.	
$\frac{4}{\omega_{*}!}\left(\sqrt{\frac{D}{\omega_{*}}}-1\right)$	$\frac{4}{\omega_{\alpha}\beta_{\alpha}}\left(1-\sqrt{\frac{\omega_{\alpha}}{D}}\right)$	le 2wc	
$\frac{2}{\boldsymbol{\omega}_{a}\boldsymbol{\beta}_{a}}\left(\frac{D}{\boldsymbol{\omega}_{a}}-1\right)$	$\frac{2}{\omega_{c}\gamma_{c}}\left(1-\frac{\omega_{c}}{D}\right)_{-}$	<u>\$0</u> 4000	
$\frac{4}{\mathbf{w}_{c}1_{c}}\left(\sqrt{\frac{D}{\mathbf{w}_{c}}}-1\right)$	$\frac{4}{\omega_{c}\beta_{c}}\left(1-\sqrt{\frac{\omega_{c}}{D}}\right)$	<u><u><u></u></u> 4000</u>	
	$\frac{\mathbf{r}_{1}}{\frac{2}{\mathbf{w}_{e}\mathbf{f}_{e}}\left(\frac{D}{\mathbf{w}_{e}}-1\right)}$ $\frac{\frac{4}{\mathbf{w}_{e}\mathbf{f}_{e}}\left(\sqrt{\frac{D}{\mathbf{w}_{e}}}-1\right)}{\frac{2}{\mathbf{w}_{e}\mathbf{f}_{e}}\left(\frac{D}{\mathbf{w}_{e}}-1\right)}$ $\frac{\frac{4}{\mathbf{w}_{e}\mathbf{f}_{e}}\left(\sqrt{\frac{D}{\mathbf{w}_{e}}}-1\right)}{\frac{4}{\mathbf{w}_{e}\mathbf{f}_{e}}\left(\sqrt{\frac{D}{\mathbf{w}_{e}}}-1\right)}$	$\frac{7_{1}}{\frac{2}{\varpi_{e}\beta_{e}}\left(\frac{D}{\varpi_{e}}-1\right)} \frac{7_{e}}{\frac{2}{\varpi_{e}\beta_{e}}\left(1-\frac{\varpi_{e}}{D}\right)}$ $\frac{4}{\frac{4}{\varpi_{e}\beta_{e}}\left(\sqrt{\frac{D}{\varpi_{e}}}-1\right)} \frac{4}{\frac{2}{\varpi_{e}\beta_{e}}\left(1-\sqrt{\frac{\varpi_{e}}{D}}\right)}$ $\frac{2}{\frac{2}{\varpi_{e}\beta_{e}}\left(\frac{D}{\varpi_{e}}-1\right)} \frac{2}{\frac{2}{\varpi_{e}\beta_{e}}\left(1-\frac{\varpi_{e}}{D}\right)}$ $\frac{4}{\frac{4}{\varpi_{e}\beta_{e}}\left(\sqrt{\frac{D}{\varpi_{e}}}-1\right)} \frac{4}{\frac{2}{\varpi_{e}\beta_{e}}\left(1-\sqrt{\frac{\varpi_{e}}{D}}\right)}$	

Let us turn to study of fluctuating errors. These errors, as it is known, are expressed through the effective bandwidth of the tracking system.

Let us calculate effective bandwidths of closed systems for all types of smoothing circuits (10.14.3). They are determined by expression

$$\Delta f_{\phi\phi} = \frac{1}{2\pi} \int_{-\infty}^{\infty} \left| \frac{K_{\pi} H(i\omega)}{1 + K_{\pi} H(i\omega)} \right|^{3} d\omega, \qquad (10.14.11)$$

where K_A - the slope of the discrimination characteristic of the radio channel. Substituting for H(im) consecutively the expressions from (10.14.3) and calculating this integral in every case, we obtain (calculated by M. M. Kreymerman): for H_T(p)

$$\Delta f_{00} = \frac{1}{4} \frac{K_{R} D^{2} T_{2}^{2} + K_{R} D (T_{1} + T_{3})}{(T_{1} + T_{3}) (1 + K_{R} D T_{2}) - T_{1} T_{3} K_{R} D}; \qquad (10.14.12)$$

for H_{TT}(p)

$$\Delta f_{ab} = \frac{1}{4} \left[(1 + 2K_{R}DT_{a})K_{R}^{2}D^{a}T_{2}^{4} + 2T_{1}T_{2}K_{R}^{2}D^{a}(T_{1} + 2T_{a}) + K_{R}DT_{1}(T_{1} + 2T_{a})(T_{a} + 2T_{1} + K_{R}DT_{2}^{2}) - K_{R}DT_{1}^{2}(1 + 2DT_{a}) \left[-T_{1}^{2}T_{a}(1 + 2K_{R}DT_{2})^{a} - K_{R}DT_{1}^{2}(T_{1} + 2T_{a})^{a} + T_{1}(T_{1} + 2T_{a})(T_{2} + 2T_{1} + K_{R}DT_{2}^{2})(1 + 2K_{R}DT_{2})^{-1}; \right]$$

(10.14.13)

ter HIII(p)

$$\Delta f_{00} = \frac{1}{4} [D^{3} K_{2}^{2} T_{2}^{2} (2T_{1} + T_{3}) - K_{R} DT_{1} T_{3} (1 + K_{R} DT_{3}) + K_{R} D (T_{1} + 2T_{3}) (2T_{1} + T_{3}] \cdot [-T_{1} T_{3} (1 + K_{R} DT_{3})^{2} - T_{3} K_{R} D (2T_{1} + T_{3})^{2} + (T_{3} + 2T_{1}) (T_{1} + 2T_{3}) \times (1 + K_{R} DT_{3})^{2} - T_{3} K_{R} D (2T_{1} + T_{3})^{2} + (T_{3} + 2T_{1}) (T_{1} + 2T_{3}) \times (1 + K_{R} DT_{3})^{2} - T_{3} K_{R} D (2T_{1} + T_{3})^{2} + (T_{3} + 2T_{1}) (T_{1} + 2T_{3}) \times (1 + K_{R} DT_{3})^{2} - T_{3} K_{R} D (2T_{1} + T_{3})^{2} + (T_{3} + 2T_{1}) (T_{1} + 2T_{3}) \times (1 + K_{R} DT_{3})^{2} - T_{3} K_{R} D (2T_{1} + T_{3})^{2} + (T_{3} + 2T_{1}) (T_{1} + 2T_{3}) \times (1 + K_{R} DT_{3})^{2} - T_{3} K_{R} D (2T_{1} + T_{3})^{2} + (T_{3} + 2T_{1}) (T_{1} + 2T_{3}) \times (1 + K_{R} DT_{3})^{2} - T_{3} K_{R} D (2T_{1} + T_{3})^{2} + (T_{3} + 2T_{1}) (T_{1} + 2T_{3}) \times (1 + K_{R} DT_{3})^{2} - T_{3} K_{R} D (2T_{1} + T_{3})^{2} + (T_{3} + 2T_{1}) (T_{1} + 2T_{3}) \times (1 + K_{R} DT_{3})^{2} - T_{3} K_{R} D (2T_{1} + T_{3})^{2} + (T_{1} + 2T_{2}) K_{R} D (2T_{1} + T_{3})^{2} + (T_{1} + 2T_{2}) K_{R} D (2T_{1} + T_{3})^{2} + (T_{1} + 2T_{3}) K_{R} D (2T_{1} + T_{3})^{2} + (T_{1} + 2T_{3}) K_{R} D (2T_{1} + T_{3})^{2} + (T_{1} + T_{3})^{2} + (T_{1}$$

(10.14.14)

for H_{IV}(p)

$$\begin{split} \mathbf{A}_{I_{0}} &= \frac{1}{4} \left\{ K_{\mathbf{x}}^{2} D^{3} T_{\mathbf{y}}^{4} \left[T_{\mathbf{x}} T_{\mathbf{y}} K_{\mathbf{x}} D - 2 \left(T_{\mathbf{x}} + T_{\mathbf{y}} \right) \left(1 + 2DT_{\mathbf{y}} \right) \right] + \\ &+ 2K_{\mathbf{x}}^{2} D^{3} T_{\mathbf{y}}^{2} T_{\mathbf{x}}^{1} T_{\mathbf{y}} \left(KDT_{\mathbf{y}}^{2} + 2T_{\mathbf{x}} + 2T_{\mathbf{y}} \right) - \\ &- 4K_{\mathbf{x}}^{2} D^{3} T_{\mathbf{y}}^{2} \left(T_{\mathbf{x}} + T_{\mathbf{y}} \right) \left(T_{\mathbf{x}}^{2} + T_{\mathbf{y}}^{2} + 4T_{\mathbf{x}} T_{\mathbf{y}} \right) - \\ &- 2K_{\mathbf{x}}^{2} D^{3} T_{\mathbf{x}}^{2} T_{\mathbf{y}}^{2} \left(T_{\mathbf{x}} + T_{\mathbf{y}} \right) + K_{\mathbf{x}} DT_{\mathbf{x}} T_{\mathbf{y}} \left(K_{\mathbf{x}} DT_{\mathbf{y}}^{2} + 2T_{\mathbf{x}} + \\ &+ 2T_{\mathbf{y}} \right)^{\mathbf{y}} + 4K_{\mathbf{x}} DT_{\mathbf{x}} \left(T_{\mathbf{x}} + T_{\mathbf{y}} \right)^{\mathbf{y}} \left(1 + 2DT_{\mathbf{y}} \right) - \\ &- 2K_{\mathbf{x}} D \left(T_{\mathbf{x}} + T_{\mathbf{y}} \right) \left(T_{\mathbf{x}}^{2} + T_{\mathbf{x}}^{2} + 4T_{\mathbf{x}} T_{\mathbf{y}} \right) \left(K_{\mathbf{x}} DT_{\mathbf{y}}^{2} + 2T_{\mathbf{x}} + \\ &+ 2T_{\mathbf{y}} \right) \left\{ -T_{\mathbf{x}}^{3} T_{\mathbf{x}}^{3} K_{\mathbf{x}}^{2} D^{2} + 4K_{\mathbf{x}} DT_{\mathbf{x}}^{2} T_{\mathbf{x}}^{2} - \\ &- 2K_{\mathbf{x}} D \left(T_{\mathbf{x}} + T_{\mathbf{y}} \right) \left(T_{\mathbf{x}}^{2} + T_{\mathbf{x}}^{2} + 4T_{\mathbf{x}} T_{\mathbf{y}} \right) \left(K_{\mathbf{x}} DT_{\mathbf{y}}^{2} + 2T_{\mathbf{x}} + \\ &+ 2T_{\mathbf{y}} \right) \left\{ -T_{\mathbf{x}}^{3} T_{\mathbf{x}}^{3} K_{\mathbf{x}}^{2} D^{2} + 4K_{\mathbf{x}} DT_{\mathbf{x}}^{2} T_{\mathbf{x}}^{2} - \\ &- T_{\mathbf{x}}^{3} \left(T_{\mathbf{x}} DT_{\mathbf{x}} \right) \left(T_{\mathbf{x}}^{2} + T_{\mathbf{x}}^{2} + 4T_{\mathbf{x}} T_{\mathbf{y}} \right) \left(K_{\mathbf{x}} DT_{\mathbf{y}}^{2} + \\ &+ 2T_{\mathbf{y}} K_{\mathbf{x}} D \right) + K_{\mathbf{x}} D T_{\mathbf{x}} \left(T_{\mathbf{x}}^{2} + T_{\mathbf{x}}^{2} + 4T_{\mathbf{x}} T_{\mathbf{x}} \right) \left(K_{\mathbf{x}} DT_{\mathbf{y}}^{2} + 2T_{\mathbf{x}} + 2T_{\mathbf{y}} \right)^{2} - \\ &- T_{\mathbf{x}}^{3} \left(1 + 2K_{\mathbf{x}} DT_{\mathbf{y}} \right) \left(K_{\mathbf{x}} DT_{\mathbf{y}}^{2} + 2T_{\mathbf{x}} + 2T_{\mathbf{x}} \right) \left(T_{\mathbf{x}}^{2} + T_{\mathbf{x}}^{2} + \\ &- 4T_{\mathbf{x}} T_{\mathbf{y}} \right) \left(1 + 2K_{\mathbf{x}} DT_{\mathbf{y}} \right) \left(K_{\mathbf{x}} DT_{\mathbf{y}}^{2} + 2T_{\mathbf{x}} + 2T_{\mathbf{x}} \right) \right\}^{-1}. \end{split}$$

$$(10.14.15)$$

Expressions (10.14.12)-(10.14.15) give exact values of the effective passbands of the closed system. It is possible to express them in parameters D, φ_c and ω_c according to the formulas (10.14.6)-(10.14.10). For large values of $K_{\mu}D/\omega_c$ ($K_{\mu}D/\omega_c \ge 10$), which practically always takes place, in Table 10.3 there are given simple approximate expressions for effective bands of a closed tracking system.

Inasmuch as $K_{\rm A}$ depends on the signal-to-noise ratio in the radio channel, $\Delta f_{\partial \Phi}$ also will depend on this ratio. We note that for large reserves of phase stability $(\Phi_{\rm C} \geq 50^{\rm O})\Delta f_{\partial \Phi}$ for all types of transfer function has the same asymptotic form

$\Delta f_{\varphi\varphi} = \frac{K_{\chi}\omega_{e}}{4}.$

Knowing the effective band of the closed tracking system, we can write an expression for fluctuating error of angle measurement, which is equal to $\sigma_{\phi\Pi}^2 = 2S_{\Theta\Theta}\Delta f_{-2\phi}$. To decrease fluctuating error it is necessary, obviously, to decrease $\Delta f_{-2\phi}$. As can be seen from Table 10.3, to decrease $\Delta f_{-2\phi}$ it is necessary to decrease ω_c and K_{Π} ; however it is necessary to consider that dynamic errors here will increase. Optimum values of parameters of smoothing circuits must be selected from

-593-
considerations of a compromise between dynamic and fluctuating errors.

H (p)	Δ]
H ₁ (p)	$\omega_{o} \frac{K_{a} + \beta_{u}/2}{4 - \beta_{c}^{2}}$
Η ₁₁ (<i>p</i>)	$\frac{K_{A} + \frac{3}{8} l_{o}}{K_{A} (4 - l_{c}^{2}) - 2l_{o}}$
H ₁₁₁ (p)	$\mathbf{\Theta}_{c} \frac{K_{A} \left(1 - \frac{\beta_{c}^{2}}{16}\right) + \frac{1}{2} \beta_{c}}{4 - \frac{\beta_{c}^{2}}{12} - \frac{1}{2} K_{A} \beta_{c}}$
H _{1V} (<i>p</i>)	$K_{A}\omega_{c} \frac{K_{A}(1-\beta_{c}^{2}/16)+\frac{3}{8}\beta_{c}}{K_{A}(4-\beta_{c}^{2})-\beta_{0}}$

Several words about the condition $DE_{\mu}/\omega_{c} \ge 10$ used by us, allowing to obtain a very simple expressions for effective bands. This condition is realized practically always if h >> 1. With decrease of h, as can be seen from expression (10.14.1), E_{μ} starts to drop, and with sufficiently shall values of h approximate expressions for effective bands of Table 10.3 will already be incorrect. This is necessary to consider during practical calculations.

As an example we shall calculate fluctuating error of measurement of an angle by a system using IAC with the circuit of Fig. 10.17 (or the method of scanning with compensation with the circuit of Fig. 10.15). We assume that smoothing circuits have a transfer function of type $H_{I}(p)$. We consider that automatic gain control is closed through the sum channel.

For calculation of fluctuating error we use formula (10.7.5) for S_{BHCB} and (10.14.1) for $K_{\rm H}$ (where h is replaced by $h_{\rm H}$ - the signal-to-noise ratio in the sum channel, and $K_{\rm OH}$ = 1). The expression for the effective band we take from Table 10.3. We have

$$\sigma_{\phi,x}^{2} = 2S_{\phi,x,y}\Delta I_{\phi\phi} = \frac{1}{\mu_{x}^{2}\Delta I_{\phi}} \left(\frac{2 + \frac{1}{x}}{h_{z}} + \frac{x + 2 + \frac{1}{x}}{h_{z}^{2}} \right) \times \frac{\omega_{e}}{2} \frac{2K_{A} + \beta_{e}}{4 - \beta_{e}^{2}}.$$
(10.14.16)

. nikulandalik (hini inu

-594-

We take in (10.14.16) x = 5, $\varphi_c = 30^\circ$. Ratio $\omega_c / 4\Delta f_c$ we will change. Curves of the dependence of σ_{MI}^2 on h_{Σ} for different $\omega_c / 4\Delta f_c$ and y shown in Fig. 10.38.



Fig. 10.38. Example of the dependence of fluctuating error of a tracking goniometer on the signal-to-noise ratio. $-\omega_c/4\Delta f_c = 0.1; - -\omega_c/4\Delta f_c =$ = 0.01.

When using these graphs it is necessary to consider that expression (10.14.16) is valid, as we already noted, when $DK_{\pi}/\omega_{c} \ge 10$, i.e., if D = 400 1/sec and $\omega_c = 10 \text{ 1/sec}$, expression (10.14.16) is valid only when $K_{\pi} \ge 1/4$. This means that the curves of Fig. 10.33 are valid only up to values $1/(1 + y/h_{\Sigma}) \ge 1/4$. For instance, at y = 5 we should have $h_{\Sigma} \ge 1.6$; at y = 100 we have considerably greater limitations on the permissible range of values of h_{Σ} ($h_{\Sigma} \ge 33$). In order to have a dependence of σ_{def}^2 on $h_{\Sigma}^{}$, valid for all values of h_{γ} , it is necessary in (10.14.16) to substitute the exact value of the effective band of the closed system (10.14.12)-(10.14.15). This would lead

indicides in a second

to very bulky expressions, and therefore we will limit ourselves to the given approximate calculations, covering, incidentally, the most interesting cases for practice. We shall now study dynamic errors of tracking goniometer systems. First of all

we shall consider an input of the form

$$\boldsymbol{\vartheta}(l) = \boldsymbol{\vartheta}_{\bullet}(l) + \sum_{i=1}^{m} \boldsymbol{\mu}_{i} \boldsymbol{\vartheta}_{i}(l), \qquad (10, 14, 17)$$

where $i_{i}(t) - known$ functions;

 μ_{i} - random normally distributed coefficients with characteristics

$\overline{\mu_i} = 0, \ \overline{\mu_i \mu_h} = M_{ih}.$

The presentation of an input in the form (10.14.17) is rather general and covers a large number of practically important cases. This we already discussed in detail in the preceding chapters.

In Chapter VI we showed that mean square dynamic error in the considered case will be expressed by formula

$$\sigma_{ABB}^{2} = \sum_{l, k=1}^{m} \mathcal{M}_{ik} \mathbf{e}_{l}(l) \mathbf{e}_{k}(l),$$

where functions $\varepsilon_k(t)$ have Laplace transforms

$$\boldsymbol{E}_{k}(\boldsymbol{p}) = \frac{\boldsymbol{\Phi}_{k}(\boldsymbol{p})}{1 + K_{k}H(\boldsymbol{p})};$$

 $\theta_{k}(p)$ - Laplace transform of functions $\hat{s}_{k}(t)$;

H(p) - transfer function of the smoothing circuits.

For smoothing circuits of any of the types considered here, having astaticism of the 1st order, for large t we have

$$\mathbf{s}_{k}(t) = \frac{\mathbf{b}_{k}(t)}{K_{k}D}.$$
 (10.14.1)

Formula (10.14.19) is most often used in practice for estimating dynamic errors of tracking systems. Relationships (10.14.18) and (10.14.19) show that the steady-state value of dynamic error only exists when functions $\hat{s}_{1}(t)$ have a bounded 1st derivative. Otherwise dynamic error grows without limit. For instance, in the case of linear functions

$$\mathbf{\Phi}_{h}(t) = a_{sh} + a_{1h}t \tag{10.14.20}$$

we have

$$a_{k}(t) = \frac{a_{1k}}{K_{k}D}$$
 (10.14.21)

Mean square dynamic error here is equal to

$$\sigma_{n=n}^2 = \frac{\sigma_0^2}{K_n D^2}, \ \sigma_0^2 = \sum_{l,k=1}^m M_{lk} a_{lk} a_{lk}$$
(10.14.22)

Thus, dynamic error is inversely proportional to the slope of the discrimination characteristic $K_{\rm d}$ and Q-factor D. With decrease of $K_{\rm d}$, taking place with decrease of the signal-to-noise ratio, dynamic error increases. For example, in Fig. 10.39 there is given the dependence of $D\sigma_{\rm dyn}/\sigma_{\rm O}$ on h. Analogously we can consider error with more complicated inputs of form (10.14.17).

In the considered simple case dynamic and fluctuating components of total tracking error are minimized independently. To lower the first component it is necessary to increase Q-factor D; to lower the second $\pm t$ is necessary to increase the cutoff frequency ω_c . However, consideration of more complicated inputs leads to more complicated conditions of a minimum of overall error, ensuing from the compromise between dynamic and fluctuating errors.

(10.14.18)

Let us consider dynamic errors with purely random inputs. As it is known, during depletion [? processing (text misprint)] of a stationary random input with



Fig. 10.39. Example of the dependence of dynamic error of a tracking goniometer on the signal-to-noise ratio with an input in the form of known functions with random coefficients.

spectral density $S_{\mu}(\omega)$ there exists dynamic error, the mean square of which is equal to

$$\sigma_{ABE}^{2} = \frac{1}{2\pi} \int_{-\infty}^{\infty} \frac{S_{\mu}(\omega)}{[1 + K_{R}H(i\omega)]^{2}} d\omega. \quad (10.14.23)$$

For calculation by this formula we approximate $S_{\mu}(\omega)$ by expression

$$S_{\mu}(\boldsymbol{\omega}) = \begin{cases} S_{\mu}, |\boldsymbol{\omega}| \leq \boldsymbol{\omega}_{\mu}, \\ 0, |\boldsymbol{\omega}| > \boldsymbol{\omega}_{\mu}. \end{cases} (10.14.24)$$

Let us consider particular cases occurring with different relationships of

^ωμ, ^T1, ^T2, ^T3.

We turn to transfer function $H_{I}(p)$. We shall use approximations

$$|H_1(i\omega)| \approx \frac{D}{\omega}, \ \omega < \frac{1}{\tilde{J}_1}, \ |H_1(i\omega)| \approx \frac{D}{\omega^2 T_2}, \ \frac{1}{T_1} < \omega < \frac{1}{T_2}, \qquad (10.14.25)$$

(which already were used in deriving formula (10.14.6)). We note that when $\omega < 1/T_2$ quantity $|H_I(i\omega)| >> 1$.

Here

$$\sigma_{R\pi\pi}^{2} = \frac{S_{\mu}}{K_{A}^{2} 2\pi} \int_{-\omega_{\mu}}^{\omega_{\mu}} \frac{d\omega}{|H(i\omega)|^{2}} = \frac{1}{3\pi} \frac{S_{\mu} \omega_{\mu}^{3}}{K_{A}^{2} D^{2}}, \quad \omega_{\mu} < \frac{1}{T_{1}}, \\ \frac{S_{\mu} \omega_{\mu}^{3}}{3\pi K_{A}^{2} D^{1}} \left(\frac{3}{5} \omega_{\mu}^{2} T_{1}^{2} - \frac{3}{5} \omega_{\mu}^{3} T_{1}^{3} + 1\right), \\ \frac{1}{T_{1}} < \omega_{\mu} < \frac{1}{T_{2}}. \qquad (10.14.26)$$

Analogously for the remaining types of transfer functions we obtain the following results: in range $T_1 \omega_{\mu} < 1$ quantity $\sigma_{\pi\nu H}^2$ in all cases is identical and is expressed by formula (10.14.26); in range $1/T_1 < \omega_{\mu} < 1/T_2$ for transfer function $H_{III}(p)$ the square of error $\sigma_{\pi\nu H}^2$ has the same form as for $H_I(p)$, i.e., is expressed by formula (10.14.20), and for functions $H_{II}(p)$ and $H_{IV}(p)$ by

 $\sigma_{x=u}^{2} = \frac{1}{3\pi} \frac{S_{\mu} \omega_{\mu}^{3}}{K_{\mu}^{2} D^{2}} \left(\frac{3}{7} \omega_{\mu}^{4} T_{1}^{4} - \frac{13}{7T_{1}^{3} \omega_{\mu}^{3}} + 1 \right), \quad \frac{1}{T_{1}} < \omega_{\mu} < \frac{1}{|T_{2}|}.$ (10.14.27)

Note that the obtained formulas occur, of course, only for the case of $K_{\rm H}$ which is not too small [so that it is valid to disregard the one in the denominator of the integrand in (10.14.23)]. Time constant T_1 in formulas (10.14.26) and (10.14.27) can be replaced, if desired, by the expression containing parameters $\omega_{\rm C}$, D and $\phi_{\rm C}$ from Table 10.3.

In Fig. 10.40 we constructed from formula (10.14.26) the dependence of $K_{I}\sigma_{JXH}/\sigma_{O}(\sigma_{O} = 2S_{I}\omega_{II} - mean square value of input) on the width of the spectrum of$



Fig. 10.40. Example of the dependence of dynamic error of a tracking goniometer on the width of the spectrum of a random input.

input ω_{μ} for different ω_{c} . During calculation of this dependence we assumed D = 400 1/sec, $\varphi_{c} = 30^{\circ}$, $T_{2} \approx 0.1$ sec. Note that a section of this dependence to $\omega_{\mu} = 1$ cps is valid when $DK_{\Pi}T_{\Gamma}$ [?] >> 1, i.e., $K_{d} >> 1.25 \cdot 10^{-3}$ [for y = 100 in formula (10.14.1) the permissible signal-to-noise ratio h >> 10^{-1}]. The section $\omega_{\mu} > 1$ cps is valid practically only at $K_{\Pi} \approx 1$, i.e., for rather low noise levels.

Let us turn to the question of optimum selection of parameters of smoothing circuits in the presence of a random input. This question we shall consider purely illustratively, inasmuch as complete analysis is very difficult to perform here. Let us assume that the random input has a square spectrum (10.14.24) with $1/T_1 < \omega_{\mu} < 1/T_2$. If the smoothing circuit has transfer function $H_I(p)$, total error of tracking is equal (when $\omega_{\mu} T_4 >> 1$)

$$\sigma^{a} = \sigma_{\phi a}^{a} + \sigma_{ABB}^{a} = 2\varpi_{0} \frac{K_{A} + \frac{1}{2} I_{0}}{4 - I_{c}^{2}} S_{\phi KB} + \frac{S_{\mu} \omega_{\mu}^{5} \cdot 4}{15 \pi \omega_{e}^{4} K_{A}^{2} I_{c}^{2}}.$$

Varying this expression with respect to ω_C and β_C , it is easy to find that the optimum cutoff frequency is equal to

(10.14.28) .

$$\Theta_{\text{CORT}} = \frac{0.7}{\Theta_{\mu}} \left(\frac{4 - \beta_c^2}{K_A + \frac{1}{2} \beta_c} \right)^{\frac{1}{5}} \left(\frac{S_{\mu}}{S_{ars}} \right)^{\frac{1}{5}} \frac{1}{\beta_c K_A}, \quad (10.14.29)$$

and the optimum reserve of phase stability for $K_{\rm II} = 1 \beta_{\rm C \ O II T} = 2/5$, i.e., $\varphi_{\rm C} = 67^{\circ}$.

Substituting this value of φ_{c} in (10.14.32), we obtain (for $K_{II} = 1$)

$$\sigma_{cont} = 0.88 \omega_{\mu} \left(\frac{S_{\mu}}{S_{ont}} \right)^{\frac{1}{5}}$$
 (10.14.30)

いたので、「「「「「「「」」」」

Thus, the optimum cutoff frequency is proportional to the bandwidth of the spectrum of the random input; with growth of the spectral density of this input ω_{c} increases, and with growth of S_{BKB} it drops; however, the dependency of ω_{c} on these quantities is rather weak.

The given example shows that there exist optimum values of parameters of smoothing circuits, depending on the type of input. Optimum values depend also on the signal-to-noise ratio (through K_{II} and equivalent spectral density S_{OKE}) and should be adjusted with change of this ratio. In an analogous way we can investigate cases of other inputs.

Let us consider now in brief the question of the influence of parametric fluctuations. As shown in Chapter VI, the presence of parametric fluctuations leads to increase of errors of measurement. With smoothing circuits with constant parameters this increase is expressed by relationship (6.2.39). As we have seen earlier, the spectral density of parametric fluctuations in almost all cases is expressed by formula (10.5.23) with (10.5.24). In Paragraph 10.14.1 it is shown that with AGC the spectral density of parametric fluctuations decreases according to relationship (10.14.2). From this we find that for the considered circuit error taking into account parametric fluctuations is expressed by formula

$\sigma^{2} = \sigma_{0}^{2} \left(1 + 2S_{\text{map}} \Delta f_{2\phi} \frac{1}{n^{2}} \right) = \sigma_{0}^{2} \left(1 + \frac{\Delta f_{2\phi}}{\Delta f_{c}} \frac{1 + 3x + x^{2}}{x(1 + x)} \frac{1}{n^{2}} \right),$

where σ_0^2 - error ignoring parametric fluctuations, and the remaining designations are the same as before [see designations for formulas (10.5.24) and (10.14.2)].

10.14.3. Optimum Smoothing Circuits and Smoothing Circuits with Variable Parameters

Smoothing circuits considered in the preceding paragraph, as already noted, are practically the most commonly used in radar goniometers.

However, it is of interest to discuss other possible types of smoothing circuits.

-599- ;

Although the problem of their synthesis in principle is solved in the rather general assumptions about the character of the input (see § 6.8), an effective solution cannot always be obtained. The simplest solutions are obtained when the input is a stationary random process or a process with stationary increments. The smoothing filter here has constant parameters, depending on the statistical properties of the input. Here the smoothing circuits synthesized for the goniometer will not differ at all from smoothing circuits of a range finder or speed meter if properties of the input in all these cases are identical. Therefore all results obtained in Chapters VII and IX regarding optimum smoothing circuits almost without change can be tranferred to the case of measurement of angles. For instance, if the angular coordinate of the target is presented in the form of the double integral of white noise with spectral density B_2 , the optimum smoothing filter is a double integrator with correction, with transfer function

$$H(p) = \frac{K_{*}(1+pT_{*})}{p^{*}}, \qquad (10.14.31)$$

where $K_{_{\rm N}}$ and $T_{_{\rm K}}$ are expressed by characteristics of the radio channel $K_{_{\rm I\!I}}$ and $S_{_{\rm SFOB}}$ in the following way:

$$K_{\rm R} = \frac{1}{K_{\rm R}} \sqrt{\frac{B_{\rm R}}{S_{\rm SRB}}}, \quad T_{\rm A} = \sqrt[4]{4S_{\rm SRB}/B_{\rm R}} \tag{10.14.32}$$

(coefficient B_2 has dimensionality $[deg^2 cps^3]$ and is equal to the mean square of the angular velocity developed by the target in 1 sec). Variance of total error of measurement in this case is equal to

$$\sigma^{*} = \frac{1}{2} (B_{*} S_{*\pi*}^{3})^{1/4}. \qquad (10.14.33)$$

Let us consider, for example, a tracking goniometer in which there is used the IAC method. The circuit of the goniometer with smoothing circuits (10.14.31) is depicted in Fig. 10.15. From formulas (10.7.5) and (10.14.33) we find that total error is equal to

$$\sigma^{2} = \sqrt{2} \left[B_{s} \frac{1}{\mu_{a}^{6} \Delta / c} \left(\frac{2 + 1/x}{h_{z}} + \frac{x + 2 + 1/x}{h_{z}^{2}} \right)^{s} \right]^{1/4}. \quad (10.14.34)$$

Considering $\mu_a = 1.4/\Delta \phi$, where $\Delta \phi$ — width of the directional pattern, we reduce formula (10.14.34) to a form convenient for calculation

$$\left(\frac{\bullet}{\Delta \varphi}\right)^{\bullet} = 0.87 \left[\frac{B_{\bullet}}{\Delta \varphi^{\bullet} \Delta f_{\bullet}^{3}} \left(\frac{2+1/x}{h_{\pm}} + \frac{x+2+1/x}{h_{\pm}^{2}}\right)\right]^{1/4}.$$

Curves of the dependence of $\sigma/\Delta \phi$ on h_{Σ} for various x are shown in Fig. 10.41. During calculation we assumed $B_{\rho} = 10^{-7} \text{ deg}^2 \text{cps}^3$, $\Delta \phi = 1^{\circ}$, $\Delta f_{\rho} = 30 \text{ cps}$.



Fig. 10.41. The dependence of total error of an tracking goniometer on the signal-to-noise ratio with an input in the form of the double integral of white noise.

Note that parameters of the optimum smoothing filter (10.14.32) depend on the signal-to-noise ratio. With practical realization of such a filter they will be selected, of course, constant, and error can be expressed by a fromula differing from (10.14.33). This and similar cases were considered in detail in Chapters VII and IX. The results obtained there are easily generalized to the case of goniometers. いいなななななななないないないであっていましたとう

「「「」」になるのないので、「」」」

In many cases a minimum of errors in radar meters is ensured with the use of smoothing circuits with variable parameters. Practically such circuits in radar goniometers are almost never used; however,

considering the prospects of development of technology, it is of interest to analyze error of radar goniometers using smoothing filters with variable parameters. Here again it is necessary to note that from the point of view of smoothing the goniometer has no individual specific character as compared to meters of other types. Therefore, results obtained for range finders and speed meters are almost completely transferable to the case of goniometers.

Let us briefly discuss basic propositions concerning smoothing circuits with variable parameters. If the angular target coordinate varies according to the law

$$\boldsymbol{\vartheta}(t) = \sum_{i=1}^{m} \mu_i \boldsymbol{\vartheta}_i(t) + \boldsymbol{\vartheta}_{\boldsymbol{\varphi}}(t),$$

where μ_{1} - normally distributed random variables; $\overline{\mu}_{1} = 0$; $\overline{\mu_{1}\mu_{k}} = M_{1k}$; $\$_{1}(t) - known functions, the optimum smoothing filter will have variable parameters. Its pulse response (with an optimum discriminator) and variance of error are expressed by formulas (6.8.54).$

If, for instance, the angular target coordinate varies according to the law

$$\mathbf{\Theta}(l) = (\mathbf{\Theta}_1 + \mathbf{\Theta}_2 l) + (\mathbf{\mu}_1 + \mathbf{\mu}_2 l),$$

where $\overline{\mu_1^2} = \sigma_1^2$, $\overline{\mu_2^2} = \sigma_2^2$, according to (6.8.54) pulse response of the smoothing filter is equal to

-601-

 $G(t, \mathbf{v}) = \frac{\mathbf{e}_{1}^{2} \left(1 - \frac{\mathbf{e}_{2}^{2}}{S_{ouv}} \frac{\mathbf{v}^{2}}{6}\right) + \mathbf{e}_{2}^{2} t \mathbf{v} \left(1 + \frac{\mathbf{e}_{1}^{2}}{S_{ouv}} \frac{\mathbf{v}}{2}\right)}{1 + \frac{1}{S_{ouv}} \left(\mathbf{e}_{1}^{2} \mathbf{v} + \mathbf{e}_{2}^{2} \frac{\mathbf{v}^{2}}{3}\right) + \frac{\mathbf{e}_{1}^{2} \mathbf{e}_{2}^{2} \mathbf{v}^{4}}{12S_{ouv}^{2}}}$

Variance of error of measurement is found in the form

$$=\frac{e_1^2+e_2^2t^2+\frac{1}{3S_{ont}}e_1^2e_2^2t^4}{1+\frac{1}{S_{ont}}e_1^2t+\frac{1}{3S_{ont}}e_2^2t^2+\frac{1}{12S_{ont}^2}e_2^2t^4}$$

With growth of t error seeks 0. For large t, obviously, we have

 $\sigma^2 = \frac{4S_{0\pi\tau}}{t}$

With arbitrary smoothing circuits with variable parameters error of tracking is determined by formula (6.2.15). No difficulties appear during use of this formula.

§ 10.15. Nonlinear Phenomena in Tracking Goniometers

The whole preceding analysis of tracking radar goniometers was based on their linearized presentation. Such a presentation is valid if tracking of angles is carried out with sufficiently high accuracy. In this case it turns out that the mean value of the output signal of the radio channel is proportional to the current angular mismatch, and the spectral density of noises does not depend on mismatch. However, these idealizations take place only with a comparatively high signal-to-noise ratio. In real conditions, especially in the case of the action of interferences, essentially decreasing the signal-to-noise, mismatch during tracking may be considerable.

Here the discrimination and fluctuation characteristics of the radio channel of the goniometer become nonlinear functions of mismatch. Study of the process of tracking and of errors of goniometers in this regime represents an important and interesting task.

The task of investigating tracking systems in a nonlinear regime was considered in § 6.3. This problem is very complicated, and results obtained upon its solution are approximate and particular. However by them we can draw certain conclusions about nonlinear phenomena in tracking goniometers and find the order of numerical characteristics of these phenomena. We then use these results to study the phenomena of breakoff of tracking occurring with a sufficiently high level of noises, when increasing mismatch becomes comparable to the width of the directional pattern.

-602-

In order to best approximate conditions in which we produced analysis in § 6.3, we shall limit ourselves henceforth to the simplified situation when the fluctuation characteristic in required limits can be considered constant and equal to the characteristic S_{HEB} used everywhere earlier. This assumption is well justified, since due to demodulating properties of automatic gain control the fluctuation characteristic is essentially leveled. Furthermore, most interesting is the case of allowing for nonlinearity of the discrimination characteristic. In the preceding chapters, conforming to requirements of linear theory, we calculated only the slope of the discrimination characteristic at zero mismatch. Now we need, in general, knowledge of all the behavior of the discrimination characteristics.

The form of the discrimination characteristics of radar goniometers essentially depends on the methods of direction finding utilized and, in general, turns out to be very complicated. We will limit ourselves to consideration, as an example, of a goniometer using the IAC method. For a goniometer with IAC, in which reception of signals is carried out by two directional patterns having form $g_1(\Phi)$ and $g_2(\Phi)$, the received signals, as it is known, can be recorded in the form of formula (10.3.3), in which n = 2, and $U_i(t, \alpha)$ are determined by relationships (10.2.7). Determining for any of the circuits of the IAC method the error signal and averaging, we obtain (assuming identity of filters in the circuits) for the discrimination characteristic the following expression:

$$a(\alpha) = \frac{P_{c_{z}}}{8\pi\Delta T_{o}} \int_{-\infty}^{\infty} \operatorname{Re} H_{1}(i\omega) H^{*}_{s}(i\omega) S_{0}(\omega) d\omega \times (10.15.1)$$

$$\times [g_{1}^{2}(\gamma_{1} + \alpha) - g_{2}^{2}(\gamma_{2} - \alpha)] = K[g_{1}^{2}(\gamma_{1} + \alpha) - g_{2}^{2}(\gamma_{n} - \alpha)].$$

Thus, the form of the discrimination characteristic essentially depends on the form of the directional pattern. We use the very wide-spread cosine approximation of the directional pattern, considering the diagrams identical:

$$g_1(\varphi) = g_1(\varphi) = \cos \frac{\pi}{2} \frac{\varphi}{4\varphi},$$

where $\Delta\phi$ - width of the directional pattern at the level of half power.

We also assume $\gamma_1 = \gamma_2 = \Delta \varphi/2$. Here,

$$a(a) = K \sin \frac{\pi a}{\Delta y}. \qquad (10, 10, 2)$$

We obtained a discrimination characteristic in the form in which it was approximated in § 6.3. Consequently, it is possible to use directly the results of § 6.3, considering the half-width of the selected region $\Delta = \Delta \varphi$.

-603-

From results of § 6.3 it follows that the critical value of the ratio of mean square error of the linearized system $\sigma_{\rm pl}$ to the half-width of the selected region $\Delta = \Delta \phi$ at which breakoff of tracking occurs is $(\sigma_{\rm pl} / \Delta \phi)_{\rm Kp} \approx 0.2$ (by the criterion of sharp increase of the average time to the first breakoff) and $(\sigma_{\rm pl} / \Delta \phi) \approx 0.1$ (by the criterion of sharp growth of mean square error).

Considering $\sigma_{\pi} = \sqrt{2\Delta f_{gh} S_{3KB}}$ (dynamic errors are absent) and using expression S_{SKB} for the investigated circuit (10.7.5), it is easy to find the critical signal-to-noise ratio, at which failure occurs

$$\left(\frac{\epsilon_{a}}{\Delta \gamma}\right)_{ap} = \frac{\sqrt{\frac{2+1/x}{h_{ap}} + \frac{x+2+1/x}{h_{ap}^{2}}}}{0.7\sqrt{\frac{\Delta f_{e}}{\Delta f_{e\phi}}}} = 0.2 \text{ or } 0.1.$$

Hence when x = 1 and $\Delta f_{c} /\Delta r_{gh} = 100$ we have $h_{Rj} = 2.3$ (by the criterion of sharp growth of mean square error). The obtained magnitudes of h_{Rj} coincide in order with those that are known from practice.

Consideration of more complicated examples, accounting for the proximity of the obtained results, does not have meaning.

§ 10.16. Nontracking Rader donioneters

We shall investigate radar goniometers not containing feedbacks and giving an estimate of the angular coordinate of a target at the output of an open circuit. There exist many tactical tasks in which application of just such goniometers is most expedient.

As a rule, nontracking goniometers work by the following scheme. Directional patterns of the antenna system of such goniometers cover the sector of possible values of the angular coordinates of the target. The goniometer consists of two basic units a unit of primary processing of the signal and a unit of secondary processing. The unit of primary processing continuously issues an estimate of the angular coordinate of a target from the observed realization of a signal of duration At. Here time At, naturally, is sufficiently small so that the angular coordinate of the target will not vary. The unit of secondary processing smoothes the data of primary processing taking into account a priori laws of change of angular coordinates.

Synthesis of an optimum nontracking meter was performed in § 6.6. In the case of a rapidly fluctuating signal and for sufficiently large signal-to-noise ratios the optimum operation is given by expression (6.6.55). The structure of the optimum nontracking meter turns out to be very close to that which was described above.

-604-

The meter contains a unit of primary processing, called in Chapter VI an "estimator unit," which issues the maximum likelihood estimate of the measured parameter from the segment of realization of the signal during a time considerably exceeding the time of correlation of the signal, but considerably smaller than the time of correlation of the measured parameter. Then there follows a smoothing filter which has a pulse response given by relationship (6.6.53). Fotential ageneacy of nontracking meters here coincides with the accuracy of tracking meters.

Then we shall study synthesis of optimum nontracking gontemeters using the IAC and DPC methods. For these methods it is pessible to completely find the optimum circuit of a nontracking gontometer (for IAC, to be sure, with a structure approximation of the directional pattern). Other methods of direction finding in nontracking gontometers are practically not used.

10,10.1. Optimum Circuit of a Montracking Gentometer with IAC

We start our consideration with the IAT method. With this method there is received a pair of signals of form (10.3.3), where $U_{ai}(t, a)$ are determined by formula (10.7.5). Correlation functions of these signals is given by expression (10.3.7). If the interval of observation of signals is (0, 4t), then the functional of distribution of probabilities of these signals has the form (10.3.8) (taking into account (10.3.10) and (10.3.11)]. So that this functional is completely determined, we need to find the matrix of inverse-correlation functions, molving equation (10.5.10) for this. The solution may again be sought in form (10.3.40), and for $v(t_1, t_2)$ we obtain equation (10.3.15), which (with identical noises) can be rewritten in the form

$$\frac{P_{1}(I_{1} - a)^{b} + P_{2}(I_{1} + a)^{b}}{2N_{0}} \int_{0}^{a} \rho(I_{1} - I_{1}) \upsilon(I_{1}, I_{1}) dI_{1} + \rho(I_{1} - I_{2}) + \upsilon(I_{1}, I_{2}) - 0,$$
(10.11.1)

Henceforth we shall consider the patterns identical, and we introduce the approximation already used in the preceding soution:

$$g_1(\mathbf{q}) = g_1(\mathbf{q}) = \operatorname{Cos} \frac{\mathbf{q}}{2} \frac{\mathbf{q}}{2\mathbf{q}}, \qquad (12, 12, 22)$$

where $\Delta \phi$ - width of the directions) pattern for the half-power level.

We also assume that $\gamma_1 = \gamma_2 = 2\pi/2$, i.e., the patterns arous at the half-power level. Here (10.10.1) takes the following form:

$$\frac{P_{e_{B}}}{2N_{e}} \int_{B}^{A_{l}} P(l_{1} - l_{a}) \upsilon(l_{a}, l_{a}) dl_{a} + P(l_{a} - l_{a}) + \upsilon(l_{1}, l_{a}) \omega \upsilon(l_{a}, l_{a}) (1 + 1 + 1)$$

$$(1 + 1 + 1)$$

- را()را •

where Po 2 - total received power.

As can be such from (10,10.3), function $v(t_1, t_2)$ will not depend on a,

We compose now the maximum likelihood equation for measurement of parameter α_{i} . Calculating by formula (20.3.11) it is easy to prove that δ_{10} K/oa = 0. Therefore the maximum likelihood equation takes the form

$$= \sum_{i,j=1}^{n} \int_{0}^{M} \int_{0}^{M} \Psi_{ij}(t_i, t_j) \Psi_i(t_j) \Psi_i(t_j) dt_i dt_j = 0.$$

$$(10, 16, h)$$

Bubatituting expression (10, 2, 14) in (10, 20, 3), it is possible to convert equation (10, 10, 4) to a rather simple form and to solve it explicitly. Omitting simple calculations we give the final results

$$\hat{a} = \frac{\Delta g}{\pi} \operatorname{arcctg} \left\{ \left[\operatorname{Re} \int_{0}^{\Delta t} \int_{0}^{\Delta t} t^{i}(t_{1}, t_{2}) u(t_{1}) u^{*}(t_{2}) e^{i\omega_{0}(t_{1} - t_{2})} \times y_{1}(t_{2}) y_{1}(t_{2}) dt_{1} dt_{2} - \operatorname{Re} \int_{0}^{\Delta t} \int_{0}^{\Delta t} v(t_{1}, t_{2}) u(t_{1}) u^{*}(t_{2}) \times v(t_{1}, t_{2}) u(t_{1}) u^{*}(t_{2}) \times v(t_{1}, t_{2}) u(t_{1}) u^{*}(t_{2}) \times u(t_{1}) u^{*}(t_{2}) dt_{1} dt_{2} \right] \left[2\operatorname{Re} \int_{0}^{\Delta t} \int_{0}^{\Delta t} v(t_{1}, t_{2}) \times u(t_{1}) u^{*}(t_{2}) e^{i\omega_{0}(t_{1} - t_{2})} y_{1}(t_{1}) y_{2}(t_{2}) dt_{1} dt_{2} \right]^{-1} \right\}.$$

$$(10.16.5)$$

In this expression we have to equations on $v(t_1, t_2)$, found from equation (10.16.3).

We shall first consider the case of capid fluctuations of the signal, when the time of correlation of fluctuations is considerably less than Δt . As it is easy to use, equation (10.10.3) in this case is identical to equation (10.3.17), and for $v(t_{j}, t_{j})$ we obtain solution (10.3.18). Substituting (10.3.18) in expression (10.16.5), we can present the optimum operation of measurement in the following form:

$$\hat{a} = \frac{A_{\varphi}}{\pi} \operatorname{arccig} \left\{ \int_{0}^{t} d\tau \left[\int_{0}^{A_{1}} h_{10\Pi \tau} (\tau - s) y_{1}(s) \times \right] \right\}$$

$$\times \left[\left\{ \int_{0}^{A_{1}} d\tau \operatorname{Re} \int_{0}^{A_{1}} h_{10\Pi \tau} (\tau - s) y_{1}(s) u(s) e^{i\omega \sigma s} ds \right]^{s} \right] \times \left[\left\{ \int_{0}^{A_{1}} d\tau \operatorname{Re} \int_{0}^{A_{1}} h_{10\Pi \tau} (\tau - s) y_{1}(s) u(s) e^{i\omega \sigma s} ds \int_{0}^{A_{1}} h_{10\Pi \tau} (\tau - s) \times \right] \right\}, \quad (10.16.6)$$

-606-

where the pulse response of the optimum filter h_{1-0HT} is determined by relationship (10.3.23) [the gain-frequency response of the filter is given by formula (10.3.25)]. Expression (10.16.6) can be easily converted to real form [analogous to the transition from (10.3.24) to (10.3.29)].

The complete block diagram of an optimum nontracking meter using IAC can be presented in the form depicted in Fig. 10.42. Let us describe briefly the work of



Fig. 10.42. Optimum circuit of a nontracking meter with IAC for rapid fluctuations of the signal; 1) optimum filters with frequency response H_{10RT} (iw); 2) square-law detectors; 3) integrators; 4) dividers; 5) nonlinear unit; 6) smoothing circuits.

this circuit. Signals from the output of the antennas first are heterodyned (It is understood that heterodyne voltage has the proper amplitude modulation), are filtered by optimum filters with frequency response (10.3.25), are detected by square-law detectors, and are subtracted. The difference is integrated and enters a divider (as the dividend). To the second

input of the divider (the divisor) there proceeds the integrated result of multiplication of signals from the output of the optimum filters (phase detection). The result of division enters an inertialess unit with nonlinear characteristic y == arc ctg x (y - output, and x - input quantities).

The output signal of the nonlinear unit, multiplied by $\Delta \phi/\pi$ ($\Delta \phi$ - width of the directional pattern), enters the smoothing circuits. Optimum smoothing circuits, as shown in § 6.6, are ε linear filter with pulse response (6.6.53).

Let us consider the question of accuracy of measurement of angles in the synthesized circuit.

First of all we investigate accuracy of the estimator unit. In general in the presence in the circuit of an estimator unit estimates of deviations of its accuracy from optimality are very difficult to calculate. However, for an optimum circuit accuracy can easily be found. It is possible to show that the operation performed by the estimator unit gives an asymptotically efficient estimate of the angular coordinate for large values of $\Delta t \Delta f_c$, where Δf_c — band of fluctuations of the signal; Δt — time of observation. Variance of the efficient estimate, as it is known, is expressed by formula

-607-

$$\sigma_{\bullet\bullet}^{3} = \frac{1}{\left(\frac{\partial \ln L(\mathbf{a})}{\partial \mathbf{a}}\right)^{3}},$$
(10.16.7)

where $L(\alpha)$ - likelihood function.

Substituting expression (10.3.8) in (10.16.7), we obtain

$$a_{j_0}^2 = \left\{ -\frac{1}{2} \int_0^M \int_0^M \sum_{l,l=1}^n W'_{ij}(t_1, t_2) R'_{ij}(t_1, t_2) dt_l dt_2 \right\}^{-1}$$
(10.16.8)

(similar transformations we already made in § 10.3 during calculation of the minimum possible equivalent spectral density, which is proportional to the variance of the efficient estimate). We substitute expressions (10.3.7), (10.3.14) and (10.3.18) in (10.16.8); then

$$\mathbf{a}_{\mathbf{s}\mathbf{0}}^{2} = \left[\frac{\Delta t h_{\Sigma}^{2} \cdot 2.5}{\Delta \mathbf{y}^{2} \cdot 2\pi} \int_{-\infty}^{\infty} \frac{S_{0}^{2}(\omega)}{1 + h_{\Sigma} S_{\mathbf{s}}(\omega)} d\omega\right]^{-1}, \qquad (10.16.9)$$

where $h_{\Sigma} = P_{C \Sigma}/2N_0 \Delta f_c$ - ratio of signal power to the power of noise in the band of signal fluctuations;

 $S_{O}(\omega)$ - spectrum of signal fluctuations.

Thus, variance of the efficient estimate decreases with growth of Δt (inversely proportional dependence). It is essential to note that the found variance does not depend on angle α , i.e., accuracy of measurement of the angular coordinate does not depend on the position of the target in the considered sector.

We shall not investigate expression (10.16.9), since with an accuracy of proportionality factors it coincides with expression (10.6.5), studied in detail in $\frac{1}{2}$ 10.5. We indicate only the order of error; for $h_{\Sigma} = 10$ and $\Delta t \Delta f_{C} = 10$ we obtain $\sigma_{aff} = 0.064 \Delta \phi$.

We shall prove now the asymptotic efficiency of the estimate issued by the estimator unit. As it is known, an estimate of a parameter is close to efficient if

$$\left(\frac{\partial^{\mathbf{a}}L(\mathbf{a})}{\partial \mathbf{a}^{\mathbf{a}}} - \frac{\partial^{\mathbf{a}}L(\mathbf{a})}{\partial \mathbf{a}^{\mathbf{a}}}\right)^{\mathbf{a}} < -\frac{\partial^{\mathbf{a}}L(\mathbf{a})}{\partial \mathbf{a}^{\mathbf{a}}} = \frac{1}{\mathbf{e}_{\mathbf{a}\mathbf{a}}^{2}}, \qquad (10.16.10)$$

i.e., if the mean square value of deviations of $\partial^2 L(\alpha)/\partial \alpha^2$ from its mean value is negligible (see [8, 13]). As it is easy to show, for sets of normal signals of type (10.3.7) we have

$$\left(\frac{\overline{\partial^{2}L(a)}}{\partial a^{2}} - \frac{\overline{\partial^{2}L(a)}}{\partial a^{2}}\right)^{2} = \frac{1}{2} \sum_{i,j,k,l=1}^{2} \int_{0}^{\Delta t} \int_{0}^{\Delta t} \int_{0}^{\Delta t} \int_{0}^{\Delta t} W''_{ij}(t_{1}, t_{2}) \times \langle W''_{kl}(t_{3}, t_{4}) R_{ik}(t_{1}, t_{3}, \alpha) R_{jl}(t_{3}, t_{4}) R_{ik}(t_{1}, t_{3}, \alpha) R_{jl}(t_{3}, t_{4}) R_{ik}(t_{3}, t_{4}) R_{ik}(t_{1}, t_{3}, \alpha) R_{jl}(t_{3}, t_{4}) R_{ik}(t_{3}, t_{4}) R_{ik}(t_{3}, t_{4}) R_{ik}(t_{3}, t_{4}) R_{ik}(t_{3}, t_{4}) R_{ik}(t_{3}, t_{3}) R_{jl}(t_{3}, t_{4}) R_{ik}(t_{3}, t_{4}) R$$

Substituting in (10.16.11) expressions (10.3.7), (10.3.14) and (10.3.18) [instead of $v(t_1, t_2)$] and using approximation (10.16.2), we obtain

$$\left(\frac{\overline{\partial^2 L(\mathbf{a})} - \overline{\partial^2 L(\mathbf{a})}}{\partial a^2}\right)^3 = \frac{h_{\Sigma}^2 \sqrt{\Delta f_{\circ}}}{2} \sqrt{\frac{1}{\pi} \int_{-\infty}^{\infty} \frac{S_{\bullet}(\omega)^4}{(1 + h_{\Sigma} S_{\bullet}(\omega))^2} d^{*y}}.$$
(10.16.12)

From which we have

$$\frac{\sqrt{\left(\frac{\partial^{2}L(\alpha)}{\partial \alpha^{1}}-\frac{\partial^{2}\overline{L}(\alpha)}{\partial \alpha^{2}}\right)^{2}}}{-\frac{\partial^{2}L(\alpha)}{\partial \alpha^{2}}} = \frac{2}{\sqrt{\Delta f_{c}}} \times \frac{\sqrt{\frac{1}{\pi} \int_{-\infty}^{+\infty} \frac{S_{0}(\omega)^{4}}{[1+h_{2}S_{0}(\omega)]^{2}} d\omega}}{\int_{-\infty}^{+\infty} \frac{S_{0}(\omega)^{4}}{1+h_{2}S_{0}(\omega)} d\omega}.$$
 (10.16.13)

The ratio of integrals has the order $1/\sqrt{\Delta f_c}$ [with a square approximation of $S_0(\omega)$ this ratio is equal to $1/\sqrt{2\pi\Delta f_c}$]. Thus, ratio (10.16.13) has the order $1/\sqrt{\Delta f_c} \Delta t$ and seeks 0 with growth of Δt (as compared to the time of correlation of fluctuations). It follows from this that for large $\Delta t\Delta f_c$ the maximum likelihood estimate becomes efficient, and operation (10.16.6) ensures measurement with the minimum possible error, given by formula (10.16.9).

Error of the goniometer after smoothing can easily be found for any linear smoothing circuits. Actually, by virtue of the asymptiotic efficiency of the estimate issued by the estimator unit the output signal of this unit will be equal to the sum of the true value of the angular coordinate and white noise with spectral density $\sigma_{\partial \Phi}^2$ At, where $\Delta t = duration$ of the section of the signal processed in the estimator unit. Quantity $\sigma_{\partial \Phi}^2 \Delta t$, as can be seen from (10.16.9), does not depend on Δt and coincides with S_{OHP} , met during the study of tracking goniometers. From this it follows that tracking and nontracking goniometers with ideal construction of radio channels and identical structure of smoothing circuits (in the case of a tracking meter we have in mind the structure of the closed-loop system) will give identical accuracy of measurement.

In [46, 47] considerable attention was also paid to the case of a slowly fluctuating signal. For this case synthesis of meters, in general, was not carried out. However, it is possible to limit ourselves to synthesis of the unit of primary

-609-

processing, which gives the stimate of an angular coordinate for a small time, when this coordinate cannot change.

For such synthesis it is necessary in expression (10.16.5), giving the maximum likelihood estimate, to substitute function $v(t_1, t_2)$, calculated with various assumptions about the character of signal fluctuation. We assume that during the time of observation Δt the amplitude and phase of the signal cannot change (due to fluctuations of the target). As it was shown in Chapter VI, equation (10.16.3) has in this case the solution

$$v(l_{s}, l_{s}) = \text{const} = -\frac{1}{\frac{P_{es}At}{1 + \frac{2N_{s}}{2N_{s}}}}$$
 (10.16.11)

Here (10.16.5) takes form

$$\mathbf{\hat{e}} = \frac{\mathbf{A} \cdot \mathbf{v}}{\mathbf{m}} \operatorname{arcctg} \left\{ \begin{bmatrix} \left| \int_{0}^{t} y_{1}(t) u(t) e^{i \omega_{0} t} dt \right|^{2} - \int_{0}^{t} y_{1}(t) u(t) e^{i \omega_{0} t} dt \end{bmatrix}^{2} \cdot \left[2\operatorname{Re} \int_{0}^{t} y_{1}(t) u(t) e^{i \omega_{0} t} dt \times \int_{0}^{t} y_{1}(t) u^{*}(t) e^{-i \omega_{0} t} dt \right]^{-1} \right\}.$$

$$(10.16.15)$$

The block diagram of a unit realizing operation (10.16.15) is depicted in Fig. 10.43. It contains a filter with pulse employee

$$h(t) = \begin{cases} 1, \ 0 < t < \Delta t, \\ 0, \ t > \Delta t. \end{cases}$$

This circuit differs from the preceding circuit in that basic accumulation in the estimator unit occurs immediately after heterodyning. Accuracy of the circuit of



Fig. 10.43. Optimum circuit of the radio channel of a nontracking meter with IAC with slow fluctuations of the signal; 1) a rilter whose low-frequency equivalent is an integrator with clearing; 2) square-law detectors; 3) dividers; 4) nonlinear unit. Fig. 10.43 is very difficult to calculate. It is possible to calculate, of course, variance of the efficient estimate for the considered case. Substituting in (10.16.8) expressions (10.3.7), (10.3.14) and (10.16.14), and also using approximation (10.16.2) for directional patterns of antennas, we can obtain a very simple expression for variance of the efficient estimate of the angular coordinate α :

$$q_{ab}^{2} = 0,2\Delta \varphi^{a} \left(\frac{1}{q} + \frac{1}{q^{2}} \right),$$
 (10.16.16)

where

$$q = \frac{P_{cz}\Delta t}{2N_{\bullet}}$$

is the ratio of the total energy of the received signal (during time Δt) to the spectral density of noise.

However it is easy to prove that the maximum likelihood estimate in this case differs from the efficient one: ratio (10.16.3) in the considered case turns out to be constant and equal to 0.36. Consequently, it is only possible to state that variance of error of the circuit of Fig. 10.43 cannot be less than σ^2_{ab} .

In order to find the accuracy of this circuit, it is necessary to directly to find the mathematical expectation and variance of quantity $\hat{\alpha}$, given by formula (10.16.15). We will not produce these calculations, but note only that as $q \rightarrow \infty$ estimate $\hat{\alpha} \rightarrow \alpha$, and with growth of q it decreases proportionally to 1/q.

10,16.2. Optimum Circuit of a Nontracking Goniometer with IFC

We now shall consider the phase method. With this method there is received a pair of signals of form (10.3.3), where $\Phi_1(t, \alpha)$ is determined by formula (10.2.6). Signal amplitudes are not modulated during reception, i.e., $U_{a,i}(t, \alpha) = 1$. Correlation functions of these signals are given by expression (10.3.7), and their functional of the distribution of probabilities in interval (0, At) has the form (10.3.8), and we need only find the matrix of inverse correlation functions. Solving equations (10.3.10) by methods already described, we can find that in the considered case the inverse correlation functions have form (10.3.14), where $v(t_1, t_2)$ satisfies equation (10.46.3). In contrast to the preceding case this is valid without any approximations.

The maximum likelihood equation, as before, is determined by expression (10.16.4). Substituting (10.3.14) in it, taking into account (16.3.2) and (10.2.8), we obtain likelihood equation

$$\frac{\partial}{\partial \alpha} \operatorname{Re} e^{-i\frac{2\pi d}{\lambda}\sin\alpha} \int_{0}^{\Delta t} \int_{0}^{\Delta t} v(t_{1}, t_{2}) y_{1}(t_{1}) y_{2}(t_{2}) \times u(t_{1}) u^{*}(t_{2}) e^{i\omega_{0}(t_{1}-t_{2})} dt_{1} dt_{2} = 0$$

From this it is already easy to obtain an explicit solution of the likelihood equation:

$$\sin \alpha = \frac{\lambda}{2\pi d} \operatorname{arcctg} \left\{ \operatorname{Re} \int_{0}^{\Delta t} \int_{0}^{\Delta t} v(t_{1}, t_{s}) y_{1}(t_{1}) y_{s}(t_{2}) \times \right. \\ \left. \times u(t_{s}) u^{\ast}(t_{s}) e^{i \alpha_{0}(t_{1}-t_{s})} dt_{s} dt_{s} \left[\operatorname{Im} \int_{0}^{\Delta t} \int_{0}^{\Delta t} v(t_{1}, t_{s}) y_{1}(t_{1}) y_{s}(t_{s}) \times \right. \\ \left. \times u(t_{s}) u^{\ast}(t_{s}) e^{i \alpha_{0}(t_{s}-t_{s})} dt_{s} dt_{s} dt_{s} \right]^{-1} \right\}.$$

(10.16.17)

The sector of uniqueness of measurement of an angle with the phase method, as shown in § 10.2, is $|\alpha| < \arcsin \lambda/2d$. When d >> λ the sector of uniqueness is small and in formula (10.16.17) sin α can be replaced by α .

Let us consider in greater detail operation (10.16.17). Here it is necessary to substitute function $v(t_1, t_2)$, found from equation (10.16.3). Let us consider the case of rapid fluctuations. Here, using expression (10.3.18) for $v(t_1, t_2)$, we can reduce the optimum operation (10.16.17) to the form

$$\sin \alpha = \frac{\lambda}{2\pi d} \operatorname{arcctg} \left\{ \int_{0}^{M} d\tau \operatorname{Re} \int_{0}^{M} h_{100\tau} (\tau - s) y_{1}(s) \times \left(s \right) e^{i m_{0} s} ds \right\} \times \left(s \right) e^{i m_{0} s} ds \int_{0}^{M} h_{100\tau} (\tau - s) y_{1}(s) u(s) e^{i m_{0} s} ds \times \left[\int_{0}^{M} d\tau \operatorname{Im} \int_{0}^{M} h_{100\tau} (\tau - s) y_{1}(s) u(s) e^{i m_{0} s} ds \int_{0}^{M} h_{100\tau} (\tau - s) \times \left(y_{1}(s) u^{*}(s) e^{-i m_{0} s} ds \right]^{-1} \right\},$$

$$(10.16.18)$$

where the pulse response of the optimum filter $h_{1-OH^{4}}(t)$ is determined by relationship (10.3.23) [its frequency response has the form (10.3.25)]. The optimum circuit of a nontracking goniometer with IAC is represented in Fig. 10.44. Here signals from the output of the antennas first are heterodyned, where the output of one of the antennas is connected to two channels (we call them for briefness the 2nd and 3rd), to which there are fed heterodyne signals, shifted in , ase $\pi/2$. All signals after heterodyning are filtered by optimum filters with gain-frequency response (10.3.25). After filtration the signals are multiplied, integrated, divided and red to an inertialess nonlinear element with response $y = \operatorname{arc} \operatorname{ctg} x$.

The output signal of the nonlinear element after multiplication by $\lambda/2\pi d$ enters

-612-

the smoothing circuits. The structure of the smoothing circuits remains as before: it is determined only by statistics of change of the angular coordinate.



Fig. 10.44. Optimum direct of a nontracking meter with IFC for fast signal fluctuations: 1) optimum fictors with frequency response $H_{1/10}$ (10); 2) integrators; 3) divider; 4) nonlinear element; 5) smoothing circuits,

Let us consider the accuracy of the phase method during fact fluctuations. Calculating by formula (10, 10, 8) the variance of the efficient estimate, it is easy to find that in the considered ease it is equal to

$$\mathbf{\sigma}_{s,\phi}^{\dagger} = \mathbf{\omega} \left[\frac{\Delta t h_{s}^{\dagger}}{\left(\frac{\omega d}{\lambda}\right)^{-1}} \cdot \frac{1}{2\pi} \int_{-\infty}^{\infty} \frac{S_{\phi}(\omega)^{\dagger}}{1 + h_{s} S_{\phi}(\omega)} d\omega \right]^{-1}$$

$$= \left(1^{t_{s}}, 5^{t_{s}}, 1^{-s}\right)$$

and differs from (10,10.9) only by a constant factor,

Repeating to conclusions of the preceding paragraph, it is possible to prove that the maximum likelihood estimate found by us, expressed by formula (10,10,1/), is asymptotically efficient for large 4t (in comparison with the time of correlation of signal fluctuations). Thus, operation (10,10,17) will ensure a minimum possible out r with variance (10,16,18). Expression (10,10,18) for σ_{pq}^2 with accuracy of a propertionality factor coincides with expression (10,10,18), which was investigated in details in § 10,0. Therefore, we shall not analyze σ_{pq}^2 . We shall only show the order of error; for $h_2 = 10$, $\Delta to f_{q_1} = 4 + error \sigma_{pq_2} = 4 + 0.01 \lambda/\pi d$.

Accuracy of the synthesized gondemeter in the smoothing direction is the name as in the proceeding case. The signal proceeding to the smoothing direction is equal to the sum of the true value of the angular coordinate and write rates with spectral density $\sigma_{\rm eff}^2$ At. This quantity coincides with $\sigma_{\rm eff}$, met during the study of a continueter with 1PC. Thus, with ideal construction of ratio chase.

.......



Pig. 10.45. The systems clouds of the ratio channel of a nontracking gentemeter with TPC and slow fluctuations of the signals 1) filters where low-frequency sources interactors with cloring P) dividers 3) monitowar clement.

rtradure of the emoothing circuits accuracy of the nontracking and tracking IPC goniometers is identical.

Let us consider now the user of slow fluctuations, when $\Delta t \Delta t_G \ll \ll \ll \ll 1$. In this case, obviously, $v(t_1, t_2)$ will again be expressed by formula (10,10,14), and the operation of singling out the maximum likelihood angle with the phase method will take the form

$$\begin{split} & \sin \mathbf{a} = \frac{\lambda}{3\pi d} \operatorname{srcctg} \left\{ \operatorname{Re}_{3}^{M} y_{i}\left(l\right) u\left(l\right) e^{i\mathbf{a} \cdot \mathbf{d}} \, dt \right\} \\ & \times \int_{3}^{M} y_{i}\left(l\right) u^{a}\left(l\right) e^{-i\mathbf{a} \cdot \mathbf{d}} \, dt \left[\operatorname{Im}_{3}^{M} y_{i}\left(l\right) u\left(l\right) e^{i\mathbf{a} \cdot \mathbf{d}} \, dt \right] \\ & \times \int_{3}^{M} y_{i}\left(l\right) u^{a}\left(l\right) u^{a}\left(l\right) e^{-i\mathbf{a} \cdot \mathbf{d}} \, dt \right]^{-1} \right\}. \end{split}$$

(10,16,20)

The block diagram of a double restling operation (10.16.20) can be presented in the form depicted in fig. 19.55. The figure in this circuit has response envelope

 $h(t) \mapsto \begin{cases} 1, & 0 < t < \Delta t, \\ 0, & t > \Delta t. \end{cases}$

With respect to accuracy of the found circuit it is possible to say the following. As also in the proceeding case, it is easy to show that the estimate issued by this circuit is not estimate. Variance of the efficient estimate can be calculated and will determine an estimate from beneath of variance of error of the circuit. By simple calculation by formula (10.10.8), already known, it is easy to find that

$$\bullet_{\bullet\bullet}^{*} \simeq 0.49 \left(\frac{\lambda}{nd}\right) \left(\frac{1}{q} + \frac{1}{q^{*}}\right), \qquad (10.16.21)$$

where q is given by expression (10.10.11).

True securacy of the considered circuit should be calculated directly. The magnitude of true accorney we do not consider, since its determination is very

-014.

complicated and requires special investigation. We shall discuss this in greater detail in the concluding section of the present chapter.

§ 10.17. Influence of Interferences on Coherent Goniometers

In practice a goniometer is often subjected to the influence not only of a fluctuating signal and internal noises, but also to active and passive interferences, distorting the process of measurement or even completely disrupting it [70]. The number of interferences which it is possible to create for radar goniometers is, very great. However, to study the influence of each form of interference in the same detail as the influence of fluctuations of the signal and natural noises is hardly useful due to the large number of particular cases which can be encountered. Therefore, we shall study the influence of interferences basically in approximation, limiting ourselves to giving the simplest formulas.

First of all we shall classify possible interferences to goniometer devices, so that we see in what order we are to study them. We consider, first, active interferences. They can be sent from the target whose angular coordinates are being measured, and also from another target, located in an essentially different direction.

In the first case interference acts on the basic lobe of the directional pattern, and in the second — on side lobes. The influence of both types of interferences sharply differs. The effect of interference acting on the basic lobe, in turn, essentially depends on the form of interference, and also on the method of direction finding utilized in the considered goniometer. Interferences with amplitude modulation will strongly affect goniometers with pattern scanning and weakly affect other goniometers. Interference with great power; suppressing the useful signals through the AGC system, will have a different affect from interference with low power.

Thus, the number of possible cases here is very great. During the study of interferences we will widely use the fact that the influence of random interferences on a coherent radar, due to narrow-bandedness of its components, is equivalent to the influence of white noise (Chapter VII).

10.17.1. Influence of Broad-Band Interferences on Side Lobes

The influence of active interferences on side lobes of the directional pattern of radar geniemeters is a very likely case. At the same time, the level of side lobes of the pattern in most cases is such that we cannot provide sufficient suppression of such interference.

-615-

The influence of random interference acting on side lobes can be estimated approximately rather simply. If the interference is broad-band, then it is equivalent to white noise, the magnitude of which does not depend on angular coordinates of the target, i.e., it is equivalent to the internal noise of the receiver. Certainly, in goniometers with pattern scanning such interference obtains amplitude modulation; however the side lobes have a more or less constant level, so that such modulation can, in most cases, be ignored. Formulas for accuracy of a goniometer under the influence of such interferences coincide with formulas for accuracy of goniometers in normal conditions (without interferences), in which the spectral density of noises $h_{\rm C}$ must be replaced by $N_{\rm O} + N_{\rm H}$, where $N_{\rm H}$ — spectral density of the equivalent noise replacing interference. Spectral density $N_{\rm H}$ should be calculated in the following way. If the influence is a broad-band noise interference with bandwidth $\Delta f_{\rm H}$,

$$2N_{\rm H} = \frac{P_{\rm n}G_{\rm n\,H}}{(4\pi)^3} \frac{\lambda^3 x}{\Delta f_{\rm m} d_{\rm n}^2},\qquad(10.17.1)$$

where P_n - mean power of jamming transmitter;

 $G_{\pi\pi}$ - gain of the antenna of the jamming transmitter;

 λ - wavelength of the considered radar;

n - level of side lobes;

 d_{π} - distance to source of interferences.

If the influence is a pulse chaotic interference, then, using formula (7.14.9) derived in Chapter VII, we obtain

$$2N_{\rm H} = \frac{P_{r_0}G_{\rm HR}\lambda^2 x^{\rm vr_{\rm H}}}{(4\pi)^2 d_{\rm H}^2} \frac{1}{\sqrt{1+4f_{\rm HRO}^2 r_{\rm H}^2}},$$
(10.17.2)

where P_{ere} - pulse power of jamming transmitter;

 ν - mean frequency of repetition of pulses of interference;

 τ_{π} - duration of interference pulse;

 $\Delta f_{M}^{2} = \hat{m} - effective$ width of the spectrum of signal modulation.

Thus, the influence of broad-band interference on the side lobe leads simply to equivalent increase of natural noises, i.e., to equivalent decrease of the signalto-noise ratio.

10.17.2. Influence of Broad-Band Interferences From the Target

Broad-band interference from the target (noise or pulse chaotic interference with a high frequency of repetition) is also a very common type of interferences. However, this kind of interference is intended primarily for suppression of such functions of radars as measurement of range or speed, and for goniometers does not present serious danger. In fact, this interference itself carries information about angular coordinates, and the goniometer can measure angular coordinates of the target from the interference.

Let us consider in greater detail the influence of such interference on goniometers. First of all we note that, in view of the considerable excess of interference power over the signal, both the useful signal, and also natural noises will be suppressed, by the AGC system and measurement will be conducted exclusively of interference. Considering the broad-bandedness of the interference, it is possible to say that accuracy of measurement of angles will be determined by the same formulas with which we determined accuracy of goniometers in normal conditions, considering $h = \infty$ (internal noises are absent) and ratio $\Delta f_{\phi} /\Delta f_{c} \ll 1$ (since the width of the interference band considerably exceeds the width of the passbands of filters in the goniometer). In both circuits with scanning (Figs. 10.9 and 10.10) there will exist fluctuating error with equivalent spectral density, equal, respectively, to

$$S_{0KB} = \frac{1}{8\mu_{a}^{2}\Delta/\phi},$$
 (10.17.3)

$$S_{\bullet KB} = -\frac{1}{4\mu_a^2 \Delta f_{\bullet}}.$$
 (10.17.4)

These formulas are obtained from formulas (10.5.17) and (10.5.22). From this it is clear that tracking of a target by interference radiated from it always occurs with limited accuracy. If, for instance, $\mu_{a} = 0.2$ 1/deg and $4f_{d} = 100$ cps, then for Ω_{SKB} we obtain the corresponding values

$S_{\text{PMB}} \approx 0.03 \text{ deg}^2/\text{cps}$, $S_{\text{DMB}} \approx 0.06 \text{ deg}^2/\text{cps}$.

In many cases such magnitudes of S_{OKB} are already impermissible; therefore the considered interference for the method of pattern scanning of diagram will be rather dangerous, especially for the circuit of Fig. 10.10.

We note that for the circuit of Fig. 10.9, for which the frequency of scanning can be selected as high as one wishes, error of tracking by interference can be made minute, if the frequency of scanning considerably exceeds the width of the spectrum of interference. Fluctuating error here remains only due to the parametric component (if there are dynamic errors of tracking). The spectral density of parametric fluctuations, as it is easy to find from formulas (10.5, 24) and (11.13.2), will be equal to

-617-

$$S_{\rm RAP} = \frac{1}{2\Delta/\phi n^3}$$
(10.17.5)

For circuits with multiplication of signals of the compensation method and, correspondingly, of IAC during tracking of a target by broad-band interference radiated from it we have $\Omega_{\rm DKB} = 0$ [see (10.7.5) as $h \rightarrow \infty$]. This will take place even in the case of nonidentity of channels. However, as it is easy to see from (10.7.2), nonidentity leads to decrease of the slope of the discrimination characteristic $K_{\rm II}$, which will lead to worsening of dyamic properties of the tracking system during tracking of interference from the target.

In the circuit with subtraction of signals for scanning with compensation the considered interference will lead to the worst results. In the case of nonidentity of channels S_{3KS} will differ already from zero. If for simplicity we consider different only the gain of the channels, from expression (10.7.8) for h $\rightarrow \sigma$ and 44 $_{\rm C}$ >> $\Delta f_{\rm c}$ it is easy to obtain

$$S_{0\times 6} = \frac{1}{8\mu_{a}^{2}\Delta/6} - \left(\frac{1-k^{2}}{1+k^{2}}\right)^{6}, \qquad (10, 17, 6)$$

where k - ratio of the squares of gain factors of guannels of the radio channel.

For the IAC circuit with subtraction of signals for nonidentical channels S_{SKB} will also be expressed by formula (10.17.6).

Furthermore, here there will exist systematic error, absent in the circuit of the method of scanning with compensation

$$\Delta = \frac{1}{4\mu_0} \frac{1 - h^2}{1 + h^2}.$$
 (10.17.7)

Thus, tracking of interference from a target will have error only in the case of nonidentity of channels. With identity of them results will be precisely the same as for circuits with multiplication of signals, i.e., $S_{BKB} = 0$ and $\Delta = 0$. However, in both circuits here there will exist fluctuating error due to parametric fluctuations. The spectral density of parametric fluctuations will as before be expressed by formula (10.3.15). As already noted, this error is insignificant.

Absolutely analogously one can prove that during tracking of interference from a target using the method of phase conter scenning there will exist very great fluctuating error with equivalent spectral density

$$\mathbf{S}_{\mathbf{0}\mathbf{H}\mathbf{0}} = \frac{1}{2\left(\frac{\mathbf{n}d}{\mathbf{\lambda}}\right)^{4} \Delta f_{\mathbf{0}}} \left(\frac{2\Delta f_{\mathbf{0}}}{\mathbf{Q}}\right)^{4}, \qquad (10.17.8)$$

-618-

where Af ... - width of the spectrum of interference.

When using a goniometer with IPC (Fig. 10.23) we always will have $S_{BHE} = 0$. Nonidentity of channels will lead only to decrease of $K_{\rm H}$ and worsening of dynamic properties of the tracking system.

When using IPC by the circuit of Fig. 10.24 interference from the target will have considerably stronger influence. For an example we shall consider the gain-frequency responses to differ by a constant $\Delta \phi$. Then, during tracking of interference from a target there will exist fluctuating error with equivalent spectral density

$$S_{\bullet H \bullet} = \frac{1}{4 \left(\frac{\pi d}{\lambda}\right)^3} \operatorname{tg}^* \Delta \varphi \left(\frac{1}{\Delta f_{\psi}} + \frac{1}{2\pi \Delta f_{\varphi}^2} \int_{-\infty}^{\infty} |H(i\omega)|^4 d\omega \right) \approx \frac{3}{8 \left(\frac{\pi d}{\lambda}\right)^3 \Delta f_{\varphi}} \operatorname{tg}^* \Delta \varphi.$$
(10.17.9)

[obtained from formula (10.9.3) as h 🕶 @].

Thus, fluctuating error here increases very fast with growth of nonidentity of phase-frequency responses. Systematic error existing in this circuit is equal to

$$\Delta = -\frac{1}{2\left(\frac{\pi d}{\lambda}\right)} \operatorname{tg} \Delta \varphi. \tag{10.17.10}$$

With identical channels this circuit behaves just as the preceding one.

Consequently tracking of broad-band random interference from a target when using methods of scanning the directional pattern and phase center occurs fluctuating error differing from zero. This error, as rough calculation shows, is considerable. In methods of scanning with compensation, IAC and IFC with identical channels $S_{\mu KB} \approx 0$, and fluctuating error (very insignificant) exists only due to parametric fluctuations. Fluctuating error appears in circuits without formation of sum and difference signals when there are nonidentical channels. In circuits with formation of sum and difference signals nonidentity of channel leads only to decrease of $K_{\rm A}$ and, consequently, to worsening of dynamic properties of tracking systems.

10.17.3. Influence of Active Interferences with L-F Modulation

As already noted, against goniometers using the method of pattern scanning there may be used interference (noise, pulse, return) with low-frequency amplitude modulation. With coincidence of frequencies of modulation of interference with the frequency of scanning this interference will, obviously, be very effective. For

-619-

goniometers using the method of phase center scanning, interference with low-frequency phase modulation will lead, obviously, to analogous results. We now pass to consideration of the influence of this kind of interference on goniometers.

Let us study the effect of interference from a target with special low-frequency amplitude modulation on goniometers using the method of pattern scanning, and also the compensation method. The fluctuating signal considered in the preceding paragraphs is not equivalent to such interference, since it contains definite phase modulation. We will produce approximate calculation of the influence of the considered form of interference, seeking, basically, qualitative conclusions. We assume that interference has the form

 $u_{\pi}[1] + \xi(t)] \cos(\omega_{\bullet} t + \psi), \qquad (10.17.11)$

where $\xi(t)$ - modulating random process with a zero mean and small variance.

Let us consider first the method of pattern scanning in conditions of application of the circuit of Fig. 10.9. For simplicity we assume that the circuit is intended for work on an unmodulated signal (there is no inversion of phase modulation and corresponding transformation of amplitude modulation of the received signal), and also that scanning is uniform and conical. The received signal, after heterodyning, will obviously have form

$$u_{n}[1 + \xi(t)](1 + \mu_{a} \alpha \cos \Omega t + ...) \cos (\omega_{np} t + \psi). \qquad (10.17.12)$$

Noises and the reflected signal we disregaried, considering interference sufficiently powerful. Considering the smallness of variance and the narrowness of the spectrum of process $\xi(t)$, for the output signal of the circuit we obtain

$$z(t) = \frac{\mu_{\pi}^{2}}{2} \left[\frac{\mu_{\pi}a}{2} + \int_{-\infty}^{t} h(t-\tau)\xi(\tau) \cos \Omega \tau d\tau \right]. \qquad (10.17.13)$$

Hence equivalent spectral density of fluctuating error is equal to

$$S_{0KB} = \frac{2S_B(Q)}{\mu_a^2} \approx \frac{\sigma_n^2}{\mu_a^2 \Delta f_{\pi}}, \qquad (10.17.14)$$

where $S_{_{\rm H}}$ (\Omega) - spectral density of $\xi(t)$ at frequency $\Omega;$

 σ_{π}^{2} - variance of $\xi(t)$;

 Δf_{π} -- width of spectrum $\xi(t)$ (one-way).

Analogously it is easy to prove that for the circuit of Fig. 10.10 equivalent spectral density during tracking of the considered interference will be precisely the same. Analysis of expression (10.3.25) shows that the more exactly the modulation of interference is tuned to the frequency of scanning (i.e., Δf_{tr} decreases), the

greater the error of tracking.

Now we consider a circuit of scanning with compensation. In the circuit of Fig. 10.15, as it is easy to show, during tracking of the considered interference we will have $S_{allB} = 0$, i.e., this interference is well compensated. Nonidentity of channels will only lead to decrease of the slope of the discrimination characteristic.

In the circuit of Fig. 10.16 with identical channels we also will have $S_{3KB} = 0$. However, with nonidentical channels there will appear fluctuating error with S_{3KB} , equal to

$$S_{\text{DRB}} = \frac{2S_{\text{R}}(\mathbf{Q})}{\mu_{\text{A}}^2} \left(\frac{1-k^2}{1+k^2}\right)^2, \qquad (10.17.15)$$

where k - the ratio of squares of the gain factors in the channels.

With identical channels k = 1 and $S_{OKB} = 0$. However, even with identical channels in both the considered circuits fluctuating error due to parametric fluctuations will remain. Simple calculation shows that for both circuits, assuming identity of their channels

$$S_{\pi a p} = \frac{S_{\alpha}(0)}{n^{2}} = \frac{\sigma_{\pi}^{2}}{n^{2} \Delta f_{\pi}}, \qquad (10, 17.16)$$

where all designations are the same as before.

If the spectrum of modulation of interference does not contain components at zero frequency, $S_{\Pi B_{I}} = 0$, and guidance from such interference is accomplished, in general, without fluctuating errors.

Thus, interference from a target with amplitude modulation is very dangerous for the method of pattern scanning. For the method of scanning with compensation it is no longer dangerous. Fluctuating error will exist here only for the circuit without formation of sum and difference signals in the case of nonidentity of channels of the circuit. In the remaining cases $S_{OKB} = 0$, and fluctuating error exists only due to parametric fluctuations, if only the spectrum of interference modulation contains components at zero frequency. This error has an insignificant relative magnitude, and for $S_{OKB} \neq 0$ we can disregard it.

Absolutely analogously to the preceding we can consider the influence of interference with low-frequency phase modulation on goniometers using the method of phase center scanning. Recording interference in the form $u_{\Pi} \cos [\omega_{0} t + \xi(t)]$, where $\xi(t) = random$ process with a zero mean and small variance, it is possible to easily show that the equivalent spectral density of fluctuating error of tracking of such interference is equal to

$$S_{\text{BRB}} \approx \frac{\sigma_{\text{B}}^2}{\left(\frac{\pi d}{\lambda}\right)^{\text{s}} \cdot \Delta f_{\text{B}}},$$

「「「「「「「」」」」「「「」」」」」」」」」

where σ_{Π}^2 - variance of $\xi(t)$; Δt_{Π} - width of spectrum $\xi(t)$.

10.17.4. Influence of Powerful Intermittent Interferences from Target

Let us consider the influence of certain types or interferences from the target, the influence of which reduces to artificial disturbance of the process of reception of the radar signal (for instance, due to excitation of transients in the receiver). We are talking about interference from the target, inasmuch as interference may cause disturbances of this type only with considerable excess of power over the signal, which can occur, basically, during influence of the interference of the main lobe.

This interforence can be return, noise, pulse chaotic, etc. Specifics connected with interruptions do not vary from this. Earation of intervals of the presence of interference and the frequency of their reposition are selected such that there does not exist a steady-state regime of work of the receiver either with respect to interference or to the signal, and the receiver is always, as it were, in a prolonged transient regime. Such an effect appears during feed of intermittent interference of high power to input of a receiver with 100. Lot us consider this process in greater detail. As already noted, in coherent correcte the main i-f amplifier, covered by the AGC system, has a sufficiently wide passbare, so that inertia of the amplifier gevored by it. As was shown in Chapter II, with cortain additional limiting assumptions during feed to an amplifier input of a powerful interference is immediately becomes finites and emerges from the limitations after time.

$$I_{BHR} \approx T_{\phi} \frac{K_{\phi} u_{\mu} - u_{nrp}}{u_{\mu} \delta K_{1} (u_{orp} - E_{s})}, \qquad (10.17.17)$$

Where:

 $T_{g} = -$ time constant of the filter of automatic gain control; $K_{1} = -$ gain in the Add Loop;

K₀ and b -- parameters of linear approximation of the dependence of pair of the UP on on control voltage of adjustment at a point, corresponding to steady-state regime for a given level of interference (see Chapter 11): u_{OPI} and E_{S} - level of limitation and voltage of delay of automatic gain control; u_{ff} - amplitude of interference at the input of the UPCh.

Here there is est ished such UPCh gain that the useful signal is completely suppressed. After turning off of interference the output signal of the UPCh is equal to zero and will rise to its normal magnitude approximately after time

$$I_{\text{SMNR}} = T_{\phi} \ln \frac{K_{e} u_{a} - B_{e}}{K_{e} u_{e} - B_{e}} \frac{u_{e} b_{i} K_{i}}{1 + u_{a} b K_{i}}, \qquad (10.17.18)$$

मिल्ली विश्वमे के विश्वम न प्रशास के समाह दिया राज्य र

where K_{01} and b_1 - parameters of linear approximation of the dependence of UPCh gain on control voltage at a point corresponding to steady-state regime for a given input level of the useful signal.

Let us assume now that the duration of the time of continuous influence of interference is equal to t_{BUN} , and the duration of time when interference is turned off is equal to t_{BUN} . Here at the UPCh output there will occur a sequence of interference pulses of amplitude u_{ij} , equal to the cutoff level of the UPCh, regardless of the form of the useful signal. It is obvious that in this case goniometers using the method of scanning the directional pattern, and also the method of scanning with compensation or IAC, working on circuits without formation of sum and difference signals will absolutely malfunction. In all these cases the output signal of the goniometer circuits will not contain information about angular coordinates of the target.

In circuits of the method of scanning with compensation and IAC in which there are formed sum and difference signals at the output of the antenna system the matter will be somewhat different. In these circuits of the method of scanning with compendation and IAC in which there are formed sum and $d^{(c)}$ ence signals at the output of the antenna system the matter will be somewhat different. In these circuits the difference signal with exact tracking of the target is equal to zero, and with sufficiently small errors of tracking this signal is close to zero.

Thus, one may assume that under the influence of interference the signal in the difference channel does not fall under limitation. The amplitude of the output signal of the UPCh in the difference channel is proportional to mismatch, but it does not remain constant, since the gain factor of the UPCh in the difference channel drops as a result of work of the AGC system, removing from limitation the signal in the sum channel. It is easy to calculate that the amplitude of the signal in the difference channel will decrease according to the law

-623-

$$u(t) = u_{\rm m} \left[K_{\bullet} - b K_{\rm i} (u_{\rm orp} - E_{\bullet}) \left(1 - e^{-F_{\bullet}} \right) \right]$$
(10.17.19)

(with the AGC circuit depicted in Fig. 2.20). After turning off of interference the output signal of the UPCh to the difference channel disappears. Therefore at the output of the circuit there will appear pulses of duration $t_{\rm RKH}$ and form $\mu_{\rm a}\alpha/2$ u(t)u₀, where u(t) is given by formula (10.17.19), and u₀ - cutoff level of the UPCh in the sum channel. Tracking of a target in this case will continue, however, inasmuch as the time interval between pulses of the error signal, equal to $t_{\rm RKHP}$ is usually great, and the tracking system will pass into a mode of discrete operation.

Similar phenomena will be observed in circuits using phase methods of direction finding. Intermittent interference in these cases also, opviously, will not lead to considerable disturbances of the process of measurement angular mismatch. Actually, que to the influence of intermittent interference at the UPCh output there appear pulses with amplitude equal to the cutoff level of the UPCh; Nowever, the phase of the filling of these pulses is not distorted during passage through UPCh. Inasmuch as useful information here is included in phase, an error signal will be produced. E.g., for IPC the signal of error will be pulses of duration t_{BKT} with amplitude $au_0^2 \pi d/\lambda$, where $u_0 = cutoff$ level of the UPCh (with identical channels).

Thus, the tracking system will continue to work; however, due to the great time interval between pulses of the error signal, equal to $t_{\rm BNNF}$ work of the tracking system will be in discrete mode. Due to the influence of such interferences dynamic properties of the tracking system change.

10.17.5. Influence of Passive Interference

On the basis of results of Chapter I we can affirm that the signal from a passive interference, received by the antenna of a goniometer, after multiplication by the reference signal is equivalent to a stationary normal process with a definite spectral density S_{Π} (ω), depending on the difference of Doppler frequencies of the signal from the target and from the interference. The spectrum of interference is a sequence of expanded spectral lines, removed from each other the frequency of repetition of the signal. The width of each line is determined by irregularity of motion of the dipoles, the wavelength of the radar, the speed of the radar with respect to the cloud of interferences and the width of its airectional pattern. As we already noted in § 7.14, the width of each spectral line is tens of cycles per second with a motionless radar and can reach thousands of cycles per second with a moving radar. Therefore with good approximation in coherent goniometers passive interference is equivalent to white noise with spectral density

$$V_{\pi} = \frac{P_{\sigma}}{2\Delta f_{\sigma}} \frac{\sigma_{\pi}}{\sigma_{\pi}} f(\Delta \omega_{\pi}), \qquad (10.17.20)$$

where σ_{ii} - reflecting surface of the target;

- σ_{π} reflecting surface of the interference in the resolution volume of the radar;
- $\Delta \omega_{\rm p} = {\rm difference}~{\rm of}~{\rm Doppler}~{\rm frequencies}~{\rm of}~{\rm the}~{\rm signal}~{\rm from}~{\rm the}~{\rm target}~{\rm and}~{\rm from}~{\rm interference};$
- $f(\omega)$ form of the spectral line of interference (f(0) = 1).

As soon as passive interference is reduced to equivalent white noise its influence on accuracy of goniometers can easily be accounted for. In goniometers with one antenna (i.e., in goniometers with scanning of the directional pattern or phase center, and also with flat scanning of the pattern) for calculation of accuracy during passive interferences we can directly use the formulas derived in the preceding paragraphs, only the signal-to-noise ratio should be replaced by the signalto-interference ratio

$$h_{\rm m} = \frac{e_{\rm m}}{e_{\rm m}} \frac{1}{f(\Delta \omega_{\rm m})}.$$

(10.17.21)

The magnitude of σ_{II}/σ_{II} is usually considerably less than one. Acceptable signal-to-interference ratios occur only with sufficiently large frequency separations of the signal and interference $\Delta \omega_{II}$, when $f(\Delta \omega_{II}) \ll 1$. With coincidence of the speeds of the target and interference $\Delta \omega_{II} \approx 0$ and $f(\Delta \omega_{II}) \approx 1$; therefore the signal-to-interference ratio in this case will be minute.

In the case of goniometers with two or more antennas the possibility of use during passive interferences of the formulas for accuracy derived for the case of natural noises is not clear. The fact is that the directional patterns receive reflected signals from the same reflectors; therefore interferences at the output of the antennas are correlated. However, it is possible to show that in the practically most interesting cases this correlation does not lead to change of the shown formula. Let us consider, for instance, a goniometer with IAC. Let us designate signals from the passive interference at the output of the antennas by n $_{\Pi}$ $_{1}(t)$ and n $_{\Pi}$ $_{2}(t)$. If we assume the cloud of reflectors uniform in a sufficiently wide angular range, and the directional patterns of antennas identical, interferences n $_{\Pi}$ $_{1}(t)$ are correlated and have identical variances σ_{Π}^{2} .

-625-

If at the output of the antennas there are formed sum and difference signals, interferences in the sum and difference channels will be equal to:

$$n_{\pi+}(t) = n_{\pi_1}(t) + n_{\pi_2}(t), n_{\pi_2}(t) = n_{\pi_1}(t) - n_{\pi_2}(t).$$

It is easy to see that the sum and difference interferences are not correlated. Actually,

$\overline{n_{\mathbf{n}+}(l)n_{\mathbf{n}-}(l)} = \overline{n_{\mathbf{n}_1}(l)^2} - \overline{n_{\mathbf{n}_2}(l)^2} = 0.$

Consequently, accuracy of IAC circuits with formation of sum and differnce signals in the presence of passive interference can be estimated by the formulas derived for the case of natural noises with replacement of the signal-to-noise ratio by the signal-to-interference ratio (10.17.21). A circuit without formation of sum and difference signals with identical channels, as we know, is identical to a circuit with formation of such signals. Consequently, accuracy of circuits without formation of sum and difference signals with passive interferences is expressed by the formulas derived for the case of natural noises.

Everything said, obviously, also pertains to the method of scanning with compensation and to the IPC method.

Thus it is rather easy to approximately allow for the influence of passive interferences: under the influence of passive interference goniometer circuits behave the same way as with natural noise with some new spectral density, which is expressed through parameters of the interference. More exact analysis is complicated, although in principle it is a clear problem. We will not continue our study of these questions.

§ 10.18. Conclusion

In the preceding sections we synthesized and studied in detail optimum radar goniometers. Basic attention was allotted to synthesis and analysis of radio channels of goniometer devices. Namely, here there is manifested to the greatest extent the specific character of the problem of measurement of angles, in many respects differing in initial prerequisites and results of the conducted investigation from that presented in chapters devoted to other radar meters. To a considerably lesser extent this specific character affects smoothing circuits of meters, since the structure of smoothing circuits does not depend on the method of encoding the measured parameter in the radar signal, but is determined, basically, by laws of change of the parameter.

-626-

Synthesized circuits of optimum radio channels of tracking goniometers were found close in basic features to circuits described in the literature, created on the basis of technical and physical intuitions. An exception are the circuits of optimum radio channels of goniometers using the method of pattern scanning or scanning with compensation. For these methods of direction finding synthesized circuits ensure considerable or even total compensation of the harmful influence of amplitude fluctuations of the signal and somewhat differ from known circuits.

the second se

where the second s

Analysis, which was allotted considerable attention, had as it purpose to reveal the criticality of circuits to various deviations from optimality in their structure. We showed, in particular, that imperfectness of heterodyning (imperfectness of inversion of amplitude and phase modulations of the received signal) is always equivalent to some decrease of the signal-to-noise ratio. In circuits for the method of scanning with compensation, IAC and IPC nonidentity of channels leads to very harmful consequences. Here there is established the evident advantage of circuits with formation of sum and difference signals, which turned out to be least critical to nonidentity of channels.

Let us enumerate the most important problems concerning radar goniometers not touched on or insufficiently illuminated in the present chapter and requiring further investigation.

The first problem is the problem of synthesis of an optimum radar goniometer when the method of direction finding is not assigned, and there are assigned, e.g., only the dimensions and geometric shape of the aperture of the antenna array.

In this chapter this problem was solved for antennas of phased grids type. For the case when the aperture of the antenna array is an arbitrary planar region, the problem of synthesis of an optimum goniometer, including processing of the field in the assigned aperture, has not yet been solved. The essence of this problem and difficulties in its formulation were discussed in detail in § 10.13.

Another important problem not yet solved is the problem of synthesis of optimum radar goniometers for the case when the time of correlation of fluctuations of the radar signal is comparable to the time of correlation of angular shifts of the target. This problem is very important, inasmuch as the shown case is often encountered in proctice. However, in the way of solution of this problem there are considerable difficulties of a mathematical character.

We note further a number of problems concerning smoothing circuits of goniometers. Here it is necessary to produce systematic study of a priori statistics of

-627-

angular shifts of the carget and to synthesize optimum smoothing circuits. It is interesting also to investigate the influence of inaccuracies in assignment of a priori statistics of angular shifts on accuracy of the goniometers.

Fically, very far from complete solution are problems connected with investigation of nonlinear regimes of tracking goniometers. Here it would be desirable to obtain more exact expressions for error of tracking with large mismatch and for an average time to breakoff of tracking with real smoothing circuits of tracking goniometers. The enumerated problems precent doubtless practical interest. However, even the problems already solved permit us to handle many questions of radar goniometry, in particular, to correctly select in each case the structure of the goniometer and to sensibly make various technical simplifications, estimating their effect beforehand.

MEASUREMENT OF ANGULAR COORDINATES WITH AN INCOHERENT SIGNAL

ХT

HAP

§ 11.1. Introductory Remarks

In radar practice incoherent pulse radiation is widely used. Let us remember that according to the terminology of Chapter I we called incoherent pulse signals with random initial phases of the high-frequency filling in each pulse.

Use of incoherent signals, as was shown in the preceding chapters, provides, in general, less possibilities both in radar detection of targets and also in measurement of coordinates. However, in many cases the loss due to incoherence is not essential, and generation of such signals is a technically simpler task than generation of coherent pulse signals. Therefore, application of incoherent signals is often fully justified and desirable. In the present chapter we will pursue a systematic study of radar goniometers using incoherent signals, or more simply, of incoherent goniometers.

In distinction from a coherent signal, statistical properties of an incoherent signal reflected from a target are very complicated. The functional of the distribution of probabilities of a fluctuating incoherent signal cannot be calculated in general. In connection with this, we also cannot synthesize the circuit of optimum incoherent goniometers. Therefore, we shall be limited here to an approximate approach to synthesize of optimum circuits, similar to that which we used, for instance, in Chapter VIII.

In contrast to the problem of synthesis, analysis of incoherent circuits can be performed, in general, without any essential limitations. As a result of mananalysis in the present chapter we tind accuracies of different circuits of

-629-
incoherent goniometers, we investigate dependences of these accuracies on different parameters, we compare real accuracy with potential. Consideration is conducted consecutively for all the methods of angular direction finding described in § 10.2.

Thus, in its structure and actual content the present chapter is close to the preceding, which was devoted to coherent goniometers. Further we shall widely use the designations of the preceding chapter.

§ 11.2. Optimum Radio Channel of Incoherent Goniometers

In examining the problem of synthesis of the optimum radio channel of incoherent goniometers we, as in the preceding chapter, start from a generalized method of direction finding, consisting in reception of a signal on several antennas with scanning directional patterns and phase centers. Preserving all the designations introduced in § 10.3, we can record output signals of the antennas of the considered goniometer in the form

 $y_{i}(t) = \sqrt{P_{oi}} U_{ai}(t, a) u_{a}(t) \{a(t) \cos [w_{o}t + \psi(t) + \frac{1}{2} \Phi_{i}(t, a) + \delta(t)] + b(t) \sin [w_{o}t + \psi(t) + \Phi_{i}(t, a) + \frac{1}{2} \Phi_{i}(t)] + \sqrt{N_{oi}} n_{i}(t) = \sqrt{P_{oi}} \operatorname{Re} U_{i}(t, a) E(t) \times (11.2.1)$

where $\delta(t)$ takes constant random and independent values in every period of repetition of the signal, uniformly distributed in interval $(0, 2\pi)$, and $U_{\rm A}(t)$ and $\psi(t)$ - periodic functions with period $T_{\rm r}$, being the intraperiod amplitude and phase modulations of the incoherent signal.

The functional of the distribution of probabilities of an incoherent signal for certain cases (for instance, for a small signal-to-noise ratio) was calculated in Chapters V and VI. A peculiarity of the considered problem is the presence of several dependent signals, for which it is necessary to obtain a joint functional of the probability density. We repeat in brief the reasonings of Chapters V and VI in reference to this more general problem. The conditional functional of distribution of probabilities of signals (11.2.1) with assigned phases $\delta(t)$ has form [see (10.3.8)]:

$$P[y_{1}(\tau), y_{n}(\tau), \dots, y_{n}(\tau), \alpha | \delta(t)] =$$

$$= K \exp\left\{-\frac{1}{2} \int_{0}^{T} \int_{0}^{T} y^{+}(t_{1}) W(t_{1}, t_{n}) y(t_{n}) dt_{1} dt_{n}\right\} =$$

$$= K \exp\left\{-\frac{1}{4} \sum_{h_{1}/m}^{m} \int_{0}^{h_{r}} \int_{0}^{t_{r}} \int_{0}^{t_{r}} \sum_{l_{1}/m}^{n} W_{ij}(t_{1}, t_{n}) y_{i}(t_{n}) \times y_{j}(t_{n}) dt_{1} dt_{n}\right\}, \quad (11, 2, 2)$$

-630-

where α - angular coordinate of the target;

m - number of periods of repetition of the signal in the interval of observation (0, T) (all designations here are the same as in § 10.3).

If in (11.2.2) we substitute for $W_{ij}(t_1, t_2)$ expression (10.3.14) (with replacement of $\psi(t)$ by $\psi(t) + \delta(t)$), functional (11.2.2) can be reduced to form

$$P[y_{i}(l), y_{i}(l), \dots, y_{n}(l), a|\delta(l)] = K' \exp\left\{-\frac{1}{2}\sum_{k, l=1}^{m} v_{kl} \sum_{i, j=1}^{n} \frac{\sqrt{P_{ei}P_{ej}}}{N_{ei}N_{ej}} \times \left\{ \operatorname{Re} U_{i}(kT_{r}, a) U^{*}_{j}(lT_{r}, a) f_{ki} f^{*}_{lj} e^{i(\mathbf{a}_{k} - \mathbf{a}_{i})} \right\},$$
(11.2.3)

where

$$K' := K e^{-\frac{1}{2} \int_{0}^{T} \left(\sum_{i=1}^{n} \frac{y_{i}(i)^{2}}{N_{0i}} \right)^{di}}; \quad o_{hi} = o(kT_{r}, lT_{r});$$

$$f_{hi} = \int_{(h-1)T_{r}}^{hT_{r}} \mu(t) y_{i}(t) e^{i\omega_{0}t} dt, \quad (11.7.4)$$

and b_k , k = 1, 2, ..., m are values of phase b(t) in different periods of repetition of the signal.

For production of an unconditional functional of distribution of the considered signal it is necessary in (11.2.3) to average phases δ_1 , δ_2 , ..., δ_m . This averaging in general is hard to produce. However, for instance, with a sufficiently small signal-to-poise ratio it is possible to expand functional (11.2.3) in a series of powers of this ratio and, being limited to the first two terms of the obtained expansion, to produce averaging. Here we will obtain

$$P[y_{i}(l), y_{i}(l), \dots, y_{n}(l) | \alpha] \approx \\ \approx K' \exp\left\{\frac{1}{2} \sum_{k=1}^{n} \left| \sum_{l=1}^{n} \frac{\gamma' \overline{\mathcal{P}_{oi}}}{N_{oi}} U_{i}(k\overline{\Gamma}_{r} | \alpha) f_{hi} \right|^{2} \right\}.$$
(11.2.1)

通行歴任

Here we also used the circumstance that by virtue of (10.3.15) for a small signal-to-noise ratio $v(t_1, t_2) \approx \rho(t_1 - t_2)$, so that $v_{kk} = 1$.

It is possible to show that the unconditional functional of probability can be of an incoherent signal will have the same form in certain other cases, for instance in the rese of rapid fluctuations of the signal, when pulses fluctuate independently. Here the signal-to-noise ratio can be anything. Subrequently, we start from expression (11.2.5) for the functional of the probability density of the signal.

-631.-

Proceeding from this expression, it is possible to construct the operation of the optimum radio channel of an incoherent goniometer. In fact, obviously, we have

$$\frac{\partial \ln P[y_1(f), y_2(f), \dots, y_n(f)] \alpha]}{\partial \alpha} = \operatorname{Re} \sum_{k=1}^{m} \left(\sum_{l=1}^{n} \frac{\sqrt{P_{kl}}}{N_{kl}} U_k(kT_r, \alpha) f_{kl} \right) \times$$

$$\times \left(\sum_{i=1}^{a} \frac{\gamma \overline{\mathcal{P}_{si}}}{N_{si}} U^{*r_{i}}(kT_{r}, a) f^{*}_{hi}\right).$$

where the stroke denotes, as always, the derivative with respect to α (here we used the fact that ln K', as it is easy to show (see § 10.2), does not depend on α). It follows from this that operation of the optimum radio channel consists of formation of signal z(t), for which

$$\mathbf{z} (\mathbf{k}T_r) = \operatorname{Re} \sum_{j=1}^{n} \sqrt{\frac{P_{ci}}{P_{cj}}} q_j U_j (\mathbf{k}T_r, \hat{\mathbf{a}}) \times \\ \times \int_{(\mathbf{k}-1)T_r}^{\mathbf{k}T_r} u(t) y_j(t) e^{i\omega_0 t} dt \sum_{j=1}^{n} \sqrt{\frac{P_{ci}}{P_{cj}}} q_j U^{\omega_j} (\mathbf{k}T_r, \hat{\mathbf{a}}) \times \\ \times \int_{(\mathbf{k}-1)T_r}^{\mathbf{k}T_r} u^{\omega_0}(t) y_j(t) e^{-i\omega_0 t} dt, \qquad (11.2.6)$$

where $q_j = \frac{P_{c,j}T_r}{2N_{o,j}}$ - ratio of signal energy for the period of repetition at the output of the j-th antenne to the spectral density of the corresponding noise (the applied normalization, as one shall see from what follows, is very convenient);

 α - measured value of the angular coordinate (in the case of geniometers with a tracking antenna, as in the preceding chapters, $\hat{\alpha}$ should be replaced by zero).

Further processing consists of simple summation, or as it is said, incoherent accumulation of quantities $z(kT_p)$.

Let us consider the question of the circuit realization of optimum operation (11,2,6). Here in the present section we limit our consideration to goniometers with a tracking antenna, for which the optimum circuit has the simplest form. Considering that $J_{nj}(t, 0) = 1$ [see (10.3,1)], we can rewrite operation (11.2,6) in the following form:

$$\mathbf{z}(kT_r) = \operatorname{Re}\sum_{j=1}^{n} \sqrt{\frac{P_{c1}}{P_{aj}}} q_j \int_{(k-1)T_r}^{kT_r} u(t) y_j(t) e^{i\omega_0 t} dt \times \\ \times \sum_{j=1}^{n} \sqrt{\frac{P_{c1}}{P_{cj}}} q_j [U'_{aj}(kT_r, 0) - i \psi'(kT_r, 0)] \times \\ \times \int_{(k-1)T_r}^{kT_r} u^*(t) y_j(t) e^{-i\omega_0 t} dt.$$
(11.2.7)

One of the variants of the block diagram of a device realizing operation (10.2.7) [sic] is shown in Fig. 11.1 (we shall not stop to discuss the transformation of (11.2.7) to real form, on the basis of which there is constructed the block diagram,



Fig. 11.1. Optimum circuit of the radio channel of an incoherent radar goniometer with a generalized method of direction finding. 1 - optimum filters with pulse response (11.2.8); 2 - adders.

since this was done in sufficient detail in the preceding chapter). In this circuit signals are first normalized to the same level (normalization is necessary in the case of different gains of antennas and a different level of noises in the channels); then they are heterodyned with simulianeous multiplication by the law of intraperiod amplitude 「「「「「「「」」」」

modulations and with inversion of phase modulation. After this the signals are filtered. The filter must have pulse response

$$h(t) = \begin{cases} \cos \omega_{np} t, \ 0 < t < T_r, \\ 0, \ t > T_r, \end{cases}$$
(11.2.5)

i.e., must realize clearing of the accumulated voltage at the end of the period of repetition of the signal. The low-frequency equivalent of this filter is an integrat r with clearing. When the signal has a large off-duty factor, i.e., the duration of the signal pulse $\tau_g \ll T_r$, as the indicated filter we can use any filter with periods of $\Delta f_{\rm d}$ which satisfies relationship

$$\frac{1}{T_r} \ll \Delta j_{\psi} \ll \frac{1}{\epsilon_{\rm m}}.$$
(11.1.1)

The signal from the output of each such filter enters two channels: in one channel it is multiplied by a function varying according to the law of scanning of the directional pattern of the corresponding antenna; in the other after a shift in phase of $\pi/2$ it is multiplied by a function varying according to the law of scanning of the phase center of this antenna. After these operations all signals are added and proceed to the phase detector. As the reference voltage of the phase detector there is used the sum of signals taken directly from the output of the filters.

This circuit may, of course, be modified. For instance, multiplication by functions varying according to laws of scanning could have been carried out before filtration. However, these modifications are nonessential. More essential is modification connected with other in erpretation of the operation of taking integrals over the period of repetition in (.2.7). Namely, it is possible to carry out the operation of heterodyning in the absence of modulation of the heterodyne voltage, and then pass the signal through a filter with pulse response

$$h(t) = \operatorname{Re} u(-t) e^{i\theta_{x}t} . \qquad (11.2.10)$$

From the output voltage of the filter it is necessary then to separate only the values corresponding to discrete moments of time kT_r . This can be carried out by gating the output voltage of the filters by gate pulses of sufficiently small duration, with period T_r . Gating may also, obviously, be carried out after all operations of the discriminator.

A filter of type (11.2.10) is called "matched with the form of the signal" or "shortening." Processing using such a filter is called "filtrational," while processing in the initial interpretation is called "correlation." These terms already were introduced in Chapter VIII, and we shall not discuss them in detail here. The block diagram of filtrational processing differs from the circuit of Fig. 11.1 insignificantly: the heterodyne signal should have the form $\cos \alpha_{\rm Hp}$ t, and filters should have pulse response (11.2.10); at the output of the circuit there should be a gating unit. In other respects the circuits coincide.

Thus, the basic difference between an incoherent circuit and the corresponding coherent one (see Fig. 10.7) is that in the incoherent circuit each period of the signal is processed separately, and results are totaled. In the coherent circuit, due to the presence of narrow-band filters there occurs joint processing of many periods of the signal. More detailed study of synthesized circuits we postpone to subsequent sections, where we shall consider concrete methods of direction finding.

Thus, with very general assumptions with respect to the method of direction finding we performed approximate synthesis of optimum circuits of incoherent goniometers. The purpose of further investigation is a detailed and systematic study of quasi-optimum circuits from the point of view of their real accuracy. In particular very interesting is the study of such circuits with a large signal-to-noise ratio, since their optimality in this case is not clear a priori. Basic attention during the study of different circuits we will allot to variants with correlation processing, since in radar goniometers circuits with correlation processing are the most common. However, here we shall not limit our consideration to filters carrying out integration of the signal during the period of repetition: characteristics of filters as far as possible will be assumed arbitrary.

Study of quasi-optimum circuits in the above-described aspect, as already repeatedly noted, is very interesting and important from the practical point of view, since only such study will allow us to solve the question of the possibility of sufficiently good appromination of properties of optimum circuits.

§ 11.3. Method of Scanning the Directional Pattern

Let us turn to a study of incoherent goniometers using the method of scanning the directional pattern. The optimum circuit of the radio channel of such a goniometer can easily be obtained after certain concretizations from the general circuit of Fig. 11.1. For this it is necessary to set n = 1 (one antenna), $\Phi(t, \alpha) =$ = 0 (the phase center of the antenna is fixed), and express $U_a(t, \alpha)$ by formula (10.2.2). The optimum circuit can easily be reduced to the form depicted in Fig. 11.2. This circuit is known; however, the conducted synthesis permits us to more exactly formulate the requirements for the circuit from the point of view of its optimality: exact processing of intraperiod modulation of the signal and use of a filter whose low-frequency equivalent is an integrator with clearing. The influence on accuracy of all possible deviations from optimality in the circuit will be studied in detail in the present section.

Before passing to analysis of real accuracy of the synthesized circuit, we note that the circuit of Fig. 11.2 is theoretical and reflects only the fundamental operations performed on the signal. The practical variant of this circuit, depicted in Fig. 11.3, will, of course, differ somewhat from its theoretical prototype.

-635-

and the second second



~;}~

Fig. 11.2. Optimum circuit of the radio channel of an incoherent goniometer with pattern scanning: 1 — optimum filter with pulse response (11.2.8); 2 — square-law detector.



Fig. 11.3. Practical circuit of the radio channel of an incoherent goniometer with pattern scanning: 1 - high-frequency amplifier-mixer; 2 - amplitude modulator; 3 - heterodyne oscillator; 4 - generator of gate pulses; 5 - UPCh; 6 - system of automatic gain control; 7 - amplitude detector; 8 - pulse detector; 9 - phase detector; 10 - GON (reference voltage generator).

The difference is caused by the necessity of introducing certain elements, necessary during practical use of the circuit. Such an element in the first place is a system of automatic gain control [AGC]. It is introduced, as we already said, to maintain a constant level of the received signal, thanks to which the amplifier in the circuit always works in linear conditions. Furthermore, in the incoherent circuit of the method of pattern scanning after the amplitude detector in practice there usually is used a so-called pulse detector. The discharge time of the capacitor in the circuit of this detector is selected sufficiently large

Manager Sitter Stands

(so that the capacitor was discharged before the moment of arrival of the following pulse, there frequently is used forced discharge of this capacitor). Use of a pulse detector in such circuits permits us to obtain a very high transmission factor of the radio channel with minimum technical expenditures (otherwise at the output of the circuit it would be necessary to substitute an amplifier with a rather high gain factor).

As we already said in the preceding chapter, an AGC system in the first approximation influences only the slope of the discrimination characteristic of the circuit, and, consequently, such characteristics of accuracy as equivalent spectral density can be calculated without taking into account the AGC system. The influence of the AGC system on the slope of the discrimination characteristic of the circuit and on accuracy of incoherent goniometers as a whole we shall consider later, in § 11.7.

The pulse detector, as calculations show [53], approximately is equivalent to a normal amplifier. Consequently, from the point of view of calculation of accuracy

-636-

of the goniometer and study of the dependence of accuracy on the signal-to-noise ratio the pulse detector is a link such as can be omitted in general.

Thus, let us turn to analysis of accuracy of the circuit of Fig. 11.2. Calculation of accuracy characteristics of this circuit we shall produce in comparative detail, so that we can subsequently omit analogous calculations. Let us note that these calculations are very close to those with which we dealt in examining coherent circuits.

The received y(t) in the considered circuit has the form

$y(t) = \sqrt{P_{o}} \operatorname{Re} U(t, a) E(t) u(t) e^{i \omega t + i \delta(t)} + \sqrt{N_{o}} n(t),$

where $V(t, \alpha)$ is given by formula (10.2.2), and the remaining parameters were introduced earlier [see the explanation with formula (11.2.1)].

At the output of the circuit, obviously, we have signal

$$z(t) = \cos \vartheta(t) \left| \int_{-\infty}^{t} \dot{h}(t-\tau) y(\tau) v(\tau) e^{i\omega_0 \tau} d\tau \right|^{t}, \qquad (11.3.1)$$

where h(t) - complex pulse response envelope of the filter in the circuit;

v(t) = complex signal amplitude envelope of the heterodyne oscillator. Ideally

$$\hat{h}(t) = \begin{cases} 1, \ 0 < t < T_r, \\ 0, \ t > T_r, \end{cases}$$

[see formula (11.2.8)], and v(t) should coincide with the complex signal amplitude envelope u(t). However, in order to allow for possible imperfectnesses of filtration and processing of intraperiod modulation of the signal, we shall consider $\dot{\mathbf{h}}(t)$ and v(t) arbitrary. We introduce only the assumption that the passband of the filter $\Delta r_{d_{\rm c}} \gg \frac{1}{T_{\rm p}}$, which always occurs in incoherent goniometers. We shall place no further limitations on characteristics of the filter.

Let us calculate the mean value $\overline{z(t)}$, necessary for calculation of the slope of the discrimination characteristic and systematic error. Obviously,

$$\overline{z(t)} = \cos \vartheta(t) \int_{-\infty}^{t} \int_{-\infty}^{t} \dot{h}(t-t_{1}) \dot{h}^{*}(t-t_{2}) \overline{y(t_{1})y(t_{2})} \times \\ \times v(t_{1}) e^{i\omega_{0}t_{1}} v^{*}(t_{2}) e^{-i\omega_{0}t_{2}} dt_{1} dt_{2} = \\ = \cos \vartheta(t) \int_{-\infty}^{t} \int_{-\infty}^{t} \dot{h}(t-t_{1}) \dot{h}^{*}(t-t_{2}) \{P_{0}\rho(t_{1}-t_{2}) \times \\ \times \operatorname{Re} U(t_{1}, a) U^{*}(t_{2}, a) u(t_{1}) u^{*}(t_{2}) \times \\ \times e^{i\omega_{0}(t_{1}-t_{0})+t(b(t_{1})-b(t_{0}))} + N_{0}\delta(t_{1}-t_{2})\} \times \\ \times v(t_{1}) v^{*}(t_{2}) e^{i\omega_{0}(t_{1}-t_{0})} dt_{1} dt_{2}.$$
(14.3)

. 2)

We use the broad-banded nature of the filter, so that with respect to functions $\rho(t)$ and U(t, a) pulse response h(t) can be considered a δ -function. Here we have (averaging also the rapidly oscillating functions under the sign of the integrals)

$$\overline{z(t)} = \cos \vartheta(t) P_{\vartheta} |U(t, \alpha)|^{2} \frac{1}{2} \int_{-\infty}^{t} \int_{-\infty}^{t} \dot{h}(t-t_{1}) \dot{h}^{\vartheta}(t-t_{1}) \times \\ \times u(t_{1}) u^{\vartheta}(t_{2}) v^{\vartheta}(t_{1}) v(t_{2}) dt_{1} dt_{2} + \\ + N_{\vartheta} \cos \vartheta(t) \int_{-\infty}^{t} |\dot{h}(t-\tau) v(t)|^{2} d\tau.$$

Averaging now $\overline{z(t)}$ in time and reducing by usual methods the integrals over time to integrals over frequency, we obtain

$$\lim_{t\to\infty}\frac{1}{T}\int_{0}^{T}\overline{z(t)}\,dt=\frac{\mu_{n}\alpha P_{c}}{4\pi}\int_{-\infty}^{\infty}|H(i\omega)|^{n}|F(i\omega)|^{n}\,d\omega,$$

where

$$F(i\omega) = \frac{1}{\sqrt{T_r}} \int_0^{T_r} u(t) v^*(t) e^{i\omega t} dt, \qquad (11.3.3)$$

and $H(i\omega)$ - frequency response of the filter (more exactly, of its low-frequency equivalent).

It follows from this that systematic error in the considered circuit is absent, since

$$\lim_{T\to\infty}\frac{1}{T}\int_{0}^{T}\overline{z(t)}\,dt\Big|_{t=0}=0.$$

The slope of the discrimination characteristic turns out to be equal to

$$K_{\mathbf{x}} = \frac{d}{d\alpha} \lim_{r \to \infty} \frac{1}{r} \int_{0}^{r} \overline{z(l)} dt \Big|_{\alpha = 0} = \frac{\mu_{\mathbf{x}} P_{\mathbf{x}}}{4\pi} \int_{-\infty}^{\infty} |H(i\alpha)|^{2} |F(i\omega)|^{2} d\omega.$$
(11, 3.4)

For equivalent spectral density it is easy to analogously obtain the following:

$$S_{\Theta RB} = \frac{1}{K_{A}^{2}} \lim_{T \to \infty} \frac{1}{T} \int_{0}^{T} dt \int_{-\infty}^{\infty} \overline{z(t) z(\tau)} d\tau \Big|_{B=U} = \pi$$

$$= \frac{1}{K_{A}^{2}} \lim_{T \to \infty} \frac{1}{T} \int_{0}^{T} dt \int_{-\infty}^{\infty} d\tau \cos \theta (t) \cos \theta (\tau) \times$$

$$\times \int_{0}^{t} \int_{0}^{t} \int_{0}^{\tau} \int_{0}^{t} \dot{h} (t-t_{1}) \dot{h}^{*} (t-t_{2}) \dot{h} (\tau-\tau_{2}) \dot{h}^{*} (\tau-\tau_{2}) \times$$

$$\times \overline{y(t_{1}) y(t_{2}) y(\tau_{2}) y(\tau_{2}) v(t_{1}) v^{*} (t_{2})} \times$$

$$\times \overline{v(\tau_{2}) v^{*} (\tau_{2}) e^{i \omega_{0} (t_{1}-t_{0}+\tau_{1}-\tau_{2})} dt_{1} dt_{2} d\tau_{2} d\tau_{2}. \qquad (11, 3, 4^{+1})$$

-638-

Producing simple, but rather bulky calculations and limiting ourselves for simplicity to the case of uniform conical scanning, we finally can obtain

$$S_{\mathbf{0}\mathbf{K}\mathbf{0}} = \frac{1}{4\mu_{\mathbf{0}}^{2}} \widetilde{S}_{\mathbf{0}}(\Omega) + \frac{T}{2\mu_{\mathbf{0}}^{2}q^{2}} \int_{-\infty}^{\infty} \int_{-\infty}^{\infty} |H(i\omega_{1})H(i\omega_{2})|^{2} \times \\ \times [2 \operatorname{Re} qF(i\omega_{1})F^{*}(i\omega_{2})V^{*}(i\omega_{1}+i\omega_{2})+ \\ + |V(i\omega_{1}+i\omega_{2})|^{2}] d\omega_{1}d\omega_{2} \Big[\int_{-\infty}^{\infty} |H(i\omega)|^{2} |F(i\omega)^{-2}, \qquad (11.3.5) \Big]$$

where

$$\widetilde{S}_{\bullet}(\Omega) = \int_{-\infty}^{\infty} \rho(t)^{3} e^{-f \Omega t} dt; \qquad (11, 3, 6)$$

 $\rho(t)$ - correlation function of fluctuations of the signal, normalized so that $\rho(0)$ = = 1;

- \mathbf{q} ratio of energy of the signal for the period of repetition and the spectral density of noise;
- Ω = angular frequency of scanning;

$$V(i\omega) = \int_{0}^{T_{r}} |v(t)|^{a} e^{i\omega t} dt. \qquad (11.3.7)$$

When $\rho(t) = e^{-2\Delta f} c |t|$

$$\widetilde{S}_{\bullet}(\Omega) = \frac{\frac{1}{2} \Delta f_{\circ}}{1 + \left(\frac{\Omega}{4\Delta f_{\circ}}\right)^{3}}.$$
(11.3.8)

Let us study the found results in more detail. The simplest formulas are obtained in the case of a rather narrow-band filter whose passband satisfies relationship (11.2.9). In this case under the sign of integrals in (11.2.9) it is possible to set $|H(i\omega)|^2 = 2\pi\delta(\omega)\Delta f_{\vec{D}}$, where $\Delta f_{\vec{D}}$ - effective width of the passband of the filter (gain of the filter is assumed to be one). Here we easily obtain the following results. The slope of the discrimination characteristic is equal to

$$F_{\rm A} = \frac{1}{2} \mu_{\rm A} P_{\rm o} T_{\rm r} \varkappa_{\rm i} \Delta f_{\rm \phi}, \qquad (11, 3, 9)$$

where

$$\varkappa_{1} = \left| \frac{1}{T_{r}} \int_{0}^{T_{r}} u(t) v^{*}(t) dt \right|^{2}$$

This coefficient was introduced already in Chapter X [see formula (10.5.9)]. It allows for imperfectness of processing of modulation of the signal. For equivalent spectral density we obtain the very simple formula



$$S_{0KB} = \frac{1}{2\mu_{0}^{2}} \widetilde{S}_{0}(\Omega) + \frac{T_{r}}{2\mu_{0}^{2}} \frac{\operatorname{Re} 2q |F(0)|^{4} V^{*}(0) + |V(0)|^{4}}{q^{2} |F(0)|^{4}} = \frac{1}{2\mu_{0}^{2}} \widetilde{S}_{0}(\Omega) + \frac{T_{r}}{2\mu_{0}^{2}} \frac{2q' + 1}{q'^{4}}, \qquad (11.3.10)$$

where

 $q' = \frac{P_{e}T_{r}}{2N_{e}} \frac{\left|\frac{1}{T_{r}} \int_{0}^{t} u(t) v^{*}(t) dt\right|^{2}}{\frac{1}{T_{r}} \int_{0}^{t} |v(t)|^{2} dt} = \frac{P_{e}T_{r} x_{1}}{2N_{e} x_{2}}.$ (11.3.11)

Ratio n_1/n_2 , as we saw in the preceding chapter, is always smaller than one and equal to 1 only with ideal processing of modulation of the signal. Thus, imperfectness of processing of modulation with a narrow-band filter is equivalent to decrease of the signal-to-noise ratio.

From formula (11.3.10) it is clear that even with total elimination of noises there exists a residual value of equivalent spectral density

$$S_{\text{exp}} = \frac{\widetilde{S}_{\text{e}}(0)}{2\mu_{\text{e}}^2},$$
 (11.3.12)

caused by fluctuations of the signal. With increase of the frequency of scanning the error caused by this factor disappears, and we obtain

$$S_{\text{ORB}} = \frac{T_r}{2\mu_a^2} \left(\frac{2}{q'} + \frac{1}{q'^2} \right). \tag{11.3.13}$$

Let us consider asymptotic cases on high and low noise levels. With high noise levels

$$S_{\text{PKB}} = \frac{T_{\text{r}}}{2\mu_{\text{a}}^2 q^{\prime 2}}.$$
 (11.3.14)

With ideal processing of signal modulation, when q' = q, this formula gives, obviously, the potential value of equivalent spectral density for a small signal-tonoise ratio, since the analyzed circuit is optimum for this case. It is therefore of interest to compare (11.3.14) with the equivalent spectral density of an optimum coherent circuit with a small signal-to-noise ratio. With approximation of the spectrum of the signal by formula (10.3.26) and for the case of uniform conical scanning it is easy to find that the equivalent spectral density of error in the optimum coherent circuit [formula (10.4.10)] is less than the equivalent spectral density in the incoherent circuit by a factor of $1/4T_r\Delta f_c$. Let us emphasize once again that this occurs only for small signal-to-noise ratios.

Consider now the case of large signal-to-noise ratios. In this case

 $S_{aRa} = \frac{T_{r}}{m' \sigma'}$

Let us compare the equivalent spectral density for the given case with the optimum spectral density of a coherent circuit.

For large signal-to-noise ratios S_{OHT} for the coherent circuit we take from formula (10.4.11). Here it is easy to find that the ratio of the equivalent spectral density of the considered incoherent circuit and the optimum spectral density of the coherent circuit (in the case of high frequencies of scanning) is equal to

$$\frac{S_{\text{one}}}{S_{\text{one}}} = \frac{T_r}{\mu_a^2 q} \cdot \mu_a^2 h \Delta f_c = 1.$$

Thus, for large values of the signal-to-noise ratio the considered incoherent circuit (with ideal processing of signal modulation) ensures the same accuracy as an optimum coherent circuit. From this we can conclude that the circuit of Fig. 11.2 realizes the potential accuracy of the method of pattern scanning also for a large signal-to-noise ratio, i.e., is a good approximation of the optimum circuit in the whole range of changes of the signal-to-noise ratio.

We repeat that everything said pertains to the case when the frequencies of scanning are sufficiently great as compared to the width of the spectrum of signal fluctuations. With low frequencies of scanning this circuit gives a component of error (11.3.12) which does not depend on the signal-to-noise ratio, and for a sufficiently large value of the signal-to-noise ratio, apparently, differing from the optimum.

Investigation of formula (11.3.5) with an arbitrary width of the filter passband is better produced for some concrete approximation of the form of modulation of the sounding and heterodyne signals and of the frequency response of the filter. Let us consider, for instance, the case of a pulse signal without phase modulation with pulses of Gaussian form:

$$u_{a}(t) = 1.15 \sqrt{\frac{\bar{T}_{r}}{\tau_{a}}} e^{-2.8 \left(\frac{1}{\tau_{a}}\right)^{2}}, \qquad (11.3.15)$$

where $\tau_{_{\rm H}}$ - duration of the pulse at level 0.5 [the coefficient is selected from condition of normalization (10.3.4)].

Gate pulses we consider to have the same form, but with different duration:

-641-

$$D_{\mathbf{a}}(l) = e^{-2.0 \left(\frac{l}{\tau_{a}}\right)^{2}}.$$
(11.3,16)

We approximate the frequency response of the filter by function

$$\frac{-2\pi \left(\frac{\alpha}{2\pi 4 f_{\odot}}\right)^{2}}{|H(l_{\odot})|^{2} = e}$$
(11.3.17)

We designate

$$\Delta f_{\text{eora}} = \frac{5.6}{2\pi} \cdot \frac{1}{s_{\text{m}}} \approx \frac{0.9}{s_{\text{m}}}$$

A filter with bandwidth $\Delta f_{\tilde{\Phi}} = \Delta f_{COTJ}$ frequently is called matched. We assume, further, that

$$x = \frac{\Delta f_{\phi}}{\Delta f_{\phi erg}} = 1, 1 \tau_{g} \Delta f_{\phi}, \quad y = \frac{\tau_{g}}{\tau_{\phi}}. \quad (11.3.18)$$

Calculation by formula (11.3.4) with such approximations gives the following result:

$$S_{9K_{B}} = \frac{1}{2k_{B}^{2}} \widetilde{S}_{9}(\Omega) + \frac{T}{2\mu_{a}^{2}} \left\{ \frac{(1+y^{3})(1+x^{3}+y^{3})}{q^{3} \cdot 4y \, y' \, \overline{x^{3}+y^{3}}} + \frac{2}{\gamma} \frac{(1+y^{3})(1+x^{3}+y^{3})}{y' \, [x^{3}+2(1+y^{3})] \, [x^{2}+2y^{3}(1+x^{3}+y^{3})]} \right\}.$$
(11.3.19)

With high frequencies of scanning the first term can be disregarded. Here S_{OKB} attains if minimum value S_{MKH} at x = 0, y = 1. This value coincides with (11.3.13), where it is necessary to set q' = q. However, already with small deviation of y from 1 the picture changes rather sharply: the minimum of S_{OKB} starts to be reached at $x \approx 1$. This one may see well from graphs of the dependence of $S_{\text{OKB}}/S_{\text{MMH}}$ on q, shown for different values of y and x (Fig. 11.4).

Thus, with nonideal gating the optimum filter is a filter, close to the matched one. With ideal gating the optimum is a narrow-band filter, integrating the signal in the period of repetition. However, all optima here lie rather closely to one another.

Let us note the very simple formulas obtained from (11.3.19) in limiting cases. For sufficiently wide gate pulses, when $y \ll 1$, we have

$$S_{0HB} := \frac{1}{2\mu_a^2} \widetilde{S}_{p}(\Omega) + \frac{T_r}{2\mu_a^2} \left\{ \frac{1+x^3}{4xyq^2} + \frac{2}{q} \cdot \frac{1+x^2}{x\sqrt{2+x^2}} \right\}.$$
 (11.3.20)

This formula with considerable widening of the band of the filter, i.e., when $x \gg 1$, takes the following very simple form

$$S_{\Theta R g} = \frac{1}{2\mu_a^2} \widetilde{S}_{\Theta}(\Omega) + \frac{T_r}{2\mu_a^2} \left(\frac{0.28\Delta f_{\Theta^{\uparrow_a}}}{q^3} + \frac{2}{q} \right)$$
(11.3.21)

-642-

We shall now discuss the question of parametric fluctuations. The spectral density of parametric fluctuations in general is difficult to calculate, in view of the very great cumbersomeness of computations. We will give results of calculation of the spectral density of parametric fluctuations for the case of high frequencies of scanning, when the component of error caused by equivalent spectral density is rather small. In this case we have the very simple relationship

$$S_{map} = \tilde{S}_{0}(0),$$
 (11.3.22)

where $\tilde{S}_0(\Omega)$ is determined by formula (11.3.6). In particular, with approximation $\rho(t) = e$ we obtain

$$S_{\pi a p} = \frac{1}{s} \Delta f_c.$$

Thus, the spectral density of parametric fluctuations does not depend on the form of modulation of the sounding signal and is inversely proportional to the width of the spectrum of signal fluctuations.





-643-

§ 11.4. Methods of Scanning with Compensation and IAC

11.4.1 Optimum Circuits and Their Characteristics

Optimum circuits for the method of scanning with compensation are easily obtained from the general circuit of Fig. 11.1, where n = 2 (two directional patterns)



Fig. 11.5. Variant of the optimum circuit of the radio channel of an incoherent goniometer for the method of scanning with compensation: 1 - optimum filters with pulse response (11.2.8); 2 - square-law detectors.

and $\Phi_i(t, \alpha) = 0$ (their phase centers are fixed and coincide). Function $U_{ai}(t, \alpha)$ must here be expressed by formula (10.2.5) for the method of scanning with compensation or by formula (10.2.8) for IAC [the latter one coincides with (10.2.5), if there we set $\$_1(t) =$ = 0, $\$_2(t) = \pi$]. If we consider

the amplifications and gain factors of the directional patterns, and also noises in channels identical, and consider the laws of scanning connected by relationship (10.6.2) ($\$_1(t) = \$(t) = \$_2(t) + \pi$), we can present the optimum circuit for the method of scanning with compensation in one of two forms, depicted in Figs. 11.5 and 11.6. By analogy with the terminology of Chapter X the circuit of Fig. 11.5 we shall subsequently call a circuit with subtraction, and the circuit of Fig. 11.6 - a circuit with multiplication of signals. Circuits for IAC have, obviously, precisely the same form, only it is necessary to set \$(t) = 0, i.e., multiplication by cos \$(t) at the output of the circuits will be absent. The obtained circuits are very similar to the corresponding coherent circuits, synthesized in Chapter X.



Fig. 11.6. Variant of the optimum circuit of the radio channel of an incoherent goniometer for the method of scanning with compensation. 1 - optimum filters with pulse response (11.2.8). 11.4.2. Investigation of Circuits with Multiplication of Signal?

Let us investigate characteristics of synthesized circuits. It is obvious that both considered variants of circuits (with multiplication and subtraction of signals) with ideal construction give identical accuracy. The difference between these circuits will appear only in their unequal criticality with respect to nonidentity of channels. Investigation of this question is very interesting from the practical point of view.

In view of cumbersomeness of calculations we shall produce a simplified allowance for nonidentity of channels of the circuits, seeking basically to find the physics of the influence of nonidentity of different parameters and obtain simple calculating formulas.

Let us consider first the circuit with multiplication of signals. The output signal of the circuit of Fig. 11.6 has, obviously, form (10.7.1) or (10.7.2) (the latter when f(t) = 0). The received signals are equal to

$$y_{i} = \sqrt{P_{oi}} \operatorname{Re} U_{i}(t, a) E(t) e^{i \omega_{o} t + i \delta(t)} + \sqrt{N_{oi}} n_{i}(t), \qquad (11.4.1)$$

where $U_i(t, \alpha)$ are expressed by formulas (10.2.5) or (10.2.8).

We assume that the width of the bands of the filters in the circuit satisfies relationship (11.2.9). We assume that the frequency responses of the filters are identical; however between the channels there is a certain constant phase shift $\Delta \varphi$. Obviously, this type of nonidentity is the most essential for circuits with multiplication of signals.

Omitting intermediate computations, we give the final results, identical for circuits of the method of scanning with compensation and IAC. The slope of the discrimination characteristic turns out to be equal to

$$K_{\mathbf{x}} = \frac{\mu_{\mathbf{x}} P_{\mathbf{c}\mathbf{z}} T_{\mathbf{y}}^{2} \mathbf{x}_{\mathbf{i}}}{2} \cos \Delta \varphi \Delta f_{\mathbf{\Phi}}, \qquad (11.4)$$

2)

where $P_{C\Sigma}$ - total power of signals; the remaining designations are the same as in formula (11.3.9).

From this it is clear that in the presence of phase shift between channels the slope of the discrimination characteristic decreases. With very great nonidentity of channels slope $K_{\rm A}$ can fall to zero or even become negative. The circuit in this case, obviously, will not work.

Systematic error in the considered circuit will always be absent. This is also understandable from very simple reasonings: if the target is in the equisignal

-645-

direction, the signal in the difference channel will be equal to zero. Consequently, the signal of error also will be equal to zero, regaraless of identity or nonidentity of the channels.

For the equivalent spectral density of error in the circuit with multiplication of signals we obtain

$$S_{9F3} = \frac{7}{\mu_{a}^{2} \cos^{4} \Delta \psi} \left(\frac{1}{q'_{z}} + \frac{1}{q'_{z}} \right), \qquad (11.4.3)$$

where q'_{Σ} is determined by formula (11.3.11) (with replacement of P_{C} by $P_{C,\Sigma}$).

From this we see that with nonidentity of channels there also occurs increase of the equivalent spectral density of error. This formula differs somewhat from the analogous formula (11.3.13) for the method of pattern scanning.

Let us assume now that the filter has a very wide band, so that there does not occur pure integration of the signal. Here we shall limit ourselves to the case of identical channels. Results will be enalogous to those which we had for the method of pattern scanning: the slope of the discrimination characteristic is determined by formula (11.3.3) (with replacement of I_c by $P_{c,Z}$), and equivalent spectral density is determined by an expression somewhat differing from (11.3.4¹):

$$S_{\theta H g} = \frac{T_r}{\mu_0^2} \cdot \frac{1}{q_t^2} \left[\int_{-\infty}^{\infty} |H(l\omega)|^2 |F(l\omega)|^2 d\omega \right]^2 \times \int_{-\infty}^{\infty} \int_{-\infty}^{\infty} |H(l\omega_1)|^2 (Re q_L F(l\omega_1) F^*(l\omega_2) V^*(l\omega_1 + l\omega_2) + |V(l\omega_1 + l\omega_2)|^2) d\omega_1 d\omega_2.$$

$$(11, 4, 4)$$

Here there is no term caused by nonlinear transformation of the userul signal, which testifies to compensation of the influence of amplitude fluctuations of the eignet. Furthermore, the term propertional to $1/q_{0}^{2}$ is decreased by a factor of c. All contents in formula (11.4.4) are the same up in formula (11.5.4). If $|B(1.0)|^{2} = 2\pi \Delta t_{\Phi}^{2} |b(\infty)$, formula (11.4.4) passes into (11.4.3), where $\Delta p = 0$.

With approximations (11, 5, 15), (11, 3, 16), (31, 5, 17) and (11, 3, 18) for S_{BKB} we obtain the expression

$$S_{000} = \frac{T_{r_{1}}}{\mu_{2}^{2}} \left[\frac{1}{q^{0}} \frac{(1 + \mu^{0})(1 + x^{0} + \mu^{0})}{4\mu V x^{0} + \mu^{0}} + \frac{1}{q} \frac{1}{q^{0}} \frac{(1 + \mu^{0})(1 + x^{0} + \mu^{0})}{4\mu V x^{0} + \mu^{0}} \right], \qquad (11, 4.5)$$

insignificantly differing from (11.3.19). All qualitative conclusions with respect to the influence on S_{HKB} of various factors remain the same as before.

Management and the second s

11.4.3. Investigation of Circuits with Subtraction of Signals

In circuits with subtraction of signals there already is a certain difference between the method of scanning with compensation and IAC.

The signal of error at the output of the circuit of Fig. 11.5 has the form (10.7.6) in the case of scanning with compensation (where $\tau_3 = 0$ in view of the broad-band nature of the filters) or form (10.7.12) in the case of IAC. The received signals, as before, are expressed by formula (11.4.1). For these circuits the most essential type of nonidentity is nonidentity of gain factors of the channels.

Let us consider the case of a rather narrow-banded filter. Proceeding from the expression for the output signal, after simple transformations we obtain for the method of scanning with compensation the following results.

The slope of the discrimination characteristic

$$K_{\rm X} = \frac{P_{\rm cz} \mu_{\rm a} T_{\rm f}^2 x_1}{4 \mu_{\rm a}} \Delta f_{\rm \Phi} (K_1^2 + K_2^2), \qquad (11.4.6)$$

where K_1 and K_2 - gain factors of the channels.

Equivalent spectral density (in the case of uniform conical scanning)

$$S_{0NB} = \frac{1}{8\mu_0^2} \cdot \frac{(1-k)^2}{1+k^2} \widetilde{S}_0(\Omega) + \frac{T_r}{2\mu_0^2} \left(\frac{1}{q'_{\Sigma}} + \frac{1}{q'_{\Sigma}^2} \right), \qquad (11.4.7)$$

where $k = K_1^2/K_2^2$ - ratio of squares of gain factors of the channels; the remaining designations are the same as before.

Systematic error in the considered case is absent.

From formula (11.4.7) it is clear that in the case of nonidentity of channels there exists a component of error caused by nonlinear transformation of the signal and not canceled even with complete elimination of noises. We had an analogous component of error in the case of the method of pattern scanning (11.3.5). With growth of the frequency of scanning this component of error disappears. With identical channels, when k = 1, this component of error is absent.

For the IAC method calculations do not differ at all from calculations for the method of scanning with compensation. Results are somewhat different. Gain factor $K_{\rm L}$ is expressed by the same formula (11.4.6). However, now there will exist systematic error equal to

$$\Delta = \frac{1}{\mu_{\bullet}} \frac{1-k}{1+k} \left(1 + \frac{1}{q'_{\Sigma}} \right). \tag{11.4.8}$$

-647-

Systematic error monotonically grows with growth of nonidentity of gain of the channels. The signal-to-noise ratio also affects systematic error.

For equivalent spectral density in the case of IAC we obtain the following expression:

$$S_{\text{DRB}} = \frac{T_{\text{r}}}{\mu_{\text{B}}^2} \left[\frac{(1-k)^2}{4(1+k^2)} + \frac{1}{q'_{\text{Z}}} + \frac{1}{q'_{\text{Z}}^{\prime 2}} \right].$$
(11.4.9)

Thus, here there also exists a residual value of error, caused by amplitude fluctuations of the signal. In the case of identical channels, as one should have expected, formulas (11.4.9) and (11.4.7) coincide, i.e., the methods of scanning with compensation and IAC give identical accuracy, regardless of the circuit of the radio channel, if the channels of the latter are completely identical. This occurs, naturally, also for arbitrary frequency responses of the filters; then for circuits with subtraction accuracy will be characterized already by the known formula (11.4.4).

Comparing circuits of the method of scanning with compensation and with IAC, we note that any of the circuits with compensation with two channels can provide measurement of angular coordinates in two planes. Circuits with IAC for measuring two angular coordinates with switching will contain three channels (one with the sum signal and two with different signals). The IAC circuit with subtraction should already contain 4 channels (two channels for measurement of the angle in one plane and two for the other). Thus, simplest are circuits of the method of scanning with compensation, then follows the IAC circuit with multiplication of signals and, finally, the IAC circuit with subtraction of signals.

From the point of view of criticality of the circuits to various nonidentities it is possible to note the following. Circuits with multiplication are critical basically to nonidentity of phase-frequency responses, where circuits of the method of scanning with compensation and IAC are equivalent. Circuits with subtraction are basically critical to nonidentity of gain of the channels. Here the circuit of the method of scanning with compensation has an obvious advantage over the IAC circuit, especially in the case of high frequencies of scanning: at high frequencies of scanning nonidentity of gain in the circuit of the method of scanning with compensation does not affect accuracy; in the IAC circuit it always leads to the appearance of a component of equivalent spectral density, not depending on the signal-to-noise ratio, and also appearance of systematic error.

-648-

§ 11.5. Method of Instantaneous Phase Comparison of Signals

Of phase methods of direction finding with an incoherent signal we can use only the method of instantaneous phase comparison of signals (IPC). The method of scanning the phase center with an incoherent signal is not useful for measurement of angular coordinates. The IPC method can be used with an incoherent signal and will give accuracy of the same order as, for example, IAC.

The optimum circuit for IPC can be obtained from the general circuit of Fig. 11.1, if in it we set n = 2, $U_{ai}(t, \alpha) = 1$, and introduce $\Phi_i(t, \alpha)$ according to formula (10.2.9).

It is easy to see that there are two identical variants of IPC circuits, depicted in Figs. 11.7 and 11.8: a circuit with formation of sum and difference signals and a circuit without formation of these. In structure these circuits are the same as for a coherent signal; however characteristic of the filters already are substantially different.



Fig. 11.7. Lariant of the optimum circuit of the radio channel of an incoherent goniometer with IPC. 1 - optimum filters with pulse response (11.2.8).



Fig. 11.8. Variant of the optimum circuit of the radio channel of an incoherent goniometer with IPC. 1 - optimum filters with pulse response (11.2.8).

Let us investigate the synthesized circuits. We consider first the case of sufficiently narrow-banded filters, carrying out exact integration of every period of the signal. We consider that channels of the circuits are nonidentical: between channels there is a constant phase shift $\Delta \varphi$. Nonidentity of gains also can be considered; however, it does not influence accuracy of the considered circuits. As also in the case of a coherent signal, the circuit of Fig. 11.8 for IPC is completely equivalent to the circuit with multiplication for IAC in the sense that accuracy of the circuits are expressed by absolutely identical formulas; only the role of the gain factor of the directional pattern in the case of IPC is played by $\pi d/\lambda$. This is eady to prove by calculating the signal of error at the output of the circuit of Fig. 11.8 and finding the characteristic of accuracy of the circuit. Calculation shows that the slope of the discrimination characteristic will be expressed by the known formula (11.4.4), and S_{OKB} is expressed by formula (10.4.5), where μ_a must be replaced by $\pi d/\lambda$. Systematic error is equal to zero.

For a circuit without formation of a sum and difference signal results are substantially different. Although the slope of the discrimination characteristic in this case is expressed by formula (11.4.4) with replacement of μ_a by $\pi d/\lambda$, here there will already exist systematic error

$$\Delta = \frac{1}{2\pi d/\lambda} \operatorname{tg} \Delta \varphi. \tag{11.5.1}$$

Equivalent spectral density, as calculation shows, is expressed in the following way:

$$S_{\text{PRB}} = \frac{T_r}{2\left(\frac{\pi d}{\lambda}\right)^2} tg^2 \Delta \varphi + \frac{T_r}{\left(\frac{\pi d}{\lambda}\right)^2} \cos^2 \Delta \varphi \left(\frac{1}{q'_2} + \frac{1}{q'_2}\right). \tag{11.5.2}$$

From (11.5.2) it is clear that with nonidentical channels there exists a component of error which is not canceled with total elimination of noises, i.e., caused exclusively by fluctuations of the signal. From this we see that the circuit with IPC without formation of the sum and difference signals is considerably more critical to nonidentity of channels than a circuit with formation of the sum and difference signals, so that the latter circuit is preferable. With identical channels, when $\Delta_{\phi} = 0$, both circuits are identical. Without dwelling on the calculation, we note that in the case of filters whose bandwidth is comparable with the broadened spectrum of modulation of the signal accuracies of both circuits are identical and are expressed by formula (11.4.4), in which it is necessary to replace μ_{a} by $\pi d/\lambda$.

We shall now stop to compare the considered methods of direction finding. It is easy to see that with a large signal-to-noise ratio and high frequencies of scanning (for the method of pattern scanning) accuracies of the considered methods of direction finding with ideal construction of the radio channel are expressed by the same formula

$S_{area} = T_r/c^2 q_r$

where $c^2 = \mu_a^2$ (for methods of scanning the directional pattern, scanning with compensation and IAC) and $c^2 = \left(\frac{\pi d}{\lambda}\right)^2$ (for the IPC method). Comparison of methods

-650-

here is very easy to carry out, and results are the same as in the coherent case: the method of pattern scanning is inferior to methods of IAC and scanning with compensation in view of a worse signal-to-noise ratio; with decrease of the frequency of scanning accuracy of the method of pattern scanning becomes still lower. Methods of IAC and scanning with compensation are equivalent in accuracy. For comparison of IPC and IAC it is necessary to compare μ_a^2 and $\left(\frac{\pi d}{\lambda}\right)^2$; in particular, for a square aperture of an antenna system of assigned dimensions use of the methods of IAC and IPC is equivalent.

§ 11.6. Method of Flat Scanning of the Directional Pattern

We shall investigate incoherent circuits of the method of flat scanning of the directional pattern (method of tracking by pulse packs). For this method the incoherent circuit will essentially differ from the coherent. From general expression (11.2.6) it is easy to obtain the optimum operation for the pack method in the following form:

$$z(kT_{r}) = \Big| \int_{(k-1)T_{r}}^{kT_{r}} u(t)y(t) e^{i a_{r} t} dt \Big|^{3} U_{a}(kT_{r}, \hat{a}) U_{a}(kT_{r}, \hat{a}).$$

Substituting for modulation $U_a(kT_r, \hat{\alpha})$ its value from (10.2.6) for the method of flat scanning, we obtain

$$z(kT_{r}) = \left| \int_{(k-1)T_{r}}^{kT_{r}} u(t) y(t) e^{i\omega_{r}t} dt \right|^{2} \sum_{n} g[\Omega(kT_{r} - \hat{\tau}_{n} + nT_{n})] g'[\Omega(kT_{r} - \hat{\tau}_{n} + nT_{n})], \qquad (11.6.1)$$

where $\hat{\tau}_{\alpha} = \hat{\alpha}/\Omega$ - measured value of delay of packs of the signal;

 $g(\phi)$ - form of the directional pattern (for power);

 Ω - angular velocity of motion of the pattern over the sector;

 T_r - period of the sector scan.

Note that for every k in formula (11.6.1) there is only one term of the sum standing there with number

$$n := \left[\frac{kT_r - c_a}{T_r}\right],$$

where [x] signifies the integer nearest to x.

The block diagram of the device realizing operation (11.6.1) is presented in Fig. 11.9. Here, after heterodyning, filtration and square detection, the signal is multiplied by gate pulses having form $g(\Omega t)$ g'(Ωt), delay of which is controlled. These gate pulses are bipolar, which is equivalent to multiplication by single-pole

gate pulses and further subtraction. The circuit of Fig. 11.9 is well-known. Our analysis will help one to establish optimum values of parameters of this circuit.

Let us turn to investigation of characteristics of the synthesized circuit. We assume, as before, that there is used a filter with possband $\Delta f_{\oplus} \gg 1/T_r$ and arbitrary frequency response, and that the neterodyne signal is not matched with the



Fig. 11.9. Optimum circuit of the radio channel of a goniometer using the method of flat scanning. 1 - optimum filter with pulse response (11.2.8).

received signal. Furthermore, we consider that the strobes by which the output signal is multiplied have a certain derivative form f(t). We introduce the hypothesis that the packs fluctuate harmoniously. Allows the for distortion of packs resulting from

fluctuations of the signal is an extraordinarily difficult and cumbersome problem. The signal of error at the output of the circuit of Fig. 11.9 has the form

$$z(t) = \left| \int_{-\infty}^{t} h(t-\tau) y(\tau) v(\tau) e^{i\omega_0 \tau} d\tau \right|^2 \sum_{n} f(t-\tau_0 + nT_n). \quad (11.6.2)$$

where the received signal y(t) is equal to

 $y(t) = \sqrt{P_o} \sum_{n} g[\Omega(t - \tau_a + nT_{yy})] \times \operatorname{Re} E(t) u(t) e^{i\omega_n t + it(t)} + \sqrt{N_o} n(t).$

Here $\tau_{\rm B} \simeq \alpha/\Omega$; α - angular coordinate of the target. The remaining designations are the same as before.

To ease calculations we assume that function f(t) is odd. Otherwise calculations and results become very cumbersome.

Averaging expression (11.6.2) over the ensemble and over time, we obtain

$$\lim_{t \to \infty} \frac{1}{T} \int_{0}^{T} \overline{z(t)} dt = \frac{P_c}{4\pi} \int_{-\infty}^{\infty} |H(i\omega)|^2 |F(i\omega)|^2 d\omega \times \frac{1}{T_{\pi}} \int_{0}^{T_{\pi}} g \cdot [\Omega(t-\tau_n)]^2 f(t-\tau_n) dt$$
$$= \frac{P_a}{4\pi} \int_{-\infty}^{\infty} |H(i\omega)|^2 |F(i\omega)|^2 d\omega \cdot \frac{1}{T_{\pi}} \int_{0}^{T_{\pi}} g(\Omega t)^2 f(t-\Delta \tau_n) dt.$$

where $\Delta :_3 = \frac{1}{3} - \frac{1}{2}$.

From this it is easy to find the expression (); the slope of the discrimination characteristic:

$$K_{R} = \frac{\sigma}{\partial \Delta \epsilon_{0}} \lim_{T \to \infty} \frac{1}{T} \int_{0}^{T} \overline{z(t)} dt =$$

$$= \frac{P_{e}}{4\pi} \int_{-\infty}^{\infty} |H(i\omega)|^{2} |F(i\omega)|^{2} d\omega \cdot \frac{1}{T_{\pi}} \int_{0}^{T_{\pi}} g(\Omega t)^{2} f'(t) dt =$$

$$= \frac{P_{e}C_{0}}{4\pi} \int_{-\infty}^{\infty} |H(i\omega)|^{2} |F(i\omega)|^{2} d\omega, \qquad (11.6.3)$$

where

$$C_{\bullet} = \frac{1}{T_{\bullet}} \int_{0}^{T_{\bullet}} g(\Omega t)^{\bullet} f'(t) dt. \qquad (11.6.4)$$

For the equivalent spectral density of error we obtain the following expression:

. t

$$S_{0RB} = \frac{\int_{-\infty}^{\infty} \int_{-\infty}^{\infty} |H(i\omega_1)|^2 |H(i\omega_2)|^2 [C_1|V(i\omega_1+i\omega_2)]^2 +}{q^2 C_0^2 \left[\int_{-\infty}^{\infty} |H(i\omega_2)|^2 |F(i\omega_2)|^2 d\omega_2\right]^4}$$
(11.6.5)

where

$$C_{1} = \frac{1}{T_{\pi}} \int_{0}^{T_{\pi}} f(t)^{2} dt, \qquad (11, 6, 5^{+})$$

$$C_{1} = \frac{1}{T_{\pi}} \int_{0}^{T_{\pi}} g(\Omega t)^{2} f(t)^{2} dt.$$

Formula (11.6.5) is very general and from it it is difficult to perceive the laws of change of S_{BHB} depending on different factors. We expland formula (11.6.5), introducing suitable approximations for parameters in it. Considering the filter sufficiently narrow-banded so that relationship (11.2.9) is carried out, we name

$$S_{0KB} = T_r \frac{C_1 + 2q'C_1}{C_0 q'^2}, \qquad (21, \dots, c)$$

where q is expressed by formula (11.3.11).

We use now a Gauddian approximation for the directional pattern (10,10,26). Here

$$g(\Omega t) = 1.15 \sqrt{\frac{T_{\pi}}{\tau_{\mu}}} e^{-2.8 \left(\frac{t}{\tau_{\mu}}\right)^{*}},$$

where $\tau_{\Pi} = \Delta \phi / \Omega$ - duration of a pack of the signal at level 0.5. We shall approximate gate pulse f(t) by function

$$f(t) = t e^{-2.8 \left(\frac{t}{\tau_{\rm m}}\right)^*},$$

i.e., gate pulses of form f(t) are obtained by differentiation of Gaussian gate pulses having duration τ_c at level 0.5. With these assumptions it is easy to calculate that



where $x = \tau_{\Pi} / \tau_{c}$.

Substituting these results in formula (11.6.6), we obtain

$$S_{\text{PRB}} = 0.07 \frac{T_{\text{R}} \Delta y^2}{\Omega^2 q_{\text{R}}^{1/2}} \left[\frac{\gamma}{x^8} + q'_{\text{R}} \frac{2.56}{(1+x^3)^{3/2}} \right] (1+0.5x^3)^2, \qquad (11.6.7)$$

where $\gamma = \frac{\tau_{\Pi}}{T_{r}}$ - number of periods of repetition of the signal in a pack (number of pulses in a pack),

$$q'_{\rm II} = q' \frac{T_{\rm I}}{T_{\rm I}} = \frac{P_{\rm o} T_{\rm II}}{2N_{\rm o}}$$

is the ratio of energy of the signal for the period of repetition of packs to the spectral density of noise.

In formula (11.6.7) factor $3/\Omega^2$ is taken for calculation of the spectral density and error of measurement of delay of the signal. To determine error of measurement of an angle we need to drop factor $1/\Omega^2$, which we shall do subsequently.

Let us study the formula for $S_{\!\!\!2HB}$ in greater detail.

For small signal-to-noise ratios

$$S_{0KB} = 0.07 \, \frac{T_{B} \Delta \varphi^{i} \gamma}{q_{B}^{\prime 2}} \left(\frac{1 + 0.5 x^{2}}{x} \right)^{2}.$$

The minimum value of S_{376} in this case occurs, obviously, at $x = \sqrt{2} \approx 1.4$, and is equal

$$S_{\rm MBH} = 0.2 \frac{T_{\rm m} \Delta q^{\rm e_{\rm T}}}{q_{\rm m}^{\prime 2}}.$$
 (11.6.8)

If we also consider processing of signal modulation ideal, so that q'_{Π} =

= $q_{\rm H}$, then $S_{\rm MMH}$ will coincide with the minimum possible value of equivalent spectral

density for high noise levels, since with the given assumptions the analyzed circuit is optimum. It is curious to compare, therefore, formula (11.6.8) with formula (10.10.28), giving the minimum possible value of equivalent spectral density S_{OUT} for the method of flat scanning with a small signal-to-noise ratio and a coherent signal. This comparison gives the following:

 $\frac{S_{02Y}}{S_{02R}} = \frac{0.2q_{\pi}^{2}}{c^{4}k^{4}T_{\pi}\Delta f_{\pi}} \left(\sum_{\sum j=1}^{\infty} p(kT_{\pi})^{4}\right) T_{\pi}\Delta p^{2}\gamma - C^{2}\Delta p^{2}\gamma \left(\sum_{k=-\infty}^{\infty} p(kT_{\pi})^{4}\right)$

Hence, i.: the case of rigidly correlated packs, when $T_{\Pi} \Delta t_{C} \ll 1$, we have (with a Gaussian approximation of the directional pattern and exponential approximation of the correlation function of fluctuation)

$$\frac{S_{\bullet n\tau}}{S_{n\pi\pi}} \approx \frac{0.3T_{\pi}\Delta/\epsilon}{\gamma}.$$

From this formula it is clear that for small values of the signal-to-noise ratio an incoherent goniometer gives considerable worsening of accuracy as compared to a coherent one. The worsening is inversely proportional to the number of packs in the interval of correlation of the signal, and proportional to the number of pulses in a pack. Physically, both circumstances are understandable.

In the case of independently fluctuating packs, when T $_{
m I}$ Δf $_{
m C}$ \gg 1, we have

$$\frac{S_{0.0.7}}{S_{\rm MEE}} = \frac{0.07}{T},$$

i.e., here there naturally occurs lowering of accuracy, proportional to the number of pulses of the signal in a pack.

Let us consider the case of large values of the signal-to-noise ratio. Here, from formula (11.6.7) we obtain

$$S_{\text{ORB}} = 0,186 \frac{T_{\text{B}} \Delta \gamma^{3}}{q'_{\text{B}}} \left[\frac{1+0.5x^{3}}{\sqrt{1+x^{3}}} \right]^{3}.$$

It is easy to see that the obtained value of S_{2103} monotonically drops with decrease of x, reaching its minimum at x = 0. Thus, for low noise levels gate pulses must be taken as wide as possible.

For an arbitrary signal-to-noise ratio there will exist a certain optimum value of z. The graph of the dependence of optimum values of x_{OHT}^2 on $q_{\rm R}^2$ for different y is shown in Fig. 11.10. From this graph one may graphically see what the optimum relationship of durations of packs and gate pulses x_{OHT} should be for different values of the signal-to-noise ratio.

-655-

In Fig. 11.11 there is shown the dependence of ratio S_{BKB}/S_{MVH} on the signal-tonoise ratio q'_{Π} , where S_{MVH} is the equivalent spectral density at optimum x. From Fig. 11.11 it is clear that when x = 1.4 the equivalent spectral density S_{BKB} is



Fig. 11.10. Graph of the dependence of optimum values of x_{OIIT}^2 on the signal-to-noise ratio q_{π^*} .

very close to its minimum value in the whole range of signal-to-noise ratios. At x = 0.5 or x = 4 quantity S_{3HB} is considerably increased.

On this we complete our study of the radio channel of incoherent, goniometers, which together with antennas comprise the discriminator of the goniometer system, and we turn to investigation of accuracy of tracking goniometers.





-656-

§ 11.7. Analysis of Accuracy of Incoherent Goniometers

Let us calculate errors of measurement of angular coordinates of a target by tracking radar goniometers using an incoherent signal. These calculations contain nothing new in principle as compared with the coherent case: it is necessary only in too mass § 10.12 to replace the characteristics of the radio channel for the coherent case by characteristics for the incoherent case. In view of this all laws relating to accuracy of measurement of angles fixed in § 10.12 are preserved and do not need repeated discussion. Therefore, we shall limit ourselves in the present chapter basically to consideration of illustrative examples.

First of all we must consider the influence of an AGC system. As we know, under the influence of an AGC system there is established a certain transmission factor of the radio channel, depending on the signal-to-noise ratio. There are no peculiarities here as compared to coherent gonicmeters. The system of automatic gain control in single-channel incoherent circuits is closed from the output of the filter of the radio channel. In the case of multichannel radio channels the system of automatic gain control is closed either from the output of one of the filters, or the input voltage of the AGC will be formed by means of summation of the output voltages of the filters. In all cases, as it is easy to see, the transmission factor of the radio channel will be expressed by formula (10.13.1), which not is best reduced to form

$$K_{R} = \frac{K_{R0}}{1 + \frac{\Delta f_{T}T_{r}}{q}} = \frac{K_{R0}}{1 + \frac{y}{q}},$$
 (11.7.1)

where q - ratio of energy of the signal for the period of repetition to the spectral density of noise at the input of the AGC system;

 ${\rm Af}_{\rm V}$ – effective bandwidth of the amplifier covered by the AGC loop.

Let us turn to consideration of examples of calculation of errors of measurement in trucking incoherent goniometers. Let us stop on calculation of fluctuating error. Assume we have a goniometer with IAC, whose radio channel is constructed according to the circuit of Fig. 11.6. (circuit with multiplication of signals) with identical channels. We assume that smoothing circuits are linear with constant parameters and have gransfer function $H_T(p)$ (10.42.3).

Fluctuating error in such a goniometer will be expressed, obviously, by formula

$$\Delta_{\phi a}^{2} = 2\Delta f_{\phi \phi} S_{BKB} = \frac{T_{r}}{\mu_{a}^{2}} \left(\frac{1}{q'_{2}} + \frac{1}{q'_{2}} \right) \sigma_{a} - \frac{\frac{1}{1 + \frac{y}{q'_{2}}}}{4 - \beta_{c}^{2}}$$

We use here formula (11.4.7) for S_{QKB} and (10.12.18) for of σ_{T}^{\pm} . The graph of the dependence of σ_{QKI}^2 on q_{Σ}^{\dagger} for different values of y is shown in Fig. 12.12 (for $\varphi_{0} = 30^{\circ}$). As can be seen from this figure, σ_{QKI}^2 rather sharply drops with growth of the signal-to-noise ratio. Here the best results are ustained for large y, i.e., for a wide passband of the filter. However, the dependence on y is very weak, especially for large signal-to-noise ratios.

Dynamic errors during processing of random or nonrandom inputs are expressed, obviously, by the same formulas as in the observed case [see formulas (10.12.21) and (10.12.25)]. Inasmuch as the slope of the discrimination characteristic $K_{\rm H}$ in incoherent circuits depends on the signal-to-noice ratio q exactly as it depended on h in coherent circuits, dynamic errors will be the same, only in the corresponding copressions it is necessary instead of h to place q.







are identically transferable to the considered ease, so that a ditional considered case, so that a ditional goniometer is completely like the coherent one with respect to nonlinear phenomena occurring during tracking of targets, inasmuch as the given discrimination characteristics of incoherent goniometers simply coincide with discrimination characteristics of coherent goniometers, and the second second and the second s

§ 11.8. Influence of Interferences on Incoherent Goniometers

We turn to a study of the influence of interferences on incoherent goniometers [70]. As compared with the coherent case there are no essential peculiarities here. True, the passbands of filters in the radio channels of incoherent goniometers are considerably wider than in coherent goniometers. Therefore, equivalence of certain interferences to white noise, fixed for concrent goniometers, here occurs with greater strains. We shall consider these questions in greater detail.

-658-

11.8.1. Influence of Broad-Band Interferences on Side Lobes Active interferences usually have a spectrum width considerably exceeding the spectrum width of the sounding signal. Consequently, even for radio channels of incoherent goniometers such interferences can be considered equivalent to white noise. If a broad-band interference influences a side lobe, it does not carry information and is equivalent simply to natural noises of the receiver. Therefore, the presence of such interference can be sufficiently accurately accounted for by means of introduction of a new spectral density of noise, equal to $N_0 + N_{\Pi}$, where N_{Π} - spectral density of interference at the input of the radar receiver. It is easy to see that the spectral density of interference N $_{\Pi}$ will be expressed by absolutely the same formulas as for the coherent case: for noise interference we have formula (10.13.1); for pulse chaotic, formula (10.13.2).

Accuracy of incoherent goniometers is given here by formulas derived in the preceding sections with replacement of signal-to-noise ratio $q = \frac{P_c T_r}{2N_o}$ by

$$q_{\rm II} = \frac{P_c T_r}{2(N_{\bullet} + N_{\rm II})}.$$

Hence for weak interferences (and weak natural noises), when $q_{\Pi} \gg 1$, noise immunity of coherent and incoherent goniometers against the considered interference. is identical. For strong interferences, when $q_{\Pi} \ll 1$, incoherent goniometers will have an equivalent spectral density of error, larger than the coherent by a factor of approximately $\frac{1}{T_{\Pi}\Delta\Gamma_{0}}$ i.e., losses here may be very considerable.

11.8.2. Influence of Broad-Band Interferences From the Target

Broad-band interference radiated from a target also is equivalent to white noise. However this interference carries in itself angular information, by which it essentially differs from interference having effect on side lobes. In view of the considerable excess of interference from the target over the signal, the latter is usually suppressed through the AGC system, and tracking is conducted only by the interference. Let us calculate the accuracy of incoherent goniometers during work on broad-band interference from the target. We must not directly use formular of the preceding sections giving accuracy of measurement by a fluctuating right here, since in priving these formulas the width of the spectrum of fluctuations war assumed to siderably narrower than the passband of the filters. For broad-band interface we have the reverse case, so that calculation here should be performed anew.

-659-

Let us consider first the method of pattern scanning. Calculation of accuracy here can be produced proceeding from formulas (11.3.4) and (11.3.4[']), if we replace the correlation function of signal fluctuations $P_{CP}(t - s)$ by N $_{\Pi}$ S(t - s), and also set u(t) = 1, N_O = 0 (i.e., natural noises are absent, and instead of the signal there acts white noise). With these replacements calculations are essentially simplified, and it is easy to obtain the following:

$$S_{\bullet KB} = \frac{T_{\star}}{2\mu_{B}^{2}\Delta f_{\Phi}^{2}V(0)^{*}} \int_{-\infty}^{\infty} \int_{-\infty}^{\infty} \int_{-\infty}^{\infty} |H(i\omega_{1})H(i\omega_{2})|^{*} \times$$

$$\times |V(i\omega_1 + i\omega_2)|^{\circ} d\omega_1 d\omega_2,$$

where $V(i\omega)$ is given by formula (11.3.7);

H(iw) is frequency response of the filter;

 $\Delta f_{\hat{m}}$ is its effective passband;

T_r is period of repetition of the signal.

If the filter band is so narrow that it integrates gate pulses proceeding to it, then, considering $|H(i\omega)|^2 = 2\pi \Lambda f_{\tilde{\omega}} - \delta(\omega)$, we obtain

$$S_{\text{eKB}} = \frac{T_r}{2\mu_a^2}.$$
 (11.8.1)

atulul tadi si ada si atu taku ta**di sa kada sa kada** s**a kada su kada su kada s**a kada su na

This formula coincides with formula (10.13.4), occurring for the coherent circuit, if we set $\Delta f_{\phi} = \frac{1}{T_r}$. If the filter is so broad-banded that it passes gate pulses with minute distortion, then instead of (11.8.1) we have an essentially different result. Considering in this case

$$|V(i\omega)|^{2} = 2\pi \delta(\omega) \int_{0}^{T_{r}} |v(t)|^{4} dt,$$

we obtain

$$S_{\Theta KB} = \frac{T_r}{2\mu_{\Phi}^2 \Delta f_{\Phi}^2 V(0)^3} \int_{0}^{T_r} |v(t)|^4 dt \cdot \frac{1}{2\pi} \int_{-\infty}^{\infty} |H(i\omega)|^4 d\omega.$$

With a square approximation of gate pulses and the frequency response of the filter we have

$$S_{\text{SRB}} = \frac{T_{\text{c}}}{2\mu_{a}^{2}\Delta f_{\phi}\tau_{o}}, \qquad (11.8.2)$$

where τ_{c} - duration of a gate pulse.

The absolute value of error when tracking a target by broad-band interference radiated from it can easily be found from the derived formulas. For instance, at

-660-

 $T_r = 10^{-3}$ sec, $\mu_a = 0.2$ 1/deg by (11.8.1) we find $S_{BBB} \approx 0.01$ deg²/cps.

Let us consider now the influence of broad-band interference from the target on goniometers using methods of scanning with compensation, IAC and IPC. The rules here are the same as for coherent goniometers. When radio channels have identical channels, tracking by interference occurs without fluctuating error (if we do not consider parametric fluctuations). With nonidentical channels in circuits with subtraction of signals there appear fluctuating and sometimes systematic errors. If we assume the filters in the circuits have a sufficiently narrow band so that gate pulses proceeding to them are integrated, the formula for equivalent spectral densities of errors can be obtained from the corresponding formulas for the coherent case (see § 10.13) by replacement of Δf_{ij} by $1/T_r$. Systematic errors have absolutely identical form for coherent and incoherent circuits.

Thus, for circuits of the method of scanning with compensation and IAC with subtraction of signals we have

$$S_{\text{DRB}} = \frac{T_{\text{P}}}{4\mu_{\text{q}}^2} \left(\frac{1-k^2}{1+k^2}\right)^2,$$

where k - ratio of squares of gain factors in the channels.

Systematic error exists only for IAC and is expressed by formula (10.13.18).

Analogously for an IPC circuit without formation of sum and difference signals, with nonidentical channels we have equivalent spectral density

$$S_{\Theta KB} = \frac{3T_r}{4\left(\frac{\pi d}{\lambda}\right)^3} \, \mathrm{tg}^2 \Delta \varphi,$$

where $\Delta \phi$ - difference of phases of signals at the output of the channels, caused by nonidentity of channels.

Systematic error, as before, is expressed by formula (10.13.21). Absolute values of errors here are of the same order as errors during work on a fluctuating signal with considerable excess of signal over noise. From what has been said we can conclude that interference radiated from a target for goniometer channels of radars does not present danger, and tracking by such interference occurs with fully permissible errors. It is necessary, of course, to mention that for more sure tracking of interference in a number of cases we should take certain special measures: widening of the gate pulses (with broad-band filters) and sutoff of noise automatic gain control (ShARU), if we have it (since ShARU would reduce the level of interference to the level of natural noises). These measures depend on concrete conditions.

--661-

11,8,3, Influence of Active Interferences with Narrow-Band Amplitude Modulation

والماليكية والعلا كالألالا الأعافية فالمصاب وال

त्यं कार्यन्तर्भवन्तं प्रसिद्धं विद्युति के प्रदर्शने के **अधिके के विद्युति कि कार्यन**्ति कि विद्युति हिन्द्य कि

Let us consider the question of the influence of active interferences with operial narrow-band amplitude modulation, having a spectrum near the frequency of scanning, on goniometers using methods of pattern scanning. The calculations and results here absolutely do not differ from what we had for coherent goniometers, and equivalent spectral density is expressed by the known formula (10,13,25). Errors here, as shown in § 10.13, may be very considerable.

Analogously for the method of scanning with compensation we can obtain results completely coinciding with results for coherent goniometers: in a circuit with multiplication of signals fluctuating error during tracking by amplitude-modulated interference is equal to zero if we disregard parametric fluctuations; the spectral density of parametric fluctuations with identical channels is given by formula (10, 3, 27). The same thing takes place in a circuit with subtraction of signals with identical channels. With nonidentical channels in a circuit with subtraction $S_{\rm 0KB}$ is different from zero and is expressed by formula (10, 3, 26). Error here depends on the degree of nonidentity of channels and usually is rather small. Consequently, interference with narrow-band amplitude modulation (3 very dangerous for radars using the method of pattern scanning and practically does not affect radars using the method of scanning with compensation.

The influence of powerful intermittent interferences, exciting in the radar receiver prolonged transients, was qualitatively studied for the coherent case in 040.13.4. Everything presented there is completely transferable to incoherent circuits. Quantitatively, of course, there will be certain differences, due to difference of the parameters of coherent and incoherent circuits; however, the influence of intermittent interferences on goniometer devices practically does not yield to theoretical analysis, and the problem of obtaining quantitative characteristics of the influence of such interferences, as also in the coherent case, has not been studied.

With respect to passive interference it is possible to say the following. It is known that noise immunity of incoherent radars with respect to passive interference is very low if there is not provided special measures of protection from this interference (see, for instance, Chapters V, VII and Paragraph 8.10.4). Incoherent radar goniometers are not an exception in this respect. The influence of passive interference on incoherent goniometers practically always lead to sharp increase

-662-

of errors of measurement and breakoff of tracking.

€ Ē

Theoretical analysis of the influence of passive interferences on incoherent goniometers therefore is deprived of practical interest, and we shall not study it.

§ 11.9. Conclusion

In the two preceding chapters we studied reder goniometers using coherent and incoherent signals. What conclusions can be drawn with respect to comparison of these two cases? Comparison of circuits of optimum radio channels of coherent and incoherent goniometers shows that these circuits are similar in general. Their structure in both cases is identical, and the whole difference reduces only to difference of the characteristics of the filters. Filters of coherent circuits are comparatively harrow-banded and ensure integration of the signal during the time of its correlation; filters of incoherent circuits usually are considerably brokderbanded and ensure integration of the period of its repetition.

We stop now to discuss the relationship of accuracy of coherent and incoherent goniometers. The preceding analysis showed that with large signal-to-noise ratios accuracies of incoherent and coherent goniometers completely coincide, i.e., with a large signal-to-noise ratio the form of the signal does not affect accuracy of goniometer. With small signal-to-noise ratios incoherent goniometers have considerable losses of accuracy as compared to coherent. The ratio of the equivalent spectral densities of coherent and incoherent circuits for small signal-to-noise ratios have the order $T_{p}\Delta f_{c}$, i.e., lowering of accuracy due to incoherence of the signal is approximately proportional to the number of periods of repetition of the signal in the interval of its correlation. From this there ensures lower noise immunity of incoherent goniometers as compared to coherent against interferences similar to natural noises of receivers (broad-band interferences in side lobes of the direction of the signal-to-noise ratio in the goniometer, which will lead in incoherent goniometers to greater decrease of accuracy than in coherent goniometers.

A very great deficiency of incoherent goniometers, greatly narrowing the domain of their application, is their very low immunity against passive interference. Here coherent goniometers have a declaive advantage over incoherent ones.

We shall now discuss certain questions concerning incoherent gonioseters when require further investigation. In the first place here one should place the problem of exact synthesis of optimum gonicmeters using incoherent radiation.

-663-

This problem was solved by us only with certain rather limiting assumptions. Exact solution of this problem is of considerable interest. The remaining unsolved questions concerning incoherent geniometers are basically the same as were formulated in the concluding section of the preceding chapter: synthesis of geniometers without preliminary assignment of the method of direction finding, synthesis of geniometers with comparable times of correlation of angular shifts of the target and of the reflected signal, study of statistical regularities of angular shifts of the target and synthesis of optimum smoothing circuits, investigation of nonlinear phenomena in tracking incoherent geniometers.

والمتحدث والمتحد والمعار والمعار والمتحصين والمتحدين والمتحدين والمحافظ والمعالية والمعالمة والمعالية

diát

العليا فتكتر تركفت الد

CHAPTER XII

JOINT MEAJUREMENT OF SEVERAL COORDINATES

§ 12,1 Introduction

In Chapters VI-XI we considered separately measurement of each radar coordinate. Theoretically this is permissible either when finding approximate solutions or when all remaining coordinates besides the measured one are constant or change in a cortain way. Therefore, of simultaneously great practical and theoretical interest are questions of analysis and synthesis of joint meters of several coordinates. Only complicated theoretical consideration can show how exact methods of analysis and synthesis developed in idealized premises of one measured coordinate are.

With development of radar technology questions of multi-dimensional measurement become more and more urgent. Thus, in early surveillance radars they usually measured two parameters, corresponding to bearing and range. For a long period they used incoherent radars of varying assignment, measuring range, bearing and elevation. With the appearance of coherent techniques as a fourth parameter there began to appear radial speed. Finally, at present they predict the appearance of n-limerational radar installations in which among the measured quantities there are included angular velocities, radial acceleration, characteristics of the form and fluctuations of the signal, and so forth.

Inasmuch as joint meters are known in radar practice, the theory of joint measurement, as in Chapter VI, is reasonably started from analysis of joint meters of assigned structure. This in brief is done in § 12.2. However, a whole series of questions, for instance, concerning interconnections of codings of measured coordinates in the signal, can be comprehended with difficulty of the basis of analysis atone, without finding adequate potential characteristics of this

-665-
interconnection. Therefore basic attention in this chapter is paid to questions of synthesis of joint meters, to which we devote §§ 12.3-12.6. We synthesize complete elecuties of optimum joint meters (§ 12.3). First we take Gaussian distribution of all parameters, but make no concrete assumptions about the structure of input signals and correlation properties of the parameters. We consider also the method of synthesis with Markovian parameters. We separately study questions of synthesis of optimum discriminators (§ 12.4) and smoothing circuits (§ 12.5). We give several examples of construction of joint meters (§ 12.6). During selection of these examples the authors were guided by considerations of their applied interest and maximum detection of the specifics of joint measurement.

Theoretical propositions mentioned below are far from complete and can be considered the direction for new investigations. However, for instance, measurement of several coordinates of a single target can be considered on the basis of the theory mentioned below fairly rigorously and in detailed form.

§ 12.2. Analysis of Multi-Dimensional Tracking Meters

Multi-dimensional meters known in practice usually are constructed in the form of multi-dimensional tracking systems. In the literature there exists a whole series of works on tracking systems with a great number of tracked quantities (e.g., [54]). However, the specific character of radar meters, of which we talked in Chapter VI, forces us to examine from a new point of view questions of analysis of tracking systems in application to meters. A number of results of analysis considering the appearing peculiarities are given below.

12.2.1. Basic Features of Construction of Circuits and Components of Measuring Errors

Generally, a multi-dimensional tracking meter for *l* parameters can be presented by the block diagram of Fig. 12.1. In it there are marked two basic types of clements - discriminators and smoothing circuits with drives. Discriminators whose outputs are numbered from 1 to *l* detect mismatch between the input and output values of some parameters and are nonlinear radio devices for processing one or several input signals. A peculiarity of the partial discriminators in a multi-dimensional meter is the fact that to each of them there proceed measured values of all *l* coordinates, and not only that by which the error signal is detected. This is explained by the fact that radar coordinates are coded in the useful component of the input mixture in such a connected form that during processing of the signal by narrowlyselective devices, close to optimum, disturbance of selection with respect to any

-666-



parameter makes it impossible to detect information immediately on all parameters.

Fig. 12.1. Multi-dimensional meter: 1 - partial discriminators; 2 - linear smoothing circuits; 3 - adders.

Shouthing elecuits corrying out accumulation of primary data are usually built from linear elements separately for all 1 coordinates. However, we also meet smoothing circuits with cross ecuplings. Most frequently these couplings are realized in the form of inertialess devices of recalculation of coordinates immediately at the cutput of the discriminator, but we can meet more complicated cases. In the general plan we have the matrix of smoothing circuits depicted in Fig. 12.1. The matrix has 1 inputs and 1 outputs, and coupling between any input and output is possible. Usually the matrix is linear, but it can also consist of inertial nonlinear elements. Nonlinear circuits have not yet received wide use in practice, and subsequently will not be studied.

Besides discriminators and smoothing circuits in meters there sometimes is provided introduction through adders of additional quantities, intended to compensate known components of the measured parameters. In order to have before oneself graphic examples of practical realization of joint meters on the part of the discriminators, it is sufficient to combine a set of circuits similar to those studied in Chapters VII-vii and pertaining to measurement of various coordinates. Here it is necessary asticipate selection in all circuits with respect to all coordinates, and tech-

We infreduce a system of designations similar to that adopted in the one-dimenstand analysis, starting with the case of one input mixture:

 $y[l; \lambda^{(1)}(l), \ldots, \lambda^{(l)}(l)] = y[l; \lambda(l)],$

where $\{\lambda^{(1)}(l), \ldots, \lambda^{(1)}(l)\} = \lambda(l)$ - set of measured parameters (coordinates),

At the output of each discriminator there will be formed a function of time (usually in the form of voltage)

$$z^{(a)}[t; \bullet(t)] (a = 1, ..., l),$$

alle ar a attach a dh' da an alle da bhairte chan

where $\mathbf{e}(t) = {\mathbf{e}^{(1)}(t), \dots, \mathbf{e}^{(1)}(t)}$ - current mismatches between $\lambda(t)$ and measured values of parameters $\hat{\lambda}_{i}(t) = {\hat{\lambda}^{(1)}(t), \dots, \hat{\lambda}^{(n)}(t)}$, so that

$$\mathbf{a}^{(a)}(l) = \lambda^{(a)}(l) - \hat{\lambda}^{(a)}(l) \quad (a = 1, ..., l).$$

The dependence in general of each of the output voltages of the discriminator on mismatch with respect to all parameters is a purely multi-dimensional effect. It is not worth explaining that, as in the one-dimensional case, values of $\lambda(t)$ and $\boldsymbol{\varepsilon}(t)$ determine input and output voltage of discriminators only in the statistical sense: concrete realizations of $y[t; \lambda(t)]$ and $z^{(\alpha)}[t; \boldsymbol{\varepsilon}(t)]$ are determined by input fluctuating disturbances.

The number of input mixtures m in general can be arbitrary. It usually is not connected with the number of measured quantities. In conditions of several mixtures recording of discriminator inputs alone is complicated, i.e., we consider set $\mathbf{y}(t) = \{ {}^{(1)}\mathbf{y}(t), \ldots, {}^{(m)}\mathbf{y}(t) \}$. Nevertheless, the remaining designations as well as the method of analysis of meters are kept constant.

Let us stop to classify different components of errors of measurement. These components are basically the same as in the one-dimensional case. One of the components is fluctuating errors due to internal noises of the receiver, natural and organized interferences, and fluctuations of the reflected radar signal. A peculiarity of the multi-dimensional case is that similar errors detected at the output of different partial discriminators sometimes turn out to be correlated. The reason for this is not so much the common source of disturbances for the discriminators (corresponding component of the input mixture) as the similar method of encoding certain coordinates in the signal observed sometimes. A thorough understanding of this effect will be possible only on the basis of the findings of § 12.3.

In multi-dimensional meters there are also dynamic errors due to change of the measured quantities themselves and fluctuating and systematic errors introduced by equipment for processing of signals. However we shall consider only fluctuating and dynamic errors. With respect to the character of fluctuating disturbances,

-668-

which sometimes are conveniently related to immaterial randomly varying parameters of the signal, we shall limit ourselves to the assumption that all of them vary considerably faster than the measured variables.

12.2.2. Characteristic of Discriminators

In conditions of rapid random disturbances the output voltage of every discriminator $z^{(\alpha)}[t, \epsilon(t)]$ can be presented in the form of two parts: the mean

$$2^{(a)}(s, t) = \overline{z^{(a)}(t, s)}$$
 (12.2.1)

and the second second

and white noise with (two-way) spectral density

$$= \int_{-\infty}^{+\infty} \frac{S^{(a)}(t, t) =}{[z^{(a)}(t, t)][z^{(a)}(t+\tau, t)-z^{(a)}(t+\tau, t)]} d\tau, \qquad (12.2.2)$$

equal to the spectral density at low frequencies of the fluctuating component of function $z^{(\alpha)}(t, \epsilon)$. Averaging in (12,2,1) and (12,2,2) is produced over the complete ensemble of fluctuations, and quantities ϵ here are considered "frozen" in accordance with the assumption of slowness of their variation. The first peculiarity of the multi-dimensional case is the presence of functional dependence of $a^{(\alpha)}(\epsilon, t)$ and $S^{(\alpha\alpha)}(\epsilon, t)$ on all mismatches. We return to this circumstance below. A second peculiarity is the necessity of introduction besides (12,2,2) of characteristics of crosscorrelation coupling of fluctuating components at the output of different partial discriminators. Coupling between the α -th and β -th output voltages is characterized by mutual spectra) density

$$S^{(n+)}(s, t) = S^{(p_n)}(s, t) = \int_{-\infty}^{+\infty} [z^{(n)}(t, s) - z^{(n)}(t, s)] [z^{(p)}(t+\tau, s) - z^{(h)}(t+\tau, s)] d\tau, (12.2.3)$$

which, depending upon character of the coupling, can be positive, negative or zero. The last case is observed in the absence of correlation coupling of the considered disturbances, by no means signifying their complete independence.

Thus, we have a square symmetric matrix $(l \times l)$ of characteristics of fluctuating disturbances, in which in all there are l(l + 1)/2 independent elements. The statistically equivalent form of recording output voltages of the discriminator is

$$\boldsymbol{z}_{\boldsymbol{\theta}}^{(\alpha)}(t, \boldsymbol{\theta}) = \boldsymbol{a}^{(\alpha)}(\boldsymbol{\theta}, t) + \boldsymbol{V} \, \overline{\boldsymbol{S}^{(\alpha\alpha)}(\boldsymbol{\theta}, t)} \, \boldsymbol{\zeta}^{(\alpha)}(t), \qquad (1 \, \mathcal{C}^{(\alpha)})^{-1} \, \boldsymbol{\zeta}^{(\alpha)}(t),$$

where $\xi^{(\alpha)}(t)$ - white noises with unit spectral density, the interconnection of which is established by relationship

$$\int_{-\infty}^{+\infty} \overline{\zeta^{(a)}(t)} \zeta^{(2)}(t+\tau) d\tau = \frac{S^{(a2)}(e, t)}{\sqrt{S^{(au)}(e, t)} S^{(b2)}(e, t)}.$$

Functions $a^{(\alpha)}(\varepsilon, t)$ we shall call discrimination characteristics, and $s^{(\alpha\beta)}(\varepsilon, t) - fluctuation$ characteristics. We shall find them by means of analysis of passage of the signal and noises through a discriminator with fixed mismatches ε .

The form of discrimination and fluctuation characteristics in an 1-dimensional parameter space is very complicated. We shall first of all give a qualitative description of these dependences in that particular case when each a-th output of the discriminator on the average is determined by mismatch with respect to the a-th parameter, and all other mismatches have an effect only if their magnitude is great, determining several scale factors. Simultaneously we shall consider absent correlation couplings between noises at the output of the discrimination character-istic for fixed $\epsilon^{(\beta)}(\beta \neq \alpha)$ is described by an odd function of $\epsilon^{(\alpha)}$ of the same form as in the one-dimensional case. Values of all other mismatches basically affect the scale of the curve. The scale is maximum at $\epsilon^{(\beta)} = 0$ ($\beta \neq \alpha$) and varies little for small $|\epsilon^{(\beta)}|$. For large values of $|\epsilon^{(\beta)}|$, $a^{(\alpha)}(\epsilon, t)$ asymptotically seeks zero independently of the value of $\epsilon^{(\alpha)}$. For the case of two parameters the first discrimination characteristic in plane $(\epsilon^{(1)}, \epsilon^{(2)})$ is graphically presented in Fig. 12.2.



Fig. 12.2. Discrimination characteristic of a twodimensional meter.

Fluctuation characteristic $s^{(\alpha\alpha)}(\boldsymbol{\epsilon}, t)$ in the same conditions with fixed $\varepsilon^{(\beta)}(\beta \neq \alpha)$ is described by a symmetric curve of a form which is known from the one-dimensional case. With growth of other mismatches the double-humped structure of the considered sections is smoothed and $s^{(\alpha\alpha)}(\boldsymbol{\epsilon}, t)$ for any values of $\varepsilon^{(\alpha)}$ changes into a hyperplane, parallel to the coordinate hyperplane $\boldsymbol{\epsilon}$. Fluctuation characteristic $s^{11}(\varepsilon^{(1)}, \varepsilon^{(2)})$ on plane $(\varepsilon^{(1)}, \varepsilon^{(2)})$ is shown in Fig. 12.3.



Fig. 12.3. Fluctuation characteristic of a two-dimensional meter.

The illustration helps us to comprehend the law that increase of mismatch $\varepsilon^{(\beta)}$ ($\beta \neq \alpha$) leads to the same change of the discrimination $a^{(\alpha)}(\varepsilon)$ and fluctuation $S^{(\alpha\alpha)}(\varepsilon)$ characteristics, constructed as a function of $\varepsilon^{(\alpha)}$, as decrease of the signal-to-noise ratio. This is understandable inasmuch as increase of mismatch for other coordinates reads to power losses, fully equivalent to decrease of the mean input signal level.

when all mismatches in the considered particular case are small, instead of general dependences (12.2.4) It is permissible to use expansions of characteristics of discriminators

$$a^{(\alpha)}(\mathbf{e}) = a_0^{(\alpha)} + K_{\mathbf{g}}^{(\alpha)} \mathbf{e}^{(\alpha)} + \sum_{\beta,\tau} Q^{(\beta\tau)} \mathbf{e}^{(\beta)} \mathbf{e}^{(\tau)} + \dots,$$

$$S^{(\alpha\alpha)}(\mathbf{e}) = S_0^{(\alpha\alpha)} + S_1^{(\alpha\alpha)} \mathbf{e}^{(\alpha)} + S_2^{(\alpha\alpha)} (\mathbf{e}^{(\alpha)})^2 + \dots,$$

where

$$a_{0}^{(a)} = a^{(a)}(0); \ K_{x}^{(a)} = \frac{\partial a^{(a)}(0)}{\partial s^{(a)}}; \ Q^{(3_{1})} = \frac{1}{2} \ \frac{\partial^{3} a^{(a)}(0)}{\partial s^{(b)} \partial s^{(1)}}; \\ S_{0}^{(aa)} = S^{(aa)}(0); \ S_{1}^{(aa)} = \frac{\partial S^{(aa)}(0)}{\partial s^{(a)}}; \ S_{2}^{(a_{1})} = \frac{1}{2} \ \frac{\partial^{3} S^{(aa)}(0)}{\partial s^{(a)}}.$$

In conditions of symmetry of the circuit $a_0^{(\alpha)}$, $S_1^{(\alpha\alpha)}$ are usually equal to zero. Quantities $K_A^{(\alpha)}$ are called gain factors of the discriminators. Coefficients $Q^{(\beta\gamma)}$ show the influence of other mismatcher, in certain approximation leading to decrease of the scale factor. During investigation of accuracy of measurements by virtue of the smallness of mismatches it is often sufficient to use formulas

$$a^{(a)}(e) \approx K_{g}^{(a)}e^{(a)},$$

$$S^{(aa)}(e) \approx S_{0}^{(aa)} + S_{2}^{(aa)}e^{(a)2},$$
(12.2.5)

which are a simple repetition of one-dimensional (6.2.9).

-671-

Thus, in the case of small errors, when linear approximations are valid, the separate analysis of meters of separate parameters given in preceding chapters is fully permissible.

An example of a multi-dimensional meter which satisfies the formulated conditions is a joint meter of range and speed in a coherent radar. If, for instance, we are interested in the range discriminator, it contains narrow-band filters (Figs. 7.7, 7.13) which are tuned from the output of the speed channel. With small errors of measurement of speed the signal frequency is within the limits of the frequency response of the filter.

In a number of cases the described dependences of $a^{(\alpha)}(\epsilon)$ and $S^{(\alpha\beta)}(\epsilon)$ on ϵ arinvalid. Roughly speaking, this distinction consists of zero shift of the α -th discriminator in the presence of mismatch for the β -th parameter ($\beta \neq \alpha$). Changes of dependences appear also with large mismatches. Inasmuch as general description, even qualitative, is hampered here, we shall only discuss the case of small mismatches. The generalization of formulas (12.2.5) then will be

$$a^{(a)}(\mathbf{t}) = a_0^{(a)} + \sum_{\beta=1}^{7} K_{\beta}^{(\alpha\beta)} \mathbf{t}^{(\beta)}, \qquad (12.2.6)$$

$$S^{(a\beta)}(\mathbf{e}) = S_0^{(a\beta)} + \sum_{\tau=1}^{l} S_{\tau}^{(a\beta)} \mathbf{e}^{(\tau)} + \sum_{\tau,\delta=1}^{l} S_{\tau\delta}^{(a\beta)} \mathbf{e}^{(\tau)} \mathbf{e}^{(\delta)}, \qquad (12.2.7)$$

where

$$a_{0}^{(\alpha)} = a^{(\alpha)}(0); \ K_{\alpha}^{(\alpha\beta)} = \frac{\partial a^{(\alpha)}(0)}{\partial \epsilon^{(\beta)}};$$

$$S_{0}^{(\alpha\beta)} = S^{(\alpha\beta)}(0); \ S_{\gamma}^{(\alpha\beta)} = \frac{\partial S^{(\alpha\beta)}(0)}{\partial \epsilon^{(\gamma)}}; \ S_{\gamma\delta}^{(\alpha\beta)} = \frac{1}{2} \frac{\partial^{4} S^{(\alpha\beta)}(0)}{\partial \epsilon^{(\gamma)} \partial \epsilon^{(\delta)}}.$$
(12.2.8)

In conditions of symmetry of circuits of the partial discriminators, $a_0^{(\alpha)}$ and $S_{\sim}^{(\alpha\beta)}$ disappear, and there remain these simpler dependences:

 $a^{(a)}(\mathbf{s}) = \sum_{\mathbf{p} \neq \mathbf{s}}^{I} K_{\mathbf{x}}^{(a\beta)} \epsilon^{(\beta)}, \qquad (12.2.7)$

$$S^{(\alpha\beta)}(\mathbf{s}) = S_0^{(\alpha\beta)} + \sum_{\mathbf{1}, \ \mathbf{s}=1}^{I} S_{\mathbf{1}\mathbf{s}}^{(\alpha\beta)} \mathbf{s}^{(\mathbf{1})} \mathbf{s}^{(\mathbf{s})}. \tag{12.2.10}$$

Coefficient $K_{\mu}^{(\alpha\beta)}$ we call the gain factor of the α -th discriminator for the β -th parameter. Quantities $S_{0}^{(\alpha\beta)}$ characterize fluctuating components not depending on mismatch, and $S_{\gamma\delta}^{(\alpha\beta)}$ - components proportional to $\varepsilon^{(\gamma)}$ ($\gamma = 1, ..., l$). Formulas (12.2.9) and (12.2.10) correspond to the following presentation of output voltage of discriminators:

$$\mathbf{x}_{\mathbf{s}}^{(\alpha)}(t, \mathbf{s}) = \sum_{\beta=1}^{l} K_{\mathbf{x}}^{(\alpha\beta)} \eta^{(\beta)}(t) + \sum_{\alpha,\beta=1}^{l} [K_{\mathbf{x}}^{(\alpha\beta)} + \mathbf{x}^{(\alpha\beta)}(t)] \mathbf{s}^{(\beta)}(t), \qquad (12.2.11)$$

where $\eta^{(\beta)}$ and $\kappa^{(\alpha\beta)}(t)$ $(\alpha, \beta = 1, ..., l)$ - random, uncorrelated, functions; $\eta^{(\beta)}(t)$ - white noises having the dimensionality of the measured parameters with a matrix of spectral densities $\mathbf{S}_{\partial \mathbf{K}\mathbf{F}}$ determined by recalculation of matrix $\mathbf{S}_{0} = ||\mathbf{S}_{0}^{(\alpha\beta)}||$ with the help of the matrix of gain factors of discriminators $\mathbf{K}_{\mathbf{H}} = ||\mathbf{K}_{\mathbf{H}}^{(\alpha\beta)}||$ by the formula

$$\mathbf{S}_{\mathbf{0}\mathbf{K}\mathbf{a}} = \| S_{\mathbf{y}}^{(a^{\beta})} \| = \mathbf{K}_{\mathbf{x}}^{-1} \mathbf{S}_{\mathbf{0}} \mathbf{K}_{\mathbf{x}}^{-1+}.$$
(12.2.12)

For proof of formula (12.2.12) it is sufficient to multiply vector $\mathbf{K}_{\mathbf{J}} \boldsymbol{\eta}$, representing noises at the discriminator output, by row $(\mathbf{K}_{\mathbf{J}} \boldsymbol{\eta})^+$, average, and, passing to spectra, use the designation for the spectral matrix of vector $\boldsymbol{\eta}$:

$\mathbf{S}_{\mathbf{0}} = \int \overline{\mathbf{K}_{\mathbf{A}} \boldsymbol{\eta}(t) \left(\mathbf{K}_{\mathbf{A}} \boldsymbol{\eta}(t-\tau)\right)^{+}} d\tau = \mathbf{K}_{\mathbf{A}} \int \overline{\boldsymbol{\eta}(t) \boldsymbol{\eta}^{+}(t+\tau)} d\tau \mathbf{K}_{\mathbf{A}}^{+} = \mathbf{K}_{\mathbf{A}} \mathbf{S}_{\mathbf{0}\mathbf{K}\mathbf{B}} \mathbf{K}_{\mathbf{A}}^{+}.$

From this (12.2.12) is obtained by two matrix multiplications. Subsequently we shall call \mathbf{S}_{3KB} the matrix of equivalent spectral densities of a multi-dimensional discriminator.

Functions $\pi^{(\alpha\beta)}(t)$ in (12.2.11) are white noises with mutual spectral densities

$$\|S_{1b}^{(ap)}\| = \|\int_{-\infty}^{+\infty} x^{(a_1)}(t) x^{(b_1)}(t+s) ds\|.$$
(12.2.13)

Subsequently we will not consider the influence of parametric fluctuation $\varkappa^{(\gamma \delta)}(t)$ on accuracy of measurement, concentrating our attention on noises $\eta^{(\alpha)}(t)$. Their level, naturally, is determined by the intensity of interfering signals at the input, not carrying information about the measured quantities. Basically, for $\gamma^{(\alpha)}(t)$ there are observed the same laws as in the one-dimensional case.

For illustration of the method of using the obtained relationships we shall consider the fairly simple example* of a discriminator for two parameters with interconnected encoding in the signal.

Let us assume that the signal has a pulse structure with fluctuations of amplitude E and phase φ , independent between periods. It is received against a background of white noise n(t). Pulses have intrapulse linear frequency modulation (LChM). The realization of the received mixture of the signal with noise in one period has the form

*Example belongs to G. F. Bugachev.

$$y(t) = \operatorname{Re} \left[\mathcal{E} e^{t \varphi} u_{\mathbf{a}} \left(t - \tau \right) e^{i V \left(t - \tau \right) t / 2 + i \left(\omega_0 + \omega_{\mathbf{A}} \right) t} \right] + n(t),$$

where V = slope of LChM;

 τ - delay of regular modulation;

 ω_{π} - Doppler frequency shift.

The last two quantities are also the measured parameters.

The circuit of a joint discriminator of quantities τ and $\omega_{\rm g}$ is depicted in Fig. 12.4. Signal y(t) is fed to mixer 1, where convolution of spectrum LChM is



Fig. 12.4. Discriminator of $\omega_{\rm pl}$ and τ : 1 - mixer; 2 - controlled heterodyne oscillator; 3, 4 bandpass filters tuned to frequencies $\omega_1 \pm 5$; 5 - bandpass filter tuned to frequency ω_1 ; 6 - amplitude detectors; 7 gated amplifiers; 8 - time discriminator; 9 - comparison circuit; 10 - generator of gate pulses.

effected by means of multiplication by the vol-

tage of heterodyne 2

Re $[e^{iV(t-\hat{\tau})^{2}+i(\omega_{0}+\omega_{1}+\hat{\omega}_{3})t]}$.

and the second second second

a serie danas.

where $\hat{\tau}$, $\hat{\omega}_{\mu}$ - measured values of parameters.

Simultaneously there is produced transfer of oscillations to a frequency close to intermediate ω_1 . Then there is produced bandpass filtration in filters 3, 4, 5, of which filter 5 is exactly tuned to ω_1 , and filters 3, 4 have detuning of ± 0 . Results of filtration are detected in detectors 6. In the upper two channels of Fig. 12.4 in elements 7 there is produced gating; the middle of the gate pulse coincides with $\hat{\tau}$. The difference of the two

formed voltages, formed in element 9, is the first output voltage of the joint discriminator $z^{(1)}$. This voltage is sensitive to frequency shift of voltage at the output of the mixer.

In the lower channel of Fig. 12.4 after the detector there is time discriminator 3. Its action reduces to multiplication by odd function i(t), whose zero also coincides with $\hat{\tau}$, and subsequent accumulation (integration) of signals.

Consequently, there will be formed the second cutput voltage $z^{(2)}$, sensitive primarily to time delay τ :

According to the above results for determination of matrix Σ_{GKB} of errors of measurement of τ and $\omega_{\underline{A}}$ with respect to a unit pulse input it is necessary to calculate the matrix of gain factors of the discriminator

$$\mathbf{K}_{\mathbf{A}} = \begin{vmatrix} \frac{\partial \overline{z^{(1)}}}{\partial \Delta_{\mathbf{o}}} & \frac{\partial \overline{z^{(1)}}}{\partial \Delta_{\mathbf{q}}} \\ \frac{\partial \overline{z^{(1)}}}{\partial \Delta_{\mathbf{o}}} & \frac{\partial \overline{z^{(1)}}}{\partial \Delta_{\mathbf{q}}} \end{vmatrix} \begin{vmatrix} \Delta_{\mathbf{q}} = 0 \\ \Delta_{\mathbf{o}} = 0 \end{vmatrix}$$
(12.2.14)

where $\Delta_{_{10}} = \omega_{_{II}} - \hat{\omega}_{_{II}}$, $\Delta_{_{T}} = \tau - \hat{\tau}$, and the matrix of mean squares of fluctuating components of outputs is

$$\Sigma = \left\| \begin{array}{cc} \overline{(z^{(1)})^3} & \overline{z^{(1)}z^{(2)}} \\ \overline{z^{(1)}z^{(1)}} & \overline{(z^{(1)})^3} \end{array} \right\|_{A_{\pi} = 0}^{A_{\pi} = 0} \cdot (12, 2.15)$$

Analogously to (12.2.12) matrix Σ_{eq} is equal to

$$\mathbf{Z}_{\mathbf{x}\mathbf{x}} := \mathbf{K}_{\mathbf{x}}^{-1} \mathbf{Z} \mathbf{K}_{\mathbf{x}}^{+-1}. \tag{12.2.16}$$

Passing to calculation, for simplicity we assume that the bandpass filters are matched with amplitude modulation, i.e., their pulse response envelope is the inversion in time of function $u_a(t)$, that amplitude detectors 6 are square-law, and that amplifiers 7 carry out ultranarrow gating. With some selection of overall gain factors in channels, which do not affect final results, we have:

$$\frac{\partial \overline{z^{(1)}}}{\partial \Delta_{\alpha}} \Big|_{\substack{A_{1} = 0 \\ A_{\alpha} = 0}} = \mathcal{S} \operatorname{Re} \left\{ i \int t_{1} u_{a}^{2}(t_{1}) e^{-ibt_{a}} dt_{1} \int u_{a}^{2}(t_{2}) e^{-ibt_{a}} dt_{2} \right\},$$

$$\frac{\partial \overline{z^{(1)}}}{\partial \Delta_{\gamma}} \Big|_{\substack{A_{\gamma} = 0 \\ A_{\alpha} = 0}} = \mathcal{S} \operatorname{Re} \left\{ \left[\int u_{a}'(t) u_{a}(t) e^{-ibt} dt + iV \int t u_{a}^{2}(t) e^{-ibt} dt \right] \int u_{a}^{2}(\tau) e^{ib\tau} d\tau \right\},$$

$$\frac{\partial \overline{z^{(1)}}}{\partial \Delta_{\gamma}} \Big|_{\substack{A_{\gamma} = 0 \\ A_{\alpha} = 0}} = \mathcal{S},$$

$$\frac{\partial \overline{z^{(1)}}}{\partial \Delta_{\gamma}} \Big|_{\substack{A_{\gamma} = 0 \\ A_{\alpha} = 0}} = \mathcal{S},$$

$$\frac{\partial \overline{z^{(1)}}}{\partial \Delta_{\gamma}} \Big|_{\substack{A_{\gamma} = 0 \\ A_{\alpha} = 0}} = \mathcal{S},$$

$$\frac{\partial \overline{z^{(1)}}}{\partial \Delta_{\gamma}} \Big|_{\substack{A_{\gamma} = 0 \\ A_{\alpha} = 0}} = \mathcal{S},$$

$$\frac{\partial \overline{z^{(1)}}}{\partial \Delta_{\gamma}} \Big|_{\substack{A_{\gamma} = 0 \\ A_{\alpha} = 0}} = \mathcal{S},$$

$$\frac{\partial \overline{z^{(1)}}}{\partial \Delta_{\gamma}} \Big|_{\substack{A_{\gamma} = 0 \\ A_{\alpha} = 0}} = \mathcal{S},$$

$$\frac{\partial \overline{z^{(1)}}}{\partial \Delta_{\gamma}} \Big|_{\substack{A_{\gamma} = 0 \\ A_{\alpha} = 0}} = \mathcal{S},$$

$$\frac{\partial \overline{z^{(1)}}}{\partial \Delta_{\gamma}} \Big|_{\substack{A_{\gamma} = 0 \\ A_{\alpha} = 0}} = \mathcal{S},$$

$$\frac{\partial \overline{z^{(1)}}}{\partial \Delta_{\gamma}} \Big|_{\substack{A_{\gamma} = 0 \\ A_{\alpha} = 0}} = \mathcal{S},$$

$$\frac{\partial \overline{z^{(1)}}}{\partial \Delta_{\gamma}} \Big|_{\substack{A_{\gamma} = 0 \\ A_{\alpha} = 0}} = \mathcal{S},$$

$$\frac{\partial \overline{z^{(1)}}}{\partial \Delta_{\gamma}} \Big|_{\substack{A_{\gamma} = 0 \\ A_{\alpha} = 0}} = \mathcal{S},$$

$$\frac{\partial \overline{z^{(1)}}}{\partial \Delta_{\gamma}} \Big|_{\substack{A_{\gamma} = 0 \\ A_{\alpha} = 0}} = \mathcal{S},$$

$$\frac{\partial \overline{z^{(1)}}}{\partial \Delta_{\gamma}} \Big|_{\substack{A_{\gamma} = 0 \\ A_{\alpha} = 0}} = \mathcal{S},$$

$$\frac{\partial \overline{z^{(1)}}}{\partial \Delta_{\gamma}} \Big|_{\substack{A_{\gamma} = 0 \\ A_{\alpha} = 0}} = \mathcal{S},$$

$$\frac{\partial \overline{z^{(1)}}}{\partial \Delta_{\gamma}} \Big|_{\substack{A_{\gamma} = 0 \\ A_{\alpha} = 0}} = \mathcal{S},$$

$$\frac{\partial \overline{z^{(1)}}}{\partial \Delta_{\gamma}} \Big|_{\substack{A_{\gamma} = 0 \\ A_{\alpha} = 0}} = \mathcal{S},$$

$$\frac{\partial \overline{z^{(1)}}}{\partial \Delta_{\gamma}} \Big|_{\substack{A_{\gamma} = 0 \\ A_{\alpha} = 0}} = \mathcal{S},$$

$$\frac{\partial \overline{z^{(1)}}}{\partial \Delta_{\gamma}} \Big|_{\substack{A_{\gamma} = 0 \\ A_{\alpha} = 0}} = \mathcal{S},$$

$$\frac{\partial \overline{z^{(1)}}}{\partial \Delta_{\gamma}} \Big|_{\substack{A_{\gamma} = 0 \\ A_{\alpha} = 0}} = \mathcal{S},$$

$$\frac{\partial \overline{z^{(1)}}}{\partial \Delta_{\gamma}} \Big|_{\substack{A_{\gamma} = 0 \\ A_{\alpha} = 0}} = \mathcal{S},$$

$$\frac{\partial \overline{z^{(1)}}}{\partial \Delta_{\gamma}} \Big|_{\substack{A_{\gamma} = 0 \\ A_{\gamma} = 0}} = \mathcal{S},$$

$$\frac{\partial \overline{z^{(1)}}}{\partial \Delta_{\gamma}} \Big|_{\substack{A_{\gamma} = 0 \\ A_{\gamma} = 0}} = \mathcal{S},$$

$$\frac{\partial \overline{z^{(1)}}}{\partial \Delta_{\gamma}} \Big|_{\substack{A_{\gamma} = 0}} = \mathcal{S},$$

$$\frac{\partial \overline{$$

where 8 - mean energy of a sending.

From formula (12.2.17) it is clear that the matrix of gain factors is nondiagonal; mismatch for τ causes the appearance of a constant component of both output voltage of the discriminator. This circumstance is easily explained. In fact, the heterodyned oscillation carries information about time shift of regular modulation both in the amplitude envelope and also in the frequency.

It is interesting that matrix Σ turns out to be diagonal:

$$\begin{split} \overline{(\mathbf{z}^{(1)})^{2}} &= N_{0}^{2} \left\{ q \left[\int u_{a}^{2} (t) e^{-i\mathbf{t}t} dt \right]^{4} \left[1 - \operatorname{Re} \int u_{a}^{2} (t) e^{2i\mathbf{t}t} dt \right] + \frac{1}{2} \left\{ 1 - \left[u_{a}^{2} (t) e^{2i\mathbf{t}t} dt \right]^{4} \right\} \right\}, \\ \overline{(\mathbf{z}^{(1)})^{2}} &= \frac{N_{0}}{2} \left\{ q \int \int I (t_{1}) I (t_{2}) \left[\int u_{a} (\mathbf{\tau} - t_{1}) u_{a} (\mathbf{\tau} - t_{2}) d\mathbf{\tau} \right] dt_{1} dt_{2} + \\ &+ \frac{1}{2} \int \int I (t_{1}) I (t_{2}) \left[\int u_{a} (\mathbf{\tau} - t_{1}) u_{a} (\mathbf{\tau} - t_{2}) d\mathbf{\tau} \right]^{4} dt_{1} dt_{2} \right\}, \\ \overline{\mathbf{z}^{(1)} \mathbf{z}^{(1)}} &= 0, \end{split}$$

$$(12.2.1)^{4}$$

^r)

-675-

where N_0 - spectral density of noise;

 $q = \frac{\delta}{2N_{\bullet}}$ - signal-to-noise ratio in one period (see Chapter VI).

Now let us take Gaussian approximations for $u_{p}(t)$ and l(t):

$$\begin{aligned} & \mathbf{R}_{a}(t) = c_{1} \exp \left\{-\frac{\pi}{2} (t/\tau_{x})^{2}\right\}, \\ & \mathbf{I}(t) = c_{1} t \exp \left\{-\frac{\pi}{2} (t/\tau_{c})^{2}\right\}, \end{aligned}$$

where c_1 , c_2 - constants;

 $\tau_{_{M}}$ - pulse duration;

 τ_{o} - equivalent duration of gates of the time discriminator.

After we integrate according to formulas (12.2.17) and (12.2.18) and carry out operations (12.2.16) on relationships (12.2.14) and (12.2.15) elements of matrix Σ_{arr} will be quantities:

$$\Sigma_{aKB}^{(\tau\tau)} = \frac{\tau_{R}^{2}}{\pi} (1+m^{2})^{3/2} \left[\frac{1}{q} + \frac{1}{q^{4}} \left(\frac{1+m^{4}}{1+2m^{4}} \right)^{3/2} \right],$$

$$\Sigma_{aKB}^{(\tau\tau)} \equiv \Sigma_{aKB}^{(\tau\tau)} = V \Sigma_{aKB}^{(\tau\tau)},$$

$$\Sigma_{aKB}^{(\tau\tau)} = V^{s} \Sigma_{aKB}^{(\tau\tau)} + \frac{\pi}{2\tau_{c}^{2}} \left(\frac{\sin n^{2}}{n^{4}} \right) \left[\frac{1}{q} + \frac{1}{q^{2}} \frac{\sin 2n^{2}}{2\sin n^{2}} \right],$$
(12.2.19)

where

$m^2 = (\tau_c/\tau_{\pi})^2; \quad n^2 = (\tau_{\pi}\delta)^2/2\pi.$

Note that matrix Σ_{SKB} in distinction from Σ is not diagonal, which is a consequence of the interconnection of encoding of parameters.

Elements (12.2.19) depend on the signal-to-noise ratio q, pulse duration τ_0 , the slope of LChM V and two parameters of the discriminator, m and n. It is easy to prove that Σ_{SNB} becomes diagonal only in the absence of frequency modulation. Remember that according to Chapter I the principal axes of the ellipsoid, which is the section of the autocorrelation function of signal, as V \rightarrow 0 are oriented along axes of coordinates τ and ω . This is a graphic expression of the absence of interdependence between the parameters of time delay and frequency shift of the input signal.

12.2.3. Accuracy of Measurement

During the analysis of the work of multi-dimensional tracking meters there arise the same problems as in the one-dimensional case. In general it is necessary to consider a nonlinear problem. This, in principle, is possible on the basis of the technique of the Fokker-Planck equation, the multi-dimensional generalization of which is known in the literature [20]. If, however, the level of input noises is

-676-

low, during analysis of accuracy linearization of the meter is permissible, which we shall realize assuming smallness of parametric fluctuations. We assume that the matrix of smoothing circuits is linear, but not necessarily with constant parameters, so that its input and output quantities are connected by matrix relationship

$$\widehat{\lambda}(t) = \int_{t_0}^{t} h(t, \tau) z(\tau) d\tau + \lambda_{t_0}'(t), \qquad (12.2.20)$$

where $\mathbf{h}(t, \tau) = \|\mathbf{h}^{(\alpha\beta)}(t, \tau)\| - \text{matrix of pulse responses of smoothing circuits;}$ $\mathbf{\lambda}_{\mathbf{B}}(t) - \text{column vector of functions introduced for compensations of known components of } \mathbf{\lambda}(t).$

Substituting in (12.2.20) according to (12.2.11) for z(t) the column vector

$$z(t) = K_{R}t(t) + K_{R}\eta(t) \qquad (12.2.21)$$

and considering that $\hat{\lambda}(t) = \lambda(t) - \varepsilon(t)$, we have equation

$$\mathbf{s}(t) + \int_{t_0}^{t} \mathbf{h}(t, \tau) \mathbf{K}_{\mathbf{R}} \mathbf{s}(\tau) d\tau = \lambda(t) - \lambda_{\mathbf{s}}(t) - \int_{t_0}^{t} \mathbf{h}(t, \tau) \mathbf{K}_{\mathbf{R}} \eta(\tau) d\tau. \qquad (12.2.22)$$

If we introduce the matrix of pulse responses $g(t, \tau)$ of the closed system, considering the output quantity vector $\hat{\lambda}(t)$, it is determined by integral equation

$$g(t, s) + \int_{s}^{t} h(t, s) K_{\mu}g(s, s) ds = h(t, s) K_{\mu}.$$
 (12.2.23)

We introduce additionally the pulse response of the system $\Psi(t, \tau)$, considering mismatches the output quantities. It satisfies equation

$$\mathbf{v}(t, \tau) + \int_{0}^{t} \mathbf{h}(t, s) \mathbf{K}_{\mathbf{R}} \mathbf{v}(s, \tau) ds = \mathbf{I}\delta(t-\tau). \qquad (12.2.2^{k})$$

It is easy to prove that the solution of equation (12.2.22) will be expressed in the form

$$\bullet(t) = -\int_{t_0}^{t} g(t, \tau) \eta(\tau) d\tau + \int_{t_0}^{t} v(t, \tau) [\lambda(\tau) - \overline{\lambda(\tau)}] d\tau + \int_{t_0}^{t} v(t, \tau) [\overline{\lambda}(\tau) - \lambda_{s}(\tau)] d\tau = \bullet_{\phi A}(t) + \bullet_{ABH_{1}}(t) + \bullet_{ABH_{2}}(t). \qquad (12.2.25)$$

According to (12.2.25) current error is determined by the joint action of interferences $\eta(t)$ passed through matrix of filters $g(t, \tau)$ and of random variation of parameters $\lambda(t) - \overline{\lambda(t)}$ and uncompensated regular measurements $\overline{\lambda(t)} - \lambda_{\rm B}(t)$ passed through the matrix of filters $\Psi(t, \tau)$. The first component in (12.2.25) naturally is called the set (vector) of fluctuating errors, and to components $\varepsilon_{\rm RZH} - \frac{1}{4}(t)$ and $\varepsilon_{\rm RZH} - \frac{2}{4}(t)$ we give the name vectors of dynamic errors.

The matrix of second moments of total errors of measurement is equal to

$$\Sigma_{abix}(t) = \overline{\mathbf{e}(t)\mathbf{e}^+(t)} = \int_{t_0}^{t} \mathbf{g}(t, s) \mathbf{S}_{abix}\mathbf{g}^+(t, s) ds + \int_{t_0}^{t} \int_{t_0}^{t} \mathbf{v}(t, s_1) \mathbf{R}_{\lambda}(s_1, s_2) \mathbf{v}^+(t, s_3) ds_1 ds_2 + \int_{t_0}^{t} \int_{t_0}^{t} \mathbf{v}(t, s_1) \Delta(s_1) \Delta^+(s_2) \mathbf{v}^+(t, s_2) ds_1 ds_2, \qquad (12.26)$$

where we use the property of 5-correlation of interferences $\eta(\tau)$, elation matrix $\mathbf{R}_{\lambda}(t_1, t_2)$ of the random part of the parameters and des₁, (*) = = $\lambda(t) - \lambda_{B}(t)$. Certain simplifications of expression (12.2.26) are possible upon concretization of the form of smoothing circuits and the nature of change of $\lambda(t)$.

Let us assume, for instance, that random components of parameters are stationary $[\mathbf{R}_{\lambda}(t_1, t_2) = \mathbf{R}_{\lambda}(t_1 - t_2)]$, regular components are completely compensated, and smoothing circuits have constant parameters, i.e., $\mathbf{h}(t, \tau) = \mathbf{h}(t - \tau)$. The Fourier transforms from functions $\mathbf{g}(t - \tau)$ and $\mathbf{v}(t - \tau)$ by (12.2.23) and (12.2.24) are easily expressed in this case through matrices of gain factors of the discriminator \mathbf{K}_{τ} and frequency responses of the smoothing circuits

$$H(i\omega) = \int_{0}^{\infty} h(\tau) e^{-i\omega\tau} d\tau$$

in the form

$$G(i\omega) = [I + H(i\omega) K_{\pi}]^{-1} H(i\omega) K_{\pi},$$

$$V(i\omega) = [I + H(i\omega) K_{\pi}]^{-1}.$$
(12.2.27)

Thus, instead of (12.2.26) we can obtain

$$\Sigma_{\mu\nu\chi} = \frac{1}{2\pi} \int_{-\infty}^{+\infty} [1 + H^*(i\omega) K_{\mu}]^{-1} H^*(i\omega) K_{\mu} S_{\mu\nu\mu} K_{\mu}^{+} H(i\omega) [1 + H(i\omega) K_{\mu}]^{-1} d\omega + \frac{1}{2\pi} \int_{-\infty}^{+\infty} [1 + H^*(i\omega) K_{\mu}]^{-1} S_{\lambda}(\omega) \times [1 + H(i\omega) K_{\mu}]^{-1} d\omega,$$
(12.2.28)

where $S_{\lambda}(\omega) = \int_{-\infty}^{+\infty} R_{\lambda}(t) e^{-i\omega t} dt$ - spectral matrix of the parameters.

Studying the first component in (12.2.28), we prove that the matrix of moments of fluctuating errors, in general, cannot be expressed through the product of the matrix of spectral densities and a certain constant matrix characterizing the generalized effective passband of the system. However, we may encounter a case when we are interested only in the sum of diagonal terms of this matrix, i.e., its trace.*

-678-

^{*}For this errors of measurement of parameters reduce to a certain unit measure, expressed in linear units of dispersion of the measured position of the object near the true position.

Then according to (12,2,28)

$$\operatorname{spur}\Sigma_{\mathfrak{s}\mathfrak{M}\mathfrak{x}} = \operatorname{spur}S_{\mathfrak{s}\mathfrak{K}\mathfrak{s}}\Delta + \operatorname{spur}\frac{1}{2\pi}\int_{-\infty}^{+\infty}S_{\lambda}(\omega)M(\omega)d\omega, \qquad (12.2.29)$$

where

$$M(m) = \{ | I + H(i\omega) K_{\mu} |^2 \}^{-1},$$

and

$$\Delta = \frac{1}{2\pi} \int_{-\infty}^{+\infty} K_{\pi}^{+} H^{+} (i\omega) \{ | I + H(i\omega) K_{\pi}|^{2} \}^{-1} H^{*} (i\omega) K_{\mu} d\omega \qquad (12.2.30)$$

is a constant matrix which is the multi-dimensional generalization of the effective passband of the system.

Certain more complicated forms of smoothing circuits will be studied in subsequent sections, devoted to questions of synthesis.

§ 12.3. Synthesis of Multi-Dimensional Meters of Several Parameters

In Chapters VI-XI we considered questions of synthesis of meters of only one paramete: Inasmuch as radars usually measure a whole set of coordinates (or parameter: of motion) of the target, there arises the question to what degree results obtained on the assumption that all other parameters besides the measured one are known can be transfered to the case of simultaneous measurement of several unknown quantities. In order to answer this question it is necessary to solve the more complicated problem of simultaneous filtration of several processparameters, randomly varying in time, from their nonadditive mixture with interterences and noises. If the conceptual side, concerning selection of the method of synthesis, remains the same here as in the one-dimensional case (see § 6.5), concrete methods of discovering the form of the optimum operator of filtration require additional study, since they are not a simple repetition of one-dimensional methods.

12.3.1. Method of Synthesis of an Optimum Meter

First of all we shall definitize the formulation of the problem of optimum filtration of several parameters. Consider a set of m random signals ${(1)}_{y}(..., (m)_{y(1)})$, all or part of which depend on *l* generally interconnected, and random (v) varying in time, parameters $\lambda^{(1)}(t)$, ..., $\lambda^{(l)}(t)$. We assume that at moments $t_1 < t_2 < \ldots < t_n$ of interval (t_0, t) , accossible to observation, signals and parameters take values

 $\{ {}^{(1)}y_1, {}^{(2)}y_1, \ldots, {}^{(m)}y_1; \ldots; {}^{(1)}y_n, {}^{(2)}y_n, \ldots, {}^{(m)}y_n \} := \mathbf{y}, \\ \{ \boldsymbol{\lambda}_1^{(1)}, {}^{\boldsymbol{\lambda}_2^{(2)}}, \ldots, {}^{\boldsymbol{\lambda}_1^{(l)}}; \ldots; {}^{\boldsymbol{\lambda}_1^{(1)}}, {}^{\boldsymbol{\lambda}_2^{(2)}}, \ldots, {}^{\boldsymbol{\lambda}_n^{(l)}} \} := \boldsymbol{\lambda},$

-679-

where by \mathbf{y} and $\boldsymbol{\lambda}$ we understand block column vectors.

The joint density of distribution of \mathbf{y} and $\mathbf{\lambda}$ has the form $P_{U}(\mathbf{\lambda}) | \mathbf{\Gamma}(\mathbf{y} | \mathbf{\lambda})$, where $V_{U}(\mathbf{\lambda}) =$ multi-dimensional (for moments of observation and parameters) a priori distribution, and $P(\mathbf{y} | \mathbf{\lambda}) =$ multi-dimensional (for moments of observation, signals and parameters) likelihood function, which we shall discuss later. The task of optimum filtration is construction of the column vector of estimate $\hat{\lambda}$, in some sense the closest to the combination of true values of the parameters. As the loss function, characterizing the degree of this proximity, we first consider the quadratic form of

$$I(\lambda, \hat{\lambda}) = (\lambda - \hat{\lambda}) + B(\lambda - \hat{\lambda}) = \sum_{i,j=1}^{n} (\lambda_i - \hat{\lambda}_i) + B_{ij}(\lambda_j - \hat{\lambda}_j) =$$
$$= \sum_{i,j=1}^{n} \sum_{\alpha,\beta=1}^{l} B_{ij}^{(\alpha\beta)}(\lambda_i^{(\alpha)} - \hat{\lambda}_i^{(\alpha)})(\lambda_j^{(\beta)} - \hat{\lambda}_j^{(\beta)}), \qquad (12.3.1)$$

where the order of co-factors in the second and third elements of the chain of equalities corresponds to operations of matrix multiplication; sign + (raised) signifies transposition; $\mathbf{B} = \|\mathbf{B}_{ij}\|$ - block matrix with submatrix elements $\mathbf{B}_{ij} = \|\mathbf{B}_{ij}^{(\alpha\beta)}\|$, determining the value of errors of the α -th and β -th parameters, allowed in the i-th and j-th moments, respectively. Mean risk

$$R(P_{\bullet}, \hat{\lambda}) = \iint (\lambda - \hat{\lambda})^{+} B(\lambda - \hat{\lambda}) P(\mathbf{y} \mid \lambda) P_{\bullet}(\lambda) d\mathbf{y} d\lambda \qquad (1 \forall .3.2)$$

here is a linear combination of second moments of errors of measurement of all parameters in all moments of time. Vary $u_{n,1} R(t_{n,1}, \hat{\lambda})$ for estimates $\hat{\lambda}$, we can prove that for any non-singular matrices E the optimum estimate, as in the one-dimensional case, is the conditional mathematical expectation

$$\hat{\lambda} = \frac{\int \lambda P(\mathbf{y} \mid \lambda) P_{\mathbf{x}}(\lambda) d\lambda}{\int P(\mathbf{y} \mid \lambda) P_{\mathbf{x}}(\lambda) d\lambda}.$$
(12.3.5)

By a method analogous to § 6.5, it is possible to prove that with symmetry of the a posteriori distribution conditional mathematical expectation is a universal optimum estimate, valid for a broader class of loss functions (namely, for functions possessing the property of symmetry). Inasmuch as direct integration of (12.3.3)in the most interesting cases cannot be accomplished, it is again necessary to polye approximately the system of equations

$$\int (\boldsymbol{\lambda} - \boldsymbol{\hat{\lambda}}) P(\boldsymbol{y} | \boldsymbol{\lambda}) P_{\boldsymbol{v}}(\boldsymbol{\lambda}) d\boldsymbol{\lambda} = 0, \qquad (12.3.4)$$

which is another notation for (12.3.3). In order to prepare conditions for solution of this system, it is necessary to turn to study of the structure of the likelihood function.

-680-

12.3.2. Likelihood Function and Its Approximations

As also in the one-dimensional problem, for the purpose of studying the likelihood function $\mathbb{P}(\mathbf{y}|\mathbf{\lambda})$ one should turn for analogies to the case of several constant parameters. Then, upon the expiration of a certain interval of observation in a parameter space, whose dimensionality is equal to the number of measured quantities l, $\mathbb{P}(\mathbf{y}|\mathbf{\lambda})$ with Gaussian noises at the input can be connected by e-monotonic relationship with a multi-dimensional autocorrelation function of the useful component of the input mixture (mixtures) and a certain multi-dimensional random function whose arguments are the measured quantities.

A multi-dimensional autocorrelation function can be introduced not only for parameters in the form of time delays and frequency shifts, as this is usually done in the literature (see Chapter I), but in a number case for the angular coordinates. This function in general is not equal to the product of oneanmensional autocorrelation functions for all l coordinates obtained on the assumption of knowing the other l = 1 coordinates. This is explained by a certain interconnection of codings of different parameters, which we shall discuss further below. Therefore, the multi-dimensional peak of $P(\mathbf{y}|\lambda)$ will be flattened in certain directions, determined by the nature of the interconnection.

If the vertex of the multi-dimensional autocorrelation function i: smooth, i.e., this function has the necessary derivatives with respect to λ , likelihood function $P(\mathbf{y}|\boldsymbol{\lambda})$ for a low level of lateral peaks of autocorrelation functions and a low level of interferences will be an *l*-dimensional peak of approximately Generation form near the true combination of values of the parameters. Flatness of the peak will be expressed in the fact that sections of constant level will be presented in the form of hyperellipsoids whose principal axes do not coincide with the coordinate axes. As also in the one-dimensional case, increase of the time of observation leads to gradual narrowing of the peak of $\Gamma(\mathbf{y}|\boldsymbol{\lambda})$ and decrease of dispersion of the position of its vertex near the combination of true values of size parameters. Here matrix **Z** of moments of minimum errors of measurement of constant togeneters is expressed through matrix **A** of mean values of second derivations of ln $P(\mathbf{y}|\boldsymbol{\lambda})$ in the form

$$\Sigma = \mathbf{A}^{-1}; \ \mathbf{A} = \left\| -\frac{\partial^{\mathbf{a}} \ln P(\mathbf{y} \mid \lambda)}{\partial \lambda^{(\mathbf{a})} \partial \lambda^{(\mathbf{\beta})}} \right\|.$$
(12.3.5)

Prediction to changing parameters in general leads to the same methematical difficulties as in the one-dimensional problem. Let us assume, newsver, that the

-681-

rate of variation of all measured parameters is small as compared to the rate of variation of parameters belonging to the class of immaterial ones, and that the level of interferences is small. Then it is again possible to separate subintervals of observation in which the measured quantity can be considered "frozen," and immaterial parameters can be considered to vary so that for them the statistical relationship between values in the beginnings and ends of intervals is negligible.

As a supplement to the shown conditions we assume that the measured quantities differ in their physical nature and are coded dissimilarly. To clearly formulate the idea of "dissimilarity" is fairly difficult, so that we shall try to illustrate it by radar examples. Like coordinates of various targets (i.e., time delays and shifts of frequencies of signals close in form, and so forth) are coded similarly in the sense that transposition of parameters in the likelinood function leads to a situation, differing little in "likelihood" from the true situation. Unlike coordinates of one or various targets are coded dissimilarly; their transposition in the likelihood function is impermissible. (Dissimilarity, however, still does not signify disconnectedness with small deviations.)

With the initial encoding of parameters the likelihood function obtains a multipeak structure. These cases are excluded from consideration henceforth.

In the shown simplified conditions, fully sufficient, for instance, for consideration of simultaneous measurement of several coordinates of a single target, the likelihood function in the shown sub-interval of observation in an l-dimensional parameter space turns out to be a single isolated peak. In the whole interval of observation it will be expressed approximately through the product of the likelihood functions in the subintervals. Its logarithm, analogously to the one-dimensional ease, can be approximated by a quadratic form which is a truncated multi-dimensional Taylor series, no that

$$P(\mathbf{y}|\boldsymbol{\lambda}) = P(\mathbf{y}|\boldsymbol{\lambda}_{0}) \exp\left\{\sum_{i=1}^{n} \sum_{\alpha=1}^{l} (\lambda_{i}^{(\alpha)} - \lambda_{i\alpha}^{(\alpha)}) \frac{\partial \ln P(\mathbf{y}|\boldsymbol{\lambda}_{0})}{\partial \boldsymbol{\lambda}_{i}^{(\alpha)}} + \frac{1}{2} \sum_{i,j=1}^{n} \sum_{\alpha,\beta=1}^{l} (\lambda_{i}^{(\alpha)} - \lambda_{i\alpha}^{(\alpha)}) (\lambda_{j}^{(\beta)} - \lambda_{j\alpha}^{(\beta)}) \frac{\partial^{2} \ln P(\mathbf{y}|\boldsymbol{\lambda}_{0})}{\partial \boldsymbol{\lambda}_{i}^{(\alpha)} \partial \boldsymbol{\lambda}_{j}^{(\beta)}}\right\}.$$

$$(12.3.6)$$

where $\lambda_0 = \{\lambda_{10}^{(1)}, \dots, \lambda_{10}^{(1)}, \dots, \lambda_{n0}^{(1)}, \dots, \lambda_{n0}^{(1)}\}$ - column vector, coordinates of which are assumed close to the true values of the measured parameters.

As in the one-dimensional case, two concrete expansions of type (12.3.6) are useful. The first is produced at the point of the multi-dimensional optimum

-682-

estimate of parameters $\lambda_0 = \hat{\lambda} (\hat{\lambda} - \text{complex column vector of the considered type}).$ Designating

$$\ln P(\mathbf{y}|\boldsymbol{\lambda}) = L(\boldsymbol{\lambda}), \ \frac{\partial L(\boldsymbol{\lambda})}{\partial \boldsymbol{\lambda}_{i}^{(\mathbf{a})}} = \mathbf{z}_{i}^{(\mathbf{a})}, \ -\frac{\partial^{\mathbf{a}} L(\boldsymbol{\lambda})}{\partial \boldsymbol{\lambda}_{i}^{(\mathbf{a})} \partial \boldsymbol{\lambda}_{i}^{(\mathbf{b})}} = A_{ij}^{(\mathbf{a}\mathbf{f})}$$
(12.3.7)

and introducing matrix notation with block column vectors $\mathbf{z} = \{\mathbf{z}_1, \ldots, \mathbf{z}_n\}$, $\boldsymbol{\lambda} = \{\boldsymbol{\lambda}_1, \ldots, \boldsymbol{\lambda}_n\}$ and block square matrix $\mathbf{A} = \|\mathbf{A}_{1,j}\|$, where a submatrix element is equal to $\mathbf{A}_{1,j} = \|\mathbf{A}_{1,j}^{(\alpha\beta)}\|$, we have instead of (12.3.6) the relationship

$$P(\mathbf{y}|\boldsymbol{\lambda}) \approx P(\mathbf{y}|\boldsymbol{\lambda}) \exp\left\{\mathbf{z}^{+}(\boldsymbol{\lambda}-\boldsymbol{\hat{\lambda}})-\frac{1}{2}(\boldsymbol{\lambda}-\boldsymbol{\hat{\lambda}})^{+}\mathbf{A}(\boldsymbol{\lambda}-\boldsymbol{\hat{\lambda}})\right\}, \quad (12,3.8)$$

formally coinciding with the one-dimensional (6.6.3).

The second type of expansion corresponds to $\lambda_0 = \hat{\lambda}$, where $\hat{\lambda}$ — the point of maximum value of $P(\mathbf{y}|\lambda)$, and has the form

$$P(\mathbf{y}|\boldsymbol{\lambda}) = P(\mathbf{y}|\boldsymbol{\lambda}) \exp\left\{-\frac{1}{2}(\boldsymbol{\lambda}-\boldsymbol{\lambda})^{+} \mathbf{\tilde{A}}(\boldsymbol{\lambda}-\boldsymbol{\lambda})\right\}.$$
(12.3.9)

Here matrix \mathbf{A} is determined through second derivatives at point λ and practically coincides with matrix \mathbf{A} .

Relationships (12.3.8) and (12.3.9) will be used for synthesis of a multidimensional optimum meter.

12.3.3. Optimum Tracking Multi-Dimensional Meter

Let us assume that the a priori distribution of parameters is Gaussian,

$$P_{\bullet}(\lambda) = (2\pi)^{-\frac{\lambda i}{2}} (\det \mathbb{R})^{-\frac{1}{2}} \exp\left\{-\frac{1}{2}(\lambda - \overline{\lambda}) + V(\lambda - \overline{\lambda})\right\}, \qquad (12.3.10)$$

where

 λ - block column vector of mean values;

 $\mathbf{R} = \|\mathbf{R}_{1,j}\| - \text{block } (nl \times nl) \text{ square matrix with submatrices } (l \times l) \mathbf{R}_{1,j} = \\ = \|\mathbf{R}^{(\alpha\beta)}(t_1, t_j)\|, \text{ elements of which } \mathbf{R}^{(\alpha\beta)}(t_1, t_j) \text{ show} \\ \text{crosscorrelation of the } \alpha \text{-th parameter at time } t_i \text{ and of the} \\ \beta \text{-th parameter at time } t_i;$

 $\mathbf{V} = \mathbf{R}^{-1}$ — block square matrix, the inverse of \mathbf{R} .

During synthesis of an optimum meter we first use an approximation of the likelihood function in the form (12.3.8). Crossmultiplying this relationship with (12.3.10), reducing the logarithm of the resulting expression to form

$$C_{\bullet} - \frac{1}{2} (\lambda - \lambda_{\bullet})^{+} C^{-1} (\lambda - \lambda_{\bullet}),$$

where C_0 , λ_0 and \mathbf{C}^{-1} do not depend on λ , and equating λ_0 to the optimum estimate, analogously to § 6.6 we obtain expression

-683-

$$\hat{\lambda}_{n} = \sum_{k=1}^{n} C_{nk} \Big[z_{k} + \sum_{i=1}^{n} A_{ki} (\hat{\lambda}_{i} - \bar{\lambda}_{i}) \Big] + \bar{\lambda}_{n}, \qquad (12.3.11)$$

where the quantities and the order of their location have matrix meaning, and $\|\mathbf{C}_{ik}\| = \mathbf{C} = [\mathbf{A} + \mathbf{V}]^{-1} - \text{matrix (ln x ln), determined in the interval of observation.}$ After transition to continuous functions at $t_{i+1} - t_i = \Delta \rightarrow 0$, $n \rightarrow \infty$ (i = 1, ..., n; $n\Delta = t - t_0$) relationship (12.3.11) takes form

$$\hat{\lambda}(l) = \int_{l_0}^{l} \mathbf{c}(l, s) \left\{ \mathbf{z}(s) + \int_{l_0}^{l} \mathbf{A}(s, \tau) \left[\hat{\lambda}(\tau) - \overline{\lambda}(\tau) \right] d\tau \right\} ds + \overline{\lambda(l)}, \qquad (12.3.12)$$

where

ALC: NO. OF

$$C(t_{n}, t_{k}) = C_{nk}, z_{k} = \int_{t_{k}-4/2}^{t_{k}+4/2} z(\tau) d\tau;$$

$$A_{ki} = \int_{t_{k}-4/2}^{t_{k}+4/2} d\tau \int_{t_{i}-4/2} ds A(\tau, s).$$
(12.3.13)

As above, in passages to the limit of (12.3.13) quantity Δ is considered limited from below by intervals exceeding the interval of correlation of the invaterial parameters of mixtures ${}^{(1)}y(t)$, and vector function $\mathbf{z}(t)$ is determined with accuracy to statistical equivalence. We also indicate that after transition to continuous time arguments matrices and columns in (12.3.12) and below have dimensionality $(l \times l)$ and $(l \times 1)$, respectively.

For the same reason of slowness of extiation of the measured parameters function $A(t, \tau)$ has filtering properties with respect to time:

$$A(t, s) = K(t)\delta(t-s),$$
 (12.3.14)

where $\mathbf{K}(t) = (l \times l)$ matrix

Taking into account (12.3.14) we have the final expression for the optimum meter:

$$\hat{\mathbf{\lambda}}(t) = \int_{0}^{t} \mathbf{c}(t, s) \{ \mathbf{z}(s) + \mathbf{K}(s) [\hat{\mathbf{\lambda}}(s) - \overline{\mathbf{\lambda}}(s)] \} ds + \overline{\mathbf{\lambda}(t)}, \qquad (12.3.15)$$

which in expanded form should be understood as

$$\hat{\lambda}^{(a)}(t) = \sum_{\substack{\beta=1\\\beta=1}}^{I} \int_{0}^{t} c^{(a\beta)}(t, s) \left\{ z^{(\beta)}(s) + \sum_{\substack{\gamma=1\\\gamma=1}}^{I} K^{(\beta\gamma)}(s) \left[\hat{\lambda}^{(\gamma)}(s) - \overline{\lambda}^{(\gamma)}(s) \right] \right\} ds + \overline{\lambda}^{(a)}(t).$$
(12.3.16)

A block diagram illustrating an optimum joint filter-meter is given in Fig. 12.5. The whole set of input mixtures proceeds to nonlinear units 1 and 2. We indicate that unit 1 issues 1 voltages, on the average proportional to current

-684-



والمنافع المرابع المناقبة والمتعادية والمنافعة والمتلك والمنافع المناقب المناقبين المناقب المنافعة المنافع المنافع المنافع والمنافع والمنافع المنافع والمنافع والمنافع

Fig. 12.5. Two-loop variant of an optimum multi-dimensional meter: 1) multi-dimensional discriminator; 2) multi-dimensional accuracy unit; 3, 4, 7, 8) adders; 5) linear filters with pulse responses $K^{(\alpha\beta)}(t, \tau)$; 6) linear filters with pulse responses $c^{(\alpha\beta)}(t, \tau)$.

mismatches between the true and measured values of parameters $\lambda^{(\alpha)}(t)$. Unit 2 issues l(l + 1)/2 voltages [according to the number of independent coefficients of symmetric matrix $\mathbf{K}(t)$, characterizing the current instantaneous accuracy of measurements. Therefore unit 1 is reasonably named a multi-dimensional (or joint) discriminator, and unit 2 - a multi-dimensional (or joint) accuracy unit. Output voltages of discriminators through adders 3 are fed to a ($l \times l$) matrix of smoothing filters 6, output quantities of which are combined by adders again into 1 groups. Thereby smoothing circuits turn out to be interconnected. After addition in 8 of output voltages of unit 6 with a priori mean values of parameters there are formed estimates of the measured parameters, which proceed to the discrimination and accuracy unit for maintaining selection. Furthermore, grouped output voltages 6 are fed to a matrix of inertialess variable-gain amplifiers 5 from whose putput they proceed to adders of internal fe dback 3. Consequently, there are formed, as it were, two basic multi-dimensional control loops: one using a discriminator and an accuracy unit, the other using adders at the output of the discriminators. As also in the one-dimensional case, we shall call this variant of meter doubleloop. Pulse responses of filters 6 and 5 are controlled by output voltages of the

-685-

accuracy unit for compensation of unequal accuracy of separate measurements (or, which is the same, parametric fluctuations).

The other, single-loop variant of the meter corresponds to formula

$$\hat{\lambda}(t) = \int_{t_0}^{t} g(t, \tau) z(\tau) d\tau + \overline{\lambda(t)}$$
(12.3.17)

or in expanded form

$$\hat{\lambda}^{(a)}(t) = \sum_{\beta=1}^{l} \int_{t_{a}}^{t} g^{(\alpha\beta)}(t, \tau) \, z^{(\beta)}(\tau) \, d\tau + \overline{\lambda}^{(\alpha)}(t) \\ (\alpha = 1, 2, ..., l).$$
(12.3.18)

These relationships are obtained from (12.3.15) and (12.3.16), if we solve them for the estimates. The new variant of circuit is illustrated in Fig. 12.6. It is simpler than the two-loop variant inasmuch as instead of two matrices of smoothing circuits it requires only one matrix of elements 3. Discriminator 1 and accuracy unit 2 are kept in constant form.

For matrices of pulse responses $c(t, \tau)$ and $g(t, \tau)$ it is easy to obtain integral matrix equations

$$c(t, \tau) + \int_{0}^{t} c(t, s) K(s) R(s, \tau) ds = R(t, \tau), \qquad (12.3.19)$$

$$c(t, \tau) + \int_{0}^{t} c(t, s) K(s) g(s, \tau) ds = g(t, \tau), \qquad (12.3.20)$$

completely analogous to the one-dimensional case.

Thus, the optimum system of measurement of several parameters is a multidimensional self-tuning tracking system with two nonlinear multi-dimensional units, issuing signals of errors for separate parameters, and indicators of current accuracy of measurement. The system is cl ed by a matrix (or two matrices) of linear smoothing circuits, determined by the correlation matrix of parameters and matrix $\mathbf{K}(t)$. Furthermore, in the system there is provided input of a priori mean values of parameters.

Correlation matrix $\mathbf{R}(t, \tau)$ characterizes the interconnection of the measured parameters, i.e., describes trajectory properties of the target, forming the set of measured quantities. Matrix $\mathbf{R}(t)$ determines the interconnection of codings of parameters in the signals, i.e., characterizes the process of encoding parameters of the trajectory in parameters of the signal. To the latter during reception there are added noises and interferences. With diagonalness of \mathbf{R} and \mathbf{K} , when parameters and their codings are not connected, matrices \mathbf{c} and \mathbf{g} also are diagonal,



Fig. 12.6. Single-loop variant of an optimum multidimensional meter: 1) multi-dimensional discriminator; 2) multi-dimensional accuracy unit; 3) linear filters with pulse responses $g^{(\alpha\beta)}(t, \tau)$; 4) adders.

and smoothing circuits are considerably simplified. Common elements for subsystem: of measurement of separate parameters remain only a multi-dimensional discriminator and an accuracy unit.

Finally, if the set of mixtures $(i)_{Y}(t)$ is divided into i independent groups, depending each on only one parameter, and **R** and **K** are diagonal, the multi-dimensional measuring system is divided into a set of i of the independent one-dimensional tracking meters investigated above. Such a completely degenerate case in radar is a rare exception. Usually in the discriminator and in smoothing circuits there are circuit couplings which complicate consideration as compared to the

one-dimensional case. However, if matrix K is diagonal (even on average), then i outputs of the discriminator and the l outputs of the accuracy unit are a simple set of outputs of one-dimensional discriminators and accuracy units, synthesized for each parameter separately, with the distinction that derivatives of the logarithm of the likelihood function are taken at a point corresponding to the set of measured values of all l measured parameters.

Remember that in the one-dimensional case derivatives with respect to the parameter were taken at the measured value of the measured quantity, and all the other coordinates were considered accurately known. The shown circumstance is fundamental, and absence of selection for even one coordinate leads to power losses for all the others. In radar practice such a law is well-known (see § 12.2). If, however, matrix K is nondiagonal, then it is necessary additionally to 21 quantities to form in general l(l - 1)/2 more quantities, showing the current interconnection of errors of separate measurements (and of the very signals of mismatches, see below) for all coordinates with connected coding. It is natural that such complication leads to more complicated technical solutions.

The dependence of $\mathbf{z}(t)$ on mismatches $\lambda - \hat{\lambda}$ in general is nonlinear and very complicated. However, at a low level of noises this linear expansion is permissible:

 $z(t) = \xi(t) + K(t) [\lambda(t) - \hat{\lambda}(t)], \qquad (12.3.21)$ $\xi(t) = \{\xi^{(1)}(t), \dots, \xi^{(1)}(t)\}; \qquad (12.3.22)$ $\xi^{(a)}_{i} = \frac{\partial_{L}(\lambda)}{\partial \lambda_{i}^{(a)}} = \int_{t_{i}-\lambda/2}^{t_{i}(a)} \xi^{(a)}(\tau) d\tau, \qquad (12.3.22)$

whore

and $\chi(t)$ is again expressed through second derivatives with respect to $\lambda_1^{(\alpha)}$, $\lambda_j^{(\beta)}$. In view of the speed of change of immaterial parameters, noises $\xi(t)$ according to relationship (12.3.14) are white with correlation matrix $\mathbf{K}(t)b(t - \tau)$. Thereby $\mathbf{K}(t)$ in its physical meaning is the average gain factor of the α -th (β -th) output of the discriminator with respect to mismatch of the β -th (α -th) parameter and simultaneously is the mutual spectral density of white noises of the α -th and β -th output. If we recalculate $\xi(t)$ to the input of the discriminator, we shall have a set of quantities $\eta(t) = \mathbf{K}(t)^{-1}\xi(t)$ of the dimensionality of the measured parameters. It is easy to prove that the matrix of spectral densities $\eta(t)$ has the form $\mathbf{K}(t)^{-1}\delta(t - \tau)$, i.e., $\mathbf{K}(t)^{-1}$, analogously to the orbitar case, has the meaning of the matrix of equivalent spectral densities of noises. This interpretation of

-688-

matrix $\overline{\mathbf{K}(t)}^{-1}$ is the most important, inasmuch as arbitrary inertialess matrix transformation of output voltages of the discriminator changes the gain factor with respect to separate parameters and the level of output noises, but does not change the matrix of equivalent noises recalculated to the input. One should compare namely this matrix with the matrix of spectral densities obtained in the practical circuit performing the same function.

e a airte da airte an airte an ann an an tharaicht i anns a starta airte 🔊 a'r an an airte a'r a'r ar ar ar airte ar



Fig. 12.7. Block diagram of an equivalent linear tracking system; 1) subtractor; 2) linear filters

with pulse responses $\sum_{i=1}^{n} f^{(i+1)}(i, i) f^{(i+1)}(i)$; 3) adders.

In Fig. 12.7 is the block diagram of a linear tracking system, in its characteristics absolutely equivalent to the optimum meter of Fig. 12.6 with small errors of reproduction of all l parameters, when approximation (12.3.21) is valid. To the input of the equivalent system there are fed mixtures of "signals" $\lambda(t) - \overline{\lambda(t)}$

-689-

and "interferences" $\eta(t)$. The matrix of pulse responses of smoothing circuits is

equal to

 $\mathbf{c}(t,\tau)\mathbf{K}(\tau) = \left\|\sum_{\boldsymbol{\beta}=1}^{l} c^{(\alpha\beta)}(t,\tau) \mathbf{K}_{1}^{(\beta\gamma)}(\tau)\right\|.$

Resultant accuracy of measurement for any variant of the circuit is determined, analogously to Paragraph 6.6.2, by relationships

$$\hat{\boldsymbol{\lambda}}(t) = \overline{\boldsymbol{\lambda}}(t), \mathbf{R}_{\mathbf{S}\mathbf{k}\mathbf{k}^{*}}(t, \mathbf{v}) = \overline{\mathbf{c}}(t, \mathbf{v}),$$

$$\boldsymbol{\Sigma}_{\mathbf{S}\mathbf{M}\mathbf{X}}(t) = \overline{\mathbf{c}}(t, t) = \overline{\mathbf{g}}(t, t),$$
(12.3.23)

where averaging of $c(t, \tau)$ is over the ensemble of input mixtures.

In the particular case when the random component of elements of matrix $\mathbf{K}(t)$ can be ignored, accuracy units and control units for smoothing circuits disappear, nonrandom pulse responses are determined by equations (12.3.19) and (12.3.20), where $\mathbf{K}(t)$ is replaced by its mean, and during the analysis of accuracy properties of the meters there is no need for additional averaging of $\mathbf{c}(t, \tau)$.

We shall also discuss the method of obtaining functions $\mathbf{z}(t)$ and $\mathbf{K}(t)$ from the form of the likelihood functional of all the input signals. As also in the one-dimensional case, the likelihood functional, completely accounting for properties of the whole set ${}^{(1)}y(t)$, can usually be expressed in the form

$$P(\mathbf{y}(t)|\boldsymbol{\lambda}(t)) = \exp \int_{0}^{1} l(\tau, \boldsymbol{\lambda}, \mathbf{y}) d\tau, \qquad (12.3.24)$$

where $\mathbf{y}(s)$, $\lambda(s)$ - vector functions, considered in interval $s \in (t_0, \tau)$. Then, analogously to § 6.6

$$\mathbf{z}^{(\mathbf{e})}(t) = \frac{\partial}{\partial \lambda^{(\mathbf{e})}} l(\mathbf{r}; \hat{\boldsymbol{\lambda}}(t), \mathbf{y}), \qquad (12.3.25)$$

$$K^{(a)}(t) = -\frac{\partial^2}{\partial \lambda^{(a)} \partial \lambda^{(b)}} l(a, \hat{\lambda}(t), \mathbf{y}). \qquad (12.3.26)$$

If all useful components of input mixtures are incoherent pulse signals, it is necessary to produce modification of the method of synthesis, analogously to § 6.6. The optimum meter executes operations according to discrete expression

$$\hat{\boldsymbol{\lambda}}_{n} = \sum_{k=1}^{n} c_{nk} \left[\boldsymbol{z}_{k} + \boldsymbol{K}_{k} \left(\hat{\boldsymbol{\lambda}}_{k} - \overline{\boldsymbol{\lambda}}_{k} \right) \right] + \overline{\boldsymbol{\lambda}}_{n}, \qquad (12.3.27)$$

Into which (12.3.11) passes, taking into account the diagonal nature of complex matrix **A** with respect to discrete time arguments. By $z_k^{(\alpha)}$ and $K_i^{(\alpha\beta)}$ one should understand quantities grouped in a period of repetition

 $Z_{\mathbf{A}}^{(\mathbf{n})} = \frac{\partial}{\partial \lambda^{(\mathbf{n})}} \int_{(\mathbf{A}-1)T_{\mathbf{r}}}^{\mathbf{A}T_{\mathbf{r}}} l_{\mathbf{A}}(\mathbf{\tau}, \mathbf{\hat{\lambda}}_{\mathbf{A}}, \mathbf{y}) d\mathbf{\tau}, \qquad (12.3.28)$

-690~

$$K_{\tau}^{(\alpha\beta)} = -\frac{\partial^{2}}{\partial\lambda^{(\alpha)}\partial\lambda^{(\beta)}} \int_{(k-1)T_{\rho}}^{kT_{\rho}} l_{k}(\tau, \hat{\lambda}_{k}, \mathbf{y}) d\tau, \qquad (12.3.29)$$

where l_k - functions in the expression for the likelihood functional

$$P(\mathbf{y}(t)|\boldsymbol{\lambda}(t)) = \exp \sum_{\mathbf{a}} \int_{(\mathbf{a}-1)T_{\mathbf{y}}}^{\mathbf{A}T_{\mathbf{y}}} l_{\mathbf{A}}(\mathbf{\tau}, \boldsymbol{\lambda}, \mathbf{y}) d\mathbf{\tau}. \qquad (12.3.30)$$

Matrices of discrete smoothing circuits are determined by equations analogous to (6.6.20) and (6.6.22):

$$\begin{array}{c} \mathbf{C} + \mathbf{C}\mathbf{K}\mathbf{R} = \mathbf{R}, \\ \mathbf{C} + \mathbf{C}\mathbf{K}\mathbf{G} = \mathbf{G} \end{array} \right\}$$
(12.3.31)

Stand of the second is subject to

ومليل فإنباع الأنعد

or in expanded form

$$C_{ik}^{(a\beta)} + \sum_{j=1}^{n} \sum_{\substack{\gamma, \ b=1}}^{l} C_{ij}^{(a\gamma)} K_{j}^{(\gamma b)} R_{jk}^{(2\beta)} = R_{ik}^{(a\beta)},$$

$$C_{ik}^{(a\beta)} + \sum_{\substack{j=k \ \gamma, \ b=1}}^{n} \sum_{\substack{\gamma, \ b=1}}^{l} C_{ij}^{(a\gamma)} K_{j}^{(\gamma b)} G_{jk}^{(b\beta)} = G_{ik}^{(a\beta)}.$$
(12.3.32)

Further concretization of the problem of optimum measurement consists in finding the concrete form of the operation of formation of z(t) and K(t), the equipment realization of these operations, and establishment of the algorithm of smoothing in the linear filters.

We discuss these questions in subsequent paragraphs.

12.3.4. Optimum Nontracking Multi-Dimensional Meter

Let us consider now an optimum nontracking meter of several parameters. The nontracking variant of the meter is synthesized analogously to § 6.6. Multiplying (12.3.10) by another approximation of likelihood function (12.3.9) and transforming the logarithm of the formed expression, we have

$$\hat{\boldsymbol{\lambda}}_{n} = \sum_{k=1}^{n} C_{nk} A_{kj} (\check{\boldsymbol{\lambda}}_{j} - \bar{\boldsymbol{\lambda}}_{j}) + \bar{\boldsymbol{\lambda}}_{n}. \qquad (12.3.33)$$

After transition to continuous observation and allowing for the diagonal nature of matrix A for time arguments, we have finally

$$\hat{\lambda}(l) = \int_{0}^{l} \mathbf{c}(l, \tau) \mathbf{K}(\tau) [\dot{\lambda}(\tau) - \overline{\lambda}(\tau)] d\tau + \overline{\lambda}(l), \qquad (12.3.34)$$

or in expanded form

$$\hat{\lambda}^{(\alpha)}(l) = \sum_{\tau=1}^{l} \int_{0}^{l} \left\{ \sum_{j=1}^{l} c^{(\alpha)}(l, \tau) K^{(3\tau)}(\tau) \right\} \times \left[\hat{\lambda}^{(\tau)}(\tau) - \hat{\lambda}^{(\tau)}(\tau) \right] d\tau + \hat{\lambda}^{(\alpha)}(l) \ (\alpha = 1, \ldots, l). \ (12.3.35)$$



Fig. 12.8. Nontracking variant of an optimum multi-dimensional meter: 1) estimator unit; 2) subtractor; 3) linear filters with pulse responses

 $\sum_{\beta=1}^{\ell} c^{(\alpha\beta)}(\ell, \tau) K^{(\beta\gamma)}(\tau) ; 4) \text{ adders.}$

A circuit executing operations (12.3.34) is given in Fig. 12.8. As also in the one-dimensional case, here there are set off a multi-dimensional estimator unit 1 which gives the set of functions $\tilde{\lambda}^{(\alpha)}(t)$ and $K^{(\alpha\beta)}(t)$ and a matrix of linear

rilters 3, 4 with pulse responses $\sum_{p=1}^{l} c^{(a_{2})}(l, \tau) K^{(p_{1})}(\tau)$. We indicate that the estimator

unit here is a very complicated device which analizes the likelihood function for all measured parameters in the whole a priori domain of their determination. As also in the one-dimensional case, the nontracking circuit somewhat weakens the requirement on the magnitude of variance of the initial distribution of error of measurement, expanding as it were the linear section of the characteristics of accuracy of measurement (12.3.23). This circumstance is very important in the stage of lock-on and transition to precision tracking. In a number of cases it will be convenient to combine tracking and nontracking circuits for separate parameters.

-692-

}

12.3.5. Optimum Meter of Linear Functionals of Parameters

Very often in radar along with measurement of parameters of input signals there must be formed several linear functionals of them (derivatives, expected values of parameters, integrals, and so forth). Sometimes measurement of linear functionals is the basic function of devices. Therefore the problem of optimum measurement of an arbitrary number q of linear functionals $\varphi^{(\beta)}(t)$ of l parameters $\lambda^{(\alpha)}(t)$ coded in input signals is important.

The consideration below in many respects is analogous to material of Paragraph 6.6.8 describing the case of one parameter and one functional. As before we assume Gaussian distribution of parameters $\lambda^{(\alpha)}(t)$ and consequently, also of $\psi^{(f, \cdot)}(t)$.

The relationship between column vectors of functionals $\varphi(t)$ and parameters $\lambda(t)$ is established by a linear matrix integral relationship

$$\Phi(T) = \int_{T_1}^{T_2} F(T, s) \lambda(s) ds \qquad (12.3.36)$$

and is wholly determined by square (q \times l) matrix function F(t, s).

After quantization in time by intervals & relationship (12.3.30) takes form

$$\mathbf{\varphi} = \mathbf{F} \mathbf{\lambda}, \qquad (12.3.37)$$

or in expanded form

$$\varphi_{m}^{(a)} = \sum_{\beta=1}^{I} \sum_{j=N_{1}}^{N_{0}} F_{mj}^{(a\beta)} \lambda_{j}^{(3)} (T_{1,2} = N_{1,2}\Delta + t_{0}, T = m\Delta + t_{0}), \qquad (12.3.3\%)$$

where ϕ , λ and F in (12.3.37) are understood as complex columns and the matrix.

Let us assume that the input signal (signals) y(t) is accessible to observation at moments t_1, t_2, \ldots, t_n from interval (t_0, t) , in no way attached to moment $T = t_m$, the argument of the functional ϕ_m . Then mean risk in the quadratic loss function with respect to errors of measurement of functionals in the m-th moment of time is recorded in the form

$$R(P_{\lambda_1}, \hat{\varphi}) = \iint \sum_{\alpha, \beta=1}^{q} B^{(\alpha\beta)}(\varphi_m^{(\alpha)} - \hat{\varphi}_m^{(\alpha)})(\varphi_m^{(\beta)} - \hat{\varphi}_m^{(\beta)}) \times P(y_1, \dots, y_n | \lambda_1^{(l)}, \dots, \lambda_n^{(l)}) \times P(y_1, \dots, y_n | \lambda_n^{(l)}) \times P(y_1, \dots, y_n | \lambda_n^{(l)}) \times P(y_1, \dots, y_n^{(l)}) \times P(y_1, \dots, y_n^{(l)}) \times P(y_1, \dots, y_n^{(l)}) \times P(y_1, \dots, y_n^{(l)}) \times P(y_1, \dots, y_n^{(l)})$$

Varying $R(F_{0}, \hat{\varphi})$ for estimates $\hat{\varphi}$, we have the system of equations

$$\int (\mathbf{\Psi}_{m} - \hat{\mathbf{\Psi}}_{m}) P_{(1,n)}(\mathbf{y}|\boldsymbol{\lambda}) P_{0(N_{s},N_{s})}(\boldsymbol{\lambda}) d\boldsymbol{\lambda} = 0, \qquad (12.3.39)$$

where subscripts in the likelihood function and the a priori distribution signify the number of moments of reading. As also for the one-dimensional case, it is easy to prove that direct formal solution of (12.3.39),

 $\Psi_{m} = \frac{\sum_{\lambda=N_{i}}^{N_{0}} F_{m\lambda} \int \lambda_{\lambda} P_{(1,n)} (\mathbf{y}|\lambda) P_{0(N_{s}, N_{i})} (\lambda) d\lambda}{\int P_{(1,n)} (\mathbf{y}|\lambda) P_{0(N_{s}, N_{i})} (\lambda) d\lambda}$

(12.3.40)

in general is unacceptable in view of disturbance of the principle of physical realizability of the optimum operator. In other words, the linear functional F of the optimum estimate of the parameters, in general, is not the optimum estimate of linear functional F.

To find a physically realizable solution of equation (12.3.46) we again assume that for $P(\mathbf{y}|\mathbf{\lambda})$ approximation (12.3.8) is permissible. Expansion of ln $P(\mathbf{y}|\mathbf{\lambda})$ we produce at the point of the current physically realizable estimate of parameters, which it is possible to demand beforenand. Permissibility of formation and use in a meter of functionals as auxiliary functions of optimum estimates of coded parameters does not raise doubts either theoretically or practically. Substituting (12.3.3) and (12.3.10) in (12.3.39) and integrating, taking into account the diagonal nature of the matrix of second derivatives, we have

$$\hat{\Psi}_{m} = \sum_{N_{1}}^{N_{1}} F_{mk} \left\{ \overline{\lambda}_{k} + \sum_{i=1}^{n} C_{ki} \left[z_{i} + K_{i} \left(\widehat{\lambda}_{i} - \overline{\lambda}_{i} \right) \right] \right\} = \sum_{k=1}^{n} B_{mk} \left[z_{k} + K_{k} \left(\widehat{\lambda}_{k} - \overline{\lambda}_{k} \right) \right] + \sum_{k=N_{1}}^{N_{1}} F_{mk} \overline{\lambda}_{k}.$$
(12.3.41)

Here $\mathbf{C} = [\mathbf{V} + \mathbf{Q}]^{-1}$ - inverse matrix; \mathbf{Q} - matrix of order [: × (N₂ - N₁)], determined by relationship

$$\mathbf{Q}_{ik} = \begin{cases} \mathbf{K}_{i} \delta_{ik} & 1 \leq ik \leq n, \\ 0 & i, k > n; i, k < 1. \end{cases}$$
(12.3.42)

Complex matrix **B** is determined in expanded form by relationship

$$\mathbf{B}_{mk} = \begin{cases} \sum_{l=N_{i}}^{N_{b}} \mathbf{F}_{ml} \mathbf{C}_{ik} \ 1 \le k \le n, \\ \mathbf{0} \ k > n; \ k < 1. \end{cases}$$
(12.3.43)

From (12.3.43), taking into account values of $\boldsymbol{\sigma}_{ik}$ and \boldsymbol{v}_{ik} , we can obtain equation

$$\mathbf{B}_{mk} + \sum_{i=1}^{n} \mathbf{B}_{mi} \mathbf{K}_{i} \mathbf{R}_{ik} = \sum_{i=N_{1}}^{N_{1}} \mathbf{F}_{mi} \mathbf{R}_{ik}, \qquad (12.3.44)$$

relating matrix \mathbf{B}_{ik} with the correlation matrix of parameters \mathbf{R}_{ik} and the matrix function of the linear functional \mathbf{F}_{ik} .

Another presentation of solution (12.3.41) has the form

-694-

$$\hat{\boldsymbol{\varphi}}_{m} = \sum_{k=1}^{n} \boldsymbol{H}_{mk} \boldsymbol{z}_{k} + \sum_{k=N_{1}}^{N_{s}} \boldsymbol{F}_{mk} \bar{\boldsymbol{\lambda}}_{k}, \qquad (12.3.4)$$

where \boldsymbol{H}_{mk} is connected with \boldsymbol{B}_{mk} by relationship

$$H_{mk} = B_{mk} + \sum_{i=k}^{n} B_{mi} K_i G_{ik}, \qquad (12.3.46)$$

and G_{1k} in turn is determined through \mathbf{R}_{1k} according to equations (12.3.32), describing the optimum meter of parameters $\lambda(\mathbf{t})$ themselves. As we see, the case of many functionals and parameters formally has great so larity with the one-dimensional case.

Transition to continuous observation gives relationship

$$\hat{\varphi}(T) = \int_{T_{1}}^{T_{2}} \mathbf{b} (T, s) \{ \mathbf{z}(s) + \mathbf{K}(s) [\hat{\lambda}(s) - \overline{\lambda}(s)] \} ds + \int_{T_{1}}^{T_{2}} \mathbf{F}(T, s) \overline{\lambda}(s) ds, \quad (12.3.47)$$

where transition from \mathbf{z}_k , \mathbf{K}_k , \mathbf{B}_{mk} and \mathbf{F}_{ik} to $\mathbf{z}(t)$, $\mathbf{K}(t)$, $\mathbf{b}(T, s)$ and $\mathbf{F}(t, s)$ in the light of results of Paragraph 12.3.3 does not require explanation; $T = m\Delta + t_0$, $t = m\Delta + t_0$; and $\mathbf{b}(t_i, t_j) = \mathbf{B}_{ij}$ - matrix function, connected with $\mathbf{R}(t, \tau)$ by an integral equation analogous to (12.3.44):

$$\mathbf{b}(T,\tau) + \int_{0}^{\tau} \mathbf{b}(T,s) \mathbf{K}(s) \mathbf{R}(s,\tau) ds = \int_{T_{A}}^{T_{A}} \mathbf{F}(T,s) \mathbf{R}(s,\tau) ds. \qquad (12.3.4\%)$$

In order not to encumber our account, we shall not give the expanded circuit of the optimum meter of functionals, constructed on the basis of relationship (12.3.47), and we only give its description. It contains the complete block diagram of the two-loop variant of an optimum meter of $\hat{\lambda}(t)$ (see Fig. 12.4). From adders of internal coupling in smoothing circuits there are made l taps to a square matrix of $(l \times q)$ special smoothing filters with pulse responses $b^{(\alpha\beta)}(t, \tau)$ ($\alpha = 1, ..., q$; $\beta = 1, ..., l$). Output signals are combined in q groups according to the number of measured functionals, after which they are adde-

to γ functions $\sum_{j=1}^{l} \int_{\tau} F^{(a_j)}(t, \tau) \overline{\lambda}^{(j)}(\tau) d\tau$, which are results of processing a priori mean

values of parameters $\mathbf{X}(t)$ by matrix function $\mathbf{F}(t, \tau)$. As a result there as formed optimum estimates of functionals. As we see, as compared to the case of measurement only of $\mathbf{A}(t)$ there are required comparatively few complications.

Another modification of the optimum circuit is obtained by means of transition to continuous observation in relationship (12.3.45):

$$\mathbf{\hat{p}}(T) = \int_{0}^{t} \mathbf{h}(T, s) \mathbf{z}(s) ds + \int_{0}^{T_{s}} \mathbf{F}(T, s) \overline{\lambda(s)} ds. \qquad (12.3.49)$$

It is conveniently combined with the single-loop variant of an optimum meter of parameters $\lambda(t)$ (Fig. 12.5). Output voltages of discriminators are fed to a matrix ($t \ge q$) of smoothing circuits with pulse response h(T, s), satisfying a relationship analogous to (12.3.46):

$$h(T, \tau) = b(T, \tau) + \int_{\tau} b(T, s) K(s) g(s, \tau) ds,$$
 (12.3.50)

where $g(t, \tau)$ is determined by the correlation function of the parameter according to equation (12.3.32).

For accuracy of measurement of the set of linear functionals we can obtain the following formulas:

$$\mathbf{R}_{\text{BMX}}(t_1, t_2) = \int_{0}^{t} \mathbf{b}(t_1, s) \mathbf{F}^+(t_2, s) ds,$$

$$\mathbf{\Sigma}_{\text{BMX}}(T) = \int_{0}^{t} \mathbf{b}(T, s) \mathbf{F}^+(T, s) ds.$$
 (12.3.51)

Thus, for optimum measurement of linear functionals of parameters it is necessary to introduce in the optimum meter directly coded parameters of two matrices of linear filters: one, the basic one, for smoothing of data, the other for transformation of a priori mean values. Matrices have l inputs (from the number of parameters) and q outputs (from the number of functionals). Adding output quantities of these two matrices, we have optimum estimates of the functionals. Finding the first of these matrices from the concrete correlation matrix of parameters and accuracy properties of discriminator is a separate problem.

12.3.6. Optimum Meter of Markovian Parameters

During construction of a meter of several Markovian parameters we use the methods offered in § 6.9. We start from the case of discrete observation (a pulse signal), when it is possible to use the relationship for final a posteriori probabilities

$$\tilde{P}(\lambda_{n+1}) = C_{n+1}^{0} P(\mathbf{y}_{n+1} | \lambda_{n+1}) \int \tilde{P}(\lambda_n) W(\lambda_{n+1} | \lambda_n) d\lambda_n, \qquad (12.3.52)$$

generalizing (6.5.24) to the vector case. Here $\lambda_k = \{\lambda_k^{(1)}, \ldots, \lambda_k^{(l)}\}$ - column vector of the measured parameters in the k-th moment of time; $\widetilde{P}(\lambda_k)$ - its a posteriori

probability; $P(\mathbf{y}_{n+1}|\mathbf{\lambda}_{n+1})$ - the likelihood function of the input mixture at the (n + 1)-th moment (period) of observation; $W(\mathbf{\lambda}_{n+1}|\mathbf{\lambda}_n)$ - multi-dimensional probability density of transition.

Assuming small a posteriori inaccuracy, we introduce the Gaussian a_{PP} oximation of $\widetilde{\mathbb{P}}(\lambda_k)$ in the form

$$\begin{split} \widetilde{P}(\lambda_{k}) &= (2\pi)^{-1/4} \left[\det \Sigma_{\mu_{k} \times k} \right]^{-1/4} \times \\ \times \exp \left\{ -\frac{1}{2} (\lambda_{k} - \hat{\lambda}_{k})^{+} \Phi_{k} (\lambda_{k} - \hat{\lambda}_{k}) \right\}, \end{split}$$
(12.3.53)

where $\Sigma_{BHX \ k}$ - matrix of second moments of output errors in the k-th moment; Φ_{k} - its inverse matrix;

 $\hat{\lambda}_{k}$ - column vector of estimates of parameters.

Approximation of the likelihood function we take in one of two forms

$$P(\mathbf{y}_{\mathbf{\lambda}}|\boldsymbol{\lambda}_{\mathbf{\lambda}}) \approx P(\mathbf{y}_{\mathbf{\lambda}}|\boldsymbol{\lambda}_{\mathbf{\lambda}}) \exp\left\{-\frac{1}{2}(\boldsymbol{\lambda}_{\mathbf{\lambda}} - \boldsymbol{\lambda}_{\mathbf{\lambda}})^{+} \mathbf{K}_{\mathbf{\lambda}}(\boldsymbol{\lambda}_{\mathbf{\lambda}} - \boldsymbol{\lambda}_{\mathbf{\lambda}})\right\}, \qquad (12.3.54)$$

$$P(\mathbf{y}_{k}|\boldsymbol{\lambda}_{k}) \approx P(\mathbf{y}_{k}|\mathbf{e}(\hat{\boldsymbol{\lambda}}_{k-1})) \exp\{\mathbf{z}_{k}^{+}(\boldsymbol{\lambda}_{k}-\mathbf{e}(\hat{\boldsymbol{\lambda}}_{k-1})) - \frac{1}{2}(\boldsymbol{\lambda}_{k}-\mathbf{e}(\hat{\boldsymbol{\lambda}}_{k-1}))^{+}\mathbf{K}_{k}(\boldsymbol{\lambda}_{k}-\mathbf{e}(\hat{\boldsymbol{\lambda}}_{k-1}))\}.$$
(12.3.55)

In the first case the expansion of ln $P(\mathbf{y}_k | \boldsymbol{\lambda}_k)$ is conducted near the maximum likelihood estimate $\boldsymbol{\tilde{\lambda}}_k$, and in the second it is at point $\mathbf{e}(\hat{\boldsymbol{\lambda}}_{k-1})$, determined by extrapolation of the estimate vector from the preceding moment ($\mathbf{e}(\boldsymbol{\lambda})$ - column vector which depends on column vector $\boldsymbol{\lambda}$ with the same number of components).

It is necessary to further definitize the form of the multi-dimensional transition function. For a diffusion vector process on small intervals of time this Gaussian approximation is permissible:

$$W'(\lambda_{n+1}|\lambda_n) = (2\pi)^{-1/2} \left[\det \Sigma_{\lambda}(\lambda_n) \right]^{-1/2} \times \left\{ -\frac{1}{2} (\lambda_{n+1} - \mathbf{e}(\lambda_n))^+ \Sigma_{\lambda}^{-1}(\lambda_n) (\lambda_{n+1} - \mathbf{e}(\lambda_n)) \right\}, \qquad (12.3.45)$$

where $\boldsymbol{\Sigma}_{\lambda}(\boldsymbol{\lambda}) = \|B^{(\alpha\beta)}(\boldsymbol{\lambda})\|_{\Delta}$ - matrix of correlation moments of random variations of parameters for interval Δ , in the first approximation of proportional multiple aimensional coefficients of diffusion $B^{(\alpha\beta)}(\boldsymbol{\lambda})$; $\boldsymbol{\Sigma}_{\lambda}^{-1}(\boldsymbol{\lambda})$ - its inverse matrix.

The result of extrapolation to the following moment in the condition of small Δ can be expressed through multi-dimensional coefficients of drift $A^{(\alpha)}(\lambda_n)$, again limiting ourselves to terms linear in time:

$$\mathbf{e}(\lambda_n) = \lambda_n + \mathbf{a}(\lambda_n), \ \mathbf{a}(\lambda) = \{A^{(1)}(\lambda) \Delta, \ldots, A^{(1)}(\lambda) \Delta\}. \tag{12.3.57}$$

We expand $a(\lambda)$ at the estimate point

 $\mathbf{a}(\lambda) = \mathbf{a}(\hat{\lambda}) + \mathbf{U}(\hat{\lambda})(\lambda - \hat{\lambda})\Delta,$

(12.3.5)

where $U(\hat{\lambda}) = \left\| \frac{\partial A^{(\alpha)}(\hat{\lambda})}{\partial \lambda^{(p)}} \right\| - \text{square matrix depending on } \lambda$.

We substitute (12.3.53), (12.3.54) and (12.3.56) taking into account (12.3.57)and (12.3.58) in (12.3.52). Then after integration and equating of logarithms of both parts of the resulting expression we obtain matrix formulas

$$\Phi_{n+1} \hat{\lambda}_{n+1} = [\mathbf{Q}^{+}(\hat{\lambda}_{n}) \Phi_{\mathbf{s}}^{-1} \mathbf{Q}(\hat{\lambda}_{n}) + \Sigma_{\lambda}(\hat{\lambda}_{n})]^{-1} [\hat{\lambda}_{n} + \mathbf{a}(\hat{\lambda}_{n})] + K_{n+1} \hat{\lambda}_{n+1}, \qquad (12.3.69)$$

$$\Phi_{n+1} = [\mathbf{Q}^{+}(\hat{\lambda}_{n}) \Phi_{\mathbf{s}}^{-1} \mathbf{Q}(\hat{\lambda}_{n}) + \Sigma_{\lambda}(\hat{\lambda}_{n})]^{-1} + K_{n+1}, \qquad (12.3.60)$$

where $\mathbf{Q}(\lambda) = \mathbf{I} + \mathbf{U}(\lambda)\Delta$ - extrapolation matrix of partial derivatives of the new [(n + 1)-st] with respect to the old (n-th) values of parameters, analogous to the scalar factor Q in (6.9.6) and (6.9.7).

Relationships (12.3.59) and (12.3.60) are completely similar to (6.9.6) and (0.9.7). Thus by (12.3.59) the vector of the (n + 1)-st estimates will be formed by weighted addition of the column of n-th estimates, anticipated a step, and the column vector of the new measurements λ_{n+1} . The matrix weight of the first term is determined by the sum of the extrapolated matrix of output errors in the preceding moment and the matrix of variances of expected variations of the parameters. The weight of the second term is equal to matrix \mathbf{K}_n , the physical meaning of which was explained in detail above. "Information" matrix $\boldsymbol{\Phi}_n$ (the inverse of correlation matrix $\boldsymbol{\Sigma}_{\text{BHX},n}$) according to (12.3.60) on every step changes due to the action of three factors - extrapolation, random changes of the parameter and new incoming information about the parameters.

When approximation of the likelihood function is conducted according to (12.3.55), relationship (12.3.59) is replaced by

$$\hat{\lambda}_{n+1} = \hat{\lambda}_n + a(\hat{\lambda}_n) + \Sigma_{\text{sux}(n+1)} z_{n+1},$$
 (12.3.61)

where $\Sigma_{BWX(n+1)} = \Phi_{n+1}^{-1}$ is calculated again by (12.3.60). Relationship (12.3.61) is the vector generalization of formulas of Chapter VI and is interpreted the same way.

It is comparatively simple to obtain from (12.3.59)-(12.3.61) differential equations for continuous observation. They have matrix form:

$$\frac{d}{dt}\,\hat{\boldsymbol{\lambda}}(t) = \mathbf{A}(\hat{\boldsymbol{\lambda}}(t)) + \boldsymbol{\Sigma}_{\mathtt{BMX}}(t)\,\mathbf{K}(t)\,[\hat{\boldsymbol{\lambda}}(t) - \hat{\boldsymbol{\lambda}}(t)], \qquad (12.3.67)$$

-698-

$$\frac{d}{dt} \Sigma_{\text{sMX}}(t) = \mathbf{B}(\hat{\lambda}(t)) - \Sigma_{\text{sMX}}(t) \mathbf{K}(t) \Sigma_{\text{sMX}}(t) + + \mathbf{U}(\hat{\lambda}(t)) \Sigma_{\text{sMX}}(t) + \Sigma_{\text{sMX}}(t) \mathbf{U}^{+}(\hat{\lambda}(t)), \qquad (12.3.63)
$$\frac{d}{dt} \hat{\lambda}(t) = \mathbf{A}(\hat{\lambda}(t)) + \Sigma_{\text{sMX}}(t) \mathbf{Z}(t), \qquad (12.3.64)$$$$

but physical interpretation of them practically does not differ from the scalar case. Here $\mathbf{z}(t)$, $\mathbf{K}(t)$ - set of output quantities of an optimum multi-dimensional discriminator and an estimator unit.

Above we nowhere said that all 1 measured parameters have to be coded in the received signal (signals); there should only be sufficient coordinates of the set of measured quantities. Parameters $\lambda^{(1)}(t)$, ..., $\lambda^{(1)}(t)$ can, for instance, be broken up into p groups, each of which characterizes one parameter actually coded in the signal $\mu^{(1)}(t)$ (i = 1, ..., p). In a particular case all $\lambda^{(1)}(t)$..., $\lambda^{(l)}(t)$ may be sufficient coordinates of one parameter. Thus, if distance of the target is a Markovian process of the third order, as sufficient (and necessary in the process of smoothing) coordinates of time (for instance, the current signal period, the preceding period, and the one preceding it). In such cases column \mathbf{z}_n (or $\hat{\mathbf{\lambda}}_n$) and matrix \mathbf{K}_n are divided into p blocks, inside which only one element is different from zero, corresponding practically to the quantity coded in the signal, and all others are equal to zero. This does not lead to any theoretical or technical difficulties during realization of the meter.

We also note that when it is permissible to replace K(t) by K, and matrices $B(\lambda)$ and $U(\lambda)$ do not depend on parameters, behavior of the output correlation matrix $\Sigma_{BHX}(t)$ ceases to depend on the realization, and equation (12.3.63) can be solved beforehand in principle. As a result the operations of smoothing reduce t (12.3.62) or (12.3.64), where $\Sigma_{BHX}(t) - a$ certain known matrix function. This, naturally, greatly simplifies the technical problem of construction of the mater.

The structure of the closed variant of a multi-dimensional meter in approximation $\mathbf{K}(t) \approx \mathbf{K}$ is shown in Fig. 12.9. The meter consists of a multi-dimensional discriminator 1, controlled amplifiers 2 coefficients of which $\mathbf{\Sigma}^{\alpha\beta}(t)$ are issued by the unit of determination of errors 3, adders 4, integrators 5 and a unit of nonlinear converters 6 introducing the mean value of drift depending upon the measured parameters. As we see, nonlinearity in smoothing circuits remains even in this cimple case.

-699-



Fig. 12.9. Optimum multi-dimensional meter of Markovian parameters: 1) multidimensional discriminators; 2) controlled amplifiers; 3) units of determination of resultant errors; 4) adders; 5) integrators; 6) nonlinear converters. Thus, for Markovian parameters smoothing circuits of the meter become in general nonlinear. At the same time elements of the first processing - multi-dimensional discriminator and accuracy unit - remain the same as for Gaussian parameters.

The statistical synthesis conducted above shows that a meter of several parameters (or linear functionals of them) is not a simple set of independent meters of separate parameters (functionals). Even if it is possible to detect in the joint discriminator and smoothing circuits separate elements relating to measurement of only one parameter there usually remain a whole series of elements necessary for the meter as a whole. Internal causes of the

appearing couplings are the dependence of input signals immediately on several parameters, interconnection of codings of these parameters in signals and statistical couplings between the parameters, themselves. Equipment couplings are realized, first, in the multi-dimensional discriminator (in the form of common input terminals, grouping of output quantities if the parameter is coded in various signals differently, and feed of the measured values of parameters immediately to all partial discriminators) and, secondly, in smoothing circuits (in the form of common input and output quantities). Concretization of the appearing couplings is given in subsequent paragraphs, where there is conducted particular synthesis of discriminators and smoothing circuits for signals and parameters with various properties.

§ 12.4. Synthesis of Multi-Dimensional Discriminators

As it was shown in § 12.3, as one of their basic elements multi-dimensional meters contain a multi-dimensional (joint) discriminator. Now it is necessary to determine general rules for finding operations of this discriminator $z^{(\alpha)}(t)$ and

~700-

its characteristics $K^{(\alpha\beta)}$ for different forms of input signals. As also for onedimensional meters, in general consideration of discriminators at the base of our classification we place statistical properties of the signal; examples of measurement of concrete parameters will be studied later.

Let us assume we have m input mixtures

$$f^{(i)}y(t) = f^{(i)}v(t, \lambda(t)) + f^{(i)}n(t)$$
 and $(i = 1, 2, ..., m)$,

consisting of useful signals ⁽¹⁾v(t), whose form for fixed $\lambda(t)$ completely is assigned, and noises ⁽¹⁾n(t). Signals depend on l parameters { $\lambda^{(1)}(t)$, ..., $\lambda^{(l)}(t)$ } = $\lambda(t)$.

Properties of noises are assigned by the correlation matrix

$$\mathbf{R}_{\mathbf{m}}(\mathbf{s}) = \overline{\|({}^{(1)}\boldsymbol{n}(t)({}^{(1)}\boldsymbol{n}(t+\mathbf{s}))\|}.$$
(12.4.1)

Following the method of § 12.3, it is necessary to separate from the logarithm of the likelihood function, during Gaussian noises equal to

$$L(\mathbf{y}(t), \lambda(t)) = -\frac{1}{2} \int_{t-\lambda/2}^{t+\lambda/2} [\mathbf{y}(t_1) - \mathbf{v}(t_1, \lambda(t_1))]^+ \times \mathbf{W}_{\mathrm{III}}(t_1, t_2) [\mathbf{y}(t_2) - \mathbf{v}(t_2, \lambda(t_2))] dt_1 dt_2$$

 $[W_{iii}(t_1, t_2) - function, the inverse of R_{iii}(t_1, t_2)]$, the integrand

$$l(\mathbf{y}, \boldsymbol{\lambda}; t) = -\frac{1}{2} \left[\mathbf{y}(t) - \mathbf{v}(t, \boldsymbol{\lambda}(t)) \right]^{+} \int_{t-\boldsymbol{\lambda}/\boldsymbol{\lambda}}^{t+\boldsymbol{\lambda}/2} W_{\mathrm{III}}(t, \tau) \times \left[\mathbf{y}(\tau) - \mathbf{v}(\tau, \boldsymbol{\lambda}(\tau)) \right] d\tau. \quad (12.4.2)$$

After differentiation of (12.4.2) with respect to $\lambda^{(\alpha)}$ we obtain two statistically equivalent notations of the α -th operation of the discriminator:

$$\boldsymbol{z}^{(\mathbf{e})}(t) = -\frac{\partial \mathbf{v}\left(t, \hat{\boldsymbol{\lambda}}\left(t\right)\right) +}{\partial \boldsymbol{\lambda}^{(\mathbf{e})}} \int_{t-\Delta/2}^{t+\Delta/2} W_{\mathrm{III}}\left(t, \tau\right) \left[\mathbf{y}\left(\tau\right) - \mathbf{v}\left(\tau, \hat{\boldsymbol{\lambda}}\left(t\right)\right)\right] d\tau, \qquad (12.1.3)$$

$$z^{(\mathbf{e})}(t) = - \left[\mathbf{y}(t) - \mathbf{v}(t, \hat{\lambda}(t)) \right]^{+} \int_{t-\hat{\mathbf{h}}/2}^{t+\hat{\mathbf{h}}/2} \mathbf{W}_{y_{1}}(t, \tau) \frac{\partial \mathbf{v}(\tau, \hat{\lambda}(t))}{\partial \lambda} d\tau, \qquad (12.4.4)$$

which can be interpreted in the form of the block diagrams of Figs. 12.10 and 12.11. According to (12.4.3) and Fig. 12.10 from the input mixtures there are subtracted known forms of signals with the value of parameters equal to the measured ones; then the differences are passed through a matrix of linear filters possedsing rejector properties with respect to interference. Finally, grouped outputs of the matrix of filters are multiplied by the derivatives of expected forms of the signal

-701-


Fig. 12.10. First variant of a discriminator with regular signals: 1) subtractor; 2) units of formation of reference signals; 3) linear filters; 4, 6) adders; 5) multipliers.



Fig. 12.11. Second variant of a discriminator with regular signals: 1) subtractor; 2) unit of formation of reference signals; 3) multipliers; 4. (.) adders; 5) linear filters.

with respect to the measured parameter $\frac{\partial (1) \gamma(t, \lambda)}{\partial \lambda(\alpha)}$, and then are added. For any of the parameters common are elements up to the output of the matrix of rejector filters; different only are derivatives with respect to $\lambda^{(\alpha)}$ and thereby the final sums. The other monification (Fig. 12.10) [sie] differs from the first in that rejection of interference is carried out in circuits of feeding of derivatives,

which is absolutely equivalent in the final result.

()

Elements of the matrix reflecting accuracy properties of the discriminator are expressed through second derivatives of (12.4.2):

$$\mathcal{K}^{(\mathfrak{a}\mathfrak{p})} = \lim_{r \to \infty} \frac{1}{r} \int_{0}^{r} \int_{0}^{t} \frac{\partial}{\partial \lambda^{(\mathfrak{a})}} \mathbf{v}^{+}(t_{1}, \lambda) \mathbf{W}_{\mathbf{m}}(t_{1}, t_{\mathfrak{a}}) \frac{\partial}{\partial \lambda^{(\mathfrak{p})}} \mathbf{v}(t_{\mathfrak{a}}, \lambda) dt_{1} dt_{\mathfrak{a}}$$
(12.4.5)

and are determined by properties of the interference and the signals. Due to possible crosscorrelatedness of interferences in various mixtures, expressed in the nondiagonalness of matrices $\mathbf{R}_{\mathrm{H}}(t_1, t_2)$ and $\mathbf{W}_{\mathrm{H}}(t_1, t_2)$, for diagonalness of matrix **K** it is still insufficient that each signal depends on only one parameter. Only if the signals are broken up into i groups $(1,1)_{\mathrm{V}}(t)$, ..., $\begin{pmatrix} (1,m_1)\\ & & \end{pmatrix}_{\mathrm{V}}(t)$, ..., $(l,1)_{\mathrm{V}}(t)$, ...,

 $(l,1)_{v(t)}, \ldots, {l,m_l \choose v(t)}$ (where $m_1 + m_2 + \ldots + m_l = m$), depending each on its own parameter, and noises between channels are not correlated, do we have

$$\mathcal{K}^{(a,b)} \stackrel{:=}{=} \mathcal{K}^{(a)} \delta_{a,b},$$

$$\mathcal{K}^{(a)} = \lim_{T \to \infty} \frac{1}{T} \int_{\partial \lambda^{(a)}}^{\partial} v^+(t_1, \lambda^{(a)}) W^{(a)}_{\mathrm{tr}}(t_1, t_2) \frac{\partial}{\partial \lambda^{(a)}} U(t_2, \lambda^{(a)}) dt_1 dt_2, \qquad (12.1, 6)$$

where $(\alpha)_{\mathbf{v}}(t) \approx \{ (\alpha, 1)_{\mathbf{v}}(t), \ldots, (\alpha, m_{\alpha})_{\mathbf{v}}(t) \} - \text{signals of the α-th group.}$

. With a pulse signal with period ${\rm T}_{\rm r}$ only the form of operation of the discriminator changes:

$$\boldsymbol{z}_{\mathbf{A}}^{(\mathbf{a})} = \int_{(\mathbf{A}-1)\Gamma_{\mathbf{y}}}^{\mathbf{A}\Gamma_{\mathbf{y}}} \frac{\partial \mathbf{v}^{+}(t_{1},\hat{\boldsymbol{\lambda}}(t_{\mathbf{A}}))}{\partial \boldsymbol{\lambda}^{(\mathbf{a})}} \mathbf{W}_{\mathbf{H}}(t_{1},t_{3}) \left[\mathbf{y}(t_{3}) - \mathbf{v}(t_{3},\hat{\boldsymbol{\lambda}}(t_{\mathbf{A}})) \right] dt_{3}, \qquad (12.4.7)$$

which includes additionally accumulation inside the k-th period. Accuracy characteristic (12.4.5) remains the same (for $T = T_n$).

In the case of stationary interferences, by introducing their spectral matrix $S_{ij}(\omega)$ it is easy to obtain

$$z_{k} = \frac{1}{2\pi} \int_{-\infty}^{+\infty} \frac{\partial}{\partial \lambda^{(n)}} V(\omega; \hat{\lambda}) S_{\underline{\omega}}^{-1}(\omega) [Y(\omega) - V(\omega; \hat{\lambda})] d\omega, \qquad (12.4.3)$$

where $\mathbf{V}(\omega)$, $\mathbf{Y}(\omega)$ - columns of spectra of signals $\mathbf{v}(t)$ and input realizations $\mathbf{y}(t)$ in the k-th period.

Formula (12.4.3) graphically illustrates rejector properties of the matrix of filters.

12.4.2. Gaussian Coherent Signals

As also in § 6.7, for generality we consider the case when each of the input voltages has the form

-703-

i.e., consists of a whole set of $p_{,j}$ differently modulated components with various correlation properties, depending in this case on several parameters $\lambda(t) = \{\lambda^{(1)}(t), \ldots, \lambda^{(1)}(t)\}$, and white noises (j)n(t). The whole set of functions (12.4.9) can be presented in the form

 $y(t) = \operatorname{Re} \left\{ \mathbf{U}^{+}(t; \lambda(t)) \,\boldsymbol{\theta}(t) \, \mathrm{e}^{t \boldsymbol{\omega}_{o} t} \right\} + \mathbf{n}(t), \qquad (12.4.10)$

where $\mathbf{U}(t)$ - complex nearly diagonal matrix of coefficients of modulation u(t), whose j-th diagonal element is column

$$\{(j)\mu_{i}(t),\ldots,(j)\mu_{p_{i}}(t)\},\$$

where $(j)_{u_k}(t) = (j)_{u_{ak}}(t) \exp \{i^{(j)}\psi_k(t)\}; \theta(t) - \text{complex column vector whose } j-th subcolumn element is set$

 $(i)\emptyset(l) = \{ (i)E_1(l) e^{i(l)\varphi_1(l)}, \dots, (i)E_{p_j}(l) e^{i(l)\varphi_{p_j}(l)} \},\$

reflecting random modulations; $\mathbf{y}(t)$, $\mathbf{n}(t) - \text{simple column vectors of order (m × 1),}$ composed of input realizations ⁽¹⁾ $\mathbf{y}(t)$ and noises ⁽¹⁾ $\mathbf{n}(t)$. The correlation matrix of the whole set of mixtures (12.4.9) is equal to

 $\mathbf{R}(t_1, t_2) = \overline{\mathbf{y}(t_1)\mathbf{y}^+(t_2)} = \mathbf{Re}\left\{\mathbf{U}^+(t_1, \lambda)\mathbf{r}(t_1 - t_2) \times \mathbf{U}^+(t_2, \lambda)e^{i\mathbf{\omega}_2 t_1 - t_2}\right\} + \mathbf{N\delta}(t_1 - t_2), (12.4.11)$ where $\mathbf{r}(\mathbf{t}_1 - \mathbf{t}_2) = \frac{1}{2}\overline{\mathbf{\theta}'(\mathbf{t}_1)\mathbf{\theta}^+(\mathbf{t}_2)} - \text{complex matrix function of order}\left(\sum_{i=1}^{m} p_i \times \sum_{i=1}^{m} p_i\right)$ with submatrix elements $(ij)\mathbf{r}(\mathbf{t}_1 - \mathbf{t}_2) = |\mathbf{i}^{(ij)}\mathbf{r}_{pq}(\mathbf{t}_1 - \mathbf{t}_2)||, \text{ where } (id)\mathbf{r}_{pq}(\mathbf{t}_1 - \mathbf{t}_2)$ is the function of crosscorrelation of random processes $(i)\mathbf{E}_p(\mathbf{t}_1) \cos((i)\mathbf{\phi}_p(\mathbf{t}_1))$ and $(j)\mathbf{E}_q(\mathbf{t}_2) \cos((j)\mathbf{\phi}_q(\mathbf{t}_2)), \text{ considered stationary}. Matrix N in (12.4.11) is the matrix of spectral densities of white noises, as also in Paragraph 12.4.1, and is not necessarily diagonal.$

Matrix $\Psi(t_1, t_2)$, the reciprocal of (12.4.11), which is necessary for construction of the Divelihood functional, as also in § 6.7 is sought in the form

$$W(t_1, t_2) = -\operatorname{Re} \{ N^{-1} U^+(t_1) W(t_1, t_2) U^*(t_3) N^{-1} e^{i\omega_0(t_1 - t_3)} \} + N^{-1} \delta(t_1 - t_2), \quad (12.4.12)$$

from which for auxiliary matrix function $W(t_1, t_2)$, structurally similar to
 $r(t_1 - t_2)$, we have integral equation

$$\frac{1}{2}\int \mathbf{r}(t_1 - s) \mathbf{B}_{q}(s) \mathbf{w}(s, t_1) ds + \mathbf{w}(t_1, t_2) = \mathbf{r}(t_1 - t_2), \qquad (12.4.13)$$

where $\mathbf{B}_{O}(s) = \mathbf{U}^{*}(s)\mathbf{N}^{-1}\mathbf{U}^{+}(s)$ — matrix, determined by the form of regular modulations and power properties of the signals.

Assuming modulation of all separate components of mixtures to be rapidly changing, it is possible to average $\mathbf{B}_{0}(s)$ over the time under the sign of the integral, introducing constant matrix

$$\mathbf{\bar{B}}_{\bullet} = \lim_{r \to \infty} \frac{1}{r} \int_{0}^{r} \mathbf{U}^{*}(s) \mathbf{N}^{-1} \mathbf{U}^{+}(s) \, ds. \qquad (12.4.1^{1})$$

Then the Fourier transform from $\mathbf{w}(t_1, t_2) = \mathbf{w}(t_1 - t_2)$ will be expressed in the form

$$\widetilde{\mathbf{w}}(\boldsymbol{\omega}) = \left[\mathbf{I} + \frac{1}{2} \mathbf{S}_{\mathbf{c}}(\boldsymbol{\omega}) \, \overline{\mathbf{B}}_{\mathbf{o}}\right]^{-1} \mathbf{S}_{\mathbf{c}}(\boldsymbol{\omega}), \qquad (12.4.15)$$

where $\mathbf{S}_{c}(w)$ - spectral matrix of fluctations of the whole set of useful signals [i.e., the Fourier transform from $\mathbf{r}(t)$]. Thus,

$$l(\mathbf{y}, \boldsymbol{\lambda}, t) = \operatorname{const} \sum_{i, j=1}^{m} \sum_{k=1}^{P_i} \sum_{l=1}^{p_j} \int_{t_0}^{t} {}^{(i)} x_k(\tau_i, \boldsymbol{\lambda}(\tau_i)) \times \\ \times {}^{ij} h_{kl}^{(1)}(t - \tau_i) \cos \omega_{\Pi p} (t - \tau_i) d\tau_i \int_{t_0}^{t} {}^{(j)} x_l(\tau_a, \boldsymbol{\lambda}(\tau_i)) \times \\ \times {}^{(ij)} h_{kl}^{(2)}(t - \tau_a) \cos \omega_{\Pi p} (t - \tau_a) d\tau_a, \qquad (12.4.16)$$

where

$${}^{(i)}x_{k}(t,\lambda) = {}^{(i)}y(t){}^{(i)}u_{ak}(t,\lambda)\cos\left[\left(\omega_{o}+\omega_{ap}\right)t+{}^{(i)}\psi_{k}(t,\lambda(t))\right]$$

is the result of correlation processing of the i-th mixture for the purpose of singling out from it the k-th component of the signal and with simultaneous transfer of oscillations to intermediate frequency $\omega_{\rm BF}$. Fulse response envelopes $^{i,j}h_{kl}^{(1)}(t)$ and $^{4,j}h_{kl}^{(2)}(t)$ are selected physically realizable and such that this equality is satisfied:

$$\int_{-\infty}^{+\infty} {}^{(ij)} h_{kl}^{(1)}(t_1) e^{-i\omega t_l} dt_1 \int_{-\infty}^{+\infty} {}^{(ij)} h_{kl}^{(2)}(t_3) e^{i\omega t_3} dt_3 = {}^{ij} \overline{w}_{kl}(\omega), \qquad (12.4.27)$$

where $(ij)\widetilde{w}_{kl}(\omega)$ - element of complex matrix $\widetilde{w}(\omega)$, determined by relationship (12.4.1!).

Operation of the discriminator when generalized signal-to-noise ratios do not depend on $\lambda(t)$ is expressed by a formula obtained by differentiation of (12.4.19) with respect to $\lambda^{(\alpha)}$

$$\mathbf{z}^{(\mathbf{e})}(t) = \operatorname{const} \sum_{i, j=1}^{m} \sum_{k=1}^{p} \sum_{l=1}^{t} \int_{t_0}^{t} \frac{\partial}{\partial \lambda^{(\mathbf{e})}} {}^{(i)} x_k [\tau_1, \hat{\lambda}(\tau_1)]^{(i)/h_{kl}^{(i)}}(t-\tau_1) \times \\ \times \cos \omega_{\mathrm{RP}} (t-\tau_1) d\tau_1 \int_{t_0}^{t} {}^{(j)} x_i [\tau_2; \hat{\lambda}(\tau_2)]^{(ij)} h_{kl}^{(2)}(t-\tau_0) \times \\ \times \cos \omega_{\mathrm{RP}} (t-\tau_2) d\tau_4.$$
(12.4.13)

atoria de la contrata de sur alterrativa de la contrata de sur de la contrata de sur de la contrata de sur

Ĭ

Examples of optimum processing of several input signals already were given in the preceding chapters; therefore in view of the excessive generality of case (12.4.18) the circuit of the discriminators is best given only in the study of concrete examples.

For matrix element $\overline{\mathbf{K}(t)} \approx \mathbf{K}$ in the most general case of a set of normal processes with correlation matrix $\mathbf{R}(t_1, t_2; \lambda)$ and its reciprocal $\mathbf{W}(t_1, t_2; \lambda)$ we have a formula, analogous to (6.7.65):

$$K^{(\mathfrak{a}\mathfrak{p})} = -\lim_{T \to \infty} \frac{1}{2T} \int_{0}^{T} \int_{0}^{T} \operatorname{spur} \left\{ \frac{\partial}{\partial \lambda^{(\mathfrak{a})}} W(t_{1}, t_{2}; \lambda) \times \frac{\partial}{\partial \lambda^{(\mathfrak{p})}} R^{+}(t_{1}, t_{2}; \lambda) \right\} dt_{1} dt_{2} = -\lim_{T \to \infty} \frac{1}{2T} \int_{0}^{T} \int_{0}^{T} \operatorname{spur} \left\{ \frac{\partial}{\partial \lambda^{(\mathfrak{p})}} W(t_{1}, t_{2}; \lambda) \frac{\partial}{\partial \lambda^{(\mathfrak{a})}} R^{+}(t_{1}, t_{2}; \lambda) \right\} dt_{1} dt_{2}.$$

$$(12.4.19)$$

In the considered case this gives

$$\begin{aligned} \mathcal{K}^{(\alpha\beta)} &= \lim_{T \to \infty} \operatorname{Re} \frac{1}{2T} \int_{0}^{T} \int_{0}^{T} \operatorname{spur} \left\{ \mathbf{B}_{2}^{(\alpha\beta)}(t_{1}) \mathbf{w} \left(t_{1} - t_{2}\right) \overline{\mathbf{B}}_{0} \mathbf{r} \left(t_{1} - t_{2}\right) + \right. \\ \left. + \left. \mathbf{B}_{1}^{(\alpha)}(t_{1}) \mathbf{w} \left(t_{1} - t_{2}\right) \mathbf{B}_{1}^{(\beta)}(t_{2}) \mathbf{r} \left(t_{1} - t_{2}\right) \right\} dt_{1} dt_{2}, \end{aligned}$$

where

$$\mathbf{B}_{2}^{(*\beta)}(t) = \frac{\partial}{\partial \lambda^{(*)}} \mathbf{U}^{*}(t,\lambda) \mathbf{N}^{-1} \frac{\partial}{\partial \lambda^{(\beta)}} \mathbf{U}^{+}(t,\lambda),$$

$$\mathbf{B}_{1}^{(*)}(t) = \mathbf{U}(t,\lambda) \mathbf{N}^{-1} \frac{\partial}{\partial \lambda^{(*)}} \mathbf{U}^{*+}(t,\lambda) \qquad (12.4.21)$$

are matrices structurally similar to matrix \mathbf{E}_{O} .

If there exist constant limits

$$\lim_{T \to \infty} \frac{1}{T} \int_{0}^{T} \mathbf{B}_{2}^{(\alpha\beta)}(t) dt = \bar{\mathbf{B}}_{2}^{(\alpha\beta)},$$

$$\lim_{T \to \infty} \frac{1}{T} \int_{0}^{T} \mathbf{B}_{1}^{(\alpha)}(t) dt = \bar{\mathbf{B}}_{1}^{(\alpha)}, \qquad (12.4.22)$$

taking into account (12.4.15) we can simplify relationship (12.4.20):

$$\mathcal{K}^{(\alpha\beta)} = \frac{1}{4\pi} \int_{-\infty}^{+\infty} \operatorname{spur}\left\{ \left[1 + \frac{1}{2} \operatorname{S}_{\alpha}(\omega) \overline{\operatorname{B}}_{\alpha} \right]^{-1} \operatorname{S}_{\alpha}(\omega) \times \left[\overline{\operatorname{B}}_{\alpha} \operatorname{S}_{\alpha}^{+}(\omega) \overline{\operatorname{B}}_{2}^{(\alpha\beta)} + \overline{\operatorname{B}}_{1}^{(\beta)} \operatorname{S}_{\alpha}^{+}(\omega) \overline{\operatorname{B}}_{1}^{(\alpha)} \right] \right\} d\omega. \quad (12.4.23)$$

For the simple example when $p_1 = 1$, $\mathbf{N} = N_0 \mathbf{I}$, and power of the i-th signal is P_1 ,

 $\mathcal{K}^{(\alpha\beta)} = J_1^{(\alpha\beta)}(U) J_{\alpha}(h_{\Sigma}, S_{0}),$

so that
$$(ik)$$
 S $_{C}(\omega) = \sqrt{\frac{P_{1}P_{k}}{\Delta f_{C}}} S_{O}(\omega) (S_{O}(0) = 1)$, we have

where

$$J_{1}^{(\alpha\beta)}(U) = [b_{2}^{(\alpha\beta)} - b_{1}^{(\alpha)}b_{1}^{(\beta)}];$$

$$J_{a} = \frac{1}{\pi} \int_{-\infty}^{+\infty} \frac{h_{2}^{2} S_{0}^{2}(\omega) d\omega}{1 + h_{2} S_{0}(\omega)};$$

$$b_{2}^{(\alpha\beta)} = \lim_{T \to \infty} \frac{1}{T} \int_{0}^{T} \sum_{h=1}^{m} a_{h} \left[\frac{\partial^{(h)}u_{a}(t)}{\partial^{\lambda^{(\alpha)}}} \frac{\partial^{(h)}u_{a}(t)}{\partial^{\lambda^{(\beta)}}} + \frac{(h)u_{a}^{2}(t)}{\partial^{\lambda^{(\alpha)}}} \frac{\partial^{(h)}\psi(t)}{\partial^{\lambda^{(\beta)}}} \right] dt;$$

$$(12.4.25)$$

$$b_{1}^{(\alpha)} = \lim_{T \to \infty} \frac{1}{T} \int_{0}^{T} \sum_{k=1}^{m} a_{k}^{(h)}u_{a}^{2}(t) \frac{\partial^{(h)}\psi(t)}{\partial^{\lambda^{(\alpha)}}} dt;$$

$$a_{k} = P_{k} / \sum_{i=1}^{m} P_{i}; \quad h_{2} = \sum_{i=1}^{m} P_{i} / 2N_{e} \Delta f_{c}.$$

(12.4.24)

A sufficient (but not necessary) condition of diagonalness of matrix K here is the presence in each signal of only amplitude or only phase modulation, depending on no more than one parameter. But this diagonalness can also be observed on average in a number of other cases.

Absolutely in that same plan we can consider the case of weakly correlated sendings of a coherent pulse signal (see § 6.7). Subsequently we shall only discuss the limiting case of incoherent independently fluctuating Gaussian sendings.

12.4.3. Gaussian Uncorrelated Sendings

Let us consider again the case of many input mixtures (i)y(t) (i = 1, ..., m), each of which can contain a whole series of useful components, possibly, with different forms of modulation. In distinction from Paragraph 12.4.2 we assume that fluctuations of useful components are independent from period to period, but incide the period are completely correlated. Interferences in all mixtures we shall consider white noises. Such a model is valid to equal measure for coherent and incoherent signals, since additional phase shift from incoherence already changes

-707-

nothing in the case of absence of correlation of fluctuations of signals in neighboring periods. With these assumptions it is possible to conduct synthesis of circuits, considering separate periods of repetition. Construction of the likelihood function in this case is conducted analogously to Paragraph 12,4.2. The column of the set of mixtures in a separate period will take form

$$\mathbf{y}(t) = \operatorname{Re} \{ \mathbf{U}_{0}^{+}(t) \ \theta e^{t \cdot \mathbf{w}_{0} t} \} + \mathbf{n}(t), \qquad (12.4.26)$$

where ${\bf U}_{\mathbb{Q}}(t)$ - nearly diagonal matrix of coefficients of regular modulation in one period;

heta - complex column-vector of their random amplitudes and phases:

 $\theta^{(h)} = \{E_1^{(h)} e^{i\varphi_1^{(h)}}, \dots, E_{p_h}^{(h)} e^{i\varphi_{p_h}^{(h)}}\} \text{ (in distinction from Paragraph 12.4.2 it is constant inside the period);}$

n(t) - column vector of white noises.

The correlation matrix of mixtures and its reciprocal can be recorded

$$R(t_{s}, t_{s}) = \operatorname{Re} \{ U_{0}^{+}(t_{s}) r U_{0}^{*}(t_{s}) e^{t \omega_{0}(t_{s} - t_{s})} \} + N\delta(t_{s} - t_{s}),$$

$$W(t_{s}, t_{s}) = -\operatorname{Re} \{ N^{-1} U_{0}^{+}(t_{s}) \left[I + \frac{1}{2} r \overline{B}_{s} \right]^{-1} r U_{0}^{*}(t_{s}) \times N^{-1} e^{t \omega_{0}(t_{s} - t_{s})} \} + N^{-1}\delta(t_{s} - t_{s}),$$
(12.4.27)

where $\mathbf{r} = \text{matrix} \left\{ \sum_{i=1}^{m} p_i \times \sum_{i=1}^{m} p_i \right\}$ of correlation between all components of mixtures; $\overline{B}_{\bullet} = \frac{1}{T_r} \int_{0}^{r} U_{0}^{\bullet}(s) \mathbf{N}^{-1} U_{0}^{+}(s) \, ds. \qquad (12.4.25)$

In the correlation variant the operation of the discriminator according to (12.4.27) has form

$$z^{(\alpha)} = \operatorname{const} \sum_{i, j=1}^{m} \sum_{k=1}^{p_i} \sum_{i=1}^{p_j} \sum_{i=1}^{(ij)} w_{ki} \left[\int_{0}^{T_r} \int_{0}^{(j)} S_k(t, \lambda) dt \right] \times \int_{0}^{T_r} \frac{\partial (i) S_i(t, \lambda)}{\partial \lambda^{(\alpha)}} + \int_{0}^{T_r} \int_{0}^{(i)} C_k(t, \lambda) dt \int_{0}^{T_r} \frac{\partial (i) C_i(t, \lambda)}{\partial \lambda^{(\alpha)}} dt,$$

where

$${}^{(i)}S_{\mathbf{A}}(t,\lambda) := {}^{(i)}y(t){}^{(i)}u_{\mathbf{a}\mathbf{k}}(t,\lambda)\sin\left[\omega_{\mathbf{0}}t+{}^{(i)}\psi_{\mathbf{A}}(t,\lambda)\right],$$
$${}^{(i)}C_{\mathbf{A}}(t,\lambda) := {}^{(i)}y(t){}^{(i)}u_{\mathbf{a}\mathbf{k}}(t,\lambda)\cos\left[\omega_{\mathbf{0}}t+{}^{(i)}\psi_{\mathbf{k}}(t,\lambda)\right]$$

are results of correlation processing.

An element of matrix \mathbf{x} by the formula (12.4.19) can easily be reduced to a form analogous to (12.4.23):

$$\mathcal{K}^{(\mathfrak{a}\mathfrak{p})} = \frac{1}{T_r} \operatorname{spur} \left\{ \left[\mathbf{I} + \frac{1}{2} \mathbf{r} \mathbf{\bar{B}}_0 \right]^{-1} \mathbf{r} \left[\mathbf{\bar{B}}_0 \mathbf{r} \mathbf{\bar{B}}_2^{(\mathfrak{a}\mathfrak{p})} + \mathbf{\bar{B}}_1^{(\mathfrak{a})} \mathbf{r} \mathbf{\bar{B}}_1^{(\mathfrak{p})} \right] \right\}, \qquad (12.4.29)$$

Ĺ

where

$$\bar{\mathbf{B}}_{1}^{(\alpha)} = \frac{1}{T_{r}} \int_{0}^{T_{r}} \frac{\partial U_{\bullet}(s)}{\partial \lambda^{(\alpha)}} \mathbf{N}^{-1} U_{0}^{*+}(s) ds, \quad \bar{\mathbf{B}}_{2}^{(\alpha\beta)} = \frac{1}{T_{r}} \int_{0}^{T_{r}} \frac{\partial U_{\bullet}(s)}{\partial \lambda^{(\alpha)}} \mathbf{N}^{-1} \frac{\partial U_{0}^{*+}(s)}{\partial \lambda^{(\beta)}} ds \quad (12.4.30)$$

are matrices, reflecting the coding of measured quantities in regular modulation of components of mixtures.

It that particular case when the mixtures contain only one useful component cash, where these components are completely correlated, and while noises are independent, $K^{(\alpha\beta)}$ again is expressed by formula (12.4.24), where J_{α} is replaced by

$$r_{a} = \frac{2q_{r}^{2}}{(1+q_{s})T_{r}},$$

(12.4.31)

and $q_2 = \frac{1}{2N_0}$ is the ratio of total energy of the useful signal in all channels to the intensity of white noise in one of them.

12.4.4. Methods of Approximation for Construction of Discriminators and the Method of Comparison of Performance of Circuits

As also in the one-dimensional case, in engineering applications permissible and inevitable are certain deviations from optimum processing during realization of discriminators, hopefully close to optimum. In particular, a natural method of forming a discriminator is replacement of the exact expression for the derivative of $F(\mathbf{y}|\mathbf{\lambda})$ with respect to $\lambda^{(\alpha)}$ by a finite difference

$$z^{(\alpha)}(t) \approx \frac{1}{\Delta \lambda^{(\alpha)}} \left[l\left(t, \mathbf{y}, \hat{\lambda} + \frac{\Delta \lambda^{(\alpha)}}{2}\right) - l\left(t, \mathbf{y}, \hat{\lambda} - \frac{\Delta \lambda^{(\alpha)}}{2}\right) \right], \qquad (12 \dots 32)$$

where detuning is implied performed only with respect to the α -th parameter by a 4-antity smaller than the width of the peak of autocorrelation of the signal with respect to $\kappa^{(\alpha)}$. If it is technically difficult to have two channels at once, in a number of cases it is permissible to use one, alternately detuning it in different directions from the measured value of some parameter. In principle during realization of a multi-dimensional discriminator of l parameters it is possible to use just one channel if we select a switching function, which as it were "examines" points of the likelihood function removed small magnitudes of $\pm \Delta \lambda^{(\alpha)}$ ($\alpha = 1, \ldots, l_{j}$ from $\hat{\ell}(t)$ consecutively for all parameters.

Hontrivial in the multi-dimensional case is the question of comparison of different streams of discriminators, close to optimum or far from them, by their accuracy characteristics. As was shown in Paragraph 12.2.3, these characteristics are described by the matrix of equivalent spectral densities \mathbf{K}^{-1} in the optimum

-709-

case and matrix $\mathbf{S}_{9\mathbf{X}\mathbf{B}} = \mathbf{K}_{\mathbf{H}}^{-1}\mathbf{S}_{\mathbf{0}}\mathbf{K}_{\mathbf{H}}^{-1+}$ in the general case. The geometric interpretation of these matrices can be, as it is known, an ellipsoid of dispersion in the parameter space with the same second \mathbf{p}^{-1} on all axes. The first rough method of comparison of geometric dimensio. The ellipsoid is comparison of moments on the axes $\lambda^{(\alpha)}$ ($\alpha = 1, \ldots, l$) without taking into account mixed moments. This is sequential comparison of diagonal elements of two matrices. If we need more precise comparison it is possible to compare volumes of the ellipsoids, expressed through determinants of the matrices. If both these methods indicate approximate coincidence of $\mathbf{S}_{\mathbf{0}\mathbf{1}\mathbf{0}}$ with \mathbf{K}^{-1} , further attempts to improve processing, naturally, are unnecessary.

§ 12.5. Synthesis of Smoothing Circuits and Resultant Accuracy of Measurement

As also in the one-dimensional case, synthesis of smoothing circuits is analogous (for the two-loop variant of the meter) to synthesis of multi-dimensional Wiener filters. However the equivalent Wiener problem is very complicated, inasmuch as in it, it is necessary to consider the interconnection not only of measured parameters, but also of interferences superimposed on them. Such questions have been insufficiently studied in the literature. Therefore, the material below is arranged in separate parts so that it will be useful to those who are interested in the purely Wiener problem. Different classes of correlation matrices studied below are fully analogous to different correlation functions of the measured parameters considered for the one-dimensional case in § $\epsilon.8$.

12.5.1. Parameters - Random Processes with Stationary Increments We start our consideration with the case of stationary random processes, included as a subclass in random processes with stationary increments. Equation (12.3.19) for $\mathbf{K}(\mathbf{t}) = \mathbf{K}$ here has the form

$$c(t, \tau) + \int_{0}^{t} c(t, s) KR(s - \tau) ds = R(t - \tau)$$
 (12.9.1)

and is an integral matrix equation (or system of integral equations) with a nucleus depending on the difference of arguments. Methods of solutions of such equations were studied, for instance, in [55]. Exact solution for an arbitrary time of observation in principle can be obtained by a method analogous to that described in § 6.4, i.e., by transition to the proper system of differential equations. However to solve these equations is complicated even in the case of two parameters. Therefore, it is useful to immediately pass to a study of limiting operators,

-710-

corresponding to a time of observation large as compared to the interval of correlation of parameters, when equation (12.5.1) passes into the matrix generalization of the Wiener-Hopf equation:

$$c(t) + \int_{0}^{\infty} c(\tau) KR(t-\tau) d\tau = R(t) (t>0),$$
 (12.5.2)

inasmuch as c(t) is sought as a function of the difference of its own arguments. Direct application of Fourier transformation to (12.5.2) leads to a solution in the class of physically unrealizable filters, for which $c(t) \neq 0$ when t < 0.*

To find a physically realizable solution in the literature there have been offered several methods. Below we give them names, not to any extent conventional.

a. The method of factoring a matrix [55, 56], having the most general

significance, is based on presentation $I + KS(\omega)$, where $S(\omega) = \int_{-\infty}^{+\infty} R(t) e^{-t\omega t} dt$ is the spectral matrix of parameters, in the form of the product

$$\mathbf{I} + \mathbf{KS}(\boldsymbol{\omega}) = \Psi_{+}(\boldsymbol{\omega}) \Psi_{-}(\boldsymbol{\omega}), \qquad (12.5.3)$$

where $\Psi_{-}(\omega)$, $\Psi_{+}(\omega)$ — matrix functions with analytic elements and determinants differing from zero correspondingly in the upper and lower half-planes of the complex variable ω .

The fundamental potential of factoring is proved with very wide assumptions about matrix $\mathbf{I} + \mathbf{KS}(\omega)$. The physically realizable solution of (12.5.2) has Fourier transform

$$C(i\omega) = [S(\omega) \Psi_{-}^{-1}(\omega)]_{+} \Psi_{+}^{-1}(\omega) = \frac{1}{2\pi} \int_{0}^{\infty} e^{-i\omega t} dt \int_{-\infty}^{+\infty} e^{i\omega t} d\mu S(u) \Psi_{-}^{-1}(u) \Psi_{+}^{-1}(\omega), \quad (12.5.4)$$

where $[]_+$ signifies the operation of taking that part of the expression in parentheses which has poles only in the upper half-plane of ω . Analytically, forming of $C(i\omega)$ from the factored matrix is presented in (12.5.4) by two integral.

Practically, however, factoring of (12.5.3) is a difficult problem. With rational spectral matrices, the most important for applications, a series of other methods turn out to be more convenient.

b. The method of factoring the determinant [57] is based on the expansion

$$\det [I + KS(\omega)] = \psi_{+}(\omega) \psi_{-}(\omega), \qquad (12.5.5)$$

where $\psi_{\perp}(\omega)\psi_{\perp}(\omega) - \text{scalar factors with the same properties, as <math>\Psi_{\perp}(\omega)$, $\Psi_{\perp}(\omega)$ in

^{*}In principle operations of "physically unrealizable" filters can be executed by means of recording the input realization and subsequent processing, but we are interested in filtration with direct delivery of the result.

(12.5.3). The solution has the form

 $\mathbf{C}(i\boldsymbol{\omega}) = [\mathbf{S}(\boldsymbol{\omega})[\mathbf{I} + \mathbf{KS}(\boldsymbol{\omega})]^{-1} \boldsymbol{\psi}_{+}(\boldsymbol{\omega})]_{+} \frac{1}{\boldsymbol{\psi}_{+}(\boldsymbol{\omega})} + \frac{1}{\boldsymbol{\psi}_{+}(\boldsymbol{\omega})} \mathbf{C}_{0}(i\boldsymbol{\omega}), \quad (12.5.6)$

where $O_0(i\omega)$ - matrix with elements in the form $\sum_{\omega = \gamma_i} \frac{A_j^{\mu\nu}}{\omega - \gamma_i}$;

 γ_{1} - the pole of matrix $[I + KS(\omega)]^{-1}\psi_{+}(\omega)\psi_{-}(\omega)$ in the upper half-plane of ω_{1}

 $A_i^{(\alpha\beta)}$ - undetermined factors, selected from condition

$$[C (i\omega) [1 + KS (v)]_{+} = [S (\omega)]_{+}. \qquad (12.5.7)$$

c. The method of undetermined coefficients [53] consists of seeking $c^{(\alpha\beta)}(i\alpha)$

in the form of rational functions of $\sum_{k=1}^{m-1} \frac{B_k^{(i)} \omega^k}{\prod_{j=1}^m (\omega - \gamma_j)}$, where γ_j - all zeroes of the

numerator of $\psi_{+}(\omega)$ from (12.5.5). Undetermined factors $B_{k}^{(\alpha\beta)}$ will be selected irom the condition of turning into zero after reduction to common denominators of all numerators of elements of matrix $C(i\omega)\{\mathbf{I} + \mathbf{KS}(\omega)\} - \mathbf{S}(\omega)$ at points $\omega = \gamma_{i}$ and $\omega = b_{i}$, where b_{i} — pole of elements $S(\omega)$ in the upper half-plane.

Formulas (12.5.3) - (12.5.7) are applicable also to the general case of parameters with stationary increments of the k-th order if we introduce their spectral densities according to [59] and during factoring expand factors a^{2k} in denominators of elements and the determinant of $I + KS(\omega)$ as $\omega^{2k} = (i\omega)^k (-i\omega)^k$. It sometimes is more convenient to consider processes with stationary increments as limits from certain stationary processes with the same orders of denominators of the spectral matrix and to pass to the limit after obtaining the solution of (12.5.4) for auxiliary processes.

As a first example we shall consider a joint meter of l parameters in the form of stationary processes of very simple form with complete correlation, where $c^{(\alpha\beta)}(\omega) = \sigma_{\alpha}\sigma_{\beta}[1 + (\omega T)^{2}]^{-1}$. Seeking our solution in the form $C(i\omega) = C(i\omega)||\sigma_{\alpha}\sigma_{\beta}||$, it is easy to obtain.

$$\mathbf{C}(i\boldsymbol{\omega}) = \frac{2T ||\sigma_{\mathbf{a}}\sigma_{\mathbf{p}}||}{(1+a)(a+i\boldsymbol{\omega}T)}, \quad \boldsymbol{\Sigma}_{\mathbf{p}\mathbf{h}\mathbf{x}} = \frac{2 ||\sigma_{\mathbf{a}}\sigma_{\mathbf{p}}||}{1+a},$$

$$\mathbf{G}(i\boldsymbol{\omega}) = \frac{2T ||\sigma_{\mathbf{a}}\sigma_{\mathbf{p}}||}{(1+a)(1+i\boldsymbol{\omega}T)}, \quad \boldsymbol{a} = \sqrt{1+2T \sum_{\mathbf{a},\mathbf{p}=1}^{i} K^{(\mathbf{a}\mathbf{p})}\sigma_{\mathbf{a}}\sigma_{\mathbf{p}}}.$$

$$(12.5.8)$$

Smoothing circuits (Fig. 12.12) consist here of l inertialess input amplifiers, an adder, a common first-order link (kC-circuit) with time constant T, met in the

-712-



Fig. 12.12. Smoothing circuits for stationary parameters with total correlation; 1, 4) inertialess amplifiers; 2) adder; 3) inertial link of the first order. spectral matrix, and l inertialess output amplifiers. As compared to separate measurement of the α -th parameter in the shown circuit we can obtain gain in variance of a factor

$$\left(1+2T\sum_{\mathbf{a},\,\beta=1}^{l}K^{(\mathbf{a}\,\beta)}\sigma_{\mathbf{a}}\sigma_{\beta}\right)^{1/2}\times\left(1+2TK^{(\mathbf{a}\mathbf{z})}\sigma_{\mathbf{a}}^{2}\right)^{-1/2}$$

considerable when

> $K^{(aa)} \sigma_a^* > 1$, i.e., when the α -th parameter has smaller variance, and

its coding in the signal is more strongly suppressed by interferences than for other parameters.

A second example pertains to the analogous case of 1 parameters in the form of Wiener processes with total correlation, when $S^{(\alpha\beta)}(\omega) = \sqrt{B_{\alpha}B_{\beta}}/\omega^2$. The same method we obtain

$$\mathbf{C}(i\boldsymbol{\omega}) = \frac{\| \vec{V} \overline{B_{\alpha}} \overline{B_{\beta}} \|}{\sum_{\mathbf{1}, \mathbf{i} = 1}^{l} \mathcal{K}^{(\mathbf{1}\mathbf{i})} \ \mathbf{V} \overline{B_{\mathbf{1}} \overline{B_{\mathbf{i}}}} + i\boldsymbol{\omega}}, \ \mathbf{G}(i\boldsymbol{\omega}) = \frac{\| \vec{V} \overline{B_{\alpha}} \overline{B_{\beta}} \|}{i\boldsymbol{\omega}},$$

$$\mathbf{\Sigma}_{a\mathbf{M}\mathbf{X}} = \frac{\| \vec{V} \overline{B_{\alpha}} \overline{B_{\beta}} \|}{\sum_{\mathbf{1}, \mathbf{i} = 1}^{l} \mathcal{K}^{(\mathbf{1}\mathbf{i})} \ \mathbf{V} \overline{B_{\mathbf{1}} \overline{B_{\mathbf{i}}}}}.$$
(12.5.9)

Change of structure of the smoothing circuits as compared to Fig. 12.12 consists only in replacement of the common RC-circuit by an ideal integrator. Great gain in accuracy as compared to separate measurement of the α -th parameter is observed when

$$\sum_{\alpha, \beta=1}^{I} \mathcal{K}^{(\alpha\beta)} \sqrt{B_{\alpha}B_{\beta}} \gg \mathcal{K}^{(\alpha\alpha)}B_{\alpha}.$$

A third example pertains to simultaneous measurement of a parameter in the form of a Wiener process and the integral of it, when the correlation matrix has the form

$$\mathbf{S}(\boldsymbol{\omega}) := \left\| \begin{array}{c} B/\boldsymbol{\omega}^{*} & iB/\boldsymbol{\omega}^{*} \\ \cdots & iB/\boldsymbol{\omega}^{*} & B/\boldsymbol{\omega}^{*} \end{array} \right\|. \qquad (12.9.19)$$

By the method of undetermined coefficients with diagonalness of matrix \mathbf{K} we can obtain



where $g_1 = BK_1$, $g_2 = (BK_2)^{1/2}$.

According to (12.5.13) smoothing circuits of the single-loop variant of the meter have the form of Fig. 12.13. The estimate of the Wiener process will be



Fig. 12.13. Smoothing circuits for meter of position and speed: 1) inertialess amplifiers; 2) integrators; 3, 5) adders; 4) integrators with correction.



Fig. 12.14. Simplified smoothing circuits: 1) inertialess amplifier; 2) integrator; 3) integrator with correction.

formed by integration (with various amplification) of both output voltages of the discriminator and addition of the results; obtaining of the estimate of the integral of the first parameter anticipates transmission of output voltages of integrators through additional integrators with correcting RC-circuits with time constants $(z_1 + 2z_2)^{-1/2}$ and $(z_1 + 2z_2)^{1/2} z_2^{-1}$ and again addition of reputs, Reduction of smoothing circuits for cases of suppression by noises of the coding of one of the parameters ($K_1 = 0$, $K_2 \neq 0$ or $K_1 \neq 0$, $K_{\phi} = 0$) is presented in Fig. 12.14. Nowev only when $K_1 = 0$, $K_p \neq 0$ do all errors of measurement remain finite; in the opposize

case $(K_1 \neq 0, K_2 = 0)$ finite only is error of measurement of the Wiener process, while error of determination of the integral of it grows infinitely in time. This corresponds to a phenomenon known in technology, consisting in the fact that integration of speed for determination of a coordinate leads to gradual uncomponent, increase of error.

In the general case $(K_1 \neq 0, K_0 \neq 0)$ the gain as compared to the case of separate closing of smoothing circuits is given by formulan

$$\frac{(e_1^2)_{paan}}{(e_1^2)_{paan}} = \frac{1 + \frac{g_1}{g_1}}{\sqrt{1 + 2\frac{g_1}{g_1}}}, \frac{(e_2^2)_{paan}}{(e_2^2)_{coan}} = \frac{1 + \frac{g_1}{g_1}}{\sqrt{1 + \frac{g_1}{g_1}}}.$$
 (12.5.14)



Depending upon ratio g_2/g_1 it is observed for the first or second parameters, and when max $\left\{\frac{g_1}{g_2}, \frac{g_2}{g_1}\right\} > 1$ it is very marked (Fig. 12.15).

We shall also consider parameters in the form of correlated Wiener processes with spectral matrix

$$\mathbf{S}(\mathbf{w}) = \left| \begin{array}{c} \frac{B_1}{\mathbf{w}^2} & \mathbf{x} \frac{\mathbf{y} \overline{B_1 B_2}}{\mathbf{w}^2} \\ \mathbf{x} \frac{\mathbf{y} \overline{B_1 B_2}}{\mathbf{w}^2} & \frac{B_1}{\mathbf{w}^2} \end{array} \right| \quad (12.5.15)$$

with arbitrary interconnection of codings

$$\mathbf{K} = \left\| \begin{array}{cc} K_1 & \mu & \sqrt{K_1 K_2} \\ \mu & \sqrt{K_1 K_2} & K_2 \end{array} \right\|.$$

Fig. 12.15. Gain in variance during joint measurement.

Coefficients of matrices $\mathbf{C}(i\omega)$, $\mathbf{\Sigma}_{\mathbf{BMX}}$ and $\mathbf{G}(i\omega)$ we find by the method of undetermined coefficients:

$$C(i \circ) = \frac{1}{(a + i \circ)(b + i \circ)} \times$$

$$\times \left\| B_{i} \left[\sqrt{\frac{B_{i}}{B_{i}}} \beta + \frac{(1 + \sqrt{\frac{B_{i}}{B_{i}}})(\omega)}{a + b} \right] \sqrt{B_{i}B_{i}} \left[-\mu\beta + \frac{\pi - \mu\beta}{a + b}(\omega) \right] \right\|$$

$$V\overline{B_{i}B_{i}} \left[-\mu\beta + \frac{\pi - \mu\beta}{a + b}(\omega) \right] B_{i} \left[\sqrt{\frac{B_{i}}{B_{i}}} \beta + \frac{(1 + \sqrt{\frac{B_{i}}{B_{i}}})(\omega)}{a + b} \right] \right\|$$

$$(12.5.1i)$$

$$\left\| B_{i} \left(1 + \sqrt{\frac{B_{i}}{B_{i}}} \beta \right) - \sqrt{B_{i}B_{i}} \left(\pi - \mu\beta \right) \right\|$$

$$\Sigma_{BMS} = \frac{1}{a+b} \begin{vmatrix} B_s (1+\sqrt{\frac{a}{g_1}}\beta) & V B_1 B_s (x-\mu\beta) \\ V \overline{B_1 B_s} (x-\mu\beta) & B_s (1+\sqrt{\frac{g}{g_1}}\beta) \end{vmatrix}, \qquad (12...17)$$

$$\mathbf{G}(l \mathbf{\omega}) = \frac{l}{(a+b)^{l}\mathbf{\omega}} \left\| \begin{array}{c} B_{i}\left(1 + \sqrt{\frac{g_{i}}{g_{i}}}\beta\right) \sqrt{B_{i}B_{s}}(\mathbf{x} - \mu\beta) \\ \sqrt{B_{i}B_{s}}(\mathbf{x} - \mu\beta) & B_{s}\left(1 + \sqrt{\frac{g_{i}}{g_{s}}}\beta\right) \end{array} \right\|, \qquad (12.11)$$



Fig. 12.16. Smoothing circuits for two correlated Wiener processes: 1, 4) inertialess amplifiers; 2, 5) adders; 3) integrators.

where $\beta = \sqrt{\frac{1-x^2}{1-\mu^2}}$; $\xi_1 = K_1 B_1$ (1 = 1, 2), and $-a^2$, $-b^2$ are roots of equation

$x^{s} + (g_{1} + g_{s} + 2\mu\varkappa\sqrt{g_{1}g_{s}})x + g_{1}g_{2}(1 - \mu^{s})(1 - \varkappa^{s}) = 0.$

Smoothing circuits of the single-loop variant according to (12.5.18) consist of two integrals [sic], to each of which there are fed through inertialess amplifiers



Fig. 12.17. Gain in mean square error during joint measurement.

and z⁽²⁾(t) (Fig. 12.16). According to (12.5.17) the gain provided b, allowance for interconnection of parameters and their codings with respect to mean

output voltages of the discriminator $z^{(1)}(t)$

square error of measurement of the first parameter comprises

 $\frac{(\sigma_{1})_{phan}}{(\sigma_{1})_{conm}} = \left[1 + \frac{v^{2}}{4\mu^{3}} + vx + \frac{v}{\mu}(1 - \mu^{2})(1 - x^{2})\right]^{1/4} \times$ $\times \left[1 + \frac{v}{2\mu} \sqrt{\frac{1-x^2}{1-\mu^2}}\right]^{-1/2}$

where it is considered that $g_2/g_1 = \nu^2/4\mu^2$ ($\nu^2 = B_2/B_1$).

If the influence of the interconnection of coding here is nonessential when $\mu < 0.1$, strong correlation of parameters (n > 0.5) ensures noticeable gains (Fig. 12.17),

especially with great strength of the second parameter ($\nu > 1$). This also proves the advantage of joint measurement.

-716-

12.5.2. Parameters - Linear Combinations of Known Functions with Random Factors

In a number of cases, especially during measurement of coordinates of bodies travelling by determined laws, presentation of measured parameters in the form of quasiregular (degenerate) random processes is permissible:

$$\mathbf{\lambda}^{(a)}(t) = \sum_{k=1}^{q_a} \mu_k^{(a)} f_k^{(a)}(t) + \overline{\lambda^{(a)}(t)}, \qquad (12.5.19)$$

where $f_k^{(\alpha)}(t)$, $\overline{\lambda^{(\alpha)}(t)}$ - known functions;

 $\mu_{k}^{(\alpha)} - \text{random normally distributed quantities } (\mu^{(\alpha)} = 0, \ \overline{\mu_{k}^{(\alpha)}\mu_{j}^{(\beta)}} = M_{k,j}^{(\alpha\beta)}, \ \alpha, \ \beta = 1, \ \dots, \ l; \ k = 1, \ \dots, \ q_{\alpha}; \ j = 1, \ \dots, \ q_{\beta}).$

The correlation matrix for such parameters in general can be expressed in

the form

$$\mathbf{R}(t, \mathbf{r}) = \mathbf{F}^+(t) \mathbf{M} \mathbf{F}(\mathbf{r}) \tag{12.5.20}$$

where $\mathbf{F}(t)$ - complex nearly diagonal matrix with elements in the form of columns $\mathbf{F}^{(\alpha\beta)}(t) = \mathbf{F}^{(\alpha)}\delta_{\alpha\beta}, \ \mathbf{F}^{(\alpha)}(t) = \{\mathbf{f}_1^{(\alpha)}(t), \ldots, \mathbf{f}_{\alpha_n}^{(\alpha)}(t)\},\$

M - complex matrix
$$\left(\sum_{a=1}^{l} q_a \times \sum_{a=1}^{l} q_a\right)$$
 with submatrices $\mathbf{M}^{(\alpha\beta)} = ||\mathbf{M}_{kj}^{(\alpha\beta)}||$.

Then from (12.3.19), (12.3.20) and (12.5.20) we can obtain

$$c(t, \tau) = F^{+}(t) [M^{-1} + U(t)]^{-1} F(\tau),$$

$$g'(t, \tau) = F^{+}(t) [M^{-1} + U(\tau)]^{-1} F(\tau),$$

$$\Sigma_{\text{sur}}(t) = F^{+}(t) [M^{-1} + U(t)]^{-1} F(t),$$

(12.5.21)

where $\mathbf{J}(t) = \int_{t_0}^{t} \mathbf{F}(s) \mathbf{K} \mathbf{F}^+(s) ds - block matrix \left(\sum_{a=1}^{l} q_a \times \sum_{a=1}^{l} q_a \right).$

If between coefficients $\mu_k^{(\alpha)}$, pertaining to different parameters, there exists complete correlational coupling, matrix **M** turns out to be singular, **M**⁻¹ does not exist, and to find $\mathbf{c}(t, \tau)$, $\boldsymbol{\Sigma}_{BHX}(t)$ and $\mathbf{g}(t, \tau)$ another method is more convenient. It is based on presentation of the column vector of parameters in the form

$$\mathbf{L}(t) = \mathbf{\Phi}(t) \mathbf{\mu} + \overline{\lambda}(t), \qquad (12.5.52)$$

where $\Phi(t)$ -- matrix $(l \times q)$;

 $q - \text{total number of coefficients } \mu, \text{ through which all parameters are expressed.}$

This leads to the correlation matrix of parameters

$$\mathbf{R}(t, \mathbf{r}) = \mathbf{\Phi}(t) \mathbf{M}_{0} \mathbf{\Phi}^{+}(\mathbf{r}), \ \mathbf{M}_{0} = (\overline{\mu} \mu^{+}), \qquad (12.5)$$



Fig. 12.18. Smoothing circuits for quasiregular processes depending on one coefficient: 1, 4) variable-gain amplifiers; 2) adder; 3) integrators; 5) adders.

and the solution of the problem is sought by a method analogous to the presented one, and gives instead of (12.5.21):

$$c(t, \tau) = \Phi(t) [M_0^{-1} + V(t)]^{-1} \Phi^+(\tau),$$

$$t(t, \tau) = \Phi(t) [M_0^{-1} + V(\tau)]^{-1} \Phi^+(\tau),$$

$$\Sigma_{BMX}(t) = \Phi(t) [M_0^{-1} + V(t)]^{-1} \Phi^+(t),$$

$$V(t) = \int_{0}^{t} \Phi^+(s) K \Phi(s) ds.$$

(12.5.24)

Let us consider the particular case when all $q_{\alpha} = 1$, and coefficients $\mu^{(\alpha)}$ are equal to one another $(\mu^{(\alpha)} = \mu, \mu^2 = \sigma^2)$. Using (12.5.24), it is easy to obtain

$$g^{(a\beta)}(l,\tau) = \frac{e^{sf^{(\alpha)}}(l)f^{(\beta)}(\tau)}{1 + e^{s}\sum_{T,\delta=1}^{l} K^{(T\delta)} \int_{0}^{l} f^{(T)}(s)f^{(\delta)}(s) ds}$$

$$\Sigma^{(a\beta)}(l) = \frac{e^{sf^{(\alpha)}}(l)f^{(\beta)}(t)}{1 + e^{s}\sum_{T,\delta=1}^{l} K^{(T\delta)} \int_{0}^{l} f^{(T)}(s)f^{(\delta)}(s) ds} \qquad (12.12.25)$$

Smoothing circuits (Fig. 12.18) consist of i multipliers of output voltages of the discriminator by function $\sigma^{a}f^{(a)}(s)\left[1+\sum_{\tau,\ b=1}^{t}K^{(\tau b)}\sigma^{a}\int_{0}^{t}f^{(\tau)}(s)f^{(b)}(s)ds\right]^{-1}$, an adder and an integrator. At the output of the integrator there will be formed the estimate of the unknown coefficient μ . After multiplication of this estimate by known laws of variation of parameters $f^{(\alpha)}(t)$ and addition with mean values there are formed

estimates $\hat{\lambda}^{(\alpha)}(t)$. According to (12.5.25) errors of measurement of all parameters approach zero with increase of the time of observation t - t_0 . In the general case

smoothing circuits contain $\sum_{i=1}^{r} q_i$ input and as many output multipliers. After multiplications of input voltages by known functions there is produced summation in groups equal to the number of coefficients $\mu_1^{(\alpha)}$, from the formed voltages the integrators form estimates of these coefficients, and from the estimates by formulas (12.5.19) and (12.5.22) there are formed the measured parameters.

§ 12.6. Examples of Multi-Dimensional Meters

12.6.1. Measurement of Range and Speed by Coherent Radar

Let us consider joint measurement of range d(t) and speed d'(t) by a coherent radar, working on carrier frequency $f_0 = \omega_0/2\pi$ (wavelength $\lambda = c/f_0$) with square width of the spectrum of modulation

$$\Delta f_{HB} = \frac{\Delta \omega_{HB}}{2\pi}, \quad \Delta \omega_{HA}^2 = \frac{1}{2\pi} \int_{-\infty}^{+\infty} \omega^2 |U(i\omega)|^2 d\omega. \qquad (12.6.1)$$

Signar-to-noise ratio h and the width of the spectrum of fluctuations $\Delta f_{\rm g}$ are determined according to Chapters VII, IX and X. Filoted targets maneuver, where radial speed can be described by a Wiener process with parameter B, expressed in m^2/\sec^3 . Range, naturally, is the integral of the Wiener process (second integral of white noise) with the same parameter B. The influence of angular coordinates of the target on measurement of d(t) and d¹(t) we shall disregard.

According to results of Chapters VII and IX during approximation of the spectrum of fluctuations of the signal by function

$$S_{0}(\omega) = [1 + (\omega/2\Delta f_{0})^{2}]^{-1} \qquad (12.(.2))$$

we have the following formulas for characteristics of optimum discriminators of a speed meter and a range finder;

$$K_{a} = \frac{1}{2\Delta f_{a}} \frac{\hbar^{a}}{\sqrt{1+h} (1+\sqrt{1+h})^{a}},$$

$$K_{a} = 2\Delta \omega_{aa}^{2} \Delta f_{a} \frac{\hbar^{a}}{\sqrt{1+h} (1+\sqrt{1+h})},$$
 (12.0.3)

where subscripts w and τ indicate that parameters of the signal are considered the (vircu ar) carrier frequency ω and time delay τ . Approximation (12.6.2) gives a minimum value of $K_{_{\rm CD}}$ for large h (with fixed band of $_{_{\rm CD}}$); with respect to $K_{_{\rm T}}$ the form of the approximation of $S_{_{\rm O}}(\omega)$ makes practically no difference.

As it was shown above, in multi-dimensional meters it is necessary to characterize the quality of a joint discriminator also by the mixed derivative of

-719-

the logarithm of the likelihood function with respect to d and d'. This derivative for a coherent signal, as one can prove by formula (12.4.20), is close to zero if many periods of reputition are correlated and the signal-to-noise ratio is not too great. Therefore, matrix **K** is completely determined by its diagonal elements (12.6.3), and the circuit of the optimum joint discriminator of range and speed consists of circuits of discriminators of parameters τ and ω simultaneously tuned to $\hat{\tau}(t)$ and $\hat{\omega}(t)$ (Fig. 12.19).

Henceforth it is convenient to consider as parameters d and d' directly. Inasmuch as

$$\mathbf{v} = \frac{2d}{c}, \quad \mathbf{\omega} = 2\mathbf{\omega}, \frac{d'}{c}, \quad (12.6.4)$$

elements of (diagonal) matrix K in new parameters will have the form

$$K_{d} = K_{\bullet} \left(\frac{2\omega_{\bullet}}{c}\right)^{2} = \left(\frac{2\omega_{\bullet}}{c}\right)^{2} \frac{1}{2\Delta f_{\bullet}} \frac{h^{2}}{\sqrt{1+h}\left(1+\sqrt{1+h}\right)^{2}},$$

$$K_{d} = K_{\bullet} \left(\frac{2}{c}\right)^{2} = \left(\frac{2}{c}\right)^{2} 2\Delta \omega_{\kappa_{\bullet}}^{2} \Delta f_{\circ} \frac{h^{2}}{\sqrt{1+h}\left(1+\sqrt{1+h}\right)}.$$
(12.0.5)

Let us turn to smoothing circuits and resultant accuracy of joint measurement. Inacimuch as the matrix of spectral densities of d(t) and d'(t) has the form



Fig. 12.19. Joint discriminator of range and speed for a coherent signal: 1) multipliers (mixers and gated amplifiers); 2) unit of formation of reference signals; 3) i-f amplifiers with detuning; 4) i-f amplifiers; 5) square-law detectors; 6) subtractor; 7) phase detector.

(12.5.10), and matrix **K** is diagonal, it is possible to directly use the solution of Paragraph 12.5.1, substituting in $(12.5.11) - (12.5.13) \in = g_1$ and $g_d = g_2 - \text{coefficients of the}$ dimensionality of the square of frequency. Due to the correlatedness of parameters smoothing circuits of the joint meter (see Fig. 12.13) are tightly entwined. The measured value of speed of a piloted object will be formed by

integration (with various weights) of both output voltages of the joint discriminator and addition of results. This processing is hardly obvious, but it leads to minimum mean square error of measurement of speed.

The measured value of range will be formed by separate transmission of output voltages of the two mentioned integrators through additional integrators, equipped

-720-

with different correcting RC-circuits, and addition of results.

The influence of one or another of the output voltages of the joint discriminator is determined by the relationship between coefficients g_d and g_d . If $g_d > g_d$, the basic role is played by the speed discriminator, and, conversely, when $g_d > g_d$, the basic role is played by the range discriminator.

In particular cases when $g_d \ll g_{,}$, or $g_{,} \ll g_d$, smoothing circuits could have been simplified (Fig. 12.14). However, with inaccurate measurements of the range finder such simplification is technically and theoretically unjustified, inasmuch as resultant error for range grows without limit in time. A multidimensional meter, losing selection for range, also loses it for speed, as a result of which all measurement ceases. In other words, inaccurate measurement of range is better than failure of this measurement.

Otherwise, with very rough primary measurements of speed (wide spectrum of fluctuations, large wavelength, and so forth) tracking by both coordinates can be performed from readings of the range finder with finite quantities of errors of selection, if the signal-to-noise ratio h is sufficiently great.

Variances of errors for speed and range in steady-state operating conditions according to (12.5.12) are equal to

$$\sigma_{d'}^{2} = \sqrt{\frac{B}{K_{d'}}} \frac{\sqrt[4]{1+2/x}}{1+1/x},$$

$$\sigma_{d}^{2} = \sqrt[4]{2} \left(\frac{B}{K_{d}^{2}}\right)^{1/4} \frac{\sqrt[4]{1+x/2}}{1+x},$$
 (1.2.6.6)

where $n = \sqrt{g_{1}/g_{d}}$.

Comparison of (12.6.6) with results of Chapter VI shows that as compared to separate closing of loops of range and speed tracking there is observed gain in mean square error of speed and range by factors of $(1 + 2/n)^{1/4}(1 + 1/n)^{1/2}$ and $(1 + n/2)^{1/4}(1 + n)^{-1/2}$, respectively. To find conditions of increase of accuracy for d or d' we investigate quantity n as a function of tactical and technical presenters of the radar. Using (12.6.5) it is easy to prove that

$$\mathbf{x} = \frac{\pi}{\sqrt{2}} \frac{\Delta f_{AOB}}{\Delta f_{0}} \frac{f_{0}}{\Delta f_{NB}} \Psi(h), \qquad (12.6.7)$$

$$\Delta f_{AOR} = \sqrt{\frac{4B}{\lambda^{2} \Delta f_{0}}}$$

where

is root mean square magnitude of drift of Doppler frequency due to maneuver of the target during the interval of correlation of fluctuations of the signal;

$\varphi(h) = \frac{h}{(1 + h)^{1/4} (1 + V + h)^{5/2}}$

is a function of the signal-to-noise ratio h, asymptotically changing with growth and decrease of h, as



and reaching a maximum equal to 0.146 when $h \approx 6.2$ (Fig. 12.20). The influence of measurements of speed is seen to the maximum degree for mean values of the signal-to-noise (h = 0.1 to 10^3), which usually are operating values. This influence is stronger, the faster the target is piloted, the more it maneuvers, the scorter the effective wavelength λ and the narrower the width of the spectra of fluctuations $\Delta f_{\rm KB}$. Concrete values of $\pi/\phi(h)$ we shall estimate for two examples.

Thus, when $\lambda = 3 \text{ cm} (f_0 = 10^{10} \text{ cps})$, $\Delta f_c = 30 \text{ cps}$, $B = 30 \text{ m}^2/\text{sec}^3$, $\Delta f_{\text{KB}} = 10^6 \text{ cps}$ we have $\pi/\phi(h) = 4.9 \cdot 10^6$.

In the other case, when $\lambda = 10$ cm, $\Delta f_{c} = 40$ cpc, B = 10 m²/sec³, $\Delta f_{KB} = -(\cdot 10^{6} \text{ cps}, \pi/\phi(h) \approx 2.2 \cdot 10^{3}$.



Fig. 12.20. Dependence of $\varphi(h)$ on the signal-to-noise ratio.

For radars of various types and in various conditions of application $\varkappa/\psi(h)$ varies from 10^3 and 10^5 . Therefore, with h = (1-50), when $\psi(h) \ge 0.1$ coefficient

n is not less than $10^2 - 10^4$, which gives a gain in mean square error σ_d of measurement of range of $(2\pi)^{1/4} = (4-12)$ times. In these conditions instead of (12.6.6) one should use formulas

$$\sigma_{d'}^{2} = \sqrt{\frac{B}{K_{d'}}} = \frac{\lambda}{2} \sqrt{\frac{B\Delta f_{c}}{\pi} \frac{(1+h)^{1/4} (1+\sqrt{1+h})^{3/2}}{h}},$$

$$\sigma_{d}^{2} = \frac{1}{\sqrt{K_{d}K_{d'}}} = \frac{\lambda c}{(4\pi)^{2}} \frac{1}{\Delta f_{HH}} \frac{(1+h)^{1/2} (1+\sqrt{1+h})^{2}}{h^{2}}.$$
 (12.6.5)

When h >> 1, as long as condition π >> 1 is preserved, these approximate formulas are valid

$$\sigma_{d'}^{2} = \frac{\lambda}{2} \sqrt{\frac{B\Delta f_{o}}{\pi}},$$

$$\sigma_{d}^{2} = \frac{\lambda c}{(4\pi)^{2}} \frac{1}{\Delta f_{\pi\pi}} \frac{1}{\sqrt{h}}.$$
(12.0.)

Ġ.,

Faradoxically, error for range does not depend in the formulated conditions on average maneuverability, determined by parameter B. The fact is that with growth of maneuverability (i.e., of the magnitude of bursts of speed) the speed discriminator, basically determining accuracy of measurement, works ever more efficiently. We also notice that change of the width of the spectrum of fluctuations $\Delta f_{\rm c}$ leads to growth of noises at the output of the speed discriminator and decrease of them at the output of the range discriminator, so that in a joint meter there occurs compensation of two factors, and $\sigma_{\rm g}$ turns out to depend on $\Delta f_{\rm c}$ only implicitly, through magnitude h. Considering for example $\lambda = 3$ cm, $\Delta f_{\rm KB} = 10^6$ cps, h = 100, we have a minute noise component of error for range

$\sigma_{\rm m} \approx 0.08 \ M$

which is hundredths of the interval of range resolution. If, however, we seek the same quality of measurement of range with separate construction of the range finder and speed meter, it is necessary to expand by many times the spectrum of reputar modulation.

Thus, during joint measurement of range and speed of a maneuvering target by a constent right it turns out to be possible to sharply increase accuracy of measurement of range due to the high accuracy of primary measurements of speca.

12.6.7 Measurement of Hange and Speed by Incoherent Radar*

ist as consider now measurement of the same parameters of motion d'(t) and d(t)by an inconcrete single-channel radar. We keep the former designations for

"The price of results of largered 12,6,2 was obtained by C. K. Smirnov.

-- 723--

wavelength, carrier frequency and mean square width of the spectrum of modulation. The pulse signal of the radar we shall consider fluctuating independently of the puriod to a period of duration T_r . In distinction from Paragraph 12.6.1 we shall consider a nonmaneuvering target, flying at the radar with constant, but unknown speed and undetermined position at the initial moment. Influence of measurement of angular coordinates, as also in Paragraph 12.6.1, we shall disregard.

In examining properties of an optimum joint discriminator in accordance with the character of the signal it is permissible to consider one period of repetition. By formula (12.4.29) in conditions of a symmetric form of the sending we have matrix K (with parameters ω , τ) in the form

$$\mathbf{K}_{\mathbf{s}\mathbf{t}} = \frac{2q^2}{T_r \left(1+q\right)} \left\| \begin{array}{c} \mathbf{\tau}_{\mathbf{k}\mathbf{s}}^2 & k \mathbf{\tau}_{\mathbf{k}\mathbf{B}} \boldsymbol{\omega}_{\mathbf{k}\mathbf{B}} \\ k \mathbf{\tau}_{\mathbf{k}\mathbf{B}} \boldsymbol{\omega}_{\mathbf{k}\mathbf{B}} & \boldsymbol{\omega}_{\mathbf{k}\mathbf{B}}^2 \end{array} \right\|,$$

where $u(t) = u_a(t)e^{i\psi(t)}$ - regular modulation; q - signal-to-noise ratio;

$$\tau_{HB} = \sqrt{\frac{1}{T_r} \int_{-\infty}^{+\infty} |u(t)|^2 t^2 dt}$$

- mean square duration of the sending;

$$k = -\frac{1}{R_{t}} \int_{-\infty}^{+\infty} u'(t) u^{*}(t) i dt \qquad (12.6.11)$$

(12.6.10)

- dimensionless coefficient of interconnection of codings of the two parameters of modulation.

It also is possible to express matrix \mathbf{K} very graphically through the derivative of the autocorrelation function of the pulse sending (see Chapter I),

$$C(\tau, \omega) = \frac{1}{T_{\tau}} \int_{-\infty}^{+\infty} u(t-\tau) u^{*}(t) e^{-i\omega t} dt \quad (C(0,0)=1), \qquad (12.0.12)$$

taken for zero arguments

$$\mathbf{K}_{\mathbf{w}\tau} = -\frac{2q^{2}}{T, (1+q)} \operatorname{Re} \left[\begin{array}{c} \frac{\partial^{2}C(0,0)}{\partial \omega^{2}} & \frac{\partial^{2}C(0,0)}{\partial \tau^{2} \partial \omega} \\ \frac{\partial^{2}C(0,0)}{\partial \omega \partial \tau} & \frac{\partial^{2}C(0,0)}{\partial \tau^{2}} \end{array} \right]. \tag{12.6.15}$$

In other words, with an accuracy of a coefficient monotonically depending on the signal-to-noise ratio q matrix $\mathbf{K}_{\omega\tau}$ consists of quantities characterizing

-724-

sharpness of the peak of the autocorrelation function in various sections. If lines of constant level $C(\tau, \omega)$ for small arguments are ellipses whose axes are oriented along axes ω , τ , the mixed derivative turns into zero, which bears witness to the absence of interconnection of codings of range and speed in the signal. Such a situation occurs, for instance, for simple pulse modulation, intrapulse triangular

frequency modulation and phase-code manipulation, when integral $\int_{-\infty}^{\infty} u'(t) u^{\bullet}(t) ildit turns$ out to be a pure imaginary quantity or zero. In the general case the interconnection of two parameters exists, which is easy to prove by the example of linear intrapulse frequency modulation (FM)

$$u(l) = u_{a}(l)e^{\frac{lVl}{2}}, \qquad (12.6.14)$$

where V - rate of rise (decrease) of the frequency of the filling.

Here coefficient k is proportional to the steepness of FM

$$k = V \frac{\varsigma_{xx}}{\omega_{xx}}, \qquad (12.6.15)$$

We shall study resultant errors for d' and d for fixed duration of the sending $\tau_{\rm KB}$ and width of its spectrum $\omega_{\rm KB}$. Transition in matrix $K_{\rm cot}$ to parameters d' and d, in accordance with (12.6.4) and (12.6.10), gives

$$\mathbf{K}_{d'd} = \begin{vmatrix} K_d, & k \sqrt{K_d, K_d} \\ k \sqrt{K_d, K_d} & K_d \end{vmatrix}, \qquad (12.6.16)$$

where $K_d = \frac{2q^2}{1+q} \frac{\epsilon_{k0}^2}{T_r} \left(\frac{2\omega_0}{c}\right)^2$; $K_d = \frac{2q^2}{1+q} \frac{\omega_{k\eta}^2}{T_r} \left(\frac{2}{c}\right)^2$, and the value of k is kept equal to (-2.0.11).

Let us turn to synthesis of smoothing circuits. Here it is convenient to use the method described by relationships (12.5.22)-(12.5.24). Actually,

where d'(t), d(t) are a priori mean values of range and radial speed, and d'_{ij} , d_{ij} are unknown deviations of the initial (i.e., moment t_{ij}) speed and position with zero mean values and the matrix of moments

$$\mathbf{M}_{\mathbf{0}} = \left| \begin{array}{c} \sigma_{1}^{2} \ \mathbf{0} \\ \mathbf{0} \ \sigma_{0}^{2} \end{array} \right|.$$

It is possible to present $\lambda(t) = \{d^{\dagger}(t), d(t)\}$ in the form (12.5.22), where

$$\Phi(t) = \left| \begin{array}{c} 1 & 0 \\ t - t_{\bullet} & 1 \end{array} \right|, \quad \mu = \left| \begin{array}{c} d_{\bullet} \\ d_{\bullet} \end{array} \right|,$$

and, directly using (12.5.24), to record the solution

$$\mathbf{g}(t,\tau) = \frac{1}{\det(\tau)} \begin{vmatrix} K_{d}(\tau-t_{0}) + \frac{1}{e_{0}^{2}} & K_{d}(t-t_{0}) + \frac{K_{d}(t-t_{0})}{2} - \frac{K_{d}(t-t_{0}) + \frac{\tau-t_{0}}{2}}{-k\sqrt{K_{d}K_{d}}(t-t_{0}) + \frac{K_{d}(t-t_{0}) + \frac{\tau-t_{0}}{2}}{-k\sqrt{K_{d}K_{d}}(t-t_{0}) + \frac{K_{d}(t-t_{0})^{2}}{2} - \frac{K_{d}(t-t_{0})^{2}}{2} - \frac{K_{d}(t-t_{0})^{2}}{2} - \frac{K_{d}(t-t_{0})(\tau-t_{0}) + \frac{1}{e_{1}^{2}}}{-k\sqrt{K_{d}K_{d'}}(t-t_{0})(\tau-t_{0}) + \frac{1}{e_{1}^{2}}} \\ \frac{K_{d}(\tau-t_{0}) + \frac{1}{e_{0}^{2}}}{K_{d}(\tau-t_{0}) + \frac{1}{e_{0}^{2}}} - \frac{K_{d}(\tau-t_{0})^{2}}{2} - \frac{K_{d}(\tau-t_{0})(\tau-t_{0}) + \frac{1}{e_{1}^{2}}}{2} \\ \frac{K_{d}(\tau-t_{0})(t-t_{0}) - \frac{K_{d}(\tau-t_{0})^{2}}{2} - \frac{K_{d}(\tau-t_{0})(\tau-t_{0})}{2} - \frac{K_{d}(\tau-t_{0})^{2}}{2} - \frac{K_{d}(\tau-t_{0})(\tau-t_{0})}{2} - \frac{K_{d}(\tau-t_{0})^{2}}{2} - \frac{K_{d}(\tau-t_{0})(\tau-t_{0}) + \frac{1}{e_{0}^{2}}}{-k\sqrt{K_{d}K_{d'}}(t-t_{0})(\tau-t_{0}) + \frac{1}{e_{0}^{2}}} \\ -k\sqrt{K_{d}K_{d'}}(\tau-t_{0}) + \frac{t-t_{0}}{2} - \frac{K_{d}(\tau-t_{0})(\tau-t_{0})}{2} - \frac{K_{d}(\tau-t_{0})(\tau-t_{0}) + \frac{1}{e_{0}^{2}}}{-k\sqrt{K_{d}K_{d'}}(t-t_{0})(\tau-t_{0}) + \frac{1}{e_{0}^{2}}} \\ -k\sqrt{K_{d}K_{d'}}(\tau-t_{0}) + \frac{(t-t_{0})(\tau-t_{0})}{e_{0}^{2}} - \frac{1}{e_{0}^{2}} - \frac{1}{$$

where

$$\det (t) = \frac{K_d^2 (t-t_0)^4}{12} + K_d, K_d (1-k^2) (t-t_0)^2 + \frac{K_d (t-t_0)}{\sigma_1^2} + \frac{1}{\sigma_0^2} \Big[K_1 (t-t_0) + k \sqrt{K_d K_{d'}} (t-t_0)^2 + \frac{K_d (t-t_0)^2}{3} \Big] + \frac{1}{\sigma_0^2 \sigma_1^2}.$$

The algorithm of joint processing of data from the output of the discriminator in the single-loop variant of meter is presented in Fig. 12.21, where the meaning of operations is intelligible in light of the descriptions of § 12.5.

From the matrix of resultant errors $\Sigma_{BHX}(t)$ we extract elements on the main diagonal, again obtained after transition to continuous time:

$$s_{a}^{2}(t) = \frac{4}{K_{a}(t-t_{a})} \times \frac{1-3kb(t)+3b^{2}(t)+3b^{2}_{0}(t)+3b^{2}_{1}(t)}{1+12b^{2}(t)(1-k^{2})+4(1+3kb(t)+3b^{2}(t))b^{2}_{0}(t)+12b^{2}_{1}(t)+12b^{2}_{0}(t)b^{2}_{1}(t)},$$

$$\sigma_{d'}^{2}(t) = \frac{12}{K_{d}(t-t_{0})^{2}} \times$$

$$\langle \frac{1+b_{1}^{2}(t)}{1+12b^{2}(t)(1-k^{2})+4(1+3kb(t)+3b^{2}(t))b_{0}^{2}(t)+12b_{1}^{2}(t)+12b_{0}^{2}(t)b_{1}^{2}(t)}, \qquad (12.6.20)$$

where

>

$$b(t) = \sqrt{\frac{Ka'}{K_{4}} \frac{1}{(t-t_{0})}}; \quad b_{0}(t) = \frac{1}{\sigma_{0} \sqrt{K_{4}(t-t_{0})}};$$

$$b_{1}(t) = \frac{1}{\sigma_{1}(t-t_{0}) \sqrt{K_{4}(t-t_{0})}} \quad (12.6.21)$$

are coefficients depending on the time of observation and approaching zero with increase of it. The first of them b(t) is actually proportional to the ratio of unit (i.e., in a period) error of the range discriminator $\sqrt{T_r/K_d}$ to the advance of error in range during the time of observation t - t₀ due to unit error for speca $\sqrt{T_r/K_d}$ with independent codings of d'(t) and d(t) in the signal:

$$b(t) = \frac{\tau_{\rm H0}}{t - t_0} \cdot \frac{f_0}{f_{\rm KB}}.$$
 (12.6.22)



Fig. 12.21. Smoothing circuits for a meter of range and radial speed: 1) integrators; 2, 5, 6) variablegain amplifiers; 3) inertialess amplifiers; 4, 7) adders.

The second and third coefficients $b_0(t)$ and $b_1(t)$ in (12.6.21) are determined by accuracy of a priori data and may be negligible, in any case, with a sufficiently large time of observation. We shall investigate namely the last case when, approximately, we have

$$\sigma_{d}^{2}(l) \approx \frac{4}{K_{d}(l-l_{0})} \frac{1-3kb(l)+3b^{2}(l)}{1+12b^{2}(l)(1-k^{2})},$$

$$\sigma_{d'}^{2}(l) \approx \frac{12}{K_{d}(l-l_{0})^{2}} \frac{1}{1+12b^{2}(l)(1-k^{2})}.$$
(12.6.23)

As relationships (12.6.23) show, for resultant errors not indifferent arc the absolute value and sign of the coefficient k of interconnection of codings of parameters in the signal. Let us discuss the dependence of range error $s_d(+)$ on k. For the two extreme and the intermediate cases (k = -1; C; 1) we have:

$$\sigma_{d}^{2}(t)|_{k=1} = \frac{4}{K_{d}(t-t_{0})} \left[1 + 3b^{2}(t) + 3b(t)\right],$$

$$\sigma_{d}^{2}(t)|_{k=0} = \frac{4}{K_{d}(t-t_{0})} \left[1 + 3b^{2}(t) + 3b(t)\right],$$

$$(1 \ge 0, \ge 4)$$

$$\sigma_{d}^{2}(t)|_{k=0} = \frac{4}{K_{d}(t-t_{0})} \left[1 + 3b^{4}(t) - 3b(t)\right],$$

÷,

('

Ē,

i.e., with increase of k from -1 to +1 error decreases. The family of dependence of ratio $a = \frac{a^2}{a^2}$ on k for several values of b is shown in Fig. 12.22.



Fig. 12.22. The dependence of range error on the interconnection of coding of range and speed.

For typical values of width of the spectrum of modulation and duration of the sending (of the order of 1 Mc and 1 millisecond), wavelength 0.5-1 m and small times of observation (of the order of 10 coc) we have:

$$\frac{\Phi_d^2(l)|_{k=-1}}{\Phi_d^2(l)|_{k=1}} = 1,5 - 1,8.$$

Howev 7, with increase of the time of observation in any case error of range asymptotically seeks $4/K_d(t - t_0)$, i.e., is determined only by primary measurements of the range finder. In Fig. 12.23 is the family of curves $\sigma_d^2(t) = \frac{K_d(t - t_0)}{4}$. Here k is a fixed parameter, $K_d/K_d = 1 \sec^2$.

These results show that the dependence of error of measurement of range on coefficient k appears only with a comparatively small time of smoothing.



Fig. 12,23. Asymptotic behavior or range error.

On the pther hand, with therease of the time of observation accuracy of measuring speed is increased. Indeed, if we use only readings of the speed meter, we have

$$[\sigma_d^2,(t)]_{\text{pass}} \rightarrow \frac{1}{\mathcal{K}_{d'}(t-t_{\bullet})},$$

and with joint measurement we have, according to (12.6.23),

$$[\sigma_{d}^{2}, (t)]_{comm} \rightarrow \frac{12}{K_{4}(t-t_{0})^{4}},$$

by a repair of which

$$\frac{[a_{d},(l)]_{\text{para}}}{[a_{d},(l)]_{\text{surv}}} \approx \frac{1}{\sqrt{12}b(l)}$$
(12.0.12)

Thus, when $K_{ij}/K_{ij} = 1$ sectorbervation for 100 sec gives a rain in man square error of approximately a factor of (). This rain is explained by the fact that in a range finder for determination of speed there is the possibility of offsetively.

using the time base of the extended interval of measurement. The speed discriminator after expiration of the period of rough measurement of frequency due to noises ceases to feel the small remainder of the unmeasured difference of the Doppler frequency and practically does not participate in the process of further definitizing of data.

A similar effect of gradual decrease of the influence of measurements of speed is also observed with coherent radiation, for which the above mentioned results preserve their force, if in (12.6.5) as b(t) we substitute

$$b(t) = \frac{1}{2(\Delta f_c t)} \left(\frac{f_o}{\Delta f_{xx}}\right) \frac{1}{1 + \sqrt{1 + h}} (t_o = 0). \qquad (12.6.20)$$

With growth of h and t function b(t) decreases, although for average times of observation (Δf_{c} t = 2-30 and h \approx 100) i may reach 15-25. This, according to the middle formula of (12.6.23) gives a gain in mean square error of range of a factor of approximately 2, inasmuch as

$$o_4^2(t) \approx \frac{1}{K_4 t}$$
 (12.6.27)

Formula (12.6.27) is intelligible inasmuch as the range finder must definitize only the initial position of the target, and its speed in the first moments turns out to be known with sufficient accuracy.

Returning to the case of an incoherent pulse signal, we shall discuss in concwhat greater detail the case of deep linear frequency modulation $(|\mathbf{k}| \approx 1)$. Here matrix **K**, becomes singular, and elements of matrices \mathbf{K}^{-1} increase infinitely. d d The latter, at first sight, bears withous to huge unit errors (in recalculation of parameters to the discriminator input). In fact the case of deep FM does not give any practical inconveniences. It is possible not always to recalculate the output volvage of the discriminator of range and speed to the input, i.e., to reduce them to quantities of the dimensionality of d' and d in such a manner that in one quantity there is not contained range, and in other there is not contained speed. We encounter this case with deep FM. With increase of V outputs contained speed discriminator $\mathbf{z}_{\mathbf{i}}^{d}$ and $\mathbf{z}_{\mathbf{i}}^{d'}$ become all the more similar. The fact is that here speed is coded in frequency shift, and range is coded in frequency shift and delay of amplibude modulation of the sending:

$$\begin{split} \mu(t,\tau,\omega_{\rm R}) &= \mu_{\rm R}(t-\tau) \, {\rm e}^{-\frac{iV(t-\tau)^{\rm s}}{2}+i\omega_{\rm R}t} \\ &\equiv \mu_{\rm R}(t-\tau) \, {\rm e}^{\frac{iVt^{\rm s}}{2}-i(V\tau+\omega_{\rm R})t} \, {\rm e}^{\frac{iV\tau^{\rm s}}{2}}. \end{split}$$

With increase of the gain factor V the channel of the range discriminator, separating mismatch in amplitude modulation, takes an ever smaller weight, so that already with a width of the FM spectrum exceeding the width of the spectrum of the sending envelope by 5-10 times the additional channel can be disregarded. Then z_1^d and z_1^d turn out to be identical accurate to a proportionality factor ω_Q/V , and for further processing it is sufficient to take only one output quantity, as a result of which the smoothing circuits are simplified somewhat.

Transition to a smaller number of output voltages of the discriminator is also useful in all cases when matrix **K** turns out to be singular, inasmuch as singuliarity appear: due to the presence of linear interconnection between output voltages of the discriminator. The number of output quantities not accounted for is equal to the number of linear couplings superimposed on the measured quantities. Here there remain in force all the above mentioned formulas; it is necessary only to manifest caution during inversion of matrix **K**. Inasmuch as this inversion is nevertheless necessary during comparison of optimum circuits with real ones, it is necessary to pass to a smaller number of unknown parameters. In particular, in the considered example as $\lambda(t)$ it is possible to consider

 $\lambda(t) = d(t) + \frac{\Theta_0}{V} d'(t)$

「「「「「「「」」」」

国は国家、地方公司市政部務協会

 \mathbf{cr}

$$\lambda(t) = d'(t) + \frac{V}{\omega_{\bullet}} d(t),$$

which formally reduces the problem to a one-dimensional one.

From the practical side investigation of accuracy of measurement of range and speed in incoherent radars showed the necessity of allowance for interconnection of these parameters in trajectory and in coding in the common signal. During motion by a definite (linear) law the role of primary measurements of speed with respect : resultant accuracy decreases in time. Accuracy for range and speed is basically determined by the range finder, where in speed there is observed great gain as compared to the case of its independent measurement on the basis of the hepder effect. For small times of observation (10-20 sec) measurements of speed increase accuracy, where the concrete gain depends on the form of modulation. Modulation with a coefficient of interconnection of errors $k \rightarrow 41$ (for instance, linear FE with increase of frequency) has the best qualities; modulation with $k \rightarrow -1$ (linear EX with increase of frequency) has the worst qualities. All modulations with $k \neq 0$ (simple ends modulation, triangular FM and phase-code intrapulse modulation) p^2 .

-731-

intermediate results. Linear FM with variable slope gives approximately the same result.

§ 12.7. Conclusion

During the analysis of tracking multi-dimensional meters known in practice, as in the one-dimensional case, it is convenient to divide interconnected discriminators and smoothing circuits. Discrimination and fluctuation characteristics of multidimensional discriminators, which for 7 measured quantities are determined in the mumber 1 and l(l + 1)/2, respectively, turn out to append in general on all l parameters. With unconnected codings of parameters and absence of coupling in smoothing circuits during the analysis of accuracy fully permissible is reduction of the multi-dimensional problem to 1 one-dimensional problems, which is the basis for results of Chapters VII-XI. In the general case with known characteristics of the discriminator and an assigned scheme of smoothing circuits resultant errors are found by solving integral matrix equations. In the case of smoothing circuits with constant parameters and stationary measured parameters calculation of errors is comparatively simple and includes algebraic operations on matrices of equivalent spectral densities of the multi-dimensional discriminator Same of spectral densities of parameters $\bm{S}_\lambda(\omega)$, the matrix of gain factors of the discriminator $\bm{K}_{I\!\!I}$ and the transmission matrix of smoothing circuits.

As also in one-dimensional theory, the most suitable apparatus of synthesis of optimum multi-dimensional meters is the theory of statistical polutions. There is performed synthesis of an optimum action of any set of antitrarily connected peraseters with Gaussian and Markovian statistics with Gaussian approximation of the multidimensional likelihood function. There are given meter circuits equivalent in quality: single-loop and two-loop tracking circuits and a nontracking circuit.

Tracking variants, practically the most convenient, contain as basic elements multi-dimensional nonlinear units - discriminators and accuracy units - and a matrix (or two matrices) of smoothing circuits which for Gaussian statistics of parameters are linear. Output quantities of the discriminator characterise current mismatch for all parameters, and output quantities of the accuracy unit characterize current accuracy of separate measurements. With interconnected codings of parameters in signals there simultaneously appears dependence of separate output quantities of the discriminator directly on several mismatches, and interconnection of arrors of measurements of the shown parameters. The latter leads to the necessity of creating a special securacy unit characterizing the current measure of

-732-

interconnection of errors. In a number of cases, however, the unit of accuracy and control of smoothing circuits from the realization of the signal may be rejected, which leads to circuit simplifications.

Without concretization of the nature of the parameters it is possible to note a cortes of laws governing synthesis of discriminators for different statistical projectics of input signals and synthesis of smoothing circuits for different correlation properties of the parameters.

The given examples of synthesis of joint meters show large gains in accuracy of measurement of one or another of the measured interconnected quantities. In particular, during joint measurement of speed and range of a body with determined character of motion there is noted considerable gain in speed, growing in time; during measurement of the speed and range of a maneuvering target with a coherent signal there is the possibility of obtaining great gain in range.

However the theory developed above cannot in any measure be called exhaustive. In particular, analysis of multi-dimensional meters is conducted on the assumption of the absence of a number of important components of errors. Absolutely ignored are questions of breakoff of tracking in multi-dimensional circuits, although in principle the device of investigation of nonlinear random phenomena in similar systems can be offered (Fokker-Planck equations).

If we pass to questions of synthesis, then it is immediately necessary to stipulate that the theory given in Chapter XII with full success can be used during synthesis of meters of coordinates of single signals, and also signals against a background of interferences of arbitrary (including, similar) structure with known parameters.

If, however, we are interested in the problem of multipurpose radar (recolution and identification of similar signals in the course of measurement), the abovementioned device, strictly speaking, can be used only partially. As before indisputable is the necessity of application of the theory of statistical colutions (filtration); however, during synthesis it is necessary to consider the multipeak structure of the likelihood function. The latter appears in view of the similarity of signals, so that transposition of like parameters of them leads to a situation, "likely" almost to the same degree as the true one. Different peaks correspond to all possible transpositions of parameters. As the signals draw close, when there appears the problem of resolution, separate peaks merge, so that a Gaussian approximation of the likelihood function turns out to be inapplicable in principle.

-733-

From the point of view of broadening the theory developed above it is necessary to continue consideration of functionals of parameters (including, nonlinear), to construct other different forms of smoothing circuits for cases of interest for practice, and so forth. is weichigh the back of a line.

In conclusion we indicate that results of analysis and synthesis of multidimensional radar meters, as also results of Chapter VI, can be used with success in diverse fields of technology: optics, infrared and ultrasonic technology, communications, broadcasting, television; everywhere where they use signals carrying information.

CHAPTER XIII

RESOLVING POWER

§ 13.1. Introductory Remarks

In connection with the expansion of the area of use of radar and the increasing complexity of problems solved by it in recent years there has arisen interest in problems connected with simultaneous separate observation (resolution) of many targets. Inasmuch as development of a theory for this problem is far from complete, there naturally exists quite a lot of different approaches to the problem of resolution of targets, in part concerning quantitative appraisal of resolving power of various instruments, and also in part concerning formulation of the problem of synthesis of systems optimum with respect to resolving power.

As it is known, the concept of resolving power was first used by Rayleigh in the theory of optical instruments. There, by resolving power there was understood the ability of an instrument to a sufficient degree to separately reproduce at its exit images of observed sources of light. As the quantitative characteristic of resolving power of an instrument Rayleigh proposed using the minimum distance $\Delta_{\rm MMH}$ between identical point sources starting from which the total response of the instrument, considered as a function of a given coordinate, has two maxima. Obviously, $\Delta_{\rm MMH}$ coincides with the width of the instrument response, property defined.

The characteristic of resolving power of optical instruments introduced by Rayleigh found application in other areas, too, including radar. Initially this characteristic was used for appraisal of the performance of systems, principles of whose construction were selected, basically, empirically. After F. Woodward and J. Davis developed the theory of optimum reception of radar signals in noise, there were conducted more general investigations of the resolving power of such a receiver.

-735-

Intesmuch as in a receiver which is optimum with respect to noises there is produced multiplication of the received signal by the expected and integration, the intensity of the response to the useful signal is proportional to the square of the modulus of the autocorrelation function of modulation $|C(\tau, \Omega)|^2$ (function of uncertainty, considered in Chapter I, Vol. I): In accordance with this Rayleigh characteristics of resolving power of such a receiver with respect to delay and Doppler frequency coincide with the width of the principal maximum of the function of uncertainty with respect to τ and Ω , respectively.

Investigation of properties of the function of uncertainty (see Chapter I, Vol. I) showed that the width of the maximum on the τ -axis is inversely proportional to the width of the spectrum of modulation, and on the Ω -axis it is inversely proportional to the duration of the signal. From this there arose the common idea that range resolving power is determined by the width of the spectrum, and speed resolving power is determined by the duration of the sounding signal.

The Rayleign characteristic of resolving power is introduced very conditionally and has meaning only in reference to resolved signals identical in strength. During detection of a weak signal against the background of a strong signal the behavior of the response of the instrument (of the function of uncertainty in the case of correlation reception) for all values of mistuning between signals, and not only within the limits of the principal maximum of the response, becomes of interest. Inasmuch as the response may not decrease monotonically with increase of detuning, and may have spurious maximum, for the case of targets of arbitrary, differing strength it is not possible to introduce in general the idea of an interval of resolution. hereiving power is characterized by the magnitude of response as a function of detuning.

In connection with the problem of resolution of targets of different strength the function of uncertainty over the whole plane τ , Ω was subjected to therough investigation [4, 36], and there were formulated requirements on modulation from the point of view of resolving power. These requirements, naturally, reduced to decrease of the values of $|C(\tau, \Omega)|^2$ outside the principal maximum ($\tau = 0$, $\Omega = 0$) and narrowing of this maximum. Increase of resolving power by selection of modulation during correlation reception is limited by integral properties of the function of uncertainty (see Chapter I). Nonethelers in this direction there were obtained a series of results very useful for practice (for instance, the creation

-736-

of phase-code manipulation).

Likewice it is possible to consider the problem of resolution of angular coordinates. Increase of angle resolving power is provided by decrease of the level of side labes of directional patterns on reception and transmission and increase of the limensions of antennas. During consideration of these questions it is consideration of the known theorem of mutuality [67], to limit oneself to consideration of the receiving pattern. Here there becomes alear the analogy between problems of resolution for angles, range and speed.

The expected signal in the problem of processing the field in the aperture constitutes a plane wave $\mathbf{E}_0 e^{i\omega t - i\mathbf{k}\mathbf{r}}$, where \mathbf{k} - wave vector; \mathbf{r} - radius vector of the point of observation. As a result of multiplication of the field in the aperture $y(\mathbf{r}, t)$ by the expected signal and integration over the aperture σ , we obtain

$$Z(k, t) = \int y(r, t) e^{-ikr} dr.$$
 (13.1.1)

If the leading edge of the expected wave is parallel to the aperture $(\mathbf{kr} = 0$ in the aperture), transformation of (15.1.1) corresponds to cophasal summation of the aperture field. Obviously, this operation ensures obtaining of maximum energy of the useful signal for given dimensions of the aperture and, consequently, is optimum with respect to natural noises of the receiver. For dimensions of the aperture which are large as compared to wavelength operation (15.1.1) is optimum energy of the with respect to a background of the type of white noise (components of the cockground arriving from different directions are identical on average in intensity and are statistically independent) [11]. Correctness of this last affirmation is clear from the space-time analogy.

An unclog of the function of uncertainty is, obviously, the dependence of the square of the modulus of response of a system with cophasal summation on a plane wave on the direction of arrival of the wave, i.e., function

$$|G(\mathbf{k})|^{\mathbf{s}} = \left| \frac{1}{\mathbf{s}} \int e^{i\mathbf{k}\mathbf{r}} d\mathbf{r} \right|^{\mathbf{s}}, \qquad (1 \le 1, 2)$$

coinciding with the sirections) pattern of the enterna with respect to power with approximation of the field in the aperture.

In practice decrease of side totes 1, proving by schedion of a method of processing the field, stiftering from coplessing them. Here, of course, there is

-1311
less in power of the received signal and, consequently, in the signal-to-noise ratio. Making an analogy between space and time processing of a signal, it is possible to say that this method of increasing resolving power corresponds to selection of processing of the time signal in the receiver, and to selection of the form of the time signal there corresponds, although not exactly, to selection of the shape and dimensions of the aperture.

一日本語を読みます。 「日本語を読む」

ale and all the parts of

In theoretical consideration of the question of possible means and limits of the increase of resolving power it is natural to use both selection of the method of processing, and also selection of the form of the signal. Complete solution of this problem should include synthesis of the system of processing the received signal, optimum in resolving power, and finding the form of signal which under the imposed limitations during optimum processing provides the best performance indices of the system. Of great interest also is investigation of the criticality of optimum processing and of the form of the signal. Two approaches to solution of the problem at hand are possible, differing in the criteria of optimulity utilized:

1. The criterion of optimality can be formulated directly for the response of the instrument, not connecting it directly with functions of the radar (detection, measurement of coordinates and so forth). Resolving power here is understood in the clussical sense as the ability of an instrument to separate signals from observed targets. Subsequently we shall call systems synthesized with this approach systems of optimum separation of signals.

2. An optimum system can, as this is done in the theory of room, be could be by methods of the theory of statistical solutions, preceeding from the quality of examples by the rader of various functions in the presence of home tangets [43, 44, 4]. The problem of separation of signals in this case is not period beforehold. The many which in the very best manner produces detection or measurement of constitutes (or execute some other function) in the presence of several elevely because tangets is considered optimum in its receiving nower. Resciving power in this approach is understood as the ability to execute some function in the oresence of rangetary tangets. This concept thus turns out to be very bread.

Each of the shown approached has its advantaged and definitions. The first convoach is preferable when there is no a pricel intermation above the matter was leadion of reacized targets. The advantage is also simulating of the content of the rection and of the obtained results, is definitely of some improved to be

-130-

difficulty of correct selection of a criterion of optimality having meaning for a broad class of multitarget problems.

The second approach is more rigorous; however, consistent use of it requires assignment of a priori information about the number and position of targets. Such an approach may be preferable, e.g., in the case of simultaneous measurement of coordinates of all detected targets. It has also great value for finding the limits of increase of quality of multitarget systems.

below we shall give examples of use of both approaches and show cases when they lead to close results.

§ 13.2. <u>Synthesis of a System of Optimus</u> Separation of Signals

13.2.1. Solving the Problem in a General Form

In accordance with the classification introduced move during synthesis of a synthem of optimum 2-paration of signals the criterion of optimulity is forsulated directly for the response of the system, which chould be small for signals interfering with respect to a given channel and large for the useful (detected) signal. The optimum is obtained as a result of a compromise between these two requirements. As examples of criteria of optimulity of systems of separation of signals it is possible to indicate the criterion of a maximum ratio of the culput power of the useful signal to the total power of interfering signals with an assigned value of response to the separated signal; the criterion of a maximum of a maximum of the signals with an assigned value of response to the separated signal; the criterion of a maximum of the signals with a solution of the signal to the total power of interfering signals with an assigned value of response to the separated signal; the criterion of a maximum of the signal to the separated signal; the criterion of a maximum of the signal of the separated signal; the criterion of a maximum of the separatement of the separated signal; the criterion of a maximum of the separatement of the separatement of the separatement of the separated signal; the criterion of a maximum of the separatement of the sepa

We shall first synthesize an optimum system in acc riance with the last of the synthesized operations on the received signal from the relationship of the synthesized operations on the received signal from the relationship of strengths of the useful and interfering signals, which in real conditions is accually data with the begins is denoted by results of synthesis bared on the orithship of the useful and interfering signals, which in real conditions is accually data with the begins of the right of synthesis bared on the orithship of the useful model of the right of synthesis bared on the orithship of the definition of the right of synthesis bared on the orithship of the definition of the right of synthesis bared on the original is and we define that for signal-to-interference with a line as compared a party months of provide with replice the original of the theory of the inclusion, which replies a difference between there are disting that is not subtacted, with replice the mean is ordered as the theory of the inclusion of the original difference between there are no ordered as of the theory of the inclusion of the ordered of the order of the theory of the inclusion of the order of the order of the order of the theory of the inclusion of the order of

-1500

very large overlaps of the resolved signals, where for receiving any reliable solutions for separate targets from the observed set there is required such an increase of power of the useful signal as compared to the case of absence of interfering signals that for many typical radar problems this is equivalent to the impossibility of resolution with the given signal parameters (width of the spectrum of modulation, dimensions of the aperture, time of observation).

During solution of the problem of synthesis we shall consider the system of separation of signals linear (which is very natural, inasmuch as we want to suppress the sum of interfering signals with unknown coefficients), and we shall limit our consideration to such times of separation of signals in which fluctuating changes of these signals are negligible.

This limitation is not essential, inasmuch as separation of signals by parameters of modulation depending on range and angular coordinates is best realized in each period of modulation, which is usually small as compared to the time of correlation of fluctuations.

Let us turn to solution of the problem at hand with the imposed limitations. Mathematically this problem can be formulated in the following way. Let us assume that in multi-dimensional* domain S_0 there is assigned a system of random signals of form Re $A_{\lambda}\phi(c, \lambda)$, where λ - parameter taking values in domain A_0 , $A_{\lambda} = independence of form complex coefficients not depending on s. For each of the signals we introduce a correlation function$

$$R_{\lambda}(s_{i}, s_{i}) = E_{\lambda} \operatorname{Re} \varphi(s_{i}, \lambda) \varphi^{*}(s_{i}, \lambda). \qquad (13.2, 1)$$

Functions $\psi(z, \lambda)$ are assumed complex and such that

$$\int_{S_{\bullet}} \varphi(s, \lambda_{i}) \varphi(s, \lambda_{i}) ds = 0$$

For all λ_1 , λ_2 . Let us separate from the introduced signals useful signal $\varphi(s, \lambda_0)$ and signals $\varphi(s, \lambda_1)$, ..., $\varphi(s, \lambda_n)$, ..., to be suppressed (interfering signals). Let us assume that in that same domain S_0 there is assigned interference with correlation function $R(s_1, s_2)$. It is necessary to find such a reference signal $\psi(s, \lambda_0)$, that the signal-to-interference ratio

"Consideration of the multi-dimensional case is not connected with any difficulties and at the same time permits us to cover the problem of angle resolution.

$$q_{\bullet} = \frac{E_{\bullet}}{2} \frac{\left| \int_{S_{\bullet}} \psi(s, \lambda_{\bullet}) \psi^{\bullet}(s, \lambda_{\bullet}) ds \right|^{2}}{\int_{S_{\bullet}} \psi(s_{1}, \lambda_{\bullet}) R(s_{1}, s_{2}) \psi^{\bullet}(s_{2}, \lambda_{\bullet}) ds_{1} ds_{2}}$$
(13.2.2)

is marinum on the condition

$$\int_{S_0} \psi(s, \lambda_0) \varphi^*(s, \lambda) ds = 0 \quad \text{when} \quad \lambda \in \Lambda_1, \qquad (13.2.3)$$

where Λ_1 - domain of variation of parameters of interfering targets, signals of which it is necessary to suppress.

Ocviously, without disturbing generality it is possible to consider

$$\int_{S_0} \psi(s, \lambda_0) \varphi^*(s, \lambda_0) ds = 1. \qquad (13.2.4)$$

Fulfillment of this condition is ensured by selection of a constant factor for $\psi(z, \lambda_0)$. Here the problem is reduced to minimization of the q - atic functional

$$\iint_{S_0} \Psi(s_1, \lambda_0) R(s_1, s_2) \Psi^*(s_1, \lambda_0) ds_1 ds_2$$
(13.2.5)

under conditions (13.2 and (13.2.4). In order to solve this problem without passing to consideration. Freal and imaginary parts of $\psi(z, \lambda_0)$, one should add to (13.2.3) and (13.2.4) complex conjugate conditions.

Let us consider the case when Λ_1 consists of discrete points $\lambda_1, \ldots, \lambda_n$. In accordance with the known Lagrange method the problem of minimization of the functional with the considered additional conditions is equivalent to the problem of finding the poslute minimum of the functional

$$\int_{S_{\bullet}} \psi(s_{1}, \lambda_{0}) R(s_{1}, s_{2}) \psi^{*}(s_{1}, \lambda_{0}) ds_{1} ds_{2} - \sum_{k=0}^{n} \left[w_{0k}^{\prime} \int_{S_{\bullet}} \psi(s, \lambda_{0}) \varphi^{*}(s, \lambda_{k}) ds + w_{0k} \int_{S_{\bullet}} \psi^{*}(s, \lambda_{0}) \varphi(s, \lambda_{k}), ds \right], \qquad (13.2.1)$$

where w_{OK} and w'_{OK} are determined from conditions (13.2.3) and (13.2.4) and their conjugates, respectively.

reputing to zero the first variations of (13.2.5) with respect to ; (s, k) and $\psi^*(s, \lambda_0)$, we find $w_{0k}^* \approx w_{0k}^*$ and

$$\int_{S_0} \Psi(s_1, \lambda_0) R(s_1, s_2) ds_1 = \sum_{k=0}^n w_{0k} \varphi(s_1, \lambda_k). \qquad (13, 7.7)$$

Multiplying both parts of (13.2.7) by function $W(s_2, s)$, determined by equation

$$\int_{\mathbf{s}_{0}} \Psi(\mathbf{s}_{1}, \mathbf{s}') R(\mathbf{s}', \mathbf{s}_{0}) d\mathbf{s}' = \delta(\mathbf{s}_{1} - \mathbf{s}_{0}), \qquad (13.2.3)$$

and integrating over sp, we obtain

$$\psi(s,\lambda_{0}) = \sum_{k=0}^{n} w_{0k} \int_{S_{0}} W(s,s_{1}) \varphi(s_{1},\lambda_{k}) ds_{1}. \qquad (13.2.9)$$

Substituting this expression in (13.2.3) and (13.2.4), it is easy to see that coefficients w_{Ok} are elements of the first line (with zero index) of a matrix, the reciprocal of matrix $\|C(\lambda_j, \lambda_k)\|$ (j, k = 0, 1, ..., n), where

$$C(\lambda_{j}, \lambda_{k}) = \iint_{S_{0}} \varphi(s_{1}, \lambda_{j}) W(s_{1}, s_{2}) \varphi^{*}(s_{3}, \lambda_{k}) ds_{1} ds_{3}. \qquad (13.2.10)$$

Obviously $C(\lambda_j, \lambda_k)$ constitutes the generalized analog of the autocorrelation function of the sounding signal. Substituting (13.2.9) in (13.2.2), we obtain the signal-to-interference ratio with optimum separation of signals:

$$q_0 = \frac{E_0}{2w_{00}}$$
 (13.2.11)

If the useful and interfering signals differ so strongly that $C(\lambda_0, \lambda_k) = 0$ when $k \neq 0$, $w_{0k} = C^{-1}(\lambda_0, \lambda_0)\delta_{0k}$ and $q_0 = q_0^{i} = \frac{E_0}{2}C(\lambda_0, \lambda_0)$. Ratio

$$\Gamma := \frac{q_0}{q_0} := w_{00} C(\lambda_0, \lambda_0) \qquad (13.2.12)$$

characterizes the loss in the signal-to-interference ratio which is the cost of separation. It is possible in principle to separate signals as close as one wishes, if only there is a sufficient reserve of power of the useful signal.

It is easy to prove that loss Γ increases with increase of the number of signals which it is necessary to suppress. Let us assume that to the considered set there is added one signal. In accordance with the property of elements of inverse matrices, already used in Chapter IV Vol. I [see (4.9.20)], we have*

$$\boldsymbol{\omega}_{jk}^{(n+1)} - \boldsymbol{\omega}_{jk}^{(n)} - \frac{\boldsymbol{\omega}_{jk+1}^{(n+1)} \boldsymbol{\omega}_{n+1k}^{(n+1)}}{\boldsymbol{\omega}_{n+1}^{(n+1)}}, \qquad (13.2.13)$$

*Formula (13.2.13) is derived in the following way. Multiplying both parts of

-742-

where the upper index signifies the order of the matrix,

From (13.2.13) we obtain

$$\boldsymbol{w}_{\bullet\bullet}^{(n+2)} - \boldsymbol{w}_{\bullet\bullet}^{(n+1)} = \frac{\boldsymbol{w}_{\bullet n+2}^{(n+2)} \boldsymbol{w}_{n+2\,0}^{(n+2)}}{\boldsymbol{w}_{n+2\,n+2}^{(n+2)}} = \frac{\left| \boldsymbol{w}_{0\,n+2}^{(n+2)} \right|^{2}}{\boldsymbol{w}_{n+2\,n+2}^{(n+2)}} \ge 0, \qquad (13.2.14)$$

cince $\|w_{jk}\|$, as $\|C(\lambda_j, \lambda_k)\|$, is a Hermitian matrix. From (13.2.14) it follows that $\Gamma(n + 1) \geq \Gamma(n)$, since $w_{n+2n+2}^{(n+2)} > 0$ by virtue of the positive definitiveness of functions $C(\lambda, \lambda')$.

Above we did not touch on the question of selection of parameters of the suppressed signals. In those cases when parameters of the available targets are known exactly, or there is assigned an a priori distribution for these parameters, we naturally equate $\lambda_1, \ldots, \lambda_n$ to parameters of the interfering targets or use the principle of least risk in some form. In the absence of a priori data, encountered most frequently, we naturally use the minimax principle, giving to parameters of the suppressed signals those values at which these signals present the greatest danger for execution by the radar of its functions.

In particular it is possible to combine suppressed signals with coordinates of spurious maxima $\lambda_{m1}, \lambda_{m2}, \ldots$ of the function of uncertainty $|\mathcal{C}(\lambda_0, \lambda_1)|^2$. Here, due to the fact that in the reference signal $\psi(s, \lambda_0)$ there are present signals with parameters $\lambda_{m1}, \lambda_{m2}, \ldots$, there appear additional spurious maxima near each point λ_{m1} . These maxima of the response will have a magnitude of the order of the square of the initial maxima. If the resulting attenuation of spurious maxima is recognized as insufficient, it is possible to repeat the whole procedure, combining

[FOOTNOTE CONT'D FROM PRECEDING FACE]

equation

$$\sum_{I} w_{II} C_{Iv} = -iv$$

by $w_{i,k}^{(n+1)}$ and summaring for ν from 1 to n, we of ain

$$w_{jk}^{(n+1)} - w_{jk}^{(n)} (1 - \delta_{k,n+1}) = - w_{n+1k}^{(n+1)} \sum_{l=1}^{n} w_{jl} C_{l,n+1}.$$

detting k = n + 1, we find

$$w_{j,n+1}^{(n+1)} = \cdots w_{n+1,n+1}^{(n+1)} \sum_{l=1}^{n} w_{jl}^{(n)} C_{l,n+1}.$$

From the last two formular when $\beta_1 \in \beta_2$ is there immediately follows (15.2.13). It is considered that conditions of extense of the corresponding inverse matrices are realized.

 $\lambda_1,\;\ldots,\;\lambda_n$ both with the principal and with additional maxima, etc.

In certain cases the described procedure leads to complete elimination of spurious maxima of the response. Thus are matters, in particular, for phase-code manipulation [see Paragraph 13.2.2].

For a number of problems more justified is consideration of the continuum of parameters of suppressed signals, i.e., the case when domain h_{\perp} , introduced in connection with formula (13.2.3), is not assumed to consist of discrete points. This case is obtained from the considered one by passage to the limit with distances between points approaching zero, and the number of points approaching infinity. As a result relationship (13.2.9) is replaced by

$$\psi(s, \lambda_0) = w_0 \int_{S_0} W(s, s_1) \varphi(s_1, \lambda_0) ds_1 + + \int_{A_1} w(\lambda_0, \lambda_1) d\lambda_1 \int_{S_0} W(s, s_1) \varphi(s_1, \lambda_1) ds_1, \qquad (13.2.15)$$

where w_0 and $w(\lambda_0, \lambda_1)$ are determined by equation

$$w_{0}C(\lambda_{0},\lambda) + \int_{\lambda_{1}} w(\lambda_{0},\lambda_{1}) C(\lambda_{1},\lambda) d\lambda_{1} = \begin{cases} 0 & \text{when } \lambda \in \Lambda_{1}, \\ 1 & \text{when } \lambda = \lambda_{0}. \end{cases}$$
(13.2.16)

It is possible to simplify these "ormulas conewhat. Determining w₀ from (13.2.17) for $\lambda = \lambda_0$, we obtain

$$\boldsymbol{w}_{\boldsymbol{y}} = \frac{1 - \int_{\lambda_{1}} \boldsymbol{w}\left(\lambda_{1}, \lambda_{1}\right) C\left(\lambda_{1}, \lambda_{0}\right) d\lambda_{1}}{C(\lambda_{0}, \lambda_{0})} \qquad (13.2.17)$$

Substituting this expression in (13.2.10) when $2 \in \Lambda_0$, we have

$$\int_{A_{1}} w(\lambda_{0}, \lambda_{1}) [C(\lambda_{1}, \lambda_{0}) C(\lambda_{0}, \lambda) - C(\lambda_{0}, \lambda_{0}) C(\lambda_{1}, \lambda)] d\lambda_{1} == (13.2.17)$$

$$= -C(\lambda_{0}, \lambda).$$

Loss in the signal-to-interference ratio, as before, is determined by formula (13,2.12), where instead of w_{00} one should substitute w_0 , determined from (13.2.17). As a result we have

$$\Gamma = 1 - \int_{\lambda_1} w(\lambda_0, \lambda_1) C(\lambda_1, \lambda_0) d\lambda_1. \qquad (13.2.19)$$

Thus, solution of the problem of separation of signals for the considered case reduces to solution of a Fredholm integral equation of the first kind (12, 2, 12).

In certain cases loss in the signal-to-interference ratio, connected with total suppression of the selected interfering signals, can be impermissibly large and even infinite. In these cases it is expedient to use partial suppression of signals. Corresponding operations are most simply found from the condition of a maximum of the ratio of the power of the useful signal to the sum of the powers of interference and the selected interfering signals with assigned strengths of interfering signals at the input of the system.

We will give here only the result of solution of the problem in this formulation, inasmuch as the method of solution does not differ from the considered one. The reference signal in this case, as before, is recorded in the form (13.2.9), but matrix $\|w_{jk}\|$ is replaced by $\|v_{jk}\|$, the reciprocal of $\|C(\lambda_j, \lambda_k) + \frac{2}{E_k}\varepsilon_{jk}\|$, where $E_k =$ energy of the k-th signal.

The formula for the signal-to-interference ratio taking into account all interfering signals has the form

$$q_0 = \frac{E_0}{2v_{00}} - 1. \tag{13.2.20}$$

In § ± 3.3 it will be shown that optimum processing of a signal when using such a criterion coincides with optimum processing in the problem of detection of a separated signal against a background of interference and interfering signals.

It is easy to see that for large values of the signal-to-interference ratio for all the considered signals at the input of the system operations of partial and total suppression of signals, and also output signal-to-interference ratios, coincide. As a confirming example we shall consider the very simple case of one interfering target. In this case from formula (13.2.11) and (13.12.20) we obtain:

with total suppression of the interfering signal

$$q_{\bullet} = q'_{\bullet} (1 - \gamma');$$
 (13.2.21)

with partial suppression of the interfering signal

$$q_{\bullet} = q'_{\bullet} \left(1 - \frac{q'_{1}}{1 + q'_{1}} \chi^{2} \right), \qquad (13.2.82)$$

where q_i and q'_1 - signal-to-interference ratios for the useful and interfering signate without suppression; $\gamma == \left| \frac{C(\lambda_1, \lambda_2)}{C(\lambda_2, \lambda_2)} \right|$.

In Fig. 13.1 there is given the dependence of the ratio γ of quantities (13.2.22) and (13.2.21) on γ^2 for different q_1^{\dagger} , correcterizing loss in the



Fig. 13.1. Dependence of loss in the signal-to-interference ratio with replacement of partial separation of signals by total.

signal-to-interference ratio due to use of total ceparation instead of partial. Assigned a permissible value of loss Γ we can find the γ at which loss starts to exceed the permissible:

$$\gamma^{2} = 1 - \frac{1}{(\Gamma - 1)(1 + q'_{1}) + 1}$$

Introducing the density of strength of the interfering signals and passing to the limit as $n \rightarrow \infty$ and with the interval between signals approaching zero, we can obtain the proper formulas for the continuum of interfering targets. Here the problem coincides with that which was considered in § 4.11 Vol. I in connection with selection of a signal against a background of passive interferences.

13.2.2. Separa for of signals by Range

As a particular case we shall use the problem of separation of signals by range in the presence of interference in the form of white noise. Here, if the signal is periodic $S_{\rm C}$ coincides with the meriod of modulation:

$$\varphi(t, \tau) = u(t - \tau) e^{i\omega_{b}t}, \quad R(t_{1}, t_{2}) = N_{0} \mathcal{E}(t_{1} - t_{2}), \\ W'(t_{1}, t_{2}) = \frac{1}{N_{0}} \mathcal{E}(t_{1} - t_{2}), \\ C(\tau_{j}, \tau_{k}) = \frac{1}{N_{0}} \int_{0}^{T_{v}} u(t - \tau_{j}) u^{*}(t - \tau_{k}) dt = \frac{T_{v}}{N_{0}} C_{0}(\tau_{k} - \tau_{j}),$$
(13.2.23)

$$\gamma(l, \tau_0) = \frac{1}{N_0} \sum_{k=0}^{n} \omega_{0k} u(l - \tau_k) e^{l\omega_0 l}. \qquad (13.2.24)$$

Thus, the reference signal in a system of optimum separation is a linear combination of all n + 1 considered signals, orthogonal to n interfering signals. Multiplication by reference signal $\varphi(t_1, t_2)$ and integration can be carried out in the usual circuit of heterodyne processing, shown in Fig. 13.2. Under certain conditions the same result can be obtained in principle in a circuit with filtration.

Let us consider first the case when there is applied a filter designed to process the whole period of modulation. This is inevitable in the case of continuous



Fig. 13.2. Block diagram of heterodyne processing: 1 - mixer; 2 - 1 - ffilter; 3 - detector. radiation and can be considered one of the variants of processing periodic pulse modulation. We consider $\psi(t, \tau_0)$ a function of difference $\tau_0 - t$

$$-\psi(t,\tau_0) = h_1(\tau_0 - t)$$

and consider $h_1(\tau_0 - t)$ equal to zero outside

interval $\tau_0 - T_r$, τ_0 . Then $h_1(t)$ it can be considered the pulse response of a physically realizable filter. It is easy to see that passage of a signal through such a filter ensures separation of signals optimum in the considered meaning, where the separated signal is obtained at the output at a moment equal to the value of delay of the separated signal. Really, if to the filter input there proceeds signal $u(t - \tau')e^{iw_0 t}$, the output signal at time t is recorded in the form

$$Q_{\phi}(t) = \int_{t-T_{r}}^{t} h_{i}(t-x) u^{*}(x-\tau') e^{-i\omega_{\phi}x} dx =$$

$$= \frac{1}{N_{\phi}} \int_{t-T_{r}}^{t} \sum_{k=0}^{n} \omega_{\phi h} u(\tau_{\phi} - \tau_{k} - t + x) u^{*}(x-\tau') dx = \frac{T_{r}}{N_{\phi}} \sum_{k=0}^{n} \omega_{\phi h} C_{\phi}(\tau_{\phi} - \tau_{k} + \tau' - t);$$

and as time $t = \tau'$, in the form

$$Q_{\oplus}(\tau') = \frac{T_r}{N_{\bullet}} \sum_{k=0}^{n} w_{ok} C_{\bullet}(\tau_o - \tau_k) = 1.$$

At the same moment all signals for which $\tau' = \tau_0 = \tau_0$, $(\nu = 1, 2, ..., n)$, are completely suppressed:

$$Q_{\phi M}(\tau') = \frac{T_r}{N_o} \sum_{k=0}^n w_{ok} C_o(\tau_v - \tau_k) = 0.$$

Thus, we obtain a system whole output signal in each moment of time t collected with the output signal, delayed this time, of an optimum directly with total suppore alon of signals, separated $\tau_0 = \tau_p$ ($\nu = 1, ..., n$). In (), with with filtratice points $\lambda_0, \lambda_1, ..., \lambda_n$ shift with change of t, because factor relative colling. Somewhat more complicated is the matter with use of filtration for processing separate pulses. For physical realizability of the filter it is necessary that $\psi(t_1, \tau_0)$ turn into zero for t larger than some value t_0 . Inasmuch as u(t) has limited duration, it is possible to set u(t) = 0 when t > 0. Then by virtue of (13.2.24) $\tau_0 = \max_k \tau_k$, and it is possible to consider $\psi(t, \tau_0) = h(\max_k \tau_k - t)$ the pulse response of a physically realizable filter. If $\max_k \tau_k \neq \tau_0$ the result of optimum processing of a signal with $\tau = \tau_0$ is obtained with delay with respect to the moment of termination of this signal by time max $\tau_k - \tau_0$.

In principle it is possible to eliminate the delay in delivery of results of processing separate pulses if we beforehand require equality $\psi(t, \tau_0) = 0$ when $t > \tau_0$. Here it is possible to preserve the whole described approach if we consider $u(\tau - \tau_k) = 0$ when $t > \tau_0$ for all the considered signals. Such "truncation" of signals should, of course, be taken into account during calculation of elements of matrices $C(\tau_j, \tau_k)$ and w_{jk} . When using filtration with "truncation" of signals at the output of the filter at every romant of time there is separated a signal with $\tau = t$, and all signals with $\tau = t - \tau_0 + \tau_k$, "Altering at the same moment of time, are suppressed. From qualitative considerations in its quite clear that "truncation" of signals can lead to increase of corresponding values of $C(\tau_j, \tau_k)$ and that this increase should cause worsening of the ultimat-to-noise ratio as compared to the "untruncated" case. However there is a constant of a cases, where there exist signals for which "truncation" does not 1...t a constant of a processing values of the signals of the signals is the same there investigations in this direction are of consideration in the same there investigations in this direction are of consideration in the same there there exist signals for which "truncation" does not 1...t a constant of the same there investigations in this direction are of consideration in the same there investigations in this direction are of consideration in the same there investigations in the same there investigations in the same of consideration in the same there investigations in the same there investigations in the same of consideration in the same there investigations in the same of consideration in the same there investigations in the same of consideration in the same of consideration in the same filler the same of consideration in the same there investigations in the same of consideration in the same of c

Let us consider two examples of signals with phase-code manipulation (see Chapter I, Vol. I). In this case, combining delays of the suppressed signals with spurious maxima of the function of uncertainty, it is possible to achieve complete elimination of the spurious maxima. It is simple to show that the response of a linear instrument with a phase-manipulated reference signal to a phase-manipulated signal with the same code interval A, but delayed time τ , is determined by a formula analogous to (1.2.22):

$$Q(\mathbf{r}) = Q(\mu\Delta) - \left(\frac{\mathbf{r}}{\Delta} - \mu\right) \{Q(\mu\Delta) - Q[(\mu+1)\Delta]\}, \qquad (13.2.25)$$

where μ - the integral part of ratio $\frac{\pi}{2}$.

Seeking absences of response for τ , which are multiples of Δ , we also provide suppression of signals with intermediate values of τ .

Example 1. Let us consider as an example a continuous signal, manipulated in phase by the code of Hoffman [?]. As it is known, for such a code $C_0(\mu\Delta) = -\frac{1}{n}$ when $\mu \neq 0$. Let us find a reference signal (pulse response of the filter) at which response of the instrument to signals $\tau - \tau_0 = \mu\Delta(\mu \neq 0)$ is equal to zero. Here it is sufficient to consider values $|\mu| \leq \frac{n-1}{2}$, where n is the number of elements of the code, since for larger $|\mu|$ the whole picture is repeated periodically. It is easy to show that

$$w_{jk} = \frac{N_0}{T_r} \frac{n}{n+1} (1+\delta_{jk}),$$

$$\psi(t,\tau_0) = \frac{N_0 e^{i\omega_0 t}}{T_r} \frac{n}{n+1} \left\{ 2u(t-\tau_0) + \frac{1}{2} (1-\tau_0) \right\}$$

$$+\sum_{1}^{(n-1)/2} [u(t-\tau_{0}-\mu\Delta)+u(t-\tau_{0}+\mu\Delta)] \}$$
(13.2.27)

In different code intervals u(t) takes value $1/\sqrt{n}$ or $-1/\sqrt{n}$ (phase is equal to 0 or π). In the reference signal $\psi(t, \tau_0)$ at certain code intervals there appear zero values (signal gaps). Thus, the reference signal turns out to be modulated in amplitude. For instance, when $n = 2^3 - 1 = 7$ the sequence of values of $u(t - \tau_0)$ has the form 1, -1, -1, +1, +1, -1, and for $\psi(t, \tau_0)$ 0, -1, -1, -1, 0, 0, -1 [for brevity of recording we rejected in the expressions for u(t) and $\psi(t, \tau_0)$ the normalizing factors].

By the formula (13,2.12) it is easy to calculate the loss in the signal-to-mineration due to separation. Substituting (13,2.23) and (13,2.26) in (13,2.12), we obtain

$$\Gamma = \frac{2n}{n+1} \cdot$$

For large n loss is approximately double.

Example 2. Let us assume we have a pulse signal, munipulated in place by the code of Barker with n = 3. Such a code has the form 1, 1, -1.

The corresponding function of $C_G(n\Delta)$ is ...; $G_{j} = \frac{1}{2j}$, $G_{j} = 0$; $f_{j} = 0$;

$$w_{j\,j+\nu} = \frac{N_{\bullet}}{\pi} \int_{0}^{\infty} \frac{\cos\nu\lambda d\lambda}{1 + \frac{2}{3}\cos 2\lambda} = \\ \begin{cases} 0, \ \nu = 2k + 1\\ (-1)^{h} \frac{3\pi}{\sqrt{5}} N_{\bullet} \left(\frac{3 - \sqrt{5}}{2}\right)^{|k|}, \ \nu = 2k. \end{cases}$$

Coefficients $w_{jj+\nu}$ rapidly decrease with growth of ν . Therefore in practice it is possible to limit the duration of the reference signal. Quantity w_{jj} in accordance with (13.2.12) determines loss in the signal-to-interference ratio during separation of signals: $\Gamma = \frac{3\pi}{\sqrt{5}} \approx 4.2$. The considered method can be used for finding processing of codes with a larger number of elements. Difficulties arising here are of a purely calculating character.

13.2.3. Separation of Signals in the Direction of Arrival

The described approach can be used for synthesis of processing of the field in the antenna aperture, providing increase of angle resolving power. Here "omain S_0 is the surface of the aperture of the antenna, and signal $\varphi(s, \lambda)$ is a plane wave $e^{-i\mathbf{k}\mathbf{r}}$, where \mathbf{r} - radius vector of a point in the aperture, which we consider flat; \mathbf{k} - wave vector (parameter of the signal). The dependence of the signal on time may be ignored if we are not concerned with joint separation with respect to angles and range.*

Interference in the problem of second tor. By angles can be considered to consist of two components: background radiation and satural noises of the receiver. We shall consider these components separately.

Background radiation we shall consider superposition of plane waves arriving from different directions with random and independent amplitudes and phases. Here the correlation function for incorporate in the aperture has the form [51]

 $R_{\phi}(\mathbf{r}_{1}-\mathbf{r}_{3}) = N_{\phi} - \frac{1}{\pi} - \int_{0}^{2\pi} \int_{0}^{\frac{1}{2}} e^{-(\mathbf{r}_{1}-\mathbf{r}_{2}) \mathbf{r}} \sin \theta \cos \theta d\varphi d\theta =$ $= N_{\phi} - \frac{2J_{1}\left(\frac{\omega}{c} | \mathbf{r}_{1}-\mathbf{r}_{3} |\right)}{\frac{\omega}{c} | \mathbf{r}_{1}-\mathbf{r}_{3} |}$

The problem of simultaneous separation with respect to unrise and range was considered by Y_{44} is thirman $\{C^{(1)}\}$ for the case of an operture in the form of an infinitely small slot.

where $J_{4}(x)$ - Bessel function of the first kind;

spectral current density of power of the background through the aperture at the considered frequency;

Low Mark Street Street

 $\frac{\omega}{c} \rho = k_{\sigma} - projection$ of the wave vector on the surface of the aperture.

The formula for R ϕ (\mathbf{r}_1 , \mathbf{r}_2) is derived in the following way. Let us consider as elementary plane wave striking the aperture at an angle of 5 to the normal and angle ϕ with respect to an arbitrarily selected direction \mathbf{x} in the aperture plane. We consider that the phase and amplitude of the wave ϕ random, where all values of phase in interval (0, 2π) are equiprobable. Then, obviously, the correlation furction of the field of this plane wave for any two points of the aperture is

$$\overline{A}\operatorname{Re} e^{i(\mathbf{r}_1-\mathbf{r}_2)\mathbf{k}} = \overline{A}\operatorname{Re} e^{i\frac{\mathbf{r}_1-\mathbf{r}_2}{c}}$$

where k - wave vector,

A-average density of the flux of power, corresponding to the given plane wave.

Inasmuch as on the receiver of radiation there usually acts part of the power flux passing through the aperture, we must multiply $\bar{\Lambda}$ by cos §. If waves arriving from different directions are not correlated, the correlation function for their sum is obtained by summation of correlation functions of the components. There are interpreted the angular density of the strength of $\frac{4}{\pi}N_{th}$ in accordance with formula

$$\overline{A} = \frac{N_{\bullet}}{\pi} \sin \vartheta d\varphi d\vartheta,$$

we obtain as a result of integration the above-mentioned expression for R $_{\oplus}$ $(\mathbf{r}_{1}, \mathbf{r}_{2})$. To correlation functions R $_{\oplus}$ $(\mathbf{r}_{1}, \mathbf{r}_{2})$ there corresponds spectral density

$$S_{\phi}(\mathbf{p}) = \begin{cases} \frac{1}{\pi} N_{\phi} \lambda^2, & |\mathbf{p}| < \frac{\omega}{c}, \\ 0 & |\mathbf{p}| \ge \frac{\omega}{c}. \end{cases}$$

In examining natural noises one should distinguish noises added to the signal in the course of space processing (output power of these noises depends on the character of processing) (noises of the first kind), and noises added to the signal after space processing (noises of the second kind). Noises of the first kind play a large role in systems with a phased antenna array, in which every element of the array is connected to a noisy receiver. Space processing of a signal in this case consists in weighted summation of output voltages of these receivers. With a large number of closely located elements summation can be replaced by integration, considering noises δ -correlated. Allowance for the influence of noises of the first kind and of background and synthesis of a system for ceparation of signals are produced in full conformity with the general method described in 13.2.1.

Noises of the second kind, added to a signal after space processing, play a basic role in those cases when space processing is produced without preliminary amplification of signals received from separate elements of the antenna. The power of these noises does not depend on the nature of space processing; therefore such a case requires special consideration.

Taking into account the presence of a δ -correlated background with spectral density $\frac{1}{\pi}N_{-\phi} = \lambda^2$, natural noises of the first kind, also assumed δ -correlated, and noises of the second kind, we write the signal-to-interference ratio, which is required to be maximized, in the form

$$q_{0} = \frac{E_{0}}{2} \frac{\left|\frac{1}{\sigma}\int\psi(\mathbf{r},\mathbf{k}_{0})e^{i\mathbf{k}_{0}\cdot\mathbf{r}}d\mathbf{r}\right|^{2}}{N_{2}+N_{0}\frac{1}{\sigma}\int|\psi(\mathbf{r},\mathbf{k}_{0})|^{2}d\mathbf{r}} \qquad (13.2.26)$$

In this formula σ - area of the aperture, N_2 - spectral density of noises of the second kind, $N_0 = N_1 + \frac{1}{\pi}N_{\tilde{\Phi}} - \lambda^2 - \sin \sigma f$ spectral densities of noises of the first kind (N_1) and the background [we consider factors with $\sigma(\mathbf{r}_1 - \mathbf{r}_2)$ in expressions for correlation functions]. In (13.2.28) by means of introduction of factor $\frac{1}{\sqrt{\sigma}}$ with $\psi(\mathbf{r}, \mathbf{k}_0)$, it is considered that in the absence of amplification the energy received from the aperture cannot be greater than the incident energy (there occurs summation of the field with respect to power, and not to voltage).

With maximization of expression (13.2.28) on function ψ there should be imposed

an additional condition, limiting amplification of the signal to addition of noises of the second kind. If this condition is absent, the result is trivial: the constant factor in $|\psi(\mathbf{r}, \mathbf{k}_{0})|$ should be so large that component N₂ in the denominator of (13.2.28) can be ignored.

If we write the shown condition in the form

$$\frac{1}{\sigma}\int_{\sigma}|\psi(\mathbf{r},\mathbf{k}_{0})|^{s}d\mathbf{r}=1,$$

i.e., assume that amplification at separate points of the aperture is possible, but on the average over the aperture it is absent, then, as it is easy to see, the expression for $\psi(\mathbf{r}, \mathbf{k}_0)$ with accuracy of a constant factor coincides with that obtained for $N_2 = 0$:

$$\Psi(\mathbf{r}, \mathbf{k}_{\bullet}) = A \sum_{j=0}^{n} w_{\bullet j} e^{-i\mathbf{k}_{j}\mathbf{r}}, \qquad (13.2.29)$$

where $k_j(j = 1, ..., n)$ - wave vectors of suppressed signals: [maximization of (13.2.28) is performed with the additional conditions (13.2.3)].

Coefficient A is determined from condition

$$\frac{|A|^2}{\sigma} \int_{\sigma} \sum_{l,l} w_{ol} w^*_{ol} e^{-l(\mathbf{k}_j - \mathbf{k}_l)\mathbf{r}} d\mathbf{r} =$$

$$= \frac{|A|^2}{\sigma} \sum_{l,l} w_{ol} w^*_{ol} C(\mathbf{k}_l, \mathbf{k}_l) = |A|^2 w_{oo} = 1,$$

$$q_o = E_o / [2w_{oo} (N_o + N_o)].$$

so that

In these formulas
$$w_{jk}$$
 - elements of a matrix, the reciprocal of $|C(\mathbf{k}_j, \mathbf{k}_l)|$,

and

$$C(\mathbf{k}_i, \mathbf{k}_i) = \frac{1}{\sigma} \int e^{-i(\mathbf{k}_i - \mathbf{k}_i)\mathbf{r}} d\mathbf{r}.$$

If amplification and attenuation of the signal are absent at every point of the aperture, $|\psi(\mathbf{r}, \mathbf{k}_0)| = 1$, and separation is carried out only by phase processing. Function $\psi(\mathbf{r}, \mathbf{k}_0)$ in this case can be recorded in the form

$$\mathbf{k}(\mathbf{r}, \mathbf{k}_{\mathbf{0}}) = \mathbf{e}^{i\mathbf{\theta}(\mathbf{r}, \mathbf{k}_{\mathbf{0}})}, \qquad (3.3.5)$$

where $\theta(\mathbf{r}, \mathbf{k}_0)$ - real function.

Substituting (13.2.30) in (13.2.8) and (13.2.3), we obtain

$$q_{0} = \frac{E_{0}}{2(N_{0} + N_{0})} \left\{ \left(\frac{1}{\sigma} \int \cos \left[\vartheta(\mathbf{r}, \mathbf{k}_{0}) + \mathbf{k}_{0} \mathbf{r} \right] d\mathbf{r} \right)^{2} + \left(\frac{1}{\sigma} \int \sin \left[\vartheta(\mathbf{r}, \mathbf{k}_{0}) + \mathbf{k}_{0} \mathbf{r} \right] d\mathbf{r} \right)^{2} \right\}.$$
(13.2.31)

The problem reduces to maximization of (13.2.31) under condition

$$\frac{1}{\sigma}\int_{\sigma}\cos\left[\theta(\mathbf{r},\,\mathbf{k}_{0})+\mathbf{k}_{j}\mathbf{r}\right]d\mathbf{r}=\frac{1}{\sigma}\int_{\sigma}\sin\left[\theta(\mathbf{r},\,\mathbf{k}_{0})+\mathbf{k}_{j}\mathbf{r}\right]d\mathbf{r}=0. \tag{13.8.32}$$

Using the method of undetermined Lagrange factors, we obtain

$$\psi(\mathbf{r}, \mathbf{k}_{0}) = e^{i \theta(\mathbf{r}, \mathbf{k}_{0})} = \frac{\sum_{j=0}^{n} a_{j} e^{-i\mathbf{k}_{j}\mathbf{r}}}{\left|\sum_{j=0}^{n} a_{j} e^{-i\mathbf{k}_{j}\mathbf{r}}\right|}, \qquad (13.2.33)$$

where coefficients a, are determined from condlitions

$$a_{\bullet} = \frac{1}{\sigma} \int_{\sigma} \frac{\sum_{j=i}^{j} a_{j} e^{-i(\mathbf{k}_{j} - \mathbf{k}_{0})\mathbf{r}}}{|\sum_{j=1}^{j} a_{j} e^{-i\mathbf{k}_{j}\mathbf{r}}|} d\mathbf{r},$$

$$\frac{1}{\sigma} \int_{\sigma} \frac{\sum_{j=1}^{j} a_{j} e^{-i(\mathbf{k}_{j} - \mathbf{k}_{0})\mathbf{r}}}{|\sum_{j=1}^{j} a_{j} e^{-i\mathbf{k}_{j}\mathbf{r}}|} - d\mathbf{r} = 0, \quad l = 1, ..., n.$$
(13.2.34)

Thus, the optimum weighting function $\psi(\mathbf{r}, \mathbf{k}_0)$ in the considered case is a linear combination of all the considered signals, multiplied by a certain function of the radius vector, ensuring constancy of $|\psi(\mathbf{r}, \mathbf{k}_0)|$. The signal-to-interference ratio with such a reference signal, as it is easy to prove by substitution of (13.2.33) and (13.2.34) in (13.2.28), has the form

$$q_{0} = \frac{B_{0}}{2(N_{2} + N_{0})} |a_{0}|^{2} = q'_{0} |a_{0}|^{2}, \qquad (13.2.35)$$

where q_0' - signal-to-interference ratio in a system without suppression of interfering signals. The obtained formulas show that calculation of optimum processing of a signal in this case is considerably more complicated than in that considered earlier. We find the gain in the signal-to-noise ratio to which replacement of processing found without taking into account absence of amplification of power in the antenna by treatment of the form just now considered leads.

In the first case there is produced multiplication of the received signal by a reference signal of form (13.2.29), normalized, however, in such a way that $|\psi(\mathbf{r}, \mathbf{k}_0)| = 1$. Energy of the signal and the spectral density of noise at the output of such a system will be recorded in the form

$$E_{01} = E_0 \left| \frac{1}{\sigma} \int_{\sigma} \frac{\psi(\mathbf{r}, \mathbf{k}_0) e^{i\mathbf{k}_0 \mathbf{r}} d\mathbf{r}}{\max \left| \psi(\mathbf{r}, \mathbf{k}_0) \right|^2} \right|^2,$$

$$N_{01} = N_0 + N_0 \frac{1}{\sigma} \frac{\int_{\sigma} |\psi(\mathbf{r}, \mathbf{k}_0)|^2 d\mathbf{r}}{\max \left| \psi(\mathbf{r}, \mathbf{k}_0) \right|^2},$$

where E_0 and N_0 - energy of the signal and spectral density of the background radiation in the absence of separation.

Substituting these expressions in (13.2.28), for the signal-to-interference ratio we obtain

$$q_{0} = \frac{E_{01}}{2N_{01}} = \frac{E_{0}}{2} \frac{1}{N_{2} \max \left| \sum_{j} w_{0j} e^{-ik_{j} t} \right|^{2} + N_{0} w_{00}}.$$
 (13.2.30)

The sought gain is characterized by the ratio of (13.2.35) and (13.2.36):

$$\Gamma_{1} = \frac{|a_{0}|^{2} \left(\max \left| \sum_{i} w_{0i} e^{-i \mathbf{k}_{i} \mathbf{r}} \right|^{2} + \frac{N_{0} w_{00}}{N_{0}} \right)}{1 + \frac{N_{0}}{N_{0}}}.$$
 (13.2.37)

Considering the problem of separation of signals in the direction of their arrival, we considered that the field in the antenna aperture is directly subjected to processing. In antennas of the image type accessible for processing usually is the uit acted field in the focus of the mirror, equivalent, as we know [65], to the diffected field at great distances from the aperture in the absence of a focusing mirror. For large dimensions of the aperture this diffracted field can be calculated by a simplified Kirchhoff formula

$$y_1(\mathbf{k}') \sim \frac{1}{\sigma} \int_{\sigma} y(\mathbf{r}) e^{i\mathbf{k}'\mathbf{r}} d\mathbf{r},$$
 (15.2.37)

where $\mathbf{k}' = \frac{\omega}{c} \rho$, and $\rho - a$ unit vector characterizing the direction for which the diffracted wave is considered.

In accordance with this formula the plane wave e^{-ikr} will be transformed with accuracy of a factor into function C(k - k'), which in connection with this can be called the response of the aperture to the plane wave.

Let us establish conformity between processing of the field in the aperture of an antenna and the diffracted field. As can be seen from (13.2.38) as a result of multiplication of the field in the aperture by a plane wave and integration there is obtained a diffracted wave for a certain direction. Thus, multiplication by reference signal (13.2.29), which is a linear combination of plane waves, and integration over the aperture can be replaced, in principle, by formation of the same linear combination composed of values of the diffracted field for the corresponding directions. However, by virtue of the inevitable presence of noises of the second kind, such an operation is unprofitable in terms of power. The fact is that $|y_1(\mathbf{k})|^2$ is proportional to the density of the power flux of the diffracted field. In order to obtain a signal proportional to $y_1(\mathbf{k})$ it is necessary to take, in general, an infinitesimal elementary solid angle near the given direction \mathbf{k} . The energy of such a signal, and consequently also the ratio of this energy to the spectral density N₀ of noise of second kind, will be infinitesimal.

It is possible to construct another operation, equivalent to the considered one from the point of view of separation of signals and allowing us to completely use the energy reaching the aperture. Let us consider expression

$$Z(\mathbf{k}) = \int y_{\mathbf{k}}(\mathbf{k}') C(\mathbf{k}' - \mathbf{k}) d\mathbf{k}' \sim \frac{1}{\sigma} \int y(\mathbf{r}) d\mathbf{r} e^{i\mathbf{k}\mathbf{r}} \times \frac{1}{\sigma} \int C(\mathbf{k}' - \mathbf{k}) e^{i(\mathbf{k}' - \mathbf{k})\mathbf{r}} d\mathbf{k}'.$$

$$(13.7.52)$$

$$|\mathbf{k}'| < \frac{\sigma}{\sigma}$$

For dimensions of the aperture, large as compared with the wavelength, $C(\mathbf{k})$ rapidly decreases with growth of $|\mathbf{k}|$. Therefore, it is possible to replace the limits of integration in the integral over \mathbf{k}' by infinite limits and to consider this integral the two-dimensional reverse Fourier transform. Inasmuch as the response of the aperture $C(\mathbf{k})$ is the Fourier transform of a function equal to $\frac{1}{\sigma}$ in the aperture and to zero outside the aperture, the considered integral for all points of the aperture has a constant value. Thus,

-756-

$$Z(\mathbf{k}) \sim \frac{1}{a} \int y(\mathbf{r}) e^{i\mathbf{k}\mathbf{r}} d\mathbf{r}$$

and multiplication of the field in the aperture by plane wave $e^{-i\mathbf{k}\mathbf{r}}$ and integration over r is equivalent to multiplication of the diffracted field $y_1(\mathbf{k}')$ by response of the aperture $C(\mathbf{k}' + \mathbf{k})$ and integration in all directions (over the whole focal plane in the case of an image antenna).

To reference signal $\sum_{j} e^{-ik_j r}$ in the aperture there corresponds, obviously, reference signal $\sum_{j} C(k' + k_j)$ for the diffracted field or $\sum_{j} C(\frac{r_{\oplus}}{d_{\oplus}} + k_j)$ for the focal plane $(d_{\oplus}$ - focal length; r_{\oplus} - radius vector in the focal plane). The obtained results determine the structure of the irradiator of an image (or lens) antenna, ensuring the required separation of signals with a maximum signal-tointerference ratio.

Example. Let us consider a linear aperture (an aperture with infinitesimal width), for which

$$C(\mathbf{k}_{1}-\mathbf{k}_{2}) = \frac{\sin\frac{\pi d}{\lambda}(\sin\theta_{1}-\sin\theta_{2})}{\frac{\pi d}{\lambda}(\sin\theta_{1}-\sin\theta_{2})} \approx \frac{\sin\frac{\pi d}{\lambda}(\theta_{1}-\theta_{2})}{\frac{\pi d}{\lambda}(\theta_{1}-\theta_{2})} = C(\theta_{1}-\theta_{2}),$$

where θ - angle between the wave vector and the line of the aperture:

d - length of the aperture;

 λ - wavelength.

j.

We demand suppressions of first spurious maxima, occurring for $\partial_{\rm In} = 4.5 \frac{\lambda}{\pi d}$, where

$$C(\theta_m) = \gamma_1 = -0.217, \quad C(2\theta_m) = \gamma_1 = -0.046.$$

With n = 2 and a symmetric location of interfering signals matrices $\|C(\lambda_1, \lambda_k)\|$ and w_{jk} have form

$$\|C(\lambda_{j}, \lambda_{k})\| = \left\| \begin{array}{c} 1 & \gamma_{i} & \gamma_{i} \\ \gamma^{*}_{i} & 1 & \gamma_{i} \\ \gamma^{*}_{a} & \gamma^{*}_{i} & 1 \end{array} \right\|,$$

$$\|w_{jk}\| = \frac{1}{\Delta} \left\| \begin{array}{c} 1 - |\gamma_{i}|^{2} & \gamma^{*}_{i}\gamma_{a} - \gamma_{i} & \gamma_{i}^{2} - \gamma_{2} \\ \gamma_{i}\gamma^{*}_{a} - \gamma^{*}_{i} & 1 - |\gamma_{a}|^{2}\gamma^{*}_{i}\gamma_{a} - \gamma \\ \gamma_{i}^{2^{*}} - \gamma^{*}_{a} & \gamma_{i}\gamma^{*}_{a} - \gamma^{*}_{i} & 1 - |\gamma_{i}|^{a} \end{array} \right\|,$$

where $\Delta = 1 - 2|\gamma_1|^2 - |\gamma_2|^2 + 2 \operatorname{Re} \gamma_1^2 \gamma_2^*$.

From this it is easy to find loss in the signal-to-interference ratio and optimum processing for the case when there is allowed amplification during processing of the field from separate sections of the aperture. Substituting w_{00} and $C(\lambda_0, \lambda_0)$ (elements on the intersection of diagonals in the above matrices) in (13.2.12), we find $\Gamma = w_{00} = (1 - |\gamma_2|^2)/\Delta \approx 1.1$. Loss comprises in all 10%. Optimum processing of the field in the aperture for the case when the selected target is on the axis of the antenna ($\theta_0 = 0$) is recorded in the form

$$\begin{aligned} \psi(x, \mathbf{k}_{0}) &= \frac{1}{\nabla w_{00}} \frac{1}{|\mathbf{v}|_{00}} \frac{|\mathbf{v}_{1}|^{2} |\mathbf{x}_{1}|^{2}}{|\mathbf{v}_{00}|^{2}} = \\ &= \frac{1}{\Delta \sqrt{w_{00}}} \left[(\gamma_{1} \gamma^{*}_{a} - \gamma^{*}_{1}) e^{j \frac{2\pi x}{\lambda} \theta_{m}} + (1 - |\gamma_{a}|^{2}) + \right. \\ &+ (\gamma^{*}_{1} \gamma_{a} - \gamma_{1}) e^{j \frac{2\pi x}{\lambda} \theta_{m}} \right] \approx 1 + 0.415 \cos \frac{2\pi \theta_{m}}{\lambda} x, \end{aligned}$$

where x - coordinate of the point in the aperture.

The field at every point of the aperture should be multiplied by $\psi(\mathbf{x}, \mathbf{k}_0)$, which corresponds to a certain amplification or attenuation, inasmuch as phase



Fig. 13.3. Directional pattern of a linear antenna: ______ with suppression of the first side lobe; _____ without suppression of the lobe.

shirt is equal to zero, and is integrated over the aperture.

Let us consider the directional pattern for power, obtained during suppression of first side lobes by the shown method:

$$g(\theta) = \left| \frac{1}{d} \int_{-d/2}^{d/2} \psi(x, \mathbf{k}_0) e^{i\frac{2\pi x}{\lambda}} dx \right|^2 \approx |C(\theta) + 0.207 [C(\theta - \theta_m) + C(\theta + \theta_m)]|^2.$$

Relationship g(9) is shown in Fig. 13.3. In the same place for comparison there is plotted the directional pattern with an uncompensated side lobe. From comparison of patterns it is clear that suppression of the first side lobe is accompanied by certain expansion of the principal lobe and increase of the second side lobe by 2 db. As already noted, by the shown processing it is possible to synthesize a system with suppression of any number of side lobes.

§ 13.3. Statistical Synthesis of Optimum Systems of Resolution

......

We shall now consider the second (of those enumerated in the introduction) approach to the problem of optimum resolution of targets, in which optimum from the point of view of resolving power is considered that system which best executes some function of the radar in the presence of many rather closely located targets.

Let us find general properties of optimum processing of the received signal in multitarget systems without specifying the solutions, taken for the set of targets. This can be done by using the concept of minimum sufficient statistics and a sufficient receiver (see § 3.7, Vol. I).

If there is no a priori information about the number and position of the targets, minimum sufficient statistics for any solutions taken for a set of targets is the set of values of likelihood ratios for all possible situations $\Lambda_1(y; \lambda_1)$, $\ell_p(y; \lambda_1, \lambda_2)$, ..., $\Lambda_n(y; \lambda_1, \ldots, \lambda_n)$, ..., where n (number of targets) has values from 1 to ∞ , and parameters of targets λ_1 , ..., λ_n pass through all possible a priori values. In accordance with this, sufficient is a receiver at whose output there will be formed the shown set. Having output signals of a sufficient receiver, it is possible to optimally find any solutions about the set of targets. The form of further transformations of these signals and what part of them is used depend on the presence of a priori information, the character of the solution and the rules utilized for solution.

Each of the likelihood ratios Λ_n is the ratio of the probability of reception of realization y in the presence of n signals with parameters $\lambda_1, \ldots, \lambda_n$ to the probability of reception of the same realization without the signals. Instead of the likelihood ratios it is possible to consider the likelihood function; however in this case there arise certain formal difficulties connected with transition from discrete realizations to continuous.

Let us consider operations connected with forming of the likelihood ratio for rapid and slow fluctuations of the reflected signal, considering signals and interference statistically independent Gaussian random processes. Here, in order to have the possibility of comparing results of synthesis with results of optimum separation of signals, we will generalize expressions for likelihood ratios obtained in Chapters I and IV Vol. I to the multi-dimensional case.

-759-

Consideration of the multi-dimensional case is produced exactly as for the one-dimensional. We assume that in multi-dimensional domain S_0 there are assigned interference with correlation function $R(s_1, s_2)$ and a signal with correlation function $R(c_1, s_2)$, where both the signal and interference are Gaussian random processes. We consider the set of points s_1, \ldots, s_N . The likelihood ratio for values $y_j = y(s_j)$ is equal to

$$\Lambda(y_{1},...,y_{N}) = \frac{\exp\left\{\frac{1}{2}\sum_{j,k=1}^{N} (W_{jk} - W_{on jk}) y_{j}y_{k}\right\}}{\left|\delta_{jk} + \sum_{l=1}^{N} W_{jl}R_{o lk}\right|^{\frac{1}{2}}}, \qquad (13.3.1)$$

where $R_{jk} = R(s_j, s_k)$; $R_{C jk} = R_C(s_j, s_k)$, and W_{jk} and $W_{C\Pi jk}$ - elements of matrices, the reciprocals of correlation matrices of the interference and the signal with interference, respectively.

Passing to processing as $N \rightarrow \infty$ and bringing to zero the distances between points, it is possible to replace the sum by integrals and, using multi-dimensional analogs of formulas (1.4.1), (1.4.2), (1.4.12), (1.3.15), (4.2.4) and (4.2.5), write the likelihood ratio for the realization in the form

$$A(y) = \exp\left\{\frac{1}{2} \iint_{S_{0}} V(s_{1}, s_{2}) y(s_{1}) y(s_{2}) ds_{1} ds_{2} - \frac{1}{2} \int_{0}^{1} da \int_{S_{0}} B(s, s, a) ds\right\},$$
(13.3.2)

where $V(s_1, s_2)$ and $B(s_1, s_2, \alpha)$ are determined by equations

$$\int_{S_{1}} V(s_{1}, s) [R(s, s_{2}) + R_{c}(s, s_{2})] ds = W_{1}(s_{1}, s_{2}), \qquad (13.3.3)$$

$$B(s_1, s_2; a) + a \int B(s_1, s; a) W_1(s_1, s_2) ds = W_1(s_1, s_2), \qquad (13.3.4)$$

$$W_1(s_1, s_2) = \int_{s_1} W(s_1, s) R_0(s, s_2) ds,$$
 (13.3.5)

$$\int_{S_{1}} W(s_{1}, s) R(s, s_{1}) ds = \delta(s_{1} - s_{2}).$$
(13.3.6)

Variables s in these formulas are multi-dimensional vectors.

We concretize the obtained relationships for the case when the signal is the

sum of n statistically independent signals with correlation functions of form (13.2.1). The radius of correlation of fluctuations of signals is considered large as compared to dimensions of domain ${\rm S}_\Omega$ (case of slow fluciuations). When

$$R_{\mathbf{o}}(s_{1}, s_{\mathbf{a}}) = \sum_{i} E_{i} \operatorname{Re} \varphi(s_{1}, \lambda_{j}) \varphi^{*}(s_{\mathbf{a}}, \lambda_{j}),$$

solutions of equations (13.3.3) and (13.3.4) are presented, as it is simple to prove, in the form

$$V(s_{1}, s_{2}) = 2 \operatorname{Re} \sum_{j, k} v_{jk} X^{*}_{j}(s_{1}) X_{k}(s_{2}), \qquad (13.3.7)$$

References in the state of the second s

「「「「「「「「」」」」

(13.3.3)

$$B(s_1, s_2; a) = 2 \operatorname{Re} \sum_{j, k} b_{jk}(a) \varphi^*(s_1, \lambda_j) X_k(s_2), \qquad (13.3.5)$$

where

$$X_{j}(s_{1}) = \int_{S} W(s_{1}, s) \varphi(s, \lambda_{j}) ds, \qquad (13.3.3)$$

and matrices $\|v_{jk}\|$ and $\|b_{jk}(\alpha)\|$ are reciprocals of matrices

$$\left\|\frac{2}{E_{h}}\delta_{jh}+C(\lambda_{j},\lambda_{h})\right\|, \quad \left\|\frac{2}{E_{h}}+aC(\lambda_{j},\lambda_{h})\right\|$$

respectively, where $C(\lambda_j, \lambda_k)$, as before, is determined by formula (13.2.10). Substituting (13.3.7) and (13.3.8) in (13.3.2), we obtain

$$\Lambda_n(y; \lambda_1, \dots, \lambda_n) = \exp\left\{\sum_{j, k} v_{jk} Q^*_{j} Q_k - - \int_0^1 \sum_{j, k} b_{jk}(u) C(\lambda_k; \lambda_j) du\right\},$$

$$Q_j = \int_0^1 y(s) X_j(s) ds.$$
(13.3.1)

where

In § 1.4 Vol. I it was shown that for an arbitrary matrix we have relationship

$$\frac{\partial \ln |R_{jk}|}{\partial \lambda} = \sum_{j,k} \frac{\partial R_{jk}}{\partial \lambda} W_{kj},$$

where $\lambda = parameter$.

Using this formula and considering properties of matrix $\|b_{jk}\|$, it is possible to rewrite (13.3.10) in the form

-761-

$$\Lambda_{n}(y; \lambda_{1}, \ldots, \lambda_{n}) = \exp \left\{ \sum_{j, k} \sigma_{jk} Q^{*}_{j} Q_{k} - \frac{1}{2} \ln \left[\delta_{jk} + \frac{E_{k}}{2} C(\lambda_{j}, \lambda_{k}) \right] \right\}.$$
(13.3.14)

Let us consider operations connected with forming the logarithm of the likelihood ratio. The first component $\ln \Lambda_n(y; \lambda_1, \dots, \lambda_n)$ can, taking into account (13.3.9) and (13.3.11), be transformed in the following way:

$$L_{n}(y) = \sum_{j,k} v_{jk} Q^{*}_{j} Q_{k} =$$

$$= \sum_{j} Q^{*}_{j} \int_{\mathbf{x}} y(s) \left(\sum_{k} v_{jk} \int_{\mathbf{s}_{*}} W(s, s_{i}) \varphi(s_{i}, \lambda_{k}) ds_{i} \right) ds. \qquad (13.3.7)$$

The function of s in parentheses under the integral coincides [see Paragraph 13.2.1] with the optimum reference signal $\psi(s, \lambda_j)$, ensuring a maximum ratio of the power of the considered signal to the sum of powers of interference and all other signals, considered as disturbing. As noted in § 13.2, for large signal-to-interference ratios for all targets this reference signal coincides with that which ensures total suppression of interfering signals.

Thus, transformations connected with obtaining the likelihood ratio include operations providing partial, and with high energies, practically total separation



Fig. 13.4. Block diagram of optimum processing a signal with slow fluctuations: 1 - multiplier; 2 - integrator over domain S₀;<math>3 - adder. of signals, the hypothesis of whose presence is being proved.* Furthermore, these transformations include multiplication by reference signals $X_j^*(s)$, optimum for separation of a signal from interference, and integration. Results of both forms of correlation processing with identical indices are multiplied and totaled. The corresponding blockdiagram is shown in Fig. 13.4.

^{*}This is the difference from the case considered in §13.2, where there was ensured suppression of signals, the presence of which was not necessarily assumed, but was considered possible.

Let us consider analogously the case of rapid fluctuations, where we consider that fluctuation are fast with respect to one variable - time. In order to separate this coordinate, in our examination we everywhere shall write instead of one variable s two variables s and t. The correlation function of the signal we she, write in this case in the form

$$R_{\mathbf{c}}(s_1, t_1, s_2, t_2) = \operatorname{Re} \sum_{j} E_{j} \varphi(s_1, t_1; \lambda_j) \varphi^*(s_2, t_2; \lambda_j) \varphi_j(t_1 - t_2).$$

We assume also that the spectrum of fluctuations of interference is considerably wider than the spectrum of fluctuations of the signal, and we replace interference by white noise, i.e., we assume

$$R(s_1, t_1, s_2, t_2) = R_1(s_1, s_2) \delta(t_1 - t_2),$$

and that $\varphi(s, t; \lambda_j)$ is a periodic function of time for all j, where we shall consider the magnitude of the period T_r , common for all λ_j , small as compared to the time of correlation of fluctuations of the signal. Solutions of equations (13.3.3) and (13.3.4), as before, we shall seek in the form of (13.3.7) and (13.3.8), but v_{jk} and b_{jk} we shall consider to depend on t_1 , t_2 . All these assumptions and the subsequent derivation are completely analogous to those used in Chapter IV Vol. I in examining signals depending only on time.

Substituting (13.3.7) in (13.3.3) and leveling term by term components in the right and left parts of the equation, with the given assumptions we obtain

$$v_{jk}(t_1, t_2) + \frac{E_k}{2} \sum_l \int_0^T v_{jl}(t_1, t) C(\lambda_l, \lambda_k) p_k(t - t_2) dt =$$

$$= \frac{E_k}{2} p_k(t_1 - t_2), \qquad (13.3.14)$$

where

$$C(\lambda_{l}, \lambda_{h}) =$$

$$= \frac{1}{7r} \int_{0}^{T_{r}} dt \int_{S_{0}} \int \varphi(s_{1}, t, \lambda_{l}) W(s_{1}, s_{2}) \varphi^{*}(s_{1}, t, \lambda_{h}) ds_{1} ds_{2}.$$

$$(.3.3.15)$$

Informuch as it is considered that $\Delta f = \int_{\mathbb{T}} \mathbb{T} \gg 1$ ($\Delta f = \int_{\mathbb{T}} - effective$ width of the spectrum of fluctuations) for all j, it is possible, disregarding fringe effects, to solve equation (13.3.14) by Fourier transformation. The matrix of spectral $V_{jk}(\alpha)$ is the reciprocal of matrix $\|\frac{2\Delta f}{\mathbb{E}_k S_k}(\alpha) + C(\lambda_j, \lambda_k)\|$, where $S_k(\alpha) - \text{spectral}$

density, corresponding to $\rho_k(t)$, normalized so that max $S_k(\omega) = 1$. Likewise we find $B_{jk}(\alpha, \omega)$.

Substituting results of reverse Fourier transformation from $V_{jk}(\omega)$ and $B_{jk}(\alpha, \omega)$ in (13.3.7), (13.3.8), and (13.3.2), we obtain

$$\Lambda(y, \lambda_{j}, \ldots, \lambda_{n}) = \exp\left\{\frac{1}{2\pi} \int_{-\infty}^{\infty} \left[\sum_{j,k} V_{jk}(\omega) Q^{*}_{j}(\omega) Q_{h}(\omega) - T \ln\left|\delta_{jk} + \frac{E_{k}S_{k}(\omega)}{2\Delta f_{ck}} C(\lambda_{j}, \lambda_{k})\right|\right] d\omega\right\},$$
(13.3.16)

where

$$Q_{j}(\omega) = \int_{0}^{t} dt \int_{S_{0}} y(s, t) X_{j}(s, t) e^{-i\omega t} ds. \qquad (13.3...)$$

As can be seen from the obtained formulas, in the case of rapid fluctuations separate spectral components of the signal obtained as a result of correlation processing must be subjected to operations of the same form as for slow fluctuations. Optimum processing is more complicated here than in the case of one target, because $V_{jk}(\omega)$ cannot, in general, be presented in the form of the product of certain frequency responses of filters: $H_{j}^{*}(i_{\tau})$ and $H_{k}(i_{\tau})$.

With a large signal-to-interference ratio we have equality $||V_{jk}(u)||\approx ||w_{jk}||$ in frequency band Δf , where the ratio of spectral densities of the signal and interference is great.

This frequency band is broadened stin increase of the signal-to-interference ratio. By analogy with results obtained for one signal (Chapters IV, VII, IX, X), it is possible to expect that expansion of band Δi with growth of the signal-tointerference ratio, after this band exceeds the width of the spectrum of fluctuations of the signal, and the form of the frequency responses do not substantially affect the performance characteristics of the system. Therefore we replace $V_{jk}(a)$ by $w_{jk}|H(i\omega)|^2$, where $H(i\omega)$ — frequency response of a filter with a passband, not considerably exceeding the width of the spectrum of fluctuations max $\Delta f | c | j$. Here it is possible, by transforming the expression for $L_{jk}(y)$ just as we did in examing one signal, to present a block diagram for processing in the form shown in Fig. 13.5.

Above we considered optimum transformations of a signal in a sufficient receiver, connected with forming the likelihood ratio for any assumed set of targets. If on the number and position of targets there are imposed limitations, optimum processing is substantially simplified. The simplest example of this kind

-71,4-



Fig. 13.5. Block diagram of quasi-optimum processing of a signal with fast fluctuations: 1 - multiplier; 2 - integrator over domain S₀; 3 - filter with frequency response H(i ω); 4 - adder; 5 - integrator during the time of observation.

is the case when parameters of n - 1 targets are known, and a solution is found only for the n-th target. The likelihood ratio for this problem equals ratio

$$\Lambda_n(y, \lambda_1, \ldots, \lambda_{n-1}, \lambda_n) / \Lambda_{n-1}(y, \lambda_1, \ldots, \lambda_{n-1}) = \Lambda(y).$$

Using the property (13.2.13) of inverse matrices, transformations of the signal connected with forming $\ln \Lambda(y)$ can be written in the form (with slow fluctuations)

$$L(y) = \frac{1}{v_{nn}} \left| \sum_{j=1}^{n} v_{nj} Q_j \right|^2.$$
(13.3.18)

In this case transformations of the signal ensure singling out of the useful signal with partial (and for large signal-to-interference ratios - total) suppression of interfering signals. Solution of the problem of detection of the n-th target against a background of n - 1 targets and interferences was conducted in [43] for finding potential possibilities of increasing resolving power.

Besides the mentioned work [43] there are a number of works in which there is posed and solved the problem of statistical synthesis of optimum resolution systems. In [44] by optimum resolution there is understood detection of a set of targets, producible by means of comparison of results of estimation for maximum likelyhood of amplitudes of signals from these targets with a threshold. We consider the case of regular signals and white noise. Operations connected with estimating amplitudes coincide with those obtained in the problem of complete separation of signals.

In [64] there is offered another formulation of the problem, leading, however, under certain conditions to the same results. Considered best in resolving power

-765-

is the system providing the best estimate of the reflecting surface (level of reflected signal) as a function of the coordinates. As the loss function there is used*

$$I(\hat{\mathbf{\varphi}}, \mathbf{\varphi}) = \int_{\mathbf{S}} \left| \hat{\mathbf{\varphi}}(s) - \mathbf{\varphi}(s) \right|^2 ds, \qquad (13.3.19)$$

아파마트 걸릴 데지도 한다. 그는

where $\varphi(s) - signal$, corresponding to the true relief of reflecting surface; $\hat{\varphi}(s) - signal$, corresponding to the estimated relief.

Minimization of mean risk is provided by selection of $\hat{\phi}(s)$ at which $I(\phi, \phi)$ is minimum, where the dash here signifies averaging with the help of the a posteriori distribution of $p(\phi|y)$. It is assumed that the signal $\sigma(s)$ is the surferposition of signals from n point targets

$$\boldsymbol{\varphi}(\boldsymbol{s}) = \operatorname{Re} \sum_{j} A_{j} \boldsymbol{\varphi}(\boldsymbol{s}, \lambda_{j}),$$

where A_i - amplitude factors (signal $\varphi(s)$ is considered regular).

Here the problem reduces to finding \hat{A}_j , $\hat{\lambda}_j$ (j = 4, ..., n). Substituting (13.3.19) with $A_j = \hat{A}_j$, $\lambda_j = \hat{\lambda}_j$ in the formula for $\overline{I(\phi, \phi)}$, differentiating with respect to A_j and equating the derivative to zero, we have

$$\hat{A}^{\bullet}_{j} = 2 \sum_{l=1}^{n} B_{ll}^{-1} \Phi_{l}, \qquad (13.3.20)$$

where B_{jl}^{-1} - element of a matrix, the inverse of $||B_{jl}||$:

$$B_{jl} = \int_{S_0} \varphi(s, \lambda_j) \varphi^*(s, \lambda_l) ds, \qquad (13.3.21)$$

$$\Phi_{j} = \int_{S} \varphi(s, \lambda_{j}) \overline{\varphi(s)} \, ds. \qquad (13.3.22)$$

Substituting (13.3.20) in the original formula for $I(\phi, \phi)$, we find that $\hat{\lambda}_j (j = 1, ..., n)$ must be selected from condition

$$\max_{\hat{\lambda}_{j}} \sum_{j,k} B_{jk}^{-1} \Psi^{*}_{j} \Psi_{k}.$$
(13.3.23)

If there is no a priori information about the position of the targets, $\overline{\varphi(s)} = y(s)$. Here the considered method of estimating the relief of the reflecting

*Here we do not hold to the designations adopted in the mentioned article.

surface coincides with the classical method of least squares. It is easy to see that transformation of (13.3.20) ensures total suppression of all signals, with the exception of that one whose amplitude is being estimated. It is not difficult to prove that this processing coincides with that considered in § 13.2, when $R(s_1, s_2) = R_0 b(s_1 - s_2)$, i.e., when interference constitutes "white noise" with respect to all measurements.

Thus, the conducted consideration of transformations of the signal in multitarget systems which are optimum from different points of view showed that these transformations always include total or partial separation of signals from targets whose presence is assumed. In certain variants of formulation of the problem crossprocessing of signals is exhausted by these operations of separation and further transformations of the separated signals are such as if each of the signals was unique (see Chapter IV Vol. 1). However in many cases this is not so, and for them it is interesting to compare characteristics of performance of an optimum system and a system in which separated signals are processed independently.

The fact is that practical use of optimum systems, in the which signal is processed in accordance with formulas (13.3.10) and (13.3.16), is possible, apparently, only with rather severe limitations (assigned in the form of a priori distribution) on the number of targets and their position. Thus are matters, e.g., during simultaneous tracking of several targets, the number and initial positions of which are sufficiently accurately known, or during detection of a group of a known number of targets, located at assigned points. In the absence of a priori information the optimum system is extraordinarily complicated, and it is advisable to use in this case a system with total suppression of assigned interfering signals and subsequent processing of the same form as for one signal.

In accordance with what has been said it is useful to take the following order of further consideration. In § 13.4 and § 13.5 we shall consider the problem of optimum detection and measurement of coordinates of an assigned set of targets. In § 13.6 we shall consider and compare with optimum ones simplified systems of detection and measurement, using the principle of total separation of signals.

§ 13.4. Optimum Detection of an Assigned Set of Targets

If possible positions of targets are fixed and it is required to determine which of these targets exist in fact, the optimum procedure of solution consists (see Chapter III, Vol I) of comparing likelihood ratios corresponding to all possible

-767-

situations with thresholds and among themselves. If losses connected with all possible errors in determination of the number and position of targets are identical, we should take such a solution in which the likelihood ratio has a maximum magnitude, exceeding the noise threshold. For instance, if the number of targets is not greater than two, it is necessary to compare with the threshold and among themselves

$\Lambda_{\mathbf{1}}(y, \lambda_{\mathbf{1}}); \quad \Lambda_{\mathbf{1}}(y, \lambda_{\mathbf{2}}); \quad \Lambda_{\mathbf{3}}(y, \lambda_{\mathbf{1}}, \lambda_{\mathbf{3}}).$

In order to completely characterize the procedure of finding a solution in this case it is necessary to calculate the probabilities of all possible errors, the number of which, obviously, equals $\frac{n(n-1)}{2}$, where n is the number of possible solutions. For calculation of these probabilities it is necessary to find the joint distribution of probabilities for likelihood ratios. All use calculations turn out to be very cumbersome. Therefore we here will limit our consideration to the case of two targets. Let us calculate first the characteristics of detection of one of the targets, when the presence of the second target is exactly established. It is clear that this characteristic has the greatest bearing on the problem of resolution of closely located targets in conditions of detection. Absence of resolution will lead to detection of one target instead of two; therefore the quality of resolution can be estimated primarily by this, characteristic.

Furthermore, we shall be interested in characteristics of detection of both targets when they are either present or closent simultaneously. Here we shall consider fluctuation of the signal slow.

In the first case the solution is found on the basis of comparision of quantity $\left|\sum_{j=1}^{2} v_{j} Q_{j}\right|^{2}$ and, inasmuch as Q_{j} is distributed by normal law, the equation of characteristics of detection has the form (see Chapter IV Vol. I)

$$D=F^{\frac{1}{1+q_0}},$$

where q_0 - the signal-to-interference ratio at the output of the system forming $\sum v_{sj}Q_j$.

This ratio already was considered in § 13.2 in connection with the problem of separation of signals. Remember that the optimum system of detection in this case is a system of partial separation of signals, supplemented with a device comparing the square of the modulus of the output signal with the threshold.

-768-

Let us consider the case of a pair of simultaneously appearing targets. The solution is found on the basis of comparing with a threshold the likelihood ratio $\Lambda_2(y; \lambda_1, \lambda_2)$ or quantity

$$L_{a}(y) = \sum_{j,k=1}^{2} \sigma_{jk} Q^{*}_{j} Q_{k},$$

Formed by the block diagram of Fig. 13.2. Writing the distribution for quantities Q_1 , Q_2 in the form (see [19], and also formula (4.11.2) in Volume I)

$$p(Q_1, Q_2) = \frac{1}{\pi^2 |R_{jk}|} \exp \left\{ -\sum_{j,k=1}^2 W_{jk} f_j f^*_k \right\},\,$$

where $\|W_{jk}\|$ - matrix, the inverse of R_{jk} , and

$$R_{i_k} = \overline{Q_j Q^*_k} = C(\lambda_j, \lambda_k) + \sum_{l=1}^2 \frac{E_l}{2} C(\lambda_j, \lambda_l) C(\lambda_l, \lambda_k), \qquad (13.4.1)$$

it is easy to calculate the characteristic function of quantity $L_p(y)$:

$$\psi(\eta) = \left| \delta_{jk} - i\eta \sum_{l} \sigma_{ll} R_{lk} \right|^{-1}.$$
 (13.4.2)

Expanding the characteristic function to simple fractions, it is easy to find the corresponding density of distribution. Assuming for simplicity that the strengths of the signals are identical, we obtain for the probabilities of correct detection and false alarm the following expressions:

$$D(c) = \frac{1+\gamma}{2\gamma} e^{-\frac{c}{1+\gamma}} - \frac{1-\gamma}{2\gamma} e^{-\frac{c}{1-\gamma}}, \qquad (13.4.3)$$

$$F(c) = \frac{1+\gamma+q'_{\bullet}(1-\gamma^{3})}{2\gamma} \exp\left[-\frac{(1+q'_{\bullet})^{2}-q'_{0}^{2}\gamma^{3}}{1+\gamma+q'_{\bullet}(1-\gamma^{3})}c\right] - \frac{1-\gamma+q'_{\bullet}(1-\gamma^{3})}{2\gamma} \exp\left[-\frac{(1+q_{\bullet})^{2}-q_{0}^{2}\gamma^{2}}{1-\gamma+q'_{\bullet}(1-\gamma^{3})}c\right], \qquad (13.4.3)$$

where c - is the magnitude of the threshold, divided by q_0^{\dagger} ; with which we compare L_2 , $\gamma = |C(\lambda_1, \lambda_2)| / \sqrt{C(\lambda_1, \lambda_2)C(\lambda_2, \lambda_2)};$

 q'_{o} - the signal-to-interference ratio in the absence of separation.

In order to simplify calculations connected with determination of threshold ratio $q_0^{'}$, corresponding to the selected probabilities F and D, it is possible to use the fast that, as one may see from (13.4.3) and (13.4.4),

$$F(c, \gamma, q'_{\bullet}) = D(c_{1}, \gamma_{1}),$$
 (13.4.5)

where

$$\mathbf{Y}_{1} = \frac{\mathbf{Y}}{1 + q'_{\bullet}(1 - \mathbf{Y}^{2})}; \quad \mathbf{c}_{1} = \mathbf{c} \frac{(1 + q'_{\bullet})^{2} - q'_{0}^{2} \mathbf{Y}^{2}}{1 + q'_{\bullet}(1 - \mathbf{Y}^{2})}. \quad (13.4.6)$$

If $\mathbf{F} \ll 1$ and D > 0.5, the threshold values of q_0 turn out to be so large that $\gamma_1 \approx 0$ when $\gamma^2 < 0.8$. One can prove this by performing the necessary numerical



Fig. 13.6. Dependence of the threshold signal-to-interference ratio on the degree of nonorthogonality of the detected signals during optimum processing.

calculations. This circumstance essentially simplifies calculation of dependence $q_0(F, D, \gamma)$, which can be produced by the following scheme. First we graphically find dependence $c(D, \gamma)$, where D's are given very small values, typical for the probability of false alarm. Incidentally, with these values we can disregard the second component in (13.4.3), if $\gamma^2 > 0.2$. The q_0' is determined from the second relationship of (13.4.6)

$$q_{0} = \frac{1}{1 - \gamma^{2}} \left\{ \frac{c(F, 0)}{2c(D, \gamma)} (1 - \gamma^{2}) - 1 + \frac{1}{\left[\frac{c(F, 0)}{2c(D, \gamma)} (1 - \gamma^{2})\right]^{2} + \gamma^{2}} \right\}.$$
(13.4.7)

Relationship $q_0'(D, \gamma)$ when $\gamma = 10^{-6}$ is shown in Fig. 13.6. As can be seen from the figure, for D's not very close to unity q_0' weakly depends on parameter γ , slowly increasing with increase of the latter. When $1 - D \ll 1$ this growth is more marked. In § 13.6 the obtained result will be used for comparison of the optimum system and a system with total separation of signals.

§ 13.5. Synthesis of Systems of Resolution In Conditions of Measurement of Coordinates

13.5.1. Formulation of the Problem

During measurement by radar of coordinates of closely located unresolved targets it turns out that due to interaction of signals reflected from these targets the mean values of the measured quantities do not coincide with their true values. With total absence of resolution the radar measures the coordinate of every target. At the same time if for an assigned sounding signal we introduce resolution of

-770-

targets with the help of the proper processing of signals in the receiver, in g_{ℓ} neral, the ratio of the signal to noise decreases and fluctuating error increases.

In connection with the shown circumstances it is useful to consider as the criterion of optimality of a resolution system during measurement of coordinates of targets absence of systematic errors of measurement of these coordinates and a minimum of fluctuating errors. The latter, as also earlier, can be estimated by their variances at each given moment of time.

If we apply this criterion for synthesis of the system of measurement of coordinates as a whole, then, as also in the case of one target, with observance of certain not too limiting conditions the system can be divided into two parts. The first of them is the discriminator in the tracking variant or the estimator unit in the nontracking variant of the meter, i.e., a system for processing the r-f signal, producing the current estimate of the measured parameter (set of parameters) λ . The second part of the system are the smoothing circuits.*

In the case of fast fluctuations of the signals the smoothing circuits have less bearing on questions of resolution. At the same time their synthesis in the presence of many targets and many parameters varying randomly in time, subject to measurement, constitutes a very complicated problem. In the present book we will not deal with this problem; we shall limit ourselves to synthesis of the device in which there is concentrated the radio part of the system. Not making any distinctions between tracking and nontracking variants of the system, we shall call this device a discriminator. General properties of optimum discriminators, corresponding to multitarget problems, will be formulated below.

Preliminarily let us note that in accordance with the formulated criterion the optimum discriminator should exceute such operations on the received signal which correspond to formation of the current efficient estimate of the measured quantity x, since the concept of efficiency includes the absence of bias and a minimum variance of the estimate. Practically, considering the approximate nature of fulfillment of optimum operations, it is sufficient to require asymptotic efficiency of the estimate.

dere it is possible to pose two problems. The first, and the simpler of them, arises when to measurement without systematic error with minimum fluctuating error

-771-

^{*}I' should be noted that conditions under which an optimum mater indeed even of divided into the shown parts for multitarget problems is far from clear.

We subject one parameter λ_1 , the carrier of which is signal $y_1(t, \lambda_1)$, but here there exist other signals $y_p(t, \lambda_p)$, (p = 2, ..., n), not orthogonal to $y_1(t)$, parameters of which λ_p are known. For solution of this problem it is necessary to find the efficient (at least asymptotically) estimate of parameter λ_1 for known λ_p . Properties of an optimum discriminator are determined by the known formula for spectral density, corresponding to variance of the efficient estimate

$$\mathbf{S}_{0\pi\tau_1} = \sigma_{\mathbf{s}\phi_1}^2 T = -T \frac{1}{\frac{\partial^4 L(\lambda_1)}{\partial \lambda_1^2}}, \qquad (13.5.1)$$

where $T = time during which parameter <math>\lambda_1$ can be considered constant;

 $L(\lambda_4)$ - logarithm of the likelihood function.

A second, more complicated problem corresponds to the case when n signals $y_p(t, \lambda_p)$ (p = 1, 2, ..., n) depend on n unknown parameters λ_p . The problem does not change whether it is required to measure with zero systematic and minimum illuctuating errors all n parameters or one of them (for instance, λ_q). In the last case it is necessary all the same to measure all n parameters for optimum rundization of compensation for extraneous signals.

For solution of this problem is is necessary to seek an asymptotically jointlyefficient estimate of parameters (f) and croperfies of the optimum system obtained as a result of the solution of the problem are determined by variances and crosscorrelation moments J_{pq} of the estimates:

$$\|J_{pq}\| = \left\| - \frac{\partial^2 L(\lambda_1, \ldots, \lambda_n)}{\partial \lambda_p \partial \lambda_q} \right\|^{-1}, \qquad (13.5.8)$$

where $L(\lambda_1, \ldots, \lambda_n) = \log arithm$ of the likelihood function of the parameters.

An ellipsoid with moments of inertia J_{pq} is the minimum among possible ones during measurement of parameters $\lambda_1, \ldots, \lambda_n$ of ellipsoids of dispersion [8, 66]. Questions connected with this problem will be considered in greater detail later.

There exist no direct methods of finding efficient or jointly-efficient estimates. At the same time we know that during measurements of one-dimensional parameters the comparatively easily found maximum likelihood estimates have the property of asymptotic efficiency with small times of correlation of the observed signal [4, 13]. There is every basis to assume that these estimates in the case of measurements of multi-dimensional parameters, too, will possess the property of joint asymptotic

-772-

efficiency.

This assumption we shall use as the basis for our further considerations. Here we shall find the maximum likelihood estimates of coordinates of many arbitrarily (frequently closely) located targets and demonstrate their joint asymptotic efficiency for a rapidly fluctuating normal signal.* The practical importance and applicability of this case was repeatedly proven earlier. It should be noted that solution of the first problem (only parameter λ_1 is unknown) will follow as a particular case from solution of the second problem.

For clarification of general properties of optimum discriminators we shall turn first of all to maximum likelihood estimates of multi-dimensional parameters. Then on this base we shall consider particular cases which it is possible to investigate completely and obtain in explicit form circuits and properties of optimum resolution systems. These are cases of n targets with a very large signal-to-noise ratio h and of two targets (n = 2) for any h.

13.5.2. Maximum Likelihood Estimates of Multi-Dimensional Parameters and Their Asymptotic Efficiency. General Properties of an Optimum Discriminator

Let us assume that in field of sight of the radar there are n targets. Then the signal received by the radar can be recorded in the form

$$y(t) = \sum_{p=1}^{n} y_p(t, \lambda_p) + n(t), \qquad (13.5.3)$$

where

n(t) - normal white noise with spectral density N_{O} ;

 $y_p(t,\,\lambda_p)$ — normal random process with a correlation function depending on the unknown parameter λ_p

$$R_{\mu}(t_{1}, t_{2}) = \operatorname{Re} P_{\mu} \mu_{\mu}(t_{1}, \lambda_{\mu}) \mu^{*}_{\mu}(t_{2}, \lambda_{\mu}) \rho_{\mu}(t_{1} - t_{2}) \times (17.5.1)$$

$$\times e^{i \omega_{\mu}(t_{1} - t_{2})}.$$

Here

P_p - mean power;

 $\rho_{\rm p}(t)$ - coefficient of correlation of fluctuations;

 $u_p(t, \lambda_p)$ - complex law of modulation of the p-th signal:

[&]quot;Unfortunately, with a slowly fluctuating signal asymptotic efficiency of joint maximum likelihood estimates utilized by us does not take place. As a result, withough the structure of the multitarget measuring system itself is found without special difficulties, the question of its optimality remains open. Therefore here we consider only the case of rapid fluctuations of signals reflected from the targets.
$u_p(t, \lambda_p) = u_{ap}(t, \lambda_p) e^{i\phi_p(t, \lambda_p)}.$

Index p in these expressions allows for the possible depertor λ_p) and $u_p(t, \lambda_p)$ on other parameters besides λ_p . We shall conside λ_p) are periodic (with period T_r) functions and that the interval of construction of fluctuations τ_{pk} [effective width of function $\rho_p(t)$] is much larger than period T_r . At the same time we shall consider this interval much smaller than time T, during which parameter λ_p is constant:

$$T_r \ll \tau_{pk} \ll T. \tag{13.5.}$$

Additionally we assume that different reflected signals are statistically independent; therefore functions of their cross-correlation are:

$$\rho_{pq}(t) = P_{p} \delta_{pq} \rho_{p}(t),$$

where δ_{pq} - Kronecker delta.

The formulated assumptions are normal and natural. A new element here is the condition of independence of fluctuations of different targets, which is practically always realized.

We introduce designation

$$\frac{1}{T_r} \int_{0}^{T_r} u^*_{\nu}(t, \lambda_{\nu}) \, \iota_q(t, \lambda_q) \, dt = C_{\nu q}(\lambda_{\nu}, \lambda_q), \qquad (13.5.6)$$

where $C_{pq}(\lambda_p, \lambda_q)$ - correlation functions of modulations of signals reflected from different targets. These functions with certain small modifications, determined by the essence of the question in connection with which they are introduced into consideration, appeared earlier. Thus, for instance, in Chapter I Vol. I we introduced functions $C_0(\tau, \Omega)$, which are a particular case of functions (13.5.6) when signals from different targets differ in delays and Doppler frequency shifts. Correlation functions of signals were introduced already in § 13.2 of the present chapter [see (13.2.10)]. But for our further purposes most suitable is definition (13.5.6). Let us note that, as a rule, C_{pp} does not depend on the index p and λ_p . We shall consider functions $u_p(t, \lambda_p)$ normalized so that $C_{pp} = 1$.

Let us turn to calculation of maximum likelihood estimates of parameters λ_p and study of their properties. Here in the first place we must obtain an expression for the likelihood functional of parameters λ_p , which for a more general case was found in § 13.3. We shall repeat briefly calculations of this paragraph in reference to the considered case.

The correlation function of signal y(t) (13.5.3) is equal to

$$\overline{y(t_1)y(t_2)} = R(t_1, t_2) = \operatorname{Re} e^{i\omega_0(t_1-t_2)} \times \sum_{p=1}^{n} u_p(t_1, \lambda_p) u^*_p(t_2, \lambda_p) P_p \rho_p(t_1-t_2) + N_0 \delta(t_1-t_2).$$
(13.5.7)

We seek function $W(t_1, t_2)$ in the form

$$W(t_1, t_2) = \operatorname{Re} e^{i \omega_0 (t_1 - t_2)} \sum_{p,q=1}^n u_p (t, \lambda_p) \times (13.5.3)$$

$$\times u^{\omega_q} (t_2, \lambda_q) v_{pq} (t_1, t_2) + \frac{1}{N_0} \delta (t_1 - t_3),$$
(13.5.3)

where $v_{pq}(t_1, t_2)$ are unknown functions which we consider to vary slowly as compared to $u_p(t, \lambda_p)$.

Substituting expressions (13.5.7) and (13.5.8) in the integral equation for function $W(t_1, t_2)$

$$\int_{0}^{T} W(t_{1}, t_{2}) R(t_{2}, t_{3}) dt_{2} = \delta(t_{1} - t_{3})$$
(13.5.2)

and considering the rapidly-varying nature of functions $u_p(t, \lambda_p)$ as compared to $\rho_p(t)$ and $v_{pq}(t_1, t_2)$, it is possible for functions $v_{pq}(t_1, t_2)$ to obtain the system of integral equations

$$\rho_{\mu q}(t_{1}-t_{s})+N_{\bullet}^{2}v_{\mu q}(t_{1},t_{s})+\frac{N_{\bullet}}{2}\int_{0}^{T}\sum_{s=1}^{n}P_{\mu}\rho_{\mu}(t_{1}-t_{s})C_{\mu s}(\lambda_{\mu},\lambda_{\epsilon})v_{s q}(t_{s},t_{s})dt_{s}=0.$$

In the case of a rapidly fluctuating signal, when the effective duration of peaks of functions $\rho_p(t)$ is considerably less than time T, we may assume that $v_{pq}(t_1, t_3) = v_{pq}(t_1 - t_3)$, and for the Fourier transforms of functions $v_{pq}(t)$ we can obtain system of equations

$$P_{p}S_{p}(\boldsymbol{\omega})\delta_{\mu q} + N_{\bullet}^{2}V_{pq}(\boldsymbol{\omega}) + \frac{N_{\bullet}}{2}\sum_{s=1}^{n}P_{\mu}S_{\mu}(\boldsymbol{\omega})C_{\mu \bullet}(\lambda_{\mu},\lambda_{s})V_{sq}(\boldsymbol{\omega}) = 0, \qquad (13.5.10)$$

where $S_p(\omega)$ and $V_{pq}(\omega)$ are Fourier transforms of functions $\rho_p(\tau)$ and $v_{pq}(\tau)$, respectively.

This system of equations is easily solved. Considering

$$\begin{split} \mathbf{S}(\boldsymbol{\omega}) &= \| P_p \mathbf{S}_p(\boldsymbol{\omega}) \, \delta_{pq} \|, \quad \mathbf{V}(\boldsymbol{\omega}) = \| \mathbf{V}_{pq}(\boldsymbol{\omega}) \|, \\ \mathbf{E} &= \| \boldsymbol{\delta}_{pq} \|, \quad \mathbf{C} = \| C_{pq}(\lambda_p, \lambda_q) \|, \end{split}$$

we rewrite system (13.5.10) in the form

$$\mathbf{S}(\boldsymbol{\omega}) + N_{\bullet}^{2} \mathbf{V}(\boldsymbol{\omega}) + \frac{N_{0}}{2} \mathbf{S}(\boldsymbol{\omega}) \mathbf{C} \mathbf{V}(\boldsymbol{\omega}) = 0,$$

ires which

$$\mathbf{V}(\boldsymbol{\omega}) = -\frac{1}{N_0^2} \left(\mathbf{E} + \frac{1}{2N_{\bullet}} \mathbf{S}(\boldsymbol{\omega}) \mathbf{C} \right)^{-1} \mathbf{S}(\boldsymbol{\omega}). \qquad (12.1)$$

Thus, basically, we find function $W(t_i, t_2)$. True, matrix inversion in expression (13.5.11) is a very difficult problem in itself.

Knowing function $W(t_1, t_2)$, we can study properties of maximum likelihood estimates of parameters λ_p . The system of maximum likelihood equations has the form

$$\frac{\partial L(\lambda_1, \ldots, \lambda_n)}{\partial \lambda_p} = 0. \tag{13.5.12}$$

Let us designate by $\lambda_1^{(0)}$, ..., $\lambda_n^{(0)}$ the set of values of the parameters which we know are close to the true values of the parameters. In particular, these may be true values of the parameters. Considering that with a softletently large time of observation colutions of equations (13.5.12) will be sufficiently close to $\lambda_n^{(0)}$, ..., $\lambda_n^{(0)}$, we can replace equation (20.5.12) by statistically equivalent equations

$$\frac{\partial L\left(\lambda_{1}^{(0)},\ldots,\lambda_{n}^{(0)}\right)}{\partial\lambda_{p}}+\sum_{q=1}^{n}\frac{\partial^{2}L\left(\lambda_{1}^{(0)},\ldots,\lambda_{n}^{(0)}\right)}{\partial\lambda_{p}\partial\lambda_{q}}\left(\lambda_{q}-\lambda_{q}^{(0)}\right)=0.$$
(13.5.121)

For study of properties of solutions of these equations of paramount importance is matrix

$$\mathbf{A} = \left\| -\frac{\partial^{\mathbf{e}} L\left(\lambda_{1}^{(0)}, \ldots, \lambda_{n}^{(0)}\right)}{\partial^{\lambda} e^{\partial \lambda_{p}}} \right\|.$$

We shall consider $\lambda_1^{(0)}$, ..., $\lambda_n^{(0)}$ true values of the parameters. We find the mean value and variance of this matrix (understanding by mean value of a matrix a matrix composed of mean values of the elements, and by variance of a matrix a matrix

-776-

composed of variances of elements). We note here that matrix

a second provide the second second

$$\bar{\mathbf{A}} = \mathbf{J}^{-1},$$
 (13.5.13)

where 3 is the matrix of variances and cross-correlations of jointly-efficient (13.5.2)

Generalizing slightly the results of Chapter I (Vol. 1 of this book), we can write the following expression

$$\frac{\partial L(\lambda_1, \dots, \lambda_n)}{\partial \lambda_p} = -\frac{1}{2} \int_0^T \int_0^T \frac{\partial R(t_1, t_2)}{\partial \lambda_p} W(t_1, t_2) dt_1 dt_2 - -\frac{1}{2} \int_0^T \int_0^T \frac{\partial W(t_1, t_2)}{\partial \lambda_p} y(t_1) y(t_2) dt_1 dt_2.$$
(13.5.14)

· · · · ·

From this, after elementary transformations, we obtain

$$\frac{\overline{\partial^2 L\left(\lambda_1^{(0)},\ldots,\lambda_n^{(0)}\right)}}{\partial \lambda_p \partial \lambda_q} = -\frac{1}{2} \int_0^T \int_0^T \frac{\partial W\left(l_1,l_2\right)}{\partial \lambda_q} \frac{\partial R\left(l_1,l_2\right)}{\partial \lambda_p} dt_1 dt_2. \qquad (13.5.14^{+})$$

Substituting in (13.5.14) expressions for R(t $_1,$ t $_2)$ and W(t $_1,$ t $_2)$ and introducing decignations

$$C_{pq}^{(10)} = C_{qp}^{(01)*} = \frac{1}{T_r} \int_0^r \frac{\partial u^*_p (t, \lambda_F^{(0)})}{\partial \lambda_p} u_q (t, \lambda_q^{(0)}) dt,$$

$$C_{pq}^{(11)} = \frac{1}{T_r} \int_0^r \frac{\partial u^*_p (t, \lambda_p^{(0)})}{\partial \lambda_p} \cdot \frac{\partial u_q (t, \lambda_q^{(0)})}{\partial \lambda_q} dt,$$
(27.10.15)

it is easy to obtain the following:

$$\frac{\partial^{2}L(\lambda_{1}^{(0)},\ldots,\lambda_{n}^{(0)})}{\partial\lambda_{p}\partial\lambda_{q}} = \operatorname{Re}\left\{\sum_{m=1}^{n} \left(C_{qp}^{(11)}C_{pm} + C_{pm}^{(10)}C_{qp}^{(10)}\right) \times \int_{0}^{T}\int_{0}^{T}\rho_{p}\left(t_{1}-t_{2}\right)\upsilon_{mq}\left(t_{1}-t_{2}\right)dt_{1}dt_{2} + \right. \\ \left. + \sum_{n,m=1}^{n}C_{pm}^{(10)}C_{mp} \times \int_{0}^{T}\int_{0}^{T}\rho_{p}\left(t_{1}-t_{2}\right)\frac{\partial\upsilon_{nm}\left(t_{1}-t_{2}\right)}{\partial\lambda_{q}}dt_{1}dt_{2}\right\} \approx T\operatorname{Re}\left\{\sum_{m=1}^{n}\left(C_{qp}^{(11)}C_{1m} + \right. \\ \left. + C_{pm}^{(10)}C_{qp}^{(10)}\right)\int_{-\infty}^{\infty}\rho_{p}\left(t\right)\upsilon_{mq}\left(t\right)dt + \sum_{n,m=1}^{n}C_{pn}^{(10)}C_{mp}\int_{-\infty}^{\infty}\rho_{p}\left(t\right)\frac{\partial\upsilon_{nm}\left(t\right)}{\partial\lambda_{q}}dt\right\}. \quad (1 \leq \dots, 1 \nu)$$

Here we used, first, the rapidly-varying nature of functions $u_{p}(t, t_{p})$ as compared

to $\rho_p(t)$ and $v_{pq}(t)$, which permitted us under signs of integrals to average expressigns with functions $u_{p}(t, \lambda_{p})$; second, we used the fact the functions $v_{pq}(t_{1} - t_{2})$ providedly immediately turn into 0 when v_1 goes beyond the limits of interval (0, T), which permitted us to replace integrals over interval (0, \mathbb{T}) from functions $v_{pq}(v_{q} - v_{pq})$ - v_{μ}) by integrals over interval (- α , ∞). Very significant is asymptotic dependence $\int_{\mathcal{O}_{i}}^{\mathcal{O}_{i}} \frac{\partial^{2}L\left(\lambda_{1}^{(0)},\ldots,\lambda_{n}^{(0)}\right)}{\partial\lambda_{n}\partial\lambda_{n}}$

$$\frac{\partial^{a}L\left(\lambda_{1}^{(0)},\ldots,\lambda_{n}^{(0)}\right)}{\partial^{\lambda}\varphi^{d\lambda}q}\sim T.$$

For finding of dispersion $D\left[\frac{\partial^{2}L(\lambda_{1}^{(0)},\ldots,\lambda_{n}^{(0)})}{\partial\lambda_{n}\partial\lambda_{n}}\right]$ we use relationship

$$D\left[\frac{\partial^2 L\left(\lambda_1^{(0)},\ldots,\lambda_n^{(0)}\right)}{\partial\lambda_p\partial\lambda_q}\right] = \frac{1}{2}\int_0^T \int_0^T \int_0^T \int_0^T \frac{\partial^2 W\left(t_1,t_2\right)}{\partial\lambda_p\partial\lambda_q} \frac{\partial^2 W\left(t_2,t_3\right)}{\partial\lambda_p\partial\lambda_q} \times R\left(t_1,t_3\right) R\left(t_2,t_4\right) dt_1 dt_2 dt_3 dt_4$$

(a relationship of this type we derived and used in Chapter X of this book). Exact calculation of this integral is rother cumbersone. For our problem exact knowledge of the magnitude of $D\left[\frac{\partial^3 L\left(\lambda_1^{(0)}, \dots, \lambda_n^{(0)}\right)}{\partial \lambda_* \partial \lambda_*}\right]$ is not required; it is necessary only to know the asymptotic dependence of this magnitude on T. It is easy to see that the succelated integral is a sum with finite coefficients of integrals of type

$$\int_{0}^{T} \int_{0}^{T} \int_{0}^{T} v_{kl} (t_{1} - t_{2}) v_{mil} (t_{2} - t_{4}) v_{r_{2}} (t_{1} - t_{2}) \rho_{tu} (t_{2} - t_{4}) dt_{4} \times dt_{2} dt_{3} dt_{4}$$

$$\int_{0}^{T} \int_{0}^{T} v_{kl} (t_{2} - t_{2}) v_{mil} (t_{3} - t_{2}) \rho_{r_{0}} (t_{1} - t_{3}) dt_{4} dt_{3} dt_{3} dt_{4}$$

$$\int_{0}^{T} \int_{0}^{T} v_{kl} (t_{1} - t_{2}) v_{mil} (t_{3} - t_{3}) dt_{4} dt_{3} dt_{4}$$

and their derivatives with respect to $\lambda_{\mathbf{p}}$ and $\lambda_{\mathbf{q}}$ up to and including the second order. coefficients of these integrals are time-averaged values of expressions containing rapioly varying functions $u_p(t, \lambda_p)$. Further it is possible to note that we have equal*t1es

$$\int_{0}^{T} \int_{0}^{T} \int_{0}^{T} \int_{0}^{T} v_{kl} (t_{1} - t_{2}) v_{min} (t_{4} - t_{4}) p_{\tau 2} (t_{4} - t_{2}) p_{tu} (t_{v} - t_{q}) \times$$

$$\times dt_{1} dt_{3} dt_{4} t_{4} \approx T \int_{-\infty}^{\infty} \int_{-\infty}^{\infty} \int_{-\infty}^{\infty} v_{kl} (t_{4}) v_{mn} (t_{3}) p_{r 3} (t_{4} - t_{5} - t_{3}) \times$$

$$\times \rho_{tu} (t_{9}) dt_{4} dt_{3} dt_{4} ,$$

$$\int_{0}^{T} \int_{0}^{T} \int_{0}^{T} v_{kl} (t_1 - t_2) v_{mn} (t_2 - t_3) \rho_{rs} (t_1 - t_3) dt_1 dt_2 dt_3 \approx$$

$$\approx T \int_{-\infty}^{\infty} \int_{-\infty}^{\infty} v_{kl} (t_1) v_{mn} (t_2) \rho_{rs} (t_1 - t_3) dt_1 dt_3,$$

$$\int_{0}^{T} \int_{0}^{T} v_{kl} (t_1 - t_2) v_{mn} (t_1 - t_2) dt_1 dt_3 \approx T \int_{-\infty}^{\infty} v_{kl} (t) v_{mn} (t) dt$$

From this it follows that

$$D\left[\frac{\frac{\partial^{2}L(\lambda_{1}^{(0)},\ldots,\lambda_{n}^{(0)})}{\partial\lambda_{p}\partial\lambda_{q}}\right] \sim T,$$

$$\frac{\sqrt{D\left[\frac{\partial^{2}L(\lambda_{1}^{(0)},\ldots,\lambda_{n}^{(0)})}{\partial\lambda_{p}\partial\lambda_{q}}\right]}{\frac{\partial^{2}L(\lambda_{1}^{(0)},\ldots,\lambda_{n}^{(0)})}{\partial\lambda_{p}\partial\lambda_{q}}} \sim \frac{1}{\sqrt{T}}.$$

The last relationship shows that for large T quantity $\frac{\partial^{3}L(\lambda_{n}^{(0)},...,\lambda_{n}^{(0)})}{\partial\lambda_{p}\partial\lambda_{q}}$ in

probability seeks its own mean value and may be replaced by it. Thus, for large T matrix $\mathbf{A} \approx \mathbf{\overline{A}}$. Now we can solve equation (13.5.12). Using (13.5.13), we obtain

$$\lambda_p - \lambda_p^{(0)} = \sum_{r=1}^n J_{pr} \frac{\partial L(\lambda_1^{(0)}, \dots, \lambda_n^{(0)})}{\partial \lambda_r}, \qquad (13.5.17)$$

where J_{pr} - elements of matrix (13.5.2).

Hence

$$\overline{\lambda}_{p} = \lambda_{p}^{(0)},$$

$$\overline{(\lambda_{p} - \lambda_{p}^{(0)})(\lambda_{q} - \lambda_{q}^{(0)})} =$$

$$= \sum_{l, q=1}^{n} J_{p, l, q_{q}} \frac{\overline{\partial L(\lambda_{1}^{(0)}, \dots, \lambda_{q}^{(0)})}}{\partial \lambda_{r}} \cdot \frac{\partial L(\lambda_{1}^{(1)}, \dots, \lambda_{n}^{(0)})}{\partial \lambda_{q}} = J_{p, q}$$

From these equalities there ensues, by definition, joint efficiency of maximum likelihood estimates taking place with a rapidly fluctuating signal and a sufficiently large time of observation.*

[&]quot;Let us note that time of observation T at which we already have approximate efficiency of the considered estimates is greater, the less the coordinates of targets differ.

We note now that all the results presented are easily generalized in the case when we observe, not a scalar, but a vector signal. We meet such a case when the radar has a multiunit antenna. In this case we observe several signals at the cutput of different elements of the antenna. We shall enumerate briefly the obvious modifications which the above-stated results will obtain in examining this more general case. Signals received by different elements of the antenna will be recorded in the form

$$y_{l}(l) = \sum_{p=1}^{n} y_{p}^{(l)}(l, \lambda_{p}) + n_{l}(l), \qquad (12.5.15)$$

يانيان الالتداريجية ، العلم الملكية م مشاهرة الله المارية المليطية الله المارية المالية المالية المالية المالي 1944 - الالتداريجية ، العلم الملكية م مشاهرة الله المارية المليطية الله المالية المالية المالية المالية المالية

tion and the second of the second

where $y_p^{(l)}(t, \lambda_p)$ - signal from the p-th target at the output of the *l*-th element of the antenna.

The correlation function of signals $y_p^{(i)}(t, \lambda_p)$ will be recorded in the form

$$R_{l_1 l_1}(t_1, t_2) = \overline{y_p^{(l_1)}(t_1, \lambda_p) y_p^{(l_1)}(t_2, \lambda_p)} =$$

= Re $P_p u_p^{(l_1)}(t_1, \lambda_p) u_p^{(l_1)*}(t_2, \lambda_p) \rho_p (t_1 - t_2) e^{i\omega_0 (t_1 - t_2)},$

where $u_p^{(l)}(t, \lambda_p)$ - complex modulation of the signal reflected from the p-th target and received by the *l*-th element of the antenna. The correlation function of the whole signal (13.5.18) will have form

$$\overline{y_{l_{a}}(t_{1})y_{l_{a}}(t_{2})} = \operatorname{Re} e^{i\omega_{0}(t_{1}-t_{2})} \sum_{p=1}^{n} P_{p} \rho_{p} (t_{1} - t_{2})$$
$$--t_{2} u_{p}^{(l_{1})}(t_{1}, \lambda_{p}) u_{p}^{(l_{0})*}(t_{2}, \lambda_{p}) + N_{0} \delta_{l_{1}l_{a}} \delta(t_{1} - t_{2})$$
(13.5.181)

(we consider noises $n_1(t)$ independent). Function $W(t_1, t_2)$ now will be equal to

where function $v_{pq}(t_1, t_2)$ will be precisely the same as in the case of a scalar signal, only by $C_{pq}(\lambda_p, \lambda_q)$ it is necessary now to understand

$$C_{pq}(\lambda_{p},\lambda_{q}) = \frac{1}{T_{r}} \int_{0}^{T_{r}} \frac{1}{N} \sum_{i=1}^{N} u_{p}^{(i)*}(t,\lambda_{p}) u_{p}^{(i)}(t,\lambda_{q}) dt. \qquad (13.5.19)$$

Forther, relationships (13.5.14) and (13.5.14) are replaced by

$$\frac{\partial L(\lambda_{1}, \dots, \lambda_{n})}{\partial \lambda_{p}} = -\frac{1}{2} \sum_{l_{s}, l_{s}=1}^{N} \int_{0}^{T} \int_{0}^{T} \frac{\partial R_{l_{s}l_{s}}(t_{1}, t_{2})}{\partial \lambda_{p}} \times \\ \times W_{l_{s}l_{s}}(t_{1}, t_{s}) dt_{1} dt_{s} - \frac{1}{2} \sum_{l_{s}, l_{s}=1}^{N} \int_{0}^{T} \int_{0}^{T} \frac{\partial W_{l_{s}l_{s}}(t_{1}, t_{2})}{\partial \lambda_{p}} \times \\ \times y_{l_{s}}(t_{1}) y_{l_{s}}(t_{s}) dt_{1} dt_{s} - \frac{1}{2} \sum_{l_{s}, l_{s}=1}^{N} \int_{0}^{T} \frac{\partial W_{l_{s}l_{s}}(t_{1}, t_{2})}{\partial \lambda_{p}} \times \\ \times \frac{\partial L(\lambda_{1}^{(0)}, \dots, \lambda_{n}^{(0)})}{\partial \lambda_{p} \partial \lambda_{q}} = -\frac{1}{2} \sum_{l_{s}, l_{s}=1}^{N} \int_{0}^{T} \frac{\partial W_{l_{s}l_{s}}(t_{1}, t_{1})}{\partial \lambda_{p}} \times \\ \times \frac{\partial R_{l_{s}l_{s}}(t_{1}, t_{2})}{\partial \lambda_{q}} dt_{s} dt_{s}$$

$$(13.5.20)$$

in the second second

(by $\lambda_p^{(0)}$ we again mean the true values of parameters λ_p). Expressions (13.5.19) and (13.5.20) are obtained by generalization of results of Chapter X. Those changes which the vector character of the received signal will entail are limited actually to these.

We proved the asymptotic efficiency of the maximum likelihood estimates. On this basis it is possible to find a system of optimum resolution in the case of parameters of targets which do not vary in time. In the case of slowly varying parameters λ_p , when intervals of approximate constancy of these parameters are great as compared to times of correlation of fluctuations of the signals, the obtained results permit us to find the operations of an optimum discriminator. Actually, let us assume that the output of the p-th channel of a certain meter is $\hat{\lambda}_p(t)$. If this quantity is close to the true value of the measured parameter in every given moment of time, then, considering in (13.5.17) $\lambda_p^{(0)} = \hat{\lambda}_p$, we obtain

$$\Delta \hat{\lambda}_{p} = \lambda_{p} - \hat{\lambda}_{p} = -\sum_{r=1}^{n} J_{pr} \frac{\partial L(\hat{\lambda}_{1}, \dots, \hat{\lambda}_{n})}{\partial \lambda_{r}}, \qquad (13.9.21)$$

where $\lambda_{\rm p}$ is the maximum likelihood estimate obtained during time T.

{ 1

Considering the unbiased nature of estimate λ_p proved above, we note that the mean $\overline{\lambda\lambda_p}$ is equal to mismatch between the true value of the parameter and the output quantity of the meter. If we now present $\Delta\lambda_p$ in the form

$$\Delta \hat{z}_{\mu} = \int_{0}^{\infty} z_{\nu}(t) dt, \qquad (13.5.22)$$

 $z_p(t)$ is a quantity which is, on the average, equal to the current mismatch between the true and measured values of parameter $\lambda_p(t)$. Considering the very great inertia of the smoothing circuits, for $z_p(t)$ we can write the statistically equivalent expression

$$\mathbf{z}_{\mathbf{p}}(l) = \Delta \lambda_{\mathbf{p}}(l) + \xi_{\mathbf{p}}(l), \qquad (43.5.23)$$

where $\Delta \lambda_{p}(t)$ - true mismatch with respect to the measured considerate; $\xi_{p}(t)$ - white noise. The circumstance that variance of quantity $\Delta \hat{\lambda}_{p}$ is minimum testifies to minimum spectral density of noise $\xi_{p}(t)$.

Thus, under the above conditions for any smoothing circuits of a meter optimum processing of the r-r signal is carried out by a discriminator forming the set of quantities $z_p(t)$ (p = 1, 2, ..., n). The quality of its operation is characterized by the matrix of spectral densities $\mathbf{S}_{OHT} = \|\mathbf{S}_{OHT}^{(pq)}\|$, where

$$S_{out} = TJ.$$
 (13.5.24)

In order to relate $S_{p(t)}^{(pq)}$ with the equivalent presentation of (13.5.23) we note that noises $f_p(t)$ are correlated, in general, among themselves. Their functions of cross-correlation have form

$$\overline{\xi_p(l)\xi_q(s)} = S_{ont}^{(pq)} \delta(t-s).$$
(13.5.25)

The problem of synthesis of smoothing circuits of meters of coordinates of closely located targets should be solved separately. In the case of rapid fluctuations it has less bearing on questions of resolution. We note that difficulties connected with matrix inversion in expression (5.5.11) permits us to obtain discernible results only for the case of two targets. But measuring conditions of radar devices usually occur with a large signal-to-noise ratio. Therefore in examining meters of coordinates of many targets it is completely natural and justified to assume a large signal-to-noise ratio. This assumption, as we shall see below, permits us to obtain sufficiently simple and physically intelligible results without limitation on the number of observed targets. The case of an arbitrary signal-to-noise ratio we will consider with the example of a two-target problem.

13.5.3. Optimum Discriminator in the Case of Many Targets with a Large Signal-To-Noise Ratio

The above results [see (13.5.11), (13.5.17)] in principle permit us to determine the general structure of an optimum multitarget discriminator with an arbitrary signal-to-noise ratio. However inversion of the corresponding matrices of high and arbitrary order in this case runs into considerable difficulties, so that the

-782-

structure of the optimum discriminator turns out not to be completely discovered. At the same time for measuring problems of special interest is the case of a large signal-to-noise ratio, for which these difficulties can be surmounted.

With a large signal-to-noise ratio elements of matrix $\frac{1}{2N_0}S(\omega)$ with a sufficiently contail ω are large, and it is possible to expand (13.5.11) for $V(\omega)$ in a sories of negative powers of this matrix. Limiting ourselves to the first two terms of this expansion, we have

$$\mathbf{V}(\boldsymbol{\omega}) \approx -\frac{2}{N_{\bullet}} \mathbf{C}^{-1} + 4 (\mathbf{CS}(\boldsymbol{\omega}) \mathbf{C})^{-1}. \qquad (13.5.26)$$

From this for $V_{pq}(\omega)$ we obtain the following expression:

$$V_{pq}(\mathbf{o}) = -\frac{2}{N_0} C_{pq}^{-1} + 4 \sum_{r=1}^{n} \frac{C_{pr}^{-1} C_{rq}^{-1}}{P_r S_r(\mathbf{o})}, \qquad (13.5.27)$$

where by C_{pq}^{-1} we denote elements of matrix C^{-1} . Subsequently, we use the first term of expansion (13.5.27). Obviously, the condition where it is possible to use such an approximation has the form

$$\left|\frac{2N_{o}}{C_{pq}^{-1}}\sum_{r=1}^{n}\frac{C_{pr}^{-1}C_{rq}^{-1}}{P_{r}S_{r}(\omega)}\right| = \left|\frac{1}{C_{pq}^{-1}}\sum_{r=1}^{n}\frac{1}{h_{r}}\frac{C_{pr}^{-1}C_{rq}^{-1}}{S_{or}(\omega)}\right| \leq 1,$$
(13.5.28)

where

 $S_{Or}(\omega) = \Delta f_{cr} S_r(\omega)$ - normalized spectrum of fluctuations of the signal reflected from r-th target;

 $\Delta f_{cr} = effective width of the spectrum of fluctuations of this signal;$

 $h_r = P_r/2N_0\Delta f_{cr}$ - ratio of power of the signal reflected from the r-th target to the power of noise in the band of fluctuations of the signal.

With a sufficiently large signal-to-noise ratio condition (13.5.23) will be carried out. It is necessary, however, to consider that condition (13.5.28) should be realized in an interval of frequencies sufficiently exceeding the width of the spectrum of fluctuations of the signal.

Thus, subsequently we use the first term of expansion (13.5.27). We shall first calculate matrix $\overline{\mathbf{A}}$ (13.5.13). Differentiating (13.5.7) and (13.5.8) with respect to parameters λ_{p} and λ_{q} , respectively, and substituting the obtained relationships in (13...4), we can obtain the following very important result:

$$\overline{A_{pq}} = -\frac{\overline{\partial^{2}L\left(\lambda_{1}^{(0)}, \dots, \lambda_{n}^{(0)}\right)}}{\partial \lambda_{p} \partial \lambda_{q}} = \frac{TP_{p}}{N_{q}} \delta_{qp} \left\{ C_{pq}^{(11)}\left(\lambda_{p}, \lambda_{q}\right) - - \sum_{r,s=1}^{n} C_{pr}^{(10)}(\lambda_{p}, \lambda_{r}) C_{rs}^{-1} C_{sq}^{(01)}\left(\lambda_{s}, \lambda_{q}\right) \right\}, \qquad (13.5.29)$$

where we use designations (13.5.15).

Thus, matrix $\overline{\mathbf{A}}$ is diagonal, and the jointly-efficient estimates are uncorrelated. Matrix inversion of $\overline{\mathbf{A}}$ causes no difficulties, and for variance of the efficient estimate of parameter $\lambda_{\mathbf{p}}$ we obtain

$$\sigma_{s\phi,p}^{2} = \frac{1}{\frac{TP_{p}}{N_{\bullet}} \left\{ C_{pp}^{(11)}(\lambda_{p},\lambda_{p}) - \sum_{r,s=1}^{n} C_{pr}^{(10)}(\lambda_{p},\lambda_{r}) C_{rs}^{-1} C_{sp}^{(01)}(\lambda_{s},\lambda_{p}) \right\}}$$
(13.1.1.1)

This is a very important formula, characterizing potential accuracy of measurement of coordinates of several targets with a large signal-to-noise ratio. The equivalent spectral density (for this parameter λ_p) is determined from (13.5.30) by multiplication by T.

$$S_{ourp} = \frac{1}{\frac{P_p}{N_o} \left\{ C_{pp}^{(11)} - \sum_{r_s \, s \neq 1}^{n} C_{pr}^{(10)} C_{rs}^{-1} C_{sp}^{(01)} \right\}}$$
(13.5.31)

It is important to note that accurate of reasonment of coordinates of the p-th target does not depend on the power of signals reflected from the other targets. This testifies to good compensation of the interfering signals in an optimum meter of coordinates of many targets.

We pase to finding the optimum processing of a signal in a meter of coordinates of many targets. Substituting expressions (13.5.7) and (13.5.8) in (13.5.14) if is easy to prove that the component not containing realizations of signals will be equal to zero. As a result we have

$$\frac{\partial L(\lambda_1,\ldots,\lambda_n)}{\partial \lambda_p} = -\frac{1}{2} \frac{\partial}{\partial \lambda_p} \int_0^T \int_0^T \operatorname{Re} e^{i\omega_0(t_1-t_0)} \times \sum_{r,s=1}^n u(t_1,\lambda_r) u^*(t_3,\lambda_s) v_{rs}(t_1-t_0) y(t_1) y(t_0) dt_1 dt_3.$$

If the signal-to-noise ratio is great, then within a frequency range sufficiently exceeding the width of the spectrum of fluctuations of the signal we have $V_{re}(n) \approx C_{re}^{-1}$, i.e., function $v_{rs}(t_1 - t_2)$ can be considered b-shaped with respect to

fluctuations of the signal. On the other hand, as we know, function $v_{rs}(t)$ is very slow in relation to functions $u_p(t, \lambda_p)$. Now $v_{pq}(t_1 - t_2)$ can be presented thus:

$$\boldsymbol{v}_{pq}\left(t_{1}-t_{p}\right)\approx C_{pq}^{-1}\,\boldsymbol{v}\left(t_{1}-t_{s}\right)\approx$$
$$\approx C_{pq}^{-1}\int_{0}^{T}h\left(t-t_{1}\right)h\left(t-t_{s}\right)dt,$$

where h(t) - pulse response of the filter integrating modulation of the receiving signal, but passing without change fluctuations of the signal.

Here the derivative of the logarithm of the likelihood functional is rewritten in the form

$$\frac{\partial L (\lambda_{1}, \dots, \lambda_{n})}{\partial \lambda_{p}} = -\frac{1}{2} \frac{\partial}{\partial \lambda_{p}} \int_{0}^{T} dt \cdot \operatorname{Re} \int_{0}^{T} \int_{0}^{T} h (t - t_{1}) \times \\ \times h(t - t_{p}) e^{i \omega_{p} (t_{p} - t_{1})} y(t_{1}) y(t_{p}) \sum_{s, r = 1}^{n} u_{r} (t_{1}, \lambda_{r}) \times \\ \times C_{rs}^{-1} u^{*}_{s} (t_{q}, \lambda_{s}) dt_{1} dt_{1}.$$

$$(43.5.32)$$

The circuit realization of operation (13.5.32) can be carried out by various methods. Here it is important to obtain the simplest circuit. For this we calculate expression

$$\frac{\partial}{\partial \lambda_{p}} \sum_{r_{e}}^{n} \frac{u_{r}}{u_{r}} (t_{1}, \lambda_{r}) u^{*}_{a} (t_{2}, \lambda_{s}) C_{rs}^{-1},$$

in (13.5.32). It, obviously, is equal to

$$\frac{\partial u_{p}(t_{1},\lambda_{p})}{\partial \lambda_{p}} \sum_{a=1}^{n} u^{*}{}_{a}(t_{a},\lambda_{a}) C_{ap}^{-1*} + \sum_{s=1}^{n} u_{s}(t_{1},\lambda_{s}) C_{ap}^{-1} \frac{\partial u^{*}{}_{p}(t_{1},\lambda_{p})}{\partial \lambda_{p}} + \sum_{r,\ s=1}^{n} u_{r}(t_{1},\lambda_{r}) u^{*}{}_{a}(t_{s},\lambda_{s}) \frac{dC_{rs}^{-1}}{d\lambda_{p}}.$$
(12.5.2)

On the other hand, we have

()

$$CC^{-1} = E, \quad \frac{dC^{-1}}{d\lambda_p} C + C^{-1} \frac{dC}{d\lambda_p} = 0,$$
$$\frac{dC^{-1}}{d\lambda_p} = -C^{-1} \frac{dC}{d\lambda_p} C^{-1}.$$

-785-

Expanding the last relationship, we obtain

$$\frac{dC_{rs}^{-1}}{d\lambda_{p}} = -\sum_{i, j=1}^{n} C_{ri}^{-1} \frac{dC_{ij}}{d\lambda_{p}} C_{js}^{-1} = -C_{rp}^{-1} \sum_{j=1}^{n} C_{pj}^{(10)} C_{js}^{-1} - C_{ps}^{-1} \left(\sum_{j=1}^{n} C_{pj}^{(10)} C_{jr}^{-1}\right)^{*}.$$

Substituting this expression in (13.5.33) and then in (13.5.32), for the derivative of the logarithm of the likelihood functional we finally obtain

$$\frac{\partial \mathcal{L}(\lambda_{1},\ldots,\lambda_{n})}{\partial\lambda_{p}} = -\frac{1}{2} \int_{0}^{T} dt \operatorname{Re} \int_{0}^{T} h\left(t-\tau\right) \left\{ \frac{\partial u_{\nu}\left(\tau,\lambda_{n}\right)}{\partial\lambda_{p}} - \sum_{s=1}^{n} u_{s}\left(\tau,\lambda_{s}\right) \left[\sum_{j=1}^{n} C_{sj}^{-1} C_{jp}^{(01)}\left(\lambda_{j},\lambda_{\nu}\right) \right] \right\} y\left(\tau\right) e^{i\omega_{0}\tau} d\tau \times \left[\int_{0}^{T} h\left(t-\tau\right) \left(\sum_{s=1}^{n} u_{s}\left(\tau,\lambda_{s}\right) C_{sp}^{-1} \right) y\left(\tau\right) e^{i\omega_{0}\tau} d\tau \right]^{*}.$$
(13.5.34)

 C_{m} integral over t in expression (13.5.34), we have an analytic expression for the operation of the optimum discriminator:

$$z(t) \approx \operatorname{Re}_{0}^{T} h\left(t-\tau\right) \left\{ \frac{\partial u_{p}\left(\tau, \lambda_{p}\right)}{\partial \lambda_{p}} - \sum_{s=1}^{n} u_{s}\left(\tau, \lambda_{s}\right) \times \left[\sum_{j=1}^{n} C_{sj}^{-1} C_{jp}^{(01)}\left(\lambda_{j}, \lambda_{p}\right) \right] \right\} y\left(\tau\right) e^{iu_{0}\tau} d\tau \times \left\{ \int_{0}^{T} h\left(t-\tau\right) \left[\sum_{s=1}^{n} u_{s}\left(\tau, \lambda_{s}\right) C_{sp}^{-1} \right] y\left(\tau\right) e^{iu_{0}\tau} d\tau \right\}_{\lambda_{1}}^{*} = \lambda_{1}^{*}.$$
(13.5.35)

The block diagram of an optimum discriminator executing operation (13.5.35) is employed in Fig. 13.7. It coincides in structure with circuits of discriminators of matters c. coordinates of one target; only the heterodyne signals by which we multiple of signals in the channels were changed. The physical meaning of introduction of namely such heterodyne signals is rather easily perceived and is very curious.

Cignal

$$\sum_{s=1}^{n} u_{s}(t, \lambda_{s}) C^{-1}{}_{s \mu} e^{i \omega_{p} t} = u(t) e^{i \omega_{p} t}, \qquad (13.5.36)$$

as it is easy to establish by means of direct calculation, is orthogonal to the interfering signals.

figral

$$\boldsymbol{v}(t) e^{i\boldsymbol{\omega}_{r}t} = \left\{ \frac{\partial u_{p}(t, \lambda_{p})}{\partial \lambda_{p}} - \sum_{\substack{j=1\\j=1}}^{n} u_{\bullet}(t, \lambda_{\bullet}) \left[\sum_{\substack{j=1\\j=1}}^{n} C_{\bullet j}^{-1} C_{j p}^{(01)}(\lambda_{j}, \lambda_{p}) \right] \right\} e^{i\boldsymbol{\omega}_{r}t}$$
(13.5.37)

turns out to be orthogonal to all signals without exception. The special form of reference signals at the output of the circuit ensures the existence of a signal only in the presence of mismatch between the input and heterodyne signals with respect



Fig. 13.7. Optimum discriminator of a meter of coordinates of many targets: 1) mixer; 2) filter; 3) multiplier.

to the measured parameter. The output signal is proportional to this mismatch. The presence of a channel with heterodyne signal (13.5.37) is caused by the essence of the measuring problem: we must tune away from interfering signals, even if parameters corresponding to these signals are not known by us exactly.

The meaning of operations of the synthesized circuit can be grasped more clearly if we analyze this circuit, i.e.,

directly calculate its characteristics. We shall calculate these characteristics; here we shall not be limited only to the case of a large signal-to-noise ratio, since this calculation is slightly complicated if we consider the signal-to-noise ratio arbitrary.

Furthermore, we consider that filters in the analyzed circuit have a frequency band comparable with the band of fluctuations of the signal. This also does not complicate calculation, but the obtained results will be more valuable from the practical point of view. The fact is that for practical realization of the considered circuit besides knowledge of its optimality for a large signal-to-noise ratio it is very desirable also to know its behavior in the whole range of signal-to-noise ratios; it is also clear that for small signal-to-noise ratios accuracy or the enalyzed circuit will essentially depend on the width of the filter passbands, and comparative analysis of all possible situations here is of extraordinary interest.

Thus, we shall consider the circuit depicted in Fig. 13.7. We consider that filters in this circuit have frequency response $H(1\infty)$. Using expression (13.8.35) for the comput signal of the considered circuit and introducing designations u(1) and v(z) of (13.5.36) and (13.5.37) for modulations of the hoterodyne signal, we can obtain the following results:

$$\overline{t(l)} = \frac{1}{2} \sum_{k=1}^{n} \operatorname{Re} b_{k} a^{*}_{k} \cdot \frac{1}{2\pi} \int_{-\infty}^{\infty} |H(i\omega_{j})|^{2} S_{k}(\omega) d\omega + N_{0} \operatorname{Re} C\Delta j_{0}, \qquad (\pm 3.5.36)$$

(13.5.30)

where



 Δr_{m} - effective width of the filter passband.

Considering true values of parameters λ_p to coincide with parameters of the signals of the heterodyne oscillator u(t) (23.5.34) and v(t) (13.5.37), it is easy to find that $\Phi_k = 0$ and systematic error

$$r_{amor} = 0.$$

With detuning $\Delta\lambda_p$ between the shown values of the parameters from (13.5.38) and (13.5.39) we obtain

$$\overline{\mathbf{r}(i)} = \Delta \lambda_{p} \frac{1}{2} \left(C_{pp}^{(11)} - \sum_{r,s=1}^{n} C_{ps}^{(10)} C_{rj}^{-1} C_{rp}^{(01)} \right) \times \\ \times \frac{1}{2\pi} \int_{-\infty}^{\infty} |H(i\omega)|^{2} S_{p}(\omega) d\omega, \qquad (12.1.5)$$

i.e., the mean value of the signal at the output of the circuit is proportional only to mismatch with respect to the measured parameter and does not depend on interfering signals even in the presence of mistuning (of course, small) between true values of their parameters and the values introduced in the heterodyne signals. Herein lies the meaning of the special and complicated form of these signals. From expression (13.5.40) we find the slope of the discrimination characteristic of the analyzed circuit:

$$K_{\mathbf{x}} = \frac{1}{2} \left(C_{pp}^{(11)} - \sum_{r, \, a=1}^{n} C_{pr}^{(10)} C_{rr}^{-1} C_{rp}^{(01)} \right) \frac{1}{2n} \int_{-\infty}^{\infty} |H(i\omega)|^2 S_{\mu}(\omega) \, d\omega.$$
(13.5.41)

Spectral density at zero frequency of signal z(t) in the absence of mistuning for all parameters is also easily calculated and turns out to be equal to

$$\int_{-\infty}^{\infty} \overline{z(t) z(t+\tau)} d\tau = \frac{N_0^2}{2} \frac{1}{T_r} \int_{0}^{T_r} |v(t)|^2 dt \frac{1}{2\pi} \times \int_{-\infty}^{\infty} |H(i\omega)|^4 \left(\frac{S_F(\omega)}{2N_0} + \frac{1}{T_r} \int_{0}^{T_r} |u(t)|^2 dt \right) d\omega.$$

From this we obtain the following final expression for equivalent spectral density with respect to parameter λ_n :

$$S_{BKB} = \frac{\frac{1}{2\pi} \int_{-\infty}^{\infty} |H(i\omega)|^4 (h_p S_{ep}(\omega) + C_{pp}^{-1}) d\omega}{2h_p^2 \left(C_{pp}^{(11)} - \sum_{l, l=1}^{n} C_{pl}^{(10)} C_{ll}^{-1} C_{lp}^{(01)} \right) \left(\frac{1}{2\pi} \int_{-\infty}^{\infty} |H(i\omega)|^2 S_{ep}(\omega) d\omega \right)^2}, \quad (13.5.52)$$

where $h_p = P_p/2N_0 \Delta f_{cp}$ - ratio of power of the signal reflected from the p-th target to the power of noise in the band of fluctuations of this signal, and

$S_{op}(\omega) = \Delta f_{op} S_p(\omega)$

is the normalized spectrum of fluctuations of the signal from this target. It is easy to see that for a large signal-to-noise ratio and widening of the passband of the filter expression (13.5.42) coincides with (13.5.31), i.e., accuracy of the analyzed circuit in these conditions, naturally, coincides with the potential accuracy. From (13.5.42) it is easy to establish the conditions in which this takes place:

$$\begin{split} h_p \Delta f_{op} \geqslant C_{pp}^{-1} \Delta f_{\phi}, \\ \Delta f_{\phi} \geqslant \Delta f_{op}. \end{split}$$

Practically, it is sufficient to have

のないので、日本の

To produce calculation of accuracies by formulas (13.5.31) and (13.5.42) is fairly difficult in view of the complexity of calculation of elements of matrix C^{-1} . Simple formulas are obtained for two targets. Here

-789-



The circuit of the optimum discriminator for measurement of parameter λ_p takes the form depicted in Fig. 13.8. Modulation of heterodyne signals in this circuit can be presented in the form



and equivalent spectral density can be presented in the form

$$S_{\text{PKB}} = \frac{1}{\frac{2P}{N_{\bullet}} \left\{ C_{11}^{(11)} - \frac{|C_{11}^{(10)}|^2 + |C_{12}^{(10)}|^2 - 2\text{Re} C_{12}^{(10)} C_{11}^{(01)} C_{12}}{1 - |C_{12}|^2} \right\}}.$$

The optimum discriminator for the case of two targets will be studied in more detail in the next paragraph. In conclusion we shall briefly consider the case when



Hig. 13.8. Optimum discriminator of a metercreated instance of two targets: 1) mixor; 2) filter; 3) multiplier. the signal has a vector character. In this case, as we have seen, matrix $V(\omega)$, characterizing the joint functional of the probability density of the observed signals, is expressed by the former formula (13.5.11); only matrix C is meatthen; its elements are determined now by formula (13.5.19). Consequently, with a large signal-to-noise ratio $V(\omega)$, as before, is found in the form of (13.5.26) or (13.5.27).

We substitute expression (13.7.48) for correlation functions of the received signals and expression (13.5.18) for functions $W_{l_1 l_2}(t_1, t_2)$ in formula (.3.7.29). Here it is easy to find that

$$\frac{\partial^* L\left(\lambda_1^{(0)},\ldots,\lambda_n^{(0)}\right)}{\partial \lambda_p \partial \lambda_q}$$

has the former form (13.5.16), only functions C_{1j} are expressed by formulas (13.5.19) and

$$C_{ij}^{(10)} = \frac{1}{T_r} \int_0^{T_r} \frac{1}{N} \sum_{l=1}^N \frac{\partial u_l^{*(l)}(t, \lambda_i)}{\partial \lambda_i} u_j^{(l)} - \lambda_j) dt, \qquad (13.5.44)$$

$$C_{ij}^{(11)} = \frac{1}{T_r} \int_0^{T_r} \frac{1}{N} \sum_{l=1}^N \frac{\partial u_i^{*(l)}(t, \lambda_i)}{\partial \lambda_i} \frac{\partial u_j^{(l)}(t, \lambda_j)}{\partial \lambda_j} dt.$$

Consequently, $\sigma_{p\phi}^2$ p and S_{OUT p} will also be expressed by the former formulas (13.5.30) and (13.5.31). Then, formula (13.5.20), we can obtain for the optimum operation in the considered case using the following expression:

$$z(t) = \operatorname{Re} \int_{0}^{t} h(t-\tau) \sum_{l=1}^{N} y^{(l)}(\tau) \left[\frac{\partial u_{p}^{(l)}(\tau, \hat{\lambda}_{p})}{\partial \lambda_{p}} \right] - -\sum_{s=1}^{n} u_{s}^{(l)}(\tau, \lambda_{s}) \left(\sum_{j=1}^{n} C_{sj}^{-1} C_{jp}^{(0)} \right) \right] e^{i\omega_{0}\tau} d\tau \times \left\{ \int_{0}^{t} h(t-\tau) \sum_{l=1}^{N} \left(\sum_{s=1}^{n} u_{s}^{(l)}(\tau, \hat{\lambda}_{s}) C^{-1} \sum_{sp} \right) p^{(l)}(\tau) e^{i\omega_{0}\tau} d\tau \right\}^{*}.$$
(13.5.45)



The block diagram of the device executing operation (13.5.45) is depicted in Fig. 13.9. This circuit coincides in structure wi analogous circuits of meters of coordinates of one target (see, for instance, \$\$ 10.2, 10.3, 10.13); however, the form of the heterodyne signal: 1 6 LO

1

Fig. 13.9. Optimum discriminstor of a mater of angular coordinates of many targets: 1) mixers; 2) adders; 3) filters; 4) multiplier. is throughd. Their structure, as it is possible to see as a result of analysit yell operation (13.5.45), obeys again the same principle: to movimally tune away from invertering signals, even if their parameters are not known exactly by us.

13.5.4. Optimum Discriminate in the Same of Two Pargets Let us consider the case of two targets (n = 2), located so close to one another that their signals are not orthogonal. The signal-to-no se ratio h we shall consider arbitrary. Let us assume that parameter λ_1 , whose carrier is signal $y_1(\tau, \lambda_1)$, is to be measured. Signal $y_2(\tau, \lambda_2)$ plays here the role of interference. We first consider that the coordinate of interference λ_2 is known (is measured with great accuracy). Subsequently we shall reject this assumption. Considering coefficients of correlation of the useful and interfering signals identical $(p_1(\tau) = p_2(\tau) = p(\tau))$ and expressing function $W(\tau, \tau)$, the inverse of the correlation function, in the form

$$W(t, \tau) = \operatorname{Re} v(t, \tau) e^{i u_{\bullet}(t-\tau)} + \frac{1}{N_{\bullet}} \delta(t-\tau), \qquad (12.1.46)$$

then, substituting (13.5.46) in equation (13.5.9), we arrive at the following equation for $v(t, \tau)$:

$$P_{\sigma}u_{s}(t)\int_{0}^{T}\rho(t-s)u^{*},(s)v(s,\tau)ds + + P_{\pi}u_{s}(t)\int_{0}^{T}\rho(t-s)u^{*},(s)v(s,\tau)ds + + \frac{2P_{e}}{N_{o}}\rho_{s}(t-\tau)u_{s}(t)u^{*},(\tau) + + \frac{2P_{\pi}!}{N_{o}}\rho(t-\tau)u_{s}(t)u^{*},(\tau) + 2N_{o}v(t,\tau) = 0, \qquad (13.5.4)$$

where $P_c = P_1$, $P_H = P_2$ - mean powers of the useful and interfering signals. Solution of equation (13.5.47) is sought in the form

$$v(l, \mathbf{s}) = \sum_{k, j=1}^{2} v_{kj}(l - \mathbf{s}) u_k(l) u^{\mathbf{s}}_j(\mathbf{s}). \qquad (13.5.48)$$

Substituting (13.5.48) in (13.5.47) and producing approximate calculations allowing for observance of condition $T_r \ll \tau_k \ll T$, for the Fourier transforms of functions $v_{k,l}(\tau - \tau)$, we obtain

$$V_{11}(\omega) = -\frac{2N_{\bullet} + P_{\pi}S(\omega)}{P_{\pi}S(\omega)} F(\omega),$$

$$V_{11}(\omega) = -\frac{2N_{\bullet} + P_{c}S(\omega)}{P_{q}S(\omega)} F(\omega),$$

$$V_{11}(\omega) = \gamma e^{-i\varphi}F(\omega),$$

$$V_{11}(\omega) = \gamma e^{i\varphi}F(\omega),$$

where

()

$$S(\omega) = \int_{-\infty}^{\infty} \rho(\tau) e^{-i\omega\tau} d\tau; \qquad (13.5.50)$$

$$F(\omega) = \frac{2P_{o}P_{B}}{N_{o}} \frac{S^{2}(\omega)}{4N_{0}^{2} + 2N_{o}(P_{c} + P_{B})S(\omega) + P_{c}P_{B}S^{2}(\omega)(1 - \gamma^{2})},$$

and γ and φ are determined by function C_{21} (13.5.6):

$$C_{11} = C^*_{11} = \gamma e^{l_{\varphi}} = \frac{1}{T_r} \int_{\delta}^{T_r} u^*_{\delta}(s) z_1(s) ds. \qquad (12.5.51)$$

We substitute (13.5.46) and (13.5.48) in the likelihood equations (13.5.12), (13.5.14). Here, considering that this equality is valid,

$$-\int_{0}^{T}\int_{0}^{T} v_{hj} (l-\tau) e^{l\omega_{h}(l-\tau)} u_{h}(t) u^{*}_{j}(\tau) y(t) y(\tau) dt d\tau =$$

$$= a_{hj} \int_{0}^{T} dt \int_{0}^{t} h_{kj} (t-s_{1}) u_{h}(s_{1}) e^{l\omega_{0}s_{1}} y(s_{1}) ds_{1} \times$$

$$\times \int_{0}^{t} h_{hj} (t-s_{2}) u^{*}_{j}(s_{2}) e^{-i\omega_{0}\tau_{2}} y(s_{2}) ds_{2}, \qquad (13.9.52)$$

the term of the likelihood equation containing the received signal can be reduced to the form

$$-\frac{\partial}{\partial \lambda_{1}}\int_{0}^{T}\int_{0}^{T}W(t,\tau)y(t)y(\tau)dt\,d\tau =$$

$$=\frac{2}{N_{0}}\frac{\partial}{\partial \lambda_{1}}\int_{0}^{T}dt\left\{\left\|\int_{0}^{t}h_{11}(t-\tau)u_{1}(\tau)e^{i\omega_{0}\tau}y(\tau)d\tau\right\|^{2}+\right.$$

$$\left.+\left|\int_{0}^{t}h_{22}(t-\tau)u_{2}(\tau)e^{i\omega_{0}\tau}y(\tau)d\tau\right\|^{2}-$$

$$\left.-2\gamma\operatorname{Re}\int_{0}^{t}h_{12}(t-\tau_{1})u_{1}(\tau_{1})e^{i\omega_{0}\tau_{1}}y(\tau_{1})d\tau_{1}\times\right.$$

$$\left.\times\int_{0}^{t}h_{12}(t-\tau_{2})u^{*}_{2}(\tau_{2})e^{-i(\omega_{0}\tau_{1}+\tau)}y(\tau_{2})d\tau_{3}\right\},$$

where $h_{k,j}(v)$ — pulse responses of filters, squares of moduli of frequency responses of which are defined as

$$|H_{11}(i\omega)|^{2} = -\frac{N_{\bullet}}{2} V_{11}(\omega); \quad |H_{22}(i\omega)|^{2} = -\frac{N_{\bullet}}{2} V_{22}(\omega),$$

$$|H_{12}(i\omega)|^{2} = \frac{N_{\bullet}}{2} \frac{V_{12}(\omega)}{\gamma e^{-i\varphi}} = \frac{N_{\bullet}}{2} \frac{V_{21}(\omega)}{\gamma e^{i\varphi}}.$$
(13.5.53)

i di di di seconda se desta de la seconda de la second

The term of the likelihood equation not containing y(t) after substitution of the value of $W(t, \tau)$ and necessary transformations takes the form

$$I = -\int_{0}^{T} \int_{0}^{T} \frac{\partial R_{T}(t,\tau)}{\partial \lambda_{1}} W(t,\tau) dt d\tau = 4T \gamma \frac{\partial \gamma}{\partial \lambda_{1}} \frac{1}{2\pi} \times \int_{-\infty}^{\infty} \frac{P_{c} P_{\pi} S^{2}(\omega)}{4N_{0}^{2} + 2N_{\bullet} (P_{c} + P_{\pi}) S(\omega) + P_{c} P_{\pi} S^{2}(\omega) (1-\gamma^{2})} d\omega. \qquad (13.5.54)$$

With small signal-to-noise ratios

$$I = \frac{P_{o}P_{u}}{N_{0}^{2}} \Upsilon \frac{\partial \Upsilon}{\partial \lambda_{1}} T \frac{1}{2\pi} \int_{-\infty}^{\infty} S^{1}(\omega) d\omega =$$

$$= \frac{4h^{2}}{q} \Upsilon \frac{\partial \Upsilon}{\partial \lambda_{1}} T \frac{1}{2\pi} \int_{-\infty}^{\infty} S_{0}^{2}(\omega) d\omega, \qquad (13.5.54^{\circ})$$

where $h = P_c / 2\Delta f_c N_0$ - the signal-to-noise ratio;

 $q = P_{\text{C}}/P_{\text{H}}$ - the signal-to-intervence ratio;

 $\Delta f_{\rm C} = \mathop{\rm effective} {\rm width} \ {\rm of} \ {\rm the \ spectrum \ of \ fluctuations \ of \ the \ signal;}$

 $\omega_{\rm er}(\omega) = \Delta f_{\rm e} S(\omega)$ - normalized spectral density of fluctuations.

For large h, considering S(m) a square function, we obtain

$$I \approx T \gamma \frac{\partial \gamma}{\partial \lambda_1} \Delta f_0 \frac{4}{1 - \gamma^2}.$$
 (13.5.54")

Using the preceding equalities, we can write the likelihood equation in the form

$$\frac{\partial}{\partial h_{1}} \int dt \left\{ \left\| \int_{0}^{t} h_{11} \left(t - \tau \right) u_{1} \left(\tau \right) e^{i \omega_{0} \tau} y\left(\tau \right) d\tau \right\|^{2} - \left\| \int_{0}^{t} h_{22} \left(t - \tau \right) u_{2} \left(\tau \right) e^{i \omega_{0} \tau} y\left(\tau \right) d\tau \right\|^{2} - \frac{2\gamma \operatorname{Re}}{\delta} \int_{0}^{t} h_{12} \left(t - \tau_{1} \right) u_{1} \left(\tau_{1} \right) e^{i \omega_{0} \tau_{1}} y\left(\tau_{1} \right) d\tau_{1} \times \int_{0}^{t} h_{12} \left(t - \tau_{2} \right) u_{2}^{*} \left(\tau_{2} \right) e^{-i \left(\omega_{0} \tau_{1} + \tau_{1} \right)} y\left(\tau_{2} \right) d\tau_{2} \right\} + \frac{N_{0}}{2} I = 0.$$

It we approximately replace the operation of differentiation by calculation of a finite difference, the optimum circuit, corresponding to (13.5.55), takes the form of Fig. 13.10.* With small detuning of channels $\Delta\lambda_1$ and not too large γ character-



Fig. 13.10. Optimum discriminator for measurement of parameter λ_1 when n = 2: 1) mixer; 2) amplifier with gain varying according to the law of amplitude modulation; 3, 4, 5, 6, 7, 8) filters with pulse responses $h_{11-}(t) \cos \omega_{np}t$, $h_{12-}(t) \cos \omega_{np}t$, $h_{22-}(t) \cos \omega_{np}t$, $h_{22+}(t) \cos \omega_{np}t$, $n_{12+}(t) \cos \omega_{np}t$, $h_{11+}(t) \cos \omega_{np}t$, respectively; 9, 10) phase shifters with phase shift φ_{-} and φ_{+} , respectively; 11) square-law detector; 12) phase detector; 13, 14) amplifiers with gain $-2\gamma_{-}$ and $-2\gamma_{+}$, respectively.

1 3

istics of filters depend little on detuning. Disregarding this dependence, the circuit can be simplified and reduced to the form of Fig. 13.11. The circuit contains two discriminator channels, intended for working out mismatch with respect to λ_q . These channels of an optimum discriminator when there is no interfering signal $y_{0}(t)$ only by characteristics of the filters. Both forms of characteristics of filters coincide when $\gamma = 0$. Furthermore, the circuit contains channel: of compensation of the interfering signal, and in it there

is provided insertion of correction I for compensation of systematic error.

It is possible to imagine other circuits corresponding to operations (13.9.95) produced on signal y(t). In particular, it is possible to create a discriminator circuit with only two channels, occurring in the absence of interfering signals buy with reference signals properly modified. The circuit of Fig. 13.11 is expedient for tracking radar meters of coordinates of two targets. In this case as the channel of compensation we can use the channel of measurement of parameter λ_2 . Properly transforming the output of this channel, we realize the required compensation. This is simpler than change of the complicated reference signals necessary upon change of distances between targets in the other variant of optimum discriminator mentioned.

*Subscript "-" in the figure corresponds to channels defuned $-\Delta\lambda_1$; and subscript "+" corresponds to defuning $+\Delta\lambda_1$ relative to the measured value of λ_1 .



Fig. 13.11. Simplified circuit of an optimum discriminator: 1) mixer; 2) amplifier with gain varying according to the law of amplitude modulation; 3, 4) filters with pulse responses $h_{11}(t) \cos \omega_{\Pi_1^+} t$ and $h_{12}(t) \cos \omega_{\Pi_1^-} t$, respectively; 5, 6) phase shifters with phase shift φ_{\perp} and φ_{\perp} , respectively; 7, 8) amplifiers with gain $2\gamma_{\perp}$ and $2\gamma_{\perp}$, respectively; 9) square-law detector; 10) phase detector.

In connection with the fact that measuring systems frequently work with a very large signal-to-noise ratio, it is interesting to discuss further simplifications of the considered circuit for large h.

From (13.5.49), (13.5.50), and (13.5.53) it follows that for large h the squares of moduli of frequency responses of all filters become identical and equal in M_{22} . Each of fluctuations of the signal:

$$|H_{11}(i\omega)|^{2} = |H_{12}(i\omega)|^{2} = |H_{12}(i\omega)|^{2} = \frac{1}{1-\gamma^{2}} \quad (|\omega| \le 2\pi\Delta f_{c}) \quad (13.5.56)$$

At the same time, in accordance with the hypotheses made in deriving the basic relationships, these filters are integrating for the period of modulation of the signal T_{μ} . It is easy to see that equation (13.5.55) here takes the form

$$\frac{2}{N_{o}}\int_{0}^{t}\frac{\partial}{\partial\lambda_{1}}\left\{\frac{1}{1-\gamma^{2}}\left[|Q_{i}|^{2}+|Q_{i}|^{2}-2\gamma\operatorname{Re}Q_{i}Q^{*}_{i}e^{-i\varphi}\right]\right\}dt+I=0, \qquad (13.5.57)$$

$$Q_{1,2} = \int_{t-T_{p}}^{t} u_{1,2}(\tau) e^{i\omega_{0}\tau} y(\tau) d\tau.$$

Equation (12.5.57) will be transformed to

$$\int_{0}^{T} \left\{ \frac{\partial}{\partial \lambda_{1}} \left[\frac{|Q_{1} - \gamma e^{l \varphi} Q_{2}|^{2}}{1 - \gamma^{2}} \right] + \frac{l N_{0}}{2T} \right\} dt = 0.$$
 (13.5.59)

With replacement of differentiation by calculation of a finite difference operations of the discriminator are executed by the circuit of Fig. 13.12. Circuits



Fig. 13.12. Optimum discriminator for a large signalto-noise ratio. 1) mixer; 2) amplifier with gain varying according to the law of amplitude modulation; 3, 4) amplifiers with gain γ_{-} and γ_{+} , respectively; 5, 6) phase shifters with phase shift $-\Delta \varphi_{-}$ and $+\Delta_{\varphi_{+}}$, respectively; 7) filter, integrating for period T_{r} ; 8) squarelaw detector; 9, 10) amplifiers with gain $(1 - \gamma_{-}^{2})^{-1}$ and $(1 - \gamma_{+}^{2})^{-1}$ respectively. of compensation in this scheme correspond to total suppression of the interfering signal (see § 13.2). This circuit differs from the circuit optimum in the absence of $y_2(t)$ and with $h \rightarrow m$, to which there are added only channels of orthogonalization (total suppression) of interference, by the presence of amplifiers with gain

(13.5.58)

factors $\frac{1}{1-\gamma_{+}^{2}}$ and $\frac{1}{1-\gamma_{+}^{2}}$ and

introduction of correction

 $I_0 = \frac{IN_0}{2T}$. For small γ , very large h and small detunings $\Delta \lambda_1$ the shown differences of circuits are absent.

The optimum circuit ensures absence of systematic error, and the equivalent spectral density of fluctuating error of measurement of parameter λ_1 is defined as $\delta_{0111-1} = \sigma_{9\phi-1}^2$, where $\sigma_{9\phi-1}^2 = variance of the efficient estimate.$

If y't) is a normal random process, S_{OHI-1} is found in accordance with (13.5.14) and (13.5.24) by the formula

$$S_{out 1}^{-1} = -\frac{1}{27} \iint_{0}^{T} \frac{\partial R_{V}(t,\tau)}{\partial \lambda_{1}} \frac{\partial W(t,\tau)}{\partial \lambda_{1}} dt d\tau. \qquad (13.5.6^{\circ})$$

Substituting in (13.5.60) values of R_y and W and producing approximate calculations, valid under the assumptions made above ($\Re_r \ll \tau_k \ll T$), we obtain

$$S_{\bullet\bullet\bullet\uparrow}^{-1} = \frac{P_o}{2} \left\{ \left[(1 - \gamma^{\bullet}) A + \frac{2N_o}{P_a} B \right] C + 2\gamma^{\bullet} \frac{\partial \phi}{\partial \lambda_1} A D - - \left| \frac{\partial}{\partial \lambda_1} (\gamma e^{i\phi}) \right|^2 A + \frac{2N_o}{P_a} \gamma \frac{\partial \gamma}{\partial \lambda_1} \frac{\partial B}{\partial \lambda_1} - \left(A + \frac{2N_o}{P_a} B \right) D^2 \right\}, \qquad (12.5.11)$$

 $A = \frac{1}{2\pi} \int_{-\infty}^{\infty} F(\omega) S(\omega) d\omega;$ $B = \frac{1}{\sqrt{2\pi}} \int_{-\infty}^{\infty} F(\omega) d\omega;$ $C = C_{11}^{(11)} = \frac{1}{T_r} \int_{-\infty}^{T_r} \left| \frac{\partial u_1(t)}{\partial \lambda_1} \right|^{-1} dt;$ $D = iC_{11}^{(01)} = i \frac{1}{T_r} \int_{-\infty}^{T_r} \frac{\partial u_{+1}(t)}{\partial \lambda_1} u_1(t) dt =$ $= \frac{1}{T_r} \int_{-\infty}^{T_r} \frac{\partial \psi_1(t)}{\partial \lambda_1} u_{a_1}^2(t) dt.$

(13.5.52)

For large signal-to-noise ratios h formula (13.8.64) is simplified, taking form

$$S_{\mathfrak{s}\mathfrak{s}\mathfrak{r}\mathfrak{s}}^{-1} = \frac{\hbar 2\Delta I_{\mathfrak{s}}}{1-\gamma^{\mathfrak{s}}} \left[(1-\gamma^{\mathfrak{s}}) C + 2\gamma^{\mathfrak{s}} \frac{\partial \varphi}{\partial \lambda_{\mathfrak{s}}} D - \left| \frac{\partial}{\partial \lambda_{\mathfrak{s}}} (\gamma e^{i\gamma}) \right|^{\mathfrak{s}} - D^{\mathfrak{s}} \right].$$
(13.5.63)

If deriving this formula we assumed that $(1 - \gamma^2)h \gg 1$. It is easy to see that this formula is obtained from (13.5.31) when n = 2.

For small h

where

$$S_{out1}^{-1} = 2h^{\mathfrak{s}}(C - D^{\mathfrak{s}}) \frac{1}{2\pi} \int_{-\infty}^{\infty} S_{\mathfrak{s}}^{2}(\mathfrak{s}) d\mathfrak{s}. \qquad (12.5.64) \quad .$$

Bo that the found optimum circuit encores accuracy of measurement of parameter λ_1 , corresponding to formulas (13.5.61)-(13.5.64), we need exact tuning of compensation channels, i.e., unerring knowledge of quantity λ_2 . Practically, however, this parameter, corresponding to the coordinate of interference, is place measured with error. From

results of Paragraph 13.5.2 it follows that the best case is when there are systems of the considered form for measurement of parameters λ_1 and λ_2 , acting jointly. Tuning of compensation channels in each of these systems is produced in accordance with the output quantity of the other system. To find fluctuating errors of such such the legend it is necessary to compose matrix of spectral densities (13.5.24). Calcu-Lating the element of it which corresponds to error of measurement of parameter λ_1 , for the equivalent spectral density we obtain expression

$$S_{i} = \frac{S_{0171}}{1 - \frac{S_{0171}S_{0171}}{S_{12}^{2}}},$$
 (13.5.65)

where $\mathcal{E}_{\text{OHT}2}$ - spectral density of fluctuating error of measurement of parameter λ_2 for known λ_3 , which is found from $\mathcal{E}_{\text{OHT}1}$ if we change the places here of P_c and P_{ff}, and also of λ_4 and λ_2 ;

 S_{10} - mutual spectral density, determined according to (13.5.14) as

$$S_{11}^{-1} = -\frac{1}{2T} \int_{0}^{T} \int_{0}^{T} \frac{\partial R_{y}(t,\tau)}{\partial \lambda_{1}} \frac{\partial W(t,\tau)}{\partial \lambda_{2}} dt d\tau. \qquad (13.5.66)$$

(alculating integral (13.5.66), we have

$$\mathbf{S}_{11}^{-1} = -N_{\mathbf{0}} \frac{\partial}{\partial \lambda_1} \left[B \frac{\partial \gamma^1}{\partial \lambda_1} \right]. \tag{33.8.661}$$

From (13.5.65) it is easy to see that when $\gamma = 0$ equality $S_1 = S_{00T1}$ is value, which one should have expected, since here there are no compensation channels, and each parameter is measured independently one from the other. As $h \to \infty$, by calculating v_{12} we prove that this quantity approaches a finite limit, while $S_{0011,2} \simeq \frac{1}{h}$. The $v_1 \to v_{001,4}$, and fluctuating errors for large signal-to-noise ratios are determined by expression (13.5.63).

For small h, using (13.5.64) and calculating C_{12} , we obtain

$$S_{1} = \frac{S_{0R+1}}{1 - \left[\frac{\partial^{2} (Y^{1})}{\partial \lambda_{1}^{2}}\right]^{1}}, \qquad (13, \dots, 7)$$

4.e., there occurs increase of fluctuating error of measurement of parameter λ_1 are to errors of measurement of parameter λ_2 .

--799--

§ 13.6. <u>Analysis of Systems of Detection and Measurement</u> with Suppression of Distuising Signals

13.6.1. System of Detection

Output signals of the separation system synthesized in $\S13.2$ can be used for finding a decision about the presence of cargets and for measurement of their



c e. 17.17. Block diagram of a system of obtention with suppression of interfering signals: a) with multiplication by reference anymous shifted in time; b) with filtration: 1) mixer, 2) filter: 3) detector; 4) relay; 5) filter matched with the law of modulation; 5) mated applifier.

coordinates exactly as output signals of systems for singling out a signer against a background of noises and interferences, considered in preceding chapters. Such a detector or meter with suppression of disturbing signals differs from the corresponding devices calculated for the presence of only one target only in the form of reference signal or in the response of the shortening filter (in the case of range). As examples, in Fig. 13.13 there are given two variants of a block diagram of a system of detection of targets in a range of distances (with filtration and with multiplication by reference si, als shifted in time).

We shall consider characteristics of detection for such a system and compare neg. with the optimum. First of all one should note that the probabilities of false there and correct detection of one target with fixed parameters for a system with coparation of signals are determined by the very same formulas as for a system without reportion of signals. The difference consists only in the powers of the signal and noise obtained after multiplication of the received signal by the reference signal and integration (see Chapter IV, Vol. I).

Thus, comparison of characteristics of detection of one target in systems with suppression and without suppression of interfering signals reduces to comparison of the controponding signal-to-noise ratios. Such comparison already was conducted in § 13.2. The same remark is also valid for an optimum system of detection of a given target against a background of disturbing targets, inasmuch as such a system differs from a system with complete suppression of interfering signals and subsequent processing of the same form as in the absence of interfering signals also basically only in the form of the reference signal. Comparison of the corresponding signal-tointerference ratios for the case of one interfering target was already conducted in 13.2.1.

The difference between characteristics of detection in systems with suppression and without suppression of interfering signals turns out to be more considerable for detection of one or several targets in a certain range of values of parameters if we use a multichannel system of detection. This difference is caused by the fact that noises in detuned channels turn out to be correlated due to the presence in the reference signals of the same components.

Let us consider characteristics of detection for a system of channels in each of which there is produced complete suppression of the signals separated by the remaining channels. Fluctuations of all signals we shall consider slow.

Output voltage of each channel is recorded in the form (see §13.2)

$$r_{j} = |z_{j}|^{2} = \left| \int_{S_{0}} y(s) \psi(s, \lambda_{j}) ds \right|^{2} = \left| \sum_{i=1}^{n} w_{j} Q_{e} \right|^{2}.$$
(13.6.1)

If the received signal y(s) is a normal random process, the real and imaginary part of complex quantities $z_j(j = 1, ..., n)$ are distributed by normal law. Therefore joint distribution of these quantities can be recorded in the form

$$p(z_1,...,z_n) = \frac{1}{\pi^n |R_{jk}|} \exp\left\{-\sum_{j,k} W_{jk} z_j z_k^{*}\right\}, \qquad (13.6..)$$

where $\|d_{jk}\| = \text{matrix}$, the inverse of $\|\mathbf{R}_{jk}\|$, and

$$R_{jk} = \overline{z_j z^*_k} = \sum_{l, m} w_{jl} w^*_{km} \overline{Q_o Q^*_m} =$$
$$= \sum_{l, m} w_{jo} w^*_{km} \left[C_{1m} + \sum_{v} \frac{E_v}{2} C_{lv} C_{vm} \right] = w_{jk} + \frac{E_k}{2} \delta_{jk}, \qquad (13.0.3)$$

where E_{ν} - energy of the signal to which the ν -th channel is tuned. If there is no such signal, $E_{\nu} = 0$.

Let us find the joint distribution for quantities r_j . For this it is necessary to pass to polar coordinates, in which $z_j = \sqrt{r_j} e^{i r_j}$, and to integrate the obtained distribution over all φ_j^{\prime} . Introducing for convenience instead of $||R_{jk}||$ the matrix or correlation coefficients

$$P_{jk} = \frac{R_{jk}}{\sqrt{R_{jj}R_{kk}}}, \qquad (13.6.4)$$

. با معالمها الشفار فالقال المعالمين.

والمنافعة والمتعادية والمنافعة المنافعة المنافعة المنافعة المنافعة المنافعة والمنافعة وال

we obtain

$$p(r_1, \ldots, r_n) = \frac{1}{(2\pi)^n |\varphi_{jk}| \overline{r_1} \cdots \overline{r_n}} \times \\ \times \int_0^{2\pi} \exp\left[-\sum_{j,k} \theta_{jk} \sqrt{\frac{r_j r_k}{\overline{r_j r_k}}} e^{i(\varphi_j - \varphi_k)}\right] d\varphi_1 \dots d\varphi_n, \qquad (13.6.5)$$

where $\|\theta_{ik}\| = \text{matrix}$, the inverse of $\|\rho_{ik}\|$;

$$\overline{r_j} = \overline{|z_j|^2} = w_{jj} + \frac{E_j}{2} = w_{jj} (1 + q_j), \qquad (13.6.6)$$

and q_j = signal-to-interference ratio at the cutput of the system of separation for the j-th signal.

We cannot take integral (13.6.5) in general. Therefore we assume that f_{jk} is small when $j \neq k$, ile., that $\|\rho_{jk}\| = quar_{jk}$ -aia_onal matrix. Inasmuch as

$$P_{jk} = \frac{w_{jk}}{\sqrt{w_{jj}w_{kk}}} \frac{1}{\sqrt{(1+q_j)(1+q_k)}}, \qquad (13.6.41)$$

We sumption about the quasi-diagonal nature of $\|\rho_{jk}\|$ is satisfied if either satisfies $\|u_{jk}\|$ and $\|c_{jk}\|$ are quasi-diagonal, or the signal-to-interference ratios of the output of the system of separation for all targets are great.

It
$$\rho_{jk} = (1 - a_{jj}) \delta_{jk} + a_{jk}$$
 and $|a_{jk}| \ll 1$,

់ក្មមា

$\theta_{jk} \approx (1 + a_{jj}) \delta_{jk} - a_{jk}.$

Substituting this expression in (13.6.5), expanding the exponential function and the determinant in powers of α_{jk} with an accuracy of the second order, after integration over ϕ_j we obtain

-802-

$$p(r_1,\ldots,r_n) = \frac{1 + \sum_{j \neq k} |\theta_{jk}|^2 \frac{r_j r_k}{\overline{r_j r_k}}}{\left(1 + \sum_{j \neq k} |\theta_{jk}|^2\right) \overline{r_1 \ldots \overline{r_n}}} \exp\left(-\sum_j \frac{r_j}{\overline{r_j}}\right). \quad (13.6.7)$$

 U_{Sing} this distribution it is possible to calculate the probability of any combimations of exceedings of thresholds in the channels.

In the case n = 2 integration of (13.6.5) is easily conducted for any $\rho = |\rho_{12}|$. During calculations it is convenient to use the expansion of this distribution in powers of ρ , which has the form

$$P(r_1, r_3) = \frac{1}{r_1 r_3 (1-\rho)^3} \exp\left[-\left(\frac{r_1}{r_1} + \frac{r_3}{r_3}\right)/(1-\rho^2)\right] \times \\ \times I_0\left(\frac{2\rho V r_1 r_3}{\sqrt{r_1 r_3 (1-\rho^2)}}\right) = \frac{1}{r_1 r_3} \sum_{k=0}^{\infty} \left(\frac{\rho^k}{k!}\right)^6 \times \\ \times \left(\frac{d^k}{dx^k} x^k e^{-x}\right)_{\substack{x=\frac{r_1}{r_1}\\r_1}} \left(\frac{d^k}{dy^k} y^k e^{-y}\right)_{\substack{y=\frac{r_3}{r_4}}}.$$
(13.6.8)

In order to have the possibility of comparing a system of detection of the considered form with an optimum system of detection of a group of targets, we shall consider the case of two targets appearing simultaneously (see § 13.4). Here, we consider that in the system with separation the simultaneity of appearance of targets is used either by means of summation of the output signals of the corresponding two channels, or by means of making a decision about the presence of both targets when in at least one of these channels we exceed the threshold.

In the first case with the threshold we compare quantity

$$r = r_{1} + r_{2} = |w_{11}Q_{1} + w_{12}Q_{2}|^{2} + + |w_{21}Q_{1} + w_{32}Q_{2}|^{2} = \sum_{j,k=1}^{2} v_{jk}Q_{j}Q^{*}_{k}, \qquad (13....9)$$
$$v_{11} = |w_{11}|^{2} + |w_{12}|^{2}; \quad v_{22} = |w_{22}|^{2} + |w_{12}|^{2}; \\v_{13} = v^{*}_{21} = w_{12}(w_{11} + w_{22}).$$

where

When calculating probabilities of correct detection it is possible to use general formula (13.6.8). However, it is simpler to start from the expression for characteristic function (13.4.2), taking into account (13.6.9). Performing calculations analogous to those which we made in deriving (13.4.3) and (13.4.4) and assuming $q_{\underline{i}}^{\dagger}=q_{\underline{i}}^{\dagger}=q_{0}^{\dagger}$ (see § 13.4), we obtain

$$F_{(c)} = \frac{1+Y}{2Y} e^{-\frac{c}{1+Y}} - \frac{1-Y}{2Y} e^{-\frac{c}{1-Y}}, \qquad (13.6.4^{\circ})$$

$$D(c) = \frac{1+q'_{\bullet}(1-Y^{2})+Y}{2Y} e^{-\frac{c}{1+q'_{\bullet}(1-Y^{2})+Y}} - \frac{1+q'_{\bullet}(1-Y^{2})-Y}{2Y} e^{-\frac{c}{1+q'_{\bullet}(1-Y^{2})-Y}}, \qquad (13.6.11)$$

where c - the threshold quantity, multiplied by $(1 + \gamma^2)/(1 - \gamma^2)$, with which we compare r, and

$$\mathbf{Y} = |C(\lambda_1, \lambda_2)|/\sqrt{C(\lambda_1, \lambda_1)C(\lambda_2, \lambda_2)}.$$

Calculation of relationship $q_0^{\dagger}(D, F, \gamma)$ in this case is conducted almost the same as for the optimum system in § 13.3. Note that

$$D(c, \gamma, q'_{v}) = F(c_{i}, \gamma_{i}), \qquad (13.6.12)$$

where

$$Y_1 = \frac{Y}{1 + q'_{\bullet}(1 - Y^2)}; \ c_1 = \frac{c}{1 + q'_{\bullet}(1 - Y^2)}.$$
 (13.6.13)

Considering that $\gamma_1 << 1$ when $\gamma^2 < \ldots$, it is possible to calculate q_0' by formula

$$\boldsymbol{q}_{\bullet}^{\prime} = \frac{1}{1-\gamma^{*}} \left[\frac{\boldsymbol{c}\left(\boldsymbol{F},\gamma\right)}{\boldsymbol{c}\left(\boldsymbol{D},\boldsymbol{u}\right)} - 1 \right]. \tag{13.1.16}$$

If the registion about the presence of targets is taken when at least one channel ∞ must the threshold, the probability of making such a decision is recorded, taking incomposition (13.6.8), in the form

$$p_{1} = 1 - \int_{0}^{c_{e}} \int_{0}^{c} p(r_{1}, r_{2}) dr_{1} dr_{2} = 2e^{-\frac{c}{r_{1}}} - e^{-\frac{2c}{r_{1}}} - e^{-\frac{2c}{r_{1}}} - \sum_{k=1}^{\infty} \left(\frac{p^{k}}{k!}\right)^{k} \left(\frac{d^{k-1}}{dx^{k-1}} x^{k} e^{-\frac{c}{r_{1}}}\right)^{k} = \frac{c}{r_{1}}$$
(13.6.15)

Constituting in this formula the corresponding values of ρ and $\overline{r_1}$, it is possible to calculate the probability of false alarm F and of correct detection D. For comparison in Fig. 13.14 we show the dependence of the threshold signal-to-interference ratio q_0^{-1} (obtained in the absence of separation) on , and on the probability of correct metertion of both targets for $F = 10^{-6}$ for different methods of processing.



From comparison of curves of this figure one may see that loss due to use of a system with separation instead of an optimum one grows with increase of y and with decrease of the probability of correct detection. For small y and $\beta = 1 - D$ separation of signals does not lead to essential losses. If, for instance, we consider double loss in the signal-to-interference ratio permissible, it is attained for $\beta = 0.5$ at $\gamma^2 \approx 0.37$; for $\beta = 0.9$ at $\gamma^2 \approx 0.05$; for $\beta = 0.999$ at $\gamma^2 \approx 0.93$. The difference between systems with sugmation and independent comparison with a threshold of the output signals in the case of two targets is immaterial.

13.6.2. System of Measurement with Complete Suppression of Selected Interfering Signals

Let us consider the discriminator of a tracking meter with total suppression of selected interfering signals. Here, as too in § 13.5, we shall consider fluctuations of the signal to be fast, and noise to be white, and we shall limit ourselves for simplicity to consideration of signals depending only on time. Such consideration



ity, 1/.10. Plock diagram of the analyted discriminator with suppression of interforing signals: 1) mixer; 2) cliter; 3) defector; 4) subtractor.

embraces problems of measurement of range and measurement of angles with scanning. Generalization of results to the base of space processing is achieved by simple replacement of matrix elements $\mathcal{O}_{jk}^{(pq)}(x_j, x_p)$ (for designations see § 10.5) obtained by integration over time by the same elements obtained by integration over time and over space.

A block diagram of the analyzed discriminator is shown in Fig. 1 (19).

-805-

Reference signals $\psi(t, \lambda)$ and $\psi(t, \lambda_0 - \Delta \lambda)$ are determined by formulas $(\pm j, 2, 9)$. In order to simplify calculations we consider detuning $\Delta \lambda$ very small, so that

$$z(t) = \left| \int_{-\infty}^{t} h(t-\tau) \psi(\tau, \lambda_{0} + \Delta \lambda) y(\tau) d\tau \right|^{2} - \frac{1}{2} \int_{-\infty}^{t} h(t-\tau) \psi(\tau, \lambda_{0} - \Delta \lambda) y(\tau) d\tau \Big|^{2} = 2\Delta \lambda \frac{\partial}{\partial \lambda_{0}} \left| \int_{-\infty}^{t} h(t-\tau) \psi(\tau, \lambda_{0}) y(\tau) d\tau \right|^{2}.$$
(13.6)

We can hope that the dependence of equivalent spectral density on detuning will in the given case, as for systems without suppression of interfering signals (Chapters VIL -XI), be sufficiently weak so that the given assumption immaterially limits the applicability of the results.

Let us consider the gain factor $K_{\rm R}$, the equivalent spectral density $S_{\rm HCB}$ and systematic error $\varepsilon_{\rm CACT}$ for a system of the considered form. Here we consider that interfering signals, if there are any, are completely suppressed, and we consider only the influence of noises. In general, complete suppression of interfering signals is obtained only for preselected values of their parameters. In certain particular cases (for instance, during phase-tode manipulation) suppression of signals at selected points ensures their suppression for all values of parameters beyond the limits of the basic maximum of the function of uncertainty. After very the argume transformations, snalogous to those used in § 13.5, we obtain

$$K_{\mathbf{R}} = \left(\frac{\delta z}{\partial \lambda_{\mathbf{0}}}\right)_{\substack{\mathbf{A}_{\mathbf{0}} = \lambda_{\mathbf{0}}}} = -2\Delta \lambda \frac{P_{\mathbf{0}}}{\Delta f_{\mathbf{0}}} \boldsymbol{u}_{\mathbf{0}\mathbf{0}} \times \\ \times \frac{1}{2\pi} \int_{-\infty}^{\infty} |H(l\boldsymbol{w})|^{2} S_{\mathbf{0}}(\boldsymbol{w}) d\boldsymbol{u} (b_{\mathbf{n}} - 2a_{\mathbf{n}}^{2}), \qquad (13.0.17)$$

$$s_{\text{CHOT}} = \frac{\frac{1}{2\pi} \int_{-\infty}^{\infty} |H(i\omega)|^2 \left[1 + \frac{h}{\varpi_{ee}} S_e(\omega)\right] d\omega}{h(b_n - 2a_n^2) \frac{1}{2\pi} \int_{-\infty}^{\infty} |H(i\omega)|^2 S_e(\omega) d\omega} a_n \sqrt{\omega_{ee}}, \qquad (13.6.12)$$

-806-

$$S_{\theta KB} = \frac{\frac{1}{2\pi} \int_{-\infty}^{\infty} |H(i\omega)|^4 \left\{ b_n + 2a_n^2 + \frac{2h^2}{w_{ee}} (b_n - 2a_n^2)^{s} \times \frac{2h^2}{w_{ee}} (b_n - 2a_n^2)^{s} \times \frac{2h^2}{w_{ee}} S_e(\omega) (b_n + 4a_n^2) + \frac{2h^2}{w_{eo}^2} S_0^2(\omega) a_n^2 \right\} d\omega}{\times \left(\frac{1}{2\pi} \int_{-\infty}^{\infty} |H(i\omega)|^2 S_e(\omega) d\omega \right)^{s}}, \qquad (13.6.19)$$

where $h = \frac{P_0}{2N_0\Delta f_c}$; $H(i\omega) - frequency response of the filter;$

$$b_n = C_{\bullet\bullet}^{(11)}(\lambda_{\bullet}, \lambda_{\bullet}) - \sum_{j,k=0}^{n} C_{\bullet j}^{(10)}(\lambda_{\bullet}, \lambda_j) w_{jk} C_{\bullet k}^{(10)*}(\lambda_{\bullet}, \lambda_k); \qquad (13.6.20)$$

$$a_n = \frac{1}{\sqrt{w_{\bullet\bullet}}} \operatorname{Re} \sum_{j=0}^{n} w_{\bullet j} \frac{\partial C^*_{\bullet j} (\lambda_{\bullet}, \lambda_j)}{\partial \lambda_{\bullet}}.$$
 (13.6.21)

In general a discriminator of the considered form gives systematic error which does not approach zero with unlimited increase of the signal-to-interference ratio, which it is necessary to specially compensate. This error disappears with symmetric location of parameters of interfering signals relative to the measured parameter.

In order to prove this we shall consider the case when $C_{jk}(\lambda_j, \lambda_k)$ depends only on difference $\lambda_j - \lambda_k (C_{jk}(\lambda_j, \lambda_k) = C(\lambda_j - \lambda_k))$, and values $\lambda_{-l}, \ldots, \lambda_{-1}, \lambda_1, \ldots, \lambda_l$ of the parameters of the interfering signals are such that $\lambda_j - \lambda_0 = \lambda_0 - \lambda_{-j} (0 \le 1 \le l)$. Here, as it is easy to show, $w_{jk} = w_{-j-k}^* (0 \le j, k \le l)$ and $a_n = 0$. Systematic error ε_{OUCT} turns into zero, and the expression for equivalent spectral density takes the following form:

$$S_{\mathbf{0}\mathbf{K}\mathbf{0}} = \frac{\frac{1}{2\pi} \int_{-\infty}^{\infty} |H(i\omega)|^4 \left(1 + \frac{h}{\omega_{ee}} S_{\mathbf{0}}(\omega)\right) d\omega}{2 \frac{h^2}{\omega_{ee}} b_n \left(\frac{1}{2\pi} \int_{-\infty}^{\infty} |H(i\omega)|^4 S_{\mathbf{0}}(\omega) d\omega\right)^2} \approx \frac{1}{\frac{P_e}{N_e} b_n}.$$
 (33.0.22)

The last equality in (13.6.22) is valid for a large signal-to-noise ratio at the output of the system of separation of signals $(\frac{h}{w_{OO}} >> 1)$ and for $\Delta f_{\oplus} >> \Delta f_{C}$ ($\Delta f_{\oplus} = -$ width of the passband of the filter).

The obtained expression coincides for large ratios $\frac{h}{w_{00}}$ with the equivalent spectral density of the optimum discriminator and for arbitrary h to the equivalent spectral density of the discriminator analyzed in § 13.5 (Fig. 13.7). This is

completely natural, since with symmetric location of interfering signals $\frac{\partial w_{OU}}{\partial \lambda_{O}} = 0$, and we have equality [see (13.2.13)]



so that for large signal-to-noise ratios processing of a signal in the considered discriminator coincides with the optimum.

It follows from this that a discriminator of the given form can be used with success in systems of measurement if parameters of the suppressed signal are located symmetrically relative to the measured parameter of the target and vary together with this parameter and $C_{jk}(\lambda_j, \lambda_k) = C(\lambda_j - \lambda_k)$. This occurs, for instance, if signals corresponding to spurious maxima of the function of uncertainty are suppressed.

Loss in accuracy connected with suppression of interfering signals is determined for large signal-to-noise ratios by formula

$$\Gamma = \frac{b_{\bullet}}{b_{\pi}} = \frac{C_{00}^{(11)}(\lambda_{\bullet}, \lambda_{\bullet}) - |C_{00}^{(10)}(\lambda_{\bullet}, \lambda_{\bullet})|^{2}}{C_{00}^{(11)}(\lambda_{\bullet}, \lambda_{\bullet}) - \sum_{j=k=0}^{\pi} C_{0j}^{(10)}(\lambda_{\bullet}, \lambda_{j}) w_{jk} C_{0k}^{(10)*}(\lambda_{\bullet}, \lambda_{k})}.$$
(13.6.23)

With an arbitrary signal-to-noise ratio the loss in separation of signals can be accounted for by increasing the spectral density for the system without separation of signals by a factor of Γ/w_{OO} and replacing the signal-to-noise ratio h by h/w_{OO} .

Note that quantity Γ is equal to one if $C(\lambda) = \text{const}$ outside the principal maximum and $C\binom{(10)}{00}(\lambda_0, \lambda_0) = 0$. This takes place, for instance, for measurement of range with a signal manipulated in phase by Hoffman's code [see 13.2.2].

The same result is always obtained when values of parameters of suppressed signals coincide with maxima of the function of uncertainty $|C(\lambda - \lambda_0)|^2$, and $C(\lambda - \lambda_0)$ is a real function. Here $C(\frac{10}{00}(\lambda_0, \lambda_0) = C^*(\lambda_0 - \lambda_0) = 0$ at all the considered points. For the shown cases the loss of accuracy for large signal-to-noise ratios is absent.

-808-

13.6.3. System of Measurement of Parameters of Motion of Two Targets

In the preceding paragraph we considered the case when we exactly know parameters of the interfering signals. Such a case occurs, for instance, during suppression of signals corresponding to maxima of the function of uncertainty. We turn now to the case of measurement of coordinates of several targets by tracking radar meters. Such systems have a number of outputs equal to the number of measured quantities. Voltages from the outputs and also from intermediate points of the measuring channels can be used for compensation of interfering signals in other channels or for tuning of compensation channels.

Let us consider a tracking meter of coordinates of two targets. Being interested in the meter of coordinate λ_1 of one of them, we assume that in the channel of



Fig. 13.16. Circuit of a discriminator with suppression of an interfering signal. 1) mixer; ?) amplifier with gain varying according to the law of amplitude modulation; 3) amplifier with gain γ ; 4) filter; 5) square-law detector.

measurement of quantity λ_1 there is realized compensation of the signal reflected from the second target, where for construction of the compensation channel to maximum extent there is used the channel of coordinate λ_p of the second target. If we apply here a form of compensation which with exact tuning corresponds to total suppression of the interfering signal, the functional circuit of the discriminator is represented by Fig. 13.16.

The discriminator contains two channels mutually defuned with respect to parameter $\lambda_1 (\pm \Delta \lambda_1)$ intended for measurement of λ_1 in the absence of an interfering signal, and also a channel for compensation of the latter. With proper selection of filters of the basic channels the discriminator is optimum in the absence of an interfering signal, strictly speaking, we would need to select part eters of the compensation channel γ and ζ

-809-
differently for the two defuned channels. However, if we are interested in tracking meters of both parameters, the channel of compensation is tuned with errors determined by the accuracy of measurement of the parameter λ_p .

For small $\Delta\lambda_1$ errors of measurement of λ_2 (quantities $\Delta\lambda_2$) turn out to be comparable with $\Delta\lambda_1$. Here separate tuning of compensation channels loses meaning and the simplest construction of the compensation channel, corresponding to the considered circuit, is desirable. Let us find systematic error and equivalent spectral density of fluctuating error of the circuit of Fig. 13.16 first on the assumption of absence of error of measurement of $\lambda_2(\Delta\lambda_2 = 0)$, and then taking this error into account.

The signal at the discriminator output can be recorded in the form

$$\Delta u_{\pi}(t) = \int_{0}^{t} \int_{0}^{t} h(t - s_{1}) h(t - s_{2}) \{u_{1-}(s_{1}) u^{*}_{1-}(s_{2}) - u_{1+}(s_{1}) u^{*}_{1-}(s_{2}) - u_{1+}(s_{1})) u^{*}_{2}(s_{2}) e^{-i\varphi} + (u^{*}_{1-}(s_{2}) - u^{*}_{1+}(s_{2})) u_{2}(s_{1}) e^{i\varphi}] \} \times e^{i\omega_{0}(s_{1}-s_{1})} y(s_{1}) y(s_{2}) ds_{1} ds_{2}, \qquad (13.6.24)$$

where

h(t) - pulse response of the low-frequency equivalent of the filter; $u_{d+}(t)$ and $u_{d+}(t)$ - reference last of modulation in the two defuned channels.

With the same simplifying assumptions at in the preceding paragraph and with coincidence of the true value of λ_4 with the average value of $\hat{\lambda}_1$ among points of tuning, of the channels for small detunings $\Delta \lambda_4$ we obtain

$$\overline{\Delta u_{\mathbf{A}}(l)}|_{\mathbf{i}\mathbf{\lambda}_{s}=0} = 8N_{\mathbf{o}}B_{\mathbf{i}}\gamma \frac{\partial \gamma}{\partial \lambda_{\mathbf{i}}} \Delta \lambda_{\mathbf{i}}, \qquad (13.0.20)$$

where $\delta \lambda_1 = \lambda_1 - \hat{\lambda_1};$

$$B_1 = \int h^3(s) ds. \qquad (13.6.26)$$

The gain factor of the discriminator $K_{\rm ff}$ can be found as

$$K_{\mathbf{A}} = \frac{d\Delta u_{\mathbf{A}}(l)}{d\lambda_{1}}\Big|_{\mathbf{A}_{1}=0}.$$
(13.6.27)

After calculations we have

$$K_{\mathrm{R}} = -\frac{4A_{1}P_{e}}{\Delta f_{e}} \left[C\left(1-\gamma^{*}\right)+\gamma^{*}\frac{\partial \varphi}{\partial \lambda_{1}}D-D^{*} \right] \Delta \lambda_{1}, \qquad (13.6.28)$$

where

$$A_{1} = \Delta f_{0} \iint_{0} h(s_{1}) h(s_{2}) p(s_{1} - s_{2}) ds_{1} ds_{2}. \qquad (13.0.29)$$

Systematic error ϵ_{cMCT} is now found as ratio

$$\mathbf{s}_{CRCT} = \frac{\overline{\Delta \alpha_{R}(b)}_{|b\lambda_{1}=0}}{\mathcal{K}_{R}} = -\frac{B_{1}\gamma \frac{\partial \gamma}{\partial \lambda_{1}}}{A_{1}h \left[C\left(1-\gamma^{1}\right)+\gamma^{1} \frac{\partial \gamma}{\partial \lambda_{1}}D-D^{2}\right]}.$$
(13.6.30)

Thus, when $\gamma \neq 0$ due to compensation there appears systematic error, inversely proportional to the signal-to-noise ratio h. The presence of such error should have been expected, since in an optimum circuit for compensation of such error there is introduced the corresponding correction, whereas in the given circuit such a measure is not provided. For large h systematic error can be disregarded.

For allowance for error $\Delta\lambda_2$ of tuning of the compensation channel it is necessary in expression (13.6.24) to replace $u_2(t)$ by a certain $u_{20}(t)$. Accordingly γ and ϕ should be replaced by γ_0 and ϕ_0 . Considering $\Delta\lambda_2$ small and limiting ourselves to the first two terms of the expansion of $u_{20}(t)$ in a Taylor series

$$u_{so}(t) = u_{s}(t) + \Delta \lambda_{s} \frac{\partial u_{s}(t)}{\partial \lambda_{s}}, \qquad (13.1.31)$$

it is easy to find

$$K_{\mathbf{A}} = -\frac{4A_{1}P_{o}}{\Delta f_{o}} \left[C\left(1 - \gamma_{o}^{2}\right) + \gamma_{o}^{2} \frac{\partial \varphi_{o}}{\partial \lambda_{1}} D - D^{*} \right] \Delta \lambda_{1}$$
(13.0.32)

and

$$\mathbf{z}_{chOT} = -\frac{\frac{B_1 \mathbf{Y}_0 \frac{\partial \mathbf{Y}_0}{\partial \lambda_1}}{A_1 h} + \frac{1}{q} \left[\mathbf{Y}^2 \frac{\partial \mathbf{y}}{\partial \lambda_1} \left(D - \frac{\partial \mathbf{y}}{\partial \lambda_1} \right) - \left(\frac{\partial \mathbf{Y}}{\partial \lambda_1} \right)^2 \right] \Delta \lambda_1}{C \left(1 - \mathbf{Y}_0^2 \right) + \mathbf{Y}_0^2 \frac{\partial \mathbf{y}_0}{\partial \lambda_1} D - D^2}$$
(13)

The difference of γ_0 and ϕ_0 from γ and ϕ , respectively, for small $\Delta\lambda_2$ can usually be disregarded. However, the presence of a second term in the numerator of (13.6.33) can be accounted for since q and h may differ greatly. For very shall q (high power of interfering signals) this term may cause substantial additional error. For sufficiently large q it can be disregarded.

It is necessary to note that for moments t, very remote from the beginning of the process, quantities A_{j} and B_{j} in (13.0.33) do not depend on t and are determined ty formulas

$$A_{1} = \frac{1}{2\pi} \int_{-\infty}^{\infty} S_{\bullet}(\omega) |H(i\omega)|^{*} d\omega,$$

$$B_{1} = \frac{1}{2\pi} \int_{-\infty}^{\infty} |H(i\omega)|^{*} d\omega,$$
(13.6.3b)

where $H(im) \rightarrow$ frequency response of a filter whose pulse response is h(t).

The equivalent spectral density of fluctuation error is found as

$$\mathbf{S}_{u \mathbf{R} \mathbf{B}} = \frac{1}{K_{\mathbf{A}}^2} \int_{-\infty}^{\infty} R_{\mathbf{A}}(\tau) d\tau, \qquad (13.0.35)$$

where

 $R_{\mathbf{A}}(\tau) = \overline{\Delta u_{\mathbf{A}}(l) \Delta u_{\mathbf{A}}(l)} \longrightarrow \overline{\Delta u_{\mathbf{A}}(l) \Delta u_{\mathbf{A}}(l-\tau)} \qquad (3.2.5.3)$

is the correlation function of the output quantity of the distriminator. If we are interested in steady-state operating conditions (large t), then on the mode assumptions $(\tau_{1}^{2} - (\tau_{2}^{2}))$ this function does not depend on t, and accordingly τ_{200}^{2} is constant.

Substituting (13.6.24) in (13.6.35) and performing calculations correspond to (13.6.37) and (13.6.35) assuming small Δx_1 and $\Delta x_2 = 0$, we obtain

$$S_{\mathfrak{p}\mathfrak{k}\mathfrak{n}} = \frac{1}{2A_{1}^{\mathfrak{q}} \left[C\left(1-\gamma^{\mathfrak{q}}\right)+\gamma^{\mathfrak{q}} \frac{\partial \mathfrak{q}}{\partial \lambda_{1}} D-D^{\mathfrak{q}} \right]^{\mathfrak{q}}} \times \left\{ \frac{1}{q} S_{\mathfrak{o}} \left(1-\gamma^{\mathfrak{q}}\right)^{\mathfrak{q}} \left| \frac{\partial}{\partial \lambda_{1}} (\gamma e^{i\eta}) \right|^{\mathfrak{q}} + \frac{1}{h} S_{\mathfrak{o}\mathfrak{m}} \left(1-\gamma^{\mathfrak{q}}\right) \left[\frac{M}{2} - D^{\mathfrak{q}} + \frac{1}{2} + 2\gamma^{\mathfrak{q}} \frac{\partial \mathfrak{q}}{\partial \lambda_{1}} D + \frac{1}{q} \left| \frac{\partial}{\partial \lambda_{1}} (\gamma e^{i\eta}) \right|^{\mathfrak{q}} \right] + \frac{1}{h^{\mathfrak{q}}} S_{\mathfrak{m}} \left[\frac{M}{2} \left(1-\gamma^{\mathfrak{q}}\right) - D^{\mathfrak{q}} + \frac{1}{2} + 2\gamma^{\mathfrak{q}} \left(\frac{\partial \mathfrak{q}}{\partial \lambda_{1}} D + \left(\frac{\partial \gamma}{\partial \lambda_{1}} \right)^{\mathfrak{q}} \right) - \gamma^{\mathfrak{q}} \left| \frac{\partial}{\partial \lambda_{1}} (\gamma e^{i\eta}) \right|^{\mathfrak{q}} \right\} \right\}, \qquad (13.4.5)$$

$$S_{c} = \frac{1}{2\pi} \int_{-\infty}^{\infty} S_{\bullet}^{2}(\omega) |H(i\omega)|^{\bullet} d\omega;$$

$$S_{em} = \frac{1}{2\pi} \int_{-\infty}^{\infty} S_{\bullet}(\omega) |H(i\omega)|^{\bullet} d\omega;$$

$$S_{m} = \frac{1}{2\pi} \int_{-\infty}^{\infty} |H(i\omega)|^{\bullet} d\omega;$$

$$M = C + E;$$

$$E = \frac{1}{T_{r}} \int_{0}^{T} \left[\mathcal{U}_{a_{1}}(s) \left(\frac{\partial \phi_{1}(s)}{\partial \lambda_{1}} \right)^{\circ} - \frac{\partial^{2} \mathcal{U}_{a_{1}}(s)}{\partial \lambda_{1}^{2}} \right] \mathcal{U}_{a_{1}}(s) ds.$$

whore

1

(13.6.30)

The expression for C_{OKB} contains three components. The first of them does not depend on h and is caused by interaction of random components of the useful and interfering signals. The second component, proportional to $\frac{1}{h}$, reflects beats of the signal and interference with noise. Finally, the third component, proportional to $\frac{1}{C_2}$, appeared due to nonlinear transformations of noise.

Thus, in distinction from the optimum circuit, in the given case fluctuating arour is different from zero even with infinitely large signal-to-noise ratios h.

Allowance for error in tuning of the compensation channel $(\Delta \lambda_p \neq 0)$ leads, as it is easy to see, to appearance of a correction to formula (13.7.37), proportional to $(\Delta \lambda_p)^2$. If we consider $\Delta \lambda_p$ a small quantity, correction of the order of $(\Delta \lambda_p)^2$ in many cases can be disregarded.

13.5.4. Meter of Range with Two Pargets for One Particular Form of Elgnal Modulation

As an example of application of the results obtained slove for systems of measurement of coordinates we shall consider the following particular case. Let us find quantities characterizing errors of a range finder measuring the of two tato one of two targets, the distance between which is small. We ensure that the sounding signal is a pulse signal with Gaussian pulse shape and, forthermore, can possess intropulse linear frequency modulation, i.e., he scaples how of modulation has the form

$$u(t) = \sqrt{\frac{T_r}{\tau_{\rm s}}} e^{-\frac{\pi t^2}{2\tau_{\rm s}^2} + t\frac{\delta t^2}{2}} \left(-\frac{T_r}{2} \le t \le \frac{T_r}{2}\right), \qquad (1.5.7.57)$$

where $f = \frac{d\omega}{dt}$ - rate of change of frequency,

 $\tau_{_{\rm H}}$ - effective duration of a pulse, equal to

$$\tau_{\rm m} = \frac{1}{u_{a\,\rm MaKC}^2} \int_{-\frac{T_r}{2}}^{\frac{T_r}{2}} u_a^2(i) \, dt \qquad (\tau_{\rm sc} \ll T_r).$$

We consider that the correlation function of fluctuations of the reflected signal is exponential, i.e.,

$$\rho(t) = e^{-\frac{n}{2} \frac{1}{1}},$$

 \mathbf{or}

$$S_{\bullet}(\boldsymbol{\omega}) = \frac{a^{1}}{a^{1} + \boldsymbol{\omega}^{1}}.$$
 (13.6.40)

Let us consider accuracy of measurement of range when using discriminators ' oull: by the optimum scheme (Figs. 13.10 and 13.11), and also by schemes of Fig. 13.⁹ and Fig. 13.16, which in contain conditions, stated above, should give results close to the best ones, peculiar to the optimum closed).

As the basic characteristic we shall use equivalent spectral density S_{GKB} . In the same cases, when there are systematic errors, we shall also estimate them.

Properties of the optimum discriminator are completely characterized by S_{OF1} , determined by expression (13.5.64).

Calculating the quantities in this formula, we have

$$C = \frac{\pi}{\varrho_{\tau_{B}^{2}}} + \frac{\beta^{4}\tau_{B}^{2}}{2\pi},$$

$$D = 0$$

$$-\frac{\pi \mu}{4\tau_{B}^{2}} \left(1 + \frac{\beta^{4}\tau_{B}^{4}}{\pi^{4}}\right),$$

$$\eta = \frac{\beta \delta m}{2},$$

$$A = \frac{\pi \alpha}{2\mu N_{0}^{2}} \left(\frac{a_{1} - a_{2}}{\alpha} - \frac{a_{1}}{\sqrt{\alpha^{2} + \xi - \mu}} + \frac{a_{3}}{\sqrt{\alpha^{2} + \xi + \mu}}\right),$$

$$B = \frac{\pi}{4\mu N_{0}^{2}} \left(\frac{1}{\sqrt{\alpha^{3} + \xi - \mu}} - \frac{1}{\sqrt{\alpha^{2} + \xi + \mu}}\right),$$

(13.6.41)

 $\mathcal{M}_{1,1} : \mathbf{1}^{\mathbf{C}}$

8 = t1 - t1,

-014-

 τ_1 , τ_2 - delays of modul tion of signals reflected from the two targets,



With intrapulse frequency modulation deviation of frequency is usually selected large, so that $\beta \tau_M^2 \gg 1$. Fulfillment of this condition permits results of calculations (both for the optimum and also for other circuits) to approximately coincide for cases of the presence and absence of frequency modulation if only we introduce in formulas quantity x, determined as $x = \tau_M$ without frequency modulation and $x = \frac{\pi}{\beta \tau_M}$ with it.

Designating $y = \sqrt{\frac{\pi}{2}} \frac{5}{x}$ - relative difference of delays of signals reflected from two targets,* we have

$$C = \frac{\pi}{2r^{1}}; \ \gamma = e^{-\frac{\mu^{2}}{2}},$$

and formula (13.5.61) takes form

$$S_{\bullet a \gamma}^{-1} = \frac{P_{\bullet}}{2} \left\{ \left[(1 - \gamma^{\bullet}) A + \frac{2N_{\bullet}}{P_{a}} B \right] C + \frac{2N_{\bullet}}{P_{a}} \gamma \gamma' B' - \gamma'' A \right\}, \qquad (13.6.42)$$

where the stroke signifies differentiation with respect to τ_1 .

Results of calculations of S_{ONT} for case $q = \frac{rc}{P_{\Pi}} = 1$ (identical powers of the base ful and interfering signals) are presented in Fig. 13.17.

In this same figure there is depicted the asymptotic dependence of $S_{\rm out}$ on y

$$S_{0\Pi T} = \frac{2x^2 (1 - e^{-y^2})}{\pi \alpha h (1 - e^{-y^2} - y^2 e^{-y^2})}, \qquad (13.0.43)$$

taking place for very large $h(1 - e_y^2)$.

*It is also possible to call this quantity the relative distance between forgets.

-,315-



Fig. 13.17. Dependence of equivalent spectral density of an optimum discriminator on the relative distance between targets y: - - by the formula (13.5.43).

With decrease of quantity y, proportional to the distance between targets, spectral density of fluctuating error grows, which is explained by two factors. First, there take place power losses connected with compensation of the interfering signal. Second, compensation of the interfering signal is inaccurate due to errors of measurement. This inaccuracy leads to eaximum errors in the region of the greatest slope of the function of uncertainty of the signal. Bith greater nearness of the targets the distance between them corresponds to the flat part of the curve of the function of supertainty, due to which equivalent spectral density, due to the second to the shown factors, decreases. With small signal-to-noise ratios h quantity $d_{\rm def}$ depends little on the distance between targets, which was clear from general formula (13.6.64).

Equivalent spectral density for a range discriminator built by the scheme of Fig. (3.4 is determined by formula (13.5.42), which for the considered case takes the form

$$S_{0KR} = \frac{hS_{oR} + \frac{S_{m}}{1 - \gamma^{2}}}{2h^{2} \left(E - \frac{\gamma^{2}}{1 - \gamma^{2}}\right) A_{1}^{2}}.$$
 (13.44)

Quantity A_1 is determined by expression (13.6.34), and quantities $S_{\mu\nu}$ (or

-816-

and E - by expressions (13.6.38) We consider that the frequency response of the filters in these expressions has the form

$$|H(i\sigma)|^{2} = \frac{hS_{0}(\omega)}{1 + hS_{0}(\omega)},$$
 (13.6.45)

i.e., filters are optimum in conditions of resolved targets.

Producing calculations, we obtain

$$A_{1} = \frac{1}{2} \frac{ah}{\sqrt{1+h}(1+\sqrt{1+h})},$$

$$S_{m} = \frac{h^{2}a}{4(1+h)^{1/a}},$$

$$S_{0m} = \frac{h^{2}a(2\sqrt{1+h}+1)}{4(1+h)^{1/a}(1+\sqrt{1+h})^{2}},$$

$$E = C = \frac{\pi}{2x^{2}}.$$
(13.0.4c)

For large h formula (13.6.44) after substitution of values of coefficients takes the simple form

$$S_{\text{DKB}} = \frac{x^2}{\pi a h} \cdot \frac{2(1 - e^{-y^2}) + \frac{1}{\sqrt{h}}}{1 - e^{-y^2} - y^2 e^{-y^2}}.$$
 (13.6.47)

From comparison of (13.6.43) and (13.6.47) we see that for large h and not too small y the considered circuit possesses the same properties as the optimum. The difference between this circuit and the optimum appears at small h and small y.

For the discriminator of Fig. 13.16 equivalent spectral density is determined by formula (13.6.37). Finding from (13.6.38) coefficients S_c and M in this formula, we obtain

$$S_{e} = \frac{h^{2} \alpha \left[(1 + \sqrt{1 + h})^{2} + \sqrt{1 + h} \right]}{4 (1 + h)^{5/4} (1 + \sqrt{1 + h})^{2}}, \qquad (13.7.48)$$
$$M = E + C = \frac{\pi}{x^{2}}.$$

In Fig. 13.18 are curves of the dependences of S_{BKB} on the relative distance between targets y for different values of the signal-to-noise ratio h for all three considered circuits. For large h and relatively small y, at which application of circuits of compensation of signals still has meaning, the circuit of Fig. 13.5 turns out to be close to optimum. For small h its properties differ from properties of the optimum circuit in a certain range of values of y more than properties of the circuit.



A STATE OF STATES

off P10. 13.46.

The large h the last circuit possesses considerably greater fluctuation error than the preceding ones.

Furthermore, in distinction from the above considered directly to possesses by a case error, determined by formula (12, 2, 2). Calculation in this formula relate of coefficients, we find that

$$e_{CROT} = e_{CROT_1} + e_{ORCT_3},$$

$$e_{CROT_1} = \sqrt{\frac{2}{\pi}} x \frac{1 + \sqrt{1 + h}}{h} \frac{y e^{-y^2}}{1 - e^{-y^2}}$$

$$e_{OROT_3} = \Delta \tau_3 \frac{y^3 e^{-y^2}}{1 - e^{-y^2}},$$
(13.6.49)

where

and

 Δt_{0} - error of measurement of delay of the signal reflected from the second target.

The curve of the dependence of $\varepsilon_{\text{CMCT 1}}$ on y is depicted in Fig. 13.19.

The ratio of the two components of systematic error is defined as

$$Z = \frac{e_{\text{source}}}{e_{\text{source}}} \frac{2x^2}{\pi} \frac{1 + \sqrt{1+h}}{h} \frac{1}{\partial \cdot \Delta \tau_1}.$$
 (13.0.50)

From this we can find the limiting value of error $\Delta \tau_2$ at which error $\epsilon_{\text{curv},2}$ can be disregarded as compared to $\epsilon_{\text{OMOT 1}}$ in the range of 5 interesting us.



The given relationships illustrate application of the general formulas obtained above and explicitly show the dependence of basic characteristics of discriminators on different physical parameters.

5 13.7. Conclusion

Featies selection of the form of the sounding signal or the form and dimensions d the mentane of the untenne of great value for increase of resolving power of

resear cycles is synthesis of optimum methods of processing signals. This synthesis can be conducted in two ways. The first of them consists in finding operations of the receivar produced on the r-f signal for which there is provided the best nower reintionships between the separated signal, interference and possible interforing signals at the output of the receiver. These operations can be found with various additional conditions, connected with the specific nature of resolution. In particular they are found on the condition of total suppression of interforing signals.

The second method consists in statistical synthesis of an optimum ruber, executed in the very best menner various functions in the presence of several closely loss is sarjets. Here, as a matter of fuct, it would be possible, in general, not to introduce the contept of resolution, since it is not necessary to turn to this idea for volution of the corresponding problems. However, constaining, on the one has: the extent of the idea of resoluting power, and on the other hand, interest in responneous of these methods of synthesis, we treated a series of solved statistical provides behave it is clear that in most cases with not too close coincidence of the eligible reflected from various targets and with not too close coincidence of the eligible for eperations optimum for realistic considered in formation of systems to be basis of superscience of signals from interficience of the possibility of complete the interesting in contrasted with its clustificity and the possibility of application for an rease of resolving power of systems, optimum without interfering there are lossed in the preceding shapeers.

We have same time, by turning to the theory of statistical solutions we obtain the dystems, optimum in conditions of detection and measurement of coordination turn in cost not only processing of r-f signals in receivers, but also subsequent confident transformations. The found colutions, besides confirming the correctness of the decementioned general basis of resolution, are also of the dependent interest. Outline nonlinear transformations of signals have in the presence of disturking therees specific penultarities which cannot be found by investigation of separate disting out of signals alone and sometimes render substantial influence on recolving power. Characteristics of the corresponding optimum and quasi-aptimum determine pescificities of resolution of targets.

-020-

Multi-target problems are often complicated and diverse. The given solutions are, obviously, only an insignificant part of the solutions of practical interest in this area. We shall show several problems whose solution, in the opinion of the authors, are necessary in connection with the problem of resolution of targets.

For meters of coordinates it would be of interest to find optimum (in the sense of absence of systematic and a minimum of fluctuating errors) systems in the case of slow fluctuations of the signal, when the maximum likelihood estimates cease to be efficient, and then it is necessary to look for other methods of construction of efficient estimates.

Both for problems of measurement and also for problems of detection it is expedient to find optimum systems for non-Gaussian fluctuating signals.

If the form of discriminators of systems of optimum resolution is found in a number of interesting cases, the form of smoothing circuits was not discussed in general. This problem deserves serious attention.

In examing questions of detection of targets we assumed that their number and possible location are known to some extent. Very important is solution of the problems of detection with an unknown number and position of targets. Furthermore, one should solve the problem of optimum search in the presence of many unresolved targets with finding of the optimum law of change of resolving power in the course of searching.

In connection with these problems there is also the question of suppression of interfering signals, not at assigned points, but in a certain range of values of parameters.

Similar problems were solved for definite forms of sounding signals. The obtain a solutions give us the possibility of carrying out selection of sounding signals while ensure the best characteristics with an optimum method of their processing. Buck a selection of signals and also solution on the basis of the obtained general results of different particular problems are practically very important.

-821--

Preceding page blank

· Participation of the state o

LITERATURE FOR VOLUME II

1. Radar technology, Vol. II. Trans. from English. Publishing House "Soviet Radio," 1949.

2. H. V. Sytina. Self-contained Doppler radio navigational instruments. Publishing House "Soviet Radio," 1957.

3. A. A. Fel'dbaum. Computers in automatic systems. Fizmatgiz, 1959.

4. F. M. Woodward. Probability theory and information theory with applications to cadar. Trans. from English, edited by G. S. Gorelik. Publishing House "Soviet Radio," 1955.

5. D. Middleton. Introduction to statistical communication theory, Vols. I-II. Trans. From English, edited by B. R. Levin. Publishing House "Soviet Radio," 1901-1962.

o. S. S. L. Chang. Informational criterion for closed systems of automatic control. Transactions I International Congress of IFAC. Statistical methods of investigation, 1961.

7. Tsien Hsüch-sên. Engineering cybernetics. Trans. from English, edited by A. K. Fel'dbaum. Publishing House of Foreign literature, 1995.

1. H. Cramer. Mathematical methods of statistics. Trans. from English, edited by A. N. Kolmogorov. Publishing House of Foreign Literature, 1948.

9. J. H. Laning and R. H. Battin. Random processes in problems of automatic control. Frans. from English. edited by V. S. Pugachev. Publishing House of Foreign Literature, 1958.

10. R. Bernstein. Investigation of angular accuracy of survey radar. In collection "Reception of signals in noise," edited by L. S. Gutkin. Publishing House of Foreign Literature, 1960.

11. P. Sverling. Maximum accuracy of determination of angular coordinates of palse radar sets. "Problems of radar technology," 1957. No. 2.

12. J. Ye. Faltkovich. Reception of radar signals against a background of fluctuating interferences. Publishing House "Soviet Radio," 1901.

13. P. A. Bakut. Maximum likelihood estimate of normal signals. "Hews of Schools of Hig.er Education," Radio engineering, 1962, No. 3.

14. J. J. Shapiro, dulculation of trajectories of tallistic missiles by regar neuspressies. Trans. From English, edited by Yu. G. Hillgram. Publishing House of

Foreign Literature, 1901.

15. L. A. Zadeh. Optimum nonlinear filters. Journ. Appl. Phys., 1953, Vol. 24, p. 300.

4...E. L. Stratonovich. Application of the theory of Markov processes for optimum filtration of signals. "Radio engineering and electronics," 1960, Vol. V. No. 11.

17. R. L. Stratonovich. Optimum reception of a narrow-band signal with unknown frequency against a background of noises. "Radio engineering and electronics," '901, Vol. VI, No. 7.

13. H. R. Stratonovich. Ceparation of a signal with indenstant fragency treat noise. "Radio engineering and electronics," 1962. Vol. VII, No. 2.

19. D. A. Vaynchteyn and V. D. Zubakov, Separation of signals against a baseground of random interferences. Publishing House "Soviet Radio," 1960.

2. R. L. Stratonovich. Selected problems of the theory of fluctuations in radio engineering. Publishing House "Soviet Radio," 1961.

21. I. A. Bol'shakov. Analysis of breakoft of tracking in automatic control systems under the influence of fluctuating interference. "Automation and teremedianner." 1959, Vol. XV, No. 12.

20. A. M. Vasil'yev. Application of the theory of Brownian movement to investigration of noise locanity of radio tracking devices (scientific reports of higher school). "Radio engineering and electronics," 1959, Vol. 2, No. 1.

53. G. F. Barret. Application of Recognic equations for investigation of technic control systems with random disturbances. Transactions I International denoress of IFAC. Statistical methods of investigation, 3901.

14. D. P. Juina and M. I. Van-Val'kenburg. Statistical analysis of systems of contactions is tracking. Transactions I International Congress of TPAC. Statistical sethods of investigation, 1961.

25. I. A. Bortshakov and V. G. Rept., Constitute of Monthews allowed by a section and telemechanics," 1951, 201, 2011, 30, 51

20. A. F. Rogonelov. Principles of reder. Publiching house ""write Radio," 1984.

17. A. C. Sybell, Principles of radar. Fublishing House "Foviet Fadio,"

C. G. P. Farlakovskiy. Radio range finder under the influence of interview and Classicalians of the reflected signal. "Journal of Engineering physics," (1999, 301, 1, No. 6.

F. I. M. Asigntov and V. I. Tikhénov. Influence of fluctuations on work of a reference range finder. "Automation and telemechanics," 1958, Vol. XIX, No. 4.

20. R. N. Ellynshev. Noise immunity of time discrimination of pulse signals. "Anote engineering and electronics," 1959, Vol. TV, No. 4.

34. G. F. Bartakovskiy. Radio range finder with frequency modulation under the influence of noises and fluctuations of the reflected signal. "Madic engineering and electronics," 1961, Vol. VI, No. 4.

A. R. N. Mityashev. Optimum method of determining the time position of pulses. "Padto engineering and electronics," 1960, Vol. V, No. 2.

33. 7. Ye. Fallkovich. Accuracy of reading of the range coordinate in rodar systems. "Radio engineering and electronics," 197, Vol. II. Nos. 4, 8.

34. P. A. Bakulev. Radar methods of selection of moving targets. Oborongiz, 1998.

35. G. L. Turin. Introduction to the theory of matched filters. IRE. Trans., June 1960, IT-6.

 \mathbb{P}^{1} . V. Zibert. General laws governing detection of targets by radar. "Problems of rear technology," 1955, No. 2.

37. W. Fishlein and O. Riltenfack. Correlation radar using pseudorandom modulation. IRE International Convention Record, 1961, Pt. 5.

38. Klauder, Price, Darlington and Elbersheim [?] Theory and designs of pulse radar sets with frequency modulation, "Foreign radio electronics," 1961, No. 1.

39. Cook. Pulse compression - key to more efficient radar transmission. Proc. of the IRE, 1960, Vol. 48, No. 3.

40. Wiesner. Source noises and their influence on radar systems. Trans. ISE, 1954, DE-1, No. 4.

4⁴. "Threshold signals." Trans. from English, edited by A. F. Sivers. Publishing House "Soviet Radio," 1952.

42. V. V. Solodovnikov. Statistical dynamics of linear automatic control systems. Fizmatgiz, 1960.

4%. Va. D Shimman. Theory of detection of the useful signal against a background of Gaussian noises and an arbitrary number of interfering signals. "Radio engineering and electronics," 1959, Vol. IV, No. 12.

44. C. Halstrom. Statistical theory of detection of signals. Trans. from English, edited by Yu. B. Kobzarev. Publishing House of Foreign Literature, 1962.

45. B. J. Duwaldt. Inverse Probability in Angular Tracking Radars. IRE Trans. on Information Theory, March, 1956, Vol. IT-2, No. 1.

46. S. Ye. Fal'kovich. Potential accuracy of reading of angular coordinates in radar systems. "Radio engineering and electronics," 1959, Vol. IV, No. 1.

47. "Servomechanism theory." Edited by H. James, N. Nichols and R. Phillips. Trans. from English, edited by Ya. Z. Tsypkin. Publishing House of Foreign Literature. 1953.

43. D. R. Rods. Introduction to single-pulse radar. Trans. from English, edited by L. D. Bakhrakh. Publishing House "Soviet Radio," 1960.

49. V. V. Migulin. Lectures on radar principles. Fublishing House of Moscow State University, 1958.

10. Brennan. Accuracy of measurement of angular coordinates with an antenna in the form of a phased array. "Foreign radio electronics," 1962, No. 1.

51. A. A. Kuriksha. Optimum use of space-time signals. "Radio engineering and electronics,' 1963, No. 4.

12. M. M. Kreymerman. Definition of parameters of correcting devices of serve systems from assigned values of their generalized parameters. "Automation and telemechanics," 1958, Vol. XIX, No. 2.

3. 3. P. Tartakovskiy and Yu. M. Sergiyenko. Influence of a sequence of pulses modulated by a random process on an inertial pulse detector. "Radio engineering," 1955, No. 1.

A. A. Colomb and D. Usdin. A theory of multidimensional cervo systems. Formar, Denklin Inst., January, 1952.

50. I. Ws. Gokhberg and M. G. Kreyn. Systems of integral equations on a holfline with nuclei, depending on the difference of arguments. UMN, 1958, Vol. XIII, Issue 2 (80).

bc. D. C. Joula. On the factorization of rational matrices. INE Transactions on IF, July, 1961.

57. H. C. Heich and C. T. Leondes. On the synthesis of multi-pole control systems in the Wiener sense. TRE National Conv. Record, 1959, Pt. 4.

- 93. A. M. Yaglom. Effective solutions of linear approximational problems for nutri-dimensional stationary processes with a rational spectrum. "Freiability theory and its application," 1960, Vol. V, Issue 3.

50. A. M. Yaglom. Correlation theory of processes with random stationary n-thincrements. "Mathematics collection," 1955, Vol. 37 (79), Issue 1.

40. I. A. Bollshakov. The problem of nonlinear filtration. "Radio engineeric and electronics," 1963, Vol. VIII, No. 6.

1. A. M. Yaglom. Examples of optimum extrapolation of stationary random processes. Report to VI All-Union conference on probability theory and mattematical statistics. Villayus, 1962.

まましをなめしておっていた。それのないのであるのであった。 マスクロボンド レトライン 仕事がなまた。そう トレイ・ストライン しょうしょう しょうせいりょう しょうせいりょう しょうしょう しょうしょう しょうしょう

2. V. G. Repin. Solution of the problem of filtration with limited knowledges of the statistics. "Automation and telemechanics," 1962, Vol. XXIII, No. 8.

67. D. Blackwell and M. A. Girshik. Sheory of games and statistical solutions. Truns. from English, edited by B. A. Sevasilyanov. Publishing House of Foreign Electature, 1958. 4. N. J. Hilsson. On the Optimum Bange Forefacion of Radar Signals in Moise. IRE Trans. on IT, October, 1961, Vol. IT-7, No. 4.

15. A. Conmerfeld. Optics. Trans. from German, edited by M. A. Illy-chreviel. Publishing House of Foreign Literature, 1993.

66. Yu. E. Linnik. Method of least squares are principled of the theory of processing observations. Fizmatgiz, 1969.

57. L. A. Vagnahteyn, Electromagnetit water. Fibitahing Heure "Boviet Badio," 1967.

ed. Mo. e. Cairman. Statistical analysis of optimum resolution. "Radio Astronomy and electronics," 1961, No. 3.

(3). L. N. Mityashev. Definition of the time position of pulses in the presence of interferences. Fublishing House "Soviet Radio," 1962.

70. Fovit et al. Methods and techniques of jamming and combatting it. "Foreign radio electronics," 1960, No. 11.